UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

Amendment No. 4 to FORM 10

GENERAL FORM FOR REGISTRATION OF SECURITIES PURSUANT TO SECTION 12(B) OR 12(G) OF THE SECURITIES EXCHANGE ACT OF 1934

The Babcock & Wilcox Company

(exact name of registrant as specified in its charter)

Delaware (State of incorporation or organization) 80-0558025 (I.R.S. Employer Identification No.)

The Harris Building
13024 Ballantyne Corporate
Place, Suite 700
Charlotte, North Carolina
(Address of principal
executive offices)

28277 (Zip code)

Registrant's telephone number, including area code: (704) 625-4900

Securities to be registered pursuant to Section 12(b) of the Act:

Title of Each Class
Registered
Common Stock, par value \$0.01 per share

Name of Each Exchange on Which Such Class will be Registered The New York Stock Exchange, Inc.

Securities to be registered pursuant to Section 12(g) of the Act:

None

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer $\,\mathbb{I}\,$

Accelerated filer [

Non-accelerated filer

Smaller reporting company [

(Do not check if a smaller reporting company)

THE BABCOCK & WILCOX COMPANY

INFORMATION INCLUDED IN INFORMATION STATEMENT AND INCORPORATED BY REFERENCE IN FORM 10

CROSS REFERENCE SHEET BETWEEN INFORMATION STATEMENT AND ITEMS OF FORM 10

We have filed our information statement as Exhibit 99.1 to this Form 10. For your convenience, we have provided below a cross-reference sheet identifying where the items required by Form 10 can be found in the information statement.

| No. Item Caption 1. Business. See "Summary," "Risk Factors," "Cautionary Statement Concerning "The Spin-Off," "Capitalization," "Selected Historical Combined I Forma Combined Financial Data," "Management's Discussion and An Results of Operations," "Business," "Relationship with McDer "Management." | |
|---|---|
| "The Spin-Off," "Capitalization," "Selected Historical Combined I Forma Combined Financial Data," "Management's Discussion and An Results of Operations," "Business," "Relationship with McDer | |
| | Financial Data," "Unaudited Pronalysis of Financial Condition and |
| 1A. Risk Factors. See "Risk Factors." | |
| Financial Information. See "Summary," "Risk Factors," "Capitalization," "Selected Histor "Unaudited Pro Forma Combined Financial Data" and "Managemer Financial Condition and Results of Operations." | |
| 3. Properties. See "Business." | |
| 4. Security Ownership of Certain Beneficial Owners and Management. See "Security Ownership of Certain Beneficial Owners and Management. | ent." |
| 5. Directors and Executive Officers. See "Management." | |
| 6. Executive Compensation. See "Management" and "Executive Compensation." | |
| 7. Certain Relationships and Related Transactions, and Director Independence. See "Summary," "Risk Factors," "Management," "Certain Relationship with McDermott After the Spin-Off." | ips and Related Transactions" and |
| 8. Legal Proceedings. See "Business—Legal Proceedings." | |
| 9. Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters. See "Summary," "Risk Factors," "The Spin-Off," "Dividend Police Stock." | cy" and "Description of Capital |
| 10. Recent Sales of Unregistered Securities. Not Applicable. | |
| 11. Description of Registrant's Securities to be Registered. See "Description of Capital Stock." | |
| 12. Indemnification of Directors and Officers. See "Indemnification of Directors and Officers." | |
| 13. Financial Statements and Supplementary Data. See "Summary," "Selected Historical Combined Financial Data," "Financial Data" and "Index to Financial Statements." | Unaudited Pro Forma Combined |

| Item No. | Item Caption | Location in Information Statement |
|-------------|--|-----------------------------------|
| 14. | Changes in and Disagreements with Accountants on | Not Applicable. |
| | Accounting and Financial Disclosure. | |

- 15. Financial Statements and Exhibits.
- (a) <u>Combined Financial Statements</u>: The following financial statements are included in the information statement and filed as part of this Registration Statement:

Report of Independent Registered Public Accounting Firm

Combined Balance Sheets as of December 31, 2009 and 2008

Combined Statements of Income for the years ended December 31, 2009, 2008 and 2007

Combined Statements of Comprehensive Income (Loss) for the years ended December 31, 2009, 2008 and 2007

Combined Statements of Parent Equity (Deficit) as of December 31, 2009, 2008 and 2007

Combined Statements of Cash Flows for the years ended December 31, 2009, 2008 and 2007

Notes to Combined Financial Statements

Condensed Combined Balance Sheets as of March 31, 2010 and December 31, 2009 (Unaudited)

Condensed Combined Statements of Income for the three months ended March 31, 2010 and 2009 (Unaudited)

Condensed Combined Statements of Comprehensive Income for the three months ended March 31, 2010 and 2009 (Unaudited)

Condensed Combined Statements of Parent Equity (Deficit) as of March 31, 2010 and 2009 (Unaudited)

Condensed Combined Statements of Cash Flows for the three months ended March 31, 2010 and 2009 (Unaudited)

Notes to Condensed Combined Financial Statements (Unaudited)

- (b) <u>Combined Financial Statements Schedules</u>: Schedule II is filed with this information statement. All other schedules for which provision is made in the applicable regulations of the SEC have been omitted because they are not required under the relevant instructions or because the required information is included in the financial statements or the related notes contained in this information statement.
- (c) Exhibits. The following documents are filed as exhibits hereto:

| Exhibit Number | Exhibit Description |
|----------------|--|
| 2.1* | Form of Master Separation Agreement |
| 3.1* | Form of Restated Certificate of Incorporation of the Registrant |
| 3.2* | Form of Amended and Restated Bylaws of the Registrant |
| 10.1* | Form of Tax Sharing Agreement |
| 10.2* | Form of Employee Matters Agreement |
| 10.3 | Form of Transition Services Agreement (McDermott International, Inc. as service provider) |
| 10.4 | Form of Transition Services Agreement (The Babcock & Wilcox Company as service provider) |
| 10.5 | Assumption and Loss Allocation Agreement dated as of May 18, 2010 by and among ACE American Insurance Company and the Ace Affiliates (as defined therein), McDermott International, Inc. and Babcock & Wilcox Holdings, Inc. |

| Exhibit Number | Exhibit Description |
|----------------|---|
| 10.6 | Novation and Assumption Agreement dated as of May 18, 2010 by and among ACE American Insurance Company and the Ace Affiliates (as defined therein), Creole Insurance Company, Ltd. and Boudin Insurance Company, Ltd. |
| 10.7 | Novation and Assumption Agreement dated as of May 18, 2010 by and among McDermott International, Inc., Babcock & Wilcox Holdings, Inc., Boudin Insurance Company, Ltd. and Creole Insurance Company, Ltd. |
| 10.8* | Form of 2010 Long-Term Incentive Plan of The Babcock & Wilcox Company |
| 10.9* | Form of The Babcock & Wilcox Company Executive Incentive Compensation Plan |
| 10.10* | Form of The Babcock & Wilcox Company Management Incentive Compensation Plan |
| 10.11* | Form of Supplemental Executive Retirement Plan of The Babcock & Wilcox Company |
| 10.12* | Restructuring Transaction Retention Agreement between McDermott International, Inc. and John A. Fees dated December 10, 2009 |
| 10.13* | Restructuring Transaction Retention Agreement between McDermott International, Inc. and Michael S. Taff dated December 10, 2009 |
| 10.14* | Form of Restructuring Transaction Retention Agreement between McDermott International, Inc. and certain executive officers (other than Messrs. Fees or Taff) dated December 10, 2009 |
| 10.15* | Form of Restructuring Transaction Retention Agreement between McDermott International, Inc. and certain other employees dated December 10, 2009 |
| 10.16 | Credit Agreement dated as of May 3, 2010, among Babcock & Wilcox Investment Company, the lenders and letter of credit issuers party thereto, and Bank of America, N.A., as administrative agent |
| 10.17 | Pledge and Security Agreement dated as of May 3, 2010, by Babcock & Wilcox Investment Company and certain of its subsidiaries in favor of Bank of America, N.A., as administrative agent |
| 10.18 | Securities Purchase Agreement, dated as of May 25, 2010, by and among USEC Inc., Toshiba Corporation and Babcock & Wilcox Investment Company |
| 21.1* | List of Subsidiaries |
| 99.1 | Information Statement, Subject to Completion, dated June 4, 2010 |

^{*} Previously filed.

SIGNATURES

Pursuant to the requirements of Section 12 of the Securities Exchange Act of 1934, the registrant has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized.

Date: June 4, 2010

THE BABCOCK & WILCOX COMPANY

By: /S/ BRANDON C. BETHARDS

Name: Brandon C. Bethards

Title: President and Chief Executive Officer

EXHIBIT INDEX

| Exhibit Number | Exhibit Description |
|----------------|--|
| 2.1* | Form of Master Separation Agreement |
| 3.1* | Form of Restated Certificate of Incorporation of the Registrant |
| 3.2* | Form of Amended and Restated Bylaws of the Registrant |
| 10.1* | Form of Tax Sharing Agreement |
| 10.2* | Form of Employee Matters Agreement |
| 10.3 | Form of Transition Services Agreement (McDermott International, Inc. as service provider) |
| 10.4 | Form of Transition Services Agreement (The Babcock & Wilcox Company as service provider) |
| 10.5 | Assumption and Loss Allocation Agreement dated as of May 18, 2010 by and among ACE American Insurance Company and the Ace Affiliates (as defined therein), McDermott International, Inc. and Babcock & Wilcox Holdings, Inc. |
| 10.6 | Novation and Assumption Agreement dated as of May 18, 2010 by and among ACE American Insurance Company and the Ace Affiliates (as defined therein), Creole Insurance Company, Ltd. and Boudin Insurance Company, Ltd. |
| 10.7 | Novation and Assumption Agreement dated as of May 18, 2010 by and among McDermott International, Inc., Babcock & Wilcox Holdings, Inc., Boudin Insurance Company, Ltd. and Creole Insurance Company, Ltd. |
| 10.8* | Form of 2010 Incentive Compensation Plan of The Babcock & Wilcox Company |
| 10.9* | Form of The Babcock & Wilcox Company Executive Incentive Compensation Plan |
| 10.10* | Form of The Babcock & Wilcox Company Management Incentive Compensation Plan |
| 10.11* | Form of Supplemental Executive Retirement Plan of The Babcock & Wilcox Company |
| 10.12* | Restructuring Transaction Retention Agreement between McDermott International, Inc. and John A. Fees dated December 10, 2009 |
| 10.13* | Restructuring Transaction Retention Agreement between McDermott International, Inc. and Michael S. Taff dated December 10, 2009 |
| 10.14* | Form of Restructuring Transaction Retention Agreement between McDermott International, Inc. and certain executive officers (other than Messrs. Fees or Taff) dated December 10, 2009 |
| 10.15* | Form of Restructuring Transaction Retention Agreement between McDermott International, Inc. and certain other employees dated December 10, 2009 |
| 10.16 | Credit Agreement dated as of May 3, 2010, among Babcock & Wilcox Investment Company, the lenders and letter of credit issuers party thereto, and Bank of America, N.A., as administrative agent |
| 10.17 | Pledge and Security Agreement dated as of May 3, 2010, by Babcock & Wilcox Investment Company and certain of its subsidiaries in favor of Bank of America, N.A., as administrative agent |
| 10.18 | Securities Purchase Agreement, dated as of May 25, 2010, by and among USEC Inc., Toshiba Corporation and Babcock & Wilcox Investment Company |
| 21.1* | List of Subsidiaries |
| 99.1 | Information Statement, Subject to Completion, dated June 4, 2010 |
| | |

Previously filed.

TRANSITION SERVICES AGREEMENT

BETWEEN

 ${\bf MCDERMOTT\ INTERNATIONAL,\ INC.}$

(as service provider)

and

THE BABCOCK & WILCOX COMPANY

(as service receiver)

Dated [], 2010

TABLE OF CONTENTS

| | | Page No. |
|--------------|--|----------|
| ARTICLE I | DEFINITIONS | 1 |
| Section 1.1 | Definitions | 1 |
| ARTICLE II | SERVICES | 2 |
| Section 2.1 | Services | 2 |
| Section 2.2 | Service Coordinators | 3 |
| Section 2.3 | Additional Services | 3 |
| Section 2.4 | Third Party Services | 3 |
| Section 2.5 | Standard of Performance; Limitation of Liability | 4 |
| Section 2.6 | Service Boundaries and Scope | 5 |
| Section 2.7 | Cooperation | 6 |
| Section 2.8 | Transitional Nature of Services; Changes | 6 |
| Section 2.9 | Access | 6 |
| ARTICLE III | SERVICE CHARGES | 6 |
| Section 3.1 | Compensation | 6 |
| ARTICLE IV | PAYMENT | 6 |
| Section 4.1 | Payment | 6 |
| Section 4.2 | Payment Disputes | 7 |
| Section 4.3 | Review of Charges; Error Correction | 7 |
| Section 4.4 | Taxes | 7 |
| Section 4.5 | Records | 8 |
| ARTICLE V | TERM | 8 |
| Section 5.1 | Term | 8 |
| ARTICLE VI | DISCONTINUATION OF SERVICES | 8 |
| Section 6.1 | Discontinuation of Services | 8 |
| Section 6.2 | Procedures Upon Discontinuation or Termination of Services | 8 |
| ARTICLE VII | DEFAULT | 9 |
| Section 7.1 | Termination for Default | 9 |
| ARTICLE VIII | INDEMNIFICATION AND WAIVER | 9 |
| Section 8.1 | Waiver of Consequential Damages | 9 |
| Section 8.2 | Services Received | 9 |
| Section 8.3 | Express Negligence | 10 |
| ARTICLE IX | CONFIDENTIALITY | 11 |
| Section 9.1 | Confidentiality | 11 |
| Section 9.2 | System Security | 11 |
| ARTICLE X | FORCE MAJEURE | 12 |
| Section 10.1 | Performance Excused | 12 |
| Section 10.2 | Notice | 12 |
| Section 10.3 | Cooperation | 12 |

i

| ARTICLE XI | MISCELLANEOUS | 12 |
|---------------|--|----|
| Section 11.1 | Entire Agreement | 12 |
| Section 11.2 | Binding Effect; No Third-Party Beneficiaries; Assignment | 12 |
| Section 11.3 | Amendment; Waivers | 13 |
| Section 11.4 | Notices | 13 |
| Section 11.5 | Counterparts | 13 |
| Section 11.6 | Severability | 13 |
| Section 11.7 | Governing Law | 13 |
| Section 11.8 | Performance | 14 |
| Section 11.9 | Relationship of Parties | 14 |
| Section 11.10 | Regulations | 14 |
| Section 11.11 | Construction | 14 |
| Section 11.12 | Effect if Separation does not Occur | 14 |
| | | |

Schedules

| Schedule A - Ta | x Services |
|-----------------|------------|
|-----------------|------------|

- Schedule B Accounting and Financial Reporting Services
 Schedule C Executive Consulting Services
 Schedule D Legal Services

- Schedule E Human Resources Services
- Schedule F Risk Management Services Schedule G Treasury Services
- Schedule H -Investor Relations Services
- Schedule I Litigation Services
- Schedule J Services for Sales Office in Singapore

- Schedule K Services for MPLS and Facility Access in Houston
 Schedule L Office Space
 Schedule M Offiste and Third Party Records Storage and Destruction
- Schedule N Office in McDermott's New Orleans Facility
- Schedule 2.4 Certain Subcontractors
- Schedule 4.1 Payment Instructions

TRANSITION SERVICES AGREEMENT

This TRANSITION SERVICES AGREEMENT (together with the Schedules hereto, this "Agreement") is entered into as of [], 2010, by and between McDermott International, Inc., a Panamanian corporation ("McDermott"), and The Babcock & Wilcox Company, a Delaware corporation ("B&W").

WHEREAS, the Board of Directors of McDermott has determined that it would be appropriate and desirable for McDermott to distribute (the "Distribution") on a pro rata basis to the holders of outstanding shares of common stock, par value \$1.00 per share, of McDermott all of the outstanding shares of common stock, par value \$0.01 per share, of B&W owned by McDermott;

WHEREAS, in order to effectuate the foregoing, McDermott and B&W have entered into a Master Separation Agreement, dated as of the date hereof (the "Master Separation Agreement"), which provides, among other things, upon the terms and subject to the conditions thereof, for the separation of the respective businesses of McDermott and B&W and the Distribution, and the execution and delivery of certain other agreements, including this Agreement, in order to facilitate and provide for the foregoing; and

WHEREAS, in order to provide for an orderly transition under the Master Separation Agreement, it will be advisable for McDermott, through members of the MII Group, to provide to B&W certain services described herein for a transitional period.

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

ARTICLE I DEFINITIONS

Section 1.1 Definitions. As used in this Agreement, the following terms shall have the meanings set forth below:

- "Additional Services" has the meaning set forth in Section 2.3.
- "Agreement" has the meaning set forth in the preamble.
- "Availed Party" has the meaning set forth in Section 9.2(a).
- "B&W" has the meaning set forth in the preamble.
- "Distribution" has the meaning set forth in the recitals.
- "Force Majeure Event" has the meaning set forth in Section 10.1.

- "Master Separation Agreement" has the meaning set forth in the recitals.
- "McDermott" has the meaning set forth in the preamble.
- "Schedules" means the Schedules attached hereto.
- "Security Regulations" has the meaning set forth in Section 9.2(a).
- "Service Coordinator" has the meaning set forth in Section 2.2.
- "Services" has the meaning set forth in Section 2.1(a).
- "Systems" has the meaning set forth in Section 9.2(a).
- "Tax" has the meaning set forth in Section 4.4.

Capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings assigned to such terms in the Master Separation Agreement.

ARTICLE II SERVICES

Section 2.1 Services.

- (a) Upon the terms and subject to the conditions of this Agreement, McDermott, acting directly and/or through its Affiliates and their respective employees, agents, contractors or independent third parties designated by any of them, agrees to use commercially reasonable efforts to provide or to cause to be provided services to the B&W Group as set forth in Schedules A through N (including any Additional Services provided in accordance with Section 2.3 hereof, all such services are collectively referred to herein as the "Services").
- (b) At all times during the performance of the Services, all Persons performing such Services (including agents, temporary employees, independent third parties and consultants) shall be construed as being independent from the B&W Group, and such Persons shall not be considered or deemed to be employees of any member of the B&W Group nor entitled to any employee benefits of B&W as a result of this Agreement. The responsibility of such Persons is to perform the Services in accordance with this Agreement and, as necessary, to advise the applicable member of the B&W Group in connection therewith, and such Persons shall not be responsible for decision-making on behalf of any member of the B&W Group. Such Persons shall not be required to report to management of any member of the B&W Group nor be deemed to be under the management or direction of any member of the B&W Group. B&W acknowledges and agrees that, except as may be expressly set forth herein as a Service (including any Additional Services provided in accordance with Section 2.3 hereof) or otherwise expressly set forth in the Master Separation Agreement or an Ancillary Agreement, no member of the MII Group shall be obligated to provide, or cause to be provided, any service or goods to any member of the B&W Group.

(c) Notwithstanding anything to the contrary in this Agreement, McDermott and members of the MII Group shall not be required to perform Services hereunder or take any actions relating thereto that conflict with or violate any applicable law, contract, license, authorization, certification or permit or McDermott's Code of Business Conduct or other governance policies, as they may be amended from time to time.

Section 2.2 Service Coordinators. Each party will nominate in writing a representative to act as the primary contact with respect to the provision of the Services and the resolution of disputes under this Agreement (each such person, a "Service Coordinator"). The initial Service Coordinators shall be Mr. Gary Brauchle (for McDermott) and Mr. Keith Robinson (for B&W) (or their designated delegates) for each of McDermott and B&W, respectively. The Service Coordinators shall meet as expeditiously as possible to resolve any dispute hereunder; and any dispute that is not resolved by the Service Coordinators within 45 days shall be resolved in accordance with the dispute resolution procedures set forth in Article V of the Master Separation Agreement. Each party hereto may treat an act of a Service Coordinator of the other party hereto which is consistent with the provisions of this Agreement as being authorized by such other party without inquiring behind such act or ascertaining whether such Service Coordinator had authority to so act; provided, however, that no such Service Coordinator shall have authority to amend this Agreement. McDermott and B&W shall advise each other promptly (in any case no more than three Business Days) in writing of any change in their respective Service Coordinators, setting forth the name of the replacement, and stating that the replacement Service Coordinator is authorized to act for such party in accordance with this Section 2.2.

Section 2.3 Additional Services. B&W may request additional Services (the "Additional Services") from McDermott by providing written notice. Upon the mutual written agreement as to the nature, cost, duration and scope of such Additional Services, McDermott and B&W shall supplement in writing the Schedules hereto to include such Additional Services. Subject to the other limitations in this Agreement, including the provisions in Section 2.6, but notwithstanding the foregoing provisions of this Section 2.3, in addition to providing the Services specified in the Schedules, McDermott, acting directly and/or through its Affiliates and their respective employees, agents, contractors or independent third parties designated by any of them, shall use commercially reasonable efforts to provide or to cause to be provided additional, de minimis administrative support services to the B&W Group as may be requested by any member of the B&W Group from time to time, at no cost beyond the amounts set forth in the Schedules (as the amounts set forth in the Schedules contemplate such additional, de minimis administrative support services); provided, however, that, for any such additional services to be considered de minimis for purposes of this sentence, such additional services shall not require the attention of (i) any one employee of any member of the McDermott Group for more than 2 hours in any single calendar month or (ii) any group of employees of any one or more members of the McDermott Group for more than 30 hours in any single calendar month. Except where the context otherwise indicates or requires, any such additional services referred to in the immediately preceding sentence shall be deemed to be "Services" under this Agreement.

Section 2.4 Third Party Services. McDermott shall have the right to hire third-party subcontractors to provide all or part of any Service hereunder; provided, that McDermott shall consult in good faith with B&W regarding the proposed hiring of any third-party subcontractor that has not previously been involved in the activities relating to such Service prior to the date

hereof; provided, further, that, in the event such subcontracting is inconsistent with the practice applied by McDermott generally from time to time within its own organization, McDermott shall give notice to B&W of its intent to subcontract any portion of the Services and B&W shall have 20 days (or such lesser period set forth in the notice as may be practicable in the event of exigent circumstances) to determine, in its sole discretion, whether to permit such subcontracting or whether to cancel such Service in accordance with Article VI hereof. If B&W opts to cancel a Service pursuant to the immediately preceding sentence, it shall not be liable to McDermott pursuant to Section 6.1 for any costs or expenses McDermott or any member of the MII Group remains obligated to pay to the third-party subcontractor identified in the notice provided by McDermott as described above. McDermott shall not be required to give notice of its intent to subcontract Services to any party listed on Exhibit 2.4 hereto, nor shall B&W have any right to cancel any Service subcontracted to any such listed party pursuant to this Section 2.4 (provided, that this sentence shall not prevent B&W from cancelling any Service pursuant to Section 6.1).

Section 2.5 Standard of Performance; Limitation of Liability.

- (a) The Services to be provided hereunder shall be performed with the same general degree of care, at the same general level and at the same general degree of accuracy and responsiveness, as when performed within the McDermott organization (including, for this purpose, B&W and its subsidiaries) prior to the date of this Agreement. It is understood and agreed that McDermott and the members of the MII Group are not professional providers of the types of services included in the Services and that McDermott personnel performing Services have other responsibilities and will not be dedicated full-time to performing Services hereunder.
- (b) In the event McDermott or any member of the MII Group fails to provide, or cause to be provided, the Services in accordance with the standard of service set forth in Section 2.5(a) or Section 2.5(c), the sole and exclusive remedy of B&W shall be, at B&W's sole discretion, within 90 days from the date that McDermott or such member of the MII Group first fails to provide such Service, to not pay for such Service; *provided* that in the event McDermott defaults in the manner described in clause (ii) of Section 7.1, B&W shall have the further rights set forth in Article VII.
- (c) EXCEPT AS EXPRESSLY SET FORTH IN THIS SECTION 2.5, NO REPRESENTATIONS OR WARRANTIES OF ANY KIND, EXPRESSED OR IMPLIED (INCLUDING THE WARRANTIES OF NON-INFRINGEMENT, MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, OR CONFORMITY TO ANY REPRESENTATION OR DESCRIPTION), ARE MADE BY MCDERMOTT OR ANY MEMBER OF THE MII GROUP WITH RESPECT TO THE SERVICES UNDER THIS AGREEMENT AND, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ALL SUCH REPRESENTATIONS OR WARRANTIES ARE HEREBY WAIVED AND DISCLAIMED. B&W (ON ITS OWN BEHALF AND ON BEHALF OF EACH OTHER MEMBER OF THE B&W GROUP) HEREBY EXPRESSLY WAIVES ANY RIGHT B&W OR ANY MEMBER OF THE B&W GROUP MAY OTHERWISE HAVE FOR ANY LOSSES, TO ENFORCE SPECIFIC PERFORMANCE OR TO PURSUE ANY OTHER REMEDY AVAILABLE IN CONTRACT, AT LAW OR IN EQUITY IN THE EVENT OF ANY NON-PERFORMANCE, INADEQUATE PERFORMANCE, FAULTY PERFORMANCE OR OTHER FAILURE OR BREACH BY MCDERMOTT OR ANY MEMBER OF THE MII

GROUP UNDER OR RELATING TO THIS AGREEMENT, NOTWITHSTANDING THE NEGLIGENCE OR GROSS NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT OR ACTIVE OR PASSIVE) OF MCDERMOTT OR ANY MEMBER OF THE MII GROUP OR ANY THIRD PARTY SERVICE PROVIDER AND WHETHER DAMAGES ARE ASSERTED IN CONTRACT OR TORT, UNDER FEDERAL, STATE OR NON U.S. LAWS OR OTHER STATUTE OR OTHERWISE; PROVIDED, HOWEVER, THAT THE FOREGOING WAIVER SHALL NOT EXTEND TO COVER, AND MCDERMOTT SHALL BE RESPONSIBLE FOR, SUCH LOSSES CAUSED BY THE WILLFUL MISCONDUCT OF MCDERMOTT OR ANY MEMBER OF THE MII GROUP. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, IN NO EVENT SHALL THE MII GROUP BE LIABLE TO THE B&W GROUP WITH RESPECT TO CLAIMS ARISING OUT OF THIS AGREEMENT FOR AMOUNTS IN THE AGGREGATE EXCEEDING THE AGGREGATE SERVICE CHARGES PAID HEREUNDER BY THE B&W GROUP.

Section 2.6 Service Boundaries and Scope. Except as provided in a Schedule for a specific Service: (a) McDermott shall be required to provide, or cause to be provided, the Services only at the locations such Services are being provided by any member of the MII Group for any member of the B&W Group immediately prior to the Distribution Date; provided, however, that, to the extent any such Service is to be provided by an employee of McDermott who works in the corporate headquarters of McDermott, such Service shall, to the extent feasible, only be provided by such employee from the corporate headquarters of McDermott; and (b) the Services shall be available only for purposes of conducting the business of the B&W Group substantially in the manner it was conducted immediately prior to the Distribution Date. Except as provided in a Schedule for a specific Service, in providing, or causing to be provided, the Services, McDermott shall not be obligated to: (i) maintain the employment of any specific employee or hire additional employees or third-party service providers; (ii) purchase, lease or license any additional equipment (including computer equipment, furniture, furnishings, fixtures, machinery, vehicles, tools and other tangible personal property), software or other assets, rights or properties; (iii) make modifications to its existing systems or software; (iv) provide any member of the B&W Group with access to any systems or software other than those to which it has authorized access immediately prior to the Distribution Date; or (v) pay any costs related to the transfer or conversion of data of any member of the B&W Group. B&W acknowledges (on its own behalf and on behalf of the other members of the B&W Group) that the employees of McDermott or any other members of the MII Group who may be assisting in the provision of Services hereunder are at-will employees and, as such, may terminate or be terminated from employment with McDermott or any of the other members of the MII Group providing Services hereunder at any time for any reason. In no event shall McDermott or any of its Affiliates or any of their respective employees or agents be required to perform any Services or take any other actions hereunder that conflict with any applicable Law. For the avoidance of doubt and except as may hereafter be designated as Additional Services in accordance with Section 2.3, the Services do not include any services required for or as the result of any business acquisitions, divestitures, start-ups or terminations by the B&W Group. To the extent B&W desires McDermott to provide any services in connection with any such acquisitions, divestitures, start-ups or terminations, B&W shall follow the procedures for requesting Additional Services pursuant to

- Section 2.7 Cooperation. McDermott and B&W shall cooperate with one another and provide such further assistance as the other party may reasonably request in connection with the provision of Services hereunder.
- Section 2.8 *Transitional Nature of Services; Changes.* Subject to Sections 2.3 and 2.5, the parties acknowledge the transitional nature of the Services and that McDermott may make changes from time to time in the manner of performing the Services.
- Section 2.9 Access. During the term of this Agreement and for so long as any Services are being provided to B&W by McDermott, B&W will provide McDermott and its authorized representatives reasonable access, during regular business hours upon reasonable notice, to B&W and its employees, representatives, facilities and books and records as McDermott and its representatives may reasonably require in order to perform such Services.

ARTICLE III SERVICE CHARGES

Section 3.1 Compensation. Subject to the specific terms of this Agreement, the compensation to be received by McDermott for each Service provided hereunder will be the fees set forth on the Schedule relating to the particular Service, subject to any escalation provided for on such Schedule. In consideration for the provision of a Service, each member of the B&W Group receiving such Service shall pay to McDermott or, at the election of McDermott, the member of the MII Group providing such Service, the applicable fee for such Service as set forth on the attached Schedules.

ARTICLE IV PAYMENT

Section 4.1 Payment. Except as otherwise provided in a Schedule for a specific Service, charges for Services shall be invoiced monthly by McDermott or, at its option, the member of the MII Group providing the Service. Except as otherwise provided in a Schedule for a specific Service, B&W shall make the corresponding payment no later than 60 days after receipt of the invoice. Unless otherwise provided in this Agreement, B&W shall remit funds in payment of invoices provided hereunder either by wire transfer or ACH (Automated Clearing House) in accordance with the payment instructions set forth in Schedule 4.1. Each invoice shall be directed to the B&W Service Coordinator or such other person designated in writing from time to time by such Service Coordinator. The invoice shall set forth in reasonable detail the Services rendered and the invoice amount for the Services rendered for the period covered by such invoice. Interest will accrue on any unpaid amounts at ten percent (10%) per annum (compounded monthly) or, if less, the maximum non-usurious rate of interest permitted by applicable law, until such amounts, together with all accrued and unpaid interest thereon, are paid in full. All timely payments under this Agreement shall be made without early payment discount. Any preexisting obligation to make payment for Services provided hereunder shall survive the termination of this Agreement. If McDermott incurs any reasonable out-of-pocket expenses (including any incremental license fees incurred by McDermott in connection with performance of the Services and any travel expenses incurred at the request or with the consent of B&W) or remits funds to a third-party on behalf of B&W, in either case in connection with

the rendering of Services, then McDermott shall include such amount on its monthly invoice to B&W, with reasonable supporting documentation, and B&W shall reimburse that amount to McDermott pursuant to this Section 4.1 as part of its next monthly payment.

- Section 4.2 Payment Disputes. B&W may object to any amounts for any Service at any time before, at the time of, or after payment is made, provided such objection is made in writing to McDermott within 120 days following the date of the disputed invoice. B&W shall timely pay the disputed items in full while resolution of the dispute is pending; provided, however, that McDermott shall pay interest at a rate of five percent (5%) per annum (compounded monthly) on any amounts it is required to return to B&W upon resolution of the dispute. Payment of any amount shall not constitute approval thereof. The Service Coordinators shall meet as expeditiously as possible to resolve any dispute. Any dispute that is not resolved by the Service Coordinators within 45 days shall be resolved in accordance with the dispute resolution and arbitration procedures set forth in Article V of the Master Separation Agreement. Neither party (or any member of its respective Group) shall have a right of set-off against the other party (or any member of its respective Group) for billed amounts hereunder. Upon written request, McDermott will provide to B&W reasonable detail and support documentation to permit B&W to verify the accuracy of an invoice.
- Section 4.3 Review of Charges; Error Correction. McDermott shall maintain accurate books and records (including invoices of third parties) related to the Services sufficient to calculate, and allow B&W to verify, the amounts owed under this Agreement. From time to time until 120 days following the termination of this Agreement, B&W shall have the right to review, and McDermott shall provide access to, such books and records to verify the accuracy of such amounts, provided that such reviews shall not occur more frequently than once per calendar quarter. Each such review shall be conducted during normal business hours and in a manner that does not unreasonably interfere with the operations of McDermott. If, as a result of any such review, B&W determines that it overpaid any amount to McDermott, then B&W may raise an objection pursuant to the provisions of Section 4.2. B&W shall bear the cost and expense of any such review. McDermott shall make adjustments to charges as required to reflect the discovery of errors or omissions in charges.
- Section 4.4 Taxes. All transfer taxes, excises, fees or other charges (including value added, sales, use or receipts taxes, but not including a tax on or measured by the income, net or gross revenues, business activity or capital of a member of the MII Group), or any increase therein, now or hereafter imposed directly or indirectly by law upon any fees paid hereunder for Services, which a member of the MII Group is required to pay or incur in connection with the provision of Services hereunder ("Tax"), shall be passed on to B&W as an explicit surcharge and shall be paid by B&W in addition to any Service fee payment, whether included in the applicable Service fee payment, or added retroactively. If B&W submits to McDermott a timely and valid resale or other exemption certificate acceptable to McDermott and sufficient to support the exemption from Tax, then such Tax will not be added to the Service fee payable pursuant to Article III; provided, however, that if a member of the MII Group is ever required to pay such Tax, B&W will promptly reimburse McDermott for such Tax, including any interest, penalties and attorney's fees related thereto. The parties will cooperate to minimize the imposition of any Taxes.

Section 4.5 Records. McDermott shall maintain true and correct records of all receipts, invoices, reports and such other documents relating to the Services hereunder in accordance with its standard accounting practices and procedures, consistently applied. McDermott shall retain such accounting records and make them available to B&W's authorized representatives and auditors for a period of not less than one year from the close of each fiscal year of McDermott; provided, however, that McDermott may, at its option, transfer such accounting records to B&W upon termination of this Agreement.

ARTICLE V TERM

Section 5.1 Term. Subject to Articles VI and VII, the MII Group shall provide the specific Services to the B&W Group pursuant to this Agreement for the time period set forth on the Schedule relating to the specific Service. In accordance with the Master Separation Agreement and Article VI of this Agreement, except as otherwise provided in a Schedule for a specific Service, B&W shall undertake to provide to itself and members of the B&W Group, and to terminate as soon as reasonably practicable, the Services provided to the B&W Group hereunder. Except as otherwise provided in a Schedule for a specific Service or group of related Services, all Services provided for hereunder shall terminate on March 31, 2011. Except as otherwise expressly agreed or unless sooner terminated, this Agreement shall commence upon the Distribution Date and shall continue in full force and effect between the parties for so long as any Service set forth in any Schedule hereto is being provided to B&W or members of the B&W Group and this Agreement shall terminate upon the cessation of all Services provided hereunder; provided that Articles I, IV, VIII, IX and XI and Section 2.5(c) will survive the termination of this Agreement and any such termination shall not affect any obligation for the payment of Services rendered prior to termination.

ARTICLE VI DISCONTINUATION OF SERVICES

Section 6.1 Discontinuation of Services. Unless otherwise provided in the relevant Schedule for a particular Service, at any time after the Distribution Date, B&W may, without cause and in accordance with the terms and conditions hereunder and the Master Separation Agreement, request the discontinuation of one or more specific Services by giving McDermott at least 30 days' prior written notice; provided, however, that any such discontinuation will not affect the amounts payable to McDermott hereunder unless (and then only to the extent that) the charges for the discontinued Services have been separately identified in the applicable Schedule. B&W shall be liable to McDermott for all costs and expenses McDermott or any member of the MII Group remains obligated to pay in connection with any discontinued Service or Services, except in the case of a Service terminated by B&W pursuant to clause (ii) of the first sentence of Section 7.1 hereof. The parties shall cooperate as reasonably required to effectuate an orderly and systematic transfer to the B&W Group of all of the duties and obligations previously performed by McDermott or a member of the MII Group under this Agreement.

Section 6.2 Procedures Upon Discontinuation or Termination of Services. Upon the discontinuation or termination of a Service hereunder, this Agreement shall be of no further force and effect with respect to such Service, except as otherwise provided in a Schedule for a specific

Service and except as to obligations accrued prior to the date of discontinuation or termination; provided, however, that Articles I, IV, VIII, IX and XI and Section 2.5(c) of this Agreement shall survive such discontinuation or termination. Each party and the applicable member(s) of its respective Group shall, within 60 days after discontinuation or termination of a Service, deliver to the other party and the applicable member(s) of its respective Group originals of all books, records, contracts, receipts for deposits and all other papers or documents in its possession which pertain exclusively to the business of the other party and relate to such Service; provided that a party may retain copies of material provided to the other party pursuant to this Section 6.2 as it deems necessary or appropriate in connection with its financial reporting obligations or internal control practices and policies.

ARTICLE VII DEFAULT

Section 7.1 Termination for Default. In the event (i) of a failure of B&W to pay for Services in accordance with the terms of this Agreement, or (ii) any party shall default, in any material respect, in the due performance or observance by it of any of the other terms, covenants or agreements contained in this Agreement, then (1) if the non-defaulting party is McDermott, McDermott shall have the right, at its sole discretion, to immediately terminate the Service with respect to which the default occurred, and (2) if the non-defaulting party is B&W, B&W shall have the right, at its sole discretion, to immediately terminate the Service with respect to which the default occurred, in either case if the defaulting party has failed to cure the default within 30 days of receipt of the written notice of such default. B&W's right to terminate this Agreement pursuant to this Article VII and the rights set forth in Section 2.5 shall constitute B&W's sole and exclusive rights and remedies for a breach by McDermott hereunder (including any breach caused by an Affiliate of McDermott or other third party providing a Service hereunder).

ARTICLE VIII INDEMNIFICATION AND WAIVER

Section 8.1 Waiver of Consequential Damages. NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY UNDER THIS AGREEMENT FOR ANY EXEMPLARY, PUNITIVE, SPECIAL, INDIRECT, CONSEQUENTIAL, REMOTE OR SPECULATIVE DAMAGES (INCLUDING IN RESPECT OF LOST PROFITS OR REVENUES), HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY (INCLUDING NEGLIGENCE OR GROSS NEGLIGENCE) ARISING IN ANY WAY OUT OF THIS AGREEMENT, WHETHER OR NOT SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES; PROVIDED, HOWEVER, THAT THE FOREGOING LIMITATIONS SHALL NOT LIMIT EACH PARTY'S INDEMNIFICATION OBLIGATIONS FOR LIABILITIES TO THIRD PARTIES AS SET FORTH IN THIS AGREEMENT, THE MASTER SEPARATION AGREEMENT OR ANY ANCILLARY AGREEMENT.

Section 8.2 Services Received. B&W hereby acknowledges and agrees that:

(a) the Services to be provided hereunder are subject to and limited by the provisions of Section 2.5, Article VII and the other provisions hereof, including the limitation of remedies available to B&W that restricts available remedies resulting from a Service not

provided in accordance with the terms hereof to non-payment and, in certain limited circumstances, the right to terminate this Agreement;

- (b) the Services are being provided solely to facilitate the transition of B&W to a separate company as a result of the Distribution, and McDermott and its Affiliates do not provide any such Services to non-Affiliates;
- (c) it is not the intent of McDermott and the other members of the MII Group to render, nor of B&W and the other members of the B&W Group to receive from McDermott and the other members of the MII Group, professional advice or opinions, whether with regard to tax, legal, treasury, finance, employment or other business and financial matters, or technical advice, whether with regard to information technology or other matters; B&W shall not rely on, or construe, any Service rendered by or on behalf of McDermott as such professional advice or opinions or technical advice; and B&W shall seek all third-party professional advice and opinions or technical advice as it may desire or need, and in any event B&W shall be responsible for and assume all risks associated with the Services, except to the limited extent set forth in Section 2.5 and Article VII;
- (d) with respect to any software or documentation within the Services, B&W shall use such software and documentation internally and for their intended purpose only, shall not distribute, publish, transfer, sublicense or in any manner make such software or documentation available to other organizations or persons, and shall not act as a service bureau or consultant in connection with such software; and
- (e) a material inducement to McDermott's agreement to provide the Services is the limitation of liability and the release provided by B&W in this Agreement.

ACCORDINGLY, EXCEPT WITH REGARD TO THE LIMITED REMEDIES EXPRESSLY SET FORTH HEREIN, B&W SHALL ASSUME ALL LIABILITY FOR AND SHALL FURTHER RELEASE, DEFEND, INDEMNIFY AND HOLD MCDERMOTT, ANY MEMBER OF THE MII GROUP AND THEIR RESPECTIVE EMPLOYEES, OFFICERS, DIRECTORS AND AGENTS (ALL AS INDEMNIFIED PARTIES) FREE AND HARMLESS FROM AND AGAINST ALL LOSSES RESULTING FROM, ARISING OUT OF OR RELATED TO THE SERVICES, HOWSOEVER ARISING AND WHETHER OR NOT CAUSED BY THE NEGLIGENCE OR GROSS NEGLIGENCE OF MCDERMOTT, ANY MEMBER OF THE MII GROUP OR ANY THIRD PARTY SERVICE PROVIDER, OTHER THAN THOSE LOSSES CAUSED BY THE WILLFUL MISCONDUCT OF MCDERMOTT OR ANY MEMBER OF THE MII GROUP.

Section 8.3 Express Negligence. THE INDEMNITY, RELEASES AND LIMITATIONS OF LIABILITY IN THIS AGREEMENT (INCLUDING ARTICLES II AND VIII) ARE INTENDED TO BE ENFORCEABLE AGAINST THE PARTIES IN ACCORDANCE WITH THE EXPRESS TERMS AND SCOPE THEREOF NOTWITHSTANDING ANY EXPRESS NEGLIGENCE RULE OR ANY SIMILAR DIRECTIVE THAT WOULD PROHIBIT OR OTHERWISE LIMIT INDEMNITIES BECAUSE OF THE NEGLIGENCE OR GROSS NEGLIGENCE (WHETHER SOLE,

JOINT OR CONCURRENT OR ACTIVE OR PASSIVE) OR OTHER FAULT OR STRICT LIABILITY OF ANY OF THE INDEMNIFIED PARTIES.

ARTICLE IX CONFIDENTIALITY

Section 9.1 Confidentiality. B&W and McDermott each acknowledge and agree that the terms of Section 6.9 of the Master Separation Agreement shall apply to information, documents, plans and other data made available or disclosed by one party to the other in connection with this Agreement. B&W and McDermott each acknowledge and agree that any third party Information (to the extent such Information does not constitute McDermott Books and Records) provided by any member of the B&W Group to any member of the MII Group after the Distribution Date in connection with the provision of the Services by any member of the MII Group after the Distribution Date, in each case that primarily relates to the B&W Business, the B&W Assets, or the B&W Liabilities, shall not be considered Privileged Information of McDermott or Confidential Information of McDermott.

Section 9.2 System Security.

- (a) If any party hereto is given access to the other party's computer systems or software (collectively, the "Systems") in connection with the Services, the party given access (the "Availed Party") shall comply with all of the other party's system security policies, procedures and requirements that have been provided to the Availed Party in advance and in writing (collectively, "Security Regulations"), and shall not tamper with, compromise or circumvent any security or audit measures employed by such other party. The Availed Party shall access and use only those Systems of the other party for which it has been granted the right to access and use.
- (b) Each party hereto shall use commercially reasonable efforts to ensure that only those of its personnel who are specifically authorized to have access to the Systems of the other party gain such access, and use commercially reasonable efforts to prevent unauthorized access, use, destruction, alteration or loss of information contained therein, including notifying its personnel of the restrictions set forth in this Agreement and of the Security Regulations.
- (c) If, at any time, the Availed Party determines that any of its personnel has sought to circumvent, or has circumvented, the Security Regulations, that any unauthorized Availed Party personnel has accessed the Systems, or that any of its personnel has engaged in activities that may lead to the unauthorized access, use, destruction, alteration or loss of data, information or software of the other party hereto, the Availed Party shall promptly terminate any such person's access to the Systems and immediately notify the other party hereto. In addition, such other party hereto shall have the right to deny personnel of the Availed Party access to its Systems upon notice to the Availed Party in the event that the other party hereto reasonably believes that such personnel have engaged in any of the activities set forth above in this Section 9.2(c) or otherwise pose a security concern. The Availed Party shall use commercially reasonable efforts to cooperate with the other party hereto in investigating any apparent unauthorized access to such other party's Systems.

ARTICLE X FORCE MAJEURE

Section 10.1 Performance Excused. Continued performance of a Service may be suspended immediately to the extent caused by any event or condition beyond the reasonable control of the party suspending such performance (and not involving any willful misconduct of such party), including acts of God, pandemics, floods, fire, earthquakes, labor or trade disturbances, strikes, war, acts of terrorism, civil commotion, electrical shortages or blackouts, breakdown or injury to computing facilities, compliance in good faith with any Law (whether or not it later proves to be invalid), unavailability of materials or bad weather (a "Force Majeure Event"). B&W shall not be obligated to pay any amount for Services that it does not receive as a result of a Force Majeure Event (and the parties hereto shall negotiate reasonably to determine the amount applicable to such Services not received). In addition to the reduction of any amounts owed by B&W hereunder, during the occurrence of a Force Majeure Event, to the extent the provision of any Service has been disrupted or reduced, during such disruption or reduction, (a) B&W may replace any such affected Service by providing any such Service for itself or engaging one or more third parties to provide such Service at the expense of B&W and (b) McDermott shall cooperate with, provide such information to and take such other actions as may be reasonably required to assist such third parties to provide such substitute Service

Section 10.2 Notice. The party claiming suspension due to a Force Majeure Event will give prompt notice to the other of the occurrence of the Force Majeure Event giving rise to the suspension and of its nature and anticipated duration.

Section 10.3 Cooperation. Upon the occurrence of a Force Majeure Event, the parties shall cooperate with each other to find alternative means and methods for the provision of the suspended Service.

ARTICLE XI MISCELLANEOUS

Section 11.1 Entire Agreement. This Agreement, together with the documents referenced herein (including the Master Separation Agreement), constitutes the entire agreement and understanding between the parties hereto with respect to the subject matter hereof and supersedes all prior written and oral and all contemporaneous oral agreements and understandings with respect to the subject matter hereof. To the extent any provision of this Agreement conflicts with the provisions of the Master Separation Agreement, the provisions of this Agreement shall be deemed to control with respect to the subject matter hereof.

Section 11.2 Binding Effect; No Third-Party Beneficiaries; Assignment. This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns; and nothing in this Agreement, express or implied, is intended to confer upon any other person or entity any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement. This Agreement may not be assigned by either party hereto, except with the prior written consent of the other party hereto.

Section 11.3 Amendment; Waivers. No change or amendment may be made to this Agreement except by an instrument in writing signed on behalf of both of the parties hereto. Either party hereto may, at any time, (i) extend the time for the performance of any of the obligations or other acts of the other, (ii) waive any inaccuracies in the representations and warranties of the other contained herein or in any document delivered pursuant hereto, and (iii) waive compliance by the other with any of the agreements, covenants or conditions contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the party to be bound thereby. No failure or delay on the part of either party hereto in the exercise of any right hereunder shall impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty, covenant or agreement contained herein, nor shall any single or partial exercise of any such right preclude other or further exercise thereof or of any other right.

Section 11.4 Notices. Unless otherwise expressly provided herein, all notices, claims, certificates, requests, demands and other communications hereunder shall be in writing and shall be deemed to be duly given (i) when personally delivered or (ii) if mailed by registered or certified mail, postage prepaid, return receipt requested, on the date the return receipt is executed or the letter is refused by the addressee or its agent or (iii) if sent by overnight courier which delivers only upon the signed receipt of the addressee, on the date the receipt acknowledgment is executed or refused by the addressee or its agent or (iv) if sent by facsimile or electronic mail, on the date confirmation of transmission is received (provided that a copy of any notice delivered pursuant to this clause (iv) shall also be sent pursuant to clause (i), (ii) or (iii)), addressed to the attention of the addressee's General Counsel at the address of its principal executive office or to such other address or facsimile number for a party hereto as it shall have specified by like notice.

Section 11.5 Counterparts. This Agreement, including the Schedules hereto and the other documents referred to herein, may be executed in multiple counterparts, each of which when executed shall be deemed to be an original but all of which together shall constitute one and the same agreement.

Section 11.6 Severability. If any term or other provision of this Agreement or the Schedules attached hereto is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to either party hereto. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the court, administrative agency or arbitrator shall interpret this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible. If any sentence in this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.

Section 11.7 Governing Law. This Agreement shall be governed by, and construed and enforced in accordance with, the substantive laws of the State of Texas, without regard to any conflicts of law provisions thereof that would result in the application of the laws of any other jurisdiction.

- Section 11.8 *Performance*. Each party hereto shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth herein to be performed by any Subsidiary or Affiliate of such party.
- Section 11.9 *Relationship of Parties*. This Agreement does not create a fiduciary relationship, partnership, joint venture or relationship of trust or agency between the parties. The parties hereto agree that McDermott (and any other member of the MII Group which performs Services hereunder) is an independent contractor in the performance of Services for the B&W Group under this Agreement.
- Section 11.10 Regulations. All employees of McDermott and the members of the MII Group shall, when on the property of B&W, conform to the rules and regulations of B&W concerning safety, health and security which are made known to such employees in advance in writing.
- Section 11.11 Construction. This Agreement shall be construed as if jointly drafted by the parties hereto and no rule of construction or strict interpretation shall be applied against either party. In this Agreement, unless the context clearly indicates otherwise, words used in the singular include the plural and words used in the plural include the singular; and if a word or phrase is defined in this Agreement, its other grammatical forms, as used in this Agreement, shall have a corresponding meaning. Whenever the context requires, the gender of all words used in this Agreement includes the masculine, feminine and the neuter. Unless the context otherwise requires, the words "include," "includes" and "including" shall be deemed to be followed by the words "without limitation," and the word "or" shall have the inclusive meaning represented by the phrase "and/or." The words "shall" and "will" are used interchangeably in this Agreement and have the same meaning. Relative to the determination of any period of time hereunder, "from" means "from and including," "to" means "to but excluding" and "through" means "through and including." All references herein to a specific time of day in this Agreement shall be based upon Central Standard Time or Central Daylight Savings Time, as applicable, on the date in question. Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. Any reference herein to any clause means such clause of such Section or definition. As used in this Agreement, the words "this Agreement," "herein," "herences," "hereof," "hereof," "hereto" and words of similar import shall be deemed references to this Agreement as a whole and not to any particular Section or other provision of this Agreement. The titles to Articles and headings of Sections contained in this Agreement, in any Schedule and in the table of contents to this Agreement have been inserted for convenience of reference only and shall not be deemed to be a part of or to aff

Section 11.12 Effect if Separation does not Occur. If the Distribution does not occur, then all actions and events that are, under this Agreement, to be taken or occur effective as of or following the Distribution Date, or otherwise in connection with the Distribution, shall not be taken or occur except to the extent specifically agreed by the parties and neither party shall have any liability or further obligation to the other party under this Agreement.

[Signature page follows.]

| IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above. | |
|---|-------------------------------|
| | MCDERMOTT INTERNATIONAL, INC. |
| | By: |
| | Name: Title: |
| | THE BABCOCK & WILCOX COMPANY |
| | Ву: |

Name: Title:

Schedule A Tax Services

Description of Services to be Provided:

Services to be provided as part of Tax Transition Services:

- (A) Subject matter expertise as it relates to B&W's tax provision through the filing of the Form 10-K for the 2010 year
- (B) Subject matter expertise as it relates to B&W's international tax operations
- (C) Miscellaneous subject matter expertise on an as needed basis

Special Provisions:

Service Fee:

Amount: \$12,744 per month.

Currency: USD

Termination Date:

All services will be terminated automatically on December 31, 2011.

Schedule B

Accounting and Financial Reporting Services

Areas of Service:

Assistance with Preparation of Quarterly/Annual Filings and Shareholder Reports

- o Assistance with recording consolidating adjustments, intercompany eliminations, capital contributions, intercompany dividends, minority interest, foreign currency translations, etc.
- o Assistance with verifying memo data (depreciation & amortization, etc).
- o Provision of support in the development of select footnotes.
- Assistance with producing monthly and quarterly debt covenant tests and reporting packages.
- o Assistance with performing various analyses during the quarterly and annual close processes to assist management with understanding current results and the latest forecast, with variance explanations and with information for the earnings calls with analysts and others.

Assistance with Application Support (HFM)

- Assistance with the maintenance of the HFM financial consolidation system, including SOx security protocols, close process protocols, and account hierarchy maintenance.
- o Assistance with providing Hyperion administration and user support, including connectivity and report creation.
- o Assistance with the management of Deloitte audit SharePoint site(s).

Additional Subject Matter Support

- o Assistance with equity accounting.
- o Assistance with SGA expense analyses and reporting (Blue Book).
- Assistance with Defense Contractor's Audit Agency audit requests.

Special Provisions:

- o Formal training courses will be provided if required subject to availability of qualified trainers at rates to be agreed (not included in Service Fee below).
- Except as necessitated by completion of the Distribution, preparation of quarterly and annual financial statements and reporting will be consistent with historical practice.
- Services will not include HFM development work, other than for metadata changes, and will not include support of upgrades or implementation of new consolidations and reporting technology.

Service Fee:

Amount:

Currency:

\$5,035 per month United States Dollars (USD) Service fee above does not include interim executive costs Note:

Termination Date:

Services provided up to one year from the Distribution Date.

Schedule C

Executive Consulting Services

Description of Services To Be Provided:

Services to be provided as part of Executive Consulting Transition Services:

Access, on an as-needed basis, to the consulting services to be provided by each of Messrs. Preston Johnson and Dennis S. Baldwin (each, a "Consultant" and, collectively, the "Consultants"), in each case pursuant to the separate consulting agreement between the Consultant and [MII], a copy of which has been provided to B&W (collectively, in the forms provided to B&W, the "Consulting Agreements").

Special Provisions:

The Services described above shall be subject to the terms and provisions of the Consulting Agreements (including the provisions thereof respecting indemnity, which B&W hereby agrees to be bound by, as if it were a party to each of the Consulting Agreements in the same capacity as MII is a party).

Service Fee:

Amount: \$47,500 (50% of the consulting fees and other compensation payable pursuant to the Consulting Agreements).

Currency: United States Dollars (USD)

Termination Date:

Services provided up to three months from the Distribution Date.

Schedule D Legal Services

Description of Services to be Provided:

Services to be provided as part of Legal Transition Services:

- (A) COBRA Subsidy Support
- (B) Medicare Part D Compliance Support
- (C) HIPAA Compliance Support
- (D) Thrift Plan VCP Support
- (E) Contract Support
- (F) Welfare Plans Support

(A) COBRA Subsidy Support

Assistance as reasonably requested for compliance with COBRA subsidy provisions of ARRA, as amended

(B) Medicare Part D Support

- o Assistance as reasonably requested for:
 - Drafting Medicare Part D Notices of Creditable Coverage
 - Filing disclosures to CMS

(C) HIPAA Compliance Support

- Assistance as reasonably requested for:
 - Compliance with HIPAA privacy and security requirements
 - Drafting HITECH amendments to business associate agreements

(D) Thrift Plan VCP Support

- o Assistance as reasonably requested for:
 - Matters related to B&W participants affected by the 2008 Thrift Plan VCP filing
 - Administration of Thrift Plan loan procedures

(E) B&W NOG Hourly Welfare Plan Contract Support

o Assistance as reasonably requested for review of ASO and Stop Loss contract

(F) Welfare Plans Support

- Assistance as reasonably requested for:

 Issues related to 2011 open enrollment

 Issues related to dependent coverage

 General welfare and cafeteria plan compliance

Service Fee:

Amount:

\$1,553 per month United States Dollars (USD) Currency:

Termination Date:

Services provided up to nine months from the Distribution Date.

Schedule E

Human Resources Services

Description of Services to be Provided:

Services to be provided as part of Human Resources Transition Services:

- (A) Assistance with Administration of Run-off of Health and Welfare Claims
- (B) Payroll Processing Support
- (C) HCM SAP Service Center Support
- (D) Expat Administration Support
- (E) Assistance with Deployment of SAP Compensation Tool

(A) Assistance with Administration of Run-off of Health and Welfare Claims

- o Continue payment of all B&W claims with dates of service prior to the Distribution Date, as directed by B&W (create an invoice with back-up and provide to B&W for reimbursement).
- o Assistance with administration of claims currently in subrogation or dispute or those which arise within 12 months after the Distribution Date, to include assistance with appeals to the Health Plan Committee.
- o Provide access to data with dates prior to the spin for the purposes of completing required audits and regulatory filings:
 - DCAA CIPR audit
 - Audits by third parties relating to current pension plans
 - NOG-B trust audit
 - Medicare secondary payer requests
 - FAS 106 data requests for Buck Consultants
 - 5500 filings
- o Provide data to support the COBRA ARRA subsidiary tax credits for B&W payrolls (premiums received for B&W COBRA participants post spin will be provided to B&W with appropriate back-up).
- o For premium payments which are made following the end of the month's payroll, process payments/finding on behalf of B&W for the immediate period following the Distribution Date (create and invoice with back-up and provide to B&W for reimbursement).

(B) Payroll Processing Support

- Assistance as reasonably requested for:
 - · Pay issue troubleshooting
 - Special payment processing (bonus, vacation, pay advances)
 - · Garnishment processing
 - · Payroll tax filings and research

- Equity compensation tax reporting
- Board of Directors payment processing and reporting
- Payroll processing back-up

(C) HCM SAP Service Center Support

- Assistance as reasonably requested for:
 - Deleting personnel numbers created in error
 - Changing entry/separation dates
 - Deleting infotypes and records
 - Org Management creation and maintenance

(D) Expat Administration Support

- o Assistance as reasonably requested for:
 - Assignment offers, hire/rehire, country reassignments and terminations
 - Change of pay processing
 - Expat leaves of absence
 - Change in positions (intra and inter region transfers)
 - Housing & utilities (H&U) and commodities & services (C&S) processing
 - Tax Equalization
 - Unified Earning Statements
 - Foreign Tax compliance (with third party tax preparer)

Special Provisions:

The number of expatriates and their locations shall not increase by more than 10% during the service period.

(E) Assistance with Deployment of SAP Compensation Tool

- Assistance as reasonably requested for:
 - Providing an overview of the compensation tool and portal to include a detailed demonstration
 - · Providing a review and briefing on the transition plan which was developed during the design phase for deploying the tool in the business units
 - Serving as a subject matter expert during planning sessions/workshops to confirm the compensation process design and tool set-up
 - Resolving issues and responding to inquiries during the deployment process

Service Fee:

Amount: \$12,272 per month Currency: US Dollars (USD)

Termination Date:

Services provided for up to twelve months from the Transaction Date.

Schedule F

Risk Management Services

Description of Services to be Provided:

Miscellaneous advice and consulting as requested concerning:

- Historic MII CIRM activities, documents and processes including, but not limited to, insurance coverages, captive operations and coverages, and RIMS systems.
- 2. Risk identification, assessment, and treatment (including financing via transfer or captive).
- 3. Removal of B&W related rights and obligations from Boudin and Non-B&W obligations from Creole.
- 4. New employee training.

Charges:

\$1,245 per month for three months from the effective date hereof. After three months, charges to be agreed if any services still are being provided.

Schedule G

Treasury Services

Description of Services to be Provided:

Services to be provided as part of Treasury Transition Services:

- (A) Cash management services
- (B) Assistance with managing temporary investments and hedging
- (C) Assistance with coordinating revolver borrowings
- (D) Assistance with managing credit relationships
- (E) Assistance with credit agreement compliance reporting

Each such service category is described below.

(A) Cash Management Services

0

- o Monitor cash activity to determine daily cash surplus or shortfall
- o Initiate intercompany transfers necessary to fund bank accounts
- o Execute vendor payments as directed by B&W
- o Upload payment files as approved by B&W
 - Convert excess foreign currencies to USD
 - Purchase foreign currency per B&W cash call requests
 - Buy/Sell foreign currency to fund operations according to B&W instructions
- o Reconcile and report prior day activity
- o Assist with opening and activating new deposit and investment accounts, closing accounts, updating signatories on accounts
- o Assistance with managing escrow accounts
- o Assistance with completing letter of credit applications, receiving, reconciling and paying letter of credit fees
- o Assistance with managing excess liquidity and longer term financial assets
 - Redeem funds as necessary to fund daily activity
 - Invest excess cash in B&W approved investment funds
- o Assistance with receiving, analyzing and paying banking fees related to Treasury Management Services
- o Assistance with managing banking projects, service and user entitlements on banking systems
- o Serve on pre- and post-acquisition teams

(B) Manage Temporary Investments and Hedging

- o Assistance with negotiating Investment Management Agreements for all B&W affiliates
 - Assistance with establishment and maintenance of investment guidelines
- Assistance with monitoring compliance with investment guidelines
- o Assistance with hedging

- (C) Assistance with Coordinating Revolver Borrowings

 o Notify B&W management of any cash shortfalls
 o Deliver borrowing notices to bank if directed by B&W
 - Report all activity reasonably promptly

(D) Assistance with managing credit relationships

- Determine credit needs o
- Negotiate credit facilities o

$(E)\ Assistance\ with\ Credit\ Agreement\ Compliance\ Reporting$

- Assistance with tracking reporting requirements
- Assistance with performance of required periodic covenant calculations and reporting same to lenders

Service Fee for All Services:

Amount: \$10,459 per month

Termination Date:

Services provided up to nine months from the Distribution Date

Schedule H

Investor Relations Services

Description of Services to be Provided:

Services that can be provided by McDermott as part of Investor Relations ("IR") Transition Services:

- (A) IR Assistance
- (B) Additional services are available on request, subject to acceptable fee negotiation

Each such service category is described below.

(A) IR Assistance

- o $\,$ JRM IR team will provide advice and assistance to B&W's CEO, CFO and IR Officer on:
 - Sell-side analysts
 - Buy-side analysts & institutions
 - Vendors
 - General IR/Corporate Communication information, issues and activities
- o Service not to exceed 10 hours per month.

(B) Additional services available

- o Data provision:
 - Available sell-side analyst reports from First Call and/or email distribution
 - Consensus Earnings Estimates through monthly Corporate Profile
 - ¡ Quarterly ownership information via BD Corporate
 - Analyst & Institution profiles via BD Corporate
- IR Review:
 - i McDermott IR team will review and comment as requested on financial community directed:
 - Press Releases
 - Conference call scripts
 - Conference call Q&A
 - Investor presentations

- o IR Preparation:
- paration:

 McDermott IR team will assist in drafting and preparing as needed:

 Press Releases

 Conference call scripts

 Conference call Q&A

 Investor presentations

 - Investor presentations
 Other documents as reasonably requested

Service Fee:

\$1,886 per month Amount:

Termination Date:

Service provided for an agreed upon time, but in no case to exceed nine months from the Distribution Date

Schedule I

Litigation Services

Areas of Service:

JRM will provide to B&W legal group support/assistance, as necessary, with regard to the following matters: B&W, et al v. McGriff, Seibels & Williams, et al; and B&W, et al v. ANI, et al. (collectively, "the Litigation")

Examples of such support/assistance may include:

- · Witness testimony
- Education re: background on the Litigation and on underlying or related matters
- · Access to and assistance with case files for underlying matters
- Advice and/or participation in legal strategizing upon request

Terms of Service:

Such assistance shall be provided through the duration of the Litigation.

Cost of Service:

\$7,783 per month.

Schedule J

Services for Sales Office in Singapore

Description of Services to be Provided:

As of the Distribution Date, McDermott will provide as reasonably needed office space, personal computer, land-line telephone, Internet access (other than through a McDermott network), IT user support (as more fully described in the following paragraph), routinely available consumable office supplies, and security (building access) services for up to six (6) B&W employees resident in McDermott's AsiaPac Headquarters office in Singapore (11 Lorong 3 Toa Payoh (Jackson Square)). Such Services will be provided at the same level they are being provided for McDermott employees in similar positions in the Singapore sales office. For the avoidance of doubt, other than in relation to these specific services, the Services do not include any employee administration functions, *e.g.*, issuing employee salaries or bonuses to such B&W employees, nor do they include access to the McDermott network.

As indicated above, McDermott will provide IT user support assistance to B&W personnel located at McDermott's AsiaPac Headquarters office in Singapore (11 Lorong 3 Toa Payoh (Jackson Square)) as requested by such B&W personnel, the B&W Help Desk, other B&W IT management, including assistance with PC hardware and software, printing, and general network connectivity ("User Support Services").

Service Fee:

\$6,850 per month for the above-listed Services.

B&W will reimburse McDermott for the actual cost incurred by McDermott for all other items and services utilized by the B&W employees in the Singapore office, including, but not limited to, travel expenses, miscellaneous marketing items, parking fees, newspapers, courier and shipping charges, transport allowances, and administration fees.

Termination Date:

The User Support Services terminate twelve (12) months after the Distribution Date or when the last B&W personnel has permanently departed McDermott's AsiaPac Headquarters office in Singapore (11 Lorong 3 Toa Payoh (Jackson Square)), whichever comes first. All other Services hereunder will terminate six (6) months after the Distribution Date, unless earlier terminated upon thirty (30) days' prior written notice provided to McDermott by B&W.

Schedule K

Services for MPLS and Facility Access in Houston

Description of Services to be Provided:

Support services described in this schedule are those required to support the overall transition of information technology services, particularly with respect to the connectivity, security and integrity of required information and telecommunications systems, including firewall structure management and the distribution, forwarding and separation of e-mail (collectively, the "MPLS Services"), and further to provide for transitional information and telecommunications support and access for B&W employees resident in McDermott's Houston facilities following the Distribution Date, including use of computers, phones, and other access requirements (collectively, the "Facility-Related Services"). Such Services will be provided at the same level they are being provided to McDermott employees in similar positions in McDermott's Houston facilities.

MPLS Services: As of the Distribution Date, McDermott will provide the following support services, with the cooperation of B&W:

• Information Systems Connectivity and Security—McDermott will work with B&W to separate the contractual relationship with Verizon under which information systems connectivity and security (the "Information Network") is provided. The parties anticipate that McDermott will acquire the coverage in the Houston and international regions provided under the Verizon relationship, and B&W will acquire coverage in all other regions. To the extent such separation is not completed as of the Distribution Date, the parties will cooperate through their respective Service Coordinators to complete such separation as commercially reasonable. To the extent either party needs access to the Information Network of the other party after the Distribution Date, then during the MPLS Transition Period, each party will provide the other party with reasonable access to their respective Information Networks to the extent necessary to facilitate the overall transition of information technology services. As each party no longer requires access to the Information Network of the other party with respect to specific functions, systems or services, such party's access will be terminated. The process for such termination will be as follows: At such time as a party no longer needs access to the Information Network of the other party with respect to a particular function, system or service, then the Service Coordinator for the party no longer requiring access will provide notice to the other party's Service Coordinator. Access management will be implemented through the relevant firewall rules. The Service Coordinators will promptly confer and reasonably determine how the firewall rules should be modified to implement the access changes. The Service Coordinator for the party owning the Information Network to which access is being restricted shall coordinate the implementation of the new firewall rules with Verizon or other service provider.

As part of the above Services, B&W will be allowed to continue its current use of the 131.184.46.n/24 subnet on the internet for a maximum period of twelve (12) months after the Distribution Date. During this use period, B&W will migrate off this and any other 131.184.0.0/16 subnet. Any public/internet routing or provider changes required by B&W during this period will be accommodated and auctioned by McDermott. To maintain application access while any IT transition services are being provided by either party pursuant to a Transition Services Agreement, both McDermott and B&W will maintain the current IP addressing space on their internal networks: B&W will maintain exclusive use of the 10.0.0.0 to 10.127.255.255 range and McDermott the 10.128.0.0 to 10.255.255.255 range. B&W will not expand its use of any 131.184.0.0 subnet internally or externally without prior consent of McDermott.

In addition, McDermott will provide connectivity between the B&W and McDermott networks, with both B&W and McDermott maintaining a back-to-back firewall solution allowing free and open access to applications covered under a Transition Services Agreement. Any rule changes made to these firewalls will be communicated in advance and agreed upon by both companies. B&W and McDermott will be responsible for maintaining the uptime of each firewall and the updates to routing tables on their respective networks. It is intended that rules sets on the firewalls will adopt a least access posture, i.e., only allowing IP address and TCP ports required to support application access. Any security violations or virus outbreaks on either company's network will be communicated to the other party to ensure the integrity of each company's network; either company may temporarily sever all access to the other during such an outbreak, until the cause has been determined and remedied.

- ADP Network Access—For a period of twelve (12) months after the Distribution Date or until all legacy date has been exported, whichever comes first, McDermott will maintain the link to the ADP legacy site systems. B&W will be allowed to use this link during such period with access and routing via the back-to-back firewalls and inter-company connectivity maintained by McDermott.
- · Electronic Mail—For a period of six (6) months after the Distribution Date, McDermott will forward all emails of its former executive-level employees.

For a period of ninety (90) days after the Distribution Date, McDermott will forward emails of all its other former employees.

The foregoing electronic mail services will only be provided for so long as such employees are B&W employees after the Distribution Date.

• *Phone Greeting*—For a period of ninety (90) days after the Distribution Date, McDermott will maintain a phone greeting on the extensions of its former employees. The phone greeting, which will be recorded by each employee on the last day of employment with McDermott, will tell callers the new phone number(s) to use. No phone calls will be forwarded, and the foregoing phone greeting services will only be provided for so long as such employees are B&W employees after the Distribution Date.

User Support—McDermott will provide IT user support assistance to B&W personnel located at 777 N. Eldridge as requested by the B&W Help Desk or
other B&W IT management, including assistance with PC hardware and software, printing, general network connectivity and application access. These user
support services will only be provided for a period of twelve (12) months after the Distribution Date or until the last B&W personnel has permanently
departed the 777 N. Eldridge building, whichever comes first.

Facility-Related Services: As of the Distribution Date, McDermott will provide as reasonably needed personal computer, land-line telephone, Internet and Network access, and security (building access) services for those B&W employees resident in McDermott's Houston facilities during the MPLS Transition Period.

Monthly Status Conferences: During the MPLS Transition Period, Louis Burkart and David Black (or their designated delegates) agree to confer on a monthly basis to consider and discuss the status of these Services and other issues relating thereto.

Service Fee:

Amount: \$5,000 per month Currency: US Dollars (USD).

Termination Date:

McDermott will provide the Services for up to eighteen (18) months from the Distribution Date, or, with respect to certain Services, for the shorter period indicated in the Services descriptions above, unless earlier terminated upon thirty (30) days' prior written notice provided to McDermott by B&W ("MPLS Transition Period").

Schedule L Office Space

Description of Services to be Provided:

Office space to be provided for B&W employees at the 777 N. Eldridge Pkwy, Houston, Texas location.

Service Fee:

Amount: \$43,143 per month US Dollars (USD).

Termination Date:

McDermott will provide the Services for up to fourteen (14) months from the Distribution Date.

Schedule M

Offsite and Third Party Records Storage and Destruction

Description of Services to be Provided:

Offsite and third party records storage and destruction

Service Fee:

Amount: \$[5,964] per month US Dollars (USD).

Termination Date:

McDermott will provide the Services for up to eighteen (18) months from the Distribution Date.

Schedule N

Office in McDermott's New Orleans Facility

Description of Services to be Provided:

As of the Distribution Date, McDermott will provide, as reasonably needed, office space, personal computer, land-line telephone, Internet access (other than through a McDermott network), IT support, routinely available consumable office supplies, and security (building access) services for one (1) B&W employee resident in McDermott's office in New Orleans. Such Services will be provided at the same level they are being provided for McDermott employees in similar positions in the McDermott's New Orleans office. For the avoidance of doubt, other than in relation to these specific services, the Services do not include any employee administration functions, *e.g.*, issuing employee salaries or bonuses to such B&W employees, nor do they include access to the McDermott network.

Service Fee:

\$400 per month for the above-listed Services.

B&W will reimburse McDermott for the actual cost incurred by McDermott for all other items and services utilized by the B&W employees in the McDermott New Orleans office, including, but not limited to, travel expenses, miscellaneous marketing items, parking fees, newspapers, courier and shipping charges, transport allowances, and administration fees.

Termination Date:

The Services will terminate twelve (12) months after the Distribution Date, unless earlier terminated upon thirty (30) days' prior written notice provided to McDermott by B&W.

Schedule 2.4 Certain Subcontractors

None.

Schedule 4.1

Payment Instructions

Until further notice, the following are wire transfer/ACH payment instructions for payment to McDermott for amounts owing under this Agreement:

McDermott International, Inc.

JP Morgan Chase Bank New York 021000021 J. Ray McDermott Inc. 304299227 Bank Name: Bank ABA #:

Account Name:

Account Number:

TRANSITION SERVICES AGREEMENT

BETWEEN

THE BABCOCK & WILCOX COMPANY

(as service provider)

and

MCDERMOTT INTERNATIONAL, INC.

(as service receiver)

Dated [], 2010

TABLE OF CONTENTS

| | | Page No. |
|--------------|--|----------|
| ARTICLE I | DEFINITIONS | 1 |
| Section 1.1 | Definitions | 1 |
| ARTICLE II | SERVICES | 2 |
| Section 2.1 | Services | 2 |
| Section 2.2 | Service Coordinators | 3 |
| Section 2.3 | Additional Services | 3 |
| Section 2.4 | Third Party Services | 3 |
| Section 2.5 | Standard of Performance; Limitation of Liability | 4 |
| Section 2.6 | Service Boundaries and Scope | 5 |
| Section 2.7 | Cooperation | 6 |
| Section 2.8 | Transitional Nature of Services; Changes | 6 |
| Section 2.9 | Access | 6 |
| ARTICLE III | SERVICE CHARGES | 6 |
| Section 3.1 | Compensation | 6 |
| ARTICLE IV | PAYMENT | 6 |
| Section 4.1 | Payment | 6 |
| Section 4.2 | Payment Disputes | 7 |
| Section 4.3 | Review of Charges; Error Correction | 7 |
| Section 4.4 | Taxes | 7 |
| Section 4.5 | Records | 8 |
| ARTICLE V | TERM | 8 |
| Section 5.1 | Term | 8 |
| ARTICLE VI | DISCONTINUATION OF SERVICES | 8 |
| Section 6.1 | Discontinuation of Services | 8 |
| Section 6.2 | Procedures Upon Discontinuation or Termination of Services | 8 |
| ARTICLE VII | DEFAULT | 9 |
| Section 7.1 | Termination for Default | 9 |
| ARTICLE VIII | INDEMNIFICATION AND WAIVER | 9 |
| Section 8.1 | Waiver of Consequential Damages | 9 |
| Section 8.2 | Services Received | 9 |
| Section 8.3 | Express Negligence | 10 |
| ARTICLE IX | CONFIDENTIALITY | 11 |
| Section 9.1 | Confidentiality | 11 |
| Section 9.2 | System Security | 11 |
| ARTICLE X | FORCE MAJEURE | 12 |
| Section 10.1 | Performance Excused | 12 |
| Section 10.2 | Notice | 12 |
| Section 10.3 | Cooperation | 12 |

i

| ADTICLE VI | MICCELLANDOLIC | 12 |
|---------------|--|----|
| ARTICLE XI | MISCELLANEOUS | 12 |
| Section 11.1 | Entire Agreement | 12 |
| Section 11.2 | Binding Effect; No Third-Party Beneficiaries; Assignment | 12 |
| Section 11.3 | Amendment; Waivers | 13 |
| Section 11.4 | Notices | 13 |
| Section 11.5 | Counterparts | 13 |
| Section 11.6 | Severability | 13 |
| Section 11.7 | Governing Law | 13 |
| Section 11.8 | Performance | 14 |
| Section 11.9 | Relationship of Parties | 14 |
| Section 11.10 | Regulations | 14 |
| Section 11.11 | Construction | 14 |
| Section 11.12 | Effect if Separation does not Occur | 14 |
| | | |

Schedules
Schedule A - Tax Services

Schedule A - 1ax Services
Schedule B - Accounting and Financial Reporting Services
Schedule C - Corporate Secretary Services
Schedule D - Legal Services
Schedule E - Human Resources Services
Schedule F - Risk Management Services

Schedule G - Treasury Services
Schedule H - Sharepoint
Schedule I - Disaster Recovery
Schedule 2.4 Certain Subcontractors
Schedule 4.1 Payment Instructions

TRANSITION SERVICES AGREEMENT

This TRANSITION SERVICES AGREEMENT (together with the Schedules hereto, this "Agreement") is entered into as of [], 2010, by and between The Babcock & Wilcox Company, a Delaware corporation ("B&W"), and McDermott International, Inc., a Panamanian corporation ("McDermott").

WHEREAS, the Board of Directors of McDermott has determined that it would be appropriate and desirable for McDermott to distribute (the "Distribution") on a pro rata basis to the holders of outstanding shares of common stock, par value \$1.00 per share, of McDermott all of the outstanding shares of common stock, par value \$0.01 per share, of B&W owned by McDermott;

WHEREAS, in order to effectuate the foregoing, McDermott and B&W have entered into a Master Separation Agreement, dated as of the date hereof (the "Master Separation Agreement"), which provides, among other things, upon the terms and subject to the conditions thereof, for the separation of the respective businesses of McDermott and B&W and the Distribution, and the execution and delivery of certain other agreements, including this Agreement, in order to facilitate and provide for the foregoing; and

WHEREAS, in order to provide for an orderly transition under the Master Separation Agreement, it will be advisable for B&W, through members of the B&W Group, to provide to McDermott certain services described herein for a transitional period.

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements herein contained, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

ARTICLE I DEFINITIONS

Section 1.1 Definitions. As used in this Agreement, the following terms shall have the meanings set forth below:

- "Additional Services" has the meaning set forth in Section 2.3.
- "Agreement" has the meaning set forth in the preamble.
- "Availed Party" has the meaning set forth in Section 9.2(a).
- "B&W" has the meaning set forth in the preamble.
- "Distribution" has the meaning set forth in the recitals.
- "Force Majeure Event" has the meaning set forth in Section 10.1.

- "Master Separation Agreement" has the meaning set forth in the recitals.
- "McDermott" has the meaning set forth in the preamble.
- "Schedules" means the Schedules attached hereto.
- "Security Regulations" has the meaning set forth in Section 9.2(a).
- "Service Coordinator" has the meaning set forth in Section 2.2.
- "Services" has the meaning set forth in Section 2.1(a).
- "Systems" has the meaning set forth in Section 9.2(a).
- "Tax" has the meaning set forth in Section 4.4.

Capitalized terms used but not otherwise defined in this Agreement shall have the respective meanings assigned to such terms in the Master Separation Agreement.

ARTICLE II SERVICES

Section 2.1 Services.

- (a) Upon the terms and subject to the conditions of this Agreement, B&W, acting directly and/or through its Affiliates and their respective employees, agents, contractors or independent third parties designated by any of them, agrees to use commercially reasonable efforts to provide or to cause to be provided services to the MII Group as set forth in Schedules A through I (including any Additional Services provided in accordance with Section 2.3 hereof, all such services are collectively referred to herein as the "Services").
- (b) At all times during the performance of the Services, all Persons performing such Services (including agents, temporary employees, independent third parties and consultants) shall be construed as being independent from the MII Group, and such Persons shall not be considered or deemed to be employees of any member of the MII Group nor entitled to any employee benefits of MII as a result of this Agreement. The responsibility of such Persons is to perform the Services in accordance with this Agreement and, as necessary, to advise the applicable member of the MII Group in connection therewith, and such Persons shall not be responsible for decision-making on behalf of any member of the MII Group. Such Persons shall not be required to report to management of any member of the MII Group nor be deemed to be under the management or direction of any member of the MII Group. McDermott acknowledges and agrees that, except as may be expressly set forth herein as a Service (including any Additional Services provided in accordance with Section 2.3 hereof) or otherwise expressly set forth in the Master Separation Agreement or an Ancillary Agreement, no member of the B&W Group shall be obligated to provide, or cause to be provided, any service or goods to any member of the MII Group.

(c) Notwithstanding anything to the contrary in this Agreement, B&W and members of the B&W Group shall not be required to perform Services hereunder or take any actions relating thereto that conflict with or violate any applicable law, contract, license, authorization, certification or permit or B&W's Code of Business Conduct or other governance policies, as they may be amended from time to time.

Section 2.2 Service Coordinators. Each party will nominate in writing a representative to act as the primary contact with respect to the provision of the Services and the resolution of disputes under this Agreement (each such person, a "Service Coordinator"). The initial Service Coordinators shall be Mr. Gary Brauchle (for McDermott) and Mr. Keith Robinson (for B&W) (or their designated delegates) for each of McDermott and B&W, respectively. The Service Coordinators shall meet as expeditiously as possible to resolve any dispute hereunder; and any dispute that is not resolved by the Service Coordinators within 45 days shall be resolved in accordance with the dispute resolution procedures set forth in Article V of the Master Separation Agreement. Each party hereto may treat an act of a Service Coordinator of the other party hereto which is consistent with the provisions of this Agreement as being authorized by such other party without inquiring behind such act or ascertaining whether such Service Coordinator had authority to so act; provided, however, that no such Service Coordinator shall have authority to amend this Agreement. McDermott and B&W shall advise each other promptly (in any case no more than three Business Days) in writing of any change in their respective Service Coordinators, setting forth the name of the replacement, and stating that the replacement Service Coordinator is authorized to act for such party in accordance with this Section 2.2.

Section 2.3 Additional Services. McDermott may request additional Services (the "Additional Services") from B&W by providing written notice. Upon the mutual written agreement as to the nature, cost, duration and scope of such Additional Services, McDermott and B&W shall supplement in writing the Schedules hereto to include such Additional Services. Subject to the other limitations in this Agreement, including the provisions in Section 2.6, but nothwithstanding the foregoing provisions of this Section 2.3, in addition to providing the Services specified in the Schedules, B&W, acting directly and/or through its Affiliates and their respective employees, agents, contractors or independent third parties designated by any of them, shall use commercially reasonable efforts to provide or to cause to be provided additional, de minimis administrative support services to the MII Group as may be requested by any member of the MII Group from time to time, at no cost beyond the amounts set forth in the Schedules (as the amounts set forth in the Schedules contemplate such additional, de minimis administrative support services); provided, however, that, for any such additional services to be considered de minimis for purposes of this sentence, such additional services shall not require the attention of (i) any one employee of any member of the B&W Group for more than 2 hours in any single calendar month or (ii) any group of employees of any one or more members of the B&W Group for more than 30 hours in any single calendar month. Except where the context otherwise indicates or requires, any such additional services referred to in the immediately preceding sentence shall be deemed to be "Services" under this Agreement.

Section 2.4 *Third Party Services.* B&W shall have the right to hire third-party subcontractors to provide all or part of any Service hereunder; provided, that B&W shall consult in good faith with McDermott regarding the proposed hiring of any third-party subcontractor that has not previously been involved in the activities relating to such Service prior to the date hereof;

provided, further, that, in the event such subcontracting is inconsistent with the practice applied by B&W generally from time to time within its own organization, B&W shall give notice to McDermott of its intent to subcontract any portion of the Services and McDermott shall have 20 days (or such lesser period set forth in the notice as may be practicable in the event of exigent circumstances) to determine, in its sole discretion, whether to permit such subcontracting or whether to cancel such Service in accordance with Article VI hereof. If McDermott opts to cancel a Service pursuant to the immediately preceding sentence, it shall not be liable to B&W pursuant to Section 6.1 for any costs or expenses B&W or any member of the B&W Group remains obligated to pay to the third-party subcontractor identified in the notice provided by B&W as described above. B&W shall not be required to give notice of its intent to subcontract Services to any party listed on Exhibit 2.4 hereto, nor shall McDermott have any right to cancel any Service subcontracted to any such listed party pursuant to this Section 2.4 (provided, that this sentence shall not prevent McDermott from cancelling any Service pursuant to Section 6.1).

Section 2.5 Standard of Performance; Limitation of Liability.

- (a) The Services to be provided hereunder shall be performed with the same general degree of care, at the same general level and at the same general degree of accuracy and responsiveness, as when performed within the McDermott organization (including, for this purpose, B&W and its subsidiaries) prior to the date of this Agreement. It is understood and agreed that B&W and the members of the B&W Group are not professional providers of the types of services included in the Services and that B&W personnel performing Services have other responsibilities and will not be dedicated full-time to performing Services hereunder.
- (b) In the event B&W or any member of the B&W Group fails to provide, or cause to be provided, the Services in accordance with the standard of service set forth in Section 2.5(a) or Section 2.5(c), the sole and exclusive remedy of McDermott shall be, at McDermott's sole discretion, within 90 days from the date that B&W or such member of the B&W Group first fails to provide such Service, to not pay for such Service; *provided* that in the event B&W defaults in the manner described in clause (ii) of Section 7.1, McDermott shall have the further rights set forth in Article VII.
- (c) EXCEPT AS EXPRESSLY SET FORTH IN THIS SECTION 2.5, NO REPRESENTATIONS OR WARRANTIES OF ANY KIND, EXPRESSED OR IMPLIED (INCLUDING THE WARRANTIES OF NON-INFRINGEMENT, MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, OR CONFORMITY TO ANY REPRESENTATION OR DESCRIPTION), ARE MADE BY B&W OR ANY MEMBER OF THE B&W GROUP WITH RESPECT TO THE SERVICES UNDER THIS AGREEMENT AND, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ALL SUCH REPRESENTATIONS OR WARRANTIES ARE HEREBY WAIVED AND DISCLAIMED. MCDERMOTT (ON ITS OWN BEHALF AND ON BEHALF OF EACH OTHER MEMBER OF THE MII GROUP) HEREBY EXPRESSLY WAIVES ANY RIGHT MCDERMOTT OR ANY MEMBER OF THE MII GROUP MAY OTHERWISE HAVE FOR ANY LOSSES, TO ENFORCE SPECIFIC PERFORMANCE OR TO PURSUE ANY OTHER REMEDY AVAILABLE IN CONTRACT, AT LAW OR IN EQUITY IN THE EVENT OF ANY NON-PERFORMANCE, INADEQUATE PERFORMANCE, FAULTY PERFORMANCE OR OTHER FAILURE OR BREACH BY B&W OR ANY MEMBER OF THE B&W GROUP

UNDER OR RELATING TO THIS AGREEMENT, NOTWITHSTANDING THE NEGLIGENCE OR GROSS NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT OR ACTIVE OR PASSIVE) OF B&W OR ANY MEMBER OF THE B&W GROUP OR ANY THIRD PARTY SERVICE PROVIDER AND WHETHER DAMAGES ARE ASSERTED IN CONTRACT OR TORT, UNDER FEDERAL, STATE OR NON U.S. LAWS OR OTHER STATUTE OR OTHERWISE; PROVIDED, HOWEVER, THAT THE FOREGOING WAIVER SHALL NOT EXTEND TO COVER, AND B&W SHALL BE RESPONSIBLE FOR, SUCH LOSSES CAUSED BY THE WILLFUL MISCONDUCT OF B&W OR ANY MEMBER OF THE B&W GROUP. NOTWITHSTANDING ANYTHING IN THIS AGREEMENT TO THE CONTRARY, IN NO EVENT SHALL THE B&W GROUP BE LIABLE TO THE MII GROUP WITH RESPECT TO CLAIMS ARISING OUT OF THIS AGREEMENT FOR AMOUNTS IN THE AGGREGATE EXCEEDING THE AGGREGATE SERVICE CHARGES PAID HEREUNDER BY THE MII GROUP.

Section 2.6 Service Boundaries and Scope. Except as provided in a Schedule for a specific Service: (a) B&W shall be required to provide, or cause to be provided, the Services only at (i) the locations such Services are being provided by any member of the B&W Group for any member of the MII Group immediately prior to the Distribution Date and (ii) the corporate headquarters of B&W in Charlotte, North Carolina; provided, however, that, to the extent any such Service is to be provided by an employee of B&W who works in the corporate headquarters of B&W, such Service shall, to the extent feasible, only be provided by such employee from the corporate headquarters of B&W; and (b) the Services shall be available only for purposes of conducting the business of the MII Group substantially in the manner it was conducted immediately prior to the Distribution Date. Except as provided in a Schedule for a specific Service, in providing, or causing to be provided, the Services, B&W shall not be obligated to: (i) maintain the employment of any specific employee or hire additional employees or third-party service providers; (ii) purchase, lease or license any additional equipment (including computer equipment, furniture, furnishings, fixtures, machinery, vehicles, tools and other tangible personal property), software or other assets, rights or properties; (iii) make modifications to its existing systems or software; (iv) provide any member of the MII Group with access to any systems or software other than those to which it has authorized access immediately prior to the Distribution Date; or (v) pay any costs related to the transfer or conversion of data of any member of the MII Group. McDermott acknowledges (on its own behalf and on behalf of the other members of the MII Group) that the employees of B&W or any other members of the B&W Group who may be assisting in the provision of Services hereunder are at-will employees and, as such, may terminate or be terminated from employment with B&W or any of the other members of the B&W Group providing Services hereunder at any time for any reason. In no event shall B&W or any of its Affiliates or any of their respective employees or agents be required to perform any Services or take any other actions hereunder that conflict with any applicable Law. For the avoidance of doubt and except as may hereafter be designated as Additional Services in accordance with Section 2.3, the Services do not include any services required for or as the result of any business acquisitions, divestitures, start-ups or terminations by the MII Group. To the extent McDermott desires B&W to provide any services in connection with any such acquisitions, divestitures, start-ups or terminations, McDermott shall follow the procedures for requesting Additional Services pursuant to Section 2.3.

- Section 2.7 Cooperation. McDermott and B&W shall cooperate with one another and provide such further assistance as the other party may reasonably request in connection with the provision of Services hereunder.
- Section 2.8 *Transitional Nature of Services; Changes.* Subject to Sections 2.3 and 2.5, the parties acknowledge the transitional nature of the Services and that B&W may make changes from time to time in the manner of performing the Services.
- Section 2.9 Access. During the term of this Agreement and for so long as any Services are being provided to McDermott by B&W, McDermott will provide B&W and its authorized representatives reasonable access, during regular business hours upon reasonable notice, to McDermott and its employees, representatives, facilities and books and records as B&W and its representatives may reasonably require in order to perform such Services.

ARTICLE III SERVICE CHARGES

Section 3.1 Compensation. Subject to the specific terms of this Agreement, the compensation to be received by B&W for each Service provided hereunder will be the fees set forth on the Schedule relating to the particular Service, subject to any escalation provided for on such Schedule. In consideration for the provision of a Service, each member of the MII Group receiving such Service shall pay to B&W or, at the election of B&W, the member of the B&W Group providing such Service, the applicable fee for such Service as set forth on the attached Schedules.

ARTICLE IV PAYMENT

Section 4.1 Payment. Except as otherwise provided in a Schedule for a specific Service, charges for Services shall be invoiced monthly by B&W or, at its option, the member of the B&W Group providing the Service. Except as otherwise provided in a Schedule for a specific Service, McDermott shall make the corresponding payment no later than 60 days after receipt of the invoice. Unless otherwise provided in this Agreement, McDermott shall remit funds in payment of invoices provided hereunder either by wire transfer or ACH (Automated Clearing House) in accordance with the payment instructions set forth in Schedule 4.1. Each invoice shall be directed to the MII Service Coordinator or such other person designated in writing from time to time by such Service Coordinator. The invoice shall set forth in reasonable detail the Services rendered and the invoice amount for the Services rendered for the period covered by such invoice. Interest will accrue on any unpaid amounts at ten percent (10%) per annum (compounded monthly) or, if less, the maximum non-usurious rate of interest permitted by applicable law, until such amounts, together with all accrued and unpaid interest thereon, are paid in full. All timely payments under this Agreement shall be made without early payment discount. Any preexisting obligation to make payment for Services provided hereunder shall survive the termination of this Agreement. If B&W incurs any reasonable out-of-pocket expenses (including any incremental license fees incurred by B&W in connection with performance of the Services and any travel expenses incurred at the request or with the consent of McDermott) or remits funds to a third-party on behalf of McDermott, in either case in

connection with the rendering of Services, then B&W shall include such amount on its monthly invoice to McDermott, with reasonable supporting documentation, and McDermott shall reimburse that amount to B&W pursuant to this Section 4.1 as part of its next monthly payment.

Section 4.2 Payment Disputes. McDermott may object to any amounts for any Service at any time before, at the time of, or after payment is made, provided such objection is made in writing to B&W within 120 days following the date of the disputed invoice. McDermott shall timely pay the disputed items in full while resolution of the dispute is pending; provided, however, that B&W shall pay interest at a rate of five percent (5%) per annum (compounded monthly) on any amounts it is required to return to McDermott upon resolution of the dispute. Payment of any amount shall not constitute approval thereof. The Service Coordinators shall meet as expeditiously as possible to resolve any dispute. Any dispute that is not resolved by the Service Coordinators within 45 days shall be resolved in accordance with the dispute resolution and arbitration procedures set forth in Article V of the Master Separation Agreement. Neither party (or any member of its respective Group) shall have a right of set-off against the other party (or any member of its respective Group) for billed amounts hereunder. Upon written request, B&W will provide to McDermott reasonable detail and support documentation to permit McDermott to verify the accuracy of an invoice.

Section 4.3 Review of Charges; Error Correction. B&W shall maintain accurate books and records (including invoices of third parties) related to the Services sufficient to calculate, and allow McDermott to verify, the amounts owed under this Agreement. From time to time until 120 days following the termination of this Agreement, McDermott shall have the right to review, and B&W shall provide access to, such books and records to verify the accuracy of such amounts, provided that such reviews shall not occur more frequently than once per calendar quarter. Each such review shall be conducted during normal business hours and in a manner that does not unreasonably interfere with the operations of B&W. If, as a result of any such review, McDermott determines that it overpaid any amount to B&W, then McDermott may raise an objection pursuant to the provisions of Section 4.2. McDermott shall bear the cost and expense of any such review. B&W shall make adjustments to charges as required to reflect the discovery of errors or omissions in charges.

Section 4.4 Taxes. All transfer taxes, excises, fees or other charges (including value added, sales, use or receipts taxes, but not including a tax on or measured by the income, net or gross revenues, business activity or capital of a member of the B&W Group), or any increase therein, now or hereafter imposed directly or indirectly by law upon any fees paid hereunder for Services, which a member of the B&W Group is required to pay or incur in connection with the provision of Services hereunder ("Tax"), shall be passed on to McDermott as an explicit surcharge and shall be paid by McDermott in addition to any Service fee payment, whether included in the applicable Service fee payment, or added retroactively. If McDermott submits to B&W a timely and valid resale or other exemption certificate acceptable to B&W and sufficient to support the exemption from Tax, then such Tax will not be added to the Service fee payable pursuant to Article III; provided, however, that if a member of the B&W Group is ever required to pay such Tax, McDermott will promptly reimburse B&W for such Tax, including any interest, penalties and attorney's fees related thereto. The parties will cooperate to minimize the imposition of any Taxes.

Section 4.5 *Records*. B&W shall maintain true and correct records of all receipts, invoices, reports and such other documents relating to the Services hereunder in accordance with its standard accounting practices and procedures, consistently applied. B&W shall retain such accounting records and make them available to McDermott's authorized representatives and auditors for a period of not less than one year from the close of each fiscal year of B&W; provided, however, that B&W may, at its option, transfer such accounting records to McDermott upon termination of this Agreement.

ARTICLE V TERM

Section 5.1 Term. Subject to Articles VI and VII, the B&W Group shall provide the specific Services to the MII Group pursuant to this Agreement for the time period set forth on the Schedule relating to the specific Service. In accordance with the Master Separation Agreement and Article VI of this Agreement, except as otherwise provided in a Schedule for a specific Service, McDermott shall undertake to provide to itself and members of the MII Group, and to terminate as soon as reasonably practicable, the Services provided to the MII Group hereunder. Except as otherwise provided in a Schedule for a specific Service or group of related Services, all Services provided for hereunder shall terminate on March 31, 2011. Except as otherwise expressly agreed or unless sooner terminated, this Agreement shall commence upon the Distribution Date and shall continue in full force and effect between the parties for so long as any Service set forth in any Schedule hereto is being provided to McDermott or members of the MII Group and this Agreement shall terminate upon the cessation of all Services provided hereunder; provided that Articles I, IV, VIII, IX and XI and Section 2.5(c) will survive the termination of this Agreement and any such termination shall not affect any obligation for the payment of Services rendered prior to termination.

ARTICLE VI DISCONTINUATION OF SERVICES

Section 6.1 Discontinuation of Services. Unless otherwise provided in the relevant Schedule for a particular Service, at any time after the Distribution Date, McDermott may, without cause and in accordance with the terms and conditions hereunder and the Master Separation Agreement, request the discontinuation of one or more specific Services by giving B&W at least 30 days' prior written notice; provided, however, that any such discontinuation will not affect the amounts payable to B&W hereunder unless (and then only to the extent that) the charges for the discontinued Services have been separately identified in the applicable Schedule. McDermott shall be liable to B&W for all costs and expenses B&W or any member of the B&W Group remains obligated to pay in connection with any discontinued Service or Services, except in the case of a Service terminated by McDermott pursuant to clause (ii) of the first sentence of Section 7.1 hereof. The parties shall cooperate as reasonably required to effectuate an orderly and systematic transfer to the MII Group of all of the duties and obligations previously performed by B&W or a member of the B&W Group under this Agreement.

Section 6.2 Procedures Upon Discontinuation or Termination of Services. Upon the discontinuation or termination of a Service hereunder, this Agreement shall be of no further force and effect with respect to such Service, except as otherwise provided in a Schedule for a specific

Service and except as to obligations accrued prior to the date of discontinuation or termination; provided, however, that Articles I, IV, VIII, IX and XI and Section 2.5(c) of this Agreement shall survive such discontinuation or termination. Each party and the applicable member(s) of its respective Group shall, within 60 days after discontinuation or termination of a Service, deliver to the other party and the applicable member(s) of its respective Group originals of all books, records, contracts, receipts for deposits and all other papers or documents in its possession which pertain exclusively to the business of the other party and relate to such Service; provided that a party may retain copies of material provided to the other party pursuant to this Section 6.2 as it deems necessary or appropriate in connection with its financial reporting obligations or internal control practices and policies.

ARTICLE VII DEFAULT

Section 7.1 Termination for Default. In the event (i) of a failure of McDermott to pay for Services in accordance with the terms of this Agreement, or (ii) any party shall default, in any material respect, in the due performance or observance by it of any of the other terms, covenants or agreements contained in this Agreement, then (1) if the non-defaulting party is B&W, B&W shall have the right, at its sole discretion, to immediately terminate the Service with respect to which the default occurred, and (2) if the non-defaulting party is McDermott, McDermott shall have the right, at its sole discretion, to immediately terminate the Service with respect to which the default occurred, in either case if the defaulting party has failed to cure the default within 30 days of receipt of the written notice of such default. McDermott's right to terminate this Agreement pursuant to this Article VII and the rights set forth in Section 2.5 shall constitute McDermott's sole and exclusive rights and remedies for a breach by B&W hereunder (including any breach caused by an Affiliate of B&W or other third party providing a Service hereunder).

ARTICLE VIII INDEMNIFICATION AND WAIVER

Section 8.1 Waiver of Consequential Damages. NEITHER PARTY SHALL BE LIABLE TO THE OTHER PARTY UNDER THIS AGREEMENT FOR ANY EXEMPLARY, PUNITIVE, SPECIAL, INDIRECT, CONSEQUENTIAL, REMOTE OR SPECULATIVE DAMAGES (INCLUDING IN RESPECT OF LOST PROFITS OR REVENUES), HOWEVER CAUSED AND ON ANY THEORY OF LIABILITY (INCLUDING NEGLIGENCE OR GROSS NEGLIGENCE) ARISING IN ANY WAY OUT OF THIS AGREEMENT, WHETHER OR NOT SUCH PARTY HAS BEEN ADVISED OF THE POSSIBILITY OF SUCH DAMAGES; PROVIDED, HOWEVER, THAT THE FOREGOING LIMITATIONS SHALL NOT LIMIT EACH PARTY'S INDEMNIFICATION OBLIGATIONS FOR LIABILITIES TO THIRD PARTIES AS SET FORTH IN THIS AGREEMENT, THE MASTER SEPARATION AGREEMENT OR ANY ANCILLARY AGREEMENT.

Section 8.2 Services Received. McDermott hereby acknowledges and agrees that:

(a) the Services to be provided hereunder are subject to and limited by the provisions of Section 2.5, Article VII and the other provisions hereof, including the limitation of remedies available to McDermott that restricts available remedies resulting from a Service not

provided in accordance with the terms hereof to non-payment and, in certain limited circumstances, the right to terminate this Agreement;

- (b) the Services are being provided solely to facilitate the transition of McDermott as a separate company as a result of the Distribution, and B&W and its Affiliates do not provide any such Services to non-Affiliates;
- (c) it is not the intent of B&W and the other members of the B&W Group to render, nor of McDermott and the other members of the MII Group to receive from B&W and the other members of the B&W Group, professional advice or opinions, whether with regard to tax, legal, treasury, finance, employment or other business and financial matters, or technical advice, whether with regard to information technology or other matters; McDermott shall not rely on, or construe, any Service rendered by or on behalf of B&W as such professional advice or opinions or technical advice; and McDermott shall seek all third-party professional advice and opinions or technical advice as it may desire or need, and in any event McDermott shall be responsible for and assume all risks associated with the Services, except to the limited extent set forth in Section 2.5 and Article VII;
- (d) with respect to any software or documentation within the Services, McDermott shall use such software and documentation internally and for their intended purpose only, shall not distribute, publish, transfer, sublicense or in any manner make such software or documentation available to other organizations or persons, and shall not act as a service bureau or consultant in connection with such software; and
- (e) a material inducement to B&W's agreement to provide the Services is the limitation of liability and the release provided by McDermott in this Agreement.

ACCORDINGLY, EXCEPT WITH REGARD TO THE LIMITED REMEDIES EXPRESSLY SET FORTH HEREIN, MCDERMOTT SHALL ASSUME ALL LIABILITY FOR AND SHALL FURTHER RELEASE, DEFEND, INDEMNIFY AND HOLD B&W, ANY MEMBER OF THE B&W GROUP AND THEIR RESPECTIVE EMPLOYEES, OFFICERS, DIRECTORS AND AGENTS (ALL AS INDEMNIFIED PARTIES) FREE AND HARMLESS FROM AND AGAINST ALL LOSSES RESULTING FROM, ARISING OUT OF OR RELATED TO THE SERVICES, HOWSOEVER ARISING AND WHETHER OR NOT CAUSED BY THE NEGLIGENCE OR GROSS NEGLIGENCE OF B&W, ANY MEMBER OF THE B&W GROUP OR ANY THIRD PARTY SERVICE PROVIDER, OTHER THAN THOSE LOSSES CAUSED BY THE WILLFUL MISCONDUCT OF B&W OR ANY MEMBER OF THE B&W GROUP.

Section 8.3 Express Negligence. THE INDEMNITY, RELEASES AND LIMITATIONS OF LIABILITY IN THIS AGREEMENT (INCLUDING ARTICLES II AND VIII) ARE INTENDED TO BE ENFORCEABLE AGAINST THE PARTIES IN ACCORDANCE WITH THE EXPRESS TERMS AND SCOPE THEREOF NOTWITHSTANDING ANY EXPRESS NEGLIGENCE RULE OR ANY SIMILAR DIRECTIVE THAT WOULD PROHIBIT OR OTHERWISE LIMIT INDEMNITIES BECAUSE OF THE NEGLIGENCE OR GROSS NEGLIGENCE (WHETHER SOLE,

JOINT OR CONCURRENT OR ACTIVE OR PASSIVE) OR OTHER FAULT OR STRICT LIABILITY OF ANY OF THE INDEMNIFIED PARTIES.

ARTICLE IX CONFIDENTIALITY

Section 9.1 Confidentiality. B&W and McDermott each acknowledge and agree that the terms of Section 6.9 of the Master Separation Agreement shall apply to information, documents, plans and other data made available or disclosed by one party to the other in connection with this Agreement. B&W and McDermott each acknowledge and agree that any third party Information (to the extent such Information does not constitute B&W Books and Records) provided by any member of the MII Group to any member of the B&W Group after the Distribution Date in connection with the provision of the Services by any member of the B&W Group after the Distribution Date, in each case that primarily relates to the MII Business, the MII Assets, or the MII Liabilities, shall not be considered Privileged Information of B&W or Confidential Information of B&W.

Section 9.2 System Security.

- (a) If any party hereto is given access to the other party's computer systems or software (collectively, the "Systems") in connection with the Services, the party given access (the "Availed Party") shall comply with all of the other party's system security policies, procedures and requirements that have been provided to the Availed Party in advance and in writing (collectively, "Security Regulations"), and shall not tamper with, compromise or circumvent any security or audit measures employed by such other party. The Availed Party shall access and use only those Systems of the other party for which it has been granted the right to access and use.
- (b) Each party hereto shall use commercially reasonable efforts to ensure that only those of its personnel who are specifically authorized to have access to the Systems of the other party gain such access, and use commercially reasonable efforts to prevent unauthorized access, use, destruction, alteration or loss of information contained therein, including notifying its personnel of the restrictions set forth in this Agreement and of the Security Regulations.
- (c) If, at any time, the Availed Party determines that any of its personnel has sought to circumvent, or has circumvented, the Security Regulations, that any unauthorized Availed Party personnel has accessed the Systems, or that any of its personnel has engaged in activities that may lead to the unauthorized access, use, destruction, alteration or loss of data, information or software of the other party hereto, the Availed Party shall promptly terminate any such person's access to the Systems and immediately notify the other party hereto. In addition, such other party hereto shall have the right to deny personnel of the Availed Party access to its Systems upon notice to the Availed Party in the event that the other party hereto reasonably believes that such personnel have engaged in any of the activities set forth above in this Section 9.2(c) or otherwise pose a security concern. The Availed Party shall use commercially reasonable efforts to cooperate with the other party hereto in investigating any apparent unauthorized access to such other party's Systems.

ARTICLE X FORCE MAJEURE

Section 10.1 Performance Excused. Continued performance of a Service may be suspended immediately to the extent caused by any event or condition beyond the reasonable control of the party suspending such performance (and not involving any willful misconduct of such party), including acts of God, pandemics, floods, fire, earthquakes, labor or trade disturbances, strikes, war, acts of terrorism, civil commotion, electrical shortages or blackouts, breakdown or injury to computing facilities, compliance in good faith with any Law (whether or not it later proves to be invalid), unavailability of materials or bad weather (a "Force Majeure Event"). McDermott shall not be obligated to pay any amount for Services that it does not receive as a result of a Force Majeure Event (and the parties hereto shall negotiate reasonably to determine the amount applicable to such Services not received). In addition to the reduction of any amounts owed by McDermott hereunder, during the occurrence of a Force Majeure Event, to the extent the provision of any Service has been disrupted or reduced, during such disruption or reduction, (a) McDermott may replace any such affected Service by providing any such Service for itself or engaging one or more third parties to provide such Service at the expense of McDermott and (b) B&W shall cooperate with, provide such information to and take such other actions as may be reasonably required to assist such third parties to provide such substitute Service.

Section 10.2 *Notice*. The party claiming suspension due to a Force Majeure Event will give prompt notice to the other of the occurrence of the Force Majeure Event giving rise to the suspension and of its nature and anticipated duration.

Section 10.3 Cooperation. Upon the occurrence of a Force Majeure Event, the parties shall cooperate with each other to find alternative means and methods for the provision of the suspended Service.

ARTICLE XI MISCELLANEOUS

Section 11.1 Entire Agreement. This Agreement, together with the documents referenced herein (including the Master Separation Agreement), constitutes the entire agreement and understanding between the parties hereto with respect to the subject matter hereof and supersedes all prior written and oral and all contemporaneous oral agreements and understandings with respect to the subject matter hereof. To the extent any provision of this Agreement conflicts with the provisions of the Master Separation Agreement, the provisions of this Agreement shall be deemed to control with respect to the subject matter hereof.

Section 11.2 Binding Effect; No Third-Party Beneficiaries; Assignment. This Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and permitted assigns; and nothing in this Agreement, express or implied, is intended to confer upon any other person or entity any rights, benefits or remedies of any nature whatsoever under or by reason of this Agreement. This Agreement may not be assigned by either party hereto, except with the prior written consent of the other party hereto.

Section 11.3 Amendment; Waivers. No change or amendment may be made to this Agreement except by an instrument in writing signed on behalf of both of the parties hereto. Either party hereto may, at any time, (i) extend the time for the performance of any of the obligations or other acts of the other, (ii) waive any inaccuracies in the representations and warranties of the other contained herein or in any document delivered pursuant hereto, and (iii) waive compliance by the other with any of the agreements, covenants or conditions contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the party to be bound thereby. No failure or delay on the part of either party hereto in the exercise of any right hereunder shall impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty, covenant or agreement contained herein, nor shall any single or partial exercise of any such right preclude other or further exercise thereof or of any other right.

Section 11.4 Notices. Unless otherwise expressly provided herein, all notices, claims, certificates, requests, demands and other communications hereunder shall be in writing and shall be deemed to be duly given (i) when personally delivered or (ii) if mailed by registered or certified mail, postage prepaid, return receipt requested, on the date the return receipt is executed or the letter is refused by the addressee or its agent or (iii) if sent by overnight courier which delivers only upon the signed receipt of the addressee, on the date the receipt acknowledgment is executed or refused by the addressee or its agent or (iv) if sent by facsimile or electronic mail, on the date confirmation of transmission is received (provided that a copy of any notice delivered pursuant to this clause (iv) shall also be sent pursuant to clause (i), (ii) or (iii)), addressed to the attention of the addressee's General Counsel at the address of its principal executive office or to such other address or facsimile number for a party hereto as it shall have specified by like notice.

Section 11.5 Counterparts. This Agreement, including the Schedules hereto and the other documents referred to herein, may be executed in multiple counterparts, each of which when executed shall be deemed to be an original but all of which together shall constitute one and the same agreement.

Section 11.6 Severability. If any term or other provision of this Agreement or the Schedules attached hereto is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to either party hereto. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the court, administrative agency or arbitrator shall interpret this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible. If any sentence in this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.

Section 11.7 Governing Law. This Agreement shall be governed by, and construed and enforced in accordance with, the substantive laws of the State of Texas, without regard to any conflicts of law provisions thereof that would result in the application of the laws of any other jurisdiction.

Section 11.8 *Performance*. Each party hereto shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth herein to be performed by any Subsidiary or Affiliate of such party.

Section 11.9 *Relationship of Parties*. This Agreement does not create a fiduciary relationship, partnership, joint venture or relationship of trust or agency between the parties. The parties hereto agree that B&W (and any other member of the B&W Group which performs Services hereunder) is an independent contractor in the performance of Services for the MII Group under this Agreement.

Section 11.10 Regulations. All employees of B&W and the members of the B&W Group shall, when on the property of McDermott, conform to the rules and regulations of McDermott concerning safety, health and security which are made known to such employees in advance in writing.

Section 11.11 Construction. This Agreement shall be construed as if jointly drafted by the parties hereto and no rule of construction or strict interpretation shall be applied against either party. In this Agreement, unless the context clearly indicates otherwise, words used in the singular include the plural and words used in the plural include the singular; and if a word or phrase is defined in this Agreement, its other grammatical forms, as used in this Agreement, shall have a corresponding meaning. Whenever the context requires, the gender of all words used in this Agreement includes the masculine, feminine and the neuter. Unless the context otherwise requires, the words "include," "includes" and "including" shall be deemed to be followed by the words "without limitation," and the word "or" shall have the inclusive meaning represented by the phrase "and/or." The words "shall" and "will" are used interchangeably in this Agreement and have the same meaning. Relative to the determination of any period of time hereunder, "from" means "from and including," "to" means "to but excluding" and "through" means "through and including." All references herein to a specific time of day in this Agreement shall be based upon Central Standard Time or Central Daylight Savings Time, as applicable, on the date in question. Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. Any reference herein to any clause means such clause of such Section or definition. As used in this Agreement, the words "this Agreement," "herein," "herences," "hereof," "hereof," "hereto" and words of similar import shall be deemed references to this Agreement as a whole and not to any particular Section or other provision of this Agreement. The titles to Articles and headings of Sections contained in this Agreement, in any Schedule and in the table of contents to this Agreement have been inserted for convenience of reference only and shall not be deemed to be a part of or to aff

Section 11.12 Effect if Separation does not Occur. If the Distribution does not occur, then all actions and events that are, under this Agreement, to be taken or occur effective as of or following the Distribution Date, or otherwise in connection with the Distribution, shall not be taken or occur except to the extent specifically agreed by the parties and neither party shall have any liability or further obligation to the other party under this Agreement.

[Signature page follows.]

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the date first written above.

THE BABCOCK & WILCOX COMPANY

| By: | |
|-----|--------------------------------------|
| | Name: |
| | Title: |
| | |
| | |
| MCD | ERMOTT INTERNATIONAL, INC. |
| MCD | Eldio II II I Eld II II or III, II o |
| | |
| By: | |
| | Name: |
| | Title: |
| | |

Schedule A

Tax Services

Description of Services to be Provided:

Services to be provided as part of Tax Transition Services:

- $(A) \quad \mbox{ Assistance with 2009 and 2010 tax compliance for the following entities / jurisdictions:} \\$
 - J. Ray Holdings, Inc. & Subsidiaries as part of the Babcock & Wilcox U.S. consolidated return
 - J. Ray McDermott, Inc. Louisiana
 - J. Ray McDermott, Inc. Alabama
 - d. J. Ray McDermott, Inc. - Mississippi
 - e.
 - J. Ray McDermott, Inc. Texas J. Ray McDermott, Inc. Florida f.
 - Sabine River Realty, Inc. Louisiana
- Assistance with the calculation of the domestic tax provision to be included in the Q2 2010 (and Q3 2010, if applicable) Form 10-Q for McDermott International, Inc.
- Miscellaneous subject matter expertise on an as needed basis (C)

Service Fee:

Amount: \$8,861 per month

United States Dollars (USD) Currency:

Termination Date:

Services provided until all 2010 tax compliance is complete (not later than November 15, 2011).

Schedule B

Accounting and Financial Reporting Services

Areas of Service:

Assistance in Preparation of Quarterly/Annual Filings and Shareholder Reports

- o Assistance with preparation and filing of MII's quarterly and annual SEC filings (10Q/10K), in accordance with the details specified on Annex A hereto.
- o Assistance with managing the accounting close and consolidations efforts.
- o Assistance with monitoring and managing collection of footnotes and coordination of SEC filings on the Securities and Exchange Commission's EDGAR system (XBRL / R.R. Donnelly).
- Assistance with evaluating stock option volatility measures and required documentation.
- o Provision of support in the accounting for equity and pension plans (FAS 123R stock expense, compensation, options, 10Q/10K disclosures, thrift plan, APIC, stock compensation).
- o Provision of support in the development of select.
- o Assistance with performing various analyses during the quarterly and annual close processes to assist MII's management with understanding current results and the latest forecast, with variance explanations and with information for earnings calls with analysts and others.

Additional Subject Matter Support

- o Assistance with accounting for SERP and Excess plan.
- o Assistance with healthcare & benefits accounting (including pensions).
- o Assistance with equity accounting.
- o Assistance with SGA expense analyses and reporting (Blue Book).
- Assistance with thrift plan funding.
- Assistance with sales and use tax compliance for pre-spin periods.

Special Provisions:

- o Formal training courses will be provided if required, subject to availability of qualified trainers at rates to be agreed (not included in Service Fee below).
- o Except as necessitated by completion of the Distribution, preparation of quarterly and annual financial statements and reporting will be consistent with historical practice.

Service Fee:

Amount: Currency:

\$13,858 per month United States Dollars (USD) Service fee above does not include interim executive costs Note:

Termination Date:

Services provided up to one year from the Distribution Date.

Schedule C

Corporate Secretary Services

Areas of Service:

- Provide general advice regarding securities and corporate governance matters.
- Provide consulting and advisory assistance regarding fiscal year proxy statement (incl. compensation discussion & analysis). 2.

Special Provisions:

- Formal training courses to be provided if required subject to availability of qualified trainers at rates to be agreed (not included in Service Fee below). Support limited to the business units and associated transactions and activities as of the Distribution Date. Preparation of annual proxies consistent with historical practice.

Service Fee:

Amount: \$3,075 per month

United States Dollars (USD) Currency:

Termination Date:

Services provided up to 9 months from the Distribution Date.

Schedule D Legal Services

Description of Services to be Provided:

Services to be provided as part of Legal Transition Services:

- (A) Qualified Plans Support
- (B) Nonqualified Deferred Compensation Plans Support
- (C) Employee Benefit Plan Committees Support
- (D) Employment and Separation Agreement Support
- (E) Bell v. J. Ray McDermott, Inc. Support
- (F) 2009 Form 5500 Support
- (G) Annual EPL and FL insurance renewal
- (H) HR Legal Support

(A) Qualified Plans Support

- o Assistance as reasonably requested for tax and legal compliance of qualified plans:
 - Plan design, amendment and drafting
 - Reporting and disclosure
 - QDRO and claims administration
 - Governmental agency filings
 - Plan interpretation
 - Acting as liaison with plan service providers

(B) Nonqualified Deferred Compensation Plans Support

- Assistance as reasonably requested for tax and legal compliance of nonqualified deferred compensation programs:
 - Plan design, amendment and drafting
 - Code Section 409A issues
 - Code Section 457A issues
 - Reporting and disclosure

(C) Employee Benefit Plan Committees Support

- o Continue to serve as nonmember secretary to the following committees:
 - U.S. and TCN Retirement Plan Investment Committees
 - Thrift Plan Committee
 - TCN Retirement Plan Administrative Committee

(D) Employment and Separation Agreement Support

- o Assistance as reasonably requested for:
 - Design and drafting of employment agreements and separation agreements
 - · Administration of existing agreements
 - · Statutory and regulatory compliance

(E) Bell v J. Ray McDermott, Inc. Support

- o Assistance as reasonably requested for:
 - · Acting as liaison with outside counsel, company employees and witnesses
 - Reviewing outside counsel bills
 - · Setting case strategy
 - Reviewing pleadings and other court filings

(F) 2009/2010 Form 5500 Support

o Oversight of preparation and filing of 2009 Form 5500 for MI/JRM employee benefit plans and final 2010 filings

(G) EPL and FL Insurance Annual Renewal Support

- o Assistance as reasonably requested for:
 - Research and preparation of responses for renewal applications
 - Compilation of supporting documentation
 - Preparation of presentation materials for underwriters
 - Responding to questions from underwriters and brokers

(H) HR Legal Support

- o Assistance as reasonably requested for:
 - · HR related statutory and regulatory compliance
 - M&A due diligence

Service Fee:

Amount: \$4,690 per month

Currency: United States Dollars (USD)

Termination Date:

Services provided up to nine months from the Distribution Date.

Schedule E

Human Resources Services

Description of Services to be Provided:

Services to be provided as part of Human Resources Transition Services:

- (A) Assistance with Payroll Management/Oversight and Processing Support
- (B) Assistance with Stock Plan Administration

(A) Assistance with Payroll Management/Oversight and Processing Support

- Provide the services of the Director, Americas Payroll Service Center for assistance with oversight of payroll processing
- o Assistance as reasonably requested for:
 - Pay issue troubleshooting
 - Special payment processing (bonus, vacation, pay advances)
 - Garnishment processing
 - · Payroll tax filings and research
 - Payroll processing back-up

(B) Assistance with Stock Plan Administration

- Assist with administration of the long-term incentive and supplemental executive retirement programs, to include:
 - · Managing Schwab service delivery
 - Resolving issues and responding to inquires from plan participants
 - Compliance and external reporting
 - Training new Stock Plan Administrator

Service Fee:

Amount: \$5,754

Currency: US Dollars (USD)

Termination Date:

Services provided for up to six months from the Transaction Date.

Schedule F

Risk Management Services

Description of Services to be Provided:

Miscellaneous advice and consulting as requested concerning:

- Historic MII CIRM activities, documents and processes including, but not limited to, insurance coverages, captive operations and coverages, and RIMS systems.
- 2. Risk identification, assessment, and treatment (including financing via transfer or captive).
- 3. Removal of B&W related rights and obligations from Boudin and Non-B&W obligations from Creole.
- 4. Support migration from current RMIS System to new RMIS system.
- 5. New employee training.

Charges:

\$1,148 per month for three months from the effective date hereof. After three months, charges to be agreed if any services still are being provided.

Schedule G

Treasury Services

Description of Services to be Provided:

Services to be provided as part of Treasury Transition Services:

(A) Assistance with managing surety bonding and guarantees

Each such service category is described below.

(A) Assistance with managing Bonding and Guarantees

- o Assistance with establishing lines with surety companies
- o Assistance with managing issuance of and termination of surety bonds
- o Assistance with managing issuance of and termination of all letters of credit
- o Assistance with managing the issuance of parent company guarantees

Service Fee

Amount: \$5,032 per month

Termination Date

Service provided up to nine months from the Distribution Date.

Schedule H

Sharepoint

Description of Services to be Provided:

Support services described in this schedule are those required to support the Sharepoint site that has been created for collaboration and sharing of documentation between the records management personnel of McDermott and B&W (the "Sharepoint Site"). The Sharepoint Site will include information such as detailed inventory lists of records forwarded to B&W and a log documenting when such information was forwarded. Records management personnel from McDermott and B&W may load new documents as well as modify and edit documents already loaded on the Sharepoint Site. B&W will routinely back-up the Sharepoint Site, consistent with B&W's back-up procedures for its other sites and applications of a similar nature, and provide a complete electronic copy of the Sharepoint Site to McDermott at the end of the Sharepoint Transition Period.

Service Fee:

Amount: 50% of all customary costs associated with maintaining the Sharepoint Site.

Currency: US Dollars (USD).

Termination Date:

B&W will provide the Services for twenty-four (24) months from the Distribution Date, and such period may be extended by an additional six (6) months at the request of either B&W or McDermott ("Sharepoint Transition Period").

Schedule I

Disaster Recovery

Description of Services to be Provided:

Support services described in this schedule are those required to provide disaster recovery for McDermott's data center in Houston (the <u>'Houston Data Center</u>") during the Disaster Recovery Transition Period. In the event of an extended interruption in service to the Houston Data Center, B&W will provide the following services to McDermott from B&W's PGG Barberton data center:

- · receiving and storage of backup tapes;
- · adequate data center floor space for two standard equipment racks;
- six 30-Amp power feeds;
- work space for up to six McDermott IT personnel;
- · landline telephone equipment/service and external internet access; and
- infrastructure support from B&W IT personnel as reasonably required to establish and make operational the McDermott recovery team in the PGG Barberton data center.

Such services are collectively referred to herein as the "Extended Interruption Services." A period of Extended Interruption Services remains in effect until McDermott removes from B&W's PGG Barberton data center all equipment installed in response to the Houston Data Center service disruption.

In the event McDermott anticipates a disruption in service to the Houston Data Center due to hurricane or other foreseeable event, B&W will provide the following services to McDermott from B&W's PGG Barberton data center:

- · receiving and storage of back up tapes; and
- · work space for up to two McDermott IT personnel.

Such services are collectively referred to herein as the "<u>Foreseeable Interruption Services</u>." A period of Foreseeable Interruption Services remains in effect until McDermott removes from B&W's PGG Barberton data center all equipment installed in response to the Houston Data Center service disruption.

Service Fee:

Amount:

- \$2,100 per month for those months in which the Extended Interruption Services are provided.
- \$1,000 per month for those months in which the Foreseeable Interruption Services are provided. If Extended Interruption Services are subsequently provided, those charges would also apply.

Currency: US Dollars (USD).

Termination Date:

B&W will provide the option for McDermott to utilize the Extended Interruption Services and Foreseeable Interruption Services from the Distribution Date until December 1, 2010, unless earlier terminated upon thirty (30) days' prior written notice provided to B&W by McDermott ("<u>Disaster Recovery Transition Period</u>").

Schedule 2.4 Certain Subcontractors

None.

Schedule 4.1

Payment Instructions

Until further notice, the following are wire transfer/ACH payment instructions for payment to B&W for amounts owing under this Agreement:

The Babcock & Wilcox Company

JP Morgan Chase Bank New York 021000021 Bank Name: Bank ABA #:

Account Number: The Babcock & Wilcox Company 304607029

ASSUMPTION AND LOSS ALLOCATION AGREEMENT

by and among

ACE American Insurance Company, acting for itself and the ACE Affiliates (as defined below)

and

MCDERMOTT INTERNATIONAL, INC.,

a Panamanian corporation

and

BABCOCK & WILCOX HOLDINGS, INC.,

a corporation organized and existing under the laws of the State of Delaware

RECITALS

THIS ASSUMPTION AND LOSS ALLOCATION AGREEMENT (the "Agreement"), is entered into and effective as of May 18, 2010 (the 'Effective Date") by and among ACE AMERICAN INSURANCE COMPANY, individually and acting for the ACE Affiliates (in such capacities, the "Company"), MCDERMOTT INTERNATIONAL, INC., a Panamanian corporation ("MII"), BABCOCK & WILCOX HOLDINGS, INC., a Delaware corporation ("B&W"), to be succeeded by The Babcock & Wilcox Company, a Delaware corporation, after the effective time of the Merger, and, solely with respect to Sections 2, 3 and 5(c), the other MII Entities signatory hereto and the other B&W Entities signatory hereto.

WHEREAS, the Company and/or the ACE Affiliates have issued the Existing Policies to one or more MII Entities and one or more B&W Entities; and

WHEREAS, in connection with the Existing Policies, the Company, the ACE Affiliates, one or more MII Entities, and/or one or more B&W Entities entered into various Existing Insurance Agreements; and

WHEREAS, pursuant to the Existing Policies and the Existing Insurance Agreements, the MII Entities and the B&W Entities are obligated, among other things, to pay or reimburse the Company and/or the ACE Affiliates for certain Obligations, which Obligations are secured by the Existing Collateral; and

WHEREAS, B&W, prior to the Separation, is a wholly owned Subsidiary of MII; and

WHEREAS, MII intends to spin-off B&W from MII through a dividend of common stock of B&W to the shareholders of MII (the Separation"); and

WHEREAS, in connection with the Separation: (a) the MII Entities desire to transfer and the B&W Entities desire to assume any B&W Obligations that were incurred by, or with respect to which there exists any obligation of, an MII Entity, whether such B&W Obligations were existing, accruing or arising before, on or after the Effective Date; and (b) the B&W Entities desire to transfer and the MII Entities desire to assume any MII Obligations that were incurred by, or with respect to which there exists any obligation of, a B&W Entity, whether such MII Obligations were existing, accruing or arising before, on or after the Effective Date; and

WHEREAS, the Company, on its own behalf and on behalf of the ACE Affiliates, is willing to consent to the transfer and assumption of the Obligations as set forth herein, subject to the terms and conditions of this Agreement;

NOW, THEREFORE, in consideration of the mutual promises set out herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, including a one-time administrative fee of \$25,000 (the "Fee"), and intending to be legally bound, the Parties agree as follows:

- 1. <u>Definitions</u>. The following terms used herein, including in the recitals and Exhibits hereto, shall have the following meanings:
- "ACE Affiliate" means each Affiliate of ACE American Insurance Company that is listed on Exhibit V attached hereto and made a part hereof that has issued an Existing Policy or is party to an Existing Insurance Agreement.
- "Affiliate" means, with respect to any specified Person, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, the specified Person. For this purpose "control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise.
 - "Agreement" has the meaning set forth in the recitals to this Agreement.
 - "Assumption Time" means midnight (New York time) on the Effective Date.
 - "B&W" has the meaning set forth in the recitals to this Agreement.
 - "B&W Assumed Obligations" has the meaning set forth in Section 5(b).
 - "B&W Assumption" has the meaning set forth in Section 2(a).
- "B&W Entity" means B&W and each of the entities listed on Exhibit I attached hereto and made a part hereof. It is acknowledged and understood that, from and after the effectiveness of the Separation, the B&W Entities will not be Subsidiaries or Affiliates of MII or any of the other MII Entities.
 - "B&W LOC" has the meaning set forth in Section 5(b).
- "B&W Obligations" means any Obligations of, or to the extent arising from the operations, business, or property of, a B&W Entity for which any MII Entity is responsible under the terms of an Existing Policy or Existing Insurance Agreement, whether arising prior to, at or after the Effective Date.
 - "B&W Retained Obligations" has the meaning set forth in Section 2(d).
 - "Cash Collateral" has the meaning set forth in Section 7.
 - "Company" has the meaning set forth in the recitals to this Agreement.
 - "Company Designation" has the meaning set forth in Section 4(a).
 - "Effective Date" has the meaning set forth in the recitals to this Agreement.
 - "ESIS" means ESIS, Inc., an Affiliate of the Company.
- "Existing Collateral" means any and all of the following forms of security held by the Company or any ACE Affiliate under the terms of any Existing Policy or Existing Insurance Agreement in order to secure any Obligations outstanding as of the date hereof: (i) any and all letters of credit outstanding as of the date hereof provided by or required to be provided by a B&W Entity or a MII Entity; (ii) any and all Cash Collateral provided by or required to be provided by a B&W Entity or a MII Entity pursuant to any Existing Insurance Agreement; or (iv) any other collateral or security previously provided by a B&W Entity or a MII Entity under the terms of any Existing Policy or Existing Insurance Agreement in order to secure any Obligations outstanding as of the date hereof.

"Existing ESIS Agreement" means any agreement relating to claims or losses under one or more Existing Policies in which ESIS is in direct contractual privity with any MII Entity or any B&W Entity.

"Existing Insurance Agreement" means any agreement entered into on or prior to the date hereof by or on behalf of (or which is otherwise binding on) any B&W Entity and/or MII Entity with the Company or an ACE Affiliate in connection with an Existing Policy, including, without limitation, any high deductible agreement, any notice of election, any collateral agreement, any agreement relating to any deductible or paid loss retrospectively rated insurance program, any agreement relating to deductibles under any of the Existing Policies, any letter or agreement relating to policy dividends, any early close-out agreement relating to any Existing Policy or Existing Insurance Agreement and any agreement described on Exhibit II and Exhibit VI attached hereto and made a part hereof.

"Existing Policy" means each policy of general liability insurance, automobile liability insurance and workers compensation insurance (other than any insurance policy that is the subject of any reinsurance agreement) issued prior to the date hereof by the Company or an ACE Affiliate to a B&W Entity or a MII Entity, as applicable, including those policies identified on Exhibit III and Exhibit VII attached hereto and made a part hereof.

"Fee" has the meaning set forth in the recitals to this Agreement.

"Foreign Insurance Agreements" means the Existing Insurance Agreements listed on Exhibit VI attached hereto and made a part hereof and any other similar written agreements entered into between the Company or any of its Affiliates and a MII Entity or a B&W Entity in connection with or relating to insurance policies issued to cover risks located primarily outside of the continental United States during the period from January 1, 1974 through May 7, 2010.

"Foreign Policies" means the Existing Policies listed on Exhibit VII attached hereto and made a part hereof and any other general liability insurance policy issued by the Company or any of its Affiliates to a MII Entity or a B&W Entity to cover risks located primarily outside of the continental United States during the period from January 1, 1974 through May 7, 2010.

"Master Separation Agreement" means a Master Separation Agreement to be entered into between MII and The Babcock & Wilcox Company in connection with the Separation.

"Merger" means the merger, to occur after the date hereof, of Babcock & Wilcox Holdings, Inc., a Delaware corporation, with and into The Babcock & Wilcox Company, a Delaware corporation and the surviving entity of such Merger.

"MII" has the meaning set forth in the recitals to this Agreement.

"MII Cash Collateral" has the meaning set forth in Section 5(a).

"MII Assumed Obligations" has the meaning set forth in Section 5(a).

"MII Assumption" has the meaning set forth in Section 2(c).

"MII Entity" means MII and each of the entities listed on Exhibit IV attached hereto and made a part hereof. It is acknowledged and understood that, from and after the effectiveness of the Separation, the MII Entities will not be Subsidiaries or Affiliates of B&W or any of the other B&W Entities.

"MII LOC" has the meaning set forth in Section 5(a).

"MII Obligations" means any Obligations of, or to the extent arising from the operations, business, or property of, a MII Entity for which any B&W Entity is responsible under the terms of an Existing Policy or Existing Insurance Agreement, whether arising prior to, at or after the Effective Time.

"MII Retained Obligations" has the meaning set forth in Section 2(b)

"Obligations" means any and all amounts, duties, liabilities and obligations, whether accrued, fixed or contingent, mature or inchoate, known or unknown, including deductibles and premium adjustments, payable by or to be performed by a MII Entity or a B&W Entity to the Company or any ACE Affiliate under the terms of any Existing Policy or any Existing Insurance Agreement.

"Organizational Documents" means (a) with respect to any corporation, its certificate or articles of incorporation or organization and its bylaws, (b) with respect to any limited partnership, its certificate of limited partnership agreement, (c) with respect to any general partnership, its partnership agreement, and (d) with respect to any limited liability company, its certificate or articles of formation or organization and its operating agreement or other organizational documents.

"Parties" means the Company, MII and B&W, collectively (and each individually is a "Party").

"Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, a union, an unincorporated organization or a governmental entity or any department, agency or political subdivision thereof.

"Separation" has the meaning set forth in the recitals to this Agreement.

"Subsidiary" means, with respect to any specified Person, any corporation, partnership, limited liability company, joint venture or other organization, whether incorporated or unincorporated, of which at least a majority of the securities or interests having by the terms thereof ordinary voting power to elect at least a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such specified Person or by any one or more of its Subsidiaries, or by such specified Person and one or more of its Subsidiaries.

"Substituted Collateral" means (i) the MII LOC and the B&W LOC and (ii) any other collateral or security to be provided on or after the date hereof by a B&W Entity or a MII Entity under the terms of any Existing Policy or Existing Insurance Agreement in order to secure any Obligations outstanding as of the date hereof.

2. Assumption.

- (a) <u>B&W Assumption</u>. Notwithstanding anything in any Existing Insurance Agreement or Existing Policy to the contrary, each MII Entity that is a signatory hereto hereby transfers and assigns, and B&W does hereby assume, effective as of the Assumption Time, the B&W Obligations; and B&W hereby agrees to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of any MII Entity under the Existing Policies and Existing Insurance Agreements in respect of the B&W Obligations (the "<u>B&W Assumption</u>"). The Company, on its own behalf and on behalf of the ACE Affiliates, hereby consents to, and agrees to give full force and effect to, the B&W Assumption. From and after the Assumption Time, the Company (and/or the applicable ACE Affiliate): (i) may enforce its rights under the Existing Policies and the Existing Insurance Agreements in respect of the B&W Obligations against B&W to the same extent such Person could, prior to the B&W Assumption, enforce such rights against the applicable MII Entity and (ii) releases each MII Entity from its obligation to observe, pay, perform, satisfy, fulfill or discharge any such B&W Obligations.
- (b) MII Retained Obligations. MII hereby agrees to continue to observe, pay, perform, satisfy, fulfill and discharge any and all of its now existing and hereafter arising Obligations (other than B&W Obligations) (the "MII Retained Obligations") in accordance with the terms of this Agreement and the applicable Existing Policy and Existing Insurance Agreement.
- (c) MII Assumption. Notwithstanding anything in any Existing Insurance Agreement or Existing Policy to the contrary, each B&W Entity that is a signatory hereto hereby transfers and assigns, and MII does hereby assume, effective as of the Assumption Time, the MII Obligations; and MII hereby agrees to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of any B&W Entity under the Existing Policies and Existing Insurance Agreements in respect of the MII Obligations (the "MII Assumption"). The Company, on its own behalf and on behalf of the ACE Affiliates, hereby consents to, and agrees to give full force and effect to, the MII Assumption. From and after the Assumption Time, the Company (and/or the applicable ACE Affiliate): (i) may enforce its rights under the Existing Policies and the Existing Insurance Agreements

in respect of the MII Obligations against MII to the same extent such Person could, prior to the MII Assumption, enforce such rights against the applicable B&W Entity and (ii) releases each B&W Entity from its obligation to observe, pay, perform, satisfy, fulfill or discharge any such MII Obligations.

- (d) <u>B&W Retained Obligations</u>. B&W hereby agrees to continue to observe, pay, perform, satisfy, fulfill and discharge any and all of its now existing and hereafter arising Obligations (other than MII Obligations) (the "<u>B&W Retained Obligations</u>") in accordance with the terms of this Agreement and the applicable Existing Policy and Existing Insurance Agreement.
- (e) Obligations of the Company and the ACE Affiliates. For the avoidance of doubt, the Parties acknowledge that nothing in this Agreement shall discharge, limit or in any way affect the obligations of the Company or the ACE Affiliates as insurers under any of the Existing Policies. Such obligations shall continue to be performed to the extent and in the manner set forth in the applicable Existing Policy by the Company and/or by the ACE Affiliates, as the case may be, for the benefit of such Persons who are entitled to such performance under the applicable Existing Policy, provided, however, that to the extent that such performance gives rise to Obligations, the responsibility for such Obligations shall be governed by this Agreement.
- (f) Existing ESIS Agreements. The Parties shall use commercially reasonable efforts to enter into an agreement with ESIS promptly after the date hereof pursuant to which ESIS shall acknowledge and consent to the B&W Assumption and the MII Assumption and the other provisions of this Agreement with respect to determining any MII Obligations, MII Retained Obligations, B&W Obligations or B&W Retained Obligations (or allocations thereof) in respect of any Existing ESIS Agreement.
- 3. Joinder. As of the Effective Date, (a) to the extent that B&W is not already a party thereto and an Existing Insurance Agreement contains any B&W Obligations, each Existing Insurance Agreement is hereby deemed amended to add B&W as an "Insured" or other such obligor thereunder solely to the extent necessary to give effect to the B&W Assumption and (b) to the extent that MII is not already a party thereto and an Existing Insurance Agreement contains any MII Obligations, each Existing Insurance Agreement is hereby deemed amended to add MII as an "Insured" or other such obligor thereunder solely to the extent necessary to give effect to the MII Assumption.

4. Allocation.

(a) Company Designations. (i) MII shall continue to pay or perform any and all Obligations constituting MII Retained Obligations pursuant to and in the manner set forth in the applicable Existing Policy and the applicable Existing Insurance Agreement giving rise to such Obligations and (ii) B&W shall continue to pay or perform any and all Obligations constituting B&W Retained Obligations pursuant to and in the manner set forth in the applicable Existing Policy and the applicable Existing Insurance Agreement giving rise to such Obligations; provided, however, that in each case, MII and B&W shall provide to the Company on a timely basis such information as the Company may request so that the Company may determine whether the Obligations constitute MII Retained Obligations. The Company shall determine whether such Obligations constitute MII Retained Obligations or B&W Retained Obligations (the "Company Designation") and shall notify the applicable Party of any such Company Designation. With respect to the Obligations arising out of or relating to the Foreign Policies or the Foreign Insurance Agreements, if the Company is unable to make a determination as to whether any such Obligations constitute MII Retained Obligations or B&W Retained Obligations. B&W agrees that, notwithstanding any dispute or disagreement it may have with respect to any Company Designation, it will pay any B&W Retained Obligation pursuant to and in the manner set forth in the applicable Existing Policy and the applicable Existing Insurance Agreement giving rise to such B&W Retained Obligation pursuant to and in the manner set forth in the applicable Existing Policy and the applicable Existing Insurance

Agreement giving rise to such MII Retained Obligation; provided, however, that such payment shall not be construed as prejudicial to either Party in any dispute between MII and B&W with respect to any such Company Designation.

(b) <u>Disputes.</u> Notwithstanding any dispute or disagreement between MII and B&W concerning a Company Designation, the applicable Party shall pay any amount payable pursuant to a Company Designation as set forth in Section 4(a), and any such dispute or disagreement between MII and B&W shall be resolved pursuant to Article V of the Master Separation Agreement; <u>provided, that</u> (i) until the Master Separation Agreement is executed and delivered by all parties thereto, the reference to such agreement herein shall be deemed to refer to the draft thereof dated as of April 28, 2010; (ii) the Company will not be made a party to any arbitration proceeding arising from such dispute or disagreement, but may be called as a witness; (iii) any costs incurred by the Company in respect of any such arbitration proceeding will be fully reimbursed to the Company equally by the Disputing Parties promptly following receipt of a reimbursement demand from the Company; (iv) under no circumstances will MII or B&W, as a result of such arbitration proceeding, require the Company to return any amount received by the Company pursuant to a prior Company Designation, whether such amount was received as a result of the Company's draw against security posted for its benefit or otherwise, and (v) the Company shall comply with the allocation or other resolution of such dispute established by any award or order of such arbitration, or settlement between the Disputing Parties; and (vi) any indemnification and reimbursement of the Company by B&W and MII pursuant to this Agreement, the Existing Policies and the Existing Insurance Agreements and any other agreement relating to the disputed Company Designation shall be in accordance with the allocation established by such award, order or settlement of such dispute.

5. Collateral and Fee.

(a) MII LOC

- (i) MII will, within fifteen (15) days after the Effective Date, provide to the Company, as beneficiary thereof, (A) cash collateral in an amount of \$687,236 in respect of its Obligations under the Existing Insurance Agreements and the Existing Policies (other than the Foreign Insurance Agreements and Foreign Policies) and (B) cash collateral in an amount of \$6,074,640 in respect of the Foreign Insurance Agreements and Foreign Policies (such cash collateral individually and collectively being referred to herein as the "MII Cash Collateral"). The Company shall return the MII Cash Collateral to MII promptly upon receipt of the MII LOC as set forth below. MII will, within fifteen (15) days after July 1, 2010, provide to the Company, as beneficiary thereof, (A) a clean, irrevocable and unconditional letter of credit in an amount of \$687,236 in respect of its Obligations under the Existing Insurance Agreements and the Existing Policies (other than the Foreign Insurance Agreements and Foreign Policies) and (B) a clean, irrevocable and unconditional letter of credit in an amount of \$6,074,640 in respect of the Foreign Insurance Agreements and Foreign Policies (each such letter of credit individually and collectively being referred to herein as the "MII LOC"), issued in a form and by a bank or other financial institution, in each case acceptable to the Company; and/or such other forms of collateral as the Company may permit from time to time. The MII LOC shall be in an aggregate amount that is less than the aggregate amount of the Existing Collateral provided by MII and shall secure the MII Retained Obligations and the Obligations assumed by MII in the MII Assumption (the "MII Assumed Obligations").
- (ii) The MII LOC shall be "evergreen," meaning that it shall provide by its terms that it will be renewed automatically each year for an additional year unless written notice of non-renewal is received by the Company at least sixty (60) days prior to the MII LOC's anniversary date. If the Company permits MII to provide collateral in a form other than the MII LOC, MII shall provide such collateral in an amount and form acceptable to the Company.
- (iii) MII shall keep the MII LOC in place (or other collateral acceptable to the Company) as security for payment of the MII Retained Obligations and the MII Assumed Obligations, until the

Company determines in its sole discretion that there is no longer any need for such collateral. If there shall be a material deterioration in the financial condition of the bank or other financial institution which has issued the MII LOC, the Company shall have the right to require MII to replace the MII LOC with a new letter of credit with similar terms issued by a bank or other financial institution then acceptable to the Company.

- (iv) The Company shall have the right to draw against the MII LOC and/or other collateral in each instance where any portion of the MII Retained Obligation or the MII Assumed Obligations for any reason is not fulfilled in the manner and within the time periods required under this Agreement or the Existing Policies or Existing Insurance Agreements giving rise thereto.
- (v) Annually, the Company shall review and redetermine the amount of the MII Retained Obligations and the MII Assumed Obligations and the amount of collateral security required pursuant to this Agreement. At such time, MII will provide its most recent audited financial statements, interim financial statements, and any other financial information reasonably requested by the Company for the purpose of evaluating the financial condition of MII. MII will provide any needed increases in the amount of the MII LOC (and/or other collateral if acceptable to the Company) within thirty (30) days of the Company's written request for any additional required amount of the MII LOC. The Company will effect any decreases in the amount of the MII LOC (and/or other collateral) promptly, provided that MII is not in breach of any of its obligations under this Agreement, the Existing Policies or any Existing Insurance Agreement.

(b) B&W LOC

- (i) B&W will, within fifteen (15) days after the Effective Date, provide to the Company, as beneficiary thereof, (A) a clean, irrevocable and unconditional letter of credit in an amount of \$33,828,776 in respect of its Obligations under the Existing Insurance Agreements and the Existing Policies (other than the Foreign Insurance Agreements and Foreign Policies) and (B) a clean, irrevocable and unconditional letter of credit in an amount of \$200,000 in respect of the Foreign Insurance Agreements and Foreign Policies (each such letter of credit individually and collectively being referred to herein as the "B&W LOC"), issued in a form and by a bank or other financial institution, in each case acceptable to the Company; and/or such other forms of collateral as the Company may permit from time to time. The B&W LOC shall be in an aggregate amount that is less than the aggregate amount of the Existing Collateral provided by B&W and shall secure the B&W Retained Obligations and the Obligations assumed by B&W in the B&W Assumption (the "B&W Assumed Obligations").
- (ii) The B&W LOC shall be "evergreen," meaning that it shall provide by its terms that it will be renewed automatically each year for an additional year unless written notice of non-renewal is received by the Company at least sixty (60) days prior to the B&W LOC's anniversary date. If the Company permits B&W to provide collateral in a form other than the B&W LOC, B&W shall provide such collateral in an amount and form acceptable to the Company.
- (iii) B&W shall keep the B&W LOC in place (or other collateral acceptable to the Company) as security for payment of the B&W Retained Obligations and the B&W Assumed Obligations, until the Company determines in its sole discretion that there is no longer any need for such collateral. If there shall be a material deterioration in the financial condition of the bank or other financial institution which has issued the B&W LOC, the Company shall have the right to require B&W to replace the B&W LOC with a new letter of credit with similar terms issued by a bank or other financial institution then acceptable to the Company.
- (iv) The Company shall have the right to draw against the B&W LOC and/or other collateral in each instance where any portion of the B&W Retained Obligations or the B&W Assumed Obligations for any reason is not fulfilled in the manner and within the time periods required under this Agreement or the Existing Policies or Existing Insurance Agreements giving rise thereto.

- (v) Annually, the Company shall review and redetermine the amount of the B&W Retained Obligations and the B&W Assumed Obligations and the amount of collateral security required pursuant to this Agreement. At such time, B&W will provide its most recent audited financial statements, interim financial statements, and any other financial information reasonably requested by the Company for the purpose of evaluating the financial condition of B&W. B&W will provide any needed increases in the amount of the B&W LOC (and/or other collateral if acceptable to the Company) within thirty (30) days of the Company's written request for any additional required amount of the B&W LOC. The Company will effect any decreases in the amount of the B&W LOC (and/or other collateral) promptly, provided that B&W is not in breach of any of its obligations under this Agreement, the Existing Policies or any Existing Insurance Agreement.
- (c) <u>Substituted Collateral</u>. Notwithstanding anything in any Existing Policy or Existing Insurance Agreement to the contrary, the Parties, each B&W Entity that is a party hereto and each MII Entity that is a party hereto hereby agree that, upon receipt of the Substituted Collateral as set forth in Section 5(a)(i) and 5(b) (i), the Existing Collateral shall be replaced with such Substituted Collateral and, accordingly, shall be released by the Company and the ACE Affiliates.
- (d) Fee. No later than fifteen (15) days after the Effective Date, MII shall pay to the Company the Fee, which shall be paid pursuant to the Company's wire instructions as provided to MII in writing prior to the date such Fee is payable.
- **6.** Existing Collateral. The Substituted Collateral required to be provided by MII and B&W hereunder shall, except to the extent provided otherwise in this Agreement, be subject to all of the terms and conditions applicable to the Existing Collateral pursuant to the Existing Insurance Agreements to the same extent that such terms and conditions applied to the Existing Collateral thereunder.
- 7. Security Interest. Each of MII and B&W will separately provide (or have provided) to the Company and ESIS, from time to time, funds to be credited to paid loss deposit funds, deductible funds and/ or loss funds (collectively, "Cash Collateral") that the Company or ESIS shall hold pursuant to the Existing Insurance Agreements with respect to the MII Retained Obligations and the MII Assumed Obligations, in the case of MII, and with respect to the B&W Retained Obligations and the B&W Assumed Obligations, in the case of B&W. Each of MII and B&W hereby grant to the Company, for its benefit and the benefit of the ACE Affiliates, a continuing first priority security interest in and lien on all of their respective right, title and interest, if any, in and to the Cash Collateral and all proceeds thereof as security for their now existing and hereafter arising Obligations to the Company or such ACE Affiliates. The Company shall hold the Cash Collateral in accordance with the terms of the applicable Existing Insurance Agreement pursuant to which such Cash Collateral was provided to the Company or the applicable ACE Affiliate. The Company shall have the sole and exclusive right, and is hereby authorized, to use the Cash Collateral to pay any and all Obligations of MII and/or B&W in accordance with the Company Designation in accordance with the terms of the applicable Existing Insurance Agreement and this Agreement.
- 8. <u>Billing.</u> On and after the Effective Date, the Company (or any third party administrator acting on behalf of the Company in respect of an Existing Policy) will, in each case in accordance with the billing procedures set forth in the applicable Existing Policy and Existing Insurance Agreement:
 - (a) bill MII directly for the MII Retained Obligations and the MII Assumed Obligations; and
 - (b) bill B&W directly for the B&W Retained Obligations and the B&W Assumed Obligations.
- 9. Amendments. Neither this Agreement nor any provision hereof may be amended, changed, waived, discharged or terminated except by a written instrument signed by the Company, B&W, MII and each other Party, if any, against whom enforcement of such amendment, change, waiver, discharge or termination is sought.

10. Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Neither this Agreement nor any right or obligation hereunder may be assigned or conveyed by any Party without the prior written consent of the other Parties, which consent shall not be unreasonably withheld.

11. [Intentionally Omitted].

- 12. No Waiver. The failure or refusal by any Party to exercise any rights granted hereunder shall not constitute a waiver of such rights or preclude the subsequent exercise thereof, and no oral communication shall be asserted as a waiver of any such rights hereunder unless such communication shall be confirmed in a writing plainly expressing an intent to waive such rights and signed by the Party against whom such waiver is asserted.
- 13. <u>Counterparts.</u> This Agreement may be executed in any number of counterparts each of which when so executed and delivered shall constitute an original, but such counterparts together shall constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile transmission or by electronic mail shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of the Parties and other Persons signatory hereto transmitted by facsimile or by electronic mail shall be deemed to be their original signatures for all purposes.
- 14. No Third Party Beneficiary. This Agreement shall not be deemed to give any right or remedy to any third party whatsoever unless otherwise specifically granted hereunder.
- 15. Parties' Representations. As of the Effective Date, each of the Parties expressly represents on its own behalf: (a) it is an entity in good standing in its jurisdiction of organization; (b) it has all requisite corporate power and authority to enter into this Agreement, and to perform its obligations hereunder; (c) the execution and delivery by it of this Agreement, and the performance by it of its obligations under this Agreement, have been duly authorized by all necessary corporate or other action; (d) this Agreement, when duly executed and delivered by it, and subject to the due execution and delivery hereof by the other Parties, will be a valid and binding obligation of it, enforceable against it, its successors and permitted assigns, in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and general equity principles; (e) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby in accordance with the respective terms and conditions hereof will not (i) violate any provision of its Organizational Documents, (ii) violate any applicable order, judgment, injunction, award or decree of any court, arbitrator or governmental or regulatory body against it, or binding upon it, or any agreement with, or condition imposed by, any governmental or regulatory body, foreign or domestic, binding upon it as of the date hereof, or (iii) violate any agreement, contract, obligation, promise or undertaking that is legally binding and to which it is a party or by which it is bound; and (f) the signatory hereto on behalf of it is duly authorized and legally empowered to enter into this Agreement on its behalf.

16. Notices. Any and all notices, requests, approvals, authorizations, consents, instructions, designations and other communications that are required or permitted to be given pursuant to this Agreement shall be in writing and may be given either by personal delivery, first class prepaid post (airmail if to another country) or by internationally recognized overnight delivery service to the following address, or to such other address and recipient as such Party may have notified in accordance with the terms of this section as being its address or recipient for notification for the purposes of this Agreement:

If to the Company ACE American Insurance Company

225 E. John Carpenter Freeway, Suite 1300

Irving, TX 75062

Attention: Underwriting Manager

ACE Risk Management

Telephone: (972) 465.7500 Facsimile: (972) 465.7826

If to any MII Entity: McDermott International, Inc.

757 N. Eldridge Parkway Houston, Texas 77079

Attention: VP and Chief Risk Officer (with copy to General Counsel)

Telephone: 281-870-5785 Telecopier: 281-870-5923

Electronic Mail: twoodard@mcdermott.com

If to any B&W Entity: Prior to the Separation:

Babcock & Wilcox Holdings, Inc.

800 Main Street

Lynchburg, Virginia 24504

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 434-522-6800

with a copy to:

McDermott International, Inc. 777 N. Eldridge Parkway Houston, Texas 77079

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 281-870-5476 Telecopier: 281-870-5923

Electronic Mail: cjryan@mcdermott.com

On and After the Separation:

The Babcock & Wilcox Company

The Harris Building

13024 Ballantyne Corporate Place, Suite 700

Charlotte, North Carolina

Attention: Director, Risk Management (with copy to General Counsel)

Any notice or communication to any Person shall be deemed to be received by that Person:

- (A) upon personal delivery; or
- (B) upon receipt if sent by mail or courier.

- 17. Governing Law. This Agreement shall be governed by and construed in accordance with the internal laws of the Commonwealth of Pennsylvania without regard to those provisions concerning conflicts of laws that would result in the application of the laws of any other jurisdiction.
- 18. Entire Agreement. This Agreement, together with the Existing Policies and Existing Insurance Agreements, constitute the entire agreement among all of the Parties and supersedes all other prior agreements and understandings, both written and oral, with respect to the subject matter hereof.
- 19. <u>Dispute Resolution.</u> If a dispute between either MII or B&W, on the one hand, and the Company or any ACE Affiliate, on the other hand, involves rights or obligations arising under this Agreement, or any of the Existing Policies or Existing Insurance Agreements, the arbitration provisions in the most recent Existing Insurance Agreement referenced in Exhibit II shall govern the resolution of the entire dispute in all respects. In any such arbitration brought by or against MII or B&W, the other of B&W or MII, as applicable, shall have right to associate effectively in the defense and/or prosecution of such arbitration.
- 20. Severability. If any term or other provision of this Agreement or the Exhibits attached hereto is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the court, administrative agency or arbitrator shall interpret this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible. If any sentence in this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- 21. Rules of Construction. The definitions of terms used herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation". The word "will" shall be construed to have the same meaning and effect as the word "shall". Unless the context requires otherwise (a) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified, (b) any reference herein to any law shall be construed as referring to such law as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, (c) any reference herein to any Person shall be construed to include such Person's successors and assigns (subject to the restrictions contained herein), (d) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Agreement in its entirety and not to any particular provision hereof, (e) with respect to the determination of any time period, the word "from" means "from and including" and the word "to" means "to and including" and (f) any reference herein to Articles, Sections and Exhibits shall be construed to refer to Articles and Sections of, and Exhibits to, this Agreement. No provision of this Agreement shall be interpreted or construed against any Person solely because such Person or its legal representative drafted such provision.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the Parties intending to be legally bound hereby have executed this Agreement, by their duly authorized representatives.

ACE AMERICAN INSURANCE COMPANY, on behalf of itself and the ACE Affiliates

| • | /s/ Laura Vest | |
|-----------|-----------------------|--|
| Name: | Laura Vest | |
| Title: | Vice President | |
| MCDERMOTT | Γ INTERNATIONAL, INC. | |
| By: | /s/ LIANE K. HINRICHS | |
| Name: | Liane K. Hinrichs | |
| Title: | Senior Vice President | |

Signature Page to Assumption and Loss Allocation Agreement

ACKNOWLEDGED AND AGREED FOR PURPOSES OF SECTIONS 2, 3 and 5(c):

B&W ENTITIES:

AMERICON EQUIPMENT SERVICES, INC.

AMERICON, INC.

APPLIED SYNERGISTICS, INC.

BABCOCK & WILCOX CANADA LTD.

BABCOCK & WILCOX CHINA HOLDINGS, INC.

BABCOCK & WILCOX CONSTRUCTION CO., INC.

BABCOCK & WILCOX DE MONTERREY, S.A. DE C.V.

 ${\bf BABCOCK~\&~WILCOX~DENMARK~HOLDINGS, INC.}$

BABCOCK & WILCOX EBENSBURG POWER, INC.

BABCOCK & WILCOX EQUITY INVESTMENTS, INC.

BABCOCK & WILCOX INDIA HOLDINGS, INC.

BABCOCK & WILCOX INTERNATIONAL SALES AND SERVICE CORPORATION

BABCOCK & WILCOX INTERNATIONAL, INC.

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash
Title: Assistant Secretary of each of the above

Signature Page to Assumption and Loss Allocation Agreement

BABCOCK & WILCOX MODULAR NUCLEAR ENERGY LLC

BABCOCK & WILCOX NUCLEAR OPERATIONS GROUP, IN $\!C$

BABCOCK & WILCOX NUCLEAR ENERGY, INC.

BABCOCK & WILCOX NUCLEAR SERVICES (U.K.) LIMITED

BABCOCK & WILCOX TECHNICAL SERVICES CLINCH RIVER, LLC

BABCOCK & WILCOX POWER GENERATION GROUP, INC.

BABCOCK & WILCOX TECHNICAL SERVICES GROUP, INC.

BABCOCK & WILCOX TECHNICAL SERVICES SAVANNAH RIVER COMPANY

BABCOCK & WILCOX VOLUND A/S

BCE PARTS LTD.

BWX TECHNOLOGIES, INC.

BWXT FEDERAL SERVICES, INC.

BWXT HANFORD COMPANY

BWXT OF IDAHO, INC.

BWXT OF OHIO, INC.

BWXT WASHINGTON, INC.

By: /S/ BENJAMIN H. BASH

Name: Benjamin H. Bash

Title: Assistant Secretary of each of the above

CREOLE INSURANCE COMPANY, LTD.

DELTA POWER SERVICES, LLC

DIAMOND OPERATING CO., INC.

DIAMOND POWER AUSTRALIA HOLDINGS, INC.

DIAMOND POWER CHINA HOLDINGS, INC.

DIAMOND POWER EQUITY INVESTMENTS, INC.

DIAMOND POWER INTERNATIONAL, INC.

DIAMOND POWER SPECIALTY (PROPRIETARY) LIMITED

DPS BERKELEY, LLC

DPS CADILLAC, LLC

DPS FLORIDA, LLC

DPS GREGORY, LLC

DPS LOWELL COGEN, LLC

DPS MECKLENBURG, LLC

DPS MICHIGAN, LLC

DPS MOJAVE, LLC

DPS SABINE, LLC

GUMBO INSURANCE COMPANY, LTD.

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash
Title: Assistant Secretary of each of the above

INTECH INTERNATIONAL INC.

INTECH, INC.

IVEY-COOPER SERVICES, L.L.C.

MARINE MECHANICAL CORPORATION

NFS HOLDINGS, INC.

NOG-ERWIN HOLDINGS, INC.

NORTH COUNTY RECYCLING, INC.

NUCLEAR FUEL SERVICES, INC.

O&M HOLDING COMPANY

SOFCO - EFS HOLDINGS LLC

By: /s/ Benjamin H. Bash

Name: Benjamin H. Bash

Title: Assistant Secretary of each of the above

BABCOCK & WILCOX TECHNICAL SERVICES (U.K.) LIMITED

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash

Title: Joint Secretary

B&W DE PANAMA, INC.

BABCOCK & WILCOX INTERNATIONAL INVESTMENTS CO., INC.

BABCOCK & WILCOX INVESTMENT COMPANY

| By: | /S/ MICHAEL S. TAFF |
|-------|---------------------|
| Name: | Michael S. Taff |
| DD1.1 | |

Title: Senior Vice President, of each of the above

Signature Page to Assumption and Loss Allocation Agreement

BABCOCK & WILCOX INDIA PRIVATE LIMITED

DIAMOND POWER SPECIALTY LIMITED

| By: | /S/ ROBERT E. STUMPF |
|--------|----------------------|
| Name: | Robert E. Stumpf |
| Title: | Assistant Secretary |
| NATIO | NAL ECOLOGY COMPANY |
| By: | /S/ ROBERT E. STUMPF |
| Name: | Robert E. Stumpf |
| Title: | Sacratary |

DIAMOND POWER DO BRASIL LIMITADA

DIAMOND POWER INTERNATIONAL, INC.

BABCOCK & WILCOX INTERNATIONAL SALES AND SERVICE CORPORATION (as Shareholders of Diamond Power do Brasil Limitada)

By: /S/ ROBERT E. STUMPF
Name: Robert E. Stumpf
Title: Assistant Secretary of each of the Shareholders

P. T. BABCOCK & WILCOX ASIA SERVICIOS DE FABRICACION DE VALLE SOLEADO, S.A. DE C.V.

SERVICOS PROFESIONALES DE VALLE SOLEADO, S.A. DE C.V.

By: /s/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary

Signature Page to Assumption and Loss Allocation Agreement

DIAMOND POWER CENTRAL & EASTERN EUROPE S.R.O.

DIAMOND POWER FINLAND OY

| By: | /s/ Juha K. Mustonen |
|--------|----------------------|
| Name: | Juha K. Mustonen |
| Title: | Managing Director |

 $Signature\ Page\ to\ Assumption\ and\ Loss\ Allocation\ Agreement$

| DIAMOND POWER MACHINE (HUBEI) CO |)., |
|----------------------------------|-----|
| INC. | |

 By:
 /s/ DAVID R. GIBBS

 Name:
 David R. Gibbs

 Title:
 Director

Signature Page to Assumption and Loss Allocation Agreement

| DIAMOND | DOWED | CWEDEN | A D |
|----------|-------|---------|-----|
| DIAVIOND | PUWER | SWEIDEN | AK |

By: /S/ MIKA J. HAIKOLA

Name: Mika J. Haikola

Title: Managing Director

Signature Page to Assumption and Loss Allocation Agreement

ACKNOWLEDGED AND AGREED FOR PURPOSES OF SECTIONS 2, 3 and 5(c):

MII ENTITIES:

CHARTERING COMPANY (SINGAPORE) PTE. LTD.

J. RAY MCDERMOTT (QINGDAO) PTE.

J. RAY MCDERMOTT ASIA PACIFIC PTE. LTD.

MALMAC SDN. BHD.

MCDERMOTT (MALAYSIA) SENDIRIAN BERHAD

By: /S/ JEFF J. HIGHTOWER

Name: Jeff J. Hightower

Title: Director of each of the above

BOUDIN INSURANCE COMPANY, LTD.

J. RAY MCDERMOTT EASTERN HEMISPHERE LIMITED

LAGNIAPPE INSURANCE COMPANY, LTD.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Assistant Secretary, of each of the above

J. RAY MCDERMOTT UK LTD.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Joint Secretary

EASTERN MARINE SERVICES, INC. GLOBAL ENERGY - MCDERMOTT LIMITED

HYDRO MARINE SERVICES, INC.

- J. RAY MCDERMOTT (CASPIAN), INC.
- J. RAY MCDERMOTT (NIGERIA) LTD.
- J. RAY MCDERMOTT (NORWAY), AS
- J. RAY MCDERMOTT CANADA HOLDING, LTD.
- J. RAY MCDERMOTT CANADA, LTD.
- J. RAY MCDERMOTT CONTRACTORS, INC.
- J. RAY MCDERMOTT DE MEXICO, S.A. DE C.V.

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary of each of the above

- J. RAY MCDERMOTT ENGINEERING SERVICES PRIVATE LIMITED
- J. RAY MCDERMOTT ENGINEERING, LLC
- J. RAY MCDERMOTT FAR EAST, INC.
- J. RAY MCDERMOTT INTERNATIONAL, INC.
- J. RAY MCDERMOTT INTERNATIONAL VESSELS, LTD.
- J. RAY MCDERMOTT INVESTMENTS B.V.
- J. RAY MCDERMOTT KAZAKHSTAN LLP
- J. RAY MCDERMOTT LOGISTIC SERVICES PVT. LIMITED
- J. RAY MCDERMOTT MIDDLE EAST, INC.
- J. RAY MCDERMOTT SOLUTIONS, INC.
- J. RAY MCDERMOTT TECHNOLOGY, INC.
- J. RAY MCDERMOTT UNDERWATER SERVICES, INC.
- J. RAY MCDERMOTT WEST AFRICA HOLDINGS, INC.
- J. RAY MCDERMOTT WEST AFRICA, INC.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Secretary of each of the above

MCDERMOTT CASPIAN CONTRACTORS, INC.

MCDERMOTT FAR EAST, INC.

MCDERMOTT GULF OPERATING COMPANY, INC.

MCDERMOTT INTERNATIONAL B.V.

MCDERMOTT INTERNATIONAL INVESTMENTS CO., INC.

MCDERMOTT INTERNATIONAL MARINE INVESTMENTS N.V.

MCDERMOTT INTERNATIONAL TRADING CO., INC.

 $\label{eq:mcdermott} \mbox{ MCDERMOTT INTERNATIONAL VESSELS, } \mbox{ INC. }$

MCDERMOTT MARINE MEXICO, S.A. DE C V

MCDERMOTT OFFSHORE SERVICES COMPANY, INC.

MCDERMOTT OLD JV OFFICE, INC.

MCDERMOTT OVERSEAS INVESTMENT CO. N.V.

MCDERMOTT OVERSEAS, INC.

MCDERMOTT TRADE CORPORATION

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary of each of the above

MENTOR SUBSEA TECHNOLOGY SERVICES, INC.

NORTH ATLANTIC VESSEL, INC.

OFFSHORE PIPELINES INTERNATIONAL, LTD.

OPI VESSELS, INC.

OPMI, LTD.

SABINE RIVER REALTY, INC.

SERVICIOS DE FABRICACION DE ALTAMIRA, S.A. DE C.V.

SERVICOS PROFESIONALES DE ALTAMIRA, S.A. DE C.V.

SPARTEC, INC.

VARSY INTERNATIONAL N.V.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Secretary of each of the above

MCDERMOTT KFT.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Managing Director

J. RAY MCDERMOTT (AUST.) HOLDING PTY. LIMITED

MCDERMOTT AUSTRALIA PTY. LTD.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Corporate Officer of each of the above

MCDERMOTT INCORPORATED

MCDERMOTT PANAMA HOLDINGS, S.A.

J. RAY MCDERMOTT HOLDINGS, LLC

J. RAY MCDERMOTT, INC.

J. RAY MCDERMOTT, S.A.

MCDERMOTT CAYMAN LTD.

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Senior Vice President of each of the above

MCDERMOTT INTERNATIONAL MARKETING, INC.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: President

| By: | |
|--|--|
| NT | /s/ Robert E. Stumpf |
| Name: | Robert E. Stumpf |
| Title: | Assistant Secretary |
| MCDERMOT | T HOLDINGS (U.K.) LIMITED |
| MCDERMOT LIMITED | T MARINE CONSTRUCTION |
| Ву: | /S/ ROBERT E. STUMPF |
| Name: | Robert E. Stumpf |
| Title: | Joint Secretary of each of the above |
| | O, LTDA. CDERMOTT INC. |
| J. RAY MO | CDERMOTT INC. |
| J. RAY MO | CDERMOTT INC. OTT OVERSEAS, INC., |
| J. RAY MO MCDERM | CDERMOTT INC. OTT OVERSEAS, INC., |
| J. RAY MO MCDERM (as Shareho | CDERMOTT INC. OTT OVERSEAS, INC., olders) |
| J. RAY MO MCDERM (as Shareho | CDERMOTT INC. OTT OVERSEAS, INC., olders) /S/ ROBERT E. STUMPF |
| J. RAY MC MCDERM (as Shareho By: Name: Title: | CDERMOTT INC. OTT OVERSEAS, INC., olders) /S/ ROBERT E. STUMPF Robert E. Stumpf |
| J. RAY MC MCDERM (as Shareho By: Name: Title: OFFSHORE P | CDERMOTT INC. OTT OVERSEAS, INC., olders) /S/ ROBERT E. STUMPF Robert E. Stumpf Assistant Secretary PIPELINES SDN. BHD. EE PIPELINES SDN. BLD. |

Robert E. Stumpf Assistant Secretary

FLOATEC, LLC

Name: Title:

| 1 | IN | ľ | T. | 1 | o | N | ۸ | 7 | Гī | • | ١ | N | Ι. | ۸ | T | 1 | 7 | r | c | c | 1 | 71 | LS | 2 | T | ח | rı | n | |
|---|-----|---|----|---|---|----|---|---|----|---|---|---|----|----|---|-----|---|----|---|---|---|------|-----|----|---|---|----|---|--|
| П | יוו | N | ľ | | ≺ | IN | Α | ч | | • | , | N | 17 | а. | | . 1 | v | ۲. | | | н | ١, ١ | 1.0 | ٠. | | | | , | |

| By: | /s/ Steven W. Roll |
|--------|--------------------|
| Name: | Steven W. Roll |
| Title: | Director |

 $Signature\ Page\ to\ Assumption\ and\ Loss\ Allocation\ Agreement$

| DТ | RATA | WAHAN | A INDONESIA |
|-----|------|--------|-------------|
| PI. | BAJA | WAHANA | a indunksia |

| By: | /s/ Scott Cummins |
|--------|--------------------|
| Name: | Scott Cummins |
| Title: | President Director |

Signature Page to Assumption and Loss Allocation Agreement

| SINCA | DODE | HUANGD | AO DTE | ITD |
|-------|------|--------|--------|-----|
| | | | | |

 By:
 /S/
 ROCKNE L. MOSELEY

 Name:
 Rockne L. Moseley

 Title:
 Director

Signature Page to Assumption and Loss Allocation Agreement

EXHIBIT I – B&W ENTITIES

| See Schedule 1.1(a) and Schedule 1.1(c) attached hereto. No B&W Entity listed on Schedule 1.1(a) shall be deemed to be an "Insured", a "Named Insured" or otherwise |
|---|
| be deemed to be insured under any Existing Policy or Existing Insurance Agreement solely by virtue of being listed on such Schedule 1.1(a). |

Schedule 1.1(a) B&W FORMER

| Reference ID | Name |
|--------------|---|
| 333 | Ahahsain Hudson Heat Transfer Co. Ltd. |
| 398 | Advanced Refractory Technologies, Inc. |
| | A.M. Lockett & Co., Limited |
| | Amcermet Corporation |
| 924 | ASEA Babcock |
| 235 | Ash Acquisition Company |
| 326 | B & W Clarion, Inc. |
| 574 | B&W Ebensburg Pa., Inc. |
| 460 | B&W Energy Investments, Inc. |
| 383 | B&W Fort Worth Power, Inc. |
| 950 | B&W Fuel Company |
| 535 | B&W Fuel, Inc. |
| 922 | B&W Mexicana, S.A. de C.V. |
| 9991 | B&W North Branch G.P., Inc. |
| 9990 | B&W North Branch L.P., Inc. |
| 537 | B&W Nuclear Service Company |
| 960 | B&W Nuclear Service Company |
| 536 | B&W Nuclear, Inc. |
| 586 | B&W Saba, Inc. |
| 591 | B&W Service Company |
| 579 | B&W SOFC G.P., Inc. |
| 578 | B&W SOFC L.P., Inc. |
| 381 | B&W Special Projects, Inc. |
| 569 | B&W Triso Corporation |
| | B&W Tubular Products Limited |
| 573 | B&W/OHM Weldon Spring, Inc. |
| 212 | Babcock & Wilcox Asia Investment Co., Inc. |
| 115 | Babcock & Wilcox Asia Limited |
| 533 | Babcock & Wilcox Canada Leasing Ltd. |
| 503 | Babcock & Wilcox Canada Ltd. |
| 215 | Babcock & Wilcox China Investment Co., Inc. |
| 594 | Babcock & Wilcox do Brasil Limitada |
| 528 | Babcock & Wilcox do Brasil Participacoes Limitada |
| | |

1

6/2/2010

| 206 | Babcock & Wilcox Egypt SAE |
|------|--|
| 169 | Babcock & Wilcox Fibras Ceramicas Limitada |
| 519 | Babcock & Wilcox Finance, Inc. |
| 557 | Babcock & Wilcox Foreign Sales Corporation |
| 175 | Babcock & Wilcox Gama Kazan Teknolojisi A.S. |
| 552 | Babcock & Wilcox General Contracting Company |
| 565 | Babcock & Wilcox Government Services Company |
| 395 | Babcock & Wilcox HRSG Company |
| | Babcock & Wilcox Industries, Ltd. |
| 531 | Babcock & Wilcox International Sales Corporation |
| 549 | Babcock & Wilcox International Sales Corporation |
| 342 | Babcock & Wilcox Investment Company |
| 305 | Babcock & Wilcox Jonesboro Power, Inc. |
| 2007 | Babcock & Wilcox Nevada, LLC |
| | Babcock & Wilcox Refractories Limited |
| 323 | Babcock & Wilcox Salt City Power, Inc. |
| 543 | Babcock & Wilcox Services, Inc. |
| 322 | Babcock & Wilcox Tracy Power, Inc. |
| 314 | Babcock & Wilcox Victorville Power, Inc. |
| 727 | Babcock & Wilcox Volund France SAS |
| 315 | Babcock PFBC, Inc. |
| 559 | Babcock Southwest Construction Corporation |
| 945 | Babcock-Brown Boveri Reaktor GmbH |
| 936 | Babcock-Ultrapower Jonesboro |
| 937 | Babcock-Ultrapower West Enfield |
| 951 | Bailey Beijing Controls Co., Ltd. |
| 516 | Bailey Controls Australia Pty. Limited |
| 517 | Bailey Controls International Sales & Services Company, Inc. |
| 954 | Bailey Controls Jordan for Process Controls Services, Ltd. |
| 563 | Bailey Controls Sales & Service (Australia) Pty. Limited |
| 564 | Bailey Controls Sales & Services Canada Inc. |
| 561 | Bailey do Brasil Instrumentos Industriais Limitada |
| 114 | Bailey International, Inc. |
| 923 | Bailey Japan Company Limited |
| 542 | Bailey Meter and Controls Company |
| | Bailey Meter Company |
| 562 | Bailey Meter Co. (Japan) Ltd. |
| | |

| | Bailey Meter Company Limited |
|------|--|
| | Bailey Meter GmbH |
| 646 | Brick Insurance Company, Ltd. |
| 590 | BWXT Protec, Inc. |
| | C.C. Moore & Company Engineers |
| 511 | Ceramatec G.P., Inc. |
| 510 | Ceramatec SOFC, Inc. |
| 329 | Clarion Energy, Inc. |
| 328 | Clarion Power Company |
| 321 | Conam Nuclear, Inc. |
| | Control Components France |
| 514 | Control Components Italy S.R.L. |
| | Control Components, Inc. (California) |
| | Control Components, Inc. (Delaware) |
| 948 | Control Components Japan |
| 1914 | CTR Solutions, LLC |
| 545 | Detroit Broach & Machine Corporation |
| 551 | Diamond Blower Company Limited |
| | Diamond Canapower Ltd. |
| 518 | Diamond Power Importação e Exportação Ltda. |
| 144 | Diamond Power Korea Inc. |
| 526 | Diamond Power Specialty (Japan) Ltd. |
| 558 | Diamond Power Specialty (Proprietary) Limited |
| 546 | Diamond Power Specialty Corporation (Delaware) |
| | Diamond Power Speciality Corporation (Ohio) |
| 529 | Diamond Power Specialty GmbH |
| | Diescher Tube Mills, Inc. |
| 332 | Ebensburg Energy, Inc. |
| 397 | Ejendomsaktieselskabet Falkevej2 |
| 961 | Enserch Environmental Management Company, Inc. |
| 968 | EPC Business Trust |
| 919 | Especialidades Termomecanicas, S.A. de C.V. |
| 550 | Ferry-Diamond Engineering Company Limited |
| 928 | Fibras Ceramicas C.A. |
| 509 | Fibras Ceramicas, Inc. |
| 547 | Globe Steel Tubes Corporation |
| | Greer Land Co. |
| | |

6/2/2010

| | Holmes Insulations Limited |
|------|--|
| 124 | Hudson Heat Transfer International, Inc. |
| 1955 | Hudson HEI Pty. Ltd. |
| 914 | Hudson Northern Industries Inc. |
| 125 | Hudson Products Aktiebolag |
| 567 | Hudson Products Corporation (1) |
| 902 | Hudson Products de Mexico, S.A. de C.V. |
| 1907 | Integran Technologies Inc. |
| 461 | International Disarmament Corporation |
| 941 | Isolite Babcock Refractories Company, Ltd. |
| 927 | Isolite Eastern Union Refractories Co., Ltd. |
| 920 | KBW Gasification Systems, Inc. |
| 512 | LT Produkter i Skutskar AB |
| 938 | Maine Power Services |
| 345 | McDermott Heat Transfer Company |
| 344 | McDermott Productos Industriales de Mexico, S.A. de C.V. |
| 946 | Medidores Bailey, S.A. de C.V. |
| 942 | Morganite Ceramic Fibres Limited |
| 943 | Morganite Ceramic Fibres Pty. Limited |
| 944 | Morganite Ceramic Fibres S. A. |
| | National Drill & Manufacturing Co. |
| 544 | National Ecology (Alabama) Incorporated |
| 575 | National Ecology (Utah) Incorporated |
| 976 | Nooter/Eriksen - Babcock & Wilcox, L.L.C. |
| 933 | North American CWF Partnership |
| 9989 | North Branch Power Company L.P. |
| 971 | North County Operations Associates |
| | Nuclear Materials and Equipment Corporation |
| 958 | Olin Pantex Inc. |
| 1153 | P. T. Heat Exchangers Indonesia |
| 934 | Palm Beach Energy Associates |
| | Piedmont Tool Machine Company |
| 576 | Power Computing Company |
| 581 | Power Systems Sunnyside Operations GP, Inc. |
| 583 | Power Systems Sunnyside Operations LP, Inc. |
| | |

(1) a Delaware Corporation

6/2/2010

| 905 | PowerSafety International, Inc. |
|------|--------------------------------------|
| 508 | Productos de Caolin, Inc. |
| 577 | PSO Caribbean, Inc. |
| 1980 | Savannah River Alliance LLC |
| 2008 | Savannah River Tactical Services LLC |
| 984 | SOFCo L.P. |
| 932 | South Point CWF |
| 556 | Sunland Construction Co., Inc. |
| 988 | Sunnyside Cogeneration Associates |
| 582 | Sunnyside II, Inc. |
| 992 | Sunnyside II, L.P |
| 303 | Sunnyside III, Inc. |
| 993 | Sunnyside Operations Associates L.P. |
| 571 | Termobloc Industria E Comercio Ltda. |
| 953 | Thermax Babcock & Wilcox Limited |
| 502 | TLT-Babcock, Inc. |
| 570 | Triso |
| 1152 | W.E. Smith Hudson Pty. Ltd. |
| 548 | W. F. and John Barnes Company |
| | |

Schedule 1.1(c) B&W CURRENT

| Reference ID | Name | |
|--------------|--|---|
| 553 | Americon Equipment Services, Inc. | |
| 554 | Americon, Inc. | |
| 732 | Applied Synergistics, Inc. | |
| 127 | B&W de Panama, Inc. | |
| 532 | Babcock & Wilcox Canada Ltd. | |
| 1570 | Babcock & Wilcox China Holdings, Inc. | |
| 555 | Babcock & Wilcox Construction Co., Inc. | |
| 2011 | Babcock & Wilcox de Monterrey, S.A. de C.V. | |
| 1571 | Babcock & Wilcox Denmark Holdings, Inc. | |
| 327 | Babcock & Wilcox Ebensburg Power, Inc. | |
| 302 | Babcock & Wilcox Equity Investments, Inc. | |
| 1965 | Babcock & Wilcox Holdings, Inc. | |
| 2028 | Babcock & Wilcox India Holdings, Inc. | |
| 598 | Babcock & Wilcox India Private Limited | |
| 126 | Babcock & Wilcox International Investments Co., Inc. | |
| 530 | Babcock & Wilcox International Sales and Service Corporation | |
| 541 | Babcock & Wilcox International, Inc. | |
| 380 | Babcock & Wilcox Investment Company | |
| 2010 | Babcock & Wilcox Modular Nuclear Energy LLC | |
| 1974 | Babcock & Wilcox Nuclear Operations Group, Inc. | |
| 1967 | Babcock & Wilcox Nuclear Energy, Inc. | (formerly Babcock & Wilcox Nuclear Power Generation Group, Inc. |
| 2018 | Babcock & Wilcox Nuclear Services (U.K.) Limited | |
| 500 | Babcock & Wilcox Power Generation Group, Inc. | |
| 1961 | Babcock & Wilcox Technical Services (U.K.) Limited | |
| 1970 | Babcock & Wilcox Technical Services Clinch River, LLC | |
| 572 | Babcock & Wilcox Technical Services Group, Inc. | (formerly BWXT Services, Inc.) |
| 587 | Babcock & Wilcox Technical Services Savannah River Company | (formerly BWXT Savannah River Company) |
| 599 | Babcock & Wilcox Volund A/S | (formerly Babcock & Wilcox Volund ApS) |
| 2000 | BCE Parts Ltd. | |
| 596 | BWX Technologies, Inc. | |
| 580 | BWXT Federal Services, Inc. | |
| 589 | BWXT Hanford Company | |
| 382 | BWXT of Idaho, Inc. | |
| | | |

1

Draft 6/2/2010

501

2006

BWXT of Ohio, Inc. BWXT Washington, Inc. 592 1576 Creole Insurance Company, Ltd. 189 1988 Delta Power Services, LLC 766 Diamond Operating Co., Inc. Diamond Power Australia Holdings, Inc.
Diamond Power Central & Eastern Europe s.r.o. 1572 1984 Diamond Power China Holdings, Inc. 1573 521 Diamond Power do Brasil Limitada 1574 Diamond Power Equity Investments, Inc. 525 Diamond Power Finland OY Diamond Power Germany GmbH 504 597 Diamond Power International, Inc. 949 Diamond Power Machine (Hubei) Co., Inc. 1908 Diamond Power Services S.E.A. Ltd. Diamond Power Specialty (Proprietary) Limited Diamond Power Specialty Limited 522 523 Diamond Power Sweden AB 524 1990 DPS Berkeley, LLC 1997 DPS Cadillac, LLC DPS Florida, LLC 1995 1993 DPS Gregory, LLC 1996 DPS Lowell Cogen, LLC DPS Mecklenburg, LLC DPS Michigan, LLC 1992 1991 1994 DPS Mojave, LLC 1998 DPS Sabine, LLC 278 Gumbo Insurance Company, Ltd. 2003 Intech International Inc. 2002 Intech, Inc. 2001 Ivey-Cooper Services, L.L.C. 1968 Marine Mechanical Corporation 540 National Ecology Company NFS Holdings, Inc.
NOG-Erwin Holdings, Inc. 2005 2004

North County Recycling, Inc.

Nuclear Fuel Services, Inc.

(formerly Diamond Power - Sturm GmbH)

Draft 6/2/2010

| 1989 | O&M Holding Company |
|------|---|
| 707 | P. T. Babcock & Wilcox Asia |
| 534 | Palm Beach Resource Recovery Corporation |
| 560 | Power Systems Operations, Inc. |
| 568 | Revloc Reclamation Service, Inc. |
| 2013 | Servicios de Fabricacion de Valle Soleado, S.A. de C.V. |
| 2012 | Servicios Profesionales de Valle Soleado, S.A. de C.V. |
| 767 | SOFCo - EFS Holdings LLC |
| 2029 | The Babcock & Wilcox Company |

EXHIBIT II – EXISTING INSURANCE AGREEMENTS See Attached.

| Residual Market | McDermott International, Inc. | CICNIAI | | | | Agreement | | |
|---|--|---|--|---|--------------------------------------|-----------|-----------------------------------|----|
| | | CIGNA Insurance Company of Texas | | | · · | 1992 | | tf |
| Assessments Deductible | | Company of Texas | | | | | | |
| WC Programs High Deductible Agreement | McDermott Incorporated | Insurance Company of North America | | | | 4/1/1981 | | tf |
| High Deductible Agreement | McDermott International | Insurance Company of North America | | | | 4/1/1985 | | tf |
| High Deductible Agreement | McDermott International | Insurance Company of North America | | | | 4/1/1986 | | tf |
| High Deductible Agreement | McDermott International, Inc. | Insurance Company of North America | INA of Texas | | | 4/1/1987 | Auto & GL only | tf |
| High Deductible Agreement | | Insurance Company of North America | | | | 4/1/1988 | | tf |
| High Deductible Agreement | McDermott International, Inc. | Insurance Company of North America | | | | 4/1/1989 | | tf |
| High Deductible Agreement | McDermott International, | Insurance Company of | | | | 4/1/1990 | | tf |
| Workers' Compensation | Inc. The Babcock & Wilcox Company | North America Insurance Company of North America | | | | 4/1/1991 | | vm |
| Agreement High Deductible Agreement | McDermott International, | Insurance Company of | | | | 4/1/1991 | | tf |
| Workers Compensation Deductible Funding | Inc. Babcock & Wilcox Company | North America CIGNA Insurance Company of Texas | Pacific Employers Insurance Company | | | 4/1/1992 | | ср |
| Agreement Agreement for Workers | Cigna Insurance Companyof Texas | Babcock & Wilson | Pacific Employers | | | 4/1/1992 | | сp |
| | McDermott International, | Company CIGNA Insurance | Insurance Company | | | 4/1/1992 | | tf |
| Deductible Funding Agreement High Deductible Agreement | Inc. McDermott International, | Company of Texas Insurance Company of | | | | 4/1/1992 | XS, Auto & GL | tf |
| Casualty Insurance Program | Inc. The Babcock & Wilcox | North America Pacific Employers | | | | 4/1/1993 | | ср |
| Agreement | Company Pacific Employers | Insurance Company Creole Insurance | | | | 4/1/1993 | | |
| Compensation Residual market Assessments | Insurance Company | Company, Ltd | | | | 4/1/1993 | | cp |
| Captive Program Casualty Insurance Program | McDermott International, | Pacific Employers | | | | 4/1/1993 | | tf |
| Agreement Collateral Agreement | Inc. McDermott International, | Insurance Company Insurance Company of | | | | 4/1/1993 | | tf |
| Casualty Insurance Program | Inc. | North America Pacific Employers | | | | 4/1/1994 | | |
| Agreement | Company | Insurance Company | | | | | | ср |
| Agreement for Workers compensation Redidual Market Assessments Captive Program | Pacific Employers Insurance Company | Creole Insurance Company, Ltd | | | | 4/1/1994 | | ср |
| High Deductible Agreement | , | Insurance Company of | | | | 4/1/1994 | | tf |
| Casualty Insurance Program | Inc. The Babcock & Wilcox | North America CIGNA Insurance | Bankers Standard | | | 4/1/1995 | | ср |
| Agreement Agreement Regarding | Company McDermott International, | Company Creole Insurance | Insurance Company Bankers Standard | | | 4/1/1995 | | ср |
| Additional Premiums | Inc. | Company, Limited | Insurance Company | | | | | |
| Casualty Insurance Program Agreement | Company | CIGNA Insurance Company | Bankers Standard Insurance Company | | | 4/1/1995 | | ср |
| High Deductible Agreement | McDermott International, Inc. | Indemnity Insurance Company of North America | | | | 4/1/1995 | | tf |
| Addendum I Casualty Insurance Program Agreement | The Babcock & Wilcox Company | CIGNA Insurance Company | Bankers Standard Insurance Company | | | 4/1/1996 | | cp |
| Casualty Insurance Program Agreement | McDermott International, Inc. | Bankers Standard Insurance Company | | | | 4/1/1996 | | tf |
| High Deductible Agreement | | | | | | 4/1/1996 | | tf |
| Reinsurance Agreement | Honore Insurance Company Limited | CIGNA Insurance | | | | 4/1/1997 | | cp |
| Reinsurance Agreement | Honore Insurance | Company CIGNA Insurance | | | | 4/1/1997 | | cp |
| Addendum III Casualty Insurance Program | Company Limited The Babcock & Wilcox Company | Company of Canada CIGNA Insurance Company | Bankers Standard Insurance Company | Pacific Employers Insurance Company | | 4/1/1997 | | ср |
| Agreement High Deductible Agreement | McDermott International, Inc. | Indemnity Insurance Company of North | | | | 4/1/1997 | | tf |
| Casualty Insurance Program Agreement | McDermott International, Inc. | America Bankers Standard Insurance Company | | | | 4/1/1997 | Addendum II to the 1996 agreement | tf |
| Cash Flow Deductible Workers' Compensation | McDermott International, Inc. | | Bankers Standard Insurance Company | | | 4/1/1997 | agreement | tf |
| Program Cash Flow High Deductible Program | McDermott International, Inc. | Indemnity Insurance Company of North | | | | 4/1/1997 | Auto & GL only | tf |
| Compensaiton & | McDermott International, Inc. | America CIGNA Insurance Company | | | | 4/1/1997 | | tf |
| Employers' Liability Final Summary of Proposal | Aon Risk Services of Texas, Inc | Pacific Employers Ins/Bankers Standard Ins. | | | | 4/1/1997 | Canadian Proposal included | dc |
| | Honore Insuance | CIGNA Insurance | | | | 4/1/1997 | | dc |
| | Company Limited Honore Insuance | Company of Canada CIGNA Insurance | | | | 4/1/1997 | | dc |
| Reinsurance Agreement Agreement Regarding Return Premiums | Company Limited Babcock & Wilcox Company | Company of Canada Creole Insurance Company, Limited | CIGNA Insurance Company | Indemnity Insurance Company of North | CIGNA Insurance Company of Canada | 7/28/1997 | | cp |
| Agreement Regarding | The Babcock & Wilcox | Creole Insurance | CIGNA Insurance | American Indemnity Insurance | CIGNA Insuarnce | 7/28/1997 | | dc |
| Return Premiums | Company The Babcock & Wilcox Company | Company, Limited Pacific Employers Insurance Company | Company | Company of NA | Company fo Canada | 4/1/1998 | | ср |

| Casualty Insurance Program | | CIGNA Insurance | Pacific Employers | | 4/1/1998 | | cp |
|--|--|--|---|----------------------|----------|--|----|
| Agreement Addendum I Reinsurance | Company Honore Insurance | Company of Texas CIGNA Insurance | Insurance Company | | 4/1/1998 | | ср |
| Agreement Casualty Insurance Program | | Company of Canada CIGNA Insurance | Pacific Employers | | 4/1/1998 | | ср |
| Agreement Casualty Insurance Program | Company McDermott International, | Company of Texas Pacific Employers | Insurance Company CIGNA Insurance | | 4/1/1998 | | tf |
| Agreement High Deductible Agreement | Inc. McDermott International | Insurance Company Pacific Employers | Company of Texas | | 4/1/1998 | | tf |
| | Inc. | Insurance Company | | | | | |
| Cash Flow Deductible Workers' Compensation Program | AON Risk Services of Texas, Inc. | CIGNA Insurance Company of Texas | Pacific Employers Insurance Company | | 4/1/1998 | | tf |
| Cash Flow High Deductible Program | AON Risk Services of Texas, Inc. | Insurance Company of North America | Pacific Employers Insurance Company | | 4/1/1998 | Auto & GL only | tf |
| Specific Excess Workers' Compensaiton & Employers' Liability | AON Risk Services of Texas, Inc. | CIGNA Insurance Company | | | 4/1/1998 | | tf |
| Assignment of Dividends | McDermott International, | Pacific Employers | | | 4/1/1998 | | tf |
| TPA Agreement | Inc. The Babcock & Wilcox | Insurance Company Pacific Employers | | | 4/1/1998 | | dc |
| Casualty Insurance Program | | Insurance Co. CIGNA Insurance | Pacific Employers | | 4/1/1998 | | dc |
| Agreement Casualty Insurance Program | Company The Babcock & Wilcox | Company of Texas CIGNA Insurance | Insurance Company Pacific Employers | | 4/1/1998 | | dc |
| Agreement Final Summary of Proposal | Company Aon Risk Services of Texas, Inc | Company of Texas Pacific Employers Insurance Co. | Insurance Company CIGNA Insurance Company | | 4/1/1998 | Included: Cash Flow Ded. WC Program and Guaranteed Cost WC | dc |
| Agreement | The Babcock & Wilcox | Pacific Employers | | | 4/1/1999 | Program | ср |
| Addendum I Casualty | Company The Babcock & Wilcox | Insurance Company Pacific Employers | ACE Insurance Company | | | This Addendum is from | ср |
| Insurance Program Agreement | Company | Insurance Company | of Texas (formerly CIGNA Insurance company of Texas | | 4/1/1999 | 04/01/99 to 04/01/00 | Ср |
| Addendum II Casualty Insurance Program Agreement | The Babcock & Wilcox Company | Pacific Employers Insurance Company | CIGNA Insurance Company of Texas | | 4/1/1999 | This Addendum is from 04/01/99 to 01/01/00 | cp |
| Addendum II to Reinsurance Agreement | Cigna Insurance Companyof Canada | Honore Insurance Company Limited | | | 4/1/1999 | This Addendum is from 04/01/99 to 04/01/00 | cp |
| Addendum III to Reinsurance Agreement | ACE INA Insurance (formely CIGNA Insurance Company | Honore Insurance Company Limited | | | 4/1/1999 | This Addendum is from 04/01/99 to 01/01/00 | ср |
| Multi-Line Deductible Cash | of Canada AON Risk Services of | Pacific Employers | CIGNA Insurance | Insurance Company of | 4/1/1999 | | tf |
| Flow Insurance High Deductible Agreement | Texas, Inc. McDermott International, Inc. | Insurance Company Pacific Employers Insurance Company | Company | North America | 4/1/1999 | | tf |
| Casualty Insurance Program | McDermott International, | Pacific Employers | CIGNA Insurance | | 4/1/1999 | | tf |
| Agreement Casualty Insurance Program Agreement - Addendum | Inc. McDermott International, Inc. | Insurance Company Pacific Employers Insurance Company | Company of Texas ACE Insurance Company of Texas (formerly | | 4/1/1999 | changing expiration to 01/01/00 | tf |
| II Assignment of Dividends | McDermott International, | Pacific Employers | CIGNA) Hudson Companies | | 4/1/1999 | | tf |
| Combined Miult-Line | Inc. Pacific Employers | Insurance Company Babcock & Wilson | | | 1/1/2000 | | ср |
| Program Agreement Addendum IV Reinsurance | Insurance Company ACE INA Insurance | Company Honore Insurance | | | 1/1/2000 | | ср |
| Agreement Combined Multi-Line | McDermott International, | Company Limited Pacific Employers | | | 1/1/2000 | Addendum I to the 2000 | tf |
| Program Agreement Assignment of Dividends | Inc. McDermott International, | Insurance Company Pacific Employers | Hudson Companies | | 1/1/2000 | term | tf |
| · · | Inc. | Insurance Company | Tradson Companies | | | | |
| Addendum II Combined Multi-line Program Agreement | Pacific Employers Insurance Company | Babcock & Wilson Company | | | 1/1/2001 | | cp |
| Addendum V to Resinsurance Agreement | ACE INA Insurance | Honore Insurance Company Limited | | | 1/1/2001 | | ср |
| Combined Multi-Line Program Agreement | McDermott International, Inc. | Pacific Employers Insurance Company | | | 1/1/2001 | | tf |
| Addendum IV to Combined Multi-Line Program Agreement | Pacific Employers Insurance Company | Babcock & Wilson Company | | | 1/1/2002 | | ср |

| Addendum VI to Reinsurance | ACE INA Insurance | Pirogue Insurance Company, Ltd | | | 1/1/2002 | |
|--|---|--|---|---|----------------------|---------------|
| Agreement Addendum V to Combined Multi- | Babcock & Wilcox Company | ACE American Insurance Company | Pacific Employers Insurance Company | | 1/1/2003 | |
| Line Program Agreement Addendum VII to Reinsurance Agreement | ACE INA Insurance | Pirogue Insurance Company, Ltd | | | 1/1/2003 | |
| Addendum VI to Combined multi- Line Program | Babcock & Wilcox Company | ACE American Insurance Company | Pacific Employers Insurance Company | ACE INA Insurance Company | 1/1/2004 | |
| Addendum VII to Combined Multi-Line Program Agreement | Babcock & Wilcox Company | ACE American Insurance Company | ACE INA Insurance Company | | 1/1/2005 | |
| Addendum II to Combined Multi- Line Program Agreement | McDermott International, Inc. | Pacific Employers Insurance Company | | | 1/1/2002 | cp ! vm |
| Addendum IV to Combined Multi- Line Program Agreement | McDermott International, Inc. | ACE American Insurance Company | Pacific Employers Insurance Company | | 1/1/2003 | |
| Addendum V to Combined Multi- Line Program Agreement | McDermott International, Inc. | ACE American Insurance Company | Pacific Employers Insurance Company | | 1/1/2004 | |
| Addendum VI to Combined Multi- Line Program Agreement | McDermott International, Inc. | ACE American Insurance Company | | | 1/1/2005 | vm |
| Casualty Program Binder Casualty Program Binder | The Babcock & Wilcox Company The Babcock & Wilcox Company | ACE American Insurance Company ACE American Insurance Company | ACE INA Insurance ACE INA Insurance | | 1/1/2006 1/1/2007 | |
| Casualty Program Binder | The Babcock & Wilcox Company | ACE American Insurance Company | ACE INA Insurance | Indemnity Insurance Company of North America | 1/1/2008 | |
| Casualty Program Proposal | The Babcock & Wilcox Company | ACE American Insurance Company | ACE INA Insurance | Indemnity Insurance Company of North America | 1/1/2009 | |
| Casualty Program Proposal | The Babcock & Wilcox Company | ACE American Insurance Company | ACE INA Insurance | Indemnity Insurance Company of North America | 1/1/2010 | |
| Casualty Program Binder Casualty Program Binder | McDermott International, Inc. McDermott International, Inc. | ACE American Insurance Company ACE American Insurance Company | | TVOTAL TAILCTICA | 1/1/2006 1/1/2007 | vm |
| Casualty Program Binder | McDermott International, Inc. | ACE American Insurance Company | Indemnity Insurance Company of North America | | 1/1/2008 | |
| Casualty Program Proposal | McDermott International, Inc. | ACE American Insurance Company | Indemnity Insurance Company of North America | | 1/1/2009 | |
| Casualty Program Proposal | McDermott International, Inc. | ACE American Insurance Company | Indemnity Insurance Company of North America | | 1/1/2010 | |

EXHIBIT III – EXISTING POLICIES

See Attached.

Exhibit 1 - Policies and Named Insured's

| | | | | Type of Policy | |
|----------------|------------------------|---------------|--|----------------|-------------------|
| Effective Date | Expiration Date | Policy Number | Named Insured | (LOB) | Issuing Company |
| 4/1/1977 | 4/1/1978 | XWC 6858 | J. Ray McDermott & Co., Inc. | XWC | INA |
| 4/1/1980 | 4/1/1981 | XCP 143599 | J. Ray McDermott & Company - See attachment 205 | XGL | INA |
| 4/1/1982 | 4/1/1983 | GLP 829719 | Babcock & Wilcox Company - See attachment 164 | GL | INA |
| 4/1/1982 | 4/1/1985 | XCP 014414 | Intermountain Power Agency Department of Water & Power City of Los Angeles | XGL | INA |
| 4/1/1983 | 4/1/1984 | RSC C20704202 | Babcock & Wilcox Company - See attachment 168 | WC | INA - TX |
| 4/1/1983 | 4/1/1984 | RSC C20704226 | Babcock & Wilcox Company - See attachment 178 | WC | Pacific Employers |
| 4/1/1983 | 4/1/1984 | SCA 5914 | The Babcock & Wilcox Company - See attachment 176 | Auto | INA - TX |
| 4/1/1984 | 4/1/1985 | RSC C22890785 | TLT Babcock, Inc., a Joint Venture | WC | Pacific Employers |
| 3/8/1985 | 3/8/1986 | GLP G05111523 | Babcock & Wilcox Company | GL | INA |
| 3/8/1985 | 9/12/1985 | NWC X1939718A | MMC - McDermott - See attachment 137 | WC | INA |
| 4/1/1985 | 4/1/1986 | RSC C19396812 | Babcock & Wilcox Company - See attachment 274 | WC | Cigna |
| 4/1/1985 | 4/1/1986 | XCP G05111730 | Commonwealth Edison Company | XGL | INA |
| 4/1/1985 | 4/1/1986 | XCP G0511326A | Intermountain Power Agency Department of Water & Power City of Los Angeles | GL | INA |
| 4/21/1985 | 5/2/1985 | GLP G05111766 | Chevron USA | GL | INA |
| 6/30/1985 | 7/10/1985 | GLP G05112588 | Chevron USA | GL | INA |
| 4/1/1986 | 4/1/1987 | CPO G03518656 | Babcock & Wilcox Company | GL | INA |
| 4/1/1986 | 4/1/1987 | CPO G05114329 | Babcock & Wilcox Company - See attachment 187 | GL | INA |

Page 1 of 25

| 4/1/1986 | 4/1/1987 CPO G05114329 | Babcock & Wilcox Company - See attachment 203 | GL | INA |
|----------|------------------------|---|------|---------------|
| 4/1/1986 | 4/1/1990 XCE G0519622A | McDermott International, Inc. | XGL | Cigna |
| 4/1/1986 | 4/1/1987 XCP G05114184 | Arco Oil & Gas Company | XGL | INA |
| 4/1/1987 | 4/1/1988 CPO G05115401 | Babcock & Wilcox Company - See attachment 50 | XGL | INA |
| 4/1/1987 | 4/1/1988 CPO G05115620 | Babcock & Wilcox Company - See attachment 51 | GL | INA ILL |
| 4/1/1987 | 4/1/1988 CPO G05115632 | Babcock & Wilcox Company - See attachment 52 | GL | INA Ohio |
| 4/1/1987 | 4/1/1988 CPO G05115978 | Babcock & Wilcox Company | GL | INA |
| 4/1/1987 | 4/1/1988 SCA 010999 | The Babcock & Wilcox Company - See attachment 119 | Auto | INA |
| 4/1/1987 | 4/1/1988 XCE G05115735 | McDermott International, Inc. | XGL | Cigna |
| 4/1/1988 | 4/1/1989 HDO G0519393A | McDermott International, Inc. | GL | INA |
| 4/1/1988 | 4/1/1989 PLI 519745 | McDermott International, Inc See attachment 209 | GL | Alaska |
| 4/1/1988 | 4/1/1989 XSL G05194167 | The Babcock & Wilcox Company - See attachment 49 | XGL | INA |
| 4/1/1988 | 4/1/1989 XSL G05194179 | The Babcock & Wilcox Company - See attachment 48 | XGL | Cigna of Ohio |
| 4/1/1989 | 4/1/1990 CGO G05195792 | Babcock & Wilcox Company - See attachment 79 | GL | INA |
| 4/1/1989 | 4/1/1990 CPO G05195809 | Babcock & Wilcox Company | GL | INA |
| 4/1/1989 | 4/1/1990 CPO G05195809 | Babcock & Wilcox Company - See attachment 94 | GL | INA |
| 4/1/1989 | 4/1/1990 HDC G05196218 | McDermott International, Inc. | GL | INA |
| 4/1/1989 | 4/1/1990 HDO G05196206 | McDermott International, Inc. | GL | INA |
| 4/1/1989 | 4/1/1990 ISA 582543 | McDermott International, Inc See attachment 103 | Auto | INA |

| 4/1/1989 | 4/1/1990 NWC C32916909 | McDermott International, Inc See attachment 152 | WC INA |
|-----------|-------------------------|---|-----------------------|
| 4/1/1989 | 4/1/1990 PLI 519746 | McDermott International, Inc See attachment 106 | GL Alaska |
| 4/1/1989 | 4/1/1990 RSC C22291312 | B & W Fuel Company | WC INA |
| 4/1/1989 | 4/1/1990 RSC C32916879 | Babcock & Wilcox Company - See attachment 131 | WC Pacific Employers |
| 4/1/1989 | 4/1/1990 RSC C32916880 | Babcock & Wilcox Company - See attachment 147 | WC Cigna |
| 4/1/1989 | 4/1/1990 RSC C32916892 | Babcock & Wilcox Company - See attachment 132 | WC Cigna - ILL |
| 4/1/1989 | 4/1/1990 RSC C32916922 | Babcock & Wilcox Company - See attachment 19 | WC Atlantic Employers |
| 4/1/1989 | 4/1/1990 SCA 012151 | The Babcock & Wilcox Company - See attachment 158 | Auto INA |
| 4/1/1989 | 4/1/1990 SCA 012152 | The Babcock & Wilcox Company - See attachment 84 | GL INA |
| 4/1/1989 | 4/1/1990 XSL G05195780 | The Babcock & Wilcox Company - See acchment 35 | XGL INA |
| 4/1/1989 | 4/1/1990 XSL G05195780 | The Babcock & Wilcox Company - See attachment 90 | XGL INA |
| 4/1/1989 | 4/1/1990 XSL G0519619A | McDermott International, Inc See attachment 99 | XGL INA |
| 4/1/1989 | 4/1/1990 XWC 11157 | McDermott International, Inc. | XWC Cigna |
| 9/17/1989 | 9/17/1990 CRL G0519698A | ASEA - Babcock PFBC Joint Venture | GL INA |
| 4/1/1990 | 4/1/1991 CGO G0519751A | Babcock & Wilcox Company - See attachment 76 | GL INA |
| 4/1/1990 | 4/1/1991 CGO G0519751A | Babcock & Wilcox Company - See attachment 83 | GL INA |
| | | - · | |

| 4/1/1990 | 4/1/1991 CPO G05197521 | Babcock & Wilcox Company - See attachment 54 | GL | INA |
|----------|------------------------|--|------|--------------------|
| 4/1/1990 | 4/1/1991 CPO G05197521 | Babcock & Wilcox Company - See attachment 80 | GL | INA |
| 4/1/1990 | 4/1/1991 CPO G05197521 | Babcock & Wilcox Company - See attachment 80 | GL | INA |
| 4/1/1990 | 4/1/1991 HDC G05197557 | McDermott International, Inc. | GL | INA |
| 4/1/1990 | 4/1/1991 HDO G05197545 | McDermott International, Inc. | GL | INA |
| 4/1/1990 | 4/1/1991 ISA 582563 | McDermott International, Inc See attachment 111 | Auto | INA |
| 4/1/1990 | 4/1/1991 NWC C22292341 | McDermott International, Inc See attachment 136 | WC | INA |
| 4/1/1990 | 4/1/1991 PLI 321018 | McDermott International, Inc See attachment 107 | GL | Alaska |
| 4/1/1990 | 4/1/1991 RSC C22292286 | Babcock & Wilcox Company - See attachment 165 | WC | Pacific Employers |
| 4/1/1990 | 4/1/1991 RSC C22292304 | Babcock & Wilcox Company - See attachment 179 | WC | Cigna - ILL |
| 4/1/1990 | 4/1/1991 RSC C22292316 | Babcock & Wilcox Company - See attachment 188 | WC | Atlantic Employers |
| 4/1/1990 | 4/1/1991 RSC C22292328 | B & W Fuel Company | WC | INA |
| 4/1/1990 | 4/1/1991 RSC C22292730 | Babcock & Wilcox Company | WC | INA |
| 4/1/1990 | 4/1/1991 RSC C2229362A | Babcock & Wilcox Company - See attachment 85 | WC | California Union |
| 4/1/1990 | 4/1/1991 SCA 012197 | The Babcock & Wilcox Company - See attachment 97 | Auto | INA |
| 4/1/1990 | 4/1/1991 SCA 012198 | The Babcock & Wilcox Company - See attachment 96 | Auto | INA |
| 4/1/1990 | 4/1/1991 XCE G05197569 | McDermott International, Inc See attachment 101 | XGL | Cigna |

| 4/1/1990 | 4/1/1991 XSL G05197508 | The Babcock & Wilcox Company - See attachment 92 | XGL | INA |
|-----------|-------------------------|--|------|--------------------|
| 4/1/1990 | 4/1/1991 XSL G05197533 | McDermott International, Inc See attachment 115 | XGL | INA |
| 4/1/1990 | 4/1/1991 XWC 011187 | McDermott International, Inc See attachment 154 | XWC | Cigna |
| 4/19/1990 | 4/1/1991 RSC C22292778 | McDermott, Incorporated | WC | Cigna -TX |
| 9/12/1990 | 9/12/1991 CPO G13210755 | Babcock & Wilcox Company Power Generation Group - See attachment 185 | GL | INA |
| 4/1/1991 | 4/1/1992 CGO G13211061 | Babcock & Wilcox Company - See attachment 75 | GL | INA |
| 4/1/1991 | 4/1/1992 CGO G13211061 | Babcock & Wilcox Company - See attachment 204 | GL | INA |
| 4/1/1991 | 4/1/1992 CPO G13211073 | Babcock & Wilcox Company - See attachment 39 | GL | INA |
| 4/1/1991 | 4/1/1992 CPO G13211073 | Babcock & Wilcox Company - See attachment 120 | GL | INA |
| 4/1/1991 | 4/1/1992 HDC G13211103 | McDermott International, Inc. | GL | INA |
| 4/1/1991 | 4/1/1992 HDO G13211097 | McDermott International, Inc. | GL | INA |
| 4/1/1991 | 4/1/1992 ISA 582584 | McDermott International, Inc See attachment 126 | Auto | INA |
| 4/1/1991 | 4/1/1992 NWC C22293667 | McDermott International, Inc See attachment 192 | WC | INA |
| 4/1/1991 | 4/1/1992 PLI 321112 | McDermott International, Inc See attachment 102 | GL | Alaska |
| 4/1/1991 | 4/1/1992 RSC C2229354 | Babcock & Wilcox Company - See attachment 133 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 RSC C22293564 | Babcock & Wilcox Company - See attachment 199 | WC | Atlantic Employers |
| 4/1/1991 | 4/1/1992 RSC C22293898 | Babcock & Wilcox Company | WC | INA |

| 4/1/1991 | 4/1/1992 | RSC C22293928 | McDermott, Incorporated | WC | Cigna - TX |
|-----------|-----------|---------------|--|------|-------------------|
| 4/1/1991 | 4/1/1992 | SCA 012272 | Babcock & Wilcox Company - See attachment 182 | Auto | INA |
| 4/1/1991 | 4/1/1992 | WLR C22293552 | Babcock & Wilcox Company - See attachment 181 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 | WLR C22293849 | Babcock & Wilcox Company - See attachment 143 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 | WLR C22293850 | Babcock & Wilcox Company - See attachment 144 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 | WLR C22293862 | Babcock & Wilcox Company - See attachment 145 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 | | Babcock & Wilcox Company - See attachment 141 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 | WLR C22293886 | Babcock & Wilcox Company - See attachment 20 | WC | Pacific Employers |
| 4/1/1991 | 4/1/1992 | XCE G13211048 | McDermott International, Inc See attachment 124 | XGL | Cigna |
| 4/1/1991 | 4/1/1992 | XSL G1321105A | The Babcock & Wilcox Company - See attachment 197 | XGL | INA |
| 4/1/1991 | 4/1/1992 | XSL G13211115 | McDermott International, Inc See attachment 118 | XGL | INA |
| 4/1/1991 | 4/1/1992 | XWC 011230 | McDermott International, Inc See attachment 134 | XWC | Cigna |
| 4/1/1991 | 4/1/1992 | XWC 011231 | Babcock & Wilcox Company - See attachment 180 | XWC | Cigna |
| 7/22/1991 | 7/22/1992 | XCP G13211437 | City of Lakeland, Florida Department of Electric & Water Utilities | XGL | INA |
| 8/24/1991 | 8/24/1992 | OGL G1321153A | PowerSafety International, Inc. | GL | INA |
| | | | | | |

| 9/12/1991 | 9/12/1992 CPO G13211541 | Babcock & Wilcox Company Power Generation Group - See attachment 190 | GL | INA |
|-----------|-------------------------|---|------|-------------------|
| 4/1/1992 | 4/1/1993 CGO G13211887 | Babcock & Wilcox Company - See attachment 41 | GL | INA |
| 4/1/1992 | 4/1/1993 CPO G13211899 | Babcock & Wilcox Company - See attachment 40 | GL | INA |
| 4/1/1992 | 4/1/1993 CPO G13211899 | Babcock & Wilcox Company - See attachment 87 | GL | INA |
| 4/1/1992 | 4/1/1993 HDO G13211930 | McDermott International, Inc. | GL | INA |
| 4/1/1992 | 4/1/1993 ISA 582609 | McDermott International, Inc See attachment 122 | Auto | INA |
| 4/1/1992 | 4/1/1993 NWC C38326296 | McDermott International, Inc. | WC | INA |
| 4/1/1992 | 4/1/1993 PLI 321195 | McDermott International, Inc See attachment 207 | GL | Alaska |
| 4/1/1992 | 4/1/1993 RSC C38326259 | Babcock & Wilcox Company - See attachment 171 | WC | Pacific Employers |
| 4/1/1992 | 4/1/1993 RSC C38326351 | Babcock & Wilcox Company | WC | INA |
| 4/1/1992 | 4/1/1993 WLR C38326260 | Babcock & Wilcox Company - See attachment 186 | WC | Pacific Employers |
| 4/1/1992 | 4/1/1993 WLR C38326338 | Babcock & Wilcox Company - See attachment 174 | WC | Cigna - TX |
| 4/1/1992 | 4/1/1993 WLR C38326363 | McDermott, Incorporated | WC | Cigna |
| 4/1/1992 | 4/1/1993 XCP G13211991 | Naheola Cogeneration Limited Partnership & Industrial Development Board | XGL | INA |
| 4/1/1992 | 4/1/1993 XSL G13211905 | The Babcock & Wilcox Company - See attachment 71 | XGL | INA |
| 4/1/1992 | 4/1/1993 XSL G13211966 | McDermott International, Inc See attachment 117 | XGL | INA |

| 4/1/1992 | 4/1/1993 XWC 011301 | Babcock & Wilcox Company - See attachment 200 | XWC | Cigna |
|----------|------------------------|---|------|-------------------|
| 4/1/1992 | 4/1/1993 XWC 011302 | McDermott International, Inc See attachment 149 | XWC | Cigna |
| 4/1/1993 | 4/1/1994 CGO G13212612 | Babcock & Wilcox Company - See attachment 73 | GL | INA |
| 4/1/1993 | 4/1/1994 CGO G13212612 | Babcock & Wilcox Company - See attachment 191 | GL | INA |
| 4/1/1993 | 4/1/1994 CPO G13212600 | Babcock & Wilcox Company - See attachment 36 | GL | INA |
| 4/1/1993 | 4/1/1994 HDC G13212648 | McDermott International, Inc. | GL | INA |
| 4/1/1993 | 4/1/1994 HDO G1321265A | McDermott International, Inc. | GL | INA |
| 4/1/1993 | 4/1/1994 ISA 711093 | McDermott International, Inc See attachment 98 | Auto | INA |
| 4/1/1993 | 4/1/1994 NWC C39770507 | McDermott International, Inc See attachment 151 | WC | INA |
| 4/1/1993 | 4/1/1994 PLI 321258 | McDermott International, Inc See attachment 208 | GL | INA |
| 4/1/1993 | 4/1/1994 RSC C36158621 | Babcock & Wilcox Company | WC | INA |
| 4/1/1993 | 4/1/1994 RSC C39770489 | Babcock & Wilcox Company - See attachment 160 | WC | Pacific Employers |
| 4/1/1993 | 4/1/1994 SCA 012527 | The Babcock & Wilcox Company - See attachment 184 | Auto | INA |
| 4/1/1993 | 4/1/1994 SCA 012528 | The Babcock & Wilcox Company - See attachment 195 | Auto | INA |
| 4/1/1993 | 4/1/1994 WLR C39770490 | Babcock & Wilcox Company - See attachment 162 | WC | Pacific Employers |
| 4/1/1993 | 4/1/1994 WLR C39770532 | McDermott, Incorporated | WC | Pacific Employers |
| 4/1/1993 | 4/1/1994 XCE G13212570 | McDermott International, Inc See attachment 100 | XGL | Cigna |

| 4/1/1993 | 4/1/1994 XCP G13212673 | Naheola Cogeneration Limited Partnership & Industrial Development Board | GL | INA |
|-----------|-------------------------|---|------|-------|
| 4/1/1993 | 4/1/1994 XSL G13212569 | McDermott International, Inc See attachment 104 | XGL | INA |
| 4/1/1993 | 4/1/1994 XSL G13212594 | The Babcock & Wilcox Company - See attachment 72 | XGL | INA |
| 4/1/1993 | 4/1/1994 XSL G13212594 | Babcock & Wilcox Company - See attachment 166 | XGL | INA |
| 4/1/1993 | 4/1/1994 XWC 011381 | McDermott International, Inc See attachment 18 | XWC | Cigna |
| 8/24/1993 | 8/24/1994 OGL G13212879 | PowerSafety International, Inc. | GL | INA |
| 9/12/1993 | 9/13/1994 CPO G13212909 | Babcock & Wilcox Company Power Generation Group - See attachment 89 | GL | INA |
| 10/1/1993 | 4/1/1994 XSL G13212983 | Hudson Companies - See attachment 183 | XGL | INA |
| 4/1/1994 | 4/1/1995 CGO G13213215 | Babcock & Wilcox Company - See attachment 38 | GL | INA |
| 4/1/1994 | 4/1/1995 CGO G13213215 | Babcock & Wilcox Company - See attachment 159 | GL | INA |
| 4/1/1994 | 4/1/1995 CPO G13213203 | Babcock & Wilcox Company - See attachment 37 | GL | INA |
| 4/1/1994 | 4/1/1995 CPO G13213203 | Babcock & Wilcox Company - See attachment 169 | GL | INA |
| 4/1/1994 | 4/1/1995 HDC G1321315A | McDermott International, Inc. | GL | INA |
| 4/1/1994 | 4/1/1995 HDO G13213161 | McDermott International, Inc. | GL | INA |
| 4/1/1994 | 4/1/1995 ISA 042894 | McDermott International, Inc See attachment 112 | Auto | INA |
| 4/1/1994 | 4/1/1995 NWC C4080951A | McDermott International, Inc See attachment 150 | WC | INA |
| | | | | |

| 4/1/1994 4/1/1994 | | PLI 321317 RSC C40809016 | McDermott International, Inc See attachment 109 | GL WC | INA INA |
|----------------------|-----------|--------------------------------|--|----------|-------------------|
| 4/1/1994 | | RSC C40809016 RSC C40809491 | Babcock & Wilcox Company Babcock & Wilcox Company - See attachment 172 | | Pacific Employers |
| 4/1/1994 | | RSC C40809491 RSC C40809521 | McDermott, Incorporated | WC | Cigna - TX |
| 4/1/1994 | | SCA 012556 | The Babcock & Wilcox Company - See attachment 121 | Auto | INA |
| 4/1/1994 | | WLR C36160159 | Babcock & Wilcox Company - See attachment 129 | WC | |
| 4/1/1994 | | WLR C40809508 | * * | | Cigna - TX |
| | | | Babcock & Wilcox Company - See attachment 22 | | Pacific Employers |
| 4/1/1994 | | XCE G13213185 | McDermott International, Inc See attachment 108 | | C |
| 4/1/1994 | | XSL G13213148 | McDermott International, Inc See attachment 116 | XGL | INA |
| 4/1/1994 | | XSL G13213197 | The Babcock & Wilcox Company - See attachment 44 | | INA |
| 4/1/1994 | | XSL G13213197 | Babcock & Wilcox Company - See attachment 170 | | INA |
| 4/1/1994 | 4/1/1995 | XSL G13213227 | Hudson Companies - See attachment 43 | XGL | INA |
| 4/1/1994 | 4/1/1995 | XSL G13213227 | Hudson Companies - See attachment 86 | XGL | INA |
| 4/1/1994 | 4/1/1995 | XWC 011467 | McDermott International, Inc See attachment 153 | XWC | Cigna |
| 8/24/1994 | 8/24/1995 | OGL G13213410 | PowerSafety International, Inc. | GL | INA |
| 9/12/1994 | 9/12/1995 | CPO G13213380 | Babcock & Wilcox Company Power Generation Group - See attachment 167 | GL | INA |
| | | | | | |

| 4/1/1995 | 4/1/1996 CGO G13213720 | Babcock & Wilcox Company - See attachment 74 | GL INA |
|----------|------------------------|---|---------------------|
| 4/1/1995 | 4/1/1996 CPO G13213719 | Babcock & Wilcox Company | GL IND |
| 4/1/1995 | 4/1/1996 HDC G13213756 | McDermott International, Inc. | GL IND |
| 4/1/1995 | 4/1/1996 HDO G13213744 | McDermott International, Inc. | GL IND |
| 4/1/1995 | 4/1/1996 ISA 042936 | McDermott International, Inc See attachment 125 | Auto IND |
| 4/1/1995 | 4/1/1996 NWC C3616248A | Babcock & Wilcox Construction Co See attachment 173 | WC Cigna |
| 4/1/1995 | 4/1/1996 PLI 321380 | McDermott International, Inc See attachment 127 | GL IND |
| 4/1/1995 | 4/1/1996 RSC C36162028 | Babcock & Wilcox Company - See attachment 189 | WC Cigna |
| 4/1/1995 | 4/1/1996 RSC C36162041 | McDermott International, Inc See attachment 17 | WC Bankers Standard |
| 4/1/1995 | 4/1/1996 RSC C36162053 | Hudson Companies | WC Bankers Standard |
| 4/1/1995 | 4/1/1996 RSC C3616234A | Babcock & Wilcox Company | WC Cigna |
| 4/1/1995 | 4/1/1996 WLR C36161991 | Babcock & Wilcox Company - See attachment 148 | WC Cigna |
| 4/1/1995 | 4/1/1996 WLR C36162004 | Babcock & Wilcox Company - See attachment 194 | WC Bankers Standard |
| 4/1/1995 | 4/1/1996 WLR C36162016 | Babcock & Wilcox Company - See attachment 196 | WC Cigna |
| 4/1/1995 | 4/1/1996 XCE G13213768 | McDermott International, Inc See attachment 114 | XGL IND |
| 4/1/1995 | 4/1/1996 XSL G13213707 | The Babcock & Wilcox Company - See attachment 45 | XGL IND |
| 4/1/1995 | 4/1/1996 XSL G13213707 | The Babcock & Wilcox Company - See attachment 91 | XGL IND |
| 4/1/1995 | 4/1/1996 XSL G13213732 | McDermott International, Inc See attachment 110 | XGL IND |

| 4/1/1995 | 4/1/1996 XSL G13213793 | Hudson Companies - See attachment 88 | XGL IND |
|-----------|-------------------------|--|---------------------|
| 4/1/1995 | 4/1/1996 XWC 011541 | McDermott International, Inc See attachment 138 | XWC Cigna |
| 8/24/1995 | 8/24/1996 CAL H06353149 | Deep Oil Technology, Inc. | Auto IND |
| 8/24/1995 | 8/24/1996 NWC C41224319 | Deep Oil Technology, Inc. | WC Cigna |
| 8/24/1995 | 8/24/1996 OGL G14232096 | Deep Oil Technology, Inc. | GL IND |
| 4/1/1996 | 4/1/1997 CGO G18967431 | Babcock & Wilcox Company - See attachment 77 | GL IND |
| 4/1/1996 | 4/1/1997 CPO G18968599 | Babcock & Wilcox Company - See attachment 46 | GL IND |
| 4/1/1996 | 4/1/1997 CPO G18968599 | Babcock & Wilcox Company - See attachment 82 | GL IND |
| 4/1/1996 | 4/1/1997 HDC G18967157 | McDermott International, Inc. | GL IND |
| 4/1/1996 | 4/1/1997 HDO G18967078 | McDermott International, Inc. | GL IND |
| 4/1/1996 | 4/1/1997 ISA H07131616 | McDermott International, Inc See attachment 113 | Auto IND |
| 4/1/1996 | 4/1/1997 PLI G19067236 | McDermott International, Inc See attachment 105 | GL IND |
| 4/1/1996 | 4/1/1997 RSC C2733278A | Babcock & Wilcox Company - See attachment 142 | WC Cigna |
| 4/1/1996 | 4/1/1997 RSC C4206726A | Babcock & Wilcox Company | WC Cigna |
| 4/1/1996 | 4/1/1997 SCA 012575 | The Babcock & Wilcox Company - See attachment 93 | Auto IND |
| 4/1/1996 | 4/1/1997 WLR C27332080 | Babcock & Wilcox Company - See attachment 201 | WC Cigna |
| 4/1/1996 | 4/1/1997 WLR C27332742 | McDermott International, Inc See attachment 155 | WC Bankers Standard |
| 4/1/1996 | 4/1/1997 WLR C27332766 | Hudson Companies | WC Bankers Standard |
| 4/1/1996 | 4/1/1997 WLR C27332778 | Babcock & Wilcox Company - See attachment 140 | WC Bankers Standard |

| 4/1/1996 | 4/1/1997 WLR C27332808 | Babcock & Wilcox Company - See attachment 146 | WC Cigna |
|-----------|-------------------------|---|----------------------|
| 4/1/1996 | 4/1/1997 XSL G18967911 | Babcock & Wilcox Company - See attachment 78 | GL INA |
| 4/1/1996 | 4/1/1997 XSL G1896823A | Hudson Companies - See attachment 55 | XGL IND |
| 4/1/1996 | 4/1/1997 XSL G18968277 | McDermott International, Inc See attachment 123 | XGL IND |
| 4/1/1996 | 4/1/1997 XWC 011614 | McDermott International, Inc See attachment 135 | XWC Cigna |
| 8/24/1996 | 8/24/1997 CAL H07132001 | Deep Oil Technology, Inc. | Auto INA |
| 8/24/1996 | 8/24/1997 NWC C42181018 | Deep Oil Technology, Inc. | WC INA |
| 8/24/1996 | 8/24/1997 OGL G18967364 | Deep Oil Technology, Inc. | GL INA |
| 4/1/1997 | 4/1/1998 RSC C42109484 | Babcock & Wilcox Company - See attachment 21 | WC Pacific Employers |
| 4/1/1997 | 4/1/1998 RSC C42109502 | Babcock & Wilcox Company | WC Cigna |
| 4/1/1997 | 4/1/1998 WLR C42109460 | Babcock & Wilcox Company - See attachment 128 | WC Pacific Employers |
| 4/1/1997 | 4/1/1998 WLR C42109472 | Babcock & Wilcox Company - See attachment 193 | WC Bankers Standard |
| 4/1/1997 | 4/1/1998 WLR C42109496 | Babcock & Wilcox Company - See attachment 81 | WC Pacific Employers |
| 4/1/1997 | 4/1/1998 WLR C42109514 | Hudson Companies | WC Bankers Standard |
| 4/1/1997 | 4/1/1998 WLR C42109526 | McDermott International, Inc See attachment 156 | WC Bankers Standard |
| 8/24/1997 | 8/24/1998 CAL H07322574 | Deep Oil Technology, Inc. | Auto INA |
| 8/24/1997 | 8/24/1998 NWC C42425497 | Deep Oil Technology, Inc. | WC INA |
| 8/24/1997 | 8/24/1998 OGL G19326792 | Deep Oil Technology, Inc. | GL INA |
| | | | |

| 12/18/1997 | 12/18/1998 | OGL G1965722A | Monmouth Couty Materials Processing & Recovery Facility; National Ecology Company & The Babcock and Wilcox Company - See attachment 157 | GL | IND |
|------------|------------|----------------|---|------|-------------------|
| 4/1/1998 | 4/1/1999 | HDC G19657280 | Babcock & Wilcox Company - See attachment 61 | GL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | HDC G19657334 | McDermott International, Inc - See attachment 267 | GL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | HDO G19657322 | McDermott International, Inc - See attachment 265 | GL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | HDO G219657292 | Babcock & Wilcox Company - See attachment 60 | GL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | ISA H07323360 | The Babcock & Wilcox Company - See attachment 57 | Auto | Pacific Employers |
| 4/1/1998 | 4/1/1999 | ISA H07323372 | McDermott International, Inc - See attachment 268 | Auto | Pacific Employers |
| 4/1/1998 | 4/1/1999 | PLI G19657346 | McDermott International, Inc See attachment 59 | GL | INA |
| 4/1/1998 | 4/1/1999 | PWC C42460187 | Babcock & Wilcox Company - See attachment 163 | WC | Pacific Employers |
| 4/1/1998 | 4/1/1999 | WLR C42460151 | Babcock & Wilcox Company - See attachment 130 | WC | Pacific Employers |
| 4/1/1998 | 4/1/1999 | WLR C42460163 | Babcock & Wilcox Company - See attachment 177 | WC | Pacific Employers |
| 4/1/1998 | 4/1/1999 | WLR C42460175 | Babcock & Wilcox Company - See attachment 175 | WC | Cigna - TX |
| 4/1/1998 | 4/1/1999 | WLR C42460199 | McDermott International, Inc See attachment 139 | WC | INA |
| 4/1/1998 | 4/1/1999 | WLR C42460205 | Hudson Companies | WC | Cigna - TX |

Page 14 of 25

| 4/1/1998 | 4/1/1999 | XSL G19657279 | The Babcock & Wilcox Company - See attachment 62 | XGL | Pacific Employers |
|------------|------------|----------------|--|------|-------------------|
| 4/1/1998 | 4/1/1999 | XSL G19657309 | Hudson Companies - See attachment 47 | XGL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | XSL G19657309 | Hudson Companies - See attachment 58 | XGL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | XSL G19657310 | McDermott International, Inc - See attachment 264 | XGL | Pacific Employers |
| 4/1/1998 | 4/1/1999 | XWC 011786 | McDermott International, Inc - See attachment 266 | XWC | Cigna |
| 8/24/1998 | 8/24/1999 | CAL H07406794 | Deep Oil Technology, Inc. | Auto | Cigna |
| 8/24/1998 | 8/24/1999 | OGL G19657905 | Deep Oil Technology, Inc. | GL | Cigna |
| 10/1/1998 | 10/1/2001 | XOO G19658193 | State Line Energy, LLC | XGL | Cigna |
| 12/18/1998 | 12/18/1999 | CAL H07407385 | National Ecology Company - See attachment 161 | Auto | Cigna |
| 12/18/1998 | 12/18/1999 | OGL G19327024 | National Ecology Company, Power Systems Operations, Inc. and The Babcock | GL | Cigna |
| | | | & Wilcox Company and Monmouth County Materials Processing and Recovery | | |
| | | | Facility | | |
| 4/1/1999 | 1/1/2000 | HDC G19657280 | Babcock & Wilcox Company - See attachment 61 | GL | Pacific Employers |
| 4/1/1999 | 1/1/2000 | HDO G219657292 | Babcock & Wilcox Company - See attachment 60 | GL | Pacific Employers |
| 4/1/1999 | 1/1/2000 | PLI G19657346 | McDermott International, Inc See attachment 59 | GL | INA |
| 4/1/1999 | 4/1/2000 | PWC C42643890 | Babcock & Wilcox Company – See attachment 2 | WC | Pacific Employers |
| 4/1/1999 | 4/1/2000 | PWC C42643890 | Babcock & Wilcox Company - See attachment 25 | WC | Pacific Employers |
| | | | | | |

| 4/1/1999 4/1/1999 4/1/1999 | 4/1/2000 WLR C42 4/1/2000 WLR C42 4/1/2000 WLR C42 | Hudson Compani Babcock & Wilco | national, Inc - See attachment 263 es - See attachment 262 ox Company - See attachment 1 | WC WC | Pacific Employers Pacific Employers Pacific Employers |
|----------------------------------|--|-----------------------------------|--|----------|---|
| 4/1/1999 | 1/1/2000 XSL G196 | | Vilcox Company - See attachment 62 | | Pacific Employers |
| 4/1/1999 | 1/1/2000 XSL G196 | | es - See attachment 58 | | Pacific Employers |
| 8/24/1999 | 8/24/2000 CAL H076 | 67823A Deep Oil Technol | logy, Inc. | Auto | Cigna |
| 8/24/1999 | 8/24/2000 NWC C42 | · · · | | WC | Cigna |
| 8/24/1999 | 8/24/2000 OGL G198 | 89521A Deep Oil Technol | logy, Inc. | GL | Cigna |
| 12/18/1999 | 12/18/2000 CAL H076 | 678903 National Ecology | Company - See attachment 272 | AL | Pacific Employers |
| 12/18/1999 | 12/18/2000 CUA-1045 | 533-0 National Ecology | Company - See attachment 271 | Umb | Westchester Fire Ins Co |
| 12/18/1999 | 12/18/2000 OGL G198 | 895993 National Ecology | Company - See attachment 270 | GL | Pacific Employers |
| 1/1/2000 | 1/1/2001 HDC G19 | 657280 Babcock & Wilco | ox Company - See attachment 61 | GL | Pacific Employers |
| 1/1/2000 | 1/1/2001 HDO G21 | 9657292 Babcock & Wilco | ox Company - See attachment 60 | GL | Pacific Employers |
| 1/1/2000 | 1/1/2001 PLI G1965 | 57346 McDermott Interr | national, Inc See attachment 59 | GL | INA |
| 1/1/2000 | 1/1/2001 PWC C42 | 645618 Babcock & Wilco | ox Company - See attachment 13 | WC | Pacific Employers |
| 1/1/2000 | 1/1/2001 WLR C42 | .64562A McDermott Interr | national, Inc See attachment 69 | WC | Pacific Employers |
| 1/1/2000 | 1/1/2001 WLR C42 | Hudson Compani Hudson Compani | es | WC | Pacific Employers |

| 1/1/2000 | 1/1/2001 WLR C42645643 | Babcock & Wilcox Company - See attachment 12 | WC Pacific Employers |
|------------|--------------------------|--|------------------------|
| 1/1/2000 | 1/1/2001 XOO G20292817 | Mid American Energy Company | XGL Pacific Employers |
| 1/1/2000 | 1/1/2001 XSL G19657279 | The Babcock & Wilcox Company - See attachment 62 | XGL Pacific Employers |
| 1/1/2000 | 1/1/2001 XSL G19657309 | Hudson Companies - See attachment 58 | XGL Pacific Employers |
| 12/11/2000 | 6/1/2001 XLG G20295624 | NRG El Segundo Operations Inc. | XGL AAI |
| 12/18/2000 | 12/18/2001 HDO G20295715 | National Ecology Company - See attachment 269 | GL Pacific Employers |
| 12/18/2000 | 12/18/2001 ISA H07668016 | National Ecology Company - See attachment 210 | Auto Pacific Employers |
| 12/18/2000 | 7/31/2001 XLG G20295661 | NRG Cabrillo Power I | XGL AAI |
| 1/1/2001 | 1/1/2002 HDC G19657280 | Babcock & Wilcox Company - See attachment 61 | GL Pacific Employers |
| 1/1/2001 | 1/1/2002 HDO G219657292 | Babcock & Wilcox Company - See attachment 60 | GL Pacific Employers |
| 1/1/2001 | 1/1/2002 PLI G19657346 | McDermott International, Inc See attachment 59 | GL INA |
| 1/1/2001 | 1/1/2002 SCF C43139988 | Babcock & Wilcox Company - See attachment 11 | WC Pacific Employers |
| 1/1/2001 | 1/1/2002 WLR C43139940 | Babcock & Wilcox Company - See attachment 10 | WC Pacific Employers |
| 1/1/2001 | 1/1/2002 WLR C43140024 | McDermott International, Inc See attachment 277 | WC Pacific Employers |
| 1/1/2001 | 1/1/2002 WLR C43140061 | Hudson Companies | WC Pacific Employers |
| 1/1/2001 | 1/1/2002 XOO G20292817 | Mid American Energy Company | XGL Pacific Employers |
| 1/1/2001 | 1/1/2002 XSL G19657279 | The Babcock & Wilcox Company - See attachment 62 | XGL Pacific Employers |
| 1/1/2001 | 1/1/2002 XSL G19657309 | Hudson Companies - See attachment 58 | XGL Pacific Employers |

| 7/2/2001 | 12/31/2001 | XLG G20298017 | Mirant Delta LLC | XGL | AAI |
|------------|------------|----------------|--|-----|-------------------|
| 7/31/2001 | 3/31/2002 | XLG G20295661 | NRG Cabrillo Power I | XGL | AAI |
| 12/18/2001 | 12/18/2002 | HDO G20298431 | National Ecology Company - See attachment 276 | GL | Pacific Employers |
| 12/18/2001 | 12/18/2002 | ISA H0766946A | National Ecology Company - See attachment 275 | AL | Pacific Employers |
| 1/1/2002 | 1/1/2003 | HDC G19657280 | Babcock & Wilcox Company - See attachment 61 | GL | Pacific Employers |
| 1/1/2002 | 1/1/2003 | HDO G219657292 | Babcock & Wilcox Company - See attachment 60 | GL | Pacific Employers |
| 1/1/2002 | 1/1/2003 | PLI G19657346 | McDermott International, Inc See attachment 59 | GL | INA |
| 1/1/2002 | 1/1/2003 | SCF C43100075 | Babcock & Wilcox Company - See attachment 27 | WC | Pacific Employers |
| 1/1/2002 | 1/1/2003 | WLR C43099991 | McDermott International, Inc See attachment 16 | WC | Pacific Employers |
| 1/1/2002 | 1/1/2003 | XSL G19657279 | The Babcock & Wilcox Company - See attachment 62 | XGL | Pacific Employers |
| 1/1/2002 | 1/1/2003 | XSL G19657309 | Hudson Companies - See attachment 58 | XGL | Pacific Employers |
| 1/1/2002 | 1/1/2003 | XWC 014129 | McDermott International, Inc See attachment 56 | XWC | AAI |
| 3/31/2002 | 4/30/2002 | XLG G20295661 | NRG Cabrillo Power I | XGL | AAI |
| 10/1/2002 | 1/1/2003 | XLG G20580161 | Mid American Energy Company | XGL | AAI |
| 12/18/2002 | 12/18/2003 | HDO G2030442A | National Ecology Company - See attachment 273 | GL | AAI |
| 12/18/2002 | 12/18/2003 | ISA H07669884 | National Ecology Company - See attachment 268 | AL | AAI |
| 1/1/2003 | 1/1/2004 | HDC G20580033 | McDermott International, Inc. | GL | AAI |
| 1/1/2003 | 1/1/2004 | HDC G20580124 | Babcock & Wilocx Company - See attachment 31 | GL | AAI |

| 1/1/2003 | 1/1/2004 HDO G20579997 | Babcock & Wilcox Company - See attachment 42 | GL AAI |
|------------|--------------------------|--|----------|
| 1/1/2003 | 1/1/2004 HDO G20580070 | McDermott International, Inc. | GL AAI |
| 1/1/2003 | 1/1/2004 ISA H07670023 | McDermott International, Inc See attachment 34 | Auto AAI |
| 1/1/2003 | 1/1/2004 ISA H07670060 | The Babcock & Wilcox Company - See attachment 30 | Auto AAI |
| 1/1/2003 | 1/1/2004 PLI G20580112 | McDermott International, Inc See attachment 33 | GL AAI |
| 1/1/2003 | 1/1/2004 SCF C43497194 | Babcock & Wilcox Company - See attachment 24 | WC AAI |
| 1/1/2003 | 1/1/2004 WLR C43497157 | Babcock & Wilcox Company - See attachment 23 | WC AAI |
| 1/1/2003 | 1/1/2004 WLR C43497169 | McDermott International, Inc See attachment 70 | WC AAI |
| 1/1/2003 | 1/1/2004 XSL G20579559 | McDermott International, Inc See attachment 32 | XGL AAI |
| 1/1/2003 | 1/1/2004 XSL G20580045 | The Babcock & Wilcox Company - See attachment 29 | XGL AAI |
| 2/1/2003 | 12/17/2003 OCP G20304479 | Pasadena ISD | GL AAI |
| 3/1/2003 | 3/1/2004 XLG G20579511 | Reliant Energy | XGL AAI |
| 12/17/2003 | 12/17/2004 OCP G20304479 | Pasadena ISD | GL AAI |
| 12/18/2003 | 12/18/2004 HDO G2030442A | National Ecology Company - See attachment 273 | GL AAI |
| 12/18/2003 | 12/18/2004 ISA H07669884 | National Ecology Company - See attachment 268 | AL AAI |
| 1/1/2004 | 1/1/2005 HDC G20300140 | Babcock & Wilcox Company - See attachment 66 | GL AAI |
| 1/1/2004 | 1/1/2005 HDO G20300103 | Babcock & Wilcox Company - See attachment 64 | GL AAI |

| 1/1/2004 | 1/1/2005 | ISA H07670485 | The Babcock & Wilcox Company - See acchment 63 | Auto | AAI |
|------------|------------|----------------|--|------|-----|
| 1/1/2004 | 1/1/2005 | SCF C43966169 | Babcock & Wilcox Company - See attachment 14 | WC | AAI |
| 1/1/2004 | 1/1/2005 | WLR C43966157 | Babcock & Wilcox Company - See attachment 15 | WC | AAI |
| 1/1/2004 | 1/1/2005 | WLR C43966303 | McDermott International, Inc See attachment 211 | WC | AAI |
| 1/1/2004 | 1/1/2005 | XSL G20300061 | The Babcock & Wilcox Company - See attachment 65 | XGL | AAI |
| 2/1/2004 | 2/1/2005 | WCU 014570 | Babcock and Wilcox Investment Company | XWC | AAI |
| 12/17/2004 | 12/17/2005 | OCP G20304479 | Pasadena ISD | GL | AAI |
| 12/18/2004 | 12/18/2005 | HDO G2030442A | National Ecology Company - See attachment 273 | GL | AAI |
| 12/18/2004 | 12/18/2005 | ISA H07669884 | National Ecology Company - See attachment 268 | AL | AAI |
| 1/1/2005 | 1/1/2006 | SCF C4396494A | Babcock & Wilcox Company - See attachment 67 | WC | AAI |
| 1/1/2005 | 1/1/2006 | WLR C43988761 | Babcock & Wilcox Company - See attachment 68 | WC | AAI |
| 1/1/2005 | 1/1/2006 | WLR C43988840 | McDermott International, Inc | WC | AAI |
| 9/19/2005 | 12/18/2005 | XSL G20299721 | American Electric Power Service Corp | XGL | AAI |
| 9/19/2005 | 12/18/2005 | XSL G20299769 | American Electric Power Service Corp | XGL | AAI |
| 12/17/2005 | 12/17/2006 | OCP G20304479 | Pasadena ISD | GL | AAI |
| 12/18/2005 | 12/18/2009 | HDO G21703618R | National Ecology Company | GL | AAI |
| 1/1/2006 | 1/1/2007 | HDC G21728007 | Babcock & Wilcox Company - See attachment 217 | GL | AAI |
| | | | | | |

| 1/1/2006 | 1/1/2007 HDC G21728032 | McDermott International, Inc See attachment 238 | GL AAI |
|------------|--------------------------|---|-----------------------------|
| 1/1/2006 | 1/1/2007 HDO G21727994 | Babcock & Wilcox Company - See attachment 215 | GL AAI |
| 1/1/2006 | 1/1/2007 HDO G21728020 | McDermott International, Inc. | GL AAI |
| 1/1/2006 | 1/1/2007 ISA H07670965 | McDermott International, Inc - See attachment 240 | Auto AAI |
| 1/1/2006 | 1/1/2007 ISA H07670977 | Babcock & Wilcox Company - See attachment 214 | Auto AAI |
| 1/1/2006 | 1/1/2007 PLI G21728044 | McDermott International, Inc. See attachment 239 | GL Illinois Union Insurance |
| | | | Company |
| 1/1/2006 | 1/1/2007 SCF C44185520 | Babcock & Wilcox Company - See attachment 5 | WC AAI |
| 1/1/2006 | 1/1/2007 WCU C44185544 | McDermott International, Inc See attachment 212 | XWC AAI |
| 1/1/2006 | 1/1/2007 WLR C44185532 | Babcock & Wilcox Company - See attachment 3 | WC AAI |
| 1/1/2006 | 1/1/2007 WLR C44185556 | McDermott International, Inc. | WC AAI |
| 1/1/2006 | 1/1/2007 XSL G21728056 | McDermott International, Inc See attachment 237 | GL AAI |
| 1/1/2006 | 1/1/2007 XSL G21728068 | Babcock & Wilcox Company - See attachment 216 | GL AAI |
| 12/18/2006 | 12/18/2007 CAL H07670928 | National Ecology Company - See attachment 267 | AL AAI |
| 1/1/2007 | 1/1/2008 HDC G21736727 | McDermott International, Inc - See attachment 242 | GL AAI |
| 1/1/2007 | 1/1/2008 HDC G21736764 | Babcock & Wilcox Company - See attachment 229 | GL AAI |
| 1/1/2007 | 1/1/2008 HDC G2372358A | BWXT Services, Inc See attachment 246 | GL AAI |

Page 21 of 25

| 1/1/2007 | 1/1/2008 HDO G21736739 | McDermott International, Inc See attachment 243 | GL | AAI |
|----------|------------------------|---|------|-----|
| 1/1/2007 | 1/1/2008 HDO G21736752 | Babcock & Wilcox Company - See attachment 228 | GL | AAI |
| 1/1/2007 | 1/1/2008 ISA H07673206 | McDermott International, Inc - See attachment 241 | Auto | AAI |
| 1/1/2007 | 1/1/2008 ISA H07673218 | Babcock & Wilcox Company - See attachment 227 | Auto | AAI |
| 1/1/2007 | 1/1/2008 ISA H07834597 | BWXT Services, Inc See attachment 245 | Auto | AAI |
| 1/1/2007 | 1/1/2008 SCF C44450653 | Babcock & Wilcox Company - See attachment 9 | WC | AAI |
| 1/1/2007 | 1/1/2008 WCU C44450550 | Babcock & Wilcox Investment Company | WC | AAI |
| 1/1/2007 | 1/1/2008 WCU C44450604 | McDermott International, Inc. | XWC | AAI |
| 1/1/2007 | 1/1/2008 WLR C44450203 | BWXT Services, Inc. | WC | AAI |
| 1/1/2007 | 1/1/2008 WLR C44450616 | Babcock & Wilcox Company - See attachment 4 | WC | AAI |
| 1/1/2007 | 1/1/2008 WLR C44450665 | McDermott International, Inc. | WC | AAI |
| 1/1/2007 | 1/1/2008 XSL G21736235 | BWTX Services, Inc See attachment 247 | GL | AAI |
| 1/1/2007 | 1/1/2008 XSL G21736740 | Babcock & Wilcox Company - See attachment 230 | GL | AAI |
| 1/1/2007 | 1/1/2008 XSL G21736776 | McDermott International, Inc See attachment 244 | GL | AAI |
| 1/1/2008 | 1/1/2009 HDC G23734758 | Babcock & Wilcox Company - See attachement 220 | GL | AAI |
| 1/1/2008 | 1/1/2009 HDC G23734783 | McDermott International, Inc See attachment 249 | GL | AAI |
| 1/1/2008 | 1/1/2009 HDO G23734746 | Babcock & Wilcox Company - See attachment 219 | GL | AAI |

| 1/1/2008 | 1/1/2009 HDO G23734795 | McDermott International, Inc See attachment 250 | Gl | AAI |
|----------|------------------------|---|------|-----|
| 1/1/2008 | 1/1/2009 ISA H0823839A | The Babcock & Wilcox Company - See attachment 221 | Auto | AAI |
| 1/1/2008 | 1/1/2009 ISA H08238418 | McDermott International, Inc See attachment 251 | Auto | AAI |
| 1/1/2008 | 1/1/2009 PLI G2373476A | McDermott International, Inc See attachment 248 | GL | AAI |
| 1/1/2008 | 1/1/2009 SCF C44479825 | The Babcock & Wilcox Company - See attachment 7 | WC | AAI |
| 1/1/2008 | 1/1/2009 WCU C44479850 | McDermott International, Inc. | XWC | AAI |
| 1/1/2008 | 1/1/2009 WLR C44478596 | The Babcock & Wilcox Company | WC | AAI |
| 1/1/2008 | 1/1/2009 WLR C44479813 | The Babcock & Wilcox Company - See attachment 213 | WC | IND |
| 1/1/2008 | 1/1/2009 WLR C44479862 | McDermott International, Inc. | WC | IND |
| 1/1/2008 | 1/1/2009 XSL G23733766 | The Babcock & Wilcox Company - See attachment 218 | GL | AAI |
| 1/1/2009 | 1/1/2010 HDC G23749154 | McDermott International, Inc See attachment 254 | GL | AAI |
| 1/1/2009 | 1/1/2010 HDC G23749270 | The Babcock & Wilcox Company - See attachment 223 | GL | AAI |
| 1/1/2009 | 1/1/2010 HDO G23749117 | McDermott International, Inc - See attachment 252 | GL | AAI |
| 1/1/2009 | 1/1/2010 HDO G23749233 | The Babcock & Wilcox Company - See attachment 225 | GL | AAI |
| 1/1/2009 | 1/1/2010 ISA H08252464 | McDermott International, Inc - See attachment 256 | Auto | AAI |
| 1/1/2009 | 1/1/2010 ISA H08252506 | The Babcock & Wilcox Company - See attachment 224 | Auto | AAI |
| 1/1/2009 | 1/1/2010 PLI G23749075 | M.D. (41.4 (* 14.6 (* 4.1 (* 42.2 | GL | AAI |
| | | McDermott International, Inc See attachment 253 | | |

| 1/1/2009 | 1/1/2010 SCF C44356909 | Babcock & Wilcox Company - See attachment 6 | WC AA | I |
|----------|------------------------|---|---------|----|
| 1/1/2009 | 1/1/2010 WCU C4435682A | McDermott International, Inc. | XWC AA | I |
| 1/1/2009 | 1/1/2010 WCU C44356867 | Babcock & Wilcox Investment Company | WC AA | ΛI |
| 1/1/2009 | 1/1/2010 WLR C44356788 | McDermott International, Inc. | WC INI | D |
| 1/1/2009 | 1/1/2010 WLR C44356946 | The Babcock & Wilcox Company - See attachment 226 | WC INI | D |
| 1/1/2009 | 1/1/2010 WLR C44356983 | The Babcock & Wilcox Company - See attachment 8 | WC AA | ΛI |
| 1/1/2009 | 1/1/2010 XSL G23749191 | McDermott International, Inc See attachment 255 | GL AA | ΛI |
| 1/1/2009 | 1/1/2010 XSL G23749312 | The Babcock & Wilcox Company - See attachment 222 | GL AA | ΛI |
| 1/1/2010 | 1/1/2011 HDC G24938463 | The Babcock & Wilcox Company - See attachment 234 | GL AA | ΛI |
| 1/1/2010 | 1/1/2011 HDC G24938499 | McDermott International, Inc See attachment 258 | GL AA | ΛI |
| 1/1/2010 | 1/1/2011 HDO G24938475 | The Babcock & Wilcox Company - See attachment 233 | GL AA | ΛI |
| 1/1/2010 | 1/1/2011 ISA H0858381A | McDermott International, Inc See attachmetn 259 | Auto AA | ΛI |
| 1/1/2010 | 1/1/2011 PLI G24938505 | McDermott International, Inc See attachment 257 | GL AA | ΛI |
| 1/1/2010 | 1/1/2011 SCF C45709516 | The Babcock & Wilcox Company - See attachment 236 | WC AA | ·Ι |
| 1/1/2010 | 1/1/2011 WCU C45709541 | McDermott International, Inc | WC AA | ΛI |
| 1/1/2010 | 1/1/2011 WLR C45709498 | The Babcock & Wilcox Company - See attachment 231 | WC AA | ·Ι |
| 1/1/2010 | 1/1/2011 WLR C45709504 | The Babcock & Wilcox Company - See attachment 232 | WC INI | D |

| Evhibit 1 | L - Policies | and Named | Incured'e |
|-----------|--------------|-----------|-----------|
| | | | |

| 1/1/2010 | 1/1/2011 WLR C4570953A | McDermott International, Inc See attachment 261 | WC IND |
|----------|------------------------|---|--------|
| 1/1/2010 | 1/1/2011 XSL G24938451 | The Babcock & Wilcox Company - See attachment 235 | GL AAI |
| 1/1/2010 | 1/1/2011 XSL G24938487 | McDermott International, Inc See attachment 260 | GL AAI |

Page 25 of 25

EXHIBIT IV – MII ENTITIES

| See Schedule 1.1(b) and Schedule 1.1(j) attached hereto. No MII Entity listed on Schedule 1.1(b) shall be deemed to be an "Insured", a "Named Insured" or otherwise |
|---|
| be deemed to be insured under any Existing Policy or Existing Insurance Agreement solely by virtue of being listed on such Schedule 1.1(b). |

Schedule 1.1(b) MII FORMER

| Reference ID | |
|--------------|---|
| 947 | Arabian General Contracting Company |
| 911 | Arabian Petroleum Marine Construction Company |
| 8000 | Associated Pipe Line Contractors, Inc. |
| 117 | Badya Builders, Inc. |
| 242 | Beheer-en Beleggingsmaatschappij Belesa B.V. |
| 9996 | BJA-Mentor Production Systems Limited |
| 164 | Brown & Root McDermott Fabricators Limited |
| 277 | Cayenne Insurance Company, Ltd. |
| 629 | CCC Fabricaciones y Construcciones S.A. de C.V. |
| 940 | Construcciones Maritimas Mexicanas, S.A. de C.V. |
| 952 | Davy McDermott Limited |
| 921 | DB/McDermott Company |
| 1952 | Deep Gulf Contractors LLC |
| 171 | Deep Sea Divers Australia Pty. Limited |
| | Dick Evans, Inc. |
| 997 | ETPM International (UK) Limited |
| 955 | ETPM International S.A.S. |
| 999 | Far East Vessels, Inc. |
| 108 | First Emirates Trading Corporation |
| | Harvey Lumber & Supply Company |
| 174 | Heavy Lift Chartering, Inc. |
| 935 | HeereMac v.o.f. |
| 956 | Heerema-McDermott (Aust.) Pty. Ltd. |
| 190 | Honore Insurance Company, Ltd. |
| 917 | Initec, Astano y McDermott International Inc., S.A. |
| | Ingenieria Petrolera Maritima, S.A. de C.V. |
| | Iranian Marine Contractors, Inc. |
| 102 | J. Ray McDermott & Co. (Nederland) B.V. |
| 150 | J. Ray McDermott (Aust.) Pty. Ltd. |
| 151 | J. Ray McDermott (U.K.) Inc. |
| 350 | J. Ray McDermott (Venezuela) C.A. |
| 325 | J. Ray McDermott Arctic Ltd. |
| | |

| 2010 | | |
|------|---|------------------------------|
| 307 | J. Ray McDermott Gulf Contractors, Inc. | (formerly the Early Company) |
| 394 | J. Ray McDermott International Services (No. 1) Limited | |
| 101 | J. Ray McDermott International Services Limited | |
| 152 | J. Ray McDermott London, Ltd. | |
| 641 | J. Ray McDermott Marine Construction AS | |
| 1125 | J. Ray McDermott Newfoundland, Ltd. | |
| 871 | J. Ray McDermott Offshore Chartering (U.S.), Inc. | |
| 372 | J. Ray McDermott Properties, Inc. | |
| 238 | J. Ray McDermott Underwater Services, Inc. | |
| 603 | J. Ray McDermott Venture Holdings (U.S.), Inc. | |
| 153 | Jaramac Associated Services, Ltd. | |
| | Jaramac Petroleum (U.K.) Limited | |
| | Khafji Contractors, Inc. | |
| 166 | Khor Contractors, Inc. | |
| 903 | KME a.s. | |
| 324 | Lan-Dermott, S.A. de C.V. | |
| 202 | Madock Shipyard Company, Ltd. | |
| 203 | Macshelf Marine Construction Company, Ltd. | |
| 106 | Mandato Pty. Limited | |
| 360 | Marine Contractors, Inc. | |
| 146 | McAntille, N.V. | |
| 168 | McDermott - ETPM V.O.F. | |
| 158 | McDermott (Holland) B.V. | |
| 304 | McDermott (Nigeria) Limited | |
| 245 | McDermott Acquisition Company, Inc. | |
| 972 | McDermott APG Services Limited | |
| 352 | McDermott Azerbaijan Pipelines, Inc. | |
| 156 | McDermott Azerbaijan, Inc. | |
| 351 | McDermott Azerbaijan, inc. McDermott Central & Eastern Europe, Inc. | |
| 331 | McDermott de Mexico, S.A. de C.V. | |
| 118 | McDermott Denmark ApS | |
| 464 | McDermott Energy Services, Inc. | |
| 178 | McDermott Engineering (Europe) Limited | |
| 170 | McDermott Enterprises France, S.A. | |
| 128 | McDermott Europe Marine Services, Ltd. | |
| 313 | McDermott Industries, Inc. | |
| 354 | McDermott Inland Services, Inc. | |
| | | |

| 201 | McDermott International Asset Management, Ltd. | |
|------|--|---|
| 343 | McDermott International Aviation, Inc. | |
| 346 | McDermott International Aviation, Inc. | (formerly McDermott International Aviation Co., Inc.) |
| | McDermott International (Deutschland) GmbH | |
| 237 | McDermott International Engineering & Construction Co., Ltd. | |
| 139 | McDermott International Engineering Investments N.V. | |
| 121 | McDermott International General Services, Inc. | |
| 123 | McDermott International Marine Services, Inc. | |
| 236 | McDermott International Marketing, Inc. | |
| 143 | McDermott Marine UK Limited | |
| 462 | McDermott Marketing Services, Inc. | |
| 103 | McDermott Middle East Trading, Ltd. | |
| 172 | McDermott Neutral Zone, Inc. | |
| | McDermott Norway, Inc. | |
| | McDermott Petroleum (Iran), Inc. | |
| 105 | McDermott Sakhalin, Inc. | |
| 347 | McDermott Shipbuilding, Inc. | |
| 111 | McDermott Singapore Pte. Ltd. | |
| 221 | McDermott Submarine Cable Ltd. | |
| 1951 | McDermott Submarine Cable Systems Limited | |
| 205 | McDermott Subsea Constructors Limited | |
| 110 | McDermott Transition Co., Inc. | |
| | McDermott UAR, Inc. | |
| 184 | McDermott West Indies Company | |
| 187 | McDermott-ETPM (Norway), Inc. | |
| 998 | McDermott-ETPM (UK) Limited | |
| 963 | McDermott-ETPM B.V. | |
| 161 | McDermott-ETPM East B.V. | |
| 167 | McDermott-ETPM East N.V. | |
| 964 | McDermott-ETPM N.V. | |
| 965 | McDermott-ETPM S.A. | |
| 962 | McDermott-ETPM West, Inc. | |
| 104 | Menck GmbH | |
| 463 | Mentor Engineering Consultants Limited | |
| 131 | MIMCO, Inc. | |
| 908 | MMC-McDermott Engineering Sdn. Berhad | |
| 317 | Mofco N.V. | |
| | 3 | |
| | 3 | |

6/2/2010

| 129 | NOLA, Servicos E Participações Ltda. |
|------------|---|
| 141 | Northern Marine Services, Inc. |
| 107 | Oceanic Red Sea Company |
| 620 | Offshore Angola, Ltd. |
| 632 | Offshore Energy Capital Corporation |
| 622 | Offshore Hyundai International Limited |
| 630 | Offshore Hyundai International, Ltd. |
| 210 | Offshore Marine Chartering N.V. |
| 610 | Offshore Petroleum Divers, Inc. |
| 614 | Offshore Pipelines (Mauritius) Ltd. |
| 616 | Offshore Pipelines Far East Limited |
| 617 | Offshore Pipelines International Gulf E.C. |
| 633 | Offshore Power Generation Ltd. |
| 635 | Offshore Production Vessels, Ltd. |
| 615 | OPI Offshore Netherlands Holding B.V. |
| 611 | OPI Sales & Service Corporation |
| 628 | OPI Towing & Supply, L.L.C. |
| 624 | OPMI, E.C. |
| 613 | P. T. Armandi Pranaupaya |
| 850 | Panama Offshore Chartering Company 1, Inc. |
| 850 859 | Panama Offshore Chartering Company 1, Inc. Panama Offshore Chartering Company 10, Inc. |
| 859 860 | Panama Offshore Chartering Company 10, Inc. Panama Offshore Chartering Company 11, Inc. |
| 861 | Panama Offshore Chartering Company 11, Inc. Panama Offshore Chartering Company 12, Inc. |
| 862 | Panama Offshore Chartering Company 12, Inc. |
| 802 | Panama Offshore Chartering Company 13, Inc. |
| 863 | Panama Offshore Chartering Company 14, Inc. |
| 864 | Panama Offshore Chartering Company 15, Inc. |
| 865 | Panama Offshore Chartering Company 16, Inc. |
| 866 | Panama Offshore Chartering Company 17, Inc. |
| 867 | Panama Offshore Chartering Company 18, Inc. |
| 868 | Panama Offshore Chartering Company 19, Inc. |
| 851 | Panama Offshore Chartering Company 2, Inc. |
| 869 | Panama Offshore Chartering Company 20, Inc. |
| 870 | Panama Offshore Chartering Company 21, Inc. |
| 852 | Panama Offshore Chartering Company 3, Inc. |
| 853 | Panama Offshore Chartering Company 4, Inc. |
| 854 | Panama Offshore Chartering Company 5, Inc. |
| 855 | Panama Offshore Chartering Company 6, Inc. |
| | |

6/2/2010

| 856 | Panama Offshore Chartering Company 7, Inc. |
|-----|--|
| 857 | Panama Offshore Chartering Company 8, Inc. |
| 858 | Panama Offshore Chartering Company 9, Inc. |
| 154 | Panama Offshore Chartering Company, Inc. |
| 990 | Personal Administrativo de Vera Cruz, S.A. de C.V. |
| 991 | Personal Ejecutivo de Veracruz, S.A. de C.V. |
| 645 | Pirogue Insurance Company, Ltd. |
| 981 | Sakhalin Energy Investment Company, Ltd. |
| 384 | Samburg Holdings, Inc. |
| 109 | Sharman, Allen, Gay & Taylor, S.A. |
| 989 | Tallares Navales del Golfo, S.A. de C.V. |
| | The Evelyn Kay Company |
| | The Roger Thomas Company |
| | The Rosalie E. Company |
| | The Walter E. Company |
| 634 | TL Marine Sdn. Bhd. |
| 621 | TL Offshore Sdn. Bhd. |
| 926 | Topside Contractors of Newfoundland |
| 925 | Topside Contractors of Newfoundland, Ltd. |
| 642 | U.S. Offshore Chartering |
| 918 | Universal Fabricators Incorporated |
| 980 | US Shipbuilding Corporation, Inc. |
| 318 | Wagley, Inc. |
| | |

Schedule 1.1(j) MII - CURRENT

| CURRENT | | |
|--------------|---|--|
| Reference ID | Name | |
| 873 | Barmada McDermott (L) Limited | |
| 1953 | Barmada McDermott Sdn. Bhd. | |
| 276 | Boudin Insurance Company, Ltd. | |
| 939 | Caspian Offshore Fabricators LLC | |
| 116 | Chartering Company (Singapore) Pte. Ltd. | |
| 142 | Eastern Marine Services, Inc. | |
| 249 | Global Energy - McDermott Limited | |
| 140 | Hydro Marine Services, Inc. | |
| 649 | International Vessels Ltd. | |
| 148 | J. Ray McDermott (Aust.) Holding Pty. Limited | |
| 1982 | J. Ray McDermott (Caspian), Inc. | (formerly J. Ray McDermott Diving International, Inc.) |
| 243 | J. Ray McDermott (Nigeria) Ltd. | |
| 2025 | J. Ray McDermott (Norway), AS | |
| 1237 | J. Ray McDermott (Qingdao) Pte. Ltd. | |
| 194 | J. Ray McDermott Asia Pacific Pte. Ltd. | (formerly McDermott South East Asia Pte. Ltd.) |
| 1972 | J. Ray McDermott Canada Holding, Ltd. | |
| 1969 | J. Ray McDermott Canada, Ltd. | |
| 204 | J. Ray McDermott Contractors, Inc. | |
| 246 | J. Ray McDermott de Mexico, S.A. de C.V. | |
| 208 | J. Ray McDermott Eastern Hemisphere Limited | (formerly J. Ray Middle East (Indian Ocean) Ltd.) |
| 1956 | J. Ray McDermott Engineering Services Private Limited | |
| 1950 | J. Ray McDermott Engineering, LLC | |
| 220 | J. Ray McDermott Far East, Inc. | |
| 211 | J. Ray McDermott Holdings, LLC | (formerly J. Ray McDermott Holdings, Inc.) |
| 627 | J. Ray McDermott International Vessels, Ltd. | (formerly OPI International, Ltd.) |
| 602 | J. Ray McDermott International, Inc. | |
| 160 | J. Ray McDermott Investments B.V. | (formerly Menck B.V.) |
| 1958 | J. Ray McDermott Kazakhstan Limited Liability Partnership | |
| 1975 | J. Ray McDermott Logistic Services Pvt. Limited | |
| 136 | J. Ray McDermott Middle East, Inc. | |
| 239 | J. Ray McDermott Solutions, Inc. | (formerly J. Ray McDermott Engineering Holdings, Inc.) |
| 625 | J. Ray McDermott Technology, Inc. | |
| 1957 | J. Ray McDermott UK Ltd. | |

1

| 1960 | J. Ray McDermott Underwater Services, Inc. | |
|------|---|--|
| 248 | J. Ray McDermott West Africa Holdings, Inc. | |
| 244 | J. Ray McDermott West Africa, Inc. | |
| 601 | J. Ray McDermott, Inc. | |
| 207 | J. Ray McDermott, S.A. | |
| 644 | Lagniappe Insurance Company, Ltd. | |
| 193 | Malmac Sdn. Bhd. | |
| 132 | McDermott (Malaysia) Sendirian Berhad | |
| 904 | McDermott Abu Dhabi Offshore Construction Company | |
| 915 | McDermott Arabia Company Limited | |
| 149 | McDermott Australia Pty. Ltd. | (formerly McDermott Industries (Aust.) Pty. Limited) |
| 183 | McDermott Azerbaijan Marine Construction, Inc. | |
| 241 | McDermott Caspian Contractors, Inc. | |
| 275 | McDermott Cayman Ltd. | |
| 181 | McDermott Far East, Inc. | |
| 133 | McDermott Gulf Operating Company, Inc. | |
| 177 | McDermott Holdings (U.K.) Limited | |
| 300 | McDermott Incorporated | |
| 147 | McDermott International B.V. | |
| 120 | McDermott International Investments Co., Inc. | |
| 119 | McDermott International Marine Investments N.V. | |
| 213 | McDermott International Marketing, Inc. | (formerly McDermott International Beijing, Inc.) |
| 130 | McDermott International Trading Co., Inc. | |
| 1971 | McDermott International Vessels, Inc. | |
| 100 | McDermott International, Inc. | |
| 1962 | McDermott Kft. | |
| 349 | McDermott Marine Construction Limited | |
| 1986 | McDermott Marine Mexico, S.A. de C.V. | |
| 135 | McDermott Offshore Services Company, Inc. | |
| 176 | McDermott Old JV Office, Inc. | |
| 162 | McDermott Overseas Investment Co. N.V. | |
| 390 | McDermott Overseas, Inc. | |
| 1966 | McDermott Panama Holdings, S.A. | |
| 170 | McDermott Servicos de Construcao, Ltda. | |
| 320 | McDermott Trade Corporation | |
| 306 | Mentor Subsea Technology Services, Inc. | |
| 138 | North Atlantic Vessel, Inc. | |
| | | |

| 2021 | North Ocean II AS | |
|------|--|--|
| | | |
| 2023 | North Ocean II KS | |
| 2022 | North Ocean V AS | |
| 612 | Offshore Pipelines International, Ltd. | |
| 618 | Offshore Pipelines Nigeria Limited | |
| 619 | Offshore Pipelines Sdn. Bhd. | |
| 626 | OPI Vessels, Inc. | |
| 623 | OPMI, Ltd. | |
| 185 | P. T. Bataves Fabricators | |
| 180 | P. T. McDermott Indonesia | |
| 901 | PT. Baja Wahana Indonesia | (formerly P.T. Babcock & Wilcox Indonesia) |
| 643 | PT. J. Ray McDermott Indonesia | (formerly PT. Jay Ray) |
| 636 | Sabine River Realty, Inc. | |
| 1977 | Servicios de Fabricacion de Altamira, S.A. de C.V. | |
| 1976 | Servicos Profesionales de Altamira, S.A. de C.V. | |
| 2019 | Singapore Huangdao Pte. Ltd. | |
| 640 | SparTEC, Inc. | (formerly J. Ray McDermott Spars, Inc.) |
| 145 | Varsy International N.V. | * |

EXHIBIT V ACE AFFILIATES

ACE American Insurance Company
ACE Fire Underwriters Insurance Company
ACE Indemnity Insurance Company
ACE Insurance Company of the Midwest
ACE Property and Casualty Insurance Company
Illinois Union Insurance Company
Indemnity Insurance Company of North America
Insurance Company of North America
Pacific Employers Insurance Company
Bankers Standard Insurance Company
Atlantic Employers Insurance Company

EXHIBIT VI -FOREIGN INSURANCE AGREEMENTS

| EFFECTIVE DATE/ YEAR | PARTIES | ТҮРЕ | PROGRAMS COVERED | POLICIES COVERED |
|----------------------------|--|---------------------------|---------------------|---------------------|
| 1985 | Mcdermott Int., Inc. and Insurance Co. of N.A. | High Deductible Agreement | General Liability | XWC11051 |
| | | | Auto Liability | ISGG05111444 |
| | | | | 92HF60007 |
| | | | | 92HF60006 |
| 1986 | Mcdermott Int., Inc. and Insurance Co. of N.A. | High Deductible Agreement | General Liability | XWC11075 |
| | | | Auto Liability | ISA480419 |
| | | | | OGLGO511414-7 |
| | | | | HDCGO511431-7 |

EXHIBIT VI (Continued)

| Name Of Agreement | Party To Agreement | Effective Date of Agreement |
|------------------------------------|---|--|
| Security Agreement | McDermott International, Inc and CIGNA Property and Casualty Insurance Company | April 1, 1992 |
| Security Agreement | McDermott International, Inc and CIGNA Property and Casualty Insurance Company | April 1, 1993 |
| Indemnity and Security Agreement | McDermott International, Inc. and CIGNA Insurance Company and Seguros CIGNA, S.A. | December 19, 1995 |
| Security Agreement | McDermott International, Inc. and CIGNA Insurance Company | April 1, 1996 |
| Pledge and Security Agreement | <u>Pledgor</u> : McDermott International, Inc. <u>Pledgees</u> :CIGNA Property and Casualty CIGNA Insurance Company | September 15, 1997 |
| Security Agreement | McDermott International, Inc. and ACE American Insurance Company | January 1, 2004 |
| INAMAR Claims Service Agreement | McDermott International, Inc. and INAMAR | Annual agreements incepting January 1, 2001 – December 31, 2005 |
| INAMAR Claims Service Agreement | McDermott International LLC, and Babcock and Wilcox and INAMAR | Effective with respect to Claims resulting from occurrences during the period beginning 12:00am January 1, 2006 |
| Security Agreement | McDermott International, Inc. and ACE American Insurance Company | January 1, 2006 |

EXHIBIT VII -FOREIGN POLICIES

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | Relating ACE Policy or Policies |
|--------------------------------------|------------------|--|--|---|---------------------------------------|
| 1 Jan 2007 - 31 December 2007 | B1003 | J. Ray McDermott Inc. and related subsidiary and affiliated companies. | General Automobile Workers Comp. | ACE American Ins. Co. ACE American Ins. Co. ACE American Ins. Co. | CMX043526 CMX043527 CMX043525 |
| 1 Jan 2007 - 31 December 2007 | B1004 | J. Ray McDermott Inc. and related subsidiary and affiliated companies. | General Workers Comp. | ACE American Ins. Co. ACE American Ins. Co. | CMX043526 CMX043525 |
| 1 Jan 2005 - 31 December 2005 | C1001 | J. Ray McDermott Inc. and related subsidiary and affiliated companies. | General Automobile Workers Comp. | ACE American Ins. Co. ACE American Ins. Co. ACE American Ins. Co. | CMX043411 CMX043412 CMX043410 |
| 1 Jan 2006 - 31 December 2006 | C1001 | J. Ray McDermott Inc. and related subsidiary and affiliated companies. | General Automobile Workers Comp. | ACE American Ins. Co. ACE American Ins. Co. ACE American Ins. Co. | CMX043526 CMX043527 CMX043525 |
| 1 Jan 2005 - 31 December 2005 | C1002 | J. Ray McDermott Inc. and related subsidiary and affiliated companies. | General Workers Comp. | ACE American Ins. Co. ACE American Ins. Co. | CMX043411 CMX043410 |
| 1 Jan 2006 - 31 December 2006 | C1002 | J. Ray McDermott Inc. and related subsidiary and affiliated companies. | General Workers Comp. | ACE American Ins. Co. ACE American Ins. Co. | CMX043411 CMX043410 |
| 1 January 1974 - 31 December 1974 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. General & Auto | CIGNA CIGNA | 50HF2295 45HF3277 |
| 1 January 1975 - 31 December 1975 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. General & Auto | CIGNA CIGNA | 50HF2296 45HF3278 |

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | RelatingACE Policy or Policies |
|--------------------------------------|------------------|---|----------------|--------------------|--------------------------------------|
| 1 January 1976 - 31 December 1976 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. | CIGNA | 50HF2296 |
| | | | General & Auto | CIGNA | 45HF3278 |
| 1 January 1977 - 31 December 1977 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. | CIGNA | 50HF2296 |
| | | | General & Auto | CIGNA | 45HF3278 |
| 1 January 1978 - 31 December 1978 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. | CIGNA | 50HF2296 |
| | | | General & Auto | CIGNA | 45HF3277 |
| 1 January 1979 - 31 December 1979 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. | CIGNA | 50HF2295 |
| | | | General & Auto | CIGNA | 45HF3277 |
| 1 January 1980 - 1 April 1981 | 23002 | Oceanic Contractors Inc. and any subsidiary, associated or affiliated companies as their interests may appear, as now or hereafter existing, acquired or constituted | Workers Comp. | CIGNA | 50HF2295 |
| | | McDermott International Inc effective March 14, 1980 | | | |
| | | | General & Auto | CIGNA | 45HF3277 |
| 1 April 1981 - | 23002 | McDermott International | Workers Comp. | CIGNA | 50HF2295 |

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | Relating ACE Policy or Policies |
|--------------------------------|------------------|--|--|---|---|
| 1 April 1982 | | Incorporated | General & Auto | CIGNA | 45HF3277 |
| 1 April 1982 - 1 April 1983 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA | 92HF0001 & ISG1163 |
| 1 April 1983 - 1 April 1984 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA | 92HF0001 & ISG351763 |
| 1 April 1984 - 1 April 1985 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA | 92HF0001 & ISG351764 |
| 1 April 1985 - 1 April 1986 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA | 92HF0001 & ISG351763 |
| 1 April 1986 - 1 April 1987 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | Aetna Insurance Company | 92AET-34599, 92AET-34598 & 28AET-41012 NO |
| 1 April 1987 - 1 April 1988 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | Aetna Insurance Company | 92AET-34599, 92AET-34598 & 28AET-41012 |
| 1 April 1988 - 1 April 1989 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | Aetna Insurance Company | 92AET-34599, 92AET-34598 & 28AET-41012 |
| 1 April 1989 - 1 April 1990 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX-12871, CMX-12870 & CMX-12871 |

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | Relating ACE Policy or Policies |
|--------------------------------|------------------|---|---|---|--|
| 1 April 1990 - 1 April 1991 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX-12871, CMX-12870 & CMX-12871 |
| 1 April 1991 - 1 April 1992 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX-12871, CMX-12870 & CMX-12871 |
| 1 April 1992 - 1 April 1993 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX-12871, CMX-12870 & CMX-12871 |
| 1 April 1993 - 1 April 1994 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043410, CMX043411& CMX043412 |
| 1 April 1994 - 1 April 1995 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043410, CMX043411& CMX043412 |
| 1 April 1995 - 1 April 1996 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043410, CMX043411& CMX043412 |
| 1 April 1996 - 1 April 1997 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043410, CMX043411& CMX043412 |
| 1 April 1997 - 1 April 1998 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043410, CMX043411& CMX043412 |

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | Relating ACE Policy or Policies |
|--------------------------------------|------------------|--|---|---|---------------------------------------|
| 1 April 1998 - 1 April 1999 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043410, CMX043411& CMX043412 |
| 1 April 1999 - 31 December 1999 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | CIGNA Property and Casualty Insurance Company | CMX043411, CMX043411& CMX043412 |
| 1 January 2000 - 31 December 2000 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | ACE American Ins. Co. | CMX043410, CMX043411& CMX043412 |
| 1 January 2001 - 31 December 2001 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | ACE American Ins. Co. | CMX043410, CMX043411& CMX043412 |
| 1 January 2002 - 31 December 2002 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | ACE American Ins. Co. | CMX043410, CMX043411& CMX043412 |
| 1 January 2003 - 31 December 2003 | 23002 | McDermott International Incorporated et al | Worker's Compensation / Employer's Liability / Comprehensive General and Comprehensive Automobile Liability | ACE American Ins. Co. | CMX043410, CMX043411& CMX043412 |
| 1 April 1986 - 1 April 1987 | 23010 | McDermott International Inc. et al | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | 92AET-34599 |
| | | | General Liability coverages Auto Liability | CIGNA CIGNA | 92AET-34598 28AET-41012 |
| 1 April 1987 - 1 April 1988 | 23010 | McDermott International Inc. et al | coverages worker's compensation, voluntary compensation of employer's | CIGNA | 87-88 92AET-34599 |
| | | | liability coverages General Liability Coverages | CIGNA | 92AET-34598 |

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | Relating ACE Policy or Policies |
|--------------------------------|------------------|------------------------------------|---|--------------------|---------------------------------------|
| 1 April 1988 - 1 April 1989 | 23010 | McDermott International Inc. et al | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | 92AET-41640 |
| | | | General Liability Coverages | CIGNA | 92AET-41639 |
| 1 April 1989 - 1 April 1990 | 23010 | McDermott International Inc. et al | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX-12871 |
| | | | General Liability Coverages | CIGNA | CMX-12869 |
| 1 April 1990 - 1 April 1991 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX-12871 |
| | | | General Liability Coverages | CIGNA | CMX-12869 |
| 1 April 1991 - 1 April 1992 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX-12871 |
| | | | General Liability Coverages | CIGNA | CMX-12869 |
| 1 April 1992 - 1 April 1993 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX-12871 |
| | | | General Liability Coverages | CIGNA | CMX-12869 |
| 1 April 1993 - 1 April 1994 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX043410 |
| | | | General Liability Coverages | CIGNA | CMX043411 |
| 1 April 1994 - 1 April 1995 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX043410 |
| | | | General Liability Coverages | CIGNA | CMX043411 |
| 1 April 1995 - 1 April 1996 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages | CIGNA | CMX043410 |

General Liability Coverages CIGNA CMX043411

| Effective Date | Policy Number | Named Insured | Type of Policy | Issuing Company | Relating ACE Policy or Policies |
|--------------------------------------|------------------|------------------------------|---|--|---------------------------------------|
| 1 April 1996 - 1 April 1997 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages General Liability | CIGNA | CMX043411 |
| 1 April 1997 - 1 April 1998 | 23010 | McDermott International Inc. | Coverages worker's compensation, voluntary compensation of employer's liability | CIGNA | CMX043410 |
| | | | coverages General Liability Coverages | CIGNA | CMX043411 |
| 1 April 1998 - 1 April 1999 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages General Liability Coverages | CIGNA CIGNA | CMX043410 CMX043411 |
| 1 April 1999 - 31 December 1999 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages General Liability Coverages | CIGNA | CMX043410 CMX043411 |
| 1 January 2000 - 31 December 2000 | 23010 | McDermott International Inc. | worker's compensation, voluntary compensation of employer's liability coverages General Liability Coverages | ACE American Ins. Co. ACE American Ins. Co. | CMX043410 CMX043411 |

GROUP II AGREEMENT- EXECUTION VERSION

NOVATION AND ASSUMPTION AGREEMENT

by and among

ACE American Insurance Company, acting for itself and its affiliates including, without limitation, Pacific Employers Insurance Company;

ACE INA Insurance Company;

ACE Insurance Company;

Insurance Company of North America

and

Creole Insurance Company, Ltd., a Bermuda company,

and

Boudin Insurance Company, Ltd., a Bermuda company

RECITALS

THIS NOVATION AND ASSUMPTION AGREEMENT (the "Agreement"), is entered into and effective as of May 18, 2010 (the "Effective Date") by and among ACE American Insurance Company, individually and acting for the ACE Affiliates (in such capacities, the "Company"), Creole Insurance Company, Ltd., a Bermuda company ("Engle"), and Boudin Insurance Company, Ltd., a Bermuda company ("Boudin").

WHEREAS, the Company and/or the ACE Affiliates have entered into the Existing Reinsurance Agreements with Creole and Boudin; and

WHEREAS, pursuant to the Existing Reinsurance Agreements, Creole and Boudin are obligated, among other things, to reinsure the Company and/or the ACE Affiliates with regard to certain Existing Policies; and

WHEREAS, Creole, prior to the Separation, is a wholly owned Subsidiary of MII; and

WHEREAS, MII intends to spin-off the B&W Entities (including Creole) from MII in connection with a dividend of common stock of B&W to the shareholders of MII (the "Separation"); and

WHEREAS, in connection with the Separation, the Parties wish to provide that Boudin be the reinsurer with respect to cessions arising from the operations, business, or property of MII and that Creole be the reinsurer with respect to cessions arising from the operations, business, or property of B&W, in each case whether such obligations were existing, accruing or arising before, on or after the Effective Date; and

WHEREAS, the Company, on its own behalf and on behalf of the ACE Affiliates, is willing to consent to the transfer, assumption, and novation of the matters as set forth herein, subject to the terms and conditions of this Agreement;

NOW, THEREFORE, in consideration of the mutual promises set out herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

1. <u>Definitions.</u> The following terms used herein, including in the recitals and Exhibits hereto, shall have the following meanings:

"ACE Affiliate" means each Affiliate of ACE that has issued an Existing Policy, including Pacific Employers Insurance Company, ACE INA Insurance Company, ACE Insurance Company and Insurance Company of North America.

"Affiliate" means, with respect to any specified Person, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, the specified Person. For this purpose "control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise.

"Agreement" has the meaning set forth in the recitals to this Agreement.

"ALAA" means the Assumption and Loss Allocation Agreement, dated as of the date hereof, among the Company, B&W, MII and certain other MII Entities and B&W Entities signatory thereto.

"Assumption Time" means midnight (New York time) on the Effective Date.

"B&W" means Babcock & Wilcox Holdings, Inc., a Delaware corporation, to be succeeded by The Babcock & Wilcox Company, a Delaware corporation, after the effective time of the merger, to occur after the date hereof, of Babcock & Wilcox Holdings, Inc., with and into The Babcock & Wilcox Company, the surviving entity of such merger.

"B&W Entity" means B&W and each of the entities listed on Exhibit I attached hereto and made a part hereof. It is acknowledged and understood that, from and after the effectiveness of the Separation, the B&W Entities will not be Subsidiaries or Affiliates of MII or any of the other MII Entities.

"Boudin Assumption and Novation" has the meaning set forth in Section 2(a).

"Boudin LOC" has the meaning set forth in Section 3(a).

"Company" has the meaning set forth in the recitals to this Agreement.

"Company B&W Obligation" means any obligation of the Company or an ACE Affiliate to or for the benefit of an Insured under an Existing Policy that arises, will arise, or has arisen from the operations, business, or property of a B&W Entity.

"Company MII Obligation" means any obligation of the Company or an ACE Affiliate to or for the benefit of an Insured under an Existing Policy that arises, will arise, or has arisen from the operations, business, or property of an MII Entity.

"Creole Assumption and Novation" has the meaning set forth in Section 2(c).

"Creole LOC" has the meaning set forth in Section 3(b).

"Effective Date" has the meaning set forth in the recitals to this Agreement.

"ESIS" means ESIS, Inc., an Affiliate of the Company.

"Existing Collateral" means any and all letters of credit or trust agreements outstanding as of the date hereof provided by or required to be provided by Creole or Boudin under the terms of any Existing Reinsurance Agreement in order to secure obligations arising thereunder.

"Existing Policy" means each policy of general liability insurance, automobile liability insurance, or workers compensation insurance issued prior to the date hereof by the Company or an ACE Affiliate to one or more B&W Entities and/or one or more MII Entities that is subject to an Existing Reinsurance Agreement.

"Existing Reinsurance Agreement" means each reinsurance agreement (whether denominated a treaty, a reinsurance policy, a reinsurance agreement, a facultative certificate, or otherwise) in which (a) Creole or Boudin is the reinsurer, (b) one or more of the Company and/or ACE Affiliates is or are the reinsureds, and (c) the ceded risk includes risk under any Existing Policy. "Existing Reinsurance Agreements" shall not mean or include any Existing Security Agreements (as defined below). Exhibit II attached hereto and made a part hereof reflects the Parties' best efforts to list all Existing Reinsurance Agreements, but the definitions in this Agreement shall control in the event of any errors or omissions on such Exhibit.

"Existing Security Agreements" shall mean and include any trust agreement, collateral agreement, pledge and security agreement or other similar contract between the Company or an ACE Affiliate and Boudin or between the Company or an ACE Affiliate and Creole, which was entered into in connection with or pursuant to any Existing Reinsurance Agreement (as defined above), and which are embodied in separately-executed written instruments.

"Go-Forward Boudin Obligations" means the obligations of Boudin under (a) the Transferable Boudin Reinsurance Agreements, (b) the Novated-to-Boudin Reinsurance Agreements and (c) the Wholly Retained Boudin Reinsurance Agreements, in each case after giving effect to the transfers, assumptions, novations, and releases effected by this Agreement.

"Go-Forward Creole Obligations" means the obligations of Creole under (a) the Transferable Creole Reinsurance Agreements, (b) the Novated-to-Creole Reinsurance Agreements and (c) the Wholly Retained Creole Reinsurance Agreements, in each case after giving effect to the transfers, assumptions, novations, and releases effected by this Agreement.

"Insured," as a noun in reference to one or more insurance policies, means any Person who is insured by such policy or policies, regardless of whether such Person is designated an "Insured" or a "Named Insured" in such policy or is otherwise expressly identified therein.

"Master Separation Agreement" means a Master Separation Agreement to be entered into between MII and The Babcock & Wilcox Company in connection with the Separation.

"MII" means McDermott International, Inc., a Panamanian corporation.

"MII Entity" means MII and each of the entities listed on Exhibit III attached hereto and made a part hereof. It is acknowledged and understood that, from and after the effectiveness of the Separation, the MII Entities will not be Subsidiaries or Affiliates of B&W or any of the other B&W Entities.

"Novated Reinsurance Agreement" means a Novated-to-Boudin Reinsurance Agreement or a Novated-to-Creole Reinsurance Agreement.

"Novated-to-Boudin Reinsurance Agreement" means, as to any Transferable Creole Reinsurance Agreement after the Assumption Time and giving effect to this Agreement, the aggregate of (a) all rights, duties, and obligations of Boudin to and in respect of the Company under such Transferable Creole Reinsurance Agreement as and to the extent novated to Boudin, and (b) all rights, duties, and obligations of the Company to and in respect of Boudin under such Transferable Creole Reinsurance Agreement as and to the extent novated to Boudin.

"Novated-to-Creole Reinsurance Agreement" means, as to any Transferable Boudin Reinsurance Agreement after the Assumption Time and giving effect to this Agreement, the aggregate of (a) all rights, duties, and obligations of Creole to and in respect of the Company under such Transferable Boudin Reinsurance Agreement as and to the extent novated to Creole, and (b) all rights, duties, and obligations of the Company to and in respect of Creole under such Transferable Boudin Reinsurance Agreement as and to the extent novated to Creole.

"Organizational Documents" means (a) with respect to any corporation, its certificate or articles of incorporation or organization and its bylaws, (b) with respect to any limited partnership, its certificate of limited partnership agreement, (c) with respect to any general partnership, its partnership agreement, and (d) with respect to any limited liability company, its certificate or articles of formation or organization and its operating agreement or other organizational documents.

"Parties" means the Company, Creole and Boudin, collectively (and each individually is a "Party").

"Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, a union, an unincorporated organization or a governmental entity or any department, agency or political subdivision thereof.

"Separation" has the meaning set forth in the recitals to this Agreement.

"Subsidiary" means, with respect to any specified Person, any corporation, partnership, limited liability company, joint venture or other organization, whether incorporated or unincorporated, of which at least a majority of the securities or interests having by the terms thereof ordinary voting power to elect at least a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such specified Person or by any one or more of its Subsidiaries, or by such specified Person and one or more of its Subsidiaries.

"Substituted Collateral" means the MII LOC and the B&W LOC.

"Transferable Boudin Reinsurance Agreements" means those Existing Reinsurance Agreements in which Boudin is the reinsurer and that reinsure, in whole or in part and whether or not including other obligations, Company B&W Obligations.

"Transferable Creole Reinsurance Agreements" means those Existing Reinsurance Agreements in which Creole is the reinsurer and that reinsures, in whole or in part and whether or not including other obligations, Company MII Obligations.

"Wholly Retained Boudin Reinsurance Agreement" means an Existing Reinsurance Agreement in which Boudin is the reinsurer and that is not a Transferable Boudin Reinsurance Agreement.

"Wholly Retained Creole Reinsurance Agreement" means an Existing Reinsurance Agreement in which Creole is the reinsurer and that is not a Transferable Creole Reinsurance Agreement.

2. Assumption and Novation.

- (a) <u>Boudin Assumption and Novation</u>. Notwithstanding anything in any Transferable Creole Reinsurance Agreement to the contrary, and effective as of the Assumption Time, Creole hereby transfers and assigns, and Boudin hereby assumes by novation, so much of each Transferable Creole Reinsurance Agreement as reinsures any Company MII Obligation. In connection with such transfer, assignment, and novation:
 - (i) Boudin hereby agrees to observe, pay, perform, satisfy, fulfill and discharge, to the extent and in the manner required under the applicable Transferable Creole Reinsurance Agreement, any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of Creole under the Transferable Creole Reinsurance Agreement with respect to the Company MII Obligations insofar as transferred above (the "Boudin Assumption and Novation"); and
 - (ii) The Company and each ACE Affiliate hereby consent to, and agree to give full force and effect to, the Boudin Assumption and Novation. From and after the Assumption Time, Boudin and not Creole shall be treated as the Company's (or applicable ACE Affiliate's) contractual counterparty with respect the contracts and mutual rights and obligations subject to the Boudin Assumption and Novation. Without limitation, the Company and each ACE Affiliate, as applicable:
 - a may enforce against Boudin its rights with respect to the Company MII Obligations under the Transferable Creole Reinsurance Agreements to the same extent such Person could, prior to the Boudin Assumption and Novation, enforce such rights against Creole, and
 - b shall perform for the benefit of Boudin any obligation with respect to the Company MII Obligations under the Transferable Creole Reinsurance Agreements to the same extent such Person was obligated, prior to the Boudin Assumption and Novation, to perform such obligations for the benefit of Creole, and
 - c releases Creole from its obligation to observe, pay, perform, satisfy, fulfill or discharge any obligations under any Transferable Creole Reinsurance Agreements with respect to any Company MII Obligation.
- (b) No Transfer or Novation of Creole Obligations arising from B&W Operations The Wholly Retained Creole Reinsurance Agreements are not novated or otherwise affected by the Boudin Assumption

and Novation. The Transferable Creole Reinsurance Agreements are novated to Boudin as set forth above only to the extent that they reinsure Company MII Obligations. To the extent that the Transferable Creole Reinsurance Agreements reinsure Company B&W Obligations, the Parties acknowledge that Creole and not Boudin shall continue to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of the reinsurer under the Transferable Creole Reinsurance Agreements.

- (c) <u>Creole Assumption and Novation</u>. Notwithstanding anything in any Transferable Boudin Reinsurance Agreement to the contrary, and effective as of the Assumption Time, Boudin hereby transfers and assigns, and Creole hereby assumes by novation so much of each Transferable Boudin Reinsurance Agreement as reinsures any Company B&W Obligation. In connection with such transfer and assignment,
 - (i) Creole hereby agrees to observe, pay, perform, satisfy, fulfill and discharge, to the extent and in the manner required under the applicable Transferable Boudin Reinsurance Agreement, any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of Boudin under the Transferable Boudin Reinsurance Agreements with respect to the Company B&W Obligations insofar as transferred above (the "Creole Assumption and Novation"); and
 - (ii) The Company and each ACE Affiliate hereby consent to, and agree to give full force and effect to, the Creole Assumption and Novation. From and after the Assumption Time, Creole and not Boudin shall be treated as the Company's (or applicable ACE Affiliate's) contractual counterparty with respect the contracts and mutual rights and obligations subject to the Creole Assumption and Novation. Without limitation, the Company and each ACE Affiliate, as applicable:
 - a may enforce against Creole its rights under the Transferable Boudin Reinsurance Agreements to the same extent such Person could, prior to the Creole Assumption and Novation, enforce such rights against Boudin, and
 - b shall perform for the benefit of Creole any obligation under the Transferable Boudin Reinsurance Agreements to the same extent such Person was obligated, prior to the Creole Assumption and Novation, to perform such obligations for the benefit of Boudin, and
 - c releases Boudin from its obligation to observe, pay, perform, satisfy, fulfill or discharge any obligations under any Transferable Boudin Reinsurance Agreements with respect to any Company B&W Obligation.
- (d) No Transfer or Novation of Boudin Obligations arising from MII Operations The Wholly Retained Boudin Reinsurance Agreements are not novated or otherwise affected by the Creole Assumption and Novation. The Transferable Boudin Reinsurance Agreements are novated to Creole as set forth above only to the extent that they reinsure Company B&W Obligations. To the extent that the Transferable Boudin Reinsurance Agreements reinsure Company MII Obligations, the Parties acknowledge that Boudin and not Creole shall continue to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of the reinsurer under the Transferable Boudin Reinsurance Agreements.
- (e) No Effect on Aggregate Limits of Liability. For the avoidance of doubt, it is understood and agreed that the aggregate liability of Creole and Boudin, taken together, under the Existing Reinsurance Agreements (or any of them) is not intended to be, and shall be deemed not to be, increased by implementation of this Agreement. In particular, and without limitation, to the extent that any Transferable Creole Reinsurance Agreement (or Transferable Boudin Reinsurance Agreement) contains an aggregate limit of liability, that aggregate limit of liability shall, after the Assumption Time, apply as a single, joint aggregate limit of liability as between (i) the resulting Novated Reinsurance Agreement and (ii) the portions of the Transferable Creole Reinsurance Agreement (or Transferable Boudin Reinsurance Agreement) that are retained pursuant to Section 2(b) (or Section 2(d)) above. Neither Creole or Boudin shall have any further liability under a Novated Reinsurance Agreement upon actual exhaustion by payment of the single, joint aggregate limit of liability thereunder; provided, however, that each of Creole and Boudin agree that

the Company may determine the order in which any such single, joint aggregate limit of liability may be payable by each of them under such Novated Reinsurance Agreement, and provided further that (x) neither Creole or Boudin may refuse to pay any amount due to the Company on the basis of any claim or contention that in determining the order in which such limits are to be paid thereunder, the Company has not acted in good faith or has acted improperly, and (y) neither Creole nor Boudin may assert any such claim or contention as a defense to liability or payment or otherwise.

3. Collateral.

(a) Boudin LOC.

- (i) Boudin (including for purposes of this Section 3, and if MII and Boudin so elect, MII on Boudin's behalf) will, within fifteen (15) days after the Effective Date, provide to the Company, as beneficiary thereof, an irrevocable letter of credit (the "Boudin LOC") in an amount of \$696.00 (which amount shall be less than the aggregate amount of the Existing Collateral), issued in a form and by a bank or other financial institution, in each case acceptable to the Company; and/or such other forms of collateral as the Company may permit in writing from time to time. The Boudin LOC shall secure the Go-Forward Boudin Obligations.
- (ii) The Boudin LOC shall be "evergreen," meaning that it shall provide by its terms that it will be renewed automatically each year for an additional year unless written notice of non-renewal is received by the Company at least sixty (60) days prior to the Boudin LOC's anniversary date. If the Company permits Boudin to provide collateral in a form other than the Boudin LOC, Boudin shall provide such collateral in an amount and form acceptable to the Company.
- (iii) Boudin shall continue to provide the Boudin LOC (or other collateral acceptable to the Company) as security for payment of the Go-Forward Boudin Obligations, until the Company determines that there is no longer any need for such collateral. If there shall be a material deterioration in the financial condition of the bank or other financial institution which has issued the Boudin LOC, the Company shall have the right to require Boudin to replace the Boudin LOC with a new letter of credit with similar terms issued by a bank or other financial institution then acceptable to the Company.
- (iv) The Company shall have the right to draw against the Boudin LOC and/or other collateral solely (a) in accordance with the terms of the applicable Wholly Retained Boudin Reinsurance Agreement or Novated-to-Boudin Reinsurance Agreement, as the case may be, and/or as required and permitted by the laws and regulations of the Commonwealth of Pennsylvania, or (b) in the event that a notice of nonrenewal is received pursuant to the evergreen clause.
- (v) Annually, the Company shall review and redetermine the amount of the Go-Forward Boudin Obligations and the amount of collateral security required pursuant to this Agreement. At such time, MII will provide its most recent audited financial statements, interim financial statements, and any other financial information reasonably requested by the Company for the purpose of evaluating the financial condition of MII. MII will provide any needed increases in the amount of the Boudin LOC (and/or other collateral if acceptable to the Company) within thirty (30) days of the Company's written request for any additional required amount of the Boudin LOC. The Company will effect any decreases in the amount of the Boudin LOC (and/or other collateral) promptly, provided that Boudin is not in breach of any of its obligations under this Agreement or the Existing Reinsurance Agreements as transferred and novated hereunder and MII is not in breach of any of its obligations to the Company under the ALAA.

(b) Creole LOC

(i) Creole (including for purposes of this Section 3, and if B&W and Creole so elect, B&W on Creole's behalf) will, within fifteen (15) days after the Effective Date, provide to the Company, as beneficiary thereof, an irrevocable letter of credit (the "Creole LOC") in an amount of \$4,171,230.00 (which amount shall be less than the aggregate amount of the Existing Collateral), issued in a form and

by a bank or other financial institution, in each case acceptable to the Company; and/or such other forms of collateral as the Company may permit in writing from time to time. The Creole LOC shall secure the Go-Forward Creole Obligations.

- (ii) The Creole LOC shall be "evergreen," meaning that it shall provide by its terms that it will be renewed automatically each year for an additional year unless written notice of non-renewal is received by the Company at least sixty (60) days prior to the Creole LOC's anniversary date. If the Company permits Creole to provide collateral in a form other than the Creole LOC, Creole shall provide such collateral in an amount and form acceptable to the Company.
- (iii) Creole shall continue to provide the Creole LOC (or other collateral acceptable to the Company) as security for payment of the Go-Forward Creole Obligations, until the Company determines that there is no longer any need for such collateral. If there shall be a material deterioration in the financial condition of the bank or other financial institution which has issued the Creole LOC, the Company shall have the right to require Creole to replace the Creole LOC with a new letter of credit with similar terms issued by a bank or other financial institution then acceptable to the Company.
- (iv) The Company shall have the right to draw against the Creole LOC and/or other collateral solely (a) in accordance with the terms of the applicable Wholly Retained Creole Reinsurance Agreement or Novated-to-Creole Reinsurance Agreement, as the case may be, and/or as required and permitted by the laws and regulations of the Commonwealth of Pennsylvania, or (b) in the event that a notice of nonrenewal is received pursuant to the evergreen clause.
- (v) Annually, the Company shall review and redetermine the amount of the Go-Forward Creole Obligations and the amount of collateral security required pursuant to this Agreement. At such time, B&W will provide its most recent audited financial statements, interim financial statements, and any other financial information reasonably requested by the Company for the purpose of evaluating the financial condition of B&W. B&W will provide any needed increases in the amount of the Creole LOC (and/or other collateral if acceptable to the Company) within thirty (30) days of the Company's written request for any additional required amount of the Creole LOC. The Company will effect any decreases in the amount of the Creole LOC (and/or other collateral) promptly, provided that Creole is not in breach of any of its obligations under this Agreement or the Existing Reinsurance Agreements as transferred and novated hereunder and B&W is not in breach of any of its obligations to the Company under the ALAA.
- (c) Existing Security Agreements: Substituted Collateral. Notwithstanding anything in any Existing Reinsurance Agreement or any Existing Security Agreement to the contrary, the Company shall release the Existing Collateral (including, without limitation, providing the necessary directions to issuing banks of letters of credit, trustees of reinsurance trusts, and/or similar third parties) upon receipt of the Substituted Collateral as set forth in Section 3(a)(i) and 3(b)(i), and the Parties shall take all actions and execute any documents necessary to terminate the Existing Security Agreements.
- (d) Adjustments of Funds Withheld. To the extent that the Company has retained funds of Creole or Boudin as funds withheld under any Existing Reinsurance Agreement (including, without limitation, by allocating reinsurance premium or other funds of the reinsurers to claim payment funds held by ESIS), such funds shall continue to be maintained following the transactions contemplated by this Agreement. If the Company desires to effect a one-time reallocation of such funds withheld as between Creole and Boudin as a result of the transactions contemplated by this Agreement, it shall provide notice of such reallocation as promptly as possible after the Assumption Time, but in any event on or prior to October 1, 2010, and Creole and Boudin shall accept the Company's determination of the amount of such reallocation.
- 4. Existing Collateral. The Substituted Collateral required to be provided by Creole and Boudin hereunder shall, except to the extent provided otherwise in this Agreement, be subject to all of the terms and conditions applicable to the Existing Collateral pursuant to the Existing Reinsurance Agreements to the same extent that such terms and conditions applied to the Existing Collateral thereunder.

5. Allocation.

- (a) <u>Company Designations</u>. The Company's determination as to whether an obligation is a Go-Forward Boudin Obligation or a Go-Forward Creole Obligation (a "<u>Company Designation</u>") shall be binding on Creole and Boudin, and they shall not delay, or make any deduction with respect to, their payment or other response to such obligation on account of any disagreement with such determination, <u>provided</u>, <u>however</u>, that such payment or other response shall not be construed as prejudicial to either Party in any dispute between Creole and Boudin with respect to any such Company Designation.
- (b) <u>Disputes</u>. Any dispute or disagreement between Creole and Boudin with respect to the correctness of a Company Designation shall be resolved pursuant to Article V of the Master Separation Agreement by MII on behalf of Boudin and B&W on behalf of Creole; <u>provided, that</u> (i) the Company will not be made a party to any arbitration proceeding arising from such dispute or disagreement, but may be called as a witness; (ii) any reasonable costs incurred by the Company in respect of any such arbitration proceeding will be fully reimbursed to the Company promptly following receipt of a reimbursement demand from the Company; (iii) under no circumstances will MII or B&W or Creole or Boudin, as a result of such arbitration proceeding, require the Company to return any amount received by the Company pursuant to a prior Company Designation, whether such amount was received as a result of the Company's draw against security posted for its benefit or otherwise, and (iv) the Company shall comply with the allocation or other resolution of such dispute.
- 6. <u>Billing</u>. On and after the Effective Date, the Company will (in each case in accordance with the billing procedures set forth in the applicable Existing Reinsurance Agreement)
 - (a) bill Creole directly for such of the Go-Forward Creole Obligations as are then due and payable, and provide Creole with appropriate reports and accounting with respect to such obligations; and
 - (b) bill Boudin directly for such of the Go-Forward Boudin Obligations as are then due and payable, and provide Boudin with appropriate reports and accounting with respect to such obligations.
- 7. <u>Amendments</u>. Neither this Agreement nor any provision hereof may be amended, changed, waived, discharged or terminated except by a written instrument signed by each Party.
- 8. <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Neither this Agreement nor any right or obligation hereunder may be assigned or conveyed by any Party without the prior written consent of the other Parties, which consent shall not be unreasonably withheld.
- 9. No Waiver. The failure or refusal by any Party to exercise any rights granted hereunder shall not constitute a waiver of such rights or preclude the subsequent exercise thereof, and no oral communication shall be asserted as a waiver of any such rights hereunder unless such communication shall be confirmed in a writing plainly expressing an intent to waive such rights and signed by the Party against whom such waiver is asserted.
- 10. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts each of which when so executed and delivered shall constitute an original, but such counterparts together shall constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile transmission shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of the Parties and other Persons signatory hereto transmitted by facsimile shall be deemed to be their original signatures for all purposes.
- 11. No Third Party Beneficiary. This Agreement shall not be deemed to give any right or remedy to any third party whatsoever unless otherwise specifically granted hereunder.

- 12. Parties' Representations. As of the Effective Date, each of the Parties expressly represents on its own behalf: (a) it is an entity in good standing in its jurisdiction of organization; (b) it has all requisite corporate power and authority to enter into this Agreement, and to perform its obligations hereunder; (c) the execution and delivery by it of this Agreement, and the performance by it of its obligations under this Agreement, have been duly authorized by all necessary corporate or other action; (d) this Agreement, when duly executed and delivered by it, and subject to the due execution and delivery hereof by the other Parties, will be a valid and binding obligation of it, enforceable against it, its successors and permitted assigns, in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and general equity principles; (e) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby in accordance with the respective terms and conditions hereof will not (i) violate any provision of its Organizational Documents, (ii) violate any applicable order, judgment, injunction, award or decree of any court, arbitrator or governmental or regulatory body against it, or binding upon it, or any agreement with, or condition imposed by, any governmental or regulatory body, foreign or domestic, binding upon it as of the date hereof, or (iii) violate any agreement, contract, obligation, promise or undertaking that is legally binding and to which it is a party or by which it is bound; and (f) the signatory hereto on behalf of it is duly authorized and legally empowered to enter into this Agreement on its behalf.
- 13. Notices. Any and all notices, requests, approvals, authorizations, consents, instructions, designations and other communications that are required or permitted to be given pursuant to this Agreement shall be in writing and may be given either by personal delivery, first class prepaid post (airmail if to another country) or by internationally recognized overnight delivery service to the following address, or to such other address and recipient as such Party may have notified in accordance with the terms of this section as being its address or recipient for notification for the purposes of this Agreement:

If to the Company ACE American Insurance Company

225 E. John Carpenter Freeway, Suite 1300

Irving, TX 75062

Attention: Underwriting Manager

ACE Risk Management

Telephone: (972) 465,7500 Facsimile: (972) 465,7826

If to Boudin: Boudin Insurance Company, Ltd., c/o McDermott International, Inc.

757 N. Eldridge Parkway

Houston, Texas 77079

VP and Chief Risk Officer (with copy to General Counsel) Attention:

Telephone: 281-870-5785 281-870-5923 Telecopier:

Electronic Mail: twoodard@mcdermott.com

with a copy to:

Boudin Insurance Company, Ltd.

Cedar House 41 Cedar Avenue P.O. Box HM 1838 Hamilton HM HX Bermuda Bermuda

Fax (441) 295-3982

Telephone: 281-870-5476 Telecopier: 281-870-5923

Electronic Mail: cjryan@mcdermott.com If to any MII Entity: McDermott International, Inc.

757 N. Eldridge Parkway Houston, Texas 77079

Attention: VP and Chief Risk Officer (with copy to General Counsel)

Telephone: 281-870-5785 281-870-5923 Telecopier:

Electronic Mail: twoodard@mcdermott.com

If to Creole: Creole Insurance Company, Ltd.

Prior to the Separation:

Creole Insurance Company, Ltd., c/o Babcock & Wilcox Holdings, Inc.

800 Main Street

Lynchburg, Virginia 24504

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 434-522-6800

with a copy to:

McDermott International, Inc. 777 N. Eldridge Parkway Houston, Texas 77079

Director, Risk Management (with copy to General Counsel) Attention:

with an additional copy to: Creole Insurance Company, Ltd.

Cedar House 41 Cedar Avenue P.O. Box HM 1838 Hamilton HM HX Bermuda Bermuda

Fax (441) 295-3982

Telephone: 281-870-5476 Telecopier: 281-870-5923 Electronic Mail: cjryan@mcdermott.com

On and After the Separation:

The Babcock & Wilcox Company

The Harris Building

13024 Ballantyne Corporate Place, Suite 700

Charlotte, North Carolina

Attention: Director, Risk Management (with copy to General Counsel)

with a copy to:

Creole Insurance Company, Ltd.

Cedar House 41 Cedar Avenue P.O. Box HM 1838 Hamilton HM HX Bermuda

Bermuda

Fax (441) 295-3982

Telephone: 281-870-5476 Telecopier: 281-870-5923 Electronic Mail: cjryan@mcdermott.com

If to any B&W Entity: Prior to the Separation:

Babcock & Wilcox Holdings, Inc. (f/k/a The Babcock & Wilcox Company)

800 Main Street

Lynchburg, Virginia 24504

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 434-522-6800

with a copy to:

McDermott International, Inc. 777 N. Eldridge Parkway Houston, Texas 77079

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 281-870-5476
Telecopier: 281-870-5923
Electronic Mail: cjryan@mcdermott.com

On and After the Separation:

The Babcock & Wilcox Company

The Harris Building

13024 Ballantyne Corporate Place, Suite 700

Charlotte, North Carolina

Attention: Director, Risk Management (with copy to General Counsel)

Any notice or communication to any Person shall be deemed to be received by that Person:

(A) upon personal delivery; or

(B) upon receipt if sent by mail or courier.

- 14. Governing Law. This Agreement shall be governed by and construed in accordance with the internal laws of the Commonwealth of Pennsylvania without regard to those provisions concerning conflicts of laws that would result in the application of the laws of any other jurisdiction.
- 15. Entire Agreement. THIS AGREEMENT CONSTITUTES THE ENTIRE AGREEMENT AMONG ALL OF THE PARTIES WITH RESPECT TO THE TRANSFERS, ASSUMPTIONS, AND NOVATIONS DESCRIBED HEREIN AND SUPERSEDES ALL OTHER PRIOR AGREEMENTS AND UNDERSTANDINGS, BOTH WRITTEN AND ORAL, WITH RESPECT TO SUCH TRANSFERS, ASSUMPTIONS, AND NOVATIONS. SOLELY FOR INTERPRETATION PURPOSES, THE PARTIES ACKNOWLEDGE THAT THIS AGREEMENT IS INTENDED TO BE READ TOGETHER WITH THE ALAA.

16. Dispute Resolution.

(a) For dispute resolution purposes as between Creole and Boudin, and all disputes between Creole and Boudin arising out of this Agreement, including without limitation as to whether a given obligation is a Go-Forward Creole Obligation or a Go-Forward Boudin Obligation, shall be resolved in accordance with the procedures set forth in the Master Separation Agreement (provided, however, that until the Master Separation Agreement is executed and delivered by all parties thereto, the reference to such agreement in this Section 16(a) shall be deemed to refer to the draft thereof dated as of April 28, 2010).

- (b) As between the Company on the one hand and Creole and/or Boudin, on the other hand, all disputes arising hereunder shall be resolved in accordance with the arbitration provisions of the most recent Existing Reinsurance Agreement. In any such arbitration, the entity named in the applicable Company Designation shall be the party formally opposed to the Company, but the other of Boudin or Creole shall have right to associate effectively in the defense and/or prosecution of such arbitration.
- 17. Severability. If any term or other provision of this Agreement or the Exhibits attached hereto is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the court, administrative agency or arbitrator shall interpret this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible. If any sentence in this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- 18. Rules of Construction. The definitions of terms used herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include", "includes" and "including" shall be deemed to be followed by the phrase "without limitation". The word "will" shall be construed to have the same meaning and effect as the word "shall". Unless the context requires otherwise (a) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified, (b) any reference herein to any law shall be construed as referring to such law as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, (c) any reference herein to any Person shall be construed to include such Person's successors and assigns (subject to the restrictions contained herein), (d) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this Agreement in its entirety and not to any particular provision hereof, (e) with respect to the determination of any time period, the word "from" means "from and including" and the word "to" means "to and including" and (f) any reference herein to Articles, Sections and Exhibits shall be construed to refer to Articles and Sections of, and Exhibits to, this Agreement. No provision of this Agreement shall be interpreted or construed against any Person solely because such Person or its legal representative drafted such provision.

[Remainder of Page Intentionally Left Blank]

| IN WITNESS WHEREOF | the Parties intending to I | be legally bound hereby have executed this | Agreement, by their duly authorized representatives. |
|----------------------|------------------------------|--|--|
| IN WILINGSS WHEKEUE. | , the Parties intending to i | be regariv bound hereby have executed this | Agreement, by their duty authorized representatives. |

ACE AMERICAN INSURANCE COMPANY

| By: | /s/ Laura Vest |
|---------------|----------------|
| Printed Name: | Laura Vest |
| Title: | Vice President |

| Consented to f | for Purposes of Section 3: |
|----------------|----------------------------|
| MCDERMOT | TT INTERNATIONAL INC. |
| By: | /S/ LIANE K. HINRICHS |
| Name: | Liane K. Hinrichs |
| Title: | Senior Vice President |
| BABCOCK & | & WILCOX HOLDINGS, INC. |
| Ву: | /s/ MICHAEL S. TAFF |
| Name: | Michael S. Taff |
| Title: | Senior Vice President |

CREOLE INSURANCE COMPANY, LTD.

BOUDIN INSURANCE COMPANY, LTD.

By: /S/ LIANE K. HINRICHS

/S/ BENJAMIN H. BASH
Benjamin H. Bash
Assistant Secretary

Liane K. Hinrichs Assistant Secretary

EXHIBIT I – B&W ENTITIES

See Schedule 1.1(a) and Schedule 1.1(c) attached hereto. No B&W Entity listed on Schedule 1.1(a) shall be deemed to be an "Insured", a "Named Insured" or otherwise be deemed to be insured under any Existing Policy or Existing Reinsurance Agreement solely by virtue of being listed on such Schedule 1.1(a).

Schedule 1.1(a) B&W FORMER

| Reference ID | Name |
|--------------|---|
| 333 | Ahahsain Hudson Heat Transfer Co. Ltd. |
| 398 | Advanced Refractory Technologies, Inc. |
| | A.M. Lockett & Co., Limited |
| | Amcermet Corporation |
| 924 | ASEA Babcock |
| 235 | Ash Acquisition Company |
| 326 | B & W Clarion, Inc. |
| 574 | B&W Ebensburg Pa., Inc. |
| 460 | B&W Energy Investments, Inc. |
| 383 | B&W Fort Worth Power, Inc. |
| 950 | B&W Fuel Company |
| 535 | B&W Fuel, Inc. |
| 922 | B&W Mexicana, S.A. de C.V. |
| 9991 | B&W North Branch G.P., Inc. |
| 9990 | B&W North Branch L.P., Inc. |
| 537 | B&W Nuclear Service Company |
| 960 | B&W Nuclear Service Company |
| 536 | B&W Nuclear, Inc. |
| 586 | B&W Saba, Inc. |
| 591 | B&W Service Company |
| 579 | B&W SOFC G.P., Inc. |
| 578 | B&W SOFC L.P., Inc. |
| 381 | B&W Special Projects, Inc. |
| 569 | B&W Triso Corporation |
| | B&W Tubular Products Limited |
| 573 | B&W/OHM Weldon Spring, Inc. |
| 212 | Babcock & Wilcox Asia Investment Co., Inc. |
| 115 | Babcock & Wilcox Asia Limited |
| 533 | Babcock & Wilcox Canada Leasing Ltd. |
| 503 | Babcock & Wilcox Canada Ltd. |
| 215 | Babcock & Wilcox China Investment Co., Inc. |
| 594 | Babcock & Wilcox do Brasil Limitada |
| 528 | Babcock & Wilcox do Brasil Participacoes Limitada |
| 206 | Babcock & Wilcox Egypt SAE |
| | |

| 169 | Babcock & Wilcox Fibras Ceramicas Limitada |
|------|---|
| 519 | Babcock & Wilcox Finance, Inc. |
| 557 | Babcock & Wilcox Foreign Sales Corporation |
| 175 | Babcock & Wilcox Gama Kazan Teknolojisi A.S. |
| 552 | Babcock & Wilcox General Contracting Company |
| 565 | Babcock & Wilcox Government Services Company |
| 395 | Babcock & Wilcox HRSG Company |
| | Babcock & Wilcox Industries, Ltd. |
| 531 | Babcock & Wilcox International Sales Corporation |
| 549 | Babcock & Wilcox International Sales Corporation |
| 342 | Babcock & Wilcox Investment Company |
| 305 | Babcock & Wilcox Jonesboro Power, Inc. |
| 2007 | Babcock & Wilcox Nevada, LLC |
| | Babcock & Wilcox Refractories Limited |
| 323 | Babcock & Wilcox Salt City Power, Inc. |
| 543 | Babcock & Wilcox Services, Inc. |
| 322 | Babcock & Wilcox Tracy Power, Inc. |
| 314 | Babcock & Wilcox Victorville Power, Inc. |
| 727 | Babcock & Wilcox Volund France SAS |
| 315 | Babcock PFBC, Inc. |
| 559 | Babcock Southwest Construction Corporation |
| 945 | Babcock-Brown Boveri Reaktor GmbH |
| 936 | Babcock-Ultrapower Jonesboro |
| 937 | Babcock-Ultrapower West Enfield |
| 951 | Bailey Beijing Controls Co., Ltd. |
| 516 | Bailey Controls Australia Pty. Limited |
| 517 | Bailey Controls International Sales & Services Company, Inc |
| 954 | Bailey Controls Jordan for Process Controls Services, Ltd. |
| 563 | Bailey Controls Sales & Service (Australia) Pty. Limited |
| 564 | Bailey Controls Sales & Services Canada Inc. |
| 561 | Bailey do Brasil Instrumentos Industriais Limitada |
| 114 | Bailey International, Inc. |
| 923 | Bailey Japan Company Limited |
| 542 | Bailey Meter and Controls Company |
| | Bailey Meter Company |
| 562 | Bailey Meter Co. (Japan) Ltd. |
| | Bailey Meter Company Limited |
| | Bailey Meter GmbH |
| | |

| 646 | Brick Insurance Company, Ltd. |
|------|--|
| 590 | BWXT Protec, Inc. |
| | C.C. Moore & Company Engineers |
| 511 | Ceramatec G.P., Inc. |
| 510 | Ceramatec SOFC, Inc. |
| 329 | Clarion Energy, Inc. |
| 328 | Clarion Power Company |
| 321 | Conam Nuclear, Inc. |
| | Control Components France |
| 514 | Control Components Italy S.R.L. |
| | Control Components, Inc. (California) |
| | Control Components, Inc. (Delaware) |
| 948 | Control Components Japan |
| 1914 | CTR Solutions, LLC |
| 545 | Detroit Broach & Machine Corporation |
| 551 | Diamond Blower Company Limited |
| | Diamond Canapower Ltd. |
| 518 | Diamond Power Importação e Exportação Ltda. |
| 144 | Diamond Power Korea Inc. |
| 526 | Diamond Power Specialty (Japan) Ltd. |
| 558 | Diamond Power Specialty (Proprietary) Limited |
| 546 | Diamond Power Specialty Corporation (Delaware) |
| | Diamond Power Speciality Corporation (Ohio) |
| 529 | Diamond Power Specialty GmbH |
| | Diescher Tube Mills, Inc. |
| 332 | Ebensburg Energy, Inc. |
| 397 | Ejendomsaktieselskabet Falkevej2 |
| 961 | Enserch Environmental Management Company, Inc. |
| 968 | EPC Business Trust |
| 919 | Especialidades Termomecanicas, S.A. de C.V. |
| 550 | Ferry-Diamond Engineering Company Limited |
| 928 | Fibras Ceramicas C.A. |
| 509 | Fibras Ceramicas, Inc. |
| 547 | Globe Steel Tubes Corporation |
| | Greer Land Co. |
| | Holmes Insulations Limited |
| 124 | Hudson Heat Transfer International, Inc. |
| 1955 | Hudson HEI Pty. Ltd. |
| | |

| 914 | Hudson Northern Industries Inc. |
|------|--|
| 125 | Hudson Products Aktiebolag |
| 567 | Hudson Products Corporation (1) |
| 902 | Hudson Products de Mexico, S.A. de C.V. |
| 1907 | Integran Technologies Inc. |
| 461 | International Disarmament Corporation |
| 941 | Isolite Babcock Refractories Company, Ltd. |
| 927 | Isolite Eastern Union Refractories Co., Ltd. |
| 920 | KBW Gasification Systems, Inc. |
| 512 | LT Produkter i Skutskar AB |
| 938 | Maine Power Services |
| 345 | McDermott Heat Transfer Company |
| 344 | McDermott Productos Industriales de Mexico, S.A. de C.V. |
| 946 | Medidores Bailey, S.A. de C.V. |
| 942 | Morganite Ceramic Fibres Limited |
| 943 | Morganite Ceramic Fibres Pty. Limited |
| 944 | Morganite Ceramic Fibres S. A. |
| | National Drill & Manufacturing Co. |
| 544 | National Ecology (Alabama) Incorporated |
| 575 | National Ecology (Utah) Incorporated |
| 976 | Nooter/Eriksen - Babcock & Wilcox, L.L.C. |
| 933 | North American CWF Partnership |
| 9989 | North Branch Power Company L.P. |
| 971 | North County Operations Associates |
| | Nuclear Materials and Equipment Corporation |
| 958 | Olin Pantex Inc. |
| 1153 | P. T. Heat Exchangers Indonesia |
| 934 | Palm Beach Energy Associates |
| ,,,, | Piedmont Tool Machine Company |
| 576 | Power Computing Company |
| 581 | Power Systems Sunnyside Operations GP, Inc. |
| 583 | Power Systems Sunnyside Operations LP, Inc. |
| | ž ž |
| | (1) a Delaware Corporation |
| 905 | PowerSafety International, Inc. |
| 508 | Productos de Caolin, Inc. |
| 577 | PSO Caribbean, Inc. |
| 1980 | Savannah River Alliance LLC |
| | |

| 2008 | Savannah River Tactical Services LLC |
|------|--------------------------------------|
| 984 | SOFCo L.P. |
| 932 | South Point CWF |
| 556 | Sunland Construction Co., Inc. |
| 988 | Sunnyside Cogeneration Associates |
| 582 | Sunnyside II, Inc. |
| 992 | Sunnyside II, L.P |
| 303 | Sunnyside III, Inc. |
| 993 | Sunnyside Operations Associates L.P. |
| 571 | Termobloc Industria E Comercio Ltda. |
| 953 | Thermax Babcock & Wilcox Limited |
| 502 | TLT-Babcock, Inc. |
| 570 | Triso |
| 1152 | W.E. Smith Hudson Pty. Ltd. |
| 548 | W. F. and John Barnes Company |
| | |

Schedule 1.1(c) B&W CURRENT

| Reference ID | Name | |
|--------------|--|---|
| 553 | Americon Equipment Services, Inc. | |
| 554 | Americon, Inc. | |
| 732 | Applied Synergistics, Inc. | |
| 127 | B&W de Panama, Inc. | |
| 532 | Babcock & Wilcox Canada Ltd. | |
| 1570 | Babcock & Wilcox China Holdings, Inc. | |
| 555 | Babcock & Wilcox Construction Co., Inc. | |
| 2011 | Babcock & Wilcox de Monterrey, S.A. de C.V. | |
| 1571 | Babcock & Wilcox Denmark Holdings, Inc. | |
| 327 | Babcock & Wilcox Ebensburg Power, Inc. | |
| 302 | Babcock & Wilcox Equity Investments, Inc. | |
| 1965 | Babcock & Wilcox Holdings, Inc. | |
| 2028 | Babcock & Wilcox India Holdings, Inc. | |
| 598 | Babcock & Wilcox India Private Limited | |
| 126 | Babcock & Wilcox International Investments Co., Inc. | |
| 530 | Babcock & Wilcox International Sales and Service Corporation | |
| 541 | Babcock & Wilcox International, Inc. | |
| 380 | Babcock & Wilcox Investment Company | |
| 2010 | Babcock & Wilcox Modular Nuclear Energy LLC | |
| 1974 | Babcock & Wilcox Nuclear Operations Group, Inc. | |
| 1967 | Babcock & Wilcox Nuclear Energy, Inc. | (formerly Babcock & Wilcox Nuclear Power Generation Group, Inc. |
| 2018 | Babcock & Wilcox Nuclear Services (U.K.) Limited | |
| 500 | Babcock & Wilcox Power Generation Group, Inc. | |
| 1961 | Babcock & Wilcox Technical Services (U.K.) Limited | |
| 1970 | Babcock & Wilcox Technical Services Clinch River, LLC | |
| 572 | Babcock & Wilcox Technical Services Group, Inc. | (formerly BWXT Services, Inc.) |
| 587 | Babcock & Wilcox Technical Services Savannah River Company | (formerly BWXT Savannah River Company) |
| 599 | Babcock & Wilcox Volund A/S | (formerly Babcock & Wilcox Volund ApS) |
| 2000 | BCE Parts Ltd. | |
| 596 | BWX Technologies, Inc. | |
| 580 | BWXT Federal Services, Inc. | |
| 589 | BWXT Hanford Company | |
| 382 | BWXT of Idaho, Inc. | |
| 592 | BWXT of Ohio, Inc. | |

| 1576 | BWXT Washington, Inc. | |
|------|---|---------------------------------------|
| 189 | Creole Insurance Company, Ltd. | |
| 1988 | Delta Power Services, LLC | |
| 766 | Diamond Operating Co., Inc. | |
| 1572 | Diamond Power Australia Holdings, Inc. | |
| 1984 | Diamond Power Central & Eastern Europe s.r.o. | |
| 1573 | Diamond Power China Holdings, Inc. | |
| 521 | Diamond Power do Brasil Limitada | |
| 1574 | Diamond Power Equity Investments, Inc. | |
| 525 | Diamond Power Finland OY | |
| 504 | Diamond Power Germany GmbH | (formerly Diamond Power - Sturm GmbH) |
| 597 | Diamond Power International, Inc. | |
| 949 | Diamond Power Machine (Hubei) Co., Inc. | |
| 1908 | Diamond Power Services S.E.A. Ltd. | |
| 522 | Diamond Power Specialty (Proprietary) Limited | |
| 523 | Diamond Power Specialty Limited | |
| 524 | Diamond Power Sweden AB | |
| 1990 | DPS Berkeley, LLC | |
| 1997 | DPS Cadillac, LLC | |
| 1995 | DPS Florida, LLC | |
| 1993 | DPS Gregory, LLC | |
| 1996 | DPS Lowell Cogen, LLC | |
| 1992 | DPS Mecklenburg, LLC | |
| 1991 | DPS Michigan, LLC | |
| 1994 | DPS Mojave, LLC | |
| 1998 | DPS Sabine, LLC | |
| 278 | Gumbo Insurance Company, Ltd. | |
| 2003 | Intech International Inc. | |
| 2002 | Intech, Inc. | |
| 2001 | Ivey-Cooper Services, L.L.C. | |
| 1968 | Marine Mechanical Corporation | |
| 540 | National Ecology Company | |
| 2005 | NFS Holdings, Inc. | |
| 2004 | NOG-Erwin Holdings, Inc. | |
| 501 | North County Recycling, Inc. | |
| 2006 | Nuclear Fuel Services, Inc. | |
| 1989 | O&M Holding Company | |
| 707 | P. T. Babcock & Wilcox Asia | |
| | | |

| 534 | Palm Beach Resource Recovery Corporation |
|------|---|
| 560 | Power Systems Operations, Inc. |
| 568 | Revloc Reclamation Service, Inc. |
| 2013 | Servicios de Fabricacion de Valle Soleado, S.A. de C.V. |
| 2012 | Servicios Profesionales de Valle Soleado, S.A. de C.V. |
| 767 | SOFCo - EFS Holdings LLC |
| 2029 | The Babcock & Wilcox Company |

EXHIBIT II - EXISTING REINSURANCE AGREEMENTS

See Attached.

| Name of Agreement | Party to Agreement | Party to Agreement | Party to Agreement | Party to Agreement | Party to Agreement | | | |
|---|-----------------------------------|--|--|--------------------|--------------------|-----------|---|-----------|
| Reinsurance Agreement (agreement # 22004) | Creole Insurance Company, Ltd. | Insurance Company of North America | McDermott Incorporated Et AI. | | | 6/1/1975 | | mcdermott |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1977 | WC only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1977 | Auto & GL only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | INA of Texas | | | 4/1/1978 | WC only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of | | | | 4/1/1978 | Auto & GL only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | North America Insurance Company of North America | INA of Texas | | | 4/1/1979 | WC only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | INA of Texas | | | 4/1/1979 | Auto & GL only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | INA of Texas | | | 4/1/1980 | WC only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | INA of Texas | | | 4/1/1980 | Auto & GL only | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1982 | Various other issuing companies included | vm |
| Addendum Number Five to Reinsurance Agreement effective 4/1/1982 | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1983 | Various other issuing companies included | vm |
| Addendum Number Six to Reinsurance Agreement effective 4/1/1982 | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1984 | Various other issuing companies included | vm |
| Addendum Number Seven to Reinsurance Agreement effective 4/1/1982 | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1985 | Various other issuing companies included | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1986 | Various other issuing companies included | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1987 | Various other issuing companies included; Amended to include CIGNA Insurance Company | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | CIGNA Insurance Company | | | | 4/1/1987 | Excess Liability | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | | 4/1/1988 | Various other issuing companies included; for Auto, WC and GL | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | CIGNA Insurance Company | | | | 4/1/1988 | for Excess Liability policy | vm |
| Trust Agreement | Creole Insurance Company, Ltd. | Pacific Employers Insurance Company | Morgan Guaranty Trust Company of New York | | | 6/24/1988 | poncy | vm |

| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | 4/1/1989 | Various other issuing companies included | vm |
|--|-----------------------------------|---|--|---|-----------|--|----|
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | 4/1/1990 | Various other issuing companies included; for Auto, WC and GL | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | 4/1/1990 | for Maritime policy | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | CIGNA Insurance Company | | | 4/18/1990 | • | tf |
| Reinsurance Agreement - Addendum I | Creole Insurance Company, Ltd. | CIGNA Insurance Company | | | 4/19/1990 | | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | 4/1/1991 | Various other issuing companies included; for Auto, WC and GL | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | | | 4/1/1991 | for Maritime policy | vm |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | Pacific Employers Insurance Company | | 4/1/1992 | princy | ср |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | Pacific Employers Insurance Company | | 4/1/1993 | | ср |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Insurance Company of North America | Pacific Employers Insurance Company | | 4/1/1994 | | ср |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | CIGNA Insurance Company of Texas | | | 4/1/1994 | | tf |
| Workers Compensation Residual Market Assessments Captive Programs | Creole Insurance Company, Ltd. | CIGNA Insurance Company of Texas | | | 4/1/1994 | | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | CIGNA Insurance Company | | | 4/1/1995 | | ср |
| Agreement for Workers Compensation Residual Market Assessments Captive Programs | Creole Insurance Company, Ltd. | CIGNA Insurance Company | | | 4/1/1995 | | ср |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | CIGNA Insurance Company | Indemnity Insurance Company of North America | CIGNA Insurance Company of Canada | 4/1/1995 | | ср |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Bankers Standard Insurance Company | | Canaua | 4/1/1995 | | tf |
| Workers Compensation Residual Market Assessments 1995 Captive Programs | Creole Insurance Company, Ltd. | Bankers Standard Insurance Company | | | 4/1/1995 | | tf |
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Indemnity Insurance Co. of N.A. | CIGNA Insurance Company | CIGNA Insurance Company of Canada | 4/1/1995 | Addendum 1 effective 4/1/96 | dc |

| Addendum I to Reinsurance Agreement | Creole Insurance Company, Ltd. | Indemnity Insurance Company of North America | CIGNA Insurance Company | CIGNA Insurance Company of Canada | | 4/1/1996 | | ср |
|---|-----------------------------------|--|-------------------------------|---|--|----------|-----------------------------------|----|
| Reinsurance Agreement | Creole Insurance Company, Ltd. | Indemnity Insurance Co. of N.A. | CIGNA Insurance Company | CIGNA Insurance Company of Canada | Pacific Employers Insurance Company | | Addendum 2 effective 4/1/97 | dc |

EXHIBIT III - MII ENTITIES

| See Schedule 1.1(b) and Schedule 1.1(j) attached hereto. No MII Entity listed on Schedule 1.1(b) shall be deemed to be an "Insured", a "Named Insured" or otherwise |
|---|
| be deemed to be insured under any Existing Policy or Existing Reinsurance Agreement solely by virtue of being listed on such Schedule 1.1(b). |

Schedule 1.1(b) MII FORMER

| Reference ID | |
|--------------|--|
| 947 | Arabian General Contracting Company |
| 911 | Arabian Petroleum Marine Construction Company |
| 8000 | Associated Pipe Line Contractors, Inc. |
| 117 | Badya Builders, Inc. |
| 242 | Beheer-en Beleggingsmaatschappij Belesa B.V. |
| 9996 | BJA-Mentor Production Systems Limited |
| 164 | Brown & Root McDermott Fabricators Limited |
| 277 | Cayenne Insurance Company, Ltd. |
| 629 | CCC Fabricaciones y Construcciones S.A. de C.V. |
| 940 | Construcciones Maritimas Mexicanas, S.A. de C.V. |
| 952 | Davy McDermott Limited |
| 921 | DB/McDermott Company |
| 1952 | Deep Gulf Contractors LLC |
| 171 | Deep Sea Divers Australia Pty. Limited |
| | Dick Evans, Inc. |
| 997 | ETPM International (UK) Limited |
| 955 | ETPM International S.A.S. |
| 999 | Far East Vessels, Inc. |
| 108 | First Emirates Trading Corporation |
| | Harvey Lumber & Supply Company |
| 174 | Heavy Lift Chartering, Inc. |
| 935 | HeereMac v.o.f. |
| 956 | Heerema-McDermott (Aust.) Pty. Ltd. |
| 190 | Honore Insurance Company, Ltd. |
| 917 | Initec, Astano y McDermott International Inc., S.A. |
| | Ingenieria Petrolera Maritima, S.A. de C.V. |
| | Iranian Marine Contractors, Inc. |
| 102 | J. Ray McDermott & Co. (Nederland) B.V. |
| 150 | J. Ray McDermott (Aust.) Pty. Ltd. |
| 151 | J. Ray McDermott (U.K.) Inc. |
| 350 | J. Ray McDermott (Venezuela) C.A. |
| 325 | J. Ray McDermott Arctic Ltd. |
| 307 | J. Ray McDermott Gulf Contractors, Inc. (formerly the Early Company) |

| 394 | J. Ray McDermott International Services (No. 1) Limited |
|------|---|
| 101 | J. Ray McDermott International Services Limited |
| 152 | J. Ray McDermott London, Ltd. |
| 641 | J. Ray McDermott Marine Construction AS |
| 1125 | J. Ray McDermott Newfoundland, Ltd. |
| 871 | J. Ray McDermott Offshore Chartering (U.S.), Inc. |
| 372 | J. Ray McDermott Properties, Inc. |
| 238 | J. Ray McDermott Underwater Services, Inc. |
| 603 | J. Ray McDermott Venture Holdings (U.S.), Inc. |
| 153 | Jaramac Associated Services, Ltd. |
| | Jaramac Petroleum (U.K.) Limited |
| | Khafji Contractors, Inc. |
| 166 | Khor Contractors, Inc. |
| 903 | KME a.s. |
| 324 | Lan-Dermott, S.A. de C.V. |
| 202 | Madock Shipyard Company, Ltd. |
| 203 | Macshelf Marine Construction Company, Ltd. |
| 106 | Mandato Pty. Limited |
| 360 | Marine Contractors, Inc. |
| 146 | McAntille, N.V. |
| 168 | McDermott - ETPM V.O.F. |
| 158 | McDermott (Holland) B.V. |
| 304 | McDermott (Nigeria) Limited |
| 245 | McDermott Acquisition Company, Inc. |
| 972 | McDermott APG Services Limited |
| 352 | McDermott Azerbaijan Pipelines, Inc. |
| 156 | McDermott Azerbaijan, Inc. |
| 351 | McDermott Central & Eastern Europe, Inc. |
| | McDermott de Mexico, S.A. de C.V. |
| 118 | McDermott Denmark ApS |
| 464 | McDermott Energy Services, Inc. |
| 178 | McDermott Engineering (Europe) Limited |
| | McDermott Enterprises France, S.A. |
| 128 | McDermott Europe Marine Services, Ltd. |
| 313 | McDermott Industries, Inc. |
| 354 | McDermott Inland Services, Inc. |
| 201 | McDermott International Asset Management, Ltd. |
| 343 | McDermott International Aviation, Inc. |
| | |

| 346 | McDermott International Aviation, Inc. | | | |
|------|--|--|--|--|
| | McDermott International (Deutschland) GmbH | | | |
| 237 | McDermott International Engineering & Construction Co., Ltd. | | | |
| 139 | McDermott International Engineering Investments N.V. | | | |
| 121 | McDermott International General Services, Inc. | | | |
| 123 | McDermott International Marine Services, Inc. | | | |
| 236 | McDermott International Marketing, Inc. | | | |
| 143 | McDermott Marine UK Limited | | | |
| 462 | McDermott Marketing Services, Inc. | | | |
| 103 | McDermott Middle East Trading, Ltd. | | | |
| 172 | McDermott Neutral Zone, Inc. | | | |
| | McDermott Norway, Inc. | | | |
| | McDermott Petroleum (Iran), Inc. | | | |
| 105 | McDermott Sakhalin, Inc. | | | |
| 347 | McDermott Shipbuilding, Inc. | | | |
| 111 | McDermott Singapore Pte. Ltd. | | | |
| 221 | McDermott Submarine Cable Ltd. | | | |
| 1951 | McDermott Submarine Cable Systems Limited | | | |
| 205 | McDermott Subsea Constructors Limited | | | |
| 110 | McDermott Transition Co., Inc. | | | |
| | McDermott UAR, Inc. | | | |
| 184 | McDermott West Indies Company | | | |
| 187 | McDermott-ETPM (Norway), Inc. | | | |
| 998 | McDermott-ETPM (UK) Limited | | | |
| 963 | McDermott-ETPM B.V. | | | |
| 161 | McDermott-ETPM East B.V. | | | |
| 167 | McDermott-ETPM East N.V. | | | |
| 964 | McDermott-ETPM N.V. | | | |
| 965 | McDermott-ETPM S.A. | | | |
| 962 | McDermott-ETPM West, Inc. | | | |
| 104 | Menck GmbH | | | |
| 463 | Mentor Engineering Consultants Limited | | | |
| 131 | MIMCO, Inc. | | | |
| 908 | MMC-McDermott Engineering Sdn. Berhad | | | |
| 317 | Mofco N.V. | | | |
| 129 | NOLA, Servicos E Participacoes Ltda. | | | |
| 141 | Northern Marine Services, Inc. | | | |
| 107 | Oceanic Red Sea Company | | | |
| | * * | | | |

(formerly McDermott International Aviation Co., Inc.)

| 620 | Offshore Angola, Ltd. |
|-----|---|
| 632 | Offshore Energy Capital Corporation |
| 622 | Offshore Hyundai International Limited |
| 630 | Offshore Hyundai International, Ltd. |
| 210 | Offshore Marine Chartering N.V. |
| 610 | Offshore Petroleum Divers, Inc. |
| 614 | Offshore Pipelines (Mauritius) Ltd. |
| 616 | Offshore Pipelines Far East Limited |
| 617 | Offshore Pipelines International Gulf E.C. |
| 633 | Offshore Power Generation Ltd. |
| 635 | Offshore Production Vessels, Ltd. |
| 615 | OPI Offshore Netherlands Holding B.V. |
| 611 | OPI Sales & Service Corporation |
| 628 | OPI Towing & Supply, L.L.C. |
| 624 | OPMI, E.C. |
| 613 | P. T. Armandi Pranaupaya |
| 850 | Panama Offshore Chartering Company 1, Inc. |
| 859 | Panama Offshore Chartering Company 10, Inc. |
| 860 | Panama Offshore Chartering Company 11, Inc. |
| 861 | Panama Offshore Chartering Company 12, Inc. |
| 862 | Panama Offshore Chartering Company 13, Inc. |
| 863 | Panama Offshore Chartering Company 14, Inc. |
| 864 | Panama Offshore Chartering Company 15, Inc. |
| 865 | Panama Offshore Chartering Company 16, Inc. |
| 866 | Panama Offshore Chartering Company 17, Inc. |
| 867 | Panama Offshore Chartering Company 18, Inc. |
| 868 | Panama Offshore Chartering Company 19, Inc. |
| 851 | Panama Offshore Chartering Company 2, Inc. |
| 869 | Panama Offshore Chartering Company 20, Inc. |
| 870 | Panama Offshore Chartering Company 21, Inc. |
| 852 | Panama Offshore Chartering Company 3, Inc. |
| 853 | Panama Offshore Chartering Company 4, Inc. |
| 854 | Panama Offshore Chartering Company 5, Inc. |
| 855 | Panama Offshore Chartering Company 6, Inc. |
| 856 | Panama Offshore Chartering Company 7, Inc. |
| 857 | Panama Offshore Chartering Company 8, Inc. |
| 858 | Panama Offshore Chartering Company 9, Inc. |
| 154 | Panama Offshore Chartering Company, Inc. |
| | |

| 990 | Personal Administrativo de Vera Cruz, S.A. de C.V. |
|-----|--|
| 991 | Personal Ejecutivo de Veracruz, S.A. de C.V. |
| 645 | Pirogue Insurance Company, Ltd. |
| 981 | Sakhalin Energy Investment Company, Ltd. |
| 384 | Samburg Holdings, Inc. |
| 109 | Sharman, Allen, Gay & Taylor, S.A. |
| 989 | Tallares Navales del Golfo, S.A. de C.V. |
| | The Evelyn Kay Company |
| | The Roger Thomas Company |
| | The Rosalie E. Company |
| | The Walter E. Company |
| 634 | TL Marine Sdn. Bhd. |
| 621 | TL Offshore Sdn. Bhd. |
| 926 | Topside Contractors of Newfoundland |
| 925 | Topside Contractors of Newfoundland, Ltd. |
| 642 | U.S. Offshore Chartering |
| 918 | Universal Fabricators Incorporated |
| 980 | US Shipbuilding Corporation, Inc. |
| 318 | Wagley, Inc. |
| | |

Schedule 1.1(j) MII - CURRENT

| CURRENT | V. | |
|---------------------|---|--|
| Reference ID 873 | Name Barmada McDermott (L) Limited | |
| 873 1953 | Barmada McDermott Sdn. Bhd. | |
| 276 | | |
| 939 | Boudin Insurance Company, Ltd. | |
| 939 116 | Caspian Offshore Fabricators LLC | |
| | Chartering Company (Singapore) Pte. Ltd. | |
| 142 | Eastern Marine Services, Inc. | |
| 249 | Global Energy - McDermott Limited | |
| 140 | Hydro Marine Services, Inc. | |
| 649 | International Vessels Ltd. | |
| 148 | J. Ray McDermott (Aust.) Holding Pty. Limited | (C 1 I D M D ((D)) I (() 1 I) |
| 1982 | J. Ray McDermott (Caspian), Inc. | (formerly J. Ray McDermott Diving International, Inc.) |
| 243 | J. Ray McDermott (Nigeria) Ltd. | |
| 2025 | J. Ray McDermott (Norway), AS | |
| 1237 | J. Ray McDermott (Qingdao) Pte. Ltd. | (C 1 M D |
| 194 | J. Ray McDermott Asia Pacific Pte. Ltd. | (formerly McDermott South East Asia Pte. Ltd.) |
| 1972 | J. Ray McDermott Canada Holding, Ltd. | |
| 1969 | J. Ray McDermott Canada, Ltd. | |
| 204 | J. Ray McDermott Contractors, Inc. | |
| 246 | J. Ray McDermott de Mexico, S.A. de C.V. | |
| 208 | J. Ray McDermott Eastern Hemisphere Limited | (formerly J. Ray Middle East (Indian Ocean) Ltd.) |
| 1956 | J. Ray McDermott Engineering Services Private Limited | |
| 1950 | J. Ray McDermott Engineering, LLC | |
| 220 | J. Ray McDermott Far East, Inc. | |
| 211 | J. Ray McDermott Holdings, LLC | (formerly J. Ray McDermott Holdings, Inc.) |
| 627 | J. Ray McDermott International Vessels, Ltd. | (formerly OPI International, Ltd.) |
| 602 | J. Ray McDermott International, Inc. | |
| 160 | J. Ray McDermott Investments B.V. | (formerly Menck B.V.) |
| 1958 | J. Ray McDermott Kazakhstan Limited Liability Partnership | |
| 1975 | J. Ray McDermott Logistic Services Pvt. Limited | |
| 136 | J. Ray McDermott Middle East, Inc. | |
| 239 | J. Ray McDermott Solutions, Inc. | (formerly J. Ray McDermott Engineering Holdings, Inc.) |
| 625 | J. Ray McDermott Technology, Inc. | |
| 1957 | J. Ray McDermott UK Ltd. | |
| 1960 | J. Ray McDermott Underwater Services, Inc. | |

| 248 | J. Ray McDermott West Africa Holdings, Inc. | |
|------|---|--|
| 244 | J. Ray McDermott West Africa, Inc. | |
| 601 | J. Ray McDermott, Inc. | |
| 207 | J. Ray McDermott, S.A. | |
| 644 | Lagniappe Insurance Company, Ltd. | |
| 193 | Malmac Sdn. Bhd. | |
| 132 | McDermott (Malaysia) Sendirian Berhad | |
| 904 | McDermott Abu Dhabi Offshore Construction Company | |
| 915 | McDermott Arabia Company Limited | |
| 149 | McDermott Australia Pty. Ltd. | (formerly McDermott Industries (Aust.) Pty. Limited) |
| 183 | McDermott Azerbaijan Marine Construction, Inc. | |
| 241 | McDermott Caspian Contractors, Inc. | |
| 275 | McDermott Cayman Ltd. | |
| 181 | McDermott Far East, Inc. | |
| 133 | McDermott Gulf Operating Company, Inc. | |
| 177 | McDermott Holdings (U.K.) Limited | |
| 300 | McDermott Incorporated | |
| 147 | McDermott International B.V. | |
| 120 | McDermott International Investments Co., Inc. | |
| 119 | McDermott International Marine Investments N.V. | |
| 213 | McDermott International Marketing, Inc. | (formerly McDermott International Beijing, Inc.) |
| 130 | McDermott International Trading Co., Inc. | |
| 1971 | McDermott International Vessels, Inc. | |
| 100 | McDermott International, Inc. | |
| 1962 | McDermott Kft. | |
| 349 | McDermott Marine Construction Limited | |
| 1986 | McDermott Marine Mexico, S.A. de C.V. | |
| 135 | McDermott Offshore Services Company, Inc. | |
| 176 | McDermott Old JV Office, Inc. | |
| 162 | McDermott Overseas Investment Co. N.V. | |
| 390 | McDermott Overseas, Inc. | |
| 1966 | McDermott Panama Holdings, S.A. | |
| 170 | McDermott Servicos de Construcao, Ltda. | |
| 320 | McDermott Trade Corporation | |
| 306 | Mentor Subsea Technology Services, Inc. | |
| 138 | North Atlantic Vessel, Inc. | |
| 2021 | North Ocean II AS | |
| 2023 | North Ocean II KS | |
| | | |

| 2022 | North Ocean V AS | |
|------|--|--|
| 612 | Offshore Pipelines International, Ltd. | |
| 618 | Offshore Pipelines Nigeria Limited | |
| 619 | Offshore Pipelines Sdn. Bhd. | |
| 626 | OPI Vessels, Inc. | |
| 623 | OPMI, Ltd. | |
| 185 | P. T. Bataves Fabricators | |
| 180 | P. T. McDermott Indonesia | |
| 901 | PT. Baja Wahana Indonesia | (formerly P.T. Babcock & Wilcox Indonesia) |
| 643 | PT. J. Ray McDermott Indonesia | (formerly PT. Jay Ray) |
| 636 | Sabine River Realty, Inc. | |
| 1977 | Servicios de Fabricacion de Altamira, S.A. de C.V. | |
| 1976 | Servicos Profesionales de Altamira, S.A. de C.V. | |
| 2019 | Singapore Huangdao Pte. Ltd. | |
| 640 | SparTEC, Inc. | (formerly J. Ray McDermott Spars, Inc.) |
| 145 | Varsy International N.V. | |

NOVATION AND ASSUMPTION AGREEMENT

by and among

MCDERMOTT INTERNATIONAL, INC.,

a Panamanian corporation,

and

BABCOCK & WILCOX HOLDINGS, INC.,

a corporation organized and existing under the laws of the State of Delaware,

and

BOUDIN INSURANCE COMPANY, LTD., a Bermuda company

and

CREOLE INSURANCE COMPANY, LTD., a Bermuda company,

RECITALS

THIS NOVATION AND ASSUMPTION AGREEMENT (the "Agreement"), is entered into and effective as of May 18, 2010 (the "Effective Date") by and among MCDERMOTT INTERNATIONAL, INC., a Panamanian corporation ("MII"), BABCOCK & WILCOX HOLDINGS, INC., a Delaware corporation ("B&W"), to be succeeded by The Babcock & Wilcox Company, a Delaware corporation, after the effective time of the Merger, CREOLE INSURANCE COMPANY, LTD., a Bermuda company ("Creole"), and BOUDIN INSURANCE COMPANY, LTD., a Bermuda company ("Boudin") and, solely with respect to Sections 2(a)(ii) and 2(c) (ii), respectively, the other MII Entities signatory hereto and the other B&W Entities signatory hereto.

WHEREAS, Boudin or Creole have issued the Existing Policies to one or more MII Entities and one or more B&W Entities; and

WHEREAS, pursuant to the Existing Policies, Boudin and Creole are obligated, among other things, to pay or reimburse MII Entities and/or B&W Entities for certain obligations; and

WHEREAS, B&W, prior to the Separation, is a wholly owned Subsidiary of MII; and

WHEREAS, MII intends to spin-off B&W from MII through a dividend of common stock of B&W to the shareholders of MII (the Separation"); and

WHEREAS, in connection with the Separation: (a) Boudin desires to transfer and Creole desires to assume any Transferable Boudin Policies insofar as such policies relate to B&W Entities; and (b) Creole desires to transfer and Boudin desires to assume any Transferable Creole Policies insofar as such policies relate to MII Entities; and

WHEREAS, MII and B&W are parties, together with ACE American Insurance Company and certain of its affiliates (collectively, "ACE") to a certain Assumption and Loss Allocation Agreement (the "ALAA") that, among other things, assists in effecting the Separation by identifying certain MII Obligations and certain B&W Obligations and allocating liability and responsibility therefor; and

WHEREAS, certain of the liabilities of MII and/or B&W and their respective Affiliates that are insured by the Existing Policies under this Agreement are MII Obligations and/or B&W Obligations under the ALAA, and the parties desire that the allocation of liability in this Agreement conform to that in the ALAA insofar as such overlap may exist; and

WHEREAS, each of MII and B&W, each for itself and for its respective Subsidiaries, is willing to consent to the transfer, assumption, and novation of the matters set forth herein, subject to the terms and conditions of this Agreement;

NOW, THEREFORE, in consideration of the mutual promises set out herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

- 1. <u>Definitions</u>. The following terms used herein, including in the recitals and Exhibits hereto, shall have the following meanings:
 - "ACE" has the meaning set forth in the recitals to this Agreement.
- "Affiliate" means, with respect to any specified Person, any other Person that directly, or indirectly through one or more intermediaries, controls, is controlled by, or is under common control with, the specified Person. For this purpose "control" of a Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of voting securities, by contract or otherwise.
 - "Agreement" has the meaning set forth in the recitals to this Agreement.
 - "ALAA" has the meaning set forth in the recitals to this Agreement.
 - "Assumption Time" means midnight (New York time) on the Effective Date.
 - "B&W" has the meaning set forth in the recitals to this Agreement.
- "B&W Entity" means B&W and each of the entities listed on Exhibit I attached hereto and made a part hereof. It is acknowledged and understood that, from and after the effectiveness of the Separation, the B&W Entities will not be Subsidiaries or Affiliates of MII or any of the other MII Entities.
- "B&W Obligations" shall have the meaning ascribed to such term in the ALAA and shall be interpreted under this Agreement to conform to interpretations under the ALAA.
 - "Boudin Assumption and Novation" has the meaning set forth in Section 2(a)(i).
 - "Creole Assumption and Novation" has the meaning set forth in Section 2(c)(i).
 - "Effective Date" has the meaning set forth in the recitals to this Agreement.
- "Existing Policies" means the Transferable Boudin Policies, the Transferable Creole Policies, the Wholly Retained Boudin Policies, and the Wholly Retained Creole Policies.
- "Insured," as a noun in reference to one or more insurance policies, means any Person who is insured by such policy or policies, regardless of whether such Person is designated an "Insured" or a "Named Insured" in such policy or is otherwise expressly identified therein.
- "Insured B&W Obligation" means an obligation of Creole or Boudin under the Existing Policies that arises from an obligation to ACE or its Affiliates that is a B&W Obligation under the ALAA.
- "Insured MII Obligation" means an obligation of Creole or Boudin under the Existing Policies that arises from an obligation to ACE or its Affiliates that is an MII Obligation under the ALAA.
 - "Insurer," as a noun in reference to one or more insurance policies, means the Person identified in such policy or policies as the insurer.
 - "Inter-Captive Reinsurance Agreements" has the meaning set forth in Section 2(e).

"Master Separation Agreement" means a Master Separation Agreement to be entered into between MII and The Babcock & Wilcox Company in connection with the Separation.

"Merger" means the merger, to occur after the date hereof, of Babcock & Wilcox Holdings, Inc., a Delaware corporation, with and into The Babcock & Wilcox Company, a Delaware corporation and the surviving entity of such Merger.

"MII" has the meaning set forth in the recitals to this Agreement.

"MII Entity" means MII and each of the entities listed on Exhibit II attached hereto and made a part hereof. It is acknowledged and understood that, from and after the effectiveness of the Separation, the MII Entities will not be Subsidiaries or Affiliates of B&W or any of the other B&W Entities.

"MII Obligations" shall have the meaning ascribed to such term in the ALAA and shall be interpreted under this Agreement to conform to interpretations under the ALAA.

"Organizational Documents" means (a) with respect to any corporation, its certificate or articles of incorporation or organization and its bylaws, (b) with respect to any limited partnership, its certificate of limited partnership agreement, (c) with respect to any general partnership, its partnership agreement, and (d) with respect to any limited liability company, its certificate or articles of formation or organization and its operating agreement or other organizational documents.

"Parties" means MII, B&W, Boudin and Creole collectively (and each individually is a "Party").

"Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, a union, an unincorporated organization or a governmental entity or any department, agency or political subdivision thereof.

"Separation" has the meaning set forth in the recitals to this Agreement.

"Subsidiary" means, with respect to any specified Person, any corporation, partnership, limited liability company, joint venture or other organization, whether incorporated or unincorporated, of which at least a majority of the securities or interests having by the terms thereof ordinary voting power to elect at least a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such specified Person or by any one or more of its Subsidiaries, or by such specified Person and one or more of its Subsidiaries.

"Transferable Boudin Policy" means each insurance policy issued prior to the date hereof by Boudin on which a B&W Entity is named or otherwise identified as an insured (including without limitation by being within a generic description such as "affiliates" or "subsidiaries").

"Transferable Creole Policy" means each insurance policy issued prior to the date hereof by Creole on which an MII Entity is named or otherwise identified as an insured (including without limitation by being within a generic description such as "affiliates" or "subsidiaries").

"Transferred Boudin Policy" means, as to any Transferable Creole Policy after the Assumption Time and giving effect to this Agreement, the aggregate of (a) all rights, duties, and obligations of Boudin vis-à-vis the MII Entities that were, prior to the Assumption Time, Insureds of Creole under such Transferable Creole Policy, and (b) all rights, duties, and obligations vis-à-vis Boudin of the MII Entities that were, prior to the Assumption Time, Insureds of Creole under such Transferable Creole Policy.

"Transferred Creole Policy" means, as to any Transferable Boudin Policy after the Assumption Time and giving effect to this Agreement, the aggregate of (a) all rights, duties, and obligations of Creole vis-à-vis the B&W Entities that were, prior to the Assumption Time, Insureds of Boudin under such Transferable Boudin Policy, and (b) all rights, duties, and obligations vis-à-vis Creole of the B&W Entities that were, prior to the Assumption Time, Insureds of Boudin under such Transferable Boudin Policy.

"Wholly Retained Boudin Policy" means an insurance policy issued by Boudin to one or more MII Entities (and possibly others) that is not a Transferable Boudin Policy.

"Wholly Retained Creole Policy" means an insurance policy issued by Creole to one or more B&W Entities (and possibly others) that is not a Transferable Creole Policy.

2. Assumption

- (a) <u>Boudin Assumption and Novation</u>. Notwithstanding anything in any Transferable Creole Policy to the contrary, and effective as of the Assumption Time, Creole hereby transfers and assigns, and Boudin hereby assumes by novation, (x) so much of each Transferable Creole Policy as relates to MII or any MII Entity as an Insured thereunder; and (y) any and all obligations of Creole under any of the Existing Policies that arise from Insured MII Obligations. In connection with and implementation of such transfer, assignment, and novation:
 - (i) Boudin hereby agrees to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of Creole under the Transferable Creole Policies and with respect to the Insured MII Obligations, both insofar as transferred above (the "Boudin Assumption and Novation"); and
 - (ii) Each MII Entity that is a signatory hereto hereby consents to, and agrees to give full force and effect to, the Boudin Assumption and Novation. From and after the Assumption Time, Boudin and not Creole shall be treated as the applicable MII Entity's contractual counterparty with respect to the contracts and mutual rights and obligations subject to the Boudin Assumption and Novation. Without limitation, each MII Entity:
 - a may enforce against Boudin its rights under the Transferable Creole Policies or with respect to any Insured MII Obligation to the same extent such Person could, prior to the Boudin Assumption and Novation, enforce such rights against Creole, and
 - b shall perform for the benefit of Boudin any obligation under the Transferable Creole Policies or with respect to any Insured MII Obligation to the same extent such Person was obligated, prior to the Boudin Assumption and Novation, to perform such obligations for the benefit of Creole, and
 - c releases Creole from its obligation to observe, pay, perform, satisfy, fulfill or discharge any obligations under any Transferable Creole Policy or with respect to any Insured MII Obligation.
 - (iii) With respect to each Transferable Creole Policy that is subject to the Boudin Assumption and Novation, the aggregate of rights, duties, and obligations set forth in subsections (a)(i) and (a)(ii) above shall be treated as the Transferred Boudin Policy relating to such Transferable Creole Policy.
- (b) No Transfer or Novation of Creole Obligations to B&W Entities. Except to the extent (which the Parties do not expect) that the Wholly Retained Creole Policies are determined to cover Insured MII Obligations notwithstanding the absence of any MII Entity as an Insured thereunder, the Wholly Retained Creole Policies are not novated or otherwise affected by the Boudin Assumption and Novation. The Transferable Creole Policies are novated to Boudin as set forth above only to the extent that one or more MII Entities is an insured thereunder. To the extent that B&W Entities are insureds under the Transferable Creole Policies, the Parties acknowledge that Creole and not Boudin shall continue to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of the Insurer under the Transferable Creole Policies. The Parties further acknowledge that Creole and not Boudin shall continue to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of the Insurer with respect to the Insured B&W Obligations.
- (c) <u>Creole Assumption and Novation</u>. Notwithstanding anything in any Transferable Boudin Policy to the contrary, and effective as of the Assumption Time, Boudin hereby transfers and assigns, and Creole hereby assumes by novation, (x) so much of each Transferable Boudin Policy as relates to B&W or any

B&W Affiliate as an Insured thereunder; and (y) any and all obligations of Boudin under any of the Existing Policies that arise from Insured B&W Obligations. In connection with and in implementation of such transfer, assignment, and novation:

- (i) Creole hereby agrees to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of Boudin under the Transferable Boudin Policies and with respect to the Insured B&W Obligations, both insofar as transferred above (the "Creole Assumption and Novation"); and
- (ii) Each B&W Entity that is a signatory hereto hereby consents to, and agrees to give full force and effect to, the Creole Assumption and Novation. From and after the Assumption Time, Creole and not Boudin shall be treated as the B&W Entity's contractual counterparty with respect the contracts and mutual rights and obligations subject to the Creole Assumption and Novation. Without limitation, each B&W Entity:
 - a may enforce against Creole its rights under the Transferable Boudin Policies or with respect to any Insured B&W Obligation to the same extent such Person could, prior to the Creole Assumption and Novation, enforce such rights against Boudin, and
 - b shall perform for the benefit of Creole any obligation under the Transferable Boudin Policies or with respect to any Insured B&W Obligation to the same extent such Person was obligated, prior to the Creole Assumption and Novation, to perform such obligations for the benefit of Boudin, and
 - c releases Boudin from its obligation to observe, pay, perform, satisfy, fulfill or discharge any obligations under any Transferable Boudin Policy or with respect to any Insured B&W Obligation.
- (iii) With respect to each Transferable Boudin Policy that is subject to the Creole Assumption and Novation, the aggregate of rights, duties, and obligations set forth in subsections (c)(i) and (c)(ii) above shall be treated as the Transferred Creole Policy relating to such Transferable Boudin Policy.
- (d) No Transfer or Novation of Boudin Obligations to MII Entities Except to the extent (which the Parties do not expect) that the Wholly Retained Boudin Policies are determined to cover Insured B&W Obligations notwithstanding the absence of any B&W Entity as an Insured thereunder, the Wholly Retained Boudin Policies are not novated or otherwise affected by the Creole Assumption and Novation. The Transferable Boudin Policies are novated to Creole as set forth above only to the extent that one or more B&W Entities is an insured thereunder. To the extent that MII Entities are insureds under the Transferable Boudin Policies, the Parties acknowledge that Boudin and not Creole shall continue to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of the Insurer under the Transferable Boudin Policies. The Parties further acknowledge that Boudin and not Creole shall continue to observe, pay, perform, satisfy, fulfill and discharge any and all now existing and hereafter arising duties, terms, provisions, covenants, obligations and liabilities of the Insurer with respect to the Insured MII Obligations.
- (e) <u>Commutation and Release of Mutual Reinsurance by Boudin and Creole</u> Effective at the Assumption Time, Creole and Boudin do hereby commute any and all contracts of reinsurance by which one of them reinsured the other (the "<u>Inter-Captive Reinsurance Agreements</u>"). In furtherance of such commutation, and also effective at the Assumption Time, Creole and Boudin do hereby fully release and discharge one another from and against any past, present, or future liabilities or obligations, known or unknown, arising under the Inter-Captive Reinsurance Agreements.
- (f) No Effect on Aggregate Limits of Liability. For the avoidance of doubt, it is understood and agreed that the aggregate liability of Creole and Boudin, taken together, is not intended to be, and shall be deemed not to be, increased by implementation of this Agreement. In particular, and without limitation, to the extent that any Transferable Creole Policy (or Transferable Boudin Policy) contains an aggregate limit of liability,

that aggregate limit of liability shall, after the Assumption Time, apply as a single, joint aggregate limit of liability as between (i) the resulting Transferred Boudin Policy (or Transferred Creole Policy) and (ii) the portions of the Transferable Creole Policy (or Transferable Boudin Policy) that are retained pursuant to Section 2(b) (or Section 2(d)) above. Disputes as to priority of claims and/or allocation of the single, joint aggregate limit of liability shall be resolved pursuant to Section 12 hereof.

- 3. Amendments. Neither this Agreement nor any provision hereof may be amended, changed, waived, discharged or terminated except by a written instrument signed by each Party.
- 4. Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns. Neither this Agreement nor any right or obligation hereunder may be assigned or conveyed by any Party without the prior written consent of the other Parties, which consent shall not be unreasonably withheld.
- 5. No Waiver. The failure or refusal by any Party to exercise any rights granted hereunder shall not constitute a waiver of such rights or preclude the subsequent exercise thereof, and no oral communication shall be asserted as a waiver of any such rights hereunder unless such communication shall be confirmed in a writing plainly expressing an intent to waive such rights and signed by the Party against whom such waiver is asserted.
- **6.** Counterparts. This Agreement may be executed in any number of counterparts each of which when so executed and delivered shall constitute an original, but such counterparts together shall constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile transmission shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes. Signatures of the Parties and other Persons signatory hereto transmitted by facsimile shall be deemed to be their original signatures for all purposes.
- 7. No Third Party Beneficiary. This Agreement shall not be deemed to give any right or remedy to any third party whatsoever unless otherwise specifically granted hereunder.
- 8. Parties' Representations. As of the Effective Date, each of the Parties expressly represents on its own behalf: (a) it is an entity in good standing in its jurisdiction of organization; (b) it has all requisite corporate power and authority to enter into this Agreement, and to perform its obligations hereunder; (c) the execution and delivery by it of this Agreement, and the performance by it of its obligations under this Agreement, have been duly authorized by all necessary corporate or other action; (d) this Agreement, when duly executed and delivered by it, and subject to the due execution and delivery hereof by the other Parties, will be a valid and binding obligation of it, enforceable against it, its successors and permitted assigns, in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar laws affecting creditors' rights generally and general equity principles; (e) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby in accordance with the respective terms and conditions hereof will not (i) violate any provision of its Organizational Documents, (ii) violate any applicable order, judgment, injunction, award or decree of any court, arbitrator or governmental or regulatory body against it, or binding upon it, or any agreement with, or condition imposed by, any governmental or regulatory body, foreign or domestic, binding upon it as of the date hereof, or (iii) violate any agreement, contract, obligation, promise or undertaking that is legally binding and to which it is a party or by which it is bound; and (f) the signatory hereto on behalf of it is duly authorized and legally empowered to enter into this Agreement on its behalf.

9. Notices. Any and all notices, requests, approvals, authorizations, consents, instructions, designations and other communications that are required or permitted to be given pursuant to this Agreement shall be in writing and may be given either by personal delivery, first class prepaid post (airmail if to another country) or by internationally recognized overnight delivery service to the following address, or to such other address and recipient as such Party may have notified in accordance with the terms of this section as being its address or recipient for notification for the purposes of this Agreement:

If to Boudin: Boudin Insurance Company, Ltd., c/o McDermott International, Inc.

757 N. Eldridge Parkway

Houston, Texas 77079

Attention: VP and Chief Risk Officer (with copy to General Counsel)

Telephone: 281-870-5785 Telecopier: 281-870-5923

Electronic Mail: twoodard@mcdermott.com

with a copy to:

Boudin Insurance Company, Ltd.

Cedar House 41 Cedar Avenue P.O. Box HM 1838 Hamilton HM HX Bermuda

Bermuda

 Fax
 (441) 295-3982

 Telephone:
 281-870-5476

 Telecopier:
 281-870-5923

 Electronic Mail:
 cjryan@mcdermott.com

If to any MII Entity: McDermott International, Inc.

757 N. Eldridge Parkway Houston, Texas 77079

Attention: VP and Chief Risk Officer (with copy to General Counsel)

Telephone: 281-870-5785 Telecopier: 281-870-5923

Electronic Mail: twoodard@mcdermott.com

If to Creole: Creole Insurance Company, Ltd.

Prior to the Separation:

Creole Insurance Company, Ltd., c/o Babcock & Wilcox Holdings, Inc.

800 Main Street

Lynchburg, Virginia 24504

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 434-522-6800

with a copy to:

McDermott International, Inc. 777 N. Eldridge Parkway Houston, Texas 77079

Attention: Director, Risk Management (with copy to General Counsel)

with an additional copy to:

Creole Insurance Company, Ltd. Cedar House

41 Cedar Avenue P.O. Box HM 1838

Hamilton HM HX Bermuda

Bermuda

Fax (441) 295-3982 Telephone: 281-870-5476 Telecopier: 281-870-5923

Electronic Mail: cjryan@mcdermott.com

On and After the Separation:

The Babcock & Wilcox Company

The Harris Building

13024 Ballantyne Corporate Place, Suite 700

Charlotte, North Carolina

Attention: Director, Risk Management (with copy to General Counsel)

with a copy to:

Creole Insurance Company, Ltd.

Cedar House 41 Cedar Avenue

P.O. Box HM 1838 Hamilton HM HX Bermuda

Bermuda

Fax (441) 295-3982 Telephone: 281-870-5476 Telecopier: 281-870-5923

Electronic Mail: cjryan@mcdermott.com

If to any B&W Entity:

Prior to the Separation:

Babcock & Wilcox Holdings, Inc.

800 Main Street

Lynchburg, Virginia 24504

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 434-522-6800

with a copy to:

McDermott International, Inc. 777 N. Eldridge Parkway Houston, Texas 77079

Attention: Director, Risk Management (with copy to General Counsel)

Telephone: 281-870-5476
Telecopier: 281-870-5923
Electronic Mail: cjryan@mcdermott.com

On and After the Separation:

The Babcock & Wilcox Company

The Harris Building

13024 Ballantyne Corporate Place, Suite 700

Charlotte, North Carolina

Attention: Director, Risk Management (with copy to General Counsel)

Any notice or communication to any Person shall be deemed to be received by that Person:

- (A) upon personal delivery; or
- (B) upon receipt if sent by mail or courier.
- 10. Governing Law. This Agreement shall be governed by and construed in accordance with the internal laws of the Commonwealth of Pennsylvania without regard to those provisions concerning conflicts of laws that would result in the application of the laws of any other jurisdiction.
- 11. Entire Agreement. THIS AGREEMENT CONSTITUTES THE ENTIRE AGREEMENT AMONG ALL OF THE PARTIES WITH RESPECT TO THE TRANSFERS, ASSUMPTIONS, AND NOVATIONS DESCRIBED HEREIN AND SUPERSEDES ALL OTHER PRIOR AGREEMENTS AND UNDERSTANDINGS, BOTH WRITTEN AND ORAL, WITH RESPECT TO THE SUBJECT MATTER HEREOF. SOLELY FOR INTERPRETATION PURPOSES, THE PARTIES ACKNOWLEDGE THAT THIS AGREEMENT IS INTENDED TO BE READ TOGETHER WITH THE ALAA, AND THE PARTIES FURTHER ACKNOWLEDGE THE CONTROLLING NATURE OF THE ALAA WITH RESPECT TO DISPUTE RESOLUTION, AS SET FORTH IN THE FOLLOWING PARAGRAPH.
- 12. <u>Dispute Resolution</u>. Any and all disputes arising out of this Agreement shall be resolved in accordance with the procedures set for in the Master Separation Agreement (or, in the event the Master Separation Agreement is not entered into by MII and The Babcock & Wilcox Company, in accordance with the procedures set forth in the draft thereof dated as of April 28, 2010). Issues as to whether a given obligation is an Insured B&W Obligation or an Insured MII Obligation shall be resolved as set forth in the ALAA, which resolution shall be binding for all purposes of this Agreement, notwithstanding (by way of example only) any assertion that the putative Insured B&W Obligation arose under a Wholly Retained Boudin Policy or the putative Insured MII Obligation arose under a Wholly Retained Creole Policy, or any other assertion that the terms of this Agreement prohibit the allocation reached under the ALAA. Any such assertions may be made, to the extent appropriate, in connection with the dispute resolution under the ALAA but may not be made as a basis for challenging such resolution.
- 13. Severability. If any term or other provision of this Agreement or the Exhibits attached hereto is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the court, administrative agency or arbitrator shall interpret this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible. If any sentence in this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.
- 14. Rules of Construction. The definitions of terms used herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "includes" and "including" shall be deemed to be followed by the phrase "without limitation". The word "will" shall be construed to have the same meaning and effect as the word "shall". Unless the context requires otherwise (a) any definition of or reference to any agreement, instrument or other document herein shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified, (b) any reference herein to any law shall be construed as referring to such law as amended, modified, codified or reenacted, in whole or in part, and in effect from time to time, (c) any reference herein to any Person shall be construed to include such Person's successors and assigns (subject to the restrictions contained herein), (d) the words "herein", "hereof" and "hereunder", and words of similar import, shall be construed to refer to this

Agreement in its entirety and not to any particular provision hereof, (e) with respect to the determination of any time period, the word "from" means "from and including" and the word "to" means "to and including" and (f) any reference herein to Articles, Sections and Exhibits shall be construed to refer to Articles and Sections of, and Exhibits to, this Agreement. No provision of this Agreement shall be interpreted or construed against any Person solely because such Person or its legal representative drafted such provision.

[Remainder of Page Intentionally Left Blank]

| By: /S/ LIANE K. HINRICHS Liane K. Hinrichs Senior Vice President |
|--|
| BOUDIN INSURANCE COMPANY, LTD. |
| By: /S/ LIANE K. HINRICHS Liane K. Hinrichs Assistant Secretary |
| BABCOCK & WILCOX HOLDINGS, INC. |
| By: /S/ MICHAEL S. TAFF Michael S. Taff Senior Vice President |
| CREOLE INSURANCE COMPANY, LTD. |
| By: /S/ BENJAMIN H. BASH Benjamin H. Bash Assistant Secretary |

 ${\bf MCDERMOTT\ INTERNATIONAL, INC.}$

IN WITNESS WHEREOF, the Parties intending to be legally bound hereby have executed this Agreement, by their duly authorized representatives.

ACKNOWLEDGED AND AGREED FOR PURPOSES OF SECTION 2:

B&W ENTITIES:

AMERICON EQUIPMENT SERVICES, INC.

AMERICON, INC.

APPLIED SYNERGISTICS, INC.

BABCOCK & WILCOX CANADA LTD.

BABCOCK & WILCOX CHINA HOLDINGS, INC.

BABCOCK & WILCOX CONSTRUCTION CO., INC.

BABCOCK & WILCOX DE MONTERREY, S.A. DE C.V.

BABCOCK & WILCOX DENMARK HOLDINGS, INC.

BABCOCK & WILCOX EBENSBURG POWER, INC.

BABCOCK & WILCOX EQUITY INVESTMENTS, INC.

BABCOCK & WILCOX INDIA HOLDINGS, INC.

BABCOCK & WILCOX INTERNATIONAL SALES AND SERVICE CORPORATION

BABCOCK & WILCOX INTERNATIONAL, INC.

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash
Title: Assistant Secretary of each of the above

BABCOCK & WILCOX MODULAR NUCLEAR ENERGY LLC

BABCOCK & WILCOX NUCLEAR OPERATIONS GROUP, INC.

BABCOCK & WILCOX NUCLEAR ENERGY, INC

BABCOCK & WILCOX NUCLEAR SERVICES (U.K.) LIMITED

BABCOCK & WILCOX TECHNICAL SERVICES CLINCH RIVER, LLC

BABCOCK & WILCOX POWER GENERATION GROUP, INC.

BABCOCK & WILCOX TECHNICAL SERVICES GROUP, INC.

BABCOCK & WILCOX TECHNICAL SERVICES SAVANNAH RIVER COMPANY

BABCOCK & WILCOX VOLUND A/S

BCE PARTS LTD.

BWX TECHNOLOGIES, INC.

BWXT FEDERAL SERVICES, INC.

BWXT HANFORD COMPANY

BWXT OF IDAHO, INC.

BWXT OF OHIO, INC.

BWXT WASHINGTON, INC.

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash

Title: Assistant Secretary of each of the above

CREOLE INSURANCE COMPANY, LTD.

DELTA POWER SERVICES, LLC

DIAMOND OPERATING CO., INC.

DIAMOND POWER AUSTRALIA HOLDINGS, INC.

DIAMOND POWER CHINA HOLDINGS, INC.

DIAMOND POWER EQUITY INVESTMENTS, INC.

DIAMOND POWER INTERNATIONAL, INC.

DIAMOND POWER SPECIALTY (PROPRIETARY) LIMITED

DPS BERKELEY, LLC

DPS CADILLAC, LLC

DPS FLORIDA, LLC

DPS GREGORY, LLC

DPS LOWELL COGEN, LLC

DPS MECKLENBURG, LLC

DPS MICHIGAN, LLC

DPS MOJAVE, LLC

DPS SABINE, LLC

GUMBO INSURANCE COMPANY, LTD.

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash
Title: Assistant Secretary of each of the above

INTECH, INC.
IVEY-COOPER SERVICES, L.L.C.
MARINE MECHANICAL CORPORATION
NFS HOLDINGS, INC.
NOG-ERWIN HOLDINGS, INC.
NORTH COUNTY RECYCLING, INC.
NUCLEAR FUEL SERVICES, INC.
O&M HOLDING COMPANY

INTECH INTERNATIONAL INC.

By: /S/ BENJAMIN H. BASH
Name: Benjamin H. Bash
Title: Assistant Secretary of each of the above

BABCOCK & WILCOX TECHNICAL SERVICES (U.K.) LIMITED

SOFCO - EFS HOLDINGS LLC

By: /s/ BENJAMIN H. BASH
Name: Benjamin H. Bash
Title: Joint Secretary

B&W DE PANAMA, INC.

BABCOCK & WILCOX INTERNATIONAL INVESTMENTS CO., INC.

BABCOCK & WILCOX INVESTMENT COMPANY

By: /S/ MICHAEL S. TAFF

Name: Michael S. Taff

Title: Senior Vice President, of each of the above

BABCOCK & WILCOX INDIA PRIVATE LIMITED

DIAMOND POWER SPECIALTY LIMITED

| By: | /s/ Robert E. Stumpf |
|--------|----------------------|
| Name: | Robert E. Stumpf |
| Title: | Assistant Secretary |
| | |

NATIONAL ECOLOGY COMPANY

 By:
 /S/
 ROBERT E. STUMPF

 Name:
 Robert E. Stumpf

 Title:
 Secretary

DIAMOND POWER DO BRASIL LIMITADA

DIAMOND POWER INTERNATIONAL, INC.

BABCOCK & WILCOX INTERNATIONAL SALES AND SERVICE CORPORATION (as Shareholders of Diamond Power do Brasil Limitada)

By: /S/ ROBERT E. STUMPF
Name: Robert E. Stumpf
Title: Assistant Secretary of each of the Shareholders

P. T. BABCOCK & WILCOX ASIA SERVICIOS DE FABRICACION DE VALLE SOLEADO, S.A. DE C.V.

SERVICOS PROFESIONALES DE VALLE SOLEADO, S.A. DE C.V.

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary

DIAMOND POWER CENTRAL & EASTERN EUROPE S.R.O.

DIAMOND POWER FINLAND OY

By: /S/ JUHA K. MUSTONEN
Name: Juha K. Mustonen
Title: Managing Director

| DIAMOND | POWER | MACHINE | (HUBED) | CO. | . INC |
|---------|-------|---------|---------|-----|-------|

By: /S/ DAVID R. GIBBS
Name: David R. Gibbs
Title: Director

| n | ١T | A | A | Æ | n | N | П | ١. | D | n | 1 | X | 71 | 5 | E | • | C | U | 47 | П | ١, | • | П | 1 | NT. | I | • |
|---|----|---|---|---|---|---|---|----|---|---|---|---|----|---|---|---|---|---|----|---|----|---|---|---|-----|---|---|
| | | | | | | | | | | | | | | | | | | | | | | | | | | | |

By: /S/ MIKA J. HAIKOLA
Name: Mika J. Haikola
Title: Managing Director

 ${\it Signature~Page~to~Group~IV~Agreement}$

ACKNOWLEDGED AND AGREED FOR PURPOSES OF SECTION 2:

MII ENTITIES:

CHARTERING COMPANY (SINGAPORE) PTE. LTD.

J. RAY MCDERMOTT (QINGDAO) PTE.

J. RAY MCDERMOTT ASIA PACIFIC PTE. LTD.

MALMAC SDN. BHD.

MCDERMOTT (MALAYSIA) SENDIRIAN BERHAD

By: /S/ JEFF J. HIGHTOWER

Name: Jeff J. Hightower

Title: Director of each of the above

BOUDIN INSURANCE COMPANY, LTD.

J. RAY MCDERMOTT EASTERN HEMISPHERE LIMITED

LAGNIAPPE INSURANCE COMPANY, LTD.

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Assistant Secretary, of each of the above

J. RAY MCDERMOTT UK LTD.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Joint Secretary

EASTERN MARINE SERVICES, INC.
GLOBAL ENERGY - MCDERMOTT LIMITED

HYDRO MARINE SERVICES, INC.

- J. RAY MCDERMOTT (CASPIAN), INC.
- J. RAY MCDERMOTT (NIGERIA) LTD.
- J. RAY MCDERMOTT (NORWAY), AS
- J. RAY MCDERMOTT CANADA HOLDING, LTD.
- J. RAY MCDERMOTT CANADA, LTD.
- J. RAY MCDERMOTT CONTRACTORS, INC.
- J. RAY MCDERMOTT DE MEXICO, S.A. DE C V

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary of each of the above

- J. RAY MCDERMOTT ENGINEERING SERVICES PRIVATE LIMITED
- J. RAY MCDERMOTT ENGINEERING, LLC
- J. RAY MCDERMOTT FAR EAST, INC.
- J. RAY MCDERMOTT INTERNATIONAL, INC.
- J. RAY MCDERMOTT INTERNATIONAL VESSELS, LTD.
- J. RAY MCDERMOTT INVESTMENTS B.V.
- J. RAY MCDERMOTT KAZAKHSTAN LLP
- J. RAY MCDERMOTT LOGISTIC SERVICES PVT. LIMITED
- J. RAY MCDERMOTT MIDDLE EAST, INC.
- J. RAY MCDERMOTT SOLUTIONS, INC.
- J. RAY MCDERMOTT TECHNOLOGY, INC.
- J. RAY MCDERMOTT UNDERWATER SERVICES, INC.
- J. RAY MCDERMOTT WEST AFRICA HOLDINGS, INC.
- J. RAY MCDERMOTT WEST AFRICA, INC.

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary of each of the above

MCDERMOTT CASPIAN CONTRACTORS, INC.

MCDERMOTT FAR EAST, INC.

MCDERMOTT GULF OPERATING COMPANY, INC.

MCDERMOTT INTERNATIONAL B.V.

MCDERMOTT INTERNATIONAL INVESTMENTS CO., INC.

MCDERMOTT INTERNATIONAL MARINE INVESTMENTS N.V.

MCDERMOTT INTERNATIONAL TRADING CO., INC.

 $\label{eq:mcdermott} \mbox{ MCDERMOTT INTERNATIONAL VESSELS, } \mbox{INC.}$

MCDERMOTT MARINE MEXICO, S.A. DE C.V.

MCDERMOTT OFFSHORE SERVICES COMPANY, INC.

MCDERMOTT OLD JV OFFICE, INC.

MCDERMOTT OVERSEAS INVESTMENT CO. N.V.

MCDERMOTT OVERSEAS, INC.

MCDERMOTT TRADE CORPORATION

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Secretary of each of the above

MENTOR SUBSEA TECHNOLOGY SERVICES, INC.

NORTH ATLANTIC VESSEL, INC.

OFFSHORE PIPELINES INTERNATIONAL, LTD.

OPI VESSELS, INC.

OPMI, LTD.

SABINE RIVER REALTY, INC.

SERVICIOS DE FABRICACION DE ALTAMIRA, S.A. DE C.V.

SERVICOS PROFESIONALES DE ALTAMIRA, S.A. DE C.V.

SPARTEC, INC.

VARSY INTERNATIONAL N.V.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Secretary of each of the above

MCDERMOTT KFT.

By: /S/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: Managing Director

J. RAY MCDERMOTT (AUST.) HOLDING PTY. LIMITED

MCDERMOTT AUSTRALIA PTY. LTD.

By: /S/ LIANE K. HINRICHS

Name: Liane K. Hinrichs

Title: Corporate Officer of each of the above

MCDERMOTT INCORPORATED

MCDERMOTT PANAMA HOLDINGS, S.A.

J. RAY MCDERMOTT HOLDINGS, LLC

J. RAY MCDERMOTT, INC.

J. RAY MCDERMOTT, S.A.

MCDERMOTT CAYMAN LTD.

By: /s/ LIANE K. HINRICHS
Name: Liane K. Hinrichs

Title: Senior Vice President of each of the above

MCDERMOTT INTERNATIONAL MARKETING, INC.

By: /s/ LIANE K. HINRICHS
Name: Liane K. Hinrichs
Title: President

| By: | /s/ Robert E. Stumpf |
|------------------------------|--------------------------------------|
| Name: | Robert E. Stumpf |
| Title: | Assistant Secretary |
| | |
| MCDERMOTT | HOLDINGS (U.K.) LIMITED |
| MCDERMOTT | MARINE CONSTRUCTION LIMITED |
| | |
| By: | /s/ Robert E. Stumpf |
| Name: | Robert E. Stumpf |
| Title: | Joint Secretary of each of the above |
| | |
| MCDERMOTT | SERVICOS DE CONSTRUCAO, LTDA. |
| J. RAY MCDEI | RMOTT INC. |
| MCDERMOTT (as Shareholder | OVERSEAS, INC., s) |
| ` | |
| Ву: | /s/ Robert E. Stumpf |
| Name: | Robert E. Stumpf |
| Title: | Assistant Secretary |
| | |
| OFFSHORE PI | PELINES SDN. BHD. |
| OFFSHORE PI | PELINES SDN. BLD. |
| (as Sole Shareho | older) |
| | |
| Ву: | /s/ Robert E. Stumpf |
| Name: | Robert E. Stumpf |
| Title: | Assistant Secretary |

FLOATEC, LLC

| 1 | T N | רד | PT7 | n | Γ. | Т | T | ^ | | т | ٨ | T | • | 7 | 7 | C | C | T | 1 | 1 | r | n | |
|---|-----|----|-----|----|----|---|---|----|---|----|---|---|-----|---|----|---|---|---|---|---|---|---|----|
| П | H) | NI | ľ | лк | IΑ | | ш | ., | 1 | ١, | 4 | | . 1 | v | H. | | | r | и | | | u | ١. |

 By:
 /S/ STEVEN W. ROLL

 Name:
 Steven W. Roll

 Title:
 Director

| PT. | BAJA | WAHAN | NA IND | ONESIA |
|-----|------|-------|--------|--------|

 By:
 /S/ SCOTT CUMMINS

 Name:
 Scott Cummins

 Title:
 President Director

| CINCA | DODE | LITTA | NGDA | ODTE | ITD |
|--------|------|-------|-------|-------|-----|
| SUNCTA | PURE | HUA | NUTDA | DPIE. | |

 By:
 /s/ ROCKNE L. MOSELEY

 Name:
 Rockne L. Moseley

 Title:
 Director

EXHIBIT I – B&W ENTITIES

See Schedule 1.1(a) and Schedule 1.1(c) attached hereto. No B&W Entity listed on Schedule 1.1(a) shall be deemed to be an "Insured", a "Named Insured" or otherwise be deemed to be insured under any Existing Policy solely by virtue of being listed on such Schedule 1.1(a).

Schedule 1.1(a) B&W FORMER

| Reference ID | Name |
|--------------|---|
| 333 | Ahahsain Hudson Heat Transfer Co. Ltd. |
| 398 | Advanced Refractory Technologies, Inc. |
| | A.M. Lockett & Co., Limited |
| | Amcermet Corporation |
| 924 | ASEA Babcock |
| 235 | Ash Acquisition Company |
| 326 | B & W Clarion, Inc. |
| 574 | B&W Ebensburg Pa., Inc. |
| 460 | B&W Energy Investments, Inc. |
| 383 | B&W Fort Worth Power, Inc. |
| 950 | B&W Fuel Company |
| 535 | B&W Fuel, Inc. |
| 922 | B&W Mexicana, S.A. de C.V. |
| 9991 | B&W North Branch G.P., Inc. |
| 9990 | B&W North Branch L.P., Inc. |
| 537 | B&W Nuclear Service Company |
| 960 | B&W Nuclear Service Company |
| 536 | B&W Nuclear, Inc. |
| 586 | B&W Saba, Inc. |
| 591 | B&W Service Company |
| 579 | B&W SOFC G.P., Inc. |
| 578 | B&W SOFC L.P., Inc. |
| 381 | B&W Special Projects, Inc. |
| 569 | B&W Triso Corporation |
| | B&W Tubular Products Limited |
| 573 | B&W/OHM Weldon Spring, Inc. |
| 212 | Babcock & Wilcox Asia Investment Co., Inc. |
| 115 | Babcock & Wilcox Asia Limited |
| 533 | Babcock & Wilcox Canada Leasing Ltd. |
| 503 | Babcock & Wilcox Canada Ltd. |
| 215 | Babcock & Wilcox China Investment Co., Inc. |
| 594 | Babcock & Wilcox do Brasil Limitada |
| 528 | Babcock & Wilcox do Brasil Participacoes Limitada |
| 206 | Babcock & Wilcox Egypt SAE |

| 169 | Babcock & Wilcox Fibras Ceramicas Limitada |
|------|--|
| 519 | Babcock & Wilcox Finance, Inc. |
| 557 | Babcock & Wilcox Foreign Sales Corporation |
| 175 | Babcock & Wilcox Gama Kazan Teknolojisi A.S. |
| 552 | Babcock & Wilcox General Contracting Company |
| 565 | Babcock & Wilcox Government Services Company |
| 395 | Babcock & Wilcox HRSG Company |
| | Babcock & Wilcox Industries, Ltd. |
| 531 | Babcock & Wilcox International Sales Corporation |
| 549 | Babcock & Wilcox International Sales Corporation |
| 342 | Babcock & Wilcox Investment Company |
| 305 | Babcock & Wilcox Jonesboro Power, Inc. |
| 2007 | Babcock & Wilcox Nevada, LLC |
| | Babcock & Wilcox Refractories Limited |
| 323 | Babcock & Wilcox Salt City Power, Inc. |
| 543 | Babcock & Wilcox Services, Inc. |
| 322 | Babcock & Wilcox Tracy Power, Inc. |
| 314 | Babcock & Wilcox Victorville Power, Inc. |
| 727 | Babcock & Wilcox Volund France SAS |
| 315 | Babcock PFBC, Inc. |
| 559 | Babcock Southwest Construction Corporation |
| 945 | Babcock-Brown Boveri Reaktor GmbH |
| 936 | Babcock-Ultrapower Jonesboro |
| 937 | Babcock-Ultrapower West Enfield |
| 951 | Bailey Beijing Controls Co., Ltd. |
| 516 | Bailey Controls Australia Pty. Limited |
| 517 | Bailey Controls International Sales & Services Company, Inc. |
| 954 | Bailey Controls Jordan for Process Controls Services, Ltd. |
| 563 | Bailey Controls Sales & Service (Australia) Pty. Limited |
| 564 | Bailey Controls Sales & Services Canada Inc. |
| 561 | Bailey do Brasil Instrumentos Industriais Limitada |
| 114 | Bailey International, Inc. |
| 923 | Bailey Japan Company Limited |
| 542 | Bailey Meter and Controls Company |
| | Bailey Meter Company |
| 562 | Bailey Meter Co. (Japan) Ltd. |
| | Bailey Meter Company Limited |
| | Bailey Meter GmbH |
| | |

| 646 | Brick Insurance Company, Ltd. |
|------|--|
| 590 | BWXT Protec, Inc. |
| | C.C. Moore & Company Engineers |
| 511 | Ceramatec G.P., Inc. |
| 510 | Ceramatec SOFC, Inc. |
| 329 | Clarion Energy, Inc. |
| 328 | Clarion Power Company |
| 321 | Conam Nuclear, Inc. |
| | Control Components France |
| 514 | Control Components Italy S.R.L. |
| | Control Components, Inc. (California) |
| | Control Components, Inc. (Delaware) |
| 948 | Control Components Japan |
| 1914 | CTR Solutions, LLC |
| 545 | Detroit Broach & Machine Corporation |
| 551 | Diamond Blower Company Limited |
| | Diamond Canapower Ltd. |
| 518 | Diamond Power Importacao e Exportacao Ltda. |
| 144 | Diamond Power Korea Inc. |
| 526 | Diamond Power Specialty (Japan) Ltd. |
| 558 | Diamond Power Specialty (Proprietary) Limited |
| 546 | Diamond Power Specialty Corporation (Delaware) |
| | Diamond Power Speciality Corporation (Ohio) |
| 529 | Diamond Power Specialty GmbH |
| | Diescher Tube Mills, Inc. |
| 332 | Ebensburg Energy, Inc. |
| 397 | Ejendomsaktieselskabet Falkevej2 |
| 961 | Enserch Environmental Management Company, Inc. |
| 968 | EPC Business Trust |
| 919 | Especialidades Termomecanicas, S.A. de C.V. |
| 550 | Ferry-Diamond Engineering Company Limited |
| 928 | Fibras Ceramicas C.A. |
| 509 | Fibras Ceramicas, Inc. |
| 547 | Globe Steel Tubes Corporation |
| | Greer Land Co. |
| | Holmes Insulations Limited |
| 124 | Hudson Heat Transfer International, Inc. |
| 1955 | Hudson HEI Pty. Ltd. |
| | - |

| 914 | Hudson Northern Industries Inc. |
|------|--|
| 125 | Hudson Products Aktiebolag |
| 567 | Hudson Products Corporation (1) |
| 902 | Hudson Products de Mexico, S.A. de C.V. |
| 1907 | Integran Technologies Inc. |
| 461 | International Disarmament Corporation |
| 941 | Isolite Babcock Refractories Company, Ltd. |
| 927 | Isolite Eastern Union Refractories Co., Ltd. |
| 920 | KBW Gasification Systems, Inc. |
| 512 | LT Produkter i Skutskar AB |
| 938 | Maine Power Services |
| 345 | McDermott Heat Transfer Company |
| 344 | McDermott Productos Industriales de Mexico, S.A. de C.V. |
| 946 | Medidores Bailey, S.A. de C.V. |
| 942 | Morganite Ceramic Fibres Limited |
| 943 | Morganite Ceramic Fibres Pty. Limited |
| 944 | Morganite Ceramic Fibres S. A. |
| | National Drill & Manufacturing Co. |
| 544 | National Ecology (Alabama) Incorporated |
| 575 | National Ecology (Utah) Incorporated |
| 976 | Nooter/Eriksen - Babcock & Wilcox, L.L.C. |
| 933 | North American CWF Partnership |
| 9989 | North Branch Power Company L.P. |
| 971 | North County Operations Associates |
| | Nuclear Materials and Equipment Corporation |
| 958 | Olin Pantex Inc. |
| 1153 | P. T. Heat Exchangers Indonesia |
| 934 | Palm Beach Energy Associates |
| ,,,, | Piedmont Tool Machine Company |
| 576 | Power Computing Company |
| 581 | Power Systems Sunnyside Operations GP, Inc. |
| 583 | Power Systems Sunnyside Operations LP, Inc. |
| | · · · |
| | (1) a Delaware Corporation |
| 905 | PowerSafety International, Inc. |
| 508 | Productos de Caolin, Inc. |
| 577 | PSO Caribbean, Inc. |
| 1980 | Savannah River Alliance LLC |
| | |

| 2008 | Savannah River Tactical Services LLC |
|------|--------------------------------------|
| 984 | SOFCo L.P. |
| 932 | South Point CWF |
| 556 | Sunland Construction Co., Inc. |
| 988 | Sunnyside Cogeneration Associates |
| 582 | Sunnyside II, Inc. |
| 992 | Sunnyside II, L.P |
| 303 | Sunnyside III, Inc. |
| 993 | Sunnyside Operations Associates L.P. |
| 571 | Termobloc Industria E Comercio Ltda. |
| 953 | Thermax Babcock & Wilcox Limited |
| 502 | TLT-Babcock, Inc. |
| 570 | Triso |
| 1152 | W.E. Smith Hudson Pty. Ltd. |
| 548 | W. F. and John Barnes Company |
| | |

Schedule 1.1(c) B&W CURRENT

| Reference ID | Name | |
|--------------|--|---|
| 553 | Americon Equipment Services, Inc. | |
| 554 | Americon, Inc. | |
| 732 | Applied Synergistics, Inc. | |
| 127 | B&W de Panama, Inc. | |
| 532 | Babcock & Wilcox Canada Ltd. | |
| 1570 | Babcock & Wilcox China Holdings, Inc. | |
| 555 | Babcock & Wilcox Construction Co., Inc. | |
| 2011 | Babcock & Wilcox de Monterrey, S.A. de C.V. | |
| 1571 | Babcock & Wilcox Denmark Holdings, Inc. | |
| 327 | Babcock & Wilcox Ebensburg Power, Inc. | |
| 302 | Babcock & Wilcox Equity Investments, Inc. | |
| 1965 | Babcock & Wilcox Holdings, Inc. | |
| 2028 | Babcock & Wilcox India Holdings, Inc. | |
| 598 | Babcock & Wilcox India Private Limited | |
| 126 | Babcock & Wilcox International Investments Co., Inc. | |
| 530 | Babcock & Wilcox International Sales and Service Corporation | |
| 541 | Babcock & Wilcox International, Inc. | |
| 380 | Babcock & Wilcox Investment Company | |
| 2010 | Babcock & Wilcox Modular Nuclear Energy LLC | |
| 1974 | Babcock & Wilcox Nuclear Operations Group, Inc. | |
| 1967 | Babcock & Wilcox Nuclear Energy, Inc. | (formerly Babcock & Wilcox Nuclear Power Generation Group, Inc. |
| 2018 | Babcock & Wilcox Nuclear Services (U.K.) Limited | |
| 500 | Babcock & Wilcox Power Generation Group, Inc. | |
| 1961 | Babcock & Wilcox Technical Services (U.K.) Limited | |
| 1970 | Babcock & Wilcox Technical Services Clinch River, LLC | |
| 572 | Babcock & Wilcox Technical Services Group, Inc. | (formerly BWXT Services, Inc.) |
| 587 | Babcock & Wilcox Technical Services Savannah River Company | (formerly BWXT Savannah River Company) |
| 599 | Babcock & Wilcox Volund A/S | (formerly Babcock & Wilcox Volund ApS) |
| 2000 | BCE Parts Ltd. | |
| 596 | BWX Technologies, Inc. | |
| 580 | BWXT Federal Services, Inc. | |
| 589 | BWXT Hanford Company | |
| 382 | BWXT of Idaho, Inc. | |
| 592 | BWXT of Ohio, Inc. | |
| | | |

| 1576 | BWXT Washington, Inc. | |
|------|---|---------------------------------------|
| 189 | Creole Insurance Company, Ltd. | |
| 1988 | Delta Power Services, LLC | |
| 766 | Diamond Operating Co., Inc. | |
| 1572 | Diamond Power Australia Holdings, Inc. | |
| 1984 | Diamond Power Central & Eastern Europe s.r.o. | |
| 1573 | Diamond Power China Holdings, Inc. | |
| 521 | Diamond Power do Brasil Limitada | |
| 1574 | Diamond Power Equity Investments, Inc. | |
| 525 | Diamond Power Finland OY | |
| 504 | Diamond Power Germany GmbH | (formerly Diamond Power - Sturm GmbH) |
| 597 | Diamond Power International, Inc. | |
| 949 | Diamond Power Machine (Hubei) Co., Inc. | |
| 1908 | Diamond Power Services S.E.A. Ltd. | |
| 522 | Diamond Power Specialty (Proprietary) Limited | |
| 523 | Diamond Power Specialty Limited | |
| 524 | Diamond Power Sweden AB | |
| 1990 | DPS Berkeley, LLC | |
| 1997 | DPS Cadillac, LLC | |
| 1995 | DPS Florida, LLC | |
| 1993 | DPS Gregory, LLC | |
| 1996 | DPS Lowell Cogen, LLC | |
| 1992 | DPS Mecklenburg, LLC | |
| 1991 | DPS Michigan, LLC | |
| 1994 | DPS Mojave, LLC | |
| 1998 | DPS Sabine, LLC | |
| 278 | Gumbo Insurance Company, Ltd. | |
| 2003 | Intech International Inc. | |
| 2002 | Intech, Inc. | |
| 2001 | Ivey-Cooper Services, L.L.C. | |
| 1968 | Marine Mechanical Corporation | |
| 540 | National Ecology Company | |
| 2005 | NFS Holdings, Inc. | |
| 2004 | NOG-Erwin Holdings, Inc. | |
| 501 | North County Recycling, Inc. | |
| 2006 | Nuclear Fuel Services, Inc. | |
| 1989 | O&M Holding Company | |
| 707 | P. T. Babcock & Wilcox Asia | |
| | | |

| 534 | Palm Beach Resource Recovery Corporation | |
|------|---|--|
| 560 | Power Systems Operations, Inc. | |
| 568 | Revloc Reclamation Service, Inc. | |
| 2013 | Servicios de Fabricacion de Valle Soleado, S.A. de C.V. | |
| 2012 | Servicios Profesionales de Valle Soleado, S.A. de C.V. | |
| 767 | SOFCo - EFS Holdings LLC | |
| 2029 | The Babcock & Wilcox Company | |
| | | |

EXHIBIT II – MII ENTITIES

| See Schedule 1.1(b) and Schedule 1.1(j) attached hereto. No MII Entity listed on Schedule 1.1(b) shall be deemed to be an "Insured", a "Named Insured" or otherwise |
|---|
| be deemed to be insured under any Existing Policy solely by virtue of being listed on such Schedule 1.1(b). |

Schedule 1.1(b) MII FORMER

| Reference ID | | |
|--------------|---|------------------------------|
| 947 | Arabian General Contracting Company | |
| 911 | Arabian Petroleum Marine Construction Company | |
| 8000 | Associated Pipe Line Contractors, Inc. | |
| 117 | Badya Builders, Inc. | |
| 242 | Beheer-en Beleggingsmaatschappij Belesa B.V. | |
| 9996 | BJA-Mentor Production Systems Limited | |
| 164 | Brown & Root McDermott Fabricators Limited | |
| 277 | Cayenne Insurance Company, Ltd. | |
| 629 | CCC Fabricaciones y Construcciones S.A. de C.V. | |
| 940 | Construcciones Maritimas Mexicanas, S.A. de C.V. | |
| 952 | Davy McDermott Limited | |
| 921 | DB/McDermott Company | |
| 1952 | Deep Gulf Contractors LLC | |
| 171 | Deep Sea Divers Australia Pty. Limited | |
| | Dick Evans, Inc. | |
| 997 | ETPM International (UK) Limited | |
| 955 | ETPM International S.A.S. | |
| 999 | Far East Vessels, Inc. | |
| 108 | First Emirates Trading Corporation | |
| | Harvey Lumber & Supply Company | |
| 174 | Heavy Lift Chartering, Inc. | |
| 935 | HeereMac v.o.f. | |
| 956 | Heerema-McDermott (Aust.) Pty. Ltd. | |
| 190 | Honore Insurance Company, Ltd. | |
| 917 | Initec, Astano y McDermott International Inc., S.A. | |
| | Ingenieria Petrolera Maritima, S.A. de C.V. | |
| | Iranian Marine Contractors, Inc. | |
| 102 | J. Ray McDermott & Co. (Nederland) B.V. | |
| 150 | J. Ray McDermott (Aust.) Pty. Ltd. | |
| 151 | J. Ray McDermott (U.K.) Inc. | |
| 350 | J. Ray McDermott (Venezuela) C.A. | |
| 325 | J. Ray McDermott Arctic Ltd. | |
| 307 | J. Ray McDermott Gulf Contractors, Inc. | (formerly the Early Company) |

| 394 | J. Ray McDermott International Services (No. 1) Limited |
|------|---|
| 101 | J. Ray McDermott International Services Limited |
| 152 | J. Ray McDermott London, Ltd. |
| 641 | J. Ray McDermott Marine Construction AS |
| 1125 | J. Ray McDermott Newfoundland, Ltd. |
| 871 | J. Ray McDermott Offshore Chartering (U.S.), Inc. |
| 372 | J. Ray McDermott Properties, Inc. |
| 238 | J. Ray McDermott Underwater Services, Inc. |
| 603 | J. Ray McDermott Venture Holdings (U.S.), Inc. |
| 153 | Jaramac Associated Services, Ltd. |
| | Jaramac Petroleum (U.K.) Limited |
| | Khafji Contractors, Inc. |
| 166 | Khor Contractors, Inc. |
| 903 | KME a.s. |
| 324 | Lan-Dermott, S.A. de C.V. |
| 202 | Madock Shipyard Company, Ltd. |
| 203 | Macshelf Marine Construction Company, Ltd. |
| 106 | Mandato Pty. Limited |
| 360 | Marine Contractors, Inc. |
| 146 | McAntille, N.V. |
| 168 | McDermott - ETPM V.O.F. |
| 158 | McDermott (Holland) B.V. |
| 304 | McDermott (Nigeria) Limited |
| 245 | McDermott Acquisition Company, Inc. |
| 972 | McDermott APG Services Limited |
| 352 | McDermott Azerbaijan Pipelines, Inc. |
| 156 | McDermott Azerbaijan, Inc. |
| 351 | McDermott Central & Eastern Europe, Inc. |
| | McDermott de Mexico, S.A. de C.V. |
| 118 | McDermott Denmark ApS |
| 464 | McDermott Energy Services, Inc. |
| 178 | McDermott Engineering (Europe) Limited |
| | McDermott Enterprises France, S.A. |
| 128 | McDermott Europe Marine Services, Ltd. |
| 313 | McDermott Industries, Inc. |
| 354 | McDermott Inland Services, Inc. |
| 201 | McDermott International Asset Management, Ltd. |
| 343 | McDermott International Aviation, Inc. |
| | |

| 346 | McDermott International Aviation, Inc. |
|------|--|
| | McDermott International (Deutschland) GmbH |
| 237 | McDermott International Engineering & Construction Co., Ltd. |
| 139 | McDermott International Engineering Investments N.V. |
| 121 | McDermott International General Services, Inc. |
| 123 | McDermott International Marine Services, Inc. |
| 236 | McDermott International Marketing, Inc. |
| 143 | McDermott Marine UK Limited |
| 462 | McDermott Marketing Services, Inc. |
| 103 | McDermott Middle East Trading, Ltd. |
| 172 | McDermott Neutral Zone, Inc. |
| | McDermott Norway, Inc. |
| | McDermott Petroleum (Iran), Inc. |
| 105 | McDermott Sakhalin, Inc. |
| 347 | McDermott Shipbuilding, Inc. |
| 111 | McDermott Singapore Pte. Ltd. |
| 221 | McDermott Submarine Cable Ltd. |
| 1951 | McDermott Submarine Cable Systems Limited |
| 205 | McDermott Subsea Constructors Limited |
| 110 | McDermott Transition Co., Inc. |
| | McDermott UAR, Inc. |
| 184 | McDermott West Indies Company |
| 187 | McDermott-ETPM (Norway), Inc. |
| 998 | McDermott-ETPM (UK) Limited |
| 963 | McDermott-ETPM B.V. |
| 161 | McDermott-ETPM East B.V. |
| 167 | McDermott-ETPM East N.V. |
| 964 | McDermott-ETPM N.V. |
| 965 | McDermott-ETPM S.A. |
| 962 | McDermott-ETPM West, Inc. |
| 104 | Menck GmbH |
| 463 | Mentor Engineering Consultants Limited |
| 131 | MIMCO, Inc. |
| 908 | MMC-McDermott Engineering Sdn. Berhad |
| 317 | Mofco N.V. |
| 129 | NOLA, Servicos E Participacoes Ltda. |
| 141 | Northern Marine Services, Inc. |
| 107 | Oceanic Red Sea Company |
| | |

(formerly McDermott International Aviation Co., Inc.)

| 620 | Offshore Angola, Ltd. |
|-----|---|
| 632 | Offshore Energy Capital Corporation |
| 622 | Offshore Hyundai International Limited |
| 630 | Offshore Hyundai International, Ltd. |
| 210 | Offshore Marine Chartering N.V. |
| 610 | Offshore Petroleum Divers, Inc. |
| 614 | Offshore Pipelines (Mauritius) Ltd. |
| 616 | Offshore Pipelines Far East Limited |
| 617 | Offshore Pipelines International Gulf E.C. |
| 633 | Offshore Power Generation Ltd. |
| 635 | Offshore Production Vessels, Ltd. |
| 615 | OPI Offshore Netherlands Holding B.V. |
| 611 | OPI Sales & Service Corporation |
| 628 | OPI Towing & Supply, L.L.C. |
| 624 | OPMI, E.C. |
| 613 | P. T. Armandi Pranaupaya |
| 850 | Panama Offshore Chartering Company 1, Inc. |
| 859 | Panama Offshore Chartering Company 10, Inc. |
| 860 | Panama Offshore Chartering Company 11, Inc. |
| 861 | Panama Offshore Chartering Company 12, Inc. |
| 862 | Panama Offshore Chartering Company 13, Inc. |
| 863 | Panama Offshore Chartering Company 14, Inc. |
| 864 | Panama Offshore Chartering Company 15, Inc. |
| 865 | Panama Offshore Chartering Company 16, Inc. |
| 866 | Panama Offshore Chartering Company 17, Inc. |
| 867 | Panama Offshore Chartering Company 18, Inc. |
| 868 | Panama Offshore Chartering Company 19, Inc. |
| 851 | Panama Offshore Chartering Company 2, Inc. |
| 869 | Panama Offshore Chartering Company 20, Inc. |
| 870 | Panama Offshore Chartering Company 21, Inc. |
| 852 | Panama Offshore Chartering Company 3, Inc. |
| 853 | Panama Offshore Chartering Company 4, Inc. |
| 854 | Panama Offshore Chartering Company 5, Inc. |
| 855 | Panama Offshore Chartering Company 6, Inc. |
| 856 | Panama Offshore Chartering Company 7, Inc. |
| 857 | Panama Offshore Chartering Company 8, Inc. |
| 858 | Panama Offshore Chartering Company 9, Inc. |
| 154 | Panama Offshore Chartering Company, Inc. |
| | |

| 990 | Personal Administrativo de Vera Cruz, S.A. de C.V. |
|-----|--|
| 991 | Personal Ejecutivo de Veracruz, S.A. de C.V. |
| 645 | Pirogue Insurance Company, Ltd. |
| 981 | Sakhalin Energy Investment Company, Ltd. |
| 384 | Samburg Holdings, Inc. |
| 109 | Sharman, Allen, Gay & Taylor, S.A. |
| 989 | Tallares Navales del Golfo, S.A. de C.V. |
| | The Evelyn Kay Company |
| | The Roger Thomas Company |
| | The Rosalie E. Company |
| | The Walter E. Company |
| 634 | TL Marine Sdn. Bhd. |
| 621 | TL Offshore Sdn. Bhd. |
| 926 | Topside Contractors of Newfoundland |
| 925 | Topside Contractors of Newfoundland, Ltd. |
| 642 | U.S. Offshore Chartering |
| 918 | Universal Fabricators Incorporated |
| 980 | US Shipbuilding Corporation, Inc. |
| 318 | Wagley, Inc. |
| | |

Schedule 1.1(j) MII - CURRENT

| CURRENT | | | |
|------------------------------------|---|--|--|
| Reference ID | Name | | |
| 873 | Barmada McDermott (L) Limited | | |
| 1953 | Barmada McDermott Sdn. Bhd. | | |
| 276 | Boudin Insurance Company, Ltd. | | |
| 939 | Caspian Offshore Fabricators LLC | | |
| 116 | Chartering Company (Singapore) Pte. Ltd. | | |
| 142 | Eastern Marine Services, Inc. | | |
| 249 | Global Energy - McDermott Limited | | |
| 140 | Hydro Marine Services, Inc. | | |
| 649 | International Vessels Ltd. | | |
| 148 | J. Ray McDermott (Aust.) Holding Pty. Limited | | |
| 1982 | J. Ray McDermott (Caspian), Inc. | (formerly J. Ray McDermott Diving International, Inc.) | |
| 243 | J. Ray McDermott (Nigeria) Ltd. | | |
| 2025 | J. Ray McDermott (Norway), AS | | |
| 1237 | J. Ray McDermott (Qingdao) Pte. Ltd. | | |
| 194 | J. Ray McDermott Asia Pacific Pte. Ltd. | (formerly McDermott South East Asia Pte. Ltd.) | |
| 1972 | J. Ray McDermott Canada Holding, Ltd. | | |
| 1969 | J. Ray McDermott Canada, Ltd. | | |
| J. Ray McDermott Contractors, Inc. | | | |
| | J. Ray McDermott de Mexico, S.A. de C.V. | | |
| 208 | J. Ray McDermott Eastern Hemisphere Limited | (formerly J. Ray Middle East (Indian Ocean) Ltd.) | |
| 1956 | J. Ray McDermott Engineering Services Private Limited | | |
| 1950 | J. Ray McDermott Engineering, LLC | | |
| 220 | J. Ray McDermott Far East, Inc. | | |
| 211 | J. Ray McDermott Holdings, LLC | (formerly J. Ray McDermott Holdings, Inc.) | |
| 627 | J. Ray McDermott International Vessels, Ltd. | (formerly OPI International, Ltd.) | |
| 602 | J. Ray McDermott International, Inc. | (2 1 1 2 1 2 1) | |
| 160 | J. Ray McDermott Investments B.V. | (formerly Menck B.V.) | |
| 1958 | J. Ray McDermott Kazakhstan Limited Liability Partnership | | |
| 1975 | J. Ray McDermott Logistic Services Pvt. Limited | | |
| 136 | J. Ray McDermott Middle East, Inc. | | |
| 239 | J. Ray McDermott Solutions, Inc. | (formerly J. Ray McDermott Engineering Holdings, Inc.) | |
| 625 | J. Ray McDermott Technology, Inc. | | |
| 1957 | J. Ray McDermott UK Ltd. | | |
| 1960 | J. Ray McDermott Underwater Services, Inc. | | |

| 248 | J. Ray McDermott West Africa Holdings, Inc. | |
|------|---|--|
| 244 | J. Ray McDermott West Africa, Inc. | |
| 601 | J. Ray McDermott, Inc. | |
| 207 | J. Ray McDermott, S.A. | |
| 644 | Lagniappe Insurance Company, Ltd. | |
| 193 | Malmac Sdn. Bhd. | |
| 132 | McDermott (Malaysia) Sendirian Berhad | |
| 904 | McDermott Abu Dhabi Offshore Construction Company | |
| 915 | McDermott Arabia Company Limited | |
| 149 | McDermott Australia Pty. Ltd. | (formerly McDermott Industries (Aust.) Pty. Limited) |
| 183 | McDermott Azerbaijan Marine Construction, Inc. | |
| 241 | McDermott Caspian Contractors, Inc. | |
| 275 | McDermott Cayman Ltd. | |
| 181 | McDermott Far East, Inc. | |
| 133 | McDermott Gulf Operating Company, Inc. | |
| 177 | McDermott Holdings (U.K.) Limited | |
| 300 | McDermott Incorporated | |
| 147 | McDermott International B.V. | |
| 120 | McDermott International Investments Co., Inc. | |
| 119 | McDermott International Marine Investments N.V. | |
| 213 | McDermott International Marketing, Inc. | (formerly McDermott International Beijing, Inc.) |
| 130 | McDermott International Trading Co., Inc. | |
| 1971 | McDermott International Vessels, Inc. | |
| 100 | McDermott International, Inc. | |
| 1962 | McDermott Kft. | |
| 349 | McDermott Marine Construction Limited | |
| 1986 | McDermott Marine Mexico, S.A. de C.V. | |
| 135 | McDermott Offshore Services Company, Inc. | |
| 176 | McDermott Old JV Office, Inc. | |
| 162 | McDermott Overseas Investment Co. N.V. | |
| 390 | McDermott Overseas, Inc. | |
| 1966 | McDermott Panama Holdings, S.A. | |
| 170 | McDermott Servicos de Construcao, Ltda. | |
| 320 | McDermott Trade Corporation | |
| 306 | Mentor Subsea Technology Services, Inc. | |
| 138 | North Atlantic Vessel, Inc. | |
| 2021 | North Ocean II AS | |
| 2023 | North Ocean II KS | |
| | | |

| 2022 | North Ocean V AS | |
|------|--|--|
| 612 | Offshore Pipelines International, Ltd. | |
| 618 | Offshore Pipelines Nigeria Limited | |
| 619 | Offshore Pipelines Sdn. Bhd. | |
| 626 | OPI Vessels, Inc. | |
| 623 | OPMI, Ltd. | |
| 185 | P. T. Bataves Fabricators | |
| 180 | P. T. McDermott Indonesia | |
| 901 | PT. Baja Wahana Indonesia | (formerly P.T. Babcock & Wilcox Indonesia) |
| 643 | PT. J. Ray McDermott Indonesia | (formerly PT. Jay Ray) |
| 636 | Sabine River Realty, Inc. | |
| 1977 | Servicios de Fabricacion de Altamira, S.A. de C.V. | |
| 1976 | Servicos Profesionales de Altamira, S.A. de C.V. | |
| 2019 | Singapore Huangdao Pte. Ltd. | |
| 640 | SparTEC, Inc. | (formerly J. Ray McDermott Spars, Inc.) |
| 145 | Varsy International N.V. | |

| Published | CUSIP Number: | |
|-----------|---------------|--|
| Published | COSIP Number: | |

CREDIT AGREEMENT

Dated as of May 3, 2010

among

BABCOCK & WILCOX INVESTMENT COMPANY (or, after the Spinoff, THE BABCOCK & WILCOX COMPANY),

as the Borrower,

BANK OF AMERICA, N.A.,

as Administrative Agent, Swing Line Lender and an L/C Issuer,

and

The Other Lenders Party Hereto

BNP PARIBAS,

as Syndication Agent

and

JPMORGAN CHASE BANK, N.A., WELLS FARGO BANK, N.A., BBVA COMPASS, and CRÉDIT AGRICOLE CORPORATE AND INVESTMENT BANK,

as Documentation Agents

BANC OF AMERICA SECURITIES LLC, BNP PARIBAS SECURITIES CORP. and J.P. MORGAN SECURITIES INC.,

as Joint Lead Arrangers and Joint Book Managers

TABLE OF CONTENTS

| Section | Pag |
|--|----------------|
| ARTICLE I. DEFINITIONS AND ACCOUNTING TERMS 1.01 Defined Terms 1.02 Other Interpretive Provisions 1.03 Accounting Terms | 30 30 |
| 1.04 Rounding 1.05 Exchange Rates; Currency Equivalents 1.06 Alternative Currencies | 33 |
| 1.06 Alternative Currencies 1.07 Times of Day 1.08 Letter of Credit Amounts | 33 33 33 |
| ARTICLE II. THE COMMITMENTS AND CREDIT EXTENSIONS 2.01 Committed Loans | 39 |
| 2.02 Borrowings, Conversions and Continuations of Committed Loans 2.03 Letters of Credit | 39 |
| 2.04 Swing Line Loans 2.05 Prepayments | 55 54 |
| 2.06 Termination or Reduction of Commitments 2.07 Repayment of Loans 2.08 Interest | 5: 5: 5: |
| 2.09 Fees 2.10 Computation of Interest and Fees | 5° 5° 5° |
| 2.11 Evidence of Debt 2.12 Payments Generally; Administrative Agent's Clawback | 58 58 |
| 2.13 Sharing of Payments by Lenders 2.14 Increase in Commitments 2.15 Cash Collateral | 60 |
| 2.15 Cash Collateral 2.16 Defaulting Lenders 2.17 Borrower Succession | 62 63 64 |
| ARTICLE III. TAXES, YIELD PROTECTION AND ILLEGALITY 3.01 Taxes | 65 |
| 3.02 Illegality 3.03 Inability to Determine Rates | 69 70 |
| 3.04 Increased Costs; Reserves on Eurocurrency Rate Loans 3.05 Compensation for Losses | 7 72 |
| 3.06 Mitigation Obligations; Replacement of Lenders 3.07 Survival | 73 73 |
| ARTICLE IV. CONDITIONS PRECEDENT TO CREDIT EXTENSIONS 4.01 Conditions of Initial Credit Extension | 7. 7. |
| 4.02 Conditions to all Credit Extensions | 7′ |
| ARTICLE V. REPRESENTATIONS AND WARRANTIES 5.01 Corporate Existence, Compliance with Law 5.02 Corporate Power: Authorization: Enforceable Obligations | 78 78 71 |

i

TABLE OF CONTENTS (continued)

| | <u>Section</u> | Page |
|-------|--|------|
| | 5.03 Ownership of Borrower; Subsidiaries | 80 |
| | 5.04 Financial Statements | 80 |
| | 5.05 Material Adverse Change | 81 |
| | 5.06 Solvency | 81 |
| | 5.07 Litigation | 81 |
| | 5.08 Taxes | 81 |
| | 5.09 Full Disclosure | 82 |
| | 5.10 Margin Regulations | 82 |
| | 5.11 No Burdensome Restrictions; No Defaults | 82 |
| | 5.12 Investment Company Act | 82 |
| | 5.13 Use of Proceeds | 82 |
| | 5.14 Insurance | 83 |
| | 5.15 Labor Matters | 83 |
| | 5.16 ERISA | 83 |
| | 5.17 Environmental Matters | 84 |
| | 5.18 Intellectual Property | 85 |
| | 5.19 Title; Real Property | 85 |
| | 5.20 Security Instruments | 86 |
| ARTIC | CLE VI. AFFIRMATIVE COVENANTS | 87 |
| | 6.01 Financial Statements | 87 |
| | 6.02 Collateral Reporting Requirements | 88 |
| | 6.03 Default and certain other Notices | 89 |
| | 6.04 Litigation | 89 |
| | 6.05 Labor Relations | 89 |
| | 6.06 Tax Returns | 90 |
| | 6.07 Insurance | 90 |
| | 6.08 ERISA Matters | 90 |
| | 6.09 Environmental Matters | 91 |
| | 6.10 Patriot Act Information | 91 |
| | 6.11 Other Information | 92 |
| | 6.12 Preservation of Corporate Existence, Etc. | 92 |
| | 6.13 Compliance with Laws, Etc. | 92 |
| | 6.14 Conduct of Business | 92 |
| | 6.15 Payment of Taxes, Etc. | 92 |
| | 6.16 Maintenance of Insurance | 92 |
| | 6.17 Access | 93 |
| | 6.18 Keeping of Books | 93 |
| | 6.19 Maintenance of Properties, Etc. | 93 |
| | 6.20 Application of Proceeds | 93 |
| | 6.21 Environmental | 93 |
| | 6.22 Additional Collateral and Guaranties | 95 |
| | 6.23 Real Property | 97 |
| | 6.24 Replacement of Borrower | 97 |
| | 6.25 BWXT Entities | 98 |
| | 6.26 Further Assurances | 98 |

TABLE OF CONTENTS (continued)

| Section | Pag |
|---|--|
| 6.27 Post-Closing Deliveries6.28 Cash Collateralization of Extended Letters of Credit6.29 Post-Closing Real Property Matters | 99 99 90 |
| ARTICLE VII. NEGATIVE COVENANTS 7.01 Indebtedness 7.02 Liens 7.03 Investments 7.04 Asset Sales 7.05 Restricted Payments 7.06 Fundamental Changes 7.07 Change in Nature of Business 7.08 Transactions with Affiliates 7.09 Burdensome Agreements 7.10 Amendments of Constituent Documents 7.11 Accounting Changes 7.12 Use of Proceeds 7.13 Sale Leasebacks 7.14 Capital Expenditures 7.15 Cancellation of Indebtedness Owed To It 7.16 No Speculative Transactions 7.17 Post-Termination Benefits 7.18 Financial Covenants 7.19 BWXT Ownership | 100 101 102 102 102 100 100 100 100 100 |
| ARTICLE VIII. EVENTS OF DEFAULT AND REMEDIES 8.01 Events of Default 8.02 Remedies Upon Event of Default 8.03 Application of Funds | 109 109 111 112 |
| ARTICLE IX. ADMINISTRATIVE AGENT 9.01 Appointment and Authority 9.02 Rights as a Lender 9.03 Exculpatory Provisions 9.04 Reliance by Administrative Agent 9.05 Delegation of Duties 9.06 Resignation of Administrative Agent 9.07 Non-Reliance on Administrative Agent and Other Lenders 9.08 No Other Duties, Etc. 9.09 Administrative Agent May File Proofs of Claim 9.10 Collateral and Guaranty Matters 9.11 Secured Cash Management Agreements and Secured Hedge Agreements | 113 114 114 114 115 115 116 116 117 117 |
| ARTICLE X. MISCELLANEOUS 10.01 Amendments, Etc. 10.02 Notices; Effectiveness; Electronic Communication 10.03 No Waiver; Cumulative Remedies; Enforcement 10.04 Expenses; Indemnity; Damage Waiver | 119 119 120 122 123 |

TABLE OF CONTENTS (continued)

| Section | Page |
|---|------|
| 10.05 Payments Set Aside | 125 |
| 10.06 Successors and Assigns | 125 |
| 10.07 Treatment of Certain Information; Confidentiality | 130 |
| 10.08 Right of Setoff | 132 |
| 10.09 Interest Rate Limitation | 132 |
| 10.10 Counterparts; Integration; Effectiveness | 132 |
| 10.11 Survival of Representations and Warranties | 133 |
| 10.12 Severability | 133 |
| 10.13 Replacement of Lenders | 133 |
| 10.14 Governing Law; Jurisdiction; Etc. | 134 |
| 10.15 Waiver of Jury Trial | 135 |
| 10.16 No Advisory or Fiduciary Responsibility | 135 |
| 10.17 Electronic Execution of Assignments and Certain Other Documents | 136 |
| 10.18 Judgment Currency | 136 |
| 10.19 Release and Reinstatement of Collateral | 136 |
| SIGNATURES | S-1 |

SCHEDULES

| 1.01(a) | Affiliate Agreements |
|-------------|--|
| 1.01(b) | Existing Letters of Credit |
| 1.01(c) | Initial Guarantors |
| 2.01 | Commitments and Applicable Percentages |
| 4.01(a)(iv) | Mortgaged Properties |
| 5.02 | Consents |
| 5.03 | Ownership of Subsidiaries |
| 5.04 | Supplement to Financial Statements |
| 5.07 | Litigation |
| 5.15 | Labor Matters |
| 5.16 | ERISA |
| 5.17 | Environmental Matters |
| 5.19(b) | Real Property |
| 7.01 | Existing Indebtedness |
| 7.02 | Existing Liens |
| 7.03 | Existing Investments |
| 10.02 | Administrative Agent's Office; Certain Addresses for Notices |
| | |

EXHIBITS

Form of

- Committed Loan Notice Swing Line Loan Notice Note Compliance Certificate Assignment and Assumption Administrative Questionnaire Guaranty Collateral Agreement Global Intercompany Note A B C D E-1 E-2
- G H

CREDIT AGREEMENT

This CREDIT AGREEMENT is entered into as of May 3, 2010, among BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation, as the borrower hereunder (or, after the effectiveness of the Spinoff (defined below) and the satisfaction of the other terms and conditions herein relating to the substitution thereof, the New Borrower (defined below) as the borrower hereunder), each lender from time to time party hereto (collectively, the "Lenders" and individually, a "Lender"), and BANK OF AMERICA, N.A., as Administrative Agent, Swing Line Lender and an L/C Issuer.

The Borrower has requested that the Lenders provide a revolving credit facility, and the Lenders are willing to do so on the terms and conditions set forth herein.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

ARTICLE I. DEFINITIONS AND ACCOUNTING TERMS

- 1.01 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth below:
- "Administrative Agent" means Bank of America in its capacity as administrative agent under any of the Loan Documents, or any successor administrative agent.
- "Administrative Agent's Office" means the Administrative Agent's address and, as appropriate, account as set forth on Schedule 10.02 or such other address or account as the Administrative Agent may from time to time notify to the Borrower and the Lenders.
- "Administrative Questionnaire" means an Administrative Questionnaire in substantially the form of Exhibit E-2 or any other form approved by the Administrative Agent.
- "Affiliate" means, with respect to any Person, any other Person, directly or indirectly, controlling or that is controlled by or is under common control with such Person, each officer, director or general partner of such Person, and each Person that is the beneficial owner of 10% or more of any class of Voting Stock of such Person. For the purposes of this definition, "control" means the possession of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise.
 - "Affiliate Agreements" means, collectively, the agreements listed on Schedule 1.01(a) hereto.
 - "Aggregate Commitments" means the Commitments of all the Lenders. As of the Closing Date, the Aggregate Commitments shall equal \$700,000,000.
 - "Agreement" means this Credit Agreement.

"Alternative Currency" means (a) with respect to Existing Letters of Credit, the currencies (other than Dollars) in which such Letters of Credit are issued (but only with respect to such Existing Letters of Credit), and (b) with respect to any other Letters of Credit, those currencies (other than Dollars) that are approved by the L/C Issuer issuing such Letters of Credit in accordance with Section 1.06.

"Alternative Currency Equivalent" means, at any time, with respect to any amount denominated in Dollars, the equivalent amount thereof in the applicable Alternative Currency as determined by the Administrative Agent or the applicable L/C Issuer, as the case may be, at such time on the basis of the Spot Rate (determined in respect of the most recent Revaluation Date) for the purchase of such Alternative Currency with Dollars.

"Alternative Currency Sublimit' means an amount equal to the lesser of the Aggregate Commitments and \$100,000,000. The Alternative Currency Sublimit is part of, and not in addition to, the Aggregate Commitments.

"Applicable Percentage" means with respect to any Lender at any time, the percentage (carried out to the ninth decimal place) of the Aggregate Commitments represented by such Lender's Commitment at such time, subject to adjustment as provided in Section 2.16. If the commitment of each Lender to make Loans and the obligation of each L/C Issuer to make L/C Credit Extensions have been terminated pursuant to Section 8.02 or if the Aggregate Commitments have expired, then the Applicable Percentage of each Lender shall be determined based on the Applicable Percentage of such Lender most recently in effect, giving effect to any subsequent assignments. The initial Applicable Percentage of each Lender is set forth opposite the name of such Lender on Schedule 2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable.

"Applicable Rate" means, from time to time, the following percentages per annum, based upon the Debt Rating as set forth below:

| Applicable Rate | | | | | |
|------------------|-----------------------------|-------------------|--|--------------|---|
| Pricing Level | Debt Ratings S&P/Moody's | Commitment Fee | Eurocurrency Rate/ Financial Letter of Credit Fees | Base Rate | Performance Letter of Credit Fees / Commercial Letter of Credit Fees |
| 1 | BBB / Baa2 or higher | 0.375% | 2.500% | 1.500% | 1.250% |
| 2 | BBB- / Baa3 | 0.375% | 2.750% | 1.750% | 1.375% |
| 3 | BB+ / Ba1 | 0.500% | 3.000% | 2.000% | 1.500% |
| 4 | BB/Ba2 | 0.500% | 3.250% | 2.250% | 1.625% |
| 5 | BB-/Ba3 or lower | 0.625% | 3.500% | 2.500% | 1.750% |

"Debt Rating" means, as of any date of determination, the rating as determined by either S&P or Moody's (collectively, the 'Debt Ratings') of the credit facilities established pursuant to this Agreement; provided that (a) if the respective Debt Ratings issued by the foregoing rating agencies differ by one level, then the Pricing Level for the higher of such Debt Ratings shall apply (with the Debt Rating for Pricing Level 1 being the highest and the Debt Rating for Pricing Level 5 being the lowest); (b) if there is a split in Debt Ratings of more than one level, then the Pricing Level that is one level lower than the Pricing Level of the higher Debt Rating shall apply; (c) if the Borrower has only one Debt Rating, the Pricing Level that is one level lower than that of such Debt Rating shall apply; (d) if the Borrower does not have any Debt Rating (other than as a result of the circumstances described in clause (e) below), Pricing Level 5 shall apply; and (e) if the Borrower does not have a Debt Rating because S&P or Moody's cease to exist or cease to provide such ratings, the Borrower and the Lenders will negotiate in good faith to agree to a substitute rating agency or rating service, or otherwise to adjust the pricing grid set forth above to address such situation, but until agreement is reached, the applicable Debt Rating of the rating agency that either ceased to exist or ceased providing ratings shall be the last Debt Rating provided by such rating agency. Initially, the Applicable Rate resulting from an announced change in the Debt Rating shall be effective during the period commencing on the date of the announcement thereof and ending on the date immediately preceding the effective date of the next such change.

"Applicable Time" means, with respect to any borrowings and payments in any Alternative Currency, the local time in the place of settlement for such Alternative Currency as may be determined by the Administrative Agent or the applicable L/C Issuer, as the case may be, to be necessary for timely settlement on the relevant date in accordance with normal banking procedures in the place of payment.

- "Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Arranger" means Banc of America Securities LLC, BNP Paribas Securities Corp., and J.P. Morgan Securities Inc., each in its capacity as a joint lead arranger and joint book manager.
 - "Asset Sale" has the meaning specified in Section 7.04.
- "Assignee Group" means two or more Eligible Assignees that are Affiliates of one another or two or more Approved Funds managed by the same investment advisor.
- "Assignment and Assumption" means an assignment and assumption entered into by a Lender and an assignee (with the consent of any party whose consent is required by Section 10.06(b)), and accepted by the Administrative Agent, in substantially the form of Exhibit E-1 or any other form approved by the Administrative Agent.
- "Availability Period" means the period from and including the Closing Date to the earliest of (a) the Maturity Date, (b) the date of termination of the Aggregate Commitments pursuant to Section 2.06, and (c) the date of termination of the commitment of each Lender to

make Loans and of the obligation of the L/C Issuer to make L/C Credit Extensions pursuant to Section 8.02.

"Bank of America" means Bank of America, N.A. and its successors.

"Base Rate" means for any day a fluctuating rate per annum equal to the highest of (a) the Federal Funds Rate plus 1/2 of 1%, (b) the rate of interest in effect for such day as publicly announced from time to time by Bank of America as its "prime rate" and (c) the Eurocurrency Rate determined in accordance with clause (b) of the definition thereof, plus 1.00%. The "prime rate" is a rate set by Bank of America based upon various factors including Bank of America's costs and desired return, general economic conditions and other factors, and is used as a reference point for pricing some loans, which may be priced at, above, or below such announced rate. Any change in the "prime rate" announced by Bank of America shall take effect at the opening of business on the day specified in the public announcement of such change.

"Base Rate Committed Loan" means a Committed Loan that is a Base Rate Loan.

"Base Rate Loan" means a Loan that bears interest based on the Base Rate. All Base Rate Loans shall be denominated in Dollars.

"Borrower" means (a) prior to the Borrower Replacement Effectiveness Time, the Original Borrower, and (b) on and after the Borrower Replacement Effectiveness Time, the New Borrower.

"Borrower Materials" has the meaning specified in Section 6.01.

"Borrower Replacement Effectiveness Time" means the date and time on which the conditions set forth in Section 6.24 are satisfied for the substitution of the New Borrower for the Original Borrower as the borrower hereunder.

"Borrower's Accountants" means Deloitte & Touche LLP or another firm of independent nationally recognized public accountants.

"Borrowing" means a Committed Borrowing or a Swing Line Borrowing, as the context may require.

"Business Day" means any day other than a Saturday, Sunday or other day on which commercial banks are authorized to close under the Requirements of Law of, or are in fact closed in, the state where the Administrative Agent's Office with respect to Obligations denominated in Dollars is located and:

(a) if such day relates to any interest rate settings as to a Eurocurrency Rate Loan denominated in Dollars, any fundings, disbursements, settlements and payments in Dollars in respect of any such Eurocurrency Rate Loan, or any other dealings in Dollars to be carried out pursuant to this Agreement in respect of any such Eurocurrency Rate Loan, means any such day that is also a London Banking Day; and

(b) if such day relates to any determination of the Spot Rate pursuant to this Agreement, means any such day on which banks are open for foreign exchange business in the principal financial center of the country of the relevant Alternative Currency for which the Spot Rate is being determined.

"BWPGG" means Babcock & Wilcox Power Generation Group, Inc., a Delaware corporation and a Wholly-Owned Subsidiary of the Borrower.

"BWXT" means BWX Technologies, Inc., a Delaware corporation and a Wholly-Owned Subsidiary of the Borrower.

"BWXT Entities" means, subject to Section 6.25(b), collectively or individually, BWXT and each of its Subsidiaries.

"Capital Expenditures" means, with respect to any Person for any period, (a) the aggregate of amounts that would be reflected as additions to property, plant or equipment on a consolidated balance sheet of such Person and its Subsidiaries prepared in conformity with GAAP, excluding interest capitalized during construction less (b) the aggregate of such amounts used to acquire assets useful in the Borrower's and its Subsidiaries' business to the extent such amounts arose from a sale or disposition of equipment described in Section 7.04(c); provided, however, that the Capital Expenditures of the Borrower shall exclude such amounts to the extent (i) financed with the proceeds of Indebtedness permitted to be incurred hereunder (other than the Loans), (ii) financed with insurance or condemnation proceeds received with respect to loss of, damage to or taking of property of the Borrower or any of its Subsidiaries, (iii) reimbursed, or promptly reimbursable, by any Governmental Authority pursuant to a contractual arrangement therefor between the Borrower and/or one or more of its Subsidiaries and such Governmental Authority or (iv) such amounts represent the purchase price of a Permitted Acquisition that in accordance with GAAP would otherwise be required to be accounted for as a Capital Expenditure.

"Capital Lease" means, with respect to any Person, any lease of (or other arrangement conveying the right to use) property by such Person as lessee that would be accounted for as a capital lease on a balance sheet of such Person prepared in conformity with GAAP.

"Capital Lease Obligations" means, with respect to any Person, the capitalized amount of all obligations of such Person or any of its Subsidiaries under Capital Leases, as determined on a consolidated basis in conformity with GAAP.

"Captive Insurance Subsidiaries" means, collectively or individually as of any date of determination, those regulated Subsidiaries of the Borrower primarily engaged in the business of providing insurance and insurance-related services to the Borrower, its other Subsidiaries and certain other Persons.

"Cash Collateralize" means to pledge and deposit with or deliver directly to an L/C Issuer or to the Administrative Agent, for the benefit of the Administrative Agent, any L/C Issuer or the Swing Line Lender (as applicable) and the Lenders, as the context may indicate, as collateral for L/C Obligations, Obligations in respect of Swing Line Loans, or obligations of Lenders to fund participations in respect of either thereof (as the context may require), cash or deposit account balances or, if the L/C Issuer or Swing Line Lender benefitting from such

collateral shall agree in its sole discretion, other credit support, in each case pursuant to documentation in form and substance satisfactory to (a) the Administrative Agent (but only if the Administrative Agent is a party to such Cash Collateral arrangement) and (b) the applicable L/C Issuer or the Swing Line Lender (as applicable). "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.

"Cash Collateralized Letter of Credit" has the meaning specified in Section 2.03(o).

"Cash Equivalents" means (a) securities issued or fully guaranteed or insured by the United States government or any agency thereof, (b) certificates of deposit, eurodollar time deposits, overnight bank deposits and bankers' acceptances of (i) any commercial bank organized under the laws of the United States, any state thereof, the District of Columbia, any foreign bank, or any branch or agency of any of the foregoing, in each case if such bank has a minimum rating at the time of investment of A-1+ by S&P or P-1 by Moody's, or (ii) any Lender or any branch or agency of any Lender, (c) commercial paper, (d) municipal issued debt securities, including notes and bonds, (e) (i) shares of any money market fund that has net assets of not less than \$500,000,000 and satisfies the requirements of rule 2a-7 under the Investment Company Act of 1940 and (ii) shares of any offshore money market fund that has net assets of not less than \$500,000,000 and a \$1 net asset mandate, (f) fully collateralized repurchase agreements and (g) demand deposit accounts; provided, however, that (i) all obligations of the type specified in clauses (c) or (d) above shall have a minimum rating of A-1 or AAA by S&P or P-1 or Aaa by Moody's, in each case at the time of acquisition thereof, and (ii) the maturities of all obligations of the type described in clause (b) above shall not exceed one year from the date of acquisition thereof.

"Cash Interest Expense" means, with respect to any Person for any period, the Interest Expense of such Person for such periodess, to the extent included in the calculation of Interest Expense of such Person for such period, (a) the amount of debt discount and debt issuance costs amortized, (b) charges relating to write-ups or write-downs in the book or carrying value of existing Financial Covenant Debt and (c) interest payable in evidences of Indebtedness or by addition to the principal of the related Indebtedness.

"Cash Management Agreement" means any agreement to provide cash management services, including treasury, depository, overdraft, credit or debit card, electronic funds transfer and other cash management arrangements in the ordinary course of business of the Borrower and its Subsidiaries, but excluding any such agreement providing for overdraft services or financing that may remain outstanding for more than three Business Days.

"Cash Management Bank" means (a) any Person that, at the time it enters into a Cash Management Agreement, is a Lender or an Affiliate of a Lender, in its capacity as a party to such Cash Management Agreement, and (b) any Person that is a party to a Cash Management Agreement at the time it or its relevant Affiliate becomes a Lender (whether on the Closing Date or at a later date pursuant to Section 10.06), in its capacity as a party to such Cash Management Agreement.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority or (c) the making or issuance of any request, guideline or directive (whether or not having the force of law) by any Governmental Authority.

"Change of Control" means an event or series of events by which:

- (a) prior to the effective time of consummation of the Spinoff in accordance with the terms of this Agreement:
- (i) MII shall cease to beneficially own and control, directly or indirectly, 100% of the issued and outstanding Voting Stock of the Borrower on a fully diluted basis; or
- (ii) any "person" or "group" (within the meaning of Rule 13d-5 of the Securities Exchange Act of 1934 as in effect on the date hereof) (excluding MII and its Subsidiaries and excluding underwriters in the course of their distribution of Voting Stock in an underwritten registered public offering provided such underwriters shall not hold such Stock for longer than five Business Days) shall (A) own directly or indirectly, beneficially or of record, Stock representing more than 30% of either the aggregate ordinary voting power or the aggregate equity value represented by the issued and outstanding Stock in MII or (B) shall have obtained the power (whether or not exercised) to elect a majority of the members of the board of directors of the Borrower or MII; or
- (iii) during any period of twelve consecutive calendar months, individuals who at the beginning of such period constituted the board of directors of either the Borrower or MII (together with any new directors whose election by the board of directors of the Borrower or MII, as applicable, or whose nomination for election by the stockholders of the Borrower, or MII, as applicable, was approved by a vote of at least a majority of the directors then still in office who either were directors at the beginning of such period or whose elections or nomination for election was previously so approved) cease for any reason other than death or disability to constitute a majority of the directors then in office and
- (b) on and after the effective time of consummation of the Spinoff in accordance with the terms of this Agreement:
- (i) any "person" or "group" (within the meaning of Rule 13d-5 of the Securities Exchange Act of 1934 as in effect on the date hereof) (excluding the Borrower and its Subsidiaries and excluding underwriters in the course of their distribution of Voting Stock in an underwritten registered public offering provided such underwriters shall not hold such Stock for longer than five Business Days) shall (A) own directly or indirectly, beneficially or of record, Stock representing more than 30% of either the aggregate ordinary voting power or the aggregate equity value represented by the issued and outstanding Stock in the Borrower or (B) shall have obtained the power (whether or

not exercised) to elect a majority of the members of the board of directors of the Borrower; or

- (ii) during any period of twelve consecutive calendar months, individuals who at the beginning of such period constituted the board of directors of the Borrower (together with any new directors whose election by the board of directors of the Borrower or whose nomination for election by the stockholders of the Borrower was approved by a vote of at least a majority of the directors then still in office who either were directors at the beginning of such period or whose elections or nomination for election was previously so approved) cease for any reason other than death or disability to constitute a majority of the directors then in office; and
- (c) for the avoidance of doubt, the consummation of the Spinoff in accordance with the terms of this Agreement shall not constitute a "Change of Control" hereunder.
 - "Closing Date" means the first date all the conditions precedent in Section 4.01 are satisfied or waived in accordance with Section 10.01.
 - "Code" means the Internal Revenue Code of 1986.
- "Collateral" means, collectively, the Pledged Interests and all other personal and real property of the Borrower, any Guarantor or any other Person in which the Administrative Agent or any Secured Party is granted a Lien under any Security Instrument as security for all or any portion of the Obligations or any other obligation arising under any Loan Document.
- "Collateral Agreement" means the Pledge and Security Agreement dated as of the date hereof by the Borrower and certain of the Guarantors to the Administrative Agent for the benefit of the Secured Parties, substantially in the form of Exhibit G.
- "Collateral Reinstatement Event" means, after a release of Collateral as provided for in Section 10.19(a), the occurrence of any of the following: (a) both (i) the corporate family rating of the Borrower and its Subsidiaries from Moody's is Ba1 and (ii) the corporate rating of the Borrower and its Subsidiaries from Moody's is Ba2 or below (regardless of the then applicable corporate rating of the Borrower and its Subsidiaries from S&P) or (c) the corporate rating of the Borrower and its Subsidiaries from S&P is BB or below (regardless of the then applicable corporate family rating of the Borrower and its Subsidiaries from S&P is BB or below (regardless of the then applicable corporate family rating of the Borrower and its Subsidiaries from Moody's); provided that for purposes of determining whether a Collateral Reinstatement Event shall have occurred, if, for any reason, only one rating agency shall maintain corporate or corporate family ratings of the Borrower and its Subsidiaries then the applicable rating provided by such rating agency (or its equivalent) shall apply for both rating agencies.
- "Collateral Release Event" means the satisfaction of each of the following conditions: (a) the corporate family rating of the Borrower and its Subsidiaries from Moody's is Baa3 or better (with a stable outlook or better), (b) the corporate rating of the Borrower and its Subsidiaries from S&P is BBB- or better (with a stable outlook or better), (c) no Default exists,

and (d) the Administrative Agent's receipt of a certificate from the Borrower with respect to the foregoing.

"Collateral Release Period" means, each period commencing with the occurrence of a Collateral Release Event and continuing until the Collateral Reinstatement Event immediately following such Collateral Release Event.

"Commitment" means, as to each Lender, its obligation to (a) make Committed Loans to the Borrower pursuant to Section 2.01, (b) purchase participations in L/C Obligations, and (c) purchase participations in Swing Line Loans, in an aggregate principal amount at any one time outstanding not to exceed the Dollar amount set forth opposite such Lender's name on Schedule 2.01 or in the Assignment and Assumption pursuant to which such Lender becomes a party hereto, as applicable, as such amount may be adjusted from time to time in accordance with this Agreement.

"Commitment Letter" means that certain commitment letter dated as of March 19, 2010 by and among the Borrower, the Arrangers, Bank of America, BNP Paribas and JPMorgan Chase Bank, N.A.

"Committed Borrowing" means a borrowing consisting of simultaneous Committed Loans of the same Type and, in the case of Eurocurrency Rate Loans, having the same Interest Period made by each of the Lenders pursuant to Section 2.01.

"Committed Loan" has the meaning specified in Section 2.01.

"Committed Loan Notice" means a notice of (a) a Committed Borrowing, (b) a conversion of Committed Loans from one Type to the other, or (c) a continuation of Eurocurrency Rate Loans, pursuant to Section 2.02(a), which, if in writing, shall be substantially in the form of Exhibit A.

"Compliance Certificate" means a certificate substantially in the form of Exhibit D.

"Consolidated Net Income" means, for any period, the net income (or loss) of the Borrower and its Subsidiaries for such period determined on a consolidated basis in accordance with GAAP.

"Constituent Documents" means, with respect to any Person, (a) the articles of incorporation, certificate of incorporation or certificate of formation (or the equivalent organizational documents) of such Person and (b) the by laws, operating agreement (or the equivalent governing documents) of such Person.

"Contaminant" means any material, substance or waste that is classified, regulated or otherwise characterized under any Environmental Law as hazardous, toxic, a contaminant or a pollutant or by other words of similar meaning or regulatory effect, including any petroleum or petroleum derived substance or waste, asbestos and polychlorinated biphenyls.

"Contractual Obligation" of any Person means any obligation, agreement, undertaking or similar provision of any Security issued by such Person or of any agreement, undertaking, contract, lease, indenture, mortgage, deed of trust or other instrument (excluding the Loan Documents) to which such Person is a party or by which it or any of its property is bound.

"Credit Extension" means each of the following: (a) a Borrowing and (b) an L/C Credit Extension.

"Customary Permitted Liens" means, with respect to any Person, any of the following Liens:

- (a) Liens with respect to the payment of taxes, assessments or governmental charges in each case that are not yet due or that are being contested in good faith by appropriate proceedings and with respect to which adequate reserves or other appropriate provisions are being maintained to the extent required by GAAP and, in the case of Mortgaged Property, there is no material risk of forfeiture of such property;
- (b) Liens of landlords arising by statute or lease contracts entered into in the ordinary course, inchoate, statutory or construction liens and liens of suppliers, mechanics, carriers, materialmen, warehousemen, producers, operators or workmen and other liens imposed by law created in the ordinary course of business for amounts not yet due or that are being contested in good faith by appropriate proceedings and with respect to which adequate reserves or other appropriate provisions are being maintained to the extent required by GAAP;
- (c) liens, pledges or deposits made in the ordinary course of business in connection with workers' compensation, unemployment insurance or other types of social security benefits, taxes, assessments, statutory obligations or other similar charges or to secure the performance of bids, tenders, sales, leases, contracts (other than for the repayment of borrowed money) or in connection with surety, appeal, customs or performance bonds or other similar instruments;
- (d) encumbrances arising by reason of zoning restrictions, easements, licenses, reservations, covenants, rights-of-way, utility easements, building restrictions and other similar encumbrances on the use of Real Property not materially detracting from the value of such Real Property and not materially interfering with the ordinary conduct of the business conducted at such Real Property;
- (e) encumbrances arising under leases or subleases of Real Property that do not, individually or in the aggregate, materially detract from the value of such Real Property or materially interfere with the ordinary conduct of the business conducted at such Real Property;
 - (f) financing statements with respect to a lessor's rights in and to personal property leased to such Person in the ordinary course of such Person's business;
- (g) liens, pledges or deposits relating to escrows established in connection with the purchase or sale of property otherwise permitted hereunder and the amounts secured thereby shall not exceed the aggregate consideration in connection with such purchase or sale (whether established for an adjustment in purchase price or liabilities, to secure indemnities, or otherwise); and

(h) bankers' Liens, rights of setoff and other similar Liens existing solely with respect to cash and Cash Equivalents on deposit in one or more accounts maintained by the Borrower or a Subsidiary, in each case granted in the ordinary course of business in favor of the bank or banks with which such accounts are maintained, securing amounts owing to such bank with respect to cash management and operating account arrangements, including those involving pooled accounts and netting arrangements; provided that, unless such Liens are non-consensual and arise by operation of law, in no case shall any such Liens secure (either directly or indirectly) the repayment of any Indebtedness.

"Debt Rating" has the meaning specified in the definition of "Applicable Rate."

"Debtor Relief Laws" means the Bankruptcy Code of the United States, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Requirements of Law of the United States or other applicable jurisdictions from time to time in effect and affecting the rights of creditors generally.

"Default" means any event or condition that constitutes an Event of Default or that, with the giving of any notice, the passage of time, or both, would be an Event of Default.

"<u>Default Rate</u>" means (a) when used with respect to Obligations other than Letter of Credit Fees, an interest rate equal to (i) the Base Rate<u>plus</u> (ii) the Applicable Rate applicable to Base Rate Loans <u>plus</u> (iii) 2% per annum; <u>provided</u>, <u>however</u>, that with respect to a Eurocurrency Rate Loan, the Default Rate shall be an interest rate equal to the interest rate (including any Applicable Rate) otherwise applicable to such Loan plus 2% per annum, and (b) when used with respect to Letter of Credit Fees, a rate equal to the Applicable Rate <u>plus</u> 2% per annum.

"Defaulting Lender" means, subject to Section 2.16(b), any Lender that, as determined by the Administrative Agent, (a) has failed to perform any of its funding obligations hereunder, including in respect of its Loans or participations in respect of Letters of Credit or Swing Line Loans, within three Business Days of the date required to be funded by it hereunder, (b) has notified the Borrower or the Administrative Agent that it does not intend to comply with its funding obligations or has made a public statement to that effect with respect to its funding obligations hereunder or generally under other agreements in which it commits to extend credit, (c) has failed, within three Business Days after request by the Administrative Agent, to confirm in a manner satisfactory to the Administrative Agent that it will comply with its funding obligations hereunder, or (d) has, or has a direct or indirect parent company that has, (i) become the subject of a proceeding under any Debtor Relief Law, (ii) had a receiver, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or a custodian appointed for it, or (iii) taken any action in furtherance of, or indicated its consent to, approval of or acquiescence in any such proceeding or appointment; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority.

"<u>Disqualified Stock</u>" means with respect to any Person, any Stock that, by its terms (or by the terms of any Security into which it is convertible or for which it is exchangeable), or upon the happening of any event, matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise, or is exchangeable for Indebtedness of such Person, or is redeemable at the option of the holder thereof, in whole or in part, on or prior to the Maturity Date.

"Dollar" and "\$" mean lawful money of the United States.

"Dollar Equivalent" means, at any time, (a) with respect to any amount denominated in Dollars, such amount, and (b) with respect to any amount denominated in any Alternative Currency, the equivalent amount thereof in Dollars as determined by the Administrative Agent or the applicable L/C Issuer, as the case may be, at such time on the basis of the Spot Rate (determined in respect of the most recent Revaluation Date) for the purchase of Dollars with such Alternative Currency.

"Domestic Subsidiary" means any Subsidiary that is organized under the laws of any political subdivision of the United States.

"EBITDA" means, for any period,

(a) Consolidated Net Income for such period;

plus

- (b) the sum of, in each case to the extent deducted in the calculation of such Consolidated Net Income but without duplication,
 - (i) any provision for income taxes,
 - (ii) Interest Expense,
 - (iii) depreciation expense,
 - (iv) amortization of intangibles or financing or acquisition costs,
 - (v) any aggregate net loss from the sale, exchange or other disposition of business units by the Borrower or its Subsidiaries, and
- (vi) all other non-cash charges (including impairment of intangible assets and goodwill) and non-cash losses for such period (excluding any non-cash item to the extent it represents an accrual of, or reserve for, cash disbursements for any period ending prior to the Maturity Date);

<u>provided</u>, that, to the extent that all or any portion of the income or gains of any Person is deducted pursuant to any of clauses (c)(v) or (c)(vi) below for a given period, any amounts set forth in any of the preceding clauses (b)(i) through (b)(vi) that are attributable to such Person shall not be included for purposes of this clause (b) for such period,

minus

(c) the sum of, in each case to the extent included in the calculation of such Consolidated Net Income but without duplication,

- (i) any credit for income tax,
- (ii) non-cash interest income.
- (iii) any other non-cash gains or other items which have been added in determining Consolidated Net Income (other than any such gain or other item that has been deducted in determining EBITDA for a prior period),
- (iv) the income of any Subsidiary or Joint Venture to the extent that the declaration or payment of dividends or similar distributions or transfers or loans by such Subsidiary or Joint Venture, as applicable, of that income is not at the time permitted by operation of the terms of its charter or any agreement, instrument, judgment, decree, statute, rule or governmental regulation applicable to such Subsidiary or Joint Venture, as applicable,
- (v) the income of any Person (other than a Subsidiary) in which any other Person (other than the Borrower or a Wholly-Owned Subsidiary or any director holding qualifying shares in accordance with applicable law) has an interest, except to the extent of the amount of dividends or other distributions or transfers or loans actually paid to the Borrower or a Wholly-Owned Subsidiary by such Person during such period, and
- (vi) any aggregate net gains from the sale, exchange or other disposition of business units by the Borrower or any of its Subsidiaries out of the ordinary course of business.

For any period of measurement that includes any Permitted Acquisition or any sale, exchange or disposition of any Subsidiary or business unit of the Borrower or any Subsidiary, EBITDA (and the relevant elements thereof) shall be computed on a <u>pro forma</u> basis for each such transaction as if it occurred on the first day of the period of measurement thereof, so long as the Borrower provides to the Administrative Agent reconciliations and other detailed information relating to adjustments to the relevant financial statements (including copies of financial statements of the acquired Person or assets in any Permitted Acquisition) used in computing EBITDA (and the relevant elements thereof) sufficient to demonstrate such <u>pro forma</u> calculations in reasonable detail.

"Eligible Assignee" means any Person that meets the requirements to be an assignee under Section 10.06(b)(iii) and (v) (subject to such consents, if any, as may be required under Section 10.06(b)(iii)).

"Eligible Line of Business" means the businesses and activities engaged in by the Borrower and its Subsidiaries on the Closing Date, any other businesses or activities reasonably related or incidental thereto and any other businesses that, when taken together with the existing businesses of the Borrower and its Subsidiaries, are immaterial with respect to the assets and liabilities of the Borrower and its Subsidiaries, taken as a whole.

"Employee Benefit Plan" means any "employee benefit plan" as defined in Section 3(3) of ERISA which is or was sponsored, maintained or contributed to by, or required to be contributed by, the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates.

"Environmental Laws" means all applicable Requirements of Law now or hereafter in effect and as amended or supplemented from time to time, relating to pollution or the regulation and protection of human health, safety, the environment or natural resources, including the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended (42 U.S.C. § 9601 et seq.); the Hazardous Material Transportation Act, as amended (49 U.S.C. § 1801 et seq.); the Federal Insecticide, Fungicide, and Rodenticide Act, as amended (7 U.S.C. § 136 et seq.); the Resource Conservation and Recovery Act, as amended (42 U.S.C. § 6901 et seq.); the Toxic Substance Control Act, as amended (15 U.S.C. § 2601 et seq.); the Clean Air Act, as amended (42 U.S.C. § 7401 et seq.); the Federal Water Pollution Control Act, as amended (33 U.S.C. § 1251 et seq.); the Occupational Safety and Health Act, as amended (29 U.S.C. § 651 et seq.); the Safe Drinking Water Act, as amended (42 U.S.C. § 300f et seq.); and each of their state and local counterparts or equivalents.

"Environmental Liabilities and Costs" means, with respect to any Person, all liabilities, obligations, responsibilities, Remedial Actions, losses, damages, punitive damages, consequential damages, treble damages, costs and expenses (including all fees, disbursements and expenses of counsel, experts and consultants and costs of investigation and feasibility studies), fines, penalties, sanctions and interest incurred as a result of any claim or demand by any other Person, whether based in contract, tort, implied or express warranty, strict liability, criminal or civil statute and arising under any Environmental Law, Permit, order or agreement with any Governmental Authority or other Person, in each case relating to and resulting from the past, present or future operations of, or ownership of property by, such Person or any of its Subsidiaries.

"Environmental Lien" means any Lien in favor of any Governmental Authority pursuant to any Environmental Law.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"ERISA Affiliate" means any trade or business (whether or not incorporated) under common control or treated as a single employer with the Borrower, any of its Subsidiaries or any Guarantor within the meaning of Section 414(b), (c), (m) or (o) of the Code. Any former ERISA Affiliate of the Borrower, any of its Subsidiaries or any Guarantor shall continue to be considered an ERISA Affiliate of the Borrower, such Subsidiary or such Guarantor within the meaning of this definition with respect to the period such entity was an ERISA Affiliate of the Borrower, such Subsidiary or such Guarantor and with respect to liabilities arising after such period for which the Borrower, such Subsidiary or such Guarantor could be liable under the Code or ERISA.

"ERISA Event" means (a) a reportable event described in Section 4043(b) or 4043(c) of ERISA with respect to a Title IV Plan, (b) the withdrawal of the Borrower, any of its Subsidiaries, any Guarantor or any ERISA Affiliate from a Title IV Plan subject to Section 4063 or Section 4064 of ERISA during a plan year in which any such entity was a "substantial employer" (as defined in Section 4001(a)(2) of ERISA) or the termination of any such Title IV Plan resulting, in either case, in a material liability to any such entity, (c) the "complete or partial withdrawal" (within the meaning of Sections 4203 and 4205 of ERISA) of the Borrower, any of its Subsidiaries, any Guarantor or any ERISA Affiliate from any Multiemployer Plan where the

Withdrawal Liability is reasonably expected to exceed \$1,000,000 (individually or in the aggregate), (d) notice of reorganization, insolvency, intent to terminate or termination of a Multiemployer Plan is received by the Borrower, any of its Subsidiaries, any Guarantor or any ERISA Affiliate, (e) the filing of a notice of intent to terminate a Title IV Plan under Section 4041(c) of ERISA or the treatment of a plan amendment as a termination under Section 4041(e) of ERISA, where such termination constitutes a "distress termination" under Section 4041(c) of ERISA, (f) the institution of proceedings to terminate a Title IV Plan by the PBGC, (g) the failure to make any required contribution to a Title IV Plan or Multiemployer Plan or to meet the minimum funding standard of Section 412 of the Code (in either case, whether or not waived in accordance with Section 412(c) of the Code), (h) the imposition of a lien under Section 430 of the Code or Section 303 of ERISA on the Borrower, any of its Subsidiaries, any Guarantor or any ERISA Affiliate, (i) any other event or condition that might reasonably be expected to constitute grounds under Section 4042 of ERISA for the termination of, or the appointment of a trustee to administer, any Title IV Plan or Multiemployer Plan or the imposition of any liability under Title IV of ERISA, other than for PBGC premiums due but not delinquent under Section 4007 of ERISA, (j) the imposition of liability on the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates pursuant to Section 4062(e) or 4069 of ERISA or by reason of the application of Section 4212(c) of ERISA, (k) the occurrence of an act or omission which would reasonably be expected to give rise to the imposition on the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates of fines, penalties, taxes or related charges under Chapter 43 of the Code or under Section 409, Section 502(c), (i) or (l), or Section 4071 of ERISA in respect of any "employee pension plan" (within the meaning of Section 3(2) of ERISA), (l) receipt from the IRS of notice of the failure of any employee pension plan that is intended to be qualified under Section 401(a) of the Code so to qualify under Section 401(a) of the Code, or the failure of any trust forming part of any such employee pension plan to qualify for exemption from taxation under Section 501(a) of the Code; or (m) the imposition of a Lien pursuant to Section 401(a)(29) of the Code or pursuant to ERISA with respect to any employee pension plan.

"Eurocurrency Rate" means:

(a) for any Interest Period with respect to a Eurocurrency Rate Loan, the rate per annum equal to (i) the British Bankers Association LIBOR Rate (BBA LIBOR"), as published by Reuters (or other commercially available source providing quotations of BBA LIBOR as may be designated by the Administrative Agent from time to time) at approximately 11:00 a.m., London time, two London Banking Days prior to the commencement of such Interest Period, for Dollar deposits (for delivery on the first day of such Interest Period) with a term equivalent to such Interest Period or (ii) if such rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the first day of such Interest Period in Same Day Funds in the approximate amount of the Eurocurrency Rate Loan being made, continued or converted and with a term equivalent to such Interest Period would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at approximately 11:00 a.m. (London time) two London Banking Days prior to the commencement of such Interest Period; and

(b) for any interest calculation of the Eurocurrency Rate with respect to a Base Rate Loan on any date, the rate per annum equal to (i) BBA LIBOR, at approximately 11:00 a.m., London time determined two London Banking Days prior to such date for Dollar deposits being delivered in the London interbank market for a term of one month commencing that day or (ii) if such published rate is not available at such time for any reason, the rate per annum determined by the Administrative Agent to be the rate at which deposits in Dollars for delivery on the date of determination in Same Day Funds in the approximate amount of the Base Rate Loan being made or maintained and with a term equal to one month would be offered by Bank of America's London Branch to major banks in the London interbank eurodollar market at their request at the date and time of determination.

"Eurocurrency Rate Loan" means a Committed Loan that bears interest at a rate based on clause (a) of the definition of "Eurocurrency Rate."

"Event of Default" has the meaning specified in Section 8.01.

"Excluded Taxes" means, with respect to the Administrative Agent, any Lender, the L/C Issuer or any other recipient of any payment to be made by or on account of any obligation of a Loan Party hereunder or under any of the other Loan Documents, (a) Taxes imposed on or measured by its overall net income or profits (however denominated), and franchise Taxes imposed on it (in lieu of net income Taxes), by the jurisdiction (or any political subdivision thereof) under the Requirements of Law of which such recipient is organized or in which its principal office is located or, in the case of any Lender, in which its applicable Lending Office is located, (b) any branch profits Taxes imposed by the United States (or any political subdivision thereof) or any similar Tax imposed by any other jurisdiction (or any political subdivision thereof) in which the Borrower is located, (c) any Taxes imposed as a result of a present or former connection between the taxing jurisdiction and such recipient, other than a connection resulting from the transactions arising under this Agreement or the other Loan Documents, (d) any backup withholding Tax that is required by the Code to be withheld from amounts payable to a Lender that has failed to comply with clause (A) of Section 3.01(e)(ii), and (e) in the case of a Foreign Lender (other than an assignee pursuant to a request by the Borrower under Section 10.13), any United States withholding Tax that (i) is required to be imposed on amounts payable to such Foreign Lender pursuant to the Requirements of Law in force at the time such Foreign Lender becomes a party hereto (or designates a new Lending Office) or (ii) is attributable to such Foreign Lender's failure or inability (other than as a result of a Change in Law) to comply with clause (B) of Section 3.01(e)(ii), except to the extent that such Foreign Lender (or its assignor, if any) was entitled, at the time of designation of a new Lending Office (or assignment), to receive additional amounts from the Borrower with respect to such w

"Existing Credit Agreements" means (a) that certain Revolving Credit Agreement dated as of December 9, 2003 among BWXT, as borrower, certain subsidiaries of BWXT, as guarantors, the initial lenders named therein, Credit Lyonnais New York Branch, as administrative agent, and Credit Lyonnais Securities, as lead arranger and sole bookrunner (as amended through the Closing Date) and (b) that certain Credit Agreement dated as of February 22, 2006, by and among BWPGG, as borrower, certain lenders and issuers party thereto, Credit

Suisse, Cayman Islands Branch, Credit Suisse Securities (USA) LLC, JPMorgan Chase Bank, National Association, Wachovia Bank, National Association and The Bank of Nova Scotia (as amended through the Closing Date).

"Existing Letters of Credit?" means those letters of credit set forth on Schedule 1.01(b), each of which shall constitute a Letter of Credit hereunder. For each Existing Letter of Credit, Schedule 1.01(b) shall include (a) the Person for whose account such letter of credit was issued, (b) the L/C Issuer, (c) the beneficiary, (d) the maximum amount thereof, (e) the expiration date thereof, (f) whether such letter of credit is a Financial Letter of Credit or a Performance Letter of Credit, and (g) any other information reasonably requested by the Administrative Agent with respect thereto.

"Extended Letter of Credit" has the meaning specified in Section 2.03(a)(ii).

"Fair Market Value" means the value that would be paid by a willing buyer to an unaffiliated willing seller in a transaction not involving distress or necessity of either party; provided that, for any determination of Fair Market Value in connection with an Asset Sale to be made pursuant to Section 7.04(i) in which the Fair Market Value of the properties disposed of in such Asset Sale exceeds \$10,000,000, the Borrower shall provide evidence reasonably satisfactory to the Administrative Agent with respect to the calculation of such Fair Market Value.

"FASB ASC" means the Accounting Standards Codification of the Financial Accounting Standards Board.

"Federal Funds Rate" means, for any day, the rate per annum equal to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System arranged by Federal funds brokers on such day, as published by the Federal Reserve Bank of New York on the Business Day next succeeding such day; provided that (a) if such day is not a Business Day, the Federal Funds Rate for such day shall be such rate on such transactions on the next preceding Business Day as so published on the next succeeding Business Day, and (b) if no such rate is so published on such next succeeding Business Day, the Federal Funds Rate for such day shall be the average rate (rounded upward, if necessary, to a whole multiple of 1/100 of 1%) charged to Bank of America on such day on such transactions as determined by the Administrative Agent.

"Fee Letters" means each of (a) the fee letter dated as of March 19, 2010 by and among the Borrower, Bank of America, JPMorgan Chase Bank, N.A., BNP Paribas, Banc of America Securities LLC, J.P. Morgan Securities Inc. and BNP Paribas Securities Corp., (b) the fee letter dated as of March 19, 2010 by and among the Borrower, Bank of America and Banc of America Securities LLC, (c) the fee letter dated as of March 19, 2010 by and among the Borrower, BNP Paribas and BNP Paribas Securities Corp., and (d) the fee letter dated as of March 19, 2010 by and among the Borrower, JPMorgan Chase Bank, N.A. and J.P. Morgan Securities Inc.

"Financial Covenant Debt" of any Person means, without duplication, Indebtedness of the type specified in clauses (a), (b), (c), (d), (e), (f), (g) and (h) of the definition of "Indebtedness". For the avoidance of doubt, the term "Financial Covenant Debt" shall not

include (a) reimbursement or other obligations with respect to unmatured or undrawn, as applicable, Performance Guarantees and (b) Indebtedness of the Borrower or any Subsidiary of the Borrower that is owed to the Borrower or any Subsidiary of the Borrower.

"Financial Letter of Credit" means (a) any Existing Letter of Credit that is identified as a "Financial Letter of Credit" on Schedule 1.01(b) and (b) any other standby Letter of Credit that is not a Performance Letter of Credit.

"First-Tier Foreign Subsidiary" mean a Foreign Subsidiary all or any portion of whose Stock is owned directly by the Borrower or a Domestic Subsidiary that is a Guarantor (other than a BWXT Entity).

"Fiscal Quarter" means the fiscal quarter of the Borrower ending on March 31, June 30, September 30 or December 31 of the applicable calendar year, as applicable.

"Fiscal Year" means the fiscal year of the Borrower, which is the same as the calendar year.

"Foreign Lender" means any Lender that is organized under the Requirements of Law of a jurisdiction other than that in which the Borrower is resident for tax purposes (including such a Lender when acting in the capacity of the L/C Issuer). For purposes of this definition, the United States, each State thereof and the District of Columbia shall be deemed to constitute a single jurisdiction.

"Foreign Subsidiary" means any Subsidiary that is organized under the laws of a jurisdiction other than the United States, a State thereof or the District of Columbia.

"Form 10" means the Form 10 filed with the SEC on March 12, 2010 (as amended on April 23, 2010) with the Original Borrower as registrant, without giving effect to any other amendments, alterations, replacements or other modifications thereto after such date of filing with the SEC.

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to each L/C Issuer, such Defaulting Lender's Applicable Percentage of the outstanding L/C Obligations with respect to Letters of Credit issued by such L/C Issuer, other than L/C Obligations as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swing Line Lender, such Defaulting Lender's Applicable Percentage of Swing Line Loans other than Swing Line Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its activities.

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States, that are applicable to the circumstances as of the date of determination, consistently applied.

"Governmental Authority" means the government of the United States or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Guarantors" means, collectively, each Wholly-Owned Domestic Subsidiary of the Borrower listed on Schedule 1.01(c) hereto, and each other Person that is or becomes a party to the Guaranty (including by execution of a Joinder Agreement pursuant to Section 6.22), but expressly excludes all Captive Insurance Subsidiaries.

"Guaranty" means the Guaranty Agreement dated as of the date hereof made by the Guarantors in favor of the Administrative Agent for the benefit of the Secured Parties, substantially in the form of Exhibit F, and any Joinder Agreement with respect thereto.

"Guaranty Obligation" means, as applied to any Person, without duplication, any direct or indirect liability, contingent or otherwise, of such Person with respect to any Indebtedness of another Person, if the purpose of such Person in incurring such liability is to provide assurance to the obligee of such Indebtedness that such Indebtedness will be paid or discharged, or that any agreement relating thereto will be complied with, or that any holder of such Indebtedness will be protected (in whole or in part) against loss in respect thereof, including (a) the direct or indirect guaranty, endorsement (other than for collection or deposit in the ordinary course of business), co-making, discounting with recourse or sale with recourse by such Person of Indebtedness of another Person and (b) any liability of such Person for Indebtedness of another Person through any agreement (contingent or otherwise) (i) to purchase, repurchase or otherwise acquire such Indebtedness or any security therefor, or to provide funds for the payment or discharge of such Indebtedness (whether in the form of a loan, advance, stock purchase, capital contribution or otherwise), (ii) to maintain the solvency or any balance sheet item, level of income or financial condition of another Person, (iii) to make take-or-pay or similar payments, regardless of non-performance by any other party or parties to an agreement, (iv) to purchase, sell or lease (as lessor or lessee) property, or to purchase or sell services, primarily for the purpose of enabling the debtor to make payment of such Indebtedness or to assure the holder of such Indebtedness against loss or (v) to supply funds to, or in any other manner invest in, such other Person (including to pay for property or services irrespective of whether such property is received or such services are rendered), if (and only if) in the case of any agreement described under clause (b)(i), (ii), (iii), (iv) or (v) above the primary purpose or intent thereof is to provide assurance to the obligee of Indebtedne

amount of any Guaranty Obligation shall be equal to the amount of the Indebtedness so guaranteed or otherwise supported or, if such amount is not stated or otherwise determinable, the maximum reasonable anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. For the avoidance of doubt, the term "Guaranty Obligation" shall not include reimbursement or other obligations with respect to unmatured or undrawn, as applicable, Performance Guarantees.

"Hedge Bank" means (a) any Person that, at the time it enters into a Secured Swap Contract, is a Lender or an Affiliate of a Lender, in its capacity as a party to such Secured Swap Contract, and (b) any Person that is a party to a Secured Swap Contract at the time it or its relevant Affiliate becomes a Lender (whether on the Closing Date or at a later date pursuant to Section 10.06), in its capacity as a party to such Secured Swap Contract.

"<u>Historical Financial Statements</u>" means the unaudited consolidated balance sheet of the Borrower and its Subsidiaries for the Fiscal Year ended December 31, 2009, and the related consolidated statements of income or operations, shareholders' equity and cash flows for such Fiscal Year.

"Immaterial Subsidiary" means any Subsidiary of the Borrower that, together with its Subsidiaries, (a) contributed less than \$2,000,000 to the EBITDA of the Borrower and its Subsidiaries during the most recently-ended four-quarter period of the Borrower (taken as a single period) and (b) as of any date of determination has assets with an aggregate net book value of \$2,000,000 or less.

"Indebtedness" of any Person means, without duplication, (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person evidenced by promissory notes, bonds, debentures or similar instruments, (c) all matured reimbursement obligations with respect to letters of credit, bankers' acceptances, surety bonds, performance bonds, bank guarantees, and other similar obligations, (d) all other obligations with respect to letters of credit, bankers' acceptances, surety bonds, performance bonds, bank guarantees and other similar obligations, whether or not matured, other than unmatured or undrawn, as applicable, obligations with respect to Performance Guarantees, (e) all indebtedness for the deferred purchase price of property or services, other than trade payables incurred in the ordinary course of business that are not overdue by more than ninety days or disputed in good faith, (f) all indebtedness of such Person created or arising under any conditional sale or other title retention agreement (other than operating leases) with respect to property acquired by such Person (even though the rights and remedies of the seller or lender under such agreement in the event of default are limited to repossession or sale of such property), (g) all Capital Lease Obligations of such Person, (h) all Guaranty Obligations of such Person, (i) all obligations of such Person to purchase, redeem, retire, defease or otherwise acquire for value any Stock or Stock Equivalents of such Person, valued, in the case of redeemable preferred stock, at the greater of its voluntary liquidation preference and its involuntary liquidation preference plus accrued and unpaid dividends, (j) net payments that such Person would have to make in the event of an early termination as determined on the date Indebtedness of such Person is being determined in respect of Swap Contracts of such Person and (k) all Indebtedness of the type referred to above secured by (or for which the holder of such Indebtedness has an existing right, continge

such Person, even though such Person has not assumed or become liable for the payment of such Indebtedness, but limited to the value of the property owned by such Person securing such Indebtedness. For the avoidance of doubt, the term "Indebtedness" shall not include reimbursement or other obligations with respect to unmatured or undrawn, as applicable, Performance Guarantees.

- "Indemnified Taxes" means Taxes other than Excluded Taxes.
- "Indemnitees" has the meaning specified in Section 10.04(b).
- "Information" has the meaning specified in Section 10.07.
- "Information Memorandum" means the Confidential Information Memorandum, dated March 2010, in respect of the credit facilities provided under this Agreement.
- "Intercompany Subordinated Debt Payment" means any payment or prepayment, whether required or optional, of principal, interest or other charges on or with respect to any Subordinated Debt of the Borrower or any Subsidiary of the Borrower, so long as (a) such Subordinated Debt is owed to the Borrower or a Subsidiary of the Borrower and (b) no Event of Default under Sections 8.01(a), (b) or (f) shall have occurred and be continuing.

"Interest Coverage Ratio" means, with respect to the Borrower and its Subsidiaries as of any day, the ratio of (a) EBITDA for the Borrower and its Subsidiaries for the last four full Fiscal Quarters ending on or prior to such day to (b) the Cash Interest Expense of the Borrower and its Subsidiaries for the last four full Fiscal Quarters ending on or prior to such day.

"Interest Expense" means, for any Person for any period, total interest expense of such Person and its Subsidiaries for such period, as determined on a consolidated basis in conformity with GAAP and including, in any event (without duplication for any period or any amount included in any prior period), (a) net costs under Interest Rate Contracts for such period, (b) any commitment fee (including, in the case of the Borrower or any of its Subsidiaries, the commitment fees hereunder) accrued, accreted or paid by such Person during such period, (c) any fees and other obligations (other than reimbursement obligations) with respect to letters of credit (including, in respect of the Borrower or any of its Subsidiaries, the Letter of Credit Fees) and bankers' acceptances (whether or not matured) accrued, accreted or paid by such Person for such period and (d) the fronting fee with respect to each Letter of Credit. For purposes of the foregoing, interest expense shall (i) be determined after giving effect to any net payments made or received by the Borrower or any Subsidiary with respect to interest rate Swap Contracts, (ii) exclude interest expense accrued, accreted or paid by the Borrower or any Subsidiary of the Borrower and (iii) exclude credits to interest expense resulting from capitalization of interest related to amounts that would be reflected as additions to property, plant or equipment on a consolidated balance sheet of the Borrower and its Subsidiaries prepared in conformity with GAAP.

"Interest Payment Date" means, (a) as to any Loan other than a Base Rate Loan, the last day of each Interest Period applicable to such Loan and the Maturity Date; provided, however, that if any Interest Period for a Eurocurrency Rate Loan exceeds three months, the respective

dates that fall every three months after the beginning of such Interest Period shall also be Interest Payment Dates; and (b) as to any Base Rate Loan (including a Swing Line Loan), the last Business Day of each March, June, September and December and the Maturity Date.

"Interest Period" means, as to each Eurocurrency Rate Loan, the period commencing on the date such Eurocurrency Rate Loan is disbursed or converted to or continued as a Eurocurrency Rate Loan and ending on the date one, two, three or six months thereafter, as selected by the Borrower in its Committed Loan Notice or such other period that is twelve months or less requested by the Borrower and consented to by all the Lenders; provided that:

(a) any Interest Period that would otherwise end on a day that is not a Business Day shall be extended to the next succeeding Business Day unless such Business Day falls in another calendar month, in which case such Interest Period shall end on the next preceding Business Day;

(b) any Interest Period that begins on the last Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last Business Day of the calendar month at the end of such Interest Period; and

(c) no Interest Period shall extend beyond the Maturity Date.

"Interest Rate Contracts" means all interest rate swap agreements, interest rate cap agreements, interest rate collar agreements and interest rate insurance.

"Investment" means, as to any Person, (a) any purchase or similar acquisition by such Person of (i) any Security issued by, (ii) a beneficial interest in any Security issued by, or (iii) any other equity ownership interest in, any other Person, (b) any purchase by such Person of all or substantially all of the assets of a business conducted by any other Person, or all or substantially all of the assets constituting what is known to the Borrower to be the business of a division, branch or other unit operation of any other Person, (c) any loan, advance (other than deposits with financial institutions available for withdrawal on demand, prepaid expenses, accounts receivable and similar items made or incurred in the ordinary course of business) or capital contribution by such Person to any other Person, including all Indebtedness of any other Person to such Person arising from a sale of property by such Person other than in the ordinary course of its business, (d) any Guaranty Obligation incurred by such Person in respect of Indebtedness of any other Person and (e) the drawn amounts of Performance Guarantees for the direct or indirect benefit of a Subsidiary or Affiliate or a contract counterparty thereof. For the avoidance of doubt, the term "Investment" shall not include reimbursement or other obligations with respect to unmatured or undrawn, as applicable, Performance Guarantees.

"Inventory" has the meaning specified in the Collateral Agreement.

"IP Security Agreement" means the Intellectual Property Security Agreement dated as of the date hereof by any Loan Party to the Administrative Agent for the benefit of the Secured Parties, in form and substance satisfactory to the Administrative Agent.

"IRS" means the United States Internal Revenue Service.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice, Inc. (or such later version thereof as may be in effect at the time of issuance).

"Issuer Documents" means with respect to any Letter of Credit, the Letter of Credit Application, and any other document, agreement and instrument entered into by the L/C Issuer and the Borrower (or any permitted Subsidiary, Joint Venture or Affiliate) or in favor of the L/C Issuer and relating to such Letter of Credit.

"Joinder Agreement" means a joinder agreement, in form and substance satisfactory to the Administrative Agent, with respect to the Guaranty or any Security Instrument or, with respect to the New Borrower upon the effectiveness of the Spinoff in accordance with Section 6.24, this Agreement.

"Joint Venture" means any Person (a) in which the Borrower, directly or indirectly, owns any Stock and Stock Equivalents of such Person and (b) that is not a Subsidiary of the Borrower; provided that (i) the Administrative Agent, on behalf of the Secured Parties, has a valid, perfected, first priority security interest in the Stock and Stock Equivalents in such joint venture owned directly by any Loan Party except where (x) the Constituent Documents of such joint venture prohibit such a security interest to be granted to the Administrative Agent or (y) such joint venture has incurred Non-Recourse Indebtedness the terms of which either (A) require security interests in such Stock and Stock Equivalents to be granted to secure such Non-Recourse Indebtedness or (B) prohibit such a security interest to be granted to the Administrative Agent, and (ii) no Loan Party shall, whether pursuant to the Constituent Documents of such joint venture or otherwise, be under any Contractual Obligation to make Investments or incur Guaranty Obligations after the Closing Date, or, if later, at the time of, or at any time after, the initial formation of such joint venture, that would be in violation of any provision of this Agreement.

"Landlord Lien Waiver" means a lien waiver signed by a landlord in such form as is reasonably satisfactory to the Administrative Agent.

"L/C Advance" means, with respect to each Lender, such Lender's funding of its participation in any L/C Borrowing in accordance with its Applicable Percentage. All L/C Advances shall be denominated in Dollars.

"L/C Borrowing" means an extension of credit resulting from a drawing under any Letter of Credit which has not been reimbursed on the date when made or refinanced as a Committed Borrowing. All L/C Borrowings shall be denominated in Dollars.

"L/C Credit Extension" means, with respect to any Letter of Credit, the issuance thereof or extension of the expiry date thereof, or the increase of the amount thereof.

"L/C Issuer" means Bank of America, each other Lender that is listed on the signature pages hereto as an "L/C Issuer" and any other Lender that becomes an L/C Issuer in accordance with Section 2.03(1) hereof, each in its respective capacity as issuer of Letters of Credit hereunder, or any successor issuer of Letters of Credit hereunder (whether pursuant to Section

2.03(l), 2.03(m), 9.06, 10.06 or otherwise), but excluding any Lender that resigns or is removed as an L/C Issuer pursuant to the terms hereof (except to the extent such Person has continuing rights and/or obligations with respect to Letters of Credit after such resignation or removal). References to the L/C Issuer herein shall, as the context may indicate (including with respect to any particular Letter of Credit, L/C Credit Extension, L/C Borrowing or L/C Obligations), mean the applicable L/C Issuer, each L/C Issuer, any L/C Issuer, or all L/C Issuers.

"L/C Issuer Sublimit" means with respect to each L/C Issuer, such amount as may be separately agreed between such L/C Issuer and the Borrower from time to time (with specific notice of such amount, and any change thereto, with respect to each L/C Issuer being promptly communicated to the Administrative Agent), provided that the L/C Issuer Sublimit with respect to any Person that ceases to be an L/C Issuer for any reason pursuant to the terms hereof shall be \$0 (subject to the Letters of Credit of such Person remaining outstanding in accordance with the provisions hereof).

"L/C Obligations" means, as at any date of determination, the aggregate amount available to be drawn under all outstanding Letters of Creditplus the aggregate of all Unreimbursed Amounts, including all L/C Borrowings. For purposes of computing the amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with Section 1.08. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn. The L/C Obligations of (a) any Lender at any time shall be its Applicable Percentage of the total L/C Obligations at such time, and (b) any particular L/C Issuer at any time shall mean the L/C Obligations allocable to Letters of Credit issued by such L/C Issuer.

"Lender" has the meaning specified in the introductory paragraph hereto and, as the context requires, includes the Swing Line Lender.

"Lending Office" means, as to any Lender, the office or offices of such Lender described as such in such Lender's Administrative Questionnaire, or such other office or offices as a Lender may from time to time notify the Borrower and the Administrative Agent.

"Letter of Credit" means any letter of credit issued hereunder and shall include the Existing Letters of Credit. A Letter of Credit may be a commercial letter of credit or a standby letter of credit, and a standby Letter of Credit may be a Performance Letter of Credit or a Financial Letter of Credit.

"Letter of Credit Application" means an application and agreement for the issuance or amendment of a Letter of Credit in the form from time to time in use by the L/C Issuer.

"Letter of Credit Expiration Date" means the day that is 30 days prior to the Maturity Date then in effect (or, if such day is not a Business Day, the next preceding Business Day).

"Letter of Credit Fee" has the meaning specified in Section 2.03(i).

"Leverage Ratio" means, with respect to the Borrower and its Subsidiaries as of any day, the ratio of (a) Financial Covenant Debt of the Borrower and its Subsidiaries determined on a consolidated basis in accordance with GAAP as of such day to (b) EBITDA for the Borrower and its Subsidiaries for the last four full Fiscal Quarters ending on or prior to such day.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, collateral assignment, charge, deposit arrangement, encumbrance, lien (statutory or other), security interest or preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever intended to assure payment of any Indebtedness or the performance of any other obligation, including any conditional sale or other title retention agreement, the interest of a lessor under a Capital Lease and any financing lease having substantially the same economic effect as any of the foregoing, and the filing of any effective financing statement under the UCC or comparable law of any jurisdiction naming the owner of the asset to which such Lien relates as debtor.

"Loan" means an extension of credit by a Lender to the Borrower under Article II in the form of a Committed Loan or a Swing Line Loan.

"Loan Documents" means this Agreement, each Note, the Guaranty, each Security Instrument, each Joinder Agreement, each Committed Loan Notice, each Issuer Document, each Fee Letter, any agreement creating or perfecting rights in Cash Collateral pursuant to the provisions of Section 2.03 or 2.15 of this Agreement and all other instruments and documents heretofore or hereafter executed or delivered to or in favor of the Administrative Agent, any Lender or any L/C Issuer in connection with the Loans made, Letters of Credit issued and transactions contemplated by this Agreement.

"Loan Parties" means, collectively, the Borrower, each Guarantor and any other Person providing Collateral pursuant to any Security Instrument.

"London Banking Day" means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank eurodollar market.

"Material Adverse Effect" means (a) a material adverse change in, or a material adverse effect upon, the operations, business, assets, properties, liabilities (actual or contingent), or condition (financial or otherwise) of the Borrower and its Subsidiaries taken as a whole; (b) a material impairment of the rights and remedies of the Administrative Agent or any Lender under any Loan Document or of the ability of any Loan Party to perform its obligations under any Loan Document to which it is a party; or (c) a material adverse effect upon the legality, validity, binding effect or enforceability against any Loan Party of any Loan Document to which it is a party.

"Material Intellectual Property" has the meaning specified in the Collateral Agreement.

"Material Real Property" means, any parcel of real property located in the United States and owned by any Loan Party that has a Fair Market Value in excess of \$3,000,000.

"Material Subsidiary" means, as of any date of determination, any Subsidiary of the Borrower that (a) has assets that represent more than 10% of the consolidated GAAP value of the

assets of the Borrower and its Subsidiaries, inclusive of the subject Subsidiary, as of such date or (b) contributed more than 10% of the EBITDA of the Borrower and its Subsidiaries, inclusive of the subject Subsidiary, during the most recently-ended four-quarter period of the Borrower (taken as a single period), or (c) with respect to any new Person acquired or created by the Borrower, (i) would have contributed more than 10% of the EBITDA of the Borrower and its Subsidiaries, inclusive of the subject Subsidiary, on a pro forma basis as of the last day of the most recently ended four-quarter period of the Borrower (taken as a single period) or (ii) held more than 10% of the consolidated GAAP value of the assets of the Borrower and its Subsidiaries, inclusive of the subject Subsidiary, as of such date, or (d) owns, directly or indirectly, Stock or Stock Equivalents in one or more other Subsidiaries of the Borrower that, when aggregated with such Subsidiary, (i) contributed more than 10% of the EBITDA of the Borrower and its Subsidiaries, inclusive of the subject Subsidiary, during the most recently ended four-quarter period of the Borrower (taken as single period) or (ii) held more than 10% of the consolidated GAAP value of the assets of the Borrower and its Subsidiaries, inclusive of the subject Subsidiary, as of such date

"Maturity Date" means May 3, 2014; provided that if such date is not a Business Day, the Maturity Date shall be the next preceding Business Day.

"MII" shall mean McDermott International, Inc. (a Panamanian corporation).

"Moody's" means Moody's Investors Service, Inc. and any successor thereto.

"Mortgagee Policies" has the meaning specified in Section 4.01(a)(iv)(B).

"Mortgaged Properties" mean, initially, each parcel of Material Real Property and the improvements thereto specified on Schedule 4.01(a)(iv), and shall include each other parcel of Material Real Property and improvements thereto with respect to which a Mortgage is granted pursuant to Section 6.23.

"Mortgages" mean the fee or leasehold mortgages or deeds of trust, assignments of leases and rents and other security documents granting a Lien on any Mortgaged Property to secure the Obligations, each in form and substance reasonably satisfactory to the Administrative Agent, as the same may be amended, supplemented, replaced or otherwise modified from time to time in accordance with this Agreement.

"Multiemployer Plan" means a multiemployer plan, as defined in Section 4001(a)(3) of ERISA, to which the Borrower, any of its Subsidiaries, any Guarantor or any ERISA Affiliate has any obligation or liability, contingent or otherwise.

"New Borrower" means The Babcock & Wilcox Company, a Delaware corporation, which effective as of the Borrower Replacement Effectiveness Time shall be a publicly-traded corporation and the direct or indirect owner of 100% of the outstanding Stock and Stock Equivalents of the Original Borrower.

"Non-Recourse Indebtedness" means Indebtedness of a Joint Venture or Subsidiary of the Borrower (in each case that is not a Loan Party) (a) that, if it is incurred by a Subsidiary of

the Borrower, is on terms and conditions reasonably satisfactory to the Administrative Agent, (b) that is not, in whole or in part, Indebtedness of any Loan Party (and for which no Loan Party has created, maintained or assumed any Guaranty Obligation) and for which no holder thereof has or could have upon the occurrence of any contingency, any recourse against any Loan Party or the assets thereof (other than (i) the Stock or Stock Equivalents issued by the Joint Venture or Subsidiary that is primarily obligated on such Indebtedness that are owned by a Loan Party and (ii) a requirement that a Loan Party make an Investment of equity in such Joint Venture in connection with the terms of such Indebtedness), (c) owing to an unaffiliated third-party (which for the avoidance of doubt does not include the Borrower, any Subsidiary thereof, any other Loan Party, any Joint Venture (or owner of any interest therein) and any Affiliate of any of them) and (d) the source of repayment for which is expressly limited to (i) the assets or cash flows of such Subsidiary or Joint Venture and (ii) the Stock and Stock Equivalents of such Subsidiary or Joint Venture securing such Indebtedness in compliance with the provisions of clause (b) above.

"Note" means a promissory note made by the Borrower in favor of a Lender evidencing Loans made by such Lender to the Borrower, substantially in the form of Exhibit C.

"Obligations" means all advances to, and debts, liabilities, obligations, covenants and duties of, any Loan Party arising under any Loan Document or otherwise with respect to any Loan, Letter of Credit, Secured Cash Management Agreement or Secured Hedge Agreement, in each case whether direct or indirect (including those acquired by assumption), absolute or contingent, due or to become due, now existing or hereafter arising and including interest and fees that accrue after the commencement by or against any Loan Party of any proceeding under any Debtor Relief Laws naming such Person as the debtor in such proceeding, regardless of whether such interest and fees are allowed claims in such proceeding.

"Original Borrower" means Babcock & Wilcox Investment Company, a Delaware corporation.

"Other Taxes" means all present or future stamp or documentary taxes or any other excise or property taxes, charges or similar levies arising from any payment made hereunder or under any other Loan Document or from the execution, delivery or enforcement of, or otherwise with respect to, this Agreement or any other Loan Document.

"Outstanding Amount" means (a) with respect to Committed Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any borrowings and prepayments or repayments of such Committed Loans occurring on such date; (b) with respect to Swing Line Loans on any date, the aggregate outstanding principal amount thereof after giving effect to any borrowings and prepayments or repayments of such Swing Line Loans occurring on such date; and (c) with respect to any L/C Obligations on any date, the Dollar Equivalent amount of the aggregate outstanding amount of such L/C Obligations on such date after giving effect to any L/C Credit Extension occurring on such date and any other changes in the aggregate amount of the L/C Obligations as of such date, including as a result of any reimbursements by the Borrower of Unreimbursed Amounts.

"Overnight Rate" means, for any day, the greater of (a) the Federal Funds Rate and (b) an overnight rate determined by the Administrative Agent, the L/C Issuer, or the Swing Line

Lender, as the case may be, in accordance with banking industry rules on interbank compensation.

"Participant" has the meaning specified in Section 10.06(d).

"PBGC" means the Pension Benefit Guaranty Corporation or any successor thereto.

"Performance Guarantee" of any Person means (a) any letter of credit, bankers acceptance, surety bond, performance bond, bank guarantee or other similar obligation issued for the account of such Person to support only trade payables or nonfinancial performance obligations of such Person, (b) any letter of credit, bankers acceptance, surety bond, performance bond, bank guarantee or other similar obligation issued for the account of such Person to support any letter of credit, bankers acceptance, surety bond, performance bond, bank guarantee or other similar obligation issued for the account of a Subsidiary or Joint Venture of such Person to support only trade payables or non-financial performance obligations of such Subsidiary or Joint Venture, and (c) any parent company guarantee or other direct or indirect liability, contingent or otherwise, of such Person with respect to trade payables or non-financial performance obligations of a Subsidiary or Joint Venture of such Person, if the purpose of such Person in incurring such liability is to provide assurance to the obligee that such contractual obligation will be performed, or that any agreement relating thereto will be complied with.

"Performance Letter of Credit" means (a) a standby Letter of Credit issued to secure ordinary course performance obligations in connection with project engineering, procurement, construction, maintenance and other similar projects (including projects about to be commenced) or bids for prospective project engineering, procurement, construction, maintenance and other similar projects, and (b) a standby Letter of Credit issued to back a bank guarantee, surety bond, performance bond or other similar obligation in each case issued to support ordinary course performance obligations in connection with project engineering, procurement, construction, maintenance and other similar projects (including projects about to be commenced) or bids for prospective project engineering, procurement, construction, maintenance and other similar projects. For the avoidance of doubt, the term "Performance Letter of Credit" includes each Existing Letter of Credit that is identified as a "Performance Letter of Credit" on Schedule 1.01(b).

"Permit" means any permit, approval, authorization, license, variance or permission required from a Governmental Authority under any applicable Requirements of Law.

"Permitted Acquisition" means, the acquisition of all or substantially all of the assets or line of business of, or all of the Stock and Stock Equivalents (other than director's qualifying shares and the like, as may be required by applicable Requirements of Law) of, a Person (referred to herein as the "Acquired Entity"); provided that:

- (a) such acquisition was approved by the board of directors of such Acquired Entity;
- (b) the Acquired Entity shall be in an Eligible Line of Business;

- (c) the Borrower and its Subsidiaries shall comply with Section 6.22 and 6.23, as applicable, within the time periods set forth in such Sections;
- (d) at the time of such transaction:
 - (i) both before and after giving effect thereto, no Default shall have occurred and be continuing;
- (ii) the Borrower would be in compliance with the Leverage Ratio set forth in Section 7.18(b) as of the last day of the most recently completed four Fiscal Quarter period ended prior to such transaction for which the financial statements and certificates required by Section 6.01(a) or 6.01(b) have been delivered, after giving pro forma effect to such transaction and to any other event occurring after such period as to which pro forma recalculation is appropriate as if such transaction had occurred as of the first day of such period (assuming, for purposes of pro forma compliance with Section 7.18(b), that the maximum Leverage Ratio permitted at the time by such Section was in fact 0.25 to 1.00 more restrictive than the Leverage Ratio actually provided for in such Section at such time);
- (iii) if the purchase price for such acquisition is in excess of \$10,000,000, the Borrower shall have delivered a certificate of a Responsible Officer, certifying as to the foregoing and containing reasonably detailed calculations in support thereof, in form and substance reasonably satisfactory to the Administrative Agent; and
- (e) if (i) the Borrower is a party to such transaction, it shall be a surviving entity thereof and shall continue as the Borrower hereunder, and (ii) if any party to any such transaction is a Guarantor, the surviving entity of such transaction shall either be a Guarantor or become a Guarantor pursuant to Section 6.22.

"Permitted BWXT Owner" has the meaning specified in Section 7.19.

"Permitted L/C Party" means (a) the Borrower, (b) any Subsidiary of the Borrower, (c) any Joint Venture, and (d) prior to the effectiveness of the Spinoff in accordance with the terms of this Agreement, any Affiliate of the Borrower.

"Person" means any natural person, corporation, limited liability company, trust, joint venture, association, company, partnership, Governmental Authority or other entity.

"Platform" has the meaning specified in Section 6.01.

"Pledged Interests" means (a) the Stock and Stock Equivalents of each of the existing or hereafter organized or acquired direct Domestic Subsidiaries of a Loan Party; and (b) 65% of the Voting Stock (or if the relevant Person shall own less than 65% of such Voting Stock, then 100% of the Voting Stock owned by such Person) and 100% of the nonvoting Stock and Stock Equivalents of each existing or hereafter organized or acquired First-Tier Foreign Subsidiary; provided that Pledged Interests shall not include any Stock or Stock Equivalents in (i) any BWXT Entity, (ii) any Captive Insurance Subsidiary, (iii) any Joint Venture to the extent that the Constituent Documents of such Joint Venture prohibit such a security interest to be granted to

the Administrative Agent, or (iv) any Subsidiary that is not a Loan Party or any Joint Venture to the extent that such Joint Venture or Subsidiary has incurred Non-Recourse Indebtedness the terms of which either (A) require security interests in such Stock and Stock Equivalents to be granted to secure such Non-Recourse Indebtedness or (B) prohibit such a security interest to be granted to the Administrative Agent.

"Projections" means those financial projections of the Borrower and its Subsidiaries covering the Fiscal Years ending in 2010 through 2014, inclusive, delivered to the Administrative Agent by the Borrower.

"Public Lender" has the meaning specified in Section 6.01.

"Real Property" means all Mortgaged Property and all other real property owned or leased from time to time by any Loan Party or any of its Subsidiaries.

"Register" has the meaning specified in Section 10.06(c).

"Related Parties" means, with respect to any Person, such Person's Affiliates and the partners, directors, officers, employees, agents, trustees and advisors of such Person and of such Person's Affiliates.

"Release" means, with respect to any Person, any release, spill, emission, leaking, pumping, injection, deposit, disposal, discharge, dispersal, leaching or migration, in each case, of any Contaminant into the indoor or outdoor environment or into or out of any property owned by such Person, including the movement of Contaminants through or in the air, soil, surface water, ground water or property and, in each case, in violation of Environmental Law.

"Remedial Action" means all actions required by any applicable Requirement of Law to (a) clean up, remove, treat or in any other way address any Contaminant in the indoor or outdoor environment, (b) prevent the Release or threat of Release or minimize the further Release so that a Contaminant does not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment or (c) perform pre remedial studies and investigations and post remedial monitoring and care.

"Request for Credit Extension" means (a) with respect to a Borrowing, conversion or continuation of Committed Loans, a Committed Loan Notice, (b) with respect to an L/C Credit Extension, a Letter of Credit Application, and (c) with respect to a Swing Line Loan, a Swing Line Loan Notice.

"Required Lenders" means, as of any date of determination, Lenders having more than 50% of the Aggregate Commitments or, if the commitment of each Lender to make Loans and the obligation of the L/C Issuer to make L/C Credit Extensions have been terminated pursuant to Section 8.02, Lenders holding in the aggregate more than 50% of the Total Outstandings (with the aggregate amount of each Lender's risk participation and funded participation in L/C Obligations and Swing Line Loans being deemed "held" by such Lender for purposes of this definition); provided that the Commitment of, and the portion of the Total Outstandings held or

deemed held by, any Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"Requirement of Law" means, with respect to any Person, the common law and all federal, state, local and foreign laws, rules and regulations, orders, judgments, decrees and other determinations of any Governmental Authority or arbitrator, applicable to or binding upon such Person or any of its property or to which such Person or any of its property is subject.

"Responsible Officer" means the chief executive officer, president, chief financial officer, treasurer or controller of a Loan Party and, solely for purposes of notices given for Credit Extensions, amendments to Letters of Credit, and continuations and conversions of Loans, any other officer or employee of the applicable Loan Party so designated by any of the foregoing officers in a notice to the Administrative Agent (which such notice shall include a specimen signature and incumbency confirmation reasonably satisfactory to the Administrative Agent). Any document delivered hereunder that is signed by a Responsible Officer of a Loan Party shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of such Loan Party and such Responsible Officer shall be conclusively presumed to have acted on behalf of such Loan Party.

"Restricted Payment" means (a) any dividend, distribution or any other payment whether direct or indirect, on account of any Stock or Stock Equivalents of the Borrower or any of its Subsidiaries now or hereafter outstanding, except a dividend payable solely in Stock or Stock Equivalents (other than Disqualified Stock) or a dividend or distribution payable solely to the Borrower or one or more Guarantors, (b) any redemption, retirement, sinking fund or similar payment, purchase or other acquisition for value, direct or indirect, of any Stock or Stock Equivalents of the Borrower or any of its Subsidiaries now or hereafter outstanding other than one payable solely to the Borrower or one or more Guarantors and (c) any payment or prepayment of principal, premium (if any), interest, fees (including fees to obtain any waiver or consent in connection with any Indebtedness) or other charges on, or redemption, purchase, retirement, defeasance, sinking fund or similar payment with respect to, any Subordinated Debt of the Borrower or any other Loan Party, other than any Intercompany Subordinated Debt Payment or any required payment, prepayment, redemption, retirement, purchases or other payments, in each case to the extent permitted to be made by the terms of such Subordinated Debt.

"Revaluation Date" means, with respect to any Letter of Credit, each of the following: (a) each date of issuance of a Letter of Credit denominated in an Alternative Currency, (b) each date of an amendment of any such Letter of Credit having the effect of increasing the amount thereof (solely with respect to the increased amount), (c) each date of any payment by an L/C Issuer under any Letter of Credit denominated in an Alternative Currency, (d) in the case of the Existing Letters of Credit denominated in an Alternative Currency, the Closing Date, and (e) such additional dates as the Administrative Agent or the applicable L/C Issuer shall determine or the Required Lenders shall require.

"S&P" means Standard & Poor's Ratings Services, a division of The McGraw-Hill Companies, Inc. and any successor thereto.

"Same Day Funds" means (a) with respect to disbursements and payments in Dollars, immediately available funds, and (b) with respect to disbursements and payments in an Alternative Currency, same day or other funds as may be determined by the Administrative Agent or the applicable L/C Issuer, as the case may be, to be customary in the place of disbursement or payment for the settlement of international banking transactions in the relevant Alternative Currency.

"SEC" means the Securities and Exchange Commission, or any Governmental Authority succeeding to any of its principal functions.

"Secured Cash Management Agreement" means any Cash Management Agreement that is entered into by and between any Loan Party and any Cash Management Bank.

"Secured Hedge Agreement" means any Secured Swap Contract that is entered into by and between any Loan Party and any Hedge Bank.

"Secured Swap Contracts" means all Swap Contracts in the nature of interest rate swap agreements, interest rate cap agreements, interest rate collar agreements, interest rate insurance, foreign exchange contracts, currency swap or option agreements, forward contracts, commodity swap, purchase or option agreements, other commodity price hedging arrangements, and all other similar agreements or arrangements designed to alter the risks of any Person arising from fluctuations in interest rates, currency values or commodity prices.

"Secured Parties" means, collectively, the Administrative Agent, the Lenders, each L/C Issuer, the Hedge Banks, the Cash Management Banks, each co-agent or sub-agent appointed by the Administrative Agent from time to time pursuant to Section 9.05, and the other Persons the Obligations owing to which are or are purported to be secured by the Collateral under the terms of the Security Instruments.

"Security" means any Stock, Stock Equivalent, voting trust certificate, bond, debenture, promissory note or other evidence of Indebtedness, whether secured, unsecured, convertible or subordinated, or any certificate of interest, share or participation in, or any temporary or interim certificate for the purchase or acquisition of, or any right to subscribe to, purchase or acquire, any of the foregoing, but shall not include any evidence of the Obligations.

"Security Instruments" means, collectively, the Collateral Agreement, the Mortgages, the IP Security Agreement, and all other agreements (including Joinder Agreements, control agreements, supplements, collateral assignments and similar agreements), instruments and other documents, whether now existing or hereafter in effect, pursuant to which the Borrower, any Subsidiary or other Person shall grant or convey to the Administrative Agent (for the benefit of the Secured Parties) a Lien in, or any other Person shall acknowledge any such Lien in, property as security for all or any portion of the Obligations or any other obligation under any Loan Document.

"Solvent" means, with respect to any Person, that the value of the assets of such Person (both at fair value and present fair saleable value) is, on the date of determination, greater than the total amount of liabilities (including contingent and unliquidated liabilities) of such Person as

of such date and that, as of such date, such Person is able to pay all liabilities of such Person as such liabilities are expected to mature and does not have unreasonably small capital for its then current business activities. In computing the amount of contingent or unliquidated liabilities at any time, such liabilities shall be computed at the amount that, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

"Spinoff" means the distribution of the Stock and Stock Equivalents of New Borrower to the public shareholders of MII through a series of transactions so long as (a) no Default exists at the time of, or after giving effect to, the completion of all such transactions (including the absence of any Default under Section 8.01(h)) and the Borrower shall be in pro forma compliance with the covenants set forth in Section 7.18, as of the last day of the most recently completed four Fiscal Quarter period ended prior to such transaction for which the financial statements and certificates required by Section 6.01(a) or 6.01(b) have been delivered, (b) the New Borrower assumes the obligations as the "Borrower" hereunder substantially simultaneously therewith in accordance with the provisions of Sections 2.17 and 6.24, (c) such transactions are consummated on substantially the same terms (with non-material changes or other or additional non-material transactions, steps or terms being considered consistent with consummation "on substantially the same terms") as are set forth in the Form 10 and the other written information provided by the Borrower to the Administrative Agent with respect to such transactions, with only such changes or other or additional terms as are either (i) not materially adverse to the Lenders and approved by the Administrative Agent (which such approval shall not be unreasonably withheld, conditioned or delayed but may, in the Administrative Agent's sole discretion, be made in conjunction with a reasonably prompt consultation with the Lenders with respect thereto, with any such consultation with the Lenders not to require approval of more than the Required Lenders and not to be unreasonably withheld, conditioned or delayed) or (ii) approved by the Required Lenders (with any approval of the Required Lenders not to be unreasonably withheld, conditioned or delayed), (d) the Administrative Agent is reasonably satisfied with the terms of each of the Affiliate Agreements and each other material agreement between or among the Borrower and its Subsidiaries (or any of them), on the one hand, and MII and its Subsidiaries (or any of them, but excluding the Borrower and its Subsidiaries), on the other hand, to be, or remain, in effect after the Spinoff, including any other shared services, intercompany debt and separation agreements (which such approval shall not be unreasonably withheld, conditioned or delayed but may, in the Administrative Agent's sole discretion, be made in conjunction with a reasonably prompt consultation with the Lenders with respect thereto, with any such consultation with the Lenders not to require approval of more than the Required Lenders and not to be unreasonably withheld, conditioned or delayed), and (e) after giving effect thereto, the representations and warranties in Article V shall be true and correct in all material respects.

"Spot Rate" for a currency means the rate determined by the applicable L/C Issuer, with notice thereof to the Administrative Agent, to be the rate quoted by the Person acting in such capacity as the spot rate for the purchase by such Person of such currency with another currency through its principal foreign exchange trading office at approximately 11:00 a.m. on the date two Business Days prior to the date as of which the foreign exchange computation is made; provided that the applicable L/C Issuer may obtain such spot rate from another financial institution

designated by such L/C Issuer if the Person acting in such capacity does not have as of the date of determination a spot buying rate for any such currency; and <u>provided further</u> that such L/C Issuer may use such spot rate quoted on the date as of which the foreign exchange computation is made in the case of any Letter of Credit denominated in an Alternative Currency.

"Stock" means shares of capital stock (whether denominated as common stock or preferred stock), partnership or membership interests, equity participations or other equivalents (regardless of how designated) of or in a corporation, partnership, limited liability company or similar business entity, whether voting or non-voting.

"Stock Equivalents" means all securities convertible into or exchangeable for Stock and all warrants, options or other rights to purchase or subscribe for any Stock, whether or not presently convertible, exchangeable or exercisable.

"Subordinated Debt" means Indebtedness of the Borrower or any of its Subsidiaries that is, by its terms, expressly subordinated to the prior payment of any of the Obligations pursuant to subordination terms and conditions reasonably satisfactory to the Administrative Agent, and, in the case of intercompany loans permitted hereunder, subject to the terms of a Global Intercompany Note substantially in the form of Exhibit Hhereto or such other documentation as the Administrative Agent shall deem appropriate. The terms of such Global Intercompany Note or such other documentation and any other Subordinated Debt may permit Intercompany Subordinated Debt Payments.

"Subsidiary" of a Person means a corporation, partnership, joint venture, limited liability company or other business entity of which a majority of the shares of securities or other interests having ordinary voting power for the election of directors or other governing body (other than securities or interests having such power only by reason of the happening of a contingency) are at the time beneficially owned, or the management of which is otherwise controlled, directly, or indirectly through one or more intermediaries, or both, by such Person; provided that any reference herein or in any other Loan Document to a "Subsidiary" of the Borrower shall exclude any Person whose financial statements are not consolidated with the financial statements of the Borrower in accordance with GAAP. Unless otherwise specified, all references herein to a "Subsidiary" or to "Subsidiaries" shall refer to a Subsidiary or Subsidiaries of the Borrower.

"Swap Contract" means (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement,

or any other master agreement (any such master agreement, together with any related schedules, a "Master Agreement"), including any such obligations or liabilities under any Master Agreement.

- "Swing Line Borrowing" means a borrowing of a Swing Line Loan pursuant to Section 2.04.
- "Swing Line Lender" means Bank of America in its capacity as provider of Swing Line Loans, or any successor swing line lender hereunder.
- "Swing Line Loan" has the meaning specified in Section 2.04(a).
- "Swing Line Loan Notice" means a notice of a Swing Line Borrowing pursuant to Section 2.04(b), which, if in writing, shall be substantially in the form of Exhibit B.
- "Swing Line Sublimit" means an amount equal to the lesser of (a) \$15,000,000 and (b) the Aggregate Commitments. The Swing Line Sublimit is part of, and not in addition to, the Aggregate Commitments.
- "Tax Affiliate" means, with respect to any Person, (a) any Subsidiary of such Person, and (b) any Affiliate of such Person with which such Person files or is eligible to file U.S. federal income tax returns.
 - "Tax Return" has the meaning specified in Section 5.08.
- "Taxes" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.
- "Title IV Plan" means a pension plan, other than a Multiemployer Plan, covered by Title IV of ERISA and to which the Borrower, any of its Subsidiaries, any Guarantor or any ERISA Affiliate has any obligation or liability (contingent or otherwise).
 - "Total Outstandings" means the aggregate Outstanding Amount of all Loans and all L/C Obligations.
 - "Type" means, with respect to a Committed Loan, its character as a Base Rate Loan or a Eurocurrency Rate Loan.
 - "UCC" has the meaning specified in the Collateral Agreement.
 - "United States" and "U.S." mean the United States of America.
 - "Unreimbursed Amount" has the meaning specified in Section 2.03(c)(i).
- "Voting Stock" means Stock of any Person having ordinary power to vote in the election of members of the board of directors, managers, trustees or similar controlling Persons, of such

Person (irrespective of whether, at the time, Stock of any other class or classes of such entity shall have or might have voting power by reason of the happening of any contingency).

"Wholly-Owned" means, in respect of any Subsidiary of any Person, a circumstance where all of the Stock of such Subsidiary (other than director's qualifying shares, and the like, as may be required by applicable law) is owned by such Person, either directly or indirectly through one or more Wholly-Owned Subsidiaries thereof.

"Withdrawal Liability" means, with respect to the Borrower, any of its Subsidiaries or any Guarantor at any time, the aggregate liability incurred (whether or not assessed) with respect to all Multiemployer Plans pursuant to Section 4201 of ERISA.

1.02 Other Interpretive Provisions. With reference to this Agreement and each other Loan Document, unless otherwise specified herein or in such other Loan Document:

- (a) The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The word "will" shall be construed to have the same meaning and effect as the word "shall." Unless the context requires otherwise, (i) any definition of or reference to any agreement, instrument or other document (including any Constituent Document) shall be construed as referring to such agreement, instrument or other document as from time to time amended, supplemented or otherwise modified (subject to any restrictions on such amendments, supplements or modifications set forth herein or in any other Loan Document), (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns, (iii) the words "hereto," "hereof" and "hereunder," and words of similar import when used in any Loan Document, shall be construed to refer to such Loan Document in its entirety and not to any particular provision thereof, (iv) all references in a Loan Document to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, the Loan Document in which such references appear, (v) any reference to any law shall include all statutory and regulatory provisions consolidating, amending, replacing or interpreting such law and any reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time to time, and (vi) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible assets and properties, including cash, securities, accounts and contract rights.
- (b) In the computation of periods of time from a specified date to a later specified date, the word 'from' means "from and including;" the words "to" and "until" each mean "to but excluding;" and the word "through" means "to and including."
- (c) Section headings herein and in the other Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other Loan Document.

1.03 Accounting Terms.

- (a) <u>Generally</u>. All accounting terms not specifically or completely defined herein shall be construed in conformity with, and all financial data (including financial ratios and other financial calculations) required to be submitted pursuant to this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, as in effect from time to time, applied in a manner consistent with that used in preparing the Historical Financial Statements, <u>except</u> as otherwise specifically prescribed herein. Notwithstanding the foregoing, for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, Indebtedness of the Borrower and its Subsidiaries shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded.
- (b) Changes in GAAP. If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either the Borrower or the Required Lenders shall so request, the Administrative Agent, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower shall provide to the Administrative Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.
- (c) Consolidation of Variable Interest Entities. All references herein to consolidated financial statements of the Borrower and its Subsidiaries or to the determination of any amount for the Borrower and its Subsidiaries on a consolidated basis or any similar reference shall, in each case, be deemed to include each variable interest entity that the Borrower is required to consolidate pursuant to FASB ASC 810 as if such variable interest entity were a Subsidiary as defined herein.
- **1.04 Rounding.** Any financial ratios required to be maintained by the Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed herein and rounding the result up or down to the nearest number (with a rounding-up if there is no nearest number).

1.05 Exchange Rates; Currency Equivalents.

(a) The applicable L/C Issuer shall determine the Spot Rates (and notify the Administrative Agent of the same) as of each Revaluation Date to be used for calculating Dollar Equivalent amounts of L/C Credit Extensions and Outstanding Amounts denominated in Alternative Currencies. Such Spot Rates shall become effective as of such Revaluation Date and shall be the Spot Rates employed in converting any amounts between the applicable currencies until the next Revaluation Date to occur. Except for purposes of financial statements delivered by Loan Parties hereunder or calculating financial covenants hereunder or except as otherwise provided herein, the applicable amount of any currency (other than Dollars) for purposes of the

Loan Documents shall be such Dollar Equivalent amount as so determined by the Administrative Agent or the applicable L/C Issuer, as applicable.

(b) Wherever in this Agreement in connection with the issuance, amendment or extension of a Letter of Credit, an amount, such as a required minimum or multiple amount, is expressed in Dollars, but such Letter of Credit is denominated in an Alternative Currency, such amount shall be the relevant Alternative Currency Equivalent of such Dollar amount (rounded to the nearest unit of such Alternative Currency, with 0.5 of a unit being rounded upward), as determined by the Administrative Agent or the applicable L/C Issuer, as the case may be.

1.06 Alternative Currencies.

- (a) The Borrower may from time to time request that one or more L/C Issuers issue and maintain Letters of Credit denominated in a currency other than Dollars. Any such request shall be subject to the approval of the L/C Issuer that will be issuing Letters of Credit in such currency.
- (b) Any such request shall be made by the Borrower to one or more L/C Issuers not later than 11:00 a.m., ten Business Days prior to the date of the desired issuance of a Letter of Credit in such currency (or such other time or date as may be agreed by any such L/C Issuer, in its sole discretion).
- (c) If any L/C Issuer consents to the issuance of Letters of Credit in such requested currency, such L/C Issuer shall so notify the Borrower and the Administrative Agent, and such currency shall thereupon be deemed for all purposes to be an Alternative Currency hereunder for purposes of any Letter of Credit issuances by each such approving L/C Issuer (but not by any L/C Issuer not approving such currency).
- (d) Prior to the Closing Date, each L/C Issuer may agree with the Borrower to issue Letters of Credit in particular currencies (other than Dollars) immediately upon, and at all times after, the Closing Date, and each L/C Issuer and the Borrower shall notify the Administrative Agent of the currencies (other than Dollars) approved by such L/C Issuer prior to or on the Closing Date.
 - 1.07 Times of Day. Unless otherwise specified, all references herein to times of day shall be references to Eastern time (daylight or standard, as applicable).
- 1.08 Letter of Credit Amounts. Unless otherwise specified herein, the amount of a Letter of Credit at any time shall be deemed to be the Dollar Equivalent of the stated amount of such Letter of Credit in effect at such time; provided, however, that with respect to any Letter of Credit that, by its terms or the terms of any Issuer Document related thereto, provides for one or more automatic increases in the stated amount thereof, the amount of such Letter of Credit shall be deemed to be the Dollar Equivalent of the maximum stated amount of such Letter of Credit after giving effect to all such increases, whether or not such maximum stated amount is in effect at such time.

ARTICLE II. THE COMMITMENTS AND CREDIT EXTENSIONS

2.01 Committed Loans. Subject to the terms and conditions set forth herein, each Lender severally agrees to make loans to the Borrower in Dollars (each such loan, a "Committed Loan") from time to time, on any Business Day during the Availability Period, in an aggregate amount not to exceed at any time outstanding the amount of such Lender's Commitment; provided, however, that after giving effect to any Committed Borrowing, (a) the Total Outstandings shall not exceed the Aggregate Commitments, and (b) the aggregate Outstanding Amount of the Committed Loans of any Lender, plus such Lender's Applicable Percentage of the Outstanding Amount of all Swing Line Loans shall not exceed such Lender's Commitment. Within the limits of each Lender's Commitment, and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.01, prepay under Section 2.05, and reborrow under this Section 2.01. Committed Loans may be Base Rate Loans or Eurocurrency Rate Loans, as further provided herein.

2.02 Borrowings, Conversions and Continuations of Committed Loans.

(a) Each Committed Borrowing, each conversion of Committed Loans from one Type to the other, and each continuation of Eurocurrency Rate Loans shall be made upon the Borrower's irrevocable notice to the Administrative Agent, which may be given by telephone. Each such notice must be received by the Administrative Agent not later than 1:00 p.m. (i) three Business Days prior to the requested date of any Borrowing of, conversion to or continuation of Eurocurrency Rate Loans or of any conversion of Eurocurrency Rate Loans to Base Rate Committed Loans, and (ii) on the requested date of any Borrowing of Base Rate Committed Loans; provided that if the Borrower wishes to request Eurocurrency Rate Loans having an Interest Period other than one, two, three or six months in duration as provided in the definition of "Interest Period," the applicable notice must be received by the Administrative Agent not later than 1:00 p.m. four Business Days prior to the requested date of such Borrowing, conversion or continuation of Eurocurrency Rate Loans, whereupon the Administrative Agent shall give prompt notice to the Lenders of such request and determine whether the requested Interest Period is acceptable to all of them. Not later than 1:00 p.m., three Business Days before the requested date of such Borrowing, conversion or continuation of Eurocurrency Rate Loans (or four Business Days in the case of a Borrowing of Eurocurrency Rate Loans having an Interest period other than one, two, three or six months in duration as provided in the definition of "Interest Period"), the Administrative Agent shall notify the Borrower (which notice may be by telephone) whether or not the requested Interest Period has been consented to by all the Lenders. Each telephonic notice by the Borrower pursuant to this Section 2.02(a) must be confirmed promptly by delivery to the Administrative Agent of a written Committed Loan Notice, appropriately completed and signed by a Responsible Officer of the Borrower. Each Borrowing of, conversion to or continuation of Eurocurrency Rate Loans shall be in a principal amount of \$5,000,000 or a whole multiple of \$1,000,000 in excess thereof. Except as provided in Sections 2.03(c) and 2.04(c), each Committed Borrowing of or conversion to Base Rate Committed Loans shall be in a principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof. Each Committed Loan Notice (whether telephonic or written) shall specify (i) whether the

Borrower is requesting a Committed Borrowing, a conversion of Committed Loans from one Type to the other, or a continuation of Eurocurrency Rate Loans, (ii) the requested date of the Borrowing, conversion or continuation, as the case may be (which shall be a Business Day), (iii) the principal amount of Committed Loans to be borrowed, converted or continued, (iv) the Type of Committed Loans to be borrowed or to which existing Committed Loans are to be converted, and (v) if applicable, the duration of the Interest Period with respect thereto. If the Borrower fails to specify a Type of Committed Loan in a Committed Loan Notice or if the Borrower fails to give a timely notice requesting a conversion or continuation, then the applicable Committed Loans shall be made as, or converted to, Base Rate Loans. Any automatic conversion to Base Rate Loans shall be effective as of the last day of the Interest Period then in effect with respect to the applicable Eurocurrency Rate Loans. If the Borrower requests a Borrowing of, conversion to, or continuation of Eurocurrency Rate Loans in any such Committed Loan Notice, but fails to specify an Interest Period, it will be deemed to have specified an Interest Period of one month.

- (b) Following receipt of a Committed Loan Notice, the Administrative Agent shall promptly notify each Lender of the amount of its Applicable Percentage of the applicable Committed Loans, and if no timely notice of a conversion or continuation is provided by the Borrower, the Administrative Agent shall notify each Lender of the details of any automatic conversion to Base Rate Loans as described in the preceding subsection. In the case of a Committed Borrowing, each Lender shall make the amount of its Committed Loan available to the Administrative Agent in Same Day Funds at the Administrative Agent's Office not later than (i) 1:00 p.m. on the Business Day specified in the applicable Committed Loan Notice for Borrowing of any Committed Loan requested in a Committed Loan Notice that was received prior to the Business Day specified for such Borrowing in the applicable Committed Loan Notice and (ii) 3:00 p.m. in the case of any Borrowing of a Committed Loan Notice. Upon satisfaction of the applicable conditions set forth in Section 4.02 (and, if such Borrowing is the initial Credit Extension, Section 4.01), the Administrative Agent shall make all funds so received available to the Borrower in like funds as received by the Administrative Agent either by (i) crediting the account of the Borrower on the books of Bank of America with the amount of such funds or (ii) wire transfer of such funds, in each case in accordance with instructions provided to (and reasonably acceptable to) the Administrative Agent by the Borrower; provided that if, on the date the Committed Loan Notice with respect to such Borrowing is given by the Borrower, there are L/C Borrowings outstanding, then the proceeds of such Borrowing, first, shall be applied to the payment in full of any such L/C Borrowings, and, second, shall be made available to the Borrower as provided above.
- (c) Except as otherwise provided herein, a Eurocurrency Rate Loan may be continued or converted only on the last day of an Interest Period for such Eurocurrency Rate Loan. During the existence of a Default, no Loans may be requested as, converted to or continued as Eurocurrency Rate Loans without the consent of the Required Lenders.
- (d) The Administrative Agent shall promptly notify the Borrower and the Lenders of the interest rate applicable to any Interest Period for Eurocurrency Rate Loans upon determination of such interest rate. At any time that Base Rate Loans are outstanding, the

Administrative Agent shall notify the Borrower and the Lenders of any change in Bank of America's prime rate used in determining the Base Rate promptly following the public announcement of such change.

(e) After giving effect to all Committed Borrowings, all conversions of Committed Loans from one Type to the other, and all continuations of Committed Loans as the same Type, there shall not be more than ten Interest Periods in effect with respect to Committed Loans.

2.03 Letters of Credit.

(a) The Letter of Credit Commitment.

(i) Subject to the terms and conditions set forth herein, (A) each L/C Issuer agrees, in reliance upon the agreements of the Lenders set forth in this Section 2.03, (1) from time to time on any Business Day during the period from the Closing Date until the Letter of Credit Expiration Date, to issue Letters of Credit denominated in Dollars or in one or more Alternative Currencies applicable to such L/C Issuer for the account of any Permitted L/C Party, and to amend or extend Letters of Credit previously issued by it, in accordance with subsection (b) below, and (2) to honor drawings under the Letters of Credit; and (B) the Lenders severally agree to participate in Letters of Credit issued for the account of any Permitted L/C Party and any drawings thereunder; provided that after giving effect to any L/C Credit Extension with respect to any Letter of Credit, (v) the aggregate amount of L/C Obligations with respect to Letters of Credit issued for the account of one or more Affiliates of the Borrower that will not be Joint Ventures or Subsidiaries of the Borrower after the Spinoff does not exceed \$22,000,000 at any time outstanding, (w) the Total Outstandings shall not exceed the Aggregate Commitments, (x) the aggregate Outstanding Amount of the Committed Loans of any Lender, plus such Lender's Applicable Percentage of the Outstanding Amount of all L/C Obligations, plus such Lender's Applicable Percentage of the Outstanding Amount of all Swing Line Loans shall not exceed such Lender's Commitment, (y) the Outstanding Amount of the L/C Obligations in Alternative Currencies shall not exceed the Alternative Currency Sublimit, and (z) the Outstanding Amount of L/C Obligations of any L/C Issuer shall not exceed the L/C Issuer Sublimit of such L/C Issuer; provided further that after the date that is three months after the Closing Date, no Letters of Credit shall be issued for the account of any Affiliate of the Borrower that is not at such time a Joint Venture or a Subsidiary of the Borrower. Each request by the Borrower for the issuance or amendment of a Letter of Credit shall be deemed to be a representation by the Borrower that the L/C Credit Extension so requested complies with the conditions set forth in the proviso to the preceding sentence. Within the foregoing limits, and subject to the terms and conditions hereof, the Borrower's ability to obtain Letters of Credit shall be fully revolving, and accordingly the Borrower may, during the foregoing period, obtain Letters of Credit to replace Letters of Credit that have expired or that have been drawn upon and reimbursed. All Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and from and after the Closing Date shall be subject to and governed by the terms and conditions hereof.

- (ii) The L/C Issuer shall not issue any Letter of Credit if the expiry date of such requested Letter of Credit would occur after the date that is seven Business Days prior to the Maturity Date (each such issued Letter of Credit, an "Extended Letter of Credit") unless the applicable L/C Issuer has approved such later expiry date, it being acknowledged and agreed that each such Extended Letter of Credit shall be Cash Collateralized in accordance with Section 6.28.
 - (iii) No L/C Issuer shall be under any obligation to issue any Letter of Credit if:
 - (A) any order, judgment or decree of any Governmental Authority or arbitrator shall by its terms purport to enjoin or restrain such L/C Issuer from issuing the Letter of Credit, or any Requirement of Law applicable to such L/C Issuer or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over such L/C Issuer shall prohibit, or request that such L/C Issuer refrain from, the issuance of letters of credit generally or the Letter of Credit in particular or shall impose upon such L/C Issuer with respect to the Letter of Credit any restriction, reserve or capital requirement (for which such the L/C Issuer is not otherwise compensated hereunder) not in effect on the Closing Date, or shall impose upon such L/C Issuer any unreimbursed loss, cost or expense which was not applicable on the Closing Date and which such L/C Issuer in good faith deems material to it:
 - (B) the issuance of the Letter of Credit would violate one or more policies of such L/C Issuer applicable to letters of credit generally;
 - (C) except as otherwise agreed by such L/C Issuer, the Letter of Credit is in an initial stated amount less than \$100,000, in the case of a commercial Letter of Credit, or \$500,000, in the case of a standby Letter of Credit;
 - (D) except as otherwise agreed by such L/C Issuer, the Letter of Credit is to be denominated in a currency other than Dollars or an Alternative Currency applicable to such L/C Issuer;
 - (E) such L/C Issuer does not, as of the issuance date of such requested Letter of Credit, issue Letters of Credit in the requested currency; or
 - (F) any Lender is at that time a Defaulting Lender, unless such L/C Issuer has entered into arrangements, including the delivery of Cash Collateral, satisfactory to such L/C Issuer (in its sole discretion) with the Borrower or such Lender to eliminate such L/C Issuer's actual or potential Fronting Exposure (after giving effect to Section 2.16(a)(iv)) with respect to the Defaulting Lender arising from either the Letter of Credit then proposed to be issued or that Letter of Credit and all other L/C Obligations as to which such L/C Issuer has actual or potential Fronting Exposure, as it may elect in its sole discretion.

- (iv) No L/C Issuer shall amend any Letter of Credit if such L/C Issuer would not be permitted at such time to issue the Letter of Credit in its amended form under the terms hereof.
- (v) No L/C Issuer shall be under any obligation to amend any Letter of Credit if (A) such L/C Issuer would have no obligation at such time to issue the Letter of Credit in its amended form under the terms hereof, or (B) the beneficiary of the Letter of Credit does not accept the proposed amendment to the Letter of Credit.
- (vi) Each L/C Issuer shall act on behalf of the Lenders with respect to any Letters of Credit issued by it and the documents associated therewith, and each L/C Issuer shall have all of the benefits and immunities (A) provided to the Administrative Agent in Article IX with respect to any acts taken or omissions suffered by such L/C Issuer in connection with Letters of Credit issued by it or proposed to be issued by it and Issuer Documents pertaining to such Letters of Credit as fully as if the term "Administrative Agent" as used in Article IX included the L/C Issuers with respect to such acts or omissions, and (B) as additionally provided herein with respect to the L/C Issuers or any of them.

(b) Procedures for Issuance and Amendment of Letters of Credit; Auto-Extension Letters of Credit

(i) Each Letter of Credit shall be issued or amended, as the case may be, upon the request of the Borrower delivered to the applicable L/C Issuer (with a copy to the Administrative Agent) in the form of a Letter of Credit Application, appropriately completed and signed by a Responsible Officer of the Borrower. Such Letter of Credit Application must be received by the L/C Issuer and the Administrative Agent not later than 11:00 a.m. at least two Business Days (or such later date and time as the Administrative Agent and the L/C Issuer may agree in a particular instance in their sole discretion) prior to the proposed issuance date or date of amendment, as the case may be. In the case of a request for an initial issuance of a Letter of Credit, such Letter of Credit Application shall specify in form and detail satisfactory to the L/C Issuer: (A) the proposed issuance date of the requested Letter of Credit (which shall be a Business Day); (B) the amount and currency thereof; (C) the expiry date thereof; (D) the name and address of the beneficiary thereof; (E) the documents to be presented by such beneficiary in case of any drawing thereunder; (F) the full text of any certificate to be presented by such beneficiary in case of any drawing thereunder; (G) whether such requested Letter of Credit is a Performance Letter of Credit, a Financial Letter of Credit or a commercial Letter of Credit; (H) the Permitted L/C Party for whom such Letter of Credit is to be issued; and (I) such other matters as the L/C Issuer may require. In the case of a request for an amendment of any outstanding Letter of Credit, such Letter of Credit Application shall specify in form and detail satisfactory to the L/C Issuer (A) the Letter of Credit to be amended; (B) the proposed date of amendment thereof (which shall be a Business Day); (C) the nature of the proposed amendment; and (D) such other matters as the L/C Issuer may reasonably require. Additionally, the Borrower shall furnish to the L/C Issuer and the Administrative Agent such other docu

requested Letter of Credit issuance or amendment, including any Issuer Documents, as the L/C Issuer or the Administrative Agent may reasonably require.

- (ii) Promptly after receipt of any Letter of Credit Application, the applicable L/C Issuer will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has received a copy of such Letter of Credit Application from the Borrower and, if not, the L/C Issuer will provide the Administrative Agent with a copy thereof. Unless the L/C Issuer has received written notice from any Lender, the Administrative Agent or any Loan Party, at least one Business Day prior to the requested date of issuance or amendment of the applicable Letter of Credit, that one or more applicable conditions contained in Article IV shall not then be satisfied, then, subject to the terms and conditions hereof, the L/C Issuer shall, on the requested date, issue a Letter of Credit for the account of the applicable Permitted L/C Party or enter into the applicable amendment, as the case may be, in each case in accordance with the L/C Issuer's usual and customary business practices. Immediately upon the issuance of each Letter of Credit, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the L/C Issuer a risk participation in such Letter of Credit in an amount equal to the product of such Lender's Applicable Percentage times the amount of such Letter of Credit.
- (iii) If the Borrower so requests in any applicable Letter of Credit Application, the applicable L/C Issuer may, in its sole discretion, agree to issue a Letter of Credit that has automatic extension provisions (each, an "Auto-Extension Letter of Credit"); provided that any such Auto-Extension Letter of Credit must permit the L/C Issuer to prevent any such extension at least once prior to the then applicable expiration date of such Letter of Credit (without giving effect to the next ensuing extension thereof) by giving prior notice to the beneficiary thereof not later than a day (the "Non-Extension Notice Date") to be agreed upon at the time such Letter of Credit is issued. Unless otherwise directed by the L/C Issuer, the Borrower shall not be required to make a specific request to the L/C Issuer for any such extension. Once an Auto-Extension Letter of Credit has been issued, the Lenders shall be deemed to have authorized (but may not require) the L/C Issuer to permit such extensions of such Letter of Credit; provided that if any such extension results in any such Letter of Credit becoming an Extended Letter of Credit the Borrower shall provide Cash Collateral therefor in accordance with Section 6.28; provided, however, that the L/C Issuer shall not permit any such extension if (A) the L/C Issuer has determined that it would not be permitted, or would have no obligation, at such time to issue such Letter of Credit in its revised form (as extended) under the terms hereof (by reason of the provisions of clause (ii) or (iii) of Section 2.03(a) or otherwise), or (B) it has received notice (which may be by telephone or in writing) on or before the day that is seven Business Days before the Non-Extension Notice Date (1) from the Administrative Agent that the Required Lenders have elected not to permit such extension or (2) from the Administrative Agent, any Lender or the Borrower that one or more of the applicable conditions specified in Section 4.02 is not then satisfied, and in each such case directing the L/C Issu

(v) Promptly after its delivery of any Letter of Credit or any amendment to a Letter of Credit to an advising bank with respect thereto or to the beneficiary thereof, the L/C Issuer will also deliver to the Borrower and the Administrative Agent a true and complete copy of such Letter of Credit or amendment.

(c) <u>Drawings and Reimbursements</u>; <u>Funding of Participations</u>.

(i) Upon receipt from the beneficiary of any Letter of Credit of any notice of a drawing under such Letter of Credit, the applicable L/C Issuer shall notify the Borrower and the Administrative Agent thereof. In the case of any draw under a Letter of Credit denominated in an Alternative Currency, the L/C Issuer shall notify the Borrower of the Dollar Equivalent of the amount of the drawing promptly following the determination thereof. The Borrower agrees to pay to the L/C Issuer of any Letter of Credit that has been drawn upon the amount of all draws thereunder, in Dollars (or the Dollar Equivalent of such payment if such payment was made in an Alternative Currency), no later than (x) the Business Day on which the L/C Issuer has provided notice thereof to the Borrower if such notice has been provided prior to 11:00 a.m. on such Business Day, or (y) no later than 10:00 a.m. on the next succeeding Business Day after the Borrower receives such notice from such L/C Issuer if such notice is not received prior to 11:00 a.m. on such day (each such date, an "Honor Date"), and such L/C Issuer shall provide prompt notice to the Administrative Agent of such reimbursement. If the Borrower fails to so reimburse the applicable L/C Issuer by such time, such L/C Issuer shall promptly notify the Administrative Agent of the Honor Date and the amount of the unreimbursed drawing (expressed in Dollars in the amount of the Dollar Equivalent thereof in the case of a Letter of Credit denominated in an Alternative Currency) (the "Unreimbursed Amount"), and the Administrative Agent shall provide such notice, along with the amount of such Lender's Applicable Percentage thereof, to each Lender. In such event, the Borrower shall be deemed to have requested a Committed Borrowing of Base Rate Loans to be disbursed on the Honor Date in an amount equal to the Unreimbursed Amount, without regard to the minimum and multiples specified in Section 2.02 for the principal amount of Base Rate Loans, but subject to the amount of the unutilized portion of the Aggregate Commitments and the conditions set forth in Section 4.02 (other than the delivery of a Committed Loan Notice). Any notice given by any L/C Issuer or the Administrative Agent pursuant to this Section 2.03(c)(i) may be given by telephone if immediately confirmed in writing; provided that the lack of such an immediate confirmation shall not affect the conclusiveness or binding effect of such notice.

(ii) Each Lender shall upon any notice pursuant to Section 2.03(c)(i) make funds available (and the Administrative Agent may apply Cash Collateral provided for this purpose) for the account of the L/C Issuer, in Dollars, at the Administrative Agent's Office for Dollar-denominated payments in an amount equal to its Applicable Percentage of the Unreimbursed Amount not later than 1:00 p.m. on the Business Day specified in such notice by the Administrative Agent, whereupon, subject to the provisions of Section 2.03(c)(iii), each Lender that so makes funds available shall be deemed to have made a Base Rate Committed Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received to the L/C Issuer in Dollars.

- (iii) With respect to any Unreimbursed Amount that is not fully refinanced by a Committed Borrowing of Base Rate Loans because the conditions set forth in Section 4.02 cannot be satisfied or for any other reason, the Borrower shall be deemed to have incurred from the L/C Issuer an L/C Borrowing in the amount of the Unreimbursed Amount that is not so refinanced, which L/C Borrowing shall be due and payable on demand (together with interest) and shall bear interest at the Default Rate. In such event, each Lender's payment to the Administrative Agent for the account of the L/C Issuer pursuant to Section 2.03(c)(ii) shall be deemed payment in respect of its participation in such L/C Borrowing and shall constitute an L/C Advance from such Lender in satisfaction of its participation obligation under this Section 2.03.
- (iv) Until each Lender funds its Committed Loan or L/C Advance pursuant to this Section 2.03(c) to reimburse the applicable L/C Issuer for any amount drawn under any Letter of Credit, interest in respect of such Lender's Applicable Percentage of such amount shall be solely for the account of the applicable L/C Issuer
- (v) Each Lender's obligation to make Committed Loans or L/C Advances to reimburse the L/C Issuer for amounts drawn under Letters of Credit, as contemplated by this Section 2.03(c), shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against any L/C Issuer, the Borrower, any Subsidiary or any other Person for any reason whatsoever; (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Committed Loans pursuant to this Section 2.03(c) is subject to the conditions set forth in Section 4.02 (other than delivery by the Borrower of a Committed Loan Notice). No such making of an L/C Advance shall relieve or otherwise impair the obligation of the Borrower to reimburse the applicable L/C Issuer for the amount of any payment made by the L/C Issuer under any Letter of Credit, together with interest as provided herein.
- (vi) If any Lender fails to make available to the Administrative Agent for the account of any L/C Issuer any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.03(c) by the time specified in Section 2.03(c)(ii), then, without limiting the other provisions of this Agreement, the applicable L/C Issuer shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the L/C Issuer at a rate per annum equal to the applicable Overnight Rate from time to time in effect, plus any administrative, processing or similar fees customarily charged by the L/C Issuer in connection with the foregoing. If such Lender pays such amount (with interest and fees as aforesaid), the amount so paid shall constitute such Lender's Committed Loan included in the relevant Committed Borrowing or L/C Advance in respect of the relevant L/C Borrowing, as the case may be. A certificate of the applicable L/C Issuer submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this clause (vi) shall be conclusive absent manifest error.

(d) Repayment of Participations.

- (i) At any time after the L/C Issuer has made a payment under any Letter of Credit and has received from any Lender such Lender's L/C Advance in respect of such payment in accordance with Section 2.03(c), if the Administrative Agent receives for the account of the L/C Issuer any payment in respect of the related Unreimbursed Amount or interest thereon (whether directly from the Borrower or otherwise, including proceeds of Cash Collateral applied thereto by the Administrative Agent), the Administrative Agent will distribute to such Lender its Applicable Percentage thereof in Dollars and in the same funds as those received by the Administrative Agent.
- (ii) If any payment received by the Administrative Agent for the account of the L/C Issuer pursuant to Section 2.03(c)(i) is required to be returned under any of the circumstances described in Section 10.05 (including pursuant to any settlement entered into by the L/C Issuer in its discretion), each Lender shall pay to the Administrative Agent for the account of the L/C Issuer its Applicable Percentage thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned by such Lender, at a rate per annum equal to the applicable Overnight Rate from time to time in effect. The obligations of the Lenders under this clause shall survive the payment in full of the Obligations and the termination of this Agreement.
- (e) Obligations Absolute. The obligation of the Borrower to reimburse the applicable L/C Issuer for each drawing under each Letter of Credit and to repay each L/C Borrowing shall be absolute, unconditional and irrevocable, and shall be paid strictly in accordance with the terms of this Agreement under all circumstances, including the following:
 - (i) any lack of validity or enforceability of such Letter of Credit, this Agreement, or any other Loan Document;
 - (ii) the existence of any claim, counterclaim, setoff, defense or other right that the Borrower or any Subsidiary may have at any time against any beneficiary or any transferee of such Letter of Credit (or any Person for whom any such beneficiary or any such transferee may be acting), the L/C Issuer or any other Person, whether in connection with this Agreement, the transactions contemplated hereby or by such Letter of Credit or any agreement or instrument relating thereto, or any unrelated transaction;
 - (iii) any draft, demand, certificate or other document presented under such Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; or any loss or delay in the transmission or otherwise of any document required in order to make a drawing under such Letter of Credit;
 - (iv) any payment by the L/C Issuer under such Letter of Credit against presentation of a draft or certificate that does not strictly comply with the terms of such Letter of Credit; or any payment made by the L/C Issuer under such Letter of Credit to any Person purporting to be a trustee in bankruptcy, debtor-in-possession, assignee for the benefit of creditors, liquidator, receiver or other representative of or successor to any

beneficiary or any transferee of such Letter of Credit, including any arising in connection with any proceeding under any Debtor Relief Law;

- (v) any adverse change in the relevant exchange rates or in the availability of the relevant Alternative Currency to the Borrower or any Subsidiary or in the relevant currency markets generally; or
- (vi) any other circumstance or happening whatsoever, whether or not similar to any of the foregoing, including any other circumstance that might otherwise constitute a defense available to, or a discharge of, the Borrower or any Subsidiary.

The Borrower shall promptly examine a copy of each Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will immediately notify the L/C Issuer. The Borrower shall be conclusively deemed to have waived any such claim against the L/C Issuer and its correspondents unless such notice is given as aforesaid, but only to the extent not prohibited by any applicable Requirement of Law.

(f) Role of L/C Issuer. Each Lender and the Borrower agree that, in paying any drawing under a Letter of Credit, no L/C Issuer shall have any responsibility to obtain any document (other than any sight draft, certificates and documents expressly required by the Letter of Credit) or to ascertain or inquire as to the validity or accuracy of any such document or the authority of the Person executing or delivering any such document. None of the L/C Issuers, the Administrative Agent, any of their respective Related Parties nor any correspondent, participant or assignee of any L/C Issuer shall be liable to any Lender for (i) any action taken or omitted in connection herewith at the request or with the approval of the Lenders or the Required Lenders, as applicable; (ii) any action taken or omitted in the absence of gross negligence or willful misconduct; or (iii) the due execution, effectiveness, validity or enforceability of any document or instrument related to any Letter of Credit or Issuer Document. The Borrower hereby assumes all risks of the acts or omissions of any beneficiary or transferee with respect to its use of any Letter of Credit; provided, however, that this assumption is not intended to, and shall not, preclude the Borrower's pursuing such rights and remedies as it may have against the beneficiary or transferee at law or under any other agreement. None of the L/C Issuers, the Administrative Agent, any of their respective Related Parties nor any correspondent, participant or assignee of any L/C Issuer shall be liable or responsible for any of the matters described in clauses (i) through (vi) of Section 2.03(e); provided, however, that anything in such clauses to the contrary notwithstanding, the Borrower may have a claim against the applicable L/C Issuer, and the applicable L/C Issuer may be liable to the Borrower, to the extent, but only to the extent, of any direct, as opposed to consequential or exemplary, damages suffered by the Borrower which the Borrower proves were caused by the L/C Issuer's willful misconduct or gross negligence or the L/C Issuer's willful failure to pay under any Letter of Credit after the presentation to it by the beneficiary of a sight draft and certificate(s) strictly complying with the terms and conditions of a Letter of Credit. In furtherance and not in limitation of the foregoing, each L/C Issuer may accept documents that appear on their face to be in order, without responsibility for further investigation, regardless of any notice or information to the contrary, and no L/C Issuer shall be responsible for the validity or sufficiency of any instrument transferring or assigning or

purporting to transfer or assign a Letter of Credit or the rights or benefits thereunder or proceeds thereof, in whole or in part, which may prove to be invalid or ineffective for any reason.

- (g) <u>Applicability of ISP and UCP</u>. Unless otherwise expressly agreed by the applicable L/C Issuer and the Borrower when a Letter of Credit is issued (including any such agreement applicable to an Existing Letter of Credit), (i) the rules of the ISP shall apply to each standby Letter of Credit, and (ii) the rules of the Uniform Customs and Practice for Documentary Credits, as most recently published by the International Chamber of Commerce at the time of issuance shall apply to each commercial Letter of Credit.
- (h) Letter of Credit Fees. The Borrower shall pay to the Administrative Agent for the account of each Lender (subject to the limitations with respect to Defaulting Lenders set forth in the proviso below) in accordance with its Applicable Percentage, in Dollars, a Letter of Credit fee (the "Letter of Credit Fee") (i) for each commercial Letter of Credit equal to the Applicable Rate for commercial Letters of Credit times the Dollar Equivalent of the daily amount available to be drawn under such Letter of Credit, and (ii) for each standby Letter of Credit equal to the Applicable Rate for such type (Financial Letter of Credit or Performance Letter of Credit) of such Letter of Credit times the Dollar Equivalent of the daily amount available to be drawn under such Letter of Credit (but in each case, excluding from the calculation thereof any portion of any Letter of Credit that has been Cash Collateralized by the Borrower pursuant to Section 2.15(a) as a result of any Defaulting Lender during the period such portion is so Cash Collateralized); provided that any Letter of Credit Fees otherwise payable for the account of a Defaulting Lender with respect to any Letter of Credit (or portion thereof) as to which neither the Borrower nor such Defaulting Lender has provided Cash Collateral satisfactory to the L/C Issuer pursuant to this Section 2.03 or Section 2.15(a) shall be payable, to the maximum extent permitted by applicable Requirements of Law, to the other Lenders in accordance with the upward adjustments in their respective Applicable Percentages allocable to such Letter of Credit pursuant to Section 2.16(a)(iv), with the balance of such fee, if any, payable to the applicable L/C Issuer for its own account. For purposes of computing the Dollar Equivalent of the daily amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with Section 1.08. Letter of Credit Fees shall be (i) due and payable on the third Business Day after the last Business Day of each March, June, September and December, commencing with the first such date to occur after the issuance of such Letter of Credit, on the Letter of Credit Expiration Date and thereafter on demand and (ii) computed on a quarterly basis in arrears. If there is any change in the Applicable Rate during any quarter, the Dollar Equivalent of the daily amount available to be drawn under each Letter of Credit shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect. Notwithstanding anything to the contrary contained herein, upon the request of the Required Lenders, while any Event of Default exists, all Letter of Credit Fees shall accrue at the Default Rate.
- (i) <u>Fronting Fee and Documentary and Processing Charges Payable to L/C Issuer.</u> The Borrower shall pay directly to the applicable L/C Issuer for its own account, in Dollars, a fronting fee (i) with respect to each commercial Letter of Credit, at a rate separately agreed to between the Borrower and such L/C Issuer, computed on the Dollar Equivalent of the amount of such Letter of Credit, and payable upon the issuance thereof, (ii) with respect to any amendment

of a commercial Letter of Credit increasing the amount of such Letter of Credit, at a rate separately agreed between the Borrower and such L/C Issuer, computed on the Dollar Equivalent of the amount of such increase, and payable upon the effectiveness of such amendment, and (iii) with respect to each standby Letter of Credit, at the rate per annum specified in the applicable Fee Letter or otherwise agreed between such L/C Issuer and the Borrower, computed on the Dollar Equivalent of the daily amount available to be drawn under such Letter of Credit on a quarterly basis in arrears. Such fronting fee with respect to standby Letters of Credit shall be due and payable on the third Business Day after the last Business Day of each March, June, September and December in respect of the then-ended quarterly period (or portion thereof, in the case of the first payment), commencing with the first such date to occur after the issuance of such Letter of Credit, on the Letter of Credit Expiration Date and thereafter on demand. Such fronting fee with respect to commercial Letters of Credit shall be due and payable as provided in subparts (i) and (ii) above. For purposes of computing the daily amount available to be drawn under any Letter of Credit, the amount of such Letter of Credit shall be determined in accordance with Section 1.08. In addition, the Borrower shall pay directly to the applicable L/C Issuer for its own account, in Dollars, the customary issuance, presentation, amendment and other processing fees, and other standard costs and charges, of the L/C Issuer relating to letters of credit as from time to time in effect. Such customary fees and standard costs and charges are due and payable on demand and are nonrefundable.

- (j) Conflict with Issuer Documents. In the event of any conflict between the terms hereof and the terms of any Issuer Document, the terms hereof shall control.
- (k) <u>Letters of Credit Issued for Subsidiaries</u>, <u>Affiliates and Joint Ventures</u>. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a Permitted L/C Party other than the Borrower, the Borrower shall be obligated to reimburse the L/C Issuer hereunder for any and all drawings under such Letter of Credit. The Borrower hereby acknowledges that the issuance of Letters of Credit for the account of Permitted L/C Parties other than the Borrower inures to the benefit of the Borrower, and that the Borrower's business derives substantial benefits from the businesses of such Permitted L/C Parties.
- (I) Additional L/C Issuers. In addition to Bank of America and each L/C Issuer listed on the signature pages hereto as an "L/C Issuer," the Borrower may from time to time, with notice to the Lenders and the consent of the Administrative Agent and the applicable Lender being so appointed, appoint additional Lenders to be L/C Issuers hereunder, provided that the total number of L/C Issuers at any time shall not exceed six Lenders (or such larger number of additional Lenders as the Administrative Agent may agree to permit from time to time, it being understood the Administrative Agent has consented to The Bank of Nova Scotia being a seventh L/C Issuer as of the Closing Date solely with respect to its Existing Letters of Credit hereunder). Upon the appointment of a Lender as an L/C Issuer hereunder such Person shall become vested with all of the rights, powers, privileges and duties of an L/C Issuer hereunder.
- (m) <u>Removal of L/C Issuers</u>. The Borrower may at any time remove Bank of America or any L/C Issuer that is appointed pursuant to subpart (l) above, if (but only if) such Person is at such time a Defaulting Lender, in each such case upon not less than 30 days prior notice to such L/C Issuer; <u>provided</u> that such removed L/C Issuer shall retain all the rights, powers, privileges

and duties of an L/C Issuer hereunder with respect to all Letters of Credit outstanding as of the effective date of its removal as L/C Issuer and all L/C Obligations with respect thereto (including the right to require the Lenders to make Base Rate Committed Loans or fund risk participations in Unreimbursed Amounts pursuant to Section 2.03(c)). Without limiting the foregoing, upon the removal of a Lender as an L/C Issuer hereunder, the Borrower may, or at the request of such removed L/C Issuer the Borrower shall use commercially reasonable efforts to, arrange for one or more of the other L/C Issuers to issue Letters of Credit hereunder in substitution for the Letters of Credit, if any, issued by such removed L/C Issuer and outstanding at the time of such removal, or make other arrangements satisfactory to the removed L/C Issuer to effectively cause another L/C Issuer to assume the obligations of the removed L/C Issuer with respect to any such Letters of Credit.

(n) Reporting of Letter of Credit Information and L/C Issuer Sublimit At any time that there is more than one L/C Issuer, then on (i) the last Business Day of each calendar month, and (ii) each date that an L/C Credit Extension occurs with respect to any Letter of Credit, each L/C Issuer (or, in the case of part (ii), the applicable L/C Issuer) shall deliver to the Administrative Agent a report setting forth in form and detail reasonably satisfactory to the Administrative Agent information with respect to each Letter of Credit issued by such L/C Issuer that is outstanding hereunder, including any auto-renewal or termination of auto-renewal provisions in such Letter of Credit. In addition, each L/C Issuer shall provide notice to the Administrative Agent of its L/C Issuer Sublimit, or any change thereto, promptly upon it becoming an L/C Issuer or making any change to its L/C Issuer Sublimit. No failure on the part of any L/C Issuer to provide such information pursuant to this Section 2.03(n) shall limit the obligation of the Borrower or any Lender hereunder with respect to its reimbursement and participation obligations, respectively, pursuant to this Section 2.03.

(o) <u>Cash Collateralized Letters of Credit</u> If the Borrower has fully Cash Collateralized the applicable L/C Issuer with respect to any Extended Letter of Credit issued by such L/C Issuer in accordance with <u>Section 6.28</u> and the Borrower and the applicable L/C Issuer have made arrangements between them with respect to the pricing and fees associated therewith (each such Extended Letter of Credit a "<u>Cash Collateralized Letter of Credit</u>"), then on the day that is 91 days after the date of notice to the Administrative Agent thereof by the applicable L/C Issuer (so long as such Cash Collateral has remained in place for the entirety of such 91-day period), and for so long as such Cash Collateral remains in place (i) such Cash Collateralized Letter of Credit shall cease to be a "Letter of Credit" hereunder, (ii) such Cash Collateralized Letter of Credit shall have any further obligation to fund participations, L/C Borrowings or Committed Loans to reimburse any drawing under any such Cash Collateralized Letter of Credit, (iv) no Letter of Credit Fee shall be due or payable to the Lenders, or any of them, hereunder with respect to such Cash Collateralized Letter of Credit, and (v) any fronting fee, issuance fee or other fee with respect to such Cash Collateralized Letter of Credit shall be as agreed separately between the Borrower and such L/C Issuer.

2.04 Swing Line Loans.

(a) The Swing Line. Subject to the terms and conditions set forth herein, the Swing Line Lender, in reliance upon the agreements of the other Lenders set forth in this Section 2.04, may in its sole discretion make loans in Dollars (each such loan, a Swing Line Loan") to the Borrower from time to time on any Business Day during the Availability Period in an aggregate amount not to exceed at any time outstanding the amount of the Swing Line Sublimit, notwithstanding the fact that such Swing Line Loans, when aggregated with the Applicable Percentage of the Outstanding Amount of Committed Loans and L/C Obligations of the Lender acting as Swing Line Lender, may exceed the amount of such Lender's Commitment; provided, however, that after giving effect to any Swing Line Loan, (i) the Total Outstandings shall not exceed the Aggregate Commitments, and (ii) the aggregate Outstanding Amount of the Committed Loans of any Lender, plus such Lender's Applicable Percentage of the Outstanding Amount of all L/C Obligations, plus such Lender's Applicable Percentage of the Outstanding Amount of all Swing Line Loans shall not exceed such Lender's Commitment, and provided, further, that the Borrower shall not use the proceeds of any Swing Line Loan to refinance any outstanding Swing Line Loan. Within the foregoing limits, and subject to the other terms and conditions hereof, the Borrower may borrow under this Section 2.04, prepay under Section 2.05, and reborrow under this Section 2.04. Each Swing Line Loan shall be a Base Rate Loan. Immediately upon the making of a Swing Line Loan, each Lender shall be deemed to, and hereby irrevocably and unconditionally agrees to, purchase from the Swing Line Lender a risk participation in such Swing Line Loan in an amount equal to the product of such Lender's Applicable Percentage times the amount of such Swing Line Loan.

(b) <u>Borrowing Procedures</u>. Each Swing Line Borrowing shall be made upon the Borrower's irrevocable notice to the Swing Line Lender and the Administrative Agent, which may be given by telephone. Each such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the requested borrowing date, and shall specify (i) the amount to be borrowed, which shall be a minimum of \$100,000, and (ii) the requested borrowing date, which shall be a Business Day. Each such telephonic notice must be confirmed promptly by delivery to the Swing Line Lender and the Administrative Agent of a written Swing Line Loan Notice, appropriately completed and signed by a Responsible Officer of the Borrower. Promptly after receipt by the Swing Line Lender of any telephonic Swing Line Loan Notice, the Swing Line Lender will confirm with the Administrative Agent (by telephone or in writing) that the Administrative Agent has also received such Swing Line Loan Notice and, if not, the Swing Line Lender will notify the Administrative Agent (by telephone or in writing) of the contents thereof. Unless the Swing Line Lender has received notice (by telephone or in writing) from the Administrative Agent (including at the request of any Lender) prior to 2:00 p.m. on the date of the proposed Swing Line Borrowing (A) directing the Swing Line Lender not to make such Swing Line Loan as a result of the limitations set forth in the first proviso to the first sentence of Section 2.04(a), or (B) that one or more of the applicable conditions specified in Article IV is not then satisfied, then, subject to the terms and conditions hereof, the Swing Line Lender will, not later than 3:00 p.m. on the borrowing date specified in such Swing Line Loan Notice, make the amount of its Swing Line Loan available to the Borrower.

(c) Refinancing of Swing Line Loans.

- (i) The Swing Line Lender at any time in its sole and absolute discretion may request, on behalf of the Borrower (which hereby irrevocably authorizes the Swing Line Lender to so request on its behalf), that each Lender make a Base Rate Committed Loan in an amount equal to such Lender's Applicable Percentage of the amount of Swing Line Loans then outstanding. Such request shall be made in writing (which written request shall be deemed to be a Committed Loan Notice for purposes hereof) and in accordance with the requirements of Section 2.02, without regard to the minimum and multiples specified therein for the principal amount of Base Rate Loans, but subject to the unutilized portion of the Aggregate Commitments and the conditions set forth in Section 4.02. The Swing Line Lender shall furnish the Borrower with a copy of the applicable Committed Loan Notice promptly after delivering such notice to the Administrative Agent. Each Lender shall make an amount equal to its Applicable Percentage of the amount specified in such Committed Loan Notice available to the Administrative Agent in Same Day Funds (and the Administrative Agent may apply Cash Collateral available with respect to the applicable Swing Line Loan) for the account of the Swing Line Lender at the Administrative Agent's Office for Dollar-denominated payments not later than 1:00 p.m. on the day specified in such Committed Loan Notice, whereupon, subject to Section 2.04(c)(ii), each Lender that so makes funds available shall be deemed to have made a Base Rate Committed Loan to the Borrower in such amount. The Administrative Agent shall remit the funds so received to the Swing Line Lender.
- (ii) If for any reason any Swing Line Loan cannot be refinanced by such a Committed Borrowing in accordance with Section 2.04(c)(i), the request for Base Rate Committed Loans submitted by the Swing Line Lender as set forth herein shall be deemed to be a request by the Swing Line Lender that each of the Lenders fund its risk participation in the relevant Swing Line Loan and each Lender's payment to the Administrative Agent for the account of the Swing Line Lender pursuant to Section 2.04(c)(i) shall be deemed payment in respect of such participation.
- (iii) If any Lender fails to make available to the Administrative Agent for the account of the Swing Line Lender any amount required to be paid by such Lender pursuant to the foregoing provisions of this Section 2.04(c) by the time specified in Section 2.04(c)(i), the Swing Line Lender shall be entitled to recover from such Lender (acting through the Administrative Agent), on demand, such amount with interest thereon for the period from the date such payment is required to the date on which such payment is immediately available to the Swing Line Lender at a rate per annum equal to the applicable Overnight Rate from time to time in effect, plus any administrative, processing or similar fees customarily charged by the Swing Line Lender in connection with the foregoing. If such Lender pays such amount (with interest and fees as aforesaid), the amount so paid shall constitute such Lender's Committed Loan included in the relevant Committed Borrowing or funded participation in the relevant Swing Line Loan, as the case may be. A certificate of the Swing Line Lender submitted to any Lender (through the Administrative Agent) with respect to any amounts owing under this clause (iii) shall be conclusive absent manifest error.

(iv) Each Lender's obligation to make Committed Loans or to purchase and fund risk participations in Swing Line Loans pursuant to this Section 2.04(c) shall be absolute and unconditional and shall not be affected by any circumstance, including (A) any setoff, counterclaim, recoupment, defense or other right which such Lender may have against the Swing Line Lender, the Borrower or any other Person for any reason whatsoever, (B) the occurrence or continuance of a Default, or (C) any other occurrence, event or condition, whether or not similar to any of the foregoing; provided, however, that each Lender's obligation to make Committed Loans pursuant to this Section 2.04(c) is subject to the conditions set forth in Section 4.02. No such funding of risk participations shall relieve or otherwise impair the obligation of the Borrower to repay Swing Line Loans, together with interest as provided herein.

(d) Repayment of Participations.

- (i) At any time after any Lender has purchased and funded a risk participation in a Swing Line Loan, if the Swing Line Lender receives any payment on account of such Swing Line Loan, the Swing Line Lender will distribute to such Lender its Applicable Percentage thereof in the same funds as those received by the Swing Line Lender.
- (ii) If any payment received by the Swing Line Lender in respect of principal or interest on any Swing Line Loan is required to be returned by the Swing Line Lender under any of the circumstances described in Section 10.05 (including pursuant to any settlement entered into by the Swing Line Lender in its discretion), each Lender shall pay to the Swing Line Lender its Applicable Percentage thereof on demand of the Administrative Agent, plus interest thereon from the date of such demand to the date such amount is returned, at a rate per annum equal to the applicable Overnight Rate. The Administrative Agent will make such demand upon the request of the Swing Line Lender. The obligations of the Lenders under this clause shall survive the payment in full of the Obligations and the termination of this Agreement.
- (e) Interest for Account of Swing Line Lender. The Swing Line Lender shall be responsible for invoicing the Borrower for interest on the Swing Line Loans. Until each Lender funds its Base Rate Committed Loan or risk participation pursuant to this Section 2.04 to refinance such Lender's Applicable Percentage of any Swing Line Loan, interest in respect of such Applicable Percentage shall be solely for the account of the Swing Line Lender.
- (f) <u>Payments Directly to Swing Line Lender</u>. The Borrower shall make all payments of principal and interest in respect of the Swing Line Loans directly to the Swing Line Lender.

2.05 Prepayments.

(a) The Borrower may, upon notice to the Administrative Agent, at any time or from time to time voluntarily prepay Committed Loans in whole or in part without premium or penalty; provided that (i) such notice must be received by the Administrative Agent not later than 11:00 a.m. (A) three Business Days prior to any date of prepayment of Eurocurrency Rate Loans, and (B) on the date of prepayment of Base Rate Committed Loans; (ii) any prepayment of Eurocurrency Rate Loans shall be in a principal amount of \$5,000,000 or a whole multiple of

\$1,000,000 in excess thereof; and (iii) any prepayment of Base Rate Committed Loans shall be in a principal amount of \$500,000 or a whole multiple of \$100,000 in excess thereof or, in each case, if less, the entire principal amount thereof then outstanding. Each such notice shall specify the date and amount of such prepayment and the Type(s) of Committed Loans to be prepaid and, if Eurocurrency Rate Loans are to be prepaid, the Interest Period(s) of such Loans. The Administrative Agent will promptly notify each Lender of its receipt of each such notice, and of the amount of such Lender's Applicable Percentage of such prepayment. If such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein. Any prepayment of a Eurocurrency Rate Loan shall be accompanied by all accrued interest on the amount prepaid, together with any additional amounts required pursuant to Section 3.05. Subject to Section 2.16, each such prepayment shall be applied to the Committed Loans of the Lenders in accordance with their respective Applicable Percentages.

- (b) The Borrower may, upon notice to the Swing Line Lender (with a copy to the Administrative Agent), at any time or from time to time, voluntarily prepay Swing Line Loans in whole or in part without premium or penalty; <u>provided</u> that (i) such notice must be received by the Swing Line Lender and the Administrative Agent not later than 1:00 p.m. on the date of the prepayment, and (ii) any such prepayment shall be in a minimum principal amount of \$100,000. Each such notice shall specify the date and amount of such prepayment. If such notice is given by the Borrower, the Borrower shall make such prepayment and the payment amount specified in such notice shall be due and payable on the date specified therein.
- (c) If the Administrative Agent notifies the Borrower at any time that the Total Outstandings at such time exceed the Aggregate Commitments then in effect, then, within two Business Days after receipt of such notice, the Borrower shall prepay Loans and/or Cash Collateralize the L/C Obligations in an aggregate amount sufficient to reduce such Outstanding Amount as of such date of payment to an amount not to exceed the Aggregate Commitments then in effect; provided, however, that the Borrower shall not be required to Cash Collateralize the L/C Obligations pursuant to this Section 2.05(c) unless after the prepayment in full of the Loans the Total Outstandings exceed the Aggregate Commitments then in effect. The Administrative Agent may, at any time and from time to time after the initial deposit of such Cash Collateral, request that additional Cash Collateral be provided in order to protect against the results of further exchange rate fluctuations.
- 2.06 Termination or Reduction of Commitments. The Borrower may, upon notice to the Administrative Agent, terminate the Aggregate Commitments, or from time to time permanently reduce the Aggregate Commitments; provided that (a) any such notice shall be received by the Administrative Agent not later than 11:00 a.m. five Business Days prior to the date of termination or reduction, (b) any such partial reduction shall be in an aggregate amount of \$10,000,000 or any whole multiple of \$1,000,000 in excess thereof, (c) the Borrower shall not terminate or reduce the Aggregate Commitments if, after giving effect thereto and to any concurrent prepayments hereunder, the Total Outstandings would exceed the Aggregate Commitments, and (d) if, after giving effect to any reduction of the Aggregate Commitments, the Alternative Currency Sublimit or the Swing Line Sublimit exceeds the amount of the Aggregate Commitments, such Sublimit shall be automatically reduced by the amount of such excess.

Except as provided in the preceding sentence, the amount of any such Aggregate Commitment reduction shall not be applied to the Alternative Currency Sublimit or the Swing Line Sublimit unless otherwise specified by the Borrower. The Administrative Agent will promptly notify the Lenders of any such notice of termination or reduction of the Aggregate Commitments. Any reduction of the Aggregate Commitment shall be applied to the Commitment of each Lender according to its Applicable Percentage. All fees accrued until the effective date of any termination of the Aggregate Commitments shall be paid on the effective date of such termination. A notice of termination of the Aggregate Commitments may state that such notice is conditioned upon the effectiveness of other credit facilities, and if any notice so states it may be revoked by the Borrower by notice to the Administrative Agent on or prior to the date specified for the termination of the Aggregate Commitments that the refinancing condition has not been met and the termination is to be revoked, provided that the Borrower will continue to be responsible for any costs or expenses pursuant to Section 3.05 in connection with the failure to prepay Loans resulting from such revocation.

2.07 Repayment of Loans.

- (a) The Borrower shall repay to the Lenders on the Maturity Date the aggregate principal amount of Committed Loans made to the Borrower outstanding on such date.
 - (b) The Borrower shall repay each Swing Line Loan on the earlier to occur of (i) the date 10 Business Days after such Loan is made and (ii) the Maturity Date.

2.08 Interest.

- (a) Subject to the provisions of subsection (b) below, (i) each Eurocurrency Rate Loan shall bear interest on the outstanding principal amount thereof for each Interest Period at a rate per annum equal to the Eurocurrency Rate for such Interest Period <u>plus</u> the Applicable Rate; (ii) each Base Rate Committed Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Base Rate <u>plus</u> the Applicable Rate; and (iii) each Swing Line Loan shall bear interest on the outstanding principal amount thereof from the applicable borrowing date at a rate per annum equal to the Base Rate <u>plus</u> the Applicable Rate.
- (b) (i) If any amount of principal of any Loan is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Requirements of Law.
 - (ii) If any amount (other than principal of any Loan) payable by the Borrower under any Loan Document is not paid when due (without regard to any applicable grace periods), whether at stated maturity, by acceleration or otherwise, then upon the request of the Required Lenders, such amount shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Requirements of Law.

- (iii) Upon the request of the Required Lenders, while any Event of Default exists, the Borrower shall pay interest on the principal amount of all outstanding Obligations hereunder at a fluctuating interest rate per annum at all times equal to the Default Rate to the fullest extent permitted by applicable Requirements of Law
 - (iv) Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be due and payable upon demand.
- (c) Interest on each Loan shall be due and payable in arrears on each Interest Payment Date applicable thereto and at such other times as may be specified herein. Interest hereunder shall be due and payable in accordance with the terms hereof before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law.
 - 2.09 Fees. In addition to certain fees described in subsections (h) and (i) of Section 2.03:
- (a) Commitment Fee. The Borrower shall pay to the Administrative Agent for the account of each Lender (subject to Section 2.16(a)(iii)(x) with respect to Defaulting Lenders) in accordance with its Applicable Percentage, a commitment fee in Dollars equal to the Applicable Rate times the actual daily amount by which the Aggregate Commitments exceed the sum of (i) the Outstanding Amount of Committed Loans and (ii) the Outstanding Amount of L/C Obligations, subject to adjustment as provided in Section 2.16. The commitment fee shall accrue at all times during the Availability Period, including at any time during which one or more of the conditions in Article IV is not met, and shall be due and payable quarterly in arrears on the third Business Day after the last Business Day of each March, June, September and December, commencing with the first such date to occur after the Closing Date, and on the last day of the Availability Period. The commitment fee shall be calculated quarterly in arrears, and if there is any change in the Applicable Rate during any quarter, the actual daily amount shall be computed and multiplied by the Applicable Rate separately for each period during such quarter that such Applicable Rate was in effect.

(b) Other Fees.

- (i) The Borrower shall pay to the Arrangers and the Administrative Agent for their own respective accounts, in Dollars, fees in the amounts and at the times specified in the Fee Letters. Such fees shall be fully earned when paid and shall not be refundable for any reason whatsoever.
- (ii) The Borrower shall pay to the Lenders, in Dollars, such fees as shall have been separately agreed upon in writing in the amounts and at the times so specified. Such fees shall be fully earned when paid and shall not be refundable for any reason whatsoever.
- **2.10 Computation of Interest and Fees**. All computations of interest for Base Rate Loans (including Base Rate Loans determined by reference to the Eurocurrency Rate) shall be made on the basis of a year of 365 or 366 days, as the case may be, and actual days elapsed. All

other computations of fees and interest shall be made on the basis of a 360-day year and actual days elapsed (which results in more fees or interest, as applicable, being paid than if computed on the basis of a 365-day year). Interest shall accrue on each Loan for the day on which the Loan is made, and shall not accrue on a Loan, or any portion thereof, for the day on which the Loan or such portion is paid, provided that any Loan that is repaid on the same day on which it is made shall, subject to Section 2.12(a), bear interest for one day. Each determination by the Administrative Agent of an interest rate or fee hereunder shall be conclusive and binding for all purposes, absent manifest error.

2.11 Evidence of Debt.

- (a) The Credit Extensions made by each Lender shall be evidenced by one or more accounts or records maintained by such Lender and by the Administrative Agent in the ordinary course of business. The accounts or records maintained by the Administrative Agent and each Lender shall be conclusive absent manifest error of the amount of the Credit Extensions made by the Lenders to the Borrower and the interest and payments thereon. Any failure to so record or any error in doing so shall not, however, limit or otherwise affect the obligation of the Borrower hereunder to pay any amount owing with respect to the Obligations. In the event of any conflict between the accounts and records maintained by any Lender and the accounts and records of the Administrative Agent in respect of such matters, the accounts and records of the Administrative Agent, the Borrower made through the Administrative Agent, the Borrower shall execute and deliver to such Lender (through the Administrative Agent) a Note, which shall evidence such Lender's Loans to the Borrower in addition to such accounts or records. Each Lender may attach schedules to a Note and endorse thereon the date, Type (if applicable), amount and maturity of its Loans and payments with respect thereto.
- (b) In addition to the accounts and records referred to in subsection (a), each Lender and the Administrative Agent shall maintain in accordance with its usual practice accounts or records evidencing the purchases and sales by such Lender of participations in Letters of Credit and Swing Line Loans. In the event of any conflict between the accounts and records maintained by the Administrative Agent and the accounts and records of any Lender in respect of such matters, the accounts and records of the Administrative Agent shall control in the absence of manifest error.

2.12 Payments Generally; Administrative Agent's Clawback.

(a) <u>General</u>. All payments to be made by the Borrower shall be made without condition or deduction for any counterclaim, defense, recoupment or setoff. Except as otherwise expressly provided herein, all payments by the Borrower hereunder shall be made to the Administrative Agent, for the account of the respective Lenders to which such payment is owed, at the Administrative Agent's Office in Dollars and in Same Day Funds not later than 2:00 p.m. on the date specified herein. Without limiting the generality of the foregoing, the Administrative Agent may require that any payments due under this Agreement be made in the United States. The Administrative Agent will promptly distribute to each Lender its Applicable Percentage (or other applicable share as provided herein) of such payment in like funds as received by wire transfer to such Lender's Lending Office. All payments received by the Administrative Agent or

the applicable L/C Issuer after 2:00 p.m. shall be deemed received on the next succeeding Business Day and any applicable interest or fee shall continue to accrue. If any payment to be made by the Borrower shall come due on a day other than a Business Day, payment shall be made on the next following Business Day, and such extension of time shall be reflected in computing interest or fees, as the case may be.

- (b) (i) Funding by Lenders: Presumption by Administrative Agent. Unless the Administrative Agent shall have received notice from a Lender prior to the proposed date of any Committed Borrowing of Eurocurrency Rate Loans (or, in the case of any Committed Borrowing of Base Rate Loans, prior to (A) 12:00 noon on the date of such Committed Borrowing if such Committed Borrowing is to be made on a Business Day other than the date the Administrative Agent received the applicable Committed Loan Notice with respect to such Borrowing and (B) 2:00 p.m. on the date of such Committed Borrowing if such Committed Borrowing is to be made on the same Business Day as the date the Administrative Agent received the applicable Commitment Loan Notice with respect to such Borrowing) that such Lender will not make available to the Administrative Agent such Lender's share of such Committed Borrowing, the Administrative Agent may assume that such Lender has made such share available on such date in accordance with Section 2.02 (or, in the case of a Committed Borrowing of Base Rate Loans, that such Lender has made such share available in accordance with and at the time required by Section 2.02) and may, in reliance upon such assumption, make available to the Borrower a corresponding amount. In such event, if a Lender has not in fact made its share of the applicable Committed Borrowing available to the Administrative Agent, then the applicable Lender and the Borrower severally agree to pay to the Administrative Agent forthwith on demand such corresponding amount in Same Day Funds with interest thereon, for each day from and including the date such amount is made available to the Borrower to but excluding the date of payment to the Administrative Agent, at (A) in the case of a payment to be made by such Lender, the Overnight Rate, plus any administrative, processing or similar fees customarily charged by the Administrative Agent in connection with the foregoing, and (B) in the case of a payment to be made by the Borrower, the interest rate applicable to Base Rate Loans. If the Borrower and such Lender shall pay such interest to the Administrative Agent for the same or an overlapping period, the Administrative Agent shall promptly remit to the Borrower the amount of such interest paid by the Borrower for such period. If such Lender pays its share of the applicable Committed Borrowing to the Administrative Agent, then the amount so paid shall constitute such Lender's Committed Loan included in such Committed Borrowing. Any payment by the Borrower shall be without prejudice to any claim the Borrower may have against a Lender that shall have failed to make such payment to the Administrative Agent.
 - (ii) <u>Payments by Borrower; Presumptions by Administrative Agent.</u> Unless the Administrative Agent shall have received notice from the Borrower prior to the date on which any payment is due to the Administrative Agent for the account of the Lenders or the L/C Issuer hereunder that the Borrower will not make such payment, the Administrative Agent may assume that the Borrower has made such payment on such date in accordance herewith and may, in reliance upon such assumption, distribute to the

Lenders or the L/C Issuer, as the case may be, the amount due. In such event, if the Borrower has not in fact made such payment, then each of the Lenders or the L/C Issuer, as the case may be, severally agrees to repay to the Administrative Agent forthwith on demand the amount so distributed to such Lender or the L/C Issuer, in Same Day Funds with interest thereon, for each day from and including the date such amount is distributed to it to but excluding the date of payment to the Administrative Agent, at the Overnight Rate.

A notice of the Administrative Agent to any Lender or the Borrower with respect to any amount owing under this subsection (b) shall be conclusive, absent manifest error.

- (c) <u>Failure to Satisfy Conditions Precedent.</u> If any Lender makes available to the Administrative Agent funds for any Loan to be made by such Lender to the Borrower as provided in the foregoing provisions of this <u>Article II</u>, and such funds are not made available to the Borrower by the Administrative Agent because the conditions to the applicable Credit Extension set forth in <u>Article IV</u> are not satisfied or waived in accordance with the terms hereof, the Administrative Agent shall return such funds (in like funds as received from such Lender) to such Lender, without interest.
- (d) Obligations of Lenders Several. The obligations of the Lenders hereunder to make Committed Loans, to fund participations in Letters of Credit and Swing Line Loans and to make payments pursuant to Section 10.04(c) are several and not joint. The failure of any Lender to make any Committed Loan, to fund any such participation or to make any payment under Section 10.04(c) on any date required hereunder shall not relieve any other Lender of its corresponding obligation to do so on such date, and no Lender shall be responsible for the failure of any other Lender to so make its Committed Loan, to purchase its participation or to make its payment under Section 10.04(c).
- (e) <u>Funding Source</u>. Nothing herein shall be deemed to obligate any Lender to obtain the funds for any Loan in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for any Loan in any particular place or manner.
- 2.13 Sharing of Payments by Lenders. If any Lender shall, by exercising any right of setoff or counterclaim or otherwise, obtain payment in respect of any principal of or interest on any of the Committed Loans made by it, or the participations in L/C Obligations or in Swing Line Loans held by it resulting in such Lender's receiving payment of a proportion of the aggregate amount of such Committed Loans or participations and accrued interest thereon greater than its <u>pro rata</u> share thereof as provided herein, then the Lender receiving such greater proportion shall (a) notify the Administrative Agent of such fact, and (b) purchase (for cash at face value) participations in the Committed Loans and subparticipations in L/C Obligations and Swing Line Loans of the other Lenders, or make such other adjustments as shall be equitable, so that the benefit of all such payments shall be shared by the Lenders ratably in accordance with the aggregate amount of principal of and accrued interest on their respective Committed Loans and other amounts owing them, <u>provided</u> that:

- (i) if any such participations or subparticipations are purchased and all or any portion of the payment giving rise thereto is recovered, such participations or subparticipations shall be rescinded and the purchase price restored to the extent of such recovery, without interest; and
- (ii) the provisions of this Section shall not be construed to apply to (x) any payment made by or on behalf of the Borrower pursuant to and in accordance with the express terms of this Agreement (including the application of funds arising from the existence of a Defaulting Lender), (y) the application of Cash Collateral provided for in Section 2.15, or (z) any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Committed Loans or subparticipations in L/C Obligations or Swing Line Loans to any assignee or participant, other than an assignment to the Borrower or any Subsidiary or Affiliate thereof (as to which the provisions of this Section shall apply).

The Borrower consents to the foregoing and agrees, to the extent it may effectively do so under applicable law, that any Lender acquiring a participation pursuant to the foregoing arrangements may exercise against the Borrower rights of setoff and counterclaim with respect to such participation as fully as if such Lender were a direct creditor of the Borrower in the amount of such participation.

2.14 Increase in Commitments.

- (a) Request for Increase. Provided there exists no Default, upon notice to the Administrative Agent (which shall promptly notify the Lenders), the Borrower may from time to time, request an increase in the Aggregate Commitments (which increase may take the form of one or more new revolving loan tranches) by an amount (for all such requests) not exceeding \$150,000,000; provided that (i) any such request for an increase shall be in a minimum amount of \$10,000,000, and (ii) the Borrower may make a maximum of three such requests (excluding any such requests that are not consummated). At the time of sending such notice, the Borrower (in consultation with the Administrative Agent) shall specify the time period within which each Lender is requested to respond (which shall in no event be less than ten Business Days from the date of delivery of such notice to the Lenders).
- (b) <u>Lender Elections to Increase</u>. Each Lender shall notify the Administrative Agent within such time period whether or not it agrees to increase its Commitment and, if so, whether by an amount equal to, greater than, or less than its Applicable Percentage of such requested increase. Any Lender not responding within such time period shall be deemed to have declined to increase its Commitment, and no Lender has any obligation to agree to increase its Commitment pursuant to this Section.
- (c) Notification by Administrative Agent; Additional Lenders. The Administrative Agent shall notify the Borrower and each Lender of the Lenders' responses to each request made hereunder. To achieve the full amount of a requested increase and subject to the approval of the Administrative Agent, each L/C Issuer and the Swing Line Lender (which approvals shall not be unreasonably withheld), the Borrower may also invite additional Eligible Assignees to become

Lenders pursuant to a joinder agreement in form and substance satisfactory to the Administrative Agent and its counsel.

- (d) <u>Effective Date and Allocations</u>. If the Aggregate Commitments are increased in accordance with this Section, the Administrative Agent and the Borrower shall determine the effective date (the "<u>Increase Effective Date</u>") and the final allocation of such increase. The Administrative Agent shall promptly notify the Borrower and the Lenders of the final allocation of such increase and the Increase Effective Date.
- (e) Conditions to Effectiveness of Increase. As a condition precedent to such increase, the Borrower shall deliver to the Administrative Agent a certificate of each Loan Party dated as of the Increase Effective Date (in sufficient copies for each Lender) signed by a Responsible Officer of such Loan Party (i) certifying and attaching the resolutions adopted by such Loan Party approving or consenting to such increase, and (ii) in the case of the Borrower, certifying that, before and after giving effect to such increase, (A) the representations and warranties contained in Article V and the other Loan Documents are true and correct in all material respects on and as of the Increase Effective Date, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they are true and correct in all material respects as of such earlier date, and except that for purposes of this Section 2.14, the representations and warranties contained in subsections (a) and (b) of Section 5.04 shall be deemed to refer to the most recent statements furnished pursuant to clauses (a) and (b), respectively, of Section 6.01, and (B) no Default exists. The Borrower shall prepay any Committed Loans outstanding on the Increase Effective Date (and pay any additional amounts required pursuant to Section 3.05) to the extent necessary to keep the outstanding Committed Loans ratable with any revised Applicable Percentages arising from any nonratable increase in the Commitments under this Section.
 - (f) Conflicting Provisions. This Section shall supersede any provisions in Section 2.13 or 10.01 to the contrary.

2.15 Cash Collateral.

- (a) <u>Certain Credit Support Events</u> Upon the request of the Administrative Agent or any L/C Issuer (i) if any L/C Issuer has honored any full or partial drawing request under any Letter of Credit and such drawing has resulted in an L/C Borrowing, the Borrower shall immediately Cash Collateralize such L/C Borrowing and, if requested by such L/C Issuer, all other Letters of Credit issued by such L/C Issuer, or (ii) if, as of the Letter of Credit Expiration Date, any L/C Obligation for any reason remains outstanding, the Borrower shall immediately Cash Collateralize the then Outstanding Amount of all L/C Obligations. At any time that there shall exist a Defaulting Lender, immediately upon the request of the Administrative Agent, the L/C Issuer or the Swing Line Lender, the Borrower shall deliver to the Administrative Agent Cash Collateral in an amount sufficient to cover all Fronting Exposure (after giving effect to Section 2.16(a)(iv) and any Cash Collateral provided by the Defaulting Lender).
- (b) Grant of Security Interest. All Cash Collateral (other than credit support not constituting funds subject to deposit) shall be maintained in blocked, non-interest bearing (unless otherwise agreed by the depositary) deposit accounts at the Administrative Agent or the relevant

L/C Issuer, as applicable. To the extent provided by the Borrower, the Borrower, and to the extent provided by any Lender, such Lender, hereby grants to (and subjects to the control of) the relevant L/C Issuer or to the Administrative Agent, for the benefit of the Administrative Agent, the L/C Issuer and the Lenders (including the Swing Line Lender), as applicable, and agrees to maintain, a first priority security interest in all such cash, deposit accounts and all balances therein, and in all proceeds of the foregoing, all as security for the obligations to which such Cash Collateral may be applied pursuant to Section 2.15(c). If at any time the Administrative Agent dermines that Cash Collateral is subject to any right or claim of any Person other than the Administrative Agent as herein provided, or that the total amount of such Collateral is less than the applicable Fronting Exposure and other obligations secured thereby, the Borrower or the relevant Defaulting Lender will, promptly (but in any event within five Business Days) after demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency.

- (c) <u>Application</u>. Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under any of this <u>Section 2.15</u> or <u>Sections 2.03</u>, <u>2.04</u>, <u>2.05</u>, <u>2.16</u> or <u>8.02</u> in respect of Letters of Credit or Swing Line Loans shall be held and applied to the satisfaction of the specific L/C Obligations, Swing Line Loans, obligations to fund participations therein (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) and other obligations for which the Cash Collateral was so provided, prior to any other application of such property as may be provided for herein.
- (d) Release. Cash Collateral (or the appropriate portion thereof) provided to reduce Fronting Exposure or other obligations shall be released promptly following (i) the elimination of the applicable Fronting Exposure or other obligations giving rise thereto (including by the termination of Defaulting Lender status of the applicable Lender (or, as appropriate, its assignee following compliance with Section 10.06(b)(vi))) or (ii) the Administrative Agent's good faith determination that there exists excess Cash Collateral; provided that (x) Cash Collateral furnished by or on behalf of a Loan Party shall not be released during the continuance of a Default (and following application as provided in this Section 2.15 may be otherwise applied in accordance with Section 8.03), and (y) the Person providing Cash Collateral and the L/C Issuer or Swing Line Lender, as applicable, may agree that Cash Collateral shall not be released but instead held to support future anticipated Fronting Exposure or other obligations.

2.16 Defaulting Lenders.

- (a) <u>Adjustments</u>. Notwithstanding anything to the contrary contained in this Agreement, if any Lender becomes a Defaulting Lender, then, until such time as that Lender is no longer a Defaulting Lender, to the extent permitted by applicable Requirements of Law:
 - (i) Waivers and Amendments. That Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in Section 10.01.
 - (ii) Reallocation of Payments. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of that Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Article VIII or

otherwise, and including any amounts made available to the Administrative Agent by that Defaulting Lender pursuant to Section 10.08), shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by that Defaulting Lender to the Administrative Agent hereunder; second, to the payment on a pro rata basis of any amounts owing by that Defaulting Lender to the L/C Issuer or Swing Line Lender hereunder; third, if so determined by the Administrative Agent or requested by the L/C Issuer or Swing Line Lender, to be held as Cash Collateral for future funding obligations of that Defaulting Lender of any participation in any Swing Line Loan or Letter of Credit; fourth, as the Borrower may request (so long as no Default exists), to the funding of any Loan in respect of which that Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, if so determined by the Administrative Agent and the Borrower, to be held in a non-interest bearing (unless otherwise agreed by the depositary) deposit account and released in order to satisfy obligations of that Defaulting Lender to fund Loans under this Agreement; sixth, to the payment of any amounts owing to the Lenders, the L/C Issuer or Swing Line Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, the L/C Issuer or Swing Line Lender against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against that Defaulting Lender as a result of that Defaulting Lender's breach of its obligations under this Agreement; and eighth, to that Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or L/C Borrowings in respect of which that Defaulting Lender has not fully funded its appropriate share and (y) such Loans or L/C Borrowings were made at a time when the conditions set forth in Section 4.02 were satisfied or waived, such payment shall be applied solely to pay the Loans of, and L/C Borrowings owed to, all non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or L/C Borrowings owed to, that Defaulting Lender. Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this Section 2.16(a)(ii) shall be deemed paid to and redirected by that Defaulting Lender, and each Lender irrevocably consents hereto.

- (iii) <u>Certain Fees</u>. That Defaulting Lender (x) shall not be entitled to receive any commitment fee pursuant to $\underline{Section 2.09(a)}$ for any period during which that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender) and (y) shall be limited in its right to receive Letter of Credit Fees as provided in $\underline{Section 2.03(h)}$.
- (iv) Reallocation of Applicable Percentages to Reduce Fronting Exposure. During any period in which there is a Defaulting Lender, for purposes of computing the amount of the obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit or Swing Line Loans pursuant to Sections 2.03 and 2.04, the "Applicable Percentage" of each non-Defaulting Lender shall be computed

without giving effect to the Commitment of that Defaulting Lender; provided that (i) each such reallocation shall be given effect only if, at the date the applicable Lender becomes a Defaulting Lender, no Default exists; and (ii) the aggregate obligation of each non-Defaulting Lender to acquire, refinance or fund participations in Letters of Credit and Swing Line Loans shall not exceed the positive difference, if any, of (1) the Commitment of that non-Defaulting Lender minus (2) the aggregate Outstanding Amount of the Committed Loans of that Lender.

- (b) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, Swing Line Lender and the L/C Issuer agree in writing in their sole discretion that a Defaulting Lender should no longer be deemed to be a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase that portion of outstanding Loans of the other Lenders or take such other actions as the Administrative Agent may determine to be necessary to cause the Committed Loans and funded and unfunded participations in Letters of Credit and Swing Line Loans to be held on a pro rata basis by the Lenders in accordance with their Applicable Percentages (without giving effect to <u>Section 2.16(a)(iv)</u>), whereupon that Lender will cease to be a Defaulting Lender; <u>provided</u> that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and <u>provided</u>, <u>further</u>, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- 2.17 Borrower Succession. Upon the effectiveness of the Spinoff in accordance with the provisions of this Agreement and the satisfaction of the requirements and conditions set forth in Section 6.24 hereof, the New Borrower shall replace the Original Borrower as the Borrower hereunder, and all references herein to the "Borrower" shall be deemed to be references to the New Borrower.

ARTICLE III. TAXES, YIELD PROTECTION AND ILLEGALITY

3.01 Taxes.

- (a) Payments Free of Taxes; Obligation to Withhold; Payments on Account of Taxes.
- (i) Any and all payments by or on account of any obligation of any Loan Party hereunder or under any other Loan Document shall to the extent permitted by applicable Requirements of Law be made free and clear of and without reduction or withholding for any Taxes. If, however, applicable Requirements of Law require such Loan Party or the Administrative Agent to withhold or deduct any Tax, such Tax shall be withheld or deducted in accordance with such Requirements of Law as determined by such Loan Party or the Administrative Agent, as the case may be, upon the basis of the information and documentation to be delivered pursuant to subsection (e) below.

- (ii) If the Borrower or the Administrative Agent shall be required by the Code to withhold or deduct any Taxes, including both United States Federal backup withholding and withholding Taxes, from any payment made hereunder or under any other Loan Document, then (A) the Administrative Agent shall withhold or make such deductions as are determined by the Administrative Agent to be required based upon the information and documentation it has received pursuant to subsection (e) below, (B) the Administrative Agent shall timely pay the full amount withheld or deducted to the relevant Governmental Authority in accordance with the Code, and (C) to the extent that the withholding or deduction is made on account of Indemnified Taxes or Other Taxes, the sum payable by the Borrower shall be increased as necessary so that after any required withholding or the making of all required deductions (including deductions applicable to additional sums payable under this Section) the Administrative Agent, Lender or L/C Issuer, as the case may be, receives an amount equal to the sum it would have received had no such withholding or deduction for Indemnified Taxes or Other Taxes been made.
- (iii) If the Borrower or the Administrative Agent shall be required by any applicable Requirements of Law other than the Code to withhold or deduct any Taxes from any payment made hereunder or under any other Loan Document, then (A) the Borrower or the Administrative Agent, as required by such Requirements of Law, shall withhold or make such deductions as are determined by it to be required based upon the information and documentation it has received pursuant to subsection (e) below, (B) the Borrower or the Administrative Agent, to the extent required by such Requirements of Law, shall timely pay the full amount so withheld or deducted by it to the relevant Governmental Authority in accordance with such Requirements of Law, and (C) to the extent that the withholding or deduction is made on account of Indemnified Taxes or Other Taxes, the sum payable by the Borrower shall be increased as necessary so that after any required withholding or the making of all required deductions (including deductions applicable to additional sums payable under this Section) the Administrative Agent, Lender or L/C Issuer, as the case may be, receives an amount equal to the sum it would have received had no such withholding or deduction for Indemnified Taxes or Other Taxes been made.
- (b) <u>Payment of Other Taxes by the Loan Parties</u>. Without limiting the provisions of subsection (a) above, the each Loan Party shall timely pay any Other Taxes to the relevant Governmental Authority in accordance with applicable Requirements of Law.

(c) Tax Indemnifications.

(i) Without limiting the provisions of subsection (a) or (b) above, each Loan Party shall, and does hereby, indemnify the Administrative Agent, each Lender and the L/C Issuer, and shall make payment in respect thereof within 10 days after demand therefor, for the full amount of any Indemnified Taxes or Other Taxes (including Indemnified Taxes or Other Taxes imposed or asserted on or attributable to amounts payable under this Section) paid by the Administrative Agent, such Lender or the L/C Issuer, as the case may be, with respect to the Obligations hereunder and any penalties, interest and reasonable expenses arising therefrom or with respect thereto, whether or not

such Indemnified Taxes or Other Taxes were correctly or legally imposed or asserted by the relevant Governmental Authority. Each Loan Party shall also, and does hereby, indemnify the Administrative Agent, and shall make payment in respect thereof within 10 days after demand therefor, for any amount which a Lender or the L/C Issuer for any reason fails to pay indefeasibly to the Administrative Agent as required by clause (ii) of this subsection. A certificate as to the amount of any such payment or liability delivered to the applicable Loan Party by a Lender or the L/C Issuer (with a copy to the Administrative Agent), or by the Administrative Agent on its own behalf or on behalf of a Lender or the L/C Issuer, shall be conclusive absent manifest error. Notwithstanding anything to the contrary in this Agreement or any other Loan Document, the Loan Parties are not indemnifying any Person for Excluded Taxes.

- (ii) Without limiting the provisions of subsection (a) or (b) above, each Lender and the L/C Issuer shall, and does hereby, indemnify the Loan Parties and the Administrative Agent, and shall make payment in respect thereof within 10 days after demand therefor, against any and all Taxes and any and all related losses, claims, liabilities, penalties, interest and expenses (including the fees, charges and disbursements of any counsel for the Borrower or the Administrative Agent) incurred by or imposed or asserted against any Loan Party or the Administrative Agent by any Governmental Authority, whether or not such Taxes were correctly or legally imposed or asserted, as a result of the failure by such Lender or the L/C Issuer, as the case may be, to deliver, or as a result of the inaccuracy, inadequacy or deficiency of, any documentation required to be delivered by such Lender or the L/C Issuer, as the case may be, to the applicable Loan Party or the Administrative Agent pursuant to subsection (e). Each Lender and the L/C Issuer hereby authorizes the Administrative Agent to set off and apply any and all amounts at any time owing to such Lender or the L/C Issuer, as the case may be, under this Agreement or any other Loan Document against any amount due to the Administrative Agent under this clause (ii). The agreements in this clause (ii) shall survive the resignation and/or replacement of the Administrative Agent, any assignment of rights by, or the replacement of, a Lender or the L/C Issuer, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all other Obligations.
- (d) Evidence of Payments. Upon request by the Borrower or the Administrative Agent, as the case may be, after any payment of Taxes by the Borrower or by the Administrative Agent to a Governmental Authority as provided in this Section 3.01, the Borrower shall deliver to the Administrative Agent or the Administrative Agent shall deliver to the Borrower, as the case may be, the original or a certified copy of a receipt issued by such Governmental Authority evidencing such payment, a copy of any return required by Requirements of Law to report such payment or other evidence of such payment reasonably satisfactory to the Borrower or the Administrative Agent, as the case may be.

(e) Status of Lenders; Tax Documentation.

(i) Each Lender shall deliver to the Borrower and to the Administrative Agent, at the time or times prescribed by applicable Requirements of Law or when reasonably requested by the Borrower or the Administrative Agent, such properly

completed and executed documentation prescribed by applicable Requirements of Law or by the taxing authorities of any jurisdiction and such other reasonably requested information as will permit the Borrower or the Administrative Agent, as the case may be, to determine (A) whether or not payments made by the Borrower hereunder or under any other Loan Document are subject to Taxes, (B) if applicable, the required rate of withholding or deduction, and (C) such Lender's entitlement to any available exemption from, or reduction of, applicable Taxes in respect of all payments to be made to such Lender by the Borrower pursuant to this Agreement or otherwise to establish such Lender's status for withholding tax purposes in the applicable jurisdictions.

- (ii) Without limiting the generality of the foregoing, if the Borrower is resident for tax purposes in the United States,
- (A) any Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the Code shall deliver to the Borrower and the Administrative Agent executed originals of IRS Form W-9 or such other documentation or information prescribed by applicable Requirements of Law or reasonably requested by the Borrower or the Administrative Agent as will enable the Borrower or the Administrative Agent, as the case may be, to determine whether or not such Lender is subject to backup withholding or information reporting requirements; and
- (B) each Foreign Lender that is entitled under the Code or any applicable treaty to an exemption from or reduction of withholding tax with respect to payments hereunder or under any other Loan Document shall deliver to the Borrower and the Administrative Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Lender under this Agreement (and from time to time thereafter upon the request of the Borrower or the Administrative Agent, but only if such Foreign Lender is legally entitled to do so), whichever of the following is applicable:
 - (I) executed originals of IRS Form W-8BEN claiming eligibility for benefits of an income tax treaty to which the United States is a party,
 - (II) executed originals of IRS Form W-8ECI,
 - (III) executed originals of IRS Form W-8IMY and all required supporting documentation,
 - (IV) in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under section 881(c) of the Code, (x) a certificate to the effect that such Foreign Lender is not (A) a "bank" within the meaning of section 881(c)(3)(A) of the Code, (B) a "10 percent shareholder" of the Borrower within the meaning of section 881(c)(3)(B) of the Code, or (C) a "controlled foreign corporation" described in section 881(c)(3)(C) of the Code and (y) executed originals of IRS Form W-

8BEN, or

- (V) executed originals of any other form prescribed by applicable Requirements of Law as a basis for claiming exemption from or a reduction in United States Federal withholding Tax together with such supplementary documentation as may be prescribed by applicable Requirements of Law to permit the Borrower or the Administrative Agent to determine the withholding or deduction required to be made.
- (iii) Each Lender shall promptly (A) notify the Borrower and the Administrative Agent of any change in circumstances which would modify or render invalid any claimed exemption or reduction, and (B) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Lender, and as may be reasonably necessary (including the re-designation of its Lending Office) to avoid any requirement of applicable Requirements of Law of any jurisdiction that the Borrower or the Administrative Agent make any withholding or deduction for taxes from amounts payable to such Lender.
- (f) Treatment of Certain Refunds Unless required by applicable Requirements of Law, at no time shall the Administrative Agent have any obligation to file for or otherwise pursue on behalf of a Lender or the L/C Issuer, or have any obligation to pay to any Lender or the L/C Issuer, any refund of Taxes withheld or deducted from funds paid for the account of such Lender or the L/C Issuer, as the case may be. If the Administrative Agent, any Lender or the L/C Issuer determines, in its sole discretion, that it has received a refund of any Taxes or Other Taxes or a direct offset against its Tax liabilities as to which it has been indemnified by a Loan Party or with respect to which any Loan Party has paid additional amounts pursuant to this Section, it shall promptly after such determination pay to the applicable Loan Party an amount equal to such refund (but only to the extent of indemnity payments made, or additional amounts paid, by such Loan Party under this Section with respect to the Taxes or Other Taxes giving rise to such refund) or such direct offset (but only to the extent of the net Tax benefit obtained by it as a result of such payment by the applicable Loan Party), net of all out-of-pocket expenses and net of any loss or gain realized in the conversion of such funds from or to another currency incurred by the Administrative Agent, such Lender or the L/C Issuer, as the case may be, and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund), provided that such Loan Party, upon the request of the Administrative Agent, such Lender or the L/C Issuer is required to repay such refund to such Governmental Authority. This subsection shall not be construed to require the Administrative Agent, any Lender or the L/C Issuer to make available its tax returns (or any other information relating to its Taxes that it deems confidential) to a Loan Party or any other Person.
- 3.02 Illegality. If any Lender determines that any Requirement of Law has made it unlawful, or that any Governmental Authority has asserted that it is unlawful, for any Lender or its applicable Lending Office to make, maintain or fund Loans whose interest is determined by reference to the Eurocurrency Rate, or to determine or charge interest rates based upon the

Eurocurrency Rate, or any Governmental Authority has imposed material restrictions on the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the applicable interbank market, then, on notice thereof by such Lender to the Borrower through the Administrative Agent, (a) any obligation of such Lender to make or continue Eurocurrency Rate Loans or to convert Base Rate Committed Loans to Eurocurrency Rate Loans shall be suspended, and (b) if such notice asserts the illegality of such Lender making or maintaining Base Rate Loans the interest rate on which is determined by reference to the Eurocurrency Rate component of the Base Rate, the interest rate on which Base Rate Loans of such Lender shall, if necessary to avoid such illegality, be determined by the Administrative Agent without reference to the Eurocurrency Rate component of the Base Rate, in each case until such Lender notifies the Administrative Agent and the Borrower that the circumstances giving rise to such determination no longer exist. Upon receipt of such notice, (x) the Borrower shall, upon demand from such Lender (with a copy to the Administrative Agent), prepay or, if applicable, convert all Eurocurrency Rate Loans of such Lender to Base Rate Loans (the interest rate on which Base Rate Loans of such Lender shall, if necessary to avoid such illegality, be determined by the Administrative Agent without reference to the Eurocurrency Rate component of the Base Rate), either on the last day of the Interest Period therefor, if such Lender may lawfully continue to maintain such Eurocurrency Rate Loans to such day, or immediately, if such Lender may not lawfully continue to maintain such Eurocurrency Rate, the Administrative Agent shall during the period of such suspension compute the Base Rate applicable to such Lender without reference to the Eurocurrency Rate component thereof until the Administrative is advised in writing by such Lender that it is no longer illegal for such Lender to determine or charge interest rates based u

3.03 Inability to Determine Rates. If the Required Lenders determine that for any reason in connection with any request for a Eurocurrency Rate Loan or a conversion to or continuation thereof that (a) Dollar deposits are not being offered to banks in the London interbank eurodollar market for the applicable amount and Interest Period of such Eurocurrency Rate Loan, (b) adequate and reasonable means do not exist for determining the Eurocurrency Rate for any requested Interest Period with respect to a proposed Eurocurrency Rate Loan or in connection with an existing or proposed Base Rate Loan, or (c) the Eurocurrency Rate for any requested Interest Period with respect to a proposed Eurocurrency Rate Loan does not adequately and fairly reflect the cost to such Lenders of funding such Loan, the Administrative Agent will promptly so notify the Borrower and each Lender. Thereafter, (x) the obligation of the Lenders to make or maintain Eurocurrency Rate Loans shall be suspended, and (y) in the event of a determination described in the preceding sentence with respect to the Eurocurrency Rate component of the Base Rate, the utilization of the Eurocurrency Rate component in determining the Base Rate shall be suspended, in each case until the Administrative Agent (upon the instruction of the Required Lenders) revokes such notice. Upon receipt of such notice, the Borrower may revoke any pending request for a Borrowing of, conversion to or continuation of Eurocurrency Rate Loans or, failing that, will be deemed to have converted such request into a request for a Committed Borrowing of Base Rate Loans in the amount specified therein.

3.04 Increased Costs; Reserves on Eurocurrency Rate Loans.

- (a) Increased Costs Generally. If any Change in Law shall:
- (i) impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge or similar requirement against assets of, deposits with or for the account of, or credit extended or participated in by, any Lender (except any reserve requirement contemplated by <u>Section 3.04(e)</u>, other than as set forth below) or the L/C Issuer:
- (ii) subject any Lender or the L/C Issuer to any tax of any kind whatsoever with respect to this Agreement, any Letter of Credit, any participation in a Letter of Credit or any Eurocurrency Rate Loan made by it, or change the basis of taxation of payments to such Lender or the L/C Issuer in respect thereof (except for Indemnified Taxes or Other Taxes covered by Section 3.01 and the imposition of, or any change in the rate of, any Excluded Tax payable by such Lender or the L/C Issuer);
- (iii) impose on any Lender or the L/C Issuer or the London interbank market any other condition, cost or expense affecting this Agreement or Eurocurrency Rate Loans made by such Lender or any Letter of Credit or participation therein;

and the result of any of the foregoing shall be to increase the cost to such Lender of making or maintaining any Loan the interest on which is determined by reference to the Eurocurrency Rate (or of maintaining its obligation to make any such Loan), or to increase the cost to such Lender or the L/C Issuer of participating in, issuing or maintaining any Letter of Credit (or of maintaining its obligation to participate in or to issue any Letter of Credit), or to reduce the amount of any sum received or receivable by such Lender or the L/C Issuer hereunder (whether of principal, interest or any other amount) then, upon request of such Lender or the L/C Issuer, the Borrower will pay to such Lender or the L/C Issuer, as the case may be, for such additional costs incurred or reduction suffered.

(b) <u>Capital Requirements</u>. If any Lender or the L/C Issuer determines that any Change in Law affecting such Lender or the L/C Issuer or any Lending Office of such Lender or such Lender's or the L/C Issuer's holding company, if any, regarding capital requirements has or would have the effect of reducing the rate of return on such Lender's or the L/C Issuer's capital or on the capital of such Lender's or the L/C Issuer's holding company, if any, as a consequence of this Agreement, the Commitments of such Lender or the Loans made by, or participations in Letters of Credit held by, such Lender, or the Letters of Credit issued by the L/C Issuer, to a level below that which such Lender or the L/C Issuer or such Lender's or the L/C Issuer's holding company could have achieved but for such Change in Law (taking into consideration such Lender's or the L/C Issuer's policies and the policies of such Lender's or the L/C Issuer's holding company with respect to capital adequacy), then from time to time the Borrower will pay to such Lender or the L/C Issuer, as the case may be, such additional amount or amounts as will compensate such Lender or the L/C Issuer's holding company for any such reduction suffered.

- (c) <u>Certificates for Reimbursement.</u> A certificate of a Lender or the L/C Issuer setting forth the amount or amounts necessary to compensate such Lender or the L/C Issuer or its holding company, as the case may be, as specified in subsection (a) or (b) of this Section and delivered to the Borrower shall be conclusive absent manifest error. The Borrower shall pay such Lender or the L/C Issuer, as the case may be, the amount shown as due on any such certificate within 10 days after receipt thereof.
- (d) <u>Delay in Requests</u>. Failure or delay on the part of any Lender or the L/C Issuer to demand compensation pursuant to the foregoing provisions of this Section shall not constitute a waiver of such Lender's or the L/C Issuer's right to demand such compensation, <u>provided</u> that the Borrower shall not be required to compensate a Lender or the L/C Issuer pursuant to the foregoing provisions of this Section for any increased costs incurred or reductions suffered more than nine months prior to the date that such Lender or the L/C Issuer, as the case may be, notifies the Borrower of the Change in Law giving rise to such increased costs or reductions and of such Lender's or the L/C Issuer's intention to claim compensation therefor (except that, if the Change in Law giving rise to such increased costs or reductions is retroactive, then the nine-month period referred to above shall be extended to include the period of retroactive effect thereof).
- (e) Additional Reserve Requirements. The Borrower shall pay to each Lender, (i) as long as such Lender shall be required to maintain reserves with respect to liabilities or assets consisting of or including Eurocurrency funds or deposits (currently known as "Eurocurrency liabilities"), additional interest on the unpaid principal amount of each Eurocurrency Rate Loan equal to the actual costs of such reserves allocated to such Loan by such Lender (as determined by such Lender in good faith, which determination shall be conclusive), and (ii) as long as such Lender shall be required to comply with any reserve ratio requirement or analogous requirement of any other central banking or financial regulatory authority imposed in respect of the maintenance of the Commitments or the funding of the Eurocurrency Rate Loans, such additional costs (expressed as a percentage per annum and rounded upwards, if necessary, to the nearest five decimal places) equal to the actual costs allocated to such Commitment or Loan by such Lender (as determined by such Lender in good faith, which determination shall be conclusive), which in each case shall be due and payable on each date on which interest is payable on such Loan, provided the Borrower shall have received at least 10 days' prior notice (with a copy to the Administrative Agent) of such additional interest or costs from such Lender. If a Lender fails to give notice 10 days prior to the relevant Interest Payment Date, such additional interest or costs shall be due and payable 10 days from receipt of such notice.
- 3.05 Compensation for Losses. Upon demand of any Lender (with a copy to the Administrative Agent) from time to time, the Borrower shall promptly compensate such Lender for and hold such Lender harmless from any loss, cost or expense incurred by it as a result of:
- (a) any continuation, conversion, payment or prepayment of any Loan other than a Base Rate Loan on a day other than the last day of the Interest Period for such Loan (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise);
- (b) any failure by the Borrower (for a reason other than the failure of such Lender to make a Loan) to prepay, borrow, continue or convert any Loan other than a Base Rate Loan on the date or in the amount notified by the Borrower;

- (c) any failure by the Borrower to make payment of any drawing under any Letter of Credit (or interest due thereon) denominated in an Alternative Currency on its scheduled due date or any payment thereof in a different currency; or
- (d) any assignment of a Eurocurrency Rate Loan on a day other than the last day of the Interest Period therefor as a result of a request by the Borrower pursuant to Section 10.13;

including any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Loan, or from fees payable to terminate the deposits from which such funds were obtained. The Borrower shall also pay any customary administrative fees charged by such Lender in connection with the foregoing. A certificate of a Lender setting forth the amount of any such loss or expense provided for in this Section and delivered to the Borrower shall be conclusive absent manifest error.

For purposes of calculating amounts payable by the Borrower to the Lenders under this <u>Section 3.05</u>, each Lender shall be deemed to have funded each Eurocurrency Rate Loan made by it at the Eurocurrency Rate for such Loan by a matching deposit or other borrowing in the offshore interbank market for such currency for a comparable amount and for a comparable period, whether or not such Eurocurrency Rate Loan was in fact so funded.

3.06 Mitigation Obligations; Replacement of Lenders.

- (a) <u>Designation of a Different Lending Office</u>. If any Lender requests compensation under <u>Section 3.04</u>, or the Borrower is required to pay any additional amount to any Lender, the L/C Issuer, or any Governmental Authority for the account of any Lender or the L/C Issuer pursuant to <u>Section 3.01</u>, or if any Lender gives a notice pursuant to <u>Section 3.02</u>, then such Lender or the L/C Issuer shall, as applicable, use reasonable efforts to designate a different Lending Office for funding or booking its Loans hereunder or to assign its rights and obligations hereunder to another of its offices, branches or affiliates, if, in the judgment of such Lender or the L/C Issuer, such designation or assignment (i) would eliminate or reduce amounts payable pursuant to <u>Section 3.01</u> or <u>3.04</u>, as the case may be, in the future, or eliminate the need for the notice pursuant to <u>Section 3.02</u>, as applicable, and (ii) in each case, would not subject such Lender or the L/C Issuer, as the case may be, to any unreimbursed cost or expense and would not otherwise be disadvantageous to such Lender or the L/C Issuer, as the case may be. The Borrower hereby agrees to pay all reasonable costs and expenses incurred by any Lender or the L/C Issuer in connection with any such designation or assignment.
- (b) <u>Replacement of Lenders</u>. If any Lender requests compensation under <u>Section 3.04</u>, or if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to <u>Section 3.01</u>, the Borrower may replace such Lender in accordance with <u>Section 10.13</u>.
- 3.07 Survival. All of the Borrower's obligations under this Article III shall survive termination of the Aggregate Commitments, repayment of all other Obligations hereunder, and resignation of the Administrative Agent.

ARTICLE IV. CONDITIONS PRECEDENT TO CREDIT EXTENSIONS

- **4.01 Conditions of Initial Credit Extension.** The effectiveness of this Agreement, and the obligation of the L/C Issuer and each Lender to make its initial Credit Extension hereunder, are each subject to satisfaction of the following conditions precedent:
- (a) The Administrative Agent's receipt of the following, each of which shall be originals or telecopies (followed promptly by originals) unless otherwise specified, each properly executed by a Responsible Officer of the signing Loan Party, each dated the Closing Date (or, in the case of certificates of governmental officials, a recent date before the Closing Date) and each in form and substance satisfactory to the Administrative Agent and each of the Lenders:
 - (i) executed counterparts of this Agreement and the Guaranty, sufficient in number for distribution to the Administrative Agent, each Lender and the Borrower;
 - (ii) a Note executed by the Borrower in favor of each Lender requesting a Note;
 - (iii) executed counterparts of each Security Instrument to be entered into by any Loan Party (other than the BWXT Entities) on the Closing Date, duly executed by each Loan Party party thereto, together with:
 - (A) certificates representing the certificated Pledged Interests pledged under the Collateral Agreement, and accompanied by undated stock or other transfer powers executed in blank,
 - (B) proper financing statements in form appropriate for filing under the Uniform Commercial Code of all jurisdictions that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the Collateral Agreement, covering the Collateral described therein,
 - (C) completed requests for information, dated on or before the Closing Date, listing all effective financing statements filed in the jurisdictions referred to in clause (B) above that name any Loan Party as debtor, together with copies of such other financing statements, and
 - (D) evidence of the completion of all other actions, recordings and filings of or with respect to the Security Instruments to be entered into on the Closing Date that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created thereby (including receipt of duly executed payoff letters and UCC-3 termination statements), and
 - (E) evidence that all action that the Administrative Agent may deem necessary or desirable in order to perfect the Liens created under the IP Security Agreement has been taken or will be taken on or after the Closing Date;

- (iv) with respect to each of the Mortgaged Properties listed on Schedule 4.01(a)(iv), except to the extent waived by the Administrative Agent (in which case Section 6.29 shall apply to any matters set forth below that are so waived), each of the following:
 - (A) evidence that counterparts of the Mortgages have been duly executed, acknowledged and delivered and are in form suitable for filing or recording in all filing or recording offices that the Administrative Agent may deem necessary or desirable in order to create a valid first and subsisting Lien on the property described therein in favor of the Administrative Agent for the benefit of the Secured Parties, excepting only Liens permitted under the Loan Documents, and that all filing, documentary, stamp, intangible and recording taxes and fees have been paid (or the Borrower has made arrangements satisfactory to the Administrative Agent for payment thereof),
 - (B) a mortgagee's title insurance policy (or policies) (the "Mortgagee Policies") or marked up unconditional binder for such insurance, with endorsements and in amounts acceptable to the Administrative Agent, issued, coinsured and reinsured by title insurers acceptable to the Administrative Agent, insuring the Mortgages to be valid first and subsisting Liens on the property described therein, free and clear of all defects (including, but not limited to, mechanics' and materialmen's Liens) and encumbrances, excepting only Liens permitted under the Loan Documents,
 - (C) evidence that all premiums in respect of the Mortgagee Policies have been paid (or the Borrower has made arrangements satisfactory to the Administrative Agent for payment thereof),
 - (D) evidence that no such Mortgaged Property is located in a special flood hazard area as designated by any federal Governmental Authority other than those for which flood insurance has been provided, and evidence of any such flood insurance, and
 - (E) evidence that all other action that the Administrative Agent may deem necessary or desirable in order to create valid first and subsisting Liens (excepting only Liens permitted under the Loan Documents) on the property described in the Mortgages has been taken;
- (v) such certificates of resolutions or other action, incumbency certificates and/or other certificates of Responsible Officers of each Loan Party as the Administrative Agent may require evidencing the identity, authority and capacity of each Responsible Officer thereof authorized to act as a Responsible Officer in connection with this Agreement and the other Loan Documents to which such Loan Party is a party or is to be a party;
- (vi) such documents and certifications as the Administrative Agent may reasonably require to evidence that each Loan Party is duly organized or formed, and that

each Loan Party is validly existing and in good standing in its jurisdiction of organization;

- (vii) a favorable opinion of (A) Baker Botts L.L.P., counsel to the Loan Parties, (B) Liane K. Hinrichs, General Counsel of the Borrower, or James Canafax, Assistant General Counsel of the Borrower, (C) Vorys, Sater, Seymour and Pease LLP, local Ohio counsel to certain of the Loan Parties (except to the extent any such opinion relates to matters required by Section 4.01(a)(iv) which are waived to a post-closing date and covered by Section 6.29), and (D) Watkins Ludlam Winter & Stennis, P.A., local Mississippi counsel to certain of the Loan Parties (except to the extent any such opinion relates to matters required by Section 4.01(a)(iv) which are waived to a post-closing date and covered by Section 6.29), in each case addressed to the Administrative Agent and each Lender, in form and substance reasonably satisfactory to the Administrative Agent and the Lenders and addressing such matters concerning the Loan Parties and the Loan Documents as the Required Lenders may reasonably request;
- (viii) a certificate of a Responsible Officer of the Borrower either (A) attaching copies of all consents, licenses and approvals required in connection with the execution, delivery and performance by each Loan Party and the validity against each Loan Party of the Loan Documents to which it is a party, and such consents, licenses and approvals shall be in full force and effect, or (B) stating that no such consents, licenses or approvals are so required;
- (ix) a certificate signed by a Responsible Officer of the Borrower certifying (A) that the conditions specified in Sections 4.02(a) and (b) have been satisfied, (B) that there has been no event or circumstance since December 31, 2008 that has had or would be reasonably expected to have, either individually or in the aggregate, a Material Adverse Effect; and (C) to (and providing such backup evidence as may reasonably be requested) the current Debt Ratings and the current corporate family and corporate ratings of the Borrower and its Subsidiaries from each of Moody's and S&P;
- (x) a duly completed Compliance Certificate signed by the Chief Financial Officer or the Treasurer of the Borrower, demonstrating compliance as of the last day of the Fiscal Year ended on December 31, 2009 with the financial covenants in Section 7.18 after giving pro forma to the incurrence and repayment of Indebtedness on the Closing Date (and providing such backup evidence as may reasonably be requested);
- (xi) evidence that all insurance required to be maintained pursuant to the Loan Documents has been obtained and is in effect, together with the certificates of insurance or other appropriate documentation, naming the Administrative Agent, on behalf of the Secured Parties, as an additional insured or loss payee, as the case may be, under all insurance policies maintained with respect to the assets and properties of the Loan Parties that constitute Collateral;
- (xii) evidence that each of the Existing Credit Agreements has been or concurrently with the Closing Date is being paid in full and terminated and all Liens securing obligations under each of the Existing Credit Agreements have been or

concurrently with the Closing Date are being released (including receipt of duly executed payoff letters and UCC-3 termination statements);

- (xiii) such documentation and other information as has been reasonably requested by the Administrative Agent or any Lender prior to the Closing Date in connection with the provisions of Section 6.10 hereof;
- (xiv) copies of the Historical Financial Statements and the audited consolidated financial statements of the Borrower and its Subsidiaries for Fiscal Years 2007 and 2008, and (if applicable) any interim unaudited financial statements for each quarterly period ended since December 31, 2009;
- (xv) a copy of the Form 10, in form and substance reasonably satisfactory to the Administrative Agent and the Lenders, with respect to the Spinoff (including the pro forma consolidated financial statements of the New Borrower required by such Form 10); and
- (xvi) such other assurances, certificates, documents, consents or opinions as the Administrative Agent, the L/C Issuer, the Swing Line Lender or the Required Lenders reasonably may require.
- (b) (i) All fees required to be paid to the Administrative Agent and the Arrangers on or before the Closing Date shall have been paid and (ii) all fees required to be paid to the Lenders on or before the Closing Date shall have been paid, in each case pursuant to the Fee Letters.
- (c) Unless waived by the Administrative Agent, the Borrower shall have paid all reasonable fees, charges and disbursements of counsel to the Administrative Agent (directly to such counsel if requested by the Administrative Agent) to the extent invoiced at least one Business Day prior to the Closing Date, plus such additional amounts of such fees, charges and disbursements as shall constitute its reasonable estimate of such fees, charges and disbursements incurred or to be incurred by it through the closing proceedings (provided that such estimate shall not thereafter preclude a final settling of accounts between the Borrower and the Administrative Agent).

Without limiting the generality of the provisions of the last paragraph of Section 9.03, for purposes of determining compliance with the conditions specified in this Section 4.01, each Lender that has signed this Agreement shall be deemed to have consented to, approved or accepted or to be satisfied with, each document or other matter required thereunder to be consented to or approved by or acceptable or satisfactory to a Lender unless the Administrative Agent shall have received notice from such Lender prior to the proposed Closing Date specifying its objection thereto.

4.02 Conditions to all Credit Extensions. The obligation of each Lender to honor any Request for Credit Extension (other than a Committed Loan Notice requesting only a conversion of Committed Loans to the other Type, or a continuation of Eurocurrency Rate Loans) is subject to the following conditions precedent:

- (a) The representations and warranties of (i) the Borrower contained in Article V and (ii) each Loan Party contained in each other Loan Document or in any document furnished at any time under or in connection herewith or therewith, shall be true and correct in all material respects on and as of the date of such Credit Extension, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct in all material respects as of such earlier date, and except that for purposes of this Section 4.02, the representations and warranties contained in subsections (a) and (b) of Section 5.04 shall be deemed to refer to the most recent statements furnished pursuant to clauses (a) and (b), respectively, of Section 6.01.
 - (b) No Default shall exist, or would result from such proposed Credit Extension or the application of the proceeds thereof.
- (c) The Administrative Agent and, if applicable, the L/C Issuer or the Swing Line Lender shall have received a Request for Credit Extension in accordance with the requirements hereof.
- (d) In the case of an L/C Credit Extension to be denominated in an Alternative Currency, there shall not have occurred any change in national or international financial, political or economic conditions or currency exchange rates or exchange controls which in the reasonable opinion of the applicable L/C Issuer would make it impracticable for such L/C Credit Extension to be denominated in the relevant Alternative Currency.

Each Request for Credit Extension (other than a Committed Loan Notice requesting only a conversion of Committed Loans to the other Type or a continuation of Eurocurrency Rate Loans) submitted by the Borrower shall be deemed to be a representation and warranty that the conditions specified in <u>Sections 4.02(a)</u> and <u>(b)</u> have been satisfied on and as of the date of the applicable Credit Extension.

ARTICLE V. REPRESENTATIONS AND WARRANTIES

To induce the Lenders, the L/C Issuers and the Administrative Agent to enter into this Agreement, the Borrower represents and warrants each of the following to the Lenders, the L/C Issuers and the Administrative Agent, on and as of the Closing Date and the making of Credit Extensions on the Closing Date and on and as of each date as required by Section 4.02 or on any other date required by any Loan Document (with references in this Article V (other than Sections 5.03, 5.04 and 5.05) to "Subsidiaries" to exclude Captive Insurance Subsidiaries):

5.01 Corporate Existence, Compliance with Law. Each of the Borrower and the Borrower's Subsidiaries (a) is duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization (other than, so long as it is an Immaterial Subsidiary, North County Recycling, Inc., a California corporation), (b) is duly qualified to do business as a foreign corporation and in good standing under the laws of each jurisdiction where such qualification is necessary, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect, (c) has all requisite corporate or other organizational power and authority and the legal right to own, pledge, mortgage and operate its properties, to lease the

property it operates under lease and to conduct its business as now or currently proposed to be conducted, (d) is in compliance with its Constituent Documents, (e) is in compliance with all applicable Requirements of Law except where the failure to be in compliance would not, in the aggregate, have a Material Adverse Effect and (f) has all necessary licenses, permits, consents or approvals from or by, has made all necessary filings with, and has given all necessary notices to, each Governmental Authority having jurisdiction, to the extent required for such ownership, operation and conduct, except for licenses, permits, consents, approvals, filings or notices that can be obtained or made by the taking of ministerial action to secure the grant or transfer thereof or the failure of which to obtain or make would not, in the aggregate, have a Material Adverse Effect.

5.02 Corporate Power; Authorization; Enforceable Obligations.

- (a) The execution, delivery and performance by each Loan Party of the Loan Documents to which it is a party and the consummation of the transactions contemplated thereby:
 - (i) are within such Loan Party's corporate, limited liability company, partnership or other organizational powers;
 - (ii) have been or, at the time of delivery thereof pursuant to <u>Article IV</u> will have been duly authorized by all necessary corporate, limited liability company or partnership action, including the consent of shareholders, partners and members where required;
 - (iii) do not and will not (A) contravene such Loan Party's or any of its Subsidiaries' respective Constituent Documents, (B) violate any other Requirement of Law applicable to such Loan Party (including Regulations T, U and X of the FRB), or any order or decree of any Governmental Authority or arbitrator applicable to such Loan Party, (C) conflict with or result in the breach of, or constitute a default under, or result in or permit the termination or acceleration of, any lawful Contractual Obligation of such Loan Party or any of its Subsidiaries, other than in the case of this clause (C) any such conflict, breach, default, termination or acceleration that could not reasonably be expected to have a Material Adverse Effect, or (D) result in the creation or imposition of any Lien upon any property of such Loan Party or any of its Subsidiaries, other than those in favor of the Secured Parties pursuant to the Security Instruments; and
 - (iv) do not require the consent of, authorization by, approval of, notice to, or filing or registration with, any Governmental Authority or any other Person, other than those listed on <u>Schedule 5.02</u> or that have been or will be, prior to the Closing Date, obtained or made, copies of which have been or will be delivered to the Administrative Agent pursuant to <u>Section 4.01</u>, and each of which on the Closing Date will be in full force and effect and, with respect to the Collateral, filings required to perfect the Liens created by the Security Instruments.
- (b) This Agreement has been, and each of the other Loan Documents will have been upon delivery thereof pursuant to the terms of this Agreement, duly executed and delivered by

each Loan Party who is a party thereto. This Agreement is, and the other Loan Documents will be, when delivered, the legal, valid and binding obligation of each Loan Party who is a party thereto, enforceable against such Loan Party in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principles of equity, regardless of whether considered in a proceeding in equity or at law.

5.03 Ownership of Borrower; Subsidiaries.

- (a) All of the outstanding capital stock of the Borrower is validly issued, fully paid and non-assessable.
- (b) Set forth on Schedule 5.03 is a complete and accurate list showing, as of the Closing Date, all Subsidiaries of the Borrower and, as to each such Subsidiary, the jurisdiction of its organization, the number of shares of each class of Stock authorized (if applicable), the number outstanding on the Closing Date, the number and percentage of the outstanding shares of each such class owned (directly or indirectly) by the Borrower. Except as set forth on Schedule 5.03, no Stock of any Subsidiary of the Borrower is subject to any outstanding option, warrant, right of conversion or purchase of any similar right. Except as set forth on Schedule 5.03, all of the outstanding Stock of each Subsidiary of the Borrower owned (directly or indirectly) by the Borrower has been validly issued, is fully paid and non-assessable (to the extent applicable) and is owned by the Borrower or a Subsidiary of the Borrower, free and clear of all Liens (other than the Lien in favor of the Secured Parties created pursuant to the Security Instruments), options, warrants, rights of conversion or purchase or any similar rights. Except as set forth on Schedule 5.03, neither the Borrower nor any such Subsidiary is a party to, or has knowledge of, any agreement restricting the transfer or hypothecation of any Stock of any such Subsidiary, other than the Loan Documents and, with respect to any Subsidiary that is not a Wholly-Owned Subsidiary, the Constituent Documents of such Subsidiary. The Borrower does not own or hold, directly or indirectly, any Stock of any Person other than such Subsidiaries and Investments permitted by Section 7.03.

5.04 Financial Statements.

- (a) The interim unaudited financial statements for the Borrower and its Subsidiaries for the most-recently ended Fiscal Quarter, copies of which have been furnished to each Lender, fairly present in all material respects, subject to the absence of footnote disclosure and normal recurring year-end audit adjustments, the consolidated financial condition of the Borrower and its Subsidiaries as at such dates and the consolidated results of the operations of the Borrower and its Subsidiaries for the period ended on such dates, all in conformity with GAAP, <u>provided</u> that this <u>Section 5.04(a)</u> shall not apply until the time for the delivery of the interim unaudited financial statements for the Fiscal Quarter ended March 31, 2010 as required hereunder.
- (b) The unaudited consolidated balance sheet of the Borrower and its Subsidiaries as of the end of the Fiscal Year ended December 31, 2009, and the related statements of income and cash flows of the Borrower and its Subsidiaries for such Fiscal Year, copies of which have been furnished to each Lender, (i) were prepared in conformity with GAAP and (ii) fairly present in all material respects, subject to the absence of footnote disclosure and normal recurring year-end

audit adjustments, the consolidated financial condition of the Borrower and its Subsidiaries as at the date indicated and the consolidated results of their operations and cash flow for the period indicated in conformity with GAAP applied on a basis consistent with prior years (except for changes with which the Borrower's Accountants shall concur and that shall have been disclosed in the notes to the financial statements).

- (c) Except as set forth on Schedule 5.04, neither the Borrower nor any of its Subsidiaries has, as of the Closing Date, any material obligation, contingent liability or liability for taxes, long-term leases (other than operating leases) or unusual forward or long-term commitment that is not reflected in the financial statements referred to in clause (b) above and not otherwise permitted by this Agreement.
- (d) The Projections have been prepared by the Borrower taking into consideration past operations of its business, and reflect projections for the period beginning approximately January 1, 2010 and ending approximately December 31, 2014 on a Fiscal Year by Fiscal Year basis. The Projections are based upon estimates and assumptions stated therein, all of which the Borrower believes, as of the Closing Date, to be reasonable in light of current conditions and current facts known to the Borrower (other than any necessary adjustments due to fees payable in accordance herewith) and, as of the Closing Date, reflect the Borrower's good faith estimates of the future financial performance of the Borrower and its Subsidiaries and of the other information projected therein for the periods set forth therein.
- **5.05 Material Adverse Change.** Since December 31, 2008, there has been no event or circumstance, either individually or in the aggregate, that has had or would reasonably be expected to result in a Material Adverse Effect.
- **5.06 Solvency.** Both before and after giving effect to (a) the Credit Extensions to be made or extended on the Closing Date or such other date as Credit Extensions requested hereunder are made or extended, (b) the disbursement of the proceeds of such Loans pursuant to the instructions of the Borrower, (c) the consummation of the transactions contemplated hereby (including, with respect to the making of this representation and warranty upon and at any time after its consummation, the Spinoff) and (d) the payment and accrual of all transaction costs in connection with the foregoing, the Loan Parties, taken as a whole, are Solvent.
- **5.07 Litigation.** Except as set forth on Schedule 5.07, there are no pending or, to the knowledge of the Borrower, threatened actions, investigations or proceedings against the Borrower or any of its Subsidiaries before any court, Governmental Authority or arbitrator other than those that, in the aggregate, would not reasonably be expected to have a Material Adverse Effect. Schedule 5.07 lists all litigation pending against any Loan Party as of the Closing Date that, if adversely determined, could be reasonably expected to have a Material Adverse Effect.
- 5.08 Taxes. All federal income and other material tax returns, reports and statements (collectively, the "Tax Returns") required to be filed by the Borrower or any of its Tax Affiliates have been filed with the appropriate Governmental Authorities in all jurisdictions in which such Tax Returns are required to be filed, all such Tax Returns are true and correct in all material respects, and all material taxes, charges and other impositions reflected therein or otherwise due and payable have been paid prior to the date on which any fine, penalty, interest, late charge or

loss may be added thereto for non-payment thereof except where contested in good faith and by appropriate proceedings if adequate reserves therefor have been established on the books of the Borrower or such Tax Affiliate in conformity with GAAP. The Borrower and each of its Tax Affiliates have withheld and timely paid to the respective Governmental Authorities all material amounts required to be withheld.

- **5.09 Full Disclosure.** The Information Memorandum and any other information prepared or furnished by or on behalf of any Loan Party and delivered to the Lenders in writing in connection with this Agreement or the consummation of the transactions contemplated hereunder or thereunder (in each case, taken as a whole) does not, as of the time of delivery of such information (with respect to the Information Memorandum, as of the Closing Date only), contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein or herein not misleading; provided that to the extent any such information was based upon, or constituted, a forecast or projection, such Loan Party represents only, in respect of such projection or forecast, that it acted in good faith and utilized reasonable assumptions and due care in the preparation of such information.
- **5.10 Margin Regulations.** The Borrower is not engaged in the business of extending credit for the purpose of purchasing or carrying margin stock (within the meaning of Regulation U of the FRB), and no proceeds of any Credit Extension will be used to purchase or carry any such margin stock or to extend credit to others for the purpose of purchasing or carrying any such margin stock in contravention of Regulation T, U or X of the FRB.

5.11 No Burdensome Restrictions; No Defaults.

- (a) Neither the Borrower nor any of its Subsidiaries (i) is a party to any Contractual Obligation (x) the compliance with which could reasonably be expected to have a Material Adverse Effect or (y) the performance of which by any thereof would result in the creation of a Lien (other than a Lien permitted under Section 7.02) on the property or assets of any thereof or (ii) is subject to any charter restriction that could reasonably be expected to have a Material Adverse Effect.
- (b) Neither the Borrower nor any of its Subsidiaries is in default under or with respect to any Contractual Obligation owed by it, other than, in either case, those defaults that would not reasonably be expected to have a Material Adverse Effect.
 - (c) No Default has occurred and is continuing.
- **5.12 Investment Company Act.** Neither the Borrower nor any of its Subsidiaries is an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended.
- **5.13 Use of Proceeds.** The (a) proceeds of the Loans are being used by the Borrower only (i) for working capital needs, capital expenditures, Permitted Acquisitions, general corporate purposes and other lawful corporate purposes of the Borrower and its Subsidiaries, (ii) to refinance obligations under the Existing Credit Agreements on the Closing Date, and (iii) to

pay fees and expenses in connection with this Agreement and the related transactions, and (b) Letters of Credit are being solely used by the Borrower to support warranties, bid bonds, payment or performance obligations and for other general corporate purposes by Permitted L/C Parties.

5.14 Insurance. All policies of insurance of any kind or nature currently maintained by the Borrower or any of its Subsidiaries, including policies of fire, theft, product liability, public liability, property damage, other casualty, employee fidelity, workers' compensation and employee health and welfare insurance, are in full force and effect and are of a nature and provide such coverage as is sufficient and as is customarily carried by businesses of the size and character of such Person.

5.15 Labor Matters.

- (a) There are no strikes, work stoppages, slowdowns or lockouts pending or, to the Borrower's knowledge, threatened against or involving the Borrower, any of its Subsidiaries or any Guarantor, other than those that, in the aggregate, would not reasonably be expected to have a Material Adverse Effect.
- (b) There are no unfair labor practices, grievances or complaints pending, or, to the Borrower's knowledge, threatened, against or involving the Borrower, any of its Subsidiaries or any Guarantor, nor, to the Borrower's knowledge, are there any unfair labor practices, arbitrations or grievances threatened involving the Borrower, any of its Subsidiaries or any Guarantor, other than those that if resolved adversely to the Borrower, such Subsidiary or such Guarantor, as applicable, would not reasonably be expected to have a Material Adverse Effect.
- (c) Except as set forth on Schedule 5.15, as of the Closing Date, there is no collective bargaining agreement covering any employee of the Borrower or its Subsidiaries. With respect to employees of the Borrower or any of its Subsidiaries not already covered by a collective bargaining agreement set forth on Schedule 5.15, as of the Closing Date no union representation question exists with respect to such employees and, to Borrower's knowledge, no union organization activity is taking place as of the Closing Date.

5.16 ERISA.

- (a) Except as set forth on Schedule 5.16, each Employee Benefit Plan that is intended to qualify under Section 401 of the Code has received a favorable determination letter from the IRS indicating that such Employee Benefit Plan is so qualified and nothing has occurred subsequent to the issuance of such determination letter which would cause such Employee Benefit Plan to lose its qualified status. Any trust created under any Employee Benefit Plan is exempt from tax under the provisions of Section 501 of the Code, except where such failures could not reasonably be expected to have a Material Adverse Effect.
- (b) The Borrower, each of its Subsidiaries, each Guarantor and each of their respective ERISA Affiliates is in material compliance with all applicable provisions and requirements of ERISA, the Code and applicable Employee Benefit Plan provisions with respect

to each Employee Benefit Plan except for non-compliances that would not reasonably be expected to have a Material Adverse Effect.

- (c) With respect to each Title IV Plan and each Multiemployer Plan, the Borrower, each of its Subsidiaries, each Guarantor and each of their respective ERISA Affiliates has made all contributions required under ERISA and the Code and are in material compliance with the minimum funding standard of Section 412 of the Code (in each case, whether or not waived in accordance with Section 412(c) of the Code).
- (d) There has been no, nor is there reasonably expected to occur, any ERISA Event other than those that would not reasonably be expected to have a Material Adverse Effect.
- (e) Except (i) to the extent required under Section 4980B of the Code or similar state laws, and (ii) with respect to which the aggregate liability, calculated on a FAS 106 basis as of December 31, 2009, does not exceed \$150,000,000, no Employee Benefit Plan provides health or welfare benefits (through the purchase of insurance or otherwise) to any retired or former employees, consultants or directors (or their dependents) of the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates. None of the Borrower, its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates has incurred or reasonably expects to incur any withdrawal liability with respect to any Multiemployer Plan. The Borrower, each of its Subsidiaries, each Guarantor and each of their ERISA Affiliates has complied with the requirements of Section 515 of ERISA with respect to each Multiemployer Plan and are not in material "default" (as defined in Section 4219(c)(5) of ERISA) with respect to payments to a Multiemployer Plan.

5.17 Environmental Matters.

- (a) Except as disclosed on <u>Schedule 5.17</u>, the operations of the Borrower and each of its Subsidiaries have been and are in compliance with all Environmental Laws, including obtaining and complying with all required environmental, health and safety Permits, other than non-compliances that, in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.
- (b) Except as disclosed on <u>Schedule 5.17</u>, none of the Borrower or any of its Subsidiaries or any Real Property currently or, to the knowledge of the Borrower, previously owned, operated or leased by or for the Borrower or any of its Subsidiaries is subject to any pending or, to the knowledge of the Borrower, threatened, claim, order, agreement, notice of violation, notice of potential liability or is the subject of any pending or threatened proceeding or governmental investigation under or pursuant to Environmental Laws other than those orders, agreements, notices, proceedings or investigations that, in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.
- (c) Except as disclosed on Schedule 5.17, to the knowledge of the Borrower, there are no facts, circumstances or conditions arising out of or relating to the operations or ownership of the Borrower or of Real Property owned, operated or leased by the Borrower or any of its Subsidiaries that are not specifically included in the financial information furnished to the

Lenders other than those that, in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.

5.18 Intellectual Property. Except where the failure to do so would not, taken as a whole, reasonably be expected to have a Material Adverse Effect, the Borrower and its Subsidiaries own or license or otherwise have the right to use all licenses, permits, patents, patent applications, trademarks, trademar

5.19 Title; Real Property.

- (a) Each of the Borrower and its Subsidiaries has valid and indefeasible title to, or valid leasehold interests in, all of its material properties and assets (including Real Property) and good title to, or valid leasehold interests in, all personal property, in each case that is purported to be owned or leased by it, including those reflected on the most recent financial statements delivered by the Borrower hereunder, and none of such properties and assets is subject to any Lien, except Liens permitted under Section 7.02. The Borrower and its Subsidiaries have received all deeds, assignments, waivers, consents, non-disturbance and recognition or similar agreements, bills of sale and other documents, and have duly effected all recordings, filings and other actions necessary to establish, protect and perfect the Borrower's and its Subsidiaries' right, title and interest in and to all such property, other than those that would not reasonably be expected to result in a Material Adverse Effect.
- (b) Set forth on Schedule 5.19(b) is a complete and accurate list, as of the Closing Date, of all (i) owned Real Property located in the United States with a reasonably estimated Fair Market Value in excess of \$3,000,000 showing, as of the Closing Date, the street address, county (or other relevant jurisdiction or state) and the record owner thereof and (ii) leased Real Property located in the United States with annual lease payments in excess of \$1,000,000 showing, as of the Closing Date, the street address and county (or other relevant jurisdiction or state) thereof.
- (c) No portion of any Real Property has suffered any material damage by fire or other casualty loss that has not heretofore been completely repaired and restored to its original condition other than those that would not reasonably be expected to have a Material Adverse Effect. As of the Closing Date, no portion of any Mortgaged Property is located in a special flood hazard area as designated by any federal Governmental Authority other than those for which flood insurance has been provided in accordance with Section 4.01(a)(iv).
- (d) Except as would not reasonably be expected to have a Material Adverse Effect, (a) each Loan Party has obtained and holds all Permits required in respect of all Real Property

and for any other property otherwise operated by or on behalf of, or for the benefit of, such person and for the operation of each of its businesses as presently conducted and as proposed to be conducted, (b) all such Permits are in full force and effect, and each Loan Party has performed and observed all requirements of such Permits, (c) no event has occurred that allows or results in, or after notice or lapse of time would allow or result in, revocation or termination by the issuer thereof or in any other impairment of the rights of the holder of any such Permit, (d) no such Permits contain any restrictions, either individually or in the aggregate, that are materially burdensome to any Loan Party, or to the operation of any of its businesses or any property owned, leased or otherwise operated by such person, (e) each Loan Party reasonably believes that each of its Permits will be timely renewed and complied with, without material expense, and that any additional Permits that may be required of such Person will be timely obtained and complied with, without material expense and (f) the Borrower has no knowledge or reason to believe that any Governmental Authority is considering limiting, suspending, revoking or renewing on materially burdensome terms any such Permit.

- (e) None of the Borrower or any of its Subsidiaries has received any notice, or has any knowledge, of any pending, threatened or contemplated condemnation proceeding affecting any Real Property or any part thereof, except those that would not reasonably be expected to have a Material Adverse Effect.
- (f) Each of the Loan Parties, and, to the knowledge of the Borrower, each other party thereto, has complied with all obligations under all leases of Real Property to which it is a party other than those the failure with which to comply would not reasonably be expected to have a Material Adverse Effect and all such leases are legal, valid, binding and in full force and effect and are enforceable in accordance with their terms other than those the failure of which to so comply with the foregoing would not reasonably be expected to have a Material Adverse Effect. No landlord Lien has been filed, and, to the knowledge of the Borrower, no claim is being asserted, with respect to any lease payment under any lease of Real Property other than those that would not reasonably be expected to have a Material Adverse Effect.
- (g) There are no pending or, to the knowledge of the Borrower, proposed special or other assessments for public improvements or otherwise affecting any material portion of the owned Real Property, nor are there any contemplated improvements to such owned Real Property that may result in such special or other assessments, other than those that would not reasonably be expected to have a Material Adverse Effect.
- **5.20 Security Instruments.** The provisions of the Security Instruments are effective to create in favor of the Administrative Agent for the benefit of the Secured Parties a legal, valid and enforceable first priority Lien (subject to Liens permitted by Section 7.02) on all right, title and interest of the respective Loan Parties (other than the BWXT Entities that are Loan Parties) in the Collateral described therein. Except for filings completed on or prior to the Closing Date and filings and other actions contemplated hereby and by the Security Instruments, no filing or other action in the United States will be necessary to perfect or protect such Liens.

ARTICLE VI. AFFIRMATIVE COVENANTS

The Borrower agrees with the Lenders, L/C Issuers and the Administrative Agent to each of the following, as long as any Obligation or any Commitment remains outstanding and, in each case, unless the Required Lenders otherwise consent in writing (<u>provided</u> that those provisions under this <u>Article VI</u> with which Subsidiaries of the Borrower are required to comply shall exclude from such compliance any Captive Insurance Subsidiary):

- **6.01 Financial Statements.** The Borrower shall furnish to the Administrative Agent each of the following:
- (a) Quarterly Reports. Within 45 days after the end of each of the first three Fiscal Quarters of each Fiscal Year (unless such period is extended pursuant to SEC guidelines), consolidated unaudited balance sheets as of the close of such quarter and the related statements of income and cash flow for such quarter and that portion of the Fiscal Year ending as of the close of such quarter, setting forth in comparative form the figures for the corresponding period in the prior year, in each case certified by a Responsible Officer of the Borrower as fairly presenting in all material respects the consolidated financial condition of the Borrower and its Subsidiaries as at the dates indicated and the results of their operations and cash flow for the periods indicated in accordance with GAAP (subject to the absence of footnote disclosure and normal year-end audit adjustments).
- (b) Annual Reports. Within 90 days after the end of each Fiscal Year (unless such period is extended pursuant to SEC guidelines), consolidated balance sheets of the Borrower and its Subsidiaries as of the end of such Fiscal Year and related statements of income and cash flows of the Borrower and its Subsidiaries for such Fiscal Year, all prepared in conformity with GAAP and certified, in the case of such consolidated financial statements, without qualification as to the scope of the audit or as to the Borrower being a going concern by the Borrower's Accountants, together with the report of such accounting firm stating that (i) such financial statements fairly present in all material respects the consolidated financial condition of the Borrower and its Subsidiaries as at the dates indicated and the results of their operations and cash flow for the periods indicated in conformity with GAAP applied on a basis consistent with prior years (except for changes with which the Borrower's Accountants shall concur and that shall have been disclosed in the notes to the financial statements) and (ii) the examination by the Borrower's Accountants in connection with such consolidated financial statements has been made in accordance with generally accepted auditing standards.
- (c) <u>Compliance Certificate</u>. Together with each delivery of any financial statement pursuant to clause (a) or (b) above, a Compliance Certificate (i) showing in reasonable detail the calculations used in determining the Leverage Ratio and demonstrating compliance with each of the other financial covenants contained in <u>Section 7.18</u>, and (ii) stating that no Default has occurred and is continuing or, if a Default has occurred and is continuing, stating the nature thereof and the action which the Borrower has taken or proposes to take with respect thereto.

The Borrower hereby acknowledges that (a) the Administrative Agent and/or one or more of the Arrangers will make available to the Lenders and the L/C Issuers materials and/or

information provided by or on behalf of the Borrower hereunder (collectively, "Borrower Materials") by posting the Borrower Materials on IntraLinks or another similar electronic system (the "Platform") and (b) certain of the Lenders (each, a "Public Lender") may have personnel who do not wish to receive material non-public information with respect to the Borrower or its Affiliates, or the respective securities of any of the foregoing, and who may be engaged in investment and other market-related activities with respect to such Persons' securities. The Borrower hereby agrees that (w) all Borrower Materials that the Borrower intends to be made available to Public Lenders shall be clearly and conspicuously marked "PUBLIC" which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the first page thereof; (x) by marking Borrower Materials "PUBLIC," the Borrower shall be deemed to have authorized the Administrative Agent, each Arranger, each L/C Issuer and the Lenders to treat the Borrower Materials as not containing any material non-public information with respect to the Borrower or its securities for purposes of United States Federal and state securities laws (provided, however, that to the extent such Borrower Materials constitute Information, they shall be treated as set forth in Section 10.07); (y) all Borrower Materials marked "PUBLIC" are permitted to be made available through a portion of the Platform designated "Public Side Information;" and (z) the Administrative Agent and the Arrangers shall be entitled to treat any Borrower Materials that are not marked "PUBLIC" as being suitable only for posting on a portion of the Platform not designated "Public Side Information."

6.02 Collateral Reporting Requirements. The Borrower shall furnish to the Administrative Agent each of the following:

- (a) <u>Updated Corporate Chart</u>. If requested by the Administrative Agent, together with each delivery of any financial statement pursuant to <u>Section 6.01(b)</u>, a corporate organizational chart or other equivalent list, current as of the date of delivery, in form and substance reasonably acceptable to the Administrative Agent and certified as true, correct and complete by a Responsible Officer of the Borrower, setting forth, for each of the Loan Parties, all Persons subject to <u>Section 6.22</u> or <u>Section 6.25</u>, all Subsidiaries of any of them and any joint venture (including Joint Ventures) entered into by any of the foregoing, (i) its full legal name, (ii) its jurisdiction of organization and organizational number (if any) and (iii) the number of shares of each class of its Stock authorized (if applicable), the number outstanding as of the date of delivery, and the number and percentage of the outstanding shares of each such class owned (directly or indirectly) by the Borrower.
- (b) Additional Information. From time to time, statements and schedules further identifying and describing the Collateral and such other reports in connection with the Collateral, all as the Administrative Agent may reasonably request, and in reasonable detail.
- (c) <u>Additional Filings.</u> At any time and from time to time, upon the reasonable written request of the Administrative Agent, and at the sole expense of the Loan Parties, duly executed, delivered and recorded instruments and documents for the purpose of obtaining or preserving the full benefits of this Agreement, each Security Instrument and each other Loan Document and of the rights and powers herein and therein granted (and each Loan Party shall take such further action as the Administrative Agent may reasonably request for such purpose, including the filing of any financing or continuation statement under the UCC or other similar Requirement of Law in effect in any domestic jurisdiction with respect to the security interest

created by the Collateral Agreement but excluding (i) the execution and delivery of any control agreements with respect to deposit accounts or securities accounts (except with respect to deposit accounts holding Cash Collateral provided hereunder), (ii) any filings to perfect Liens on intellectual property, other than any such filings under the UCC or with the U.S. Patent and Trademark Office or U.S. Copyright Office and (iii) any filings or actions in any jurisdiction outside the United States.

The reporting requirements set forth in this <u>Section 6.02</u> are in addition to, and shall not modify and are not in replacement of, any rights and other obligation set forth in any Loan Document (including notice and reporting requirements) and satisfaction of the reporting obligations in this <u>Section 6.02</u> shall not, by itself, operate as an update of any Schedule or any schedule of any other Loan Document and shall not cure, or otherwise affect in any way, any Default, including any failure of any representation or warranty of any Loan Document to be correct in any respect when made.

- **6.03 Default and certain other Notices.** Promptly and in any event within five Business Days after a Responsible Officer of the Borrower obtains actual knowledge thereof, the Borrower shall give the Administrative Agent notice:
 - (a) of the occurrence of any Default or Event of Default;
- (b) of any announcement by Moody's or S&P of any change in a Debt Rating, corporate rating or corporate family rating that has not been publicly announced or is not otherwise publicly available; and
 - (c) of the issuance of a notice of proposed debarment or notice of proposed suspension by a Governmental Authority or Governmental Authorities.

Each notice pursuant to this <u>Section 6.03</u> (other than <u>Section 6.03(b)</u>) shall be accompanied by a statement of a Responsible Officer of the Borrower setting forth details of the occurrence referred to therein, the anticipated effect thereof, and stating what action the Borrower has taken and proposes to take with respect thereto. Each notice pursuant to <u>Section 6.03(a)</u> shall describe with particularity any and all provisions of this Agreement and any other Loan Document that have been breached. Any notice pursuant to this <u>Section 6.03</u>, if given by telephone, shall be promptly confirmed in writing on the next Business Day.

6.04 Litigation. Promptly after a Responsible Officer of the Borrower obtains actual knowledge of the commencement thereof, the Borrower shall give the Administrative Agent written notice of the commencement of all actions, suits and proceedings before any domestic or foreign Governmental Authority or arbitrator, regarding the Borrower, any of its Subsidiaries or any Joint Venture that (i) seeks injunctive or similar relief that, in the reasonable judgment of the Borrower, if adversely determined, would reasonably be expected to result in a Material Adverse Effect or (ii) in the reasonable judgment of the Borrower would expose the Borrower, such Subsidiary or such Joint Venture to liability in an amount aggregating \$20,000,000 or more or that, if adversely determined, would reasonably be expected to have a Material Adverse Effect.

- **6.05 Labor Relations.** Promptly after a Responsible Officer of the Borrower has actual knowledge of the same, the Borrower shall give the Administrative Agent written notice of (a) any material labor dispute to which the Borrower, any of its Subsidiaries, any Guarantors or any Joint Venture is a party, including any strikes, lockouts or other material disputes relating to any of such Person's plants and other facilities, provided that such dispute, strike or lockout involves a work stoppage exceeding 30 days, (b) any material Worker Adjustment and Retraining Notification Act or related liability incurred with respect to the closing of any plant or other facility of any such Person affecting 300 or more employees of the Borrower and its Subsidiaries and (c) any union organization activity with respect to employees of the Borrower or any of its Subsidiaries not covered by a collective bargaining agreement as of the Closing Date.
- 6.06 Tax Returns. Upon the reasonable request of any Lender, through the Administrative Agent, the Borrower shall provide copies of all federal, state, local and foreign tax returns and reports filed by the Borrower, any of its Subsidiaries or any Joint Venture in respect of taxes measured by income (excluding sales, use and like taxes).
- **6.07 Insurance.** As soon as is practicable and in any event within 90 days after the end of each Fiscal Year, the Borrower shall furnish the Administrative Agent with a report on the standard "Acord" form outlining all material insurance coverage maintained as of the date of such report by the Borrower, its Subsidiaries and Joint Ventures and the duration of such coverage.
 - 6.08 ERISA Matters. The Borrower shall furnish the Administrative Agent each of the following:
- (a) promptly and in any event within 30 days after a Responsible Officer of the Borrower knows, or has reason to know, that any ERISA Event has occurred that, alone or together with any other ERISA Event, would reasonably be expected to result in liability of the Borrower, any Subsidiary, any Guarantor and/or any ERISA Affiliate in an aggregate amount exceeding \$15,000,000, written notice describing the nature thereof, what action the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates has taken, is taking or proposes to take with respect thereto and, when known by such Responsible Officer, any action taken or threatened by the IRS, the Department of Labor or the PBGC with respect to such event;
- (b) promptly and in any event within 10 days after a Responsible Officer of the Borrower knows, or has reason to know, that a request for a minimum funding waiver under Section 412 of the Code has been filed with respect to any Title IV Plan, a written statement of a Responsible Officer of the Borrower describing such waiver request and the action, if any, the Borrower, its Subsidiaries and ERISA Affiliates propose to take with respect thereto and a copy of any notice filed with the PBGC or the IRS pertaining thereto;
- (c) simultaneously with the date that the Borrower, any of its Subsidiaries or any ERISA Affiliate files with the PBGC a notice of intent to terminate any Title IV Plan, if, at the time of such filing, such termination would require material additional contributions in order to

be considered a standard termination within the meaning of Section 4041(b) of ERISA, a copy of each notice; and

- (d) promptly, copies of (i) each Schedule B (Actuarial Information) to the annual report (Form 5500 Series) filed by the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates with the IRS with respect to each Title IV Plan; (ii) all notices received by the Borrower, any of its Subsidiaries, any Guarantor or any of their respective ERISA Affiliates from a Multiemployer Plan sponsor concerning an ERISA Event that would reasonably be expected to result in liability of the Borrower, any Subsidiary, any Guarantor and/or any ERISA Affiliate in an aggregate amount exceeding \$7,500,000; and (iii) copies of such other documents or governmental reports or filings relating to any Employee Benefit Plan as the Administrative Agent shall reasonably request.
- **6.09 Environmental Matters.** The Borrower shall provide the Administrative Agent promptly, and in any event within 10 Business Days after any Responsible Officer of the Borrower obtains actual knowledge of any of the following, written notice of each of the following:
- (a) that any Loan Party is or may be liable to any Person as a result of a Release or threatened Release that would reasonably be expected to subject such Loan Party to Environmental Liabilities and Costs of \$10,000,000 or more;
- (b) the receipt by any Loan Party of notification that any material real or personal property of such Loan Party is or is reasonably likely to be subject to any Environmental Lien;
- (c) the receipt by any Loan Party of any notice of violation of or potential liability under, or knowledge by a Responsible Officer of the Borrower that there exists a condition that would reasonably be expected to result in a violation of or liability under, any Environmental Law, except for violations and liabilities the consequence of which, in the aggregate, would not be reasonably likely to subject the Loan Parties collectively to Environmental Liabilities and Costs of \$10,000,000 or more; and
- (d) promptly following reasonable written request by any Lender, through the Administrative Agent, a report providing an update of the status of any environmental, health or safety compliance, hazard or liability issue identified in any notice or report delivered pursuant to this <u>Section 6.09</u>.
- 6.10 Patriot Act Information. Each Lender and the Administrative Agent (for itself and not on behalf of any Lender) hereby notifies the Borrower that pursuant to the requirements of the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Patriot Act"), it is required to obtain, verify and record information that identifies the Borrower and each other Loan Party, which information includes the name and address of the Borrower and each other Loan Party and other information that will allow such Lender or the Administrative Agent, as applicable, to identify the Borrower and each other Loan Party in accordance with the Patriot Act. The Borrower shall promptly, following a request by the Administrative Agent or any Lender, provide all documentation and other information that the Administrative Agent or such Lender reasonably requests in order to comply with its ongoing

obligations under applicable "know your customer" and anti-money laundering rules and regulations, including, without limitation, the Patriot Act.

- **6.11 Other Information.** The Borrower shall provide the Administrative Agent or any Lender with such other information respecting the business, properties, condition, financial or otherwise, or operations of the Borrower, any of its Subsidiaries or any Joint Venture as the Administrative Agent or such Lender, through the Administrative Agent, may from time to time reasonably request. The Administrative Agent shall provide copies of any written information provided to it pursuant to Sections 6.01 through 6.10 above to any Lender requesting the same.
- 6.12 Preservation of Corporate Existence, Etc. The Borrower shall, and shall cause each of its Subsidiaries to, preserve and maintain its legal existence, rights (charter and statutory) and franchises, except as permitted by Sections 7.03, 7.04 and 7.06 and except if, in the reasonable business judgment of the Borrower, it is in the business interest of the Borrower or such Subsidiary not to preserve and maintain such rights (charter and statutory) and franchises, and such failure to preserve the same would not reasonably be expected to have a Material Adverse Effect and would not reasonably be expected to materially affect the interests of the Secured Parties under the Loan Documents or the rights and interests of any of them in the Collateral.
- **6.13 Compliance with Laws, Etc.** The Borrower shall, and shall cause each of its Subsidiaries to, comply with all applicable Requirements of Law, Contractual Obligations and Permits, except where the failure so to comply would not reasonably be expected to have a Material Adverse Effect.
- **6.14 Conduct of Business.** The Borrower shall, and shall cause each of its Subsidiaries to, (a) conduct its business in the ordinary course (except for non-material changes in the nature or conduct of its business as carried on as of the Closing Date) and (b) use its reasonable efforts, in the ordinary course, to preserve its business and the goodwill and business of the customers, suppliers and others having business relations with the Borrower or any of its Subsidiaries, except where the failure to comply with the covenants in each of clauses (a) and (b) above would not reasonably be expected to have a Material Adverse Effect.
- **6.15 Payment of Taxes, Etc.** The Borrower shall, and shall cause each of its Subsidiaries to, pay and discharge before the same shall become delinquent, all lawful governmental claims, taxes, assessments, charges and levies, except where (a) contested in good faith, by proper proceedings and adequate reserves therefor have been established on the books of the Borrower or the appropriate Subsidiary in conformity with GAAP or (b) the failure to so pay and discharge would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.
- **6.16 Maintenance of Insurance.** The Borrower shall, and shall cause each of its Subsidiaries to, (a) maintain insurance with responsible and reputable insurance companies or associations in such amounts and covering such risks as, in the reasonable determination of the Borrower, is usually carried by companies engaged in similar businesses and owning similar properties in the same general areas in which the Borrower or such Subsidiary operates and (b) cause all property and general liability insurance to name the Administrative Agent on behalf

of the Secured Parties as additional insured (with respect to liability and property policies), loss payee (with respect to property policies) or lender's loss payee (with respect to property policies), as appropriate, and to provide that no cancellation, material addition in amount or material change in coverage shall be effective until after 30 days' written notice thereof to the Administrative Agent.

- 6.17 Access. The Borrower shall from time to time during normal business hours, and subject to national security and defense requirements of any Governmental Authority, permit the Administrative Agent, the L/C Issuers and the Lenders, or any agents or representatives thereof, within five Business Days after written notification of the same (except that during the continuance of an Event of Default, no such notice shall be required) to (a) examine and make copies of and abstracts from the records and books of account of the Borrower and each of its Subsidiaries, (b) visit the properties of the Borrower and each of its Subsidiaries with any of their respective officers or directors; provided that the Borrower will not be required to permit any examination or visit as set forth in clauses (a) and (b) above with respect to each of the Administrative Agent, the L/C Issuers and the Lenders (or any agents or representatives thereof) (i) within the twelve-month period following the date of the most recent examination or visit by any L/C Issuer, any Lender or the Administrative Agent (or any agents or representatives thereof), as applicable, unless an Event of Default has occurred and is continuing and (ii) unless such visit is coordinated through the Administrative Agent.
- **6.18 Keeping of Books.** The Borrower shall, and shall cause each of its Subsidiaries to keep, proper books of record and account, in which full and correct entries shall be made in conformity with GAAP of the financial transactions and assets and business of the Borrower and each such Subsidiary.
- **6.19 Maintenance of Properties, Etc.** The Borrower shall, and shall cause each of its Subsidiaries to, maintain and preserve (a) in good working order and condition (ordinary wear and tear excepted) all of its properties necessary in the conduct of its business, (b) all rights, permits, licenses, approvals and privileges (including all Permits) necessary in the conduct of its business and (c) all Material Intellectual Property, except where failure to so maintain and preserve the items set forth in clauses (a), (b) and (c) above would not reasonably be expected to have a Material Adverse Effect.
 - 6.20 Application of Proceeds. The Borrower shall use the entire amount of the proceeds of the Loans as provided in Section 5.13.

6.21 Environmental.

- (a) The Borrower shall, and shall cause each of its Subsidiaries to, exercise reasonable due diligence in order to comply in all material respects with all Environmental Laws.
- (b) The Borrower agrees that the Administrative Agent may, from time to time, retain, at the expense of the Borrower, an independent professional consultant reasonably acceptable to the Borrower to review any report relating to Contaminants prepared by or for the Borrower and to conduct its own investigation (the scope of which investigation shall be

reasonable based upon the circumstances) of any property currently owned, leased, operated or used by the Borrower or any of its Subsidiaries, if (x) a Default or an Event of Default shall have occurred and be continuing, or (y) the Administrative Agent reasonably believes (1) that an occurrence relating to such property is likely to give rise to any Environmental Liabilities and Costs or (2) that a violation of an Environmental Law on or around such property has occurred or is likely to occur, which could, in either such case, reasonably be expected to result in Environmental Liabilities and Costs in excess of \$10,000,000, provided that, unless an Event of Default shall have occurred and be continuing, such consultant shall not drill on any property of the Borrower or any of its Subsidiaries without the Borrower's prior written consent. Borrower shall use its reasonable efforts to obtain for the Administrative Agent and its agents, employees, consultants and contractors the right, upon reasonable notice to Borrower, to enter into or on to the facilities currently owned, leased, operated or used by Borrower or any of its Subsidiaries to perform such tests on such property as are reasonably necessary to conduct such a review and/or investigation. Any such investigation of any property shall be conducted, unless otherwise agreed to by Borrower and the Administrative Agent, during normal business hours and shall be conducted so as not to unreasonably interfere with the ongoing operations at any such property or to cause any damage or loss at such property. Borrower and the Administrative Agent hereby acknowledge and agree that any report of any investigation conducted at the request of the Administrative Agent pursuant to this subsection will be obtained and shall be used by the Administrative Agent and the Lenders for the purposes of the Lenders' internal credit decisions, to monitor the Obligations and to protect the Liens created by the Loan Documents, and the Administrative Agent and the Lenders hereby acknowledge and agree any such report will be kept confidential by them to the extent permitted by law except as provided in the following sentence. The Administrative Agent agrees to deliver a copy of any such report to Borrower with the understanding that Borrower acknowledges and agrees that (i) it will indemnify and hold harmless the Administrative Agent and each Lender from any costs, losses or liabilities relating to Borrower's use of or reliance on such report, (ii) neither Administrative Agent nor any Lender makes any representation or warranty with respect to such report, and (iii) by delivering such report to Borrower, neither the Administrative Agent nor any Lender is requiring or recommending the implementation of any suggestions or recommendations contained in such

(c) Promptly after a Responsible Officer of the Borrower obtains actual knowledge thereof, the Borrower shall advise the Administrative Agent in writing and in reasonable detail of (i) any Release or threatened Release of any Contaminants required to be reported by Borrower or its Subsidiaries, to any Governmental Authorities under any applicable Environmental Laws and which would reasonably be expected to have Environmental Liabilities and Costs in excess of \$10,000,000, (ii) any and all written communications with respect to any pending or threatened claims under Environmental Law in each such case which, individually or in the aggregate, have a reasonable possibility of giving rise to Environmental Liabilities and Costs in excess of \$10,000,000, (iii) any Remedial Action performed by Borrower or any other Person in response to (x) any Contaminants on, under or about any property, the existence of which has a reasonable possibility of resulting in Environmental Liabilities and Costs in excess of \$10,000,000, or (y) any other Environmental Liabilities and Costs in excess of \$10,000,000, (iv) discovery by Borrower or its Subsidiaries of any occurrence or condition on any material property that could

cause Borrower's or its Subsidiaries' interest in any such property to be subject to any material restrictions on the ownership, occupancy, transferability or use thereof under any applicable Environmental Laws or Environmental Liens, and (v) any written request for information from any Governmental Authority that fairly suggests such Governmental Authority is investigating whether Borrower or any of its Subsidiaries may be potentially responsible for a Release or threatened Release of Contaminants which has a reasonable possibility of giving rise to Environmental Liabilities and Costs in excess of \$10,000,000.

- (d) Borrower shall promptly notify the Administrative Agent of (i) any proposed acquisition of Stock, assets, or property by Borrower or any of its Subsidiaries that would reasonably be expected to expose Borrower or any of its Subsidiaries to, or result in Environmental Liabilities and Costs in excess of \$10,000,000 and (ii) any proposed action to be taken by Borrower or any of its Subsidiaries to commence manufacturing, industrial or other similar operations that would reasonably be expected to subject Borrower or any of its Subsidiaries to additional Environmental Laws, that are materially different from the Environmental Laws applicable to the operations of Borrower or any of its Subsidiaries as of the Closing Date.
- (e) Borrower shall, at its own expense, provide copies of such documents or information as the Administrative Agent may reasonably request in relation to any matters disclosed pursuant to this subsection.
- (f) To the extent required by Environmental Laws or Governmental Authorities under applicable Environmental Laws, Borrower shall promptly take, and shall cause each of its Subsidiaries promptly to take, any and all necessary Remedial Action in connection with the presence, handling, storage, use, disposal, transportation or Release or threatened Release of any Contaminants on, under or affecting any property in order to comply in all material respects with all applicable Environmental Laws and Permits. In the event Borrower or any of its Subsidiaries undertakes any Remedial Action with respect to the presence, Release or threatened Release of any Contaminants on or affecting any property, Borrower or any of its Subsidiaries shall conduct and complete such Remedial Action in material compliance with all applicable Environmental Laws, and in material accordance with the applicable policies, orders and directives of all relevant Governmental Authorities except when, and only to the extent that, Borrower or any such Subsidiaries' liability for such presence, handling, storage, use, disposal, transportation or Release or threatened Release of any Contaminants is being contested in good faith by Borrower or any of such Subsidiaries. In the event Borrower fails to take required actions to address such Release or threatened Release of Contaminants or to address a violation of or liability under Environmental Law, the Administrative Agent may, upon providing the Borrower with 5 Business Days' prior written notice, enter the property and, at Borrower's sole expense, perform whatever action the Administrative Agent reasonably deems prudent to rectify the situation.
- **6.22 Additional Collateral and Guaranties.** Notify the Administrative Agent promptly after any Person (i) becomes a Wholly-Owned Domestic Subsidiary (including in connection with the Spinoff) that is not an Immaterial Subsidiary (including a Wholly-Owned Domestic Subsidiary that ceases for any reason to satisfy the definition of "Immaterial Subsidiary" at any time), (ii) becomes a First-Tier Foreign Subsidiary, or (iii) is required to become a Guarantor and/or grant Collateral in compliance with Section 6.24 or 6.25, and

promptly thereafter (and in any event within 30 days, or such longer period of time permitted by the Administrative Agent in its sole discretion):

- (a) if such Person is a Wholly-Owned Domestic Subsidiary and is not a Captive Insurance Subsidiary:
- (i) cause such Wholly-Owned Domestic Subsidiary to become a Guarantor by executing and delivering to the Administrative Agent a Joinder Agreement or such other document as the Administrative Agent shall deem reasonably appropriate for such purpose; and
- (ii) without duplication of clause (b)(iii) below, cause such Person to deliver to the Administrative Agent documents of the types referred to in clauses (v), (vi) and (viii) of Section 4.01(a) and favorable opinions of counsel to such Person (which shall cover, among other things, the legality, validity, binding effect and enforceability of the documentation referred to in clause (a)(i)), all in form, content and scope reasonably satisfactory to the Administrative Agent;
- (b) except during a Collateral Release Period, if such Person is a Wholly-Owned Domestic Subsidiary other than a BWXT Entity or a Captive Insurance Subsidiary:
 - (i) cause such Person to deliver to the Administrative Agent for the benefit of the Secured Parties, Security Instruments (or supplements thereto), as specified by and in form and substance reasonably satisfactory to the Administrative Agent (including delivery of all certificated Pledged Interests in and of such Subsidiary, and other instruments of the type specified in Section 4.01(a)(iii) and (iv)), securing payment of all the Obligations and constituting Liens on all such real and personal properties,
 - (ii) take whatever action (including the filing of Uniform Commercial Code financing statements and the giving of notices) as may be necessary or advisable in the reasonable opinion of the Administrative Agent to vest in the Administrative Agent (or in any representative of the Administrative Agent designated by it) valid and subsisting Liens on the properties purported to be subject to the Security Instruments (or supplements thereto) delivered pursuant to this Section 6.22, enforceable against all third parties in accordance with their terms (subject to Liens permitted by the Loan Documents), provided that no such actions shall be required in any jurisdiction outside the United States; and
 - (iii) without duplication of clause (a)(ii) above, cause such Person to deliver to the Administrative Agent documents of the types referred to in clauses (v), (vi) and (viii) of Section 4.01(a) and, at the request of the Administrative Agent, favorable opinions of counsel to such Person (which shall cover, among other things, the legality, validity, binding effect and enforceability of the documentation referred to in clause (b)(i)), all in form, content and scope reasonably satisfactory to the Administrative Agent; and
- (c) except during a Collateral Release Period, if such Person is a First-Tier Foreign Subsidiary any of whose Stock is owned by a Loan Party (or a Person becoming a Loan Party

pursuant to this Section), cause such Loan Party to deliver to the Administrative Agent for the benefit of the Secured Parties all certificated Pledged Interests in and of such First-Tier Foreign Subsidiary, and any Security Instruments (or supplements thereto), as specified by and in form and substance reasonably satisfactory to the Administrative Agent, in each case securing payment of all the Obligations and constituting Liens on all such Pledged Interests.

6.23 Real Property. Except during a Collateral Release Period, with respect to any fee interest in any Material Real Property that is acquired or any lease of domestic Real Property that is leased for more than \$10,000,000 annually, in either case after the Closing Date by the Borrower or any other Loan Party (other than a BWXT Entity), the Borrower or the applicable Loan Party shall promptly (and, in any event, within thirty days following the date of such acquisition, unless such date is extended by the Administrative Agent in its sole discretion) (i) in the case of any Material Real Property, execute and deliver a first priority Mortgage (subject only to Liens permitted by this Agreement and such Mortgage) in favor of the Administrative Agent, for the benefit of the Secured Parties, covering such Real Property and complying with the provisions herein and in the Security Instruments, (ii) in the case of any leased domestic Real Property that is leased for more than \$10,000,000 annually, if requested by the Administrative Agent, execute and deliver a first priority Mortgage (subject only to Liens permitted by this Agreement and such Mortgage) in favor of the Administrative Agent, for the benefit of the Secured Parties, covering such Real Property and complying with the provisions herein and in the Security Instruments, (iii) provide the Secured Parties with title insurance in an amount at least equal to the purchase price of such Real Property (or such other amount as the Administrative Agent shall reasonably specify) described in clauses (i) or (ii) above, and if applicable, flood insurance and lease estoppel certificates, all in accordance with the standards for deliveries contemplated on the Closing Date, as described in Section 4.01(a)(iv) hereof, (iv) if requested by the Administrative Agent, deliver to the Administrative Agent legal opinions relating to the matters described above, which opinions shall be in form and substance, and from counsel, reasonably satisfactory to the Administrative Agent, and (v) if requested by the Administrative Agent, use commercially reasonable efforts to obtain Landlord Lien Waivers for each domestic Real Property leasehold interest on which a manufacturing facility or warehouse or other facility where Collateral is stored or held (but excluding any office lease that does not include manufacturing or warehouse facilities), provided that no such landlord Lien Waiver shall be required for any location at which $Collateral\ is\ stored\ or\ located\ unless\ the\ aggregate\ value\ of\ Collateral\ stored\ or\ held\ at\ such\ location\ exceeds\ \$10,000,000.$

6.24 Replacement of Borrower. Substantially simultaneously with, and in no event later than five Business Days after, the effectiveness of the Spinoff in accordance with the provisions of this Agreement, the New Borrower shall become the Borrower hereunder, the Original Borrower and each Wholly-Owned Domestic Subsidiary of the New Borrower that is not an Immaterial Subsidiary and is not already a Guarantor shall become Guarantors hereunder, and each such Person (as applicable) shall: (a) deliver to the Administrative Agent and the Lenders (i) a Joinder Agreement executed and delivered by the New Borrower, the Original Borrower and the Administrative Agent with respect to the New Borrower becoming the Borrower hereunder, (ii) a Joinder Agreement executed and delivered by the Original Borrower, each Wholly-Owned Domestic Subsidiary of the New Borrower (other than Immaterial Subsidiaries) that is not already a Guarantor, and the Administrative Agent with respect to such

Persons becoming Guarantors hereunder; and (iii) a reaffirmation agreement from each Guarantor reaffirming such Person's obligations under the Loan Documents to which it is a party, (b) cause the New Borrower and each Wholly-Owned Domestic Subsidiary of the New Borrower (other than Immaterial Subsidiaries) that is not already a Guarantor to deliver to the Administrative Agent documents of the types referred to in clauses (v), (vi) and (viii) of Section 4.01(a) and, at the request of the Administrative Agent, favorable legal opinions of counsel to each such Person (which shall cover, among other things, the legality, validity, binding effect and enforceability of the documentation referred to in clauses (a) and, if applicable, (c) of this Section), all in form, content and scope reasonably satisfactory to the Administrative Agent, (c) except during a Collateral Release Period, cause the New Borrower and each Wholly-Owned Domestic Subsidiary of the New Borrower (other than Immaterial Subsidiaries and BWXT Entities) to deliver documents of the type described in Section 6.22(b)(ii) and take actions of the type described in Section 6.22(b)(ii) and (d) take all such other actions reasonably requested by the Administrative Agent to give effect toSection 2.17.

6.25 BWXT Entities. If, on or after the Closing Date, any BWXT Entity shall pledge its assets or properties in support of or otherwise create or suffer to exist any Lien upon or with respect to any of their respective properties or assets, whether now owned or hereafter acquired, to secure any Indebtedness described in clause (a) or (b) of such definition incurred on or after the Closing Date (other than the Obligations) then such Person shall immediately cease to be a BWXT Entity and the Borrower shall promptly cause such BWXT Entity to pledge its assets and properties as Collateral pursuant to the Security Instruments and take all such other actions of the type described in Section 6.22, 6.23 and 6.26 with respect to Wholly-Owned Domestic Subsidiaries that are required to provide Collateral pursuant to the Security Instruments (including without limitation, the execution and delivery of any intercreditor agreement or other applicable documentation reasonably requested by the Administrative Agent and reasonably satisfactory to the Administrative Agent to ensure that the Administrative Agent's Lien on behalf of the Secured Parties with respect to the properties and assets securing such other Financial Covenant Debt will rank equal and ratable with such the Liens securing such other Financial Covenant Debt).

6.26 Further Assurances. Promptly upon request by the Administrative Agent, or any Lender through the Administrative Agent, the Borrower or the applicable Loan Party shall (a) correct any material defect or error that may be discovered in any Loan Document or in the execution, acknowledgment, filing or recordation thereof, and (b) do, execute, acknowledge, deliver, record, re-record, file, re-file, register and re-register any and all such further acts, deeds, certificates, assurances and other instruments as the Administrative Agent, or any Lender through the Administrative Agent, may reasonably require from time to time in order to (i) carry out more effectively the purposes of the Loan Documents, (ii) except during a Collateral Release Period, to the fullest extent permitted by applicable law, subject any Loan Party's (other than any BWXT Entity's) or any of its Subsidiaries' properties, assets, rights or interests to the Liens now or hereafter intended to be covered by any of the Security Instruments, (iii) except during a Collateral Release Period, perfect and maintain the validity, effectiveness and priority of any of the Security Instruments and any of the Liens intended to be created thereunder and (iv) except during a Collateral Release Period, assure, convey, grant, assign, transfer, preserve, protect and confirm more effectively unto the Secured Parties the rights granted or now or hereafter intended

to be granted to the Secured Parties under any Loan Document or under any other instrument executed in connection with any Loan Document to which any Loan Party (other than any BWXT Entity) or any of its Subsidiaries is or is to be a party, and cause each of its Subsidiaries to do so. Notwithstanding anything to the contrary contained in this Section 6.26 or any Loan Document, no Loan Party shall be required to (i) execute or deliver any control agreements with respect to deposit accounts (other than with respect to Cash Collateral), commodities accounts or securities accounts, (ii) make any filings to perfect Liens on intellectual property, other than any such filings under the UCC or with the U.S. Patent and Trademark Office or U.S. Copyright Office, and (iii) make any filings or take any actions in any jurisdiction outside the United States to create or perfect any Liens created by the Security Instruments.

- 6.27 Post-Closing Deliveries. In the event the Spinoff has not been consummated in full on or prior to September 30, 2010, the Original Borrower shall deliver consolidated balance sheets of the Original Borrower and its Subsidiaries as of the end of the Fiscal Year ended December 31, 2009, and related statements of income and cash flows of the Original Borrower and its Subsidiaries for such Fiscal Year, all prepared in conformity with GAAP and certified, in the case of such consolidated financial statements, without qualification as to the scope of the audit or as to the Original Borrower being a going concern by the Borrower's Accountants, together with the report of such accounting firm stating that (i) such financial statements fairly present in all material respects the consolidated financial position of the Original Borrower and its Subsidiaries as at the dates indicated and the results of their operations and cash flow for the periods indicated in conformity with GAAP applied on a basis consistent with prior years (except for changes with which the Borrower's Accountants shall concur and that shall have been disclosed in the notes to the financial statements) and (ii) the examination by the Borrower's Accountants in connection with such consolidated financial statements has been made in accordance with generally accepted auditing standards.
- 6.28 Cash Collateralization of Extended Letters of Credit. The Borrower shall provide Cash Collateral (in an amount equal to 105% of the maximum face amount of each Extended Letter of Credit, calculated in accordance with Section 1.08) to each applicable L/C Issuer with respect to each Extended Letter of Credit issued by such L/C Issuer by a date that is no earlier than 120 days prior to the Maturity Date, but no later than 95 days prior to the Maturity Date (or, if such Letter of Credit is issued on or after the date that is 95 days prior to the Maturity Date, on the date of issuance thereof); provided that if the Borrower fails to provide Cash Collateral with respect to any such Extended Letter of Credit by such time, such event shall be treated as a drawing under such Extended Letter of Credit (in an amount equal to 105% of the maximum face amount of each such Letter of Credit, calculated in accordance with Section 1.08), which shall be reimbursed (or participations therein funded) in accordance with Section 2.03(c), with the proceeds being utilized to provide Cash Collateral for such Letter of Credit.
- **6.29 Post-Closing Real Property Matters.** To the extent not delivered on or prior to the Closing Date pursuant to Section 4.01(a)(iv), then not later than the date that is 30 days after the Closing Date (which date may be extended by an additional 30 days by the Administrative Agent in its discretion) the Borrower will deliver each of the following to the Administrative Agent with respect to each of the Mortgaged Properties listed on Schedule 4.01(a)(iv):

- (a) evidence that counterparts of the Mortgages have been duly executed, acknowledged and delivered and are in form suitable for filing or recording in all filing or recording offices that the Administrative Agent may deem necessary or desirable in order to create a valid first and subsisting Lien on the property described therein in favor of the Administrative Agent for the benefit of the Secured Parties, excepting only Liens permitted under the Loan Documents, and that all filing, documentary, stamp, intangible and recording taxes and fees have been paid (or the Borrower has made arrangements satisfactory to the Administrative Agent for payment thereof);
- (b) a Mortgagee Policy, with endorsements and in amounts acceptable to the Administrative Agent, issued, coinsured and reinsured by title insurers acceptable to the Administrative Agent, insuring the Mortgages to be valid first and subsisting Liens on the property described therein, free and clear of all defects (including, but not limited to, mechanics' and materialmen's Liens) and encumbrances, excepting only Liens permitted under the Loan Documents;
 - (c) evidence that all premiums in respect of the Mortgagee Policies have been paid;
- (d) evidence that no such Mortgaged Property is located in a special flood hazard area as designated by any federal Governmental Authority other than those for which flood insurance has been provided, and evidence of any such flood insurance;
- (e) evidence that all other action that the Administrative Agent may deem necessary or desirable in order to create valid first and subsisting Liens (excepting only Liens permitted under the Loan Documents) on the property described in the Mortgages has been taken; and
- (f) to the extent applicable to any Mortgaged Property and the documents being delivered in connection therewith, a favorable opinion of (i) Vorys, Sater, Seymour and Pease LLP, local Ohio counsel to certain of the Loan Parties, and (ii) Watkins Ludlam Winter & Stennis, P.A., local Mississippi counsel to certain of the Loan Parties, in each case addressed to the Administrative Agent and each Lender, in form and substance reasonably satisfactory to the Administrative Agent and the Lenders and addressing such matters concerning the relevant Loan Parties, Loan Documents and Mortgaged Properties as the Required Lenders may reasonably request.

Notwithstanding anything to the contrary contained in this <u>Article VI</u>, nothing in this <u>Article VI</u> shall prohibit the consummation of any one or more individual transactions entered into in connection with the Spinoff in accordance with and subject to compliance with each of the conditions set forth in the definition of "Spinoff".

ARTICLE VII. NEGATIVE COVENANTS

The Borrower agrees with the Lenders and the Administrative Agent to each of the following, as long as any Obligation or any Commitment remains outstanding and, in each case, unless the Required Lenders otherwise consent in writing (<u>provided</u> that references herein to "Subsidiaries"

shall exclude any Captive Insurance Subsidiary for all Sections under this Article VII except Sections 7.01 and 7.02):

- **7.01 Indebtedness.** The Borrower shall not, and shall not permit any of its Subsidiaries to, directly or indirectly create, incur, assume or otherwise become or remain directly or indirectly liable with respect to any Indebtedness except for the following:
 - (a) Indebtedness under the Loan Documents;
 - (b) Indebtedness outstanding on the date hereof and listed on Schedule 7.01;
- (c) Guaranty Obligations incurred by the Borrower or any Guarantor in respect of Indebtedness of the Borrower or any Guarantor that is permitted by this Section 7.01 (other than clause (g) below);
- (d)(i) Indebtedness in respect of Capital Lease Obligations and purchase money obligations for tangible property, (ii) Indebtedness in respect of sale and leaseback transactions permitted by Section 7.13 and (iii) other secured Indebtedness (including secured Indebtedness incurred or assumed by the Borrower and its Subsidiaries in connection with a Permitted Acquisition); provided, however, that the aggregate principal amount of all such Indebtedness permitted by this subsection (d) at any one time outstanding shall not exceed \$200,000,000 and the Liens securing such Indebtedness shall be within the limitations set forth in Sections 7.02(d), 7.02(e) or 7.02(k);
- (e) renewals, extensions, refinancings and refundings of Indebtedness permitted by clause (b) or (d) above or this clause (e) provided, however, that any such renewal, extension, refinancing or refunding is in an aggregate principal amount not greater than the principal amount of (plus reasonable fees, expenses and any premium incurred in connection with the renewal, extension, refinancing or refunding of such Indebtedness), and is on terms that in the aggregate are not materially less favorable to the Borrower or such Subsidiary, including as to weighted average maturity, than the Indebtedness being renewed, extended, refinanced or refunded;
- (f) Indebtedness arising from intercompany loans (i) from the Borrower to any Guarantor; (ii) from any Subsidiary of the Borrower to the Borrower or any Guarantor; (iii) from any Subsidiary of the Borrower that is not a Loan Party to any other Subsidiary of the Borrower that is not a Loan Party; (iv) from the Borrower or any Guarantor to any Subsidiary of the Borrower that is not a Guarantor; or (v) prior to the Spinoff, from MII or any Affiliate of MII (other than the Borrower or a Subsidiary of the Borrower) to the Borrower or any Subsidiary of the Borrower; provided, however, that (x) all such Indebtedness (other than the Indebtedness described in clause (iii) or (v) of this clause (f)) shall be evidenced by promissory notes and all such notes shall be subject to a first priority Lien pursuant to the Collateral Agreement (if the payee is a Loan Party that is a party to the Collateral Agreement), (y) all such Indebtedness (other than the Indebtedness described in clauses (i), (iii) and (iv) of this clause (f)) shall be Subordinated Debt and, in the case of Indebtedness described in clause (v) only, shall not permit any cash payments of any kind prior to the Maturity Date, and (z) any payment by any such

Guarantor under any guaranty of the Obligations shall result in a pro tanto reduction of the amount of any Indebtedness owed by such Subsidiary to the Borrower or to any of its Subsidiaries for whose benefit such payment is made; provided, further that, in the case of Indebtedness described in clauses (i), (ii), (iii) and (iv) above, the Investment in the intercompany loan by the lender thereof is permitted under Section 7.03;

- (g) Non-Recourse Indebtedness;
- (h) Indebtedness under or in respect of Swap Contracts that are not speculative in nature;
- (i) unsecured Indebtedness (including unsecured Indebtedness incurred or assumed by the Borrower and its Subsidiaries in connection with a Permitted Acquisition) in an aggregate principal amount not to exceed \$200,000,000 at any time outstanding;
- (j) Indebtedness in respect of any insurance premium financing for insurance being acquired by the Borrower or any Subsidiary under customary terms and conditions and not in connection with the borrowing of money;
 - (k) Indebtedness under or in respect of Cash Management Agreements; and
 - (1) Cash Collateralized Letters of Credit.
- **7.02 Liens.** The Borrower shall not, and shall not permit any of its Subsidiaries to, create or suffer to exist any Lien upon or with respect to any of their respective properties or assets, whether now owned or hereafter acquired, or assign, or permit any of its Subsidiaries to assign, any right to receive income, except for the following:
 - (a) Liens created pursuant to any Loan Document;
 - (b) Liens existing on the date hereof and listed on Schedule 7.02;
 - (c) Customary Permitted Liens;
- (d) Liens granted by the Borrower or any Subsidiary of the Borrower under a Capital Lease and Liens to which any property is subject at the time, on or after the Closing Date, of the Borrower's or such Subsidiary's acquisition thereof in accordance with this Agreement, in each case securing Indebtedness permitted under Section 7.01(d) and limited to the property purchased (and proceeds thereof) with the proceeds subject to such Capital Lease or Indebtedness;
- (e) purchase money security interests in real property, improvements thereto or equipment (including any item of equipment purchased in connection with a particular construction project that the Borrower or a Subsidiary expects to sell to its customer with respect to such project and that, pending such sale, is classified as inventory) hereafter acquired (or, in the case of improvements, constructed) by the Borrower or any of its Subsidiaries; provided, however, that (i) such security interests secure purchase money Indebtedness permitted under Section 7.01(d) and are limited to the property purchased with the proceeds of such purchase

money Indebtedness (and proceeds thereof), (ii) such security interests are incurred, and the Indebtedness secured thereby is created, within ninety days of such acquisition or construction, and (iii) the Indebtedness secured thereby does not exceed the lesser of the cost or Fair Market Value of such real property, improvements or equipment at the time of such acquisition or construction;

- (f) any Lien securing the renewal, extension, refinancing or refunding of any Indebtedness secured by any Lien permitted by clause (b), (d) or (e) above, this clause (f) or clause (k) below, without any material change in the assets subject to such Lien;
 - (g) Liens in favor of lessors securing operating leases permitted hereunder;
- (h) Liens securing Non-Recourse Indebtedness permitted under Section 7.01(g) on (i) the assets of the Subsidiary or Joint Venture financed by such Non-Recourse Indebtedness and (ii) the Stock of the Joint Venture or Subsidiary financed by such Non-Recourse Indebtedness;
 - (i) Liens arising out of judgments or awards and not constituting an Event of Default underSection 8.01(g);
- (j) Liens encumbering inventory, work-in-process and related property in favor of customers or suppliers securing obligations and other liabilities to such customers or suppliers (other than Indebtedness) to the extent such Liens are granted in the ordinary course of business and are consistent with past business practices;
- (k) Liens not otherwise permitted hereunder securing Indebtedness permitted by <u>Section 7.01(d)(ii)</u> or <u>(iii)</u> and encumbering assets of (i) Foreign Subsidiaries or (ii) Domestic Subsidiaries that are not (and are not required to be) Guarantors, in each case that do not constitute Collateral;
- (l) Liens with respect to foreign exchange netting arrangements to the extent incurred in the ordinary course of business and consistent with past business practices; provided, that the aggregate outstanding amount of all such obligations and liabilities secured by such Liens shall not exceed \$15,000,000 at any time;
- (m) Liens securing insurance premium financing permitted under <u>Section 7.01(j)</u> under customary terms and conditions; provided, that no such Lien may extend to or cover any property other than the insurance being acquired with such financing, the proceeds thereof and any unearned or refunded insurance premiums related thereto;
- (n) Liens not otherwise permitted by this Section securing obligations or other liabilities (other than Indebtedness for borrowed money) of the Borrower or its Subsidiaries; provided however, that the aggregate outstanding amount of all such obligations and liabilities secured by such Liens shall not exceed \$20,000,000 at any time; and
 - (o) Liens on Cash Collateral securing only Cash Collateralized Letters of Credit.

- **7.03 Investments.** The Borrower shall not, and shall not permit any of its Subsidiaries to, directly or indirectly make or maintain any Investment except for the following:
- (a) Investments existing on the Closing Date and disclosed on <u>Schedule 7.03</u>, and any refinancings of such Investments to the extent constituting Indebtedness otherwise permitted under <u>Section 7.01(b)</u>, <u>provided</u> such refinancing complies with the provisions of <u>Section 7.01(e)</u>;
 - (b) Investments held by the Borrower or such Subsidiary in the form of cash or Cash Equivalents;
- (c) Investments in accounts, contract rights and chattel paper (each as defined in the UCC), notes receivable and similar items arising or acquired from the sale of Inventory in the ordinary course of business consistent with the past practice of the Borrower and its Subsidiaries;
 - (d) Investments received in settlement of amounts due to the Borrower or any Subsidiary of the Borrower effected in the ordinary course of business;
- (e) Investments by (i) the Borrower in any Guarantor or by any Guarantor in the Borrower or another Guarantor, (ii) a Subsidiary of the Borrower that is not a Guarantor in the Borrower or any of its Subsidiaries, or (iii) the Borrower or any Subsidiary of the Borrower in (A) Joint Ventures; (B) Subsidiaries that are not Guarantors; or (C) or an Affiliate of the Borrower that is neither a Guarantor nor a Joint Venture; provided that, the aggregate outstanding amount of all such Investments pursuant to this clause (iii) (including Letters of Credit and other credit support obligations from the Borrower or its Subsidiaries, and including obligations to make Investments of equity in Joint Ventures or Subsidiaries in connection with the terms of Non-Recourse Indebtedness) shall not exceed \$200,000,000 at any time;
- (f) loans or advances to employees of the Borrower or any of its Subsidiaries (or guaranties of loans and advances made by a third party to employees of the Borrower or any of its Subsidiaries) in the ordinary course of business; provided, that the aggregate principal amount of all such loans and advances and guaranties of loans and advances shall not exceed \$1,000,000 at any time;
 - (g) Investments constituting Guaranty Obligations permitted by Section 7.01;
 - (h) Investments in connection with a Permitted Acquisition;
- (i) Investments in that certain joint venture between Thermax Ltd., an entity organized under the laws of India, and BWPGG or any of its Subsidiaries for the design, manufacture and supply of equipment, including supercritical boilers, to the Indian energy and power sector in an aggregate amount not exceed \$150,000,000 at any time outstanding; and
- (j) Investments not otherwise permitted hereby; <u>provided</u>, however, that the aggregate outstanding amount of all such Investments shall not exceed \$25,000,000 at any time.

For purposes of covenant compliance, the amount of any Investment shall be the original cost of such Investment, minus the amount of any portion of such Investment repaid to the investor as a dividend, repayment of loan or advance, release or discharge of a guarantee or other obligation or other transfer of property or return of capital, as the case may be, but without any other adjustments for increases or decreases in value, or write-ups, write-downs or write-offs with respect to such Investment or interest earned on such Investment.

- **7.04 Asset Sales.** The Borrower shall not, and shall not permit any of its Subsidiaries to, sell, convey, transfer, lease or otherwise dispose of any of their respective assets or any interest therein (including the sale or factoring at maturity of any accounts) to any Person, or permit or suffer any other Person to acquire any interest in any of their respective assets or, in the case of any Subsidiary, issue or sell any shares of such Subsidiary's Stock or Stock Equivalent (any such disposition being an "Asset Sale") except for the following:
 - (a) the sale or disposition of inventory in the ordinary course of business;
- (b) transfers resulting from any taking or condemnation of any property of the Borrower or any of its Subsidiaries (or, as long as no Default exists or would result therefrom, deed in lieu thereof);
- (c) as long as no Default exists or would result therefrom, the sale or disposition of equipment that the Borrower reasonably determines is no longer useful in its or its Subsidiaries' business, has become obsolete, damaged or surplus or is replaced in the ordinary course of business;
- (d) as long as no Default exists or would result therefrom, the sale or disposition of assets of any Subsidiary that is not a Wholly-Owned Subsidiary that, both at the time of such sale and as of the Closing Date, do not constitute, in the aggregate, all or a material part of the assets of such Subsidiary;
- (e) as long as no Default exists or would result therefrom, the lease or sublease of Real Property not constituting a sale and leaseback, to the extent not otherwise prohibited by this Agreement or the Mortgages;
- (f) as long as no Default exists or would result therefrom, non-exclusive assignments and licenses of intellectual property of the Borrower and its Subsidiaries in the ordinary course of business;
- (g) as long as no Default exists or would result therefrom, discounts, adjustments, settlements and compromises of Accounts and contract claims in the ordinary course of business;
 - (h) any Asset Sale (i) to the Borrower or any Guarantor or (ii) by any Subsidiary that is not a Loan Party to another Subsidiary that is not a Loan Party;
- (i) as long as no Default exists or would result therefrom, any other Asset Sale for Fair Market Value, at least 75% of which is payable in cash or Cash Equivalents upon such sale;

provided, however, that with respect to any such Asset Sale in accordance with this clause (i), the aggregate consideration received for the sale of all assets sold in accordance with this clause (i) during any Fiscal Year, including such Asset Sale, shall not exceed \$20,000,000 in the aggregate;

- (j) any single transaction or series of related transactions so long as neither such single transaction nor such series of related transactions involves assets having a Fair Market Value of more than \$1,000,000; and
 - (k) Asset Sales permitted by Section 7.13.
- 7.05 Restricted Payments. The Borrower shall not, and shall not permit any of its Subsidiaries to, directly or indirectly, declare, order, pay or make any sum for any Restricted Payment except for:
 - (a) Restricted Payments by the Borrower to any Guarantor;
- (b) Restricted Payments by (i) any Subsidiary of the Borrower to the Borrower or any Guarantor or (ii) any Subsidiary that is not a Loan Party to another Subsidiary that is not a Loan Party;
- (c) Restricted Payments by any Subsidiary that is not a Wholly-Owned Subsidiary to the Borrower or any Guarantor and to any other direct or indirect holders of equity interests in such Subsidiary to the extent (i) such Restricted Payments are made pro rata among the holders of the equity interests in such Subsidiary or (ii) pursuant to the terms of the joint venture or other distribution agreement for such Subsidiary in form and substance approved by the Administrative Agent (such approval not to be unreasonably withheld or delayed);
- (d) any redemption, retirement, sinking fund or similar payment, purchase or other acquisition for value, direct or indirect, of any Stock or Stock Equivalents of the Borrower or any of its Subsidiaries (i) made solely with the proceeds received from the exercise of any warrant or option or (ii) that is deemed to occur upon the cashless exercise of stock options or warrants;
- (e) the repurchase, redemption or other acquisition or retirement for value of any Stock or Stock Equivalents of the Borrower or any Subsidiary held by any current or former officer, director or employee pursuant to any equity-based compensation plan, equity subscription agreement, stock option agreement, shareholders' agreement or similar agreement in an aggregate amount not to exceed \$10,000,000 in any Fiscal Year; and
- (f) so long as no Default exists, or would result therefrom, the Borrower may make Restricted Payments of the type described in clause (a) and (b) of the definition thereof (including Restricted Payments of the type described in clause (e) of this Section that are in excess of the aggregate amount permitted in clause (e) of this Section), in an aggregate amount not to exceed \$50,000,000 in any Fiscal Year.
- 7.06 Fundamental Changes. Except in connection with a Permitted Acquisition, the Borrower shall not, and shall not permit any of its Subsidiaries to, (a) merge or consolidate with

any Person (<u>provided</u> that, if at the time thereof and immediately after giving effect thereto no Default exists (i) any Wholly-Owned Subsidiary may merge into the Borrower so long as the Borrower is the surviving company, (ii) any Subsidiary may merge into or consolidate with any other Subsidiary in a transaction in which the surviving entity is a Subsidiary and no Person other than the Borrower or a Subsidiary receives any consideration (provided that if any party to any such transaction is a Loan Party, the surviving entity of such transaction shall be a Loan Party) and (iii) any Subsidiary of the Borrower may merge with another Person in a transaction constituting an Asset Sale permitted hereunder), (b) acquire all or substantially all of the Stock or Stock Equivalents of any Person (other than the acquisition of the Stock or Stock Equivalents of a Person that is a Subsidiary of the Borrower prior to such acquisition, <u>provided</u> that if the acquired Subsidiary is a Loan Party, the acquiring Subsidiary must be either the Borrower or a Wholly-Owned Domestic Subsidiary that will comply with <u>Sections 6.22</u> and <u>6.23</u>), (c) acquire all or substantially all of the assets of any Person or all or substantially all of the assets constituting what is known by the Borrower to be the business of a division, branch or other unit operation of any Person (other than an Asset Sale permitted by <u>Section 7.04(h)</u>), (d) enter into any joint venture or partnership with any Person that is not a Loan Party other than any Joint Venture or (e) acquire or create any Subsidiary unless, after giving effect to such acquisition or creation, (i) the Borrower is in compliance with <u>Sections 6.22</u> and <u>6.23</u> and (ii) the Investment in such Subsidiary is permitted under <u>Section 7.03</u>.

7.07 Change in Nature of Business. The Borrower shall not, and shall not permit any of its Subsidiaries to, engage in any business other than the Eligible Line of Business.

7.08 Transactions with Affiliates. The Borrower shall not, and shall not permit any of its Subsidiaries to, enter into any transaction of any kind with any Affiliate of the Borrower, whether or not in the ordinary course of business, other than on fair and reasonable terms substantially as favorable to the Borrower or such Subsidiary as would be obtainable by the Borrower or such Subsidiary at the time in a comparable arm's length transaction with a Person other than an Affiliate except (a) transactions among the Borrower and its Subsidiaries otherwise permitted under the Loan Documents, (b) Restricted Payments and Investments otherwise permitted by this Agreement, and (c) transactions in accordance with the Affiliate Agreements.

7.09 Burdensome Agreements. Other than pursuant to the Loan Documents and any agreements governing any Non-Recourse Indebtedness, or any Indebtedness permitted by Section 7.01(b), (d), (e) or (g) (in the case of any such Indebtedness, so long as any prohibition or limitation is only effective against the assets financed thereby), the Borrower shall not, and shall not permit any of its Subsidiaries to, (a) other than for any Subsidiary that is not a Wholly-Owned Subsidiary, agree to enter into or suffer to exist or become effective any consensual encumbrance or consensual restriction of any kind on the ability of such Subsidiary to pay dividends or make any other distribution or transfer of funds or assets or make loans or advances to or other Investments in, or enter into any Guaranty Obligation or pay any Indebtedness owed to, the Borrower or any other Subsidiary of the Borrower or (b) other than customary non-assignment provisions in contracts entered into in the ordinary course of business, enter into or permit to exist or become effective any enforceable agreement prohibiting or limiting the ability of the Borrower or any Subsidiary to create, incur, assume or permit to exist any Lien upon any of its property, assets or revenues, whether now owned or hereafter acquired, to secure the

Obligations, including any agreement requiring any other Indebtedness or Contractual Obligation to be equally and ratably secured with the Obligations.

- **7.10 Amendments of Constituent Documents**. The Borrower shall not, and shall not permit any of its Subsidiaries to, change its capital structure (including the terms of its outstanding Stock) or otherwise amend its Constituent Documents, except for changes and amendments that do not materially and adversely affect the rights and privileges of the Borrower or any of its Subsidiaries and do not materially and adversely affect the interests of the Secured Parties under the Loan Documents or the rights and interests of any of them in the Collateral.
- **7.11 Accounting Changes**. The Borrower shall not, and shall not permit any of its Subsidiaries to, (a) make any material change in its accounting treatment and reporting practices or tax reporting treatment, except as required by GAAP or any Requirement of Law and disclosed to the Lenders and the Administrative Agent or (b) change its Fiscal Year.
- **7.12** Use of Proceeds. The Borrower shall not, and shall not permit any of its Subsidiaries to, use all or any portion of the proceeds of any credit extended hereunder to purchase or carry margin stock (within the meaning of Regulation U of the FRB) in contravention of Regulation U of the FRB.
- **7.13 Sale Leasebacks.** The Borrower shall not, and shall not permit any of its Subsidiaries to, enter into any sale and leaseback transaction unless the proceeds of such transaction received by the Loan Parties equal the Fair Market Value of the properties subject to such transaction and, after giving effect to such sale and leaseback transaction, the aggregate Fair Market Value of all properties covered at any one time by all sale and leaseback transactions permitted hereunder (other than any sale and leaseback transaction of property entered into within 90 days of the acquisition of such property) does not exceed \$20,000,000.
- **7.14 Capital Expenditures.** The Borrower shall not make or incur, or permit any of its Subsidiaries to make or incur, Capital Expenditures that, in the aggregate, exceed \$150,000,000 during any Fiscal Year; provided that up to \$75,000,000 of the amount permitted for any Fiscal Year, if not expended in the Fiscal Year for which it is so permitted, may be carried over for expenditure only in the next following Fiscal Year (with amounts available to be carried over to a subsequent fiscal year being calculated by assuming amounts expended in any Fiscal Year first utilized the \$150,000,000 basket for such Fiscal Year, and only then utilized any amounts carried over from a prior Fiscal Year).
- 7.15 Cancellation of Indebtedness Owed To It. The Borrower shall not, and shall not permit any of its Subsidiaries to, cancel any material claim or Indebtedness owed to any of them except in the ordinary course of business.
- **7.16 No Speculative Transactions.** The Borrower shall not, and shall not permit any of its Subsidiaries to, engage in any material speculative transaction or in any material transaction involving the entry into of Swap Contracts by such Person except for the sole purpose of hedging in the normal course of business.

7.17 Post-Termination Benefits. Except to the extent required under Section 4980B of the Code or similar state laws, the Borrower shall not, and shall not permit any of its Subsidiaries to, without the consent of the Required Lenders, adopt any Employee Benefit Plan that provides health or welfare benefits (through the purchase of insurance or otherwise) to any retired or former employees, consultants or directors (or their dependents) of the Borrower or any of its Subsidiaries.

7.18 Financial Covenants.

- (a) <u>Interest Coverage Ratio</u>. The Borrower shall not permit the Interest Coverage Ratio as of the end of any Fiscal Quarter (commencing with the Fiscal Quarter ending June 30, 2010, but with effect prior thereto with respect to any required <u>pro forma</u> compliance calculations made prior to such time) to be less than 4.00 to 1.00.
- (b) <u>Leverage Ratio</u>. The Borrower shall not permit the Leverage Ratio as of the end of any Fiscal Quarter (commencing with the Fiscal Quarter ending June 30, 2010, but with effect prior thereto with respect to any required <u>pro forma</u> compliance calculations made prior to such time) to be greater than 2.50 to 1.00.
- 7.19 BWXT Ownership. The Borrower shall not permit any Stock or Stock Equivalents in BWXT to be owned (a) prior to the Spinoff, by any Person other than the Original Borrower, and (b) at any time after the Spinoff, by any Person other than the New Borrower or a Permitted BWXT Owner. A "Permitted BWXT Owner" is any Wholly Owned Subsidiary of the New Borrower all of whose Stock or Stock Equivalents are owned, directly or indirectly, only by the New Borrower and Subsidiaries of the New Borrower that are holding companies without material assets other than Stock or Stock Equivalents in other Subsidiaries of the New Borrower and without material operations other than those related to such ownership, provided that the Original Borrower may also have assets in the form of Investments in cash and Cash Equivalents in addition to Stock and Stock Equivalents.

Notwithstanding anything to the contrary contained in this <u>Article VII</u>, nothing in this <u>Article VII</u> shall prohibit the consummation of any one or more individual transactions entered into in connection with the Spinoff in accordance with and subject to compliance with each of the conditions set forth in the definition of "Spinoff".

ARTICLE VIII. EVENTS OF DEFAULT AND REMEDIES

8.01 Events of Default. Any of the following shall constitute an "Event of Default":

- (a) Non-Payment of Principal. the Borrower shall fail to pay any principal of any Loan or any L/C Obligation when the same becomes due and payable; or
- (b) Non-Payment of Interest and Other Amounts. the Borrower shall fail to pay any interest on any Loan, any fee under any of the Loan Documents or any other Obligation (other than one referred to in clause (a) above) and such non-payment continues for a period of three Business Days after the due date therefor; or

- (c) <u>Representations and Warranties</u>, any representation or warranty made or deemed made by any Loan Party in any Loan Document shall prove to have been incorrect in any material respect when made or deemed made; or
- (d) <u>Failure to Perform Covenants</u>. any Loan Party shall fail to perform or observe (i) any term, covenant or agreement contained in <u>Sections 6.03(a)</u>, <u>6.12</u>, <u>6.17</u>, <u>6.28</u>, <u>6.29</u> or <u>Article VII</u> or (ii) any other term, covenant or agreement contained in this Agreement or in any other Loan Document if such failure under this clause (ii) shall remain unremedied for 30 days after the earlier of (A) the date on which a Responsible Officer of the Borrower obtains actual knowledge of such failure and (B) the date on which written notice thereof shall have been given to the Borrower by the Administrative Agent, any Lender or any L/C Issuer; or
- (e) <u>Cross-Default</u>. (i) the Borrower or any of its Material Subsidiaries shall fail to make any payment on any recourse Indebtedness of the Borrower or any such Material Subsidiary (other than the Obligations) or any Guaranty Obligation in respect of Indebtedness of any other Person, and, in each case, such failure relates to Indebtedness having a principal amount of \$10,000,000 or more when the same becomes due and payable (whether by scheduled maturity, required prepayment, acceleration, demand or otherwise), (ii) any other event shall occur or condition shall exist under any agreement or instrument relating to any such Indebtedness, if the effect of such event or condition is to accelerate, or to permit the acceleration of, the maturity of such Indebtedness or (iii) any such Indebtedness shall become or be declared to be due and payable, or required to be prepaid or repurchased (other than by a regularly scheduled required prepayment), prior to the stated maturity thereof; provided that clauses (ii) and (iii) above shall not apply to secured Indebtedness that becomes due as a result of the voluntary sale or transfer of the property or assets securing such Indebtedness; or
- (f) Insolvency Proceedings. Etc. (i) the Borrower or any of its Material Subsidiaries shall generally not pay its debts as such debts become due, shall admit in writing its inability to pay its debts generally or shall make a general assignment for the benefit of creditors, (ii) any proceeding shall be instituted by or against the Borrower or any of its Material Subsidiaries seeking to adjudicate it a bankrupt or insolvent, or seeking liquidation, winding up, reorganization, arrangement, adjustment, protection, relief or composition of it or its debts, under any Requirement of Law relating to bankruptcy, insolvency or reorganization or relief of debtors, or seeking the entry of an order for relief or the appointment of a custodian, receiver, trustee or other similar official for it or for any substantial part of its property; provided, however, that, in the case of any such proceedings instituted against the Borrower or any of its Material Subsidiaries (but not instituted by the Borrower or any of its Subsidiaries), either such proceedings shall remain undismissed or unstayed for a period of 60 days or more or an order or decree approving or ordering any of the foregoing shall be entered, or (iii) the Borrower or any of its Material Subsidiaries shall take any corporate action to authorize any action set forth in clauses (i) or (ii) above; or
- (g) <u>Judgments</u>. one or more judgments, injunctions or orders (or other similar process) involving, in the case of a money judgment, an amount in excess of \$10,000,000 in the aggregate (to the extent not covered by insurance as to which a solvent and unaffiliated insurance company has acknowledged coverage), shall be rendered against one or more of the Borrower and its Material Subsidiaries and shall remain unpaid and either (x) enforcement proceedings

shall have been commenced by any creditor upon such judgment, injunction or order or (y) there shall be any period of 30 consecutive days during which a stay of enforcement of such judgment, injunction or order, by reason of a pending appeal or otherwise, shall not be in effect; or

- (h) <u>ERISA</u>. (i) one or more ERISA Events shall occur and the amount of all liabilities and deficiencies resulting therefrom imposed on or which could reasonably be expected to be imposed directly on the Borrower, any of its Subsidiaries or any Guarantor, whether or not assessed, when taken together with amounts of all such liabilities and deficiencies for all other such ERISA Events exceeds \$20,000,000 in the aggregate, or (ii) there exists any fact or circumstance that reasonably could be expected to result in the imposition of a Lien or security interest under Section 430 of the Code or under ERISA; or
 - (i) Invalidity of Loan Documents. either:
 - (i) any provision of any Security Instrument or the Guaranty after delivery thereof pursuant to this Agreement or any other Loan Document shall for any reason, except as permitted by the Loan Documents, cease to be valid and binding on, or enforceable against, any Loan Party which is a party thereto, or any Loan Party shall so state in writing; or
 - (ii) any Security Instrument shall for any reason fail or cease to create a valid Lien on any Collateral with an aggregate value of \$5,000,000 or more purported to be covered thereby or, except as permitted by the Loan Documents, such Lien shall fail or cease to be a perfected and first priority Lien or any Loan Party shall so state in writing; or
 - (j) Change of Control. there occurs any Change of Control.
- **8.02 Remedies Upon Event of Default** If any Event of Default occurs and is continuing, the Administrative Agent shall, at the request of, or may, with the consent of, the Required Lenders, take any or all of the following actions:
- (a) declare the commitment of each Lender to make Loans and any obligation of the L/C Issuers to make L/C Credit Extensions to be terminated, whereupon such commitments and obligation shall be terminated;
- (b) declare the unpaid principal amount of all outstanding Loans, all interest accrued and unpaid thereon, and all other amounts owing or payable hereunder or under any other Loan Document to be immediately due and payable, without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower;
 - (c) require that the Borrower Cash Collateralize the L/C Obligations (in an amount equal to the then Outstanding Amount thereof); and
 - (d) exercise on behalf of itself, the Lenders and the L/C Issuers all rights and remedies available to it, the Lenders and the L/C Issuers under the Loan Documents;

provided, however, that upon the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under the Bankruptcy Code of the United States, the obligation of each Lender to make Loans and any obligation of the L/C Issuers to make L/C Credit Extensions shall automatically terminate, the unpaid principal amount of all outstanding Loans and all interest and other amounts as aforesaid shall automatically become due and payable, and the obligation of the Borrower to Cash Collateralize the L/C Obligations as aforesaid shall automatically become effective, in each case without further act of the Administrative Agent or any Lender.

8.03 Application of Funds. After the exercise of remedies provided for in <u>Section 8.02</u> (or after the Loans have automatically become immediately due and payable and the L/C Obligations have automatically been required to be Cash Collateralized as set forth in the proviso to <u>Section 8.02</u>), any amounts received on account of the Obligations shall, subject to the provisions of <u>Sections 2.15</u> and <u>2.16</u>, be applied by the Administrative Agent in the following order:

First, to payment of that portion of the Obligations constituting fees, indemnities, expenses and other amounts (including fees, charges and disbursements of counsel to the Administrative Agent and amounts payable under Article III) payable to the Administrative Agent in its capacity as such;

Second, to payment of that portion of the Obligations constituting fees, indemnities and other amounts (other than principal, interest and Letter of Credit Fees) payable to the Lenders and the L/C Issuers (including fees, charges and disbursements of counsel to the respective Lenders and the L/C Issuers arising under the Loan Documents and amounts payable under Article III), ratably among them in proportion to the respective amounts described in this clause Second payable to them;

Third, to payment of that portion of the Obligations constituting accrued and unpaid Letter of Credit Fees and interest on the Loans, L/C Borrowings and other Obligations arising under the Loan Documents, ratably among the Lenders and the L/C Issuers in proportion to the respective amounts described in this clause Third payable to them;

<u>Fourth</u>, to payment of that portion of the Obligations constituting unpaid principal of the Loans, L/C Borrowings and Obligations then owing under Secured Hedge Agreements and Secured Cash Management Agreements, ratably among the Lenders, the L/C Issuers, the Hedge Banks and the Cash Management Banks in proportion to the respective amounts described in this clause <u>Fourth</u> held by them;

 $\underline{\text{Fifth}}, \text{ to the Administrative Agent for the account of the L/C Issuers, to Cash Collateralize that portion of L/C Obligations composed of the aggregate undrawn amount of Letters of Credit to the extent not otherwise Cash Collateralized by the Borrower pursuant to <math>\underline{\text{Sections 2.03}}$ and $\underline{2.15}$; and

Last, the balance, if any, after all of the Obligations have been indefeasibly paid in full, to the Borrower or as otherwise required by any applicable Requirement of Law.

Subject to Sections 2.03(c) and 2.15, amounts used to Cash Collateralize the aggregate undrawn amount of Letters of Credit pursuant to clause ithen be applied to satisfy drawings under such Letters of Credit as they occur. If any amount remains on deposit as Cash Collateral after all Letters of Credit have either been fully drawn or expired, such remaining amount shall be applied to the other Obligations, if any, in the order set forth above.

Notwithstanding the foregoing, Obligations arising under Secured Cash Management Agreements and Secured Hedge Agreements shall be excluded from the application described above if the Administrative Agent has not received written notice thereof, together with such supporting documentation as the Administrative Agent may reasonably request, from the applicable Cash Management Bank or Hedge Bank, as the case may be. Each Cash Management Bank or Hedge Bank not a party to the Credit Agreement that has given the notice contemplated by the preceding sentence shall, by such notice, be deemed to have acknowledged and accepted the appointment of the Administrative Agent pursuant to the terms of Article IX hereof for itself and its Affiliates as if a "Lender" party hereto.

ARTICLE IX. ADMINISTRATIVE AGENT

9.01 Appointment and Authority.

(a) Each of the Lenders and each L/C Issuer hereby irrevocably appoints Bank of America to act on its behalf as the Administrative Agent hereunder and under the other Loan Documents and authorizes the Administrative Agent to take such actions on its behalf and to exercise such powers as are delegated to the Administrative Agent by the terms hereof or thereof, together with such actions and powers as are reasonably incidental thereto. The provisions of this Article are solely for the benefit of the Administrative Agent, the Lenders and the L/C Issuers, and the Borrower shall not have any rights as a third party beneficiary of any of such provisions.

(b) The Administrative Agent shall also act as the "collateral agent" under the Loan Documents, and each of the Lenders (including in its capacities as a potential Hedge Bank and a potential Cash Management Bank) and the L/C Issuer hereby irrevocably appoints and authorizes the Administrative Agent to act as the agent of such Lender and the L/C Issuer for purposes of acquiring, holding and enforcing any and all Liens on Collateral granted by any of the Loan Parties to secure any of the Obligations, together with such powers and discretion as are reasonably incidental thereto. In this connection, the Administrative Agent, as "collateral agent" and any coagents, sub-agents and attorneys-in-fact appointed by the Administrative Agent pursuant to Section 9.05 for purposes of holding or enforcing any Lien on the Collateral (or any portion thereof) granted under the Security Instruments, or for exercising any rights and remedies thereunder at the direction of the Administrative Agent, shall be entitled to the benefits of all provisions of this Article IX and Article X (including Section 10.04(c), as though such co-agents, sub-agents and attorneys-in-fact were the "collateral agent" under the Loan Documents) as if set forth in full herein with respect thereto. Without limiting the generality of the foregoing, the Administrative Agent is further authorized on behalf of all the Lenders, without the necessity of any notice to or further consent from the Lenders, from time to time to take any action, or permit the any co-agents, sub-agents and attorneys-in-fact appointed by the Administrative Agent to

take any action, with respect to any Collateral or the Loan Documents which may be necessary to perfect and maintain perfected the Liens upon any Collateral granted pursuant to any Loan Document.

- 9.02 Rights as a Lender. The Person serving as the Administrative Agent hereunder shall have the same rights and powers in its capacity as a Lender as any other Lender and may exercise the same as though it were not the Administrative Agent and the term "Lender" or "Lenders" shall, unless otherwise expressly indicated or unless the context otherwise requires, include the Person serving as the Administrative Agent hereunder in its individual capacity. Such Person and its Affiliates may accept deposits from, lend money to, act as the financial advisor or in any other advisory capacity for and generally engage in any kind of business with the Borrower or any Subsidiary or other Affiliate thereof as if such Person were not the Administrative Agent hereunder and without any duty to account therefor to the Lenders.
- **9.03 Exculpatory Provisions.** The Administrative Agent shall not have any duties or obligations except those expressly set forth herein and in the other Loan Documents. Without limiting the generality of the foregoing, the Administrative Agent:
 - (a) shall not be subject to any fiduciary or other implied duties, regardless of whether a Default has occurred and is continuing;
- (b) shall not have any duty to take any discretionary action or exercise any discretionary powers, except discretionary rights and powers expressly contemplated hereby or by the other Loan Documents that the Administrative Agent is required to exercise as directed in writing by the Required Lenders (or such other number or percentage of the Lenders as shall be expressly provided for herein or in the other Loan Documents), <u>provided</u> that the Administrative Agent shall not be required to take any action that, in its opinion or the opinion of its counsel, may expose the Administrative Agent to liability or that is contrary to any Loan Document or applicable law; and
- (c) shall not, except as expressly set forth herein and in the other Loan Documents, have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Affiliates that is communicated to or obtained by the Person serving as the Administrative Agent or any of its Affiliates in any capacity.

The Administrative Agent shall not be liable for any action taken or not taken by it (i) with the consent or at the request of the Required Lenders (or such other number or percentage of the Lenders as shall be necessary, or as the Administrative Agent shall believe in good faith shall be necessary, under the circumstances as provided in <u>Sections 10.01</u> and <u>8.02</u>) or (ii) in the absence of its own gross negligence or willful misconduct. The Administrative Agent shall be deemed not to have knowledge of any Default unless and until notice describing such Default is given to the Administrative Agent by the Borrower, a Lender or the L/C Issuer.

The Administrative Agent shall not be responsible for or have any duty to ascertain or inquire into (i) any statement, warranty or representation made in or in connection with this Agreement or any other Loan Document, (ii) the contents of any certificate, report or other document delivered hereunder or thereunder or in connection herewith, (iii) the

performance or observance of any of the covenants, agreements or other terms or conditions set forth herein or therein or the occurrence of any Default, (iv) the validity, enforceability, effectiveness or genuineness of this Agreement, any other Loan Document or any other agreement, instrument or document, or the creation, perfection or priority of any Lien purported to be created by the Security Instruments, (v) the value or the sufficiency of any Collateral, or (vi) the satisfaction of any condition set forth in Article IV or elsewhere herein, other than to confirm receipt of items expressly required to be delivered to the Administrative Agent.

9.04 Reliance by Administrative Agent.

The Administrative Agent shall be entitled to rely upon, and shall not incur any liability for relying upon, any notice, request, certificate, consent, statement, instrument, document or other writing (including any electronic message, Internet or intranet website posting or other distribution) believed by it to be genuine and to have been signed, sent or otherwise authenticated by the proper Person. The Administrative Agent also may rely upon any statement made to it orally or by telephone and believed by it to have been made by the proper Person, and shall not incur any liability for relying thereon. In determining compliance with any condition hereunder to the making of a Loan, or the issuance of a Letter of Credit, that by its terms must be fulfilled to the satisfaction of a Lender or the L/C Issuer, the Administrative Agent may presume that such condition is satisfactory to such Lender or the L/C Issuer unless the Administrative Agent shall have received notice to the contrary from such Lender or the L/C Issuer prior to the making of such Loan or the issuance of such Letter of Credit. The Administrative Agent may consult with legal counsel (who may be counsel for the Borrower), independent accountants and other experts selected by it, and shall not be liable for any action taken or not taken by it in accordance with the advice of any such counsel, accountants or experts.

9.05 Delegation of Duties. The Administrative Agent may perform any and all of its duties and exercise its rights and powers hereunder or under any other Loan Document by or through any one or more sub agents appointed by the Administrative Agent. The Administrative Agent and any such sub agent may perform any and all of its duties and exercise its rights and powers by or through their respective Related Parties. The exculpatory provisions of this Article shall apply to any such sub agent and to the Related Parties of the Administrative Agent and any such sub agent, and shall apply to their respective activities in connection with the syndication of the credit facilities provided for herein as well as activities as Administrative Agent.

9.06 Resignation of Administrative Agent. The Administrative Agent may at any time give notice of its resignation to the Lenders, the L/C Issuers and the Borrower. Upon receipt of any such notice of resignation, the Required Lenders shall have the right, in consultation with the Borrower, to appoint a successor, which shall be a bank with an office in the United States, or an Affiliate of any such bank with an office in the United States. If no such successor shall have been so appointed by the Required Lenders and shall have accepted such appointment within 30 days after the retiring Administrative Agent gives notice of its resignation, then the retiring Administrative Agent may on behalf of the Lenders and the L/C Issuers, appoint a successor Administrative Agent meeting the qualifications set forth above; provided that if the Administrative Agent shall notify the Borrower and the Lenders that no qualifying Person has accepted such appointment, then such resignation shall nonetheless become effective in accordance with such notice and (1) the retiring Administrative Agent shall

be discharged from its duties and obligations hereunder and under the other Loan Documents (except that in the case of any Collateral held by the Administrative Agent on behalf of the Secured Parties under any of the Loan Documents, the retiring Administrative Agent shall continue to hold such Collateral until such time as a successor Administrative Agent is appointed) and (2) all payments, communications and determinations provided to be made by, to or through the Administrative Agent shall instead be made by or to each Lender and the applicable L/C Issuer directly, until such time as the Required Lenders appoint a successor Administrative Agent as provided for above in this Section. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring (or retired) Administrative Agent, and the retiring Administrative Agent shall be discharged from all of its duties and obligations hereunder or under the other Loan Documents (if not already discharged therefrom as provided above in this Section). The fees payable by the Borrower to a successor Administrative Agent shall be the same as those payable to its predecessor unless otherwise agreed between the Borrower and such successor. After the retiring Administrative Agent's resignation hereunder and under the other Loan Documents, the provisions of this Article and Section 10.04 shall continue in effect for the benefit of such retiring Administrative Agent, its sub agents and their respective Related Parties in respect of any actions taken or omitted to be taken by any of them while the retiring Administrative Agent was acting as Administrative Agent.

Any resignation by Bank of America as Administrative Agent pursuant to this Section shall also constitute its resignation as an L/C Issuer and as the Swing Line Lender. Upon the acceptance of a successor's appointment as Administrative Agent hereunder, (a) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Swing Line Lender and, unless the Borrower elects not to have such Person serve in such capacity, a retiring L/C Issuer, (b) the retiring L/C Issuer and Swing Line Lender shall be discharged from all of their respective duties and obligations hereunder or under the other Loan Documents, and (c) the L/C Issuers hereunder (including such successor, if applicable) shall issue letters of credit in substitution for the Letters of Credit, if any, outstanding at the time of such succession or make other arrangements satisfactory to the retiring L/C Issuer to effectively assume the obligations of the retiring L/C Issuer with respect to such Letters of Credit.

9.07 Non-Reliance on Administrative Agent and Other Lenders. Each Lender and each L/C Issuer acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. Each Lender and each L/C Issuer also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender or any of their Related Parties and based on such documents and information as it shall from time to time deem appropriate, continue to make its own decisions in taking or not taking action under or based upon this Agreement, any other Loan Document or any related agreement or any document furnished hereunder or thereunder.

9.08 No Other Duties, Etc. Anything herein to the contrary notwithstanding, none of the Bookrunners, Arrangers, Syndication Agent or Documentation Agents listed on the cover

page hereof shall have any powers, duties or responsibilities under this Agreement or any of the other Loan Documents, except in its capacity, as applicable, as the Administrative Agent, a Lender or an L/C Issuer hereunder.

- 9.09 Administrative Agent May File Proofs of Claim. In case of the pendency of any proceeding under any Debtor Relief Law or any other judicial proceeding relative to any Loan Party, the Administrative Agent (irrespective of whether the principal of any Loan or L/C Obligation shall then be due and payable as herein expressed or by declaration or otherwise and irrespective of whether the Administrative Agent shall have made any demand on the Borrower) shall be entitled and empowered, by intervention in such proceeding or otherwise:
- (a) to file and prove a claim for the whole amount of the principal and interest owing and unpaid in respect of the Loans, L/C Obligations and all other Obligations that are owing and unpaid and to file such other documents as may be necessary or advisable in order to have the claims of the Lenders, the L/C Issuers and the Administrative Agent (including any claim for the reasonable compensation, expenses, disbursements and advances of the Lenders, the L/C Issuers and the Administrative Agent and their respective agents and counsel and all other amounts due the Lenders, the L/C Issuers and the Administrative Agent under Sections 2.03(h) and (i), 2.09 and 10.04) allowed in such judicial proceeding; and
 - (b) to collect and receive any monies or other property payable or deliverable on any such claims and to distribute the same;

and any custodian, receiver, assignee, trustee, liquidator, sequestrator or other similar official in any such judicial proceeding is hereby authorized by each Lender and each L/C Issuer to make such payments to the Administrative Agent and, in the event that the Administrative Agent shall consent to the making of such payments directly to the Lenders and the L/C Issuers, to pay to the Administrative Agent any amount due for the reasonable compensation, expenses, disbursements and advances of the Administrative Agent and its agents and counsel, and any other amounts due the Administrative Agent under Sections 2.09 and 10.04.

Nothing contained herein shall be deemed to authorize the Administrative Agent to authorize or consent to or accept or adopt on behalf of any Lender or any L/C Issuer any plan of reorganization, arrangement, adjustment or composition affecting the Obligations or the rights of any Lender or any L/C Issuer to authorize the Administrative Agent to vote in respect of the claim of any Lender or any L/C Issuer in any such proceeding.

- **9.10 Collateral and Guaranty Matters.** Each of the Lenders (including in its capacities as a potential Cash Management Bank and a potential Hedge Bank, and on behalf of their Affiliates in such capacities) and each L/C Issuer irrevocably authorize the Administrative Agent, at its option and in its discretion,
- (a) to release any Lien on any property granted to or held by the Administrative Agent under any Loan Document (i) upon termination of the Aggregate Commitments and payment in full of all Obligations (other than (A) contingent indemnification obligations and (B) obligations and liabilities under Secured Cash Management Agreements and Secured Hedge Agreements either (x) as to which arrangements satisfactory to the applicable Cash Management

Bank or Hedge Bank shall have been made or (y) notice has not been received by the Administrative Agent from the applicable Cash Management Bank or Hedge Bank that such amounts are then due and payable) and the expiration or termination of all Letters of Credit (other than Letters of Credit as to which other arrangements satisfactory to the Administrative Agent and the applicable L/C Issuer shall have been made), (ii) that is sold or to be sold as part of or in connection with any sale permitted hereunder or under any other Loan Document, (iii) in connection with the release of the Collateral provided in Section 10.19(a), or (iv) subject to Section 10.01, if approved, authorized or ratified in writing by the Required Lenders;

- (b) to subordinate or release any Lien on any property granted to or held by the Administrative Agent under any Loan Document to the holder of any Lien on such property that is permitted by Section 7.02(b), (d), (e) or (f) or (h), and to enter into any intercreditor agreement, subordination agreement or similar agreement with respect to any such property; and
 - (c) to release any Guarantor from its obligations under the Guaranty if such Person ceases to be a Subsidiary as a result of a transaction permitted hereunder.

Upon request by the Administrative Agent at any time, the Required Lenders will confirm in writing the Administrative Agent's authority to release or subordinate its interest in particular types or items of property, or to release any Guarantor from its obligations under the Guaranty pursuant to this Section 9.10. In each case as specified in this Section 9.10, the Administrative Agent will, at the Borrower's expense, execute and deliver to the applicable Loan Party such documents as such Loan Party may reasonably request to evidence the release of such item of Collateral from the assignment and security interest granted under the Security Instruments or to subordinate its interest in such item, or to release such Guarantor from its obligations under the Guaranty, in each case in accordance with the terms of the Loan Documents and this Section 9.10.

9.11 Secured Cash Management Agreements and Secured Hedge Agreements. No Cash Management Bank or Hedge Bank that obtains the benefits of Section 8.03, any Guaranty or any Collateral by virtue of the provisions hereof or of any Guaranty or any Security Instrument shall have any right to notice of any action or to consent to, direct or object to any action hereunder or under any other Loan Document or otherwise in respect of the Collateral (including the release or impairment of any Collateral) other than in its capacity as a Lender and, in such case, only to the extent expressly provided in the Loan Documents. Notwithstanding any other provision of this Article IX to the contrary, the Administrative Agent shall not be required to verify the payment of, or that other satisfactory arrangements have been made with respect to, Obligations arising under Secured Cash Management Agreements and Secured Hedge Agreements unless the Administrative Agent has received written notice of such Obligations, together with such supporting documentation as the Administrative Agent may request, from the applicable Cash Management Bank or Hedge Bank, as the case may be.

ARTICLE X. MISCELLANEOUS

- 10.01 Amendments, Etc. No amendment or waiver of any provision of this Agreement or any other Loan Document, and no consent to any departure by the Borrower or any other Loan Party therefrom, shall be effective unless in writing signed by the Required Lenders and the Borrower or the applicable Loan Party, as the case may be, and acknowledged by the Administrative Agent, and each such waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; provided, however, that no such amendment, waiver or consent shall:
 - (a) waive any condition set forth in Section 4.01 (other than Section 4.01(b)(i) or (c)) without the written consent of each Lender;
- (b) extend or increase the Commitment of any Lender (or reinstate any Commitment terminated pursuant to <u>Section 8.02</u>) without the written consent of such Lender;
- (c) postpone any date fixed by this Agreement or any other Loan Document for any payment (excluding mandatory prepayments, if any) of principal, interest, fees or other amounts due to the Lenders (or any of them) hereunder or under any other Loan Document without the written consent of each Lender entitled to such payment;
- (d) reduce the principal of, or the rate of interest specified herein on, any Loan or L/C Borrowing, or (subject to clause (iv) of the second proviso to this Section 10.01) any fees or other amounts payable hereunder or under any other Loan Document without the written consent of each Lender entitled to such amount; provided, however, that only the consent of the Required Lenders shall be necessary to amend the definition of "Default Rate" or to waive any obligation of the Borrower to pay interest, commitment fees or Letter of Credit Fees at the Default Rate;
 - (e) change Section 8.03 in a manner that would alter the pro rata sharing of payments required thereby without the written consent of each Lender;
 - (f) amend Section 1.06 or the definition of "Alternative Currency" without the written consent of the Administrative Agent and each affected L/C Issuer;
- (g) change any provision of this Section or the definition of "Required Lenders" or any other provision hereof specifying the number or percentage of Lenders required to amend, waive or otherwise modify any rights hereunder or make any determination or grant any consent hereunder without the written consent of each Lender; or
- (h) release all or substantially all of the Collateral in any transaction or series of related transactions, or release all or substantially all of the value of the Guaranty, in each case without the written consent of each Lender, except to the extent the release of any Collateral or any Guarantor is permitted pursuant to <u>Section 10.19(a)</u> and/or <u>Section 9.10</u> (in which case such release may be made by the Administrative Agent acting alone);

and, provided further, that (i) no amendment, waiver or consent shall, unless in writing and signed by the applicable L/C Issuer in addition to the Lenders required above, affect the rights or duties of such L/C Issuer under this Agreement or any Issuer Document relating to any Letter of Credit issued or to be issued by it; (ii) no amendment, waiver or consent shall, unless in writing and signed by the Swing Line Lender in addition to the Lenders required above, affect the rights or duties of the Swing Line Lender under this Agreement; (iii) no amendment, waiver or consent shall, unless in writing and signed by the Administrative Agent in addition to the Lenders required above, affect the rights or duties of the Administrative Agent under this Agreement or any other Loan Document; and (iv) each Fee Letter may be amended, or rights or privileges thereunder waived, in a writing executed only by the parties thereto. Notwithstanding anything to the contrary herein, (x) no Defaulting Lender shall have any right to approve or disapprove any amendment, waiver or consent hereunder (and any amendment, waiver or consent which by its terms requires the consent of all Lenders or each affected Lender may be effected with the consent of the applicable Lenders other than Defaulting Lenders), except that (1) the Commitment of any Defaulting Lender may not be increased or extended without the consent of such Lender and (2) any waiver, amendment or modification requiring the consent of all Lenders or each affected Lender that by its terms affects any Defaulting Lender more adversely than other affected Lenders shall require the consent of such Defaulting Lender and (y) the Administrative Agent, the Borrower and the applicable L/C Issuer may, without the consent of any other Lender or L/C Issuer, make such changes as may be necessary to incorporate provisions with respect to the issuance of Letters of Credit in any Alternative Currency approved by such L/C Issuer. Notwithstanding anything to the contrary contained in this Section, if the Administrative Agent and the Borrower shall have jointly identified (each in its sole discretion) an obvious error or omission of a technical or immaterial nature, in each case, in any provision of the Loan Documents, then the Administrative Agent and the applicable Loan Parties shall be permitted to amend such provision and such amendment shall become effective without any further action or consent of any other party to any Loan Document if the same is not objected to in writing by the Required Lenders within five Business Days following the posting of such amendment to the Lenders.

If any Lender does not consent to a proposed amendment, waiver, consent or release with respect to any Loan Document that requires the consent of such Lender and that has been approved by the Required Lenders, the Borrower may replace such non-consenting Lender in accordance with Section 10.13; provided that such amendment, waiver, consent or release can be effected as a result of the assignment contemplated by such Section (together with all other such assignments required by the Borrower to be made pursuant to this paragraph).

10.02 Notices: Effectiveness: Electronic Communication.

(a) <u>Notices Generally</u>. Except in the case of notices and other communications expressly permitted to be given by telephone (and except as provided in subsection (b) below), all notices and other communications provided for herein shall be in writing and shall be delivered by hand or overnight courier service, mailed by certified or registered mail or sent by telecopier as follows, and all notices and other communications expressly permitted hereunder to be given by telephone shall be made to the applicable telephone number, as follows:

- (i) if to the Borrower, the Administrative Agent, Bank of America as an L/C Issuer or the Swing Line Lender, to the address, telecopier number, electronic mail address or telephone number specified for such Person on Schedule 10.02; and
- (ii) if to any other Lender or any other L/C Issuer, to the address, telecopier number, electronic mail address or telephone number specified in its Administrative Questionnaire (including, as appropriate, notices delivered solely to the Person designated by a Lender on its Administrative Questionnaire then in effect for the delivery of notices that may contain material non-public information relating to the Borrower).

Notices and other communications sent by hand or overnight courier service, or mailed by certified or registered mail, shall be deemed to have been given when received; notices and other communications sent by telecopier shall be deemed to have been given when sent (except that, if not given during normal business hours for the recipient, shall be deemed to have been given at the opening of business on the next business day for the recipient). Notices and other communications delivered through electronic communications to the extent provided in subsection (b) below, shall be effective as provided in such subsection (b).

(b) <u>Electronic Communications</u>. Notices and other communications to the Lenders and the L/C Issuer hereunder may be delivered or furnished by electronic communication (including e-mail and Internet or intranet websites) pursuant to procedures approved by the Administrative Agent, <u>provided</u> that the foregoing shall not apply to notices to any Lender or any L/C Issuer pursuant to <u>Article II</u> if such Lender or such L/C Issuer, as applicable, has notified the Administrative Agent that it is incapable of receiving notices under such Article by electronic communication. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic communications pursuant to procedures approved by it, <u>provided</u> that approval of such procedures may be limited to particular notices or communications.

Unless the Administrative Agent otherwise prescribes, (i) notices and other communications sent to an e-mail address shall be deemed received upon the sender's receipt of an acknowledgement from the intended recipient (such as by the "return receipt requested" function, as available, return e-mail or other written acknowledgement), <u>provided</u> that if such notice or other communication is not sent during the normal business hours of the recipient, such notice or communication shall be deemed to have been sent at the opening of business on the next business day for the recipient, and (ii) notices or communications posted to an Internet or intranet website shall be deemed received upon the deemed receipt by the intended recipient at its e-mail address as described in the foregoing clause (i) of notification that such notice or communication is available and identifying the website address therefor.

(c) The Platform. THE PLATFORM IS PROVIDED "AS IS" AND "AS AVAILABLE." THE AGENT PARTIES (AS DEFINED BELOW) DO NOT WARRANT THE ACCURACY OR COMPLETENESS OF THE BORROWER MATERIALS OR THE ADEQUACY OF THE PLATFORM, AND EXPRESSLY DISCLAIM LIABILITY FOR ERRORS IN OR OMISSIONS FROM THE BORROWER MATERIALS. NO WARRANTY OF ANY KIND, EXPRESS, IMPLIED OR STATUTORY, INCLUDING ANY WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NON-INFRINGEMENT

OF THIRD PARTY RIGHTS OR FREEDOM FROM VIRUSES OR OTHER CODE DEFECTS, IS MADE BY ANY AGENT PARTY IN CONNECTION WITH THE BORROWER MATERIALS OR THE PLATFORM. In no event shall the Administrative Agent or any of its Related Parties (collectively, the "Agent Parties") have any liability to the Borrower, any Lender, any L/C Issuer or any other Person for losses, claims, damages, liabilities or expenses of any kind (whether in tort, contract or otherwise) arising out of the Borrower's or the Administrative Agent's transmission of Borrower Materials through the Internet, except to the extent that such losses, claims, damages, liabilities or expenses are determined by a court of competent jurisdiction by a final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Agent Party; provided, however, that in no event shall any Agent Party have any liability to the Borrower, any Lender, the L/C Issuer or any other Person for indirect, special, incidental, consequential or punitive damages (as opposed to direct or actual damages).

(d) Change of Address, Etc. Each of the Borrower, the Administrative Agent, the L/C Issuers and the Swing Line Lender may change its address, telecopier or telephone number for notices and other communications hereunder by notice to the other parties hereto. Each other Lender may change its address, telecopier or telephone number for notices and other communications hereunder by notice to the Borrower, the Administrative Agent, each L/C Issuer and the Swing Line Lender. In addition, each Lender and each L/C Issuer agrees to notify the Administrative Agent from time to time to ensure that the Administrative Agent has on record (i) an effective address, contact name, telephone number, telecopier number and electronic mail address to which notices and other communications may be sent and (ii) accurate wire instructions for such Lender or L/C Issuer. Furthermore, each Public Lender agrees to cause at least one individual at or on behalf of such Public Lender to at all times have selected the "Private Side Information" or similar designation on the content declaration screen of the Platform in order to enable such Public Lender or its delegate, in accordance with such Public Lender's compliance procedures and applicable Requirements of Law, including United States Federal and state securities laws, to make reference to Borrower Materials that are not made available through the "Public Side Information" portion of the Platform and that may contain material non-public information with respect to the Borrower or its securities for purposes of United States Federal or state securities laws.

(e) Reliance by Administrative Agent, L/C Issuer and Lenders The Administrative Agent, the L/C Issuer and the Lenders shall be entitled to rely and act upon any notices (including telephonic Committed Loan Notices and Swing Line Loan Notices) purportedly given by or on behalf of the Borrower even if (i) such notices were not made in a manner specified herein, were incomplete or were not preceded or followed by any other form of notice specified herein, or (ii) the terms thereof, as understood by the recipient, varied from any confirmation thereof. The Borrower shall indemnify the Administrative Agent, each L/C Issuer, each Lender and the Related Parties of each of them from all losses, costs, expenses and liabilities resulting from the reliance by such Person on each notice purportedly given by or on behalf of the Borrower. All telephonic notices to and other telephonic communications with the Administrative Agent may be recorded by the Administrative Agent, and each of the parties hereto hereby consents to such recording.

10.03 No Waiver; Cumulative Remedies; Enforcement. No failure by any Lender or the Administrative Agent to exercise, and no delay by any such Person in exercising, any right, remedy, power or privilege hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege. The rights, remedies, powers and privileges herein provided are cumulative and not exclusive of any rights, remedies, powers and privileges provided by law.

Notwithstanding anything to the contrary contained herein or in any other Loan Document, the authority to enforce rights and remedies hereunder and under the other Loan Documents against the Loan Parties or any of them shall be vested exclusively in, and all actions and proceedings at law in connection with such enforcement shall be instituted and maintained exclusively by, the Administrative Agent in accordance with Section 8.02 for the benefit of all the Lenders and the L/C Issuer; provided, however, that the foregoing shall not prohibit (a) the Administrative Agent from exercising on its own behalf the rights and remedies that inure to its benefit (solely in its capacity as Administrative Agent) hereunder and under the other Loan Documents, (b) any L/C Issuer or the Swing Line Lender from exercising the rights and remedies that inure to its benefit (solely in its capacity as an L/C Issuer or Swing Line Lender, as the case may be) hereunder and under the other Loan Documents, (c) any Lender from exercising setoff rights in accordance with Section 10.08 (subject to the terms of Section 2.13), or (d) any Lender from filing proofs of claim or appearing and filing pleadings on its own behalf during the pendency of a proceeding relative to any Loan Party under any Debtor Relief Law; and provided, further, that if at any time there is no Person acting as Administrative Agent hereunder and under the other Loan Documents, then (i) the Required Lenders shall have the rights otherwise ascribed to the Administrative Agent pursuant to Section 8.02 and (ii) in addition to the matters set forth in clauses (b), (c) and (d) of the preceding proviso and subject to Section 2.13, any Lender may, with the consent of the Required Lenders, enforce any rights and remedies available to it and as authorized by the Required Lenders.

10.04 Expenses; Indemnity; Damage Waiver.

(a) Costs and Expenses. The Borrower shall pay (i) all reasonable out of pocket expenses incurred by the Administrative Agent and its Affiliates (including the reasonable fees, charges and disbursements of counsel for the Administrative Agent) and, to the extent provided in the Commitment Letter, each Arranger, in connection with the syndication of the credit facilities provided for herein, the preparation, negotiation, execution, delivery and administration of this Agreement and the other Loan Documents or any amendments, modifications or waivers of the provisions hereof or thereof (whether or not the transactions contemplated hereby or thereby shall be consummated), (ii) all reasonable out of pocket expenses incurred by each L/C Issuer in connection with the issuance, amendment, renewal or extension of any Letter of Credit or any demand for payment thereunder and (iii) all out of pocket expenses incurred by the Administrative Agent, any Lender or any L/C Issuer (including the fees, charges and disbursements of any counsel for the Administrative Agent, any Lender or the L/C Issuer) in connection with the enforcement or protection of its rights (A) in connection with this Agreement and the other Loan Documents, including its rights under this Section, or (B) in connection with the Loans made or Letters of Credit issued hereunder, including all such out of

pocket expenses incurred during any workout, restructuring or negotiations in respect of such Loans or Letters of Credit.

(b) Indemnification by the Borrower. The Borrower shall indemnify the Administrative Agent (and any sub-agent thereof), each Arranger, each Lender and each L/C Issuer, and each Related Party of any of the foregoing Persons (each such Person being called an "Indemnitee") against, and hold each Indemnitee harmless from, any and all losses, claims, damages, liabilities and related expenses (including the reasonable fees, charges and disbursements of any counsel for any Indemnitee), incurred by any Indemnitee or asserted against any Indemnitee by any third party or by the Borrower or any other Loan Party arising out of, in connection with, or as a result of (i) the execution or delivery of this Agreement, any other Loan Document or any agreement or instrument contemplated hereby or thereby, the performance by the parties hereto of their respective obligations hereunder or thereunder, the consummation of the transactions contemplated hereby or thereby, or, in the case of the Administrative Agent (and any sub-agent thereof) and its Related Parties only, the administration of this Agreement and the other Loan Documents (including in respect of any matters addressed in Section 3.01), (ii) any Loan or Letter of Credit or the use or proposed use of the proceeds therefrom (including any refusal by any L/C Issuer to honor a demand for payment under a Letter of Credit if the documents presented in connection with such demand do not strictly comply with the terms of such Letter of Credit), (iii) any actual or alleged presence or release of Contaminants on or from any property owned or operated by the Borrower or any of its Subsidiaries, or any Environmental Liability related in any way to the Borrower or any of its Subsidiaries, or (iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, whether based on contract, tort or any other theory, whether brought by a third party or by the Borrower or any other Loan Party, and regardless of whether any Indemnitee is a party thereto; provided that such indemnity shall not, as to any Indemnitee, be available to the extent that such losses, claims, damages, liabilities or related expenses (x) are determined by a court of competent jurisdiction by final and nonappealable judgment to have resulted from the gross negligence or willful misconduct of such Indemnitee, (y) are solely between or among Indemnified Parties (other than the Administrative Agent acting in such capacity), or (z) result from a claim brought by the Borrower or any other Loan Party against an Indemnitee for breach in bad faith of such Indemnitee's obligations hereunder or under any other Loan Document, if the Borrower or such Loan Party has obtained a final and nonappealable judgment in its favor on such claim as determined by a court of competent jurisdiction.

(c) Reimbursement by Lenders. To the extent that the Borrower for any reason fails to indefeasibly pay (and without limiting any obligation of the Borrower so to pay) any amount required under subsection (a) or (b) of this Section to be paid by it to the Administrative Agent (or any sub-agent thereof), any L/C Issuer or any Related Party of any of the foregoing, each Lender severally agrees to pay to the Administrative Agent (or any such sub-agent), the applicable L/C Issuer or such Related Party, as the case may be, such Lender's Applicable Percentage (determined as of the time that the applicable unreimbursed expense or indemnity payment is sought) of such unpaid amount, provided that the unreimbursed expense or indemnified loss, claim, damage, liability or related expense, as the case may be, was incurred by or asserted against the Administrative Agent (or any such sub-agent) or the applicable L/C Issuer in its capacity as such, or against any Related Party of any of the foregoing acting for the

Administrative Agent (or any such sub-agent) or such L/C Issuer in connection with such capacity. The obligations of the Lenders under this subsection (c) are subject to the provisions of Section 2.12(d).

- (d) Waiver of Consequential Damages, Etc. To the fullest extent permitted by applicable law, the Borrower shall not assert, and the Borrower hereby waives, any claim against any Indemnitee, on any theory of liability, for special, indirect, consequential or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, this Agreement, any other Loan Document or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby, any Loan or Letter of Credit or the use of the proceeds thereof. No Indemnitee referred to in subsection (b) above shall be liable for any damages arising from the use by unintended recipients of any information or other materials distributed to such unintended recipients by such Indemnitee through telecommunications, electronic or other information transmission systems in connection with this Agreement or the other Loan Documents or the transactions contemplated hereby or thereby other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnitee as determined by a final and nonappealable judgment of a court of competent jurisdiction.
 - (e) Payments. All amounts due under this Section shall be payable not later than ten Business Days after demand therefor.
- (f) <u>Survival</u>. The agreements in this Section shall survive the resignation of the Administrative Agent, any L/C Issuer and/or the Swing Line Lender, the replacement of any Lender, the termination of the Aggregate Commitments and the repayment, satisfaction or discharge of all the other Obligations.

10.05 Payments Set Aside. To the extent that any payment by or on behalf of the Borrower is made to the Administrative Agent, any L/C Issuer or any Lender, or the Administrative Agent, any L/C Issuer or any Lender exercises its right of setoff, and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by the Administrative Agent, such L/C Issuer or such Lender in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Law or otherwise, then (a) to the extent of such recovery, the obligation or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such setoff had not occurred, and (b) each Lender and each L/C Issuer severally agrees to pay to the Administrative Agent upon demand its applicable share (without duplication) of any amount so recovered from or repaid by the Administrative Agent, plus interest thereon from the date of such demand to the date such payment is made at a rate per annum equal to the applicable Overnight Rate from time to time in effect, in the applicable currency of such recovery or payment. The obligations of the Lenders and the L/C Issuers under clause (b) of the preceding sentence shall survive the payment in full of the Obligations and the termination of this Agreement.

10.06 Successors and Assigns.

(a) <u>Successors and Assigns Generally</u>. The provisions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns permitted hereby, except that, other than as permitted and required by <u>Sections 2.17</u> and <u>6.24</u>, the Borrower may not assign or otherwise transfer any of its rights or obligations hereunder without the prior written consent of the Administrative Agent and each Lender and no Lender may assign or otherwise transfer any of its rights or obligations hereunder except (i) to an assignee in accordance with the provisions of subsection (b) of this Section, (ii) by way of participation in accordance with the provisions of subsection (d) of this Section, or (iii) by way of pledge or assignment of a security interest subject to the restrictions of subsection (f) of this Section (and any other attempted assignment or transfer by any party hereto shall be null and void). Nothing in this Agreement, expressed or implied, shall be construed to confer upon any Person (other than the parties hereto, their respective successors and assigns permitted hereby, Participants to the extent provided in subsection (d) of this Section and, to the extent expressly contemplated hereby, the Related Parties of each of the Administrative Agent, the L/C Issuers and the Lenders) any legal or equitable right, remedy or claim under or by reason of this Agreement.

(b) <u>Assignments by Lenders</u>. Any Lender may at any time assign to one or more assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Commitment and the Loans (including for purposes of this subsection (b), participations in L/C Obligations and in Swing Line Loans) at the time owing to it); provided that any such assignment shall be subject to the following conditions:

(i) Minimum Amounts.

- (A) in the case of an assignment of the entire remaining amount of the assigning Lender's Commitment and the Loans at the time owing to it or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (B) in any case not described in subsection (b)(i)(A) of this Section, the aggregate amount of the Commitment (which for this purpose includes Loans outstanding thereunder) or, if the Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment, determined as of the date the Assignment and Assumption with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption, as of the Trade Date, shall not be less than \$5,000,000 unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); provided, however, that concurrent assignments to members of an Assignee Group and concurrent assignments from members of an Assignee Group to a single Eligible Assignee (or to an Eligible Assignee and members of its Assignee Group) will be treated as a single assignment for purposes of determining whether such minimum amount has been met.

- (ii) <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loans or the Commitment assigned, except that this clause (ii) shall not apply to the Swing Line Lender's rights and obligations in respect of Swing Line Loans;
- (iii) Required Consents. No consent shall be required for any assignment except to the extent required by subsection (b)(i)(B) of this Section and, in addition:
 - (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (1) an Event of Default has occurred and is continuing at the time of such assignment or (2) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund, provided that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within five (5) Business Days after having received notice thereof;
 - (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required if such assignment is to a Person that is not a Lender, an Affiliate of such Lender or an Approved Fund with respect to such Lender;
 - (C) the consent of each L/C Issuer (such consent not to be unreasonably withheld or delayed) shall be required for any assignment that increases the obligation of the assignee to participate in exposure under one or more Letters of Credit (whether or not then outstanding); and
 - (D) the consent of the Swing Line Lender (such consent not to be unreasonably withheld or delayed) shall be required for any assignment.
- (iv) <u>Assignment and Assumption</u>. The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption, together with a processing and recordation fee in the amount of \$3,500; <u>provided</u>, <u>however</u>, that the Administrative Agent may, in its sole discretion, elect to waive such processing and recordation fee in the case of any assignment. The assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire.
- (v) No Assignment to Certain Persons No such assignment shall be made (A) to the Borrower or any of the Borrower's Affiliates or Subsidiaries, or (B) to any Defaulting Lender or any of its Subsidiaries, or any Person who, upon becoming a Lender hereunder, would constitute any of the foregoing Persons described in this clause (B), or (C) to a natural person, or (D) to any competitor of the Borrower or any of its Subsidiaries that is primarily engaged in an Eligible Line of Business and that has been previously identified as such by the Borrower to the Administrative Agent.
- (vi) <u>Certain Additional Payments</u>. In connection with any assignment of rights and obligations of any Defaulting Lender hereunder, no such assignment shall be

effective unless and until, in addition to the other conditions thereto set forth herein, the parties to the assignment shall make such additional payments to the Administrative Agent in an aggregate amount sufficient, upon distribution thereof as appropriate (which may be outright payment, purchases by the assignee of participations or subparticipations, or other compensating actions, including funding, with the consent of the Borrower and the Administrative Agent, the applicable pro rata share of Loans previously requested but not funded by the Defaulting Lender, to each of which the applicable assignee and assignor hereby irrevocably consent), to (x) pay and satisfy in full all payment liabilities then owed by such Defaulting Lender to the Administrative Agent or any Lender hereunder (and interest accrued thereon) and (y) acquire (and fund as appropriate) its full pro rata share of all Loans and participations in Letters of Credit and Swing Line Loans in accordance with its Applicable Percentage. Notwithstanding the foregoing, in the event that any assignment of rights and obligations of any Defaulting Lender hereunder shall become effective under applicable Requirements of Law without compliance with the provisions of this paragraph, then the assignee of such interest shall be deemed to be a Defaulting Lender for all purposes of this Agreement until such compliance occurs.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to subsection (c) of this Section, from and after the effective date specified in each Assignment and Assumption, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 3.01, 3.04, 3.05, and 10.04 with respect to facts and circumstances occurring prior to the effective date of such assignment. Upon request, the Borrower (at its expense) shall execute and deliver a Note to the assignee Lender. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this subsection shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with subsection (d) of this Section.

(c) Register. The Administrative Agent, acting solely for this purpose as an agent of the Borrower (and such agency being solely for tax purposes), shall maintain at the Administrative Agent's Office a copy of each Assignment and Assumption delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Commitments of, and principal amounts of the Loans and L/C Obligations owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. In addition, the Administrative Agent shall maintain on the Register information regarding the designation, and revocation of designation, of any Lender as a Defaulting Lender. The Register shall be available for inspection

by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

(d) <u>Participations</u>. Any Lender may at any time, without the consent of, or notice to, the Borrower or the Administrative Agent, sell participations to any Person (other than a natural person, a Defaulting Lender or the Borrower or any of the Borrower's Affiliates or Subsidiaries) (each, a "<u>Participant</u>") in all or a portion of such Lender's rights and/or obligations under this Agreement (including all or a portion of its Commitment and/or the Loans (including such Lender's participations in L/C Obligations and/or Swing Line Loans) owing to it); <u>provided</u> that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations and (iii) the Borrower, the Administrative Agent, the Lenders and the L/C Issuer shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement.

Any agreement or instrument pursuant to which a Lender sells such a participation shall provide that such Lender shall retain the sole right to enforce this Agreement and to approve any amendment, modification or waiver of any provision of this Agreement; <u>provided</u> that such agreement or instrument may provide that such Lender will not, without the consent of the Participant, agree to any amendment, waiver or other modification described in the first proviso to <u>Section 10.01</u> that affects such Participant. Subject to subsection (e) of this Section, the Borrower agrees that each Participant shall be entitled to the benefits of <u>Sections 3.01</u>, <u>3.04</u> and <u>3.05</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to subsection (b) of this Section. To the extent permitted by law, each Participant also shall be entitled to the benefits of <u>Section 10.08</u> as though it were a Lender.

- (e) <u>Limitations upon Participant Rights</u>. A Participant shall not be entitled to receive any greater payment under <u>Section 3.01</u> or <u>3.04</u> than the applicable Lender would have been entitled to receive with respect to the participation sold to such Participant, unless the sale of the participation to such Participant is made with the Borrower's prior written consent. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of <u>Section 3.01</u> unless the Borrower is notified of the participation sold to such Participant and such Participant agrees, for the benefit of the Borrower, to comply with <u>Section 3.01(e)</u> as though it were a Lender.
- (f) <u>Certain Pledges</u>. Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement (including under its Note(s), if any) to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; <u>provided</u> that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.

(g) Resignation as L/C Issuer or Swing Line Lender after Assignment

(i) Notwithstanding anything to the contrary contained herein, if at any time Bank of America or any other L/C Issuer assigns all of its Commitment and Loans

pursuant to subsection (b) above, then (i) Bank of America or such other L/C Issuer may, upon 30 days' notice to the Borrower and the Lenders, resign as an L/C Issuer and/or (ii) Bank of America may, upon 30 days' notice to the Borrower, resign as the Swing Line Lender. In the event of any such resignation of an L/C Issuer or the Swing Line Lender, the Borrower shall be entitled to appoint from among the Lenders a successor L/C Issuer or Swing Line Lender hereunder; provided, however, that no failure by the Borrower to appoint any such successor shall affect the resignation of Bank of America or the applicable L/C Issuer as an L/C Issuer or of Bank of America as the Swing Line Lender, as the case may be.

- (ii) If Bank of America or any other L/C Issuer resigns as an L/C Issuer, it shall retain all the rights, powers, privileges and duties of an L/C Issuer hereunder with respect to all Letters of Credit issued by it and outstanding as of the effective date of its resignation as an L/C Issuer and all L/C Obligations with respect thereto (including the right to require the Lenders to make Base Rate Committed Loans or fund risk participations in Unreimbursed Amounts pursuant to Section 2.03(c)). Upon the appointment of a successor L/C Issuer with respect to such resigning L/C Issuer (x) such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring L/C Issuer and (y) such successor L/C Issuer (or another of the L/C Issuers, as may be arranged by the Borrower) shall issue letters of credit in substitution for the Letters of Credit, if any, issued by the resigning L/C Issuer and outstanding at the time of such succession, or make other arrangements satisfactory to Bank of America or such other resigning L/C Issuer to effectively assume the obligations of Bank of America or such other resigning L/C Issuer to appoint and remove L/C Issuers pursuant to Sections 2.03(1) and (m).
- (iii) If Bank of America resigns as Swing Line Lender, it shall retain all the rights of the Swing Line Lender provided for hereunder with respect to Swing Line Loans made by it and outstanding as of the effective date of such resignation, including the right to require the Lenders to make Base Rate Committed Loans or fund risk participations in outstanding Swing Line Loans pursuant to Section 2.04(c). Upon the appointment of a successor Swing Line Lender, such successor shall succeed to and become vested with all of the rights, powers, privileges and duties of the retiring Swing Line Lender.
- 10.07 Treatment of Certain Information; Confidentiality. Each of the Administrative Agent, the Lenders and each L/C Issuer agrees to maintain the confidentiality of the Information (as defined below), except that Information may be disclosed (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, trustees, advisors and representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential), (b) to the extent requested by any regulatory authority purporting to have jurisdiction over it (including any self-regulatory authority, such as the National Association of Insurance Commissioners), (c) to the extent required by applicable laws or regulations or by any subpoena or similar legal process, (d) to any other party hereto, (e) in

connection with the exercise of any remedies hereunder or under any other Loan Document or any action or proceeding relating to this Agreement or any other Loan Document or the enforcement of rights hereunder or thereunder, (f) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any assignee of or Participant in, or any prospective assignee of or Participant in, any of its rights or obligations under this Agreement or any Eligible Assignee invited to be a Lender pursuant to Section 2.14(c) or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations, (g) with the consent of the Borrower or (h) to the extent such Information (x) becomes publicly available other than as a result of a breach of this Section or (y) becomes available to the Administrative Agent, any Lender, any L/C Issuer or any of their respective Affiliates on a nonconfidential basis from a source other than the Borrower. For purposes of this Section, "Information" means all information received from the Borrower, any Subsidiary or any Affiliate of the Borrower relating to the Borrower, any Subsidiary or any Affiliate of the Borrower or any of their respective businesses, other than any such information that is (i) available to the Administrative Agent, any Lender or any L/C Issuer on a nonconfidential basis prior to disclosure by the Borrower, any Subsidiary or any Affiliate of the Borrower, or (ii) is clearly and conspicuously marked "PUBLIC" by the Borrower, which, at a minimum, shall mean that the word "PUBLIC" shall appear prominently on the page thereof. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

The Administrative Agent, the Lenders and the L/C Issuers acknowledge that the Borrower and its Subsidiaries perform classified contracts funded by or for the benefit of the United States Federal government and, accordingly, neither the Borrower nor any Subsidiary will be obligated to release, disclose or otherwise make available to the Administrative Agent, any Lender or any L/C Issuer any classified or special nuclear material to any parties not in possession of a valid security clearance and authorized by the appropriate agency of the United States Federal government to receive such material. The Administrative Agent, the Lenders and the L/C Issuers agree that in connection with any exercise of a right or remedy the United States Federal government may remove classified information or government-issued property. Upon notice from the Borrower, the Administrative Agent, the Lenders and the L/C Issuers shall take such steps in accordance with this Agreement as may reasonably be requested by the Borrower to enable the Borrower or any Subsidiary thereof to comply with the Foreign Ownership Control or Influence requirements of the United States Federal government imposed from time to time.

Each of the Administrative Agent, the Lenders and the L/C Issuers acknowledges that (a) the Information may include material non-public information concerning the Borrower or a Subsidiary, as the case may be, (b) it has developed compliance procedures regarding the use of material non-public information and (c) it will handle such material non-public information in accordance with applicable Requirements of Law, including United States Federal and state securities laws.

10.08 Right of Setoff. If an Event of Default shall have occurred and be continuing, each Lender, each L/C Issuer and each of their respective Affiliates is hereby authorized at any time and from time to time, to the fullest extent permitted by applicable Requirements of Law to set off and apply any and all deposits (general or special, time or demand, provisional or final, in whatever currency) at any time held and other obligations (in whatever currency) at any time owing by such Lender, such L/C Issuer or any such Affiliate to or for the credit or the account of the Borrower against any and all of the obligations of the Borrower now or hereafter existing under this Agreement or any other Loan Document to such Lender or such L/C Issuer, irrespective of whether or not such Lender or such L/C Issuer shall have made any demand under this Agreement or any other Loan Document and although such obligations of the Borrower may be contingent or unmatured or are owed to a branch or office of such Lender or such L/C Issuer different from the branch or office holding such deposit or obligated on such indebtedness; provided that in the event that any Defaulting Lender shall exercise any such right of setoff, (x) all amounts so set off shall be paid over immediately to the Administrative Agent for further application in accordance with the provisions of Section 2.16 and, pending such payment, shall be segregated by such Defaulting Lender from its other funds and deemed held in trust for the benefit of the Administrative Agent and the Lenders, and (y) the Defaulting Lender shall provide promptly to the Administrative Agent a statement describing in reasonable detail the Obligations owing to such Defaulting Lender as to which it exercised such right of setoff. The rights of each Lender, each L/C Issuer or their respective Affiliates under this Section are in addition to other rights and remedies (including other rights of setoff) that such Lender, such L/C Issuer or their respective Affiliates may have. Each Lender and each

10.09 Interest Rate Limitation. Notwithstanding anything to the contrary contained in any Loan Document, the interest paid or agreed to be paid under the Loan Documents shall not exceed the maximum rate of non-usurious interest permitted by applicable Requirements of Law (the "Maximum Rate"). If the Administrative Agent or any Lender shall receive interest in an amount that exceeds the Maximum Rate, the excess interest shall be applied to the principal of the Loans or, if it exceeds such unpaid principal, refunded to the Borrower. In determining whether the interest contracted for, charged, or received by the Administrative Agent or a Lender exceeds the Maximum Rate, such Person may, to the extent permitted by applicable Requirements of Law, (a) characterize any payment that is not principal as an expense, fee, or premium rather than interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread in equal or unequal parts the total amount of interest throughout the contemplated term of the Obligations hereunder.

10.10 Counterparts; Integration; Effectiveness This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement and the other Loan Documents constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Section 4.01, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts

hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by telecopy or other electronic imaging means shall be effective as delivery of a manually executed counterpart of this Agreement.

- 10.11 Survival of Representations and Warranties. All representations and warranties made hereunder and in any other Loan Document or other document delivered pursuant hereto or thereto or in connection herewith or therewith shall survive the execution and delivery hereof and thereof. Such representations and warranties have been or will be relied upon by the Administrative Agent and each Lender, regardless of any investigation made by the Administrative Agent or any Lender or on their behalf and notwithstanding that the Administrative Agent or any Lender may have had notice or knowledge of any Default at the time of any Credit Extension, and shall continue in full force and effect as long as any Loan or any other Obligation hereunder shall remain unpaid or unsatisfied or any Letter of Credit shall remain outstanding.
- 10.12 Severability. If any provision of this Agreement or the other Loan Documents is held to be illegal, invalid or unenforceable, (a) the legality, validity and enforceability of the remaining provisions of this Agreement and the other Loan Documents shall not be affected or impaired thereby and (b) the parties shall endeavor in good faith negotiations to replace the illegal, invalid or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the illegal, invalid or unenforceable provisions. The invalidity of a provision in a particular jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction. Without limiting the foregoing provisions of this Section 10.12, if and to the extent that the enforceability of any provisions in this Agreement relating to Defaulting Lenders shall be limited by Debtor Relief Laws, as determined in good faith by the Administrative Agent, any L/C Issuer or the Swing Line Lender, as applicable, then such provisions shall be deemed to be in effect only to the extent not so limited.
- 10.13 Replacement of Lenders. If any Lender requests compensation under Section 3.04, or if the Borrower is required to pay any additional amount to any Lender or any Governmental Authority for the account of any Lender pursuant to Section 3.01, or if any Lender is a Defaulting Lender, or if any Lender is subject to replacement pursuant to the last paragraph of Section 10.01, then the Borrower may, at its sole expense and effort, upon notice to such Lender and the Administrative Agent, require such Lender to assign and delegate, without recourse (in accordance with and subject to the restrictions contained in, and consents required by, Section 10.06), all of its interests, rights and obligations under this Agreement and the related Loan Documents to an assignee that shall assume such obligations (which assignee may be another Lender, if a Lender accepts such assignment), provided that:
 - (a) the Borrower shall have paid to the Administrative Agent the assignment fee specified in Section 10.06(b);
- (b) such Lender shall have received payment of an amount equal to 100% of the outstanding principal of its Loans and L/C Advances, accrued interest thereon, accrued fees and all other amounts payable to it hereunder and under the other Loan Documents (including any

amounts under Section 3.05) from the assignee (to the extent of such outstanding principal and accrued interest and fees) or the Borrower (in the case of all other amounts):

- (c) in the case of any such assignment resulting from a claim for compensation under Section 3.04 or payments required to be made pursuant to Section 3.01, such assignment will result in a reduction in such compensation or payments thereafter; and
 - (d) such assignment does not conflict with applicable Requirements of Law.

A Lender shall not be required to make any such assignment or delegation if, prior thereto, as a result of a waiver by such Lender or otherwise, the circumstances entitling the Borrower to require such assignment and delegation cease to apply.

10.14 Governing Law; Jurisdiction; Etc.

- (a) <u>GOVERNING LAW</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK.
- (b) <u>SUBMISSION TO JURISDICTION</u>. THE BORROWER IRREVOCABLY AND UNCONDITIONALLY SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE NONEXCLUSIVE JURISDICTION OF THE COURTS OF THE STATE OF NEW YORK SITTING IN NEW YORK COUNTY AND OF THE UNITED STATES DISTRICT COURT OF THE SOUTHERN DISTRICT OF NEW YORK AND ANY APPELLATE COURT FROM ANY THEREOF, IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, OR FOR RECOGNITION OR ENFORCEMENT OF ANY JUDGMENT, AND EACH OF THE PARTIES HERETO IRREVOCABLY AND UNCONDITIONALLY AGREES THAT ALL CLAIMS IN RESPECT OF ANY SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN SUCH NEW YORK STATE COURT OR, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, IN SUCH FEDERAL COURT. EACH OF THE PARTIES HERETO AGREES THAT A FINAL JUDGMENT IN ANY SUCH ACTION OR PROCEEDING SHALL BE CONCLUSIVE AND MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR IN ANY OTHER MANNER PROVIDED BY LAW. NOTHING IN THIS AGREEMENT OR IN ANY OTHER LOAN DOCUMENT SHALL AFFECT ANY RIGHT THAT THE ADMINISTRATIVE AGENT, ANY LENDER OR ANY L/C ISSUER MAY OTHERWISE HAVE TO BRING ANY ACTION OR PROCEEDING RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT AGAINST THE BORROWER OR ITS PROPERTIES IN THE COURTS OF ANY JURISDICTION.
- (c) <u>WAIVER OF VENUE</u>. THE BORROWER IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY OBJECTION THAT IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT IN ANY COURT REFERRED TO IN PARAGRAPH (B) OF THIS SECTION. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT

PERMITTED BY APPLICABLE LAW, THE DEFENSE OF AN INCONVENIENT FORUM TO THE MAINTENANCE OF SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.

(d) <u>SERVICE OF PROCESS</u>. EACH PARTY HERETO IRREVOCABLY CONSENTS TO SERVICE OF PROCESS IN THE MANNER PROVIDED FOR NOTICES IN <u>SECTION 10.02</u>. NOTHING IN THIS AGREEMENT WILL AFFECT THE RIGHT OF ANY PARTY HERETO TO SERVE PROCESS IN ANY OTHER MANNER PERMITTED BY APPLICABLE LAW.

10.15 Waiver of Jury Trial. EACH PARTY HERETO HEREBY IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY LEGAL PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

10.16 No Advisory or Fiduciary Responsibility. In connection with all aspects of each transaction contemplated hereby (including in connection with any amendment, waiver or other modification hereof or of any other Loan Document), the Borrower acknowledges and agrees that: (i) (A) the arranging and other services regarding this Agreement provided by the Administrative Agent and the Arrangers are arm's-length commercial transactions between the Borrower and its Affiliates, on the one hand, and the Administrative Agent and the Arrangers, on the other hand, (B) the Borrower has consulted its own legal, accounting, regulatory and tax advisors to the extent it has deemed appropriate, and (C) the Borrower is capable of evaluating, and understands and accepts, the terms, risks and conditions of the transactions contemplated hereby and by the other Loan Documents; (ii) (A) each of the Administrative Agent and the Arrangers is and has been acting solely as a principal and, except as expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for the Borrower, any other Loan Party or any of their respective Affiliates, or any other Person and (B) neither the Administrative Agent nor any Arranger has any obligation to the Borrower or any other Loan Documents; and (iii) the Administrative Agent and the Arrangers and their respective Affiliates may be engaged in a broad range of transactions that involve interests that differ from those of the Borrower, the other Loan Parties and their respective Affiliates. To the fullest extent permitted by law, each of the Borrower hereby waives and releases any claims that it may have

against the Administrative Agent or any Arranger with respect to any breach or alleged breach of agency or fiduciary duty in connection with any aspect of any transaction contemplated hereby.

10.17 Electronic Execution of Assignments and Certain Other Documents. The words "execution," "signed," "signature," and words of like import in any Assignment and Assumption or in any amendment or other modification hereof (including waivers and consents) shall be deemed to include electronic signatures or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any applicable law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act.

10.18 Judgment Currency. If, for the purposes of obtaining judgment in any court, it is necessary to convert a sum due hereunder or any other Loan Document in one currency into another currency, the rate of exchange used shall be that at which in accordance with normal banking procedures the Administrative Agent could purchase the first currency with such other currency on the Business Day preceding that on which final judgment is given. The obligation of the Borrower in respect of any such sum due from it to the Administrative Agent or any Lender hereunder or under the other Loan Documents shall, notwithstanding any judgment in a currency (the "Judgment Currency") other than that in which such sum is denominated in accordance with the applicable provisions of this Agreement (the 'Agreement Currency'), be discharged only to the extent that on the Business Day following receipt by the Administrative Agent or such Lender, as the case may be, of any sum adjudged to be so due in the Judgment Currency, the Administrative Agent or such Lender, as the case may be, may in accordance with normal banking procedures purchase the Agreement Currency with the Judgment Currency. If the amount of the Agreement Currency so purchased is less than the sum originally due to the Administrative Agent or such Lender, as the case may be, against such loss. If the amount of the Agreement Currency so purchased is greater than the sum originally due to the Administrative Agent or any Lender in such currency, the Administrative Agent or such Lender, as the case may be, agrees to return the amount of any excess to the Borrower (or to any other Person who may be entitled thereto under applicable law).

10.19 Release and Reinstatement of Collateral.

(a) Notwithstanding anything to the contrary contained in this Agreement, any Loan Document or any other document executed in connection herewith, if at any time (including after a Collateral Reinstatement Event shall have occurred) a Collateral Release Event shall have occurred and be continuing, then all Collateral (other than Cash Collateral) and the Security Instruments (other than Security Instruments entered into in connection with Cash Collateral) shall be released automatically and terminated without any further action. In connection with the foregoing, the Administrative Agent shall, at Borrower's expense and at the Borrower's request, promptly execute and file in the appropriate location and deliver to Borrower such termination and full or partial release statements or confirmation thereof, as applicable, and do such other

things as are reasonably necessary to release the liens to be released pursuant hereto promptly upon the effectiveness of any such release.

(b) Notwithstanding clause (a) above, if a Collateral Reinstatement Event shall have occurred all Collateral and Security Instruments shall, at the Borrower's sole cost and expense, be reinstated and all actions reasonably necessary, or reasonably requested by the Administrative Agent, to provide to the Administrative Agent for the benefit of the Secured Parties valid, perfected, first priority security interests in the Collateral (including without limitation the delivery of documentation and taking of actions of the type described in clauses (a), (b) and (c) of Section 6.22) shall be taken within 30 days of such event, which 30 day period may be extended by the Administrative Agent in its sole discretion.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed as of the date first above written.

BABCOCK & WILCOX INVESTMENT COMPANY

| By: /s | s/ James C. Lewis |
|--------|-------------------|
| Name: | James C. Lewis |
| Title: | Treasurer |

BANK OF AMERICA, N.A., as Administrative Agent

By: /s/ Bridgett J. Manduk

Name: Bridgett J. Manduk

Title: Assistant Vice President

 \boldsymbol{BANK} \boldsymbol{OF} $\boldsymbol{AMERICA},$ $\boldsymbol{N.A.},$ as a Lender, an L/C Issuer and Swing Line Lender

By: /s/ Mathew Griesbach

Name: Mathew Griesbach

Title: Vice President

BNP PARIBAS, as a Lender and an L/C Issuer

By: /s/ Nicolas Rabier

Name: Nicolas Rabier

Title: Director

By: /s/ John Treadwell, Jr.

Name: John Treadwell, Jr.

Title: Director

 $\boldsymbol{JPMORGAN}$ CHASE BANK, N.A., as a Lender and an L/C Issuer

By: /s/ Patrick S. Thornton

Name: Patrick S. Thornton Title: Senior Vice President

CRÉDIT AGRICOLE CORPORATE AND INVESTMENT BANK, as a Lender and an L/C Issuer

By: /s/ Page Dillehunt

Name: Page Dillehunt

Title: Managing Director

By: /s/ Michael Willis

Name: Michael Willis

Title: Managing Director

WELLS FARGO BANK, N.A., as a Lender

By: /s/ J.C. Hernandez

Name: J.C. Hernandez

Title: Vice President

COMPASS BANK, as a Lender and an L/C Issuer

By: /s/ Stuart Murray

Name: Stuart Murray

Title: Senior Vice President

THE BANK OF NOVA SCOTIA, as a Lender and an L/C Issuer

By: /s/ David G. Mills

Name: David G. Mills

Title: Managing Director and Head of Energy Execution

US BANK, NATIONAL ASSOCIATION, as a Lender

By: /s/ Michael P. Dickman

Name: Michael P. Dickman

Title: Vice President

REGIONS BANK, as a Lender

By: /s/ Randy Petersen

Name: Randy Petersen

Title: Senior Vice President

UNION BANK, N.A., as a Lender

By: /s/ Peter C. Thompson

Name: Peter C. Thompson

Title: Vice President

BRANCH BANKING AND TRUST CO, as a Lender

By: /s/ Stuart M. Jones

Name: Stuart M. Jones

Title: Senior Vice President

 $\mbox{\bf PNC BANK, NATIONAL ASSOCIATION}$, as a Lender and an L/C Issuer

By: /s/ Dale A. Stein

Name: Dale A. Stein

Title: Senior Vice President

FIFTH THIRD BANK, as a Lender

By: /s/ Mary J. Ramsey

Name: Mary J. Ramsey

Title: Vice President

COMERICA BANK, as a Lender

By: /s/ De Von J. Lang

Name: De Von J. Lang

Title: Assistant Vice President

WHITNEY NATIONAL BANK, as a Lender

By: /s/ Paul W. Cole

Name: Paul W. Cole

Title: Vice President

SUMITOMO MITSUI BANKING CORPORATION, as a Lender

By: /s/ William M. Ginn

Name: William M. Ginn

Title: Executive Officer

ALLIED IRISH BANKS, p.l.c., as a Lender

By: <u>/s/ Shreya Shah</u>
Name: Shreya Shah

Title: Vice President

By: /s/ Gregory J. Wiske

 Name:
 Gregory J. Wiske

 Title:
 Senior Vice President

Date: ___

FORM OF COMMITTED LOAN NOTICE

| 10 | . Dan | ik of Afficitea, IV.A., as | Administrative Agent | | | |
|-----|-------------------|--|--|--|---|------------------------|
| La | dies and | Gentlemen: | | | | |
| CC | m time MPAN | to time, the "Credit Y, a Delaware corporate | Agreement;" the terms defined thereintion, as the borrower thereunder (or, after | n being used her or the effectiveness | amended, restated, extended, supplemented or otherwise modification as therein defined), among BABCOCK & WILCOX IN ass of the Spinoff and the satisfaction of the other terms and condituders, the Administrative Agent, the Swing Line Lender and each | VESTMENT tions therein |
| | The | Borrower hereby requ | ests (select one): | | | |
| | A Co | mmitted Borrowing | | | A conversion of [Type] to [Type] A continuation of Eurocurrency Rate Loans | |
| | 1. | On , | (a Business Day). | | | |
| | 2. | In the amount of \$ | . [principal amount to be borrowed, con | verted or continue | ned] | |
| | 3. | Comprised of | [Type of Committed Borrowing reques | sted or to which a | an existing Committed Borrowing is to be converted] | |
| | 4. | For Eurocurrency R | tate Loans: with an Interest Period of | months. | | |
| | 5. | For conversions or | continuations of Eurocurrency Rate Loan | ns: Loan Number_ | | |
| | [The | e Committed Borrowin | ng requested herein complies with the pro | oviso to the first se | sentence of Section 2.01 of the Credit Agreement.] | |
| | | | | | BABCOCK & WILCOX INVESTMENT COMPANY | 72 |
| | | | | | Ву: | |
| 1 2 | Applica Update | able if requesting a Cor to reflect the New Bor | mmitted Borrowing. crower post Spinoff. | | | |
| | | | | | | |

| Name: | |
|--------|--|
| Title: | |

| FYH | ΠR | IT | R |
|-----|----|----|---|
| | | | |

Date: ______, ____

FORM OF SWING LINE LOAN NOTICE

| To: | | k of America, N.A., k of America, N.A., | | ving Line Lender dministrative Agent | | |
|------|---------------|--|-------|--|------------------------------|--|
| Ladi | es and | Gentlemen: | | | | |
| COM | time IPANY | to time, the "Creaty, a Delaware corporate | lit A | rtain Credit Agreement, dated as of May 3, 2010 (as amend greement;" the terms defined therein being used herein a n, as the borrower thereunder (or, after the effectiveness of the New Borrower, as the borrower thereunder), the Lenders, | s therein de e Spinoff ar | fined), among BABCOCK & WILCOX INVESTMENT d the satisfaction of the other terms and conditions therein |
| | The | undersigned hereby | requ | ests a Swing Line Borrowing: | | |
| | 1. | On , | | (a Business Day). | | |
| | 2. | In the amount of | \$ | | | |
| | The | Swing Line Borrow | ing r | equested herein complies with the requirements of the provise | to the first | sentence of Section 2.04(a) of the Credit Agreement. |
| | | | | | BABCO | OCK & WILCOX INVESTMENT COMPANY ³ |
| | | | | | By: | |
| | | | | | Name: | |
| | | | | | Title: | |
| | | | | | | |

³ Update to reflect the New Borrower post Spinoff.

FORM OF NOTE

FOR VALUE RECEIVED, the undersigned (the "Borrower") hereby promises to pay to or its registered assigns (the 'Lender''), in accordance with the provisions of the Credit Agreement (as hereinafter defined), the principal amount of each Loan (as defined in the Credit Agreement) from time to time made by the Lender to the Borrower under that certain Credit Agreement, dated as of May 3, 2010 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Credit Agreement;" the terms defined therein being used herein as therein defined), among BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation, as the borrower thereunder (or, after the effectiveness of the Spinoff and the satisfaction of the other terms and conditions therein relating to the substitution thereof, the New Borrower, as the borrower thereunder), the Lenders, the Administrative Agent, the Swing Line Lender and each L/C Issuer.

The Borrower promises to pay interest on the unpaid principal amount of each Loan from the date of such Loan until such principal amount is paid in full, at such interest rates and at such times as provided in the Credit Agreement. Except as otherwise provided in Section 2.04(f) of the Credit Agreement with respect to Swing Line Loans, all payments of principal and interest shall be made to the Administrative Agent for the account of the Lender in Dollars in immediately available funds at the Administrative Agent's Office. If any amount is not paid in full when due hereunder, such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Credit Agreement for such unpaid amount.

This Note is one of the Notes referred to in the Credit Agreement, is entitled to the benefits thereof and may be prepaid in whole or in part subject to the terms and conditions provided therein. This Note is also entitled to the benefits of the Guaranty and is secured by the Collateral. Upon the occurrence and continuation of one or more of the Events of Default specified in the Credit Agreement, all amounts then remaining unpaid on this Note shall become, or may be declared to be, immediately due and payable all as provided in the Credit Agreement. Loans made by the Lender shall be evidenced by one or more loan accounts or records maintained by the Lender in the ordinary course of business. The Lender may also attach schedules to this Note and endorse thereon the date, amount and maturity of its Loans and payments with respect thereto.

In accordance with the Credit Agreement, the Borrower, for itself, its successors and assigns, hereby waives diligence, presentment, protest and demand and notice of protest, demand, dishonor and non-payment of this Note.

| THIS NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH | THE LAWS OF THE STATE OF NEW YORK. |
|---|--|
| | BABCOCK & WILCOX INVESTMENT COMPANY ⁴ |
| | By: |
| | Name: |
| | Title: |

⁴ Update to reflect the New Borrower post Spinoff.

EXHIBIT D

FORM OF COMPLIANCE CERTIFICATE

| Financial | Statement | Date: | |
|-------------|-----------|-------|--|
| Tillaliciai | Statement | Date. | |

To: Bank of America, N.A., as Administrative Agent

Ladies and Gentlemen:

Reference is made to that certain Credit Agreement, dated as of May 3, 2010 (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "<u>Credit Agreement</u>;" the terms defined therein being used herein as therein defined), among BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation, as the borrower thereunder (or, after the effectiveness of the Spinoff and the satisfaction of the other terms and conditions therein relating to the substitution thereof, the New Borrower, as the borrower thereunder), the Lenders, the Administrative Agent, the Swing Line Lender and each L/C Issuer.

The undersigned Responsible Officer hereby certifies as of the date hereof that he/she is the [Chief Financial Officer/Treasurer] of the Borrower, and that, as such, he/she is authorized to execute and deliver this Compliance Certificate to the Administrative Agent on behalf of the Borrower in his or her capacity as a Responsible Officer of the Borrower and not in his or her individual capacity, and that:

1

[Use following paragraph 1 for fiscal year-end financial statements]

The Borrower has delivered the year-end consolidated audited financial statements required by Section 6.01(b) of the Credit Agreement for the Fiscal Year ended as of the above date, together with the report and opinion of Borrower's Accountant required by such section.

[Use following paragraph 1 for fiscal quarter-end financial statements]

The Borrower has delivered the consolidated unaudited financial statements required by Section 6.01(a) of the Credit Agreement for the Fiscal Quarter ended as of the above date. Such financial statements fairly present in all material respects the consolidated financial position of the Borrower and its Subsidiaries as at such date and the results of operations and cash flows of the Borrower and its Subsidiaries for the periods indicated in accordance with GAAP (subject only to normal year-end audit adjustments and the absence of footnotes).

2. The undersigned has reviewed and is familiar with the terms of the Credit Agreement and has made, or has caused to be made under his/her supervision, a reasonably detailed review of the transactions and consolidated condition (financial or otherwise) of the Borrower and its Subsidiaries during the accounting period covered by such financial statements.

3. A review of the activities of the Borrower and its Subsidiaries during such fiscal period has been made under the supervision of the undersigned with a view to determining whether during such fiscal period the Borrower and its Subsidiaries performed and observed all their respective Obligations under the Loan Documents, and

[select one:]

[to the best knowledge of the undersigned, during such fiscal period each of the Borrower and its Subsidiaries performed and observed each covenant and condition of the Loan Documents applicable to it, and no Default has occurred and is continuing.]

-or-

[to the best knowledge of the undersigned, during such fiscal period the following covenants or conditions have not been performed or observed and the following is a list of each Default and its nature and status:]

4. The financial covenant analyses and information set forth on Annex A attached hereto are true and accurate on and as of the date of this Compliance Certificate.

| IN WITNESS WHEREOF, the undersigned has executed this Compliance Certificate as of | , . |
|--|--|
| | BABCOCK & WILCOX INVESTMENT COMPANY ⁵ |
| | Ву: |
| | Name: |
| | Title: |

⁵ Update to reflect the New Borrower post Spinoff.

ASSIGNMENT AND ASSUMPTION

This Assignment and Assumption (this "<u>Assignment and Assumption</u>") is dated as of the Effective Date set forth below and is entered into by and between the Assignor identified in item 1 below (the "<u>Assignor</u>") and the Assignee identified in item 2 below (the "<u>Assignee</u>"). Capitalized terms used but not defined herein shall have the meanings given to them in the Credit Agreement identified below (the "<u>Credit Agreement</u>"), receipt of a copy of which is hereby acknowledged by the Assignee. The Standard Terms and Conditions set forth in Annex 1 attached hereto are hereby agreed to and incorporated herein by reference and made a part of this Assignment and Assumption as if set forth herein in full.

For an agreed consideration, the Assignor hereby irrevocably sells and assigns to the Assignee, and the Assignee hereby irrevocably purchases and assumes from the Assignor, subject to and in accordance with the Standard Terms and Conditions and the Credit Agreement, as of the Effective Date inserted by the Administrative Agent as contemplated below (a) all of the Assignor's rights and obligations in its capacity as a Lender under the Credit Agreement and any other documents or instruments delivered pursuant thereto to the extent related to the amount and percentage interest identified below of all of such outstanding rights and obligations of the Assignor under the respective facilities identified below (including, without limitation, the Letters of Credit and the Swing Line Loans included in such facilities) and (b) to the extent permitted to be assigned under applicable law, all claims, suits, causes of action and any other right of the Assignor (in its capacity as a Lender) against any Person, whether known or unknown, arising under or in connection with the Credit Agreement, any other documents or instruments delivered pursuant thereto or the loan transactions governed thereby or in any way based on or related to any of the foregoing, including, but not limited to, contract claims, tort claims, malpractice claims, statutory claims and all other claims at law or in equity related to the rights and obligations sold and assigned pursuant to clause (a) above (the rights and obligations sold and assigned by the Assignor to the Assignee pursuant to clauses (a) and (b) above being referred to herein collectively as the "Assigned Interest"). Each such sale and assignment is without recourse to the Assignor and, except as expressly provided in this Assignment and Assumption, without representation or warranty by the Assignor.

| 1. | Assignors: | <u></u> |
|----|-------------------------------|--|
| 2. | Assignee: [for each Assignee, | indicate [Affiliate][Approved Fund] of [identify Lender]] |
| 3. | Borrower: | Babcock and Wilcox Investment Company ⁶ |
| 4. | Administrative Age | ent: Bank of America, N.A., as the administrative agent under the Credit Agreement |

⁶ Update to reflect the New Borrower post Spinoff.

| 5. | | " the terms defined th | s of May 3, 2010 (as amended, restated, exte erein being used herein as therein defined), a | | |
|-----|---|------------------------|---|--|--------------------|
| 6. | Assigned Interests in th | e Commitment: | | | |
| | Aggregate Amount of Commitment/Loans for all Lenders ⁷ | | Amount of Commitment/ Loans Assigned | Percentage Assigned of <u>Commitment/Loans⁸</u> | CUSIP Number |
| [7. | Trade Date: |]9 | | | |
| | ctive Date: , FRANSFER IN THE REC | | TED BY ADMINISTRATIVE AGENT AND] | WHICH SHALL BE THE EFFECTIVE DA | ATE OF RECORDATION |
| | The terms set forth in the | nis Assignment and As | ssumption are hereby agreed to: | | |
| | | | | ASSIGNOR | |
| | | | | [NAME OF ASSIGNOR] | |
| | | | | Ву: | |
| | | | | Name: | |
| | | | | Title: | |
| | | | | <u>ASSIGNEE</u> | |
| | | | | By: | |

Name: Title:

Amounts in this column and in the column immediately to the right to be adjusted by the counterparties to take into account any payments or prepayments made between the Trade Date and the Effective Date.

Set forth, to at least 9 decimals, as a percentage of the Commitment/Loans of all Lenders thereunder.

⁹ To be completed if the Assignor and the Assignee intend that the minimum assignment amount is to be determined as of the Trade Date.

| [Consented to and] ¹⁰ Accepted: | |
|--|----------|
| BANK OF AMERICA, N.A., as Administrative Agent | |
| Ву: | <u></u> |
| Name: | |
| Title: | <u> </u> |
| Consented to: | |
| BANK OF AMERICA, N.A., as L/C Issuer and Swing Line Lender | |
| Ву: | <u> </u> |
| Name: | <u></u> |
| Title: | <u> </u> |
| [Consented to: | |
| BABCOCK & WILCOX INVESTMENT COMPANY ¹¹ | |
| Ву: | <u></u> |
| Name: | |
| Title: |] |
| 10 | |

To be added only if the consent of the Administrative Agent is required by the terms of the Credit Agreement. Update to reflect the New Borrower post Spinoff.

STANDARD TERMS AND CONDITIONS FOR ASSIGNMENT AND ASSUMPTION

1. Representations and Warranties.

- 1.1. Assignor. The Assignor (a) represents and warrants that (i) it is the legal and beneficial owner of the Assigned Interest, (ii) the Assigned Interest is free and clear of any lien, encumbrance or other adverse claim and (iii) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby; and (b) assumes no responsibility with respect to (i) any statements, warranties or representations made in or in connection with the Credit Agreement or any other Loan Document, (ii) the execution, legality, validity, enforceability, genuineness, sufficiency or value of the Loan Documents or any collateral thereunder, (iii) the financial condition of the Borrower, any of its Subsidiaries or Affiliates or any other Person obligated in respect of any Loan Document or (iv) the performance or observance by the Borrower, any of its Subsidiaries or Affiliates or any other Person of any of their respective obligations under any Loan Document.
- 1.2. <u>Assignee</u>. The Assignee (a) represents and warrants that (i) it has full power and authority, and has taken all action necessary, to execute and deliver this Assignment and Assumption and to consummate the transactions contemplated hereby and to become a Lender under the Credit Agreement, (ii) it meets all the requirements to be an assignee under <u>Section 10.06(b)</u> of the Credit Agreement (subject to such consents, if any, as may be required under <u>Section 10.06(b)</u> of the Credit Agreement), (iii) from and after the Effective Date, it shall be bound by the provisions of the Credit Agreement as a Lender thereunder and, to the extent of the Assigned Interest, shall have the obligations of a Lender thereunder, (iv) it is sophisticated with respect to decisions to acquire assets of the type represented by the Assigned Interest and either it, or the Person exercising discretion in making its decision to acquire the Assigned Interest, is experienced in acquiring assets of such type, (v) it has received a copy of the Credit Agreement, and has received or has been accorded the opportunity to receive copies of the most recent financial statements delivered pursuant to <u>Section 6.01</u> thereof, as applicable, and such other documents and information as it deems appropriate to make its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, (vi) it has, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Assignment and Assumption and to purchase the Assigned Interest, and (vii) if it is a Foreign Lender, attached hereto is any documentation required to be delivered by it pursuant to the terms of the Credit Agreement, duly completed and executed by the Assignee; and (b) agrees that (i) it will, independently and without reliance upon the Administrative Agent, the Assignor or any other Lender, a

- 2. <u>Payments</u>. From and after the Effective Date, the Administrative Agent shall make all payments in respect of the Assigned Interest (including payments of principal, interest, fees and other amounts) to the Assigner for amounts which have accrued to but excluding the Effective Date and to the Assignee for amounts which have accrued from and after the Effective Date.
- 3. <u>General Provisions</u>. This Assignment and Assumption shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and assigns. This Assignment and Assumption may be executed in any number of counterparts, which together shall constitute one instrument. Delivery of an executed counterpart of a signature page of this Assignment and Assumption by telecopy shall be effective as delivery of a manually executed counterpart of this Assignment and Assumption. This Assignment and Assumption shall be governed by, and construed in accordance with, the law of the State of New York.

FORM OF ADMINISTRATIVE QUESTIONNAIRE

EXHIBIT F

FORM OF GUARANTY

GUARANTY AGREEMENT

This GUARANTY AGREEMENT (as amended, restated, supplemented or otherwise modified from time to time, this "Guaranty"), dated as of May 3, 2010, is made by certain Subsidiaries of the Borrower, as identified on the signature pages hereto, and any Additional Guarantor who may become a party to this Guaranty (such signatories and the Additional Guarantors, collectively, the "Guarantors" and individually, a "Guarantor"), in favor of BANK OF AMERICA, N.A., as administrative agent (in such capacity, the "Administrative Agent") for the ratable benefit of the Administrative Agent, the Lenders, each L/C Issuer, the Hedge Banks, the Cash Management Banks, each co-agent or sub-agent appointed by the Administrative Agent from time to time pursuant to Section 9.05 of the Credit Agreement described below and the other Persons to whom the Guaranteed Obligations are owed (collectively, the "Guaranteed Parties").

Pursuant to that certain Credit Agreement dated as of the date hereof (as amended, restated, supplemented or otherwise modified from time to time, the <u>Credit Agreement</u>") among BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation, as the borrower thereunder or, after the effectiveness of the Spinoff and the satisfaction of the other terms and conditions in the Credit Agreement relating to the replacement thereof, the New Borrower as the borrower thereunder (the "<u>Borrower</u>"), the Lenders party thereto, the Administrative Agent, the Swing Line Lender and each L/C Issuer, the Lenders have agreed to make Credit Extensions to the Borrower upon the terms and subject to the conditions set forth therein.

Each Guarantor will materially benefit from the Credit Extensions made and to be made under the Credit Agreement.

Certain of the Guarantors are required to enter into this Guaranty pursuant to the terms of the Credit Agreement.

For good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged by the parties hereto, and to induce the Administrative Agent and the other Guaranteed Parties to enter into, and make their respective Credit Extensions and other accommodations under, the Loan Documents, the Secured Cash Management Agreements or the Secured Hedge Agreements, as applicable, the Guaranters hereby agree with the Administrative Agent, for the ratable benefit of the Guaranteed Parties, as follows:

- 1. <u>Defined Terms</u>. Capitalized terms used and not otherwise defined herein shall have the meanings herein that are assigned to such terms in the Credit Agreement. The following terms when used herein shall have the meanings set forth below:
- "Additional Guarantor" means each Person which hereafter becomes a Guarantor pursuant to Section 19 hereof and, if applicable, Section 6.22 of the Credit Agreement.

"Contribution Share" means, for any Guarantor in respect of any Excess Payment made by any other Guarantor, the ratio (expressed as a percentage) as of the date of such Excess Payment of (a) the amount by which the aggregate present fair salable value of all of its assets and properties exceeds the amount of all debts and liabilities of such Guarantor (including probable contingent, subordinated, unmatured, and unliquidated liabilities, but excluding the

obligations of such Guarantor hereunder) to (b) the amount by which the aggregate present fair salable value of all assets and other properties of the Guarantors other than the maker of such Excess Payment exceeds the amount of all of the debts and liabilities (including probable contingent, subordinated, unmatured, and unliquidated liabilities, but excluding the obligations of the Guarantors hereunder) of the Guarantors other than the maker of such Excess Payment; provided that for purposes of calculating the Contribution Shares of the Guarantors in respect of any Excess Payment, any Guarantor that became a Guarantor subsequent to the date of any such Excess Payment shall be deemed to have been a Guarantor on the date of such Excess Payment and the financial information for such Guarantor as of the date such Guarantor became a Guarantor shall be utilized for such Guarantor in connection with such Excess Payment.

"Excess Payment" means the amount paid by any Guarantor in excess of its Ratable Share of any Guaranteed Obligations.

"Guaranteed Obligations" has the meaning set forth in Section 2.

"Ratable Share" means, for any Guarantor in respect of any payment of Guaranteed Obligations, the ratio (expressed as a percentage) as of the date of such payment of Guaranteed Obligations of (a) the amount by which the aggregate present fair salable value of all of its assets and properties exceeds the amount of all debts and liabilities of such Guarantor (including probable contingent, subordinated, unmatured, and unliquidated liabilities, but excluding the obligations of such Guarantor hereunder) to (b) the amount by which the aggregate present fair salable value of all assets and other properties of all of the Guarantors exceeds the amount of all of the debts and liabilities (including probable contingent, subordinated, unmatured, and unliquidated liabilities, but excluding the obligations of the Guarantors hereunder) of the Guarantors; provided that for purposes of calculating the Ratable Shares of the Guarantors in respect of any payment of Guaranteed Obligations, any Guarantor that became a Guarantor subsequent to the date of any such payment shall be deemed to have been a Guarantor on the date of such payment and the financial information for such Guarantor as of the date such Guarantor became a Guarantor shall be utilized for such Guarantor in connection with such payment.

2. <u>Guaranty</u>. Each Guarantor hereby absolutely and unconditionally guarantees, as a guaranty of payment and performance and not merely as a guaranty of collection, prompt payment when due, whether at stated maturity, by required prepayment, upon acceleration, demand or otherwise, and at all times thereafter, of all Obligations, including any and all existing and future indebtedness and liabilities of every kind, nature and character, direct or indirect, absolute or contingent, liquidated or unliquidated, voluntary or involuntary and whether for principal, interest, premiums, fees indemnities, damages, costs, expenses or otherwise, of the Borrower to any Guaranteed Party arising under the Credit Agreement or any other Loan Document (including all renewals, extensions, amendments, restatements and other modifications thereof and all costs, attorneys' fees and expenses incurred by the Administrative Agent or any other Guaranteed Party in connection with the collection or enforcement thereof), and whether recovery upon such indebtedness and liabilities may be or hereafter become unenforceable or shall be an allowed or disallowed claim under any proceeding or case commenced by or against any Guaranter or the Borrower under any Debtor Relief Laws, and including interest that accrues after the commencement by or against the Borrower of any

proceeding under any Debtor Relief Laws (collectively, the "Guaranteed Obligations"). The books and records of the Administrative Agent and the books and records of each Guaranteed Party showing the amount of the Guaranteed Obligations shall be admissible in evidence in any action or proceeding, and shall be conclusive absent manifest error of the amount of the Credit Extensions and the interest and payments thereon. This Guaranty shall not be affected by the genuineness, validity, regularity or enforceability of the Guaranteed Obligations or any instrument or agreement evidencing any Guaranteed Obligations, or by the existence, validity, enforceability, perfection, non-perfection or extent of any collateral therefor, or by any fact or circumstance relating to the Guaranteed Obligations which might otherwise constitute a defense to the obligations of each Guarantor under this Guaranty, and such Guarantor hereby irrevocably waives any defenses it may now have or hereafter acquire in any way relating to any or all of the foregoing. Anything contained herein to the contrary notwithstanding, the obligations of each Guarantor hereunder at any time shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance as a fraudulent transfer or conveyance under Section 548 of the Bankruptcy Code (Title 11, United States Code) or any comparable provisions of any similar federal or state law.

- 3. No Setoff or Deductions; Taxes; Payments. Each Guarantor shall make all payments hereunder without setoff or counterclaim and free and clear of and without deduction for any taxes, levies, imposts, duties, charges, fees, deductions, withholdings, compulsory loans, restrictions or conditions of any nature now or hereafter imposed or levied by any jurisdiction or any political subdivision thereof or taxing or other authority therein unless such Guarantor is compelled by Requirement of Law to make such deduction or withholding and each Guarantor shall, jointly and severally, pay and indemnify each Guaranteed Party for Indemnified Taxes and Other Taxes to the extent the Borrower would be required to do so pursuant to Section 3.01 of the Credit Agreement. The obligations of each Guarantor under this paragraph shall survive the payment in full of the Guaranteed Obligations and termination of this Guaranty as to such Guarantor.
- 4. Rights of Guaranteed Parties. Each Guarantor consents and agrees that, to the extent permitted by the Credit Agreement and the other Loan Documents, as applicable, the Guaranteed Parties may, at any time and from time to time, without notice or demand, and without affecting the enforceability or continuing effectiveness hereof: (a) amend, extend, renew, compromise, discharge, accelerate or otherwise change the time for payment or the terms of the Guaranteed Obligations or any part thereof, (b) take, hold, exchange, enforce, waive, release, fail to perfect, sell, or otherwise dispose of any security for the payment of this Guaranty or any Guaranteed Obligations, (c) apply such security and direct the order or manner of sale thereof as the Guaranteed Parties in their sole discretion may determine and (d) release or substitute one or more of any endorsers or other guarantors of any of the Guaranteed Obligations. Without limiting the generality of the foregoing, such Guarantor consents to the taking of, or failure to take, any action which might in any manner or to any extent vary the risks of such Guarantor under this Guaranty or which, but for this provision, might operate as a discharge of such Guarantor.
- 5. <u>Certain Waivers</u>. Each Guarantor waives (a) any defense arising by reason of any disability or other defense of the Borrower or any other Guarantor, or the cessation from any cause whatsoever (including any act or omission of any Guaranteed Party) of the liability of the

Borrower other than payment and performance in full of the Guaranteed Obligations, (b) any defense based on any claim that such Guarantor's obligations exceed or are more burdensome than those of the Borrower, (c) the benefit of any statute of limitations affecting such Guarantor's liability hereunder, (d) any right to require any Guaranteed Party to proceed against the Borrower, proceed against or exhaust any security for the Guaranteed Obligations, or pursue any other remedy in any Guaranteed Party's power whatsoever, (e) any benefit of and any right to participate in any security now or hereafter held by any Guaranteed Party and (f) to the fullest extent permitted by law, any and all other defenses or benefits that may be derived from or afforded by Requirement of Law limiting the liability of or exonerating guarantors or sureties. Each Guarantor expressly waives all setoffs and counterclaims and all presentments, demands for payment or performance, notices of nonpayment or nonperformance, protests, notices of dishonor and all other notices or demands of any kind or nature whatsoever with respect to the Guaranteed Obligations, and all notices of acceptance of this Guaranty or of the existence, creation or incurrence of new or additional Guaranteed Obligations.

- 6. <u>Obligations Independent</u>. The obligations of each Guarantor hereunder are those of primary obligor, and not merely as surety, and are independent of the Guaranteed Obligations and the obligations of any other Guarantor, and a separate action may be brought against such Guarantor to enforce this Guaranty whether or not the Borrower or any other person or entity is joined as a party.
- 7. <u>Subrogation</u>. Each Guarantor shall not exercise any right of subrogation, contribution, indemnity, reimbursement or similar rights with respect to any payments it makes under this Guaranty until the termination of this Guaranty in accordance with its terms. If any amounts are paid to any Guarantor in violation of the foregoing limitation, then such amounts shall be held in trust for the benefit of the Guaranteed Parties and shall forthwith be paid to the Administrative Agent (for the benefit of itself and the other Guaranteed Parties) to reduce the amount of the Guaranteed Obligations, whether matured or unmatured.
- 8. Contribution. Subject to Section 7, each Guarantor hereby agrees with each other Guarantor that if any Guarantor shall make an Excess Payment, such Guarantor shall have a right of contribution from each other Guarantor in an amount equal to such other Guarantor's Contribution Share of such Excess Payment. The payment obligations of any Guarantor under this Section shall be subordinate and subject in right of payment to the Guaranteed Obligations until such time as the Guaranteed Obligations have been paid and performed in full, and no Guarantor shall exercise any right or remedy under this Section against any other Guarantor until constitute an asset in favor of the party entitled to such contribution. This Section shall not be deemed to affect any right of subrogation, indemnity, reimbursement or contribution that any Guarantor may have under Requirement of Law against the Borrower in respect of any payment of Guaranteed Obligations.
- 9. <u>Termination</u>; <u>Reinstatement</u>. This Guaranty is a continuing and irrevocable guarantee of all Guaranteed Obligations, now or hereafter existing, and shall remain in full force and effect with respect to each of the Guarantors not otherwise released from their obligations hereunder pursuant to Section 22(b), until the termination of this Guaranty in accordance with its

terms. Notwithstanding anything to the contrary, this Guaranty shall continue in full force and effect or be revived, as the case may be, if any payment by or on behalf of the Borrower or any Guarantor is made, or any Guaranteed Party exercises its right of setoff, in respect of the Guaranteed Obligations and such payment or the proceeds of such setoff or any part thereof is subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by any Guaranteed Party in its discretion) to be repaid to a trustee, receiver or any other party, in connection with any proceeding under any Debtor Relief Laws or otherwise, all as if such payment had not been made or such setoff had not occurred and whether or not any Guaranteed Party is in possession of or has released this Guaranty and regardless of any prior revocation, rescission, termination or reduction. The obligations of each Guarantor under this paragraph shall survive termination of this Guaranty.

- 10. <u>Subordination</u>. Each Guarantor hereby subordinates the payment of all obligations and indebtedness of the Borrower owing to such Guarantor, whether now existing or hereafter arising, including but not limited to any obligation of the Borrower to such Guarantor as subrogee of any Guaranteed Party or resulting from such Guarantor's performance under this Guaranty, to the payment in full in cash of all Guaranteed Obligations; <u>provided</u> that, unless an Event of Default has occurred and is continuing, the Borrower may make payments with respect to obligations and indebtedness of the Borrower owing to such Guarantor as permitted by the Credit Agreement and ordinary course payments pursuant to the Borrower's and its Subsidiaries' cash management system. If the Administrative Agent so requests when an Event of Default has occurred and is continuing, any such obligation or indebtedness of the Borrower to any Guarantor shall be enforced and performance received by such Guarantor as trustee for the Administrative Agent and the proceeds thereof, as well as any other amounts received by such Guarantor in violation of this Section, shall be paid over to the Administrative Agent on account of the Guaranteed Obligations, but without reducing or affecting in any manner the liability of such Guarantor under this Guaranty.
- 11. Stay of Acceleration. In the event that acceleration of the time for payment of any of the Guaranteed Obligations is stayed, in connection with any case commenced by or against the Borrower or any Guarantor under any Debtor Relief Laws, or otherwise, all such amounts shall nonetheless be payable by such Guarantor immediately upon demand by the Administrative Agent.
- 12. <u>Condition of Borrower</u>. Each Guarantor acknowledges and agrees that it has the sole responsibility for, and has adequate means of, obtaining from the Borrower and any other Guarantor such information concerning the financial condition, business and operations of the Borrower and any such other Guarantor as such Guarantor requires, and that no Guaranteed Party has a duty, and such Guarantor is not relying on any Guaranteed Party at any time, to disclose to such Guarantor any information relating to the business, operations or financial condition of the Borrower or any other Guarantor (such Guarantor waiving any duty on the part of any Guaranteed Parties to disclose such information and any defense relating to the failure to provide the same).
- 13. Representations and Warranties. Each Guarantor represents and warrants that each representation and warranty contained in Article V of the Credit Agreement to the extent such representation and warranty relates to such Guarantor is true and correct to the extent set

forth therein and except for such of those representations and warranties expressly stated to relate to a specific earlier date, in which case such representations and warranties shall be true and correct to the extent set forth therein as of such earlier date, as if made by such Guarantor herein; provided that each reference in each such representation and warranty to the Borrower's knowledge shall, for the purposes of this Section 13, be deemed to be a reference to such Guarantor's knowledge.

- 14. <u>Amendments; Etc.</u> None of the terms or provisions of this Guaranty may be waived, amended, supplemented or otherwise modified, nor any consent be given, except in accordance with <u>Section 10.01</u> of the Credit Agreement.
- 15. Notices. All notices and communications hereunder or under any Joinder Agreement as set forth in Section 19 shall be given to the addresses and otherwise made in accordance with Section 10.02 of the Credit Agreement; provided that notices and communications to the Guarantors shall be directed to the Guarantors at the address of the Borrower set forth in Section 10.02 of the Credit Agreement.
- 16. Expenses; Indemnification and Survival. Without limitation on any other obligations of each Guarantor or remedies of the Administrative Agent or any other Guaranteed Party under this Guaranty, each Guarantor shall, to the fullest extent permitted by Requirement of Law, indemnify, defend and save and hold harmless the Administrative Agent and each other Guaranteed Party from and against, and shall pay on demand, any and all damages, losses, liabilities and out-of-pocket expenses (including attorneys' fees and expenses) that may be suffered or incurred by the Administrative Agent or any other Guaranteed Party in connection with or as a result of any Guaranteed Obligations to be the legal, valid and binding obligations of the Borrower enforceable against the Borrower in accordance with their terms. The obligations of such Guarantor under this paragraph shall survive the payment in full of the Guaranteed Obligations and termination of this Guaranty in accordance with its terms.
- 17. Right of Setoff; Governing Law; Submission to Jurisdiction; Venue; WAIVER OF JURY TRIAL; Judgment Currency. THIS GUARANTY SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK. Without limiting the general applicability of the foregoing and the terms of the other Loan Documents to this Guaranty and the parties hereto, the terms of Sections 10.08, 10.14, 10.15 and 10.18 of the Credit Agreement are incorporated herein by reference, mutatis mutandis, with each reference to the "Borrower" therein (whether express or by reference to the Borrower as a "party" thereto) being a reference to the Guarantors, and the parties hereto agree to such terms.
- 18. <u>Counterparts</u>; <u>Electronic Execution</u>. This Guaranty may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Delivery of an executed counterpart of a signature page of this Guaranty by telecopy or other electronic imaging means shall be effective as delivery of a manually executed counterpart of this Guaranty.
- 19. Additional Guarantors. At any time after the date of this Guaranty, one or more additional Persons may become a party hereto by executing and delivering to the Administrative

Agent a Joinder Agreement pursuant to Section 6.22 of the Credit Agreement. Immediately upon such execution and delivery of such Joinder Agreement (and without any further action), each such additional Person will become a party to this Guaranty as a "Guarantor" and have all of the rights and obligations of a Guarantor hereunder and this Guaranty shall be deemed amended by such Joinder Agreement. Attached hereto as Exhibit A is a form of Joinder Agreement.

- 20. <u>Miscellaneous</u>. No failure by any Guaranteed Party to exercise, and no delay in exercising, any right, remedy or power hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy or power hereunder preclude any other or further exercise thereof or the exercise of any other right, power or remedy. The remedies herein provided are cumulative and not exclusive of any remedies provided by law or in equity. The unenforceability or invalidity of any provision of this Guaranty shall not affect the enforceability or validity of any other provision herein. Unless otherwise agreed by the Administrative Agent and each Guarantor in writing, this Guaranty is not intended to supersede or otherwise affect any other guaranty now or hereafter given by any Guarantor or any other guarantor for the benefit of the Guaranteed Parties or any term or provision thereof.
- 21. Acknowledgments. Each Guarantor hereby acknowledges that (a) it has been advised by counsel in the negotiation, execution and delivery of this Guaranty and the other Loan Documents to which it is a party and (b) it has received a copy of the Credit Agreement and the other Loan Documents and has reviewed and understands the same.

22. Termination; Release.

- (a) At such time as the Loans and the other Obligations (other than (i) contingent indemnification obligations and (ii) Obligations in respect of Secured Cash Management Agreements and Secured Hedge Agreements either (A) as to which arrangements satisfactory to the applicable Cash Management Bank or Hedge Bank shall have been made or (B) notice has not been received by the Administrative Agent from the applicable Cash Management Bank or Hedge Bank that such amounts are then due and payable) shall have been paid in full, the Commitments under the Credit Agreement have been terminated or expired and each Letter of Credit issued under the Credit Agreement shall be Cash Collateralized or no longer outstanding (other than Letters of Credit as to which other arrangements satisfactory to the Administrative Agent and the applicable L/C Issuer shall have been made), this Guaranty and all obligations (other than those expressly stated to survive such termination or as may be reinstated after such termination) of the Administrative Agent and each Guarantor hereunder shall terminate, all without delivery of any instrument or performance of any act by any party.
- (b) At the request and sole expense of the Borrower, a Guarantor shall be released from its obligations hereunder in the event that all the Stock and Stock Equivalents in such Guarantor shall be sold or otherwise disposed of in a transaction permitted by the Credit Agreement; provided that the Borrower shall have delivered to the Administrative Agent, at least three Business Days (or such lesser period permitted in writing by the Administrative Agent) prior to the date of the proposed release, a written request for such release identifying the relevant Guarantor and the terms of the relevant sale or other disposition in reasonable detail, including the price thereof and any expenses incurred in connection therewith, together with a certification by the Borrower stating that such transaction is in compliance with the Credit Agreement and the other Loan Documents.

[Signature Pages Follow]

| IN WITNESS WHEREOF, each of the parties hereto has caused this Guaranty to be duly executed as of the date first above written. | | |
|---|-----------------------------------|--|
| GUARANTORS: | [GUARANTORS] | |
| | By: Name: Title: Babcock & Wilcox | |
| | Guaranty Agreement | |

| | Acknowled | ged and accepted: | |
|------------------|-----------|---------------------------------|--|
| | | AMERICA, N.A., trative Agent | |
| | By: | | |
| | Name: | | |
| | Title: | | |
| Babcock & Wilcox | | | |

Babcock & Wilcox Guaranty Agreement

EXHIBIT A

FORM OF JOINDER AGREEMENT

| This JOINDER AGREEMENT (as amended, restated, supplemented or otherwise modified from time to time, this 'Agreement') dated as of, 20 |
|---|
| between, a(the "New Subsidiary"), and BANK OF AMERICA, N.A., in its capacity as Administrative Agent (the "Administrative Agent") |
| under that certain Credit Agreement, dated as of May 3, 2010 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), |
| among BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation, as the borrower thereunder, or, after the effectiveness of the Spinoff and the |
| satisfaction of the other terms and conditions in the Credit Agreement relating to the replacement thereof, the New Borrower as the borrower thereunder (the |
| "Borrower"), the Lenders party thereto, the Administrative Agent, the Swing Line Lender and each L/C Issuer (each as defined therein). All capitalized terms used and |
| not defined herein shall have the meanings given thereto in the Credit Agreement or the applicable Loan Document referred to herein. |

The Borrower desires to or is required by Section 6.22 of the Credit Agreement to cause the New Subsidiary to become a "Guarantor".

Accordingly, the New Subsidiary hereby agrees as follows with the Administrative Agent, for the benefit of the Guaranteed Parties:

- 1. The New Subsidiary hereby agrees that by execution of this Agreement it is a Guarantor (as defined in the Guaranty) under the Guaranty as if a signatory thereof on the Closing Date, and the New Subsidiary (a) shall comply with, and be subject to, and have the benefit of, all of the terms, conditions, covenants, agreements and obligations set forth in the Guaranty and (b) hereby makes each representation and warranty of a Guarantor, as set forth in the Guaranty. The New Subsidiary hereby agrees that (i) each reference to a "Guarantor" or the "Guarantors" in the Guaranty and the other Loan Documents shall include the New Subsidiary and (ii) each reference to the "Guaranty" as used therein shall mean the Guaranty as supplemented hereby and as otherwise amended, restated, supplemented or otherwise modified prior to the date hereof. Without limiting the generality of the foregoing terms of this paragraph 1, the New Subsidiary hereby, jointly and severally together with the other Guarantors, guarantees to the Administrative Agent, for the benefit of the Guaranteed Parties, as provided in the Guaranty, the prompt payment and performance of the Guaranteed Obligations in full when due (whether at stated maturity, as a mandatory prepayment, by acceleration or otherwise) strictly in accordance with the terms thereof.
- 2. [The New Subsidiary hereby agrees that by execution of this Agreement it is a Grantor (as defined in the Collateral Agreement) under the Collateral Agreement as if a signatory thereof on the Closing Date, and the New Subsidiary (a) shall comply with, and be subject to, and have the benefit of, all of the terms, conditions, covenants, agreements and obligations set forth in the Collateral Agreement and (b) hereby makes each representation and warranty of a Grantor, as set forth in the Collateral Agreement. The New Subsidiary hereby agrees that (i) each reference to a "Grantor" or the "Grantors" in the Collateral Agreement and the other Loan

Documents shall include the New Subsidiary, (ii) each reference to the "Collateral Agreement" as used therein shall mean the Collateral Agreement as supplemented hereby and as otherwise amended, restated, modified or supplemented as of the date hereof and (iii) each reference to a "Collateral" in the Collateral Agreement and the other Loan Documents shall include all Collateral (as defined in the Collateral Agreement) of the New Subsidiary (other than any of New Subsidiary's Excluded Assets). Without limiting the generality of the foregoing terms of this paragraph 2, the New Subsidiary hereby grants to the Administrative Agent, for the benefit of the Secured Parties, a continuing security interest in, and a right of setoff against, any and all right, title and interest, whether now or hereafter owned or acquired, of the New Subsidiary in and to the Collateral of the New Subsidiary.]

- 3. Attached hereto as <u>Annex A</u> are supplements to <u>Schedules 5.03</u> and <u>5.19(b)</u> of the Credit Agreement [and each of the Schedules to the Collateral Agreement to the extent such Schedules have or will change after the execution and delivery hereof]² (which supplements include, as of the date hereof, all information required to be provided therein with respect to the New Subsidiary).
- 4. All notices and communications to the New Subsidiary shall be given to the address of the Borrower set forth in, and otherwise made in accordance with, Section 10.02 of the Credit Agreement.
- 5. The New Subsidiary hereby waives acceptance by the Administrative Agent and the Guaranteed Parties of the guarantee by the New Subsidiary under the Guaranty upon the execution of this Agreement by the New Subsidiary.
- 6. The New Subsidiary hereby acknowledges that (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Loan Documents to which it is deemed a party and (b) it has received a copy of the Credit Agreement and the other Loan Documents and has reviewed and understands the same.
- 7. This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. Delivery of an executed counterpart of a signature page of this Agreement by telecopy or other electronic imaging means shall be effective as delivery of a manually executed counterpart of this Agreement.
 - 8. This Agreement shall be governed by and construed and interpreted in accordance with the laws of the State of New York.

[Signature Pages Follow]

- If the New Subsidiary is a Grantor under the Collateral Agreement
- If the New Subsidiary is a Grantor under the Collateral Agreement.

| as Guarantor |
|--------------|
| Ву: |
| Name: |
| Title: |
| |
| |
| |

IN WITNESS WHEREOF, each of the parties hereto has caused this Agreement to be duly executed as of the date first above written.

| Acknowledged and accepted: |
|--|
| BANK OF AMERICA, N.A., as Administrative Agent |
| Ву: |
| Name: |
| Title: |
| |

ANNEX A Supplemental Schedules

Schedules to Credit Agreement

[Schedules to Collateral Agreement]

EXHIBIT G

FORM OF COLLATERAL AGREEMENT

EXHIBIT H

FORM OF GLOBAL INTERCOMPANY NOTE

GLOBAL INTERCOMPANY NOTE

New York, New York May 3, 2010

FOR VALUE RECEIVED, Babcock & Wilcox Investment Company, a Delaware corporation, as Borrower, or, after the effectiveness of the Spinoff (all capitalized terms used herein and not defined herein shall have the meaning ascribed in the Credit Agreement referred to below) and the satisfaction of the other terms and conditions therein relating to the substitution thereof, the New Borrower, as the borrower thereunder (the "Borrower") and each of its Subsidiaries (each, together with the Borrower, a "Payor"), which is a party to this Global Intercompany Note (this "Note"), promises to pay to the order of each applicable Payee (as defined below), at such location as the Payee shall from time to time designate, the aggregate unpaid principal amount of all loans and extensions of credit ("Advances") made by each of the Borrower or its Subsidiaries (each, a "Payee").

Each Payor promises also to pay interest on the unpaid principal amount hereof at said office from the date hereof until paid at such rate per annum as shall be agreed upon from time to time by any Payor and any Payee.

Each Payor shall pay the unpaid principal balance of this Note and all accrued and unpaid interest thereon, if any, on demand of any Payee.

Whenever any payment on this Note shall be stated to be due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and such extension of time shall be included in the computation of the payment of interest, if any, on this Note.

Each Payor shall have the right to pay or prepay all or any part of the unpaid principal amount outstanding hereunder without premium or penalty at any time; provided, that (i) no Event of Default under clauses (a), (b) and (f) of Section 8.01 of the Credit Agreement shall have occurred and be continuing and (ii) interest, if any, shall be paid on the amount prepaid to and including the date of prepayment.

This Note is one of the promissory notes contemplated by Section 7.01(f) of that certain Credit Agreement, dated as of May 3, 2010 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), among the Borrower, the Lenders, the Swing Line Lender, the L/C Issuer and Bank of America, N.A., as administrative agent for the Lenders (the "Administrative Agent").

Each Payor shall be entitled to deem and treat each Payee, or such person who has been so identified by the transferor in writing to any Payor as the holder of this Note, as the owner and holder of this Note. This Note will, forthwith upon its issuance by any Payor, be endorsed to the order of and pledged to Administrative Agent, for the ratable benefit of the Secured Parties.

This Note shall be binding upon each Payor and its successors and assigns, and the terms and provisions of this Note shall inure to the benefit of each Payee and their respective successors and assigns, including subsequent holders hereof.

In addition to, and not in limitation of, the foregoing, each Payor agrees to pay all out-of-pocket costs and expenses, including reasonable attorneys' fees, incurred in connection with the collection and enforcement of this Note.

Each Payor waives diligence, presentment, protest, demand and notice of every kind and, to the full extent permitted by law, the right to plead any statute of limitations as a defense to any demand hereunder.

No delay on the part of any Payee in the exercise of any right or remedy shall operate as a waiver thereof, and no single or partial exercise by any Payee, of any right or remedy shall preclude any other or further exercise of any other right or remedy.

No amendment, modification or waiver of any term or provision of this Note, nor consent to any departure by any Payor herefrom, shall be effective unless the same shall be in writing and signed by the applicable Payee, and then such waiver, modification or consent shall be effective only in the specific instance and for the specific purpose for which given.

Nothing in this Note, expressed or implied, shall give or be construed to give any person, firm or corporation, other than the parties hereto, any legal or equitable right, remedy or claim under or in respect of this Note, or under any covenant, condition or provision herein contained; all its covenants, conditions and provisions being for the sole benefit of any Payor and any Payee of this Note.

This Note and any amendments, waivers, consents or supplements hereto or in connection herewith may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed an original, but all such counterparts together shall constitute but one and the same instrument; signature pages may be detached from multiple separate counterparts and attached to a single counterpart so that all signature pages are physically attached to the same document.

Subject to the provisions of paragraph 5 of this Note, and when the Payee is not a Loan Party, payment of the principal of, and interest on, this Note is expressly subordinated and subject in right of payment to the prior payment in full of all obligations of Payor (whether as borrower, guarantor or pledgor) under any of the Loan Documents (as such term is defined in the Credit Agreement). By acceptance of this Note, the holder agrees to be bound by the subordination provisions of this paragraph.

THIS NOTE AND THE RIGHTS AND OBLIGATIONS OF EACH PAYOR AND EACH PAYEE HEREUNDER SHALL BE GOVERNED BY, AND SHALL BE CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE INTERNAL LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAWS PRINCIPLES.

This Note amends and restates (but does not novate or extinguish) that certain Global Intercompany Note dated as of February 22, 2006, among certain of the Payees and Payors.

In case any provision in or obligation under this Note shall be invalid, illegal or unenforceable in any jurisdiction, the validity, legality and enforceability of the remaining provisions or obligations, or of such provision or obligation in any other jurisdiction, shall not in any way be affected or impaired thereby.

| BABCOCK & WILCOX INVESTMENT COMPANY |
|-------------------------------------|
| By: Name: Title: |
| [GUARANTORS AND SUBSIDIARIES] |
| By: Name: Title: |

IN WITNESS WHEREOF, each Payor has caused this Note to be duly executed and delivered by its officer thereunto duly authorized as of the date first above

written.

ENDORSEMENT

Each Payee hereby collaterally assigns and transfers to the order of BANK OF AMERICA, N.A., as Administrative Agent, the attached Global Intercompany Note, dated May 3, 2010, made by Babcock & Wilcox Investment Company and each of its Subsidiaries (as defined in the Credit Agreement) signatory thereto, to the order of each Payee signatory hereto, as security for the Obligations, as defined in, and pursuant to, the Pledge and Security Agreement dated as of May 3, 2010, executed by each of the undersigned and certain other parties.

| DADCOC | K & WILCOX INVESTMENT COMITANT |
|---------|--------------------------------|
| By: | |
| Name: | |
| Title: | |
| [GUARA] | NTORS AND SUBSIDIARIES] |
| By: | |
| Name: | |
| Title: | |

PLEDGE AND SECURITY AGREEMENT

made by

BABCOCK & WILCOX INVESTMENT COMPANY

and certain Subsidiaries of the Borrower

in favor of

BANK OF AMERICA, N.A., as Administrative Agent, for the ratable benefit of the Secured Parties

Dated as of May 3, 2010

TABLE OF CONTENTS

| | | Pag |
|--------------|--|-----|
| SECTION 1. | DEFINED TERMS | |
| 1.1. | Definitions | |
| 1.2. | Other Definitional Provisions | |
| SECTION 2. | GRANT OF SECURITY INTEREST; CONTINUING LIABILITY UNDER COLLATERAL | |
| 2.1. | Grant of Security Interest | |
| 2.2. | Continuing Liability Under Collateral | |
| 2.3. | Foreign Action | : |
| SECTION 3. | REPRESENTATIONS AND WARRANTIES | : |
| 3.1. | Representations in Credit Agreement | |
| 3.2. | Title; No Other Liens | 9 |
| 3.3. | Perfected First Priority Liens | 9 |
| 3.4. | Name; Jurisdiction of Organization, etc. | 9 |
| 3.5. | Inventory and Equipment | 10 |
| 3.6. | Types of Collateral | 10 |
| 3.7. | Investment Property | 10 |
| 3.8. | Receivables | 1 |
| 3.9. | Intellectual Property | 1 |
| 3.10. | Commercial Tort Claims | 1: |
| 3.11. | Contracts | 1. |
| SECTION 4. | COVENANTS | 1: |
| 4.1. | Covenants in Credit Agreement | 1: |
| 4.2. | Delivery and Control of Instruments, Chattel Paper, Negotiable Documents and Investment Property | 1: |
| 4.3. | Maintenance of Insurance | 10 |
| 4.4. | Payment of Obligations | 14 |
| 4.5. 4.6. | Maintenance of Perfected Security Interest; Further Documentation | 1. |
| 4.6. 4.7. | Changes in Locations, Name, Jurisdiction of Incorporation, etc. Notices | 1: |
| 4.7. | Investment Property | 10 |
| 4.9. | Receivables | 11 |
| 4.10. | Intellectual Property | 13 |
| 4.11. | Contracts | 20 |
| 4 12 | Commercial Tort Claims | 2 |

| | | Page |
|------------|---|------|
| SECTION 5. | REMEDIAL PROVISIONS | 21 |
| 5.1. | Certain Matters Relating to Receivables | 21 |
| 5.2. | Communications with Obligors; Grantors Remain Liable | 21 |
| 5.3. | Pledged Securities | 22 |
| 5.4. | Proceeds to be Turned Over To Administrative Agent | 23 |
| 5.5. | Application of Proceeds | 23 |
| 5.6. | Code and Other Remedies | 23 |
| 5.7. | Private Sales, etc. | 25 |
| 5.8. | Deficiency | 26 |
| 5.9. | BWXT Entities | 26 |
| SECTION 6. | THE ADMINISTRATIVE AGENT | 26 |
| 6.1. | Administrative Agent's Appointment as Attorney-in-Fact, etc. | 26 |
| 6.2. | Duty of Administrative Agent | 28 |
| 6.3. | Execution of Financing Statements | 28 |
| 6.4. | Authority of Administrative Agent | 29 |
| 6.5. | Appointment of Co-Administrative Agents | 29 |
| SECTION 7. | MISCELLANEOUS | 29 |
| 7.1. | Amendments in Writing | 29 |
| 7.2. | Notices | 29 |
| 7.3. | No Waiver by Course of Conduct; Cumulative Remedies | 29 |
| 7.4. | Enforcement Expenses; Indemnification | 29 |
| 7.5. | Successors and Assigns | 30 |
| 7.6. | Set-off; Governing Law; Submission to Jurisdiction; Venue; WAIVER OF JURY TRIAL | 30 |
| 7.7. | Counterparts | 30 |
| 7.8. | Severability | 30 |
| 7.9. | Section Headings | 31 |
| 7.10. | Integration | 31 |
| 7.11. | Acknowledgments | 31 |
| 7.12. | Additional Grantors | 31 |
| 7.13. | Releases; Termination of this Agreement | 31 |

Schedule 3.3 – Perfected First Priority Liens

Schedule 3.4 – Name; Jurisdiction of Organization, etc

 $Schedule \ 3.5-Inventory \ and \ Equipment$

Schedule 3.7 – Investment Property

Schedule 3.9 - Intellectual Property

 $Schedule\ 3.10-Commercial\ Tort\ Claims$

Exhibit A – Intellectual Property Notices

This PLEDGE AND SECURITY AGREEMENT, dated as of May 3, 2010, made by each of the signatories hereto (together with any other grantor that may become a party hereto as provided herein, the "Grantors"), in favor of BANK OF AMERICA, N.A., as administrative agent (in such capacity and together with its successors in such capacity, the "Administrative Agent") for the benefit of the Secured Parties in connection with that certain Credit Agreement dated as of the date hereof (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement") among BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation, as the borrower thereunder (or, after the effectiveness of the Spinoff and the satisfaction of the other terms and conditions therein relating to the replacement thereof, the New Borrower, as the borrower thereunder), the Lenders, the Administrative Agent, the Swing Line Lender and the L/C Issuers.

Pursuant to the Credit Agreement, the Lenders have severally agreed to make Credit Extensions to the Borrower.

This Agreement is required by the terms of the Credit Agreement.

In consideration of the mutual covenants and agreements contained herein and in the other Loan Documents, the parties hereto covenant and agree as follows:

SECTION 1. DEFINED TERMS

1.1. Definitions.

(a) Unless otherwise defined herein, terms defined in the Credit Agreement and used herein shall have the meanings given to them in the Credit Agreement, and the following terms are used herein as defined in the New York UCC: Accounts, Account Debtor, As-Extracted Collateral, Certificated Security, Chattel Paper, Commercial Tort Claim, Commodity Account, Commodity Contract, Commodity Intermediary, Consumer Goods, Deposit Account, Documents, Electronic Chattel Paper, Equipment, Farm Products, Financial Asset, Fixtures, General Intangibles, Goods (as defined in Article 9 of the New York UCC), Instruments, Inventory, Letter-of-Credit Rights, Manufactured Homes, Money, Payment Intangibles, Securities Account, Securities Intermediary, Security, Security Entitlement, Supporting Obligations, Tangible Chattel Paper and Uncertificated Security.

(b) The following terms shall have the following meanings:

"Administrative Agent" shall have the meaning assigned to such term in the preamble.

"After-Acquired Intellectual Property" shall have the meaning assigned to such term in Section 4.10(i).

"Agreement" shall mean this Pledge and Security Agreement, as the same may be amended, restated, supplemented or otherwise modified from time to time.

"Collateral" shall have the meaning assigned to such term in Section 2.1.

"Collateral Account" shall mean any collateral account established by the Administrative Agent as provided in Sections 5.1 or 5.4.

"Collateral Account Funds" shall mean, collectively, the following: all funds (including all trust monies) and investments (including all cash equivalents) credited to, or purchased with funds from, any Collateral Account and all certificates and instruments from time to time representing or evidencing such investments; all Money, notes, certificates of deposit, checks and other instruments from time to time hereafter delivered to or otherwise possessed by the Administrative Agent for or on behalf of any Grantor in substitution for, or in addition to, any or all of the Collateral; and all interest, dividends, cash, instruments and other property from time to time received in, receivable or otherwise distributed to the Collateral Account in respect of or in exchange for any or all of the items constituting Collateral.

"Contracts" shall mean all contracts and agreements between any Grantor and any other Person (in each case, whether written or oral, or third party or intercompany) as the same may be amended, assigned, extended, restated, supplemented, replaced or otherwise modified from time to time including (a) all rights of any Grantor to receive moneys due and to become due to it thereunder or in connection therewith, (b) all rights of any Grantor to receive proceeds of any insurance, indemnity, warranty or guaranty with respect thereto, (c) all rights of any Grantor to damages arising thereunder and (d) all rights of any Grantor to terminate and to perform and compel performance of, such Contracts and to exercise all remedies thereunder.

"Copyright Licenses" shall mean any agreement, whether written or oral, naming any Grantor as licensor or licensee (including those listed in schedule 3.9(a) (as such schedule may be amended or supplemented from time to time)), granting any right in, to or under any Copyright, including the grant of rights to publicly perform, display, copy, prepare derivative works or distribute under any Copyright. This term shall exclude implied licenses and any rights obtained or granted under a copyright pursuant to the doctrines of first sale or estoppel.

"Copyrights" shall mean (a) all copyrights arising under the laws of the United States, any other country, or union of countries, or any political subdivision of any of the foregoing, whether registered or unregistered and whether published or unpublished (including those listed in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time)), all registrations and recordings thereof, and all applications in connection therewith and rights corresponding thereto throughout the world, including all registrations, recordings and applications in the United States Copyright Office, and all Mask Works (as defined in 17 USC 901), (b) the right to, and to obtain, all extensions and renewals thereof, and the right to sue for past, present and future infringements of any of the foregoing, (c) all proceeds of the foregoing, including license, royalties, income, payments, claims, damages, and proceeds of suit and (d) all other rights of any kind whatsoever accruing thereunder or pertaining thereof

"Credit Agreement" shall have the meaning assigned to such term in the preamble.

"Excluded Assets" shall mean:

(a) any lease, license, contract, property right or agreement to which any Grantor is a party or any of its rights or interests thereunder if, and only for so long as, the grant

of a security interest hereunder shall constitute or result in a breach, termination or default under any such lease, license, contract, property right or agreement (other than to the extent that any such term would be rendered ineffective pursuant to Sections 9-406, 9-407, 9-408 or 9-409 of the UCC of any relevant jurisdiction or any other applicable law or principles of equity); <u>provided</u>, <u>however</u>, that such security interest shall attach immediately to any portion of such lease, license, contract, property rights or agreement that does not result in any of the consequences specified above;

- (b) all Security Entitlements, Securities Accounts, Deposit Accounts, Financial Assets, Letter-of-Credit Rights (other than Letter-of-Credit Rights constituting a Supporting Obligation), Commodity Contracts and Commodity Accounts to which any Grantor has any right, title or interest;
 - (c) the Excluded Stock; and
 - (d) all cars, trucks, trailers and other vehicles covered by a certificate of title under the laws of any state to which any Grantor has any right, title or interest.

"Excluded Stock" shall mean:

- (e) the Voting Stock of any Foreign Subsidiary in excess of 65% of the outstanding Voting Stock of such Foreign Subsidiary;
- (f) the Stock and Stock Equivalents of any BWXT Entity;
- (g) the Stock and Stock Equivalents of any Captive Insurance Subsidiary;
- (h) the Stock and Stock Equivalents of any Joint Venture to the extent that the Constituent Documents of such Joint Venture prohibit such a security interest to be granted to the Administrative Agent; and
- (i) the Stock and Stock Equivalents of (i) any Subsidiary that is not a Loan Party or (ii) any Joint Venture, to the extent that such Subsidiary or Joint Venture has incurred Non-Recourse Indebtedness the terms of which either (A) require security interests in such Stock and Stock Equivalents to be granted to secure such Non-Recourse Indebtedness or (B) prohibit such a security interest to be granted to the Administrative Agent.

"Grantors" shall have the meaning assigned to such term in the preamble.

"Insurance" shall mean all insurance policies covering any or all of the Collateral (regardless of whether the Administrative Agent is the loss payee thereof).

"Intellectual Property" shall mean the collective reference to all intellectual property rights whether arising under United States, multinational or foreign laws or otherwise, including the Copyrights, the Copyright Licenses, the Patents, the Patent Licenses, the Trademarks, the Trademark Licenses, the Trade Secrets and the Trade Secret Licenses.

"Intellectual Property Security Agreement" shall mean a Notice of Grant of Security Interest in substantially the form of Exhibit A or such other form as may be approved by the Administrative Agent and the applicable Grantor.

"Intercompany Note" shall mean any promissory note evidencing Indebtedness permitted to be incurred pursuant to Section 7.01(f) of the Credit Agreement with respect to any outstanding intercompany obligations and advances owed by or to a Loan Party.

"Investment Property" shall mean the collective reference to (a) all "investment property" as such term is defined in Section 9-102(a)(49) of the New York UCC (other than any Excluded Stock), including all Certificated Securities and Uncertificated Securities and (b) whether or not otherwise constituting "investment property," all Pledged Notes and all Pledged Equity Interests.

"Licensed Intellectual Property" shall have the meaning assigned to such term in Section 3.9(a).

"Material Intellectual Property" shall have the meaning assigned to such term in Section 3.9(b).

"New York UCC" shall mean the Uniform Commercial Code as from time to time in effect in the State of New York.

"Owned Intellectual Property" shall have the meaning assigned to such term in Section 3.9(a).

"Patent License" shall mean all agreements, whether written or oral, providing for the grant by or to any Grantor of any right to make, use, import, offer for sale, or sell any invention covered in whole or in part by a Patent, including any of the foregoing listed in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time). This term shall exclude implied licenses and any rights obtained or granted under a patent pursuant to the doctrines of exhaustion or estoppel.

"Patents" shall mean (a) all United States patents, patents issued by any other country, union of countries or any political subdivision of any of the foregoing, and all reissues and extensions thereof, including any of the foregoing listed in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time), (b) all patent applications pending in the United States or any other country or union of countries or any political subdivision of any of the foregoing and all divisions, continuations and continuations-in-part thereof, including any of the foregoing listed in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time), (c) all rights to, and to obtain, any reissues or extensions of the foregoing and (d) all proceeds of the foregoing, including licenses, royalties, income, payments, claims, damages and proceeds of suit.

"<u>Pledged Equity Interests</u>" means the Pledged Interests, including the Stock and Stock Equivalents of the Subsidiaries owned by such Grantor as set forth on <u>Schedule 3.7(a)</u> (as such schedule may be amended or supplemented from time to time), in each case together with the certificates (or other agreements or instruments), if any, representing such shares, and all options

and other rights, contractual or otherwise, with respect thereto, including, but not limited to, the following:

- (j) all Stock and Stock Equivalents representing a dividend thereon, or representing a distribution or return of capital upon or in respect thereof, or resulting from a stock split, reclassification or other exchange therefor, and any subscriptions, warrants, rights or options issued to the holder thereof, or otherwise in respect thereof; and
- (k) in the event of any consolidation or merger involving the issuer thereof and in which such issuer is not the surviving Person, all shares of each class of the Stock and Stock Equivalents of the successor Person formed by or resulting from such consolidation or merger, to the extent that such successor Person is a direct Subsidiary of a Grantor.
- "Pledged LLC/Partnership Interests" means, with respect to any Grantor, the entire partnership, membership interest or limited liability company interest, as applicable, of such Grantor in each partnership, limited partnership or limited liability company owned thereby, including, without limitation, such Grantor's capital account, its interest as a partner or member, as applicable, in the net cash flow, net profit and net loss, and items of income, gain, loss, deduction and credit of any such partnership, limited partnership or limited liability company, as applicable, such Grantor's interest in all distributions made or to be made by any such partnership, limited partnership or limited liability company, as applicable, of such Grantor as a partner or member, as applicable, of any such partnership, limited partnership or limited liability company, as applicable, whether set forth in the partnership agreement or membership agreement, as applicable, of such partnership, limited partnership or limited liability company, as applicable, by separate agreement or otherwise.
- "<u>Pledged Notes</u>" shall mean all promissory notes now owned or hereafter acquired by any Grantor, including those listed on <u>Schedule 3.7(b)</u> (as such schedule may be amended or supplemented from time to time) and all Intercompany Notes at any time issued to or held by any Grantor (other than (a) promissory notes in an aggregate principal amount not to exceed \$1,000,000 at any time outstanding issued in connection with extensions of trade credit by any Grantor in the ordinary course of business and (b) promissory notes constituting Cash Equivalents that are held by any Grantor).
 - "Pledged Securities" shall mean the collective reference to the Pledged Notes and the Pledged Equity Interests.
- "Proceeds" shall mean all "proceeds" as such term is defined in Section 9-102(a)(64) of the New York UCC and, in any event, shall include all dividends or other income from the Investment Property, collections thereon or distributions or payments with respect thereto.
- "Receivable" shall mean all Accounts and any other right to payment for goods or other property sold, leased, licensed or otherwise disposed of or for services rendered, whether or not such right is evidenced by an Instrument or Chattel Paper or classified as a Payment Intangible and whether or not it has been earned by performance. References herein to Receivables shall include any Supporting Obligation or collateral securing such Receivable.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Trademark License" shall mean any agreement, whether written or oral, providing for the grant by or to any Grantor of any right in, to or under any Trademark, including any of the foregoing referred to in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time). This term shall exclude implied licenses and any rights obtained or granted under a trademark pursuant to the doctrines of first sale or estoppel.

"Trademarks" shall mean (a) all trademarks, trade names, corporate names, company names, business names, fictitious business names, trade styles, service marks, logos, designs and other source or business identifiers, and all goodwill associated therewith, now existing or hereafter adopted or acquired, all registrations and recordings thereof, and all applications in connection therewith, whether in the United States Patent and Trademark Office or in any similar office or agency of the United States, any State thereof or any other country, union of countries, or any political subdivision of any of the foregoing, or otherwise, and all common-law rights related thereto, including any of the foregoing listed in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time), (b) the right to, and to obtain, all renewals thereof, (c) the goodwill of the business symbolized by the foregoing and (d) the right to sue for past, present and future infringements or dilution of any of the foregoing or for any injury to goodwill, and all proceeds of the foregoing, including royalties, income, payments, claims, damages and proceeds of suit.

"Trade Secret License" shall mean any agreement, whether written or oral, providing for the grant by or to any Grantor of any right in, to or under any Trade Secret, including any of the foregoing listed in Schedule 3.9(a) (as such schedule may be amended or supplemented from time to time). This term shall exclude implied licenses and any rights obtained or granted under a trade secret pursuant to the doctrine of estoppel.

"Trade Secrets" shall mean (a) all trade secrets and all other confidential or proprietary information and know how whether or not reduced to a writing or other tangible form, (b) all documents and things embodying, incorporating or describing such Trade Secrets, and (c) the right to sue for past, present and future misappropriations of any Trade Secret and all proceeds of the foregoing, including royalties, income, payments, claims, damages and proceeds of suit.

1.2. Other Definitional Provisions. Without limiting the general applicability of the terms of the other Loan Documents to this Agreement and the parties hereto, the terms of Sections 1.02 of the Credit Agreement are incorporated herein by reference, mutatis mutandis, and the parties hereto agree to such terms.

SECTION 2. GRANT OF SECURITY INTEREST; CONTINUING LIABILITY UNDER COLLATERAL

2.1. Grant of Security Interest. Each Grantor hereby grants and pledges to the Administrative Agent, for the ratable benefit of the Secured Parties, a security interest in all of such Grantor's right, title and interest in and to the following property, in each case, wherever located and whether now owned or at any time hereafter acquired by such Grantor or in which such Grantor now has or at any time in the future may acquire any right, title or interest

| vely, the "Collateral"), as collateral security for the prompt and complete payment and performance when due (whether at the stated maturity, by acceleration of all Obligations: |
|---|
| (a) all Accounts; |
| (b) all As-Extracted Collateral; |
| (c) all Chattel Paper; |
| (d) all Collateral Accounts and all Collateral Account Funds; |

(e) all Commercial Tort Claims from time to time specifically described on Schedule 3.10;

- (f) all Contracts;
- (g) all Documents;
- (h) all Equipment;
- (i) all Fixtures;
- (j) all General Intangibles;
- (k) all Goods;
- (l) all Instruments;
- (m) all Insurance;
- (n) all Intellectual Property;
- (o) all Inventory;
- (p) all Investment Property;
- (q) all books, records, ledger cards, files, correspondence, customer lists, blueprints, technical specifications, manuals, computer software, computer printouts, tapes, disks and other electronic storage media and related data processing software and similar items that at any time pertain to or evidence or contain information relating to any of the Collateral or are otherwise necessary or helpful in the collection thereof or realization thereupon; and
- (r) to the extent not otherwise included, all Proceeds, goodwill, products, accessions, rents and profits of any and all of the foregoing and all collateral security, Supporting Obligations and guarantees given by any Person with respect to any of the foregoing;

provided that, notwithstanding any other provision set forth in this Section 2.1, this Agreement shall not, at any time, constitute a grant of a security interest in any property that is,

at such time, an Excluded Asset, and the term "Collateral" and each of the defined terms incorporated therein shall exclude the Excluded Assets.

- 2.2. Continuing Liability Under Collateral. Notwithstanding anything herein to the contrary, (a) each Grantor shall remain liable for all obligations under and in respect of the Collateral and nothing contained herein is intended or shall be a delegation of duties to the Administrative Agent or any other Secured Party, (b) each Grantor shall remain liable under and each of the agreements included in the Collateral, including any Receivables, any Contracts and any agreements relating to Pledged LLC/Partnership Interests, to perform all of the obligations undertaken by it thereunder all in accordance with and pursuant to the terms and provisions thereof and neither the Administrative Agent nor any other Secured Party shall have any obligation or liability under any of such agreements by reason of or arising out of this Agreement or any other document related hereto nor shall the Administrative Agent nor any other Secured Party have any obligation to make any inquiry as to the nature or sufficiency of any payment received by it or have any obligation to take any action to collect or enforce any rights under any agreement included in the Collateral, including any agreements relating to any Receivables, any Contracts or any agreements relating to Pledged LLC/Partnership Interests and (c) the exercise by the Administrative Agent of any of its rights hereunder shall not release any Grantor from any of its duties or obligations under the contracts and agreements included in the Collateral, including any agreements relating to any Receivables, any Contracts and any agreements relating to Pledged LLC/Partnership Interests.
- 2.3. Foreign Action. Notwithstanding anything to the contrary herein, to the extent any Collateral is located in any jurisdiction outside the United States, or the creation or perfection of a lien in any Collateral requires any action or documentation outside the United States, no such action or documentation outside the United States shall be required with respect to such Collateral.

SECTION 3. REPRESENTATIONS AND WARRANTIES

To induce the Administrative Agent and the Lenders to enter into the Credit Agreement and make their respective Credit Extensions, each Grantor hereby represents and warrants to the Secured Parties that:

3.1. Representations in Credit Agreement.

In the case of each Grantor, the representations and warranties set forth in Article V of the Credit Agreement as they relate to such Grantor or to the Loan Documents to which such Grantor is a party, each of which is hereby incorporated herein by reference, are true and correct, in all material respects, except for representations and warranties expressly stated to relate to a specific earlier date, in which case such representations and warranties shall be true and correct in all material respects as of such earlier date, and the Secured Parties shall be entitled to rely on each of them as if they were fully set forth herein, provided that each reference in each such representation and warranty to the Borrower's knowledge shall, for the purposes of this Section 3.1, be deemed to be a reference to such Grantor's knowledge.

- 3.2. <u>Title: No Other Liens.</u> Such Grantor owns or licenses or otherwise has the right to use each item of the Collateral free and clear of any and all Liens, including Liens arising as a result of such Grantor becoming bound (as a result of merger or otherwise) as grantor under a security agreement entered into by another Person, except for Liens expressly permitted by <u>Section 7.02</u> of the Credit Agreement. No effective financing statement, mortgage or other public notice indicating the existence of a Lien with respect to all or any part of the Collateral is on file or of record in any public office, except such as have been filed in favor of the Administrative Agent, for the ratable benefit of the Secured Parties, pursuant to this Agreement or as are expressly permitted by <u>Section 7.02</u> of the Credit Agreement.
- 3.3. Perfected First Priority Liens. The security interests granted pursuant to this Agreement (a) upon completion of the filings and other actions specified on Schedule 3.3 (all of which, in the case of all filings and other documents referred to on said Schedule, have been delivered to the Administrative Agent in duly completed and duly executed form, as applicable, and may be filed by the Administrative Agent at any time) and payment of all filing fees, will constitute valid fully perfected security interests in all of the Collateral in favor of the Administrative Agent, for the ratable benefit of the Secured Parties, as collateral security for such Grantor's Obligations, enforceable in accordance with the terms hereof, to the extent such security interest in such Collateral can be perfected by (i) the filing of a financing statement under the Uniform Commercial Code of any jurisdiction, (ii) the filing with the United States Patent and Trademark Office or the United States Copyright Office of an Intellectual Property Security Agreement, or (iii) the possession of such Collateral, and (b) are prior to all other Liens on the Collateral, except for Liens expressly permitted by Section 7.02 of the Credit Agreement. Without limiting the foregoing, each Grantor has taken all actions necessary or desirable under all Requirements of Law of the United States and of any state, territory or possession thereof, including those specified in Section 4.2 to (i) establish the Administrative Agent's "control" (within the meanings of Sections 8-106 and 9-106 of the New York UCC) over any portion of the Investment Property constituting Certificated Securities, Uncertificated Securities (each as defined in the New York UCC), other than any such Investment Property issued by a Foreign Subsidiary to the extent establishing "control" over such Investment Property would require actions under the Requirements of Law of a jurisdiction other than the United States or any state, territory or possession thereof, (ii) establish the Admi
- 3.4. Name; Jurisdiction of Organization, etc. On the date hereof, such Grantor's exact legal name (as indicated on the public record of such Grantor's jurisdiction of formation or organization), jurisdiction of organization, organizational identification number, if any, United States taxpayer identification number, if any, and the location of such Grantor's chief executive office or sole place of business are specified on Schedule 3.4. Each Grantor is organized solely under the law of the jurisdiction so specified and has not filed any certificates of domestication, transfer or continuance in any other jurisdiction. Except as otherwise indicated on Schedule 3.4, the jurisdiction of each such Grantor's organization of formation is required to maintain a public record showing the Grantor to have been organized or formed. Except as specified on Schedule 3.4, as of the Closing Date (or the date of any applicable Joinder Agreement hereto in the case of

an Additional Grantor) no such Grantor has changed its name, jurisdiction of organization, chief executive office or sole place of business or its corporate structure in any way (e.g., by merger, consolidation, change in corporate form or otherwise) within the past five years and has not within the last five years become bound (whether as a result of merger or otherwise) as a grantor under a security agreement entered into by another Person, which has not heretofore been terminated.

3.5. Inventory and Equipment.

- (a) On the date hereof, the material Inventory, Fixtures and Equipment (other than mobile goods, Inventory in transit, and Inventory, Fixtures and Equipment located outside the United States of America) that is included in the Collateral are kept at the locations listed on <u>Schedule 3.5</u>.
- (b) Any Inventory now or hereafter produced by any Grantor included in the Collateral have been and will be produced in compliance in all material respects with the requirements of all applicable laws and regulations, including the Fair Labor Standards Act, as amended.
- (c) No material portion of the Inventory, Fixtures or Equipment that is included in the Collateral is in the possession of an issuer of a negotiable document (as defined in Section 7-104 of the New York UCC) therefor or is otherwise in the possession of any bailee or warehouseman.
- 3.6. Types of Collateral. None of the Collateral constitutes, or is the Proceeds of (a) Farm Products, (b) As-Extracted Collateral, (c) Consumer Goods, (d) Manufactured Homes, (e) standing timber, or (f) as of the Closing Date, aircraft, airframe, aircraft engine, aircraft lease or any other related property.

3.7. Investment Property.

- (a) Schedule 3.7(a) hereto sets forth under the heading "Pledged Equity Interests" all of the Pledged Equity Interests as of the Closing Date, and such Pledged Equity Interests constitute the percentage of issued and outstanding shares of stock, percentage of membership interests, percentage of partnership interests or percentage of beneficial interest of the respective issuers thereof indicated on such schedule. Schedule 3.7(b) sets forth under the heading "Pledged Notes" all of the Pledged Notes owned by any Grantor as of the Closing Date, and all of such Pledged Notes have been duly authorized, authenticated or issued, and delivered and are the legal, valid and binding obligation of the issuers thereof enforceable in accordance with their terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principals of equity, regardless of whether considered in a proceeding in equity or at law, and constitute all of the issued and outstanding indebtedness evidenced by an instrument or certificated security of the respective issuers thereof owing to such Grantor.
- (b) The shares of Pledged Equity Interests pledged by such Grantor hereunder constitute all of the issued and outstanding shares of all classes of Stock and Stock Equivalents owned by such Grantor in each issuer thereof (other than Excluded Stock).

- (c) The Pledged Equity Interests have been duly and validly issued and, except as set forth on Schedule 3.7(a) hereto, are fully paid and nonassessable (to the extent applicable).
- (d) Such Grantor is the record and beneficial owner of, and has good and marketable title to, the Investment Property pledged by it hereunder, free of any and all Liens or options in favor of, or claims of, any other Person, except Liens expressly permitted by Section 7.02 of the Credit Agreement, and there are no outstanding warrants, options or other rights to purchase, or shareholder, voting trust or similar agreements outstanding with respect to, or property that is convertible into, or that requires the issuance or sale of, any Pledged Equity Interests.

3.8. Receivables.

- (a) No amount payable to such Grantor under or in connection with any Receivable in excess of \$1,000,000 that is included in the Collateral is evidenced by any Instrument or Tangible Chattel Paper which has not been delivered to the Administrative Agent or constitutes Electronic Chattel Paper that has not been subjected to the control (within the meaning of Section 9-105 of the New York UCC) of the Administrative Agent.
- (b) Each Receivable that is included in the Collateral (i) is and will be the legal, valid and binding obligation of the Account Debtor in respect thereof, representing an unsatisfied obligation of such Account Debtor, (ii) is and will be enforceable in accordance with its terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium or other laws affecting creditors' rights generally and subject to general principals of equity, regardless of whether considered in a proceeding in equity or at law, (iii) is not and will not be subject to any setoffs, defenses, taxes or counterclaims (except with respect to refunds, returns and allowances in the ordinary course of business) and (iv) is and will be in compliance with all applicable laws and regulations, except where the failure to comply with this Section 3.8(b) with respect to each Receivable would not reasonably be expected to have a Material Adverse Effect.

3.9. Intellectual Property.

(a) Schedule 3.9(a) lists all Copyrights, Patents, and Trademarks which are registered with the U.S. Patent and Trademark Office or the U.S. Copyright Office or are the subject of an application for registration with any such Governmental Authority, in each case which is owned by such Grantor in its own name on the date hereof (collectively, the "Owned Intellectual Property"). Except as set forth in Schedule 3.9(a), such Grantor is the exclusive owner of the entire and unencumbered right, title and interest in and to all material Owned Intellectual Property and is otherwise entitled to grant to others the right to use (and, where applicable, itself use) all such material Owned Intellectual Property. Such Grantor has a valid and enforceable right to use all material Intellectual Property used by, or licensed to others by, such Grantor which is not Owned Intellectual Property either pursuant to one of the written material Copyright Licenses, Patent Licenses, Trademark Licenses, and/or Trade Secret Licenses listed on Schedule 3.9(a) and subject to the terms thereof (collectively, the "Licensed Intellectual Property") or otherwise.

- (b) On the date hereof all Owned Intellectual Property and all Licensed Intellectual Property, in each case, which is material to such Grantor's business (collectively, the "Material Intellectual Property"), is valid, subsisting, unexpired and enforceable and has not been abandoned. The operation of such Grantor's business as currently conducted or as contemplated to be conducted does not infringe, constitute a misappropriation of, dilute, or otherwise violate the Intellectual Property rights of any other Person where the same would have a Material Adverse Effect.
- (c) No claim has been asserted that the use of the Material Intellectual Property does or may infringe upon or constitute a misappropriation of the rights of any other Person.
- (d) To such Grantor's knowledge, no decision or judgment has been rendered by any Governmental Authority or arbitrator in the United States or outside the United States which would materially limit or cancel the validity or enforceability of, or such Grantor's rights in, any Material Intellectual Property. Such Grantor is not aware of any uses of any item of Material Intellectual Property that could reasonably be expected to lead to such item becoming invalid or unenforceable including unauthorized trademark uses by third parties and uses which were not supported by the goodwill of the business connected with Trademarks and Trademark Licenses.
- (e) No action or proceeding is pending, or, to such Grantor's knowledge, threatened, on the date hereof (i) seeking to limit, cancel or invalidate any Owned Intellectual Property, (ii) alleging that any services provided by, processes used by, or products manufactured or sold by such Grantor infringe any Patent, Trademark, Copyright, or misappropriate any Trade Secret or violate any other right of any other Person, or (iii) alleging that any Material Intellectual Property (A) owned by such Grantor or (B) licensed by such Grantor (to such Grantor's knowledge), is being licensed or sublicensed in violation of any intellectual property or any other right of any other Person, in each case, which, if adversely determined, would reasonably be expected to have a Material Adverse Effect. To such Grantor's knowledge, no Person is engaging in any activity that infringes upon or misappropriates, or is otherwise an unauthorized use of, any Material Intellectual Property owned by Grantor. The consummation of the transactions contemplated by this Agreement and the other Loan Documents will not result in the termination of any of the Material Intellectual Property.
- (f) With respect to each Copyright License, Trademark License, Trade Secret License and Patent License which license constitutes Material Intellectual Property or the loss of which could otherwise have a Material Adverse Effect: (i) such license is binding and enforceable against the other party thereto; (ii) such license will not cease to be valid and binding and in full force and effect on terms identical to those currently in effect as a result of the rights and interests granted herein (including, but not limited to, the enforceability of such rights and interests with respect to each such license), nor will the grant of such rights and interests (or the enforceability thereof) constitute a breach or default under such license or otherwise give the license or licensee a right to terminate such license; (iii) such Grantor has not received any notice of termination or cancellation under such license; (iv) such Grantor has not received any notice of a breach or default under such license, which breach or default has not been cured; and (v) such Grantor is not in breach or default in any material respect, and no event has occurred

that, with notice and/or lapse of time, would constitute such a breach or default or permit termination, modification or acceleration under such license.

- (g) Except as set forth on Schedule 3.9(g), such Grantor has made all filings and recordations and paid all required fees and taxes to maintain each and every item of registered Material Intellectual Property in full force and effect and to protect and maintain its interest therein.
- (h) To the knowledge of such Grantor, (i) none of the Trade Secrets that constitute Material Intellectual Property of such Grantor have been used, divulged, disclosed or appropriated to the detriment of such Grantor for the benefit of any other Person without permission of such Grantor; and (ii) no employee, independent contractor or agent of such Grantor has misappropriated any Trade Secrets of any other Person in the course of the performance of his or her duties as an employee, independent contractor or agent of such Grantor where the same would reasonably be expected to have a Material Adverse Effect.
 - (i) Such Grantor has taken commercially reasonable steps to exercise quality control over any licensee of such Grantor's Trademarks.
- 3.10. Commercial Tort Claims. No Grantor has knowledge that it has any Commercial Tort Claims as of the date hereof individually or in the aggregate in excess of \$1,000,000, except as set forth on Schedule 3.10.
- 3.11. Contracts. No amount payable to such Grantor under or in connection with any Contract which has a value in excess of \$1,000,000 individually or \$5,000,000 in the aggregate is evidenced by any Instrument or Tangible Chattel Paper which has not been delivered to the Administrative Agent or constitutes Electronic Chattel Paper that is not under the control (within the meaning of Section 9-105 of the New York UCC) of the Administrative Agent.

SECTION 4. COVENANTS

Each Grantor covenants and agrees with the Secured Parties that, as of the date hereof and until the termination of this Agreement in accordance with its terms:

- 4.1. <u>Covenants in Credit Agreement</u>. Each Grantor shall take, or shall refrain from taking, as the case may be, each action that is within its control and is necessary to be taken or not taken, as the case may be, so that no Default or Event of Default is caused by the failure to take such action or to refrain from taking such action by such Grantor or any of its Subsidiaries.
 - 4.2. Delivery and Control of Instruments, Chattel Paper, Negotiable Documents and Investment Property.
- (a) If any of the Collateral having a value in excess of \$1,000,000 individually or \$5,000,000 in the aggregate is or shall become evidenced or represented by any Instrument, Certificated Security, Negotiable Document or Tangible Chattel Paper, such Instrument (other than checks received in the ordinary course of business), Certificated Security, Negotiable Documents or Tangible Chattel Paper shall be promptly delivered to the Administrative Agent, duly endorsed in a manner reasonably satisfactory to the Administrative Agent, to be held as

Collateral pursuant to this Agreement, and all of such property owned by any Grantor as of the Closing Date and represented in such form shall be delivered on the Closing Date.

- (b) If any of the Collateral having a value in excess of \$1,000,000 individually or \$5,000,000 in the aggregate is or shall become "Electronic Chattel Paper" such Grantor shall ensure that (i) a single authoritative copy shall exist which is unique, identifiable, unalterable (except as provided in clauses (iii), (iv) and (v) of this paragraph), (ii) such authoritative copy identifies the Administrative Agent as the assignee and is communicated to and maintained by the Administrative Agent or its designee, (iii) copies or revisions that add or change the assignee of the authoritative copy can only be made with the participation of the Administrative Agent, (iv) each copy of the authoritative copy and any copy of a copy is readily identifiable as a copy and that is not the authoritative copy; (v) any revision of the authoritative copy is readily identifiable as an authorized or unauthorized revision, and (vi) the Administrative Agent has "control" within the meaning of the New York UCC of such Electronic Chattel Paper.
- (c) If any Collateral having a value in excess of \$1,000,000 individually or \$5,000,000 in the aggregate is or shall become an Uncertificated Security, such Grantor shall cause the issuer thereof, if such issuer is a Subsidiary of the Borrower, either (i) to register the Administrative Agent as the registered owner of such Uncertificated Security, upon original issue or registration of transfer or (ii) to agree in writing with such Grantor and the Administrative Agent that such issuer will comply with instructions with respect to such Uncertificated Security originated by the Administrative Agent without further consent of such Grantor and such actions shall be taken on or prior to the Closing Date with respect to any such Uncertificated Securities owned as of the Closing Date by any Grantor and the Grantor shall take or cause to be taken all such other actions as may be necessary for the Administrative Agent to have "control" defined in Article 8 of the New York UCC.

4.3. Maintenance of Insurance.

- (a) Such Grantor will maintain insurance in accordance with Section 6.16 of the Credit Agreement, and furnish to the Administrative Agent, upon written request, of a copy of such insurance policies.
- (b) Such Grantor will deliver to the Administrative Agent on behalf of the Secured Parties, (i) on the Closing Date, a certificate dated as of a recent date showing the amount and types of insurance coverage as of such date, (ii) upon reasonable request of the Administrative Agent from time to time, reasonably detailed information as to the insurance carried, (iii) promptly following receipt of notice from any insurer, a copy of any notice of cancellation or material change in coverage from that existing on the Closing Date and (iv) forthwith, notice of any cancellation or nonrenewal of coverage by such Grantor. To the extent applicable, the Administrative Agent shall be named as additional insured on all such liability insurance policies of such Grantor and the Administrative Agent shall be named as loss payee (and, where applicable, mortgagee) on all property and casualty insurance policies of such Grantor.
- 4.4. Payment of Obligations. Such Grantor shall pay and discharge or otherwise satisfy at or before maturity or before they become delinquent, as the case may be, all taxes,

assessments and governmental charges or levies imposed upon the Collateral or in respect of income or profits therefrom, as well as all claims of any kind (including claims for labor, materials and supplies) against or with respect to the Collateral, except that no such tax, assessment or charge need be paid if (a) the amount or validity thereof is currently being contested in good faith by appropriate proceedings, reserves in conformity with GAAP with respect thereto have been provided on the books of such Grantor and such proceedings could not reasonably be expected to result in the sale, forfeiture or loss of any material portion of the Collateral or any interest therein, or (b) the failure to so pay and discharge would not, in the aggregate, reasonably be expected to have a Material Adverse Effect.

4.5. Maintenance of Perfected Security Interest; Further Documentation.

- (a) Except as otherwise expressly permitted by the Credit Agreement, such Grantor shall maintain each of the security interests created by this Agreement as a perfected security interest under all Requirements of Law of the United States and of any state, territory or possession thereof, having at least the priority described in Section 3.3 and shall defend such security interest against any claims and demands of any Persons (other than the Secured Parties), subject to the provisions of Section 7.13.
- (b) Such Grantor shall furnish to the Secured Parties from time to time statements and schedules further identifying and describing the Collateral and such other reports in connection with the assets and property of such Grantor as the Administrative Agent may reasonably request, all in reasonable detail.
- (c) At any time and from time to time, upon the written request of the Administrative Agent, and at the sole expense of such Grantor, such Grantor shall promptly and duly authorize, execute and deliver, and have recorded, such further instruments and documents and take such further actions as the Administrative Agent may reasonably request to be taken in the United States for the purpose of obtaining or preserving the full benefits of this Agreement and of the rights and powers herein granted, including, the filing of any financing or continuation statements under the Uniform Commercial Code (or other similar laws) in effect in the United States or any State, territory or possession thereof with respect to the security interests created hereby and in the case of Investment Property and any other relevant Collateral, taking any actions necessary to enable the Administrative Agent to obtain "control" (within the meaning of the applicable Uniform Commercial Code) with respect thereto.
- 4.6. Changes in Locations, Name, Jurisdiction of Incorporation, etc. Such Grantor shall not, except upon at least 10 days' prior written notice (or such shorter period consented to by the Administrative Agent in writing), in each case, to the Administrative Agent and delivery to the Administrative Agent of duly authorized and, where required, executed copies of all additional financing statements and other documents reasonably requested by the Administrative Agent to maintain the validity, perfection and priority of the security interests provided for herein:
- (a) change its legal name, jurisdiction of organization or the location of its chief executive office or sole place of business from that referred to in Section 3.4; or

- (b) change its legal name, identity or structure to such an extent that any financing statement filed by the Administrative Agent in connection with this Agreement would become misleading; provided, however, that no such notices shall be required in connection with the Spinoff so long as (i) the Grantors comply with each of their respective obligations under Sections 6.22 and 6.24 of the Credit Agreement and (ii) such notice is given to the Administrative Agent within 30 days after the applicable event necessitating such notice.
 - 4.7. Notices. Such Grantor shall advise the Administrative Agent promptly, in reasonable detail, of:
- (a) any Lien (other than any Lien expressly permitted by Section 7.02 of the Credit Agreement) on any of the Collateral which would adversely affect the ability of the Administrative Agent to exercise any of its remedies hereunder;
- (b) the occurrence of any other event of which such Grantor becomes aware that would reasonably be expected to have a Material Adverse Effect on the aggregate value of the Collateral or on the security interests created hereby; and
- (c) the acquisition or ownership by any Grantor of any aircraft, airframe, aircraft engine, aircraft lease or any other related property with a value in excess of \$3,000,000 individually or in the aggregate.

4.8. Investment Property.

(a) If such Grantor shall become entitled to receive or shall receive any stock or other ownership certificate (including any certificate representing a stock dividend or a distribution in connection with any reclassification, increase or reduction of capital or any certificate issued in connection with any reorganization), option or rights in respect of Stock and Stock Equivalents in any issuer thereof, whether in addition to, in substitution of, as a conversion of, or in exchange for, any shares of or other ownership interests in the Pledged Securities, or otherwise in respect thereof, such Grantor shall accept the same as the agent of the Secured Parties, hold the same in trust for the Secured Parties and promptly deliver the same to the Administrative Agent in the exact form received (other than Excluded Stock), duly endorsed by such Grantor to the Administrative Agent, if required, together with an undated stock power or similar instrument of transfer covering such certificate duly executed in blank by such Grantor and with, if the Administrative Agent so requests, signature guaranteed, to be held by the Administrative Agent, subject to the terms hereof, as additional collateral security for the Obligations. Any sums paid upon or in respect of the Pledged Securities upon the liquidation or dissolution of any issuer thereof shall be paid over to the Administrative Agent to be held by it hereunder as additional collateral security for the Obligations if an Event of Default then exists, and in case any distribution of capital shall be made on or in respect of the Pledged Securities or any property shall be distributed upon or with respect to the Pledged Securities pursuant to the recapitalization or reclassification of the capital of any issuer thereof or pursuant to the reorganization thereof, the property so distributed shall, if an Event of Default then exists, and unless otherwise subject to a perfected security interest in favor of the Administrative Agent, be delivered to the Administrative Agent to be

Pledged Securities shall be received by such Grantor in violation of the immediately preceding sentence, such Grantor shall, until such money or property is paid or delivered to the Administrative Agent, hold such money or property in trust for the Secured Parties, segregated from other funds of such Grantor, as additional collateral security for the Obligations.

(b) Without the prior written consent of the Administrative Agent, such Grantor shall not (i) vote to enable, or take any other action to permit, any Subsidiary of the Borrower that is an issuer of Pledged Securities to issue any stock, partnership interests, limited liability company interests or other equity securities of any nature or to issue any other securities convertible into or granting the right to purchase or exchange for any stock, partnership interests, limited liability company interests or other equity securities of any nature of any such issuer (except, in each case, pursuant to a transaction expressly permitted by the Credit Agreement), (ii) sell, assign, transfer, exchange, or otherwise dispose of, or grant any option with respect to, any of the Investment Property or Proceeds thereof or any interest therein (except, in each case, pursuant to a transaction expressly permitted by the Credit Agreement), (iii) create, incur or permit to exist any Lien or option in favor of, or any claim of any Person with respect to, any of the Investment Property or Proceeds thereof, or any interest therein, except for the security interests created by this Agreement or any Lien expressly permitted thereon pursuant to Section 7.02 of the Credit Agreement, (iv) enter into any agreement or undertaking restricting the right or ability of such Grantor or the Administrative Agent to sell, assign or transfer any of the Investment Property or Proceeds thereof or any interest therein or (v) without the prior written consent of the Administrative Agent, cause or permit any Subsidiary of the Borrower that is an issuer of any Pledged LLC/Partnership Interests which are not securities for purposes of the New York UCC) on the date hereof to elect or otherwise take any action to cause such Pledged LLC/Partnership Interests to be treated as Securities for purposes of the New York UCC; provided, however, notwithstanding the foregoing, if any issuer of any Pledged LLC/Partnership Interests takes any such action in violation of t

(c) In the case of each Grantor which is an issuer of Pledged Securities, such issuer agrees that (i) it shall be bound by the terms of this Agreement relating to the Pledged Securities issued by it and shall comply with such terms insofar as such terms are applicable to it, (ii) it shall notify the Administrative Agent promptly in writing of the occurrence of any of the events described in Section 4.8(a) with respect to the Pledged Securities issued by it and (iii) the terms of Sections 5.3(c) and 5.7 shall apply to it, mutatis mutandis, with respect to all actions that may be required of it pursuant to Section 5.3(c) or 5.7 with respect to the Pledged Securities issued by it. In addition, each Grantor which is either an issuer or an owner of any Pledged Security hereby consents to the grant by each other Grantor of the security interest hereunder in favor of the Administrative Agent and to the transfer of any Pledged Security to the Administrative Agent or its nominee following the occurrence and during the continuance of an Event of Default and to the substitution of the Administrative Agent or its nominee as a partner, member or shareholder of the issuer of the related Pledged Security.

4.9. <u>Receivables</u>. Other than in the ordinary course of business, such Grantor shall not (i) grant any extension of the time of payment of any Receivable, (ii) compromise or settle any Receivable for less than the full amount thereof, (iii) release, wholly or partially, any Person liable for the payment of any Receivable, (iv) allow any credit or discount whatsoever on any Receivable or (v) amend, supplement or modify any Receivable in any manner that could adversely affect the value thereof.

4.10. Intellectual Property.

- (a) Such Grantor (either itself or through licensees) shall, in the exercise of its reasonable business judgment, taking into account the Secured Parties' interests under this Agreement, (i) continue to use each owned Trademark material to its business, (ii) maintain commercially reasonable quality of products and services offered under such Trademarks and take all necessary steps to ensure that all licensed users of such Trademarks comply with such Grantor's quality control requirements and maintain reasonable quality, (iii) not adopt or use any mark which is confusingly similar or a colorable imitation of such Trademarks unless the Administrative Agent, for the ratable benefit of the Secured Parties, shall obtain a perfected security interest in such mark pursuant to this Agreement and an Intellectual Property Security Agreement, and (iv) not (and not permit any licensee or sublicensee thereof to) do any act or knowingly omit to do any act whereby such Trademark may become invalidated or impaired in any way.
- (b) Such Grantor (either itself or through licensees), subject to the exercise of its reasonable business judgment, taking into account the Secured Parties' interests under this Agreement, shall not do any act, or omit to do any act, whereby any Patent owned by such Grantor material to its business may become forfeited, abandoned or dedicated to the public.
- (c) Such Grantor (either itself or through licensees), subject to the exercise of its reasonable business judgment, taking into account the Secured Parties' interests under this Agreement, shall not (and shall not permit any licensee or sublicensee thereof to) do any act or knowingly omit to do any act whereby any material portion of Copyrights owned by such Grantor and material to its business may become invalidated or otherwise impaired. Such Grantor shall not (either itself or through licensees) do any act whereby any material portion of such Copyrights may fall into the public domain.
- (d) Such Grantor shall notify the Administrative Agent promptly if it knows or suspects that any application or registration relating to any Material Intellectual Property owned by such Grantor may become forfeited, abandoned or dedicated to the public, or of any adverse determination (including the institution of, or any such determination in, any proceeding in the United States Patent and Trademark Office, the United States Copyright Office or any court or tribunal in any country) regarding such Grantor's ownership of, or the validity of, any such Material Intellectual Property or such Grantor's right to register the same or to own and maintain the same
- (e) Promptly after such Grantor, either by itself or through any agent, employee, licensee or designee, shall file an application for the registration of any Intellectual Property that is material to the business of such Grantor with the United States Patent and

Trademark Office or the United States Copyright Office, such Grantor shall report such filing to the Administrative Agent (i) in the case of any Copyrights, within five Business Days (or such longer period of time permitted by the Administrative Agent in its sole discretion) after applying for a registration and again within five Business Days (or such longer period of time permitted by the Administrative Agent in its sole discretion) after receiving a registration and (ii) in the case of material Patents, Trademarks or other Intellectual Property, within five Business Days after the last day of the fiscal quarter in which such filing occurs. Upon request of the Administrative Agent, such Grantor shall execute and deliver, and have recorded in the United States Patent and Trademark office or the United States Copyright Office, as applicable, any and all agreements, instruments, documents, and papers as the Administrative Agent may request to evidence the Secured Parties' security interest in any Copyright, Patent, Trademark or other Intellectual Property of such Grantor.

- (f) Such Grantor, subject to the exercise of its reasonable business judgment, taking into account the Secured Parties' interests under this Agreement, shall take reasonable and necessary steps, including in any proceeding before the United States Patent and Trademark Office or the United States Copyright Office, to maintain and pursue each application (and to obtain the relevant registration) and to maintain each registration of Intellectual Property material to its business, including the payment of required fees and taxes, the filing of responses to office actions issued by the United States Patent and Trademark Office and the United States Copyright Office, the filing of applications for renewal or extension, the filing of affidavits of use and affidavits of incontestability, the filing of divisional, continuation, continuation-in-part, reissue and renewal applications or extensions, the payment of maintenance fees, and the participation in interference, reexamination, opposition, cancellation, infringement and misappropriation proceedings.
- (g) Such Grantor (either itself or through licensees), subject to the exercise of its reasonable business judgment, taking into account the Secured Parties' interests under this Agreement, shall not, without the prior written consent of the Administrative Agent, discontinue use of or otherwise abandon any of its registered Owned Intellectual Property, or abandon any application or any right to file an application for any patent, trademark, or copyright, unless such Grantor shall have previously determined that such use or the pursuit or maintenance of such Intellectual Property is no longer desirable in the conduct of such Grantor's business and that the loss thereof could not reasonably be expected to have a Material Adverse Effect.
- (h) In the event that any Intellectual Property material to its business is infringed, misappropriated or diluted by a third party, such Grantor shall (i) take such actions as such Grantor shall reasonably deem appropriate under the circumstances to protect such Intellectual Property and (ii) if such Intellectual Property is of material economic value, promptly notify the Administrative Agent after it learns thereof and sue for infringement, misappropriation or dilution, to seek injunctive relief where appropriate and to recover any and all damages for such infringement, misappropriation or dilution.
- (i) Such Grantor agrees that, should it obtain an ownership interest in any item of intellectual property which is not, as of the Closing Date, a part of the Intellectual Property Collateral (the "After-Acquired Intellectual Property"), (i) the provisions of Section 2.1 shall automatically apply thereto, (ii) any such After-Acquired Intellectual Property, and in the

case of trademarks, the goodwill of the business connected therewith or symbolized thereby, shall automatically become part of the Intellectual Property Collateral, (iii) in the case of any Copyrights, within five Business Days (or such longer period of time permitted by the Administrative Agent in its sole discretion) after obtaining such ownership, provide written notice thereof, (iv) in the case of material Patents, Trademarks or other Intellectual Property, within five Business Days after the last day of the fiscal quarter in which such ownership is obtained, and (v) promptly after the Administrative Agent's request, it shall provide the Administrative Agent with an amended Schedule 3.9(a) and take the actions specified in clauses (j) and (k) of Section 4.10.

- (j) Such Grantor agrees to execute an Intellectual Property Security Agreement with respect to its Intellectual Property in order to record the security interest granted herein to the Administrative Agent for the ratable benefit of the Secured Parties with the United States Patent and Trademark Office and the United States Copyright Office.
- (k) Such Grantor agrees to execute an Intellectual Property Security Agreement with respect to its After-Acquired Intellectual Property in order to record the security interest granted herein to the Administrative Agent for the ratable benefit of the Secured Parties with the United States Patent and Trademark Office and the United States Copyright Office.
- (l) Such Grantor shall take commercially reasonable steps to protect the secrecy of all trade secrets or confidential information material to its business, including entering into confidentiality agreements with employees and labeling and restricting access to secret information and documents.

4.11. Contracts.

- (a) Such Grantor shall perform and comply in all material respects with all its obligations under the Contracts, except where the failure to so perform and comply would not reasonably be expected to have a Material Adverse Effect.
- (b) Such Grantor shall not amend, modify, terminate, waive or fail to enforce any provision of any Contract in any manner which would reasonably be expected to have a Material Adverse Effect.
- (c) Such Grantor shall exercise promptly and diligently each and every material right which it may have under each contract (other than any right of termination), except where the failure to so exercise would not reasonably be expected to have a Material Adverse Effect.
- (d) Such Grantor shall not permit to become effective in any document creating, governing or providing for any permit, lease, license or contract, a provision that would limit the creation, perfection or scope of, or exercise or enforcement of remedies in connection with, a Lien on such permit, lease, license or contract in favor of the Administrative Agent for the ratable benefit of the Secured Parties unless such Grantor believes, in its reasonable judgment, that such prohibition is usual and customary in transactions of such type.

4.12. Commercial Tort Claims. Such Grantor shall advise the Administrative Agent promptly after such Grantor becomes aware of any Commercial Tort Claim held by such Grantor individually or in the aggregate in excess of \$1,000,000 and shall promptly execute a supplement to this Agreement in form and substance reasonably satisfactory to the Administrative Agent to grant a security interest in such Commercial Tort Claim to the Administrative Agent for the ratable benefit of the Secured Parties.

SECTION 5. REMEDIAL PROVISIONS

5.1. Certain Matters Relating to Receivables.

- (a) The Administrative Agent shall have the right (but shall in no way be obligated), at its own expense if an Event of Default does not then exist, to make test verifications of the Receivables that are included in the Collateral in any manner and through any medium that it reasonably considers advisable, and each Grantor shall furnish all such assistance and information as the Administrative Agent may reasonably require in connection with such test verifications.
- (b) Subject to the rights of the Administrative Agent under Section 5.2(b), each Grantor hereby agrees to use its commercially reasonable efforts to continue to collect all amounts due or to become due to such Grantor under the Receivables and any Supporting Obligation and diligently exercise each material right it may have under any Receivable and any Supporting Obligation, in each case, at its own expense. If required by the Administrative Agent at any time after the occurrence and during the continuance of an Event of Default, any payments of Receivables, when collected by any Grantor, (i) shall be promptly (and, in any event, within two Business Days) deposited by such Grantor in the exact form received, duly endorsed by such Grantor to the Administrative Agent if required, in a Collateral Account maintained under the sole dominion and control of the Administrative Agent, subject to withdrawal by the Administrative Agent for the account of the Secured Parties only as provided in Section 5.5, and (ii) until so turned over, shall be held by such Grantor in trust for the Secured Parties, segregated from other funds of such Grantor. Each such deposit of Proceeds of Receivables shall be accompanied by a report identifying in reasonable detail the nature and source of the payments included in the deposit.
- (c) At the Administrative Agent's request but subject to the confidentiality provisions set forth in the Credit Agreement, during the continuance of an Event of Default each Grantor shall make available to the Administrative Agent original and other documents evidencing, and relating to, the agreements and transactions which gave rise to the Receivables that are included in the Collateral, including original orders, invoices and shipping receipts.

5.2. Communications with Obligors; Grantors Remain Liable

(a) The Administrative Agent in its own name or in the name of others may at any time after the occurrence and during the continuance of an Event of Default communicate with obligors under the Receivables and parties to the Contracts to verify with them to the Administrative Agent's satisfaction the existence, amount and terms of any Receivables or Contracts.

- (b) The Administrative Agent may at any time after the occurrence and during the continuance of an Event of Default notify, or require any Grantor to so notify, the Account Debtor or counterparty on any Receivable or Contract of the security interest of the Administrative Agent therein. In addition, after the occurrence and during the continuance of an Event of Default, the Administrative Agent may upon written notice to the applicable Grantor, notify, or require any Grantor to notify, the Account Debtor or counterparty to make all payments under the Receivables and/or Contracts directly to the Administrative Agent.
- (c) Anything herein to the contrary notwithstanding, each Grantor shall remain liable under each of the Receivables and Contracts to observe and perform all the conditions and obligations to be observed and performed by it thereunder, all in accordance with the terms of any agreement giving rise thereto. No Secured Party shall have any obligation or liability under any Receivable (or any agreement giving rise thereto) or Contract by reason of or arising out of this Agreement or the receipt by any Secured Party of any payment relating thereto, nor shall any Secured Party be obligated in any manner to perform any of the obligations of any Grantor under or pursuant to any Receivable (or any agreement giving rise thereto) or Contract, to make any payment, to make any inquiry as to the nature or the sufficiency of any payment received by it or as to the sufficiency of any performance by any party thereunder, to present or file any claim, to take any action to enforce any performance or to collect the payment of any amounts which may have been assigned to it or to which it may be entitled at any time or times.

5.3. <u>Pledged Securities</u>.

- (a) Unless an Event of Default shall have occurred and be continuing and the Administrative Agent shall have given notice to the relevant Grantor of the Administrative Agent's intent to exercise its corresponding rights pursuant to Section 5.3(b), each Grantor shall be permitted to receive all cash dividends paid in respect of the Pledged Equity Interests and all payments made in respect of the Pledged Notes, to the extent not prohibited by the Credit Agreement, and to exercise all voting, corporate and other ownership (or other similar) rights with respect to the Pledged Securities; provided, however, that no vote shall be cast or corporate or other ownership (or other similar) right exercised or other action taken which would materially impair the Collateral or which would be inconsistent with or result in any violation of any provision of the Credit Agreement, this Agreement or any other Loan Document.
- (b) If an Event of Default shall occur and be continuing and the Administrative Agent shall have given notice to the relevant Grantor of the Administrative Agent's intent to exercise its rights pursuant to this Section 5.3(b): (i) all rights of each Grantor to exercise or refrain from exercising the voting and other consensual rights which it would otherwise be entitled to exercise pursuant hereto shall cease and all such rights shall thereupon become vested in the Administrative Agent who shall thereupon have the sole right, but shall be under no obligation, to exercise or refrain from exercising such voting and other consensual rights; (ii) the Administrative Agent shall have the right, without notice to any Grantor (where permitted by applicable law), to transfer all or any portion of the Investment Property to its name or the name of its nominee or agent; and (iii) the Administrative Agent shall have the right, without notice to any Grantor, to exchange any certificates or instruments representing any Investment Property for certificates or instruments of smaller or larger denominations. In order

to permit the Administrative Agent to exercise the voting and other consensual rights which it may be entitled to exercise pursuant hereto and to receive all dividends and other distributions which it may be entitled to receive hereunder each Grantor shall promptly execute and deliver (or cause to be executed and delivered) to the Administrative Agent all proxies, dividend payment orders and other instruments as the Administrative Agent may from time to time reasonably request and each Grantor acknowledges that the Administrative Agent may utilize the power of attorney set forth herein.

- (c) Each Grantor hereby authorizes and instructs each issuer of any Pledged Securities pledged by such Grantor hereunder to (i) comply with any instruction received by it from the Administrative Agent in writing that (A) states that an Event of Default has occurred and is continuing and (B) is otherwise in accordance with the terms of this Agreement, without any other or further instructions from such Grantor, and each Grantor agrees that each such issuer shall be fully protected in so complying, and (ii) upon any such instruction following the occurrence and during the continuance of an Event of Default, pay any dividends or other payments with respect to the Investment Property, including Pledged Securities, directly to the Administrative Agent.
- 5.4. Proceeds to be Turned Over To Administrative Agent. In addition to the rights of the Secured Parties specified in Section 5.1 with respect to payments of Receivables, if an Event of Default shall occur and be continuing, all Proceeds received by any Grantor consisting of cash, cash equivalents, checks and other near-cash items shall, if requested in writing by the Administrative Agent, be held by such Grantor in trust for the Secured Parties, segregated from other funds of such Grantor, and shall, forthwith upon receipt by such Grantor, be turned over to the Administrative Agent in the exact form received by such Grantor (duly endorsed by such Grantor to the Administrative Agent, if required). All Proceeds received by the Administrative Agent hereunder shall be held by the Administrative Agent in a Collateral Account maintained under its sole dominion and control. All Proceeds while held by the Administrative Agent in a Collateral Account (or by such Grantor in trust for the Secured Parties) shall continue to be held as collateral security for all the Obligations and shall not constitute payment thereof until applied as provided in Section 5.5.
- 5.5. <u>Application of Proceeds</u>. At such intervals as may be agreed upon by the Borrower and the Administrative Agent, or, if an Event of Default shall have occurred and be continuing, at any time at the Administrative Agent's election, the Administrative Agent may apply all or any part of the net Proceeds (after deducting fees and reasonable out-of-pocket expenses as provided in <u>Section 5.6</u>) constituting Collateral realized through the exercise by the Administrative Agent of its remedies hereunder, whether or not held in any Collateral Account, and any proceeds of the guarantee set forth in the Guaranty, in payment of the Obligations in accordance with the Credit Agreement.

5.6. Code and Other Remedies.

(a) If an Event of Default shall occur and be continuing, the Administrative Agent, on behalf of the Secured Parties, may exercise, in addition to all other rights and remedies granted to it in this Agreement and in any other instrument or agreement securing, evidencing or relating to the Obligations, all rights and remedies of a secured party under the New York UCC

(whether or not the New York UCC applies to the affected Collateral) or its rights under any other applicable law or in equity. Without limiting the generality of the foregoing, the Administrative Agent, without demand of performance or other demand, presentment, protest, advertisement or notice of any kind (except any notice required by law referred to below) to or upon any Grantor or any other Person (all and each of which demands, defenses, advertisements and notices are hereby waived), may in such circumstances collect, receive, appropriate and realize upon the Collateral, or any part thereof, and/or may sell, lease, license, assign, give option or options to purchase, or otherwise dispose of and deliver the Collateral or any part thereof (or contract to do any of the foregoing), in one or more parcels at public or private sale or sales, at any exchange, broker's board or office of any Secured Party or elsewhere upon such terms and conditions as it may deem advisable and at such prices as it may deem best, for cash or on credit or for future delivery without assumption of any credit risk. Each Secured Party shall have the right upon any such public sale or sales, and, to the extent permitted by law, upon any such private sale or sales, to purchase the whole or any part of the Collateral so sold, free of any right or equity of redemption in any Grantor, which right or equity is hereby waived and released. Each purchaser at any such sale shall hold the property sold absolutely free from any claim or right on the part of any Grantor, and each Grantor hereby waives (to the extent permitted by applicable law) all rights of redemption, stay and/or appraisal which it now has or may at any time in the future have under any rule of law or statute now existing or hereafter enacted. Each Grantor agrees that, to the extent notice of sale shall be required by law, at least ten days notice to such Grantor of the time and place of any public sale or the time after which any private sale is to be made shall constitute reasonable notification. The Administrative Agent shall not be obligated to make any sale of Collateral regardless of notice of sale having been given. The Administrative Agent may adjourn any public or private sale from time to time by announcement at the time and place fixed therefor, and such sale may, without further notice, be made at the time and place to which it was so adjourned. The Administrative Agent may sell the Collateral without giving any warranties as to the Collateral. The Administrative Agent may specifically disclaim or modify any warranties of title or the like. This procedure will not be considered to adversely effect the commercial reasonableness of any sale of the Collateral. Each Grantor agrees that it would not be commercially unreasonable for the Administrative Agent to dispose of the Collateral or any portion thereof by using Internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capability of doing so, or that match buyers and sellers of assets. To the extent permitted by applicable law, each Grantor hereby waives any claims against the Administrative Agent arising by reason of the fact that the price at which any Collateral may have been sold at such a private sale was less than the price which might have been obtained at a public sale, even if the Administrative Agent accepts the first offer received and does not offer such Collateral to more than one offeree. Each Grantor further agrees, at the Administrative Agent's request, to assemble the Collateral and make it available to the Administrative Agent at places which the Administrative Agent shall reasonably select, whether at such Grantor's premises or elsewhere. To the extent permitted by applicable law, and so long as an Event of Default is continuing, the Administrative Agent shall have the right to enter onto the property where any Collateral is located and take possession thereof with or without judicial process.

(b) The Administrative Agent shall apply the net proceeds of any action taken by it pursuant to this <u>Section 5.6</u>, after deducting all reasonable costs and expenses of every kind incurred in connection therewith or incidental to the care or safekeeping of any of the Collateral

or in any way relating to the Collateral or the rights of the Secured Parties hereunder, including reasonable attorneys' fees and disbursements, to the payment in whole or in part of the Obligations and only after such application and after the payment by the Administrative Agent of any other amounts required by any provision of law, including Section 9-615(a) of the New York UCC, need the Administrative Agent account for the surplus, if any, to any Grantor. If the Administrative Agent sells any of the Collateral upon credit, the Grantor will be credited only with payments actually made by the purchaser and received by the Administrative Agent and applied to indebtedness of the purchaser. In the event the purchaser fails to pay for the Collateral, the Administrative Agent may resell the Collateral and the Grantor shall be credited with proceeds of the sale. To the extent permitted by applicable law, each Grantor waives all claims, damages and demands it may acquire against any Secured Party arising out of the exercise by any Secured Party of any rights hereunder.

(c) In the event of any disposition of any of the Intellectual Property, the goodwill of the business connected with and symbolized by any Trademarks subject to such disposition shall be included, and the applicable Grantor shall, to the extent commercially reasonable and feasible under the circumstances, supply the Administrative Agent or its designee with such Grantor's know-how and expertise, and with documents and things embodying the same, relating to the manufacture, distribution, advertising and sale of products or the provision of services relating to any Intellectual Property subject to such disposition, and such Grantor's customer lists and other records and documents relating to such Intellectual Property and to the manufacture, distribution, advertising and sale of such products and services.

5.7. Private Sales, etc.

- (a) Each Grantor recognizes that the Administrative Agent may be unable to effect a public sale of any or all the Pledged Equity Interests, by reason of certain prohibitions contained in the Securities Act and applicable state securities laws or otherwise, and may be compelled to resort to one or more private sales thereof to a restricted group of purchasers which will be obliged to agree, among other things, to acquire such securities for their own account for investment and not with a view to the distribution or resale thereof. Each Grantor acknowledges and agrees that any such private sale may result in prices and other terms less favorable than if such sale were a public sale and, notwithstanding such circumstances, agrees that any such private sale shall be deemed to have been made in a commercially reasonable manner. The Administrative Agent shall be under no obligation to delay a sale of any of the Pledged Equity Interests for the period of time necessary to permit the issuer thereof to register such securities for public sale under the Securities Act, or under applicable state securities laws, even if such issuer would agree to do so.
- (b) Each Grantor agrees to use commercially reasonable efforts to do or cause to be done all such other acts as may be necessary to make such sale or sales of all or any portion of the Pledged Equity Interests pursuant to this <u>Section 5.7</u> valid and binding and in compliance with any and all other applicable Requirements of Law. Each Grantor further agrees that a breach of any of the covenants contained in this <u>Section 5.7</u> will cause irreparable injury to the Secured Parties, that the Secured Parties have no adequate remedy at law in respect of such breach and, as a consequence, that each and every covenant contained in this <u>Section 5.7</u> shall be specifically enforceable against such Grantor, and such Grantor hereby waives and agrees not to

assert any defenses against an action for specific performance of such covenants except for a defense that no Event of Default has occurred and is continuing under the Credit Agreement or a defense of payment.

- 5.8. <u>Deficiency</u>. Each Grantor shall remain liable for any deficiency if the proceeds of any sale or other disposition of the Collateral are insufficient to pay its Obligations and the reasonable fees and disbursements of any outside attorneys employed by any Secured Party to collect such deficiency.
- 5.9. <u>BWXT Entities</u>. Notwithstanding anything contained herein to the contrary, the Administrative Agent will not take any action with respect to any pledge of Stock or Stock Equivalents of any Person that directly or indirectly owns Stock or Stock Equivalents in any BWXT Entity if such action would constitute or result in the change of ownership of any Person that directly or indirectly owns Stock in a BWXT Entity if such change of ownership would require under then-existing law or any material contract, the prior approval of the U.S. Navy, the U.S. Department of Energy or any other Governmental Authority, without first obtaining such approval. Each Grantor covenants that, after the occurrence and during the continuance of an Event of Default, it will take all actions as may be requested by the Administrative Agent to obtain such approval.

SECTION 6. THE ADMINISTRATIVE AGENT

6.1. Administrative Agent's Appointment as Attorney-in-Fact, etc.

- (a) Each Grantor hereby irrevocably constitutes and appoints the Administrative Agent, with full power of substitution, as its true and lawful attorney-infact with full irrevocable power and authority in the place and stead of such Grantor and in the name of such Grantor or in its own name, for the purpose of carrying out the terms of this Agreement, to take any and all appropriate action and to execute any and all documents and instruments which may be necessary or desirable to accomplish the purposes of this Agreement, and, without limiting the generality of the foregoing, each Grantor hereby gives the Administrative Agent the power and right, on behalf of such Grantor, without notice to or assent by such Grantor, to do any or all of the following:
 - (i) in the name of such Grantor or its own name, or otherwise, take possession of and endorse and collect any checks, drafts, notes, acceptances or other instruments for the payment of moneys due under any Receivable or Contract or with respect to any other Collateral and file any claim or take any other action or proceeding in any court of law or equity or otherwise deemed appropriate by the Administrative Agent for the purpose of collecting any and all such moneys due under any Receivable or Contract or with respect to any other Collateral whenever payable;
 - (ii) in the case of any Intellectual Property, execute and deliver, and have recorded, any and all agreements, instruments, documents and papers as the Administrative Agent may request to evidence the Secured Parties' security interest in such Intellectual Property and the goodwill and general intangibles of such Grantor relating thereto or represented thereby;

- (iii) pay or discharge taxes and Liens levied or placed on or threatened against the Collateral, effect any repairs or any insurance called for by the terms of this Agreement and pay all or any part of the premiums therefor and the costs thereof;
- (iv) execute, in connection with any sale provided for in <u>Section 5.7</u>, any endorsements, assignments or other instruments of conveyance or transfer with respect to the Collateral; and
- (v) (1) direct any party liable for any payment under any of the Collateral to make payment of any and all moneys due or to become due thereunder directly to the Administrative Agent or as the Administrative Agent shall direct; (2) ask or demand for, collect, and receive payment of and receipt for, any and all moneys, claims and other amounts due or to become due at any time in respect of or arising out of any Collateral; (3) sign and endorse any invoices, freight or express bills, bills of lading, storage or warehouse receipts, drafts against debtors, assignments, verifications, notices and other documents in connection with any of the Collateral; (4) commence and prosecute any suits, actions or proceedings at law or in equity in any court of competent jurisdiction to collect the Collateral or any portion thereof and to enforce any other right in respect of any Collateral; (5) defend any suit, action or proceeding brought against such Grantor with respect to any Collateral; (6) settle, compromise or adjust any such suit, action or proceeding and, in connection therewith, give such discharges or releases as the Administrative Agent may deem appropriate; (7) assign any Copyright, Patent or Trademark (along with the goodwill of the business to which any such Copyright, Patent or Trademark pertains), throughout the world for such term or terms, on such conditions, and in such manner, as the Administrative Agent shall in its sole discretion determine; and (8) generally, sell, transfer, pledge and make any agreement with respect to or otherwise deal with any of the Collateral as fully and completely as though the Administrative Agent were the absolute owner thereof for all purposes, and do, at the Administrative Agent's option and such Grantor's expense, at any time, or from time to time, all acts and things which the Administrative Agent necessary to protect, preserve or realize upon the Collateral and the Secured Parties' security interests therein and to effect the intent of this Agreement, all as fully and effec

Anything in this <u>Section 6.1(a)</u> to the contrary notwithstanding, the Administrative Agent agrees that, except as provided in <u>Section 6.1(b)</u>, it will not exercise any rights under the power of attorney provided for in this <u>Section 6.1(a)</u> unless an Event of Default shall have occurred and be continuing.

(b) If any Grantor fails to perform or comply with any of its agreements contained herein, the Administrative Agent, at its option, but without any obligation so to do, may perform or comply, or otherwise cause performance or compliance, with such agreement; <u>provided, however</u>, that unless an Event of Default has occurred and is continuing or time is of the essence, the Administrative Agent shall not exercise this power without first making demand on the Grantor and the Grantor failing to promptly comply therewith.

- (c) The expenses of the Administrative Agent incurred in connection with actions undertaken as provided in this Section 6.1, together with interest thereon at a rate per annum equal to the rate per annum at which interest would then be payable on past due Committed Loans that are Base Rate Loans under the Credit Agreement, from the date of payment by the Administrative Agent to the date reimbursed by the relevant Grantor, shall be payable by such Grantor to the Administrative Agent on demand.
- (d) Each Grantor hereby ratifies all that said attorneys shall lawfully do or cause to be done by virtue hereof. All powers, authorizations and agencies contained in this Agreement are coupled with an interest and are irrevocable until this Agreement is terminated and the security interests created hereby are released.
- 6.2. <u>Duty of Administrative Agent</u>. The Administrative Agent's sole duty with respect to the custody, safekeeping and physical preservation of the Collateral in its possession, under Section 9-207 of the New York UCC or otherwise, shall be to deal with it in the same manner as the Administrative Agent deals with similar property for its own account. Neither the Administrative Agent, nor any other Secured Party nor any of their respective officers, directors, partners, employees, agents, attorneys and other advisors, attorneys-in-fact or affiliates shall be liable for failure to demand, collect or realize upon any of the Collateral or for any delay in doing so or shall be under any obligation to sell or otherwise dispose of any Collateral upon the request of any Grantor or any other Person or to take any other action whatsoever with regard to the Collateral or any part thereof. The powers conferred on the Secured Parties hereunder are solely to protect the Secured Parties' interests in the Collateral and shall not impose any duty upon any Secured Party to exercise any such powers. The Secured Parties shall be accountable only for amounts that they actually receive as a result of the exercise of such powers, and neither they nor any of their officers, directors, partners, employees, agents, attorneys and other advisors, attorneys-in-fact or affiliates shall be responsible to any Grantor for any act or failure to act hereunder, except to the extent that any such act or failure to act is found by a final and nonappealable decision of a court of competent jurisdiction to have resulted from their own gross negligence or willful misconduct in breach of a duty owed to such Grantor.
- 6.3. Execution of Financing Statements. Each Grantor acknowledges that pursuant to Section 9-509(b) of the New York UCC and any other applicable law, each Grantor authorizes the Administrative Agent to file or record financing or continuation statements, and amendments thereto, and other filing or recording documents or instruments with respect to the Collateral, without the signature of such Grantor, in such form and in such offices as the Administrative Agent reasonably determines appropriate to perfect or maintain the perfection of the security interests of the Administrative Agent under this Agreement. Each Grantor agrees that such financing statements may describe the collateral in the same manner as described in this Agreement or as "all assets," "all personal property" or words of similar effect, regardless of whether or not the Collateral includes all assets or all personal property of such Grantor, or such other description as the Administrative Agent, in its sole judgment, determines is necessary or advisable that is of an equal or lesser scope or with greater detail. A photographic or other reproduction of this Agreement shall, where permitted by applicable law, be sufficient as a financing statement or other filing or recording document or instrument for filing or recording in any jurisdiction.

- 6.4. <u>Authority of Administrative Agent</u>. Each Grantor acknowledges that the rights and responsibilities of the Administrative Agent under this Agreement with respect to any action taken by the Administrative Agent or the exercise or non-exercise by the Administrative Agent of any option, voting right, request, judgment or other right or remedy provided for herein or resulting or arising out of this Agreement shall, as between the Administrative Agent and the other Secured Parties, be governed by the Credit Agreement and by such other agreements with respect thereto as may exist from time to time among them, but, as between the Administrative Agent and the Grantors, the Administrative Agent shall be conclusively presumed to be acting as agent for the Secured Parties with full and valid authority so to act or refrain from acting, and no Grantor shall be under any obligation, or entitlement, to make any inquiry respecting such authority.
- 6.5. <u>Appointment of Co-Administrative Agents</u>. At any time or from time to time, in order to comply with any applicable requirement of law, the Administrative Agent may appoint another bank or trust company or one of more other Persons, either to act as co-agent or agents on behalf of the Secured Parties with such power and authority as may be necessary for the effectual operation of the provisions hereof and which may be specified in the instrument of appointment (which may, in the discretion of the Administrative Agent, include provisions for indemnification and similar protections of such co-agent or separate agent).

SECTION 7. MISCELLANEOUS

- 7.1. Amendments in Writing. None of the terms or provisions of this Agreement may be waived, amended, supplemented or otherwise modified except by a written instrument executed by each affected Grantor and the Administrative Agent, subject to any consents required under Section 10.01 of the Credit Agreement; provided that any provision of this Agreement imposing obligations on any Grantor may be waived by the Administrative Agent in a written instrument executed thereby.
- 7.2. <u>Notices</u>. All notices and communications hereunder shall be given to the addresses and otherwise made in accordance with <u>Section 10.02</u> of the Credit Agreement; <u>provided</u> that notices and communications to any Grantor other than the Borrower shall be directed to such Grantor, at the address of the Borrower.
- 7.3. No Waiver by Course of Conduct; Cumulative Remedies. No Secured Party shall by any act (except by a written instrument pursuant to Section 7.1), delay, indulgence, omission or otherwise be deemed to have waived any right or remedy hereunder or to have acquiesced in any Default or Event of Default. No failure to exercise, nor any delay in exercising, on the part of any Secured Party, any right, power or privilege hereunder shall operate as a waiver thereof. No single or partial exercise of any right, power or privilege hereunder shall preclude any other or further exercise thereof or the exercise of any other right, power or privilege. A waiver by any Secured Party of any right or remedy hereunder on any one occasion shall not be construed as a bar to any right or remedy which such Secured Party would otherwise have on any future occasion. The rights and remedies herein provided are cumulative, may be exercised singly or concurrently and are not exclusive of any other rights or remedies provided by law.
 - 7.4. Enforcement Expenses; Indemnification.

- (a) Each Grantor agrees to pay or reimburse each Secured Party for its reasonable out-of-pocket costs and expenses incurred in collecting against such Grantor under the guarantee contained in the Guaranty or otherwise enforcing or preserving any rights under this Agreement and the other Loan Documents to which such Grantor is a party, including the reasonable fees and disbursements of outside counsel to each Secured Party and outside counsel to the Administrative Agent.
- (b) Each Grantor agrees to pay, and to hold the Secured Parties harmless from, any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits and reasonable out-of-pocket costs, expenses or disbursements of any kind or nature whatsoever with respect to, or resulting from any delay in paying, any and all stamp, excise, sales or other taxes (other than Excluded Taxes) which may be payable or determined to be payable with respect to any of the Collateral or in connection with any of the transactions contemplated by this Agreement.
- (c) Each Grantor agrees to pay, and to hold the Secured Parties harmless from, any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits and reasonable out-of-pocket costs, expenses or disbursements of any kind or nature whatsoever with respect to the execution, delivery, enforcement, performance and administration of this Agreement to the extent the Borrower would be required to do so pursuant to Section 10.04 of the Credit Agreement.
- (d) The agreements in this Section shall survive repayment of the Obligations and all other amounts payable under the Credit Agreement and the other Loan Documents.
- 7.5. Successors and Assigns. This Agreement shall be binding upon the successors and assigns of each Grantor and shall inure to the benefit of the Secured Parties and their permitted successors and assigns; provided that, except as otherwise permitted by the Credit Agreement, no Grantor may assign, transfer or delegate any of its rights or obligations under this Agreement without the prior written consent of the Administrative Agent, and any attempted assignment without such consent shall be null and void.
- 7.6. Set-off; Governing Law; Submission to Jurisdiction; Venue; WAIVER OF JURY TRIAL. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAW OF THE STATE OF NEW YORK. Without limiting the general applicability of the foregoing and the terms of the other Loan Documents to this Agreement and the parties hereto, the terms of Sections 10.08, 10.14 and 10.15 of the Credit Agreement are incorporated herein by reference, mutatis mutandis, with each reference to the "Borrower" therein (whether express or by reference to the Borrower as a "party" thereto) being a reference to the Grantors, and the parties hereto agree to such terms.
- 7.7. Counterparts. This Agreement may be executed by one or more of the parties to this Agreement on any number of separate counterparts (including by facsimile), and all of said counterparts taken together shall be deemed to constitute one and the same instrument.
- 7.8. Severability. Any provision of this Agreement which is prohibited or unenforceable in any jurisdiction shall, as to such jurisdiction, be ineffective to the extent of such

prohibition or unenforceability without invalidating the remaining provisions hereof, and any such prohibition or unenforceability in any jurisdiction shall not invalidate or render unenforceable such provision in any other jurisdiction.

- 7.9. Section Headings. The Section headings used in this Agreement are for convenience of reference only and are not to affect the construction hereof or be taken into consideration in the interpretation hereof.
- 7.10. <u>Integration</u>. This Agreement and the other Loan Documents represent the agreement of the Grantors, the Administrative Agent and the other Secured Parties with respect to the subject matter hereof and thereof, and there are no promises, undertakings, representations or warranties by any Secured Party relative to subject matter hereof and thereof not expressly set forth or referred to herein or in the other Loan Documents.
 - 7.11. Acknowledgments. Each Grantor hereby acknowledges that:
 - (a) it has been advised by counsel in the negotiation, execution and delivery of this Agreement and the other Loan Documents to which it is a party;
- (b) no Secured Party has any fiduciary relationship with or duty to any Grantor arising out of or in connection with this Agreement or any of the other Loan Documents, and the relationship between the Grantors, on the one hand, and the Secured Parties, on the other hand, in connection herewith or therewith is solely that of debtor and creditor; and
- (c) no joint venture is created hereby or by the other Loan Documents or otherwise exists by virtue of the transactions contemplated hereby among the Secured Parties or among the Grantors and the Secured Parties.
- 7.12. <u>Additional Grantors</u>. Each Subsidiary of the Borrower that is required to become a party to this Agreement pursuant to <u>Section 6.22</u> of the Credit Agreement shall become a Grantor for all purposes of this Agreement upon execution and delivery by such Subsidiary of a Joinder Agreement.

7.13. Releases; Termination of this Agreement.

(a) At such time as the Loans and the other Obligations (other than (i) contingent indemnification obligations and (ii) Obligations in respect of Secured Cash Management Agreements and Secured Hedge Agreements either (A) as to which arrangements satisfactory to the applicable Cash Management Bank or Hedge Bank shall have been made or (B) notice has not been received by the Administrative Agent from the applicable Cash Management Bank or Hedge Bank that such amounts are then due and payable) shall have been paid in full, the Commitments under the Credit Agreement have been terminated or expired and each Letter of Credit issued under the Credit Agreement shall be Cash Collateralized or no longer outstanding (other than Letters of Credit as to which other arrangements satisfactory to the Administrative Agent and the applicable L/C Issuer shall have been made), the Collateral shall be released from the Liens created hereby, and this Agreement and all obligations (other than those expressly stated to survive such termination) of the Administrative Agent and each Grantor hereunder shall terminate, all without delivery of any instrument or performance of any

act by any party, and all rights to the Collateral shall revert to the Grantors. At the request and sole expense of any Grantor following any such termination, the Administrative Agent shall deliver to such Grantor any Collateral held by the Administrative Agent hereunder, and execute and deliver to such Grantor such documents as such Grantor shall reasonably request to evidence such termination.

- (b) If any of the Collateral shall be sold or otherwise disposed of by any Grantor in a transaction permitted by the Credit Agreement, then the Administrative Agent, at the request and sole expense of such Grantor, shall execute and deliver to such Grantor all releases or other documents reasonably necessary for the release of the Liens created hereby on such Collateral. At the request and sole expense of the Borrower, a Grantor shall be released from its obligations hereunder in the event that all the Stock and Stock Equivalents in such Grantor shall be sold or otherwise disposed of in a transaction permitted by the Credit Agreement; provided that the Borrower shall have delivered to the Administrative Agent, at least three Business Days (or such lesser period permitted in writing by the Administrative Agent) prior to the date of the proposed release, a written request for such release identifying the relevant Grantor and the terms of the relevant sale or other disposition in reasonable detail, including the price thereof and any expenses incurred in connection therewith, together with a certification by the Borrower stating that such transaction is in compliance with the Credit Agreement and the other Loan Documents.
- (c) After the occurrence and during the continuance of Collateral Release Event and in accordance with Section 10.19(a) of the Credit Agreement, the Administrative Agent, at the request and sole expense of the Borrower, shall promptly execute and deliver to the Borrower all releases and other documents, and take such other action, reasonably necessary for the release of the Liens created hereby or by any other Security Instrument on the applicable Collateral
- (d) Each Grantor acknowledges that it is not authorized to file any financing statement or amendment or termination statement with respect to any financing statement originally filed in connection herewith without the prior written consent of the Administrative Agent, subject to such Grantor's rights under Sections 9-509(d) (2) and 9-518 of the New York UCC.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, each of the undersigned has caused this Pledge and Security Agreement to be duly executed and delivered as of the date first above written.

BABCOCK AND WILCOX INVESTMENT COMPANY

/s/ James C. Lewis Bv: Name: James C. Lewis Title: Treasurer

AMERICON EQUIPMENT SERVICES, INC.

AMERICON, INC.

APPLIED SYNERGISTICS, INC.

BABCOCK & WILCOX CHINA HOLDINGS, INC.

BABCOCK & WILCOX CONSTRUCTION CO., INC.

BABCOCK & WILCOX DENMARK HOLDINGS, INC.

BABCOCK & WILCOX EBENSBURG POWER, INC.

BABCOCK & WILCOX EQUITY INVESTMENTS, INC.

BABCOCK & WILCOX INTERNATIONAL SALES AND

SERVICE CORPORATION

BABCOCK & WILCOX INTERNATIONAL, INC.

BABCOCK & WILCOX MODULAR NUCLEAR ENERGY LLC

BABCOCK & WILCOX NUCLEAR ENERGY, INC.

BABCOCK & WILCOX POWER GENERATION GROUP, INC.

BABCOCK & WILCOX TECHNOLOGY, INC.

DELTA POWER SERVICES, LLC

DIAMOND OPERATING CO., INC.

DIAMOND POWER AUSTRALIA HOLDINGS, INC.

DIAMOND POWER CHINA HOLDINGS, INC.

DIAMOND POWER EQUITY INVESTMENTS, INC.

DIAMOND POWER INTERNATIONAL, INC.

DPS BERKELEY, LLC

DPS CADILLAC, LLC

DPS FLORIDA, LLC

DPS GREGORY, LLC

DPS LOWELL COGEN, LLC

DPS MECKLENBURG, LLC

DPS MICHIGAN, LLC

DPS MOJAVE, LLC

DPS SABINE, LLC

INTECH, INC

INVEY-COOPER SERVICES, L.L.C.

O&M HOLDING COMPANY

PALM BEACH RESOURCE RECOVERY CORPORATION

POWER SYSTEMS OPERATIONS, INC.

REVLOC RECLAMATION SERVICE, INC.

SOFCO - EFS HOLDINGS LLC

By: /s/ James C. Lewis Name: James C. Lewis Title: Treasurer

NATIONAL ECOLOGY CAMPANY

By: /s/ James C. Lewis James C. Lewis Name: Title: Authorized Signatory

Babcock and Wilcox Pledge & Security Agreement

BANK OF AMERICA, as Administrative Agent

By: /s/ Bridgett J. Manduk Name: Bridgett J. Manduk Title: Assistant Vice President

Babcock and Wilcox Pledge & Security Agreement

EXHIBIT A

NOTICE OF GRANT OF SECURITY INTEREST IN PATENTS

United States Patent and Trademark Office

Ladies and Gentlemen:

Please be advised that pursuant to the Pledge and Security dated as of May 3, 2010 (as amended, restated, supplemented or otherwise modified from time to time, the "Agreement") among the Grantor (as defined below), the other grantors party thereto and the Administrative Agent for the Secured Parties referenced therein, the undersigned Grantor has granted a continuing security interest in and continuing lien upon the patents and patent applications on Schedule 1 to the Administrative Agent for the ratable benefit of the Secured Parties.

The Grantors and the Administrative Agent, on behalf of the Secured Parties, hereby acknowledge and agree that the security interest in such patents and patent applications (a) may only be terminated in accordance with the terms of the Agreement and (b) is not to be construed as an assignment of any patent or patent application.

| GRANTOR: | Very truly yours, | | |
|-----------------------|----------------------------|--|--|
| [Address] | [GRANTOR] | | |
| | By: Name: | | |
| | Title: | | |
| ADMINISTRATIVE AGENT: | Acknowledged and accepted: | | |
| [Address] | BANK OF AMERICA, N.A., | | |
| | By: | | |
| | Name: | | |
| | Title: | | |
| | Rahcock and Wilcox | | |

Babcock and Wilcox Pledge & Security Agreement SCHEDULE 1

PATENTS

Description of Patent No. Patent Item Date of Patent

PATENT APPLICATIONS

Description of Date of Patent Applications No. Patent Applied for Patent Applications

NOTICE OF GRANT OF SECURITY INTEREST IN TRADEMARKS

United States Patent and Trademark Office

Ladies and Gentlemen:

Please be advised that pursuant to the Pledge and Security dated as of May 3, 2010 (as amended, restated, supplemented or otherwise modified from time to time, the "Agreement") among the Grantor (as defined below), the other grantors party thereto and the Administrative Agent for the Secured Parties referenced therein, the undersigned Grantor has granted a continuing security interest in and continuing lien upon the trademarks and trademark applications on Schedule 1 to the Administrative Agent for the ratable benefit of the Secured Parties.

The Grantors and the Administrative Agent, on behalf of the Secured Parties, hereby acknowledge and agree that the security interest in such trademarks and trademark applications (a) may only be terminated in accordance with the terms of the Agreement and (b) is not to be construed as an assignment of any trademark or trademark application.

| GRANTOR: | Very truly yours, | |
|-----------------------|----------------------------|--|
| [Address] | [GRANTOR] | |
| | By: Name: | |
| | Title: | |
| ADMINISTRATIVE AGENT: | Acknowledged and accepted: | |
| [Address] | BANK OF AMERICA, N.A., | |
| | Ву: | |
| | Name: | |
| | Title: | |

SCHEDULE 1

TRADEMARKS

Description of Trademark No. Trademark Item

Date of Trademark

TRADEMARK APPLICATIONS

Trademark Applications No.

Description of Trademark Applied for Date of Trademark Applications

NOTICE OF GRANT OF SECURITY INTEREST IN COPYRIGHTS

United States Copyright Office

Ladies and Gentlemen:

Please be advised that pursuant to the Pledge and Security dated as of May 3, 2010 (as amended, restated, supplemented or otherwise modified from time to time, the "Agreement") among the Grantor (as defined below), the other grantors party thereto and the Administrative Agent for the Secured Parties referenced therein, the undersigned Grantor has granted a continuing security interest in and continuing lien upon the copyrights and copyright applications on Schedule 1 to the Administrative Agent for the ratable benefit of the Secured Parties.

The Grantors and the Administrative Agent, on behalf of the Secured Parties, hereby acknowledge and agree that the security interest in such copyrights and copyright applications (a) may only be terminated in accordance with the terms of the Agreement and (b) is not to be construed as an assignment of any copyright or copyright application.

| GRANTOR: | Very truly yours, | |
|-----------------------|----------------------------|--|
| [Address] | [GRANTOR] | |
| | Ву: | |
| | Name: | |
| | Title: | |
| ADMINISTRATIVE AGENT: | Acknowledged and accepted: | |
| [Address] | BANK OF AMERICA, N.A., | |
| | By: | |
| | Name: | |
| | Title: | |

SCHEDULE 1

COPYRIGHTS

Description of Copyright No. Date of Copyright Item

COPYRIGHT APPLICATIONS

Copyright Applications Description of Date of Copyright Applied for Copyright Applications

PERFECTED FIRST PRIORITY LIENS

UCC Filings

A UCC1 Financing Statement listing each Grantor, as debtor, and the Collateral Agent, as secured party, should be filed in the applicable governmental offices set forth below.

Jurisdiction of Filing Grantor Americon Equipment Services, Inc. Delaware Secretary of State Delaware Secretary of State Americon, Inc. Applied Synergistics, Inc. Delaware Secretary of State Babcock & Wilcox China Holdings, Inc. Delaware Secretary of State Babcock & Wilcox Construction Co., Inc. Delaware Secretary of State Babcock & Wilcox Denmark Holdings, Inc. Delaware Secretary of State Babcock & Wilcox Ebensburg Power, Inc. Delaware Secretary of State Babcock & Wilcox Equity Investments, Inc. Delaware Secretary of State Babcock & Wilcox International Sales and Service Corporation Delaware Secretary of State Babcock & Wilcox International, Inc. Delaware Secretary of State Babcock & Wilcox Investment Company Delaware Secretary of State Babcock & Wilcox Modular Nuclear Energy LLC Delaware Secretary of State Babcock & Wilcox Nuclear Energy, Inc. Delaware Secretary of State Babcock & Wilcox Power Generation Group, Inc. Delaware Secretary of State Babcock & Wilcox Technology, Inc. Delaware Secretary of State Delaware Secretary of State Delta Power Services, LLC Diamond Operating Co., Inc. Delaware Secretary of State Diamond Power Australia Holdings, Inc. Delaware Secretary of State Diamond Power China Holdings, Inc. Delaware Secretary of State Diamond Power Equity Investments, Inc. Delaware Secretary of State Delaware Secretary of State Diamond Power International, Inc. DPS Berkeley, LLC Delaware Secretary of State DPS Cadillac, LLC Delaware Secretary of State DPS Florida, LLC Delaware Secretary of State DPS Gregory, LLC Delaware Secretary of State DPS Lowell Cogen, LLC Delaware Secretary of State DPS Mecklenburg, LLC Delaware Secretary of State DPS Michigan, LLC Delaware Secretary of State DPS Mojave, LLC Delaware Secretary of State DPS Sabine, LLC Delaware Secretary of State Tennessee Secretary of State Intech, Inc. Ivey-Cooper Services, L.L.C. Tennessee Secretary of State National Ecology Company Delaware Secretary of State O&M Holding Company Delaware Secretary of State Palm Beach Resource Recovery Corporation Florida Secured Transaction Registry Power Systems Operations, Inc. Delaware Secretary of State Revloc Reclamation Service, Inc. Delaware Secretary of State SOFCo - EFS Holdings LLC Delaware Secretary of State

Schedule 3.3 – 1

To perfect the Lien in Fixtures, a UCC1 Financing Statement listing the applicable Grantor, as debtor, and the Collateral Agent, as secured party, should be filed in the real property records of the county in which such Fixtures are located. Each such UCC1 Financing Statement will need to include a legal description of the real property upon which such Fixtures are located.

Actions with respect to Pledged Securities

The original of all Pledged Securities evidence by either a Certificated Security or Instrument should be delivered to the Collateral Agent, together with an undated stock or note power, as applicable, duly executed in blank by the applicable Grantor.

Actions with respect to Patents and Trademarks

For Collateral consisting of Patents and Trademarks, the applicable Grantors should execute an Intellectual Property Security Agreement, and such agreement should be recorded with the United States Patent and Trademark Office.

Actions with respect to Copyrights

For Collateral consisting of Copyrights, the applicable Grantors should execute an Intellectual Property Security Agreement, and such agreement should be recorded with the United States Copyright Office.

Schedule 3.3 - Page 2

NAME; JURISDICTION OF ORGANIZATION, ETC.

| Exact Legal Name of Grantor | Type of Organization | Jurisdiction of Organization | Chief Executive Office | Organizational Identification Number |
|--|---------------------------|------------------------------------|---|--|
| Americon Equipment Services, Inc. | Corporation | Delaware | 74 Robinson Avenue Barberton, OH 44203 | 2077236 |
| Americon, Inc. | Corporation | Delaware | 74 Robinson Avenue Barberton, OH 44203 | 2058172 |
| Applied Synergistics, Inc. | Corporation | Delaware | 1019 Dillard Drive Lynchburg, VA 24502-0158 | 3596853 |
| Babcock & Wilcox China Holdings, Inc. | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 3563042 |
| Babcock & Wilcox Construction Co., Inc. | Corporation | Delaware | 74 Robinson Avenue Barberton, OH 44203 | 2058181 |
| Babcock & Wilcox Denmark Holdings, Inc. | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 3563041 |
| Babcock & Wilcox Ebensburg Power, Inc. | Corporation | Delaware | 20 S. Van Buren Avenue Barberton, OH 44203 | 2109431 |
| Babcock & Wilcox Equity Investments, Inc. | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 2050183 |
| Babcock & Wilcox International Sales and Service Corporation | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 0795023 |
| Babcock & Wilcox International, Inc. | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 0914783 |
| Babcock & Wilcox Investment Company | Corporation | Delaware | 800 Main Street Lynchburg, VA 24504* | 2235817 |
| Babcock & Wilcox Modular Nuclear Energy LLC | Limited Liability Company | Delaware | 2016 Mount Athos Road Lynchburg, VA 24504-5447 | 4697877 |
| Babcock & Wilcox Nuclear Energy, Inc. | Corporation | Delaware | 2016 Mount Athos Road Lynchburg, VA 24504-5447 | 4293945 |
| Babcock & Wilcox Power Generation Group, Inc. | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 0847234 |
| Babcock & Wilcox Technology, Inc. | Corporation | Delaware | 2016 Mount Athos Road Lynchburg, VA 24504-5447 | 2725506 |
| Delta Power Services, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 3363308 |

Schedule 3.4 – Page 1

| Exact Legal Name of Grantor | Type of Organization | Jurisdiction of Organization | Chief Executive Office | Organizational Identification Number |
|---|---------------------------|------------------------------------|---|--|
| Diamond Operating Co., Inc. | Corporation | Delaware | 185 Great Valley Parkway Malvern, PA 19355-1321 | 3498044 |
| Diamond Power Australia Holdings, Inc. | Corporation | Delaware | 2600 E. Main Street Lancaster, OH 43130 | 3563045 |
| Diamond Power China Holdings, Inc. | Corporation | Delaware | 2600 E. Main Street Lancaster, OH 43130 | 3563044 |
| Diamond Power Equity Investments, Inc. | Corporation | Delaware | 2600 E. Main Street Lancaster, OH 43130 | 3563043 |
| Diamond Power International, Inc. | Corporation | Delaware | 2600 E. Main Street Lancaster, OH 43130 | 2725505 |
| DPS Berkeley, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 3685555 |
| DPS Cadillac, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 4111828 |
| DPS Florida, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 4050781 |
| DPS Gregory, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 3879537 |
| DPS Lowell Cogen, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 4060363 |
| DPS Mecklenburg, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 3860060 |
| DPS Michigan, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 3768604 |
| DPS Mojave, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 3950476 |
| DPS Sabine, LLC | Limited Liability Company | Delaware | 363 N. Sam Houston Parkway E, Suite 350 Houston, TX 77060 | 4138849 |
| Intech, Inc. | Corporation | Tennessee | 2802 Belle Arbor Drive Chattanooga, TN 37406 | 282042 |
| Ivey-Cooper Services, L.L.C | Limited Liability Company | Tennessee | 2815 Belle Arbor Drive Chattanooga, TN 37406 | 0401423 |
| National Ecology Company | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 2209966 |

Schedule 3.4 – Page 2

| Exact Legal Name of Grantor | Type of Organization | Jurisdiction of Organization | Chief Executive Office | Organizational Identification Number |
|---|---------------------------|------------------------------------|--|--|
| O&M Holding Company | Corporation | Delaware | 20 S. Van Buren Ave. Barberton, OH 44203-0351 | 4568185 |
| Palm Beach Resource Recovery Corporation | Corporation | Florida | 6101 West 45th Street West Palm Beach, FL 33410 | H27351 |
| Power Systems Operations, Inc. | Corporation | Delaware | 20 S. Van Buren Avenue Barberton, OH 44203-0351 | 2074007 |
| Revloc Reclamation Service, Inc. | Corporation | Delaware | 20 S. Van Buren Avenue Barberton, OH 44203-0351 | 2234938 |
| SOFCo – EFS Holdings LLC | Limited Liability Company | Delaware | 20 S. Van Buren Ave. Barberton, Ohio 44203 | 3494936 |

^{*} After the effective date of the Spinoff, the Chief Executive Office will relocate to The Harris Building, 13024 Ballantyne Corporate Place, Suite 700, Charlotte, North Carolina 28277

Prior Names during last 5 years:

On November 20, 2007, The Babcock & Wilcox Company changed its name to Babcock & Wilcox Power Generation Group, Inc.

On November 20, 2007, The Babcock & Wilcox Companies changed its name to The Babcock & Wilcox Company, and then on March 8, 2010, changed its name to Babcock & Wilcox Investment Company

On April 20, 2010, McDermott Technology, Inc. changed its name to Babcock & Wilcox Technology, Inc.

On April 29, 2010, Babcock & Wilcox Nuclear Power Generation Group, Inc. changed its name to Babcock & Wilcox Nuclear Energy, Inc.

Prior Addresses during last 5 years:

The former chief executive office of American Equipment Services, Inc. was located at: 3333 Copley Road, Copley, OH 44321.

The former chief executive office of the following entities was located at 90 E. Tuscarawas Ave., Barberton, OH 44203:

Americon, Inc. Babcock & Wilcox Construction Co., Inc.

Schedule 3.4 - Page 3

The former chief executive office of the following entities was located at 1450 Poydras Street, New Orleans, LA 70112:

Babcock & Wilcox Equity Investments, Inc.

Babcock & Wilcox Holdings, Inc.

Babcock & Wilcox International Sales and Service Corporation

Babcock & Wilcox International, Inc.

Babcock & Wilcox Investment Company

Babcock & Wilcox Power Generation Group, Inc.

The former chief executive office of the following entities was located at 1415 Louisiana Street, Suite 3500, Houston, TX 77002:

Delta Power Services, LLC DPS Cadillac, LLC DPS Florida, LLC DPS Mojave, LLC

The former chief executive office of the following entities was located at 67 Park Place East, Morristown, NJ 07960:

DPS Berkeley, LLC DPS Michigan, LLC

The former chief executive office of the following entities was located at 7022 Wild Violet Drive, Humble, TX 77346:

DPS Gregory, LLC DPS Mecklenburg, LLC

The former chief executive office of DPS Lowell Cogen, LLC was located at: 282 Western Avenue, Lowell, MA 01852

The former chief executive office of DPS Sabine, LLC was located at: 4647 FM1006 at: 4647 FM 1006 at Lanxess Plant, Orange, TX

Schedule 3.4 - Page 4

INVENTORY AND EQUIPMENT

Grantor
Babcock and Wilcox Power Generation Group, Inc.

Babcock and Wilcox Power Generation

Group, Inc.
Babcock and Wilcox Power Generation Group, Inc.

Babcock and Wilcox Power Generation Group, Inc.

Diamond Power International, Inc.

Address/City/State/Zip Code 20 South Van Buren Ave Barberton, OH 44203 East Half Mile St. West Point, MS 39773 3333 Copley Road Copley, OH 44202 600 Ford Avenue West Barberton, OH 44203 2600 E. Main Street Lancaster, OH 43130

INVESTMENT PROPERTY

Pledged Stock:

| Grantor | Issuer | Type of Organization | # of Shares Owned | Total Shares Outstanding | % of Interest Pledged ¹ | Certificate No. | Par Value per share | |
|--|---|-------------------------|-------------------------|--------------------------------|--|--------------------|------------------------------|--|
| Americon, Inc. | Americon Equipment Services, Inc. | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 | |
| Babcock & Wilcox Power Generation Group, Inc. | Americon, Inc. | Corporation | 100 | 100 | 100% | 2 | \$10.00 | |
| Diamond Power International, Inc. | Applied Synergistics, Inc. | Corporation | 1,000 | 1,000 | 100% | 2 | 1.00 | |
| Babcock & Wilcox Power Generation Group, Inc. | Babcock & Wilcox Canada Ltd. ² | Corporation | 500,000 | 500,000 | 65% | 1 | None | |
| Babcock & Wilcox Equity Investments, Inc. | Babcock & Wilcox China Holdings, Inc. | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 | |
| Americon, Inc. | Babcock & Wilcox Construction Co., Inc. | Corporation | 100 | 100 | 100% | 2 | \$10.00 | |
| Babcock & Wilcox Equity Investments, Inc. | Babcock & Wilcox Denmark Holdings, Inc. | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 | |

¹Notwithstanding any Grantor's delivery to the Collateral Agent of any Certificated Security evidencing more than 65% of the Voting Stock of any Foreign Subsidiary, the Collateral does not include any Excluded Stock.

Schedule 3.7 - Page 1

² The stock certificate lists the issuing entity as Babcock & Wilcox Industries Ltd., which is the prior legal name of Babcock & Wilcox Canada Ltd.

| Grantor | Issuer | Type of Organization | # of Shares Owned | Total Shares Outstanding | % of Interest Pledged ¹ | Certificate No. | Par Value per share |
|---|--|-------------------------|-------------------------|--------------------------------|--|--------------------|------------------------------|
| Babcock & Wilcox Equity Investments, Inc. | Babcock & Wilcox Ebensburg Power, Inc. | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |
| Babcock & Wilcox Power Generation Group, Inc. | Babcock & Wilcox Equity Investments, Inc. | Corporation | 1,000 | 1,000 | 100% | 3 | \$1.00 |
| Babcock & Wilcox Investment Company ³ | Babcock & Wilcox India Private Limited | Corporation | 675,000 | 675,020 | 65% | 05 | 10 Indian Rupees |
| Babcock & Wilcox International, Inc. | | | 20 | | 65% | 03 & 04 | 10 Indian Rupees |
| Babcock & Wilcox Power Generation Group, Inc. | Babcock & Wilcox International Sales and Service Corporation | Corporation | 1,000 | 1,000 | 100% | 5 | \$1.00 |
| Babcock & Wilcox Power Generation Group, Inc. | Babcock & Wilcox International, Inc. | Corporation | 1,000 | 1,000 | 100% | 3 | \$10,000 |

Formerly known as "The Babcock & Wilcox Company".

Schedule 3.7 – Page 2

| Grantor | Issuer | Type of Organization | # of Shares Owned | Total Shares Outstanding | % of Interest Pledged ¹ | Certificate No. | Par Value per share |
|---------------------------------|-----------------------------|-------------------------|-------------------------|--------------------------------|--|--------------------|------------------------------|
| Babcock & | Babcock & | Corporation | 1,000 | 1,000 | 100% | 3 | \$1.00 |
| Wilcox | Wilcox Nuclear | | | | | | |
| Investment Company | Energy, Inc. | | | | | | |
| (6 | (formerly, | | | | | | |
| (formerly, The Babcock & | Babcock & Wilcox Nuclear | | | | | | |
| Wilcox Company) | | | | | | | |
| wheek Company) | Group, Inc.) | | | | | | |
| Babcock & | Babcock & | Corporation | 100,100 | 100,100 | 100% | 5 | \$1.00 |
| Wilcox | Wilcox Power | corporation | 100,100 | 100,100 | 10070 | | Ψ1.00 |
| Investment | Generation | | | | | | |
| Company | Group, Inc. | | | | | | |
| (formerly, The | (formerly, The | | | | | | |
| Babcock & | Babcock & | | | | | | |
| Wilcox Company) | | | | | | | |
| Babcock & | Babcock & | Corporation | 1,000 | 1,000 | 100% | 5 | \$1.00 |
| Wilcox | Wilcox | 1 | , | , | | | |
| Investment Company | Technology, Inc. | | | | | | |
| • • | (formerly, | | | | | | |
| (formerly, The | McDermott | | | | | | |
| Babcock & | Technology, Inc.) | | | | | | |
| Wilcox Company) | | | | | | | |
| Babcock & | Babcock & | Corporation | 100,000 | 100,000 | 65% | uncertificated | 100 |
| Wilcox Denmark | Wilcox Volund | | | | | | Danish Kroners |
| Holdings, Inc. Diamond Power | ApS Diamond | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |
| International, Inc. | Operating Co., | Corporation | 1,000 | 1,000 | 10070 | 1 | φ1.00 |
| micmational, mc. | Inc. | | | | | | |
| Diamond Power | Diamond Power | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |
| Equity | Australia | £ | × ** | , . | | | |
| Investments, Inc. | Holdings, Inc. | | | | | | |
| | | | | | | | |

Schedule 3.7 – Page 3

| Grantor | Issuer | Type of Organization | # of Shares Owned | Total Shares Outstanding | % of Interest Pledged ¹ | Certificate No. | Par Value per share |
|--|-----------------------------|-------------------------|-------------------------|--------------------------------|--|--------------------|------------------------------|
| Diamond Power | Diamond Power | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |
| Equity | China Holdings, | | | | | | |
| Investments, Inc. | Inc. | | | | | | |
| Diamond Power | Diamond Power | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |
| International, Inc. | Equity Investments, Inc. | | | | | | |
| Babcock & | Diamond Power | Corporation | 1,000 | 1,000 | 100% | 2 | \$1.00 |
| Wilcox Power Generation Group, Inc. | International, Inc. | | | | | | |
| Diamond Power | Diamond Power | Corporation | 623 Ordinary shares | | 100% | 21 & 22 | 3.700 |
| International, Inc. | Services S.E.A. | | | shares and 496** | | | Thai |
| | Ltd. | | | Preference shares | | | Baht |
| Diamond Power | Diamond Power | Corporation | 1 | 1 | 65% | 2 | 1.00 |
| International, Inc. | Specialty | | | | | | South African |
| | (Proprietary) Limited | | | | | | Rand |
| Diamond Power | Diamond Power | Corporation | 500,000 | 500,000 | 65% | 14 & 21 | 1.00 |
| International, Inc.4 | | Corporation | 300,000 | 300,000 | 0370 | 14 & 21 | Pound |
| micmational, me. | specially Emilieu | | | | | | Sterling |
| Babcock & | Intech, Inc. | Corporation | 2,500 | 2,500 | 100% | 15 | No par |
| Wilcox Nuclear | | Corporation | 2,500 | 2,000 | 10070 | 15 | rio par |
| Energy, Inc. | | | | | | | |
| | | | | | | | |
| (formerly, Babkcock & Wilcox Nuclear Power Generation Group, Inc.) | | | | | | | |
| Babcock & | National Ecology | Corporation | 1,000 | 1,000 | 100% | 4 | \$1.00 |
| Wilcox Power | Company | | | | | | |
| Generation | | | | | | | |
| Group, Inc. | | | | | | | |

The stock certificates list the registered proprietor as Babcock & Wilcox International Sales and Service Corporation. Certificates have not been reissued since the transfer of ownership to Diamond Power International, Inc.

Schedule 3.7 – Page 4

| Grantor | Issuer | Type of Organization | # of Shares Owned | Total Shares Outstanding | % of Interest Pledged ¹ | Certificate No. | Par Value per share |
|--|---|-------------------------|-------------------------|--------------------------------|--|--------------------|------------------------------|
| Babcock & Wilcox Power Generation Group, Inc. | North County Recycling, Inc. | Corporation | 300 | 300 | 100% | 5 | \$1.00 |
| Power Systems Operations, Inc. | O&M Holding Company | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |
| Babcock & Wilcox Equity Investments, Inc. | P. T. Babcock & Wilcox Asia | Corporation | 792 | 800 | 65% | 0001 - 0297 | \$500.00 |
| Babcock & Wilcox International Sales and Service Corporation | | | 8 | | 65% | 0298 - 0300 | \$500.00 |
| Power Systems Operations, Inc. | Palm Beach Resource Recovery Corporation | Corporation | 60 | 60 | 100% | 2 | No par |
| Babcock & Wilcox Power Generation Group, Inc. | Power Systems Operations, Inc. | Corporation | 1,000 | 1,000 | 100% | 2 | \$10.00 |
| Power Systems Operations, Inc. | Revloc Reclamation Service, Inc. | Corporation | 1,000 | 1,000 | 100% | 1 | \$1.00 |

Footnotes

Schedule 3.7 - Page 5

^{**} The 623 Ordinary shares owned by Diamond Power International constitute a majority of the Ordinary shares, but only account for a 48.67% effective ownership. Phothi-Ratana Engineering Co., Ltd. owns all 496 preference shares, which constitute a 50.94% effective ownership.

Pledged LLC Interests:

| Grantor | Issuer | Type of Organization | # of Shares Owned | Total Shares Outstanding | % of Interest Pledged ⁵ | Certificate No. | Par Value per share |
|--|--|---------------------------------------|-------------------------|--------------------------------|--|--------------------|------------------------------|
| Babcock & Wilcox Investment Company | Babcock & Wilcox Modular Nuclear Energy LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| O&M Holding Company | Delta Power Services, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| Diamond Power International, Inc. | Diamond Power do Brasil Limitada | Limited Liability Company (Brazil) | 297,000 quotas | 300,000 quotas | 65% | uncertificated | 1.00 Brazilian Real |
| Babcock & Wilcox International Sales and Service Corporation | | | 3,000 quotas | | 65% | | |
| Delta Power Services, LLC | DPS Berkely, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| Delta Power Services, LLC | DPS Cadillac, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| Delta Power Services, LLC | DPS Florida, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| Delta Power Services, LLC | DPS Gregory, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| Delta Power Services, LLC | DPS Lowell Cogan, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |
| Delta Power Services, LLC | DPS Mecklenburg, LLC | Limited Liability Company | N/A | N/A | 100% | uncertificated | N/A |

⁵ Notwithstanding any Grantor's delivery to the Collateral Agent of any Certificated Security evidencing more than 65% of the Voting Interest of any Foreign Subsidiary, the Collateral does not include any Excluded Stock.

 $Schedule\ 3.7-Page\ 6$

| | | Type of | # of Shares | Total Shares | % of Interest | Certificate | Par Value per |
|-------------------|------------------|-------------------|----------------|-----------------|----------------------|----------------|---------------------|
| Grantor | Issuer | Organization | Owned | Outstanding | Pledged ⁵ | No. | share |
| Delta Power | DPS Michigan, | Limited Liability | N/A | N/A | 100% | uncertificated | N/A |
| Services, LLC | LLC | Company | | | | | |
| Delta Power | DPS Mojave, | Limited Liability | N/A | N/A | 100% | uncertificated | N/A |
| Services, LLC | LLC | Company | | | | | |
| Delta Power | DPS Sabine, LLC | Limited Liability | N/A | N/A | 100% | uncertificated | N/A |
| Services, LLC | | Company | | | | | |
| Babcock & | Ivey Cooper | Limited Liability | N/A | N/A | 100% | uncertificated | N/A |
| Wilcox Equity | Services, L.L.C. | Company | | | | | |
| Investments, Inc. | | | | | | | |
| Babcock & | SOFCo-EFS | Limited Liability | N/A | N/A | 100% | uncertificated | N/A |
| Wilcox | Holdings, LLC | Company | | | | | |
| Technology, Inc. | | | | | | | |

Pledged Partnership Interests:

Babcock & Wilcox Ebensburg Power, Inc. owns a 49.5% general partnership interest in Ebensburg Power Company (which general partnership interest is uncertificated). Babcock & Wilcox Ebensburg Power, Inc. owns a 1% general partnership interest in Ebensburg Investors Limited Partnership (an unaffiliated third party), which in turn owns the other 50.5% general partnership interest in Ebensburg Power Company.

<u>Pledged Trust Interests:</u>

None.

Pledged Notes:

Global Intercompany Note, dated as of the Closing Date

<u>Pledged Commodities Contracts</u>:

None.

 $Schedule\ 3.7-Page\ 7$

INTELLECTUAL PROPERTY

(A) Material Copyright Licenses

None.

(B) Copyrights

See "Babcock Copyright Listing" chart attached hereto.

- (C) Material Patent Licenses
 - 1. License between Siemens AG of Erlangen, Germany and Babcock and Wilcox Power Generation Group, Inc. for the Benson Boiler Technology.
 - 2. License between Niro A/S of Soeborg, Denmark and Babcock and Wilcox Power Generation Group, Inc. for the Dry Flue Gas Desulfurization Technology.
 - 3. License between Babcock & Wilcox Power Generation Group, Inc. and Enviroserv GmbH for Circulating Fluidized Bed Flue Gas Desulfurization.
 - 4. License between Babcock & Wilcox Power Generation Group, Inc. and MeadWestvaco Corporation for Overfire Air System for Enhanced Black Liquor Combustion.
 - 5. License between Babcock & Wilcox Power Generation Group, Inc. and AEP ProServ Inc. for SO3 Technology that Utilizes Trona as the Reagent.
- (D) Patents

See "Active U.S. Patent Cases" charts attached hereto.

(E) Material Trademark Licenses

None.

(F) Trademarks

See "U.S. Trademarks and Domain Names" charts attached hereto.

(G) Material Trade Secret Licenses

None.

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|-------------|-----------|----------|---------|
| The Babcock & Wilcox Company | | | | |
| 544083E-01 - Technical Drawing | VAU-627-668 | 8-Jun-04 | BWSC | B&W |
| 544084E-01 - Technical Drawing | VAU-632-091 | 8-Jun-04 | BWSC | B&W |
| 544085E-01 - Technical Drawing | VAU-641-537 | 8-Jun-04 | BWSC | B&W |
| 544086E-01 - Technical Drawing | VAU-627-669 | 8-Jun-04 | BWSC | B&W |
| 5440873-01 - Technical Drawing | VAU-627-667 | 8-Jun-04 | BWSC | B&W |
| 544088E-01 - Technical Drawing | VAU-641-538 | 8-Jun-04 | BWSC | B&W |
| 544497E-00 - Technical Drawing | VAU-632-111 | 24-Jun-04 | BWSC | B&W |
| ACOUSTIC LEAK LOCATOR ALL USERS MANUAL | TX4-658-909 | 4-Nov-97 | SIDT | B&W |
| ACOUSTIC LEAK LOCATOR ALLVIEW SOFTWARE USERS MANUAL | TX4-665-982 | 4-Nov-97 | SIDT | B&W |
| ACTIGRAF | TXU-487-508 | 11-Jun-91 | EISD | B&W |
| ADLPIPE | 590-486 | 1-Dec-80 | PSI | B&W |
| AET 5500 OPERATOR'S GUIDE | TX4-658-931 | 4-Nov-97 | SIDT | B&W |
| AGASS | TX2-953-534 | 19-Nov-90 | FPD | B&W |
| APPLIED ENERGY SERVICES DEEPWATER COGENERATION PROJECT UNIT ONE, PASADENA, TX | 198-896 | 19-Aug-85 | PSI | B&W |
| ASSOCIATED ELECTRIC COOPERATIVE, THOMAS HILL UNIT CO. 3, MOBERLY, MISSOURI | VA-93-348 | 22-Oct-81 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010

Page 56 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|-------------|-----------|----------|---------|
| ASSURED STOCK PROGRAM TM (BOOKLET) | TX2-114-746 | 17-Jul-87 | DFO | B&W |
| ASSURED STOCK PROGRAM TM (BROCHURE) | TX2-113-817 | 17-Jul-87 | DFO | B&W |
| AUSTRALIA IRON & STEEL PTY., LTD, PORT KEMBLA WORKS, PORT KEMBLA, N.S.W. AUSTRALIA | VA-104-544 | 19-Jul-82 | FPD | B&W |
| B.F. GOODRICH, CHEMICAL GROUP, LOUISVILLE, KENTUCKY | VA-93-346 | 22-Oct-81 | FPD | B&W |
| BABCOCK & WILCOX, EL PULVERIZER MANUAL, DESCRIPTION, OPERATION, CHECKS | TX1-145-814 | 5-Jul-83 | FPD | B&W |
| BABCOCK & WILCOX, EL-56 PULVERIZER MANUAL DESCRIPTION, OPERATION, CHECKS | TX1-05-595 | 21-Apr-83 | FPD | B&W |
| BAR/TUBE HOT GAGE OPERATORS GUIDE | TXU-298-622 | 24-Sep-87 | TPD | B&W |
| BAR/TUBE MILL HOT GAGE | TXU-298-623 | 24-Sep-87 | TPD | B&W |
| BIG RIVERS ELECTRIC CORPORATION, REID STATION – UNIT NO. 2, SEBREE, KENTUCKY | 454-903 | 4-Mar-80 | FPD | B&W |
| BOARD OF MUNICIPAL UTILITIES, SIKESTON POWER STATION UNIT 1 | 505-125 | 18-Jun-80 | FPD | B&W |
| BOWATER CAROLINA COMPANY, NO. 3 RECOVERY BOILER, CATAWBA, SOUTH CAROLINA | VA-130-282 | 5-Jul-83 | FPD | B&W |
| BYPASS FLOW SCHEMATIC | VA-138-736 | 27-Oct-83 | FPD | B&W |
| BYPASS FLOW SCHEMATIC, INTERMOUNTAIN POWER PROJECT | VA-145-245 | 9-Jan-84 | FPD | B&W |
| CAJUN ELECTRIC POWER COOPERATIVE, INC., BIG CAJUN NO. 2, POWER STATION UNIT NO. 3, NEW ROADS, LOUISIANA | VA-116-843 | 6-Dec-82 | FPD | B&W |
| CENTRAL ILLINOIS PUBLIC SERVICE COMPANY, COFFEEN POWER STATION, UNIT NO. 1, | | | | |
| COFFEEN, ILLINOIS | VA-122-776 | 28-Mar-83 | FPD | B&W |
| CENTRAL ILLINOIS PUBLIC SERVICE COMPANY, COFFEEN POWER STATION, UNIT NO. 2, COFFEEN, ILLINOIS | VA-122-777 | 28-Mar-83 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010 Page 57 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| CERA-VAM CERAMIC LINING EXTENDS WEAR LIFE OF COAL- HANDLING SYSTEMS | TX-795-956 | 4-Nov-81 | FPD | B&W |
| CINCINNATI GAS & ELECTRIC COMPANY, THE; DAYTON POWER & LIGHT COMPANY, THE; EAST BEND STATION, UNIT NO. 2, RABBIT, KENTUCKY (RB555) | VA-198-895 | 19-Aug-85 | FPD | B&W |
| COMMONWEALTH EDISON COMPANY, KINCAID GENERATING STATION, UNIT NOS. 1&2, KINCAID, ILLINOIS | VA-157-295 | 11-Jun-84 | FPD | B&W |
| CONSUMERS POWER COMPANY, JAMES H. CAMPBELL PLANT, UNIT NO. 2, WEST OLIVE, MICHIGAN | TX-710-115 | 12-May-81 | FPD | B&W |
| COOLING CONTROL PROGRAM FOR SHAPE MELTING, A | TXU-374-472 | 3-Apr-89 | FPD | B&W |
| DAYTON POWER & LIGHT COMPANY, CINCINNATI GAS & ELECTRIC COMPANY, KILLEN ELECTRIC GENERATING STATION | VA-104-897 | 14-May-82 | FPD | B&W |
| DEPARTMENT OF ELECTRIC AND WATER UTILITIES, C.D. MCINTOSH PLANT, UNIT NO. 3, CITY OF LAKELAND, FLORIDA, BABCOCK & WILCOX, RADIANT REHEAT BOILER | TX-827-464 | 7-Jan-82 | FPD | B&W |
| DEPT. OF ELECTRIC AND WATER UTILITIES C.D. MCINTOSH PLAT, UNIT NO. 3 CITY OF LAKELAND, FLORIDA | TX-827-464 | 7-Jan-82 | FPD | B&W |
| DETROIT EDISON COMPANY, THE, BELLE RIVER POWER PLANT, UNITS NOS. 1 & 2, ST. CLAIR, MICHIGAN | VA-145-244 | 9-Jan-84 | FPD | B&W |
| DUQUESNE LIGHT COMPANY ELRAMA POWER STATION, UNIT NO. 3 ELRAMA, PENNSYLVANIA | VA-161-319 | 29-May-84 | UPG | B&W |
| DUQUESNE LIGHT COMPANY ELRAMA POWER STATION, UNITS NO. 1 & 2 ELRAMA, | | | | |
| PENNSYLVANIA | VA-161-318 | 29-May-84 | UPG | B&W |
| DUQUESNE LIGHT COMPANY, ELRAMA POWER STATION, UNIT NO. 4, ELRAMA, PENNSYLVANIA | VA-153-136 | 6-Apr-84 | UPG | B&W |
| EL PULVERIZER MAINTENANCE MANUAL | TX-1-267-005 | 16-Jan-84 | FPD | B&W |
| EL-56 PULVERIZER MAINTENANCE MANUAL FOR CITY OF GRAND HAVEN, BOARD OF POWER & LIGHT, GRAND HAVEN, MICHIGAN | TX-1-105-697 | 21-Apr-83 | FPD | B&W |
| FEATHER RIVER PROJECT, MARYSVILLE, CALIFORNIA, CIRCULATING FLUID BED BOILER | VA-237-840 | 11-Sep-86 | PSI | B&W |
| FEEDWATER TREATMENT FOR RB BOILERS | 504-142 | 3-Jul-80 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010

Page 58 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|--|--------------|-----------|----------|---------|
| FLORIDA POWER CORPORATION, CRYSTAL RIVER PLANT, UNIT NO. 4, CRYSTAL RIVER, | NA 06 701 | 20 M 92 | EDD | D 0 W |
| FLORIDA | VA-96-701 | 29-Mar-82 | FPD | B&W |
| FLUE GAS DESULFURIZATION (FGD) LEXICON | TX-6-849-972 | 16-Oct-07 | FPD | B&W |
| FOSSIL POWER GENERATION DIVISION, BABCOCK & WILCOX, B&W BAG FILTERHOUSE, BROCHURE NO. E101-3066 | TX-621-165 | 30-Jan-81 | FPD | B&W |
| FOSSIL-FUEL-FIRED BOILERS: FUNDAMENTALS AND ELEMENTS * | TXU-405-775 | 16-Feb-90 | FPD | B&W |
| GEORGIA-PACIFIC CORPORATION, CROSSETT ARKANSAS, PROJECT CIP-84-10A BOILER | VA-237-841 | 11-Sep-86 | PSI | B&W |
| GRAPHICS GUIDE | A-721387 | 2-Mar-76 | FPD | B&W |
| GREENWOOD MILLS, EDISTO PLANT, ORANGEBURG, SOUTH CAROLINA | TX-686-378 | 12-May-81 | FPD | B&W |
| HAMMERMILL PAPER COMPANY, ERIE, PENNSYLVANIA, BABCOcK & WILCOX PROCESS RECOVERY BOILER | 587-953 | 25-Nov-80 | FPD | B&W |
| ILLINOIS POWER COMPANY, BALDWIN POWER STATION, UNIT NO. 1, BALDWIN, ILLINOIS | VA-93-347 | 22-Oct-81 | FPD | B&W |
| ILLINOIS POWER COMPANY, BALDWIN POWER STATION, UNIT NO. 2, BALDWIN, ILLINOIS | VA-93-345 | 22-Oct-81 | FPD | B&W |
| INTERMOUNTAIN POWER PROJECT, UNIT NO. 1, LYNNDYL, UTAH | VA-145-242 | 9-Jan-84 | FPD | B&W |
| INTERNATIONAL PAPER COMPANY, MANSFIELD, LOUISIANA, BABCOCK & WILCOX 900 TON BLACK LIQUOR RECOVERY BOILER | TX- 660-220 | 26-Mar-81 | FPD | B&W |
| INTERNATIONAL PAPER COMPANY, MANSFIELD, LOUISIANA, BABCOCK & WILCOX STIRLING POWER BOILER | TX- 741-297 | 22-Jun-81 | FPD | B&W |
| IOWA-ILLINOIS GAS & ELECTRIC COMPANY, LOUISA GENERATING STATION, UNIT NO. 1, LOUISA COUNTY, IOWA | VA-93-349 | 22-Oct-81 | FPD | B&W |
| KANSAS CITY POWER & LIGHT COMPANY, ST. JOSEPH LIGHT & POWER COMPANY, IATAN PLANT – UNIT NO. 1, IATAN, MO, RADIANT REHEAT BOILER TUBE METAL | 405 522 | 10 I 00 | EDD | D e W |
| IDENTIFICATION | 495-522 | 18-Jun-80 | FPD | B&W |
| LOGICAL MACHINE | TX-2-165-288 | 2-Oct-87 | PCC | B&W |

Prepared by: AMSaus, 4/26/2010 Page 59 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| MAINTENANCE TRAINING MANUAL, MPS GEARBOX FOR ESTAL HOOGOVENS BV, IJMUIDEN, HOLLAND | TX-1-105-704 | 21-Apr-83 | FPD | B&W |
| MERCK & CO., INC., STONEWALL PLANT, ELKTON, VIRGINIA | VA-104-280 | 16-Jul-82 | FPD | B&W |
| MICHIGAN SOUTH CENTRAL POWER AGENCY, LITCHFIELD, MICHIGAN, BABCOCK & WILCOX STIRLING POWER BOILER | VA-116-842 | 6-Dec-82 | FPD | B&W |
| MICHIGAN SOUTH CENTRAL POWER AGENCY, LITCHFIELD, MICHIGAN, FLUE GAS SCRUBBER SYSTEM | VA-116-846 | 6-Dec-82 | FPD | B&W |
| MIRROR ALL-METAL REFLECTIVE INSULATION VALVE AND FLANGE COVERS, BROCHURE NO. A952-3006 | 582-823 | 14-Nov-80 | DPS | B&W |
| MONONGAHELA POWER COMPANY, WEST PENN POWER COMPANY, THE POTOMAC EDISON CO., FORT MARTIN UNIT NO. 2, MONONGAHELA, W. VA. B&W UNIVERSAL PRESSURE BOILER | 504-141 | 3-Jul-80 | FPD | B&W |
| MPS PULVERIZER | 504-140 | 3-Jul-80 | FPD | B&W |
| NEBRASKA PUBLIC POWER DISTRICT, GERALD GENTLEMAN STATION, UNIT NO. 2, SUTHERLAND, NEBRASKA, BABCOCK & WILCOX RADIANT REHEAT BOILER | TX-660-219 | 26-Mar-81 | FPD | B&W |
| NETWORK PLOTTING SYSTEM | TX-76-124 | 28-Jul-78 | FPD | B&W |
| NORTHERN INDIANA PUBLIC SERVICE COMPANY, ROLLIN M. SCHAHFER GENERATING STATION, UNIT NO. 14, WHEATFIELD, INDIANA, B&W UNIVERSAL PRESSURE BOILER | VA-116-845 | 6-Dec-82 | FPD | B&W |
| OHIO EDISON COMPANY, THE, W.H. SAMMIS PLANT, UNIT NO. 5, STRATTON, OHIO | VA-198-894 | 19-Aug-85 | FPD | B&W |
| OIL & GAS FIRED BOILER, OPERATIONS AND MAINTENANCE GUIDE | TX-1-255-647 | 17-Oct-83 | FPD | B&W |
| OPERATING INSTRUCTIONS FOR THE AET MODELS 204B AND 204G MINIATURE BATTERY-POWERED ACOUSTIC EMISSIONS SYSTEMS | TX4-658-979 | 4-Nov-97 | SIDT | B&W |
| OPERATOR TRAINING MANUAL CAJUN ELECTRIC POWER COOPERATIVE, INC. NEW ROADS, LOUISIANA | TX-505-639 | 7-Jul-80 | FPD | B&W |
| OPERATOR TRAINING MANUAL EL PULVERIZER MAINTENANCE MANUAL | TX-505-640 | 7-Jul-80 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR APPLETON PAPERS, INCORPORATED, ROARING SPRINGS, PENNSYLVANIA | TX-1-105-703 | 21-Apr-83 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010 Page 60 of 12

| ED ELECTRIC COOPERATIVE, MOBERLY, TX2-126-700 24-Jul-87 DFO B&W |
|---|
| 1112 120 700 2. 00. 01 |
| NEWSPRINT COMPANY, NIXON, GEORGIA TX-1-105-705 21-Apr-83 FPD B&W |
| RICH CHEMICAL COMPANY, LOUISVILLE, TX-787-881 22-Oct-81 FPD B&W |
| /EATCH CONSULTING ENGINEERS TX-868-573 12-Mar-82 FPD B&W |
| CAROLINA COMPANY, CATAWBA, SOUTH TX-868-571 12-Mar-82 FPD B&W |
| ECTRIC POWER COOPERATIVE, INC., NEW TX-1-024-037 6-Dec-82 FPD B&W |
| NCORPORATED, CEDAR RAPIDS PLANT, TX-433-699 4-Mar-80 FPD B&W |
| NCORPORATED, DAYTON, OHIO TX-584-419 18-Nov-80 FPD B&W |
| ILLINOIS PUBLIC SERVICE COMPANY, UNITS TX-1-208-776 17-Oct-83 FPD B&W |
| MES ELECTRIC AMES, IOWA TX-787-882 22-Oct-81 FPD B&W |
| RAND HAVEN, BOARD OF POWER & LIGHT, TX-1-105-706 21-Apr-83 FPD B&W |
| RS POWER COMPANY, WEST OLIVE, TX-2-121-500 24-Jul-87 DFO B&W |
| KANSAS CITY, MISSOURI TX-685-278 8-May-81 FPD B&W |
| EDISON COMPANY, BELLE RIVER POWER TX-1-267-004 16-Jan-84 FPD B&W |
| T DE NEMOURS & CO., FLORENCE, SOUTH TX-1-024-039 6-Dec-82 FPD B&W |
| OWER CORPORATION, UNIT NO. 4, TX-1-147-400 6-Jul-83 FPD B&W |
| TX-787-881 22-Oct-81 FPD E VEATCH CONSULTING ENGINEERS TX-868-573 12-Mar-82 FPD E CAROLINA COMPANY, CATAWBA, SOUTH TX-868-571 12-Mar-82 FPD E CCTRIC POWER COOPERATIVE, INC., NEW TX-1-024-037 6-Dec-82 FPD E NCORPORATED, CEDAR RAPIDS PLANT, TX-433-699 4-Mar-80 FPD E NCORPORATED, DAYTON, OHIO TX-584-419 18-Nov-80 FPD E ILLINOIS PUBLIC SERVICE COMPANY, UNITS TX-1-208-776 17-Oct-83 FPD E MES ELECTRIC AMES, IOWA TX-787-882 22-Oct-81 FPD E RAND HAVEN, BOARD OF POWER & LIGHT, TX-1-105-706 21-Apr-83 FPD E RS POWER COMPANY, WEST OLIVE, TX-2-121-500 24-Jul-87 DFO E KANSAS CITY, MISSOURI TX-685-278 8-May-81 FPD E CDISON COMPANY, BELLE RIVER POWER TX-1-267-004 16-Jan-84 FPD E TO DE NEMOURS & CO., FLORENCE, SOUTH TX-1-024-039 6-Dec-82 FPD E COWER CORPORATION, UNIT NO. 4, |

Prepared by: AMSaus, 4/26/2010 Page 61 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| OPERATOR TRAINING MANUAL FOR GILMAN PAPER COMPANY, ST. MARYS, GEORGIA | TX-1-023-949 | 6-Dec-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR GREENWOOD MILLS, INC., ORANGEBURG, SOUTH CAROLINA | TX-713-291 | 19-Jun-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR GULF STATES UTILITIES COMPANY, BEAUMONT, TEXAS | TX1-024-035 | 6-Dec-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR ILLINOIS POWER COMPANY, BALDWIN POWER STATION, BALDWIN, ILLINOIS | TX-879-662 | 29-Mar-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR INTERNATIONAL PAPER COMPANY, MANSFIELD, LOUISIANA | TX-685-277 | 8-May-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR INTERNATIONAL PAPER COMPANY, MANSFIELD, LOUISIANA (STERLING POWER BOILER) | TX-713-290 | 19-Jun-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR JACKSONVILLE ELECTRIC AUTHORITY, NORTHSIDE GENERATING STATION, JACKSONVILLE, FL. | TX-587-160 | 24-Nov-80 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR KOREA ELECTRIC COMPANY, GOJEONG POWER PLANT, SEOUL, KOREA | TX-1-024-036 | 6-Dec-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR MERCK & CO., INC., STONEWALL PLANT, ELKTON, VIRIGINA | TX-868-572 | 12-Mar-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR MICHIGAN SOUTH CENTRAL POWER AGENCY, LITCHFIELD, MICHIGAN | TX-905-504 | 14-May-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR NEBRASKA PUBLIC POWER DISTRICT, GERALD GENTLEMAN STATION, SUTHERLAND, NEBRASKA | TX-660-248 | 26-Mar-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR OIL & GAS FIRED UTILITY BOILERS | TX-1-208-774 | 17-Oct-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR PACKAGING CORPORATION OF AMERICA, FILER CITY, MICHIGAN | TX-1-145-818 | 5-Jul-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR PENNTECH PAPERS, INC., JOHNSONBURG, PENNSYLVANIA | TX-1-024-034 | 6-Dec-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR PENNTECH PAPERS, INC., JOHNSONBURG, PENNSYLVANIA | TX-1-145-817 | 5-Jul-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR PUBLIC SERVICE OF NEW HAMPSHIRE MANCHESTER, NEW HAMPSHIRE | TX-827-483 | 7-Jan-82 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010 Page 62 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| OPERATOR TRAINING MANUAL FOR REPUBLIC STEEL CORPORATION, WARREN, OHIO | TX-905-504 | 14-May-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR SUNFLOWER ELECTRIC COOPERATIVE, GARDEN CITY, KANSAS | TX-1-145-816 | 5-Jul-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR TENNECO OIL COMPANY, SWEETWATER COUNTY, WYOMING | TX-660-247 | 26-Mar-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR THE ALABAMA POWER COMPANY, MILLER STEAM PLANT, UNIT NO. 2, WEST JEFFERSON, ALABAMA | TX-1-267-006 | 16-Jan-84 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR THE FM PACKAGE BOILER | TX-1-208-775 | 17-Oct-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR TOLEDO EDISON COMPANY BAYSHORE STATION OREGON, OHIO | TX-1-145-815 | 5-Jul-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR TOLEDO EDISON COMPANY BAYSHORE STATION TOLEDO, OHIO | TX-1-024-038 | 6-Dec-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR UNIVERSITY OF NORTHERN IOWA, CEDAR FALLS, IOWA | TX-433-231 | 14-Mar-80 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR UTAH POWER & LIGHT COMPANY HUNTER STATION UNITS 3&4 CASTLE DALE, UTAH | TX-1-105-698 | 21-Apr-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR WESTERN FARMERS ELECTRIC COOPERATIVE, FT. TOWSON, OKLAHOMA | TX-2-121-504 | 24-Jul-87 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR WESTERN KRAFT PAPER GROUP | TX-685-276 | 8-May-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL FOR WISCONSIN POWER & LIGHT EDGEWATER GENERATING STATION, SHEBOYGAN, WISCONSIN | TX-1-208-683 | 17-Oct-83 | FPD | B&W |
| OPERATOR TRAINING MANUAL, BALTIMORE GAS & ELECTRIC COMPANY, BALTIMORE, MARYLAND | TX-504-105 | 3-Jul-80 | FPD | B&W |
| OPERATOR TRAINING MANUAL, BALTIMORE GAS & ELECTRIC COMPANY, BALTIMORE, MARYLAND | TX-502-894 | 2-Jul-80 | FPD | B&W |
| OPERATOR TRAINING MANUAL, CITY OF LAKELAND, FL. | TX-787-883 | 22-Oct-81 | FPD | B&W |
| OPERATOR TRAINING MANUAL, IOWA-ILLINOIS GAS & ELECTRIC COMPANY, LOUISA GENERATING STATION MUSCATINE, IOWA | TX-827-334 | 6-Jan-82 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010 Page 63 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| OPERATOR TRAINING MANUAL, PUBLIC SERVICE CO. OF NEW MEXICO UNITS 3&4 SAN JUAN STATION WATERFLOW, NM | TX-827-335 | 6-Jan-82 | FPD | B&W |
| OPERATOR TRAINING MANUAL, SAN MIGUEL ELECTRIC COOPERATIVE, SAN MIGUEL PLANT, CHRISTINE, TEXAS | TX-543-600 | 7-Jul-80 | FPD | B&W |
| PACKAGING CORPORATION OF AMERICA FILER CITY, MICHIGAN | VA-122-780 | 28-Mar-83 | FPD | B&W |
| PCPOWER | TX-2-098-911 | 26-Jun-87 | RDD | B&W |
| PCPOWER | TX-2-140-588 | 26-Aug-87 | RDD | B&W |
| PCPOWER | TX-2-190-950 | 27-Nov-87 | RDD | B&W |
| PDD-1000 PARTIAL DISCHARGE DETECTOR OPERATOR'S GUIDE | TX4-658-842 | 4-Nov-97 | SIDT | B&W |
| PENNSYLVANIA ELECTRIC COMPANY, NEW YORK ELECTRIC AND GAS COMPANY, HOMER CITY STATION UNIT NO. 3, HOMER CITY, PA | TX-467-228 | 5-Mar-80 | FPD | B&W |
| PENNTECH PAPERS, INC., UNITS 81&82, JOHNSONBURG, PENNSYLVANIA | VA-122-775 | 28-Mar-83 | FPS | B&W |
| PORTABLE ALL ACOUSTIC LEAK LOCATOR MODEL LD-180P OPERATING MANUAL | TX4-660-370 | 4-Nov-97 | SIDT | B&W |
| POTOMAC ELECTRIC POWER COMPANY GENERATING STATION "E"- UNITS NO. 1 AND 2 CHALK POINT, MARYLAND | VA-122-778 | 28-Mar-83 | FPD | B&W |
| PUBLIC SERVICE COMPANY OF NEW MEXICO, SAN JUAN STATION, UNIT NOS. 3&4, WATERFLOW, NEW MEXICO | VA-198-897 | 19-Aug-85 | PSI | B&W |
| PUBLIC SERVICE COMPANY OF NEW MEXICO, SAN JUAN, UNIT NOS. 3&4, WATERLFOW, NEW MEXICO | VA-93-350 | 22-Oct-81 | FPD | B&W |
| QUEENSLAND NICKEL PTY., LTD. YABULU, QUEENSLAND, AUSTRALIA (TECHNICAL DRAWING) | VA-104-281 | 16-Jul-82 | FPD | B&W |
| REPUBLIC STEEL CORPORATION, MAHONING VALLEY PLANT, WARREN, OHIO | VA-104-470 | 14-May-82 | FPD | B&W |
| SAN MIGUEL ELECTRIC POWER COOPERATIVE, INC., SAN MIGUEL PLANT, UNIT NO. 1, CHRISTINE, TEXAS B&W RADIANT REHEAT BOILER | 495-613 | 18-Jun-80 | FPD | B&W |
| Prepared by: AMSaus, 4/26/2010 Page 64 of 12 | | | | |

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| SECONDARY AIR HEATER, DETROIT EDISON COMPANY, BELLE RIVER POWER PLANT | VA-154-188 | 3-May-84 | FPD | B&W |
| SIERRA PACIFIC POWER COMPANY, NORTH VALMY STATION, UNIT NO. 1, VALMY, NEVADA B&W RADIANT REHEAT BOILER | VA-116-844 | 6-Dec-82 | FPD | B&W |
| SO2 REMOVAL PROJECT FOR ARIZONA PUBLIC SERVICE COMPANY FOUR CORNERS UNITS 4&5 FRUITLAND, NEW MEXICO | TX-1-267-003 | 16-Jan-84 | FPD | B&W |
| STEAM ITS GENERATION AND USE | TX-6-468-898 | 11-Aug-06 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | TX3-820-289 | 8-Mar-94 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | TX-36-602 | 13-May-78 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | A-693057 | 1-Dec-75 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | A-638308 | 25-Jul-63 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | AA-447247 | 3-Jun-60 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | AA-406366 | 20-Jul-59 | FPD | B&W |
| STEAM, ITS GENERATION AND USE | AA-214504 | 20-Dec-55 | FPD | B&W |
| SWAGE DRAW EXPERT SYSTEM | TXU-617-134 | 28-Feb-94 | FPD | B&W |
| TECHNICAL PROPOSAL TO JACKSONVILLE ELECTRIC AUTHORITY (PURCHASER) FOR REHEATER PURCHASE AT NORTHSIDE UNIT NO. 1 (ORIGINAL UP-34) | TX2-380-986 | 18-Aug-88 | FPD | B&W |
| TEXAS UTILITIES SERVICES, INC. MONTICELLO STEAM ELECTRIC STATION UNIT NO. 3, MT. PLEASANT, TEXAS BABCOCK & WILCOX UNIVERSAL PRESSURE BOILER | TX-660-218 | 26-Mar-81 | FPD | B&W |
| TUBE METAL IDENTIFICATION, INTERMOUNTAIN POWER PROJECT, UNIT NO. 1, LYNNDYL, | | | | |
| UTAH | VA-145-241 | 9-Jan-84 | FPD | B&W |
| TUBE METAL IDENTIFICATION, NORTHERN INDIANA PUBLIC SERVICE COMPANY, ROLLIN M. SCHAHFER GENERATING STATION, UNIT NO. 14 WHEATFIELD, ILLINOIS | VA-137-557 | 17-Oct-83 | FPD | B&W |

Prepared by: AMSaus, 4/26/2010 Page 65 of 12

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------------|-----------|----------|---------|
| TUBE METAL IDENTIFICATION, THE CLEVELAND ELECTRIC ILLUMINATING COMPANY, AVON LAKE PLANT – UNIT NO. 9, AVON LAKE, OHIO | VA-198-550 | 16-Aug-85 | FPD | B&W |
| TUBE METAL IDENTIFICATION, THE DETROIT EDISON COMPANY BELLE RIVER PLANT, UNIT NOS. 1 & 2 ST. CLAIR, MICHIGAN | VA-145-243 | 9-Jan-84 | FPD | B&W |
| TUBE METAL IDENTIFICATION, WISCONSIN POWER & LIGHT COMPANY, EDGEWATER GENERATING STATION, UNIT NO. 5 SHEBOYGAN, WISCONSIN | VA-122-774 | 28-Mar-83 | FPD | B&W |
| UNIVERSITY OF NORTHERN IOWA PLANT NO. 2, UNIT NO. 3 CEDAR FALLS, IOWA | TX-467-229 | 5-Mar-80 | FPD | B&W |
| USEFUL TABLES | TX-35-729 | 9-May-78 | FPD | B&W |
| UTAH POWER & LIGHT, HUNTER STATION, UNITS NO. 3 & 4, CASTLEDALE, UTAH | VA-122-773 | 28-Mar-83 | FPD | B&W |
| UTILITY MAINTENANCE GUIDE FOR WESTERN FARMERS ELECTRIC COOPERATIVE, FT. TOWSON, OKLAHOMA | TX2-122-729 | 24-Jul-87 | DFO | B&W |
| WESTERN FARMERS ELECTRIC COOPERATIVE, HUGO PLANT – UNIT NO. 1, HUGO, OKLAHOMA, BABCOCK & WILCOX RADIANT REHEAT BOILER | TX-741-298 | 22-Jun-81 | FPD | B&W |
| WISCONSIN POWER & LIGHT COMPANY, EDGEWATER GENERATING STATION, UNIT NO. 5, SHEBOYGAN, WISCONSIN | VA-122-779 | 28-Mar-83 | FPD | B&W |
| Babcock & Wilcox Industries Ltd. | | | | |
| ROUTE SHEET GENERATIVE PROGRAM | CANADIAN 409561 | 7-Oct-91 | BWCL | BWIL |
| McDermott International, Inc. (used by B&W) | | | | |
| REENGINEERING EXECUTIVE PRESENTATION | Txu-812-355 | 14-Aug-97 | MIT | MII |
| REENGINEERING WORKSHOP | TXU-666-137 | 27-Dec-94 | EISD | MII |
| REENGINEERING WORKSHOP INSTRUCTOR MANUAL | TXU-827-210 | 13-Aug-97 | MIT | MII |
| TOTAL QUALITY MANAGEMENT REFERENCE MANUAL | TXU-800-804 | 14-Aug-97 | EISD | MII |
| Prepared by: AMSaus, 4/26/2010 Page 66 of 12 | | | | |

| Copyright Title | Reg. No. | Reg. Date | Division | Company |
|---|--------------|-----------|----------|---------|
| McDermott Technology, Inc. (name change effective April 21, 2 | 010) | | | |
| DETROIT EDISON CO., THE, BELLE RIVER POWER PLANT ST. CLAIR MICHIGAN | VA-161-320 | 29-May-84 | RDD | MTI |
| ELEMENTS OF TWO-PHASE FLOW IN FOSSIL BOILERS | TX-2-150-696 | 10-Sep-87 | CRD | MTI |
| HELIX | TX-2-641-197 | 31-Jul-89 | RDD | MTI |
| PCFAD | TX-2-831-081 | 8-May-90 | RDD | MTI |
| PERSONAL COMPUTER CONTROL PROGRAM PC/CP | TX-2-147-035 | 2-Sep-87 | ARC | MTI |
| pH AND CONDUCTIVITY SOLVER | TXU-387-279 | 23-Aug-89 | RDD | MTI |
| PLATE LAMINATE | TXU-496-596 | 14-Jun-91 | RDD | MTI |
| SOFTWARE DEVELOPMENT GUIDELINES | TX-2-327-698 | 8-Jun-88 | RDD | MTI |
| STANDARD PRACTICE MANUAL | TXU-292-232 | 17-Aug-87 | RDD | MTI |

Prepared by: AMSaus, 4/26/2010

Page 67 of 12

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | |
|--------------------|--|--------------|-------------|------------|---------|-------------|---------|-----------------------------------|--|--|--|
| 0340 | SPRING | LOADED BRAK | KE ASSEMBLY | FOR INDE | XING SO | OOTBLOWER | | | | | |
| | Granted | 24-Jan-1991 | 07/645,220 | | | 19-Nov-1991 | 5065472 | Diamond Power International, Inc. | | | |
| 0343 | HUB ASS | SEMBLY FOR S | OOTBLOWER | ₹ | | | | | | | |
| | Granted | 12-Apr-1991 | 07/685464 | | | 25-Feb-1992 | 5090087 | Diamond Power International, Inc. | | | |
| 0346 | FLOATI | NG CABLE LOC | OP CONTROL | ASSEMBLY | 7 | | | | | | |
| | Granted | 29-Jul-1991 | 07/737,459 | | | 01-Dec-1992 | 5167307 | Diamond Power International, Inc. | | | |
| 0348 | METHOD AND APPARATUS FOR CONSTANT PROGRESSION OF A CLEANING JET ACROSS HEATED SURFACES | | | | | | | | | | |
| | Granted | 02-May-1992 | 07/877987 | | | 16-Aug-1994 | 5337438 | Diamond Power International, Inc. | | | |
| 0348A | METHOD AND APPARATUS FOR CONSTANT PROGRESSION OF A CLEANING JET ACROSS HEATED SURFACES | | | | | | | | | | |
| | Granted | 20-May-1994 | 08/247,001 | | | 01-Aug-1995 | 5437295 | Diamond Power International, Inc. | | | |
| 0353B | SOOTBLOWER HAVING VARIABLE DISCHARGE | | | | | | | | | | |
| 02 | Granted | 25-Apr-1994 | 08/232322 | | | 23-May-1995 | 5416946 | Diamond Power International, Inc. | | | |
| 0355 | METHOD AND APPARATUS OF PREHEATING A SOOTBLOWER LANCE | | | | | | | | | | |
| | Granted | 03-Feb-1993 | 08/013770 | | | 14-Jun-1994 | 5320073 | Diamond Power International, Inc. | | | |
| 0357 | OPEN BI | EAM SOOTBLO | WER | | | | | | | | |
| | Granted | 22-Mar-1993 | 08/034251 | | | 05-Apr-1994 | 5299533 | Diamond Power International, Inc. | | | |
| 0357A | OPEN BI | EAM SOOTBLO | WER | | | | | | | | |
| 01 | Granted | 29-Sep-1993 | 08/128840 | | | 22-Nov-1994 | 5365890 | Diamond Power International, Inc. | | | |
| 0357B | OPEN BI | EAM SOOTBLO | WER | | | | | | | | |
| 01 | Granted | 28-Jan-1994 | 08/189301 | | | 04-Jul-1995 | 5429076 | Diamond Power International, Inc. | | | |
| 0357C | OPEN BI | EAM SOOT BLO | OWER | | | | | | | | |
| 01 | Granted | 12-May-1995 | 08/440317 | | | 27-Aug-1996 | 5549079 | Diamond Power International, Inc. | | | |
| 0373 | RETRAC | TABLE CHEM | ICAL INJECT | ION SYSTEM | М | | | | | | |
| | Granted | 27-Jan-1995 | 08/379974 | | | 17-Dec-1996 | 5585072 | Diamond Power International, Inc. | | | |
| 0374 | PARAMI | ETER SENSING | SOOTBLOWI | ER | | | | | | | |
| | Granted | 03-Apr-1997 | 08/832309 | | | 13-Jul-1999 | 5920951 | Diamond Power International, Inc. | | | |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Thursday, April 29, 2010 Prepared by AMSaus

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | | |
|--------------------|--|---------------------------|------------|------------|---------|---------------|----------|---|--|--|--|--|
| 4029 | LA MON | T CYCLE FEE | DWATER A | RRANGEM | ENT | | | | | | | |
| | Pending | 13-Jun-1974 | 05/479076 | | | | | McDermott Technology, Inc. | | | | |
| 4814 | | D OF CHROM R, POSITION | IZING A WO | ORKPIECE 1 | BY APPL | YING A COATI | NG CONTA | INING CHROMIUM PARTICLES ONTO A CERAMIC | | | | |
| | Granted | 28-Feb-1990 | 07/486481 | | | 20-Aug-1991 | 5041309 | McDermott Technology, Inc. | | | | |
| 5069 | SOx, NOx AND PARTICULATE REMOVAL SYSTEM | | | | | | | | | | | |
| 01 | Granted | 03-Apr-1990 | 504192 | | | 17-Dec-1996 | 5585081 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5086 | NUCLEAR FUEL ELEMENT FOR A PARTICLE BED REACTOR | | | | | | | | | | | |
| | Granted | 21-Nov-1990 | 616951 | | | 10-Sep-1991 | 5047204 | BWX Technologies, Inc. | | | | |
| 5093 | CASCADING PRESSURE CONTINUOUS BLOW BOTTLE | | | | | | | | | | | |
| | Granted | 02-Jun-1992 | 889929 | | | 30-Nov-1993 | 5265983 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5096 | FLUID VELOCITY MEASUREMENT DEVICE FOR A GENERATING BANK TUBE | | | | | | | | | | | |
| | Granted | 17-Aug-1990 | 569315 | | | 15-Dec-1992 | 5170672 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5098 | NON-WELDED ATTACHMENT TUBE SUPPORT LUG CASTING | | | | | | | | | | | |
| | Granted | 31-Aug-1990 | 576305 | | | 03-Mar-1992 | 5092278 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5104 | METAL | DISINTEGRAT | FOR TUBE C | CUTTER | | | | | | | | |
| | Granted | 15-Oct-1990 | 597962 | | | 31-Dec-1991 | 5077456 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5108 | SOX, NO | X AND PARTI | CULATE RE | EMOVAL SY | STEM | | | | | | | |
| 01 | Granted | 02-Oct-1990 | 593546 | | | 22-Oct-1996 | 5567394 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5110 | LOW PR | ESSURE LOSS | S/REDUCER | DEPOSITIO | ON ATOM | IIZER | | | | | | |
| | Granted | 21-Mar-1991 | 672820 | | | 14-Jul-1992 | 5129583 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5115 | FURNAC | CE CLEANLIN | ESS MONITO | OR FOR HI | GH REFI | LECTIVITY ASI | H | | | | | |
| | Granted | 03-Dec-1990 | 621416 | | | 10-Mar-1992 | 5094695 | Diamond Power International, Inc. | | | | |
| 5116 | ADVANO | CED WATER L | ANCE CON | TROL SYST | EM BAS | ED ON PEAK F | URNACE W | ALL EMISSIVITY | | | | |
| | Granted | 03-Dec-1990 | 621418 | | | 17-Mar-1992 | 5096502 | Diamond Power International, Inc. | | | | |
| 5131 | BI-META | ALLIC EXTRU | SION BILLE | T PREFOR | MS & MI | ETHOD & APPA | RATUS FO | R PRODUCING SAME | | | | |
| | Granted | 04-Oct-1991 | 771906 | | | 04-May-1993 | 5207776 | Babcock & Wilcox Power Generation Group, Inc. | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 69 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | | |
|--------------------|---|-------------|----------|-----------|----------|-------------|---------|---|--|--|--|--|
| 5134 | VIDEO 7 | TEMPERATUF | | OR | | | | | | | | |
| | Granted | 05-Jun-1991 | 711469 | | | 06-Jul-1993 | 5225883 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5136 | PLASMA ARC WELDING ADAPTED TO ID ELLIPSOID PROCESS | | | | | | | | | | | |
| | Granted | 17-Jun-1991 | 715956 | | | 30-Mar-1993 | 5198635 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5140 | ACOUSTIC EMISSION LEAK SIMULATOR | | | | | | | | | | | |
| | Granted | 08-Jul-1991 | 726465 | | | 04-Aug-1992 | 5134876 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5141 | METHOD FOR DIFFUSION COATING A WORKPIECE WITH Cr, Si, AI OR B BY PLACING COATED CERAMIC ALUMINO-SILICATE FIBERS NEXT TO | | | | | | | | | | | |
| 01 | Granted | 25-Apr-1991 | 691182 | | | 04-Aug-1992 | 5135777 | McDermott Technology, Inc. | | | | |
| 5145 | SINGLE SPRAY LEVEL FOR FLUE GAS DESULFURIZATION SYSTEM | | | | | | | | | | | |
| | Granted | 26-Jul-1991 | 736334 | | | 22-Dec-1992 | 5173093 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5151 | PRESSURE PULSE CLEANING FOR ABSORPTION TOWER DISTRIBUTORS | | | | | | | | | | | |
| | Granted | 18-Dec-1991 | 809377 | | | 09-Nov-1993 | 5259446 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5152 | BOILER | TOP SUPPOR | T CASINO | G CHANNE | L SLIT R | ING | | | | | | |
| | | 01-Jul-1991 | 724258 | | | 25-Feb-1992 | | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5156 | ACCESS | DOOR WITH | SAFETY I | LATCH FO | R PRESS | URIZED AIR | CHAMBER | RS | | | | |
| | Granted | 25-Jul-1991 | 735568 | | | 30-Mar-1993 | 5197226 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5164 | LOW PF | RESSURE DRO | P DRY SC | RUBBER | | | | | | | | |
| | | 23-Oct-1991 | 781465 | | | | | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5165 | | | | TIC ACOUS | TIC TRA | | | INSPECTION | | | | |
| | | 07-Oct-1991 | 772316 | | | Č | 5237874 | BWX Technologies, Inc. | | | | |
| 5174 | | ERABLE FLUE | | SULPHURIZ | ZATION S | | | | | | | |
| | | 12-Nov-1991 | 790900 | | | | | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5181 | | LATED SLUD | GE LANC | E WITH A | MOVEAE | | | | | | | |
| | | 10-Jan-1992 | 819249 | | | | | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5184 | | D FOR ERECT | | BSORBER | TOWER | | | | | | | |
| | Granted | 30-Oct-1991 | 784860 | | | 26-Apr-1994 | 5305574 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| | | | | | | | | | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 70 of 34

| CASE NO | COT A TOTAL | EH E DATE | A DD NO | DUD DATE | DUD NO | ICCLIE DATE | DATNO | OWNER | | | |
|---------|--|--------------|-----------|-----------|---------|---------------|-----------|---|--|--|--|
| SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | <u>OWNER</u> | | | |
| 5188 | | V/DOWNFLOW | | TUBE CIF | RCULATI | | | | | | |
| 01 | Granted | 29-Oct-1991 | 785147 | | | 13-Apr-1993 | 5201282 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5202 | FEEDWATER CONTROL FOR DRUM TYPE STEAM GENERATORS | | | | | | | | | | |
| | Granted | 22-Jan-1992 | 823503 | | | 22-Sep-1992 | 5148775 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5203 | THERMIONIC REACTOR | | | | | | | | | | |
| | Granted | 17-Jan-1992 | 822391 | | | 21-Sep-1993 | 5247548 | Babcock & Wilcox Nuclear Operations Group, Inc. | | | |
| 5207 | SUPPOR | T OF INTERSE | PACED OP | POSED FE | ED HEAI | DERS FOR FGI | O SYSTEMS | | | | |
| | Granted | 21-May-1992 | 886581 | | | 21-Dec-1993 | 5271873 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5220 | HYDRO | GEN PEROXID | E FOR FL | UE GAS D | ESULFUI | RIZATION | | | | | |
| | Granted | 08-Sep-1994 | 303311 | | | 21-Jan-1997 | 5595713 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5231 | LOW NOX BURNER SYSTEM | | | | | | | | | | |
| | Granted | 13-Mar-1992 | 850600 | | | 27-Apr-1993 | 5205226 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5232 | DRY SCRUBBER WITH INTEGRAL PARTICULATE COLLECTION DEVICE | | | | | | | | | | |
| | Granted | 24-Jun-1992 | 904304 | | | 01-Jun-1993 | 5215557 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5239 | CONTAI | NMENT DEVI | CE FOR C | ONTAMIN | ATED BU | JILDING DEMO | OLITION | | | | |
| | Granted | 31-Jul-1992 | 922881 | | | 13-Apr-1993 | 5201152 | BWX Technologies, Inc. | | | |
| 5242 | FLUID B | BED COMBUST | ION REHI | EAT STEAL | м темрі | ERATURE CON | NTROL | | | | |
| | Granted | 01-Jun-1992 | 891096 | | | 11-May-1993 | 5209188 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5243 | метно | D FOR ALUM | INIZING A | A FERRITI | C WORI | KPIECE BY C | OATING IT | WITH AN AQUEOUS ALUMINA SLURRY, ADDING A | | | |
| | HALIDE | ACTIVATOR | | | | | | | | | |
| 01 | Granted | 24-Apr-1992 | 873925 | | | 04-May-1993 | 5208071 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5258 | ENHANG | CED REAGENT | UTILIZA | TION FOR | DRY OR | R SEMI-DRY SO | CRUBBER | | | | |
| | Granted | 08-Jul-1992 | 910062 | | | 28-Dec-1993 | 5273721 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5264 | PURIFIC | CATION OF FG | D GYPSU | M PRODU | CT | | | | | | |
| | Granted | 01-Sep-1992 | 937775 | | | 01-Jun-1993 | 5215672 | Babcock & Wilcox Power Generation Group, Inc. | | | |
| 5273 | CRACK | ORIENTATION | N DETERN | MINATION | AND DE | TECTION USI | NG HORIZO | NTALLY POLARIZED SHEAR WAVES | | | |
| | Granted | 17-Sep-1992 | 947123 | | | 24-Jan-1995 | 5383365 | McDermott Technology, Inc. | | | |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Thursday, April 29, 2010 Prepared by AMSaus

Page 71 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | | |
|--------------------|---|--------------|-------------|------------|----------|---------------|-----------|---|--|--|--|--|
| 5274 | SO2 CO | NTROL USING | MOVING GF | RANULAR B | EDS | | | , — — — — — — — — — — — — — — — — — — — | | | | |
| | Granted | 07-Aug-1992 | 926492 | | | 30-Nov-1993 | 5266288 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5282 | APPAR | ATUS FOR DET | ECTING CAI | RRYOVER P | ARTICLI | ES IN THE INT | ERIOR OF | A FURNACE | | | | |
| | Granted | 08-May-1990 | 521082 | | | 30-Apr-1991 | 5010827 | Diamond Power International, Inc. | | | | |
| 5289 | PRESSURE PULSE CLEANING FOR ABSORPTION TOWER DISTRIBUTORS | | | | | | | | | | | |
| 01 | Granted | 03-Sep-1992 | 940087 | | | 28-Dec-1993 | 5273590 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5294 | REVERSIBLE, WEAR-RESISTANT ASH SCREW COOLER SECTION | | | | | | | | | | | |
| | Granted | 29-Sep-1992 | 953631 | | | 10-May-1994 | 5309848 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5307 | CYCLONE FURNACE WITH INCREASED TUBE WALL MATERIAL | | | | | | | | | | | |
| 01 | Granted | 04-Nov-1992 | 972179 | | | 28-Dec-1993 | 5273003 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5311 | BENEFICIATION OF DRY SCRUBBER PRODUCT | | | | | | | | | | | |
| | Granted | 08-Feb-1993 | 015956 | | | 07-Jun-1994 | 5318227 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5313 | COST R | EDUCTION OF | WET FGD S | YSTEMS WI | TH A TR. | AY | | | | | | |
| | Granted | 27-Jan-1993 | 009698 | | | 25-Jan-1994 | 5281402 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5315 | ASYMN | IETRIC PULVE | RIZER TIRE | | | | | | | | | |
| | Granted | 16-Mar-1993 | 033694 | | | 13-Sep-1994 | 5346148 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5317 | METHO | DD AND APPAR | ATUS FOR G | AS LIQUID | CONTAC | T | | | | | | |
| | Granted | 30-Nov-1992 | 983109 | | | 21-Sep-1993 | 5246471 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5323 | CHEMI | CAL CONVERS | ION PLUG T | O ACHIEVE | GAS TIC | GHTNESS | | | | | | |
| | Granted | 03-May-1993 | 055097 | | | 21-Dec-1993 | 5271550 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5324 | FGD PE | RFORMANCE 1 | ENHANCEM | ENT BY HYI | DROCLO | NE AND RECY | CLING ST | EPS | | | | |
| | Granted | 08-Mar-1993 | 028895 | | | 03-May-1994 | 5308509 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5336 | CIRCUI | LATING FLUID | IZED BED RI | EACTOR WI | TH INTE | RNAL PRIMAF | RY PARTIC | CLE SEPARATION AND RETURN | | | | |
| | Granted | 25-Mar-1993 | 08/037986 | | | 06-Sep-1994 | 5343830 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5342 | HIGH E | FFICIENCY AD | OVANCED DE | RY SCRUBBI | ER | | | | | | | |
| | Granted | 23-Jun-1993 | 081765 | | | 11-Oct-1994 | 5354364 | Babcock & Wilcox Power Generation Group, Inc. | | | | |

 $^{^{\}star}\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,\ 2010$

Thursday, April 29, 2010 Prepared by AMSaus

| CASE NO | om . mv.o | | , pp vo | | nun vo | ************************************** | | 0.000 | | | | |
|---------|---|--|------------|-----------|----------|--|----------|---|--|--|--|--|
| SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | | |
| 5343 | | | S BETWEEN | MODULES | OF HEAT | RECOVERY S | TEAM GE | NERATORS | | | | |
| | Granted | 02-Apr-1993 | 042328 | | | 01-Nov-1994 | 5359823 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5344 | INTEGI | INTEGRAL SHIPPING TRUSS ASSEMBLY FOR HEAT RECOVERY STEAM GENERATOR MODULES | | | | | | | | | | |
| | Granted | 07-Apr-1993 | 044992 | | | 06-Dec-1994 | 5370239 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5352 | METHOD OF REAGENT AND OXIDATION AIR DELIVERY | | | | | | | | | | | |
| | Granted | 16-Aug-1993 | 107387 | | | 19-Mar-1996 | 5500196 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5353 | DRY SCRUBBER WITH CONDENSING HEAT EXCHANGER FOR CYCLE EFFICIENCY IMPROVEMENT | | | | | | | | | | | |
| | Granted | 10-Aug-1993 | 104974 | | | 23-Aug-1994 | 5339755 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5363 | COMBI | NED HEAT EX | CHANGER A | AND AMMO | NIA INJE | ECTION PROCI | ESS | | | | | |
| | Granted | 15-Jun-1993 | 08/076994 | | | 10-Jan-1995 | 5380499 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5367 | METHOD AND APPARATUS FOR CONTROLLING THE BED TEMPERATURE IN A CIRCULATING FLUIDIZED BED REACTOR | | | | | | | | | | | |
| | Granted | 18-Feb-1994 | 198694 | | | 15-Nov-1994 | 5363812 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5379 | SHIPPING CONTAINER FOR HIGHLY ENRICHED URANIUM | | | | | | | | | | | |
| | Granted | 21-Jul-1993 | 095642 | | | 19-Apr-1994 | 5303836 | Babcock & Wilcox Nuclear Operations Group, Inc. | | | | |
| 5388 | ARTICU | JLATED ANNU | JLAR SLUDO | GE LANCE | | | | | | | | |
| | Granted | 24-Sep-1993 | 08/126453 | | | 02-May-1995 | 5411043 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5391 | BALL A | ND SOCKET I | LOATING S | EAL ASSEM | IBLY | | | | | | | |
| | Granted | 27-Aug-1993 | 113026 | | | 15-Feb-1994 | 5286063 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5393 | ELECT | ROMAGNETIC | CACOUSTIC | TRANSDUC | CER FOR | BOLT TENSIO | N AND LO | AD MEASUREMENT | | | | |
| | Granted | 15-Feb-1994 | 196917 | | | 19-Mar-1996 | 5499540 | McDermott Technology, Inc. | | | | |
| 5409 | CLEAN | ING METHOD | FOR PULVE | RIZED COA | AL INJEC | TION SYSTEM | EQUIPMI | ENT USING COKE BREEZE | | | | |
| | Granted | 07-Mar-1994 | 206990 | | | 05-Sep-1995 | 5447571 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5410 | NON-BA | AFFLED LOW | PRESSURE I | OROP VACU | UM COO | LED INSERTE | D SMELT | SPOUT | | | | |
| | Granted | 28-Oct-1993 | 144921 | | | 01-Aug-1995 | 5437768 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5412 | TITANI | UM TUBE SEA | L WELDING | HEAD ENG | CLOUSRE | E | | | | | | |
| | Granted | 14-Feb-1994 | 08/194953 | | | 26-Mar-1996 | 5501389 | Babcock & Wilcox Power Generation Group, Inc. | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 73 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|---------|--------------|-------------|------------|----------|--------------|-----------|---|
| 5415 | TUBE BI | EVELLING MA | | H ADJUSTA | ABLE BE | VELLING HEA | | |
| | Granted | 26-Jan-1994 | 188530 | | | 28-Feb-1995 | 5393176 | Babcock & Wilcox Power Generation Group, Inc. |
| 5417 | GAS-GA | S HEATER PR | OTECTION S | SYSTEM AN | ND METH | IOD | | |
| | Granted | 02-Dec-1993 | 161107 | | | 23-Apr-1996 | 5509461 | Babcock & Wilcox Power Generation Group, Inc. |
| 5419 | CONDE | NSING HEAT E | XCHANGER | SCRUBBIN | NG SYSTI | EM | | |
| | Granted | 02-Dec-1993 | 161109 | | | 29-Nov-1994 | 5368096 | Babcock & Wilcox Power Generation Group, Inc. |
| 5432 | REMOT | E TONE BURS | T ELECTRO | MAGNETIC | CACOUS | TIC TRANSDU | CER PULS | ER |
| | Granted | 15-Feb-1994 | 08/196662 | | | 20-Jun-1995 | 5426388 | McDermott Technology, Inc. |
| 5433 | REMOT | E PREAMPLIF | IER AND IM | PEDANCE 1 | MATCHI | NG CIRCUIT I | FOR ELEC | TROMAGNETIC ACOUSTIC TRANSDUCER |
| | Granted | 15-Feb-1994 | 196661 | | | 30-Apr-1996 | 5511424 | McDermott Technology, Inc. |
| 5443 | ELECTE | ROSTATIC SHI | ELD FOR EL | ECTROMA | GNETIC | ACOUSTIC T | RANSDUCI | ER NOISE REJECTION |
| | Granted | 31-May-1994 | 251542 | | | 25-Jul-1995 | 5436873 | McDermott Technology, Inc. |
| 5444 | DIODE I | EXPANDER FO | R ELECTRO | MAGNETIO | C ACOUS | STIC TRANSDU | UCER ELE | CTRONICS |
| | Granted | 08-Jul-1994 | 272897 | | | 12-Sep-1995 | 5449958 | McDermott Technology, Inc. |
| 5476 | USE OF | SINGLE-LEAD | AND MULT | I-LEAD RIE | BBED TU | BING FOR SLI | DING PRE | SSURE ONCE-THROUGH BOILERS |
| | Granted | 25-May-1994 | 249183 | | | 21-Feb-1995 | 5390631 | Babcock & Wilcox Power Generation Group, Inc. |
| 5477 | METHO | D AND SYSTE | M FOR SO2 A | AND SO3 CO | ONTROL | BY DRY SORE | BENT/REAC | GENT INJECTION AND WET SCRUBBING |
| | Granted | 29-Apr-1994 | 08/235597 | | | 07-Nov-2000 | 6143263 | Babcock & Wilcox Power Generation Group, Inc. |
| 5482 | WET SC | RUBBER INLE | T AWNING | WITH LOW | PRESSU | RE DROP | | |
| | Granted | 02-May-1994 | 236617 | | | 04-Apr-1995 | 5403523 | The Babcock & Wilcox Company |
| 5496 | METHO | D OF CONVER | T A DOUBL | E-LOOP FL | UE GAS I | DESULFURIZA | ATION TO | A SINGLE-LOOP SYSTEM |
| | Granted | 11-May-1994 | 241266 | | | 19-Sep-1995 | 5451250 | Babcock & Wilcox Power Generation Group, Inc. |
| 5501 | DUAL F | LUID ATOMIZ | ER FOR HIG | H SOLIDS | SOIL PAS | STE CONTAIN | ING PEBBI | LES OF AGGLOMERATES |
| | Granted | 02-Jun-1994 | 252805 | | | 30-Apr-1996 | 5511728 | Babcock & Wilcox Power Generation Group, Inc. |
| 5506 | COMBIN | NED HEAT EX | CHANGER A | MMONIA I | NJECTIO | ON PROCESS (| CHEAP) | |
| | Granted | 27-May-1994 | 250187 | | | 01-Aug-1995 | 5437851 | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

| CASE NO SubCase | | ISSUE DATE F | PAT NO | OWNER |
|--------------------|---|----------------|---------|---|
| 5509 | METHOD & APPARATUS FOR INJECTING NOX INHIBIT TO A SENSED | TING LIQUID RE | EAGENT | TINTO THE FLUE GAS OF A BOILER IN RESPONSE |
| | Granted 28-Jun-1994 08/266809 20 | 0-Feb-2001 61 | 190628 | Diamond Power International, Inc. |
| 5519 | SEGMENTED HEAT EXCHANGER FLUE GAS TREATME | ENT | | |
| | Granted 05-Jul-1994 270705 09 | 9-Jul-1996 55 | 534230 | Babcock & Wilcox Power Generation Group, Inc. |
| 5521 | BOILER BANK SURFACE TEMPERATURE PROFILER | | | |
| | Granted 25-Jul-1994 279736 01 | 1-Apr-1997 56 | 615953 | Diamond Power International, Inc. |
| 5527 | REGENERABLE MAGNESIUM DRY SCRUBBING | | | |
| | Granted 21-Jun-1994 263120 08 | 8-Aug-1995 54 | 439658 | Babcock & Wilcox Power Generation Group, Inc. |
| 5528 | EXTENDED WEAR LIFE LOW PRESSURE DROP, REPLACEABLE WEAR | AIGHT ANGLE, | SINGLE | E EXIT ORIFICE DUAL-FLUID ATOMIZER WITH |
| | Granted 06-Jul-1994 271336 06 | 6-Aug-1996 55 | 542609 | Babcock & Wilcox Power Generation Group, Inc. |
| 5530 | EXTENDED WEAR LIFE LOW PRESSURE DROP, R REPLACEABLE WEAR | RIGHT ANGLE, | MULTI | EXIT ORIFICE DUAL-FLUID ATOMIZER WITH |
| | Granted 06-Jul-1994 271293 14 | 4-May-1996 55 | 516046 | Babcock & Wilcox Power Generation Group, Inc. |
| 5539 | DISTRIBUTION CONE FOR PULVERIZED COAL BURNE | ERS | | |
| | Granted 02-Nov-1994 333394 18 | 8-Jun-1996 55 | 526758 | Babcock & Wilcox Power Generation Group, Inc. |
| 5540 | CATALYST OUTAGE PROTECTION SYSTEM | | | |
| | Granted 20-Jan-1995 375884 08 | 8-Apr-1997 56 | 618499 | Babcock & Wilcox Power Generation Group, Inc. |
| 5543 | ENHANCED HEAT EXCHANGER FLUE GAS TREATMEN | NT USING STEAM | M INJEC | TION |
| | Granted 12-Sep-1994 08/304741 22 | 2-Oct-1996 55 | 567215 | Babcock & Wilcox Power Generation Group, Inc. |
| 5554 | VERTICAL ARRANGEMENT FLUIDIZED/NON-FLUIDIZI | ED BED CLASSII | FIER CO | OOLER |
| | Granted 07-Oct-1994 320077 18 | 8-Jun-1996 55 | 526938 | Babcock & Wilcox Power Generation Group, Inc. |
| 5573 | SPRING RELEASE MECHANISM FOR ELECTROMAGNE | ETIC ACOUSTIC | TRANSI | DUCER (EMAT) PROBE |
| | Granted 24-Feb-1995 394683 2 | 1-Apr-1998 57 | 741973 | McDermott Technology, Inc. |
| 5580 | METHOD OF REAGENT AND OXIDATION AIR DELIVER | RY | | |
| | Granted 19-Dec-1994 359147 28 | 8-May-1996 55 | 520897 | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 75 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|---------|---------------|-----------|------------|---------|--------------|-----------|---|
| 5590 | | E LANCE NOZZ | | | | | | |
| | Granted | 08-Nov-1994 | 335922 | | | 23-Apr-1996 | 5509609 | Babcock & Wilcox Power Generation Group, Inc. |
| 5591 | HIGH T | EMPERATURE | ELECTRO | OMAGNETI | C ACOUS | STIC TRANSDI | UCER (EM. | AT) PROBE AND COIL ASSEMBLIES |
| | Granted | 24-Jul-1995 | 506413 | | | 18-Nov-1997 | 5689070 | McDermott Technology, Inc. |
| 5597 | MEDIC | AL ISOTOPE PI | RODUCTIO | ON REACTO | OR | | | |
| | Granted | 10-Nov-1994 | 339264 | | | 21-Jan-1997 | 5596611 | BWX Technologies, Inc. |
| 5599 | FLUE G | SAS CONDITION | NING FOR | THE REMO | OVAL OF | ACID GASES, | AIR TOXIO | CS AND TRACE METALS |
| | Granted | 22-Nov-1994 | 343285 | | | 04-Feb-1997 | 5599508 | Babcock & Wilcox Power Generation Group, Inc. |
| 5601 | STACK | ED INTERSPAC | IAL SPRA | Y HEADER | FOR FG | D WET SCRUB | BER | |
| | Granted | 10-Feb-1995 | 386834 | | | 15-Apr-1997 | 5620144 | Babcock & Wilcox Power Generation Group, Inc. |
| 5606 | ARTICU | JLATED ANNUI | LAR SLUD | GE LANCE | | | | |
| | Granted | 12-Dec-1994 | 354394 | | | 07-May-1996 | 5514219 | Babcock & Wilcox Power Generation Group, Inc. |
| 5614 | COMPO | SITE WATERL | ANCE ANI | D CAVITY (| CONNEC | TION | | |
| | Granted | 02-Mar-1995 | 398202 | | | 05-Nov-1996 | 5570969 | Babcock & Wilcox Power Generation Group, Inc. |
| 5620 | AUTON | IATED SLUDGE | LANCE | | | | | |
| | Granted | 23-Feb-1995 | 392651 | | | 05-Nov-1996 | 5570660 | Babcock & Wilcox Power Generation Group, Inc. |
| 5621 | AUTON | IATED SLUDGE | LANCE | | | | | |
| | Granted | 23-Feb-1995 | 392652 | | | 12-Nov-1996 | 5572957 | Babcock & Wilcox Power Generation Group, Inc. |
| 5627 | FLUE G | SAS DESULFUR | IZATION N | METHOD A | ND APPA | RATUS | | |
| | Granted | 08-Mar-1996 | 611605 | | | 18-Aug-1998 | 5795548 | Babcock & Wilcox Power Generation Group, Inc. |
| 5628 | IMPRO | VED SOX NOX | AND PART | TICULATE I | REMOVA | L SYSTEM | | |
| | Granted | 27-Feb-1995 | 394980 | | | 30-Jul-1996 | 5540897 | Babcock & Wilcox Power Generation Group, Inc. |
| 5637 | BURNE | R PATTERN TO | MINIMIZ | E SIDEWA | LL CORF | ROSION POTE | NTIAL | |
| | Granted | 02-May-1995 | 433635 | | | 09-Apr-1996 | 5505146 | Babcock & Wilcox Power Generation Group, Inc. |
| 5643 | DIFFUS | ER FOR COAL | NOZZLE F | BURNER | | | | |
| | Granted | 23-May-1995 | 448206 | | | 31-Dec-1996 | 5588380 | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|-----------------|---------------------------|------------|------------|----------|---------------|-----------|---|
| 5648 | AUTOM | IATED SLUDGE | LANCE | | | | | |
| | Granted | 03-Apr-1995 | 416590 | | | 17-Sep-1996 | 5555851 | Babcock & Wilcox Power Generation Group, Inc. |
| 5654 | METHO | DD FOR SO2 AN | D SO3 CONT | ROL BY DR | Y SORBE | ENT/REAGEN | Γ INJECTI | ION AND WET SCRUBBING |
| | Granted | 23-May-1995 | 08/448060 | | | 16-Oct-2001 | 6303083 | Babcock & Wilcox Power Generation Group, Inc. |
| 5668 | EMAT V | WITH INTEGRA | L ELECTRO | STATIC SH | IELD | | | |
| | Granted | 18-Jul-1995 | 503777 | | | 04-Mar-1997 | 5608691 | McDermott Technology, Inc. |
| 5669 | METHO | D AND APPAR | ATUS FOR W | ELD JOINI | NG PIPE | SECTIONS | | |
| | Granted | 26-Sep-1995 | 533896 | | | 29-Apr-1997 | 5624067 | Babcock & Wilcox Power Generation Group, Inc. |
| 5673 | ENHAN | CED HEAT EXC | CHANGER FI | LUE GAS TI | REATME | NT USING STI | EAM INJE | CTION |
| | Granted | 02-Jun-1995 | 459592 | | | 04-Feb-1997 | 5599382 | Babcock & Wilcox Power Generation Group, Inc. |
| 5682 | ONCE-T MATER | | AM GENERA | ATOR FURN | NACE OU | TLET FLUID | MIX TO | MINIMIZE THE NUMBER OF HEADERS AND RISER |
| | Granted | 13-Feb-1996 | 08/600616 | | | 23-Sep-1997 | 5669333 | Babcock & Wilcox Power Generation Group, Inc. |
| 5689 | METHO | D OF REPLACI | NG PRIMAR | Y DIVIDER | PLATE I | IN A STEAM G | GENERATO | OR |
| | Granted | 01-Aug-1995 | 509906 | | | 29-Apr-1997 | 5623763 | Babcock & Wilcox Power Generation Group, Inc. |
| 5694 | SEGME | NTED AUTOMA | ATED SLUDO | GE LANCE | | | | |
| | Granted | 20-Sep-1995 | 530814 | | | 21-Jul-1998 | 5782209 | Babcock & Wilcox Power Generation Group, Inc. |
| 5708 | | ARTICULATE A I ENHANCE | ND AEROSO | L REMOVA | L TECHN | VIQUE IN A CO | ONDENSIN | IG HEAT EXCHANGER USING AN ELECTROSTATIC |
| | Granted | 01-Dec-1995 | 08/566381 | | | 11-Aug-1998 | 5792238 | Babcock & Wilcox Power Generation Group, Inc. |
| 5711 | DETEC | TION OF CORR | OSION FATI | GUE CRAC | KS IN MI | EMBRANE BO | ILER TUE | BES |
| | Granted | 03-Aug-1995 | 511049 | | | 18-Jun-1996 | 5526691 | McDermott Technology, Inc. |
| 5712 | LOW PI | RESSURE DROP | , TURBULE | NT MIXING | ZONE DI | RY SCRUBBEI | R | |
| | Granted | 29-Sep-1995 | 536866 | | | 29-Jul-1997 | 5651948 | Babcock & Wilcox Power Generation Group, Inc. |
| 5720 | CORNE | R FOILS FOR H | YDRAULIC | MEASUREN | MENT | | | |
| | Granted | 18-Jan-1996 | 588495 | | | 18-Nov-1997 | 5687768 | Babcock & Wilcox Power Generation Group, Inc. |
| 5721 | EMAT P | PROBE AND TE | CHNIQUE FO | OR WELD I | NSPECTI | ON | | |
| | Granted | 07-Sep-1995 | 531573 | | | 02-Jun-1998 | 5760307 | McDermott Technology, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Thursday, April 29, 2010 Prepared by AMSaus

Page 77 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|---------|--------------|------------|------------|---------|--------------|-----------|---|
| 5726 | ENERGY | ABSORBING T | UBE SUPPO | RT FOR FU | RNACE | | | |
| | Granted | 11-Jan-1996 | 585220 | | | 02-Dec-1997 | 5692457 | Babcock & Wilcox Power Generation Group, Inc. |
| R | Granted | 30-Nov-1999 | 453602 | | | 29-May-2001 | RE37192 | Babcock & Wilcox Power Generation Group, Inc. |
| 5729 | BLACK I | LIQUOR GASIF | IER | | | | | |
| | Granted | 02-Nov-1995 | 548798 | | | 03-Jun-1997 | 5634950 | Babcock & Wilcox Power Generation Group, Inc. |
| 5731 | INLET A | WNING FOR H | IGH VELOCI | TY ABSOR | BERS | | | |
| | Granted | 16-Jan-1996 | 587180 | | | 12-Aug-1997 | 5656046 | Babcock & Wilcox Power Generation Group, Inc. |
| 5733 | MIST EL | IMINATION/AI | R TOXIC CO | NTROL IN | A WET S | CRUBBER USI | NG A CON | DENSING HEAT EXCHANGER |
| | Granted | 01-Dec-1995 | 08/566192 | | | 14-Aug-2001 | 6273940 | Babcock & Wilcox Power Generation Group, Inc. |
| 5734 | APPARA | TUS FOR THE | INJECTION, | DISTRIBUT | ION AN | D DISPERSION | OF SORBI | ENT IN A UTILITY BOILER FURNACE |
| | Granted | 30-Jan-1996 | 594052 | | | 11-Nov-1997 | 5685243 | Babcock & Wilcox Power Generation Group, Inc. |
| 5748 | KEY AD | VANCED LINEA | AR KINETIC | ABSORBER | SYSTEM | M PARTICULAT | ΓE ARRES | TING DEVICE |
| | Granted | 13-Feb-1996 | 600708 | | | 14-Oct-1997 | 5676715 | Babcock & Wilcox Power Generation Group, Inc. |
| 5753 | FASTEN | ER CHARACTE | RIZATION V | VITH AN EI | LECTRO | MAGNETIC AC | COUSTIC T | RANSDUCER |
| | Granted | 30-Jan-1996 | 594051 | | | 07-Oct-1997 | 5675087 | McDermott Technology, Inc. |
| 5759 | HYDRO | GEN PEROXIDE | FOR FLUE | GAS DESUL | FURIZA | TION | | |
| | Granted | 15-May-1996 | 648261 | | | 07-Oct-1997 | 5674459 | Babcock & Wilcox Power Generation Group, Inc. |
| 5761 | METHO | O OF MAKING | CONTINUOU | S CERAMI | C FIBER | COMPOSITE H | IOT GAS F | ILTER |
| | Granted | 23-Jan-1997 | 08/788717 | | | 02-Mar-1999 | 5876537 | Babcock & Wilcox Power Generation Group, Inc. |
| 5792 | METHO | O AND APPARA | TUS TO REC | GULATE A V | OLTAG | E CONTROLLE | ER | |
| | Granted | 11-Jan-1996 | 584195 | | | 18-Nov-1997 | 5689177 | Babcock & Wilcox Power Generation Group, Inc. |
| 5795 | ACTIVA | TED CARBON F | LUE GAS DI | ESULFURIZ | ATION S | SYSTEMS FOR | MERCURY | REMOVAL |
| | Granted | 22-Apr-1996 | 635532 | | | 30-Sep-1997 | 5672323 | Babcock & Wilcox Power Generation Group, Inc. |
| 5798 | DRAINA | BLE DISCHARO | GE PAN FOR | IMPACT T | YPE PAR | TICLE SEPARA | ATOR | |
| | Granted | 17-Jun-1996 | 664755 | | | 01-Sep-1998 | 5799593 | Babcock & Wilcox Power Generation Group, Inc. |
| 5805 | DRY SCI | RUBBER WITH | FORCED RE | CIRCULAT | ION | | | |
| | Granted | 03-Jun-1996 | 656747 | | | 19-Nov-1996 | 5575834 | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|------------------|--------------|------------|-----------|----------|---------------|-----------|---|
| 5807 | APPARA' | TUS FOR AIR | SPARGED S | LURRY TA | NKS | | | |
| | Granted | 14-Aug-1996 | 08/702425 | | | 11-Jan-2000 | 6013120 | Babcock & Wilcox Power Generation Group, Inc. |
| 5818 | WET SCI | RUBBER OXID | ATION AIR | SPARGER | ARRANG | EMENT | | * |
| | Granted | 12-Jul-1996 | 08/679495 | | | 27-Jul-1999 | 5928615 | Babcock & Wilcox Power Generation Group, Inc. |
| 5820 | FLUE GA | S DESULFUR | IZATION MI | ETHOD AN | D APPAR | ATUS | | |
| | Granted | 26-Jun-1996 | 669762 | | | 29-Sep-1998 | 5814288 | Babcock & Wilcox Power Generation Group, Inc. |
| 5828 | FLOODI | NG REDUCTIO | ON ON A TU | BULAR HEA | AT EXCH | IANGER | | |
| | Granted | 02-Aug-1996 | 691725 | | | 11-Aug-1998 | 5791404 | Babcock & Wilcox Power Generation Group, Inc. |
| 5832 | TWO PH | ASE FLOW DI | SPERSION I | DEVICE | | | | |
| | Granted | 20-Aug-1996 | 08/699764 | | | 23-May-2000 | 6065486 | Babcock & Wilcox Power Generation Group, Inc. |
| 5851 | ELECTR | OMAGNETIC | ACOUSTIC | TRANSDUC | CER (EMA | AT) FOR ULTRA | ASONIC IN | NSPECTION OF LIQUIDS IN CONTAINERS |
| | Granted | 29-Aug-1996 | 704967 | | | 17-Nov-1998 | 5837898 | BWX Technologies, Inc. |
| 5852 | METHOI (EMAT) |) FOR DETE | RMINING L | IQUID LE | VEL IN A | A CONTAINER | USING A | AN ELECTROMAGNETIC ACOUSTIC TRANSDUCER |
| | Granted | 29-Aug-1996 | 08/704956 | | | 23-Jan-2001 | 6176132 | BWX Technologies, Inc. |
| 5860 | LOW PR | ESSURE DROI | P, TURBULE | NT MIXINO | G ZONE D | ORY SCRUBBER | Ł | |
| | Granted | 02-Oct-1996 | 720855 | | | 17-Jun-1997 | 5639430 | Babcock & Wilcox Power Generation Group, Inc. |
| 5862 | TRAPEZ | OIDAL DEFLE | CTORS FOR | R HEAT EX | CHANGE | R TUBES | | |
| | Granted | 22-Jul-1997 | 898607 | | | 01-Sep-1998 | 5799724 | Babcock & Wilcox Power Generation Group, Inc. |
| 5864 | CIRCUL | ATING FLUID | IZED BED F | URNACE/R | EACTOR | WITH AN INTI | EGRAL SE | CONDARY AIR PLENUM |
| | Granted | 03-Dec-1996 | 08/759885 | | | 17-Nov-1998 | 5836257 | Babcock & Wilcox Power Generation Group, Inc. |
| 5875 | PULVER | IZED COAL B | URNER | | | | | |
| | Granted | 12-Nov-1996 | 08/747319 | | | 03-Nov-1998 | 5829369 | Babcock & Wilcox Power Generation Group, Inc. |
| 5876 | AN IMPR | OVED STRAI | N RELIEF M | IAINSHAFT | ASSEME | BLY | | |
| | Granted | 01-Nov-1996 | 08/742327 | | | 17-Oct-2000 | 6132135 | Babcock & Wilcox Power Generation Group, Inc. |
| 5881 | PRIMAR | Y DIVIDER PL | ATE | | | | | |
| | Granted | 15-Nov-1996 | 749590 | | | 07-Apr-1998 | 5735341 | Babcock & Wilcox Power Generation Group, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Thursday, April 29, 2010 Prepared by AMSaus

Page 79 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | |
|--------------------|---|---------------------------|------------|-----------|---------|--------------|----------|---|--|
| 5882 | | ESSURE DROP | | | | | PAT NO | OWNER | |
| 5882 | | | | BURNERS | AND NO | | | | |
| | Granted | 15-Nov-1996 | 749883 | | | 09-Sep-1997 | 5664944 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5883 | CAST AF | BRASION RESIS | | OW BALLS | 8 | | | | |
| | Granted | 27-Nov-1996 | 757990 | | | 14-Oct-1997 | 5676193 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5887 | FURNAC | EE GAS PYROM | ETER (GAST | EMP ALTE | RNATIV | E) | | | |
| | Granted | 19-Dec-1996 | 08/770609 | | | 11-May-2004 | 6733173 | Diamond Power International, Inc. | |
| D | Granted | 09-Feb-2004 | 10/774565 | | | 21-Feb-2006 | 7001067 | Diamond Power International, Inc. | |
| 5894 | IMPROV | ED MAIN SHAI | FT | | | | | | |
| | Granted | 22-Jul-1997 | 08/898606 | | | 19-Feb-2002 | 6347758 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5895 | IMPROV | ED MAIN SHAI | T ASSEMBL | Y | | | | | |
| | Granted | 05-Jun-1997 | 08/868867 | | | 07-Aug-2001 | 6270026 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5898 | DIESEL OR DUAL-FUEL ENGINE AND BLACK LIQUOR GASIFIER COMBINED CYCLE | | | | | | | | |
| | Granted | 14-Feb-1997 | 08/801836 | | | 20-Apr-1999 | 5895507 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5910 | THERMO | OPHOTOVOLT | AIC ELECTR | IC GENER | ATOR | | | | |
| | Granted | 19-May-1997 | 08/858335 | | | 03-Aug-1999 | 5932885 | McDermott Technology, Inc. | |
| 5911 | IMPROV | ED EMAT/TRA | NSMIT/RECI | EIVE SWIT | СН | | | | |
| | Granted | 19-Aug-1999 | 09/377453 | | | 27-Feb-2001 | 6192760 | McDermott Technology, Inc. | |
| 5926 | METHO | D FOR CHROM | IZING SMAL | L PARTS | | | | | |
| | Granted | 26-Sep-1997 | 08/938319 | | | 15-Jun-1999 | 5912050 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5927 | | RTICULATE AN ENHANCEME | | L REMOVAI | TECHN | IQUE IN A CO | NDENSING | HEAT EXCHANGER USING AN ELECTROSTATIC | |
| | Granted | 05-Jun-1997 | 869335 | | | 08-Dec-1998 | 5846301 | Babcock & Wilcox Power Generation Group, Inc. | |
| 5934 | | ATED REBURN | | D NOv CON | TDOL E | | | ** | |
| 3934 | | | | K NOX COP | VIKOL F | | | | |
| | Granted | 21-Nov-1997 | 975725 | | | 09-Mar-1999 | | Babcock & Wilcox Power Generation Group, Inc. | |
| 5938 | COMPOS | SITE INSULATO | | IMENSION | AL WEA | | | | |
| | Granted | 24-Jul-1997 | 08/899995 | | | 28-Nov-2000 | 6153831 | Babcock & Wilcox Nuclear Operations Group, Inc. | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 80 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | |
|--------------------|--|--------------|-------------|------------|---------|-------------|-----------|---|--|--|
| 5942 | PRESSUI | RE EQUALIZA | TION VALV | E | | | | | | |
| | Granted | 04-Aug-1997 | 08/905176 | | | 14-Sep-1999 | 5950677 | Diamond Power International, Inc. | | |
| 5949 | COAL SI | PREADER WIT | H SWIRL V | ANES | | | | | | |
| | Granted | 29-Oct-1997 | 09/959795 | | | 02-May-2000 | 6055913 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5951 | RETROFIT OF A CENTER INLET TYPE SCRUBBER WITH ABSORPTION/GAS DISTRIBUTION TRAY TO IMPROVE GAS-LIQUI CONTACT IN THE ABSORPTION ZONE | | | | | | | | | |
| | Granted | 13-Jan-1998 | 09/006376 | | | 14-Mar-2000 | 6036756 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5958 | CYCLON | IE FURNACE I | OR RETRO | FIT APPLIC | CATIONS | \$ | | | | |
| | Granted | 07-Nov-1997 | 08/966110 | | | 08-Feb-2000 | 6021724 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5960 | MODUL | AR HIGH TEM | PERATURE | SUPERCO | NDUCTIN | NG DOWN LEA | D WITH SA | AFETY LEAD | | |
| | Granted | 12-Nov-1997 | 08/968432 | | | 07-Mar-2000 | 6034324 | BWX Technologies, Inc. | | |
| 5968 | IMPROV | ED INTERNAI | L OIL FLOW | PATH FOR | GEARB | OX BEARINGS | | | | |
| | Granted | 20-Jan-1998 | 009366 | | | 08-May-2001 | 6227712 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5969 | LOW PR | ESSURE DROI | P INLET DES | SIGN TO PR | OMOTE | GOOD GAS FL | OW PATT | ERNS IN HIGH VELOCITY ABSORBERS | | |
| | Granted | 11-Feb-1998 | 09/022136 | | | 03-Dec-2002 | 6488899 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5972 | EMAT IN | SPECTION O | F HEADER T | UBE STUB | 5 | | | | | |
| | Granted | 29-Apr-1998 | 09/069445 | | | 27-Jun-2000 | 6079273 | Babcock & Wilcox Power Generation Group, Inc. McDermott Technology, Inc. | | |
| 5973 | RETAIN | ER SHEILD FO | OR SPLIT RI | NG CASTIN | GS | | | | | |
| | Granted | 11-Jun-1998 | 09/096045 | | | 28-Dec-1999 | 6006702 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5976 | STEAM (| GENERATOR 1 | FOR GASIFY | ING COAL | | | | | | |
| 01 | Granted | 27-Mar-2000 | 09/114743 | | | 06-Nov-2001 | 6312482 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5977 | FUME SI | ENSOR SYSTE | M AND MET | HODS FOR | BACKG | ROUND NOISE | SUPPRES | SION | | |
| | Granted | 11-Aug-2000 | 09/637717 | | | 03-Dec-2002 | 6490040 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5979 | VARIAB | LE PRESSURE | ONCE-THR | OUGH STE | AM GEN | ERATOR UPPE | R FURNA | CE HAVING NON-SPLIT FLOW CIRCUITRY | | |
| | Granted | 16-Apr-1998 | 09/061480 | | | 10-Aug-1999 | 5934227 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 5980 | FURNAC | E FLOOR | | | | | | | | |
| 01 | Granted | 15-Apr-1999 | 09/292722 | | | 30-May-2000 | 6067944 | Babcock & Wilcox Power Generation Group, Inc. | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | | |
|--------------------|--|--------------|------------|--------------|-------------|--------------|----------|---|--|--|--|--|
| 5981 | SLUDGE | LANCE | | | | | | | | | | |
| | Granted | 29-Jul-1998 | 09/124130 | | | 02-Jul-2002 | 6412449 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 5994 | INTERC | ONNECT FOR | SOLID OXII | DE FUEL CEL | LS | | | | | | | |
| | Granted | 01-Mar-1999 | 09/259711 | | | 24-Jul-2001 | 6265095 | SOFCo-EFS Holdings, LLC | | | | |
| 5996 | DETECT | ION OF CORR | OSION FAT | IGUE IN BOIL | LER TUBES U | SING A SPIKE | EMAT PU | ILSER | | | | |
| | Granted | 26-Jun-1998 | 09/105514 | | | 03-Oct-2000 | 6125703 | McDermott Technology, Inc. | | | | |
| 5999 | EDGE DETECTION AND SEAM TRACKING WITH EMATS | | | | | | | | | | | |
| | Granted | 18-Mar-1999 | 09/271742 | | | 05-Dec-2000 | 6155117 | McDermott Technology, Inc. | | | | |
| 6002 | AIR HEA | TER GAS INL | ET PLENUM | Í | | | | | | | | |
| | Granted | 22-Jul-1999 | 09/359098 | | | 18-Jun-2002 | 6405791 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| C | Granted | 19-Mar-2002 | 10/101645 | 25-Jul-2002 | 02-0096316 | 17-Dec-2002 | 6494255 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6003 | WATER JACKETED, HIGH TEMPERATURE, BOILER, STRETCHER, ACCESS DOOR | | | | | | | | | | | |
| | Granted | 21-Jan-1999 | 09/234687 | | | 09-Jul-2002 | 6415724 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6004 | MAIN BA | NG RECOVE | RY EMAT | | | | | | | | | |
| | Granted | 24-Aug-1998 | 09/138620 | | | 26-Sep-2000 | 6122969 | McDermott Technology, Inc. | | | | |
| 6006 | MERCUI | RY REMOVAL | IN UTILITY | WET SCRUB | BER USING A | A CHELATING | AGENT | | | | | |
| | Granted | 31-Mar-1999 | 09/282816 | | | 11-Dec-2001 | 6328939 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6012 | CIRCUL | ATING FLUIDI | ZED BED R | EACTOR WIT | TH FLOORED | INTERNAL PI | RIMARY P | PARTICLE SEPARATOR | | | | |
| | Granted | 07-Dec-1998 | 09/206353 | | | 01-Aug-2000 | 6095095 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6013 | APPARA | TUS FOR CON | TROL OF M | IERCURY | | | | | | | | |
| | Granted | 31-Mar-1999 | 09/282817 | | | 04-Sep-2001 | 6284199 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6014 | OXIDAT | ION RESISTAN | NT LOW AL | LOY ATTACH | IMENT FOR E | BOILER COMP | PONENTS | | | | | |
| | Granted | 14-Jan-1999 | 09/231914 | | | 27-Nov-2001 | 6321691 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6016 | ALKALI | NE SORBENT | INJECTION | FOR MERCU | RY CONTROL | L | | | | | | |
| | Granted | 07-Dec-1998 | 09/206465 | | | 16-Apr-2002 | 6372187 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| C | Granted | 16-Nov-2001 | 09/993230 | 01-Aug-2002 | 02-0102189 | 04-Mar-2003 | 6528030 | McDermott Technology, Inc. | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 82 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|---------|--------------|------------|------------|---------------|-------------|------------|---|
| 6017 | | | | | | | | ES FOR SURFACE DEFECTS |
| 0017 | Granted | 03-Feb-1999 | 09/243592 | 1011102001 | 311 (23.71.12 | 26-Dec-2000 | | McDermott Technology, Inc. |
| 6018 | | ED FLUE GAS | | ZATION AN | D CARBO | | | |
| | Granted | 04-Jun-1999 | 09/326450 | | | 04-Jun-2002 | | Babcock & Wilcox Power Generation Group, Inc. |
| 6019 | LOW PR | ESSURE DROP | INLET FOR | HIGH VELO | OCITY AI | BSORBERS WI | | ** |
| | Granted | 18-Jun-1999 | 09/336412 | | | 31-Jul-2001 | 6267358 | Babcock & Wilcox Power Generation Group, Inc. |
| 6024 | METHO | D OF REDUCIN | G NOx EMIS | SIONS WIT | H MINIM | AL INCREASE | ES IN UNBU | JRNED CARBON AND WATERWALL CORROSION |
| | Granted | 13-Sep-1999 | 09/394163 | | | 20-Nov-2001 | 6318277 | Babcock & Wilcox Power Generation Group, Inc. |
| 6029 | MACHIN | NE AND METHO | D FOR PREV | VENTING F | LANGE I | LEAKAGE | | |
| | Granted | 31-Mar-1999 | 09/281891 | | | 20-Mar-2001 | 6203019 | Babcock & Wilcox Power Generation Group, Inc. |
| 6030 | WALL P | ROTECTION FI | ROM DOWN | WARD FLO | WING SO | LIDS | | |
| | Granted | 06-May-1999 | 09/305962 | | | 04-Apr-2000 | 6044805 | Babcock & Wilcox Power Generation Group, Inc. |
| 6036 | ULTRA- | HIGH PARTICU | LATE COLL | ECTION O | F SUB-MI | CRON AEROS | OLS | |
| | Granted | 23-Apr-1999 | 09/298778 | | | 29-May-2001 | 6238459 | McDermott Technology, Inc. |
| 6040 | APPARA | TUS TO RECOV | VER SULFUR | R FROM CO | NCENTR | ATED ACID G | AS INTO A | LKALINE SOLUTION |
| | Granted | 08-Nov-1999 | 09/435745 | | | 27-Aug-2002 | 6440379 | The Babcock & Wilcox Company |
| 6050 | HORIZO | NTAL SPIRAL | TUBE BOILE | CR CONVEC | CTION PA | SS ENCLOSU | RE DESIGN | I |
| A | Granted | 16-Dec-2002 | 10/320342 | | | 13-Apr-2004 | 6718915 | Babcock & Wilcox Power Generation Group, Inc. |
| 6052 | SCALLO | PED LATTICE | BAR NUCLE | AR STEAM | GENERA | TOR TUBE SU | PPORTS | |
| | Granted | 21-Mar-2000 | 09/531625 | | | 09-Apr-2002 | 6367430 | Babcock & Wilcox Canada Ltd. |
| 6057 | SOLAR | THERMAL ROC | CKET | | | | | |
| | Granted | 26-Oct-1999 | 09/427844 | | | 18-Sep-2001 | 6290185 | Babcock & Wilcox Nuclear Operations Group, Inc. |
| 6058 | SOLAR | THERMAL ROC | CKET | | | | | |
| | Granted | 01-May-2000 | 09/563046 | | | 05-Feb-2002 | 6343464 | BWX Technologies, Inc. |

 $^{{}^{\}star}\;\;McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,2010$

Thursday, April 29, 2010 Prepared by AMSaus

Page 83 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | | | |
|--------------------|--|--------------|------------|--------------|-------------|---------------|------------|---|--|--|--|--|
| 6061 | | | | | | MERCURY FRO | | | | | | |
| | Granted | 17-Dec-1999 | 09/464806 | | | 07-Jan-2003 | 6503470 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| C1 | Granted | 20-Dec-2002 | 10/324833 | 20-Dec-2002 | 03-0091490 | 02-May-2006 | 7037474 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6064 | METHOI | O AND APPAR | ATUS FOR A | AIR SPARGED | SLURRY TA | NKS | | _ | | | | |
| | Granted | 04-Aug-1999 | 09/366844 | | | 11-Jul-2000 | 6086658 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6067 | METHOD FOR PRODUCING CHROMIUM CARBIDE COATINGS | | | | | | | | | | | |
| | Granted | 02-Aug-2000 | 09/631207 | | | 07-Jan-2003 | 6503340 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6072 | FINE SO | LIDS RECYCL | E IN CIRCU | LATING FLUI | DIZED BED | | | | | | | |
| | Granted | 17-Dec-1999 | 09/464258 | | | 07-Aug-2001 | 6269778 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6074 | HEAT EX | KCHANGER TU | UBE SUPPO | RT STRUCTUI | RE | | | | | | | |
| | Granted | 01-Nov-1999 | 09/431589 | | | 24-Dec-2002 | 6498827 | Babcock & Wilcox Canada Ltd. | | | | |
| C | Granted | 31-Oct-2002 | 10/285178 | 29-Apr-2004 | 04-0081268 | 26-Oct-2004 | 6810101 | Babcock & Wilcox Canada Ltd. | | | | |
| D | Granted | 08-Jan-2004 | 10/753733 | | | 05-Jul-2005 | 6914955 | Babcock & Wilcox Canada Ltd. | | | | |
| 6075 | HYDRAU | JLIC EXPANSI | ON PRE-ST | RAINING OF I | HEAT EXCHA | NGER TUBING | | | | | | |
| | Granted | 01-Nov-1999 | 09/432564 | | | 19-Mar-2002 | 6357114 | Babcock & Wilcox Canada Ltd. | | | | |
| 6086 | MODIFIC | CATION OF DI | FFUSION C | OATING GRA | IN STRUCTU | RE BY NITRIDI | NG | | | | | |
| | Granted | 13-Dec-1999 | 09/460129 | | | 06-Mar-2001 | 6197125 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6088 | FURNAC | E FLOOR | | | | | | | | | | |
| | Granted | 30-Mar-2000 | 09/539257 | | | 27-Mar-2001 | 6205634 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6092 | DRUMLE | ESS NATURAL | CIRCULAT | TION BOILER | | | | | | | | |
| | Granted | 01-Jun-2000 | 09/585878 | | | 08-Jan-2002 | 6336429 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6093 | PARTICU | ULATE COLLE | ECTOR CHA | NNEL WITH | COOLING IN | NER ELEMENT | S IN A CFE | B BOILER | | | | |
| | Granted | 10-Jul-2000 | 09/613132 | | | 27-Nov-2001 | 6322603 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6094 | SINGLE | SPRAY LEVEI | FOR FLUE | GAS DESULF | URIZATION | SYSTEM WITH | INTERNA | L MAIN FEED HEADERS | | | | |
| | Granted | 01-Oct-2001 | | 03-Apr-2003 | | 02-Sep-2003 | 6613133 | Babcock & Wilcox Power Generation Group, Inc. | | | | |
| 6096 | INTEGR | | | MONIA INJEC | TION GRID F | OR SCR SYSTE | | | | | | |
| | Granted | 23-Jun-2000 | 09/602483 | | | 03-May-2005 | 6887435 | Babcock & Wilcox Power Generation Group, Inc. | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 84 of 34

| CASE NO SubCase | CT A TITE | FILE DATE | APP NO | PUB DATE | PUB NO | ICCUE DATE | DATNO | OWNER |
|--------------------|-----------|--------------|------------|--------------|--------------|---------------|-----------|---|
| 6098 | STATUS | ROTECTION FI | | | | ISSUE DATE | PAT NO | OWNER |
| 0090 | | | | WARDFLOW | ING SOLIDS | 10 D 2002 | (401000 | |
| (100 | Granted | 06-Nov-2001 | 09/979615 | L CTOD WITH | I ODI DOMINE | 10-Dec-2002 | 6491000 | Babcock & Wilcox Power Generation Group, Inc. |
| 6102 | | | | ACTOR WITE | 1 SELECTIVE | CATALYTIC RE | | |
| | Granted | 13-Feb-2000 | 09/503218 | | | 28-May-2002 | 6395237 | Babcock & Wilcox Power Generation Group, Inc. |
| 6103 | IMPROV | EMENTS IN EL | TYPE PULV | VERIZERS | | | | |
| | Granted | 27-Oct-2000 | 09/698558 | | | 05-Apr-2005 | 6874715 | Babcock & Wilcox Power Generation Group, Inc. |
| 6110 | VARIAB | LE RESISTANC | E DEVICE F | OR A HIGH P | RESSURE AIR | SUPPLY SYSTE | M | |
| | Granted | 22-Jun-2000 | 09/599326 | | | 14-Aug-2001 | 6273141 | Babcock & Wilcox Power Generation Group, Inc. |
| 6111 | COOLEI | TUBES ARRA | NGED TO FO | ORM IMPACT | TYPE PARTIC | CLE SEPARATO | RS | |
| | Granted | 25-May-2001 | 09/865609 | 21-Feb-2002 | 0020158-A1 | 31-Dec-2002 | 6500221 | Babcock & Wilcox Power Generation Group, Inc. |
| 6112 | CFB IMP | ACT TYPE PAR | RTICLE COI | LECTION EL | EMENTS ATT | ACHED TO COO | OLED SUPI | PORTS |
| | Granted | 25-May-2001 | 09/865332 | | | 24-Sep-2002 | 6454824 | Babcock & Wilcox Power Generation Group, Inc. |
| 6117 | SYSTEM | FOR CONTRO | LLING FLUI | E GAS EXIT T | EMPERATURI | E FOR OPTIMAI | L SCR OPE | RATIONS |
| | Granted | 27-Feb-2002 | 10/085715 | 28-Aug-2003 | 03-0159662 | 26-Aug-2003 | 6609483 | Babcock & Wilcox Power Generation Group, Inc. |
| 6119 | SOLAR 7 | THERMAL ROC | CKET | | | | | |
| | Granted | 02-Jun-2000 | 09/586057 | | | 02-Jul-2002 | 6412274 | Babcock & Wilcox Nuclear Operations Group, Inc. |
| 6121 | METHO | D FOR CONTRO | DLLING ELE | EMENTAL ME | RCURY EMIS | SIONS | | |
| | Granted | 05-Dec-2000 | 09/730100 | 06-Jun-2002 | 02-0068030 | 15-Feb-2005 | 6855859 | Babcock & Wilcox Power Generation Group, Inc. |
| 6128 | TAPERE | D CORROSION | PROTECTION | ON OF TUBES | AT MUD DRU | M LOCATION | | |
| | Granted | 28-Sep-2000 | 09/670897 | | | 17-Dec-2002 | 6495268 | Babcock & Wilcox Power Generation Group, Inc. |
| D | Granted | 31-Oct-2002 | 10/284625 | 20-Mar-2003 | 03/0051779 | 05-Oct-2004 | 6800149 | Babcock & Wilcox Power Generation Group, Inc. |
| 6130 | SPACER | BAR WITH TU | BE SLEEVE | AND TAB | | | | |
| | Granted | 09-Oct-2000 | 09/680725 | | | 14-Aug-2001 | 6273030 | Babcock & Wilcox Power Generation Group, Inc. |
| 6156 | FUEL CE | ELL COLUMN F | IEAT EXCH. | ANGER MATE | ED MODULE | | | |
| | Granted | 02-Jan-1996 | 581989 | | | 18-Mar-1997 | 5612149 | SOFCo-EFS Holdings, LLC |
| 6157 | FUEL CE | ELL MODULE | | | | | | |
| | Granted | 04-Feb-1994 | 192416 | | | 02-Jan-1996 | 5480738 | SOFCo-EFS Holdings, LLC |

 $^{{}^{\}star}\;\;McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,2010$

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|---------|--------------|------------|--------------|--------------|----------------|-----------|---|
| 6158 | THERMA | ALLY INTEGR | ATED REFO | ORMER FOR SO | LID OXIDE F | UEL CELLS | | |
| | Granted | 06-Oct-1993 | 08/133444 | | | 22-Nov-1994 | 5366819 | SOFCo-EFS Holdings, LLC |
| 6159 | THERMA | ALLY INTEGR | ATED HEAT | Γ EXCHANGE S | YSTEM FOR | SOLID OXIDE I | ELECTRO | LYTE SYSTEMS |
| | Granted | 29-Sep-1993 | 128522 | | | 23-Aug-1994 | 5340664 | SOFCo-EFS Holdings, LLC |
| 6165 | INCINER | RATION GRAT | E WITH DU | CTS FOR CONV | EYING A HE | AT TRANSMIS | SION MED | IUM |
| | Granted | 13-Jun-1997 | 08/860245 | | | 04-May-1999 | 5899149 | Babcock & Wilcox Volund A/S |
| 6170 | MAIN BA | NG RECOVE | RY EMAT | | | | | |
| | Granted | 20-Jul-2000 | 09/620083 | | | 22-May-2001 | 6234026 | McDermott Technology, Inc. |
| 6174 | BIO-MAS | SS STRAW BAI | LE SHREDD | ER-FEEDER | | | | |
| | Granted | 14-Feb-2000 | 09/503322 | | | 25-Jun-2002 | 6409110 | Babcock & Wilcox Volund A/S |
| 6185 | CFB WIT | TH CONTROLI | LABLE IN-B | ED HEAT EXC | IANGER | | | |
| | Granted | 17-Jul-2001 | 09/906993 | 23-Jan-2003 | 03-0015150 | 18-Mar-2003 | 6532905 | Babcock & Wilcox Power Generation Group, Inc. |
| 6187 | NON ME | TALLIC HYDI | RAULIC EXI | PANSION MANI | DREL | | | |
| | Granted | 19-Feb-2002 | 10/078751 | | | 25-Mar-2003 | 6536252 | Babcock & Wilcox Canada Ltd. |
| 6207 | ATTACH | ABLE HEAT F | FLUX MEAS | URING DEVICE | Ē | | | |
| | Granted | 27-Oct-2000 | 09/697866 | | | 26-Nov-2002 | 6485174 | Babcock & Wilcox Power Generation Group, Inc. |
| 6208 | PROCES | S AND COMPO | OSITION FO | R CHROMIZIN | G 400-SERIES | S STAINLESS ST | TEELS | |
| | Granted | 20-Feb-2001 | 09/788742 | | | 14-May-2002 | 6387194 | Babcock & Wilcox Power Generation Group, Inc. |
| 6217 | BOILER | INTERNAL FL | LUE GAS BY | -PASS DAMPER | R FOR FLUE (| GAS TEMPERA | TURE CON | TROL |
| | Granted | 19-Dec-2001 | 10/024874 | 27-Jun-2002 | 02-0079374 | 15-Jun-2004 | 6748880 | Babcock & Wilcox Power Generation Group, Inc. |
| 6219 | HIGH TE | EMPERATURE | GASEOUS | OXIDATION FO | R PASSIVAT | ION OF AUSTE | NITIC ALI | LOY |
| | Granted | 30-Mar-2001 | 09/821873 | | | 03-Dec-2002 | 6488783 | Babcock & Wilcox Canada Ltd. |
| C | Granted | 18-Nov-2002 | 10/298681 | 26-Jun-2003 | 03-0116229 | 06-Jul-2004 | 6758917 | Babcock & Wilcox Canada Ltd. |
| 6223 | SUPERC | ONDUCTING (| CABLE-IN-C | CONDUIT COND | OUCTOR LOW | RESISTANCE | SPLICE JO | DINT |
| | Granted | 11-Jan-2002 | 10/045813 | | | 24-Jun-2003 | 6583351 | Babcock & Wilcox Nuclear Operations Group, Inc. |
| 7000 | CIRCUL | ATING FLUID | IZED BED R | EACTOR WITH | SELECTIVE | CATALYTIC R | EDUCTIO | N |
| | Granted | 06-Feb-2001 | 09/958622 | 02-May-2005 | 02-0182122 | 22-Nov-2005 | 6967005 | Babcock & Wilcox Power Generation Group, Inc. |
| | | | | | | | | |

Page 86 of 34

* McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

| SubCase ST | ΓATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|------------|--------|--------------|-------------|---------------|-------------|--------------|----------|---|
| | | ARRIER | | | | | - | |
| | | 13-Aug-2001 | 09/928323 | 13-Feb-2003 | 03-0029583 | 24-Jun-2003 | 6581667 | Babcock & Wilcox Canada Ltd. |
| | | C | | THE PRODUCT | | | | |
| | | 29-Oct-2001 | 10/039856 | 01-May-2003 | 03-0079343 | | | Babcock & Wilcox Nuclear Operations Group, Inc. |
| 7004 CH | IANNEI | LIZED SCR IN | LET FOR IM | PROVED AMMO | ONIA INJECT | TION AND EFF | | 1 |
| Gra | anted | 29-Jun-2001 | 09/896169 | 02-Jan-2003 | 03-0003029 | 14-Jun-2005 | 6905658 | Babcock & Wilcox Power Generation Group, Inc. |
| 7005 AR | RTICUL | ATING WATE | ER MONITOR | CLEANING DE | VICE | | | • |
| Gra | anted | 26-Dec-2001 | 10/036173 | 03-Oct-2002 | 02-0139402 | 02-Dec-2003 | 6655397 | Diamond Power International, Inc. |
| 7007 CO |)MPAR | TMENTED GA | AS OUTLET F | OR NOX CONT | ROL AND MI | XING FOR SC | R | |
| Gra | anted | 23-Jan-2002 | 10/055317 | | | 28-Jan-2003 | 6510820 | Babcock & Wilcox Power Generation Group, Inc. |
| 7009 HE | EAT EX | CHANGER TU | BE SUPPOR | ΓBAR | | | | |
| Gra | anted | 23-Apr-2002 | 10/127912 | 23-Oct-2003 | 03-0196786 | 10-Aug-2004 | 6772832 | Babcock & Wilcox Canada Ltd. |
| 7010 FL | OOREI | IMPACT TY | PE SOLIDS S | EPARATOR USI | NG DOWNW | ARD EXPAND | ING SEPA | RATOR ELEMENTS |
| Gra | anted | 18-Oct-2002 | 10/273917 | | | 27-Jan-2004 | 6681722 | Babcock & Wilcox Power Generation Group, Inc. |
| 7012 SO | OTBLC | OWER NOZZL | E ASSEMBLY | Y WITH AN IMP | ROVED DOV | VNSTREAM NO | OZZLE | |
| Gra | anted | 02-Jan-2002 | 10/039430 | 05-Sep-2002 | 02/0121563 | 20-Jul-2004 | 6764030 | Diamond Power International, Inc. |
| 7013 SO | OTBLC | OWER MECHA | ANISM PROV | IDING VARYIN | G LANCE RO | TATION SPEE | ED | |
| Gra | anted | 20-Dec-2001 | 10/029,900 | 26-Jun-2002 | 02-0078982 | 10-Aug-2004 | 6772775 | Diamond Power International, Inc. |
| 7015 SO | OTBLC | OWER LANCE | TUBE FOR I | DUAL CLEANIN | G MEDIA | | | |
| Gra | anted | 05-Feb-2002 | 10/049303 | 06-Nov-2003 | 0205261 | 31-Aug-2004 | 6782902 | Diamond Power International, Inc. |
| 7017 TO | OOL FO | R LOCALLY I | FORMING A | HORIZONTAL V | WORKPIECE | | | |
| Gra | anted | 26-Jun-2002 | 10/180694 | | | 26-Aug-2003 | 6609404 | Babcock & Wilcox Canada Ltd. |
| 7020 HY | /DRAUI | LIC LOADING | S SYSTEM FO | OR BALL AND R | ING PULVER | RIZERS | | |
| | | 07-Sep-2001 | 09/948823 | 13-Mar-2003 | 03/0047629 | 26-Aug-2003 | 6609669 | Babcock & Wilcox Power Generation Group, Inc. |
| | | TING SOOTBI | | | | | | |
| Gra | anted | 19-Jul-2002 | 10/199626 | 23-Jan-2003 | 03-0015151 | 10-Jun-2003 | 6575122 | Diamond Power International, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 87 of 34

| 70 Parameter Function Func | CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--|--------------------|----------|----------------|-------------|--------------|-------------|---------------|----------|---|
| Act Granted 0-Ro-e-1999 09/45638 ** 29-May-2001 6239831 Diamond Power International, Inc. A1 Granted 0-Go-ti-2000 09/65087 ** 17-Aug-2004 6778209 Diamond Power International, Inc. A2 Granted 10-Jan-2001 09/75809 16-Aug-2001 018982 19-Oct-2004 680690 Diamond Power International, Inc. 7026 FURA SET | 7025 | FURNAC | E VIDEO CAME | ERA APPAR | ATUS | | | | |
| Part Figure Fig | | Granted | 26-Sep-1997 | 08/938195 | | | 30-May-2000 | 6069652 | Diamond Power International, Inc. |
| A2 Granted 10-Jan-2001 09/758019 16-Aug-2001 09/1892 19-Oct-2004 680690 Diamord Power International, Inc. 702 Granted 14-May-1909 252/3684 - Team 01-Jan-1901 498108 Diamord Power International, Inc. 702 LOW PUBLIC, IN-LIUS EVELEUS - Team 29-Jun-2004 675510 Babcock Wilcox Canada Ltd. 702 Cranted 29-May-2003 10/44774 - Team 29-Jun-2004 675510 Babcock Wilcox Canada Ltd. 702 EXTER-VISE USE USE USE USE USE USE USE USE USE U | A | Granted | 08-Dec-1999 | 09/456638 | | | 29-May-2001 | 6239831 | Diamond Power International, Inc. |
| 7026 Grante I-May-1990 S2864 February S2864 | A1 | Granted | 06-Oct-2000 | 09/680578 | | | 17-Aug-2004 | 6778209 | Diamond Power International, Inc. |
| Figure F | A2 | Granted | 10-Jan-2001 | 09/758019 | 16-Aug-2001 | 0013892 | 19-Oct-2004 | 6806900 | Diamond Power International, Inc. |
| 7027 Garded 29-May-2003 10/44774 2004 29-Jun-2004 675510 36-dock Wilcox Canada Ltd. 7028 EXTENTIAL SUSTED BUSING Garded 22-Apr-2004 10/82951 27-Oct-2005 05-023571 06-Feb-2007 717183 36-dock Wilcox Canada Ltd. 7029 LASEN FULDED BOILET TUBE WITTOWN W | 7026 | SLAG EL | IMINATOR FO | R FURNACI | E VIEWING SY | STEM | | | |
| 7028 29-May-2003 10/447748 29-Jun-2004 675510 Backer & Wilcox Canada Ltd. 7028 EXTERY & SUEVEY ASSETTED TUSE WED TO TUSE WED TUSE WED TO TUSE WED TO TUSE WED TO TUSE WED TUSE WED TO TUSE WED T | | Granted | 14-May-1990 | 523684 | | | 01-Jan-1991 | 4981088 | Diamond Power International, Inc. |
| 7028 EXTERN JEEVE ASSITED TUSE BENDING Granted 22-Apr-2004 10/829514 27-Oct-2005 05-0235719 06-Feb-2007 7171834 Babcock & Wilcox Canada Ltd. 7029 LASER WEDDROILER TUBE WELL PANELS Granted 19-Jun-2002 10/175297 25-Dec-2003 03-0234241 08-Feb-2005 6852945 Babcock & Wilcox Power Generation Group, Inc. TO30 DIAPH ZGR FOR SEALING OPENINGS IN PRESSURY VESSURY Granted 26-Apr-2002 10/13314 30-Oct-2003 03-0201607 26-Jul-2005 692109 Babcock & Wilcox Canada Ltd. TO33 TWO-FUTD HOUSING LINER Granted 25-Jan-2002 10/056825 31-Jul-2003 03-0141396 31-Aug-2004 6783091 Babcock & Wilcox Power Generation Group, Inc. TO35 DIAPH SUNGER FUTD HELD HELD HELD HELD HELD HELD HELD HEL | 7027 | LOW PR | OFILE, IN-LINE | BEVELER | | | | | |
| Franted Record | | Granted | 29-May-2003 | 10/447748 | | | 29-Jun-2004 | 6755101 | Babcock & Wilcox Canada Ltd. |
| Part | 7028 | EXTERN | AL SLEEVE ASS | SISTED TUE | BE BENDING | | | | |
| Frante | | Granted | 22-Apr-2004 | 10/829514 | 27-Oct-2005 | 05-0235719 | 06-Feb-2007 | 7171834 | Babcock & Wilcox Canada Ltd. |
| 7030 DIAPHR → GR SEA | 7029 | LASER V | VELDED BOILE | R TUBE WA | ALL PANELS | | | | |
| Granted 26-Apr-2002 10/133514 30-Oct-2003 03-0201607 26-Jul-2005 6921090 Babcock & Wilcox Canada Ltd. 7035 TWO-FLUTED HOUSING LINER | | Granted | 19-Jun-2002 | 10/175297 | 25-Dec-2003 | 03-0234241 | 08-Feb-2005 | 6852945 | Babcock & Wilcox Power Generation Group, Inc. |
| 7033 TWO-F LINER Granted 25-Jan-2002 10/056825 31-Jul-2003 03-0141396 31-Aug-2004 6783091 Babcock & Wilcox Power Generation Group, Inc. 7035 DUAL F L. BURNER FOLK A SHORT ENED FLAME AND REDUCED POLLUTANTEMISSUMS 7036 Granted 21-May-2003 10/443287 25-Nov-2004 040234912 04-Oct-2005 695145 Babcock & Wilcox Power Generation Group, Inc. 7036 COMPACT FOOTPRINT CFB WITH MECHANICAL DUST COLLECTOR 7039 ANTI-VISATION SUPPERS TOR SUPPERS TOR SUPPERS TOR HEAT TOR HEAT TRANSFER TUBES AND ETHOD FOR MAKING SAME 7039 Granted 28-Oct-2003 10/695154 12-Jan-2006 0005787 14-Feb-2006 699714 Babcock & Wilcox Canada Ltd. 7040 FEEDW JETER APPARATUS (was IMPERS) Was IMPERS TOR HEAT | 7030 | DIAPHR | AGM FOR SEAL | ING OPENI | NGS IN PRESS | SURE VESSEI | LS | | |
| Granted 25-Jan-2002 10/056825 31-Jul-2003 03-0141396 31-Aug-2004 6783091 Babcock & Wilcox Power Generation Group, Inc. | | Granted | 26-Apr-2002 | 10/133514 | 30-Oct-2003 | 03-0201607 | 26-Jul-2005 | 6921090 | Babcock & Wilcox Canada Ltd. |
| 7035 DUAL FUEL BURNER FOR A SHOR TENED FLAME AND REDUCED POLLUTANT EMISSIONS Granted 21-May-2003 10/443287 25-Nov-2004 040234912 04-Oct-2005 6951454 Babcock & Wilcox Power Generation Group, Inc. 7036 COMPACT FOOTPRINT UFB WITH WECHANICAL DUST COLUCTOR Granted 30-Apr-2002 10/135432 30-Oct-2003 03-0202913 08-Mar-2005 6863703 Babcock & Wilcox Power Generation Group, Inc. 7039 ANTI-VIBRATION SUPPORT FOR STEAM GENERATOR HEAT TRANSFER TUBES AND METHOD FOR MAKING SAME Granted 28-Oct-2003 10/695154 12-Jan-2006 0005787 14-Feb-2006 6997141 Babcock & Wilcox Canada Ltd. 7040 FEEDWARATUS (was IMPER) SUPEN SUPEN SUPEN SUPEN SUZZLE & SPARGER ASSEMBLY) | 7033 | TWO-FL | UTED HOUSING | G LINER | | | | | |
| Granted 21-May-2003 10/443287 25-Nov-2004 040234912 04-Oct-2005 6951454 Babcock & Wilcox Power Generation Group, Inc. 7036 COMPACT FOOTPRINT CFB WITH WECHANICAL DUST COLLECTOR Granted 30-Apr-2002 10/135432 30-Oct-2003 03-0202913 08-Mar-2005 6863703 Babcock & Wilcox Power Generation Group, Inc. 7039 ANTI-VIBATION SUPPORT FOR SUPPORT FOR SUPPORT FOR MAKING SAME Granted 28-Oct-2003 10/695154 12-Jan-2006 0005787 14-Feb-2006 6997141 Babcock & Wilcox Canada Ltd. 7040 FEEDWAT APPARATUS (was IMPORT) DESIGN OF FEEDWAT PIPE, NOZZLE & SLEEVE- SPARGER ASSEMBLY) | | Granted | 25-Jan-2002 | 10/056825 | 31-Jul-2003 | 03-0141396 | 31-Aug-2004 | 6783091 | Babcock & Wilcox Power Generation Group, Inc. |
| 7036 COMPAUT FOOTPRINT UFB WITH MECHANICAL DUST COLLECTOR Granted 30-Apr-2002 10/135432 30-Oct-2003 03-0202913 08-Mar-2005 6863703 Babcock & Wilcox Power Generation Group, Inc. 7039 ANTI-VIBRATION SUPPORT FOR STEAM GENERATOR HEAT TRANSFER TUBES AND METHOD FOR MAKING SAME Granted 28-Oct-2003 10/695154 12-Jan-2006 0005787 14-Feb-2006 6997141 Babcock & Wilcox Canada Ltd. 7040 FEEDWATER APPARATUS (was IMPROVED DESIGNOS) OF FEEDWATER PIPE, NOZZLE & SLEEVE- SPARGER ASSEMBLY) | 7035 | DUAL FU | EL BURNER FO | OR A SHOR | TENED FLAM | E AND REDU | CED POLLUTA | NT EMISS | IONS |
| Granted 30-Apr-2002 10/135432 30-Oct-2003 03-0202913 08-Mar-2005 6863703 Babcock & Wilcox Power Generation Group, Inc. | | Granted | 21-May-2003 | 10/443287 | 25-Nov-2004 | 040234912 | 04-Oct-2005 | 6951454 | Babcock & Wilcox Power Generation Group, Inc. |
| 7039 ANTI-VIBRATION SUPPORT FOR STEAM GENERATOR HEAT TRANSFER TUBES AND METHOD FOR MAKING SAME Granted 28-Oct-2003 10/695154 12-Jan-2006 0005787 14-Feb-2006 6997141 Babcock & Wilcox Canada Ltd. 7040 FEEDWATER APPARATUS (was IMPROVED DESIGN OF FEEDWATER PIPE, NOZZLE & SLEEVE- SPARGER ASSEMBLY) | 7036 | COMPAC | CT FOOTPRINT | CFB WITH | MECHANICA | L DUST COLI | LECTOR | | |
| Granted 28-Oct-2003 10/695154 12-Jan-2006 0005787 14-Feb-2006 6997141 Babcock & Wilcox Canada Ltd. 7040 FEEDWATER APPARATUS (was IMPROVED DESIGN OF FEEDWATER PIPE, NOZZLE & SLEEVE- SPARGER ASSEMBLY) | | Granted | 30-Apr-2002 | 10/135432 | 30-Oct-2003 | 03-0202913 | 08-Mar-2005 | 6863703 | Babcock & Wilcox Power Generation Group, Inc. |
| 7040 FEEDWATER APPARATUS (was IMPROVED DESIGN OF FEEDWATER PIPE, NOZZLE & SLEEVE- SPARGER ASSEMBLY) | 7039 | ANTI-VII | BRATION SUPP | ORT FOR S | TEAM GENER | ATOR HEAT | TRANSFER TU | BES AND | METHOD FOR MAKING SAME |
| | | Granted | 28-Oct-2003 | 10/695154 | 12-Jan-2006 | 0005787 | 14-Feb-2006 | 6997141 | Babcock & Wilcox Canada Ltd. |
| Granted 05-May-2003 10/429360 23-Mar-2004 6708651 Babcock & Wilcox Canada Ltd. | 7040 | FEEDWA | TER APPARAT | US (was IMI | PROVED DESI | GN OF FEED | WATER PIPE, N | OZZLE & | SLEEVE- SPARGER ASSEMBLY) |
| | | Granted | 05-May-2003 | 10/429360 | | | 23-Mar-2004 | 6708651 | Babcock & Wilcox Canada Ltd. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 88 of 34

| The Babcock & Wilcox Company US Active Patents and Applications | The Babcock & | Wilcox | Company | US Active | Patents and | Applications |
|---|---------------|--------|---------|-----------|-------------|---------------------|
|---|---------------|--------|---------|-----------|-------------|---------------------|

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|---------|---------------|------------|--------------|-------------|---------------|----------|--|
| 7041 | | | | | | | | ION (WFGD) SYSTEM TO AN UPFLOW |
| /041 | | LOOP WFGD SY | | TTLO W/CITI | OW WEITE | OE GAS DESCI | TURIZATI | ion (wrdb) 5151EM TO AN OFFEOW |
| | Granted | 16-Jul-2002 | 10/196597 | 22-Jan-2004 | 04-0011200 | 24-Apr-2004 | 6726748 | Babcock & Wilcox Power Generation Group, Inc. |
| 7042 | IMPROV | ED COLLECTION | ON SCOOP I | FOR FGD BLE | ED STREAM | OR EX SITU OX | IDATION | |
| | Granted | 10-Jun-2002 | 10/166589 | 11-Dec-2003 | 03-0228247 | 24-Feb-2004 | 6695018 | Babcock & Wilcox Power Generation Group, Inc. |
| D1 | Granted | 16-Dec-2003 | 10/736914 | 08-Jul-2004 | 040131526 | 09-Aug-2005 | 6926870 | Babcock & Wilcox Power Generation Group, Inc. |
| D2 | Granted | 16-Dec-2003 | 10/737499 | 08-Jul-2004 | 040131525 | 13-Nov-2007 | 7294322 | Babcock & Wilcox Power Generation Group, Inc. |
| 7043 | BUBBLE | CAP ASSEMBL | Y | | | | | |
| | Granted | 30-May-2003 | 10/447895 | 02-Dec-2004 | 040237858 | 22-Mar-2005 | 6868795 | Blasch Precision Ceramics, Inc. Babcock & Wilcox Power Generation Group, Inc. |
| 7044 | STEAM (| GENERATOR TU | UBE SUPPO | RT WITH SLO | TTED DISC S | PRINGS | | |
| | Granted | 26-Mar-2003 | 10/397910 | | | 06-Jan-2004 | 6672260 | Babcock & Wilcox Canada Ltd. |
| 7045 | CHILLIN | G SLEEVE FOR | R EXPANSIO | N-FITTING H | OLLOW CYL | INDERS | | |
| | Granted | 13-Feb-2003 | 10/367242 | | | 03-Aug-2004 | 6769262 | Babcock & Wilcox Canada Ltd. |
| 7046 | METHOI | O FOR CLEANIN | NG TAR-BEA | ARING WASTI | E WATER AN | D APPARATUS | FOR PERF | ORMING SAID METHOD |
| | Granted | 19-May-2004 | 10/480588 | 23-Sep-2004 | 0186338 | 08-Jul-2008 | 7396454 | Babcock & Wilcox Volund A/S |
| C | Granted | 12-Feb-2008 | 12/029845 | | | 12-Dec-2009 | 7626069 | Babcock & Wilcox Volund A/S |
| 7047 | EMISSIV | TTY PROBE | | | | | | |
| | Granted | 13-Aug-2002 | 10/218067 | 19-Feb-2004 | 04-0032583 | 21-Jun-2005 | 6909495 | Diamond Power International, Inc. |
| 7048 | FLUE GA | AS DESULFURIZ | ZATION SYS | STEM WITH A | STEPPED TR | RAY | | |
| | Granted | 24-Oct-2002 | 10/279425 | 29-Apr-2004 | 04-0079235 | 02-Aug-2005 | 6923852 | Babcock & Wilcox Power Generation Group, Inc. |
| 7052 | PASSIVE | SYSTEM FOR O | OPTIMAL N | Ox REDUCTION | ON VIA SELE | CTIVE CATALY | TIC REDU | JCTION |
| | Granted | 06-Sep-2002 | 10/236854 | 11-Mar-2004 | 04-0045513 | 04-Apr-2006 | 7021248 | Babcock & Wilcox Power Generation Group, Inc. |
| 7053 | SEGMEN | TED RING-NUT | FOR GASK | KETED PRESSI | URE VESSEL | OPENING | | |
| | Granted | 02-Dec-2002 | 10/307756 | | | 23-Dec-2003 | 6666637 | Babcock & Wilcox Canada Ltd. |
| 7057 | REACTO | R HEAD WITH | INTEGRAL | NOZZLES | | | | |
| | Granted | 18-Nov-2002 | 10/298856 | 21-Apr-2005 | 05-0084056 | 03-May-2005 | 6888908 | Babcock & Wilcox Canada Ltd. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 89 of 34

| The Babcock & Wilcox Company US Active Patents and Application | The Ba | ibcock & | Wilcox | Company | US Active | Patents and | d Applications |
|--|--------|----------|--------|---------|-----------|-------------|----------------|
|--|--------|----------|--------|---------|-----------|-------------|----------------|

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|----------|---------------|-------------|--------------|--------------|--------------|----------|---|
| 7060 | INTERM | ITTENT MIXEI | R WITH LOV | V PRESSURE I | OROP | | | |
| | Granted | 18-Mar-2003 | 10/391393 | 23-Sep-2004 | 040182052 | 20-Sep-2005 | 6946011 | ABBR Babcock & Wilcox Power Generation Group, Inc. |
| D1 | Granted | 03-Aug-2005 | 11/196084 | 15-Dec-2005 | 050274413 | 06-Nov-2007 | 7291194 | Babcock & Wilcox Power Generation Group, Inc. |
| D2 | Granted | 03-Aug-2005 | 11/196075 | 08-Dec-2005 | 050268964 | 30-Oct-2007 | 7288128 | Babcock & Wilcox Power Generation Group, Inc. |
| 7062 | UNBURN | ED CARBON A | ND OTHER | COMBUSTIBL | ES | | | |
| C1 | Granted | 06-Oct-1997 | 08/944120 | | | 30-Jun-1998 | 5774176 | Diamond Power International, Inc. |
| C2 | Granted | 08-Sep-1997 | 08/925340 | | | 23-Nov-1999 | 5988079 | Diamond Power International, Inc. |
| 7063 | COMBUS | STION OPTIMIZ | ZATION SYS | STEM | | | | |
| | Granted | 25-Jan-1996 | 08/591012 | | | 18-Aug-1998 | 5794549 | Diamond Power International, Inc. |
| 7064 | SOOTBL | OWING OPTIM | IIZATION SY | YSTEM | | | | |
| | Granted | 09-Nov-1999 | 09/436944 | | | 04-Dec-2001 | 6325025 | Diamond Power International, Inc. |
| D | Granted | 01-Aug-2001 | 09/920697 | 10-Jan-2002 | 0002956 | 30-Jul-2002 | 6425352 | Diamond Power International, Inc. |
| 7065 | ASH HAN | NDLING SYSTE | M | | | | | |
| | Granted | 18-Oct-2000 | 09/691562 | | | 15-Jan-2002 | 6338306 | Diamond Power International, Inc. |
| 7067 | IMPACT | TYPE PARTIC | LE SEPARA | FOR MADE OF | MUTUALLY | INVERTED U | -SHAPED | ELEMENTS |
| | Granted | 29-May-2003 | 10/447688 | 16-Dec-2004 | 040250521 | 22-Mar-2005 | 6869459 | Babcock & Wilcox Power Generation Group, Inc. |
| 7068 | SOOTBL | OWER CONTR | OL BASED (| ON BOILER TH | IERMAL EFF | ICIENCY OPT | IMIZATI(| ON |
| | Granted | 18-Dec-2003 | 10/739858 | 19-Aug-2004 | 0159270 | 16-Aug-2005 | 6928937 | Diamond Power International, Inc. |
| 7070 | DEVICE | FOR REGULAT | TING AND C | LEANING AN | AIR INTAKE | | | |
| | Granted | 05-Sep-2000 | 09/623450 | | | 13-Aug-2002 | 6431125 | Diamond Power International, Inc. |
| 7071 | ANTI-VI | BRATION BARS | S FOR BOILI | ER TUBES WIT | TH PROTECT | TIVE SHIELDS | | |
| | Granted | 17-Mar-2004 | 10/802688 | 26-Jan-2006 | 060016402 | 02-Oct-2007 | 7275589 | Babcock & Wilcox Power Generation Group, Inc. |
| 7072 | DUAL PF | RESSURE RECO | OVERY BOIL | ER | | | | |
| | Granted | 20-Oct-2004 | 10/969125 | 20-Apr-2006 | 06-0081199 | 17-Jul-2007 | 7243619 | Babcock & Wilcox Power Generation Group, Inc. |
| 7073 | SPLIT RI | ING CASTING I | OR BOILER | TUBES WITH | I TUBE SHIEI | LDS | | |
| | Granted | 17-Mar-2004 | 10/802474 | 22-Sep-2005 | 050205018 | 27-Feb-2007 | 7182045 | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 90 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|-----------|---------------|------------|--------------|---------------|--------------|----------|---|
| 7075 | | | | | | | | JLVERIZED COAL-FIRED BURNERS |
| | Granted | 29-Jan-2004 | 10/767978 | 04-Aug-2005 | 05-0166867 | 01-Dec-2009 | 7624707 | The Babcock & Wilcox Company |
| 7077 | CLAD TU | BE FOR NUCLI | EAR FUEL | | | | | |
| | Granted | 08-Dec-2003 | 10/730612 | 10-Jul-2008 | 08-0165912 | 15-Jul-2008 | 7400697 | Babcock & Wilcox Nuclear Operations Group, Inc. |
| 7079 | EXTERNA | AL TUBE DEFO | RMING EXT | TRACTION DE | VICE | | | |
| | Granted | 30-Jun-2004 | 10/880796 | 05-Jan-2006 | 06-0000074 | 30-Jan-2007 | 7168143 | Babcock & Wilcox Canada Ltd. |
| 7080 | INTERNA | L TUBE EXTR | ACTION DE | VICE USING A | CYLINDRICAI | COLLAPSING | WEDGE | |
| | Granted | 07-Jun-2004 | 10/862632 | 08-Dec-2005 | 0268451 | 27-Mar-2007 | 7194800 | Babcock & Wilcox Canada Ltd. |
| 7081 | EXTERNA | AL TUBE EXTR | ACTION DE | VICE WITH A | CYLINDRICAL | COLLAPSING | WEDGE | |
| | Granted | 08-Jun-2004 | 10/863187 | 08-Dec-2005 | 050268452 | 12-Dec-2006 | 7146716 | Babcock & Wilcox Canada Ltd. |
| 7082 | TUBE EX | TRACTING DE | VICE | | | | | |
| | Granted | 22-Apr-2004 | 10/830123 | 27-Oct-2005 | 050235486 | 11-Dec-2007 | 7305756 | Babcock & Wilcox Canada Ltd. |
| 7086 | SOOTBLO | OWER NOZZLE | E ASSEMBLY | WITH NOZZ | LES HAVING D | IFFERENT GEO | OMETRIES | S |
| | Granted | 24-Mar-2004 | 10/808047 | 11-Nov-2004 | 0222324 | 18-Apr-2006 | 7028926 | Diamond Power International, Inc. |
| 7087 | ASSEMBL | Y AND METHO | OD FOR REM | MOVING A DE | VICE FROM A I | BOILER | | |
| | Granted | 15-Jun-2005 | 11/153671 | 16-Feb-2006 | 060032468 | 29-Dec-2009 | 7637002 | Diamond Power International, Inc. |
| 7088 | BAFFLE I | OR INCREASE | ED CAPTURI | E OF POPCOR | N ASH IN ECON | OMIZER HOPI | PERS | |
| | Granted | 10-Jan-2005 | 11/033085 | 14-Jul-2005 | 05-0150439 | 05-Sep-2006 | 7100521 | Babcock & Wilcox Power Generation Group, Inc. |
| 7089 | EXPLOSI | VE TUBE REM | OVAL DEVI | CE | | | | |
| | Granted | 19-Jan-2005 | 11/038842 | 20-Jul-2006 | 06-0156527 | 29-Jan-2008 | 7322090 | Babcock & Wilcox Canada Ltd. |
| 7095 | LOW LOS | S POPPET VAI | LVE | | | | | |
| | Published | 30-Aug-2005 | 11/216591 | 01-Mar-2007 | 07-0045584 | | | Diamond Power International, Inc. |
| 7097 | IN-FURNA | ACE REDUCTION | ON OF NITR | OGEN OXIDE | BY MIXED FUE | LS INVOLVING | G A BIOM | ASS DERIVATIVE |
| | Published | 27-Jun-2006 | 11/426906 | 15-Feb-2007 | 07-0034126 | | | The Babcock & Wilcox Company Ole Miss |
| 7098 | SNCR DIS | TRIBUTION G | RID | | | | | |
| | Published | 12-Apr-2007 | 11/577171 | 05-Mar-2009 | 20090060806 | | | The Babcock & Wilcox Company |

Page 91 of 34

* McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | |
|--------------------|---|---------------|------------|-------------|----------------|--------------|---------|---|--|--|
| 7100 | BURNER W | VITH CENTER F | TRED AIR J | ET | | | | | | |
| | Granted | 30-Jun-2005 | 11/171027 | 04-Jan-2007 | 070003889 | 07-Oct-2008 | 7430970 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 7101 | SOOTBLO | WER FRAME AS | SEMBLY | | | | | | | |
| | Published | 20-Dec-2005 | 10/561569 | 05-Apr-2007 | 07-0079001 | | | Diamond Power International, Inc. | | |
| 7105 | OXY-FUEL | REBURN: A MI | ETHOD FOR | NOx REDUCT | TION BY FUEL I | REBURNING WI | TH OXYG | EN | | |
| | Granted | 18-Jul-2005 | 11/183512 | 16-Nov-2006 | 06-0257800 | 17-Feb-2009 | 7491055 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 7106 | CARRIER A | AIR HEATING S | YSTEM FOR | R SCR | | | | | | |
| | Granted | 13-Apr-2005 | 11/105053 | 19-Oct-2006 | 06-0234173 | 15-Sep-2009 | 7588440 | The Babcock & Wilcox Company | | |
| 7107 | WETTED P | ARTICLE AND | DROPLET I | MPINGEMENT | Γ | | | | | |
| | Granted | 23-Sep-2004 | 10/948478 | 23-Mar-2006 | 06-0060088 | 16-Oct-2007 | 7282087 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 7110 | PORT ROD | DER WITH VEI | OCITY DAN | MPER | | | | | | |
| | Granted | 26-May-2005 | 11/138574 | 01-Dec-2005 | 050263047 | 01-Jul-2008 | 7392751 | Diamond Power International, Inc. | | |
| 7111 | RETRACTA | ABLE LIQUOR (| GUN FOR A | RECOVERY F | URNACE | | | | | |
| | Granted | 26-May-2005 | 11/138044 | 01-Dec-2005 | 050263108 | 13-Jan-2009 | 7475645 | Diamond Power International, Inc. | | |
| 7112 | DETONATI | ION / DEFLAGR | ATION SOO | TBLOWER | | | | | | |
| | Granted | 14-Jun-2005 | 11/152577 | 12-Jan-2006 | 060005786 | 22-Apr-2008 | 7360508 | Diamond Power International, Inc. | | |
| 7113 | HOOD/TOV | WER INTERFAC | EE | | | | | | | |
| | Granted | 10-Feb-2005 | 11/055760 | 24-Aug-2006 | 06-0185320 | 04-May-2010 | 7707782 | The Babcock & Wilcox Company | | |
| 7115 | SYSTEM U | SING OVER FIR | E ZONE SE | NSORS AND D | ATA ANALYSIS | 1 | | | | |
| | Published | 25-Feb-2008 | 12/036639 | 27-Aug-2009 | 20090214993 | | | Babcock & Wilcox Power Generation Group, Inc. | | |
| 7117 | BOILER W | ALL BOX COOL | LING SYSTE | M | | | | | | |
| | Granted | 24-Jun-2004 | 10/876132 | | | 09-Aug-2005 | 6925969 | Diamond Power International, Inc. | | |
| 7118 | TUNNELEI | O MULTI-BLADI | E SWIRLER | FOR LIQUID | FUEL ATOMIZA | ATION | | | | |
| | Granted | 08-Jun-2005 | 11/100731 | 14-Dec-2006 | 06-0281036 | 06-May-2008 | 7367798 | Babcock & Wilcox Power Generation Group, Inc. | | |
| 7119 | CO-EXTRU | DED GENERAT | ING BANK | SWAGED TUB | ING | | | | | |
| | Granted | 31-Mar-2005 | 11/095915 | 05-Oct-2006 | 06-0218792 | 02-Feb-2010 | 7653999 | Babcock & Wilcox Canada Ltd. | | |
| * McDer | * McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010 | | | | | | | | | |

Page 92 of 34

| The Babcock & Wilcox Company US Active Patents and Applications | The Babcock & | Wilcox | Company | US Active | Patents | and A | pplications |
|---|---------------|--------|---------|-----------|----------------|-------|-------------|
|---|---------------|--------|---------|-----------|----------------|-------|-------------|

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|-----------|-----------------|------------|--------------|---------------|-----------------|----------|---|
| 7122 | HEAT EXC | HANGER FRAM | IEWORK | | | | | |
| | Granted | 17-Jul-2006 | 11/487624 | 31-Jan-2008 | 08-0022949 | 16-Dec-2008 | 7464671 | Babcock & Wilcox Power Generation Group, Inc. |
| 7124 | MATRIX M | IEANS FOR RED | OUCING COM | MBUSTION VO | DLUME | | | |
| | Granted | 11-Jul-2007 | 11/775919 | 15-Jan-2009 | 20090017402 | 24-Feb-2009 | 7493876 | The Babcock & Wilcox Company |
| 7126 | STEAM/WA | ATER CONICAL | CYCLONE | SEPARATOR | | | | |
| | Granted | 05-Jul-2007 | 11/753335 | 08-Jan-2009 | 09-0010721 | 29-Dec-2009 | 7637699 | The Babcock & Wilcox Company |
| 7127 | NATURAL | CIRCULATION | INDUSTRIA | L BOILER FO | R STEAM ASSIS | STED GRAVITY | DRAINAG | E (SAGD) PROCESS |
| | Granted | 01-May-2007 | 11/742870 | 22-Nov-2007 | 07-0266962 | 19-May-2009 | 7533632 | Babcock & Wilcox Canada Ltd. |
| 7128 | LOX NOx C | CYCLONE FURN | ACE STEAN | I GENERATO | R | | | |
| | Published | 30-May-2007 | 11/720506 | 05-Jun-2008 | 080127869 | | | The Babcock & Wilcox Company |
| 7136 | ENHANCE | MENT OF CONV | ENTIONAL | SCR AND SNO | CR PROCESSES | WITH AMMON | A DESTRI | UCTION CATALYST |
| | Published | 07-Apr-2008 | 12/098623 | 08-Oct-2009 | 20090252665 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7143 | NUBBED U | -BEND TUBE SU | PPORT | | | | | |
| | Published | 18-Jul-2007 | 11/779653 | 22-Jan-2009 | 20090020273 | | | Babcock & Wilcox Canada Ltd. |
| 7144 | STEAM GE | NERATOR TO | CONTAIN A | ND COOL SYN | THESIS GAS | | | |
| | Published | 27-Jul-2005 | 11/191183 | 17-Jan-2008 | 08-0011247 | | | The Babcock & Wilcox Company |
| 7146 | LARGE DIA | AMETER MID-Z | ONE AIR SE | PARATION C | ONE FOR EXPA | ANDING IRZ | | |
| | Published | 01-Jun-2006 | 11/444779 | 06-Dec-2007 | 07-0281265 | | | The Babcock & Wilcox Company |
| 7148 | CIRCULAT | TION SYSTEM F | OR SLIDING | F PRESSURE S | TEAM GENERA | ATOR | | |
| | Granted | 07-Jun-2006 | 11/448648 | 13-Dec-2007 | 07-0283906 | 15-Sep-2009 | 7587996 | The Babcock & Wilcox Company |
| 7149 | LINK TYPE | E SEISMIC TIE I | FOR BOILER | RS | | | | |
| | Published | 03-May-2007 | 11/743722 | 06-Nov-2008 | 08-0271686 | | | The Babcock & Wilcox Company |
| 7150 | ASSEMBLY | Y FOR ASH SEPA | ARATION FI | ROM FLUE GA | AS . | | | |
| | Granted | 27-Feb-2007 | 11/711509 | 28-Aug-2008 | 08-0202077 | 10-Nov-2009 | 7615086 | The Babcock & Wilcox Company |
| 7151 | A BOILER | PRODUCING ST | TEAM FROM | I FLUE GASES | UNDER OPTIM | IISED CONDITION | ONS | |
| | Pending | 27-Mar-2008 | 12/088256 | | | | | Babcock & Wilcox Volund A/S |
| | | | | | | | | |

Page 93 of 34

 ${\rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

| The Babcock & Wilcox Company US Active Patents and Application | The | Babcock & | Wilcox | Company | US Active | Patents and | Applications |
|--|-----|-----------|--------|---------|-----------|-------------|--------------|
|--|-----|-----------|--------|---------|-----------|-------------|--------------|

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|-----------|---------------|-------------|---------------|---------------|--------------|-----------|---|
| 7152 | | | | | | | | MPROVED SLAG QUALITY |
| | Pending | 25-Mar-2008 | 12/067968 | | | | | Babcock & Wilcox Volund A/S |
| 7155 | RADIANT | SYNGAS COOLI | ER | | | | | |
| | Granted | 26-Oct-2006 | 11/588045 | 31-May-2007 | 070119577 | 15-Sep-2009 | 7587995 | The Babcock & Wilcox Company |
| 7159 | HYBRID W | ET ELECTROS | TATIC PREC | CIPITATOR (HV | WESP) | | | |
| | Granted | 27-Mar-2008 | 12/056417 | 01-Oct-2009 | 20090241781 | 15-Dec-2009 | 7632341 | Babcock & Wilcox Power Generation Group, Inc. |
| 7160 | APPARATI | US FOR CLEANI | NG A SMEL | T SPOUT OF A | COMBUSTION | DEVICE (BLAD | ERUNNEI | R) |
| | Published | 24-May-2006 | 11/439674 | 29-Nov-2007 | 07-0272130 | ` | | Diamond Power International, Inc. |
| 7164 | MOISTURI | E REMOVAL FR | OM FLUE G | AS IN WFGD IN | N THE OXY-COM | MBUSTION TEC | CHNOLOG | SY |
| | Granted | 11-Apr-2007 | 11/733942 | 18-Oct-2007 | 0243119 | 08-Sep-2009 | 7585476 | The Babcock & Wilcox Company |
| 7166 | INTEGRAT | TED FLUIDIZED | BED ASH C | OOLER | | | | |
| | Granted | 19-Apr-2006 | 11/406765 | 13-Dec-2007 | 07-0283902 | 16-Dec-2008 | 7464669 | Babcock & Wilcox Power Generation Group, Inc. |
| 7167 | STRESS RE | EDUCTION FOR | PULVERIZI | ER MAIN SHAF | T VIA THRUST | BEARING MOI | DIFICATIO | ON |
| | Published | 13-Mar-2007 | 11/717345 | 20-Sep-2007 | 070215730 | | | The Babcock & Wilcox Company |
| 7170 | COMBUST | ION SYSTEM A | ND PROCES | S | | | | |
| | Published | 15-Nov-2007 | 11/940575 | 02-Apr-2009 | 0084294 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7171 | END SUPPO | ORT CONFIGUR | RATION FOR | R STEAM TUBE | S OF A SUPERH | EATER OR RE | HEATER | |
| | Granted | 26-Apr-2007 | 11/740340 | 30-Oct-2008 | 080264358 | 14-Jul-2009 | 7559294 | The Babcock & Wilcox Company |
| 7174 | SELF SUPP | PORTING REINE | ORCED HE | ADER | | | | |
| | Published | 15-Mar-2007 | 11/686437 | 27-Sep-2007 | 07-0221279 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7176 | CLOSED C | YCLE MHD-FAI | RADAY GEN | ERATION OF I | ELECTRIC POW | ER USING STE | AM AS TE | IE GASEOUS MEDIUM |
| | Published | 19-Jul-2007 | 11/780130 | 22-Jan-2009 | 20090021010 | | | The Babcock & Wilcox Company |
| 7178 | METHOD I | FOR CONTROLI | LING AIR DI | STRIBUTION I | N A CYCLONE | FURNACE | | |
| | Granted | 04-Oct-2006 | 11/538578 | 28-Feb-2008 | 20080050684 | 03-Feb-2009 | 7484955 | Diamond Power International, Inc. |
| 7180 | MULTIPLE | E PASS ECONOM | MIZER AND | METHOD FOR | SCR TEMPERA | TURE CONTRO | DL | |
| | Granted | 09-May-2006 | 11/430761 | 15-Nov-2007 | 07-0261646 | 25-Aug-2009 | 7578265 | The Babcock & Wilcox Company |
| C | Granted | 03-Oct-2006 | 11/542413 | 15-Nov-2007 | 07-0261647 | 29-Dec-2009 | 7637233 | The Babcock & Wilcox Company |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 94 of 34

| The Babcock & Wilcox Company US Active Patents and Applications | The Babcock & | Wilcox | Company | US Active | Patents and | Applications |
|---|---------------|--------|---------|-----------|-------------|---------------------|
|---|---------------|--------|---------|-----------|-------------|---------------------|

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|-----------|------------------------------|------------|--------------|---------------|-------------|---------|--|
| 7181 | POLYGON | TUMBLE ASSE | MBLER | | | | | |
| | Published | 17-May-2007 | 11/749967 | 05-Jun-2008 | 080128580 | | | Babcock & Wilcox Canada Ltd. Babcock & Wilcox Power Generation Group, Inc. |
| 7183 | INDUCTIO | N HEAT-AND-P | ULL TUBE S | TUB REMOVA | L TECHNIQUE | | | |
| | Granted | 13-Jun-2007 | 11/762104 | 03-Jan-2008 | 080000066 | 28-Jul-2009 | 7565727 | Babcock & Wilcox Canada Ltd. |
| 7189 | | OF CONTROLLI AKA GASIFICA | | | GENERATING EI | LECTRIC POV | VER ANA | D APPARATUS FOR USE OF SAID |
| | Pending | 31-Aug-2009 | 12/306573 | | | | | Babcock & Wilcox Volund A/S |
| 7190 | STEPPED I | OOWN GAS MIX | ING DEVIC | E | | | | |
| | Pending | 03-Feb-2010 | 12/699407 | | | | | The Babcock & Wilcox Company |
| 7191 | COMPACT | RADIAL PLATI | EN ARRANG | EMENT FOR F | RADIANT SYNG | AS COOLER | | |
| | Published | 15-Aug-2007 | 11/839285 | 21-Feb-2008 | 080041572 | | | The Babcock & Wilcox Company |
| 7193 | SYNGAS C | OOLER CLEAN | NG DEVICE | Ē | | | | |
| | Published | 14-Aug-2007 | 11/838580 | 21-Feb-2008 | 080041322 | | | The Babcock & Wilcox Company |
| 7194 | SEALING A | ARRANGEMENT | WITH A SE | EGMENTED SE | AL AND PRESSU | JRE RELIEF | | |
| | Published | 15-Aug-2007 | 11/839357 | 21-Feb-2008 | 080042373 | | | The Babcock & Wilcox Company |
| 7195 | CATALYST | T PARTICULATI | E DIVERTE | ₹ | | | | |
| | Published | 07-Sep-2007 | 11/851443 | 12-Mar-2009 | 20090065414 | | | The Babcock & Wilcox Company |
| 7197 | ULTRASON | NIC INSPECTION | N METHOD | | | | | |
| | Published | 29-Nov-2006 | 11/564346 | 29-May-2008 | 080121040 | | | BWX Technologies, Inc. |
| 7200 | EXTENDE | O WATER LEVE | L RANGE S | ΓΕΑΜ / WATEI | R CONICAL CYC | CLONE SEPAR | ATOR | |
| | Published | 19-Sep-2007 | 11/857898 | 20-Mar-2008 | 08-0069646 | | | The Babcock & Wilcox Company |
| 7202 | BOTTOM A | ASH INJECTION | FOR ENHA | NCING SPRAY | DRYER ABSOR | BER PERFOR | MANCE | |
| | Published | 13-Sep-2007 | 11/854781 | 19-Mar-2009 | 20090074629 | | | The Babcock & Wilcox Company |
| 7206 | RAPPER A | LIGNMENT PLU | G | | | | | |
| | Published | 27-Jun-2007 | 11/768941 | 01-Jan-2009 | 090000770 | | | The Babcock & Wilcox Company |
| 7207 | SPRAY DR | YER ABSORBEF | R AND RELA | TED PROCESS | SES | | | |
| | Published | 13-Sep-2007 | 11/854795 | 19-Mar-2009 | 20090074642 | | | The Babcock & Wilcox Company |
| | | | | | | | | |

 $^{{}^{\}star}\;\;McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,2010$

Page 95 of 34

| | | | The Bab | cock & Wilcox | Company US Acti | ive Patents and A | Application | 18 |
|--------------------|---------------------|---------------|------------|---------------|-----------------|-------------------|-------------|--|
| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
| 7212 | CONTOUR | ED FLAT STUD | AND STUD | ARRANGEME | NT FOR CYCLO | NE SLAG TAPS | S | |
| | Published | 18-Jul-2008 | 12/175524 | 21-Jan-2010 | 20100012005 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7214 | IN-SITU RE | EGENERATION | OF A CATAI | LYST MASKE | D BY CALCIUM | SULFATE | | |
| | Published | 26-Mar-2008 | 12/055439 | 01-Oct-2009 | 20090247392 | | | The Babcock & Wilcox Company |
| 7215 | INDUCTIO | N HEATED, HO | r wire we | LDING | | | | |
| C | Published | 16-Nov-2009 | 12/619032 | 11-Mar-2010 | 20100059493 | | | Babcock & Wilcox Technical Services Group, Inc. |
| 7219 | LOW-TEM | PERATURE, MO | VING BED | CATALYTIC F | REACTOR FOR O | CONTROL OF 1 | NOx EMIS | SIONS FROM COMBUSTION |
| | Published | 18-Jul-2008 | 12/175540 | 05-Feb-2009 | 20090035200 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7222 | COMBINAT | TORIAL HETRO | GENEOUS - | HOMOGENE | OUS REACTOR | | | |
| | Published | 03-Mar-2009 | 12/397070 | 10-Sep-2009 | 2009-0225923 | | | Babcock & Wilcox Technical Services Group, Inc. |
| 7223 | SEGREGAT COMBUST | | RCED OXID | ATION WET I | FLUE GAS DESU | LFURIZATION | N FOR OX | YGEN- FIRED FOSSIL FUEL |
| | Pending | 12-Oct-2009 | 12/577456 | | | | | The Babcock & Wilcox Company |
| 7224 | SYSTEM A | ND METHOD FO | OR MINIMIZ | ZING NITROG | EN OXIDE (NOx | EMISSIONS I | N CYCLO | NE COMBUSTORS |
| | Published | 29-May-2008 | 12/129052 | 08-Jan-2009 | 09-0007827 | | | American Air Liquide, Inc. Babcock & Wilcox Power Generation Group, Inc. |
| 7227 | METHOD A | AND APPARATU | S FOR PREI | PARING PULV | ERIZED COAL | USED TO PROI | DUCE SYN | NTHESIS GAS |
| | Published | 01-Apr-2008 | 12/060459 | 09-Oct-2008 | 080245076 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7228 | STEAM GE | NERATOR ARR | ANGEMENT | Γ | | | | |
| | Published | 11-Apr-2008 | 12/101513 | 16-Oct-2008 | 080251037 | | | The Babcock & Wilcox Company |
| 7230 | COIN RETI | RIEVAL FROM I | REFUSE | | | | | |
| | Published | 06-Aug-2007 | 11/834247 | 12-Feb-2009 | 20090038996 | | | The Babcock & Wilcox Company |
| 7231 | TRI-SECTO | OR REGENERAT | TIVE OXIDA | NT PREHEAT | ER FOR OXY-FI | RED PULVERI | ZED COA | L COMBUSTION |
| | Published | 08-Jul-2008 | 12/169168 | 15-Jan-2009 | 20090013941 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7232 | ECONOMIZ | ZER ARRANGE | MENT FOR | STEAM GENE | RATOR | | | |
| | Published | 17-May-2007 | 11/750271 | 20-Nov-2008 | 08-0282997 | | | The Babcock & Wilcox Company |
| 7233 | METHOD A | AND APPARATU | S FOR THE | PRODUCTION | N AND EXTRAC | TION OF MOLY | YBDENUN | 1-99 |
| | Granted | 23-Feb-1998 | 09/028183 | | | 08-Jun-1999 | 5910971 | BWXT Services, Inc. |
| * M.D | 44 T b 1 | T | :4 4- D | -11- 0 XX/21 | Tkl I | - 4 | | |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 96 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|--------------|------------------|---------------|----------------|-------------------|-------------------|---------|---|
| 7234 | CARBON C | OXIDE AND/OR | SULFUR OX | IDE CAPTURE | E IN A LIQUID E | CNVIRONMEN | T | |
| | Pending | 13-Oct-2009 | 12/577806 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7235 | RETRACT | ABLE ARTICUL | ATING ROB | OTIC SOOTBI | LOWER | | | |
| | Pending | 26-Feb-2009 | 12/393441 | | | | | Diamond Power International, Inc. |
| 7237 | METHOD (| OF FORMING, I | NSERTING, | AND PERMAN | ENTLY BONDI | NG RIBS IN BO | OILER T | UBES |
| | Published | 06-Jun-2008 | 12/134295 | 10-Dec-2009 | 20090301159 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7238 | BLACK PL | ANT STEAM FU | RNACE INJ | ECTION | | | | |
| | Published | 17-Jul-2008 | 12/175102 | 29-Jan-2009 | 20090025658 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7245 | SOOTBLO | WER ISOLATIO | N WALL BO | X | | | | |
| | Pending | 17-Sep-2009 | 12/561488 | | | | | Babcock & Wilcox Power Generation Group, Inc. Diamond Power International, Inc. |
| 7247 | HEAT EXC | CHANGER | | | | | | |
| | Published | 14-Jan-2009 | 12/353716 | 16-Jul-2009 | 20090178779 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7249 | ENHANCE | D STEAM CYCL | E UTILIZIN | G A DUAL PR | ESSURE RECO | VERY BOILER | WITH F | REHEAT |
| | Published | 26-Mar-2008 | 12/055832 | 01-Oct-2009 | 20090241860 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7251 | TUBE SUP | PORT SYSTEM I | FOR NUCLE | AR STEAM GI | ENERATORS | | | |
| | Published | 25-Jul-2008 | 12/180478 | 28-Jan-2010 | 20100018688 | | | Babcock & Wilcox Canada Ltd. |
| 7252 | TUBE SUP | PORT SYSTEM I | FOR NUCLE | AR STEAM GI | ENERATORS | | | |
| | Published | 25-Jul-2008 | 12/180491 | 28-Jan-2010 | 20100018689 | | | Babcock & Wilcox Canada Ltd. |
| 7255 | TUBE STU | B REMOVAL AP | PARATUS A | AND METHOD | | | | |
| | Published | 15-Jan-2009 | 12/354468 | 17-Sep-2009 | 20090229128 | | | The Babcock & Wilcox Company |
| 7256 | OXY-COM | BUSTION COAL | FIRED BOI | LER AND ME | THOD OF TRAN | SITIONED BE | TWEEN | AIR AND OXYGEN FIRING |
| | Published | 13-Apr-2009 | 12/422685 | 15-Oct-2009 | 20090255450 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7257 | ERECTION | METHOD FOR | SOLAR REC | CEIVER & SUI | PPORT TOWER | | | |
| | Published | 04-May-2009 | 12/435062 | 12-Nov-2009 | 20090276993 | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7261 | WET SCRU | JBBER TRAY | | | | | | |
| | Published | 13-Jun-2009 | 12/484187 | 17-Dec-2009 | 20090309245 | | | Babcock & Wilcox Power Generation Group, Inc. |
| * McDer | mott Technol | ogy Inc. changed | its name to B | abcock & Wilco | ox Technology, Ir | nc April 20, 2010 | 0 | |

Page 97 of 34

| CASE NO | | | | | | | | |
|---------|-----------|--------------|-------------|------------|----------|-------------|---------|---|
| SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
| 7265 | ACCESS I | OOOR WITH IN | ISIDE LATC | H RELEASI | E | | | |
| | Pending | 22-Jan-2009 | 12/357743 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7266 | SHOP-ASS | SEMBLED SOL | AR RECEIV | ER HEAT E | EXCHANG | GER | | |
| | Pending | 23-Oct-2009 | 12/605241 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7268 | BLADED | BULLET COAI | DIFFUSER | WITH COA | L LINE E | BALANCING | DEVICE | |
| | Pending | 09-Nov-2009 | 12/614480 | | | | | |
| 7281 | IN-BED SO | OLIDS CONTR | OL VALVE | | | | | |
| | Pending | 30-Sep-2009 | 12/570823 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7283 | PRIMARY | OXIDANT FE | ED TO OXY | -FIRED CIR | CULATI | NG FLUIDIZ | ED BED | (CFB) |
| | Pending | 30-Sep-2009 | 12/571187 | | | | | |
| 7284 | CIRCULA | TING FLUIDIZ | ZED BED (CI | FB) WITH I | N-FURNA | CE SECOND | ARY AIF | RNOZZLES |
| | Pending | 30-Sep-2009 | 12/571279 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7287 | SYSTEM A | AND METHOD | FOR SOFTE | ENING WAT | TER FOR | USE IN A SC | RUBBER | t . |
| | Pending | 19-Jan-2010 | 12/689313 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7288 | SYSTEM A | AND METHOD | FOR PROTI | ECTION OF | SCR CA | TALYST ANI | CONTR | ROL OF MULTIPLE EMISSIONS |
| | Pending | 21-Jan-2010 | 12/691527 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| P | Pending | 22-Apr-2009 | 61/171619 | | | | | |
| 7289 | INTEGRA | TED SPLIT ST | REAM WAT | ERCOIL AI | RHEATE | R & ECONO | MIZER (| IWE) |
| | Pending | 19-Oct-2009 | 12/581637 | | | | | |
| P | Pending | 10-Mar-2009 | 61/158774 | | | | | |
| 7290 | OXY-FUE | L COMBUSTIC | ON OXIDAN | Γ HEATER 1 | INTERNA | AL ARRANGI | EMENT | |
| | Pending | 25-Feb-2010 | 12/712268 | | | | | |
| 7292 | TUBE SUI | PPORT STRUC | TURE | | | | | |
| | Pending | 29-Apr-2009 | 12/432644 | | | | | Babcock & Wilcox Canada Ltd. |
| 7293 | FEEDWA | TER DEBRIS T | RAP FOR A | NUCLEAR | ONCE TH | HROUGH STI | EAM GE | NERATOR |
| | Pending | 29-Apr-2009 | 12/432653 | | | | | Babcock & Wilcox Canada Ltd. |
| | | | | | | | | |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Thursday, April 29, 2010 Prepared by AMSaus

Page 98 of 34

| | | | The B | abcock & W | ilcox Com | ipany US Acti | ive Patent | s and Applications | | |
|--------------------|---|--------------|-------------|------------|-----------|---------------|------------|---|--|--|
| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER | | |
| 7294 | BIOMASS | CENTER AIR | EJET(R) BUI | RNER | | | | | | |
| | Pending | 26-Apr-2010 | 12/766991 | | | | | | | |
| P | Pending | 29-Apr-2009 | 61/173659 | | | | | | | |
| 7295 | SOOTBLO | WER WITH P | ROGRESSIV | E CLEANII | NG ARC | | | | | |
| | Pending | 18-Dec-2009 | 12/642210 | | | | | Diamond Power International, Inc. | | |
| 7296 | INTEGRA | L PRESSURIZ | ED WATER | NUCLEAR | REACTO | R | | | | |
| P | Pending | 10-Jun-2009 | 61/185878 | | | | | | | |
| 7297 | CONTROL ROD DRIVE MECHANISM FOR NUCLEAR REACTOR | | | | | | | | | |
| | Pending | 12-Mar-2010 | 12/722662 | | | | | | | |
| P | Pending | 10-Jun-2009 | 61/185887 | | | | | | | |
| 7298 | NUCLEAR | REACTOR | | | | | | | | |
| P | Pending | 10-Jun-2009 | 61/185897 | | | | | | | |
| 7299 | SYSTEM I | FOR CONTRO | LLING COA | L FLOW IN | A COAL | PULVERIZE | ER | | | |
| P | Pending | 22-Jun-2009 | 61/219097 | | | | | | | |
| 7302 | SYSTEM A | AND METHOD | FOR PROTI | ECTION OF | SCR CA | TALYST | | | | |
| | Pending | 25-Aug-2009 | 12/547040 | | | | | | | |
| 7303 | REDOX M | ETHOD FOR | CAPTURE O | F TOTAL G | ASEOUS | MERCURY | BY WET | FGD | | |
| P | Pending | 16-Jul-2009 | 61/226132 | | | | | | | |
| 7305 | CO2 SCRU | JBBING SYSTI | EM | | | | | | | |
| P | Pending | 15-Oct-2009 | 61/252102 | | | | | | | |
| 7308 | RADIOAC | TIVE DEBRIS | TRAP | | | | | | | |
| | Pending | 30-Oct-2009 | 12/609026 | | | | | | | |
| 7313 | CHEMICA | AL SOLVENTS | FOR THE R | EMOVAL C | F CARBO | ON DIOXIDE | FROM F | LUE GASES | | |
| P | Pending | 26-Mar-2010 | 61/318044 | | | | | Babcock & Wilcox Power Generation Group, Inc. | | |
| 7314 | CONTRO | L ROD DRIVE | MECHANIS | M FOR NUC | CLEAR RI | EACTOR | | | | |
| | Pending | 12-Mar-2010 | 12/722696 | | | | | Babcock & Wilcox Nuclear Operations Group, Inc. | | |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Thursday, April 29, 2010 Prepared by AMSaus

Page 99 of 34

| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
|--------------------|-----------|--------------|------------|------------|----------|-------------|-----------|---|
| 7315 | | | | | | | | L ISOTOPES PRODUCTION SYSTEM |
| P | Pending | 19-Feb-2010 | 61/306363 | | | | | Babcock & Wilcox Technical Services Group, Inc. |
| 7319 | Ü | VATER TREAT | MENT FOR | HIGH TEM | PERATU | RE STEAM G | ENERATO | ** |
| | Pending | 17-Mar-2010 | 12/725777 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7328 | MULTIPL | E RAPPER CO | NTROL FOR | ELECTRO | STATIC I | PRECIPITATO | OR | 1, |
| | Granted | 27-Jul-1990 | 07/558819 | | | 22-Dec-1992 | 5173867 | Babcock & Wilcox Power Generation Group, Inc. |
| 7331 | | CAL CONTROL | L SYSTEM F | OR ELECT | | | ATOR | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
| | Granted | 27-Jul-1990 | 07/558827 | | | 26-Nov-1991 | 5068811 | Babcock & Wilcox Power Generation Group, Inc. |
| 7332 | | | | Γ LIMITING | | | | FOR ELECTROSTATIC PRECIPITATOR |
| | Granted | 22-Nov-1994 | 08/338631 | | | 06-Jan-1998 | 5705923 | Babcock & Wilcox Power Generation Group, Inc. |
| 7333 | | OR PLATE FO | | OSTATIC PE | | | | |
| | Granted | 28-Sep-1995 | 08/530300 | | | 09-Sep-1997 | 5665147 | Babcock & Wilcox Power Generation Group, Inc. |
| 7334 | METHOD | | TUS FOR CO | ONTROLLIN | | | GAS CON | IDITIONING SYSTEM |
| | Granted | 10-Oct-1997 | 08/948912 | | | 14-Sep-1999 | 5950441 | Babcock & Wilcox Power Generation Group, Inc. |
| D | Granted | 17-Sep-1998 | 09/156789 | | | 06-Apr-1999 | 5890369 | Babcock & Wilcox Power Generation Group, Inc. |
| 7335 | METHOD | AND SYSTEM | FOR IMPRO | OVED RAPP | | • | | 1, |
| | Granted | 06-Jul-2001 | 09/900082 | | | 01-Apr-2003 | 6540812 | Babcock & Wilcox Power Generation Group, Inc. |
| 7336 | AN APPAR | RATUS AND MI | ETHOD FOR | R FILTERIN | | • | ELECTRO | OSTATIC PRECIPATOR |
| | Granted | 19-Mar-2002 | 10/063099 | | | 26-Aug-2003 | 6611440 | Babcock & Wilcox Power Generation Group, Inc. |
| 7338 | USE OF PA | ARTIAL DISCH | IARGE ANA | LYSIS TEC | | C | | MONITORING OF ESPs POWER SUPPLY |
| | Granted | 29-Dec-2005 | 11/321832 | | | 18-Nov-2008 | 7452403 | Babcock & Wilcox Power Generation Group, Inc. |
| 7339 | AN ELECT | ROSTATIC PI | RECIPITATO | OR DESIGN | USING SI | PARK CURRI | ENT LIMIT | TING RESISTORS |
| | Granted | 27-Feb-2007 | 11/679513 | | | 27-Apr-2010 | 7704302 | Babcock & Wilcox Power Generation Group, Inc. |
| 7341 | RIGID DIS | CHARGE ELE | CTRODDE I | ENHANCEM | | · · | | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |
| - | Pending | 04-Jan-2006 | 11/326306 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| C | Pending | 19-Dec-2008 | 12/339332 | | | | | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 100 of 34

| | | | | | • | • | | •• |
|--------------------|--------------------|---------------|-------------|------------|-----------|-------------|-----------|--|
| CASE NO SubCase | STATUS | FILE DATE | APP NO | PUB DATE | PUB NO | ISSUE DATE | PAT NO | OWNER |
| 7343 | METHOD RADIATIO | | TO LOCATE | SPARK IN | ELECTRO | OSTATIC PRE | CIPITATO | OR USING ELECTRO- MAGNETIC FIELD WAVE |
| | Pending | 12-Nov-2008 | 12/269104 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7348 | DEVICE T | O ACHIEVE NE | EAR DC WAV | EFORM IN | kV RANG | GE FROM SIN | GLE PHA | SE INPUT |
| | Pending | 20-Aug-2009 | 12/544608 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7353 | METHOD | FOR ACCURAT | TELY DETER | MINING T | HE LIFT I | HEIGHT ON A | AN ESP RA | PPER TO OBTAIN RAPPING INTENSITY |
| | Pending | 03-Sep-2008 | 12/203713 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7355 | ESP COLI | LECTING PLATI | E CLEANING | G USING TU | NABLE F | REQUENCY | | |
| | Pending | 17-Dec-2009 | 12/640253 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7360 | Hg CALIB | RATION/SPAN | GAS HUMID | IFICATION | USING V | APORIZER T | ECHNOL | OGY |
| | Pending | 17-May-2007 | 11/803617 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7361 | PNEUMA' GAS | FIC FLOW DELI | IVERY OF W | ATER TO I | HUMIDIF | ICATION PER | RMEATIO | N TUBE OR VAPORIZER FOR ELEMENTATL Hg |
| | Pending | 29-Jun-2007 | 11/824089 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7365 | OXYGEN | MONITORING S | SYSTEM DES | SIGN | | | | |
| | Pending | 08-Aug-2008 | 12/188672 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7367 | METHOD | OF USING MID | -IR LASER M | 1EASURE A | AND MON | ITOR EXHAU | ST POLL | UTANT |
| | Pending | 09-Dec-2009 | 12/633862 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7368 | FIELDS O | | FOR PARAM | IETER INP | UT AND T | | | ATABASE BY DYNAMICALLY CREATING INPUT QL COMMAND PROGRAMMICATICALLY BASED |
| | Pending | 25-Jun-2009 | 12/491324 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7371 | GYRO-ST | ABILIZER EMIS | SSIONS MON | ITORING I | PROBE | | | |
| | Pending | 20-Jan-2010 | 12/690744 | | | | | Babcock & Wilcox Power Generation Group, Inc. |
| 7372 | CLEANIN | G SYSTEM AND | METHOD F | OR CONTI | NUOUS E | MISSIONS M | ONITORIN | NG EQUIPMENT |
| | Granted | 02-Oct-2006 | 11/541970 | | | 21-Jul-2009 | 7562556 | Babcock & Wilcox Power Generation Group, Inc. |
| 7373 | SYSTEM, | METHODS AND | COMPUTE | R PROGRA | M FOR CO | ONTINUOUS | EMISSION | S MONITORING ON- LINE (NetDAHS) |
| | Granted | 12-May-2004 | 10/709543 | | | 20-Feb-2007 | 7181320 | Babcock & Wilcox Power Generation Group, Inc. |
| | | | | | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Thursday, April 29, 2010 Prepared by AMSaus

Page 101 of 34

| | | | The Babcock & Wilco | ox Company US Act | ive Trademark | ks and Domain Nai | mes |
|---------------|---------|--------------|---------------------|-------------------|---------------|-------------------|---|
| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
| ABSORPTION | PLUS (| Hg) | | | | | |
| TM-750 | | Published | 77/371049 | 14-Jan-2008 | | | Babcock & Wilcox Power Generation Group, Inc. |
| TM-750 | 1 | Filed | 77/371146 | 14-Jan-2008 | | | Babcock & Wilcox Power Generation Group, Inc. |
| AireJet | | | | | | | |
| TM-747 | | Registered | 78/891260 | 24-May-2006 | 3535538 | 18-Nov-2008 | Babcock & Wilcox Power Generation Group, Inc. |
| ALLEN-SHERI | MAN-H | OFF (Word Ma | ark) | | | | |
| TM-485 | | Registered | 75/379906 | 27-Oct-1997 | 2218648 | 19-Jan-1999 | Diamond Power International, Inc. |
| A-S-H & DESIG | GN | | | | | | |
| TM-490 | | Registered | 71/226183 | 21-Jan-1926 | 0221415 | 30-Nov-1926 | Diamond Power International, Inc. |
| a-s-h.biz | | | | | | | |
| DN-047 | | Registered | DOMAIN NAME | 16-Mar-1996 | | 08-Nov-2001 | Diamond Power International, Inc. |
| n-s-h.com | | | | | | | |
| DN-001 | | Registered | DOMAIN NAME | 27-Jun-1996 | | 27-Jun-1996 | Diamond Power International, Inc. |
| -s-h.net | | | | | | | |
| DN-031 | | Registered | DOMAIN NAME | 04-Dec-2000 | | 04-Dec-2000 | Diamond Power International, Inc. |
| n-s-h.org | | | | | | | |
| DN-032 | | Registered | DOMAIN NAME | 04-Dec-2000 | | 04-Dec-2000 | Diamond Power International, Inc. |
| a-s-h.us | | | | | | | |
| DN-109 | | Registered | DOMAIN NAME | 31-May-2005 | | 31-May-2005 | Diamond Power International, Inc. |
| ASHANDLER (| (Word N | Iark) | | | | | |
| TM-581 | | Registered | 78/027752 | 26-Sep-2000 | 2601135 | 30-Jul-2002 | Diamond Power International, Inc. |
| ASHCOLITE (S | STYLIZ | • | | | | | |
| TM-491 | | Registered | 72/181733 | 22-Nov-1963 | 774841 | 11-Aug-1964 | Diamond Power International, Inc. |
| ASHCORE | | | | | | | |
| TM-550 | | Registered | 75/485036 | 14-May-1998 | 2444743 | 17-Apr-2001 | Diamond Power International, Inc. |
| | | | | | | | |

14-May-1998

2394639

17-Oct-2000

75/485035

Thursday, April 29, 2010 Prepared by AMSaus

Registered

ASHFLO (Word Mark)

TM-492

Page 102 of 23

Diamond Power International, Inc.

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|--------------|---------|------------|-------------|-------------|---------|-------------|--|
| ASHseal | | | | | | | |
| TM-742 | | Filed | 78/596117 | 28-Mar-2005 | | | Diamond Power International, Inc. |
| ASHVAC (Wor | rd Mark |) | | | | | |
| TM-580 | | Registered | 76/117101 | 28-Aug-2000 | 2581199 | 18-Jun-2002 | Diamond Power International, Inc. |
| ashwp.com | | | | | | | |
| DN-027 | | Registered | DOMAIN NAME | 27-Jul-2000 | | 27-Jul-2000 | Diamond Power International, Inc. |
| ash-wp.com | | | | | | | |
| DN-028 | | Registered | DOMAIN NAME | 08-Aug-2000 | | 08-Aug-2000 | Diamond Power International, Inc. |
| ASP (Word Ma | rk) | | | | | | |
| TM-465 | | Registered | 74/649783 | 21-Mar-1995 | 2079738 | 15-Jul-1997 | Babcock & Wilcox Power Generation Group, Inc. |
| ASSURED STO | OCK PR | OGRAM (W | ord Mark) | | | | |
| TM-231 | | Registered | 73624366 | 03-Oct-1986 | 1443711 | 16-Jun-1987 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-231 | 2 | Registered | 74/418432 | 28-Jun-1993 | 1846943 | 26-Jul-1994 | Babcock & Wilcox Power Generation Group, Inc. |
| ave.dk | | | | | | | |
| DN-100 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 12-Jul-2004 | Babcock & Wilcox Volund A/S |
| B&W & HERO | ENGIN | NE DESIGN | | | | | |
| TM-058 | | Registered | 71/470170 | 12-May-1944 | 419052 | 05-Feb-1946 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-058 | 2 | Registered | 71/470169 | 12-May-1944 | 0415177 | 31-Jul-1945 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-058 | 3 | Registered | 71/491800 | 19-Nov-1945 | 0422229 | 09-Jul-1946 | Babcock & Wilcox Power Generation Group, Inc. |
| B&W (Word M | lark) | | | | | | |
| TM-074 | | Registered | 71/000828 | 08-Apr-1905 | 0044904 | 25-Jul-1905 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-074 | 2 | Registered | 71/470553 | 24-May-1944 | 415862 | 21-Aug-1945 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-074 | 3 | Registered | 73/782064 | 21-Feb-1989 | 1570257 | 05-Dec-1989 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-074 | 4 | Registered | 73/782065 | 21-Feb-1989 | 1573977 | 26-Dec-1989 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-074 | 6 | Registered | 75/747600 | 30-Jun-1999 | 2350120 | 16-May-2000 | Babcock & Wilcox Power Generation Group, Inc. |
| B&W mPower | | | | | | | |
| TM-757 | | Filed | 77/634702 | 17-Dec-2008 | | | Babcock & Wilcox Nuclear Power Gen. Group Inc. |
| B&W ROLL W | HEEL | | | | | | |
| TM-715 | | Registered | 78/440093 | 23-Jun-2004 | 3330614 | 06-Nov-2007 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-715 | 2 | Registered | 78/440113 | 23-Jun-2004 | 3319508 | 23-Oct-2007 | Babcock & Wilcox Power Generation Group, Inc. |

 $^{{}^{*}\;\;}McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,2010$

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|----------------|-------|--------------|--------------------|-------------|---------|-------------|---|
| BABCOCK & V | VILCO | K (Word Mark |) | | | | |
| TM-075 | | Registered | 71/000829 | 08-Apr-1905 | 0044905 | 25-Jul-1905 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-075 | 2 | Registered | 70/025550 | 18-Aug-1894 | 0025550 | 20-Nov-1894 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-075 | 4 | Registered | 73/782044 | 21-Feb-1989 | 1572421 | 19-Dec-1989 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-075 | 5 | Registered | 73/782055 | 21-Feb-1989 | 1571386 | 12-Dec-1989 | Babcock & Wilcox Power Generation Group, Inc. |
| BABCOCK & V | VILCO | X M BULLSEY | YE LOGO | | | | |
| TM-422 | | Registered | 74/499978 | 14-Mar-1994 | 1900543 | 20-Jun-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-422 | 2 | Registered | 74/499799 | 14-Mar-1994 | 1901922 | 27-Jun-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-422 | 3 | Registered | 74/499800 | 14-Mar-1994 | 1903181 | 04-Jul-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-422 | 4 | Registered | 74/501629 | 18-Mar-1994 | 1900934 | 20-Jun-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| BABCOCK & V | VILCO | K SERVICE SI | PECIALISTS (Word M | ark) | | | |
| TM-451 | | Registered | 74/591047 | 27-Oct-1994 | 1948476 | 16-Jan-1996 | Babcock & Wilcox Power Generation Group, Inc. |
| babcock.asia | | | | | | | |
| DN-175 | | Pending | DOMAIN NAME | | | | Babcock & Wilcox Canada Ltd. |
| babcock.co.in | | | | | | | |
| DN-299 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | The Babcock & Wilcox Company |
| babcock.co.kr | | | | | | | |
| DN-285 | | Registered | DOMAIN NAME | | | 01-Feb-2010 | Babcock & Wilcox Power Generation Group, Inc. |
| babcock.com | | | | | | | |
| DN-002 | | Registered | DOMAIN NAME | 24-Mar-1995 | | 24-Mar-1995 | The Babcock & Wilcox Company |
| babcock.com.hk | | | | | | | |
| DN-260 | | Registered | DOMAIN NAME | | | 04-Nov-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| babcock.com.sg | | | | | | | |
| DN-287 | | Registered | DOMAIN NAME | | | 01-Feb-2010 | Babcock & Wilcox Power Generation Group, Inc. |
| babcock.com.tw | , | | | | | | |
| DN-259 | | Registered | DOMAIN NAME | | | 04-Nov-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| babcock.dk | | | | | | | |
| DN-099 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 14-Jun-2004 | Babcock & Wilcox Volund A/S |
| babcock.jobs | | | | | | | |
| DN-110 | | Registered | DOMAIN NAME | 08-Jul-2005 | | 08-Jul-2005 | The Babcock & Wilcox Company |

 $^{{}^{*}\;\;}McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,\,2010$

Page 104 of 23

| CASE NO. | CLID | COT A PONT IC | , pp NO | EH E DATE | DEC NO. | DEC DATE | OWNER |
|----------------|-----------|---------------|-------------|-------------|---------|-------------|---|
| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
| DN-284 | | Registered | DOMAIN NAME | | | 01-Feb-2010 | Babcock & Wilcox Power Generation Group, Inc. |
| babcock.sg | | | | | | | |
| DN-286 | | Registered | DOMAIN NAME | | | 01-Feb-2010 | Babcock & Wilcox Power Generation Group, Inc. |
| babcockandwile | cock.com | | | | | | |
| DN-134 | | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| babcockandwile | cox.biz | | | | | | |
| DN-058 | | Registered | DOMAIN NAME | 25-Apr-2000 | | 13-Dec-2001 | The Babcock & Wilcox Company |
| babcockandwile | cox.co.in | | | | | | |
| DN-297 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | The Babcock & Wilcox Company |
| babcockandwile | cox.com.ı | nx | | | | | |
| DN-249 | | Registered | DOMAIN NAME | | | 20-Aug-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| babcockandwile | cox.in | | | | | | |
| DN-295 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | The Babcock & Wilcox Company |
| babcockandwile | coxcanad | a.com | | | | | |
| DN-117 | | Registered | DOMAIN NAME | 29-Nov-2005 | | 29-Nov-2005 | Babcock & Wilcox Canada Ltd. |
| babcockandwile | coxvolun | d.com | | | | | |
| DN-024 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 01-Jun-2000 | Babcock & Wilcox Volund A/S |
| babcockandwile | coxvolun | d.dk | | | | | |
| DN-026 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 01-Jun-2000 | Babcock & Wilcox Volund A/S |
| babcockwilcock | c.com | | | | | | |
| DN-130 | | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| babcockwilcox. | asia | | | | | | |
| DN-176 | | Pending | DOMAIN NAME | | | | Babcock & Wilcox Canada Ltd. |
| babcockwilcox. | biz | | | | | | |
| DN-059 | | Registered | DOMAIN NAME | 25-Apr-2000 | | 13-Dec-2001 | The Babcock & Wilcox Company |
| babcock-wilcox | .biz | | | | | | |
| DN-060 | | Registered | DOMAIN NAME | 25-Apr-2000 | | 13-Dec-2001 | The Babcock & Wilcox Company |
| | | | | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 105 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|----------|------------|-------------|-------------|--------|-------------|---|
| oabcockwilcox. | co.in | | | | | | |
| DN-300 | | Registered | DOMAIN NAME | 01-Apr-2010 | | 01-Apr-2010 | The Babcock & Wilcox Company |
| oabcock-wilcox. | .co.in | | | | | | |
| DN-298 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | The Babcock & Wilcox Company |
| oabcockwilcox. | com | | | | | | |
| DN-004 | | Registered | DOMAIN NAME | 19-May-1999 | | 19-May-1999 | The Babcock & Wilcox Company |
| oabcock-wilcox. | .com | | | | | | |
| DN-003 | | Registered | DOMAIN NAME | 19-May-1999 | | 19-May-1999 | The Babcock & Wilcox Company |
| oabcockwilcox. | com.mx | | | | | | |
| DN-248 | | Registered | DOMAIN NAME | | | 19-Aug-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| oabcockwilcox.o | dk | | | | | | |
| DN-097 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 14-Jun-2004 | Babcock & Wilcox Volund A/S |
| oabcockwilcox.i | in | | | | | | |
| DN-301 | | Registered | DOMAIN NAME | 01-Apr-2010 | | 01-Apr-2010 | The Babcock & Wilcox Company |
| oabcock-wilcox. | .in | | | | | | |
| DN-296 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | The Babcock & Wilcox Company |
| oabcockwilcox.j | jobs | | | | | | |
| DN-111 | | Registered | DOMAIN NAME | 08-Jul-2005 | | 08-Jul-2005 | The Babcock & Wilcox Company |
| oabcockwilcox.i | net | | | | | | |
| DN-272 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| oabcockwilcox. | us | | | | | | |
| DN-079 | | Registered | DOMAIN NAME | 09-Apr-2002 | | 25-Apr-2002 | The Babcock & Wilcox Company |
| oabcock-wilcox. | .us | | | | | | |
| DN-078 | | Registered | DOMAIN NAME | 09-Apr-2002 | | 25-Apr-2002 | The Babcock & Wilcox Company |
| oabcockwilcoxc | anada.co | om | | | | | |
| DN-116 | | Registered | DOMAIN NAME | 29-Nov-2005 | | 29-Nov-2005 | Babcock & Wilcox Canada Ltd. |
| oabcockwilcoxn | npower.c | com | | | | | |
| DN-220 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 106 of 23

| CASE NO | SUB STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|------------|-------------|-------------|--------|-------------|-----------------------------------|
| babcockwilcoxn | npower.net | | | | | |
| DN-234 | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| babcockwilcoxn | npower.org | | | | | |
| DN-232 | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| babcockwilcoxv | olund.com | | | | | |
| DN-023 | Registered | DOMAIN NAME | 01-Jun-2000 | | 01-Jun-2000 | Babcock & Wilcox Volund A/S |
| babcockwilcoxv | olund.dk | | | | | |
| DN-025 | Registered | DOMAIN NAME | 01-Jun-2000 | | 01-Jun-2000 | Babcock & Wilcox Volund A/S |
| babcoxandwilco | ox.com | | | | | |
| DN-135 | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| babcoxwilcox.co | om | | | | | |
| DN-131 | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| babockandwilco | ox.com | | | | | |
| DN-133 | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| babockwilcox.co | om | | | | | |
| DN-129 | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| bceparts.ca | | | | | | |
| DN-196 | Registered | DOMAIN NAME | | | 12-Jun-2008 | Diamond Canapower Inc. |
| bceparts.com | | | | | | |
| DN-194 | Registered | DOMAIN NAME | | | 11-Jun-2008 | Diamond Power International, Inc. |
| bceparts.net | | | | | | |
| DN-195 | Registered | DOMAIN NAME | | | 11-Jun-2008 | Diamond Power International, Inc. |
| beaumontbirch. | | | | | | |
| DN-076 | Registered | DOMAIN NAME | 11-Apr-2002 | | 11-Apr-2002 | Diamond Power International, Inc. |
| beaumontbirch. | | | | | | |
| DN-074 | Registered | DOMAIN NAME | 12-Apr-2002 | | 12-Apr-2002 | Diamond Power International, Inc. |
| beaumontbirch. | | | | | | |
| DN-075 | Registered | DOMAIN NAME | 12-Apr-2002 | | 12-Apr-2002 | Diamond Power International, Inc. |
| | | | | | | |

 $^{{}^{\}star}\;\;McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,\,2010$

Page 107 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | <u>OWNER</u> |
|----------------|---------|--------------------|-------------|-------------|---------|-------------|---|
| beaumontbirch. | org | | | | | | |
| DN-077 | | Registered | DOMAIN NAME | 12-Mar-2002 | | 12-Mar-2002 | Diamond Power International, Inc. |
| BLACK LIQUO | OR CLUE | (Word Mark) |) | | | | |
| TM-437 | | Registered | 74/492512 | 22-Feb-1994 | 1942067 | 19-Dec-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-437 | 3 | Registered | 76/383313 | 15-Mar-2002 | 2728790 | 24-Jun-2003 | Babcock & Wilcox Power Generation Group, Inc. |
| BLACK LIQUO | R CLUE | LOGO | | | | | |
| TM-433 | | Registered | 74/484952 | 31-Jan-1994 | 1939207 | 05-Dec-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-433 | 4 | Registered | 76/382991 | 18-Mar-2002 | 2728787 | 24-Jun-2003 | Babcock & Wilcox Power Generation Group, Inc. |
| BOS (Word Ma | rk) | | | | | | |
| TM-370 | | Registered | 74/274290 | 11-May-1992 | 1745641 | 12-Jan-1993 | Diamond Power International, Inc. |
| bs-teknik.dk | | | | | | | |
| DN-101 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 15-Jun-2004 | Babcock & Wilcox Volund A/S |
| b-w.us | | | | | | | |
| DN-082 | | Registered | DOMAIN NAME | 09-Apr-2002 | | 25-Apr-2002 | The Babcock & Wilcox Company |
| bwcanada.biz | | | | | | | |
| DN-051 | | Registered | DOMAIN NAME | 24-Mar-2001 | | 13-Dec-2001 | Babcock & Wilcox Canada Ltd. |
| bwcanada.com | | | | | | | |
| DN-041 | | Registered | DOMAIN NAME | 23-Mar-2001 | | 23-Mar-2001 | Babcock & Wilcox Canada Ltd. |
| bwcos.biz | | | | | | | |
| DN-136 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.co.uk | | | | | | | |
| DN-138 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.com | | | | | | | |
| DN-139 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.eu | | | | | | | |
| DN-140 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.info | | | | | | | |
| DN-141 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.name | | D 14 1 | DOMARINA | 16 11 2006 | | 1631 2006 | DWV T. I. I. ' |
| DN-142 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|--------|-----------------|--------------|-------------|--------|-------------|---|
| bwcos.net | | | | | | | |
| DN-143 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.org | | | | | | | |
| DN-144 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.tv | | | | | | | |
| DN-145 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.us | | | | | | | |
| DN-146 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwcos.ws | | | | | | | |
| DN-147 | | Registered | DOMAIN NAME | 16-Nov-2006 | | 16-Nov-2006 | BWX Technologies, Inc. |
| bwdemonterrey.c | com | | | | | | |
| DN-252 | | Registered | DOMAIN NAME | | | 03-Sep-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| bwdemonterrey.c | com.mx | | | | | | |
| DN-251 | | Registered | DOMAIN NAME | | | 03-Sep-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| bwdemonterrey. | net | | | | | | |
| DN-253 | | Registered | DOMAIN NAME | | | 03-Sep-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| bwempower.com | | | | | | | |
| DN-227 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwempower.net | | | | | | | |
| DN-225 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwempower.org | | | | | | | |
| DN-219 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmichoud.com | | | | | | | |
| DN-197 | | Registered | DOMAIN NAME | | | 25-Jun-2008 | Babcock & Wilcox Technical Services Group, Inc. |
| bwmichoud.net | | | | | | | |
| DN-198 | | Registered | DOMAIN NAME | | | 25-Jun-2008 | Babcock & Wilcox Technical Services Group, Inc. |
| bwmne.com | | | | | | | |
| DN-218 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| * MaDawmatt To | 1 1 | In a shanged to | 4- D-bl- 6 W | /! T I I | | 10 | |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 109 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|-----------|------------|-------------|-----------|--------|-------------|---|
| bwmne.net | | | | | | | |
| DN-217 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmne.org | | | | | | | |
| DN-236 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmodularnucle | arenergy. | com | | | | | |
| DN-235 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmodularnucle | arenergy. | net | | | | | |
| DN-224 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmodularnucle | arenergy. | org | | | | | |
| DN-223 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmpower.com | | | | | | | |
| DN-230 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bw-mpower.com | | | | | | | |
| DN-233 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmpower.net | | | | | | | |
| DN-228 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bw-mpower.net | | | | | | | |
| DN-231 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwmpower.org | | | | | | | |
| DN-226 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bw-mpower.org | | | | | | | |
| DN-229 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| bwnevadallc.com | ı | | | | | | |
| DN-199 | | Registered | DOMAIN NAME | | | 25-Jun-2008 | Babcock & Wilcox Technical Services Group, Inc. |
| bwnevadallc.net | | | | | | | Group, me. |
| DN-200 | | Registered | DOMAIN NAME | | | 25-Jun-2008 | Babcock & Wilcox Technical Services Group, Inc. |
| bwrsllc.com | | | | | | | |
| DN-201 | | Registered | DOMAIN NAME | | | 08-Jul-2008 | Babcock & Wilcox Technical Services Group, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 110 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------------|-----|------------|-----------------|-------------|--------|-------------|---|
| bwrsllc.net | | | | | | | |
| DN-202 | | Registered | DOMAIN NAME | | | 08-Jul-2008 | Babcock & Wilcox Technical Services Group, Inc. |
| bwservice.biz | | | | | | | |
| DN-061 | | Registered | DOMAIN NAME | 25-Apr-2000 | | 13-Dec-2001 | The Babcock & Wilcox Company |
| bwservice.com | | | | | | | |
| DN-005 | | Registered | DOMAIN NAME | 12-Jul-1999 | | 12-Jul-1999 | BWX Technologies, Inc. |
| bwvolund.asia | | | | | | | |
| DN-274 | | Registered | DOMAIN NAME | | | 18-Nov-2008 | Babcock & Wilcox Volund A/S |
| bwvolund.com | | | | | | | |
| DN-273 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| bwvolund.dk | | | | | | | |
| DN-277 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| bwvolund.eu | | | | | | | |
| DN-275 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| owvolund.net | | | | | | | |
| DN-276 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| bwx.com | | | | | | | |
| DN-006 | | Registered | DOMAIN NAME | 24-Jun-1997 | | 24-Jun-1997 | BWX Technologies, Inc. |
| bwxt.biz | | | B0141 B1111 B | | | | |
| DN-084 | | Registered | DOMAIN NAME | | | 08-May-2002 | BWX Technologies, Inc. |
| bwxt.com | | D 1 . 1 | DOM A DANA A CE | 24.1 1007 | | 24.1 1005 | DVIV.T. I. I. I. I. |
| DN-007 | | Registered | DOMAIN NAME | 24-Jun-1997 | | 24-Jun-1997 | BWX Technologies, Inc. |
| bwxt.us DN-072 | | D ! | DOMAININAME | 26 M 2002 | | 10. 4 2002 | DWV Teslevelesies Inc |
| | | Registered | DOMAIN NAME | 26-Mar-2002 | | 19-Apr-2002 | BWX Technologies, Inc. |
| bwxtech.com DN-008 | | Dagistarad | DOMAIN NAME | 24 Jun 1007 | | 24 Jun 1007 | DWV Tashnalagias Inc |
| | _ | Registered | DOMAIN NAME | 24-Jun-1997 | | 24-Jun-1997 | BWX Technologies, Inc. |
| bwxtechnologies.con | ı | Danistanad | DOMAINI NIAME | 10 Eak 2001 | | 21 Man 2002 | DWV Tashualasias Inc |
| DN-071 | | Registered | DOMAIN NAME | 19-Feb-2001 | | 21-Mar-2002 | BWX Technologies, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 111 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|--------------------|----------|-----------------|-------------|--------------|---------|--------------|---|
| bwx-technologie | | | <u></u> | <u></u> | | | <u> </u> |
| DN-070 | | Registered | DOMAIN NAME | 19-Feb-2001 | | 21-Mar-2002 | BWX Technologies, Inc. |
| bwxtechnologies | .net | Ü | | | | | <u> </u> |
| DN-121 | | Registered | DOMAIN NAME | | | 25-Jan-2006 | BWX Technologies, Inc. |
| bwxtechnologies | inc.com | | | | | | |
| DN-120 | | Registered | DOMAIN NAME | | | 25-Jan-2006 | BWX Technologies, Inc. |
| bwxtechnologies | inc.net | | | | | | |
| DN-119 | | Registered | DOMAIN NAME | | | 25-Jan-2006 | BWX Technologies, Inc. |
| bwxtned.com | | | | | | | |
| DN-112 | | Registered | DOMAIN NAME | 25-Oct-2005 | | 25-Oct-2005 | BWX Technologies, Inc. |
| bwxt-pantex.com | 1 | | | | | | |
| DN-067 | | Registered | DOMAIN NAME | 19-Feb-2001 | | 21-Mar-2002 | BWX Technologies, Inc. |
| bwxt-y12.com | | | | | | | |
| DN-068 | | Registered | DOMAIN NAME | 19-Feb-2001 | | 21-Mar-2002 | BWX Technologies, Inc. |
| bwxt-y-12.com | | | | | | | |
| DN-069 | | Registered | DOMAIN NAME | 19-Feb-2001 | | 21-Mar-2002 | BWX Technologies, Inc. |
| bwxty-12.com | | | | | | | |
| DN-066 | | Registered | DOMAIN NAME | 19-Feb-2001 | | 21-Mar-2002 | BWX Technologies, Inc. |
| CEMSCAN | | | | | | | |
| TM-789 | | Registered | 74/605141 | 01-Dec-1994 | 2016261 | 12-Nov-1996 | Babcock & Wilcox Power Generation Group, Inc. |
| CFS | | | | | | | Group, me. |
| TM-756 | | Filed | 77/602954 | 29-Oct-2008 | 3711230 | 17-Nov-2009 | Babcock & Wilcox Power Generation |
| 1141-750 | | Tiled | 777002754 | 29-001-2000 | 3711230 | 17-1407-2009 | Group, Inc. |
| chuihuigi.com.cr | (pingyin | for sootblower) | | | | | |
| DN-214 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| CPM | | | | | | | |
| TM-769 | | Registered | 75/318638 | 02-Jul-1997 | 2347485 | 02-May-2000 | Babcock & Wilcox Power Generation |
| Cl- ACII | | | | | | | Group, Inc. |
| CycloASH TM 710 | | Darietana i | 79/246410 | 06 Mars 2002 | 21/0220 | 07 N 2006 | Diamand Daman International I |
| TM-719 | | Registered | 78/246419 | 06-May-2003 | 3169328 | 07-Nov-2006 | Diamond Power International, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 112 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|------------------|------------|---------------------------------------|-------------|-------------|---------|-------------|---|
| deltapowerservic | es.com | · · · · · · · · · · · · · · · · · · · | | | | | |
| DN-203 | | Registered | DOMAIN NAME | | | 13-Aug-2003 | Babcock & Wilcox Power Generation Group, Inc. |
| DIAMOND (Wo | rd Mark) | | | | | | |
| TM-258 | | Registered | 71/304190 | 04-Aug-1930 | 279300 | 13-Jan-1931 | Diamond Power International, Inc. |
| TM-258 | 2 | Registered | 71/096578 | 15-Jul-1916 | 113950 | 14-Nov-1916 | Diamond Power International, Inc. |
| DIAMOND POV | VER (Wo | rd Mark) | | | | | |
| TM-261 | | Registered | 72/139604 | 12-Mar-1962 | 0746554 | 12-Mar-1963 | Diamond Power International, Inc. |
| TM-261 | 2 | Registered | 72/139605 | 12-Mar-1962 | 0755020 | 20-Aug-1963 | Diamond Power International, Inc. |
| diamond.cn (Chi | nese char | acters) | | | | | |
| DN-209 | | Registered | DOMAIN NAME | | | 17-Jun-2006 | Diamond Power International, Inc. |
| diamond.com (C | hinese cha | aracters) | | | | | |
| DN-208 | | Registered | DOMAIN NAME | | | 30-Nov-2007 | Diamond Power International, Inc. |
| diamond.net (Ch | inese cha | racters) | | | | | |
| DN-211 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| diamond.net.cn (| Chinese c | haracters) | | | | | |
| DN-210 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| diamondelectron | ics.biz | | | | | | |
| DN-055 | | Registered | DOMAIN NAME | 16-Mar-1996 | | 13-Dec-2001 | Diamond Power International, Inc. |
| diamondelectron | ics.in | | | | | | |
| DN-289 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | Diamond Power International, Inc. |
| diamondelectron | ics.net | | | | | | |
| DN-045 | | Filed | DOMAIN NAME | 16-Mar-2000 | | | Diamond Power International, Inc. |
| diamondelectron | ics.org | | | | | | |
| DN-044 | | Registered | DOMAIN NAME | 03-Oct-2001 | | 03-Oct-2001 | Diamond Power International, Inc. |
| diamondpower.b | oiz | | | | | | |
| DN-054 | | Registered | DOMAIN NAME | 16-Mar-1996 | | 13-Dec-2001 | Diamond Power International, Inc. |
| diamondpower.c | a | | | | | | |
| DN-205 | | Registered | DOMAIN NAME | | | 26-May-2009 | Diamond Canapower Inc. |
| diamondpower.c | o.uk | | | | | | |
| DN-029 | | Registered | DOMAIN NAME | 12-Sep-1996 | | 12-Sep-1996 | Diamond Power International, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 113 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------------------|---------|---------------------|-------------|-------------|--------|-------------|-----------------------------------|
| diamondpower.com | | | | | | | |
| DN-012 diamondpower.eu | | Registered | DOMAIN NAME | 16-Mar-1996 | | 16-Mar-1996 | Diamond Power International, Inc. |
| DN-180 diamondpower.in | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| DN-290 diamondpower.jobs | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | Diamond Power International, Inc. |
| DN-113 diamondpower.net | | Registered | DOMAIN NAME | 18-Aug-2005 | | 18-Aug-2005 | Diamond Power International, Inc. |
| DN-033 diamondpower.org | | Registered | DOMAIN NAME | 04-Dec-2000 | | 04-Dec-2000 | Diamond Power International, Inc. |
| DN-034 diamondpower.us | | Registered | DOMAIN NAME | 04-Dec-2000 | | 04-Dec-2000 | Diamond Power International, Inc. |
| DN-083 diamondpower-cee. | ez | Registered | DOMAIN NAME | 09-Apr-2002 | | 25-Apr-2002 | Diamond Power International, Inc. |
| DN-206 diamondpower-cee. | eu | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| DN-207 diamondpowerinter | nationa | Pending al.co.in | DOMAIN NAME | | | | Diamond Power International, Inc. |
| DN-291 diamondpowerinter | nationa | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | Diamond Power International, Inc. |
| DN-171 diamondpowerinter | nationa | Registered | DOMAIN NAME | | | 09-Apr-2007 | Diamond Power International, Inc. |
| DN-292 diamondpowerinter | nationa | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | Diamond Power International, Inc. |
| DN-172 diamondpowerinter | nationa | Registered | DOMAIN NAME | | | 10-Apr-2007 | Diamond Power International, Inc. |
| DN-173 | | Registered | DOMAIN NAME | | | 10-Apr-2007 | Diamond Power International, Inc. |
| | | | | | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 114 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|-------------|------------|-------------|-------------|---------|-------------|--|
| diamondpowerk | orea.com | | | | | | |
| DN-181 | | Registered | DOMAIN NAME | | | 09-Apr-2004 | Diamond Power International, Inc. |
| diamondpowersp | ecialty.co | .in | | | | | |
| DN-293 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | Diamond Power International, Inc. |
| diamondpowersp | ecialty.co | m | | | | | |
| DN-118 | | Registered | DOMAIN NAME | 29-Nov-2005 | | 29-Nov-2005 | Diamond Power International, Inc. |
| diamondpowers | oecialty.in | | | | | | |
| DN-294 | | Registered | DOMAIN NAME | 31-Mar-2010 | | 31-Mar-2010 | Diamond Power International, Inc. |
| DIAMONIZED | | | | | | | |
| TM-449 | 1 | Registered | 78/570907 | 18-Feb-2005 | 3455244 | 24-Jun-2008 | Diamond Power International, Inc. |
| DIAMONIZED | (Word Ma | rk) | | | | | |
| TM-449 | 2 | Registered | 75/652227 | 26-Feb-1999 | 2448661 | 08-May-2001 | Diamond Power International, Inc. |
| DP & Design (Co | olor) (NEV | V logo) | | | | | |
| TM-709 | | Registered | 78/143464 | 12-Jul-2002 | 3018397 | 22-Nov-2005 | Diamond Power International, Inc. |
| DP (STYLIZED | DESIGN) | | | | | | |
| TM-262 | | Registered | 72/139733 | 13-Mar-1962 | 741246 | 27-Nov-1962 | Diamond Power International, Inc. |
| TM-262 | 2 | Registered | 72/139732 | 13-Mar-1962 | 741270 | 27-Nov-1962 | Diamond Power International, Inc. |
| DRB-4Z (Word | Mark) | | | | | | |
| TM-531 | 2 | Registered | 75/545775 | 01-Sep-1998 | 2446782 | 24-Apr-2001 | Babcock & Wilcox Power Generation Group, Inc. |
| DRB-XCL (Wor | d Mark) | | | | | | |
| TM-368 | | Registered | 74/239934 | 24-Jan-1992 | 1741145 | 22-Dec-1992 | Babcock & Wilcox Power Generation Group, Inc. |
| DSVS (Word Ma | ırk) | | | | | | |
| TM-552 | | Registered | 75/506297 | 22-Jun-1998 | 2393999 | 10-Oct-2000 | Babcock & Wilcox Power Generation Group, Inc. |
| ELVERITE (Wo | ord Mark) | | | | | | |
| TM-084 | 1 | Registered | 76/447851 | 09-Sep-2002 | 2741528 | 29-Jul-2003 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-084 | 2 | Registered | 77/587246 | 07-Oct-2008 | 3619276 | 12-May-2009 | Babcock & Wilcox Power Generation Group, Inc. |
| enertec.com | | | | | | | |
| DN-306 | | Registered | DOMAIN NAME | | | 22-Oct-1995 | Babcock & Wilcox Power Generation Group, Inc. |
| EXTREME | | | | | | | |
| TM-767 | | Pending | 85/006295 | 05-Apr-2010 | | | Diamond Power International, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 115 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|------------|------------|-------------|-------------|---------|-------------|---|
| FlexRod | | | | | | | |
| TM-720 | | Registered | 78/246422 | 06-May-2003 | 2978785 | 26-Jul-2005 | Diamond Power International, Inc. |
| FST-GAGE (Wo | rd Mark) | | | | | | |
| TM-479 | 2 | Registered | 75/295372 | 20-May-1997 | 2162018 | 02-Jun-1998 | Babcock & Wilcox Power Generation Group, Inc. |
| GASTEMP & D | ESIGN | | | | | | |
| TM-403 | | Registered | 74/415759 | 23-Jul-1993 | 1873659 | 17-Jan-1995 | Diamond Power International, Inc. |
| GEMINI (Word | Mark) | | | | | | |
| TM-598 | | Registered | 76/215,395 | 20-Feb-2001 | 2687243 | 11-Feb-2003 | Diamond Power International, Inc. |
| GenClad | | | | | | | |
| TM-704 | | Registered | 78/233349 | 03-Apr-2003 | 2886366 | 21-Sep-2004 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-704 | 4 | Registered | 78/233339 | 03-Apr-2003 | 2889004 | 28-Sep-2004 | Babcock & Wilcox Power Generation Group, Inc. |
| GenClad (Word | Mark) | | | | | | |
| TM-704 | 2 | Registered | 78/233356 | 03-Apr-2003 | 2817469 | 24-Feb-2004 | Babcock & Wilcox Power Generation Group, Inc. |
| generatingpower | fulsolutio | ns.com | | | | | |
| DN-132 | | Registered | DOMAIN NAME | | | 24-Mar-2006 | The Babcock & Wilcox Company |
| HONE & GLOW | (Word I | Mark) | | | | | |
| TM-287 | | Registered | 73/723107 | 18-Apr-1988 | 1516861 | 13-Dec-1988 | Babcock & Wilcox Power Generation Group, Inc. |
| HYDROBIN | | | | | | | |
| TM-500 | | Registered | 72/181735 | 22-Nov-1963 | 773659 | 21-Jul-1964 | Diamond Power International, Inc. |
| HYDRO-EJECT | OR (Wor | d Mark) | | | | | |
| TM-501 | | Registered | 73/374330 | 12-Jul-1982 | 1249599 | 30-Aug-1983 | Diamond Power International, Inc. |
| HYDROJET (W | ord Mark | () | | | | | |
| TM-700 | | Registered | 76/230869 | 12-Mar-2001 | 2599299 | 23-Jul-2002 | Diamond Power International, Inc. |
| HYDROVACTO | R | | | | | | |
| TM-506 | | Registered | 72/181734 | 22-Nov-1963 | 774939 | 11-Aug-1964 | Diamond Power International, Inc. |
| IK-700 | | | | | | | |
| TM-754 | | Registered | 77/496333 | 11-Jun-2008 | 3642266 | 23-Jun-2009 | Diamond Power International, Inc. |
| IMPAK | | | | | | | |
| TM-774 | | Registered | 75/048111 | 24-Jan-1996 | 2232541 | 16-Mar-1999 | Babcock & Wilcox Power Generation Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 116 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|------------|-------------|----------------------|-------------|---------|-------------|---|
| iveycooper.com | | | | | | | |
| DN-204 | | Pending | DOMAIN NAME | | | | Babcock & Wilcox Power Generation Group, Inc. |
| JUICECAN | | | | | | | |
| TM-777 | | Registered | 76/449434 | 13-Sep-2002 | 2736486 | 15-Jul-2003 | Babcock & Wilcox Power Generation Group, Inc. |
| KVB | | | | | | | |
| TM-790 | | Registered | 73/291272 | 29-Dec-1980 | 1189025 | 02-Feb-1982 | Babcock & Wilcox Power Generation Group, Inc. |
| kvb-cems.com | | | | | | | |
| DN-303 | | Registered | DOMAIN NAME | | | 05-Aug-1996 | Babcock & Wilcox Power Generation Group, Inc. |
| kvb-enertec.com | | | | | | | |
| DN-302 | | Registered | DOMAIN NAME | | | 29-Oct-1998 | Babcock & Wilcox Power Generation Group, Inc. |
| LINKS CONNE | CTING LI | EADERSHIP & | vision (design) | | | | |
| TM-708 | | Registered | 78/158693 | 28-Aug-2002 | 2972786 | 19-Jul-2005 | BWX Technologies, Inc. |
| LINKS CONNE | CTING LI | EADERSHIP A | AND VISION (word mar | rk) | | | |
| TM-707 | | Registered | 78/158553 | 28-Aug-2002 | 2968108 | 12-Jul-2005 | BWX Technologies, Inc. |
| LM2100 (Word | Mark) | | | | | | |
| TM-588 | | Registered | 76/158574 | 03-Nov-2000 | 2490831 | 18-Sep-2001 | Babcock & Wilcox Power Generation Group, Inc. |
| marinemech.con | n | | | | | | |
| DN-192 | | Registered | DOMAIN NAME | | | 22-Jun-2008 | BWX Technologies, Inc. |
| marinemech.net | | | | | | | |
| DN-193 | | Registered | DOMAIN NAME | | | 22-Jun-2008 | BWX Technologies, Inc. |
| MILLCARE (W | ord Mark) |) | | | | | |
| TM-538 | 3 | Registered | 76/360419 | 17-Jan-2002 | 2736043 | 15-Jul-2003 | Babcock & Wilcox Power Generation Group, Inc. |
| modularnuclear | energy.net | | | | | | |
| DN-222 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| modularnuclear | energy.org | | | | | | |
| DN-221 | | Registered | DOMAIN NAME | | | 15-Jun-2009 | B&W Modular Nuclear Energy, LLC |
| mtichaos.com | | | | | | | |
| DN-030 | | Registered | DOMAIN NAME | 28-Aug-2000 | | 28-Aug-2000 | McDermott Technology, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 117 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|------------------|------------|------------|-------------|-------------|---------|-------------|---|
| mtiresearch.com | 1 | | | | | | |
| DN-017 | | Registered | DOMAIN NAME | 13-Jun-1997 | | 13-Jun-1997 | McDermott Technology, Inc. |
| MULTICLONE | (Word Ma | ark) | | | | | |
| TM-510 | | Registered | 72/421977 | 20-Apr-1972 | 0978546 | 12-Feb-1974 | Babcock & Wilcox Power Generation Group, Inc. |
| MYB&W (stand | lard chara | cters) | | | | | |
| TM-733 | | Registered | 78/433112 | 10-Jun-2004 | 3034676 | 27-Dec-2005 | Babcock & Wilcox Power Generation Group, Inc. |
| mybwbabcock.c | om | | | | | | |
| DN-182 | | Pending | DOMAIN NAME | | | | The Babcock & Wilcox Company |
| mybwstore.com | | | | | | | |
| DN-123 | | Registered | DOMAIN NAME | | | 17-Feb-2006 | The Babcock & Wilcox Company |
| NETDAHS | | | | | | | |
| TM-787 | | Pending | COMMON LAW | | | | Babcock & Wilcox Power Generation Group, Inc. |
| netdahs.com | | | | | | | |
| DN-305 | | Registered | DOMAIN NAME | | | 30-Nov-2001 | Babcock & Wilcox Power Generation Group, Inc. |
| NOTIS (Word M | Iark) | | | | | | |
| TM-249 | 2 | Registered | 73/715221 | 07-Mar-1988 | 1543964 | 13-Jun-1989 | Babcock & Wilcox Power Generation Group, Inc. |
| ntdahs.com | | | | | | | |
| DN-304 | | Registered | DOMAIN NAME | | | 31-Oct-2000 | Babcock & Wilcox Power Generation Group, Inc. |
| nuclearfuelservi | ces.com | | | | | | |
| DN-288 | | Registered | DOMAIN NAME | | | 22-May-2000 | Nuclear Fuel Services, Inc. |
| OBSERVER | | | | | | | |
| TM-788 | | Pending | COMMON LAW | | | | Babcock & Wilcox Power Generation Group, Inc. |
| ONE WAY (Wo | rd Mark) | | | | | | |
| TM-455 | | Registered | 75/422315 | 19-Jan-1998 | 2290894 | 09-Nov-1999 | Diamond Power International, Inc. |
| ourbw.net | | | | | | | |
| DN-237 | | Registered | DOMAIN NAME | | | 09-Jul-2009 | The Babcock & Wilcox Company |
| ourbw.org | | | | | | | |
| DN-238 | | Registered | DOMAIN NAME | | | 09-Jul-2009 | The Babcock & Wilcox Company |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 118 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|----------|------------|-------------|-------------|---------|-------------|---|
| ourbwbabcock.co | | STATUS | ATT NO | FILE DATE | REGINO | REGDATE | OWNER |
| DN-239 | | Registered | DOMAIN NAME | | | 09-Jul-2009 | The Babcock & Wilcox Company |
| ourbwbabcock.ne | et | | | | | | |
| DN-240 | | Registered | DOMAIN NAME | | | 09-Jul-2009 | The Babcock & Wilcox Company |
| ourbwbabcock.o | rg | | | | | | |
| DN-241 | | Registered | DOMAIN NAME | | | 09-Jul-2009 | The Babcock & Wilcox Company |
| ourdiamond.net | | | | | | | |
| DN-242 | | Registered | DOMAIN NAME | | | 10-Jul-2009 | Diamond Power International, Inc. |
| ourdiamond.org | | | | | | | |
| DN-243 | | Registered | DOMAIN NAME | | | 10-Jul-2009 | Diamond Power International, Inc. |
| ourdiamondbabo | cock.com | | | | | | |
| DN-244 | | Registered | DOMAIN NAME | | | 10-Jul-2009 | Diamond Power International, Inc. |
| ourdiamondbabo | cock.net | | | | | | |
| DN-245 | | Registered | DOMAIN NAME | | | 10-Jul-2009 | Diamond Power International, Inc. |
| ourdiamondbabo | ock.org | | | | | | |
| DN-246 | | Registered | DOMAIN NAME | | | 10-Jul-2009 | Diamond Power International, Inc. |
| PARTNERSHIP | STOCKI | NG PROGRAM | ſ | | | | |
| TM-347 | | Registered | 74/062676 | 25-May-1990 | 1720593 | 29-Sep-1992 | Diamond Power International, Inc. |
| PARTS PLUS (V | Vord Mar | k) | | | | | |
| TM-432 | | Registered | 74/485889 | 02-Feb-1994 | 1951818 | 23-Jan-1996 | Babcock & Wilcox Power Generatio Group, Inc. |
| PCFAD | | | | | | | Group, me. |
| TM-349 | | Registered | 74/078735 | 16-Jul-1990 | 1652236 | 30-Jul-1991 | McDermott Technology, Inc. |
| PowerASH | | Registered | 14/0/0/33 | 10-341-1990 | 1032230 | 30-Jul-1991 | Medermon recliniology, inc. |
| TM-739 | | Registered | 78/513189 | 08-Nov-2004 | 3065460 | 07-Mar-2006 | Diamond Power International, Inc. |
| POWERTEC | | 8 | , | | | ., | , |
| TM-794 | | Pending | COMMON LAW | | | | Babcock & Wilcox Power Generation Group, Inc. |
| powertec.com | | | | | | | |
| DN-307 | | Registered | DOMAIN NAME | | | 01-Sep-1994 | Babcock & Wilcox Power Generatio Group, Inc. |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 119 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-------------------|-----------|-----------------|-------------|-------------|---------|-------------|---|
| PowerTrain | | | | | | | |
| TM-714 | | Registered | 78/208057 | 28-Jan-2003 | 2897116 | 26-Oct-2004 | Diamond Power International, Inc. |
| PRC-1000 | | | | | | | |
| TM-778 | | Registered | 74/012213 | 18-Dec-1989 | 1614727 | 25-Sep-1990 | Babcock & Wilcox Power Generation Group, Inc. |
| PRECIPTECH | AND DI | ESIGN | | | | | |
| TM-780 | | Registered | 73/583729 | 19-Feb-1986 | 1410236 | 23-Sep-1986 | Babcock & Wilcox Power Generation Group, Inc. |
| PRECISION CI | LEAN (V | Vord Mark) | | | | | |
| TM-567 | | Registered | 76/042361 | 24-Apr-2000 | 2529275 | 15-Jan-2002 | Diamond Power International, Inc. |
| PrecisionJet (sta | andard c | haracters) | | | | | |
| TM-722 | 2 | Registered | 78/470198 | 19-Aug-2004 | 3147543 | 26-Sep-2006 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-722 | 3 | Registered | 78/470249 | 19-Aug-2004 | 3013278 | 08-Nov-2005 | Babcock & Wilcox Power Generation Group, Inc. |
| PrecisionJet (sty | ylized an | d/or with desig | gn) | | | | |
| TM-722 | | Registered | 78/470266 | 19-Aug-2004 | 3147544 | 26-Sep-2006 | Babcock & Wilcox Power Generation Group, Inc. |
| PRIMA | | | | | | | |
| TM-752 | | Filed | 77/585184 | 03-Oct-2008 | | | Babcock & Wilcox Canada Ltd. |
| RACER (Word | Mark) | | | | | | |
| TM-140 | | Registered | 72/267795 | 29-Mar-1967 | 0855089 | 20-Aug-1968 | Babcock & Wilcox Power Generation Group, Inc. |
| RSAT | | | | | | | |
| TM-755 | | Published | 77/508936 | 26-Jun-2008 | | | Babcock & Wilcox Power Generation Group, Inc. |
| SENTRYSERIE | ES | | | | | | |
| TM-743 | | Registered | 78/665617 | 07-Jul-2005 | 3589020 | 10-Mar-2009 | Diamond Power International, Inc. |
| sentryseries.con | 1 | | | | | | |
| DN-124 | | Registered | DOMAIN NAME | 02-Feb-2006 | | 02-Feb-2006 | Diamond Power International, Inc. |
| sentryseries.net | | | | | | | |
| DN-127 | | Registered | DOMAIN NAME | 01-Mar-2006 | | 01-Mar-2006 | Diamond Power International, Inc. |
| sentryseriescont | trols.com | 1 | | | | | |
| DN-125 | | Registered | DOMAIN NAME | 02-Feb-2006 | | 02-Feb-2006 | Diamond Power International, Inc. |
| sentryseriescont | trols.net | | | | | | |
| DN-126 | | Registered | DOMAIN NAME | 01-Mar-2006 | | 01-Mar-2006 | Diamond Power International, Inc. |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 120 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-----------------|----------|-----------------|--------------------------|-------------|---------|-------------|---|
| sentryseriescon | trols.us | | | | | | |
| DN-174 | | Registered | DOMAIN NAME | | | 04-Dec-2007 | Diamond Power International, Inc. |
| SMARTREPOL | RTS | | | | | | |
| TM-786 | | Pending | COMMON LAW | | | | Babcock & Wilcox Power Generation Group, Inc. |
| SmeltCam | | | | | | | |
| TM-730 | | Registered | 78/341285 | 16-Dec-2003 | 3415516 | 22-Apr-2008 | Diamond Power International, Inc. |
| sootblower.co.u | ık | | | | | | |
| DN-178 | | Registered | DOMAIN NAME | | | 04-Jun-2004 | Diamond Power International, Inc. |
| sootblower.com | (Chines | e characters) | | | | | |
| DN-212 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| sootblower.com | .cn (Chi | iese characters |) | | | | |
| DN-213 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| sootblower.eu | | | | | | | |
| DN-179 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| sootblower.net | | | | | | | |
| DN-177 | | Registered | DOMAIN NAME | | | 26-Feb-2004 | Diamond Power International, Inc. |
| sootblower.net | (Chinese | characters) | | | | | |
| DN-215 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| sootblower.net. | en (Chin | ese characters) | | | | | |
| DN-216 | | Pending | DOMAIN NAME | | | | Diamond Power International, Inc. |
| SQ-300 | | | | | | | |
| TM-781 | | Registered | 75/718164 | 24-May-1999 | 2399532 | 31-Oct-2000 | Babcock & Wilcox Power Generation Group, Inc. |
| STIRLING | | | | | | | • * |
| TM-078 | | Registered | 71/029843 | 10-Sep-1907 | 0089810 | 14-Jan-1913 | Babcock & Wilcox Power Generation Group, Inc. |
| thebwstore.com | 1 | | | | | | |
| DN-122 | | Registered | DOMAIN NAME | | | 17-Feb-2006 | The Babcock & Wilcox Company |
| THERMO-PRO | OBE (Wo | rd Mark) | | | | | |
| TM-272 | | Registered | 72/115223 | 09-Mar-1961 | 723589 | 07-Nov-1961 | Diamond Power International, Inc. |
| * ** ** | | | l ita nama ta Dahasalı (| | | | |

^{*} McDermott Technology Inc. changed its name to Babcock & Wilcox Technology, Inc April 20, 2010

Page 121 of 23

| G. OP NO | av.n | OT - TV-10 | . PP VO | | PROMO | | OVA 777 |
|-------------------|---------------|------------------|-------------|-------------|---------|-------------|---|
| CASE NO TOWERDAY | SUB Word M | STATUS order) | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
| TOWERPAK (*TM-352 | woru wi | | 74/086326 | 00 Aug 1000 | 1652309 | 30-Jul-1991 | Babcock & Wilcox Power Generation |
| 1101-332 | | Registered | 74/080320 | 09-Aug-1990 | 1632309 | 30-Jul-1991 | Group, Inc. |
| TRUFLUX | | | | | | | |
| TM-721 | | Registered | 78/247391 | 08-May-2003 | 2830175 | 06-Apr-2004 | Babcock & Wilcox Power Generation Group, Inc. |
| TSSG | | | | | | | |
| TM-749 | | Filed | 77/601817 | 28-Oct-2008 | | | Babcock & Wilcox Canada Ltd. |
| UtiliCam | | | | | | | |
| TM-738 | | Registered | 78/494808 | 05-Oct-2004 | 3410329 | 08-Apr-2008 | Diamond Power International, Inc. |
| V P A S (Word | Mark) | | | | | | |
| TM-192 | | Registered | 73/395364 | 30-Sep-1982 | 1291929 | 28-Aug-1984 | Babcock & Wilcox Power Generation Group, Inc. |
| VAM (Word M | ark) | | | | | | |
| TM-150 | 2 | Registered | 73/409839 | 17-Jan-1983 | 1299266 | 09-Oct-1984 | Babcock & Wilcox Power Generation Group, Inc. |
| TM-150 | 3 | Registered | 73/737264 | 30-Jun-1988 | 1564170 | 07-Nov-1989 | Babcock & Wilcox Power Generation Group, Inc. |
| volund.asia | | | | | | | |
| DN-271 | | Registered | DOMAIN NAME | | | 21-Jul-2008 | Babcock & Wilcox Volund A/S |
| volund.biz | | | | | | | |
| DN-102 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 16-Jun-2004 | Babcock & Wilcox Volund A/S |
| volund.cn | | | | | | | |
| DN-262 | | Registered | DOMAIN NAME | | | 10-Oct-2006 | Babcock & Wilcox Volund A/S |
| volund.com.cn | | | | | | | |
| DN-263 | | Registered | DOMAIN NAME | | | 10-Oct-2006 | Babcock & Wilcox Volund A/S |
| volund.com.hk | | | | | | | |
| DN-264 | | Registered | DOMAIN NAME | | | 23-Jan-2008 | Babcock & Wilcox Volund A/S |
| volund.com.tw | | | | | | | |
| DN-270 | | Registered | DOMAIN NAME | | | 30-Jun-2008 | Babcock & Wilcox Volund A/S |
| volund.dk | | | | | | | |
| DN-098 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 22-Jul-2004 | Babcock & Wilcox Volund A/S |
| volund.eu | | | | | | | |
| DN-261 | | Registered | DOMAIN NAME | | | 07-Sep-2006 | Babcock & Wilcox Volund A/S |

 $^{{\}rm *\ McDermott\ Technology\ Inc.\ changed\ its\ name\ to\ Babcock\ \&\ Wilcox\ Technology,\ Inc\ April\ 20,2010}$

Page 122 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|------------------|----------|------------|---------------|--------------|--------|-------------|---|
| volund.fr | | | | | | | |
| DN-104 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 30-Jun-2004 | Babcock & Wilcox Volund A/S |
| volund.hk | | | | | | | |
| DN-265 | | Registered | DOMAIN NAME | | | 03-Jan-2008 | Babcock & Wilcox Volund A/S |
| volund.in | | | | | | | |
| DN-268 | | Registered | DOMAIN NAME | | | 30-Jun-2008 | Babcock & Wilcox Volund A/S |
| volund.net.cn | | | | | | | |
| DN-266 | | Registered | DOMAIN NAME | | | 30-Jun-2008 | Babcock & Wilcox Volund A/S |
| volund.nu | | | | | | | |
| DN-103 | | Registered | DOMAIN NAME | 01-Jun-2000 | | 17-Aug-2004 | Babcock & Wilcox Volund A/S |
| volund.org.cn | | | | | | | |
| DN-267 | | Registered | DOMAIN NAME | | | 27-Jun-2008 | Babcock & Wilcox Volund A/S |
| volund.tw | | | | | | | |
| DN-269 | | Registered | DOMAIN NAME | | | 30-Jun-2008 | Babcock & Wilcox Volund A/S |
| volundsystems.as | ia | | | | | | |
| DN-280 | | Registered | DOMAIN NAME | | | 18-Nov-2008 | Babcock & Wilcox Volund A/S |
| volundsystems.co | m | | | | | | |
| DN-279 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| volundsystems.dk | ζ. | | | | | | |
| DN-283 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| vølundsystems.dk | K | | | | | | |
| DN-278 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| volundsystems.eu | l | D 1 . 1 | DOM AND LANGE | | | 07.11 2000 | D. 1. 0. W. 1. 1. 1. 1. (G |
| DN-281 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| volundsystems.ne | et | D 1 . 1 | DOM AND LANGE | | | 07.11 2000 | D. 1. 0. W. 1. 1. 1. 1. (G |
| DN-282 | | Registered | DOMAIN NAME | | | 07-Nov-2008 | Babcock & Wilcox Volund A/S |
| V-TEMP | | D 1' | 77/579266 | 25 F.1. 2000 | | | DI I O WIL D. C. C. |
| TM-760 | | Pending | 77/678266 | 25-Feb-2009 | | | Babcock & Wilcox Power Generation Group, Inc. |
| TM-760 | 1 | Pending | 77/678285 | 25-Feb-2009 | | | Babcock & Wilcox Power Generation Group, Inc. |

 $^{{}^{*}\;\;}McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,\,2010$

Page 123 of 23

| CASE NO | SUB | STATUS | APP NO | FILE DATE | REG NO | REG DATE | OWNER |
|-------------------|-------|------------|-------------|-------------|---------|-------------|---|
| WEARESISTOR | (Word | Mark) | | | | | |
| TM-461 | | Registered | 74/611124 | 14-Dec-1994 | 2169816 | 30-Jun-1998 | Babcock & Wilcox Power Generation Group, Inc. |
| westernprecip.biz | | | | | | | |
| DN-046 | | Registered | DOMAIN NAME | 16-Mar-1996 | | 08-Nov-2001 | Diamond Power International, Inc. |
| westernprecip.com | n | | | | | | |
| DN-019 | | Registered | DOMAIN NAME | 02-Nov-1999 | | 02-Nov-1999 | Diamond Power International, Inc. |
| westernprecip.net | | | | | | | |
| DN-035 | | Registered | DOMAIN NAME | 04-Dec-2000 | | 04-Dec-2000 | Diamond Power International, Inc. |
| westernprecip.org | ; | | | | | | |
| DN-036 | | Registered | DOMAIN NAME | 04-Dec-2000 | | 04-Dec-2000 | Diamond Power International, Inc. |
| WINDAC | | | | | | | |
| TM-783 | | Registered | 75/658874 | 12-Mar-1999 | 2335456 | 28-Mar-2000 | Babcock & Wilcox Power Generation Group, Inc. |
| WINRAP | | | | | | | |
| TM-785 | | Registered | 76/176810 | 07-Dec-2000 | 2571146 | 21-May-2002 | Babcock & Wilcox Power Generation Group, Inc. |
| XCL-S (Word Ma | ırk) | | | | | | |
| TM-527 | | Registered | 75/175105 | 05-Sep-1996 | 2175559 | 21-Jul-1998 | Babcock & Wilcox Power Generation Group, Inc. |

 $^{{}^{*}\;\;}McDermott\;Technology\;Inc.\;changed\;its\;name\;to\;Babcock\;\&\;Wilcox\;Technology,Inc\;April\;20,2010$

Page 124 of 23

SCHEDULE 3.10 TO PLEDGE AND SECURITY AGREEMENT

COMMERCIAL TORT CLAIMS

The Babcock & Wilcox Company v. McGriff Seibels & Williams, Inc., McGriff Seibels & Williams of Texas, Inc. and XYZ Insurance Company(ies), U.S.D.C. Eastern District of Louisiana, Civil action no. 05-4202, consolidated with Civil Action 06-223, Section J, Judge Carl Barbier, Magistrate Sally Shushan. Filed September 2, 2005.—Damages for declaratory judgment, negligence, breach of contract and breach of fiduciary duty in connection with excess insurance coverage for the Citgo litigation based on alleged untimely notice from broker. The plaintiff in this action is Babcock & Wilcox Investment Company (formerly, The Babcock & Wilcox Company).

SECURITIES PURCHASE AGREEMENT

among

USEC INC.

TOSHIBA CORPORATION

and

BABCOCK & WILCOX INVESTMENT COMPANY

as of

May 25, 2010

1. PURCHASE AND SALE OF STOCK

- 1.1 Certificates of Designation
- 1.2 Phased Sale and Issuance of Series B-1 12.75% Preferred Stock and Series B-2 11.5% Preferred Stock
- 1.3 Closing

2. REPRESENTATIONS AND WARRANTIES OF THE COMPANY

- 2.1 Organization, Good Standing and Qualification
- 2.2 Financial Statements
- 2.3 Corporate Power; Authorization; Enforceable Agreement
- 2.4 Indebtedness; Agreements
- 2.5 Litigation
- 2.6 Title
- 2.7 Taxes
- 2.8 Subsidiaries
- 2.9 Governmental Consents
- 2.10 Permits and Licenses
- 2.11 ERISA
- 2.12 Valid Issuance of Preferred Stock and Common Stock
- 2.13 Capitalization
- 2.14 Investment Company Act
- 2.15 Compliance with Other Instruments
- 2.16 Environmental Matters
- 2.17 Compliance with Laws
- 2.18 No Material Adverse Effect
- 2.19 Registration Rights; Voting Rights
- 2.20 Reports
- 2.21 Brokers and Finders
- 2.22 Intellectual Property
- 2.23 Foreign Corrupt Practices

2.24 Insurance

3. REPRESENTATIONS AND WARRANTIES OF THE INVESTORS

- 3.1 Private Placement
- 3.2 Organization
- 3.3 Power and Authority
- 3.4 Authorization; Enforceability
- 3.5 No Default or Violation
- 3.6 Financial Capability
- 3.7 Person

4. CONDITIONS TO THE FIRST CLOSING

- 4.1 Conditions to the Obligations of Each Party
- 4.2 Conditions to the Obligations of Each Investor
- 4.3 Conditions to the Obligations of the Company

5. CONDITIONS TO THE SECOND CLOSING

- 5.1 Conditions to the Obligations of Each Party.
- 5.2 Conditions to the Obligations of Each Investor
- 5.3 Conditions to the Obligations of the Company

6. CONDITIONS TO THE THIRD CLOSING.

- 6.1 Conditions to the Obligations of Each Party
- 6.2 Conditions to the Obligations of Each Investor
- 6.3 Conditions to the Obligations of the Company

COVENANTS

- 7.1 State Securities Laws
- 7.2 Submittal of Update to Loan Guarantee Application; Approval of Regulatory Bodies
- 7.3 Negation Plans
- 7.4 Antitrust
- 7.5 Other Government Approvals
- 7.6 Negative Covenants Prior to First Closing
- 7.7 Use of Proceeds
- 7.8 Reservation of Common Stock; Issuance of Shares of Common Stock

- 7.9 Filing of Certificates of Designation
- 7.10 Special Approval Rights
- 7.11 Stockholder Meeting; Proxy Materials
- 7.12 [Reserved]
- 7.13 Foreign Ownership Provisions
- 7.14 Tax Matters
- 7.15 Disclosure Updates
- 7.16 Investor Directors
- 7.17 ACM
- 8. [RESERVED]
- 9. RESTRICTIONS ON TRANSFER
 - 9.1 Transfer of Preferred Stock and Warrants
 - 9.2 Orderly Sale of Common Stock
 - 9.3 No Transfer to Competitors; Block Transfers
 - 9.4 Restriction on Hedging, Etc.
- 10. TERMINATION
 - 10.1 Termination of Agreement Prior to First Closing
 - 10.2 Termination of Agreement Prior to Second Closing
 - 10.3 Termination of Agreement Prior to Third Closing
- 11. INDEMNIFICATION
 - 11.1 Survival
 - 11.2 Company Indemnified Parties
 - 11.3 Investor Indemnified Parties
 - 11.4 Waiver
- 12. PUBLICITY
- 13. MISCELLANEOUS
 - 13.1 Governing Law
 - 13.2 Dispute Resolution
 - 13.3 Enforcement of Agreement
 - 13.4 Successors and Assigns
 - 13.5 No Third Party Beneficiaries

- 13.6 No Personal Liability of Directors, Officers, Owners, Etc.
- 13.7 Entire Agreement
- 13.8 Notices, Etc.
- 13.9 Delays or Omissions
- 13.10 Expenses
- 13.11 Amendments and Waivers
- 13.12 Rights of the Parties
- 13.13 Counterparts
- 13.14 Severability
- 13.15 Titles and Subtitles; Interpretation
- 13.16 Several Liability of Investors
- 13.17 Limitation on Damages

SCHEDULES AND EXHIBITS

| Schedule A | Schedule of Investors |
|------------|------------------------|
| Schedule B | Regulatory Approvals |
| Schedule C | List of Broker Dealers |

Exhibit A Definitions

Exhibit B Form of Series B-1 Certificate of Designation Exhibit C Form of Series B-2 Certificate of Designation Exhibit D Form of Series C Certificate of Designation

Exhibit E Form of Warrant

Exhibit F

Form of Investor Rights Agreement
Form of Foreign Ownership Restrictions Determination Exhibit G

Form of Foreign Ownership Result
Form of First Closing Opinions
Form of Second Closing Opinions
Form of Press Release Exhibit H Exhibit I Exhibit J Exhibit K

Exhibit L Form of Charter Amendment

SECURITIES PURCHASE AGREEMENT

This Securities Purchase Agreement (this "Agreement") is made as of the 25th day of May, 2010, among USEC INC., a Delaware corporation (the "Company"), and TOSHIBA CORPORATION, a corporation organized under the laws of Japan ("Toshiba"), and BABCOCK & WILCOX INVESTMENT COMPANY, a Delaware corporation ("B&W") (each, an "Investor" and, collectively, the "Investors"; the Investors together with any assignee or transferee of the Series B-1 12.75% Preferred Stock, Series B-2 11.5% Preferred Stock or Warrants (each as defined below) in accordance with the terms hereof, the "Holders" and the Investors together with the Company, the "Parties"). Capitalized terms used herein, but not otherwise defined, have the meanings set forth in Exhibit A hereto.

WITNESSETH:

WHEREAS, the Company wants to sell, and the Investors want to buy, shares of the Company's Series B-1 12.75% Preferred Stock, Series B-2 11.5% Preferred Stock and Warrants, on the terms and conditions contained herein;

WHEREAS, in connection with such sale and purchase, the Company is willing to make certain representations and warranties and to agree to observe certain covenants set forth herein for the benefit of the Investors, and the Investors will rely on such representations, warranties and covenants as a material inducement to their purchase of the Series B-1 12.75% Preferred Stock, Series B-2 11.5% Preferred Stock and Warrants; and

WHEREAS, in connection with such sale and purchase, the Investors are willing to make certain representations and warranties and to agree to observe certain covenants set forth herein for the benefit of the Company, and the Company will rely on such representations, warranties and covenants as a material inducement to its sale of the Series B-1 12.75% Preferred Stock, Series B-2 11.5% Preferred Stock and Warrants.

NOW THEREFORE, in consideration of the premises and of the respective representations, warranties, covenants and conditions contained herein, the Parties agree as follows:

1. Purchase and Sale of Stock.

1.1 Certificates of Designation. The Company shall adopt and file with the Secretary of State of the State of Delaware, on or before the First Closing (as defined below in Section 1.3(a)), the Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock in the form attached hereto as Exhibit B (the "Series B-1 Certificate of Designation"), the Certificate of Designation of Series B-2 11.5% Convertible Preferred Stock in the form attached hereto as Exhibit C (the "Series B-2 Certificate of Designation") and the Certificate of Designation of Series C Participating Preferred Stock in the form attached hereto as Exhibit D (the "Series C Certificate of Designation") and together with the Series B-1 Certificate of Designation and the Series B-2 Certificate of Designation, the "Certificates of Designation").

1.2 Phased Sale and Issuance of Series B-1 12.75% Preferred Stock and Series B-2 11.5% Preferred Stock.

(a) Subject to the terms and conditions of this Agreement, each Investor agrees, severally and not jointly, to purchase at the First Closing, and the Company agrees to sell and issue to each Investor at the First Closing, that number of shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share (the "Series B-1 12.75% Preferred Stock"), at a purchase price of \$1,000.00 per share, and Warrants to purchase Class B Common Stock or Series C Preferred Stock in the form attached hereto as Exhibit E, each as set forth opposite such Investor's name on Schedule A hereto. The Series B-1 12.75% Preferred Stock will have the rights, preferences, privileges and restrictions set forth in the Series B-1 Certificate of Designation.

(b) Subject to the terms and conditions of this Agreement, each Investor agrees, severally and not jointly, to purchase at the Second Closing, and the Company agrees to sell and issue to each Investor at the Second Closing, that number of shares of the Series B-1 12.75% Preferred Stock, at a purchase price of \$1,000.00 per share, set forth opposite such Investor's name on Schedule A hereto.

(c) Subject to the terms and conditions of this Agreement, each Investor agrees, severally and not jointly, to purchase at the Third Closing, and the Company agrees to sell and issue to each Investor at the Third Closing, that number of shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share (the "Series B-2 11.5% Preferred Stock"), at a purchase price of \$1,000.00 per share, and Warrants (such Warrants together with the Warrants referenced in Section 1.2(a), the "Warrants") to purchase Class B Common Stock, par value \$0.10 per share, of the Company (the "Class B Common Stock"), to be authorized by the Company pursuant to the Charter Amendment as provided herein, each as set forth opposite such Investor's name on Schedule A hereto. The Series B-2 11.5% Preferred Stock will have the rights, preferences, privileges and restrictions set forth in the Series B-2 Certificate of Designation. The Class B Common Stock will have the rights, preferences, privileges and restrictions set forth in the Charter Amendment.

1.3 Closing

- (a) First Closing.
- (i) The consummation of the purchase and sale of the Series B-1 12.75% Preferred Stock and Warrants in accordance with Section 1.2(a) and the other transactions contemplated hereby at the first closing hereunder (the "First Closing") shall take place at the offices of Latham & Watkins LLP, 555 Eleventh Street, NW, Washington, D.C. 20004, at 10:00 a.m. New York City time, as promptly as practicable (but no more than three (3) Business Days) following the first date on which all conditions set forth in Section 4 hereof have been satisfied or waived (other than those conditions that by their nature are to be satisfied by actions taken at the First Closing), or at such other time and place as the Company and the Investors shall mutually agree.
 - (ii) At the First Closing, the Company shall deliver to each Investor:
 - (A) a certificate or certificates representing the shares of Series B-1 12.75% Preferred Stock and Warrants to be issued to such Investor at the First Closing, as set forth in Section 1.2(a) of this Agreement,
 - (B) each Transaction Document to be executed by the Company other than this Agreement, the Strategic Relationship Agreement and Transaction Documents to be executed at the Second Closing or the Third Closing, executed by the Company, and
 - (C) such other documents and instruments to be delivered by the Company to the Investors at the First Closing pursuant to Section 4.
 - (iii) At the First Closing, each Investor shall deliver to the Company:
 - (A) the purchase price for the shares of Series B-1 12.75% Preferred Stock and Warrants issued to such Investor at the First Closing, by wire transfer of immediately available funds or such other form of payment as may be approved by the Company;
 - (B) each Transaction Document to be executed by such Investor other than this Agreement, the Strategic Relationship Agreement and Transaction Documents to be executed at the Second Closing or the Third Closing, executed by such Investor; and
 - (C) such other documents and instruments to be delivered by such Investor to the Company at the First Closing pursuant to Section 4.

(b) Second Closing.

(i) The consummation of the purchase and sale of the Series B-1 12.75% Preferred Stock in accordance with Section 1.2(b) and the other transactions contemplated hereby at the second closing hereunder (the "Second Closing") shall take place at the offices of Latham & Watkins LLP, 555 Eleventh Street, NW, Washington, D.C. 20004, at 10:00 a.m. New York City time, as promptly as practicable (but no more than three (3) Business Days) following the first date on which all conditions set forth in Section 5 hereof have been satisfied or waived (other than those conditions that by their nature are to be satisfied by actions taken at the Second Closing), or at such other time and place as the Company and the Investors shall mutually agree.

(ii) At the Second Closing, the Company shall deliver to each Investor (A) a certificate or certificates representing the shares of Series B-1 12.75% Preferred Stock to be issued to such Investor at the Second Closing, as set forth in Section 1.2(b) of this Agreement, and (B) such other documents and instruments to be delivered by the Company to the Investors at the Second Closing pursuant to Section 5.

(iii) At the Second Closing, each Investor shall pay the purchase price for the shares of Series B-1 12.75% Preferred Stock issued to such Investor at the Second Closing, by wire transfer of immediately available funds or such other form of payment as may be approved by the Company, and shall deliver to the Company such other documents and instruments to be delivered by such Investor to the Company at the Second Closing pursuant to Section 5.

(c) Third Closing.

- (i) The consummation of the purchase and sale of the Series B-2 11.5% Preferred Stock and Warrants in accordance with Section 1.2(c) and the other transactions contemplated hereby at the third closing hereunder (the "Third Closing") shall take place at the offices of Latham & Watkins LLP, 555 Eleventh Street, NW, Washington, D.C. 20004, at 10:00 a.m. New York City time, as promptly as practicable (but no more than three (3) Business Days) following the first date on which all conditions set forth in Section 6 hereof have been satisfied or waived (other than those conditions that by their nature are to be satisfied by actions taken at the Third Closing), or at such other time and place as the Company and the Investors shall mutually agree.
- (ii) At the Third Closing, the Company shall deliver to each Investor (A) a certificate or certificates representing the shares of Series B-2 11.5% Preferred Stock and Warrants to be issued to such Investor at the Third Closing, each as set forth in Section 1.2(c) of this Agreement, and (B) such other documents and instruments to be delivered by the Company to the Investors at the Third Closing pursuant to Section 6.
- (iii) At the Third Closing, each Investor shall pay the purchase price for the shares of Series B-2 11.5% Preferred Stock and Warrants issued to such Investor at the Third Closing, by wire transfer of immediately available funds or such other form of payment as may be approved by the Company, and shall deliver to the Company such other documents and instruments to be delivered by such Investor to the Company at the Third Closing pursuant to Section 6.
- 2. Representations and Warranties of the Company. The Company hereby represents and warrants to the Investors as of the date hereof that, except (a) as otherwise disclosed or incorporated by reference in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2009 or its other reports and forms filed or furnished (and publicly available) by the Company with the Securities and Exchange Commission (the "Commission") under the Securities Exchange Act of 1934, as amended (the "Exchange Act") after December 31, 2009 (excluding disclosures of risks included in any forward-looking statement disclaimers or other statements that are similarly nonspecific and are predictive and forward-looking in nature) (all such reports and forms covered by this clause (a) collectively, the "SEC Reports") and before the date of this Agreement, and (b) as set forth in the disclosure letter dated as of the date hereof provided to the Investors separately, specifically identifying the relevant subparagraph(s) hereof (the "Disclosure Letter") (provided, that disclosure in any subparagraph of such Disclosure Letter shall apply to any section or subparagraph hereof to the extent it is reasonably apparent that such disclosure is relevant to such section or subparagraph of this Agreement):
- 2.1 <u>Organization, Good Standing and Qualification</u>. Each of the Company and its Subsidiaries is duly organized, validly existing and in good standing under the laws of the state of its incorporation; has all corporate power and authority to own its properties and conduct its business as presently conducted; and is duly qualified to do business and in good standing in each and every state in the United States of America where its business requires such qualification, except where failure to qualify would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect. True and accurate copies of the Company's Certificate of Incorporation, as amended (the "Certificate of Incorporation"), and Amended and Restated Bylaws, as amended (the "Bylaws"), each as currently in effect, have been made available to the Investors.

2.2 Financial Statements.

(a) The financial statements of the Company and its Subsidiaries on a consolidated basis

for each of the periods included in the SEC Reports fairly present in all material respects, in accordance with Generally Accepted Accounting Principles consistently applied, the financial condition, results of operations and cash flows of the Company and its Subsidiaries as of the dates and for the periods indicated (subject, in the case of unaudited quarterly statements, to normal year-end adjustments and presentation of footnotes).

(b) The Company and its Subsidiaries do not have any liabilities or obligations that would be required under Generally Accepted Accounting Principles, as in effect on the date of this Agreement, to be reflected on a consolidated balance sheet of the Company, other than liabilities or obligations (i) reflected on, reserved against, or disclosed in the notes to, the Company's consolidated balance sheet included in the Company's Annual Report on Form 10-K for the Fiscal Year ended December 31, 2009, as updated with the Company's Quarterly Report on Form 10-Q for the fiscal quarter ended March 31, 2010, (ii) that were incurred in the ordinary course of business and would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, or (iii) that were not incurred in the ordinary course of business and are not, individually or in the aggregate, material to the Company.

2.3 Corporate Power; Authorization; Enforceable Agreement.

(a) The Company has all corporate power and authority to execute and deliver each Transaction Document to which it is a party, and to perform its obligations thereunder. Other than the Stockholder Approvals and the filing of the Certificates of Designation and the Charter Amendment with the Secretary of State of the State of Delaware, all corporate action on the part of the Company, its officers, directors and stockholders necessary for the due authorization, valid execution and delivery of each Transaction Document, the performance of all obligations of the Company thereunder, and the authorization, issuance (or reservation for issuance), sale and delivery of the Preferred Stock and Warrants being sold hereunder (and the shares of Preferred Stock issuable in respect of dividends thereon from time to time in accordance with the Certificates of Designation) and the Common Stock issuable upon conversion of the Preferred Stock and Series B Common Stock and exercise of Warrants, has been taken. The Company has duly and validly executed and delivered this Agreement and the Strategic Relationship Agreement to which the Company will be a party will be duly and validly executed and delivered by the Company. This Agreement and the Strategic Relationship Agreement are, and each other Transaction Document to which the Company will be a party (when executed and delivered by the Company) will be, assuming due authorization, execution and delivery by the Investors, valid and legally binding obligations of the Company, enforceable in accordance with their respective terms, subject to: (i) laws limiting the availability of specific performance, injunctive relief and other equitable remedies; (ii) bankruptcy, insolvency, reorganization, moratorium or other similar laws now or hereafter in effect generally relating to or affecting creditors' rights generally; and (iii) public policy limitations on the enforceability of indemnification provisions (clause (i) – (iii) collectively, the "Enforceability Exceptions"). The sale of the

(b) On or prior to the date hereof, the Board has duly adopted resolutions (i) approving each Transaction Document and the Transactions, including the issuance of Preferred Stock and Common Stock hereunder; (ii) approving the Foreign Ownership Restrictions Determination; (iii) adopting the Certificates of Designation; (iv) declaring the Charter Amendment advisable; (v) directing that the adoption of the Charter Amendment and the authority of the Company to issue shares of Common Stock be submitted to the Company's stockholders for approval at a meeting of the Company's stockholders; and (vi) recommending that the Company's stockholders adopt the Charter Amendment and approve the Company's issuance of Common Stock and, as of the date hereof, such resolutions have not been rescinded, modified or withdrawn in any way. Other than the provisions set forth in the Certificates of Designation, the Foreign Ownership Restrictions Determination or, following the Stockholder Approvals, the Charter Amendment, no provision of the Certificate of Incorporation or the Bylaws of the Company would, directly or indirectly, restrict or impair the ability of the Investors to vote, or otherwise to exercise the rights of a stockholder with respect to, the Preferred Stock (or any shares of Common Stock issuable upon conversion of the Preferred Stock).

(c) Prior to the First Closing, the Company shall have taken all actions necessary to: (i) render the Company Rights Agreement inapplicable to this Agreement and the transactions contemplated by this Agreement, and (ii) ensure that in connection with the transactions contemplated by this Agreement (A) none of the Investors is or will be an "Acquiring Person" as a result of the transactions contemplated by this Agreement (as defined in the Company Rights Agreement) and (B) none of a "Stock Acquisition Date," a "Distribution Date," or a "Section 11(a)(ii) Event" (as such terms are defined in the Company Rights Agreement) occurs, in each case of clauses (A) and (B), by reason of the execution of this Agreement or the consummation of the transactions contemplated by this Agreement.

(d) The Board has approved the acquisition of securities pursuant to the Transaction Documents for purposes of Section 203 of the Delaware General Corporation Law.

- 2.4 <u>Indebtedness</u>; <u>Agreements</u>. Neither the Company nor any of its Subsidiaries is, immediately prior to this Agreement, or will be, at the time of each Closing after giving effect thereto, in default in the payment of any Indebtedness or in default under any agreement relating to its material Indebtedness or under any other material agreement (other than the 2002 Agreement, representations as to which are instead set forth in <u>Section 2.10(e)</u>) or mortgage, deed of trust, security agreement or lease to which it is a party, except for defaults that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect
- 2.5 <u>Litigation</u>. There is no action, suit, claim or proceeding pending or, to the Knowledge of the Company, threatened in writing against, nor any outstanding judgment, order or decree against, the Company, any of its Subsidiaries, any director or officer of the Company or its Subsidiaries (in their capacity as a director or officer) or any Property or assets of the Company or its Subsidiaries before or by any Governmental Authority which, individually or in the aggregate have, or would reasonably be expected to have, a Material Adverse Effect. Neither the Company nor any of its Subsidiaries is in default with respect to any judgment, order or decree of any Governmental Authority with respect to the operation of its business or otherwise, except as, individually or in the aggregate, would not reasonably be expected to have a Material Adverse Effect.
- 2.6 <u>Title</u>. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect: each of the Company and its Subsidiaries has good and marketable title to, or valid leasehold interests in, its Property that is real property, and good and valid title to, or valid leasehold interests in, all of its other Property (other than negligible assets not material to the operations of the Company or any of its Subsidiaries), in each case free and clear of all Liens except for Incidental Liens.

2.7 Taxes

- (a) Each of the Company and its Subsidiaries have timely and properly filed all tax returns required to have been filed and timely paid and deposited all taxes (including all related assessments, interest, fines and penalties) required to have been paid and deposited, except (i) those tax returns for which extensions have been obtained; (ii) where such failure to timely pay or deposit or timely and properly file would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect; and (iii) for those taxes which are being contested in good faith and by appropriate proceedings and in respect of which adequate reserves with respect thereto are maintained in accordance with Generally Accepted Accounting Principles.
- (b) Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, there are no pending audits, examinations, adjustments, assessments or administrative or judicial proceedings of or with respect to any income tax returns of the Company or any of its Subsidiaries. The federal income tax returns of the Company and its Subsidiaries have been examined by the Internal Revenue Service for all periods through and including 2005. Since July 1, 2005, the Company has not been at any time a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code. Since January 1, 2008, the Company has not distributed stock of another Person, or had its stock distributed by another Person, in a transaction that was intended to be governed in whole or in part by Section 355 of the Code.
 - 2.8 Subsidiaries. As of the date hereof, the Company has no Subsidiaries other than as listed in the SEC Reports.

2.9 Governmental Consents.

(a) No Permit from, or registration, qualification, declaration or filing with, any federal, state, local or other Governmental Authority (other than a Regulatory Body, representations as to which are instead set forth

in Section 2.9(b)) on the part of the Company is necessary or required in connection with the offer, sale or issuance of the Preferred Stock or the Warrants (and the Common Stock issuable upon conversion of the Preferred Stock and exercise of the Warrants), the execution, delivery or performance of the Transaction Documents or the consummation of the Transactions, except for the following: (i) the filing of the Certificates of Designation with the office of the Secretary of State of the State of Delaware, which with respect to the Series B-1 Certificate of Designation, will be filed by the Company prior to the First Closing; (ii) following the Stockholder Approvals, if any, the filing of the Charter Amendment with the office of the Secretary of State of the State of Delaware, which, together with the Series B-2 Certificate of Designation, will be filed by the Company prior to the Third Closing; (iii) the compliance with applicable state securities laws, which compliance will have occurred within the appropriate time periods therefor; (iv) to the extent applicable to either the Second Closing or the Third Closing, the compliance with the applicable requirements of the Hart-Scott-Rodino-Antitrust Improvements Act of 1976 (the "HSR Act"); (v) the filing with the Commission of such reports under the Exchange Act as may be required in connection with this Agreement and the transactions contemplated by this Agreement; and (vi) the filing with the Commistion of such reports under the Exchange Act as may be warranted under FINSA and applicable regulations, as amended, in connection with this Agreement and the transactions contemplated by this Agreement and the transactions contemplated by this Agreement and the transactions contemplated by this Agreement and the repistration requirements of Section 5 of the Securities Act of 1933, as amended (the "Securities Act"), and all applicable state securities laws, and neither the Company nor any authorized agent acting on its behalf will take any action hereafter that would ca

(b) Other than the Regulatory Approvals set forth on Schedule B (and the implementation of any Negation Plans as described in Section 7.3), neither the execution, delivery and performance of the Transaction Documents or the consummation of the Transactions will require any Permit from, or registration, qualification, declaration or filing with, any Regulatory Body.

2.10 Permits and Licenses.

- (a) The Company and each of its Subsidiaries possess all Permits that are required to conduct its business, except for such Permits the absence of which would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (b) Subject to making the filings and obtaining the Permits set forth in Section 2.9, the consummation of the transactions contemplated by this Agreement, including the issuance and sale of Preferred Stock and Warrants (and the issuance of Common Stock upon the conversion of Preferred Stock or exercise of Warrants) shall not result in the termination, withdrawal, rescission or material modification of any Permit issued by any Governmental Authority that is necessary for the operation of the Company's (and its Subsidiaries') business as presently conducted, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(c)

(i) As of the date hereof and as of the First Closing, to the Knowledge of the Company, no action or proceeding has been initiated by DOE under or with respect to Article 3 of the 2002 Agreement that has resulted, or could reasonably be expected to result, in (A) a determination by the Assistant Secretary for Nuclear Energy or Secretary of Energy, substantially to the effect that (x) the Company has not met a milestone set out in the 2002 Agreement, (y) such failure to meet such milestone has had or will have a material adverse impact on the Company's ability to begin commercial operations at the ACP on schedule, and (z) such failure was within the Company's control or the result of the Company's fault or negligence, or (B) a recommendation or determination by the Assistant Secretary for Nuclear Energy or Secretary of Energy to exercise any remedy available under Article 3 of the 2002 Agreement.

(ii) As of the Second Closing and the Third Closing, to the Knowledge of the Company, no action or proceeding has been initiated by DOE under or with respect to Article 3 of the 2002 Agreement that has resulted, or could reasonably be expected to result, in (A) a determination by the Assistant Secretary for Nuclear Energy or Secretary of Energy, substantially to the effect that (x) the Company has not met a milestone set out in the 2002 Agreement, (y) such failure to meet such milestone has had or will have a material adverse impact on the Company's ability to begin commercial operations at the ACP on schedule, and (z) such failure was within the Company's control or the result of the Company's fault or negligence, and (B) a recommendation or determination by the Assistant Secretary for Nuclear Energy or Secretary of Energy to exercise any remedy available under Article 3 of the 2002 Agreement, except as such remedy would not, individually or in the aggregate, result in a Material Adverse Effect.

- 2.11 ERISA. No Reportable Event (as defined in Section 4043(c) of ERISA but excluding those events as to which the 30-day notice period is waived by applicable regulations as in effect on the date of this Agreement, without regard to any subsequent elimination of such waiver period) has occurred with respect to any Plan. Each Plan complies in all material respects with all applicable provisions of ERISA and the Code, and the Company and each of its Subsidiaries have filed all reports required by ERISA and the Code to be filed with any Governmental Authority with respect to each Plan. Each of the Company and its Subsidiaries and each ERISA Affiliate has fulfilled its obligations under the minimum funding standards of ERISA and the Code with respect to each Plan and is in compliance in all material respects with the presently applicable provisions of ERISA and the Code with respect to each Plan, and has not incurred any liability under Title IV of ERISA (a) to the Pension Benefit Guaranty Corporation other than a liability to the Pension Benefit Guaranty Corporation for premiums under Section 4007 of ERISA or (b) in respect of a Multiemployer Plan which has not been discharged in full when due. No event has occurred that constitutes grounds for a Plan to be terminated under circumstances which would cause the lien provided under Section 4068 of ERISA to attach to any Property of the Company or any of its Subsidiaries. No event has occurred that would cause the lien provided under Section 303 of ERISA or Section 430 of the Code to attach to any Property of the Company or any of its Subsidiaries. No Plan is, or is reasonably believed by the Company to be within one year, in "at-risk" status (as defined in Section 303(i)(4) of ERISA or Section 430(i)(4) of the Code). Neither the Company, any Subsidiary nor any ERISA Affiliate participates in, contributes to or is otherwise required to contribute to, any Multiemployer Plan that, to the Knowledge of the Code or Section 305 of ERISA. The consummation of the transaction
- 2.12 Valid Issuance of Preferred Stock and Common Stock. The Series B-1 12.75% Preferred Stock being purchased by the Investors hereunder at the First Closing and Second Closing, the Series B-2 11.5% Preferred Stock being purchased by the Investors hereunder at the Third Closing, and the Series C Preferred Stock and the Common Stock that may be issued on conversion of the Preferred Stock when issued, sold and delivered in accordance with the terms of this Agreement for the consideration expressed herein, will be duly and validly issued, fully paid and nonassessable and will be free of all Liens and restrictions on transfer other than restrictions under this Agreement, the Investor Rights Agreement, the Certificate of Incorporation, or the Certificates of Designation and under applicable state and religious reserved for issuance and, upon issuance in accordance with the terms of the Series B-1 Certificate of Designation or the Warrants, as applicable, will be duly and validly issued, fully paid and nonassessable and will be free of all Liens and restrictions on transfer other than restrictions on transfer under this Agreement, the Investor Rights Agreement or the Company's Certificate of Incorporation and under applicable state and federal laws.
- 2.13 <u>Capitalization</u>. As of the end of the most recently completed calendar month, the authorized capital stock of the Company consists of 250,000,000 shares of common stock, par value \$0.10 per share ("*Ordinary Common Stock*"), of which 114,134,710 shares of Ordinary Common Stock were issued and outstanding as of the end of the most recently completed calendar month, and 25,000,000 shares of preferred stock, par value \$1.00 per share, of which 200,000 shares are designated Series A Junior Participating Preferred Stock ("*Series A Preferred Stock*") and are reserved for issuance in connection with the rights issued pursuant to the Company's Rights Agreement, dated April 24, 2001 (the "*Company Rights Agreement*"). The Company has reserved an aggregate of 14,100,000 shares of Ordinary Common Stock for issuance to employees and consultants pursuant to the Company's 1999 Equity Incentive Plan, as amended, under which as of the end of the most recently completed calendar month, (a) 3,720,149 shares have been issued and are reflected in the currently outstanding Ordinary Common Stock; (b) options to purchase 2,985,959 shares are outstanding; (c) restricted stock units to purchase 355,280 shares are outstanding; and (d) no shares remain available for future grant. The Company has reserved an aggregate of 4,500,000 shares of Ordinary Common Stock plus the number of shares of Ordinary Common Stock, if any, underlying grants under the Company's 1999 Equity Incentive Plan that are forfeited, canceled, terminated or are settled in cash without the delivery of shares of Ordinary Common Stock on or after April 30, 2009 for issuance to employees and consultants pursuant to the Company's 2009 Equity Incentive Plan, under which as of the end of the most recently completed calendar month, (i) 1,871,048 shares have been issued and are reflected in the currently outstanding; and (iv) 2,009,646 shares plus the number of shares of Ordinary Common Stock, if any, underlying grants under the Company's 1999 Equity Incentive Plan that

reserved an aggregate of 2,750,000 shares of Ordinary Common Stock for sale to employees pursuant to the Company's 1999 Employee Stock Purchase Plan, as amended, under which as of the end of the most recently completed calendar month, (a) 2,693,102 shares have been issued and are reflected in the currently outstanding Ordinary Common Stock; and (b) 56,898 shares remain available for future sale. The Company has reserved an aggregate of 1,000,000 shares of Ordinary Common Stock for sale to employees pursuant to the Company's 2009 Employee Stock Purchase Plan, under which as of the end of the most recently completed calendar month, (a) no shares have been issued and are reflected in the currently outstanding Ordinary Common Stock; and (b) 1,000,000 shares remain available for future sale. As of the end of the most recently completed calendar month, the Company has reserved an aggregate of 48,093,000 shares of Ordinary Common Stock for issuance upon conversion of its 3.0% Convertible Senior Notes due 2014 (the "Convertible Notes"), none of which have been issued upon conversion of Convertible Notes, and solely upon the occurrence of the conditions described in the indenture governing the Convertible Notes the outstanding Convertible Notes are entitled to convert into 48,093,000 shares of Ordinary Common Stock (subject to adjustment as provided in the indenture governing the Convertible Notes). All issued and outstanding shares of Ordinary Common Stock have been duly authorized and validly issued and are fully paid and nonassessable. When issued, the Company will have reserved that number of shares of Common Stock sufficient for issuance upon conversion or redemption of the Preferred Stock, subject to the Share Issuance Limitation (as such term is defined in the Certificates of Designation), if any, or the exercise of the Warrants being issued and sold pursuant to this Agreement. Other than as provided in this Agreement (including this Section 2.13) and the Investor Rights Agreement, there are no other outstanding rights, options, warrants, preemptive rights, rights of first offer or similar rights for the purchase or acquisition from the Company of any securities of the Company, nor are there any commitments to issue or execute any such rights, options, warrants, preemptive rights or rights of first offer, nor are there any declared but unpaid dividends. Except as otherwise provided in this Agreement (including this Section 2.13), the Investor Rights Agreement or the Certificates of Designation, there are no outstanding rights or obligations of the Company to repurchase or redeem any of its securities. The respective rights, preferences, privileges and restrictions of the Series A Preferred Stock and the Ordinary Common Stock are as stated in the Certificates of Designation and the Company's Certificate of Incorporation. All outstanding securities have been issued in compliance with state and federal securities laws. Notwithstanding the foregoing, if at least five Business Days have not passed since the completion of the last calendar month, all references in this Section 2.13 to the "most recently completed calendar month" shall mean the completion of the month prior to the most recently completed calendar month.

- 2.14 <u>Investment Company Act</u>. Neither the Company nor any of its Subsidiaries is, or to the Knowledge of the Company ever has been, an investment company within the meaning of the Investment Company Act of 1940, as amended, or, directly or indirectly, controlled by or acting on behalf of any Person which is an investment company, within the meaning of said Act.
- 2.15 Compliance with Other Instruments. The Company is not in violation or default of any provision of its Certificate of Incorporation or Bylaws. The execution, delivery and performance of and compliance with the Transaction Documents by the Company, and the issuance and sale of the Preferred Stock and Warrants will not (a) result in any default or violation of the Company's Certificate of Incorporation (including, with respect to the First Closing and the Second Closing, the Series B-1 Certificate of Designation, and solely with respect to the Third Closing, the Certificates of Designation) or Bylaws, each as amended through the applicable Closing, (b) upon the receipt of the Regulatory Approvals and the filing of the Certificates of Designation and, following the Charter Amendment Approval, the Charter Amendment with the Secretary of State of the State of Delaware, result in any violation of applicable Law, (c) require the approval of any Regulatory Body, other than the Regulatory Approvals, (d) constitute or result in a default, or an event that with or without the passage of time or giving of notice would become a default, breach or violation of, or give rise to any rights of termination, amendment, acceleration or cancellation of, any material agreement or any agreement relating to its Indebtedness or under any mortgage, deed of trust, security agreement or lease to which it is a party or any other agreement, instrument or arrangement filed or required to be filed as an exhibit to the SEC Reports, or in any default or violation of any judgment, order or decree of any Governmental Authority, or (e) be in conflict with or constitute, with or without the passage of time or giving of notice, a default under any such provision, require any consent or waiver under any such provision, or result in the creation of any Lieu upon any of the properties or assets of the Company pursuant to any such provision, except in the case of clauses (b), (d) and (e), which would not, individually or in the aggregate, reasonably
- 2.16 Environmental Matters. No activity of the Company or any of its Subsidiaries requires any Environmental Permit which has not been obtained and which is not now in full force and effect, except to the extent failure to have any such Environmental Permit would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Company

and its Subsidiaries are and have been in compliance with all applicable Requirements of Environmental Law and Environmental Permits including applicable limitations, restrictions, conditions, standards, prohibitions, requirements, obligations, schedules and timetables contained in any applicable Requirement of Environmental Law or Environmental Permit, except where failure to be in such compliance would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. The Company and its Subsidiaries (a) including with respect to their Property are not subject to any (i) Environmental Claims or (ii) Environmental Liabilities, in either case arising from or based upon any act, omission, event, condition or circumstance occurring or existing on or prior to the date hereof or the date of any Closing, which, in either case, would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, and (b) have not received individually or collectively any written notice of any violation or alleged violation of any Requirements of Environmental Law or Environmental Permit or any Environmental Claim in connection with their respective Property which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

- 2.17 <u>Compliance with Laws</u>. The Company and its Subsidiaries have at all times since December 31, 2007 complied, and are currently in compliance, with any and all applicable federal, state, local, foreign or other laws, statutes, regulations, rules, ordinances, codes, conventions, directives, orders, judgments or other legal requirements (collectively, "*Laws*") of any Governmental Authority, except where such violation would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- 2.18 No Material Adverse Effect. Since the later of (a) March 31, 2010, and (b) the most recently completed Closing, no event or circumstance has occurred that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect.
- 2.19 <u>Registration Rights: Voting Rights</u>. Except as provided in the Investor Rights Agreement, (a) the Company has not granted or agreed to grant, and is not presently under any obligation to provide, any rights to register under the Securities Act any of its presently outstanding securities or any of its securities that may be issued subsequently, and (b) to the Company's Knowledge, no stockholder of the Company has entered into any agreement with respect to the voting of equity securities of the Company.

2.20 Reports.

- (a) Since December 31, 2008, the Company has timely filed all documents required to be filed with the Commission pursuant to Sections 13(a), 14(a) or 15(d) of the Exchange Act, except where the failure to so file would not reasonably be expected to materially and adversely affect an Investor's ability to complete a sale of Common Stock pursuant to an Orderly Sale Arrangement.
- (b) The SEC Reports, after giving effect to any amendments thereto filed prior to the date of the making of this representation, complied as to form in all material respects with the requirements of the Exchange Act, as applicable, and the rules and regulations of the Commission thereunder, in each case as in effect at such time, and, after giving effect to any amendments thereto filed prior to the date of the making of this representation, none of such documents contained an untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make such statements, in the light of the circumstances in which they were made, not misleading.
- 2.21 <u>Brokers and Finders</u>. Except for Credit Suisse, the fees and expenses of which will be paid by the Company, neither the Company nor any of its Subsidiaries nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any financial advisory fees, brokerage fees, commissions or finder's fees, and no broker or finder has acted directly or indirectly for the Company or any of its Subsidiaries in connection with this Agreement or the transactions contemplated hereby.
- 2.22 Intellectual Property. Each of the Company and its Subsidiaries owns, or is licensed to use, all trademarks, trade names, copyrights, patents and other intellectual property material to its business, and the use thereof by the Company and its

Subsidiaries does not infringe upon the rights of any other Person, except for any such infringements that, individually or in the aggregate, would not reasonably be expected to result in a Material Adverse Effect.

- 2.23 Foreign Corrupt Practices. Since January 1, 2005, none of (i) the Company, (ii) any Subsidiary of the Company or (iii) to the Knowledge of the Company, any director, officer, agent, employee or other person acting on behalf of the Company or any Subsidiary, has violated or is in violation of any provision of the U.S. Foreign Corrupt Practices Act of 1977. To the Knowledge of the Company, the Company, its Subsidiaries and its Affiliates have not been and are not now under any administrative, civil or criminal investigation or indictment involving alleged violations of the Foreign Corrupt Practices Act or any other applicable anticorruption or antibribery laws.
- 2.24 <u>Insurance</u>. The Company and its Subsidiaries maintain insurance underwritten by insurers of recognized financial responsibility, of the types and in the amounts that the Company reasonably believes is adequate for its and its Subsidiaries' business, with such deductibles and exclusions as are customary for companies in the same or similar business, all of which insurance is in full force and effect.
- 3. <u>Representations and Warranties of the Investors</u>. Each Investor, severally and not jointly, hereby represents and warrants as of the date hereof and as of the date of each Closing (or such other date specified herein) as follows:

3.1 Private Placement.

- (a) Such Investor is (i) an "accredited investor" within the meaning of Rule 501 of Regulation D promulgated under the Securities Act; (ii) aware that the sale of the Preferred Stock and Warrants (collectively, including the Common Stock issuable upon conversion of the Preferred Stock and exercise of Warrants, the "Securities") to it is being made in reliance on a private placement exemption from registration under the Securities Act and (iii) acquiring the Securities for its own account.
- (b) Such Investor understands and agrees that the Securities are being offered in a transaction not involving any public offering within the meaning of the Securities Act, that such Securities have not been and, except as contemplated by the Investor Rights Agreement, will not be registered under the Securities Act and that such Securities may be offered, resold, pledged or otherwise transferred only (i) in a transaction not involving a public offering, (ii) pursuant to an exemption from registration under the Securities Act provided by Rule 144 thereunder (if available), (iii) pursuant to an effective registration statement under the Securities Act, or (iv) to the Company or one of its Subsidiaries, in each of cases (i) through (iv) in accordance with any applicable securities laws of any State of the United States and Section 9 of this Agreement, and that it will notify any subsequent purchaser of Securities from it of the resale restrictions referred to above, as applicable.
- (c) Such Investor understands that, unless sold pursuant to a registration statement that has been declared effective under the Securities Act or, to the reasonable satisfaction of the Company, in compliance with Rule 144 thereunder, the Company may require that the Securities bear a legend or other restriction substantially to the following effect (it being agreed that if the Securities are not certificated, other appropriate restrictions shall be implemented to give effect to the following):

"THIS SECURITY WAS ORIGINALLY ISSUED IN A TRANSACTION EXEMPT FROM REGISTRATION UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR AS MAY BE HELD BY A PERSON DEEMED AN "AFFILIATE" (AS DEFINED IN RULE 144 UNDER THE SECURITIES ACT) OF THE ISSUER OF THIS SECURITY, AND THIS SECURITY MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN APPLICABLE EXEMPTION THEREFROM. THE HOLDER OF THIS SECURITY AGREES FOR THE BENEFIT OF THE COMPANY THAT (A) THIS SECURITY MAY BE OFFERED, RESOLD, PLEDGED OR OTHERWISE TRANSFERRED, ONLY (I) IN A TRANSACTION NOT INVOLVING A PUBLIC OFFERING, (II) PURSUANT TO ANY OTHER

EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, INCLUDING RULE 144 UNDER THE SECURITIES ACT (IF AVAILABLE), (III) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR (IV) TO THE COMPANY OR ANY OF ITS SUBSIDIARIES, IN EACH OF CASES (I) THROUGH (IV) IN ACCORDANCE WITH ANY APPLICABLE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES, AND (B) THE HOLDER WILL NOTIFY ANY SUBSEQUENT PURCHASER OF THIS SECURITY FROM IT OF THE RESALE RESTRICTIONS REFERRED TO IN (A) ABOVE. THIS SECURITY MAY ONLY BE TRANSFERRED IN ACCORDANCE WITH THE TERMS OF THE CERTIFICATE OF DESIGNATION OF SERIES [] CONVERTIBLE PREFERRED STOCK OF USEC INC. (THE "COMPANY"), AS AMENDED.

THIS SECURITY IS SUBJECT TO THE RESTRICTIONS (INCLUDING THE VOTING AND TRANSFER RESTRICTIONS) SET FORTH IN ARTICLES FOURTH AND ELEVENTH OF USEC INC.'S CERTIFICATE OF INCORPORATION, AS AMENDED.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON TRANSFER, CONVERSION AND REDEMPTION) STATED IN, AND ARE TRANSFERABLE ONLY IN ACCORDANCE WITH, THE PROVISIONS OF SECTION 9 OF THE SECURITIES PURCHASE AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA CORPORATION ("TOSHIBA") AND BABCOCK & WILCOX INVESTMENT COMPANY ("B&W"), DATED AS OF MAY 25, 2010.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON THE DISPOSITION OF SECURITIES) STATED IN THE PROVISIONS OF SECTION 4.7 OF THE INVESTOR RIGHTS AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA AND B&W, DATED AS OF , 2010."

(d) Such Investor:

- (i) has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of its prospective investment in the Securities; and
 - (ii) has the ability to bear the economic risks of its prospective investment and can afford the complete loss of such investment.
- (e) Such Investor acknowledges that (i) it has conducted its own investigation of the Company and the terms of the Securities, (ii) it has had access to the Company's public filings with the Commission and to such financial and other information as it deems necessary to make its decision to purchase the Securities, and (iii) has been offered the opportunity to conduct such review and analysis of the business, assets, condition, operations and prospects of the Company and its Subsidiaries and to ask questions of the Company and received answers thereto, each as it deemed necessary in connection with the decision to purchase the Securities. Each Investor further acknowledges that it has had such opportunity to consult with its own counsel, financial and tax advisors and other professional advisers as it believes is sufficient for purposes of the purchase of the Securities. The foregoing, however, does not limit or modify the representations and warranties of the Company in Section 2 of this Agreement, the right of the Investors to rely thereon or any rights or remedies available to the Investors in this Agreement, the other Transaction Documents or otherwise.
- (f) Such Investor understands that the Company will rely upon the truth and accuracy of the foregoing representations, acknowledgements and agreements.
- (g) Except for the representations and warranties contained in Section 2 of this Agreement or in any other Transaction Document, each Investor acknowledges that neither the Company nor any Person on behalf of the Company makes, and such Investor has not relied upon, any other express or implied representation or warranty with respect to the Company or any of its Subsidiaries or with respect to any other information provided to the Investors in connection with the transactions contemplated by this Agreement.
 - 3.2 Organization. Such Investor has been duly organized and is validly existing as a corporation under the laws of its jurisdiction of incorporation.

- 3.3 <u>Power and Authority</u>. Such Investor has full right, power, authority and capacity to enter into each Transaction Document to which it is a party and to consummate the Transactions and has taken all necessary action to authorize the execution, delivery and performance of each Transaction Document to which it is a party.
- 3.4 <u>Authorization</u>; <u>Enforceability</u>. The execution, delivery and performance of each Transaction Document to which such Investor is a party has been duly authorized by all necessary action on the part of such Investor. Such Investor has duly and validly executed and delivered this Agreement and the Strategic Relationship Agreement. On the First Closing date, each other Transaction Document to which such Investor will be a party will be duly and validly executed and delivered by such Investor. This Agreement and the Strategic Relationship Agreement, and each other Transaction Document to which such Investor is or will be a party (when executed and delivered by such Investor), assuming due authorization, execution and delivery by each other party thereto, constitutes and will constitute valid and legally binding obligations of such Investor, enforceable in accordance with their respective terms, subject to the Enforceability Exceptions.
- 3.5 No Default or Violation. Such Investor is not in violation or default of any provision of its organizational documents. Such Investor's execution, delivery, and performance of and compliance with each Transaction Document to which it is a party, and such Investor's purchase of the Preferred Stock or Warrants (or upon the conversion of Preferred Stock or exercise of Warrants, the receipt of Common Stock) as contemplated by this Agreement will not (a) result in any default or violation of the certificate of incorporation and bylaws (or similar organizational documents), of such Investor, (b) result in any violation of applicable Law, (c) constitute or result in a default, or an event that with or without the passage of time or giving of notice would become a default, breach or violation of, or give rise to any rights of termination, amendment, acceleration or cancellation of, any material agreement or any agreement relating to its Indebtedness or under any mortgage, deed of trust, security agreement or lease or any other material agreement to which such Investor is a party or in any default or violation of any material judgment, order or decree of any Governmental Authority, or (d) be in conflict with or constitute, with or without the passage of time or giving of notice, a default under any such provision, require any consent or waiver under any such provision, or result in the creation of any mortgage, pledge, lien, encumbrance, or charge upon any of the properties or assets of such Investor pursuant to any such provision, or the suspension, revocation, impairment or forfeiture of any material permit, license, authorization or approval applicable to such Investor, its business or operations, or any of its assets or properties pursuant to any such provision, except in the case of clauses (b), (c) and (d), for any matters that would not, individually or in the aggregate, materially adversely affect such Investor's performance of its obligations under any Transaction Document.
- 3.6 <u>Financial Capability</u>. Such Investor currently has, and at each Closing will have, available and unconditional funds necessary to purchase the Preferred Stock and Warrants issued at such Closing on the terms and conditions contemplated by this Agreement.
- 3.7 Person. At or prior to the First Closing, (a) B&W represents that it will be a U.S. Person, and (b) Toshiba represents that it will have assigned its rights to purchase Preferred Stock and Warrants hereunder to a U.S. Person.

4. Conditions to the First Closing.

4.1 <u>Conditions to the Obligations of Each Party.</u> The obligations of each Party to consummate the transactions contemplated hereby at the First Closing are subject to the fulfillment or waiver by such Party (to the extent permitted by Law) on or before the First Closing of each of the following conditions:

- (a) No Law/Injunction. No federal or state court of competent jurisdiction or other Governmental Authority shall have enacted, issued, promulgated, enforced or entered any order, decree, judgment, injunction or other ruling (whether temporary, preliminary or permanent), in any case which is in effect and which prevents or prohibits the consummation of the transactions contemplated hereby at the First Closing.
 - (b) Regulatory Approvals. The Company shall have received all Regulatory Approvals set forth on Schedule B under the heading "First Closing".
- (c) <u>CFIUS/FINSA</u>. (i) The period of time for any applicable review process by CFIUS under FINSA (including, if applicable, any investigation commenced thereunder) shall have expired or been terminated, (ii) CFIUS shall have provided a written notice to the effect that its review of the transactions contemplated by this Agreement has been concluded and that a determination has been made that there are no issues of national security sufficient to warrant investigation under FINSA, or (iii) the President of the United States shall have made a determination under FINSA not to suspend or prohibit the transactions contemplated by this Agreement.
- (d) <u>Transaction Documents</u>. Each Transaction Document other than this Agreement, the Strategic Relationship Agreement and any other Transaction Document to be executed and delivered at the Second Closing or the Third Closing shall have been executed and delivered by each party thereto other than such Party.
- (e) <u>Listing of Ordinary Common Stock</u>. The shares of Ordinary Common Stock issuable upon the conversion of Series B-1 12.75% Preferred Stock issued at the First Closing or the exercise of Warrants received at the First Closing shall have been approved for listing on the NYSE, subject to official notice of issuance
- 4.2 <u>Conditions to the Obligations of Each Investor</u>. The obligations of each Investor to consummate the transactions contemplated hereby at the First Closing are subject to the fulfillment or waiver by such Investor (to the extent permitted by Law) on or before the First Closing of each of the following conditions:
- (a) Representations and Warranties. The representations and warranties of the Company contained in Section 2 shall have been true and correct as of the date hereof, except as would not result in a Material Adverse Effect. The Closing Representations made by the Company at the First Closing that constitute Designated Representations and the representations and warranties set forth in Section 2.20(b) shall be true and correct at the First Closing, after giving effect to the Signing Disclosures. The Closing Representations made by the Company at the First Closing that do not constitute Designated Representations or the representations and warranties set forth in Section 2.20(b) shall, after giving effect to the Signing Disclosures and the Updated Disclosures, be true and correct at the First Closing, except for such breaches as would not reasonably be expected to have a Material Adverse Effect. For the avoidance of doubt, any representations and warranties of the Company referenced in this Section 4.2(a) which address matters only as of a particular date need only be true and correct as of such date.
- (b) <u>Updated Disclosure</u>. No Updated Disclosure shall reflect a Company Effect that has had or is reasonably likely to have a Material Adverse Effect.
- (c) <u>Covenants</u>. The Company shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or before the First Closing.
- (d) Officer's Certificate. The Company shall have delivered to each Investor a certificate, dated the date of the First Closing and signed by its duly authorized officer, to the effect that the conditions set forth in Sections 4.2(a) and 4.2(c) have been satisfied.
- (e) <u>Solicitation/Reprogramming</u>. The 45-day notice period specified under the Consolidated Appropriations Act of 2008, Division C Energy and Water Development and Related Agencies, "Title 17 Innovative Technology Loan Guarantee Program" (Public Law 110-161 Dec. 26, 2007 (at 121 STAT. 1961)) shall have expired with respect to the notice provided by the United States Department of Energy, and any successor Governmental Authority thereto (the "*DOE*"), on March 25, 2010 of its intent to use up to \$2 billion in loan guarantee authority made available to DOE under the Revised Continuing Appropriations Resolution, 2007, as amended, for front-end nuclear fuel facilities under the solicitation for Federal Loan Guarantees for Front-End Nuclear Facilities issued by DOE in 2008 (solicitation number DE-PS01-08LG00003) (the "2008 Solicitation") (the "Reprogramming"), and no federal law shall have been enacted, and Congress shall not have taken any formal action, prohibiting such Reprogramming.

- (f) Loan Guarantee Commitment Availability. As of the date of the First Closing, there shall be at least \$2 billion available under the 2008 Solicitation or another solicitation for front-end nuclear fuel projects under the Loan Guarantee Program. For purposes of this Section 4.2(f), the amount of funds "available" under a solicitation shall equal the aggregate amount of loan guarantee authority allocated to such solicitation (and not subsequently reallocated from such solicitation) less the amount committed under any conditional commitment (as defined in 10 CFR 609.2) and, without duplication, with respect to a project, Loan Guarantee Agreements (as defined in 10 CFR 609.2, "Loan Guarantee Agreements") entered into by DOE under such solicitation.
- (g) ACM Operating Agreement. The operating agreement of American Centrifuge Manufacturing LLC ("ACM") by and between B&W (or an Affiliate of B&W) and the Company (or an Affiliate of the Company), substantially in the form of the draft dated April 1, 2010, shall have been executed by the parties thereto, and such parties shall have agreed upon a term sheet with terms and conditions, including pricing, for the long-term supply by ACM LLC of centrifuges and related equipment for the ACP.
- (h) <u>Series B-1 and Series C Certificates of Designation</u>. The Company shall have filed the Series B-1 Certificate of Designation and the Series C Certificate of Designation with the Secretary of State of the State of Delaware.
- (i) Foreign Ownership Restrictions Determination. The Foreign Ownership Restrictions Determination shall be in full force and effect, and the Board shall have taken no action contrary thereto.
- (j) <u>Election of Directors</u>. The Board shall have taken all actions necessary and appropriate to permit the two directors designated by the Investors pursuant to <u>Section 7.16</u> to be elected to the Board effective immediately upon the delivery to the Company of a written consent to such effect of the Holders of the outstanding Series B-1 12.75% Preferred Stock following the First Closing.
- (k) Opinions of Counsel. The Investors shall have received from the General Counsel of the Company, Latham & Watkins LLP, special counsel for the Company, and Morgan Lewis & Bockius LLP, regulatory counsel for the Company, opinions, dated as of the First Closing, in the forms attached hereto as or as to the respective matters set forth in Exhibit H.
- (1) Concurrent Investment. The Company shall have received proceeds from the sale of Series B-1 12.75% Preferred Stock and Warrants at the First Closing to the other Investor of not less than \$37,500,000.00.
 - (m) No Material Adverse Effect. Since the date of this Agreement, there shall not have occurred any Material Adverse Effect.
- 4.3 <u>Conditions to the Obligations of the Company</u>. The obligations of the Company to consummate the transactions contemplated hereby at the First Closing with respect to an Investor are subject to the fulfillment or waiver by the Company (to the extent permitted by Law) on or before the First Closing of each of the following conditions:
- (a) <u>Representations and Warranties</u>. The representations and warranties of such Investor contained in <u>Section 3</u> shall be true and correct in all material respects as of the date hereof and as of the date of the First Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date).
- (b) <u>Covenants</u>. Such Investor shall have performed or complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by it on or before the First Closing.
- (c) Officer's Certificate. Such Investor shall have delivered to the Company a certificate, dated the date of the First Closing and signed by a duly authorized officer of such Investor, to the effect that the conditions set forth in Sections 4.3(a) and 4.3(b) have been satisfied with respect to such Investor.
- (d) <u>Investment</u>. The Company shall have received proceeds from the sale of Series B-1 12.75% Preferred Stock and Warrants at the First Closing to the Investor(s) of not less than \$75,000,000.00.
 - (e) Opinions of Counsel. The Company shall have received from special counsel to such

Investor an opinion, dated as of the First Closing, as to the matters set forth in Exhibit H.

- 5. Conditions to the Second Closing
- 5.1 <u>Conditions to the Obligations of Each Party</u>. The obligations of each Party to consummate the transactions contemplated hereby at the Second Closing are subject to the fulfillment or waiver by such Party (to the extent permitted by Law) on or before the Second Closing of each of the following conditions:
 - (a) First Closing. The First Closing shall have occurred.
- (b) No Law/Injunction. No federal or state court of competent jurisdiction or other Governmental Authority shall have enacted, issued, promulgated, enforced or entered any order, decree, judgment, injunction or other ruling (whether temporary, preliminary or permanent), in any case which is in effect and which prevents or prohibits the consummation of the transactions contemplated hereby at the Second Closing.
- (c) <u>Regulatory Approvals</u>. The Company shall have received all Regulatory Approvals set forth on <u>Schedule B</u> under the heading "Second Closing".
- (d) <u>HSR Act Approval</u>. To the extent the transactions contemplated by the Second Closing are subject to the notification requirements under the HSR Act, the applicable waiting period, together with any extensions thereof, under the HSR Act shall have expired or been terminated.
- (e) <u>Listing of Ordinary Common Stock</u>. The shares of Ordinary Common Stock issuable upon the conversion of Series B-1 12.75% Preferred Stock issued at the Second Closing shall have been approved for listing on the NYSE, subject to official notice of issuance.
- 5.2 <u>Conditions to the Obligations of Each Investor</u>. The obligations of each Investor to consummate the transactions contemplated hereby at the Second Closing is subject to the fulfillment or waiver by such Investor (to the extent permitted by Law) on or before the Second Closing of each of the following conditions:
- (a) <u>Representations and Warranties</u>. The representations and warranties of the Company contained in <u>Section 2</u> shall have been true and correct as of the date hereof, except as would not result in a Material Adverse Effect. The Closing Representations that constitute Designated Representations made by the Company at the Second Closing shall be true and correct at the Second Closing, after giving effect to the Signing Disclosures. The Closing Representations made by the Company at the Second Closing that do not constitute Designated Representations, after giving effect to the Signing Disclosures and the Updated Disclosures shall be true and correct at the Second Closing, except for such breaches as would not reasonably be expected to have a Material Adverse Effect. For the avoidance of doubt, any representations and warranties of the Company referenced in this <u>Section 5.2(a)</u> which address matters only as of a particular date need only be true and correct as of such date.
- (b) <u>Updated Disclosure</u>. No Updated Disclosure shall reflect a Company Effect that has had or is reasonably likely to have a Material Adverse Effect.
- (c) <u>Covenants</u>. The Company shall have performed or complied in all material respects with all agreements and covenants required by the Transaction Documents to be performed or complied with by it on or before the Second Closing.
- (d) Officer's Certificate. The Company shall have delivered to each Investor a certificate, dated the date of the Second Closing and signed by its duly authorized officer, to the effect that the conditions set forth in Sections 5.2(a) and 5.2(c) have been satisfied.
 - (e) <u>DOE Conditional Commitment</u>. The Company shall have entered into the Conditional Commitment with DOE.
 - (f) Foreign Ownership Restrictions Determination. The Foreign Ownership Restrictions

Determination shall be in full force and effect, and the Board shall have taken no action contrary thereto.

- (g) Compliance with Strategic Plan. Other than due to a breach of the Transaction Documents by an Investor, since the First Closing, the Company shall have conducted the business of the Company in material compliance with the Strategic Plan, as the same may be from time to time amended or modified with the consent of the Investors, such consent not to be unreasonably withheld, conditioned or delayed.
- (h) Opinions of Counsel. The Investors shall have received from the General Counsel of the Company, Latham & Watkins LLP, special counsel for the Company, and Morgan Lewis & Bockius LLP, regulatory counsel for the Company, opinions, dated as of the Second Closing, as to the respective matters set forth in Exhibit I.
- (i) <u>Concurrent Investment</u>. The Company shall have received proceeds from the sale of Series B-1 12.75% Preferred Stock at the Second Closing to the other Investor of not less than \$25,000,000.00.
- (j) No Material Adverse Effect. Since the date of the First Closing there shall not have occurred any Material Adverse Effect (for the avoidance of doubt, taking into account applicable Company Effects arising since the date hereof but not giving rise to a Material Adverse Effect prior to the First Closing).
- 5.3 <u>Conditions to the Obligations of the Company</u>. The obligations of the Company to consummate the transactions contemplated hereby at the Second Closing with respect to an Investor are subject to the fulfillment or waiver by the Company (to the extent permitted by Law) on or before the Second Closing of each of the following conditions:
- (a) <u>Representations and Warranties</u>. The representations and warranties of such Investor contained in <u>Section 3</u> shall be true and correct in all material respects as of the date hereof and as of the date of the Second Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date).
- (b) <u>Covenants</u>. Such Investor shall have performed or complied in all material respects with all agreements and covenants required by each Transaction Document to be performed or complied with by such Investor on or before the Second Closing.
- (c) Officer's Certificates. Such Investor shall have delivered to the Company a certificate, dated the date of the Second Closing and signed by a duly authorized officer of such Investor, to the effect that the conditions set forth in Sections 5.3(a) and 5.3(b) have been satisfied with respect to such Investor.
- (d) <u>Investment</u>. The Company shall have received proceeds from the sale of Series B-1 12.75% Preferred Stock at the Second Closing to the Investor(s) of not less than \$50,000,000.00.
- (e) Opinions of Counsel. The Company shall have received from special counsel to such Investor an opinion, dated as of the Second Closing as to the matters set forth in Exhibit I.

6. Conditions to the Third Closing.

- 6.1 Conditions to the Obligations of Each Party. The obligations of each Party to consummate the transactions contemplated hereby at the Third Closing are subject to the fulfillment or waiver by such Party (to the extent permitted by Law) on or before the Third Closing of each of the following conditions:
 - (a) Second Closing. The Second Closing shall have occurred.
- (b) No Law/Injunction. No federal or state court of competent jurisdiction or other Governmental Authority shall have enacted, issued, promulgated, enforced or entered any order, decree, judgment, injunction or other ruling (whether temporary, preliminary or permanent), in any case which is in effect and which prevents or prohibits the consummation of the transactions contemplated hereby at the Third Closing.
 - (c) Regulatory Approvals. The Company shall have received all Regulatory Approvals set forth on Schedule B under the heading "Third Closing".

- (d) <u>HSR Act Approval</u>. To the extent the transactions contemplated by the Third Closing are subject to the notification requirements under the HSR Act, the applicable waiting period, together with any extensions thereof, under the HSR Act shall have expired or been terminated.
- (e) <u>Listing of Ordinary Common Stock</u>. The shares of Ordinary Common Stock issuable upon the conversion of Series B-2 11.5% Preferred Stock issued at the Third Closing or the exercise of Warrants received at the Third Closing shall have been approved for listing on an Approved Market, subject to official notice of issuance.
- (f) Charter Amendment Approval. At a meeting of the Company's stockholders, duly called and held, the Company shall have obtained (i) the affirmative vote of the holders of a majority of the outstanding shares of Common Stock to approve and adopt the Charter Amendment, including the authorization of Common Stock thereunder in the form attached hereto as Exhibit L (the "Charter Amendment Approval"), which Charter Amendment at the Company's option may take the form of a restated Certificate of Incorporation so long as such restated Certificate of Incorporation incorporates the Charter Amendment in the form attached hereto as Exhibit L, and (ii) the approval of the stockholders of the Company necessary to approve the conversion or redemption of all the Preferred Stock, and the exercise of all the Warrants, for Common Stock for purposes of Section 312.03 of the New York Stock Exchange Listed Company Manual, or if shares of the Ordinary Common Stock become listed and traded on another Approved Market, so as to permit all such conversions, redemptions and exercises (the "Share Issuance Approval") and the Charter Amendment Approval, the "Stockholder Approvals") and the Company shall have filed the Series B-2 Certificate of Designation and the Charter Amendment (or the restated Certificate of Incorporation as permitted by this subsection) with the Secretary of State of the State of Delaware.
 - (g) DOE Financial Closing. The DOE Financial Closing shall have been consummated.
- 6.2 <u>Conditions to the Obligations of Each Investor</u>. The obligations of each Investor to consummate the transactions contemplated hereby at the Third Closing are subject to the fulfillment or waiver by such Investor (to the extent permitted by Law) on or before the Third Closing of each of the following conditions:
- (a) <u>Representations and Warranties</u>. The representations and warranties of the Company contained in <u>Section 2</u> shall have been true and correct as of the date hereof, except as would not result in a Material Adverse Effect. The Closing Representations made by the Company at the Third Closing that constitute Designated Representations shall be true and correct at the Third Closing, after giving effect to the Signing Disclosures. The Closing Representations made by the Company at the Third Closing that do not constitute Designated Representations, after giving effect to the Signing Disclosures and the Updated Disclosures, shall be true and correct at the Third Closing, except for such breaches as would not reasonably be expected to have a Material Adverse Effect. For the avoidance of doubt, any representations and warranties of the Company referenced in this <u>Section 6.2(a)</u> which address matters only as of a particular date need only be true and correct as of such date.
- (b) <u>Updated Disclosure</u>. No Updated Disclosure shall reflect a Company Effect that has had or is reasonably likely to have a Material Adverse Effect.
- (c) <u>Covenants</u>. The Company shall have performed or complied in all material respects with all agreements and covenants required by the Transaction Documents to be performed or complied with by it on or before the Third Closing.
- (d) Officer's Certificate. The Company shall have delivered to each Investor a certificate, dated the date of the Third Closing and signed by its duly authorized officer, to the effect that the conditions set forth in Sections 6.2(a) and 6.2(c) have been satisfied.
- (e) <u>Alternative Sourcing Agreement</u>. The Company shall have executed and delivered one or more agreements for the supply of separative work units to the Company on terms with respect to quantity, pricing and duration not less favorable to the Company than those set forth in that certain letter dated as of the date of this Agreement, from the Company to the Investors, or such other alternative supply arrangements reasonably acceptable to each Investor.
- (f) Foreign Ownership Restrictions Determination. The Foreign Ownership Restrictions Determination shall be in full force and effect, and the Board shall have taken no action contrary thereto.
 - (g) Compliance with Strategic Plan. Other than due to a breach of the Transaction

Documents by an Investor, since the Second Closing, the Company shall have conducted the business of the Company in material compliance with the Strategic Plan, as the same may be from time to time amended or modified with the consent of the Investors, such consent not to be unreasonably withheld, conditioned or delayed.

- (h) Opinions of Counsel. The Investors shall have received from the General Counsel of the Company, Latham & Watkins LLP, special counsel for the Company, and Morgan Lewis & Bockius LLP, regulatory counsel for the Company, opinions, dated as of the Third Closing, as to the respective matters set forth on Exhibit J.
- (i) <u>Concurrent Investment</u>. The Company shall have received proceeds from the sale of Series B-2 11.5% Preferred Stock and Warrants at the Third Closing to the other Investor of not less than \$37,500,000.00.
- (j) No Material Adverse Effect. Since the date of the Second Closing, there shall not have occurred any Material Adverse Effect (for the avoidance of doubt, taking into account applicable Company Effects arising since the date hereof but not giving rise to a Material Adverse Effect prior to the Second Closing).
- 6.3 <u>Conditions to the Obligations of the Company</u>. The obligations of the Company to consummate the transactions contemplated hereby at the Third Closing with respect to an Investor are subject to the fulfillment or waiver by the Company (to the extent permitted by Law) on or before the Third Closing of each of the following conditions:
- (a) <u>Representations and Warranties</u>. The representations and warranties of such Investor contained in <u>Section 3</u> shall be true and correct in all material respects as of the date hereof and as of the date of the Third Closing (except that those representations and warranties which address matters only as of a particular date need only be true and correct as of such date).
- (b) <u>Covenants</u>. Such Investor shall have performed or complied in all material respects with all agreements and covenants required by each Transaction Document to be performed or complied with by such Investor on or before the Third Closing.
- (c) Officer's Certificates. Such Investor shall have delivered to the Company a certificate, dated the date of the Third Closing and signed by a duly authorized officer of such Investor, to the effect that the conditions set forth in Sections 6.3(a) and 6.3(b) have been satisfied with respect to such Investor.
- (d) <u>Investment</u>. The Company shall have received the proceeds from the sale of Series B-2 11.5% Preferred Stock and Warrants at the Third Closing to the Investor(s) of not less than \$75,000,000.00.
- (e) Opinions of Counsel. The Company shall have received from special counsel to such Investor an opinion dated as of the Third Closing, as to the matters set forth on Exhibit J.

7. Covenants

7.1 State Securities Laws. The Company shall use all commercially reasonable efforts to (a) obtain all necessary permits and qualifications, if any, or secure an exemption therefrom, required by any state or country prior to the offer and sale of the Series B Preferred Stock and Warrants, and (b) cause such authorization, approval, permit or qualification to be effective as of the applicable Closing.

7.2 Submittal of Update to Loan Guarantee Application; Approval of Regulatory Bodies,

(a) As promptly as reasonably practicable following execution of this Agreement, the Company, after considering in good faith and reasonably reflecting appropriate input from the Investors, shall update the Company's application for a Loan Guarantee Agreement under the 2008 Solicitation and submit such update to the application to the DOE. The Company shall use commercially reasonable efforts to take all other actions reasonably necessary, proper or advisable to cause the DOE to grant the Conditional Commitment no later than June 30, 2011 and to consummate the DOE Financial Closing no later than December 31, 2011. Each Investor shall, at the Company's

request reasonably cooperate with the Company on such Loan Guarantee Agreement application consistent with this Agreement or any other Transaction Document, including (i) participating in discussions or meetings where requested to do so (and reasonable advance notice is provided) by the Company and (ii) timely providing information, data, and documents with respect to such Investor to the Company or Governmental Authorities where reasonably required by the Company in connection with the preparation of such application or requested by a Regulatory Body in connection with consideration thereof.

- (b) Promptly following execution of this Agreement, the Company shall use its commercially reasonable efforts to seek the Regulatory Approvals set forth on Schedule B under the heading "First Closing", or written confirmation from the relevant Regulatory Body that any such approvals are not required. Promptly following the First Closing, the Company shall use its commercially reasonable efforts to seek the Regulatory Approvals set forth on Schedule B under the heading "Second Closing", or written confirmation from the relevant Regulatory Body that any such approvals are not required. Promptly following the Second Closing, the Company shall use its commercially reasonable efforts to seek the Regulatory Approvals set forth on Schedule B under the heading "Third Closing", or written confirmation from the relevant Regulatory Body that any such approvals are not required. Each Investor shall, at the Company's request reasonably cooperate with the Company on the Regulatory Approvals set forth on Schedule B, including (i) participating in discussions or meetings where requested to do so (and reasonable advance notice is provided) by the Company and (ii) timely providing information, data, and documents with respect to such Investor to the Company or Governmental Authorities where reasonably required by the Company in connection with the preparation of such application or requested by a Regulatory Body in connection with consideration thereof.
- (c) To the extent permitted by Law, the Company agrees to keep the Investors informed regarding the Conditional Commitment, the DOE Financial Closing and the approval of the Transactions, on a timely basis, including with respect to any material communications received by the Company from, or given by the Company to, the Regulatory Bodies, or confirmations that any such approvals are not required (and including, to the extent permitted by Law, providing the Investors with copies of any material written correspondence). In connection with the foregoing, the Company shall, to the extent reasonably practicable and otherwise permitted by Law, (i) provide the Investors with reasonable advance notice of any scheduled meetings with the Regulatory Bodies at which the Company reasonably anticipates approval of the Transactions will be discussed; and (ii) seek to allow the Investors to participate in such meetings with the Regulatory Bodies.
- (d) Following the First Closing, the Company agrees that the BSRs (as defined in the Strategic Relationship Agreement) appointed by the Investors may, to the extent permitted by Law, participate in any discussions and correspondence with the Regulatory Bodies of the kind described in Section 7.2(c).
- 7.3 Negation Plans. To the extent necessary to obtain the approval of any of the Regulatory Bodies in connection with any of the Closings, the Company shall (i) prepare and submit such foreign ownership, control, domination or influence negation action plans relating to the Company or other mitigation measures as may be required to gain the approval or concurrence of any such Regulatory Body as soon as practicable and within the time periods set forth in this Agreement or to negate any material adverse impact upon the Permits held by the Company and its Affiliates as a result of the Transactions (each such plan or measure, a "Negation Plan"); and (ii) comply at the earliest practicable date with any requests from the Regulatory Bodies for additional information, documents or other materials related to the negation of any foreign ownership, control, domination or influence. Each Investor shall reasonably cooperate with such actions by the Company. Any Negation Plan proposed to be applied to an Investor must be acceptable to such Investor, but such Investor will not unreasonably withhold its approval of any such plan that it reasonably determines (i) will not materially or adversely affect any of the rights intended to be provided to such Investor pursuant to the Transaction Documents and (ii) does not restrict or limit such Investor's business activities other than with respect to the Company and its Affiliates. For avoidance of doubt, and without limiting the foregoing it is understood and agreed that an Investor may withhold its consent under the foregoing in the event the Negation Plan materially restricts any of the following essential rights of an Investor in accordance with the Transaction Documents: (1) the senior executive consultations as provided in Article III of the Strategic Relationship Agreement; (2) the appointment of directors including the appointment by Toshiba of a non-US national or U.S. national employed by an Agreed Affiliates, or other Person who is not fully dis-interested relative to Toshiba, and their full participation on the Board (without access to Classified Information or Export Controlled Information); (3) the Investor Director information rights provided in Section 1.1 of the Investor Rights Agreement; (4) the seconding of BSRs by each Investor under Article II of the Strategic Relationship Agreement, including the appointment by Toshiba of a non-US national or U.S. national employed by an Agreed Affiliates (and the rights of such BSR provided under the Strategic Relationship Agreement); (5) the Investor Informational Rights provided in Section 3.3 of the Strategic Relationship Agreement; (6) the rights under Section 3.6 of the Strategic Relationship Agreement; (7) the Special Approval Rights provided in Article II of the Investor Rights Agreement and Section 7.10 of this Agreement; (8) the rights related to the supply of SWU to Toshiba and its Prospective Customers as

provided in Article I of the Strategic Relationship Agreement, and (9) the purchase, ownership, holding and conversion, and disposition rights with respect to Securities as provided in the Transaction Documents subject to the restrictions on voting, holding, redemption and orderly sale as provided in the Transaction Documents. Each Investor shall also execute any agreements ancillary to, and required for the enforcement of, any Negation Plan acceptable to the Investor. Capitalized terms that are used in this Section 7.3 and not defined in this Agreement shall have the meanings ascribed to such terms in the Strategic Relationship Agreement or the Investor Rights Agreement, as applicable.

7.4 Antitrust.

(a) To the extent the transactions contemplated by the Second Closing or the Third Closing are subject to the notification requirements under the HSR Act, in each case as determined by each Investor with respect to its investment, the Company and such Investor shall use all commercially reasonable efforts to (i) make an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the acquisition of the Preferred Stock and Warrants contemplated by such Closing as promptly as practicable, and supply as promptly as practicable any additional information and documentary material that may be requested pursuant to the HSR Act, (ii) subject to the provisions of Section 7.4(b), take all other actions reasonably necessary, proper or advisable to cause the expiration or termination of the applicable waiting periods under the HSR Act, as promptly as practicable, (iii) subject to applicable Law, keep the other Parties hereto informed of any communication received by the Company or such Investor (as the case may be) from, or given by the Company or such Investor (as the case may be) to, the Federal Trade Commission (the "FTC"), the Antitrust Division of the Department of Justice (the "DOF") or any other Governmental Authority and of any communication received or given in connection with any legal, administrative, arbitral or other proceeding by a private party, in each case regarding the issuance and sale of the Preferred Stock or Warrants, (iv) subject to applicable Law and all applicable privileges, including the attorney-client privilege, permit any other Party to review in advance any communication intended to be given by it to, and consult with it in advance of any meeting or conference with, the FTC, the DOJ or any such other Governmental Authority, and to the extent permitted by the FTC, the DOJ or such other applicable Governmental Authority, give any other Party the opportunity to attend and participate in such meetings and conferences, (v) subject to applicable Law, consult and cooperate with the other Parties in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any Party hereto relating to proceedings under the HSR Act or other antitrust Laws, and (vi) comply at the earliest practicable date with any request under the HSR Act or other antitrust Laws for additional information, documents or other materials received by each of them or any of their respective Subsidiaries from the FTC, the DOJ or any other Governmental Authority in respect of such filings or such transactions.

(b) If any objections are asserted with respect to the transactions contemplated hereby under the HSR Act, or any other antitrust Law, each of the Parties shall use all commercially reasonable efforts to resolve such objections as such Governmental Authority may have to such transactions under antitrust Law, through the completion of the investigative processes of the Governmental Authority. In connection therewith, if any legal proceeding is instituted (or threatened to be instituted) challenging any transaction contemplated by this Agreement as in violation of any antitrust Laws, whether by a Governmental Authority or private party, nothing herein shall require any Party to contest or resist the objections beyond the completion of the investigative processes of the Governmental Authority. If mutually agreed by the Parties to contest or resist the objections beyond such investigative processes, then each of the Investors and the Company shall cooperate and use their respective commercially reasonable efforts to contest and resist any such legal proceeding, and to have vacated, lifted, reversed or overturned any order of any Governmental Authority, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the transactions contemplated by this Agreement. For purposes of this Section 7.4(b), the term "commercially reasonable efforts" shall not require any Party to provide for the license, sale or other disposition or holding separate of any shares or assets of such Party, or to agree to the imposition of any material limitation on the ability of any of them to freely conduct their business or effectuate the business purposes of this Agreement; and in no event shall the Company engage in such actions without the prior written consent of the Investors.

7.5 Other Government Approvals.

(a) The Investors and the Company shall cooperate with each other and shall cause their agents and representatives to cooperate with each other to determine and timely make any pre-Closing and post-Closing submissions, applications or notices for approvals or other consents from any Governmental Authority, other than those set forth in Sections 7.2, 7.3, 7.4, or 7.5(b)-(d). For the avoidance of doubt, such Sections (and not this Section 7.5(a)) shall apply to the approvals and other matters referenced in such Sections.

(b) The Company and Toshiba shall: (i) make or cause to be made a joint voluntary notice to CFIUS with respect to the transactions contemplated hereby as promptly as practicable, (ii) comply at the earliest practicable date with any requests from CFIUS or other Governmental Authority for additional information, documents or other materials required or related to the CFIUS notice and any subsequent investigation, and (iii) cooperate with each other in connection with any such filing (including, to the extent permitted by applicable Law, providing copies of drafts of such filing to the non-filing Parties prior to filing and considering all reasonable additions, deletions or changes suggested in connection therewith) and in connection with resolving any inquires from CFIUS or any other Governmental Authority with respect to the CFIUS filing or any investigation with respect to any such filing.

(c) If (A) following consultation with CFIUS, Toshiba and the Company, B&W shall reasonably determine that its failure to make a voluntary notice to CFIUS with respect to the Transactions or to be a party to the joint voluntary notice to CFIUS referenced in Section 7.5(b) could cause the satisfaction of the condition set forth in Section 4.1(c) hereof to be materially delayed or fail, or (B) CFIUS has formally requested that B&W make a voluntary notice to CFIUS or be a party to the joint voluntary notice to CFIUS with respect to the Transactions, or CFIUS determines that in the absence of the foregoing, it will initiate an investigation with respect to the Transactions (and after consultation with B&W, CFIUS has not withdrawn such request or determination), B&W shall (i) after further consultation with Toshiba and the Company, as promptly as practicable, within 14 days thereafter, unless the 14th day is prior to July 15, 2010, in which case, on July 15, 2010 (x) make a voluntary notice to CFIUS, (y) make a voluntary joint notice to CFIUS with Toshiba and the Company, or (z) if such notice has already been submitted, submit such documentation as may be necessary to become a party to the notice, and otherwise provide sufficient information to CFIUS to enable CFIUS to review B&W's proposed involvement in the transactions contemplated hereby, and (ii) the terms set forth in Sections 7.5(b)(ii) and 7.5(b)(iii) shall apply to B&W as if B&W were specifically mentioned therein.

(d) Each Party shall use its commercially reasonable efforts to furnish to the other Parties all information required for any CFIUS notice or other filing to be made pursuant to such notice under any applicable Law in connection with the transactions contemplated by this Agreement. Each Party shall promptly inform the other Parties hereto of any oral communication with, and provide copies of written communications with, any Governmental Authority regarding any CFIUS notice or investigation. No Party hereto shall independently participate in any meeting or discussion, either in person or by telephone, with any Governmental Authority in respect of any such filings, investigation or other inquiry without giving the other Parties hereto prior notice of the meeting and, to the extent permitted by such Governmental Authority, the opportunity to attend and/or participate. Subject to applicable Law, the Parties hereto will consult and cooperate with one another in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any Party hereto relating to the CFIUS notice and any investigation. Notwithstanding anything to the contrary contained in this Agreement, the Investors and the Company each reserve the right to limit disclosure of any document, or portion thereof, submitted to any Governmental Authority with respect to any CFIUS notice to the Investors' and the Company's outside counsel only. Notwithstanding anything to the contrary contained in this Agreement, the Investors and the Company each reserve the right to limit disclosure of any document, or portion thereof, submitted to any Governmental Authority with respect to any CFIUS notice to the Investors' and the Company's outside counsel only.

7.6 Negative Covenants Prior to First Closing From the date of this Agreement through the First Closing, the Company agrees that except as expressly permitted by any Transaction Document, or as required by applicable Law or the regulations and requirements of the New York Stock Exchange, unless each Investor otherwise grants its prior written approval, the Company will, and will cause each of its Subsidiaries to: (i) conduct its operations in the ordinary and usual course of its business; (ii) use its commercially reasonable efforts to keep available the services of the current officers of the Company and preserve the goodwill and current relationships of the Company with customers, suppliers and other Persons with which the Company has significant business relations; (iii) use its commercially reasonable efforts to preserve substantially intact its business organization; and (iv) comply in all material respects with applicable Law. Without limiting the foregoing, and as a non-exclusive extension thereof, except as expressly permitted by any Transaction Document, as required by applicable Law or the regulations and requirements of the New York Stock Exchange or with the prior written approval of each Investor (which approval shall not be unreasonably withheld, conditioned or delayed), from the date of this Agreement through the First Closing, the Company shall not:

- (a) Declare, or make payment in respect of, any dividend or other distribution upon any shares of capital stock of the Company;
- (b) Reclassify, combine, split, subdivide or amend the terms of, or, other than with respect to capital stock issued pursuant to a Company Plan, redeem, repurchase or acquire any capital stock of the Company

or any of its Subsidiaries;

- (c) Amend the Company's Certificate of Incorporation or Bylaws;
- (d) Other than capital stock pursuant to a Company Plan in existence as of the date of this Agreement, as may be amended, establish a new Company Plan, authorize or issue any capital stock, or debt securities convertible into capital stock, of the Company;
 - (e) except as required by the Board's fiduciary duties, merge or consolidate the Company with any Person; or
 - (f) make or authorize any capital expenditures in excess of \$175,000,000 in the aggregate prior to December 31, 2010.
- 7.7 <u>Use of Proceeds</u>. Prior to the Third Closing, the Company shall apply the net proceeds from the issuance and sale of the Series B-1 12.75% Preferred Stock for general corporate purposes and for expenditures on ACP, and in a manner consistent with the Strategic Plan.
- 7.8 Reservation of Common Stock; Issuance of Shares of Common Stock For as long as any Preferred Stock, Class B Common Stock or Warrants remain outstanding, subject to the Stockholder Approvals, the Company shall at all times reserve and keep available, free from preemptive rights, out of its authorized but unissued Series C Preferred Stock, Class B Common Stock and Ordinary Common Stock or shares of Series C Preferred Stock, Class B Common Stock and Ordinary Common Stock held in treasury by the Company, for the purpose of effecting the conversion or redemption of the Preferred Stock and the Class B Common Stock or permitting the exercise of Warrants, as subject to the Share Issuance Limitation (as such term is defined in the Certificates of Designation), if any, the full number of shares of Ordinary Common Stock, Class B Common Stock and Series C Preferred Stock then issuable upon the conversion or redemption of all Preferred Stock and Class B Common Stock and the full exercise of all outstanding Warrants (in each case, after giving effect to all anti-dilution adjustments) then outstanding. For the purposes hereof, reservations hereunder shall be based on the Base Price equal to the closing price of the Company's Ordinary Common Stock on the New York Stock Exchange on the second to last Trading Day prior to the date of this Agreement, provided, however, if the Base Price for the date that is four Trading Days prior to the First Closing, the Second Closing or the Third Closing or on June 30 of any year is less than such amount, then that lower amount shall be used as the Base Price for purposes of this calculation. To the extent that the number of shares reserved or authorized by the Company is not sufficient to meet its obligations under this Section 7.8 at the date set forth in the immediately preceding sentence, the Company shall use its commercially reasonable efforts to take such actions as are necessary to reserve or authorize such shares in sufficient amounts to meet the Company's obligations hereunder, including to the extent necessary and as expeditiously as reasonably possible, amendments to the Company's Certificate of Incorporation. All shares of Common Stock and Series C Preferred Stock delivered upon conversion or redemption of the Preferred Stock or the Class B Common Stock or the exercise of Warrants shall have been duly authorized and validly issued and shall be fully paid and nonassessable, and shall be free from preemptive rights, other than established by the Transaction Documents, and free of any Lien or adverse claim. The Company shall use commercially reasonable efforts to cause the shares of Ordinary Common Stock issuable upon the conversion or redemption of the Preferred Stock and the Class B Common Stock and the exercise of the Warrants received at the Closings to be approved for listing by an Approved Market in connection with the applicable Closing, subject to official notice of issuance.
- 7.9 Filing of Certificates of Designation. The Company shall file with the Secretary of State for the State of Delaware (i) the Series B-1 Certificate of Designation and the Series C Certificate of Designation and (ii) the Series B-2 Certificate of Designation, at or prior to the time that the closing conditions for the First Closing and the Second Closing, respectively, are satisfied (other than the closing condition related to such filing).

7.10 Special Approval Rights.

(a) The Company agrees that, between the First Closing and the earlier of (i) the Third Closing or (ii) the termination of this Agreement pursuant to Section 10, so long as either Investor maintains its Minimum

Preferred Holdings the Company shall not (and shall not permit any of its Subsidiaries to) take or agree to take any of the following actions without the prior written approval of the holders of a majority of the Preferred Stock then outstanding (voting together as a single class) (which decisions shall not be unreasonably delayed):

- (i) any issuance of equity securities of the Company senior to or pari passu with the Preferred Stock, except as the Board determines is necessary to obtain the Conditional Commitment or to consummate the DOE Financial Closing;
 - (ii) the undertaking of any new business inconsistent with the Strategic Plan;
- (iii) any incurrence of Indebtedness by the Company or any of its Subsidiaries, except as permitted under the terms of the Credit Facility or a Transaction Document or as provided in the Strategic Plan, and excluding debt incurred effective upon, and for purposes of funding, the DOE Financial Closing;
 - (iv) any redemption by the Company of the Preferred Stock, except for any Permitted Redemption;
- (v) any payment of Indebtedness (other than Indebtedness that can be reborrowed under the terms of the applicable credit facility, or any such payment made in order to consummate the DOE Financial Closing or a refinancing of the Credit Facility complying with clause (iii)) by the Company or any of its Subsidiaries prior to the stated maturity thereof; or
- (vi) except as provided in the Strategic Plan, any (A) disposition of assets or business of the Company or any of its Subsidiaries, except as permitted under the terms of the Credit Facility, (B) acquisition, binding capital commitment, guarantee of third-party obligations or investment by the Company or any of its Subsidiaries, except as permitted under the terms of the Credit Facility, or (C) joint venture involving the Company or any of its Subsidiaries; or
 - (vii) Any liquidation or dissolution of the Company.

For purposes of this Section 7.10, the exceptions based upon the Credit Facility shall apply whether or not such facility remains outstanding.

(b) An Investor will be deemed to maintain its 'Minimum Preferred Holdings' for purposes of this Section 7.10 if such Investor continues to hold shares of Preferred Stock representing no less than 75% of the aggregate initial liquidation preference of the Preferred Stock issued to such Investor in the First Closing and, if the Second Closing has occurred, the Second Closing (excluding for purposes of this definition shares of Preferred Stock issued to the Investors as payment of dividends on the Preferred Stock).

7.11 Stockholder Meeting; Proxy Materials.

- (a) As promptly as reasonably practicable following the Second Closing (or earlier at the Company's election), the Company shall prepare and file with the Commission a proxy statement relating to the meeting of the Company's stockholders to be held in connection with the Stockholder Approvals (together with any amendments thereof or supplements thereto, the "Proxy Statement"). Prior to filing the Proxy Statement, the Company shall provide each Investor with a reasonable opportunity to review and comment thereon, and the Company shall reasonably consider such comments in finalizing the Proxy Statement. The Company shall use commercially reasonable efforts to respond to any comments made by the Commission with respect to the Proxy Statement. The Investors shall furnish the Company with all information concerning the Investors as the Company may reasonably request in connection with the preparation of the Proxy Statement. Subject to the Board's right to make any disclosure to the Company's stockholders as the Board determines in good faith should be disclosed pursuant to applicable Law, the Proxy Statement shall include the determination of the Board that the Charter Amendment Approval and the Share Issuance Approval are advisable.
- (b) The Company shall call and hold a meeting of its stockholders as promptly as reasonably practicable following the Second Closing (or earlier at the Company's election) for the purpose of obtaining the Charter Amendment Approval and the Share Issuance Approval.
- (c) The Company shall file the Charter Amendment with the Secretary of State of the State of Delaware promptly after the Charter Amendment Approval.

7.12 [Reserved].

7.13 Foreign Ownership Provisions.

(a) Except as required as a result of changes in applicable Law administered by a Regulatory Body, where the failure to so act could reasonably be expected to (i) have a Material Adverse Effect, (ii) result in criminal liability for an officer, director or employee of the Company or their banning for any period from employment in any licensed nuclear facility, or (iii) result in the violation of a directive, order or decree to the Company from a Governmental Authority, from the date of this Agreement until (x) the termination of this Agreement pursuant to Section 10 or (y) should the First Closing occur, the date on which the Investors cease to hold Common Stock (or Preferred Stock or Warrants convertible into or exercisable for Common Stock) equal to less than five percent (5%) of the outstanding Common Stock (based on the Base Price then used to reserve shares under Section 7.8), the Company shall not (A) take any action to revoke the Foreign Ownership Restrictions Determination, or (B) amend or adopt any foreign ownership provisions in its Certificate of Incorporation or Bylaws (including in Article Eleventh of the Company's Certificate of Incorporation), in each case without the prior written consent of any Investor adversely affected thereby (each a "Company Action").

(b) If the Company reasonably anticipates such a change in applicable Law that could require a Company Action, or determines that such a change in applicable Law has occurred, the Company will promptly provide each Investor with written notice thereof, describing in reasonable detail the relevant change or anticipated change in Law, the reason such change would require the Company to take a Company Action, and the Company Action anticipated. At the request of an Investor who might be affected by such Company Action made within ten (10) Business Days after the delivery of such notice, the Company will meet with such Investor, promptly and in any event within the next ten (10) Business Days, to discuss the matter with the goal of agreeing in good faith on a Company Action reasonably intended to address the actual or anticipated change in Law while minimizing the effect on such Investor's rights under the Transaction Documents, including with respect to the Securities then held by such Investor. The Company shall not take any Company Action during such additional ten (10) Business Day period to allow for such consultations. After such period, the Company may, without limitation of the Investor's rights under Section 7.13(c), take the Company Action agreed upon in discussions with such Investor, or failing any such agreement, such Company Action as the Company may reasonably determine is required to address such change of applicable Law while still giving effect to the fullest extent possible to the original intent of the Investor and the Company in entering into the Transaction Documents.

(c) If, notwithstanding compliance with Section 7.13(b), the Company takes a Company Action which would adversely impact any rights of the Investor under any of Transaction Documents or with respect to its Securities, the Company and the Investor shall negotiate in good faith to amend this Agreement and the other Transaction Documents so as to effect the original intent of the Company and the Investor as closely as possible. If the Company and the Investor fail to reach agreement on such amendments within thirty (30) days after such Company Action (or such longer period as such Parties may agree), then the Company shall take all actions necessary to facilitate the Investor's prompt sale of its interest in the Company, including, as appropriate, without regard to, or subject to amended, volume limitations on block or brokered sales (and the other Investor will reasonably cooperate with respect thereto).

7.14 Tax Matters.

U.S. Person.

(a) Prior to the First Closing, upon the reasonable request of the Company or as otherwise required by applicable Law, each Investor shall deliver to the Company a properly completed and executed IRS Form W-9 (or successor form), evidencing a complete exemption from United States withholding tax (including back-up withholding tax). To the extent required by law, the Company is hereby authorized to withhold taxes from any distributions to an Investor and any amounts so withheld shall be treated as otherwise paid to the applicable Investor. In no event shall any distribution to an Investor be reduced by the withholdings applicable to the other Investor.

(b) Prior to the First Closing, Toshiba shall assign its rights to purchase Preferred Stock and Warrants under this Agreement to an Affiliate that is a

(c) The Parties agree to reasonably and mutually agree upon the allocation of the purchase price at each Closing to the Series B Preferred Stock and Warrants (as applicable) as promptly as reasonably

practicable following such Closing.

- 7.15 <u>Disclosure Updates</u>. The Company may deliver to the Investors amendments to the Disclosure Letter setting forth in reasonable detail such information with respect to developments since the date hereof and prior to such Closing as are required to make the Closing Representations made at a Closing true and correct as of such Closing (the Signing Disclosures and the matters disclosed in such amendments to the Disclosure Letter or in SEC Reports filed after the date hereof are collectively as of the applicable Closing, the "*Updated Disclosures*").
- 7.16 <u>Investor Directors</u>. Not later than 14 days after the date of this Agreement, the Investors shall notify the Company of the two Persons whom the Investors intend to elect as directors to the Board. Thereafter, to the extent any such Person is not reasonably acceptable to the Nominating and Corporate Governance Committee of the Board, the Investors shall promptly notify the Company of a replacement.
 - 7.17 ACM. The Company and B&W shall endeavor in good faith to satisfy the conditions set forth in Section 4.2(g).
 - 8. [Reserved].
 - 9. Restrictions on Transfer.
- 9.1 Transfer of Preferred Stock and Warrants. Each Investor hereby agrees that it will not transfer any Preferred Stock or any Warrants to any Person, other than to (i) its Wholly-Owned Affiliate, or Westinghouse Electric Company, LLC if controlled by Toshiba, in each case that is a U.S. Person (or a special purpose entity jointly formed and controlled by the Investors) that agrees to be bound by the terms of, and become a party to, this Agreement and the Investor Rights Agreement, (ii) the other Investor or its Wholly-Owned Affiliate, or Westinghouse Electric Company, LLC if controlled by Toshiba, in each case that is a U.S. Person, or (iii) the Company and its Subsidiaries, but in the case of either clause (i) or (ii) above, only to the extent such transfer, if it would require filings with or approvals from any Regulatory Body in order not to adversely affect the Permits or regulatory status of the Company or its Subsidiaries, has received such approvals. Any such transfer shall not occur until the Investor provides the Company at least 20 Business Days prior notice of the planned transfer and obtains any such approvals. The Company and the Investors shall cooperate in good faith to obtain any such approvals. Any transfer of Preferred Stock, Series C Preferred Stock or Warrants pursuant to this Section shall be subject to any Negation Plan to which the applicable Investor is then a party. Any transfer of Preferred Stock, Series C Preferred Stock or Warrants other than pursuant to this section shall be null and void, *ab initio*.

9.2 Orderly Sale of Common Stock.

- (a) Notwithstanding anything to the contrary in this Agreement, the Certificate of Incorporation, the Certificates of Designation, the Warrants, the Investor Rights Agreement or the Strategic Relationship Agreement, at any time an Investor is required to or otherwise determines to sell any shares of Common Stock upon the conversion or redemption of Preferred Stock, the conversion of the Class B Common Stock or the exercise of any Warrant, (i) the Company shall use its commercially reasonable efforts to assist the Holder in promptly disposing of such shares upon receipt thereof pursuant to an Orderly Sale Arrangement, including complying with all applicable covenants in the Investor Rights Agreement, and (ii) the Holder shall cooperate with such efforts by the Company and comply with the applicable provisions of the Orderly Sale Arrangement including complying with all covenants in the Investor Rights Agreement.
- (b) The term "Orderly Sale Arrangement" shall mean sales pursuant to any of the following methods acting reasonably and in good faith, that does not result in the Holder having record ownership, possession or voting power over Ordinary Common Stock (such as, for illustrative purposes only, directing the Company (or

its transfer agent) to deliver the shares of Ordinary Common Stock to a brokerage account for the benefit of the purchaser without such shares being issued in the name of the Holder):

- (i) the sale of Common Stock in an underwritten offering pursuant to the Investor Rights Agreement;
- (ii) the sale of Common Stock by a Holder or its Affiliates in compliance with Rule 144(f) promulgated under the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder, as if such sales were effected pursuant to such rule; provided that under this clause (ii), on a daily basis on any Trading Day, the sales of Common Stock by any Holder or its Affiliates shall not exceed 10% of an average daily trading volume of the Ordinary Common Stock for the four-week period immediately preceding the week during which such Trading Day occurs, as reported on Bloomberg Financial L.P. (or any successor thereof) using the HP function (or any equivalent thereof), or (b) if greater, 10% of the daily trading volume of the Ordinary Common Stock for that particular Trading Day; provided, that any additional sales made based on this clause 9.2(b)(ii) must be made at a price above the volume weighted average price for the immediately preceding Trading Day;
- (iii) a block sale of Common Stock, at a price per share no less than 90% of the volume weighted average price of the Company's Ordinary Common Stock on an Approved Market on the immediately preceding Trading Day, to a transferee that would not violate the restrictions set forth in Section 9.3 below, such sale not to exceed at any one time that number of shares equal to the lesser of (A) the average daily trading volume of the Ordinary Common Stock for the four-week period immediately preceding the week during which such sale occurs, or (B) 10% of the Ordinary Common Stock outstanding at such time and, in either case, and to be executed through a broker dealer that, immediately prior to such sale, represents one of the institutions listed on Schedule C hereto, which list may be updated from time to time by the Company upon notice to the Holders to replace any Person thereon with any of the five largest market makers for the Company's Ordinary Common Stock; provided that no Holder shall be permitted to undertake more than one block sale of Common Stock in any three-month period provided, further that the Holder delivers a certification to the Company stating that to the knowledge of such Holder, after due inquiry, such transfer will not violate the transfer restrictions set forth in Section 9.3 and the Person acquiring such shares is not a Contravening Person (as defined in the Certificate of Incorporation); or
 - (iv) any other method of sale mutually agreed by the Company and such Holder, acting reasonably and in good faith;

Notwithstanding anything herein or in any of the other Transaction Documents to the contrary, no Investor shall have any obligation to take any actions or make any efforts (commercially reasonable or otherwise) related to selling any shares of Common Stock or entering into a Sales Plan (as defined in the Certificates of Designations) during the pendency of the period that an Investor agrees not to sell any shares of Common Stock pursuant to Section 4.7 of the Investors Rights Agreement.

- 9.3 No Transfer to Competitors; Block Transfers Each Investor hereby agrees that, with respect to the Common Stock referred to in Section 9.2(a), it will not directly or indirectly:
- (a) in a transaction or a series of transactions, knowingly transfer any Common Stock to any Person that is a Competitor or any controlled Affiliate of a Competitor, other than (absent willful misconduct by the Investors) through brokers' transactions (except for such transactions relying upon the exemptions in clauses (i) or (ii) of Rule 144(g) (3) promulgated under the Securities Act); or
- (b) individually or acting together with any other Holders, at any time transfer shares of Common Stock to any Person (except as permitted under Section 9.1) that has filed a Schedule 13D or a Schedule 13G with respect to the Company's Common Stock that has not been terminated, or to the knowledge of such Investor, would be required to file a Schedule 13D or Schedule 13G as a result of the purchase of Common Stock from the transferring Investor, in each case other than through brokers' transactions. Notwithstanding the foregoing sentence, an Investor shall be permitted to sell Common Stock to a Person that at such time has filed on Schedule 13G with the Commission with respect to Ordinary Common Stock, provided, based upon such filed Schedule 13G (i) such Person beneficially owns less than 10% of the Company's outstanding Ordinary Common Stock, and (ii) as a result of the purchase of Common Stock from the transferring Investor, such Person will beneficially own in the aggregate less than 10% of the Company's outstanding Common Stock.

9.4 Restriction on Hedging, Etc.

Neither Investor nor any other Holder shall enter into any swap, hedge, forward contract or other arrangement or transaction or series of transactions that hedges or transfers, in whole or in part, directly or indirectly, any of the economic consequences of ownership of the Preferred Stock or Warrants, or of the shares of Common Stock issuable upon conversion or exercise thereof. Neither Investor, nor any of their respective Affiliates, shall directly or indirectly sell any Ordinary Common Stock that such Investor does not at the time of sale own. The restrictions in this Section 9.4 shall terminate as to each Investor when such Investor no longer owns any Common Stock referred to in Section 9.2(a), Preferred Stock or Warrants.

10. Termination.

- 10.1 Termination of Agreement Prior to First Closing. This Agreement may be terminated at any time prior to the First Closing:
- (a) by each Investor (as to such Investor's obligations hereunder) or the Company if the First Closing shall not have occurred by December 31, 2010; provided, however, that the right to terminate this Agreement under this Section 10.1(a) shall not be available to any Party whose failure to fulfill any obligation under this Agreement shall have been the cause of, or shall have resulted in, the failure of the First Closing to occur on or prior to such date;
- (b) by each Investor (as to such Investor's obligations hereunder) or the Company in the event that any Governmental Authority shall have issued an order, decree or ruling or taken any other action restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such order, decree, ruling or other action shall have become final and nonappealable; provided, however, that the right to terminate this Agreement under this Section 10.1(b) shall not be available to any Party whose failure to fulfill any obligation under this Agreement shall have been the cause of, or shall have resulted in, such action of a Governmental Authority;
- (c) by each Investor (as to such Investor's obligations hereunder) upon the occurrence of a material breach of this Agreement by the Company, which breach is incapable of being cured by December 31, 2010 or is not cured within 30 days of written notice from an Investor;
- (d) by the Company upon the occurrence of a material breach of this Agreement by an Investor, which breach is incapable of being cured by December 31, 2010 or is not cured within 30 days of written notice from the Company, unless within 30 days of written notice from the Company, the non-breaching Investor irrevocably agrees in writing to assume the obligations of such breaching Investor to purchase shares of Preferred Stock and Warrants at each Closing; or
 - (e) by the mutual written consent of the Investors (acting together) and the Company.

In the event of termination of this Agreement as provided in this <u>Section 10.1</u>, except for <u>Section 13</u>, which shall survive the termination of this Agreement, this Agreement shall forthwith become void with respect to the applicable Parties and there shall be no liability on the part of any such Party, except that nothing in this <u>Section 10.1</u> shall relieve any Party from liability for any willful breach of this Agreement.

- 10.2 Termination of Agreement Prior to Second Closing. This Agreement may be terminated at any time prior to the Second Closing:
- (a) by each Investor (as to such Investor's obligations hereunder) or the Company if the Second Closing shall not have occurred by June 30, 2011; provided, however, that the right to terminate this Agreement under this Section 10.2(a) shall not be available to any Party whose failure to fulfill any obligation under this Agreement shall have been the cause of, or shall have resulted in, the failure of the Second Closing to occur on or prior to such date;
- (b) by each Investor (as to such Investor's obligations hereunder) or the Company in the event that any Governmental Authority shall have issued an order, decree or ruling or taken any other action restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such order, decree, ruling or other action shall have become final and nonappealable; provided, however, that the right to terminate this Agreement under this Section 10.2(b) shall not be available to any Party whose failure to fulfill any obligation under this Agreement shall have been the cause of, or shall have resulted in, such action of a Governmental Authority;

(c) by each Investor (as to such Investor's obligations hereunder) upon the occurrence of a material breach of this Agreement or any other Transaction Document by the Company, which breach is incapable of being cured by June 30, 2011 or is not cured within 30 days of written notice from an Investor;

(d) by the Company upon the occurrence of a material breach of this Agreement or any other Transaction Document by an Investor, which breach is incapable of being cured by June 30, 2011 or is not cured within 30 days of written notice from the Company, unless within 30 days of written notice from the Company, the non-breaching Investor irrevocably agrees in writing to assume the obligations of such breaching Investor to purchase shares of Preferred Stock and Warrants at the Second Closing and the Third Closing; or

(e) by the mutual written consent of the Investors (acting together) and the Company.

In the event of termination of this Agreement as provided in this Section 10.2, except for Sections 7.13, 7.14(b), 9, 11, 12 and 13, which shall survive the termination of this Agreement, this Agreement shall forthwith become void with respect to the applicable Parties and there shall be no liability on the part of any such Party, except that nothing in this Section 10.2 shall relieve any Party from liability for any breach of this Agreement.

- 10.3 Termination of Agreement Prior to Third Closing This Agreement may be terminated at any time prior to the Third Closing:
- (a) by each Investor (as to such Investor's obligations hereunder) or the Company if the Third Closing shall not have occurred by December 31, 2011 (as such date may be extended as provided herein, the "*Third Closing Termination Date*"); provided, however, if all conditions to the Third Closing have been satisfied or waived, or are reasonably capable of being satisfied by December 31, 2011, other than the condition set forth in Section 6.1(g), and such condition its reasonably capable of being satisfied, the Investors may, by written notice to the Company, extend the Third Closing Termination Date up to December 31, 2012, and provided, further, that the right to terminate this Agreement under this Section 10.3(a) shall not be available to any Party whose failure to fulfill any obligation under this Agreement shall have been the cause of, or shall have resulted in, the failure of the Third Closing to occur on or prior to such date;
- (b) by each Investor (as to such Investor's obligations hereunder) or the Company in the event that any Governmental Authority shall have issued an order, decree or ruling or taken any other action restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement and such order, decree, ruling or other action shall have become final and nonappealable; provided, however, that the right to terminate this Agreement under this Section 10.3(b) shall not be available to any Party whose failure to fulfill any obligation under this Agreement shall have been the cause of, or shall have resulted in, such action of a Governmental Authority;
- (c) by each Investor (as to such Investor's obligations hereunder) upon the occurrence of a material breach of this Agreement or any other Transaction Document by the Company, which breach is incapable of being cured by December 31, 2011 or is not cured within 30 days of written notice from an Investor;
- (d) by the Company upon the occurrence of a material breach of this Agreement or any other Transaction Document by an Investor, which breach is incapable of being cured by December 31, 2011 or is not cured within 30 days of written notice from the Company, unless within 30 days of written notice from the Company, the non-breaching Investor irrevocably agrees in writing to satisfy the obligations of such breaching Investor to purchase shares of Series B-2 11.5% Preferred Stock and Warrants at the Third Closing; or
 - (e) by the mutual written consent of the Investors (acting together) and the Company.

In the event of termination of this Agreement as provided in this Section 10.3, except for Sections 7.13, 7.14(b), 9, 11, 12 and 13, which shall survive the termination of this Agreement, this Agreement shall forthwith become void with respect to the applicable Parties and there shall be no liability on the part of any such Party, except that nothing in this Section 10.3 shall relieve any Party from liability for any breach of this Agreement.

11. Indemnification.

11.1 Survival

(a) The representations and warranties contained in this Agreement made at the date hereof shall survive for one (1) year following the First Closing, and the Closing Representations made at a Closing shall survive for one (1) year following such Closing, and thereafter shall have no further force and effect; provided, however, that the representations and warranties contained in Section 2.7 (Taxes) (including when made as Closing Representations at any Closing) shall survive the execution of this Agreement and the completion of the Transactions until the expiration of the applicable statute of limitations, and provided, further, that the representations and warranties of the Company contained in Sections 2.3 (Corporate Power; Authorization; Enforceable Agreement) and 2.12 (Valid Issuance of Preferred Stock and Common Stock) (including when made as Closing Representations at any Closing) and the representations and warranties of each Investor contained in Sections 3.3 (Power and Authority) and 3.4 (Authorization; Enforceability) shall survive the execution of this Agreement and the completion of the Transactions indefinitely.

(b) All covenants, agreements and obligations of each Party contained in this Agreement shall survive each Closing indefinitely in accordance with their respective terms.

11.2 Company Indemnified Parties.

- (a) Following the First Closing, the Company hereby agrees to indemnify, pay and hold each Investor, and each of its officers, directors and employees (collectively, the "Company Indemnified Parties") harmless from and against any and all liabilities, costs, expenses, obligations, losses, damages, penalties, actions, judgments, suits, claims and disbursements of any kind or nature whatsoever (but including only the reasonable fees and expenses of one counsel) actually incurred by a Company Indemnified Party, in any manner relating to or arising out of any of the following (collectively, "Company Indemnified Liabilities"):
- (i) the failure of any of the representations and warranties set forth in Section 2 of this Agreement to be true and correct as of the date of this Agreement;
- (ii) the failure of any of the Closing Representations that constitute Designated Representations made at a Closing that is consummated to be true and correct as of the date of such Closing, after giving effect to the Signing Disclosures (and for the avoidance of doubt, without giving effect to any Updated Disclosures); or
- (iii) the failure of any of the Closing Representations that do not constitute Designated Representations made at a Closing that is consummated to be true and correct as of the date of such Closing, after giving effect to the Signing Disclosures and the Updated Disclosures as of such Closing.
- (b) Each Company Indemnified Party shall give the Company prompt written notice of any third-party claim (a "Third-Party Claim") that might give rise to Company Indemnified Liabilities setting forth a description of those elements of such Third-Party Claim of which such Indemnified Party has knowledge; provided, that any delay or failure to give such notice shall not affect the obligations of the Company unless (and then solely to the extent) the Company is prejudiced by such delay or failure. The Company shall have the right at any time during which such Third-Party Claim is pending to select counsel to defend and control the defense thereof and settle any claims for which it is responsible for indemnification hereunder (provided, that the Company will not settle any such Third-Party Claim without (i) the appropriate Company Indemnified Party's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed or (ii) obtaining an unconditional release of the appropriate Company Indemnified Party from all claims arising out of or in any way relating to the circumstances involving such Third-Party Claim) so long as in any such event the Company shall have stated in a writing delivered to the Company Indemnified Party that, as between the Company and the Company Indemnified Party, the Company is responsible to the Company Indemnified Party with respect to such Third-Party Claim to the extent and subject to the limitations set forth herein; provided, that with respect to any Third-Party Claim as to which the Company Indemnified Party is controlling the defense, the Company will not be liable to any Company Indemnified Party for any settlement of any Third-Party Claim pursuant to this Section 11.2 that is effected without its prior written consent, not to be unreasonably withheld, conditioned or delayed. To the extent that the undertaking to indemnify, pay and hold harmless set forth in the preceding sentence may be unenforceable because it is violative of any Law or public policy, the Company shall contribute the maximum portion which it is permitted to pay and satisfy under applicable law to the payment and satisfaction of all Company Indemnified Liabilities incurred by the Company Indemnified Parties or any of them. The obligations of the Company set forth in this Section 11.2 shall survive for the survival periods set forth in Section 11.1 and, with respect to any Third-Party Claim for Company Indemnified Liabilities made and for which proper notice has been provided hereunder prior to the date set forth in Section 11.1, until the final resolution thereof. The indemnity provided in this Section 11.2 shall be the sole and exclusive remedy of the Company Indemnified Parties for any inaccuracy or breach of the representations and warranties set forth in this Agreement.

- 11.3 Investor Indemnified Parties. Following the First Closing, each Investor (an "Indemnifying Investor") hereby agrees, severally and not jointly, to indemnify, pay and hold the Company and each of its officers, directors and employees (collectively, the "Investor Indemnified Parties") harmless from and against any and all liabilities, costs, expenses, obligations, losses, damages, penalties, actions, judgments, suits, claims and disbursements of any kind or nature whatsoever (but including only the reasonable fees and expenses of one counsel) actually incurred by an Investor Indemnified Party, in any manner relating to or arising out of the failure of any of the representations and warranties of the Indemnifying Investor set forth in Section 3 of this Agreement to be true and correct as of the date of this Agreement or any Closing (the "Investor Indemnified Liabilities"). Each Investor Indemnified Party shall give the Indemnifying Investor prompt written notice of any Third-Party Claim that might give rise to Investor Indemnified Liabilities setting forth a description of those elements of such Third-Party Claim of which such Investor Indemnified Party has knowledge; provided, that any delay or failure to give such notice shall not affect the obligations of the Indemnifying Investor unless (and then solely to the extent) the Indemnifying Investor is prejudiced by such delay or failure. The Indemnifying Investor shall have the right at any time during which such Third-Party Claim is pending to select counsel to defend and control the defense thereof and settle any claims for which it is responsible for indemnification hereunder (provided, that the Indemnifying Investor will not settle any such Third-Party Claim without (i) the appropriate Investor Indemnified Party's prior written consent, which consent shall not be unreasonably withheld, conditioned or delayed, or (ii) obtaining an unconditional release of the appropriate Investor Indemnified Party from all claims arising out of or in any way relating to the circumstances involving such Third-Party Claim) so long as in any such event the Indemnifying Investor shall have stated in a writing delivered to the Investor Indemnified Party that, as between the Indemnifying Investor and the Investor Indemnified Party, the Indemnifying Investor is responsible to the Investor Indemnified Party with respect to such Third-Party Claim to the extent and subject to the limitations set forth herein; provided, that with respect to any Third-Party Claim as to which the Investor Indemnified Party is controlling the defense, the Indemnifying Investor will not be liable to any Investor Indemnified Party for any settlement of any claim pursuant to this Section 11.3 that is effected without its prior written consent, not to unreasonably withheld, conditioned or delayed. To the extent that the undertaking to indemnify, pay and hold harmless set forth in the preceding sentence may be unenforceable because it is violative of any Law or public policy, the Indemnifying Investor shall contribute the maximum portion which it is permitted to pay and satisfy under applicable law to the payment and satisfaction of all Investor Indemnified Liabilities incurred by the Investor Indemnified Parties or any of them. The obligations of each Investor set forth in this Section 11.3 shall survive for the survival periods set forth in Section 11.1 and, with respect to any Third-Party Claim for Investor Indemnified Liabilities made and for which proper notice has been provided hereunder prior to the date set forth in Section 11.1, until the final resolution thereof. The indemnity provided in this Section 11.3 shall be the sole and exclusive remedy of the Investor Indemnified Parties for any inaccuracy or breach of the representations and warranties set forth in this Agreement.
- 11.4 Waiver. Except for the recourse set forth in this Section 11, the Investors and the Company hereby waive any and all remedies which may be available to them, at law or equity, for any breach or inaccuracy or alleged breach or inaccuracy of the representations or warranties of any other Party hereto; provided, however, that nothing herein shall limit in any way the remedies of either Investor or the Company in respect of fraud or willful misconduct in connection with the Transactions.
- 12. <u>Publicity</u>. On the date hereof, the Company and the Investors shall issue a joint press release substantially in the form of Exhibit K hereto. No other written public release or written announcement concerning the purchase of the Series B-1 12.75% Preferred Stock, Series B-2 11.5% Preferred Stock or Warrants contemplated hereby shall be issued by any Party without the prior written consent of the other Parties (which consent shall not be unreasonably withheld, conditioned or delayed), except as such release or announcement may be required by Law or the rules or regulations of any securities exchange, in which case the Party required to make the release or announcement shall, to the extent reasonably practicable, allow the other Parties reasonable time to comment on such release or announcement in advance of such issuance. The provisions of this Section shall not restrict the ability of a Party to summarize or describe the Transactions in any prospectus or similar offering document so long as the other Parties are provided a reasonable opportunity to review such disclosure in advance.

13. Miscellaneous.

13.1 Governing Law

.. This Agreement shall be governed in all respects by the laws of the State of New York without regard to choice of laws or conflict of laws provisions thereof that would require the application of the Laws of any other jurisdiction.

13.2 Dispute Resolution.

- (a) Executive Meetings. Prior to submitting any dispute or controversy arising from or in connection with this Agreement, including the breach, termination or invalidity thereof (a "Dispute"), to arbitration pursuant to Section 13.2(b), upon written request of any Party, each Party shall appoint a designated representative whose task it will be to meet promptly for the purpose of endeavoring to resolve such Dispute. The designated representatives shall meet, in person or by telephone or video conference as deemed appropriate by the Parties, as often as the Parties reasonably deem necessary to discuss the Dispute in an effort to resolve the Dispute without the necessity of any further proceeding. The Parties agree to negotiate, in good faith, in an attempt to resolve the Dispute for a period of not greater than thirty (30) days after notice of the Dispute is received by the Parties.
- (b) <u>Arbitration; Rules; Location</u>. Any Dispute that is not resolved pursuant to Section 13.2(a) shall be referred to and finally determined under the Rules of Arbitration of the International Chamber of Commerce then in effect (the "*ICC Rules*"). The place of arbitration shall be San Francisco, California, or such other location as the Parties may agree in writing.
- (c) <u>Arbitrators</u>. There shall be three (3) arbitrators, nominated in accordance with the ICC Rules. Each arbitrator on the arbitral tribunal shall be disinterested in the Dispute and shall have no connection to any Party thereto.
- (d) <u>Award</u>. The arbitral award shall be in writing, state the reasons for the award, and be the sole and exclusive binding remedy with respect to the Dispute between and among the Parties. Judgment on the award rendered may be entered in any court having jurisdiction thereof. The Parties hereby waive any right to refer any question of law and right of appeal on the law and/or merits to any court, except as provided by the Convention on the Recognition and Enforcement of Foreign Arbitral Awards of 1958. For purposes of such convention, the award shall be deemed an award of the United States, the relationship between the Parties shall be deemed commercial in nature, and any Dispute arbitrated pursuant to this <u>Section 13.2</u> shall be deemed commercial. The arbitrators shall have the authority to grant any equitable or legal remedies that would be available in any judicial proceeding intended to resolve a Dispute.
 - (e) Language of Proceedings. The language of the arbitral proceedings shall be English.
- (f) <u>Confidentiality of Proceedings</u>. The Parties agree that any arbitration hereunder shall be kept confidential, and that the existence of the Dispute, the proceeding and all of its elements (including any pleadings, briefs or other documents submitted or exchanged, any testimony or other oral submissions, and any awards) shall be (i) prior to the First Closing, deemed Confidential Information for purposes of the confidentiality agreements referenced in <u>Section 13.7</u>, and (ii) following the First Closing, subject to the provisions of Section 7.4 of the Investor Rights Agreement, and in either case, without limiting the foregoing, shall not be disclosed beyond the tribunal, the International Court of Arbitration, the parties to the Dispute, their counsel, and any Person necessary to the conduct of the proceeding, except as and to the extent required to enforce any arbitral award, or as otherwise contemplated by such confidentiality agreements.
 - (g) Expenses. Each party hereto to a Dispute shall bear its own legal fees and costs in connection therewith.
- (h) Notwithstanding the foregoing, any Party shall be entitled to seek preliminary injunctive relief from any court of competent jurisdiction to preserve the status quo, pending the final decision or award of the arbitrators.
- (i) Each of the Parties hereby irrevocably consents to jurisdiction of any court in the United States of America (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding pursuant to Section 13.2(h), or enforcing any award under Section 13.2(d), and irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. Process in any such action or proceeding may be served on any Party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 13.8 shall be deemed effective service of process on such Party.

(j) To the extent of concurrent Disputes (as such term is used in each of the Transaction Documents) under multiple Transaction Documents, the Parties agree to consolidate any and all such Disputes into a single proceeding pursuant to the procedures set forth in this <u>Section 13.2</u>.

- 13.3 Enforcement of Agreement. The Parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with its specific terms or were otherwise breached. It is accordingly agreed that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court, subject to the provisions of Section 13.2, this being in addition to any other remedy to which they are entitled at Law or in equity, including monetary damages. Additionally, each Party hereto irrevocably waives any defenses based on adequacy of any other remedy, whether at Law or in equity, that might be asserted as a bar to the remedy of specific performance of any of the terms or provisions hereof or injunctive relief in any action brought therefor.
- 13.4 Successors and Assigns. Except as otherwise provided herein, the provisions hereof shall inure to the benefit of, and be binding upon, the successors, assigns, heirs, executors, and administrators of the Parties hereto. The rights of each Party under this Agreement shall not be assignable to any Person without the consent of each other Party; provided, however, that the Investors shall be permitted, without the consent of the Company, to assign all or a portion of their respective rights and obligations to purchase Preferred Stock or Warrants at any Closing to one or more controlled Affiliates, but only if such controlled Affiliate would, on such purchase, constitute a permitted transferee under Section 9.1 hereof (including obtaining required approvals therefore), and such Affiliate shall thereafter constitute an "Investor" for all purposes hereunder as if it were an Investor as of the date hereof, and Schedule A hereto shall be modified to reflect such assignment of rights and obligations accordingly; and provided further, that any assignment pursuant to the preceding proviso shall not relieve the assigning Investor of any obligation under this Agreement.
- 13.5 No Third Party Beneficiaries. Notwithstanding anything contained in this Agreement to the contrary, nothing in this Agreement, expressed or implied, is intended to confer on any Person other than the Parties hereto any rights, remedies, obligations or liabilities under or by reason of this Agreement, and no Person that is not a Party to this Agreement (including any partner, member, stockholder, director, officer, employee or other beneficial owner of any Party hereto, in its own capacity as such or in bringing a derivative action on behalf of a Party hereto) shall have any standing as third party beneficiary with respect to this Agreement or the transactions contemplated hereby.
- 13.6 No Personal Liability of Directors, Officers, Owners, Etc. No director, officer, employee, incorporator, stockholder, managing member, member, general partner, limited partner, principal or other agent of any of the Investors or the Company shall have any liability for any obligations of the Investors under this Agreement or for any claim based on, in respect of, or by reason of, the respective obligations of the Investors or the Company hereunder. Each Party hereto hereby waives and releases all such liability. This waiver and release is a material inducement to each Party's entry into this Agreement.
- 13.7 Entire Agreement. This Agreement and the other Transaction Documents (including the schedule and exhibits attached hereto and thereto, which are incorporated herein by reference), together with, prior to the First Closing, the Confidentiality Agreement between the Company and Toshiba, dated October 8, 2009 (in the case of Toshiba and the Company) and the Confidentiality Agreement between the Company and The Babcock & Wilcox Company, dated October 12, 2009 (in the case of B&W and the Company) constitute the entire agreement among the Parties with respect to their subject matter. This Agreement and the other Transaction Documents, together with the confidentiality agreements referenced in the foregoing sentence (prior to the First Closing and subject to Section 4.1(b) of the Strategic Relationship Agreement) supersede all previous, contemporaneous and inconsistent agreements, negotiations, representations and promises between or among the Parties, written or oral, regarding the subject matter hereof. There are no oral or written collateral representations, agreements or understandings except as provided herein and in the other Transaction Documents.

13.8 Notices, Etc.

Except as otherwise provided in this Agreement, all notices, requests, claims, demands, waivers and other communications required or permitted hereunder shall be in writing and shall be mailed by registered or certified mail, postage prepaid, return receipt requested, sent by reputable overnight courier service, delivery fees prepaid, or otherwise delivered by hand or by messenger, addressed:

(a) if to Toshiba, to:

Toshiba Corporation
1-1 Shibaura 1-chome
Minato-ku, Tokyo 105-8001
Japan
Telephone: 011-81-3457-3706
Attention: General Manager
Legal Affairs Division
Power Systems Company

With a copy to:

Morrison & Foerster LLP Shin-Marunouchi Building, 29th floor 5-1, Marunouchi 1-chome Chiyoda-ku, Tokyo 100-6529 Japan

Telephone: 011-81-3-3214-6522 Attention: Ken Siegel, Esq.

if to B&W, to:

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502 U.S.A.

Telephone: (434) 522-5589 Attention: James D. Canafax, Esq.

With a copy to:
Baker Botts LLP
The Warner
1299 Pennsylvania Avenue, NW
Washington, DC 20004-2400
U.S.A.
Telephone: (202) 639-7724

Telephone: (202) 639-7724 Attention: Michael Gold, Esq.

(b) if to any other Holder (other than the Investors), at such address as such Holder shall have furnished to the Company in writing or, until any such Holder so furnishes an address to the Company, then to and at the address of the last Holder of such Preferred Stock who has so furnished an address to the Company, and

(c) if to the Company, to:

USEC Inc. 6903 Rockledge Drive Bethesda, Maryland 20817 Telephone: (301) 564-3200 Attention: General Counsel

With a copy to:

Latham & Watkins LLP 555 Eleventh Street, NW

Suite 1000

Washington, D.C. 20004-1304 Telephone: (202) 637-2200 Attention: Scott Herlihy

or in any such case to such other address as a Party may, from time to time, designate in a written notice given in a like manner. If notice is provided by hand, mail, messenger or overnight courier service, it shall be deemed to be delivered upon actual delivery or refusal of delivery.

- 13.9 <u>Delays or Omissions</u>. No delay or omission to exercise any right, power, or remedy accruing to a Party upon any breach or default by another Party under this Agreement shall impair any such right, power, or remedy of such Party, nor shall it be construed to be a waiver of any such breach or default, or an acquiescence therein, or of or in any similar breach or default thereafter occurring; nor shall any waiver of any single breach or default be deemed a waiver of any other breach or default thereafter occurring. Any waiver, permit, consent or approval of any kind or character of any breach or default under this Agreement, or any waiver of any provisions or conditions of this Agreement, must be in writing, specifically referring to this Agreement and the rights to be waived, and shall be effective only to the extent specifically set forth in such writing and as to the specific events or rights that are waived (and not for any other or subsequent or other event giving rise to rights hereunder). All remedies, either under this Agreement or otherwise afforded to a Party, shall be cumulative and not alternative.
- 13.10 Expenses. Each Party shall bear its own expenses and legal fees incurred on its behalf with respect to this Agreement and the transactions contemplated hereby.
- 13.11 Amendments and Waivers. Any term of this Agreement may be amended only with the written consent of the Company and each Investor. Any waiver by a Party of an instance of another Party's noncompliance with any obligation or responsibility contained herein shall be in writing and signed by the waiving Party, shall only be effective to the extent set forth in such waiver, and shall not be deemed a waiver of other instances of such other Party's noncompliance hereunder. Any amendment or waiver effected in accordance with this Section 13.11 shall be binding upon each Holder of any securities purchased under this Agreement at the time outstanding (including securities into which such securities are convertible), each future holder of all such securities, and the Company.
- 13.12 <u>Rights of the Parties</u>. Each Party to this Agreement shall have the absolute right to exercise or refrain from exercising any right or rights that such Party may have by reason of this Agreement, including the right to consent to the waiver or modification of any obligation under this Agreement, and such Party shall not incur any liability to any other Party or other holder of any securities of the Company as a result of exercising or refraining from exercising any such right or rights.
- 13.13 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts and signatures may be delivered by facsimile or in electronic format (i.e., "PDF"), each of which may be executed by less than all Parties, each of which shall be enforceable against the Parties actually executing such counterparts, and all of which together shall constitute one instrument.
- 13.14 Severability. If any provision of this Agreement becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, portions of such provision, or such provision in its entirety, to the extent necessary, shall be severed from this Agreement and the balance of this Agreement shall be enforceable in accordance with its terms.
- 13.15 <u>Titles and Subtitles; Interpretation</u>. The titles and subtitles used in this Agreement are used for convenience only and are not to be considered in construing or interpreting this Agreement. When a reference is made in this Agreement to an Article, Section, Schedule or Exhibit, such reference shall be to an Article, Section, Schedule or Exhibit of this Agreement unless otherwise

indicated. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The definitions contained in Exhibit A to this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to in this Agreement means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes. Each of the Parties has participated in the drafting and negotiation of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if it is drafted by each of the Parties, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of authorship of any of the provisions of this Agreement.

- 13.16 Several Liability of Investors. Each Investor shall be severally (and not jointly and severally or jointly with any other Person) liable for its own actions, representations, warranties, covenants and obligations under this Agreement and the other Transaction Documents.
- 13.17 <u>Limitation on Damages</u>. In no event shall any Party have any liability for loss of profits, revenue or goodwill, loss or interruption of business, loss of data, or for any indirect, incidental, special, consequential or punitive damages, arising out of or relating to this Agreement or any breach hereof, no matter what theory of liability, and even if advised of the possibility or probability of such damages.

[THIS SPACE LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, the Parties have executed this Securities Purchase Agreement as of the date first above written.

USEC INC.

By: /s/ John K. Welch

Name: John K. Welch

Title: President and Chief Executive Officer

Signature page to the USEC Inc. Securities Purchase Agreement

TOSHIBA CORPORATION

By: /s/ Yasuharu Igarashi Name: Yasuharu Igarashi Title: Corporate Senior Vice President

President & CEO Power Systems Company

BABCOCK & WILCOX INVESTMENT COMPANY

By: /s/ David S. Black

Name: David S. Black Title: Vice President and Controller

Signature page to the USEC Inc. Securities Purchase Agreement

SCHEDULE A

SCHEDULE OF INVESTORS

First Closing

| Investor Toshiba B&W | Number of Shares of Series 8-1 12.75% Preferred Stock 37,500 37,500 | Aggregate Purchase Price \$ 37,500,000.00 \$ 37,500,000.00 | Warrants to Purchase Common Stock 3,125,000 3,125,000 |
|----------------------------|--|--|--|
| | Second Closing | , , , | , , |
| | Second Closing | | |
| Investor | Number of Shares of Series B-1 12,75% Preferred Stock | Aggregate Purchase Price | Warrants to Purchase Common Stock |
| Investor Toshiba | 25,000 | \$ 25,000,000.00 | 0 |
| B&W | 25,000 25,000 | \$ 25,000,000.00 | 0 |
| | Third Closing | | |
| | . | | |
| Investor | Number of Shares of Series | A gamagata Dunah asa Duias | Warrants to Purchase |
| Investor Toshiba | B-2 11.5% Preferred Stock 37,500 | Aggregate Purchase Price \$ 37,500,000.00 | 3,125,000 |
| B&W | 37,500 | \$ 37,500,000.00 | 3,125,000 |
| DC W | 37,300 | \$ 57,500,000.00 | 3,123,000 |

SCHEDULE B

(a) First Closing

(i) NRC Review

- a. NRC written concurrence that the First Closing does not constitute a direct or indirect transfer of control of the Company's ACP licenses or Gaseous Diffusion Plant certificates ("NRC Licenses") to the Investors, which transfer would otherwise be subject to NRC prior written consent under the Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including the implementation by the Company of a Negation Plan as may be required by NRC; or
- b. in the event the foregoing NRC written concurrence is not obtained, NRC prior written consent to the Company's application for a direct or indirect transfer of control of the NRC Licenses to the Investors pursuant to the Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including the implementation by the Company of a Negation Plan as may be required by NRC. Such NRC prior written consent may be obtained separately or in conjunction with the Company's application to transfer the NRC Licenses to American Centrifuge Operating LLC or other Company-affiliated entities to facilitate the DOE Loan Closing.

(ii) DOE Review

- a. DOE written concurrence that the First Closing does not result in foreign ownership, control or influence under the NISPOM ("FOCI") regarding the Company, which approval may be evidenced by receipt of an NRC written determination (in consultation with DOE) that the First Closing will not result in FOCI; or
- b. DOE written approval of a Negation Plan to address FOCI, which approval may be evidenced by receipt of an NRC written approval (in consultation with DOE) of such Negation Plan.

(b) Second Closing

(i) NRC Review1

- a. NRC written concurrence that the Second Closing does not constitute a direct or indirect transfer of control of the NRC Licenses, which transfer would otherwise be subject to NRC prior written consent under Section Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including the implementation by the Company of a Negation Plan as may be required by NRC; or
- b. in the event the foregoing NRC written concurrence is not obtained, NRC prior written consent to the Company's application for a direct or indirect transfer of control of the NRC Licenses to the Investors pursuant to the Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including the implementation by the Company of a Negation Plan as may be required by NRC. Such NRC prior written consent may be obtained separately or in conjunction with the Company's application to transfer the NRC Licenses to American Centrifuge Operating LLC or other Company-affiliated entities to facilitate the DOE Loan Closing.

(ii) DOE Review²

- a. DOE written concurrence that the Second Closing does not result in FOCI regarding the Company, which approval may be evidenced by receipt of an NRC written determination (in consultation with DOE) that the Second Closing will not result in FOCI; or
- b. DOE written approval of a Negation Plan to address FOCI, which approval may be evidenced by

receipt of an NRC written determination (in consultation with DOE) that the Second Closing will not result in FOCI.

(c) Third Closing

(i) NRC Review³

- a. NRC prior written consent to the Company's application for a direct or indirect transfer of control of the NRC Licenses to the Investors, pursuant to the Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including the implementation by the Company of a Negation Plan as may be required by NRC. Such NRC prior written consent may be obtained separately or in conjunction with the Company's application to transfer the NRC Licenses to American Centrifuge Operating LLC or other Company-affiliated entities to facilitate the DOE Loan Closing; or
- b. in the event that the foregoing prior written consent is not required, NRC written concurrence that the Third Closing does not constitute a direct or indirect transfer of control of the NRC Licenses, which transfer would otherwise be subject to NRC prior written consent under the Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including the implementation by the Company of a Negation Plan approved by NRC.

(ii) DOE Review4

- a. DOE written concurrence that the Third Closing does not result in FOCI regarding the Company which approval may be evidenced by receipt of an NRC written determination (in consultation with DOE) that the Third Closing will not result in FOCI; or
- b. DOE written approval of a Negation Plan to address FOCI and written concurrence with the Negation Plan, which may be combined with the approval of such Negation Plan, as may be required by NRC in connection with the prior written consent under C.1. (a) of this Schedule B.

Schedule B Definitions

"Atomic Energy Act" means the Atomic Energy Act of 1954, as amended (42 U.S.C. 2011 et seq.).

"Classified Information" means (i) information classified as either Restricted Data or Formerly Restricted Data or (ii) National Security Information.

"DOE Loan Closing" means the closing of a DOE loan guarantee arrangement between the DOE and the Company, pursuant to the Conditional Commitment, providing not less than \$2 billion of governmental financing for the ACP and contemporaneous funding, or current availability to the Company pursuant to definitive loan agreements, of Third Party Financing.

"Export Controlled Information" means any unclassified information, the export of which is controlled by law or regulation under the International Traffic in Arms Regulations (22 CFR Part 120 et seq.), the Export Administration Regulations (14 CFR Part 730 et seq.), or the U.S. Department of Energy regulations (10 CFR Part 810), and including Unclassified Controlled Nuclear Information, Safeguards Information (Section 147 of the Atomic Energy Act of 1954 as amended), Sensitive Unclassified Non-Safeguards Information (NRC Regulatory Information Summary 2005-31), and Official Use Only information (DOE Order 471.3), except to the extent that the export or deemed export thereof arising by virtue of a disclosure hereunder has been licensed or approved under, or is authorized by, Applicable Law.

"Formerly Restricted Data" means information jointly determined by DOE and the Department of Defense to be related primarily to the military utilization of nuclear weapons and removed (by transclassification) from the Restricted Data category pursuant to section 142(d) of the Atomic Energy Act.

"National Security Information" means information that has been determined pursuant to Executive Order 12958, as amended (68 Federal Register 15315 (March 28, 2003)), or prior or subsequent Executive Orders to require protection against unauthorized disclosure and is marked to indicate its classification status when in document form. National Security Information is referred to as 'defense information' in the Atomic Energy Act.

"Negation Plan" means (i) a plan to negate foreign ownership, control or influence required by the DOE

or the NRC, as necessary, for the protection of Classified Information and Export Controlled Information in accordance with applicable regulations and the NISPOM or (ii) a plan to negate foreign ownership, control or domination as may be required by the NRC in connection with a license transfer application.

"NISPOM" means the National Industrial Security Program Operating Manual as required by Executive Order 12829 and under the authority of Department of Defense Directive 5220.22, "National Industrial Security Program (NISP)" or subsequent Executive Orders, directives, or regulations for the protection of classified information released or disclosed to industry in connection with classified contracts under the NISP.

"NRC" means the U.S. Nuclear Regulatory Commission, and any successor Governmental Authority thereto.

"Restricted Data" means a kind of classified information that consists of all data concerning the following, but not including data declassified or removed from the Restricted Data category pursuant to section 142 of the Atomic Energy Act: (i) design, manufacture, or utilization of atomic weapons; (ii) production of special nuclear material; or (iii) use of special nuclear material in the production of energy.

"Unclassified Controlled Nuclear Information" means certain unclassified Government information concerning nuclear facilities, materials, weapons, and components whose dissemination is controlled under section 148 of the Atomic Energy Act and 10 CFR 1017.

- May be included in NRC action in connection with First Closing as set forth in Section A.1. of this Schedule B.
- ² May be included in DOE action in connection with First Closing as set forth in Section A.2. of this Schedule B.
- May be included in NRC action in connection with First Closing or Second Closing as set forth in Section A.1 or B.1 of this Schedule B.
- May be included in DOE action in connection with First Closing or Second Closing as set forth in Section B.1 or B.2. of this Schedule B.

SCHEDULE C

- Bank of Am\Merrill Lynch
- Citigroup
- Credit Suisse
- JP Morgan
- UBS
- Morgan Stanley

EXHIBIT A

DEFINITIONS

The following terms shall have the respective meanings for all purposes of the Agreement:

- "2002 Agreement" shall mean that certain Agreement between the Company and the DOE, dated July 17, 2002, as amended.
- "2008 Solicitation" shall have the meaning ascribed to it in Section 4.2(e).
- "ACM" shall have the meaning ascribed to it in Section 4.2(g).
- "ACP" shall mean the design, manufacture, construction, development, startup, completion, operation, financing, maintenance and improvement of a front-end nuclear fuel facility utilizing U.S. gas centrifuge enrichment technology and related infrastructure assets and properties.
- "Affiliate" shall mean any Person controlling, controlled by or under common control with any other Person. For purposes of this definition, "control" (including "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of securities, partnership or other ownership interests, by contract or otherwise.
 - "Agreement" shall have the meaning ascribed to it in the Preamble.
- "Approved Market" shall mean, if the Common Stock is then listed or quoted thereon, the New York Stock Exchange, The NASDAQ Stock Market or the American Stock Exchange.
 - "Base Price" shall have the meaning set forth in the applicable Certificate of Designation.
 - "Base Price Determination Date" shall have the meaning set forth in the applicable Certificate of Designation.
 - "Board" shall mean the board of directors of the Company or any duly authorized committee thereof.
 - "brokers' transaction" shall have the meaning ascribed to such term under Rule 144(g) under the Securities Act.
- "Business Day" shall mean any calendar day other than (1) a Saturday or Sunday or (2) a calendar day on which banking institutions in either the City of New York or Tokyo, Japan are authorized by Law, regulation or executive order to remain closed.
 - "Bylaws" shall have the meaning ascribed to it in Section 2.1.
 - "B&W" shall have the meaning ascribed to it in the Preamble.
- "Capital Lease Obligations" shall mean the obligations of the Company and its Subsidiaries on a consolidated basis to pay rent or other amounts under a lease of (or other agreement conveying the right to use) real and/or personal Property which obligations are required to be classified and accounted for as a capital lease on a consolidated balance sheet of the Company and its Subsidiaries under Generally Accepted Accounting Principles (including Statement of Financial Accounting Standards No. 13 of the Financial Accounting Standards Board, as amended) and, for purposes of this Agreement, the amount of such obligations shall be the capitalized amount thereof, determined in accordance with Generally Accepted Accounting Principles (including such Statement No. 13).
 - "Certificate of Incorporation" shall have the meaning ascribed to it in Section 2.1.
 - "Certificates of Designation" shall have the meaning ascribed to it in Section 1.1.
 - "CFIUS" shall have the meaning ascribed to it in Section 2.9.
 - "Charter Amendment" shall mean that certain amendment to the Certificate of Incorporation, substantially in the

form attached hereto as Exhibit L.

- "Charter Amendment Approval" shall have the meaning ascribed to it in Section 6.1(f).
- "Class B Common Stock" shall have the meaning ascribed to it in Section 1.2(c).
- "Closing" shall mean the First Closing, the Second Closing or the Third Closing, as applicable.
- "Closing Representations" shall mean those representations and warranties contained in Section 2 as made by the Company at a Closing.
- "Code" shall mean the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder by the Internal Revenue Service.
 - "Commission" shall have the meaning ascribed to it in Section 2.
 - "Common Stock" shall mean collectively, the Ordinary Common Stock and the Class B Common Stock.
 - "Company" shall have the meaning ascribed to it in the Preamble.
 - "Company Action" shall have the meaning ascribed to it in 7.13(a).
 - "Company Effect" shall have the meaning ascribed to it in the definition of "Material Adverse Effect."
 - "Company Indemnified Liabilities" shall have the meaning ascribed to it in Section 11.2.
 - "Company Indemnified Parties" shall have the meaning ascribed to it in Section 11.2.
- "Company Plans" shall mean the Company's 1999 Equity Incentive Plan, as amended, and the Company's 2009 Equity Incentive Plan, as may be amended, the Company's 1999 Employee Stock Purchase Plan, as may be amended, and any similar plans entered into after the date hereof, and any inducement grants.
 - "Company Rights Agreement" shall have the meaning ascribed to it in Section 2.13.
 - "Competitor" shall have the meaning ascribed to it in the Strategic Relationship Agreement.
- "Conditional Commitment" shall mean a conditional commitment (as defined in 10 CFR 609.2) from DOE to the Company in an amount not less than \$2 billion, and specifying the detailed conditions to be satisfied for the DOE Financial Closing.
- "Contingent Obligations" shall mean, as to any Person, without duplication, any obligation of such Person guaranteeing or intended to guarantee the payment or performance of any Indebtedness, leases, dividends or other obligations (collectively "primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including any obligation of the Person for whom Contingent Obligations is being determined, whether or not contingent, (a) to purchase any such primary obligation or other property constituting direct or indirect security therefor, (b) assume or contingently agree to become or be secondarily liable in respect of any such primary obligation, (c) to advance or supply funds (i) for the purchase or payment of any such primary obligation or (ii) to maintain working capital or equity capital for the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (d) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation, or (e) otherwise to assure or hold harmless the owner of such primary obligation against loss in respect thereof; provided, however, that the term "Contingent Obligations" shall not include endorsements of checks or other negotiable instruments in the ordinary course of business. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determinable amount of the primary obligation in respect of which such Contingent Obligation is made or, if not stated or determinable, the maximum anticipated liability in respect thereof (assuming the Person for whom Contingent Obligations is being determined is required to perform thereunder).
 - "Convertible Notes" shall have the meaning ascribed to it in Section 2.13.
 - "Credit Facility" shall mean the Second Amended and Restated Revolving Credit Agreement, dated as of February

26, 2010, among the Company and United States Enrichment Corporation, as joint and several co-borrowers; the Lenders party thereto from time to time; J.P. Morgan Chase Bank, N.A., as Administrative and Collateral Agent; J.P. Morgan Securities, Inc., Wachovia Capital Finance Corporation (New England) and UBS Securities LLC, as Joint Book Managers and Joint Lead Arrangers; Wachovia Capital Finance Corporation (New England), as Syndication Agent; and UBS Securities LLC, as Documentation Agent (each of the foregoing capitalized terms not otherwise defined herein to be defined as set forth in the Credit Facility) as amended, restated, modified, renewed, refunded, replaced in any manner (whether upon termination or otherwise) or refinanced (including by means of sales of debt securities) in whole or in part from time to time (except for purposes of clause (a)(iii) of Section 7.10, the funds borrowed thereunder cannot exceed \$350,000,000 outstanding).

- "Designated Representations" shall mean the Company's representations and warranties contained in Sections 2.3, 2.10(c) and 2.12 made as of the date hereof, and each of the foregoing representations and warranties made as Closing Representations at any Closing.
 - "Disclosure Letter" shall have the meaning ascribed to it in Section 2.
 - "Dispute" shall have the meaning ascribed to it in Section 13.2(a).
 - "DOE" shall have the meaning ascribed to it in Section 4.2(e).
- "DOE Financial Closing" shall mean the closing of a Loan Guarantee Agreement, as defined in 10 CFR 609.2, between DOE, an eligible lender, and the Company, pursuant to the Conditional Commitment, guaranteeing a loan or other debt obligation in an amount not less than \$2 billion for the ACP and there shall have been an initial draw of the funds guaranteed pursuant to the Loan Guarantee Agreement in an amount not less than the minimum amount permitted thereunder.
 - "DOJ" shall have the meaning ascribed to it in Section 7.4(a).
 - "Enforceability Exceptions" shall have the meaning ascribed to it in Section 2.3(a).
- "Environmental Claim" shall mean any third party (including any Governmental Authority) action, lawsuit, claim, investigation, or proceeding (including claims or proceedings at common law) which seeks to impose, allege or investigate liability for (i) preservation, protection, conservation, pollution, contamination of, or releases or threatened releases of Hazardous Substances into the air, surface water, ground water or land or the clean-up, abatement, removal, remediation or monitoring of such pollution, contamination or Hazardous Substances; (ii) generation, recycling, reclamation, handling, treatment, storage, disposal or transportation of Hazardous Substances or solid waste (as defined under the Resource Conservation and Recovery Act and its regulations, as amended from time to time); (iii) exposure to Hazardous Substances; (iv) the safety or health of employees or other Persons in connection with any of the activities specified in any other subclause of this definition; or (v) the manufacture, processing, distribution in commerce, presence or use of Hazardous Substances. An "Environmental Claim" includes a common law action, as well as a proceeding to issue, modify or terminate an Environmental Permit, or to adopt or amend a regulation to the extent that the Company or its Subsidiaries are parties to such a proceeding and such a proceeding attempts to redress violations of the applicable permit, license, or regulation as alleged by any Governmental Authority.
- "Environmental Liabilities" shall mean all liabilities arising from any Environmental Claim, Environmental Permit or Requirement of Environmental Law under any theory of recovery, at law or in equity, and whether based on negligence, strict liability or otherwise, including: remedial, removal, response, abatement, restoration (including natural resources) investigative, or environmental or medical monitoring liabilities, personal injury, and damage to property, natural resources, or injuries or increased risks or injuries to persons, and any other related costs, expenses, losses, damages, penalties, fines, liabilities and obligations, and all costs and expenses necessary to cause the issuance, re-issuance or renewal of any Environmental Permit including attorney's fees and court costs. "Environmental Liability" shall mean any one of them.
- "Environmental Permit" shall mean any permit, license, approval or other authorization under any applicable law, regulation and other requirement of the United States or of any state, municipality or other subdivision thereof relating to pollution or protection of health or the environment, including laws, regulations or other requirements relating to emissions, discharges, releases or threatened releases of pollutants, contaminants or Hazardous Substances or toxic materials or wastes into ambient air, surface water, ground water or land, or otherwise relating to the manufacture, processing, distribution, recycling, presence, use, treatment, storage, disposal, transport or handling of, wastes, pollutants, contaminants or Hazardous Substances.
- "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and all rules, regulations, rulings and interpretations adopted by the Internal Revenue Service or the Department of Labor thereunder.

- "ERISA Affiliate" shall mean any trade or business (whether or not incorporated) that, together with the Company or any of its Subsidiaries, is treated as a single employer under Section 414(b) or (c) of the Code or, solely for purposes of Section 302 of ERISA and Section 412 of the Code, is treated as a single employer under Section 414 of the Code.
 - "Exchange Act" shall have the meaning ascribed to it in Section 2.
- "FINSA" shall mean the Foreign Investment and National Security Act of 2007, 50 U.S.C. App. 2061, amending the Defense Production Act of 1950, 50 U.S.C. App. 2170.
 - "First Closing" shall have the meaning ascribed to it in Section 1.3(a).
- "Foreign Ownership Restrictions Determination" shall mean that certain irrevocable determination pursuant to Article Eleventh of the Certificate of Incorporation attached as Exhibit G.
 - "FTC" shall have the meaning ascribed to it in Section 7.4(a).
- "Generally Accepted Accounting Principles" shall mean those United States generally accepted accounting principles and practices which are (a) recognized as such by the Financial Accounting Standards Board or successor organization, and (b) applied by the Company and its Subsidiaries in a manner consistent with the manner in which such principles and practices were applied to the audited financial statements of the Company set forth in its Annual Report filed on Form 10-K for the year ended December 31, 2009.
- "Governmental Authority" shall mean any foreign governmental authority, the United States of America, any state of the United States and any political subdivision of any of the foregoing, and any agency, instrumentality, department, commission, board, bureau, central bank, authority, court, arbitral body or other tribunal, in each case whether executive, legislative, judicial, regulatory or administrative, having jurisdiction over the Investor, any of the Holders or the Company, any of the Company's Subsidiaries or their respective Property.
- "Hazardous Substance" shall mean any hazardous or toxic waste, substance or product or material defined or regulated by any applicable law, rule, regulation or order described in the definition of "Requirements of Environmental Law," including solid waste (as defined under the Resource Conservation and Recovery Act of 1976 or its regulations, as amended), asbestos, petroleum and any fraction thereof, and any radioactive materials and waste.
- "Hedging Agreements" shall mean any transaction (including an agreement with respect thereto) now or hereafter existing which is a rate swap, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, bond option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collar transaction, forward transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any of these transactions) or any combination thereof, whether linked to one or more interest rates, foreign currencies, commodity prices, equity prices or other financial measures.
 - "Holders" shall have the meaning ascribed to it in the Preamble.
 - "HSR Act" shall have the meaning ascribed to it in Section 2.9.
 - "ICC Rules" shall have the meaning ascribed to it in Section 13.2(b).

"Incidental Liens" shall mean (i) Liens for taxes, assessments, levies or other governmental charges (but not Liens for clean up expenses arising pursuant to Requirements of Environmental Law) not yet due (subject to applicable grace periods) or which are being contested in good faith and by appropriate proceedings if adequate reserves with respect thereto are maintained on the books of the Company in accordance with Generally Accepted Accounting Principles; (ii) carriers', warehousemen's, mechanics', landlords', vendors', materialmen's, repairmen's, sureties' or other like Liens (other than Liens for clean up expenses arising pursuant to Requirements of Environmental Law) arising in the ordinary course of business (or deposits to obtain the release of any such Lien) and securing amounts not yet due or which are being contested in good faith and by appropriate proceedings if, in the case of such contested Liens, adequate reserves with respect thereto are maintained on the books of the Company in accordance with Generally Accepted Accounting Principles; (iii) pledges or deposits in connection with workers' compensation, unemployment insurance and other social security legislation; (iv) easements, rights-of-way, covenants, reservations, exceptions, encroachments, zoning and similar restrictions and other similar encumbrances or title defects incurred in the ordinary course of business which, in the aggregate, are not substantial in amount, and which do not in any case singly or in the aggregate materially detract from the value or usefulness of the

property subject thereto or materially interfere with the ordinary conduct of the business of the Company and its Subsidiaries, taken as a whole; (v) bankers' liens arising by operation of law; (vi) Liens arising pursuant to any order of attachment, distraint or similar legal process arising in connection with any court proceeding the payment of which is covered in full (subject to customary deductibles) by insurance; (vii) inchoate Liens arising under ERISA to secure contingent liabilities of the Company; (viii) rights of lessees and sublessees in assets leased by the Company or any Subsidiary not prohibited elsewhere herein and (ix) Liens arising under or related to the Credit Facility.

"Indebtedness" shall mean, as to any Person, without duplication: (a) all indebtedness (including principal, interest, fees and charges) of such Person for borrowed money or for the deferred purchase price of Property or services; (b) any other indebtedness which is evidenced by a promissory note, bond, debenture or similar instrument; (c) any obligation under or in respect of outstanding letters of credit, acceptances and similar obligations created for the account of such Person; (d) all Capital Lease Obligations of such Person; (e) all indebtedness, liabilities, and obligations secured by any Lien on any Property owned by such Person even though such Person has not assumed or has not otherwise become liable for the payment of any such indebtedness, liabilities or obligations secured by such Lien; (f) any obligation under or in respect of Hedging Agreements and (g) all Contingent Obligations and Synthetic Indebtedness of such Person.

- "Indemnifying Investor" shall have the meaning ascribed to it in Section 11.3.
- "Investor" shall have the meaning ascribed to it in the Preamble.
- "Investor Indemnified Liabilities" shall have the meaning ascribed to it in Section 11.3.
- "Investor Indemnified Parties" shall have the meaning ascribed to it in Section 11.3.
- "Investor Rights Agreement" shall mean an Investor Rights Agreement by and among the Parties substantially in the form attached as Exhibit F.
- "Investors" shall have the meaning ascribed to it in the Preamble.
- "Knowledge" of the Company shall mean the actual knowledge of any of the following individuals: John Welch, Chief Executive Officer; John Barpoulis, Chief Financial Officer; Peter Saba, Senior Vice President, General Counsel and Secretary; Philip Sewell, Senior Vice President, American Centrifuge and Russian HEU; Robert Van Namen, Senior Vice President, Uranium Enrichment; and Dennis J. Scott, Assistant General Counsel and Director, Corporate Compliance.
 - "Laws" shall have the meaning ascribed to it in Section 2.17.
- "Lien" shall mean any mortgage, pledge, charge, encumbrance, security interest, collateral assignment or other lien or restriction of any kind, whether based on common law, constitutional provision, statute or contract, and shall include reservations, exceptions, encroachments, easements, rights of way, covenants, conditions, restrictions, leases and other title exceptions.
 - "Loan Guarantee Agreement" shall have the meaning ascribed to it in Section 4.2(f).
- "Loan Guarantee Program" shall mean the loan guarantee program at DOE implementing Title XVII Incentives for Innovative Technologies of the Energy Policy Act of 2005 (42 U.S.C. 16511-16514).
- "Material Adverse Effect" means any change, development, occurrence or event (each, a "Company Effect") that, when considered either individually or together with all other Company Effects, is or would reasonably be expected to be materially adverse to (a) the business, properties, assets, liabilities, consolidated results of operations or financial condition of the Company and its Subsidiaries, taken as a whole or (b) the ability of the Company to consummate the Transactions; provided that any such Company Effect shall not be deemed to include the effects of the following when determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur:
- (i) any change, development, occurrence or event generally affecting the businesses or industries in which the Company and its Subsidiaries operate;
- (ii) any conditions affecting the United States general economy or the general economy in any geographic area in which the Company or its Subsidiaries operate or developments or changes therein or the financial and securities markets and credit markets in the United States or elsewhere in the world;
 - (iii) any conditions resulting from natural disasters;

(iv) global or regional political conditions, including acts of war (whether or not declared), armed hostilities and terrorism, or similar

developments;

(v) any decision by DOE to enter into a Conditional Commitment or a Loan Guarantee Agreement with any Person other than the Company or any of its Subsidiaries;

- (vi) changes in any Laws or Generally Accepted Accounting Principles;
- (vii) any action taken or omitted to be taken by or at the written request or with the written consent of the Investor;
- (viii) any announcement of the execution or filing of any Transaction Documents, or consummation of any Closing, in each case, solely to the extent due to such announcement;
- (ix) changes in the market price or trading volume of Common Stock or any other equity, equity-related or debt securities of the Company or its Affiliates (it being understood that the underlying circumstances, events or reasons giving rise to any such change may be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur); or
- (x) any failure to meet any internal or public projections, forecasts, estimates or guidance for any period (it being understood that the underlying circumstances, events or reasons giving rise to any such failure may be taken into account in determining whether a Material Adverse Effect has occurred or would reasonably be expected to occur).

provided, however, that Company Effects set forth in clauses (i), (ii), (iii), (iv) and (vi) above may be taken into account in determining whether there has been or is a Material Adverse Effect if and only to the extent such Company Effects have a disproportionate impact on the Company and its Subsidiaries, taken as a whole, relative to other companies in the Company's industry.

- "Minimum Preferred Holdings" shall have the meaning ascribed to it in Section 7.10(b).
- "Multiemployer Plan" shall mean a multiemployer plan as defined in Section 4001(a)(3) of ERISA.
- "Negation Plan" shall have the meaning ascribed to it in Section 7.3.
- "Orderly Sale Arrangement' shall have the meaning ascribed to it in Section 9.2(b).
- "Ordinary Common Stock" shall have the meaning ascribed to it in Section 2.13.
- "Parties" shall have the meaning ascribed to it in the Preamble.
- "Permit" shall mean any approval, authorization, certificate, consent, license or permit of or from any Governmental Authority.
- "Permitted Redemption" shall mean the Company's redemption of shares of Preferred Stock in accordance with the Certificates of Designation.
- "Person" shall mean any individual, corporation, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization, Governmental Authority or any other entity.
- "Plan" shall mean any employee pension benefit plan (as defined in Section 3(2)(A) of ERISA) subject to Title IV of ERISA and maintained for employees of the Company or of any member of a "controlled group," as such term is defined in Section 4001(a)(14) of ERISA, of which the Company or any of its Subsidiaries it may acquire from time to time is a part, or any such employee pension benefit plan to which the Company or any of its Subsidiaries it may acquire from time to time is required to contribute on behalf of its employees.
 - "Preferred Stock" shall mean the Series B-1 12.75% Preferred Stock, the Series B-2 11.5% Preferred Stock and the Series C Preferred Stock.
 - "Property" shall mean any interest in any kind of property or asset, whether real, personal or mixed, tangible or

intangible.

- "Proxy Statement' shall have the meaning ascribed to it in Section 7.11(a).
- "Regulatory Approvals" shall mean the approvals or consents of the Governmental Authorities set forth on Schedule B, including any required Negation Plans.
- "Regulatory Bodies" shall mean the DOE and the U.S. Nuclear Regulatory Commission, and any successor Governmental Authorities thereto.
- "Regulatory Restrictions" shall have the meaning ascribed to it in the Certificate of Incorporation (in effect at the date hereof).
- "Reprogramming" shall have the meaning ascribed to it in Section 4.2(e).
- "Requirements of Environmental Law" shall mean all requirements imposed by any law (including the Resource Conservation and Recovery Act, the Comprehensive Environmental Response, Compensation, and Liability Act, the Clean Water Act, the Clean Air Act, and any state analogues of any of the foregoing), rule, regulation or order of any Governmental Authority which relate to (i) noise; (ii) pollution, protection or clean-up of the air, surface water, ground water or land; (iii) solid, gaseous or liquid waste or Hazardous Substance generation, recycling, reclamation, release, threatened release, treatment, storage, disposal or transportation; (iv) exposure of Persons or property to Hazardous Substances; (v) the safety or health of employees or other Persons or (vi) the manufacture, presence, processing, distribution in commerce, use, discharge, releases, threatened releases, emissions or storage of Hazardous Substances into the environment. Requirement of Environmental Law shall mean any one of them.
 - "Second Closing" shall have the meaning ascribed to it in Section 1.3(b).
 - "SEC Reports" shall have the meaning ascribed to it in Section 2.
 - "Securities" shall have the meaning ascribed to it in Section 3.1(a).
 - "Securities Act" shall have the meaning ascribed to it in Section 2.9.
 - "Series A Preferred Stock" shall have the meaning ascribed to it in Section 2.13.
 - "Series B Preferred Stock" shall mean the Series B-1 12.75% Preferred Stock and the Series B-2 11.5% Preferred Stock.
 - "Series B-1 12.75% Preferred Stock" shall have the meaning ascribed to it in Section 1.2(a).
 - "Series B-1 Certificate of Designation" shall have the meaning ascribed to it in Section 1.1.
 - "Series B-2 11.5% Preferred Stock" shall have the meaning ascribed to it in Section 1.2(c).
 - "Series B-2 Certificate of Designation" shall have the meaning ascribed to it in Section 1.1.
 - "Series C Certificate of Designation" shall have the meaning ascribed to it in Section 1.1.
 - "Series C Preferred Stock" shall mean the Company's Series C Convertible Participating Preferred Stock, par value \$1.00 per share.
 - "Share Issuance Approval" shall have the meaning ascribed to it in Section 6.1(f).
- "Signing Disclosures" shall mean matters disclosed in the Disclosure Letter as of the date hereof and the SEC Reports filed through the date hereof (and for the avoidance of doubt, without taking into account any amendments to the Disclosure Letter or matters disclosed in SEC Reports filed after the date hereof).
 - "Stockholder Approvals" shall have the meaning ascribed to it in Section 6.1(f).
- "Strategic Plan" shall mean the strategic plan then in effect for the operation of the Company's business, initially in the form attached to the Strategic Relationship Agreement.

- "Strategic Relationship Agreement" shall mean the Strategic Relationship Agreement by and among Toshiba, The Babcock & Wilcox Company and the Company, dated as of the date hereof.
- "Subsidiary" of any Person shall mean any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than fifty percent (50%) of (a) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (b) the interest in the capital or profits of such partnership, joint venture or limited liability company or (c) the beneficial interest in such trust or estate is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries. Notwithstanding the foregoing, American Centrifuge Manufacturing, LLC, a Delaware limited liability company, shall not be considered a Subsidiary of B&W or the Company.
- "Synthetic Indebtedness" shall mean the monetary obligation of a Person under (a) a so-called synthetic, off-balance sheet or tax retention lease, or (b) an agreement for the use or possession of property creating obligations that do not appear on the balance sheet of such Person (excluding operating leases) but which upon the insolvency or bankruptcy of such Person, to the extent functioning as debt for borrowed money, would be characterized as the indebtedness of such Person (without regard to accounting treatment).
 - "Third Closing" shall have the meaning ascribed to it in Section 1.3(c).
 - "Third Closing Termination Date" shall have the meaning ascribed to it in Section 10.3(a).
 - "Third-Party Claim" shall have the meaning ascribed to it in Section 11.2.
 - "Toshiba" shall have the meaning ascribed to it in the Preamble.
- "Trading Day" shall mean any day on which shares of the Company's equity securities are traded, or able to be traded, on the Approved Market on which shares of the Company's equity securities are listed or traded.
- "Transaction Documents" shall mean this Agreement, the Certificates of Designation, the Charter Amendment, the Investor Rights Agreement, and the Strategic Relationship Agreement.
 - "Transactions" shall mean the transactions contemplated by the Transaction Documents.
- "U.S. Person" shall mean any person that is treated as a "United States Person" under Code Section 7701(a)(30) and that provides an IRS Form W-9 (or successor form), evidencing a complete exemption from United States withholding tax (including backup withholding tax), on or before the time at which it acquires securities pursuant to this Agreement.
 - "Warrants" shall have the meaning ascribed to it in Section 1.2(a).
- "Wholly-Owned Affiliate" shall mean, as to any Person, any Affiliate that, directly or indirectly, is wholly-owned and controlled (other than by contract) by a Person.

EXHIBIT B

FORM OF SERIES B-1 CERTIFICATE OF DESIGNATION

FORM OF

CERTIFICATE OF DESIGNATION OF

SERIES B-1 12.75% CONVERTIBLE PREFERRED STOCK

ωf

USEC INC.

Pursuant to Section 151 of the General Corporation Law

of the State of Delaware

We, James R. Mellor, Chairman of the Board, and Peter B. Saba, Secretary, of USEC Inc., a corporation organized and existing under the General Corporation Law (the "DGCL") of the State of Delaware (the "Corporation"), in accordance with the provisions of Section 103 thereof, DO HEREBY CERTIFY:

That pursuant to the authority conferred upon the Board of Directors by the Certificate of Incorporation, the Board of Directors on May 24, 2010 adopted the following resolution creating a series of 300,000 shares of Preferred Stock, par value \$1.00 per share, designated as Series B-1 12.75% Convertible Preferred Stock:

RESOLVED, that pursuant to the authority vested in the Board of Directors in accordance with the provisions of the Certificate of Incorporation, a series of Preferred Stock of the Corporation be and it hereby is created, and that the designation and amount thereof and the powers, preferences and relative, participating, optional and other rights of the shares of such series, and the qualifications, limitations and restrictions thereof are as follows:

Section 1. Designation. The designation of this series of Preferred Stock, par value \$1.00 per share, of the Corporation is "Series B-1 12.75% Convertible Preferred Stock" ("Series B-1 12.75% Preferred Stock"). Each share of Series B-1 12.75% Preferred Stock shall be identical in all respects to every other share of Series B-1 12.75% Preferred Stock.

Section 2. Number of Shares. The authorized number of shares of Series B-1 12.75% Preferred Stock is 300,000. Shares of Series B-1 12.75% Preferred Stock that are redeemed, purchased or otherwise acquired by the Corporation, or converted into another series of Preferred Stock or into Common Stock, shall revert to authorized but unissued shares of Preferred Stock and shall not be reissued as shares of Series B-1 12.75% Preferred Stock.

Section 3. Definitions. As used herein with respect to Series B-1 12.75% Preferred Stock:

- (a) "ACP" shall mean the design, manufacture, construction, development, startup, completion, operation, financing, maintenance and improvement of a frontend nuclear fuel facility utilizing U.S. gas centrifuge enrichment technology and related infrastructure assets and properties.
- (b) "Affiliate" shall mean any Person controlling, controlled by or under common control with any other Person. For purposes of this definition, "control" (including "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of securities, partnership or other ownership interests, by contract or otherwise.
- (c) "Aggregate Outstanding Value" shall mean, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock, plus (2) for each share of Series C Preferred Stock then held by the Permitted Holders, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which the Permitted Holders' acquisition of such share was calculated, plus (3) for each share of Common Stock then held by the Permitted Holders, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Ordinary Common Stock purchased in the market, the Base Price upon which the Permitted Holders' acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock which have been added to the Liquidation Preference pursuant to Section 5(a).

- (d) "Approved Market" shall have the meaning ascribed to it in the definition of "Base Price."
- (e) "Automatic Redemption" shall mean an automatic redemption pursuant to Section 7(g) of this Certificate of Designation subsequent to a Conversion Election, Section 8(c) of this Certificate of Designation or Section 8(c) of the Series B-2 Certificate of Designation.
- (f) "Automatic Redemption Adjustment" shall mean, for purposes of determining the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value, that if an Automatic Redemption has been effected prior to the date of determining such values, (1) the aggregate amount of the Liquidation Preference, as of the date of redemption, of a Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Aggregate Outstanding Value and Permitted Holder Outstanding Value and (2) the aggregate amount of the Liquidation Preference, as of the date of redemption, of such Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Original Issue Value; provided, however, that, if at any time after any Automatic Redemption, such Permitted Holder's Deemed Holder Percentage is less than 8%, then such adjustment to the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value shall not be made.
 - (g) "B&W" shall mean Babcock & Wilcox Investment Company, a Delaware corporation.
- (h) "Base Price" shall mean for any date, the price determined by the first of the following clauses that applies: (1) if the Ordinary Common Stock is then listed or quoted on the New York Stock Exchange, The NASDAQ Stock Market or the American Stock Exchange (each an "Approved Market"), the arithmetic average of the daily volume weighted average prices per share of the Ordinary Common Stock for each of the 20 consecutive Trading Days immediately preceding (but not including) such date, as reported for the regular trading session (including any extensions thereof) on the primary Approved Market on which the Ordinary Common Stock is then listed or quoted (without regard to pre-open or after hours trading outside of such regular trading session on such Trading Day), as reported by Bloomberg Financial L.P. (or any successor thereof) using the HP function (or any equivalent thereof); (2) if the Ordinary Common Stock has not been listed or quoted on an Approved Market for a minimum of 20 consecutive Trading Days immediately preceding (but not including) such date and if prices for the Ordinary Common Stock are then quoted on the OTC Bulletin Board, the arithmetic average of the daily volume weighted average prices per share of the Ordinary Common Stock for each of the 20 consecutive Trading Days immediately preceding (but not including) such date, as quoted for the regular trading session on the OTC Bulletin Board; (3) if the Ordinary Common Stock has not been listed or quoted on the OTC Bulletin Board for a minimum of 20 consecutive Trading Days immediately preceding (but not including) such date and if prices for the Ordinary Common Stock are then reported in the "Pink Sheets" published by the National Quotation Bureau Incorporated (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Ordinary Common Stock so reported; or (4) in all other cases, the fair market value of a share of Ordinary Common Stock as determined by the Board of Directo
- (i) "Beneficially Own" shall mean "beneficially own" as defined in Rule 13d-3 promulgated under Section 13(d) of the Exchange Act or any successor provisions thereto, and "Beneficial Ownership" shall have a correlative meaning.
 - (j) "Board of Directors" shall mean the board of directors of the Corporation or any duly authorized committee thereof.
- (k) "Business Day" shall mean any calendar day other than (1) a Saturday or Sunday or (2) a calendar day on which banking institutions in either the City of New York or Tokyo, Japan are authorized by law, regulation or executive order to remain closed.
 - (I) "Bylaws" shall mean the Amended and Restated Bylaws of the Corporation, as amended from time to time.
- (m) "Certificate of Designation" shall mean this Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Corporation, as amended from time to time.
 - (n) "Certificate of Incorporation" shall mean the Certificate of Incorporation of the Corporation, as amended from time to time.
 - (o) "Change of Control" shall mean the occurrence of any of the following:

- (1) any Person shall Beneficially Own, directly or indirectly, through a merger, business combination, purchase, or other transaction or series of transactions, shares of the Corporation's capital stock entitling such Person at such time to exercise 50% or more of the total voting power of the issued and outstanding capital stock having ordinary voting power to elect a majority of the Board of Directors, other than as a result of an acquisition of such stock by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this subsection (1), "Person" shall include any syndicate or group that would be deemed to be a "person" under Section 13(d)(3) of the Exchange Act);
- (2) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Stock is exchanged for or converted into cash, securities or other property, in each case other than a merger or consolidation:
 - (i) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Stock;
 - (ii) that is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding shares of the Common Stock solely into shares of any class or series of common stock of the surviving entity; or
 - (iii) where the issued and outstanding capital stock having voting power to vote generally to elect a majority of the Board of Directors outstanding immediately prior to such transaction is converted into or exchanged for such voting stock of the surviving or transferee Person constituting a majority of the outstanding shares of such voting stock of such surviving or transferee Person (immediately after giving effect to such issuance).
- (p) "Charter Amendment Approval" shall mean the approval of the stockholders of the Corporation necessary to amend the Corporation's Certificate of Incorporation to approve the authorization of Class B Common Stock and the proper filing of such amendment with the Secretary of State of the State of Delaware.
- (q) "Class B Common Stock" shall mean the Class B Common Stock of the Corporation, par value \$.10 per share, to be authorized by the Charter Amendment Approval.
- (r) "Closing Deadline Failure" shall mean, unless waived in writing (1) by the Corporation if such Closing Deadline Failure is as a result of breach by a Permitted Holder, (2) by the Permitted Holders if such Closing Deadline Failure is as a result of breach by the Corporation, or (3) by the Permitted Holders and the Corporation if such Closing Deadline Failure is not as a result of a breach by the Permitted Holders or the Corporation, either, (A) with respect to the Second Closing (as defined in the Securities Purchase Agreement), that the Second Closing shall not have occurred by June 30, 2011 and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.2 thereof, or (B) with respect to the Third Closing (as defined in the Securities Purchase Agreement), that the Third Closing shall not have occurred by the Third Closing Termination Date (as defined in the Securities Purchase Agreement) and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.3 thereof.
- (s) "Code" shall mean the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder by the Internal Revenue Service.
 - (t) "Common Stock" shall mean collectively, the Ordinary Common Stock and the Class B Common Stock.
- (u) "Conditional Commitment" shall mean a conditional commitment (as defined in 10 CFR 609.2) from DOE to the Corporation in an amount not less than \$2 billion, and specifying the detailed conditions to be satisfied for the DOE Financial Closing.
- (v) "Conversion Cap", with respect to a conversion hereunder, shall mean that the total number of shares of Class B Common Stock received upon such conversion shall not, when combined with the total number of shares of Class B Common Stock (1) issued or issuable upon the exercise of the Warrants and (2) issued by the Corporation upon conversion of securities issued pursuant to the Transactions (as defined in the Securities Purchase Agreement) exceed 49.99% of the total number of outstanding shares of Ordinary Common Stock and Class B Common Stock at the time of any such conversion, subject to adjustments for stock splits, stock dividends, reorganizations or similar transactions.

- (w) "Conversion Election" shall have the meaning ascribed to it in Section 7(a).
- (x) "Corporation" shall have the meaning ascribed to it in the recitals.
- (y) "Corporation Plans" shall mean the Corporation's 1999 Equity Incentive Plan, as amended, and the Corporation's 2009 Equity Incentive Plan, as may be amended, the Corporation's 2009 Employee Stock Purchase Plan, as may be amended, and any similar plans entered into after the date hereof, and any inducement grants.
- (z) "Deemed Holder Percentage" shall mean, as to any Permitted Holder, the percentage resulting from the following calculation, (1)(A) the number of shares of Ordinary Common Stock equal to the quotient of (w) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on the outstanding shares of Series B Preferred Stock held by such Permitted Holder from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (x) the Base Price for the date of such calculation, plus (B) the number of outstanding of shares of (y) Series C Preferred Stock multiplied by 1000 plus, (z) if then outstanding, Class B Common Stock, in each case held by such Permitted Holder divided by (2)(A) the total number of shares of Ordinary Common Stock equal to the quotient of (v) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on all outstanding shares of Series B Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (w) the Base Price for the date of such calculation, plus (B) the total number of all outstanding shares of (x) Series C Preferred Stock multiplied by 1000 plus (y) if then outstanding, Class B Common Stock, plus (z) Ordinary Common Stock.
 - (aa) "DGCL" shall have the meaning ascribed to it in the Preamble.
 - (bb) "Dividend" shall have the meaning ascribed to it in Section 5(a).
- (cc) "Dividend Payment Date" shall mean January 1, April 1, July 1 and October 1 of each year, commencing on April 1, 2010; provided that if any such Dividend Payment Date would otherwise occur on a day that is not a Business Day, such Dividend Payment Date shall instead be (and any Dividend payable on Series B-1 12.75% Preferred Stock on such Dividend Payment Date shall instead be payable on) the immediately succeeding Business Day.
- (dd) "Dividend Period" shall mean the period commencing on and including a Dividend Payment Date and shall end on and include the calendar day preceding the next Dividend Payment Date; <u>provided</u>, <u>however</u>, that with respect to any shares of Series B-1 12.75% Preferred Stock not outstanding for the entirety of any such Dividend Period, there shall be an initial pro-rated Dividend Period for such shares that shall commence on and include the issue date of such shares.
 - (ee) "Dividend Rate" shall mean 12.75% per annum.
 - (ff) "DOE" shall mean the United States Department of Energy.
- (gg) "DOE Financial Closing" shall mean the closing of a Loan Guarantee Agreement (as defined in 10 CFR 609.2), between DOE, an eligible lender, and the Corporation, pursuant to the Conditional Commitment, guaranteeing a loan or other debt obligation in an amount not less than \$2 billion for the ACP and there shall have been an initial draw of the funds guaranteed pursuant to the Loan Guarantee Agreement in an amount not less than the minimum amount permitted thereunder.
 - (hh) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
 - (ii) "Exchange Property" shall have the meaning ascribed to it in Section 10(a).
- (jj) "Excluded Lender" shall mean a bank or other financial institution providing indebtedness for borrowed money which is guaranteed by the Loan Guarantee Agreement (as defined in 10 CFR 609.2) pertaining to the DOE Financial Closing; provided, however "Excluded Lender" shall not include a Person providing funding or committed funding (pursuant to definitive binding agreements) for debt or equity of the Corporation in an amount of at least \$100,000,000 that is not guaranteed by such Loan Guarantee Agreement.
 - (kk) "Factor" shall be the Factor established in accordance with the provisions of Section 7(h)(1).
- (II) "Final Determination" shall mean the earlier to occur of (1) the conclusion of the litigation or binding arbitration (as applicable), including any and all appeals (whether by final determination or the expiration of any applicable appeal periods), regarding the dispute between the Permitted Holders and the Corporation, or (2) a written agreement between the Corporation and the appropriate Permitted Holder or Permitted Holders resolving such dispute.

- (mm) "Governmental Authority" shall mean any foreign governmental authority, the United States of America, any state of the United States and any political subdivision of any of the foregoing, and any agency, instrumentality, department, commission, board, bureau, central bank, authority, court, arbitral body or other tribunal, in each case whether executive, legislative, judicial, regulatory or administrative, having jurisdiction over any of the Permitted Holders, the Corporation, any of the Corporation's Subsidiaries or their respective Property.
 - (nn) "Initial Liquidation Preference" shall mean \$1,000 per share of Series B Preferred Stock.
 - (00) "Initial Preferred Director" shall have the meaning ascribed to it in Section 9(b)(1).
 - (pp) "Internal Reorganization Event" shall have the meaning ascribed to it in Section 10(c).
- (qq) "Investor Rights Agreement" shall mean that certain Investor Rights Agreement, dated [—], 2010 among the Corporation, Toshiba and B&W, as amended from time to time.
- (rr) "Junior Stock" shall mean the Common Stock and any other class or series of capital stock of the Corporation that ranks junior to the Series B Preferred Stock (1) as to the priority of payment of dividends and/or (2) as to the distribution of assets on any liquidation, dissolution or winding up of the Corporation. For the avoidance of doubt, Junior Stock shall include the Series A Preferred Stock and the Series C Preferred Stock.
- (ss) "<u>Liquidation Preference</u>" shall initially mean \$1,000 per share of Series B Preferred Stock; <u>provided, however</u>, that to the extent that the Corporation does not declare and pay a Dividend on a Dividend Payment Date pursuant to Section 5(a), an amount equal to the Dividend shall be added to the Liquidation Preference of such share on the applicable Dividend Payment Date.
 - (tt) "Orderly Sale Arrangement" shall have the meaning set forth in the Securities Purchase Agreement.
- (uu) "Ordinary Common Stock" shall mean the common stock of the Corporation, par value \$.10 per share. For the avoidance of doubt, the Ordinary Common Stock shall not include the Class B Common Stock.
- (vv) "Original Issuance Date" shall mean, with respect to each share of Series B Preferred Stock issued to the Permitted Holders, the date on which such share was issued by the Corporation.
- (ww) "Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, the aggregate Initial Liquidation Preference of all the shares of Series B Preferred Stock issued to the Permitted Holders excluding those shares issued as a Dividend.
- (xx) "Parity Stock" shall mean any class or series of stock of the Corporation that ranks equally with Series B-1 12.75% Preferred Stock (1) in the priority of payment of dividends and/or (2) in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation (in each case, without regard to whether dividends accrue cumulatively or non-cumulatively). For the avoidance of doubt, Parity Stock shall include the Series B-2 11.5% Preferred Stock.
 - (yy) "Permit" shall mean any approval, authorization, certificate, consent, license or permit of or from any Governmental Authority.
- (zz) "Permitted Holder Material Breach" shall mean a material breach of the Securities Purchase Agreement or the Investor Rights Agreement by any Permitted Holder.
- (aaa) "Permitted Holder Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, for any Permitted Holder, the aggregate Initial Liquidation Preference of all shares of Series B Preferred Stock issued to such Permitted Holder excluding those shares issued as a Dividend.
- (bbb) "Permitted Holder Outstanding Value" shall mean, as to any Permitted Holder, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock then held by such Permitted Holder, plus, (2) for each share of Series C Preferred Stock then held by a Permitted Holder, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which such Permitted Holder's acquisition of such share was calculated, plus (3) for each share of Common Stock then held by such Permitted Holder, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Ordinary Common Stock purchased in the market, the Base Price upon which such Permitted Holder's

acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock, which have been added to the Liquidation Preference pursuant to Section 5(a).

- (ccc) "Permitted Holders" shall mean (1) [Toshiba U.S. Co.] and its Wholly-Owned Affiliates, (2) B&W and its Wholly-Owned Affiliates, (3) a special purpose entity jointly and wholly controlled by Toshiba and B&W and (4) Westinghouse Electric Company, LLC, to the extent it is controlled by Toshiba or a Permitted Holder described under (1) above; provided, however, that each Permitted Holder must be a U.S. Person.
- (ddd) "Person" shall mean any individual, corporation, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization, Governmental Authority or any other entity.
 - (eee) "Preferred Director" shall have the meaning ascribed to it in Section 9(b)(2).
 - (fff) "Preferred Stock" shall mean any and all series of preferred stock, par value \$1.00 per share, of the Corporation, including the Series B Preferred Stock.
 - (ggg) "Property" shall mean any interest in any kind of property or asset, whether real, personal or mixed, tangible or intangible.
 - (hhh) "Qualified Director" shall mean any individual reasonably acceptable to the Nominating and Governance Committee of the Board of Directors.
 - (iii) "Regulatory Bodies" shall mean the DOE and the U.S. Nuclear Regulatory Commission, and any successor Governmental Authorities thereto.
 - (iji) "Reorganization Event" shall have the meaning ascribed to it in Section 10(a).
 - (kkk) "Sale Plan" shall have the meaning ascribed to it in Section 7(c)(1).
- (III) "Securities Purchase Agreement" shall mean that certain Securities Purchase Agreement, dated May 25, 2010, among the Corporation, Toshiba and B&W, as amended from time to time.
- (mmm) "Senior Stock" shall mean any class or series of capital stock of the Corporation that ranks senior to the Series B Preferred Stock (1) as to the priority of dividends and/or (2) as to the distribution of assets on any liquidation, dissolution or winding up of the Corporation.
- (nnn) "Series A Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series A Junior Participating Preferred Stock."
 - (000) "Series B Preferred Stock" shall mean the Series B-1 12.75% Preferred Stock together with the Series B-2 11.5% Preferred Stock.
 - $\textbf{(ppp)} \ ``\underline{Series\ B-1\ 12.75\ \%\ Preferred\ Stock''} \ shall\ have\ the\ meaning\ ascribed\ to\ it\ in\ Section\ 1.$
- (qqq) "Series B-2 11.5% Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series B-2 11.5% Convertible Preferred Stock."
- (rrr) "Series B-2 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-2 11.5% Preferred Stock, as filed with the Secretary of State of the State of Delaware.
- (sss) "Series C Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series C Participating Convertible Preferred Stock."
- (ttt) "Share Issuance Approval" shall mean the approval of the stockholders of the Corporation necessary to approve the conversion of all the Series B Preferred Stock and the Series C Preferred Stock, and the exercise of all the Warrants, for Common Stock for purposes of Section 312.03 of the New York Stock Exchange Listed Company Manual, or if shares of the Ordinary Common Stock become listed and traded on another Approved Market, the approval required by such Approved Market, or the time at which all such approvals shall for any reason become inapplicable or not required so as to permit all such conversions and exercises.

- (uuu) "Share Issuance Limitation" shall mean the total number of shares of Common Stock or securities convertible into Common Stock that can be issued by the Corporation upon conversion or exercise of securities issued pursuant to the Transactions (as defined in the Securities Purchase Agreement) in accordance with the rules and regulations of the Approved Market on which shares of the Corporation's equity securities are listed or traded prior to receipt of the Share Issuance Approval.
- (vvv) "Subsidiary" of any Person shall mean any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than 50% of (1) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (2) the interest in the capital or profits of such partnership, joint venture or limited liability company or (3) the beneficial interest in such trust or estate is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries.

 Notwithstanding the foregoing, American Centrifuge Manufacturing, LLC, a Delaware limited liability company, shall not be considered a Subsidiary of B&W or the Corporation.
 - (www) "SWU" shall have the meaning ascribed to it in the definition of "SWU Consideration."
- (xxx) "SWU Consideration" shall mean the fair market value of separative work units with respect to low enriched uranium ("SWU") (as determined reasonably and in good faith by the Board of Directors, taking into account the applicable volume of SWU, the then-current market price for SWU and other relevant factors) provided by the Corporation to the Permitted Holders minus any consideration paid by the Permitted Holders for such SWU.
- (yyy) "Third Party Financing" shall mean the funding or committed funding (pursuant to definitive binding agreements) for debt or equity of the Corporation in an amount of at least \$100,000,000 from a third party that is not an Affiliate of the Corporation, a Japanese export credit agency, a U.S. Governmental Authority or an Excluded Lender where (1) such funds, together with such other additional funds available to the Corporation at such time, is necessary and sufficient to consummate the DOE Financial Closing, and (ii) the third-party requires, as a condition to the funding, that the Preferred Stock be converted in accordance with the terms hereof.
- (zzz) "Third Party Transfer" shall mean an irrevocable Transfer in compliance with Section 11 of all legal ownership, Voting Control and Beneficial Ownership of any share or shares of Series B-1 12.75% Preferred Stock to a Person other than a Permitted Holder or its Affiliates.
 - (aaaa) "Toshiba" shall mean Toshiba Corporation, a corporation organized under the laws of Japan.
- (bbbb) "Trading Day" shall mean any day on which shares of the Corporation's equity securities are traded, or able to be traded, on the Approved Market on which shares of the Corporation's equity securities are listed or traded.
- (cccc) "Transfer" shall mean, with respect to any shares of Series B-1 12.75% Preferred Stock, any direct or indirect assignment, sale, exchange, transfer, tender or other disposition of such shares or any interest therein, whether voluntary or involuntary, by operation of law or otherwise (and includes any sale or other disposition in any one transaction or series of transactions and the grant or transfer of an option or derivative security covering such shares), and any agreement, arrangement or understanding, whether or not in writing, to effect any of the foregoing; provided, however, that a "Transfer" shall not occur simply as a result of the grant of a proxy in connection with a solicitation of proxies subject to the provisions of Section 14 of the Exchange Act.
- (dddd) "<u>U.S. Person</u>" shall mean any person that is treated as a "United States Person" under Code Section 7701(a)(30) and that provides an IRS Form W-9 (or successor form), evidencing a complete exemption from United States withholding tax (including backup withholding tax), on or before the time at which it acquires securities pursuant to this Certificate of Designation.
- (eeee) "Voting Control" shall mean, with respect to a share or shares of Series B-1 12.75% Preferred Stock, the power, whether exclusive or shared, revocable or irrevocable, to vote or direct the voting of such share or shares of Series B-1 12.75% Preferred Stock, by proxy, voting agreement or otherwise.
- (ffff) "Warrants" shall mean those warrants to purchase Class B Common Stock or Series C Preferred Stock originally issued by the Corporation to the Permitted Holders pursuant to the Securities Purchase Agreement.
- (gggg) "Wholly-Owned Affiliate" shall mean, as to any Person, any Affiliate that, directly or indirectly, is wholly-owned and controlled (other than by contract) by a Person, or any other Affiliate to which the Corporation, in its sole

discretion, consents.

Section 4. Titles and Subtitles; Interpretation. The titles and subtitles used in this Certificate of Designation are used for convenience only and are not to be considered in construing or interpreting this Certificate of Designation. The definitions contained in this Certificate of Designation are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such terms.

Section 5. Dividends.

(a) Rate. Holders of Series B-1 12.75% Preferred Stock shall be entitled to receive, when, as and if declared by the Board of Directors, out of funds or assets available therefor, subject to the provisions of the DGCL, on each share of Series B-1 12.75% Preferred Stock, Dividends with respect to each Dividend Period in an amount equal to the Dividend Rate on the Liquidation Preference per share of Series B-1 12.75% Preferred Stock, payable, at the Corporation's election, in (1) cash, (2) additional shares (including fractional shares) of Series B-1 12.75% Preferred Stock having a deemed value of \$1,000 per share for purposes of the number of such additional shares or (3) any combination of (1) and (2) (the "Dividend"). If and to the extent that the Corporation does not pay the entire Dividend for a particular Dividend Period on the applicable Dividend Payment Date for such period, the amount of such Dividend not paid shall be added to the Liquidation Preference in accordance with the definition thereof. Dividends payable at the Dividend Rate shall begin to accrue (whether or not earned or declared, whether or not there are funds legally available for the payment thereof and whether or not restricted by the terms of any of the Corporation's indebtedness outstanding at any time) and be cumulative from the Original Issuance Date, shall compound on each Dividend Payment Date (i.e., no Dividends shall accrue on other Dividends unless and until the first Dividend Payment Date for such other Dividends has passed without such other Dividends having been paid on such date) and shall be payable in arrears on the first Dividend Payment Date after such Dividend Period. Dividends that are payable on Series B-1 12.75% Preferred Stock in the form of additional shares of such stock shall, except as specifically provided in this Certificate of Designation, have all rights granted hereunder, including the payment of Dividends. Dividends that are payable on Series B-1 12.75% Preferred Stock on any Dividend Payment Date shall be payable to holders of record of Series B-1 12.75% Preferred Stock as they appear on the stock register of the Corporation on the record date for such Dividend, which shall be the date 10 Business Days prior to the applicable Dividend Payment Date, or such other date as determined by the Board of Directors. The Corporation shall elect the form of such payment by giving notice at least 5 Business Days prior to the applicable Dividend Payment Date. If no such notice is given, the Corporation shall be deemed to have elected a payment through the issuance of shares of Series B-1 12.75% Preferred Stock. Dividends paid on the shares of Series B-1 12.75% Preferred Stock in an amount less than accumulated and unpaid Dividends payable thereon shall be allocated pro rata on a share-by-share basis among all such shares at the time outstanding.

Dividends payable at the Dividend Rate on Series B-1 12.75% Preferred Stock in respect of any Dividend Period shall be computed on the basis of a 360-day year consisting of twelve 30-day months. The amount of Dividends payable at the Dividend Rate on Series B-1 12.75% Preferred Stock on any date prior to the end of a Dividend Period, and for the initial Dividend Period, shall be computed on the basis of a 360-day year consisting of twelve 30-day months, and actual days elapsed over such 30-day months

- **(b) Priority of Dividends.** Subject to any approvals required pursuant to Section 9, such Dividends (payable in cash, securities or other property) as may be determined by the Board of Directors may be declared and paid on any capital stock, including Common Stock and other Junior Stock, from time to time out of any funds legally available for such payment.
- (c) Payment in Shares. Any shares of Series B-1 12.75% Preferred Stock paid as a Dividend pursuant to this Section 5 shall be duly authorized, validly issued, fully paid and non-assessable, and shall be free of preemptive rights and free of any lien or adverse claim.

Section 6. Liquidation Rights.

(a) Voluntary or Involuntary Liquidation. In the event of any liquidation, dissolution or winding up of the affairs of the Corporation, whether voluntary or involuntary, holders of Series B-1 12.75% Preferred Stock shall be entitled to receive on par with each share of Parity Stock ranking equally with Series B-1 12.75% Preferred Stock in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation for each share of Series B-1 12.75% Preferred Stock, out of the assets of the Corporation or proceeds thereof available for distribution to stockholders of the Corporation, and after satisfaction of all liabilities and obligations to creditors of the Corporation, before any distribution of such assets or proceeds is made to or set aside for the holders of Junior Stock, an amount equal to the greater of (1) the Liquidation Preference per share of Series B-1 12.75% Preferred Stock *plus* an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference from and including the immediately preceding Dividend Payment Date to, but excluding, the date fixed for such liquidation, dissolution or winding up of the Corporation and (2) the

per share amount of all cash, securities and other property (such securities or other property having a value equal to its fair market value as reasonably and in good faith determined by the Board of Directors) to be distributed in respect of the Common Stock such holder would have been entitled to receive had it converted such Series B-1 12.75% Preferred Stock (without regard to the Conversion Cap or Share Issuance Limitation) immediately prior to the date fixed for such liquidation, dissolution or winding up of the Corporation. To the extent that such amount is paid in full to all holders of Series B-1 12.75% Preferred Stock and all holders of Parity Stock ranking equally with Series B-1 12.75% Preferred Stock in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation, the holders of other capital stock of the Corporation shall be entitled to receive all remaining assets of the Corporation (or proceeds thereof) according to their respective rights and preferences.

- (b) Partial Payment. If, in connection with any distribution described in Section 6(a) above, the assets of the Corporation or proceeds thereof are not sufficient to pay the Liquidation Preferences in full to all holders of Series B-1 12.75% Preferred Stock and all holders of Parity Stock ranking equally with Series B-1 12.75% Preferred Stock in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation, then the amounts paid to the holders of Series B-1 12.75% Preferred Stock and to the holders of all such other Parity Stock shall be paid pro rata in accordance with the respective aggregate Liquidation Preferences of the holders of Series B-1 12.75% Preferred Stock and the holders of all such other Parity Stock.
- (c) Merger, Consolidation and Sale of Assets Not Liquidation. For purposes of this Section 6, the merger or consolidation of the Corporation with any other corporation or other Person, including a merger or consolidation in which the holders of Series B-1 12.75% Preferred Stock receive cash, securities or other property for their shares, or the sale, lease or exchange (for cash, securities or other property) of all or substantially all of the assets of the Corporation, shall not constitute a liquidation, dissolution or winding up of the Corporation, but shall instead be subject to the provisions of Section 10.

Section 7. Mandatory Redemption or Conversion on a Closing Deadline Failure

- (a) Permitted Holder Election on a Closing Deadline Failure. Within 20 Business Days of a Closing Deadline Failure, each Permitted Holder shall deliver a written notice to the Corporation stating, with respect to all of its outstanding shares of Series B-1 12.75% Preferred Stock, whether such Permitted Holder elects to convert such shares pursuant to Section 7(b) (a "Conversion Election") or sell such shares pursuant to Section 7(c) (a "Sale Election"). If any Permitted Holder does not make such election by such deadline, such Permitted Holder shall be deemed to have irrevocably made a Sale Election with respect to all of its outstanding shares of Series B-1 12.75% Preferred Stock.
- **(b) Conversion Election Procedures.** Within 40 Business Days of a Closing Deadline Failure, the Corporation shall, with respect to the outstanding shares of Series B-1 12.75% Preferred Stock held by all Permitted Holders that made a Conversion Election, convert such shares (i) if the Charter Amendment Approval has been obtained and subject to the making of any filing or receipt of any approval from any Regulatory Body in order not to adversely affect the Permits or regulatory status of the Corporation or its Subsidiaries, into Class B Common Stock, or (ii) if the Charter Amendment Approval has not been obtained or such regulatory approvals cannot be obtained, into Series C Preferred Stock, in either case into the number of shares of the Class B Common Stock or Series C Preferred Stock, as applicable, equal to the product of (A) the quotient of (1) the Liquidation Preference *plus* an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion, and (2) the Base Price for such conversion date and (B) the Factor. Any outstanding shares of Series B-1 12.75% Preferred Stock not converted pursuant to this Section 7(b) as a result of the limitations set forth in Section 7(d) shall remain outstanding until the earlier of (x) the date on which the Share Issuance Approval is obtained (on which date any then outstanding shares of Series B-1 12.75% Preferred Stock held by all Permitted Holders that made a Conversion Election shall be converted pursuant to this Section 7(b) using the Base Price for such conversion date) and (y) such time the outstanding shares of Series B-1 12.75% Preferred Stock are redeemed pursuant to Section 7(g).

(c) Sale Election Procedures.

(1) If a Permitted Holder makes or is deemed to make a Sale Election, such Permitted Holder shall use commercially reasonable efforts to sell and dispose of such Permitted Holder's shares of Series B-1 12.75% Preferred Stock in accordance with Section 11 and the Orderly Sale Arrangement. In furtherance of the foregoing and without limitation, such Permitted Holder shall use commercially reasonable efforts to, as promptly as is practicable, either (x) enter into an agreement with a broker dealer that represents one of the institutions listed on Schedule C to the Securities Purchase Agreement as of such date pursuant to which all of the shares of Ordinary Common Stock into which the outstanding shares of Series B-1 12.75% Preferred Stock held by such Permitted Holder shall be converted and sold (the "Sale Plan") or (y) sell pursuant to such other method as shall be mutually agreed upon between the Corporation and the Permitted Holder. The Sale Plan shall, inter alia:

- (A) constitute a written binding contract between such Permitted Holder and such broker dealer pursuant to which such Permitted Holder instructs the broker dealer to sell such shares on its account;
- (B) result in the sale as promptly as practicable and in brokers transactions of the shares of Ordinary Common Stock into which such Permitted Holder's outstanding shares of Series B-1 12.75% Preferred Stock shall be converted, as provided below pursuant to the Orderly Sale Arrangement;
- (C) permit such Permitted Holder no influence over when or whether to effect the sale of such shares of Ordinary Common Stock underlying such Permitted Holder's outstanding shares of Series B-1 12.75% Preferred Stock (other than initiating a separate block trade undertaken in accordance with the Orderly Sale Arrangement); and
- (D) except as provided in clause (C), require that such shares of Ordinary Common Stock underlying such Permitted Holder's outstanding shares of Series B-1 12.75% Preferred Stock are sold pursuant to the terms of the Sale Plan;
 - (2) Upon a Third Party Transfer, shares of Series B-1 12.75% Preferred Stock when sold pursuant to a Sale Plan shall automatically convert into the number of shares of Ordinary Common Stock equal to the following: (x) if sold pursuant to a brokers transaction, the product of (A) the quotient of (1) the Liquidation Preference *plus* an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion, and (2) the price per share reported for the sale of the underlying Ordinary Common Stock and (B) the Factor; or (y) if sold other than through a brokers transaction, the product of (A) the quotient of (1) the Liquidation Preference *plus* an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion, and (2) the price per share at which the underlying Ordinary Common Stock is agreed to be sold in such transaction and (B) the Factor.
- (d) Share Issuance Limitation. Notwithstanding anything in this Certificate of Designation to the contrary, any issuance of Common Stock or Series C Preferred Stock shall be limited to the total number of shares that may be issued in compliance with the Share Issuance Limitation to the extent applicable.
- (e) Deferred Implementation of Sale Plan. In the event that a Permitted Holder at the time of a Sale Election advises the Corporation that it in good faith believes that it is in possession of material non-public information concerning the Corporation, such Permitted Holder may defer implementation of the Sale Plan until the next period of time during which directors and executive officers of the Corporation are permitted to purchase and sell shares of Ordinary Common Stock in a trading "window" or similar period pursuant to the Corporation's trading policies in effect at such time. Without limiting a Permitted Holder's obligation to do so, if a Permitted Holder shall fail to enter into and initiate a Sale Plan within 180 days of a Sale Election and shall at any time thereafter fail to use commercially reasonable efforts to implement a Sale Plan (tolling such period to the extent it is prevented from doing so pursuant to the provisions of Section 4.7 of the Investor Rights Agreement) the Corporation may, if the Permitted Holder fails to resume and maintain such commercially reasonable efforts within ten (10) Business Days after notice of such failure from the Corporation, convert such Permitted Holder's shares of Series B-1 12.75% Preferred Stock pursuant to the provisions of Section 7(b) (without regard for the deadline or notice provided for therein) as of the date of such failure applying the Base Price as of the date of such conversion. With respect to a Permitted Holder's obligation in the immediately preceding sentence, such commercially reasonable efforts shall include such Permitted Holder causing any Initial Preferred Director or the Preferred Director, as applicable, to resign within 90 days of the delivery of the notice pursuant to Section 7(a) if such Initial Preferred Director's or the Preferred Director's access, as applicable, to material non-public information concerning the Corporation is preventing the Permitted Holder from entering into the Sale Plan or otherwise disposing of its shares in ac
- (f) Optional Redemption. Notwithstanding any Sale Plan, at any time from and after a Sale Election, the Corporation may, subject to the provisions of the DGCL, and from time to time, upon 10 Business Days prior written notice, redeem all or any portion of the outstanding shares of Series B-1 12.75% Preferred Stock for, at the Corporation's sole discretion, cash or SWU Consideration in an amount equal to the product of (i) the Liquidation Preference of such shares plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of redemption and (ii) the Factor.
- (g) Automatic Redemption. If a Closing Deadline Failure occurs and shares of Series B-1 12.75% Preferred Stock remain outstanding on the later of (i) December 31, 2012 or (ii) the one-year anniversary of such Closing Deadline Failure,

the Corporation shall, subject to the provisions of the DGCL, redeem all outstanding shares of Series B-1 12.75% Preferred Stock for, at the Corporation's sole discretion, cash or SWU Consideration in an amount equal to (i) the product of (A) the Liquidation Preference of such shares *plus* an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of redemption and (B) the Factor.

(h) Determination of Factor.

- (1) The Factor shall be (i) if at the time of the Closing Deadline Failure the Securities Purchase Agreement was not terminable pursuant to Section 10.2(c) or (d) or Section 10.3(c) or (d) thereof, 1.0 (one); (ii) if at the time of the Closing Deadline Failure the Securities Purchase Agreement was terminable pursuant to Sections 10.2(c) or 10.3(c) thereof, 1.1 (one and one-tenth); or (iii) if at the time of the Closing Deadline Failure the Securities Purchase Agreement was terminable as to such Permitted Holder pursuant to Sections 10.2(d) or 10.3(d) thereof, 0.9 (nine-tenths).
- (2) Together with the notice delivered by each Permitted Holder pursuant to Section 7(a), each Permitted Holder shall state the Factor to be applied with respect to the Conversion Election or the Sale Election. If any Permitted Holder does not make such determination in its notice, the Factor deemed noticed and applicable to such Permitted Holder shall be 1.0. Within 20 Business Days of receipt by the Corporation of a notice pursuant to Section 7(a), the Corporation may deliver a written notice to a Permitted Holder disputing such Permitted Holder's determination of the Factor or, if the Permitted Holder did not include a Factor in its notice, the deemed Factor. If the Corporation does not timely provide such notice, such Permitted Holder's determination of the Factor or the deemed Factor, as the case may be, shall be final and binding on such Permitted Holder and the Corporation. If the Corporation timely objects to a Permitted Holder's determination of the Factor, the Factor shall be initially 1.0 for purposes of such conversion or redemption and all of such Permitted Holder's outstanding shares of Series B-1 12.75% Preferred Stock shall be converted pursuant to Section 7(b), sold pursuant to Section 7(c) or redeemed pursuant to Sections 7(f) or (g) based upon such Factor and either the Corporation or such Permitted Holder may seek a Final Determination pursuant to the procedures set forth in Section 13.2 of the Securities Purchase Agreement, and following any such Final Determination, such final Factor shall be applied hereunder.

Section 8. Other Conversion.

(a) Conversion by the Corporation.

- (1) Conversion Upon Third Party Financing. Effective upon the DOE Financial Closing that follows or is contemporaneous with a Third Party Financing or immediately prior thereto, the Corporation may convert all of the outstanding shares of Series B-1 12.75% Preferred Stock (i) if the Charter Amendment Approval has been obtained, into Class B Common Stock, or (ii) if the Charter Amendment Approval has not been obtained, into Series C Preferred Stock, in either case into the number of shares of the Class B Common Stock or Series C Preferred Stock, as applicable, equal to the quotient of (A) 120% of the sum of (i) the Liquidation Preference plus (ii) an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (B) the Base Price for the date that the Corporation provides notice pursuant to Section 8(a)(2).
- (2) Conversion Timing. If the Corporation elects to convert pursuant to this Section 8(a), the Corporation shall provide written notice to the Permitted Holders of record at their respective last addresses appearing on the books of the Corporation. Such notice shall state the conversion date of Series B-1 12.75% Preferred Stock, which date shall be no less than 5 days and no more than 60 days from the date of such notice; provided, however, that the effectiveness of the conversion (and the Corporation's right and obligation to effect the conversion) shall be conditioned upon the DOE Financial Closing. Notwithstanding the foregoing, if, after delivery of such notice, the Corporation desires to specify a different conversion date, the Corporation shall not be required to notify the Permitted Holders of such change until after the conversion is effected unless such changed conversion date is more than 15 days prior to or after the original conversion date. The conversion date shall be the date specified in such written notice or such different date as specified by the Corporation in accordance with this Section 8(a)(2).

(b) Conversion by the Permitted Holders.

(1) Post-Third Closing Conversion. At any time and from time to time after the Third Closing (as defined in the Securities Purchase Agreement), any Permitted Holder's shares of Series B-1 12.75% Preferred

Stock shall be converted, in whole or in part, upon the request of such Permitted Holder, subject to the Conversion Cap, into the number of shares of Class B Common Stock equal to the quotient of (A) the Liquidation Preference *plus* an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion and (B) the Base Price for the conversion date specified in the written notice provided by such Permitted Holder pursuant to Section 8(b)(2). Shares of Series B-1 12.75% Preferred Stock not converted as a result of the foregoing limitations shall remain outstanding except as provided berein

- (2) Conversion Timing. If a Permitted Holder elects to convert pursuant to this Section 8(b), such Permitted Holder shall provide written notice to the Corporation. Such notice shall state the conversion date of Series B-1 12.75% Preferred Stock, which date shall be no less than 5 days and no more than 60 days from the date of such notice. The conversion date shall be the date specified in such written notice.
- (c) Automatic Conversion and Redemption. On December 31, 2016, all outstanding shares of Series B-1 12.75% Preferred Stock shall be automatically converted, without any action on the part of the holder and subject to the Conversion Cap, into the number of shares of Class B Common Stock equal to the quotient of (i) the Liquidation Preference plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion and (ii) the Base Price for December 31, 2016. Shares of Series B-1 12.75% Preferred Stock not converted as a result of the foregoing limitation shall remain outstanding except as provided herein. If shares of Series B-1 12.75% Preferred Stock remain outstanding on February 28, 2017 due to the Conversion Cap, the Corporation shall, subject to the provisions of the DGCL, redeem all outstanding shares of Series B-1 12.75% Preferred Stock for cash in an amount equal to the Liquidation Preference of such shares plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of redemption. If the Corporation fails to pay such redemption amount by March 15, 2017, the Conversion Cap shall no longer apply and all outstanding shares of Series B-1 12.75% Preferred Stock shall be automatically converted, without any action on the part of the holder, into the number of shares of Class B Common Stock equal to the quotient of (A) the Liquidation Preference plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-1 12.75% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversi
- (d) Conversion Mechanics. A Permitted Holder shall cease to be a record holder of each share of Series B-1 12.75% Preferred Stock on the date such share is converted. As promptly as practicable on or after the conversion date (and in any event no later than three Trading Days thereafter), the Corporation or its agent, including its transfer agent, shall issue the number of shares of Class B Common Stock or Series C Preferred Stock (including fractional shares) issuable pursuant to Section 8(a), (b) or (c). Any such certificate or certificates shall be delivered by the Corporation or its agent, including its transfer agent, to the appropriate holder on a book-entry basis or by mailing certificates evidencing the shares to the holders at their respective addresses as set forth in the records of the Corporation, subject in each case to the provisions of Section 9 of the Securities Purchase Agreement.
- (e) Reservation of Class B Common Stock Subject to receiving the Charter Amendment Approval and for as long as any shares of Series B Preferred Stock remain outstanding, the Corporation shall at all times reserve and keep available, free from preemptive rights, out of its authorized but unissued Ordinary Common Stock or Class B Common Stock, or shares of Ordinary Common Stock or Class B Common Stock the full number of shares of Ordinary Common Stock or Class B Common Stock then issuable upon the conversion of all the shares of Series B Preferred Stock then outstanding. For purposes hereof, reservations hereunder shall be at the Base Price equal to the closing price of the Corporation's Ordinary Common Stock on the New York Stock Exchange on the second to last Trading Day prior to the date of the Securities Purchase Agreement; provided, however, if the Base Price for the date four Trading Days prior to the First Closing, the Second Closing or the Third Closing (each as defined in the Securities Purchase Agreement) or on June 30 of any year is less than such amount, then that lower amount shall be used as the Base Price for purposes of this calculation. All shares of Class B Common Stock delivered upon conversion of Series B Preferred Stock shall have been duly authorized and validly issued and shall be fully paid and nonassessable, and shall be free from preemptive rights and free of any lien or adverse claim.
- (f) Partial Conversion. In case of any conversion of any of the shares of Series B Preferred Stock at the time outstanding, the shares to be converted shall be selected pro rata among the shares of Series B Preferred Stock held by each Permitted Holder and among each such Permitted Holder's shares of Series B-1 12.75% Preferred Stock and Series B-2

11.5% Preferred Stock. If fewer than all of the shares represented by any certificate are converted, a new certificate shall be issued representing the unconverted shares without charge to the holder thereof.

(g) Taxes. The Corporation shall pay any and all taxes that may be payable in respect of the issue or delivery of shares of Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock on conversion of Series B-1 12.75% Preferred Stock. The Corporation shall not, however, be required to pay any tax that may be payable in respect of any Transfer involved in the issue and delivery of shares of Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock in a name other than that in which Series B-1 12.75% Preferred Stock so converted was registered, and no such issue or delivery shall be made unless and until the Person requesting such issue has paid to the Corporation that such tax has been paid.

Section 9. Voting Rights.

(a) General. The holders of shares of Series B-1 12.75% Preferred Stock shall not be entitled to vote, except as otherwise provided herein or required by applicable law.

(b) Election of Directors.

- (1) Effective as of the first Original Issuance Date, the number of directors constituting the Board of Directors shall be increased by two Persons and the holders of a majority of the outstanding Series B Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall be entitled to elect two Qualified Directors to the Board of Directors (each such director, an "Initial Preferred Director") until the earliest to occur of (i) a Closing Deadline Failure as a result of a Permitted Holder Material Breach at a time when the Securities Purchase Agreement is terminable pursuant to Sections 10.2(d) and 10.3(d) thereof, (ii) a Change of Control or (iii) such time as the Permitted Holders' Aggregate Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of the Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the outstanding Series B Preferred Stock to elect the Initial Preferred Directors shall cease, (B) the term of office of the Initial Preferred Directors shall immediately and automatically terminate, (C) the Initial Preferred Directors will no longer be qualified to serve and (D) the number of directors constituting the Board of Directors shall be immediately and automatically reduced by two Persons.
- (2) Effective as of the first Original Issuance Date and at such time as when the Permitted Holders do not have the right to elect the Initial Preferred Directors pursuant to Section 9(b)(1)(iii) and any Permitted Holder's Permitted Holder Outstanding Value is greater than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or, (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, the number of directors constituting the Board of Directors shall be increased by one Person and the holders of a majority of the outstanding Series B Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall be entitled to elect one Qualified Director to the Board of Directors (such director, the "Preferred Director") until the earliest to occur of (i) an event described in Section 9(b)(1)(i) or (ii) or (ii) such time as each Permitted Holder's Permitted Holder Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the outstanding Series B Preferred Stock to elect the Preferred Director shall cease, (B) the term of office of the Preferred Director shall immediately and automatically reduced by one Person.
- (3) For the avoidance of doubt, except for the increase or decrease in the number of directors provided for herein, nothing in this Section 9(b) shall prohibit the Board of Directors from fixing the number of directors constituting the Board of Directors pursuant to the Bylaws.
- (4) Term. Subject to the provisions of this Section 9(b), each Initial Preferred Director or the Preferred Director, as applicable, shall serve until the next annual meeting of the stockholders of the Corporation and until his or her successor is elected and qualified in accordance with this Section 9(b) and the Bylaws, unless any such Initial Preferred Director or the Preferred Director, as applicable, is earlier removed in accordance with the Bylaws, resigns or is otherwise unable to serve; provided, however, that only the holders of a majority of the outstanding shares of the Series B Preferred Stock may remove any such Initial Preferred Director or the Preferred Director, as applicable, without cause at any time and the holders of a majority of the voting power of the outstanding shares of the capital stock of the Corporation entitled to vote on the matter may remove any such Initial Preferred Director or the Preferred Director, as applicable, with cause at any time. Subject to the provisions of this Section 9(b), in the event any Initial Preferred Director or the Preferred Director, as applicable, is removed, resigns or is unable to serve as a member of the Board of Directors, the holders of a majority of the

outstanding shares of Series B Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall have the right to fill such vacancy. Each Initial Preferred Director or the Preferred Director, as applicable, may only be elected to the Board of Directors by the holders of the Series B Preferred Stock in accordance with this Section 9(b), and any such Initial Preferred Director's or the Preferred Director's seat, as applicable, shall otherwise remain vacant.

- (c) Class Voting Rights as to Particular Matters. In addition to any other vote or consent of stockholders required by law or by the Certificate of Incorporation, the vote or consent of the holders of at least a majority of the outstanding shares of Series B Preferred Stock, voting together as a single class to the exclusion of the holders of the Common Stock and any other series of Preferred Stock, then outstanding and entitled to vote on the matter, given in person or by proxy, either in writing without a meeting or by vote at any meeting called for the purpose, shall be necessary for effecting or validating any of the actions described in (x) Section 9(c)(1) if any shares of Series B Preferred Stock are outstanding and (y) Sections 9(c)(2) and 9(c)(3) if the number shares of Series B Preferred Stock outstanding is greater than 10% of all of the shares of Series B Preferred Stock issued to the Permitted Holders, in each case excluding shares issued as a Dividend.
- (1) Amendment of Series B Preferred Stock. Any amendment, alteration or repeal (by merger, consolidation or otherwise) of any provision of the Certificate of Incorporation or this Certificate of Designation so as to adversely affect the powers, preferences and relative participating, optional and other rights of Series B-1 12.75% Preferred Stock.

(2) Dividends, Repurchase and Redemption.

- (A) The declaration or payment of any dividend or distribution of Common Stock or other Junior Stock (other than a dividend payable solely in Junior Stock <u>provided</u> such dividend is not treated as a distribution of property for purposes of Section 305 of the Code, the Treasury Regulations promulgated thereunder or any successor provision); or
- (B) the purchase, redemption or other acquisition for consideration by the Corporation, directly or indirectly, of any Common Stock, other Junior Stock or Parity Stock, (except as necessary (i) to effect a reclassification of Junior Stock for or into other Junior Stock, (ii) to effect a reclassification of Parity Stock into Junior Stock, (iv) to effect the exchange or conversion of one share of Junior Stock for or into another share of Junior Stock, (v) to effect the exchange or conversion of one share of Parity Stock into Junior Stock for or into another share of Parity Stock with the same or lesser per share liquidation amount (vi) to effect the exchange or conversion of one share of Parity Stock into Junior Stock or (vii) pursuant to the Corporation Plans).
- (3) Issuance of Senior Stock or Parity Stock. Prior to the Third Closing (as defined in the Securities Purchase Agreement), the issuance of any Senior Stock or Parity Stock, except as specifically provided for herein or in the Certificate of Designation for the Series B-2 11.5% Preferred Stock.
- (d) Changes after Provision for Redemption or Conversion. No vote or consent of the holders of Series B-1 12.75% Preferred Stock shall be required pursuant to Section 9(c)(x) or Section 9(c)(y) if, at or prior to the time when any such vote or consent would otherwise be required pursuant to such Section, all outstanding shares of Series B-1 12.75% Preferred Stock (1) shall have been redeemed or converted, or (2) shall have been irrevocably elected for redemption or conversion in accordance with Sections 7(f), 7(g) or 8(a) and will, subject to the passage of time, be redeemed or converted; provided, that if, on or before the redemption date specified by the Corporation, all funds required for the redemption of the shares called for redemption have been deposited by the Corporation in trust for the benefit of the Permitted Holders with a bank or trust company doing business in the City of New York having a capital and surplus of at least \$500 million and selected by the Board of Directors, so as to be and continue to be available solely therefor, then, notwithstanding that any certificate for any share so called for redemption has not been surrendered for cancellation, on and after the redemption date, Dividends shall cease to accrue on all shares so called for redemption, all shares so called for redemption shall no longer be deemed outstanding and all rights with respect to such shares shall forthwith on such redemption date cease and terminate, except only the right of the Permitted Holders thereof to receive the amount payable on such redemption from such bank or trust company, without interest. Any funds unclaimed at the end of three years from the redemption date shall, to the extent permitted by law, be released to the Corporation, after which time the Permitted Holders of the shares so called for redemption shall look only to the Corporation for payment of the redemption price of such shares.

Section 10. Reorganization Events.

- (a) In the event of:
 - (1) any consolidation or merger of the Corporation with or into another Person or of another

Person with or into the Corporation; or

(2) any sale, transfer, lease or conveyance to another Person of all or substantially all of the property and assets of the Corporation,

in each case in which holders of Ordinary Common Stock would be entitled to receive cash, securities or other property for their shares of Ordinary Common Stock (any such event specified in this Section 10(a), a "<u>Reorganization Event</u>"), the outstanding shares of Series B Preferred Stock shall be deemed for the purposes of this Section 10 only to be converted into the number of shares of Ordinary Common Stock equal to the quotient of (x) the Liquidation Preference *plus* an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding the date of conversion and (y) the Base Price for the date of effectiveness of such Reorganization Event and each such share shall, (A) become convertible into securities and other property receivable in such Reorganization Event by and in the same relative amounts as a holder of Ordinary Common Stock other than securities issued or other property distributed by such holder or its Affiliates; provided, however, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Corporation or any other Person, then the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ only to the extent that the then existing designation and powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of the Ordinary Common Stock and Series B Preferred Stock differ as provided in this Certificate of Designation or the Certificate of Designation for the Series B-2 11.5% Preferred Stock (including, without limitation, with respect to the voting rights and conversion provisions thereof) or, at the Corporation's sole discretion, (B) be redeemed

- (b) Subject to the restrictions set forth in Section 10(a), in the event that holders of the shares of the Ordinary Common Stock have the opportunity to elect the form of Exchange Property to be received in such transaction, the Exchange Property that holders of the Series B Preferred Stock shall be entitled to receive shall be determined by the holders of a majority of the outstanding shares of Series B Preferred Stock.
- (c) Notwithstanding anything in this Certificate of Designation to the contrary, Section 10(a) shall not apply to a merger, consolidation, asset sale, reorganization or statutory share exchange (1) among the Corporation and its direct and indirect Subsidiaries or (2) between the Corporation and any Person for the primary purpose of changing the domicile of the Corporation (an "Internal Reorganization Event") and no such transaction shall be deemed to be a Reorganization Event. Without limiting the rights of the holders of the Series B Preferred Stock set forth in Section 9(c)(1), the Corporation shall not effectuate an Internal Reorganization Event without the consent of the holders of a majority of the outstanding shares of the Series B Preferred Stock unless the Series B Preferred Stock shall be outstanding as a class or series of preferred stock of the surviving entity having the same rights, terms, preferences, liquidation preference and accrued and unpaid Dividends as the Series B Preferred Stock in effect immediately prior to such Internal Reorganization Event, as adjusted for such Internal Reorganization Event pursuant to this Certificate of Designation after giving effect to any such Internal Reorganization Event.
- (d) The Corporation (or any successor) shall, within 20 days after the occurrence of any Reorganization Event or Internal Reorganization Event, provide written notice to the holders of the Series B Preferred Stock of the occurrence of such event and, in the case of a Reorganization Event, of the kind and amount of the cash, securities or other property that constitutes the Exchange Property. Failure to deliver such notice shall not affect the operation of this Section 10 or the validity of any Reorganization Event or Internal Reorganization Event.

Section 11. Restrictions.

- (a) Notwithstanding anything in this Certificate of Designation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, Permitted Holders may not Transfer shares of Series B-1 12.75% Preferred Stock if such Transfer would require approvals from or filings with any Regulatory Bodies in order not to adversely affect the Permits or regulatory status of the Corporation or its Subsidiaries, unless such approvals and/or filings have been made and received; provided, however, this Section 11(a) shall not apply to any transfer where the transferee received Ordinary Common Stock pursuant to the terms hereof.
- (b) Notwithstanding anything in this Certificate of Designation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, the conversion of Series B-1 12.75% Preferred Stock for Ordinary Common Stock shall also be subject to the requirements of Section 9.2 of the Securities Purchase Agreement.

(c) Any purported conversion or Transfer of Series B-1 12.75% Preferred Stock in violation of these restrictions shall be null and voidab initio.

Section 12. Record Holders. To the fullest extent permitted by applicable law, the Corporation may deem and treat the record holder of any share of Series B-1 12.75% Preferred Stock as the true and lawful owner thereof for all purposes.

Section 13. No Standing to Bring Derivative Action. Notwithstanding any provision of the DGCL, the Rules of the Court of Chancery of the State of Delaware or any other applicable law, rule or regulation which would otherwise confer such standing or empower a holder of Series B-1 12.75% Preferred Stock to take such action, no holder of any share of Series B-1 12.75% Preferred Stock shall have standing to bring an action, suit or proceeding derivatively or otherwise in the right of the Corporation.

Section 14. Legends. All certificates representing shares of Series B-1 12.75% Preferred Stock shall bear a legend or other restriction substantially to the following effect (it being agreed that if such shares are not certificated, other appropriate restrictions shall be implemented to give effect to the following):

"THIS SECURITY WAS ORIGINALLY ISSUED IN A TRANSACTION EXEMPT FROM REGISTRATION UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR AS MAY BE HELD BY A PERSON DEEMED AN "AFFILIATE" (AS DEFINED IN RULE 144 UNDER THE SECURITIES ACT) OF THE ISSUER OF THIS SECURITY, AND THIS SECURITY MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN APPLICABLE EXEMPTION THEREFROM. THE HOLDER OF THIS SECURITY AGREES FOR THE BENEFIT OF THE COMPANY THAT (A) THIS SECURITY MAY BE OFFERED, RESOLD, PLEDGED OR OTHERWISE TRANSFERRED, ONLY (I) IN A TRANSACTION NOT INVOLVING A PUBLIC OFFERING, (II) PURSUANT TO ANY OTHER EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, INCLUDING RULE 144 UNDER THE SECURITIES ACT (IF AVAILABLE), (III) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR (IV) TO THE COMPANY OR ANY OF ITS SUBSIDIARIES, IN EACH OF CASES (I) THROUGH (IV) IN ACCORDANCE WITH ANY APPLICABLE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES, AND (B) THE HOLDER WILL NOTIFY ANY SUBSEQUENT PURCHASER OF THIS SECURITY FROM IT OF THE RESALE RESTRICTIONS REFERRED TO IN (A) ABOVE. THIS SECURITY MAY ONLY BE TRANSFERRED IN ACCORDANCE WITH THE TERMS OF THE CERTIFICATE OF DESIGNATION OF SERIES B-1 CONVERTIBLE PREFERRED STOCK OF USEC INC. (THE "COMPANY"), AS AMENDED.

THIS SECURITY IS SUBJECT TO THE RESTRICTIONS (INCLUDING THE VOTING AND TRANSFER RESTRICTIONS) SET FORTH IN ARTICLES FOURTH AND ELEVENTH OF USEC INC.'S CERTIFICATE OF INCORPORATION, AS AMENDED.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON TRANSFER, CONVERSION AND REDEMPTION) STATED IN, AND ARE TRANSFERABLE ONLY IN ACCORDANCE WITH, THE PROVISIONS OF SECTION 9 OF THE SECURITIES PURCHASE AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA CORPORATION ("TOSHIBA") AND BABCOCK & WILCOX INVESTMENT COMPANY ("B&W"), DATED AS OF MAY 25, 2010.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON THE DISPOSITION OF SECURITIES) STATED IN THE PROVISIONS OF SECTION 4.7 OF THE INVESTOR RIGHTS AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA AND B&W, DATED AS OF , 2010."

Section 15. Written Consent. Any action as to which a class vote of the holders of Preferred Stock, or the holders

of Preferred Stock and Class B Common Stock voting together, is required pursuant to the terms of this Certificate of Designation or the Securities Purchase Agreement may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation.

Section 16. Notices. All notices or communications in respect of Series B-1 12.75% Preferred Stock shall be sufficiently given if given in writing and delivered in person or by first class mail, postage prepaid, or if given in such other manner as may be permitted in this Certificate of Designation, in the Certificate of Incorporation or Bylaws or by applicable law or regulation. Notwithstanding the foregoing, if Series B-1 12.75% Preferred Stock is issued in book-entry form through The Depository Trust Corporation or any similar facility, such notices may be given to the holders of Series B-1 12.75% Preferred Stock in any manner permitted by such facility.

Section 17. Other Rights. The shares of Series B-1 12.75% Preferred Stock shall not have any rights, preferences, privileges or voting powers or relative, participating, optional or other special rights, or qualifications, limitations or restrictions thereof, other than as set forth herein or in the Certificate of Incorporation or as provided by applicable law and regulation.

[THIS SPACE LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Designation to be duly executed and acknowledged by its undersigned duly authorized officers this [•] day of [•], 2010.

USEC INC.

Bv

Name: James R. Mellor Title: Chairman of the Board

Attest:

Ву:

Name: Peter B. Saba Title: Secretary

SIGNATURE PAGE TO THE CERTIFICATE OF DESIGNATION OF SERIES B-1 CONVERTIBLE PREFERRED STOCK OF USEC INC.

EXHIBIT C

FORM OF SERIES B-2 CERTIFICATE OF DESIGNATION

FORM OF

CERTIFICATE OF DESIGNATION OF

SERIES B-2 11.5% CONVERTIBLE PREFERRED STOCK

of

USEC INC.

Pursuant to Section 151 of the General Corporation Law

of the State of Delaware

We, James R. Mellor, Chairman of the Board, and Peter B. Saba, Secretary, of USEC Inc., a corporation organized and existing under the General Corporation Law (the "DGCL") of the State of Delaware (the "Corporation"), in accordance with the provisions of Section 103 thereof, DO HEREBY CERTIFY:

That pursuant to the authority conferred upon the Board of Directors by the Certificate of Incorporation, the Board of Directors on May 24, 2010 adopted the following resolution creating a series of 150,000 shares of Preferred Stock, par value \$1.00 per share, designated as Series B-2 11.5% Convertible Preferred Stock:

RESOLVED, that pursuant to the authority vested in the Board of Directors in accordance with the provisions of the Certificate of Incorporation, a series of Preferred Stock of the Corporation be and it hereby is created, and that the designation and amount thereof and the powers, preferences and relative, participating, optional and other rights of the shares of such series, and the qualifications, limitations and restrictions thereof are as follows:

Section 1. Designation. The designation of this series of Preferred Stock, par value \$1.00 per share, of the Corporation is "Series B-2 11.5% Convertible Preferred Stock" ("Series B-2 11.5% Preferred Stock"). Each share of Series B-2 11.5% Preferred Stock shall be identical in all respects to every other share of Series B-2 11.5% Preferred Stock.

Section 2. Number of Shares. The authorized number of shares of Series B-2 11.5% Preferred Stock is 150,000. Shares of Series B-2 11.5% Preferred Stock that are redeemed, purchased or otherwise acquired by the Corporation, or converted into another series of Preferred Stock or into Common Stock, shall revert to authorized but unissued shares of Preferred Stock and shall not be reissued as shares of Series B-2 11.5% Preferred Stock.

Section 3. Definitions. As used herein with respect to Series B-2 11.5% Preferred Stock:

- (a) "ACP" shall mean the design, manufacture, construction, development, startup, completion, operation, financing, maintenance and improvement of a frontend nuclear fuel facility utilizing U.S. gas centrifuge enrichment technology and related infrastructure assets and properties.
- (b) "Affiliate" shall mean any Person controlling, controlled by or under common control with any other Person. For purposes of this definition, "control" (including "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of securities, partnership or other ownership interests, by contract or otherwise.
- (c) "Aggregate Outstanding Value" shall mean, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock, plus (2) for each share of Series C Preferred Stock then held by the Permitted Holders, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which the Permitted Holders' acquisition of such share was calculated, plus (3) for each share of Common Stock then held by the Permitted Holders, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Ordinary Common Stock purchased in the market, the Base Price upon which the Permitted Holders' acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock which have been added to the Liquidation Preference pursuant to Section 5(a).

- (d) "Approved Market" shall have the meaning ascribed to it in the definition of "Base Price."
- (e) "Automatic Redemption" shall mean an automatic redemption pursuant to Section 8(c) of this Certificate of Designation, Section 7(g) of the Series B-1 Certificate of Designation subsequent to a Conversion Election (as defined in the Series B-1 Certificate of Designation) or Section 8(c) of the Series B-1 Certificate of Designation.
- (f) "Automatic Redemption Adjustment" shall mean, for purposes of determining the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value, that if an Automatic Redemption has been effected prior to the date of determining such values, (1) the aggregate amount of the Liquidation Preference, as of the date of redemption, of a Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Aggregate Outstanding Value and Permitted Holder Outstanding Value and (2) the aggregate amount of the Liquidation Preference, as of the date of redemption, of such Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Original Issue Value and Permitted Holder Original Issue Value; provided, however, that, if at any time after any Automatic Redemption, such Permitted Holder's Deemed Holder Percentage is less than 8%, then such adjustment to the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value shall not be made.
 - (g) "B&W" shall mean Babcock & Wilcox Investment Company, a Delaware corporation.
- (h) "Base Price" shall mean for any date, the price determined by the first of the following clauses that applies: (1) if the Ordinary Common Stock is then listed or quoted on the New York Stock Exchange, The NASDAQ Stock Market or the American Stock Exchange (each an "Approved Market"), the arithmetic average of the daily volume weighted average prices per share of the Ordinary Common Stock for each of the 20 consecutive Trading Days immediately preceding (but not including) such date, as reported for the regular trading session (including any extensions thereof) on the primary Approved Market on which the Ordinary Common Stock is then listed or quoted (without regard to pre-open or after hours trading outside of such regular trading session on such Trading Day), as reported by Bloomberg Financial L.P. (or any successor thereof) using the HP function (or any equivalent thereof); (2) if the Ordinary Common Stock has not been listed or quoted on an Approved Market for a minimum of 20 consecutive Trading Days immediately preceding (but not including) such date and if prices for the Ordinary Common Stock are then quoted on the OTC Bulletin Board, the arithmetic average of the daily volume weighted average prices per share of the Ordinary Common Stock for each of the 20 consecutive Trading Days immediately preceding (but not including) such date, as quoted for the regular trading session on the OTC Bulletin Board; (3) if the Ordinary Common Stock has not been listed or quoted on the OTC Bulletin Board for a minimum of 20 consecutive Trading Days immediately preceding (but not including) such date and if prices for the Ordinary Common Stock are then reported in the "Pink Sheets" published by the National Quotation Bureau Incorporated (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Ordinary Common Stock so reported; or (4) in all other cases, the fair market value of a share of Ordinary Common Stock as determined by the Board of Directo
- (i) "Beneficially Own" shall mean "beneficially own" as defined in Rule 13d-3 promulgated under Section 13(d) of the Exchange Act or any successor provision thereto, and "Beneficial Ownership" shall have a correlative meaning.
 - (j) "Board of Directors" shall mean the board of directors of the Corporation or any duly authorized committee thereof.
- (k) "Business Day" shall mean any calendar day other than (1) a Saturday or Sunday or (2) a calendar day on which banking institutions in either the City of New York or Tokyo, Japan are authorized by law, regulation or executive order to remain closed.
 - (I) "Bylaws" shall mean the Amended and Restated Bylaws of the Corporation, as amended from time to time.
- (m) "Certificate of Designation" shall mean this Certificate of Designation of Series B-2 11.5% Convertible Preferred Stock of the Corporation, as amended from time to time.
 - (n) "Certificate of Incorporation" shall mean the Certificate of Incorporation of the Corporation, as amended from time to time.
 - (o) "Change of Control" shall mean the occurrence of any of the following:
 - (1) any Person shall Beneficially Own, directly or indirectly, through a merger, business

combination, purchase, or other transaction or series of transactions, shares of the Corporation's capital stock entitling such Person at such time to exercise 50% or more of the total voting power of the issued and outstanding capital stock having ordinary voting power to elect a majority of the Board of Directors, other than as a result of an acquisition of such stock by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this subsection (1), "Person" shall include any syndicate or group that would be deemed to be a "person" under Section 13(d)(3) of the Exchange Act);

- (2) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Stock is exchanged for or converted into cash, securities or other property, in each case other than a merger or consolidation:
 - (i) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Stock;
 - (ii) that is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding shares of the Common Stock solely into shares of any class or series of common stock of the surviving entity; or
 - (iii) where the issued and outstanding capital stock having voting power to vote generally to elect a majority of the Board of Directors outstanding immediately prior to such transaction is converted into or exchanged for such voting stock of the surviving or transferee Person constituting a majority of the outstanding shares of such voting stock of such surviving or transferee Person (immediately after giving effect to such issuance).
- (p) "Charter Amendment Approval" shall mean the approval of the stockholders of the Corporation necessary to amend the Corporation's Certificate of Incorporation to approve the authorization of Class B Common Stock and the proper filing of such amendment with the Secretary of State of the State of Delaware.
- (q) "Class B Common Stock" shall mean the Class B Common Stock of the Corporation, par value \$.10 per share, to be authorized by the Charter Amendment Approval.
- (r) "Closing Deadline Failure" shall mean, unless waived in writing (1) by the Corporation if such Closing Deadline Failure is as a result of breach by a Permitted Holder, (2) by the Permitted Holders if such Closing Deadline Failure is as a result of breach by the Corporation, or (3) by the Permitted Holders and the Corporation if such Closing Deadline Failure is not as a result of a breach by the Permitted Holders or the Corporation, either, (A) with respect to the Second Closing (as defined in the Securities Purchase Agreement), that the Second Closing shall not have occurred by June 30, 2011 and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.2 thereof, or (B) with respect to the Third Closing (as defined in the Securities Purchase Agreement), that the Third Closing shall not have occurred by the Third Closing Termination Date (as defined in the Securities Purchase Agreement) and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.3 thereof.
- (s) "Code" shall mean the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder by the Internal Revenue Service.
 - (t) "Common Stock" shall mean collectively, the Ordinary Common Stock and the Class B Common Stock.
- (u) "Conditional Commitment" shall mean a conditional commitment (as defined in 10 CFR 609.2) from DOE to the Corporation in an amount not less than \$2 billion, and specifying the detailed conditions to be satisfied for the DOE Financial Closing.
- (v) "Conversion Cap", with respect to a conversion hereunder, shall mean that the total number of shares of Class B Common Stock received upon such conversion shall not, when combined with the total number of shares of Class B Common Stock (1) issued or issuable upon the exercise of the Warrants and (2) issued by the Corporation upon conversion of securities issued pursuant to the Transactions (as defined in the Securities Purchase Agreement) exceed 49.99% of the total number of outstanding shares of Ordinary Common Stock and Class B Common Stock at the time of any such conversion, subject to adjustments for stock splits, stock dividends, reorganizations or similar transactions.
 - (w) "Corporation" shall have the meaning ascribed to it in the recitals.

- (x) "Corporation Plans" shall mean the Corporation's 1999 Equity Incentive Plan, as amended, and the Corporation's 2009 Equity Incentive Plan, as may be amended, the Corporation's 2009 Employee Stock Purchase Plan, as may be amended, and any similar plans entered into after the date hereof, and any inducement grants.
- (y) "Deemed Holder Percentage" shall mean, as to any Permitted Holder, the percentage resulting from the following calculation, (1)(A) the number of shares of Ordinary Common Stock equal to the quotient of (w) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on the outstanding shares of Series B Preferred Stock held by such Permitted Holder from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (x) the Base Price for the date of such calculation, plus (B) the number of outstanding of shares of (y) Series C Preferred Stock multiplied by 1000 plus, (z) if then outstanding, Class B Common Stock, in each case held by such Permitted Holder divided by (2)(A) the total number of shares of Ordinary Common Stock equal to the quotient of (v) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on all outstanding shares of Series B Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (w) the Base Price for the date of such calculation, plus (B) the total number of all outstanding shares of (x) Series C Preferred Stock multiplied by 1000 plus (y) if then outstanding, Class B Common Stock, plus (z) Ordinary Common Stock
 - (z) "DGCL" shall have the meaning ascribed to it in the Preamble
 - (aa) "Dividend" shall have the meaning ascribed to it in Section 5(a).
- (bb) "Dividend Payment Date" shall mean January 1, April 1, July 1 and October 1 of each year, commencing on April 1, 2010; provided that if any such Dividend Payment Date would otherwise occur on a day that is not a Business Day, such Dividend Payment Date shall instead be (and any Dividend payable on Series B-2 11.5% Preferred Stock on such Dividend Payment Date shall instead be payable on) the immediately succeeding Business Day.
- (cc) "<u>Dividend Period</u>" shall mean the period commencing on and including a Dividend Payment Date and shall end on and include the calendar day preceding the next Dividend Payment Date; <u>provided</u>, <u>however</u>, that with respect to any shares of Series B-2 11.5% Preferred Stock not outstanding for the entirety of any such Dividend Period, there shall be an initial pro-rated Dividend Period for such shares that shall commence on and include the issue date of such shares.
 - (dd) "Dividend Rate" shall mean 11.50% per annum.
 - (ee) "DOE" shall mean the United States Department of Energy.
- (ff) "DOE Financial Closing" shall mean the closing of a Loan Guarantee Agreement (as defined in 10 CFR 609.2), between DOE, an eligible lender, and the Corporation, pursuant to the Conditional Commitment, guaranteeing a loan or other debt obligation in an amount not less than \$2 billion for the ACP and there shall have been an initial draw of the funds guaranteed pursuant to the Loan Guarantee Agreement in an amount not less than the minimum amount permitted thereunder.
 - (gg) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
 - (hh) "Exchange Property" shall have the meaning ascribed to it in Section 10(a).
- (ii) "Excluded Lender" shall mean a bank or other financial institution providing indebtedness for borrowed money which is guaranteed by the Loan Guarantee Agreement (as defined in 10 CFR 609.2) pertaining to the DOE Financial Closing; provided, however "Excluded Lender" shall not include a Person providing funding or committed funding (pursuant to definitive binding agreements) for debt or equity of the Corporation in an amount of at least \$100,000,000 that is not guaranteed by such Loan Guarantee Agreement.
 - (jj) "Factor" shall be the Factor established in accordance with the provisions of Section 7(h)(1).
- (kk) "Final Determination" shall mean the earlier to occur of (1) the conclusion of the litigation or binding arbitration (as applicable), including any and all appeals (whether by final determination or the expiration of any applicable appeal periods), regarding the dispute between the Permitted Holders and the Corporation, or (2) a written agreement between the Corporation and the appropriate Permitted Holder or Permitted Holders resolving such dispute.
- (II) "Governmental Authority" shall mean any foreign governmental authority, the United States of America, any state of the United States and any political subdivision of any of the foregoing, and any agency, instrumentality, department, commission, board, bureau, central bank, authority, court, arbitral body or other tribunal, in each case whether executive,

legislative, judicial, regulatory or administrative, having jurisdiction over any of the Permitted Holders, the Corporation, any of the Corporation's Subsidiaries or their respective Property.

- (mm) "Initial Liquidation Preference" shall mean \$1,000 per share of Series B Preferred Stock.
- (nn) "Initial Preferred Director" shall have the meaning ascribed to it in Section 9(b)(1).
- (00) "Internal Reorganization Event" shall have the meaning ascribed to it in Section 10(c).
- (pp) "Investor Rights Agreement" shall mean that certain Investor Rights Agreement, dated [—], 2010 among the Corporation, Toshiba and B&W, as amended from time to time.
- (qq) "Junior Stock" shall mean the Common Stock and any other class or series of capital stock of the Corporation that ranks junior to the Series B Preferred Stock (1) as to the priority of payment of dividends and/or (2) as to the distribution of assets on any liquidation, dissolution or winding up of the Corporation. For the avoidance of doubt, Junior Stock shall include the Series A Preferred Stock and the Series C Preferred Stock.
- (rr) "<u>Liquidation Preference</u>" shall initially mean \$1,000 per share of Series B Preferred Stock; <u>provided, however</u>, that to the extent that the Corporation does not declare and pay a Dividend on a Dividend Payment Date pursuant to Section 5(a), an amount equal to the Dividend shall be added to the Liquidation Preference of such share on the applicable Dividend Payment Date.
 - (ss) "Orderly Sale Arrangement" shall have the meaning set forth in the Securities Purchase Agreement.
- (tt) "Ordinary Common Stock" shall mean the common stock of the Corporation, par value \$.10 per share. For the avoidance of doubt, the Ordinary Common Stock shall not include the Class B Common Stock.
- (uu) "Original Issuance Date" shall mean, with respect to each share of Series B Preferred Stock issued to the Permitted Holders, the date on which such share was issued by the Corporation.
- (vv) "Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, the aggregate Initial Liquidation Preference of all the shares of Series B Preferred Stock issued to the Permitted Holders excluding those shares issued as a Dividend.
- (ww) "Parity Stock" shall mean any class or series of stock of the Corporation that ranks equally with Series B-2 11.5% Preferred Stock (1) in the priority of payment of dividends and/or (2) in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation (in each case, without regard to whether dividends accrue cumulatively or non-cumulatively). For the avoidance of doubt, Parity Stock shall include the Series B-1 12.75% Preferred Stock.
 - (xx) "Permit" shall mean any approval, authorization, certificate, consent, license or permit of or from any Governmental Authority.
- (yy) "Permitted Holder Material Breach" shall mean a material breach of the Securities Purchase Agreement or the Investor Rights Agreement by any Permitted Holder.
- (zz) "Permitted Holder Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, for any Permitted Holder, the aggregate Initial Liquidation Preference of all shares of Series B Preferred Stock issued to such Permitted Holder excluding those shares issued as a Dividend.
- (aaa) "Permitted Holder Outstanding Value" shall mean, as to any Permitted Holder, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock then held by such Permitted Holder, plus (2) for each share of Series C Preferred Stock then held by a Permitted Holder, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which such Permitted Holder's acquisition of such share was calculated, plus (3) for each share of Common Stock then held by such Permitted Holder, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Ordinary Common Stock purchased in the market, the Base Price upon which such Permitted Holder's acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock, which have been added to the Liquidation Preference pursuant to Section 5(a).
- (bbb) "Permitted Holders" shall mean (1) [Toshiba U.S. Co.] and its Wholly-Owned Affiliates, (2) B&W and its Wholly-Owned Affiliates, (3) a special purpose entity jointly and wholly controlled by Toshiba and B&W and (4)

Westinghouse Electric Company, LLC, to the extent it is controlled by Toshiba or a Permitted Holder described under (1) above; <u>provided</u>, <u>however</u>, that each Permitted Holder must be a U.S. Person.

- (ccc) "Person" shall mean any individual, corporation, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization. Governmental Authority or any other entity.
 - (ddd) "Preferred Director" shall have the meaning ascribed to it in Section 9(b)(2).
 - (eee) "Preferred Stock" shall mean any and all series of preferred stock, par value \$1.00 per share, of the Corporation, including the Series B Preferred Stock.
 - (fff) "Property" shall mean any interest in any kind of property or asset, whether real, personal or mixed, tangible or intangible.
 - (ggg) "Qualified Director" shall mean any individual reasonably acceptable to the Nominating and Governance Committee of the Board of Directors.
 - (hhh) "Regulatory Bodies" shall mean the DOE and the U.S. Nuclear Regulatory Commission, and any successor Governmental Authorities thereto.
 - (iii) "Reorganization Event" shall have the meaning ascribed to it in Section 10(a).
 - (jjj) "Sale Plan" shall have the meaning ascribed to it in Section 7(c)(1).
- (kkk) "Securities Purchase Agreement" shall mean that certain Securities Purchase Agreement, dated May 25, 2010, among the Corporation, Toshiba and B&W, as amended from time to time.
- (III) "Senior Stock" shall mean any class or series of capital stock of the Corporation that ranks senior to the Series B Preferred Stock (1) as to the priority of dividends and/or (2) as to the distribution of assets on any liquidation, dissolution or winding up of the Corporation.
- (mmm) "Series A Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series A Junior Participating Preferred Stock."
 - (nnn) "Series B Preferred Stock" shall mean the Series B-1 12.75% Preferred Stock together with the Series B-2 11.5% Preferred Stock.
- (000) "Series B-1 12.75% Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series B-1 12.75% Convertible Preferred Stock."
- (ppp) "Series B-1 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-1 12.75% Preferred Stock, as filed with the Secretary of State of the State of Delaware.
 - (qqq) "Series B-2 11.5 % Preferred Stock" shall have the meaning ascribed to it in Section 1.
- (rrr) "Series B-2 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-2 11.5% Preferred Stock, as filed with the Secretary of State of the State of Delaware.
- (sss) "Series C Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series C Participating Convertible Preferred Stock."
- (ttt) "Share Issuance Approval" shall mean the approval of the stockholders of the Corporation necessary to approve the conversion of all the Series B Preferred Stock and the Series C Preferred Stock, and the exercise of all the Warrants, for Common Stock for purposes of Section 312.03 of the New York Stock Exchange Listed Company Manual, or if shares of the Ordinary Common Stock become listed and traded on another Approved Market, the approval required by such Approved Market, or the time at which all such approvals shall for any reason become inapplicable or not required so as to permit all such conversions and exercises.
- (uuu) "Share Issuance Limitation" shall mean the total number of shares of Common Stock or securities convertible into Common Stock that can be issued by the Corporation upon conversion or exercise of securities issued pursuant to the Transactions (as defined in the Securities Purchase Agreement) in accordance with the rules and regulations of the Approved

Market on which shares of the Corporation's equity securities are listed or traded prior to receipt of the Share Issuance Approval.

- (vvv) "Subsidiary" of any Person shall mean any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than 50% of (1) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (2) the interest in the capital or profits of such partnership, joint venture or limited liability company or (3) the beneficial interest in such trust or estate is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries. Notwithstanding the foregoing, American Centrifuge Manufacturing, LLC, a Delaware limited liability company, shall not be considered a Subsidiary of B&W or the Corporation.
 - (www) "SWU" shall have the meaning ascribed to it in the definition of "SWU Consideration."
- (xxx) "SWU Consideration" shall mean the fair market value of separative work units with respect to low enriched uranium ("SWU") (as determined reasonably and in good faith by the Board of Directors, taking into account the applicable volume of SWU, the then-current market price for SWU and other relevant factors) provided by the Corporation to the Permitted Holders minus any consideration paid by the Permitted Holders for such SWU.
- (yyy) "Third Party Financing" shall mean the funding or committed funding (pursuant to definitive binding agreements) for debt or equity of the Corporation in an amount of at least \$100,000,000 from a third party that is not an Affiliate of the Corporation, a Japanese export credit agency, a U.S. Governmental Authority or an Excluded Lender where (1) such funds, together with such other additional funds available to the Corporation at such time, is necessary and sufficient to consummate the DOE Financial Closing, and (ii) the third-party requires, as a condition to the funding, that the Preferred Stock be converted in accordance with the terms hereof.
- (zzz) "Third Party Transfer" shall mean an irrevocable Transfer in compliance with Section 11 of all legal ownership, Voting Control and Beneficial Ownership of any share or shares of Series B-2 11.5% Preferred Stock to a Person other than a Permitted Holder or its Affiliates.
 - (aaaa) "Toshiba" shall mean Toshiba Corporation, a corporation organized under the laws of Japan.
- (bbbb) "Trading Day" shall mean any day on which shares of the Corporation's equity securities are traded, or able to be traded, on the Approved Market on which shares of the Corporation's equity securities are listed or traded.
- (cccc) "Transfer" shall mean, with respect to any shares of Series B-2 11.5% Preferred Stock, any direct or indirect assignment, sale, exchange, transfer, tender or other disposition of such shares or any interest therein, whether voluntary or involuntary, by operation of law or otherwise (and includes any sale or other disposition in any one transaction or series of transactions and the grant or transfer of an option or derivative security covering such shares), and any agreement, arrangement or understanding, whether or not in writing, to effect any of the foregoing; provided, however, that a "Transfer" shall not occur simply as a result of the grant of a proxy in connection with a solicitation of proxies subject to the provisions of Section 14 of the Exchange Act.
- (dddd) "<u>U.S. Person</u>" shall mean any person that is treated as a "United States Person" under Code Section 7701(a)(30) and that provides an IRS Form W-9 (or successor form), evidencing a complete exemption from United States withholding tax (including backup withholding tax), on or before the time at which it acquires securities pursuant to this Certificate of Designation.
- (eeee) "Voting Control" shall mean, with respect to a share or shares of Series B-2 11.5% Preferred Stock, the power, whether exclusive or shared, revocable or irrevocable, to vote or direct the voting of such share or shares of Series B-2 11.5% Preferred Stock, by proxy, voting agreement or otherwise.
- (ffff) "Warrants" shall mean those warrants to purchase Class B Common Stock or Series C Preferred Stock originally issued by the Corporation to the Permitted Holders pursuant to the Securities Purchase Agreement.
- (gggg) "Wholly-Owned Affiliate" shall mean, as to any Person, any Affiliate that, directly or indirectly, is wholly-owned and controlled (other than by contract) by a Person, or any other Affiliate to which the Corporation, in its sole discretion, consents.

Section 4. Titles and Subtitles; Interpretation. The titles and subtitles used in this Certificate of Designation are

used for convenience only and are not to be considered in construing or interpreting this Certificate of Designation. The definitions contained in this Certificate of Designation are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such terms.

Section 5 Dividends

(a) Rate. Holders of Series B-2 11.5% Preferred Stock shall be entitled to receive, when, as and if declared by the Board of Directors, out of funds or assets available therefor, subject to the provisions of the DGCL, on each share of Series B-2 11.5% Preferred Stock, Dividends with respect to each Dividend Period in an amount equal to the Dividend Rate on the Liquidation Preference per share of Series B-2 11.5% Preferred Stock, payable, at the Corporation's election, in (1) cash, (2) additional shares (including fractional shares) of Series B-2 11.5% Preferred Stock having a deemed value of \$1,000 per share for purposes of the number of such additional shares or (3) any combination of (1) and (2) (the "Dividend"). If and to the extent that the Corporation does not pay the entire Dividend for a particular Dividend Period on the applicable Dividend Payment Date for such period, the amount of such Dividend not paid shall be added to the Liquidation Preference in accordance with the definition thereof. Dividends payable at the Dividend Rate shall begin to accrue (whether or not earned or declared, whether or not there are funds legally available for the payment thereof and whether or not restricted by the terms of any of the Corporation's indebtedness outstanding at any time) and be cumulative from the Original Issuance Date, shall compound on each Dividend Payment Date (i.e., no Dividends shall accrue on other Dividends unless and until the first Dividend Payment Date for such other Dividends has passed without such other Dividends having been paid on such date) and shall be payable in arrears on the first Dividend Payment Date after such Dividend Period. Dividends that are payable on Series B-2 11.5% Preferred Stock in the form of additional shares of such stock shall, except as specifically provided in this Certificate of Designation, have all rights granted hereunder, including the payment of Dividends. Dividends that are payable on Series B-2 11.5% Preferred Stock on any Dividend Payment Date shall be payable to holders of record of Series B-2 11.5% Preferred Stock as they appear on the stock register of the Corporation on the record date for such Dividend, which shall be the date 10 Business Days prior to the applicable Dividend Payment Date, or such other date as determined by the Board of Directors. The Corporation shall elect the form of such payment by giving notice at least 5 Business Days prior to the applicable Dividend Payment Date. If no such notice is given, the Corporation shall be deemed to have elected a payment through the issuance of shares of Series B-2 11.5% Preferred Stock. Dividends paid on the shares of Series B-2 11.5% Preferred Stock in an amount less than accumulated and unpaid Dividends payable thereon shall be allocated pro rata on a share-by-share basis among all such shares at the time outstanding.

Dividends payable at the Dividend Rate on Series B-2 11.5% Preferred Stock in respect of any Dividend Period shall be computed on the basis of a 360-day year consisting of twelve 30-day months. The amount of Dividends payable at the Dividend Rate on Series B-2 11.5% Preferred Stock on any date prior to the end of a Dividend Period, and for the initial Dividend Period, shall be computed on the basis of a 360-day year consisting of twelve 30-day months, and actual days elapsed over such 30-day months.

- **(b) Priority of Dividends.** Subject to any approvals required pursuant to Section 9, such Dividends (payable in cash, securities or other property) as may be determined by the Board of Directors may be declared and paid on any capital stock, including Common Stock and other Junior Stock, from time to time out of any funds legally available for such payment.
- (c) Payment in Shares. Any shares of Series B-2 11.5% Preferred Stock paid as a Dividend pursuant to this Section 5 shall be duly authorized, validly issued, fully paid and non-assessable, and shall be free of preemptive rights and free of any lien or adverse claim.

Section 6. Liquidation Rights.

(a) Voluntary or Involuntary Liquidation. In the event of any liquidation, dissolution or winding up of the affairs of the Corporation, whether voluntary or involuntary, holders of Series B-2 11.5% Preferred Stock shall be entitled to receive on par with each share of Parity Stock ranking equally with Series B-2 11.5% Preferred Stock in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation for each share of Series B-2 11.5% Preferred Stock, out of the assets of the Corporation or proceeds thereof available for distribution to stockholders of the Corporation, and after satisfaction of all liabilities and obligations to creditors of the Corporation, before any distribution of such assets or proceeds is made to or set aside for the holders of Junior Stock, an amount equal to the greater of (1) the Liquidation Preference per share of Series B-2 11.5% Preferred Stock plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference from and including the immediately preceding Dividend Payment Date to, but excluding, the date fixed for such liquidation, dissolution or winding up of the Corporation and (2) the per share amount of all cash, securities and other property (such securities or other property having a value equal to its fair market value as reasonably and in good faith determined by the Board of Directors) to be distributed in respect of the Common Stock such holder would have been entitled to receive had it converted such Series B-2 11.5% Preferred Stock (without regard to the Conversion Cap or

Share Issuance Limitation) immediately prior to the date fixed for such liquidation, dissolution or winding up of the Corporation. To the extent that such amount is paid in full to all holders of Series B-2 11.5% Preferred Stock and all holders of Parity Stock ranking equally with Series B-2 11.5% Preferred Stock in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation, the holders of other capital stock of the Corporation shall be entitled to receive all remaining assets of the Corporation (or proceeds thereof) according to their respective rights and preferences.

- (b) Partial Payment. If, in connection with any distribution described in Section 6(a) above, the assets of the Corporation or proceeds thereof are not sufficient to pay the Liquidation Preferences in full to all holders of Series B-2 11.5% Preferred Stock and all holders of Parity Stock ranking equally with Series B-2 11.5% Preferred Stock in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation, then the amounts paid to the holders of Series B-2 11.5% Preferred Stock and to the holders of all such other Parity Stock shall be paid pro rata in accordance with the respective aggregate Liquidation Preferences of the holders of Series B-2 11.5% Preferred Stock and the holders of all such other Parity Stock.
- (c) Merger, Consolidation and Sale of Assets Not Liquidation For purposes of this Section 6, the merger or consolidation of the Corporation with any other corporation or other Person, including a merger or consolidation in which the holders of Series B-2 11.5% Preferred Stock receive cash, securities or other property for their shares, or the sale, lease or exchange (for cash, securities or other property) of all or substantially all of the assets of the Corporation, shall not constitute a liquidation, dissolution or winding up of the Corporation, but shall instead be subject to the provisions of Section 10.

Section 7. [Reserved].

Section 8. Other Conversion.

(a) Conversion by the Corporation.

- (1) Conversion Upon Third Party Financing. Effective upon the DOE Financial Closing that follows or is contemporaneous with a Third Party Financing or immediately prior thereto, the Corporation may convert all of the outstanding shares of Series B-2 11.5% Preferred Stock (i) if the Charter Amendment Approval has been obtained, into Class B Common Stock, or (ii) if the Charter Amendment Approval has not been obtained, into Series C Preferred Stock, in either case into the number of shares of the Class B Common Stock or Series C Preferred Stock, as applicable, equal to the quotient of (A) 120% of the sum of (i) the Liquidation Preference plus (ii) an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-2 11.5% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (B) the Base Price for the date that the Corporation provides notice pursuant to Section 8(a)(2).
- (2) Conversion Timing. If the Corporation elects to convert pursuant to this Section 8(a), the Corporation shall provide written notice to the Permitted Holders of record at their respective last addresses appearing on the books of the Corporation. Such notice shall state the conversion date of Series B-2 11.5% Preferred Stock, which date shall be no less than 5 days and no more than 60 days from the date of such notice; <u>provided, however</u>, that the effectiveness of the conversion (and the Corporation's right and obligation to effect the conversion) shall be conditioned upon the DOE Financial Closing. Notwithstanding the foregoing, if, after delivery of such notice, the Corporation desires to specify a different conversion date, the Corporation shall not be required to notify the Permitted Holders of such change until after the conversion is effected unless such changed conversion date is more than 15 days prior to or after the original conversion date. The conversion date shall be the date specified in such written notice or such different date as specified by the Corporation in accordance with this Section 8(a)(2).

(b) Conversion by the Permitted Holders.

(1) Post-Third Closing Conversion. At any time and from time to time after the Third Closing (as defined in the Securities Purchase Agreement), any Permitted Holder's shares of Series B-2 11.5% Preferred Stock shall be converted, in whole or in part, upon the request of such Permitted Holder, subject to the Conversion Cap, into the number of shares of Class B Common Stock equal to the quotient of (A) the Liquidation Preference plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-2 11.5% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion and (B) the Base Price for the conversion date specified in the written notice provided by such Permitted Holder pursuant to Section 8(b)(2). Shares of Series B-2 11.5% Preferred Stock not converted as a result of the foregoing limitations shall remain outstanding except as provided herein.

(2) Conversion Timing. If a Permitted Holder elects to convert pursuant to this Section 8(b), such Permitted Holder shall provide written notice to the Corporation. Such notice shall state the conversion date of Series B-2 11.5% Preferred Stock, which date shall be no less than 5 days and no more than 60 days from the date of such notice. The conversion date shall be the date specified in such written notice.

(c) Automatic Conversion and Redemption. On December 31, 2016, all outstanding shares of Series B-2 11.5% Preferred Stock shall be automatically converted, without any action on the part of the holder and subject to the Conversion Cap, into the number of shares of Class B Common Stock equal to the quotient of (i) the Liquidation Preference plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-2 11.5% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion and (ii) the Base Price for December 31, 2016. Shares of Series B-2 11.5% Preferred Stock not converted as a result of the foregoing limitation shall remain outstanding except as provided herein. If shares of Series B-2 11.5% Preferred Stock remain outstanding on February 28, 2017 due to the Conversion Cap, the Corporation shall, subject to the provisions of the DGCL, redeem all outstanding shares of Series B-2 11.5% Preferred Stock for cash in an amount equal to the Liquidation Preference of such shares plus an amount per share equal to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-2 11.5% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of redemption. If the Corporation fails to pay such redemption amount by March 15, 2017, the Conversion Cap shall no longer apply and all outstanding shares of Series B-2 11.5% Preferred Stock shall be automatically converted, without any action on the part of the holder, into the number of shares of Class B Common Stock equal to the quotient of (A) the Liquidation Preference plus an amount per share equal

to accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B-2 11.5% Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, such date of conversion and (B) the Base Price for March 15, 2017. In the event of any automatic conversion or redemption pursuant to this Section 8(c), the conversion or redemption shall be deemed to have been effected at the time that the event triggering such automatic conversion or redemption occurred.

- (d) Conversion Mechanics. A Permitted Holder shall cease to be a record holder of each share of Series B-2 11.5% Preferred Stock on the date such share is converted. As promptly as practicable on or after the conversion date (and in any event no later than three Trading Days thereafter), the Corporation or its agent, including its transfer agent, shall issue the number of shares of Class B Common Stock or Series C Preferred Stock (including fractional shares) issuable pursuant to Section 8(a), (b) or (c). Any such certificate or certificates shall be delivered by the Corporation or its agent, including its transfer agent, to the appropriate holder on a book-entry basis or by mailing certificates evidencing the shares to the holders at their respective addresses as set forth in the records of the Corporation, subject in each case to the provisions of Section 9 of the Securities Purchase Agreement.
- (e) Reservation of Class B Common Stock. Subject to receiving the Charter Amendment Approval and for as long as any shares of Series B Preferred Stock remain outstanding, the Corporation shall at all times reserve and keep available, free from preemptive rights, out of its authorized but unissued Ordinary Common Stock or Class B Common Stock, or shares of Ordinary Common Stock or Class B Common Stock held in treasury by the Corporation, for the purpose of effecting the conversion of the Series B Preferred Stock, the full number of shares of Ordinary Common Stock or Class B Common Stock then issuable upon the conversion of all the shares of Series B Preferred Stock then outstanding. For purposes hereof, reservations hereunder shall be at the Base Price equal to the closing price of the Corporation's Ordinary Common Stock on the New York Stock Exchange on the second to last Trading Day prior to the date of the Securities Purchase Agreement; provided, however, if the Base Price for the date four Trading Days prior to the First Closing, the Second Closing or the Third Closing (each as defined in the Securities Purchase Agreement) or on June 30 of any year is less than such amount, then that lower amount shall be used as the Base Price for purposes of this calculation. All shares of Class B Common Stock delivered upon conversion of Series B Preferred Stock shall have been duly authorized and validly issued and shall be fully paid and nonassessable, and shall be free from preemptive rights and free of any lien or adverse claim.
- (f) Partial Conversion. In case of any conversion of any of the shares of Series B Preferred Stock at the time outstanding, the shares to be converted shall be selected pro rata among the shares of Series B Preferred Stock held by each Permitted Holder and among each such Permitted Holder's shares of Series B-2 11.5% Preferred Stock and Series B-1 12.75% Preferred Stock. If fewer than all of the shares represented by any certificate are converted, a new certificate shall be issued representing the unconverted shares without charge to the holder thereof.
- (g) Taxes. The Corporation shall pay any and all taxes that may be payable in respect of the issue or delivery of shares of Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock on conversion of Series B-2 11.5% Preferred Stock. The Corporation shall not, however, be required to pay any tax that may be payable in respect of any Transfer involved in the issue and delivery of shares of Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock in a name other than that in which Series B-2 11.5% Preferred Stock so converted was registered, and no

such issue or delivery shall be made unless and until the Person requesting such issue has paid to the Corporation the amount of any such tax, or has established to the satisfaction of the Corporation that such tax has been paid.

Section 9. Voting Rights.

(a) General. The holders of shares of Series B-2 11.5% Preferred Stock shall not be entitled to vote, except as otherwise provided herein or required by applicable law.

(b) Election of Directors.

- (1) Effective as of the first Original Issuance Date, the number of directors constituting the Board of Directors shall be increased by two Persons and the holders of a majority of the outstanding Series B Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall be entitled to elect two Qualified Directors to the Board of Directors (each such director, an "Initial Preferred Director") until the earliest to occur of (i) a Closing Deadline Failure as a result of a Permitted Holder Material Breach at a time when the Securities Purchase Agreement is terminable pursuant to Sections 10.2(d) and 10.3(d) thereof, (ii) a Change of Control or (iii) such time as the Permitted Holders' Aggregate Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of the Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the outstanding Series B Preferred Stock to elect the Initial Preferred Directors shall cease, (B) the term of office of the Initial Preferred Directors shall immediately and automatically terminate, (C) the Initial Preferred Directors will no longer be qualified to serve and (D) the number of directors constituting the Board of Directors shall be immediately and automatically reduced by two Persons.
- (2) Effective as of the first Original Issuance Date and at such time as when the Permitted Holders do not have the right to elect the Initial Preferred Directors pursuant to Section 9(b)(1)(iii) and any Permitted Holder's Permitted Holder Outstanding Value is greater than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or, (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, the number of directors constituting the Board of Directors shall be increased by one Person and the holders of a majority of the outstanding Series B Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall be entitled to elect one Qualified Director to the Board of Directors (such director, the "Preferred Director") until the earliest to occur of (i) an event described in Section 9(b)(1)(i) or (ii) or (ii) such time as each Permitted Holder's Permitted Holder Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the outstanding Series B Preferred Stock to elect the Preferred Director shall cease, (B) the term of office of the Preferred Director shall immediately and automatically terminate, (C) the Preferred Director will no longer be qualified to serve and (D) the number of directors constituting the Board of Directors shall be immediately and automatically reduced by one Person.
- (3) For the avoidance of doubt, except for the increase or decrease in the number of directors provided for herein, nothing in this Section 9(b) shall prohibit the Board of Directors from fixing the number of directors constituting the Board of Directors pursuant to the Bylaws.
- (4) Term. Subject to the provisions of this Section 9(b), each Initial Preferred Director or the Preferred Director, as applicable, shall serve until the next annual meeting of the stockholders of the Corporation and until his or her successor is elected and qualified in accordance with this Section 9(b) and the Bylaws, unless any such Initial Preferred Director or the Preferred Director, as applicable, is earlier removed in accordance with the Bylaws, resigns or is otherwise unable to serve; provided, however, that only the holders of a majority of the outstanding shares of the Series B Preferred Stock may remove any such Initial Preferred Director or the Preferred Director, as applicable, without cause at any time and the holders of a majority of the voting power of the outstanding shares of the capital stock of the Corporation entitled to vote on the matter may remove any such Initial Preferred Director or the Preferred Director, as applicable, with cause at any time. Subject to the provisions of this Section 9(b), in the event any Initial Preferred Director or the Preferred Director, as applicable, is removed, resigns or is unable to serve as a member of the Board of Directors, the holders of a majority of the outstanding shares of Series B Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall have the right to fill such vacancy. Each Initial Preferred Director or the Preferred Director, as applicable, may only be elected to the Board of Directors by the holders of the Series B Preferred Stock in accordance with this Section 9(b), and any such Initial Preferred Director's or the Preferred Director's seat, as applicable, shall otherwise remain vacant.
- (c) Class Voting Rights as to Particular Matters. In addition to any other vote or consent of stockholders required by law or by the Certificate of Incorporation, the vote or consent of the holders of at least a majority of the outstanding shares

of Series B Preferred Stock, voting together as a single class to the exclusion of the holders of the Common Stock and any other series of Preferred Stock, then outstanding and entitled to vote on the matter, given in person or by proxy, either in writing without a meeting or by vote at any meeting called for the purpose, shall be necessary for effecting or validating any of the actions described in (x) Section 9(c)(1) if any shares of Series B Preferred Stock are outstanding and (y) Sections 9(c)(2) and 9(c)(3) if the number shares of Series B Preferred Stock outstanding is greater than 10% of all of the shares of Series B Preferred Stock issued to the Permitted Holders, in each case excluding shares issued as a Dividend.

(1) Amendment of Series B Preferred Stock. Any amendment, alteration or repeal (by merger, consolidation or otherwise) of any provision of the Certificate of Incorporation or this Certificate of Designation so as to adversely affect the powers, preferences and relative participating, optional and other rights of Series B-2 11.5% Preferred Stock.

(2) Dividends, Repurchase and Redemption.

- (A) The declaration or payment of any dividend or distribution of Common Stock or other Junior Stock (other than a dividend payable solely in Junior Stock <u>provided</u> such dividend is not treated as a distribution of property for purposes of Section 305 of the Code, the Treasury Regulations promulgated thereunder or any successor provision); or
- (B) the purchase, redemption or other acquisition for consideration by the Corporation, directly or indirectly, of any Common Stock, other Junior Stock or Parity Stock, (except as necessary (i) to effect a reclassification of Junior Stock for or into other Junior Stock, (ii) to effect a reclassification of Parity Stock into Junior Stock, (iv) to effect the exchange or conversion of one share of Junior Stock for or into another share of Junior Stock, (v) to effect the exchange or conversion of one share of Parity Stock into Junior Stock for or into another share of Parity Stock with the same or lesser per share liquidation amount (vi) to effect the exchange or conversion of one share of Parity Stock into Junior Stock or (vii) pursuant to the Corporation Plans).
- (3) Issuance of Senior Stock or Parity Stock. Prior to the Third Closing (as defined in the Securities Purchase Agreement), the issuance of any Senior Stock or Parity Stock, except as specifically provided for herein or in the Certificate of Designation for the Series B-1 12.75% Preferred Stock.
- (d) Changes after Provision for Redemption or Conversion. No vote or consent of the holders of Series B-2 11.5% Preferred Stock shall be required pursuant to Section 9(c)(x) or Section 9(c)(y) if, at or prior to the time when any such vote or consent would otherwise be required pursuant to such Section, all outstanding shares of Series B-2 11.5% Preferred Stock (1) shall have been redeemed or converted, or (2) shall have been irrevocably elected for redemption or conversion in accordance with Sections 7(f), 7(g) or 8(a) and will, subject to the passage of time, be redeemed or converted; provided, that if, on or before the redemption date specified by the Corporation, all funds required for the redemption of the shares called for redemption have been deposited by the Corporation in trust for the benefit of the Permitted Holders with a bank or trust company doing business in the City of New York having a capital and surplus of at least \$500 million and selected by the Board of Directors, so as to be and continue to be available solely therefor, then, notwithstanding that any certificate for any share so called for redemption has not been surrendered for cancellation, on and after the redemption date, Dividends shall cease to accrue on all shares so called for redemption, all shares so called for redemption shall no longer be deemed outstanding and all rights with respect to such shares shall forthwith on such redemption date cease and terminate, except only the right of the Permitted Holders thereof to receive the amount payable on such redemption from such bank or trust company, without interest. Any funds unclaimed at the end of three years from the redemption date shall, to the extent permitted by law, be released to the Corporation, after which time the Permitted Holders of the shares so called for redemption shall look only to the Corporation for payment of the redemption price of such shares.

Section 10. Reorganization Events.

- (a) In the event of:
 - (1) any consolidation or merger of the Corporation with or into another Person or of another Person with or into the Corporation; or
 - (2) any sale, transfer, lease or conveyance to another Person of all or substantially all of the property and assets of the Corporation,

in each case in which holders of Ordinary Common Stock would be entitled to receive cash, securities or other property for their shares of Ordinary Common Stock (any such event specified in this Section 10(a), a "Reorganization Event"), the outstanding shares of Series B Preferred Stock shall be deemed for the purposes of this Section 10 only to be converted into

the number of shares of Ordinary Common Stock equal to the quotient of (x) the Liquidation Preference *plus* an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on such shares of Series B Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding the date of conversion and (y) the Base Price for the date of effectiveness of such Reorganization Event and each such share shall, (A) become convertible into securities and other property receivable in such Reorganization Event by and in the same relative amounts as a holder of Ordinary Common Stock other than securities issued or other property distributed by such holder or its Affiliates; <u>provided, however</u>, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Corporation or any other Person, then the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ only to the extent that the then existing designation and powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of the Ordinary Common Stock and Series B Preferred Stock differ as provided in this Certificate of Designation or the Certificate of Designation for the Series B-1 12.75% Preferred Stock (including, without limitation, with respect to the voting rights and conversion provisions thereof) or, at the Corporation's sole discretion, (B) be redeemed by the Corporation for a cash price equal to 105% of the fair value of the consideration that would have otherwise been received under subsection (A), as determined by the Board of Directors acting reasonably and in good faith (such cash, securities and other property, the "Exchange Property").

- (b) Subject to the restrictions set forth in Section 10(a), in the event that holders of the shares of the Ordinary Common Stock have the opportunity to elect the form of Exchange Property to be received in such transaction, the Exchange Property that holders of the Series B Preferred Stock shall be entitled to receive shall be determined by the holders of a majority of the outstanding shares of Series B Preferred Stock.
- (c) Notwithstanding anything in this Certificate of Designation to the contrary, Section 10(a) shall not apply to a merger, consolidation, asset sale, reorganization or statutory share exchange (1) among the Corporation and its direct and indirect Subsidiaries or (2) between the Corporation and any Person for the primary purpose of changing the domicile of the Corporation (an "Internal Reorganization Event") and no such transaction shall be deemed to be a Reorganization Event. Without limiting the rights of the holders of the Series B Preferred Stock set forth in Section 9(c)(1), the Corporation shall not effectuate an Internal Reorganization Event without the consent of the holders of a majority of the outstanding shares of the Series B Preferred Stock unless the Series B Preferred Stock shall be outstanding as a class or series of preferred stock of the surviving entity having the same rights, terms, preferences, liquidation preference and accrued and unpaid Dividends as the Series B Preferred Stock in effect immediately prior to such Internal Reorganization Event, as adjusted for such Internal Reorganization Event pursuant to this Certificate of Designation after giving effect to any such Internal Reorganization Event.
- (d) The Corporation (or any successor) shall, within 20 days after the occurrence of any Reorganization Event or Internal Reorganization Event, provide written notice to the holders of the Series B Preferred Stock of the occurrence of such event and, in the case of a Reorganization Event, of the kind and amount of the cash, securities or other property that constitutes the Exchange Property. Failure to deliver such notice shall not affect the operation of this Section 10 or the validity of any Reorganization Event or Internal Reorganization Event.

Section 11. Restrictions.

- (a) Notwithstanding anything in this Certificate of Designation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, Permitted Holders may not Transfer shares of Series B-2 11.5% Preferred Stock if such Transfer would require approvals from or filings with any Regulatory Bodies in order not to adversely affect the Permits or regulatory status of the Corporation or its Subsidiaries, unless such approvals and/or filings have been made and received; provided, however, this Section 11(a) shall not apply to any transfer where the transferee received Ordinary Common Stock pursuant to the terms hereof.
- (b) Notwithstanding anything in this Certificate of Designation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, the conversion of Series B-2 11.5% Preferred Stock for Ordinary Common Stock shall also be subject to the requirements of Section 9.2 of the Securities Purchase Agreement.
 - (c) Any purported conversion or Transfer of Series B-2 11.5% Preferred Stock in violation of these restrictions shall be null and voidab initio.
- Section 12. Record Holders. To the fullest extent permitted by applicable law, the Corporation may deem and treat the record holder of any share of Series B-2 11.5% Preferred Stock as the true and lawful owner thereof for all purposes.

 $\textbf{Section 13. No Standing to Bring Derivative Action}. \ \textbf{Notwithstanding any provision of the DGCL}, the Rules of \textbf{Section 13. No Standing to Bring Derivative Action}. \\$

the Court of Chancery of the State of Delaware or any other applicable law, rule or regulation which would otherwise confer such standing or empower a holder of Series B-2 11.5% Preferred Stock to take such action, no holder of any share of Series B-2 11.5% Preferred Stock shall have standing to bring an action, suit or proceeding derivatively or otherwise in the right of the Corporation.

Section 14. Legends. All certificates representing shares of Series B-2 11.5% Preferred Stock shall bear a legend or other restriction substantially to the following effect (it being agreed that if such shares are not certificated, other appropriate restrictions shall be implemented to give effect to the following):

"THIS SECURITY WAS ORIGINALLY ISSUED IN A TRANSACTION EXEMPT FROM REGISTRATION UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR AS MAY BE HELD BY A PERSON DEEMED AN "AFFILIATE" (AS DEFINED IN RULE 144 UNDER THE SECURITIES ACT) OF THE ISSUER OF THIS SECURITY, AND THIS SECURITY MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN APPLICABLE EXEMPTION THEREFROM. THE HOLDER OF THIS SECURITY AGREES FOR THE BENEFIT OF THE COMPANY THAT (A) THIS SECURITY MAY BE OFFERED, RESOLD, PLEDGED OR OTHERWISE TRANSFERRED, ONLY (I) IN A TRANSACTION NOT INVOLVING A PUBLIC OFFERING, (II) PURSUANT TO ANY OTHER EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, INCLUDING RULE 144 UNDER THE SECURITIES ACT (IF AVAILABLE), (III) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR (IV) TO THE COMPANY OR ANY OF ITS SUBSIDIARIES, IN EACH OF CASES (I) THROUGH (IV) IN ACCORDANCE WITH ANY APPLICABLE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES, AND (B) THE HOLDER WILL NOTIFY ANY SUBSEQUENT PURCHASER OF THIS SECURITY FROM IT OF THE RESALE RESTRICTIONS REFERRED TO IN (A) ABOVE. THIS SECURITY MAY ONLY BE TRANSFERRED IN ACCORDANCE WITH THE TERMS OF THE CERTIFICATE OF DESIGNATION OF SERIES B-2 CONVERTIBLE PREFERRED STOCK OF USEC INC. (THE "COMPANY"), AS AMENDED.

THIS SECURITY IS SUBJECT TO THE RESTRICTIONS (INCLUDING THE VOTING AND TRANSFER RESTRICTIONS) SET FORTH IN ARTICLES FOURTH AND ELEVENTH OF USEC INC.'S CERTIFICATE OF INCORPORATION, AS AMENDED.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON TRANSFER, CONVERSION AND REDEMPTION) STATED IN, AND ARE TRANSFERABLE ONLY IN ACCORDANCE WITH, THE PROVISIONS OF SECTION 9 OF THE SECURITIES PURCHASE AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA CORPORATION ("TOSHIBA") AND BABCOCK & WILCOX INVESTMENT COMPANY ("B&W"), DATED AS OF MAY 25, 2010.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON THE DISPOSITION OF SECURITIES) STATED IN THE PROVISIONS OF SECTION 4.7 OF THE INVESTOR RIGHTS AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA AND B&W, DATED AS OF , 2010."

Section 15. Written Consent. Any action as to which a class vote of the holders of Preferred Stock, or the holders of Preferred Stock and Class B Common Stock voting together, is required pursuant to the terms of this Certificate of Designation or the Securities Purchase Agreement may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation.

Section 16. Notices. All notices or communications in respect of Series B-2 11.5% Preferred Stock shall be sufficiently given if given in writing and delivered in person or by first class mail, postage prepaid, or if given in such other

manner as may be permitted in this Certificate of Designation, in the Certificate of Incorporation or Bylaws or by applicable law or regulation. Notwithstanding the foregoing, if Series B-2 11.5% Preferred Stock is issued in book-entry form through The Depository Trust Corporation or any similar facility, such notices may be given to the holders of Series B-2 11.5% Preferred Stock in any manner permitted by such facility.

Section 17. Other Rights. The shares of Series B-2 11.5% Preferred Stock shall not have any rights, preferences, privileges or voting powers or relative, participating, optional or other special rights, or qualifications, limitations or restrictions thereof, other than as set forth herein or in the Certificate of Incorporation or as provided by applicable law and regulation.

[THIS SPACE LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Designation to be duly executed and acknowledged by its undersigned duly authorized officers this [—] day of [—], 20[—].

USEC INC.

By:

Name: James R. Mellor Title: Chairman of the Board

Attest:

By:

Name: Peter B. Saba Title: Secretary

SIGNATURE PAGE TO THE CERTIFICATE OF DESIGNATION OF SERIES B-2 CONVERTIBLE PREFERRED STOCK OF USEC INC.

EXHIBIT D

FORM OF SERIES C CERTIFICATE OF DESIGNATION

FORM OF

CERTIFICATE OF DESIGNATION OF

SERIES C CONVERTIBLE PARTICIPATING PREFERRED STOCK

of

USEC INC.

Pursuant to Section 151 of the General Corporation Law

of the State of Delaware

We, James R. Mellor, Chairman of the Board, and Peter B. Saba, Secretary, of USEC Inc., a corporation organized and existing under the General Corporation Law ("DGCL") of the State of Delaware (the "Corporation"), in accordance with the provisions of Section 103 thereof, DO HEREBY CERTIFY:

That pursuant to the authority conferred upon the Board of Directors by the Certificate of Incorporation, the Board of Directors on May 24, 2010 adopted the following resolution creating a series of 25,000 shares of Preferred Stock, par value \$1.00 per share, designated as Series C Convertible Participating Preferred Stock:

RESOLVED, that pursuant to the authority vested in the Board of Directors in accordance with the provisions of its Certificate of Incorporation, a series of Preferred Stock of the Corporation be and it hereby is created, and that the designation and amount thereof and the powers, preferences and relative, participating, optional and other rights of the shares of such series, and the qualifications, limitations and restrictions thereof are as follows:

Section 1. Designation. The designation of this series of Preferred Stock, par value \$1.00 per share, of the Corporation is "Series C Convertible Participating Preferred Stock" ("Series C Preferred Stock"). Each share of Series C Preferred Stock shall be identical in all respects to every other share of Series C Preferred Stock.

Section 2. Number of Shares. The authorized number of shares of Series C Preferred Stock is 25,000. Shares of Series C Preferred Stock that are redeemed, purchased or otherwise acquired by the Corporation, or converted into another series of Preferred Stock or into Common Stock, shall revert to authorized but unissued shares of Preferred Stock and shall not be reissued as shares of Series C Preferred Stock.

Section 3. Definitions. As used herein with respect to Series C Preferred Stock:

- (a) "Affiliate" shall mean any Person controlling, controlled by or under common control with any other Person. For purposes of this definition, "control" (including "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of securities, partnership or other ownership interests, by contract or otherwise.
- (b) "Aggregate Outstanding Value" shall mean, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock, plus, (2) for each share of Series C Preferred Stock then held by the Permitted Holders, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price (as defined in the Series B-1 Certificate of Designation) upon which the Permitted Holders' acquisition of such share was calculated, plus, (3) for each share of Common Stock then held by the Permitted Holders, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Common Stock purchased in the market, the Base Price upon which the Permitted Holders' acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock which have been added to the Liquidation Preference pursuant to Section 5(a).
- (c) "Approved Market" shall mean, if the Ordinary Common Stock is then listed or quoted on the New York Stock Exchange, The NASDAQ Stock Market or the American Stock Exchange, such market or exchange.
 - $\textbf{(d)} \ ``\underline{Automatic\ Redemption}" \ shall\ mean\ an\ automatic\ redemption\ pursuant\ to\ Section\ 7(g)\ of\ the\ Series\ B-1$

Certificate of Designation subsequent to a Conversion Election (as defined in the Series B-1 Certificate of Designation), Section 8(c) of the Series B-1 Certificate of Designation or Section 8(c) of the Series B-2 Certificate of Designation.

- (e) "Automatic Redemption Adjustment" shall mean, for purposes of determining the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value, that if an Automatic Redemption has been effected prior to the date of determining such values, (1) the aggregate amount of the Liquidation Preference, as of the date of redemption, of a Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Aggregate Outstanding Value and Permitted Holder Outstanding Value and (2) the aggregate amount of the Liquidation Preference, as of the date of redemption, of such Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Original Issue Value; provided, however, that, if at any time after any Automatic Redemption, such Permitted Holder's Deemed Holder Percentage is less than 8%, then such adjustment to the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value shall not be made.
 - (f) " $\underline{B\&W}$ " shall mean Babcock & Wilcox Investment Company, a Delaware corporation.
- (g) "Beneficially Own" shall mean "beneficially own" as defined in Rule 13d-3 promulgated under Section 13(d) of the Exchange Act or any successor provisions thereto, and "Beneficial Ownership" shall have a correlative meaning.
 - (h) "Board of Directors" shall mean the board of directors of the Corporation or any duly authorized committee thereof.
 - (i) "Bylaws" shall mean the Amended and Restated Bylaws of the Corporation, as amended from time to time.
- (j) "Certificate of Designation" shall mean this Certificate of Designation of Series C Convertible Participating Preferred Stock of the Corporation, as amended from time to time.
 - (k) "Certificate of Incorporation" shall mean the Certificate of Incorporation of the Corporation, as amended from time to time.
 - (I) "Change of Control" shall mean the occurrence of any of the following:
- (1) any Person shall Beneficially Own, directly or indirectly, through a merger, business combination, purchase, or other transaction or series of transactions, shares of the Corporation's capital stock entitling such Person at such time to exercise 50% or more of the total voting power of the issued and outstanding capital stock having ordinary voting power to elect a majority of the Board of Directors, other than as a result of an acquisition of such stock by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this subsection (1), "Person" shall include any syndicate or group that would be deemed to be a "person" under Section 13(d)(3) of the Exchange Act);
- (2) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Stock is exchanged for or converted into cash, securities or other property, in each case other than a merger or consolidation:
 - (i) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Stock;
 - (ii) that is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding shares of the Common Stock solely into shares of any class or series of common stock of the surviving entity; or
 - (iii) where the issued and outstanding capital stock having ordinary voting power to vote generally to elect a majority of the Board of Directors outstanding immediately prior to such transaction is converted into or exchanged for such voting stock of the surviving or transferee Person constituting a majority of the outstanding shares of such voting stock of such surviving or transferee Person (immediately after giving effect to such issuance).

- (m) "Charter Amendment Approval" shall mean the approval of the stockholders of the Corporation necessary to amend the Corporation's Certificate of Incorporation to approve the authorization of Class B Common Stock and the proper filing of such amendment with the Secretary of State of the State of Delaware.
- (n) "Class B Common Stock" shall mean the Class B Common Stock of the Corporation, par value \$.10 per share, to be authorized by the Charter Amendment Approval.
- (o) "Closing Deadline Failure" shall mean, unless waived in writing (1) by the Corporation if such Closing Deadline Failure is as a result of breach by a Permitted Holder, (2) by the Permitted Holders if such Closing Deadline Failure is as a result of breach by the Corporation, or (3) by the Permitted Holders and the Corporation if such Closing Deadline Failure is not as a result of a breach by the Permitted Holders or the Corporation, either, (A) with respect to the Second Closing (as defined in the Securities Purchase Agreement), that the Second Closing shall not have occurred by June 30, 2011 and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.2 thereof, or (B) with respect to the Third Closing (as defined in the Securities Purchase Agreement), that the Third Closing shall not have occurred by the Third Closing Termination Date (as defined in the Securities Purchase Agreement) and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.3 thereof.
- (p) "Code" shall mean the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder by the Internal Revenue Service.
 - (q) "Common Stock" shall mean collectively, the Ordinary Common Stock and the Class B Common Stock.
 - (r) "Converted Series C Preferred Stock" shall have the meaning ascribed to it in Section 10(b).
 - (s) "Corporation" shall have the meaning ascribed to it in the recitals.
- (t) "Deemed Holder Percentage" shall mean, as to any Permitted Holder, the percentage resulting from the following calculation, (1)(A) the number of shares of Ordinary Common Stock equal to the quotient of (w) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on the outstanding shares of Series B Preferred Stock held by such Permitted Holder from and including the immediately preceding Dividend Payment Date (as defined in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable) to, but excluding, the date of conversion and (x) the Base Price for the date of such calculation, plus (B) the number of outstanding of shares of (y) Series C Preferred Stock multiplied by 1000plus, (z) if then outstanding, Class B Common Stock, in each case held by such Permitted Holder divided by (2)(A) the total number of shares of Ordinary Common Stock equal to the quotient of (v) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on all outstanding shares of Series B Preferred Stock from and including the immediately preceding Dividend Payment Date (as defined in the Series B-1 Certificate of Designation) to, but excluding, the date of conversion and (w) the Base Price for the date of such calculation, plus (B) the total number of all outstanding shares of (x) Series C Preferred Stock multiplied by 1000 plus (y) if then outstanding, Class B Common Stock, plus (z) Ordinary Common Stock.
 - (u) "DGCL" shall have the meaning ascribed to it in the recitals.
 - (v) "Dividend" shall have the meaning ascribed to it in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable.
 - (w) "DOE" shall mean the United States Department of Energy.
 - (x) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (y) "Governmental Authority" shall mean any foreign governmental authority, the United States of America, any state of the United States and any political subdivision of any of the foregoing, and any agency, instrumentality, department, commission, board, bureau, central bank, authority, court, arbitral body or other tribunal, in each case whether executive, legislative, judicial, regulatory or administrative, having jurisdiction over any of the Permitted Holders, the Corporation, any of the Corporation's Subsidiaries or their respective Property.
 - (z) "Initial Investor Director" shall have the meaning ascribed to it in Section 5(b)(1) hereof.
 - (aa) "Initial Liquidation Preference" shall mean \$1,000 per share of each of Series B-1 12.75% Preferred Stock and Series B-2 11.5% Preferred Stock.

- (bb) "Investor Director" shall have the meaning ascribed to it in Section 5(b)(2).
- (cc) "Investor Rights Agreement" shall mean that certain Investor Rights Agreement, dated [], 2010 among the Corporation, Toshiba and B&W, as amended from time to time
- (dd) "Junior Stock" shall mean the Common Stock and any other class or series of capital stock of the Corporation that ranks junior to the Series C Preferred Stock (1) as to the priority of payment of dividends and/or (2) as to the distribution of assets on any liquidation, dissolution or winding up of the Corporation. For the avoidance of doubt, Junior Stock shall include the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series A Junior Participating Preferred Stock."
 - (ee) "Liquidation Preference" shall mean \$.01 per share of Series C Preferred Stock.
- (ff) "Ordinary Common Stock" shall mean the common stock of the Corporation, par value \$.10 per share. For the avoidance of doubt, the Ordinary Common Stock shall not include the Class B Common Stock.
- (gg) "Original Issuance Date" shall mean, with respect to each share of Series C Preferred Stock issued to the Permitted Holders, the date on which such share was issued by the Corporation.
- (hh) "Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, the aggregate Initial Liquidation Preference of all the shares of Series B Preferred Stock issued to the Permitted Holders excluding those shares issued as a Dividend.
- (ii) "Parity Stock" shall mean any class or series of stock of the Corporation that ranks equally with Series C Preferred Stock (1) in the priority of payment of dividends and/or (2) in the distribution of assets upon any liquidation, dissolution or winding up of the Corporation (in each case, without regard to whether dividends accrue cumulatively or non-cumulatively).
 - (jj) "Permit" shall mean any approval, authorization, certificate, consent, license or permit of or from any Governmental Authority.
- (kk) "Permitted Holder Material Breach" shall mean the material breach of the Securities Purchase Agreement or the Investor Rights Agreement by any Permitted Holder.
- (II) "<u>Permitted Holder Original Issue Value</u>" shall mean, subject to the Automatic Redemption Adjustment, if any, for any Permitted Holder, the aggregate Initial Liquidation Preference of all shares of Series B Preferred Stock issued to such Permitted Holder excluding those shares issued as a Dividend and those shares acquired upon exercise of the Warrants.
- (mm) "Permitted Holder Outstanding Value" shall mean, as to any Permitted Holder, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock then held by such Permitted Holder, plus, (2) for each share of Series C Preferred Stock then held by a Permitted Holder, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which such Permitted Holder's acquisition of such share was calculated, plus, (3) for each share of Common Stock then held by such Permitted Holder, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Ordinary Common Stock purchased in the market, the Base Price (as defined in the Series B-1 Certificate of Designation) upon which such Permitted Holder's acquisition of such share was calculated, plus, (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock which have been added to the Liquidation Preference pursuant to Section 5(a) of the Series B-1 Certificate of Designation or Series B-2 Certificate of Designation, as applicable.
- (nn) "Permitted Holders" shall mean (1) [Toshiba U.S. Co.] and its Wholly-Owned Affiliates, (2) B&W and its Wholly-Owned Affiliates, (3) a special purpose entity jointly and wholly controlled by Toshiba and B&W and (4) Westinghouse Electric Company, LLC, to the extent that it is controlled by Toshiba or a Permitted Holder described under (1) above; provided, however, that each Permitted Holder must be a U.S. Person.
- (00) "Person" shall mean any individual, corporation, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization, Governmental Authority or any other entity.
 - (pp) "Preferred Stock" shall mean any and all series of preferred stock, par value \$1.00 per share, of the Corporation, including the Series C Preferred Stock.

- (qq) "Property" shall mean any interest in any kind of property or asset, whether real, personal or mixed, tangible or intangible.
- (rr) "Qualified Director" shall mean any individual reasonably acceptable to the Nominating and Governance Committee of the Board of Directors.
- (ss) "Regulatory Bodies" shall mean the DOE and the U.S. Nuclear Regulatory Commission and any successor Governmental Authorities thereto.
- (tt) "Securities Purchase Agreement" shall mean that certain Securities Purchase Agreement, dated May 25, 2010, among the Corporation, Toshiba and B&W, as amended from time to time.
- (uu) "Senior Stock" shall mean any class or series of capital stock of the Corporation that ranks senior to the Series C Preferred Stock (1) as to the priority of dividends and/or (2) as to the distribution of assets on any liquidation, dissolution or winding up of the Corporation. For the avoidance of doubt, Senior Stock shall include the Series B Preferred Stock.
 - (vv) "Series B Preferred Stock" shall mean the Series B-1 12.75% Preferred Stock together with the Series B-2 11.5% Preferred Stock.
- (ww) "Series B-1 12.75 % Preferred Stock" shall mean the series of Preferred Stock of the Corporation, par value \$1.00 per share, designated as "Series B-1 12.75% Convertible Preferred Stock."
- (xx) "Series B-1 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Corporation as filed with the Secretary of State of the State of Delaware.
- (yy) "Series B-2 11.5% Preferred Stock" shall mean the series of preferred stock of the Corporation, par value \$1.00 per share, designated as "Series B-2 11.5% Convertible Preferred Stock."
- (zz) "Series B-2 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-2 11.5% Convertible Preferred Stock of the Corporation as filed with the Secretary of State of the State of Delaware.
 - (aaa) "Series C Preferred Stock" shall have the meaning ascribed to it in Section 1.
 - (bbb) "Series C Preferred Stock Automatic Conversion Time" shall have the meaning ascribed to it in Section 10(b).
- (ccc) "Share Issuance Approval" shall mean the approval of the stockholders of the Corporation necessary to approve the conversion of all the Series B Preferred Stock and the Series C Preferred Stock, and the exercise of all the Warrants, for Common Stock for purposes of Section 312.03 of the New York Stock Exchange Listed Company Manual, or if shares of the Ordinary Common Stock become listed and traded on another Approved Market, the approval required by such Approved Market, or the time at which all such approvals shall for any reason become inapplicable or not required so as to permit all such conversions and exercises.
- (ddd) "Subsidiary" of any Person shall mean any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than 50% of (1) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (2) the interest in the capital or profits of such partnership, joint venture or limited liability company or (3) the beneficial interest in such trust or estate is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries. Notwithstanding the foregoing, American Centrifuge Manufacturing, LLC, a Delaware limited liability company, shall not be considered a Subsidiary of B&W or the Corporation.
- (eee) "Third Party Transfer" shall mean an irrevocable Transfer in compliance with Section 11 of all legal ownership, Voting Control and Beneficial Ownership of any share or shares of Series C Preferred Stock to a Person other than a Permitted Holder or its Affiliates.
 - (fff) "Toshiba" shall mean Toshiba Corporation, a corporation organized under the laws of Japan.
 - (ggg) "Transfer" shall mean, with respect to any shares of Series C Preferred Stock, any direct or indirect

assignment, sale, exchange, transfer, tender or other disposition of such shares or any interest therein, whether voluntary or involuntary, by operation of law or otherwise (and includes any sale or other disposition in any one transaction or series of transactions and the grant or transfer of an option or derivative security covering such shares), and any agreement, arrangement or understanding, whether or not in writing, to effect any of the foregoing; provided, however, that a "Transfer" shall not occur simply as a result of the grant of a proxy in connection with a solicitation of proxies subject to the provisions of Section 14 of the Exchange Act.

- (hhh) "<u>U.S. Person</u>" shall mean any person that is treated as a "United States Person" under Code Section 7701(a)(30) and that provides an IRS Form W-9 (or successor form), evidencing a complete exemption from United States withholding tax (including backup withholding tax), on or before the time at which it acquires securities pursuant to this Certificate of Designation.
- (iii) "Voting Control" shall mean, with respect to a share or shares of Series C Preferred Stock, the power, whether exclusive or shared, revocable or irrevocable, to vote or direct the voting of such share or shares of Series C Preferred Stock, by proxy, voting agreement or otherwise.
- (jjj) "Warrants" shall mean those warrants to purchase Class B Common Stock or Series C Preferred Stock originally issued by the Corporation to the Permitted Holders pursuant to the Securities Purchase Agreement.
- (kkk) "Wholly-Owned Affiliate" shall mean, as to any Person, any Affiliate that, directly or indirectly, is wholly-owned and controlled (other than by contract) by a Person, or any other Affiliate to which the Corporation, in its sole discretion, consents.
- Section 4. Titles and Subtitles; Interpretation. The titles and subtitles used in this Certificate of Designation are used for convenience only and are not to be considered in construing or interpreting this Certificate of Designation. The definitions contained in this Certificate of Designation are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term.

Section 5. Voting Rights.

(a) General. The holders of shares of Series C Preferred Stock shall not be entitled to vote, except as otherwise provided herein or required by applicable law on any matter as to which the shares of Series C Preferred Stock shall be entitled to vote, each share shall entitle the holder thereof to 1,000 votes. In the event the Corporation shall at any time declare or pay any dividend on the Common Stock payable in shares of Common Stock, or effect a subdivision, combination or consolidation of the outstanding shares of Common Stock (by reclassification or otherwise than by payment of a dividend in shares of Common Stock) into a greater or lesser number of shares of Common Stock, then in each such case the number of votes per share to which holders of shares of Series C Preferred Stock were entitled immediately prior to such event shall be adjusted by multiplying such number by a fraction, the numerator of which is the number of shares of Common Stock outstanding immediately after such event and the denominator of which is the number of shares of Common Stock that were outstanding immediately prior to such event.

(b) Election of Directors.

- (1) Effective at the time no Series B Preferred Stock shall be outstanding, the number of directors constituting the Board of Directors shall be increased by two Persons and the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of any other Common Stock and any other series of Preferred Stock, shall be entitled to elect two Qualified Directors to the Board of Directors (each such director, an "Initial Investor Director") until the earliest to occur of (i) a Closing Deadline Failure as a result of a Permitted Holder Material Breach at a time when the Securities Purchase Agreement is terminable pursuant to Sections 10.2(d) and 10.3(d) thereof, (ii) a Change of Control or (iii) such time as the Permitted Holders' Aggregate Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of the Original Issue Value or, (y) after December 31, 2016, 50% of the Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the voting power of the outstanding Series B Preferred Stock to elect the Initial Investor Directors shall cease, (B) the term of office of the Initial Investor Directors shall be immediately and automatically reduced by two Persons.
- (2) Effective as of the first Original Issuance Date and at such time as when the Permitted Holders do not have the right to elect the Initial Investor Directors pursuant to Section 5(b)(1)(iii) and any Permitted Holder's Permitted Holder Outstanding Value is greater than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original

Issue Value, the number of directors constituting the Board of Directors shall be increased by one Person and the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of Ordinary Common Stock and any other series of Preferred Stock, shall be entitled to elect one Qualified Director to the Board of Directors (such director, the "Investor Director") until the earliest to occur of (i) an event described in Section 5(b)(1)(i) or (ii) or (ii) such time as each Permitted Holder's Permitted Holder Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock to elect the Investor Director shall cease, (B) the term of office of the Investor Director shall immediately and automatically terminate, (C) the Investor Director will no longer be qualified to serve and (D) the number of directors constituting the Board of Directors shall be immediately and automatically reduced by one Person.

- (3) For the avoidance of doubt, except for the increase or decrease in the number of directors provided for herein, nothing in this Section 5(b) shall prohibit the Board of Directors from fixing the number of directors constituting the Board of Directors pursuant to the Bylaws.
- (4) Term. Subject to the provisions of this Section 5(b), each Initial Investor Director or the Investor Director, as applicable, shall serve until the next annual meeting of the stockholders of the Corporation and until his or her successor is elected and qualified in accordance with this Section 5(b) and the Bylaws, unless any such Initial Investor Director or the Investor Director, as applicable, is earlier removed in accordance with the Bylaws, resigns or is otherwise unable to serve; provided, however, that only the holders of a majority of the voting power of the outstanding Class B Common Stock and the Series C Preferred Stock may remove any such Initial Investor Director or the Investor Director, as applicable, without cause at any time, and the holders of a majority of the voting power of the outstanding shares of the capital stock of the Corporation entitled to vote on the matter may remove any such Initial Investor Director or the Investor Director, as applicable, with cause at any time. Subject to the provisions of this Section 5(b), in the event any Initial Investor Director or the Investor Director, as applicable, is removed, resigns or is unable to serve as a member of the Board of Directors, the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of any other Common Stock and any other series of Preferred Stock, shall have the right to fill such vacancy. Each Initial Investor Director or the Investor Director is seat, as applicable, shall otherwise remain vacant.
- (d) Notwithstanding Section 5(a), the holders of Series C Preferred Stock and Class B Common Stock shall be entitled to vote together with the holders of Common Stock (and any other class or series of capital stock entitled to vote on the matter with the Common Stock) as a single class with respect to any transactions involving a merger of the Corporation or sale of substantially all of the Corporation's assets, which must be submitted to the Corporation's stockholders pursuant to the DGCL; provided, however, that each holder of Class B Common Stock shall be entitled to (A) one vote for each outstanding share of Class B Common Stock held of record by such holder as of the applicable record date, but only to the extent that the aggregate voting power of all of the outstanding Series C Preferred Stock and Class B Common Stock does not exceed 20% of the total voting power of all outstanding shares of all classes and series of capital stock entitled to vote thereon or (B) if pursuant to clause (A) the aggregate voting power of all of the outstanding Series C Preferred Stock and Class B Common Stock would exceed 20% of the total voting power of all outstanding shares of apital stock entitled to vote on the matter, such fraction of one vote for (i) each one-one thousandth (1/1000) of a share of Series C Preferred Stock and (ii) each share of Class B Common Stock held of record by such holder as of the applicable record date such that the aggregate voting power of all of the outstanding Series C Preferred Stock and Class B Common Stock equaled 20% of the total voting power of all outstanding shares of all classes and series of capital stock entitled to vote thereon.
- (e) Notwithstanding Section 5(a), the vote or consent of the holders of at least a majority of the outstanding shares of Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of the Common Stock and any other series of Preferred Stock then outstanding and entitled to vote thereon, given in person or by proxy, either in writing without a meeting or by vote at any meeting called for the purpose, shall be necessary for effecting or validating any amendment, alteration or repeal of the Certificate of Incorporation or this Certificate of Designation (by merger, consolidation or otherwise) so as to adversely affect the powers, preferences and relative participating, optional and other rights of Series C Preferred Stock.

Section 6. Dividends and Distributions. Subject to applicable law and the rights, if any, of the holders of any Senior Stock, the holders of Series C Preferred Stock then outstanding shall first receive, or simultaneously receive, a dividend on each outstanding share of Series C Preferred Stock in an amount equal to the product of (a) the aggregate per

share amount of all dividends declared and paid on the Ordinary Common Stock out of the assets of the Corporation that are by law available therefor at such times and in such amounts as the Board of Directors in its discretion shall determine and (b) 1000. Except as otherwise required by the DGCL and Section 7(a), in any circumstance where the Corporation may declare dividends or otherwise make distributions (including, without limitation, any distribution on liquidation, dissolution or winding-up of the Corporation) on the Common Stock, the Corporation shall declare the same per share dividends or make the same per share distributions, as the case may be, on the Series C Preferred Stock; provided, however, that if any such dividends or distributions are declared with respect to the Common Stock in the form of additional shares of Common Stock (or rights to acquire Common Stock), such dividends or distributions shall be made with respect to the Series C Preferred Stock in the form of an equivalent number of shares of Series C Preferred Stock in the form of additional shares of Series C Preferred Stock (or rights to acquire Series C Preferred Stock), such dividends or distributions shall be made with respect to the Common Stock in the form of an equivalent number of shares of Common Stock (or rights to acquire Ordinary Common Stock).

Section 7. Liquidation Rights.

- (a) Voluntary or Involuntary Liquidation Subject to the rights of the holders of any Senior Stock outstanding at any time, in the event of any liquidation, dissolution or winding up of the affairs of the Corporation, whether voluntary or involuntary, holders of the Series C Preferred Stock shall be entitled to receive for each outstanding share of Series C Preferred Stock, out of the assets of the Corporation or proceeds thereof available for distribution to stockholders of the Corporation, and after satisfaction of all liabilities and obligations to creditors of the Corporation, before any distribution of such assets or proceeds is made to or set aside for the holders of Junior Stock, an amount equal to the per share amount of all cash and other property to be distributed in respect of the Common Stock into which the Series C Preferred Stock is then convertible.
- (b) Partial Payment. If, in connection with any distribution described in Section 7(a) above, the assets of the Corporation or proceeds thereof are not sufficient to pay the Liquidation Preferences, plus an amount equal to any dividends declared but unpaid thereon, in full to all holders of Series C Preferred Stock and all holders of Parity Stock, then the amounts paid to the holders of Series C Preferred Stock and to the holders of all such other capital stock ranking equally on liquidation shall be paid pro rata in accordance with the respective aggregate Liquidation Preferences, plus any dividends declared but unpaid thereon, of the holders of Series C Preferred Stock and the holders of all such other Parity Stock.
- (c) Merger, Consolidation and Sale of Assets Not Liquidation For purposes of this Section 7, the merger or consolidation of the Corporation with any other corporation or other Person, including a merger or consolidation in which the holders of Series C Preferred Stock receive cash, securities or other property for their shares, or the sale, lease or exchange (for cash, securities or other property) of all or substantially all of the assets of the Corporation, shall not constitute a liquidation, dissolution or winding up of the Corporation, but shall instead be subject to the provisions of Section 9.
- **Section 8.** Subdivision or Combination. If the Corporation in any manner subdivides or combines the outstanding shares of any of the Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock, then the outstanding shares of the Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock, as applicable, will be subdivided or combined in the same manner.
- Section 9. Equal Status. Except as expressly provided in this Certificate of Designation, shares of Ordinary Common Stock and Series C Preferred Stock shall have the same rights, powers, preferences and restrictions and rank equally, share ratably and be identical in all respect as to all matters. In any merger, consolidation, reorganization or other business combination, the consideration received per share by the holders of the Ordinary Common Stock and per 1/1000 of a share of Series C Preferred Stock in such merger, consolidation, reorganization or other business combination shall be identical; provided, however, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Corporation or any other corporation, partnership, limited liability company or other entity, then the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ to the extent that the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of the Ordinary Common Stock and the Series C Preferred Stock differ as provided herein or in the Certificate of Incorporation (including, without limitation, with respect to the voting rights and conversion provisions hereof) if and to the extent necessary due to regulatory requirements or restrictions applicable to the entity surviving such merger, consolidation, reorganization or other business combination that are similar in nature to those applicable to the Corporation; and provided, further, that if the holders of the Ordinary Common Stock or Series C Preferred Stock are granted the right to elect to receive one of two or more alternative forms of consideration, the foregoing provision shall be deemed satisfied if holders of the other class or series are granted identical election rights, s

Section 10. Automatic Conversion.

- (a) Subject to Section 11, a share of the Series C Preferred Stock shall be automatically converted, without any action on the part of the Corporation (other than the subsequent exchange of Series C Preferred Stock certificates for Ordinary Common Stock certificates or, in the case of uncertificated shares of Series C Preferred Stock, upon receipt of proper transfer instructions from the registered holder of the shares of Series C Preferred Stock or by his, her or its attorney lawfully constituted in writing, and upon payment of all necessary transfer taxes and compliance with appropriate procedures for transferring shares in uncertificated form), or any holder of the Series C Preferred Stock or any other Person, into 1000 fully paid and nonassessable shares of Ordinary Common Stock upon a Third Party Transfer of such share.
- (b) In the event of any automatic conversion pursuant to the terms of Section 10(a), the conversion shall be deemed to have been effected upon such Third-Party Transfer (the "Series C Preferred Stock Automatic Conversion Time"). At the Series C Preferred Stock Automatic Conversion Time, the certificates that represented the shares of Series C Preferred Stock that were so converted immediately prior to such conversion (the "Converted Series C Preferred Stock") shall, automatically and without further action, represent 1000 fully paid and non-assessable shares of Ordinary Common Stock per share of Series C Preferred Stock. Permitted Holders of the Converted Series C Preferred Stock shall deliver their certificates, duly endorsed in blank or accompanied by proper instruments of transfer, in form reasonably satisfactory to the Corporation, duly executed by such Permitted Holder or such Permitted Holder's authorized attorney to the principal office of the Corporation (or such other office or agency (including the transfer agent, if applicable) of the Corporation as it may designate by notice in writing to the registered Permitted Holder at the address of such Permitted Holder appearing on the books of the Corporation), together with a written notice stating the name or names (with addresses) and denominations in which the certificate or certificates representing such shares of Ordinary Common Stock are to be issued and including instructions for delivery thereof. Upon such delivery, the Corporation or its agent shall promptly issue and deliver at such stated address to such holder of shares of Ordinary Common Stock a certificate or certificates representing the number of shares of Ordinary Common Stock to which such holder is entitled by reason of such conversion, and shall cause such shares of Ordinary Common Stock to be registered in the name of such holder. The Person entitled to receive the shares of Ordinary Common Stock issuable upon such conversion shall be treated for all purposes as the record holder of such shares of Ordinary Common Stock at and as of the Series C Preferred Stock Automatic Conversion Time, and the rights of such Person as a holder of shares of the Series C Preferred Stock that have been converted shall cease and terminate at and as of the Series C Preferred Stock Automatic Conversion Time, in each case without regard to any failure by such Permitted Holder to deliver the certificates or the notice required by this Section.

Section 11. Restrictions.

- (a) Notwithstanding anything in this Certificate of Designation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, Permitted Holders may not Transfer shares of Series C Preferred Stock if such Transfer would require approvals from or filings with any Regulatory Bodies in order not to adversely affect the Permits or regulatory status of the Corporation or its Subsidiaries, unless such approvals and/or filings have been made and received; provided, however, this Section 11(a) shall not apply to any transfer where the transferee received Ordinary Common Stock pursuant to the terms hereof.
- (b) Notwithstanding anything in this Certificate of Designation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, the conversion of Series C Preferred Stock for Ordinary Common Stock shall also be subject to the requirements of Section 9.2 of the Securities Purchase Agreement.
 - (c) Any purported conversion or Transfer of Series C Preferred Stock in violation of these restrictions shall be null and voidab initio.
- Section 12. Record Holders. To the fullest extent permitted by applicable law, the Corporation may deem and treat the record holder of any share of the Series C Preferred Stock as the true and lawful owner thereof for all purposes.
- Section 13. Legends. All certificates representing shares of Series C Preferred Stock shall bear a legend or other restriction substantially to the following effect (it being agreed that if such shares are not certificated, other appropriate restrictions shall be implemented to give effect to the following):

"THIS SECURITY WAS ORIGINALLY ISSUED IN A TRANSACTION EXEMPT FROM REGISTRATION UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR AS MAY BE HELD BY A PERSON DEEMED AN "AFFILIATE" (AS DEFINED IN RULE 144 UNDER THE SECURITIES ACT) OF THE ISSUER OF THIS SECURITY, AND THIS SECURITY MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED IN THE

ABSENCE OF SUCH REGISTRATION OR AN APPLICABLE EXEMPTION THEREFROM. THE HOLDER OF THIS SECURITY AGREES FOR THE BENEFIT OF THE COMPANY THAT (A) THIS SECURITY MAY BE OFFERED, RESOLD, PLEDGED OR OTHERWISE TRANSFERRED, ONLY (I) IN A TRANSACTION NOT INVOLVING A PUBLIC OFFERING, (II) PURSUANT TO ANY OTHER EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, INCLUDING RULE 144 UNDER THE SECURITIES ACT (IF AVAILABLE), (III) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR (IV) TO THE COMPANY OR ANY OF ITS SUBSIDIARIES, IN EACH OF CASES (I) THROUGH (IV) IN ACCORDANCE WITH ANY APPLICABLE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES, AND (B) THE HOLDER WILL NOTIFY ANY SUBSEQUENT PURCHASER OF THIS SECURITY FROM IT OF THE RESALE RESTRICTIONS REFERRED TO IN (A) ABOVE. THIS SECURITY MAY ONLY BE TRANSFERRED IN ACCORDANCE WITH THE TERMS OF THE CERTIFICATE OF DESIGNATION OF SERIES C CONVERTIBLE PARTICIPATING PREFERRED STOCK OF USEC INC. (THE "COMPANY"), AS AMENDED.

THIS SECURITY IS SUBJECT TO THE RESTRICTIONS (INCLUDING THE VOTING AND TRANSFER RESTRICTIONS) SET FORTH IN ARTICLES FOURTH AND ELEVENTH OF USEC INC.'S CERTIFICATE OF INCORPORATION, AS AMENDED.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON TRANSFER, CONVERSION AND REDEMPTION) STATED IN, AND ARE TRANSFERABLE ONLY IN ACCORDANCE WITH, THE PROVISIONS OF SECTION 9 OF THE SECURITIES PURCHASE AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA CORPORATION ("TOSHIBA") AND BABCOCK & WILCOX INVESTMENT COMPANY ("B&W"), DATED AS OF MAY 25, 2010.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON THE DISPOSITION OF SECURITIES) STATED IN THE PROVISIONS OF SECTION 4.7 OF THE INVESTOR RIGHTS AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA AND B&W, DATED AS OF , 2010."

Section 14. Written Consent. Any action as to which a class vote of the holders of Series C Preferred Stock, or the holders of Series C Preferred Stock and Class B Common Stock voting together, is required pursuant to the terms of this Certificate of Designation or the Securities Purchase Agreement may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation.

Section 15. Notices. All notices or communications in respect of Series C Preferred Stock shall be sufficiently given if given in writing and delivered in person or by first class mail, postage prepaid, or if given in such other manner as may be permitted in this Certificate of Designation, in the Certificate of Incorporation or Bylaws or by applicable law or regulation. Notwithstanding the foregoing, if Series C Preferred Stock is issued in book-entry form through The Depository Trust Corporation or any similar facility, such notices may be given to the holders of the Series C Preferred Stock in any manner permitted by such facility.

Section 16. Other Rights. The shares of Series C Preferred Stock shall not have any rights, preferences, privileges or voting powers or relative, participating, optional or other special rights, or qualifications, limitations or restrictions thereof, other than as set forth herein or in the Certificate of Incorporation or as provided by applicable law and regulation.

[THIS SPACE LEFT BLANK INTENTIONALLY]

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Designation to be duly executed and acknowledged by its undersigned duly authorized officers this [—] day of [—], 20[—].

USEC INC.

By:

Name: James R. Mellor Title: Chairman of the Board

Attest:

By:

Name: Peter B. Saba Title: Secretary

SIGNATURE PAGE TO THE CERTIFICATE OF DESIGNATION OF SERIES C PARTICIPATING PREFERRED STOCK OF USEC INC.

EXHIBIT E

FORM OF WARRANT

THIS WARRANT AND THE SECURITIES ISSUED PURSUANT TO THIS WARRANT HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT") AND MAY NOT BE OFFERED, SOLD, ASSIGNED, PLEDGED, TRANSFERRED OR OTHERWISE DISPOSED OF IN THE ABSENCE OF AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR PURSUANT TO AN AVAILABLE EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT, AND UPON DELIVERY OF AN OPINION OF COUNSEL REASONABLY SATISFACTORY TO THE COMPANY THAT THE PROPOSED TRANSFER IS EXEMPT FROM THE SECURITIES ACT OR THAT THE PROSPECTUS DELIVERY REQUIREMENTS HAVE BEEN MET.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON TRANSFER, CONVERSION AND REDEMPTION) STATED IN, AND ARE TRANSFERABLE ONLY IN ACCORDANCE WITH, THE PROVISIONS OF SECTION 9 OF THE SECURITIES PURCHASE AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA CORPORATION AND BABCOCK & WILCOX INVESTMENT COMPANY, DATED MAY 25, 2010.

THIS WARRANT AND THE SECURITIES ISSUED PURSUANT TO THIS WARRANT ARE SUBJECT TO THE RESTRICTIONS (INCLUDING THE VOTING AND TRANSFER RESTRICTIONS) SET FORTH IN ARTICLE ELEVENTH OF USEC INC.'S CERTIFICATE OF INCORPORATION, AS AMENDED.

THIS WARRANT AND THE SECURITIES ISSUED PURSUANT TO THIS WARRANT ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON THE DISPOSITION OF SECURITIES) STATED IN THE PROVISIONS OF SECTION 4.7 OF THE INVESTOR RIGHTS AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA AND B&W, DATED AS OF , 2010.

WARRANT NO. [—]
DATE OF ISSUANCE: [—], 20[—]

WARRANT TO PURCHASE

[X] SHARES OF CLASS B COMMON STOCK

٥r

[X/1000] SHARES OF SERIES C PREFERRED STOCK

OF USEC INC.

THIS STOCK PURCHASE WARRANT (this "Warrant") certifies that, for value received, [INVESTOR] (the "Holder") is entitled, upon the terms and subject to the limitations on exercise and the conditions hereinafter set forth, at any time on or after January 1, 2015 (the "Initial Exercise Date") and on or prior to the close of business, New York City time, on the Termination Date but not thereafter (the "Exercise Period"), to subscribe for and purchase from USEC Inc., a Delaware corporation (the "Company"), up to [X] shares of Class B Common Stock, par value \$0.10 per share of the Company ("Class B Common Stock"), to be authorized by the Company pursuant to the Charter Amendment, or in lieu thereof, up to [X/1000] shares of Series C Preferred Stock (together with the Class B Common Stock, the "Stock") (such shares of Class B Common Stock or Series C Preferred Stock, as applicable, the "Warrant Shares"), as provided in Section 2(c). The purchase price for each share of Class B Common Stock under this Warrant shall be U.S. \$7.50 (the "Common Exercise Price") (such Common Exercise Price or Preferred Exercise Price, as applicable, the "Exercise Price"); the Exercise Price and the number of Warrant Shares for which the Warrant is exercisable shall be subject to adjustment as provided herein. The term "Holder" shall refer to the Holder identified above or any subsequent permitted transferee of this Warrant. Capitalized terms used but not otherwise defined herein shall have the meanings set forth in that certain Securities Purchase Agreement, dated May 25, 2010, among the Company, Toshiba Corporation , a corporation organized under the laws of Japan ("Toshiba"), and Babcock & Wilcox Investment Company, a Delaware corporation ("B&W"), as amended from time to time (the "Securities Purchase Agreement").

1. <u>Authorization of Warrant Shares</u>. The Company shall at all times keep reserved for issuance and delivery upon exercise of this Warrant such number of its authorized and unissued shares of Stock of the Company from time to time

issuable upon exercise of this Warrant as will be sufficient to permit the exercise in full of this Warrant in accordance with the terms set forth herein. The Company represents and warrants that all shares of Stock delivered upon the exercise of this Warrant shall have been duly authorized and validly issued, shall be fully paid and nonassessable, and shall be free from preemptive rights and free of any lien or adverse claim.

2. Exercise of Warrant.

(a) Except as provided in Section 9 of the Securities Purchase Agreement and Sections 2(c) and 2(e) herein, exercise of the purchase rights represented by this Warrant may be made at any time or times on or after the Initial Exercise Date and before or on the Termination Date by (i) surrendering this Warrant, with the Notice of Exercise Form annexed hereto completed and duly executed, to the offices of the Company (or such other office or agency (including the transfer agent, if applicable) of the Company as it may designate by notice in writing to the registered Holder at the address of such Holder appearing on the books of the Company) and (ii) delivering payment of the Exercise Price for the shares of Stock thereby purchased by wire transfer of immediately available funds in accordance with written wire instructions to be provided by the Company promptly on the Holder's request. Subject to the restrictions of Section 9 of the Securities Purchase Agreement, the Holder exercising its purchase rights in accordance with the preceding sentence shall be entitled to receive a certificate (or designate to whom such certificate shall be issued) for the number of Warrant Shares so purchased; certificates for shares so purchased hereunder shall be issued and delivered to the Holder (or issued at its direction) within three Trading Days after the date on which this Warrant shall have been exercised as aforesaid. This Warrant shall be deemed to have been exercised and such certificate or certificates shall be deemed to have been issued, and the Holder shall be deemed to no longer hold this Warrant with respect to such shares, as of the date this Warrant has been exercised by payment to the Company of the Exercise Price (including by exercising the Net Exercise Right pursuant to Section 2(b)) and all taxes required to be paid by the Holder, if any, pursuant to Section 5 prior to the issuance of such shares, have been paid, notwithstanding that the stockholder books or records of the Company may be closed or certificates r

(b) Notwithstanding the foregoing, subject to the restrictions of Section 9 of the Securities Purchase Agreement, the Holder may, it its sole discretion, satisfy its obligation to pay the Exercise Price through a "cashless exercise" (the "Net Exercise Right"), in which case the Company shall issue to the Holder the number of Warrant Shares determined as follows:

X = Y[(A-B)/A]

where:

X = the number of Warrant Shares to be issued to the Holder;

Y = the total number of Warrant Shares with respect to which this Warrant is being exercised;

A = the Warrant Base Price as of the date of exercise; and

B = the Exercise Price then in effect for the applicable Warrant Shares at the time of such exercise.

(c) If, at the time this Warrant is exercised, the Charter Amendment Approval or the Regulatory Approvals have not been obtained, the Holder of this Warrant shall be entitled to exercise this Warrant only for shares of Series C Preferred Stock. If, however, the Charter Amendment Approval and Regulatory Approvals have been obtained, the Holder of this Warrant shall be entitled, subject to the Share Issuance Limitation (if any), to exercise this Warrant only for shares of Class B Common Stock.

(d) Subject to Section 2(e) and Section 2(e) herein and Section 9 of the Securities Purchase Agreement, the Holder may exercise all or any portion of this Warrant during the Exercise Period. In the event that this Warrant is not exercised in full, the number of Warrant Shares shall be reduced by the number of such Warrant Shares for which this Warrant is exercised and/or surrendered, and the Company, if requested by the Holder and at its expense, shall within 10 Business Days issue and deliver to the Holder a new Warrant of like tenor in the name of the Holder or as the Holder may direct, to a Permitted Holder transferee (upon payment by Holder of any applicable transfer taxes pursuant to Section 5), reflecting such adjusted Warrant Shares.

- (e) For the avoidance of doubt, this Warrant shall be exercisable for either Class B Common Stock or Series C Preferred Stock, but not both.
- 3. <u>Definitions</u>. As used herein with respect to this Warrant:
- "Approved Market" shall have the meaning ascribed to it in the definition of "Warrant Base Price."
- "B&W" shall have the meaning ascribed to it in the Preamble.

- "Business Day" shall mean any calendar day other than (1) a Saturday or Sunday or (2) a calendar day on which banking institutions in either the City of New York or Tokyo, Japan are authorized by law, regulation or executive order to remain closed.
- "Charter Amendment Approval" shall mean, collectively, (i) the approval of the stockholders of the Company necessary to adopt that certain amendment to the Company's Certificate of Incorporation, as amended, substantially in the form attached to the Securities Purchase Agreement, which amendment shall approve the authorization of Class B Common Stock, and (ii) the proper filing of such amendment with the Secretary of State of the State of Delaware.
 - "Class B Common Stock" shall have the meaning ascribed to it in the Preamble.
 - "Common Exercise Price" shall have the meaning ascribed to it in the Preamble.
 - "Company" shall have the meaning ascribed to it in the Preamble.
 - "Exchange Property" shall have the meaning ascribed to it in Section 12.
 - "Exercise Period" shall have the meaning ascribed to it in the Preamble.
 - "Exercise Price" shall have the meaning ascribed to it in the Preamble.
 - "Holder" shall have the meaning ascribed to it in the Preamble.
 - "Initial Exercise Date" shall have the meaning ascribed to it in the Preamble.
 - "Net Exercise Right' shall have the meaning ascribed to it in Section 2(b).
 - "Ordinary Common Stock" shall mean the Common Stock of the Company, par value \$0.10 per share.
- "Permitted Holders" shall mean (a) [Toshiba U.S. Subsidiary] and its Wholly-Owned Affiliates, (b) B&W and its Wholly-Owned Affiliates, (c) a special purpose entity jointly and wholly controlled by [Toshiba U.S. Subsidiary] and B&W and (d) Westinghouse Electric Company, LLC, to the extent it is controlled by Toshiba or a Permitted Holder described under (a) above; provided, however, that each Permitted Holder must be a U.S. Person.
 - "Preferred Exercise Price" shall have the meaning ascribed to it in the Preamble.
- "Preferred Stock" shall mean the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, the Company's Series B-2 11.5% Preferred Stock, par value \$1.00 per share, and the Company's Series C Convertible Participating Preferred Stock, par value \$1.00 per share.
- "Pro Rata Repurchases" means any purchase of shares of Ordinary Common Stock by the Company or any Affiliate thereof pursuant to (a) any tender offer or exchange offer subject to Section 13(e) or 14(e) of the Exchange Act or Regulation 14E promulgated thereunder, or (b) any other offer available to substantially all holders of Ordinary Common Stock, in the case of (a) or (b), whether for cash, shares of Ordinary Common Stock of the Company, other securities of the Company or any other Person, assets or any other property (including, without limitation, shares of common stock, other securities or evidences of indebtedness of a Subsidiary), or any combination thereof, effected while this Warrant is outstanding. The "effective date" of a Pro Rata Repurchase shall mean the date of acceptance of shares for purchase or exchange by the Company under any tender or exchange offer which is a Pro Rata Repurchase or the date of purchase with respect to any Pro Rata Repurchase that is not a tender or exchange offer.
 - "Reorganization Event" shall have the meaning ascribed to it in Section 12.
 - "Securities Purchase Agreement" shall have the meaning ascribed to it in the Preamble.
 - "Series C Preferred Stock" shall mean the Company's Series C Convertible Participating Preferred Stock, par value \$1.00 per share.
- "Share Issuance Limitation" shall mean the total number of shares of Common Stock or securities convertible into Common Stock that can be issued by the Corporation upon conversion or exercise of securities issued pursuant to the Transactions (as defined in the Securities Purchase Agreement) in accordance with the rules and regulations of the Approved

Market on which shares of the Corporation's equity securities are listed or traded prior to receipt of the Share Issuance Approval.

- "Stock" shall have the meaning ascribed to it in the Preamble.
- "Termination Date" shall mean December 31, 2016.
- "Toshiba" shall have the meaning ascribed to it in the Preamble.
- "Trading Day" shall mean any day on which shares of the Company's equity securities are traded, or able to be traded, on the Approved Market on which shares of the Company's equity securities are listed or traded.
 - "Warrant" shall have the meaning ascribed to it in the Preamble.
- "Warrant Base Price" shall mean for any date, the price determined by the first of the following clauses that applies: (a) if the Ordinary Common Stock is then listed or quoted on the New York Stock Exchange, The NASDAQ Stock Market or the American Stock Exchange (each an "Approved Market"), the daily volume-weighted average price per share of the Ordinary Common Stock for the Trading Day immediately preceding such date, as reported for the regular trading session (including any extensions thereof) on the primary Approved Market on which the Ordinary Common Stock is then listed or quoted (without regard to pre-open or after-hours trading outside of such regular trading session on such Trading Day), as reported by Bloomberg Financial L.P. (or any successor thereof) using the HP function (or any equivalent thereof); (b) if the Ordinary Common Stock has not been listed or quoted on an Approved Market and if prices for the Ordinary Common Stock are then quoted on the OTC Bulletin Board, the daily volume-weighted average price per share of the Ordinary Common Stock for the Trading Day immediately preceding such date, as quoted for the regular trading session on the OTC Bulletin Board; (c) if the Ordinary Common Stock has not been listed or quoted on the OTC Bulletin Board and if prices for the Ordinary Common Stock are then reported in the "Pink Sheets" published by the National Quotation Bureau Incorporated (or a similar organization or agency succeeding to its functions of reporting prices), the most recent bid price per share of the Ordinary Common Stock so reported; or (d) in all other cases, the fair market value of a share of Ordinary Common Stock as determined by the Company's Board of Directors acting reasonably and in good faith; provided, however, for any calculation pertaining to Series C Preferred Stock, the result of the calculations in any of (a) through (d) shall be multiplied by 1,000 to arrive at the Warrant Base Price.
 - "Warrant Shares" shall have the meaning ascribed to it in the Preamble.
- "Wholly-Owned Affiliate" shall mean, as to any Person, any Affiliate that, directly or indirectly, is wholly-owned and controlled (other than by contract) by a Person, or any other Affiliate to which the Company, in its sole discretion, consents.
- 4. <u>Fractional Shares or Scrip.</u> No fractional shares or scrip representing fractional shares of Class B Common Stock shall be issued upon the exercise of this Warrant. As to any fraction of a share which the Holder would otherwise be entitled to purchase upon such exercise, the Company shall pay a cash adjustment in respect of such final fraction in an amount equal to such fraction multiplied by the Warrant Base Price on the date of exercise. Fractional shares of Series C Preferred Stock shall be issued, if necessary.
- 5. <u>Taxes</u>. The Company shall pay any and all taxes that may be payable in respect of the issue or delivery of shares of Class B Common Stock or Series C Preferred Stock, as applicable, on exercise of this Warrant. The Company shall not, however, be required to pay any tax which may be payable in respect of any transfer involved in the issue and delivery of shares of Class B Common Stock or Series C Preferred Stock, as applicable, in a name other than that in which this Warrant was issued, and no such issue or delivery shall be made unless and until the Person requesting such issue has paid to the Company the amount of any such tax or has established to the satisfaction of the Company that such tax has been paid.
- 6. <u>Closing of Books</u>. The Company shall not close its stockholder books or records in any manner which prevents a timely exercise of this Warrant permitted hereby.
 - 7. Division and Combination.
- (a) This Warrant may be divided or combined with other Warrants upon presentation hereof to the Company, together with a written notice specifying the denominations in which new Warrants are to be issued and signed by the Holder or its agent or attorney. The Company shall execute and deliver a new Warrant or Warrants in exchange for the Warrant or Warrants to be divided or combined in accordance with such notice.

(b) The Company shall prepare, issue and deliver at its own expense (other than transfer taxes as provided in Section 5) the new Warrant or Warrants under this Section 7.

- 8. No Rights as Stockholder until Exercise. This Warrant does not entitle the Holder to any voting rights or other rights as a stockholder of the Company. Upon the surrender of this Warrant and the payment of the aggregate Exercise Price (or satisfaction thereof through the Net Exercise Right), the Warrant Shares so purchased shall be and be deemed to be issued to such Holder or its designated transferee, as required pursuant to Section 9 of the Securities Purchase Agreement, as the record owner of such shares as of the close of business on the later of the date of such surrender or payment/satisfaction, and this Warrant shall no longer be issuable with respect to such Warrant Shares.
- 9. Loss, Theft, Destruction or Mutilation of Warrant. The Company covenants that upon receipt by the Company of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Warrant or any stock certificate relating to the Warrant Shares, and in case of loss, theft or destruction, of indemnity or security reasonably satisfactory to it, and upon surrender and cancellation of such Warrant or stock certificate, if mutilated, the Company shall make and deliver a new Warrant or stock certificate of like tenor, and dated as of such cancellation, in lieu of such Warrant or stock certificate.
- 10. <u>Business Days</u>. If the last or appointed day for the taking of any action or the expiration of any right required or granted herein shall not be a Business Day, then such action may be taken or such right may be exercised on the next succeeding Business Day.
- 11. <u>Adjustments of Exercise Price and Number / Kind of Warrant Shares</u>. The Exercise Price and the number and kind of securities purchasable upon the exercise of this Warrant shall be subject to adjustment from time to time upon the happening of any of the following:
- (a) In case the Company shall (i) pay a dividend in shares of Ordinary Common Stock or make a distribution in shares of Ordinary Common Stock, (ii) subdivide its outstanding shares of Ordinary Common Stock into a greater number of shares, (iii) combine its outstanding shares of Ordinary Common Stock into a smaller number of shares of Ordinary Common Stock, or (iv) issue any shares of its capital stock in a reclassification of the Ordinary Common Stock, then the number of Warrant Shares purchasable upon exercise of this Warrant immediately prior thereto shall be proportionately adjusted so that the Holder shall be entitled to receive the kind and number of Warrant Shares or other shares of capital stock of the Company which it would have owned or have been entitled to receive had such Warrant been exercised in advance thereof; provided, however, that if any such dividend is paid in the form of Ordinary Common Stock or rights to acquire Ordinary Common Stock, at a time this Warrant is exercisable for Series C Preferred Stock, the Holder shall receive at the time of exercise shares of Series C Preferred Stock or rights to acquire shares of Series C Preferred Stock, as the case may be, and at a time this Warrant is exercisable for Class B Common Stock, the Holder shall receive at the time of exercise shares of Class B Common Stock or rights to acquire shares of Class B Common Stock, as the case may be. Upon each such adjustment of the kind and number of Warrant Shares or other securities of the Company which are purchasable hereunder, the Holder shall thereafter be entitled to purchase the number of Warrant Shares or other shares of capital stock resulting from such adjustment at an Exercise Price per Warrant Share or other security obtained by multiplying the Exercise Price in effect immediately prior to such adjustment and the Company purchasable pursuant hereto as a result of such adjustment made pursuant to this Section 11 shall become effective immediately after the effective date of such event retroactive to
- (b) If the Company shall distribute to all holders of Ordinary Common Stock (and not to the Holder of this Warrant) securities, evidences of the indebtedness of the Company or any other Person, assets (including cash and cash dividends), rights, options, warrants or other property (excluding dividends of its Ordinary Common Stock and other distributions referred to in Section 11(a)), in each such case, (i) the Exercise Price shall be adjusted by multiplying the Exercise Price in effect immediately prior to the record date fixed for determination of stockholders entitled to receive such distribution by a fraction, the denominator of which shall be the Warrant Base Price determined as of such record date, and the numerator of which shall be such Warrant Base Price on the date following such record date, and (ii) the number of Warrant Shares issuable upon the exercise of this Warrant shall be increased to the number obtained by multiplying the number of Warrant Shares issuable upon the exercise of the Warrant immediately prior to such adjustment by a fraction, the numerator of which is the Exercise Price in effect immediately prior to such adjustment and the denominator of which is the new Exercise Price determined in accordance with subsection (i) above. Such adjustments shall be made whenever any such distribution is made and shall become effective immediately after the record date mentioned above. Notwithstanding the foregoing, the Exercise Price shall never be adjusted below the par value of the shares of Class B Common Stock or Series C Preferred Stock, as applicable, for which this Warrant is then exercisable.
 - (c) If the Company shall effect a Pro Rata Repurchase of Ordinary Common Stock prior to the Exercise

Period, then (i) the Exercise Price shall be adjusted to the price determined by multiplying the Exercise Price in effect immediately prior to the effective date of such Pro Rata Repurchase by a fraction, the numerator of which shall be the product of the number of shares of Ordinary Common Stock outstanding immediately before such Pro Rata Repurchase and the Warrant Base Price determined as of the date of the first public announcement by the Company or any of its Affiliates of the intent to effect such Pro Rata Repurchase, minus the aggregate purchase price of the Pro Rata Repurchase, and the denominator of which shall be the product of the number of shares of Ordinary Common Stock outstanding immediately prior to such Pro Rata Repurchase minus the number of shares of Ordinary Common Stock so repurchased and the Warrant Base Price determined as of the date of the first public announcement by the Company or any of its Affiliates of the intent to effect such Pro Rata Repurchase, and (ii) the number of Warrant Shares issuable upon the exercise of this Warrant shall be adjusted to the number obtained by multiplying the number of Warrant Shares issuable upon the exercise of the Warrant immediately prior to such adjustment by a fraction, the numerator of which is the Exercise Price in effect immediately prior to such adjustment and the denominator of which is the new Exercise Price determined in accordance with subsection (i) above. Such adjustments shall be made whenever any such Pro Rata Repurchase is made and shall become effective immediately after the effective date mentioned above. Notwithstanding the foregoing, the Exercise Price shall never be adjusted below the par value of the shares of Stock for which this Warrant is then exercisable.

12. Reorganization Events. In the event of:

- (a) any consolidation or merger of the Company with or into another Person or of another Person with or into the Company; or
- (b) any sale, transfer, lease or conveyance to another Person of all or substantially all of the property and assets of the Company;

in each case in which holders of Ordinary Common Stock would be entitled to receive cash, securities or other property for their shares of Ordinary Common Stock (any such event specified in this Section 12, a "Reorganization Event"), this Warrant shall, following the effective time of such Reorganization Event, without the consent of the Holder and at the sole discretion of the Company, (i) be converted into the cash, securities and other property receivable in such Reorganization Event by and in the same relative amounts as a holder of Ordinary Common Stock (other than securities issued or other property distributed by such holder or its Affiliates) holding, immediately prior to the Reorganization Event, a number of shares of Ordinary Common Stock equal to the number of Warrant Shares which would be issuable for Class B Common Stock or Series C Preferred Stock, as applicable, under this Warrant (disregarding the Share Issuance Limitation, if any) immediately prior to such Reorganization Event; provided, however, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Company or any other Person, then the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ only to the extent that the then existing designation and powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of the Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock differ as provided in the Certificate of Incorporation (including, without limitation, with respect to the voting rights and conversion provisions thereof) if and to the extent necessary due to regulatory requirements or restrictions applicable to the entity surviving the Reorganization Event that are similar in nature to those applicable to the Company; and provided, further, that, if the holders of the Ordinary Common Stock, Class B Common Stock or Series C Preferred Stock are granted the right to elect to receive one of two or more alternative forms of consideration, the foregoing provision shall be deemed satisfied if holders of the other class are granted identical election rights, subject to the previous proviso, or (ii) be redeemed by the Company for a cash price equal to 100% of the fair value of the consideration as determined by the Board of Directors, acting reasonably and in good faith, received under clause (i) (such cash, securities or other property, the "Exchange Property"); provided, however, that the Exchange Property shall be reduced by an amount equal to the Exercise Price that the Holder would have paid had the Holder exercised this Warrant on the date of such Reorganization Event; and provided, further, that, if the Exercise Price would exceed the value of the Exchange Property, then the Warrant shall be cancelled for no consideration in connection with the Reorganization Event. Notwithstanding anything to the contrary, this Section 12 shall not apply in the case of, and a Reorganization Event shall not be deemed to be, a merger, consolidation, reorganization or statutory share exchange (A) among the Company and its direct and indirect Subsidiaries or (B) between the Company and any Person for the primary purpose of changing the domicile of the Company.

13. Notice to Holder.

(a) Whenever the number of Warrant Shares or number or kind of securities or other property purchasable upon the exercise of this Warrant or the Exercise Price is adjusted, as provided herein, the Company shall give notice thereof to the Holder, which notice shall state the number of Warrant Shares (and other securities or property) purchasable upon the exercise of this Warrant and the Exercise Price of such Warrant Shares (and other securities or property) after such

adjustment, setting forth a brief statement of the facts requiring such adjustment and setting forth the computation by which such adjustment was made.

- (b) If at any time during the Exercise Period (i) the Company shall take a record of the holders of its Ordinary Common Stock and Class B Common Stock for the purpose of entitling them to receive a dividend or other distribution, or any right to subscribe for or purchase any evidences of its indebtedness, any shares of capital stock or any other securities or property or to receive any other right; (ii) there shall be any capital reorganization of the Company, any reclassification or recapitalization of the capital stock of the Company or any consolidation or merger of the Company with, or any sale, transfer or other disposition of all or substantially all of the property, assets or business of the Company to, another Person; or (iii) there shall be a voluntary or involuntary dissolution, liquidation or winding up of the Company, then the Company shall give to the Holder at least five Business Days' prior written notice of the date on which a record date shall be selected for such dividend, distribution or right or for determining rights to vote in respect of any such reorganization, reclassification, merger, consolidation, sale, transfer, disposition, liquidation or winding up, and in the case of any such reorganization, reclassification, merger, consolidation, dissolution, liquidation or winding up, at least 10 Business Days' prior written notice of the date when the same shall take place. Failure to provide such notice, or any defect therein, shall not affect the legality or validity of any such action.
- 14. <u>Transfer Restrictions</u>. Except for transfers to Permitted Holders, consistent with Section 9 of the Securities Purchase Agreement, the Company shall refuse to register any attempted transfer of this Warrant and any such purported transfer shall be null and void.
- 15. <u>Termination of Warrant</u>. If the Securities Purchase Agreement is terminated by the Company pursuant to Sections 10.2(d) or 10.3(d) thereof, or is subject to termination thereto at the time of termination as a result of a breach of the Securities Purchase Agreement by the Holder, this Warrant shall become immediately, automatically and irrevocably unexercisable and shall expire without any action required of the Company.

16. Miscellaneous.

(a) Governing Law. This Warrant shall be governed in all respects by the laws of the State of New York without regard to choice of laws or conflict of laws provisions thereof that would require the application of the Laws of any other jurisdiction.

(b) Dispute Resolution.

- (i) Executive Meetings. Prior to submitting any dispute or controversy arising from or in connection with this Agreement, including the breach, termination or invalidity thereof (a "Dispute"), to arbitration pursuant to Section 16(b)(ii), upon written request of any Party, each Party shall appoint a designated representative whose task it will be to meet promptly for the purpose of endeavoring to resolve such Dispute. The designated representatives shall meet, in person or by telephone or video conference as deemed appropriate by the Parties, as often as the Parties reasonably deem necessary to discuss the Dispute in an effort to resolve the Dispute without the necessity of any further proceeding. The Parties agree to negotiate, in good faith, in an attempt to resolve the Dispute for a period of not greater than thirty (30) days after notice of the Dispute is received by the Parties.
- (ii) <u>Arbitration; Rules; Location</u>. Any Dispute that is not resolved pursuant to Section 16(b)(i) shall be referred to and finally determined under the Rules of Arbitration of the International Chamber of Commerce then in effect (the "*ICC Rules*"). The place of arbitration shall be San Francisco, California, or such other location as the Parties may agree in writing.
- (iii) <u>Arbitrators</u>. There shall be three (3) arbitrators, nominated in accordance with the ICC Rules. Each arbitrator on the arbitral tribunal shall be disinterested in the Dispute and shall have no connection to any Party thereto.
- (iv) Award. The arbitral award shall be in writing, state the reasons for the award, and be the sole and exclusive binding remedy with respect to the Dispute between and among the Parties. Judgment on the award rendered may be entered in any court having jurisdiction thereof. The Parties hereby waive any right to refer any question of law and right of appeal on the law and/or merits to any court, except as provided by the Convention on the Recognition and Enforcement of Foreign Arbitral Awards of 1958. For purposes of such convention, the award shall be deemed an award of the United States, the relationship between the Parties shall be deemed commercial in nature, and any Dispute arbitrated pursuant to this Section 16(b) shall be deemed commercial. The arbitrators shall have the authority to grant any equitable or legal remedies that would be available in any judicial proceeding intended to resolve a Dispute.

- (v) Language of Proceedings. The language of the arbitral proceedings shall be English.
- (vi) Confidentiality of Proceedings. The Parties agree that any arbitration hereunder shall be kept confidential, and that the existence of the Dispute, the proceeding and all of its elements (including any pleadings, briefs or other documents submitted or exchanged, any testimony or other oral submissions, and any awards) shall be deemed Confidential Information for purposes of the confidentiality agreements referenced in Section 13.7 of the Securities Purchase Agreement, and shall not be disclosed beyond the tribunal, the International Court of Arbitration, the parties to the dispute, their counsel, and any Person necessary to the conduct of the proceeding, except as and to the extent required to enforce any arbitral award, or as otherwise contemplated by such confidentiality agreements.
 - (vii) Expenses. Each party hereto to a Dispute shall bear its own legal fees and costs in connection therewith.
- (viii) Notwithstanding the foregoing, any Party shall be entitled to seek preliminary injunctive relief from any court of competent jurisdiction to preserve the status quo, pending the final decision or award of the arbitrators. Each of the Parties hereby irrevocably consents to jurisdiction of any court in the United States of America (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding pursuant to this Section 16(b)(viii), or enforcing any award under Section 16(b)(iv), and irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. Process in any such action or proceeding may be served on any Party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 16(d) shall be deemed effective service of process on such Party.
- (ix) To the extent of concurrent Disputes (as such term is used in each of the Transaction Documents) under multiple Transaction Documents, the Parties agree to consolidate any and all such Disputes into a single proceeding pursuant to the procedures set forth in this Section 16(b).
- (c) <u>Nonwaiver and Expenses</u>. No course of dealing or any delay or failure to exercise any right hereunder on the part of the Holder or the Company shall operate as a waiver of such right or otherwise prejudice the Holder's or the Company's rights, powers or remedies; <u>provided</u>, <u>however</u>, that all rights hereunder shall terminate on the Termination Date, or such earlier termination of this Warrant pursuant to Section 15 hereof.
- (d) Notices. All notices, requests, consents and other communications provided for herein shall be in writing and shall be mailed by registered or certified mail, postage prepaid, return receipt requested, or otherwise delivered by hand or by messenger, addressed:
 - (i) If to the Holder, to:

[For Toshiba:

Toshiba U.S. Co 1-1 Shibaura 1-chome Minato-ku, Tokyo 105-8001 JAPAN Telephone: 011-81-3457-3706 Attention: General Manager Legal Affairs Division, Power Systems Company

With a copy to:

Morrison & Foerster LLP Shin-Marunouchi Building, 29th floor 5-1, Marunouchi 1-chome Chiyoda-ku, Tokyo 100-6529 JAPAN Telephone: 011-81-3-3214-6522

Telephone: 011-81-3-3214-6522 Attention: Ken Siegel, Esq.]

[For B&W:

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502 U.S.A.

Telephone: (434) 522-5589 Attention: James D. Canafax, Esq.

With a copy to:

Baker Botts
The Warner
1299 Pennsylvania Avenue, NW
Washington, D.C. 20004-2400
U.S.A.
Telephone: (202) 639-7724

Attention: Michael A. Gold, Esq.]

(ii) If to the Company, to:

USEC Inc. 6903 Rockledge Drive Bethesda, MD 20817 Telephone: (301) 564-3200 Attention: General Counsel

With a copy to:

Latham & Watkins LLP 555 Eleventh Street, NW Suite 1000

Washington, D.C. 20004-1304 Telephone: (202) 637-2200 Attention: Scott Herlihy

or in any such case to such other address, facsimile number or telephone as either Party may, from time to time, designate in a written notice given in a like manner. If notice is provided by mail, it shall be deemed to be delivered five Business Days following proper deposit in a mailbox, and if notice is delivered by hand, messenger or overnight courier service, it shall be deemed to be delivered upon actual delivery.

- (e) Amendment. This Warrant may be modified or amended or the provisions hereof waived, only upon the written consent of the Company and the Holder.
- (f) Severability. If any provision of this Warrant becomes or is declared by a court of competent jurisdiction to be illegal, unenforceable or void, portions of such provision, or such provision in its entirety, to the extent necessary, shall be severed from this Warrant and the balance of this Warrant shall be enforceable in accordance with its terms.
 - (g) Headings. The headings used in this Warrant are for the convenience of reference only and shall not, for any purpose, be deemed a part of this Warrant.
- 17. Entire Agreement. This Warrant, the form attached hereto, and the Securities Purchase Agreement and the exhibits thereto contain the entire agreement between the parties with respect to the subject matter hereof and supersede all prior and contemporaneous arrangements or undertakings with respect thereto.

[THIS SPACE LEFT BLANK INTENTIONALLY]

| USEC INC. |
|-----------------|
| Ву: |
| Name: Title: |

IN WITNESS WHEREOF, the Company has caused this Warrant to be executed by its officer thereunto duly authorized.

AGREED AND ACKNOWLEDGED:

[WARRANT HOLDER]

By:

Name: Title:

SIGNATURE PAGE TO WARRANT

FORM OF NOTICE OF EXERCISE

The undersigned hereby elects to purchase Warrant Shares of USEC Inc. pursuant to the terms of the attached Warrant, and tenders herewith payment, in lawful money of the United States, of the Exercise Price in full, together with all applicable transfer taxes, if any; OR the undersigned hereby elects to exercise the Net Exercise Right to purchase Warrant Shares of USEC Inc. pursuant to the terms of the attached Warrant. Please issue a certificate or certificates representing said Warrant Shares in the name of the undersigned. The Warrant Shares shall be delivered to the following: The undersigned is an "accredited investor" as defined in Regulation D under the Securities Act. If applicable, a new Warrant evidencing the remaining Warrant Shares covered by the attached Warrant shall be issued in the name of the undersigned. The new Warrant shall be delivered to the following: [PURCHASER]

By:

Dated:

Name: Title:

EXHIBIT F

FORM OF INVESTOR RIGHTS AGREEMENT

INVESTOR RIGHTS AGREEMENT

THIS INVESTOR RIGHTS AGREEMENT (this "Agreement") is made as of , 2010, by and among Toshiba Corporation, a Japanese corporation ("Toshiba"), Babcock & Wilcox Investment Company, a Delaware corporation ("B&W", and together with Toshiba, the "Investors"), and USEC Inc., a Delaware corporation (the "Company"). The Investors and the Company are referred to herein collectively as the "Parties", and each individually as a "Party".

RECITALS

WHEREAS, the Investors and the Company are parties to a Securities Purchase Agreement, dated as of May 25, 2010 (the 'SPA"), providing for the Company's issuance and sale to the Investors of shares of Series B-1 12.75% Convertible Preferred Stock of the Company, par value \$1.00 per share and the Series B-2 11.5% Convertible Preferred Stock of the Company, par value \$1.00 per share (together, the "Preferred Stock"), and certain other transactions, in each case on the terms set forth therein (collectively, the "Transaction"); and

WHEREAS, the Parties are entering into this Agreement concurrently with the consummation of the First Closing under the SPA, in accordance therewith.

NOW, THEREFORE, in consideration of the mutual covenants set forth herein, the Parties agree as follows:

ARTICLE I INVESTOR DIRECTORS

1.1 Information Rights. For so long as an Investor Director continues to serve as a Director, the Company shall provide such Investor Director with all information made available to other Directors, as and when so made available to such other Directors; *provided, however*, that the Company shall not provide to (a) any Investor Director that is or is appointed or elected by a non-U.S. Person, any Classified Information or Export Controlled Information, except as permitted by an approved Negation Plan or applicable Laws, or (b) any Investor Director, any Competitively Sensitive Information.

1.2 Indemnification and Insurance.

- (a) For so long as an Investor Director continues to serve as a Director, and for a period of six (6) years thereafter:
- (i) (x) The Company shall, to the extent permitted by applicable Laws, indemnify and hold harmless all current and former Directors on terms that are no less favorable than the provisions contained in the Certificate of Incorporation and Bylaws of the Company as of the date hereof, and (y) such provisions shall not, except as required to comply with changes in applicable Laws, be amended, repealed or otherwise modified in any manner that would adversely affect the rights of such Directors: and
- (ii) The Company shall maintain in full force and effect Directors' and officers' liability insurance ('**D&O Insurance**') in reasonable amounts from established and reputable insurers on terms no less favorable in the aggregate than those now applicable to Directors and officers of the Company, to the extent commercially reasonably available.
- (b) In all D&O Insurance policies, each Investor Director shall be covered as an insured in such a manner as to provide the Investor Director with rights and benefits no less favorable than provided to the other Directors.
- (c) Notwithstanding the foregoing, the Company shall not be obligated to make annual premium payments for D&O Insurance to the extent that such premiums exceed three hundred percent (300%) of the annual premiums paid as of the date hereof by the Company for such insurance (such three hundred percent (300%) amount, the "Maximum Premium"). If such insurance coverage cannot be obtained at all, or can only be obtained at an annual premium in excess of the Maximum Premium, the Company shall maintain the most advantageous policies of D&O Insurance obtainable for an annual premium equal to the Maximum Premium; provided, that in all events the coverage for the Investor Directors shall be no less favorable than that accorded to any other Director of the Company.

1.3 Qualified Directors.

- (a) Notwithstanding the right of the Investors to elect the Investor Directors, no Investor Director shall be, and no Investor shall have the right to elect an Investor Director who is a director, officer or employee of, works for or on, or is assigned to any Competitor Affiliate (as defined in the Strategic Relationship Agreement dated as of May 25, 2010, by and among the Parties) of an Investor (any such Person engaged in such activities or having such position or capacity shall be considered "Unqualified").
- (b) Each Investor hereby covenants that for a period of eighteen (18) months after such Investor Director no longer serves as a Director, such former Investor Director will not be permitted to serve as a director, officer or employee of, or work for, or be assigned to any Competitor Affiliate of such Investor.
 - (c) For the avoidance of doubt, the Parties agree that Westinghouse Electric Company, LLC, is not a Competitor Affiliate of Toshiba.
- (d) If at any time an Investor Director becomes Unqualified, the Investors hereby agree to promptly cause such Investor Director to resign from the Board and take all other actions necessary to effect such removal, including acting by written consent, to remove such Investor Director from the Board.

ARTICLE II APPROVAL RIGHTS

- 2.1 Special Approval Rights. Following the Third Closing, for so long as either Investor maintains its Minimum Equity Holdings, the Company shall not (and shall not permit any of its Subsidiaries to) adopt a plan of complete or partial dissolution or liquidation of the Company (other than in connection with a merger, sale of substantially all of the Company's assets or other business combination transaction) without the prior written approval of the holders of a majority of the Preferred Stock then outstanding (voting together as a single class).
 - 2.2 Minimum Equity Holdings. An Investor will be deemed to maintain its 'Minimum Equity Holdings':
- (a) (1) from the date of the First Closing until December 31, 2016, so long as, at any time and from time to time, (1) the Original Issue Value (as defined below) of all of the then-outstanding shares of the Preferred Stock acquired by such Investor in the Closings (excluding PIK Shares and shares acquired upon exercise of the Warrants), plus (2) for each share of Series C Preferred Stock then held by such Investor, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which such Investor's acquisition of such shares was calculated, plus (3) the aggregate Base Price for each share of the Class B Common Stock received upon conversion of the Preferred Stock then held by such Investor, plus (4) the aggregate amount of accrued and unpaid Dividends on such Investor's outstanding shares of Series B Preferred Stock which have been added to the Liquidation Preference pursuant to Section 5(a) of the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable (collectively, such Investor's "Aggregate Outstanding Value") exceeds seventy-five percent (75%) of the aggregate Initial Liquidation Preference of all of the shares of the Preferred Stock issued in the Closings to such Investor, excluding PIK Shares and shares of Class B Common Stock and Series C Preferred Stock acquired on the exercise of Warrants (such Investor's "Original Issue Value"); and
 - (b) after December 31, 2016, so long as such Investor's Aggregate Outstanding Value exceeds fifty percent (50%) of such Investor's Original Issue Value.

In each case, for purposes of determining Aggregate Outstanding Value and Original Issue Value, if there has been an automatic redemption pursuant to Section 7(g) of the Series B-1 Certificate of Designation, subsequent to a Conversion Election, Section 8(c) of the Series B-1 Certificate of Designation or Section 8(c) of the Series B-2 Certificate of Designation (each, an "Automatic Redemption"), the aggregate amount of the Liquidation Preference as of the date of redemption of such Investor's Securities (excluding PIK Shares and shares acquired upon exercise of the Warrants) so redeemed shall be added to such Investor's Aggregate Outstanding Value and the aggregate amount of the Liquidation Preference as of the date of redemption of such Investor's Securities so redeemed shall be added to such Investor's Original Issue Value; provided, however, that, if at any time after any Automatic Redemption, such Investor's Deemed Holder Percentage is less than eight percent (8%), then such adjustment to Aggregate Outstanding Value and Original Issue Value shall not be made for purposes of determining such Investor's Investor including any other rights of such Investor, including any

rights of such Investor under Article II of the Strategic Relationship Agreement.

ARTICLE III SUBSCRIPTION RIGHTS

3.1 Subscription Rights.

- (a) If the Company proposes to sell shares of any preferred stock (other than Preferred Stock or in exchange therefore) ('Additional Securities''), including in a private placement, a public offering, as part of an acquisition, share exchange or otherwise, the Company shall, at least thirty (30) days prior to issuing such Additional Securities, notify each Investor in writing of such proposed issuance specifying the material terms and conditions thereof, including: (i) the number and description of such Additional Securities proposed to be issued and the percentage of the Company's outstanding equity interests that such issuance would represent; (ii) the proposed issuance date; and (iii) the form of consideration and the proposed purchase price per share (such notice, the "Subscription Right Notice"), and shall, subject to the receipt by the Company of any required Nuclear and National Security Approvals (and to the Company's compliance with Section 3.7 in respect thereof), offer to sell such Additional Securities to the Investors in the amounts set forth in Section 3.1(c), upon the terms and subject to the conditions set forth in the Subscription Right Notice and at the Purchase Price as set forth in Section 3.1(d) (the "Subscription Rights"); provided, that, if the purchase price for, or any of the other material terms and conditions of, the proposed issuance change or are not known at the time of provision of the Subscription Right Notice, the Company shall provide the Subscription Right Notice specifying that the price or other such terms and conditions are not yet available, and shall provide a supplemental notice (the "Additional Notice"), adding the missing terms, to the Investors as soon as they are known to the Company, and in no event later than ten (10) Business Days prior to such issuance.
- (b) If an Investor wishes to subscribe for a number of Additional Securities less than the number to which it is entitled under this Section 3.1, such Investor may do so (but not less than 10% of the number to which it is entitled) and shall, in the notice of exercise of the offer, specify the number of Additional Securities that it wishes to purchase, which shall not be less than 10% of the shares to which it is entitled.
- (c) The Company shall offer each Investor all, or any portion specified by the Investor in accordance with Section 3.1(b), of an amount of such Additional Securities such that, after giving effect to the proposed issuance (including the issuance to the Investor pursuant to the Subscription Rights and including any related issuance resulting from the exercise of preemptive or similar rights by any unrelated Person with respect to the same issuance that gave rise to the exercise of the Subscription Rights by the Investor), the Investor's Equity Interest after such issuance would equal the Investor's Equity Interest immediately prior to such issuance, such number of Additional Securities to constitute the "Subscription Share Amount". If, at the time of the determination of any Subscription Share Amount under this Section 3.1(c), any other Person has subscription or other equity purchase rights similar to the Subscription Rights, such Subscription Share Amount shall be recalculated to take into account the amount of Additional Securities such Persons have committed to purchase, rounding up such Subscription Share Amount to the nearest whole Additional Security.
- (d) The "Purchase Price" for the Additional Securities to be issued pursuant to the exercise of Subscription Rights shall be payable to the extent practicable, in the same form of consideration set forth in the Subscription Right Notice (unless otherwise agreed by the Company and the applicable Investor) and, except as otherwise set forth below, shall equal per Additional Security the per Security issuance price for the Additional Securities giving rise to such Subscription Right. In the case of any issuance of Additional Securities other than solely for cash, the Company and each Investor making a non-cash payment shall in good faith seek to agree upon the value of the non-cash consideration; *provided*, that the value of any publicly traded securities shall be deemed to be the closing price of such securities on the applicable national securities exchange as of the Trading Day immediately prior to the consummation of such issuance. If the Company and such Investor fail to agree on the value of such non-cash consideration during the period contemplated by the first sentence of Section 3.2, then the Company shall refer the items in dispute to a nationally recognized investment banking firm that is selected by the Board and acceptable to the Investor and that shall be instructed to make a final and binding determination of the fair market value of such items within ten (10) Business Days. If such a determination is required, the deadline for an Investor's exercise of its Subscription Rights with respect to such issuance pursuant to Section 3.1(b) shall be extended until the fifth (5th) Business Day following the date of such determination. Whichever of the Company or the Investor whose last estimate communicated in writing to the other party differed in dollar value the most from that finally determined by such investment banking firm shall be responsible for and pay all of the fees and expenses of such investment banking firm. All determinations made by such investment banking firm shall be final and binding on
 - 3.2 Exercise Period. The Subscription Rights set forth in Section 3.1 must be exercised by acceptance in writing of

an offer referred to in Section 3.1(a) deliverable to the Company within ten (10) Business Days of receiving notice from the Company of its intention to sell Additional Securities, or, if applicable and later, within ten (10) days of any Additional Notice. The closing of any purchase of Additional Securities pursuant to the exercise by the Investor of Subscription Rights hereunder shall occur on the closing of the transaction triggering such Subscription Rights, subject to the receipt of any necessary Governmental Approvals to which the issuance of Additional Securities is subject (and the Company shall use its commercially reasonable efforts to obtain such Governmental Approvals).

- 3.3 Sales to the Prospective Buyer. If an Investor fails to elect to purchase all or part of its allotment of the Additional Securities described in the Subscription Right Notice within the time period described in Section 3.2, the Company shall be free to complete the proposed issuance or sale of Securities described in the Subscription Right Notice at a price and on other terms no less favorable to the Company than those set forth in the Subscription Right Notice. If the Company does not enter into such an agreement for the sale of such Securities within thirty (30) Business Days after the expiration of the time period described in Section 3.2, or if such agreement is not consummated within ninety (90) days after the execution thereof, such Subscription Rights shall be deemed to be revived and such Additional Securities shall not be issued or sold unless first reoffered to the Investors in accordance with this Article III.
 - 3.4 Survival of Rights. The rights of an Investor set forth in this Article III shall continue for so long as such Investor maintains its Minimum Preferred Holdings.
- 3.5 No Subscription Rights for Third Party Financing. Notwithstanding anything to the contrary set forth herein, the Investors shall not be entitled to their respective Subscription Rights with respect to issuances of securities in a Third Party Financing.
- 3.6 No Additional Funding Obligations. For the avoidance of doubt, the Investors shall not have any obligation to fund or provide financial support of any kind (whether fixed, contingent, conditional or otherwise) to the Company beyond their respective obligations as to the Closings as set forth in the SPA.
- 3.7 Nuclear and National Security Approvals. The Company shall use all commercially reasonable efforts to obtain any Nuclear and National Security Approvals required with respect to the Investors' exercise of their respective rights under this Agreement, and each Investor shall reasonably cooperate with the Company in such efforts.

ARTICLE IV REGISTRATION RIGHTS

4.1 Shelf Registration Statement.

- (a) The Company shall, no later than the Filing Date, file with the SEC a Shelf Registration Statement (the **Initial Shelf**') relating to the offer and sale of the Registrable Securities by the Investors from time to time to permit the sale of Registrable Securities by the Investors pursuant to the Orderly Sale Arrangement set forth in Section 9 of the SPA and, thereafter, shall use its best efforts to cause the Initial Shelf to be declared effective under the Securities Act no later than ninety (90) calendar days following the date first filed with the SEC. None of the Company's securityholders (other than the Investors) shall have the right to include any Securities of the Company on the Initial Shelf.
- (b) For so long as either Investor holds any Registrable Securities, the Company shall use its best efforts to maintain an effective Shelf Registration Statement registering all unsold Registrable Securities. Subject to Section 4.3, the Company shall use its best efforts to keep a Shelf Registration Statement continuously effective, supplemented and amended as required by the provisions of Section 4.3(i), in order to permit the Prospectus forming a part thereof to be usable by the Investors until the date that the Investors no longer hold Registrable Securities (such period, the "Resale Period"). The Company shall be deemed not to have used its best efforts to keep a Shelf Registration Statement effective during the Resale Period if the Company voluntarily takes any action that would result in the Investors' not being able to offer and sell any of such Registrable Securities under such Shelf Registration Statement during that period, unless such action is (x) required by applicable Laws and the Company thereafter promptly complies with the requirements of Section 4.3 or (y) permitted pursuant to Section 4.3.
- (c) For the avoidance of doubt, nothing in this Agreement shall require the Company to undertake or otherwise support an underwritten offering of Registrable Securities unless the Company determines to do so in its sole discretion.
 - (d) For the avoidance of doubt, any sale made pursuant to the Initial Shelf shall be subject to the Orderly

Sale Arrangement set forth in Section 9 of the SPA.

4.2 Piggyback Rights.

- (a) If at any time an Investor is in possession of Registrable Securities and the Company proposes to effect an underwritten registration of any of its securities (other than in a Third Party Financing) under the Securities Act (other than any registration of Securities on Form S-4 or Form S-8 or any successor forms), for its own account, or for the account of one or more stockholders of the Company (other than pursuant to the Initial Shelf) (each, a "Proposed Registration"), the Company shall give prompt written notice to the Investors of the Company's intention to do so. If an Investor's Registrable Securities have not been included in the Proposed Registration, and within ten (10) Business Days of the receipt of any such notice such Investor delivers to the Company a written notice requesting to have any or all of its Registrable Securities included in such Proposed Registration (such notice to include the number of Registrable Securities that the Investor wishes to be included in the Proposed Registration), the Company shall use its commercially reasonable efforts to cause such shares to be registered as requested in such notice. Notwithstanding any other provision of this Section 4.2(a), if the managing underwriter advises the Company that marketing factors require a limitation of the number of shares to be underwritten, the Company may limit the number of shares of Registrable Securities to be included in the Proposed Registration without requiring any limitation in the number of shares to be registered on behalf of the Company; provided, however, that the number of Registrable Securities included in the Proposed Registration pursuant to this Section 4.2(a) may not be reduced to less than twenty-five percent (25%) of the total number of shares requested by the Investors to be included in the Proposed Registration (the "Cut Back Limit"), and any such cut back will be implemented on a pro rata basis according to the number of shares requested by each Investor to be included in the Proposed Registration; provided, further, that nothing herein shall prevent the Company from canceling or withdrawing any Proposed Registration prior to the filing or effectiveness thereof. Registrable Securities held by the Investors proposed to be included on a Proposed Registration shall have priority over all securities proposed to be included on such Registration Statement other than (i) securities to be sold by the Company unless the following clause (ii) applies, or (ii) if the Proposed Registration is pursuant to contractual demand rights of another Person, securities proposed to be included by such Person, which shall, subject to the Cut Back Limit, have priority over the Registrable Securities on such Registration Statement.
- (b) No Investor's Registrable Securities shall be registered unless such Investor accepts the terms of the underwriting as approved by the Company for the offering; provided that the Investor may independently negotiate with the underwriters for the offering any representations and warranties that the Investor shall give to such underwriters in connection with the offering. In the event that an Investor is unable to agree with such underwriters on such representations and warranties or does not accept the terms of such underwriting, then the Company may proceed with the Proposed Registration without the participation of such Investor or the inclusion of any of such Investor's Registrable Securities.
 - **4.3 Registration Procedures.** In connection with the Registration Statements, the following provisions shall apply:
- (a) The Company shall furnish to each Investor, prior to the Effective Time, a copy of any Registration Statement filed with the SEC (or to be filed in the case of an automatically effective Registration Statement), and shall furnish to each Investor, prior to filing with the SEC, copies of each amendment thereto and each amendment or supplement, if any, to the Prospectus included therein, and shall in good faith consider the reasonable comments suggested by such counsel, including consideration of inclusion thereof in the Registration Statement or Prospectus. The Company shall deliver promptly to counsel to the Investors and to each underwriter, if any, participating in the offering of the Registrable Securities, copies of all correspondence between the SEC and the Company, its counsel or its auditors with respect to such Registration Statement.
- (b) The Company shall promptly take such action as may be necessary so that (i) each of the Registration Statements and any amendment thereto and the Prospectus forming a part thereof and any amendment or supplement thereto (and each report or other document incorporated therein by reference in each case) complies in all material respects with the Securities Act and the Exchange Act and the respective rules and regulations thereunder, as in effect at any relevant time, (ii) each of the Registration Statements and any amendment thereto does not, when it becomes effective, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading, and (iii) each Prospectus forming a part of any Registration Statement, and any amendment or supplement to such Prospectus, in the form delivered to purchasers of the Registrable Securities during the Resale Period, does not include an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.
 - (c) The Company shall promptly advise each Investor, and shall confirm such advice in writing if requested by either Investor:
 - (i) when any Registration Statement has been filed with the SEC and when any Registration

Statement has become effective;

- (ii) when any supplement to the Prospectus, Registration Statement or post-effective amendment to a Registration Statement has been filed with the SEC and, with respect to a Registration Statement or any post-effective amendment thereto, when the same has been declared effective by the SEC;
- (iii) of any SEC comments on, or request by the SEC for amendments or supplements to, any Registration Statement or the Prospectus included therein or for additional information;
- (iv) of the issuance by the SEC of any stop order suspending the effectiveness of any Registration Statement or the initiation of any proceedings for such purpose;
- (v) of the receipt by the Company of any notification with respect to the suspension of the qualification of the securities included in any Registration Statement for sale in any jurisdiction or the initiation of any proceeding for such purpose; and
- (vi) of the happening of any event or the existence of any state of facts that requires the making of any changes in any Registration Statement or the Prospectus included therein so that, as of such date, such Registration Statement and Prospectus do not contain an untrue statement of a material fact and do not omit to state a material fact required to be stated therein or necessary to make the statements therein (in the case of the Prospectus, in the light of the circumstances under which they were made) not misleading (which advice shall be accompanied by an instruction to the Investors to suspend the use of the Prospectus until the requisite changes have been made).
- (d) The Company shall use its commercially reasonable efforts to prevent the issuance, and if issued to obtain the withdrawal at the earliest possible time, of any order suspending the effectiveness of any Registration Statement.
- (e) As promptly as reasonably practicable, the Company shall furnish to each Investor, upon their request and without charge, one (1) conformed copy of any Registration Statement and any amendment thereto, including financial statements but excluding schedules, all documents incorporated or deemed to be incorporated therein by reference and all exhibits (unless reasonably requested in writing to the Company by such Investor).
- (f) The Company shall, during the period that any Registration Statement is effective, deliver to each Investor, without charge, as many copies of each Prospectus included in the applicable Registration Statement and any amendment or supplement thereto as such Investor may reasonably request; and the Company consents (except during a Suspension Period or during the continuance of any event described in Section 4.3(c)(iii)-(vi)) to the use of the Prospectus and any amendment or supplement thereto by each of the Investors in connection with the offering and sale of the Registrable Securities covered by the Prospectus and any amendment or supplement thereto during the period that any Registration Statement is effective.
- (g) Prior to any offering of Registrable Securities pursuant to a Registration Statement, the Company shall use commercially reasonable efforts to (i) register or qualify or cooperate with the Investors and their respective counsel in connection with the registration or qualification (or exemption from such registration or qualification) of such Registrable Securities for offer and sale under the securities or "blue sky" laws of such jurisdictions within the United States as any Investor may reasonably request in writing, (ii) keep such registrations or qualifications or exemption therefrom in effect and comply with such Laws so as to permit the continuance of offers and sales in such jurisdictions for so long as may be necessary to enable any Investor or underwriter, if any, to complete its distribution of Registrable Securities pursuant to such Registration Statement, and (iii) take any and all other actions reasonably necessary to enable the disposition in such jurisdictions of such Registrable Securities; provided, however, that in no event shall the Company be obligated to (A) qualify as a foreign corporation or as a dealer in securities in any jurisdiction where it would not otherwise be required to so qualify but for this Section 4.3(g) or (B) subject itself or its Affiliates to general or unlimited service of process or to taxation in any such jurisdiction if they are not now so subject.
- (h) Unless any Registrable Securities shall be in book-entry only form, if requested, the Company shall cooperate with the Investors to facilitate the timely preparation and delivery of certificates representing Registrable Securities to be sold pursuant to any Registration Statement, which certificates, if so required by any Approved Market upon which any Registrable Securities are listed, shall be free of any restrictive legends other than those required by Article ELEVENTH of the Certificate of Incorporation, and in such permitted denominations and registered in such names as the Investors may request in connection with the sale of Registrable Securities pursuant to such Registration Statement.
- (i) Upon the occurrence of any fact or event contemplated by Section 4.3(c)(vi), subject to Section 4.3(j), the Company shall use its best efforts to promptly, but in any event within ten (10) Business Days following such occurrence,

prepare and file (and have declared effective) a post-effective amendment to any Registration Statement or an amendment or supplement to the related Prospectus included therein or file any other document with the SEC so that, as thereafter delivered to purchasers of the Registrable Securities, the Prospectus will not include an untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading. If the Company notifies the Investors of the occurrence of any fact or event contemplated by Section 4.3(c)(vi), the Investors shall suspend the use of the Prospectus until the requisite changes to the Prospectus have been made, and shall thereafter distribute the updated Prospectus to purchasers of Registrable Securities to ensure that the Prospectus received by such purchaser does not include an untrue statement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

- (j) After the Effective Time of any Registration Statement, the Company may suspend the use of any Prospectus by written notice to the Investors for a period not to exceed an aggregate of forty-five (45) calendar days in any ninety (90) calendar day period (each such period, a "Suspension Period") if:
- (i) an event has occurred and is continuing as a result of which the Registration Statement would, in the Company's judgment, contain an untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading; and
- (ii) the Company determines in good faith that the disclosure of such event at such time would have a material adverse effect on the Company and its Subsidiaries taken as a whole;

provided, that in the event that such disclosure relates to a previously undisclosed proposed or pending material business transaction, the disclosure of which would impede the Company's ability to consummate such transaction, the Company may extend the applicable Suspension Period from forty-five (45) calendar days to ninety (90) calendar days; provided, however, that Suspension Periods (including, without limitation, any such extension of a Suspension Period) shall not exceed an aggregate of one hundred twenty (120) calendar days in any three hundred sixty (360) calendar day period. Each Investor shall keep confidential any communications received by it from the Company regarding the suspension of the use of the Prospectus, except as required by applicable Laws.

- (k) The Company shall enter into such customary agreements and take all such other necessary actions in connection therewith (including those reasonably requested by any Investor) in order to expedite or facilitate disposition of such Registrable Securities; *provided*, that the Company shall not be required to take any action in connection with an underwritten offering without the Company's consent (such consent not to be unreasonably withheld, conditioned or delayed).
- (1) To the extent permitted by Law, the Company shall (i) make reasonably available for inspection by any Investor and by any underwriter participating in any disposition pursuant to any Registration Statement, and by any attorney, accountant or other agent retained by such Investor or any such underwriter, all relevant financial and other records, pertinent corporate documents and properties of the Company and its subsidiaries, and (ii) cause the Company's officers, Directors and employees to make reasonably available for inspection all information reasonably requested by any Investor or any such underwriter, attorney, accountant or agent in connection with such Registration Statement, in each case, as is customary for similar due diligence examinations; provided, however, that such Persons shall, at the Company's request, first enter into the Company's standard confidentiality agreement, which, among other things, requires that confidential information shall be used solely for the purposes of exercising rights under this Agreement, unless such disclosure is made in connection with a court proceeding or required by Law, or such records, information or documents become available to the public generally or through a third party without an accompanying obligation of confidentiality; provided, further, that, if the foregoing inspection and information gathering would otherwise disrupt the Company's conduct of its business, such inspection and information gathering shall, to the greatest extent possible, be coordinated on behalf of the Investors and the other parties entitled thereto by one counsel designated by and on behalf of the Investors and other such parties.
- (m) The Company shall use its best efforts to furnish to each Investor and, in any underwritten offering, each underwriter, a signed counterpart of (i) an opinion of counsel for the Company and (ii) a "comfort" letter signed by the independent public accountants who have certified the Company's financial statements included or incorporated by reference in such Registration Statement, in each case addressed to each Investor and, if applicable, each underwriter, covering matters with respect to such Registration Statement (and the Prospectus included therein) as the managing underwriter, if any, and the Investors shall request.
- (n) The Company shall use its best efforts to cause the Registrable Securities to be listed on the NYSE or other market or stock exchange on which the Ordinary Common Stock primarily trades on or prior to the Effective Time of each Registration Statement hereunder.

- (o) The Company shall cooperate and assist in any filings required to be made with Financial Industry Regulatory Authority, Inc.
- (p) The Company shall use its best efforts to take all other steps necessary to effect the registration, offering and sale of the Registrable Securities covered by each Registration Statement contemplated hereby, including by taking all appropriate steps with respect to the marketing of an offering of Registrable Securities and making its management available to participate in road shows and other marketing activities in connection therewith.
- (q) Notwithstanding anything to the contrary set forth herein, the Initial Shelf and any subsequent Shelf Registration Statement shall be filed as an automatically effective registration statement if the Company is eligible to do so.
- 4.4 Registration Expenses. The Company shall bear all fees and expenses incurred in connection with the performance by the Company of its obligations under this Article IV whether or not any of the Registration Statements are declared effective. Such fees and expenses shall include, without limitation, (i) all registration and filing fees (including fees and expenses (x) with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc. and (y) of compliance with United States federal and state securities or "blue sky" laws to the extent that such filings or compliance are required pursuant to this Agreement (including reasonable fees and disbursements of the counsel specified in the next sentence in connection with "blue sky" qualifications of the Registrable Securities under the laws of such jurisdictions as any Investor may designate)), (ii) printing expenses, (iii) duplication expenses relating to copies of any Registration Statement or Prospectus delivered to any Investor hereunder, (iv) fees and disbursements of counsel for the Company in connection with any Registration Statement, (v) reasonable fees and disbursements of the transfer agent for the Common Stock, and (vi) all fees for any accountants, including in connection with any comfort letter. In addition, the Company shall bear or reimburse the Investors for the reasonable fees and disbursements of legal counsel for the Investors, which amount shall not exceed \$100,000 in the aggregate. In addition, the Company shall pay the internal expenses of the Company (including all salaries and expenses of officers and employees performing legal or accounting duties), the expense of any annual audit, the fees and expenses incurred in connection with the listing of the Registrable Securities on any securities exchange on which similar securities of the Company are then listed and the fees and expenses of any Person, including special experts, retained by the Company.

4.5 Indemnification and Contribution.

(a) Indemnification by the Company. The Company shall indemnify and hold harmless each Investor, each underwriter, if any, and each Person, if any, who controls any such Investor or underwriter within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act (for purposes of this Section 4.5(a), each an "indemnified party"), from and against any loss, claim, damage, liability or expense whatsoever as incurred (including reasonable attorneys' fees and any and all expenses whatsoever incurred in investigating, preparing or defending against any litigation, commenced or threatened, or any claim whatsoever, and any and all amounts paid in settlement of any claim or litigation), joint or several, to which they or any of them may become subject under the Securities Act, the Exchange Act or otherwise, insofar as any such loss, claim, damage, liability or expense (or action in respect thereof) arises out of, or is based upon, any untrue statement or alleged untrue statement of a material fact contained in any Registration Statement or any amendment thereto or any related preliminary prospectus or the Prospectus or any amendment thereto of supplement thereof, or arises out of, or is based upon, the omission or alleged omission to state therein any material fact required to be stated therein or necessary to make the statements therein not misleading; provided, however, that the Company shall not be liable to any such indemnified party in any such case to the extent that any such loss, claim, damage, liability or expense arises out of, or is based upon, (i) any such untrue statement or alleged untrue statement or omission or alleged omission made therein in reliance upon and in conformity with written information furnished to the Company by or on behalf of an indemnified party specifically for use therein; (ii) an offer or sale by an Investor of Registrable Securities during a Suspension Period, if such indemnified party is an Investor or an Affiliate of an Investor that, in either case, received from the Company a notice of the commencement of such Suspension Period prior to the making of such offer or sale; or (iii) in a case where a copy of a Prospectus required to be delivered by such indemnified party, is based upon an untrue statement or alleged untrue statement or omission or alleged omission made in a preliminary Prospectus and a copy of the amended or updated Prospectus was not sent or given by or on behalf of such indemnified party to the Person asserting an such loss, claim, damage, liability or expense (if required by Law so to have been delivered) at or prior to the written confirmation of the sale of Registrable Securities as required by the Securities Act and the Prospectus would have corrected such untrue statement or omission or alleged untrue statement or omission; provided that in clause (iii) the Company had promptly notified the Investors of such untrue statement or alleged untrue statement or omission or alleged omission in compliance with Section 4.3(i). The foregoing indemnity agreement is in addition to any liability that the Company may otherwise have to any indemnified party. The Company shall not be liable under this Section 4.5(a) for any settlement of any action effected without its written consent, which shall not be unreasonably withheld, conditioned or delayed; provided, however, that with respect to actions pursuant to clauses (i), (ii) and (iii) of Section 4.6(c), no such consent shall be required.

(b) Indemnification by the Investors. Each Investor, severally and not jointly, shall indemnify and hold harmless the Company, and each Person, if any, who controls the Company within the meaning of Section 15 of the Securities Act or Section 20 of the Exchange Act (for purposes of this Section 4.5(b), each an "indemnified party"), from and against any loss, claim, damage, liability or expense whatsoever as incurred (including reasonable attorneys' fees and any and all expenses whatsoever incurred in investigating, preparing or defending against any litigation, commenced or threatened, or any claim whatsoever, and any and all amounts paid in settlement of any claim or litigation), joint or several, to which they or any of them may become subject under the Securities Act, the Exchange Act or otherwise, insofar as any such loss, claim, damage, liability or expense (or action in respect thereof) arises out of, or is based upon, any untrue statement or alleged untrue statement of a material fact contained in any Registration Statement or any amendment thereto or any related preliminary prospectus or the Prospectus or any amendment thereto or supplement thereof, or arises out of, or is based upon, the omission or alleged omission to state therein any material fact required to be stated therein or necessary to make the statements therein not misleading, in each case to the extent, but only to the extent, that such untrue statement or alleged untrue statement or omission or alleged omission made therein was made in reliance upon and in conformity with written information furnished to the Company by or on behalf of such Investor specifically for use therein. In no event shall the liability of any Investor hereunder be greater in amount than the dollar amount of the proceeds received by such Investor upon the sale of the Registrable Securities pursuant to the Registration Statement giving rise to such indemnification obligation, except in the case of fraud, willful misconduct or gross negligence by an Investor. The fore

(c) Notices of Claims, Etc. Promptly after receipt by an indemnified party under this Section 4.5 of notice of any claim or the commencement of any action, the indemnified party shall, if a claim in respect thereof is to be made against the indemnifying party under this Section 4.5, notify the indemnifying party in writing of the claim or the commencement of such action; provided, however, that the failure to notify the indemnifying party shall not relieve it of any liability that it may have under this Section 4.5, unless the indemnifying party is materially prejudiced by such delay or failure. If any such claim or action shall be brought against an indemnified party, and it shall notify the indemnifying party thereof, the indemnifying party shall be entitled to participate therein and, to the extent that it wishes, jointly with any other similarly notified indemnifying party, to assume the defense thereof with counsel satisfactory to the indemnified party. After notice from the indemnifying party to the indemnified party of its election to assume the defense of such claim or action, the indemnifying party shall not be liable to the indemnified party under this Section 4.5 for any legal or other expenses subsequently incurred by the indemnified party in connection with the defense thereof other than reasonable costs of investigation; provided, however, that the indemnified party shall have the right to employ counsel to represent jointly the indemnified party and its respective directors, employees, officers and controlling persons who may be subject to liability arising out of any claim in respect of which indemnity may be sought by the indemnified party against the indemnifying party under this Section 4.5 if (i) employment of such counsel has been authorized in writing by the indemnifying party, or (ii) such indemnifying party shall not have employed counsel to have charge of the defense of such proceeding within thirty (30) days of the receipt of notice thereof, or (iii) such indemnified party shall have reasonably concluded, based upon written advice of such indemnified party's counsel, that the representation of such indemnified party and those directors, employees, officers and controlling persons by the same counsel representing the indemnifying party would be inappropriate under applicable standards of professional conduct due to actual or potential differing interests between them or where there may be one or more defenses available to them that are different from or in conflict with those available to the indemnifying party, and in any such event ((i), (ii) or (iii)) the fees and expenses of such separate counsel shall be paid by the indemnifying party as incurred. It is understood that the indemnifying party shall not be liable for the fees and expenses of more than one separate firm (in addition to local counsel in each jurisdiction) for all indemnified parties in connection with any proceeding or related proceedings. No indemnifying party shall, without the prior written consent of the indemnified parties (which shall not be unreasonably withheld, conditioned or delayed), effect any settlement or compromise of, or consent to the entry of judgment with respect to, any pending or threatened claim, investigation, action or proceeding in respect of which indemnity or contribution may be or could have been sought hereunder (whether or not the indemnified party or parties are actual or potential parties thereto) unless (1) such settlement, compromise or judgment (x) includes an unconditional release of such indemnified party from all liability arising out of such claim, action, suit or proceeding and (y) does not include a statement as to or an admission of fault, culpability or failure to act by or on behalf of any indemnified party, and (2) the indemnifying party confirms in writing its indemnification obligations hereunder with respect to such settlement, compromise or judgment.

(d) <u>Contribution</u>. To the extent the indemnification provided for in this <u>Section 4.5</u> is unavailable to or unenforceable an indemnified party under <u>Section 4.5(a)</u> or <u>Section 4.5(b)</u>, then (i) each indemnifying party shall contribute to the amount paid or payable by such indemnified party as a result of the losses, claims, damages, expenses or liabilities (or actions in respect thereof) referred to in <u>Section 4.5(a)</u> or <u>Section 4.5(b)</u> in such proportion as is appropriate to reflect the relative benefits received by the indemnifying party or parties on the one hand and the indemnified party on the other from the registration of the Registrable Securities pursuant to the applicable Registration Statement, or (ii) if the allocation provided by the foregoing clause (i) is not permitted by applicable Laws, in such proportion as is appropriate to reflect not

only the relative benefits referred to in clause (i) above but also the relative fault of the indemnifying party or parties on the one hand and the indemnified party on the other in connection with the statements or omissions that resulted in such losses, claims, damages or liabilities (or actions in respect thereof) as well as any other relevant equitable considerations. The relative fault of the parties shall be determined by reference to, among other things, whether the untrue or alleged untrue statement of a material fact or the omission or alleged omission to state a material fact relates to information supplied by the Company, on the one hand, or such Investor or such other indemnified party, as the case may be, on the other hand, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The amount paid by an indemnified party as a result of the losses, claims, damages or liabilities referred to in the first sentence of this Section 4.5(d) shall be deemed to include any legal or other expenses reasonably incurred by such indemnified party in connection with investigating or defending any action or claim that is the subject of this Section 4.5(d). The Company and the Investors agree that it would not be just and equitable if contribution pursuant to this Section 4.5(d) were determined by pro rata allocation or by any other method of allocation that does not take account of the equitable considerations referred to herein. Notwithstanding any other provision of this Section 4.5(d), no Investor of the Registrable Securities shall be required to contribute any amount in excess of the amount by which the gross proceeds received by such Investor from the sale of the Registrable Securities pursuant to applicable Registration Statement exceeds the amount of damages that such Investor has otherwise been required to pay by reason of such untrue or alleged untrue statement or omission or alleged omission. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation. For purposes of this Section 4.5(d), each Person, if any, who controls such indemnified party within the meaning of the Securities Act or the Exchange Act shall have the same rights to contribution as such indemnified party and each Person, if any, who controls the Company within the meaning of the Securities Act or the Exchange Act shall have the same rights to contribution as the Company. The Investors' respective obligations to contribute pursuant to this Section 4.5(d) are several in proportion to the respective amount of Registrable Securities that they have sold pursuant to a Registration Statement and not joint. The remedies provided for in this Section 4.5(d) are not exclusive and shall not limit any rights or remedies that may otherwise be available to any indemnified party at law or in equity.

- (e) <u>Survival</u>. The indemnity and contribution provisions contained in this <u>Section 4.5</u> shall remain operative and in full force and effect regardless of (i) any termination of this Agreement, (ii) any investigation made by or on behalf of any Investor or any Person controlling any Investor, or by or on behalf of the Company, its officers or Directors or any Person controlling the Company, and (iii) any sale of Registrable Securities pursuant to any Registration Statement.
- **4.6 Rule 144 Reporting.** With a view to making available the benefits of certain rules and regulations of the SEC that may at times permit the sale of Registrable Securities to the public in the United States without registration after the Filing Determination Date, the Company shall use its commercially reasonable efforts to:
 - (a) make and keep public information available, as those terms are understood and defined in Rule 144 under the Securities Act;
 - (b) file, as and when applicable, with the SEC in a timely manner all reports and other documents required of the Company under the Exchange Act; and
- (c) furnish to each Investor forthwith upon request a written statement by the Company as to its compliance with the reporting requirements of Rule 144 and the Exchange Act.
- 4.7 Lock Up. In connection with any underwritten offering of the Company's securities, each Investor and Permitted Holder of Registrable Securities agrees that upon request of the Company or the underwriters managing any underwritten offering of the Company's securities, that it will (a) not offer, sell, contract to sell, loan, grant any option to purchase, make any short sale or otherwise dispose of, hedge or transfer any of the economic interest in (or offer, agree or commit to do any of the foregoing) any shares of Common Stock, or any options or warrants to purchase any shares of Common Stock, or any securities convertible into, exchangeable for or that represent the right to receive shares of Common Stock, whether now owned or hereinafter acquired, owned directly (including holding as a custodian) or with respect to which such Person has beneficial ownership within the rules and regulations of the Commission (other than those included by such Person in the offering in question, if any) without the prior written consent of the Company or such underwriters, as the case may be, for up to fourteen (14) days prior to, and during the ninety (90) day period following, the effective date of the registration statement for such underwritten offering, and (b) enter into and be bound by such form of agreement with respect to the foregoing as the Company or such managing underwriter may reasonably request; provided that each executive officer and director of the Company also agrees to substantially similar restrictions. The restrictions set forth in this Section 4.7 shall terminate, with respect to an Investor and its Permitted Holders, when such Investor and its Permitted Holders own less than 7.5% of the outstanding Common Stock, after giving effect to such Investor's and its Permitted Holders' Preferred Stock, Series C Preferred Stock and Class B Common Stock on an as converted basis using the Base Price on the date of the request of the Company or the underwriters mentioned above of in calculating the amount of out

that, unless the SPA is earlier terminated, no such termination shall be effective until the later of (i) the Third Closing and (ii) 90 days following completion of a Third Party Financing.

ARTICLE V OTHER AGREEMENTS

- **5.1 Standstill.** From the date hereof until the later of (i) such time as it ceases to own any Securities and (ii) nine (9) months after the Investors are no longer entitled to appoint an Investor Director pursuant to the Certificate of Incorporation (such period, the "**Standstill Period**"), each Investor agrees that it shall not and shall cause its Affiliates not to:
- (a) acquire, or propose to acquire, beneficial ownership of any Securities or assets, or rights or options to acquire any Securities or assets, of the Company, including derivative securities representing the right to vote or economic benefits of any such Securities, other than pursuant to a Permitted Offer; *provided*, that the transfer of Securities among the Investors and to or from any special purpose company formed to hold the beneficial ownership of such Securities, to the extent in compliance with the transfer restrictions and procedures set forth in Section 9.1 of the SPA, shall not be deemed a violation of this Section 5.1(a), provided any such special purpose company is owned exclusively by the Investors and their controlled Affiliates;
- (b) make, or effect or commence, any tender or exchange offer, merger or other business combination involving the Company, other than pursuant to a Permitted Offer;
- (c) commence or complete, or propose to commence or complete, any recapitalization, restructuring, liquidation, dissolution or other extraordinary transaction with respect to the Company, other than pursuant to a Permitted Offer;
- (d) make, or in any way participate in, any Solicitation of proxies to vote or consent, or seek to advise or influence any Person with respect to the voting of, any Securities of the Company, or to become a Participant in any Election Contest with respect to the Company or grant a proxy to any other Person to vote any Securities held by such Investor;
- (e) form, join or in any way participate in a 13D Group with respect to, or otherwise act in concert with any Person in respect of, any Securities of the Company; *provided*, that the Investors' formation of a 13D Group among themselves and any special purpose company formed to hold the beneficial ownership of such Securities shall not be deemed a violation of this Section 5.1(e);
- (f) otherwise act, alone or in concert with others, to seek representation on or to control or influence the management, the Board or the policies of the Company, except as expressly granted pursuant to the definitive agreements for the Transaction or by the Board;
- (g) negotiate with or provide any information to any Person with respect to, or make and statement or proposal to any Person with respect to, or make any public announcement or proposal or offer with respect to, or act as a financing source for or otherwise invest in any Person in connection with, or otherwise solicit, seek or offer to effect any transactions or actions that are prohibited pursuant to this <u>Section 5.1</u>; or
 - (h) advise, assist or encourage any other Person in connection with any transactions or actions prohibited pursuant to the foregoing (a)-(g).

Notwithstanding the foregoing, nothing in this Section 5.1 shall restrict in any way the actions of any Investor Director in such person's capacity as a Director.

- **5.2** Use of Proceeds from the Third Closing The Company shall use the proceeds of the Third Closing in accordance with the funding plan and related provisions set forth in or established under the definitive agreements for the DOE Financial Closing.
- **5.3 Waiver of Appraisal and Dissenter's Rights** Each Investor shall waive any appraisal or dissenter's rights to which they may be entitled under applicable Laws in connection with any Change of Control of the Company.

5.4 Voting Agreement.

(a) At any time at which the holders of Series C Preferred Stock and Class B Common Stock shall be

entitled to vote together with the holders of Common Stock (and any other class or series of capital stock entitled to vote on the matter with the Common Stock) as a single class with respect to any transactions involving a merger of the Corporation or sale of all or substantially all of the Corporation's assets which must be submitted to the Corporation's stockholders pursuant to the Delaware General Corporation Law (a "Transaction Approval"), to the extent any Investor and its Affiliates in the aggregate beneficially own shares entitled to exercise voting power exceeding 10% of the aggregate voting power of such class ("Excess Voting Share") on such matter, the Investor agrees to vote (or exercise a written consent in favor of), and to cause any of its Affiliates to vote (or exercise a written consent in favor of), all such Excess Voting Shares on such matter as recommended by the Board to stockholders generally.

(b) Each Investor hereby grants to the Company, and each officer of the Company holding the office of Vice President or Secretary, with full power of substitution, an irrevocable proxy (the "Proxy") from the date hereof until the termination of this Agreement to act as the Investors' proxy to vote the Excess Voting Shares, and to exercise all voting, consent and similar rights of the undersigned with respect to the Excess Voting Shares (including, without limitation, the power to execute and deliver written consents), at every annual, special, adjourned or postponed meeting of the stockholders of the Company and in every written consent in lieu of such a meeting, in each case as and to the extent provided in Section 5.4(a) in connection with a Transaction Approval. Each Investor hereby affirms that the Proxy is given in connection with the Securities Purchase Agreement and this Agreement and that such Proxy is given to secure the performance of the duties of Investor under Section 5.4(a). Each Investor hereby further affirms that the Proxy is coupled with an interest and may under no circumstances be revoked. Each Investor hereby ratifies and confirms all that such Proxy may lawfully do or cause to be done by virtue hereof. Without limiting the generality of the foregoing, such Proxy is executed and intended to be irrevocable in accordance with the provisions of Section 212 of the Delaware General Corporation Law. If for any reason the Proxy granted herein is not irrevocable, Investor agrees to vote the Excess Voting Shares in accordance with Section 5.4(a) hereof and to cause any of its Affiliates to vote any Excess Voting Shares in accordance with Section 5.4(a) hereof and to cause any of its Affiliates to execute irrevocable proxies consistent with the irrevocable Proxies given by each Investor pursuant to this Section 5.4(b).

ARTICLE VI TERM

This Agreement is effective as of the date hereof. Except as otherwise expressly provided herein with respect to specific provisions of this Agreement, this Agreement may be terminated only upon the mutual written agreement of the Parties; provided, however that the rights of an Investor pursuant to Article II hereof shall terminate at any time that such Investor fails to satisfy the Minimum Equity Holdings. If this Agreement is terminated pursuant to this Article VI, all further obligations of each Party shall terminate without further liability or obligation of such Party to any other Party, including liability for damages; provided, however, that no such termination shall relieve any Party from any liability for any breach of this Agreement arising prior to the termination date; provided, further, that Section 3.6 and Section 7.4 shall survive any termination of this Agreement.

ARTICLE VII MISCELLANEOUS

- 7.1 Certain Definitions. Capitalized terms used herein, but not otherwise defined, in the body of this Agreement have the meanings set forth in Schedule 1 or, if not defined therein, the meanings set forth in the SPA.
- 7.2 Titles and Subtitles; Interpretation. The titles and subtitles used in this Agreement are used for convenience only and are not to be considered in construing or interpreting this Agreement. When a reference is made in this Agreement to an Article, Section, Schedule or Exhibit, such reference shall be to an Article, Section, Schedule or Exhibit of this Agreement unless otherwise indicated. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." The definitions contained in Schedule 1 to this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to in this Agreement means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes. Each of the Parties has participated in the drafting and negotiation of this Agreement. If an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if it is drafted by each of the Parties, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of authorship of any of the provisions of this

Agreement.

- 7.3 Reasonable Efforts. When used in this Agreement (including in any amendment to this Agreement), the terms "best efforts," "reasonable efforts," "commercially reasonable efforts" and "reasonable best efforts" shall mean efforts that are considered to be commercially reasonable under the circumstances prevailing at the time, taking into account the economic condition of the Party making such efforts, the information actually available to such Party at the time the efforts are made, and the internal resources and employees available to such Party to make such efforts. In no event shall it require a Party to undertake measures, which in its reasonable judgment, could materially jeopardize its ability to perform its other legal or contractual obligations to others (including to the other Parties) or to comply with applicable Laws or could adversely impact the licenses, permits or regulatory status of the Company or its Subsidiaries. In all cases, such efforts shall be exercised diligently and in good faith and, in the case of the Company's efforts to receive approvals that may be required under applicable Laws, consistent with its own efforts to obtain its most critical such approvals, on its own behalf, from time to time.
- **7.4 Confidentiality.** All Confidential Information (as defined in the Strategic Relationship Agreement) disclosed by a Party to another Party pursuant to this Agreement will be subject to the terms of Section 4.4 of the Strategic Relationship Agreement. For clarity, (a) each Investor Director is a Representative (as defined in the Strategic Relationship Agreement) of the Investor appointing the Investor Director and (b) nothing herein shall be deemed to reduce or waive the fiduciary duties of a director of Company.
 - 7.5 Representations and Warranties of Toshiba. Toshiba represents and warrants to the Company as follows:
 - (a) Toshiba is a corporation duly organized and validly existing under the laws of Japan.
- (b) Toshiba has all requisite corporate power, authority and capacity to execute and deliver this Agreement and to consummate the transactions contemplated to be performed by it hereby. The execution, delivery and performance by Toshiba of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of Toshiba. No approval of Toshiba's stockholders is required in connection with Toshiba's execution, delivery and performance of this Agreement and the consummation by Toshiba of the transactions contemplated hereby. This Agreement has been duly executed and delivered by Toshiba and, assuming the due authorization, execution and delivery of this Agreement by B&W and the Company, constitutes the legal, valid and binding agreement of Toshiba enforceable against it in accordance with its terms, except as may be limited by the Enforceability Exceptions.
 - **7.6 Representations and Warranties of B&W**. B&W represents and warrants to the Company as follows:
 - (a) B&W is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) B&W has all requisite corporate power, authority and capacity to execute and deliver this Agreement and to consummate the transactions contemplated to be performed by it hereby. The execution, delivery and performance by B&W of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of B&W. This Agreement has been duly executed and delivered by B&W and, assuming the due authorization, execution and delivery of this Agreement by Toshiba and the Company, constitutes the legal, valid and binding agreement of B&W enforceable against it in accordance with its terms, except as may be limited by the Enforceability Exceptions.
 - 7.7 Representations and Warranties of the Company. The Company represents and warrants to the Investors as follows:
 - (a) The Company is a corporation organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) The Company has all requisite corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated to be performed by it hereby. The execution, delivery and performance by the Company of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary corporate action on the part of the Company other than the Stockholder Approvals. This Agreement has been duly executed and delivered by the Company and, assuming the due authorization, execution and delivery of this Agreement by Investor, constitutes the legal, valid and binding agreement of the Company enforceable against it in accordance with its terms, except as may be limited by the Enforceability Exceptions.

- **7.8 Further Assurances.** Each Party shall execute and deliver such additional instruments and other documents and shall take such further actions as may be necessary or appropriate to effectuate, carry out and comply with all of its obligations under this Agreement.
- **7.9 No Assignment or Transfer.** No Party shall assign this Agreement or its rights hereunder to any Person without the written consent of each other Party; provided, that (a) each Investor may assign its rights hereunder to any permitted transferee of its Securities in accordance with Section 9.1 of the SPA, and (b) the registration rights under Article IV shall be exercisable by any Permitted Holder of Registrable Securities, including after transfers of Registrable Shares provided such Permitted Holder becomes a party to and bound by this Agreement. Except as expressly provided herein, any purported assignment by any Party shall be null and void.
- 7.10 Injunctive Relief. The Parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with its specific terms or was otherwise breached. It is accordingly agreed that the Parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in any court without the necessity of posting any bond or other security, this being in addition to any other remedy to which they are entitled at Law or in equity. Additionally, each Party hereto irrevocably waives any defenses based on adequacy of any other remedy, whether at Law or in equity, that might be asserted as a bar to the remedy of specific performance of any of the terms or provisions hereof or injunctive relief in any action brought therefor.
- **7.11 Severability.** If any provision of this Agreement, or the application thereof to any Person, place or circumstance, is held by a court of competent jurisdiction or in an arbitration under <u>Section 7.16</u> to be invalid, void or otherwise unenforceable, such provision shall be enforced to the maximum extent possible so as to effect the intent of the Parties, or, if incapable of such enforcement, shall be deemed to be deleted from this Agreement, and the remainder of this Agreement and such provisions as applied to other Persons, places and circumstances shall remain in full force and effect and, in such event, the Parties shall negotiate in good faith in an attempt to agree to another provision (in lieu of the term or application held to be invalid or unenforceable) that will be valid and enforceable and will carry out the Parties' intentions hereunder.
- **7.12 Waivers**. The waiver by a Party of a breach of or a default under any provision of this Agreement (a) shall not be effective unless such waiver is in writing, expressly states that it is a waiver hereunder, and identifies the breach or default to be waived and (b) shall not be construed as effective against or with respect to any other Party. No waiver hereunder shall, in any event, be construed as a waiver of any subsequent breach of, or default under, the same or any other provision of this Agreement, nor shall any delay or omission on the part of a Party in exercising or availing itself of any right or remedy, or any course of dealing hereunder, operate as a waiver of any right or remedy.
- **7.13 Amendments.** This Agreement may be amended only by written document, expressly stating that it is an amendment to this Agreement, identifying the provisions of this Agreement to be amended, and duly executed on behalf of each of the Parties. No delay or omission on the part of a Party in exercising or availing itself of any right or remedy, or any course of dealing hereunder, shall operate as an amendment with respect to any provision hereof.
- **7.14 Governing Law**. This Agreement is to be construed in accordance with and governed by the internal laws of the State of New York without giving effect to any choice of law rule that would cause the application of the laws of any jurisdiction other than the internal laws of the State of New York to the rights and duties of the Parties.

7.15 Conversion or Redemption Dispute.

- (a) <u>Resolution of Conversion or Redemption Dispute</u>. If a dispute between the Company and an Investor shall arise as to the appropriate Factor under Section 7 of the Series B-1 Certificate of Designation, upon a Final Determination (as defined in the Series B-1 Certificate of Designation) of the Factor, the obligations of the Parties shall be as follows:
- (i) If the Factor is determined to be 1.1, the Corporation shall, within 20 Business Days, provide to the Investor, but only in respect of Series B-1 Preferred Stock redeemed or converted prior to such payments, (i) in the case of a redemption effected pursuant to Section 7(g) of Series B-1 Certificate of Designation, subject to the provisions of the Delaware General Corporation Law, cash in an amount equal to the difference between the amount that would have been paid to such Investor had such final Factor been applied at the time of redemption and the amount previously paid to such Investor pursuant to Section 7(h) of the Series B-1 Certificate of Designation, (ii) in the case of a Conversion Election, (x) the difference between the number of shares of Class B Common Stock or Series C Preferred Stock (as applicable) as would have been issued to such Investor pursuant to Sections 7(b) of the Series B-1 Certificate of Designation had such final Factor been applied at the time of conversion and the number of shares previously issued to such Investor pursuant to Section 7(h)

of the Series B-1 Certificate of Designation, or (y) subject to the Delaware General Corporation Law, in the sole discretion of the Company, cash in an amount equal to the value of the shares as would have been issued under clause (x) based upon the Base Price as of the date of payment, or (iii) in the case of a Sale Election (as defined in the Series B-1 Certificate of Designation), an amount in cash equal to the difference between (A) the product of the final Factor and the amount calculated pursuant to clause (2)(x)(A) of Section 7.2 of the Series B-1 Certificate of Designation and (B) the amount calculated pursuant to clause (2)(x)(A) of Section 7.2 of the Series B-1 Certificate of Designation, in accordance with Section (A) of the Series B-1 Certificate of Designation.

(ii) If the Factor is determined to be 0.9, such Investor shall, within 20 Business Days, provide to the Company, but only in respect of Series B-1 Preferred Stock redeemed or converted prior to such payments, (i) in the case of a redemption effected pursuant to Section 7(g) of the Series B-1 Certificate of Designation, cash in an amount equal to the difference between the amount previously paid to such Investor pursuant to Section 7(h) of the Series B-1 Certificate of Designation and the amount that would have been paid to such Investor thereunder had such Factor been applied, (ii) in the case of a Conversion Election, either (x) cash in an amount equal to the value of the shares issuable under Clause (y) hereof (based upon the Base Price at date of payment, or (y) a number of shares of Class B Common Stock or Series C Preferred Stock (as applicable) equal to, the difference between the number of shares of Class B Common Stock or Series C Preferred Stock (as applicable) as were issued to such Investor pursuant to Section 7(h) of the Series B-1 Certificate of Designation) at such conversion and the number of shares which have been issued had such final Factor been applied at such time, or (iii) in the case of a Sale Election, an amount in cash equal to the difference between the (A) the amount previously calculated pursuant to clause (2)(x)(A) of Section 7.2 of the Series B-1 Certificate of Designation and (B) the product of the final Factor and the amount calculated pursuant to clause (2)(x)(A) of Section 7.2 of the Series B-1 Certificate of Designation; or

(iii) if the Factor is determined to be 1.0, then no transfers of cash or shares need be made.

7.16 Dispute Resolution.

- (a) Executive Meetings. Prior to submitting any dispute or controversy arising from or in connection with this Agreement, including the breach, termination or invalidity thereof (a "Dispute"), to arbitration pursuant to Section 7.16(b), upon written request of any Party, each Party shall appoint a designated representative whose task it will be to meet promptly for the purpose of endeavoring to resolve such Dispute. The designated representatives shall meet, in person or by telephone or video conference as deemed appropriate by the Parties, as often as the Parties reasonably deem necessary to discuss the Dispute in an effort to resolve the Dispute without the necessity of any further proceeding. The Parties agree to negotiate, in good faith, in an attempt to resolve the Dispute for a period of not greater than thirty (30) days after notice of the Dispute is received by the Parties.
- (b) <u>Arbitration; Rules; Location</u>. Any Dispute that is not resolved pursuant to <u>Section 7.16(a)</u> shall be referred to and finally determined under the Rules of Arbitration of the International Chamber of Commerce then in effect (the "*ICC Rules*"). The place of arbitration shall be San Francisco, California, or such other location as the Parties may agree in writing
- (c) <u>Arbitrators</u>. There shall be three (3) arbitrators, nominated in accordance with the ICC Rules. Each arbitrator on the arbitral tribunal shall be disinterested in the Dispute and shall have no connection to any Party thereto
- (d) Award. The arbitral award shall be in writing, state the reasons for the award, and be the sole and exclusive binding remedy with respect to the Dispute between and among the Parties. Judgment on the award rendered may be entered in any court having jurisdiction thereof. The Parties hereby waive any right to refer any question of law and right of appeal on the law and/or merits to any court, except as provided by the Convention on the Recognition and Enforcement of Foreign Arbitral Awards of 1958. For purposes of such convention, the award shall be deemed an award of the United States, the relationship between the Parties shall be deemed commercial in nature, and any Dispute arbitrated pursuant to this Section 7.16 shall be deemed commercial. The arbitrators shall have the authority to grant any equitable or legal remedies that would be available in any judicial proceeding intended to resolve a Dispute.
 - (e) Language of Proceedings. The language of the arbitral proceedings shall be English.
- (f) Confidentiality of Proceedings. The Parties agree that any arbitration hereunder shall be kept confidential, and that the existence of the Dispute, the proceeding and all of its elements (including any pleadings, briefs or other documents submitted or exchanged, any testimony or other oral submissions, and any awards) shall be deemed Confidential Information (as such term is used in the Strategic Relationship Agreement), and shall not be disclosed beyond the tribunal, the International Court of Arbitration, the parties to the Dispute, their counsel, and any Person necessary to the conduct of the proceeding, except as and to the extent required to enforce any arbitral award, or as otherwise contemplated by such confidentiality agreements.

- (g) Expenses. Each party hereto to a Dispute shall bear its own legal fees and costs in connection therewith.
- (h) Injunctive Relief. Notwithstanding the foregoing, any Party shall be entitled to seek preliminary injunctive relief from any court of competent jurisdiction to preserve the status quo, pending the final decision or award of the arbitrators.
- (i) Consent to Jurisdiction. Each of the Parties hereby irrevocably consents to jurisdiction of any court State or Federal in the United States of America (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding pursuant to Section 7.16(h) or enforcing any award under Section 7.16(d), and irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum. Process in any such action or proceeding may be served on any Party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party by notice as provided in Section 7.20 shall be deemed effective service of process on such Party.
- (j) Concurrent Disputes. To the extent of concurrent Disputes (as such term is used in each of the Transaction Documents) under multiple Transaction Documents, the Parties agree to consolidate any and all such Disputes into a single proceeding pursuant to the procedures set forth in this Section 7.16.
- 7.17 Limitations on Damages. In no event shall any Party have any liability for loss of profits, revenue or goodwill, loss or interruption of business, loss of data, or for any indirect, incidental, special, consequential or punitive damages, arising out of or relating to this Agreement or the subject matter hereof, no matter what theory of liability, and even if advised of the possibility or probability of such damages.
- 7.18 Independent Contractors. Each Party is an independent contractor and no Party's personnel are employees or agents of any other Party for federal, state or other taxes or any other purposes whatsoever, and are not entitled to compensation or benefits of the other. Nothing hereunder shall be deemed to constitute, create, give effect to or otherwise recognize a joint venture, partnership or business entity of any kind, nor shall anything in this Agreement be deemed to constitute any Party the agent or representative of any other Party.
- 7.19 No Third-Party Beneficiaries. Notwithstanding anything contained in this Agreement to the contrary, nothing in this Agreement, expressed or implied, is intended to confer on any Person other than the Parties hereto any rights, remedies, obligations or liabilities under or by reason of this Agreement, and no Person that is not a Party to this Agreement (including any partner, member, stockholder, director, officer, employee or other beneficial owner of either Party, in its own capacity as such or in bringing a derivative action on behalf of a Party hereto) shall have any standing as third party beneficiary with respect to this Agreement or the transactions contemplated hereby.
- 7.20 Notices. All notices and other communications under this Agreement shall be in writing and shall be deemed given (a) when delivered by hand or upon confirmed receipt of a facsimile transmission, (b) when received if sent by an internationally recognized overnight courier service (receipt requested), or (c) ten (10) Business Days after mailing, postage prepaid, by registered or certified mail, return receipt requested, to the below address or such other addresses as a Party shall specify in a written notice to the other Parties provided as contemplated herein.

To Toshiba: To B&W:

Toshiba Corporation 1-1 Shibaura 1-chome Minato-ku, Tokyo 105-8001 JAPAN Attn: General Manager Legal Affairs Division, Power Systems Company Fax: +81 3 5444 9183

With a copy (which shall not constitute notice) to:

Morrison & Foerster Shin-Marunouchi Building, 29th floor Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502 U.S.A. Attn: James D. Canafax, Esq. Fax: +1 434 522 6793

With a copy (which shall not

Baker Botts The Warner

constitute notice) to:

5-1, Marunouchi 1-chome Chiyoda-ku, Tokyo 100-6529 JAPAN

Attn: Ken Siegel, Esq. Fax: +81 3 3214 6512

To the Company:

USEC Inc. 2 Democracy Center 6903 Rockledge Drive Bethesda, MD 20817 U.S.A.

Attn: General Counsel Fax: +1 301 564 3206

With a copy (which shall not constitute notice) to:

Latham & Watkins 555 11th Street, N.W. Washington, DC 20004 U.S.A. Attn: Scott C. Herlihy, Esq. Fax: +1 202 637 2201 1299 Pennsylvania Avenue, NW Washington, DC 20004-2400 U.S.A. Attn: Michael Gold, Esq. Fax: +1 202 585 1024

7.21 Entire Agreement. This Agreement and the other Transaction Documents (including the schedule and exhibits attached hereto and thereto, which are incorporated herein by reference) constitute the entire agreement among the Parties with respect to its subject matter. This Agreement and the other Transaction Documents supersede all previous, contemporaneous and inconsistent agreements, negotiations, representations and promises between among the Parties, written or oral, regarding the subject matter hereof. There are no oral or written collateral representations, agreements or understandings except as provided herein and in the other Transaction Documents.

7.22 Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be binding as of the date first written above, and all of which shall constitute one and the same instrument. Each such counterpart shall be deemed an original, and it shall not be necessary in making proof of this Agreement to produce or account for more than one such counterpart. The exchange of copies of this Agreement and of signature pages by facsimile transmission, portable document format (.pdf) or other electronic format shall be deemed to be their original signatures for all purposes.

[Remainder of page left intentionally blank.]

Name: Title:

Signature Page to the Investor Rights Agreement

SCHEDULE 1

DEFINITIONS

Unless otherwise defined herein, terms used herein that are defined in the SPA shall have the meanings set forth for such terms in the SPA.

- "13D Group" means any partnership, limited partnership, syndicate or other group, as those terms are used within the meaning of Section 13(d)(3) of the Exchange Act.
- "ACP" means the design, manufacture, construction, development, startup, completion, operation, financing, maintenance and improvement of a front-end nuclear fuel facility utilizing U.S. gas centrifuge enrichment technology and related infrastructure assets and properties.
 - "Atomic Energy Act" means the Atomic Energy Act of 1954, as amended (42 U.S.C. 2011 et seq.).
 - "Beneficial Owner" means any Person who has beneficial ownership.
 - "Base Price" has the meaning ascribed to it in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable.
- "beneficial ownership" shall mean any ownership that would be determined to be "beneficial ownership" pursuant to Rule 13d-3 promulgated pursuant to the Exchange Act.
 - "Board" means the board of directors of the Company or any duly authorized committee thereof.
- "Business Day" means any calendar day other than (i) a Saturday or Sunday or (ii) a calendar day on which banking institutions in either the City of New York or Tokyo, Japan are authorized by Law, regulation or executive order to remain closed.
 - "Certificate of Incorporation" means the Company's Certificate of Incorporation, as amended, from time to time.
- "CFIUS" means the Committee on Foreign Investment in the United States of the United States Treasury Department and any successor Governmental Authority thereto.
 - "Change of Control" shall mean the occurrence of any of the following:
- (a) any Person shall Beneficially Own, directly or indirectly, through a merger, business combination, purchase, or other transaction or series of transactions, shares of the Corporation's capital stock entitling such Person at such time to exercise 50% or more of the total voting power of the issued and outstanding capital stock having ordinary voting power to elect a majority of the Board of Directors, other than as a result of an acquisition of such stock by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this subsection (1), "Person" shall include any syndicate or group that would be deemed to be a "person" under Section 13(d)(3) of the Exchange Act);
- (b) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Stock is exchanged for or converted into cash, securities or other property, in each case other than a merger or consolidation:
 - (i) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Stock;
- (ii) that is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding shares of the Common Stock solely into shares of any class or series of common stock of the surviving entity; or
 - (iii) there the issued and outstanding capital stock having voting power to vote generally to elect a

majority of the Board of Directors outstanding immediately prior to such transaction is converted into or exchanged for such voting stock of the surviving or transferee Person constituting a majority of the outstanding shares of such voting stock of such surviving or transferee Person (immediately after giving effect to such issuance).

- "Class B Common Stock" means shares of the class B common stock of the Company, par value \$0.10 per share to be authorized by the Charter Amendment.
- "Classified Information" means (i) information classified as either Restricted Data or Formerly Restricted Data or (ii) National Security Information.
- "Closings" means the First Closing, the Second Closing and the Third Closing, each as defined in the SPA.
- "Common Stock" means collectively, the Ordinary Common Stock and the Class B Common Stock.
- "Competitively Sensitive Information" means any non-public information the Company possesses relating to (i) the prices, quantities and other terms of sale on which the Company supplies or offers to supply products or services to Toshiba's competitors; and (ii) the prices, quantities and other terms of service on which a Toshiba competitor or B&W competitor supplies or offers to supply products or services to a customer, including the Company.
- "Consent" means any consent, approval, authorization, waiver, permit, grant, franchise, concession, agreement, license, certificate, exemption, order, registration, declaration, filing, report or notice of, with or to any Person.
 - "Conversion Election" has the meaning ascribed to it in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable.
- "Deemed Holder Percentage" shall mean, as to any Permitted Holder, the percentage resulting from the following calculation, (1)(A) the number of shares of Ordinary Common Stock equal to the quotient of (w) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on the outstanding shares of
- Series B Preferred Stock held by such Permitted Holder from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (x) the Base Price for the date of such calculation, plus (B) the number of outstanding of shares of (y) Series C Preferred Stock multiplied by 1000 plus, (z) if then outstanding, Class B Common Stock, in each case held by such Permitted Holder divided by (2)(A) the total number of shares of Ordinary Common Stock equal to the quotient of (v) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on all outstanding shares of Series B Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (w) the Base Price for the date of such calculation, plus (B) the total number of all outstanding shares of (x) Series C Preferred Stock multiplied by 1000 plus (y) if then outstanding, Class B Common Stock, plus (z) Ordinary Common Stock.
 - "Director" means a member of the Board.
 - "Dividends" has the meaning ascribed to it in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable.
 - "DOE" means the U.S. Department of Energy and any successor Governmental Authority thereto.
- "Effective Time" means the time at which the SEC declares any Registration Statement effective or at which any Registration Statement otherwise becomes effective.
 - "Election Contest" means solicitations subject to Rule 14a-12(c) promulgated under the Exchange Act.
- "Equity Interest" means, as to each Investor at any time, a percentage represented by a fraction, the numerator of which is the number of shares of preferred stock then held by the Investor and the denominator of which is the total number of shares of preferred stock of the Company then outstanding.
 - "Excess Voting Share" has the meaning ascribed to it in Section 5.4(a).
 - "Exchange Act" mean the United States Securities Exchange Act of 1934, as amended.
- "Excluded Lender" shall mean a bank or other financial institution providing indebtedness for borrowed money which is guaranteed by the Loan Guarantee Agreement (as defined in 10 CFR 609.2) pertaining to the DOE Financial

Closing; provided, however "Excluded Lender" shall not include a Person providing funding or committed funding (pursuant to definitive binding agreements) for debt or equity of the Corporation in an amount of at least \$100,000,000 that is not guaranteed by such Loan Guarantee Agreement.

"Export Controlled Information" means any unclassified information, the export of which is controlled by law or regulation under the International Traffic in Arms Regulations (22 CFR Part 120 et seq.), the Export Administration Regulations (14 CFR Part 730 et seq.), or the U.S. Department of Energy regulations (10 CFR Part 810), and including Unclassified Controlled Nuclear Information, Safeguards Information (Section 147 of the Atomic Energy Act of 1954 as amended), Sensitive Unclassified Non-Safeguards Information (NRC Regulatory Information Summary 2005-31), and Official Use Only information (DOE Order 471.3), except to the extent that the export or deemed export thereof arising by virtue of a disclosure hereunder has been licensed or approved under, or is authorized by, Applicable Law.

"Factor" has the meaning ascribed to it in the Series B-1 Certificate of Designation.

"Filing Date" means the earlier of (i) April 30, 2011, if the Second Closing has not occurred prior to that date, (ii) October 30, 2011, if the Third Closing has not occurred prior to that date and (iii) the date of the Third Closing.

"FINSA" means the Foreign Investment and National Security Act of 2007, 50 U.S.C. App. 2061, amending the Defense Production Act of 1950, 50 U.S.C. App. 2170.

"Formerly Restricted Data" means classified information jointly determined by DOE and the Department of Defense to be related primarily to the military utilization of nuclear weapons and removed (by transclassification) from the Restricted Data category pursuant to section 142(d) of the Atomic Energy Act.

"Governmental Approval" means any Consent of any Governmental Authority.

"Initial Liquidation Preference" means \$1,000.00 per share of Series B Preferred Stock.

"Investor Director" means a Director elected by the holders of Preferred Stock or the Class B Common Stock and the Series C Preferred Stock, as applicable, pursuant to the Certificate of Incorporation.

"Liquidation Preference" has the meaning ascribed to it in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable.

"Mitigation Agreement" means an agreement or other undertaking entered into with CFIUS or one or more of its members to address any national security concerns raised by CFIUS in connection with the transactions contemplated hereby.

"National Security Information" means information that has been determined pursuant to Executive Order 12958, as amended (68 Federal Register 15315 (March 28, 2003)), or prior Executive Orders to require protection against unauthorized disclosure and is marked to indicate its classification status when in document form. National Security Information is referred to as 'defense information' in the Atomic Energy Act.

"Negation Plan" has the meaning set forth in the SPA.

"NISPOM" means the National Industrial Security Program Operating Manual as required by Executive Order 12829 and under the authority of Department of Defense Directive 5220.22, "National Industrial Security Program (NISP)" for the protection of classified information released or disclosed to industry in connection with classified contracts under the NISP.

"NRC" means the U.S. Nuclear Regulatory Commission and any successor Governmental Authority thereto.

"Nuclear and National Security Approvals" means approvals, licenses, permits, or other authorizations required: (i) from the NRC under of the Atomic Energy Act, 10 CFR Part 70 and 10 CFR Part 76, including any applicable Negation Plans; (ii) from the NRC and DOE pursuant to the NISPOM, including any applicable Negation Plans; and (iii) from DOE regarding access to Restricted Data pursuant to 10 CFR Part 725.

"Orderly Sale Arrangement" has the meaning set forth in the SPA.

"Ordinary Common Stock" means the shares of the common stock of the Company, par value \$0.10 per share.

"Participant" has the meaning set forth in Instruction 3 to Item 4 of Schedule 14A promulgated under the Exchange

Act.

- "Permitted Holder" means (i) Toshiba and its controlled Affiliates and (ii) B&W and its controlled Affiliates.
- "Permitted Offer" means (a) an offer made to all of the holders of Common Stock in cash for any or all of the outstanding Common Stock that has been approved by the Directors of the Company other than the Investor Directors or (b) a private offer to acquire all of the Company's Common Stock made to the Board in response to a third party's offer to acquire a majority of the Company's equity securities or an indication by the Board to the Investors that they would be interested in receiving such an offer. The transaction proposed in a Permitted Offer must be approved or accepted by the holders of a majority of the outstanding shares of Ordinary Common Stock.
 - "PIK Shares" means the shares of Preferred Stock paid as in-kind dividends on Preferred Stock.
 - "Preferred Stock" has the meaning set forth in the recitals hereto.
- "Prospectus" means the prospectus included in any Registration Statement, as amended or supplemented by any prospectus supplement with respect to the terms of the offering of any portion of the Registrable Securities covered by any Registration Statement and by all other amendments and supplements to such prospectus, including all material incorporated by reference in such prospectus and all documents filed after the date of such prospectus by the Company under the Exchange Act and incorporated by reference therein.
 - "Proxy" has the meaning ascribed to it in Section 5.4(b).
- "Registrable Securities" means, at any time, all of the then issued and outstanding, or issuable, as the case may be, (a) shares of Common Stock issued or issuable to the Investors upon conversion of (x) the shares of Preferred Stock purchased by the Investors pursuant to the SPA or (y) any shares of Series B Common Stock, (b) all other shares of Common Stock issued or issuable to the Investors upon exercise of Warrants, (c) shares of any class of capital stock or other securities into which or for which any such shares of Common Stock shall have been converted or exchanged pursuant to any recapitalization, reorganization, merger or consolidation of the Company or sale of all or substantially all of the assets of the Company and (d) shares of capital stock issued with respect to the foregoing pursuant to a stock split or stock dividend. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when (i) they have been distributed to the public pursuant to an offering registered under the Securities Act (ii) they have been sold to the public through a broker, dealer or market maker in compliance with Rule 144 under the Securities Act (or any similar rule then in force) or pursuant to the Orderly Sale Arrangement.
 - "Registration Statements" means the Initial Shelf and any subsequent Shelf Registration Statement registering the Registrable Securities.
- "Restricted Data" means a kind of classified information that consists of all data concerning the following, but not including data declassified or removed from the Restricted Data category pursuant to section 142 of the Atomic Energy Act: (i) design, manufacture, or utilization of atomic weapons; (ii) production of special nuclear material; or (iii) use of special nuclear material in the production of energy.
 - "SEC" means the United States Securities and Exchange Commission.
- "Securities" means any shares of Common Stock or preferred stock of the Company and any securities that are convertible into, or any option or right to subscribe for or acquire, any shares of Common Stock or preferred stock of the Company.
 - "Securities Act" means the Securities Act of 1933 as amended.
 - "Series B-1 Certificate of Designation" means the Company's Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock.
- "Series B-2 Certificate of Designation" shall mean the Company's Certificate of Designation of series of preferred stock of Series B-2 11.5% Convertible Preferred Stock.
 - "Series C Certificate of Designation" shall mean the Certificate of Designation of the Series C Participating Convertible Preferred Stock.
 - "Shelf Registration Statement" means a "shelf" registration statement filed under the Securities Act providing for

the registration of, and the sale on a continuous or delayed basis by the Investors of, all of the Registrable Securities pursuant to Rule 415 under the Securities Act and/or any similar rule that may be adopted by the SEC, filed by the Company pursuant to the provisions of Article IV of this Agreement, including the Prospectus contained therein, any amendments and supplements to such registration statement, including post-effective amendments, and all exhibits and all material incorporated by reference in such registration statement, and any additional "shelf" registration statements filed under the Securities Act to permit the registration and sale of Registrable Securities pursuant to Article IV.

"Solicitation" has the meaning set forth in Regulation 14A promulgated under the Exchange Act.

"Subsidiary" of any Person shall mean any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than fifty percent (50%) of (a) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (b) the interest in the capital or profits of such partnership, joint venture or limited liability company or (c) the beneficial interest in such trust or estate is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries.

Notwithstanding the foregoing, American Centrifuge Manufacturing, LLC shall not be considered a Subsidiary of B&W or the Company.

"Third Party Financing" means the funding or committed funding (pursuant to definitive binding agreements) for debt or equity of the Company in an amount of at least \$100,000,000 from a third party that is not an Affiliate of the Company, a Japanese export credit agency, a U.S. Governmental Authority or an Excluded Lender where (1) such funds, together with such other additional funds available to the Company at such time, is necessary and sufficient to consummate the DOE Financial Closing, and (2) the third-party requires, as a condition to the funding, that the Preferred Stock be converted in accordance with the terms hereof.

"Transaction Documents" means this Agreement, the Certificates of Designation, the Charter Amendment, the Investor Rights Agreement, and the Strategic Relationship Agreement.

"Unclassified Controlled Nuclear Information" means certain unclassified Government information concerning nuclear facilities, materials, weapons, and components whose dissemination is controlled under section 148 of the Atomic Energy Act and 10 CFR 1017.

EXHIBIT G

FORM OF FOREIGN OWNERSHIP RESTRICTIONS DETERMINATION

EXHIBIT G

Form of Foreign Ownership Restrictions Determination

WHEREAS, as of the date hereof, Toshiba Corporation ("Toshiba") and Babcock & Wilcox Investment Company ("B&W"), or their ultimate corporate parents, constitute foreign persons:

WHEREAS, in connection with the Transactions as defined in that certain Securities Purchase Agreement, to be entered into by and among the Corporation, Toshiba and B&W (the "SPA"; capitalized terms used and not defined herein have the meanings ascribed thereto in the SPA), Toshiba and B&W will acquire beneficial ownership of equity securities of the Corporation that entitle Toshiba and B&W to control the appointment and tenure of two directors on the Corporation's Board, as well as certain other rights, all as set forth in the SPA and the other Transaction Documents, and which, in the aggregate, shall constitute a "Foreign Ownership Review Event" under Section B of Article ELEVENTH of the Certificate of Incorporation of the Corporation;

WHEREAS, pursuant to Section I of Article ELEVENTH of the Corporation's Certificate of Incorporation, the Board shall have the exclusive right to interpret all issues arising under such Article ELEVENTH, including determining whether an "Adverse Regulatory Occurrence", as defined in Section J of such Article ELEVENTH, or a Foreign Ownership Review Event has occurred;

WHEREAS, Toshiba and B&W are only prepared to execute the Transaction Documents if the Board shall have determined that, upon the consummation of the Transactions contemplated under the Transaction Documents in accordance with their terms and conditions, the acquisition of beneficial ownership by Toshiba and B&W of equity securities of the Corporation, director appointment rights and other rights specified therein, on the terms and subject to the conditions of the Transaction Documents, will not constitute an Adverse Regulatory Occurrence; and

WHEREAS, the Corporation has, under the direction and oversight of the Board, participated in extensive negotiations with Toshiba and B&W, and the Board has received all of the information from Toshiba and B&W that it requires in order to make a fully-informed decision with respect to the Transactions contemplated in the Transaction Documents, and Toshiba and B&W's proposed investment in the Corporation thereunder.

NOW, THEREFORE, BE IT RESOLVED, that the Board hereby determines that (i) the consummation of the Transactions, on the terms and subject to the conditions set forth in the Transaction Documents, including, without limitation, the acquisition by Toshiba and B&W as they and their ownership are currently constituted (or, in each case, any permitted transferee pursuant to Section 9.1 of the SPA) of beneficial ownership, or continued beneficial ownership, of certain equity securities of the Corporation, Board appointment rights and other rights, all as set forth in the Transaction Documents, and the exercise of such rights in accordance with the Transaction Documents by Toshiba or B&W, as they and their ownership are currently constituted (or any permitted transferee pursuant to Section 9.1 of the SPA), will not constitute an Adverse Regulatory Occurrence, and (ii) the Corporation shall not request, and neither Toshiba, B&W nor any permitted transferee pursuant to Section 9.1 of the SPA shall be under any obligation to furnish, any information pursuant to Section C of Article ELEVENTH of the Certificate of Incorporation of the Corporation.

EXHIBIT H-1

FORM OF OPINION OF USEC INC. INTERNAL COUNSEL

 Peter B. Saba
 (301) 564-3327 phone

 General Counsel
 (301) 564-3205 fax

[date], 2010

Toshiba Corporation 1-1 Shibaura 1-chome Minato-ku, Tokyo 105-8001 Japan

And

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502

Re: Securities Purchase Agreement

Ladies and Gentlemen:

I am the General Counsel of USEC Inc., a Delaware corporation (the "Company"). This opinion is being delivered to you pursuant to Section 4.2(k) of that certain Securities Purchase Agreement dated as of May 25, 2010 among the Company, Toshiba Corporation and Babcock & Wilcox Investment Company (the "Securities Purchase Agreement") in connection with the contemplated issuance and sale of 125,000 shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, 75,000 shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share, and Warrants to purchase 12,500,000 shares of the Company's Class B Common Stock, par value \$0.10 per share, to be authorized pursuant to the Charter Amendment, and/or the Company's Series C Preferred Stock, par value \$1.00 per share, pursuant to the Securities Purchase Agreement. Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings ascribed to them in the Securities Purchase Agreement.

In rendering this opinion, I or attorneys on my staff have examined originals or copies (including facsimile transmissions), certified or otherwise identified to us as true and correct copies of originals, of the following documents:

- (1) the Securities Purchase Agreement;
- (2) the Investor Rights Agreement dated as of May , 2010 between the Company and you (the 'Investor Rights Agreement');
- (3) the Strategic Relationship Agreement dated as of May 25, 2010 between the Company and you (the "Strategic Relationship Agreement," and together with the Securities Purchase Agreement and Investor Rights Agreement, the "Transaction Agreements");
- (4) the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on June 29, 1998, the Certificate of Amendment to the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2008, the Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2001 and the Certificate of Increase of Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on September 21, 2007 (collectively, the "Certificate of Incorporation") and the Amended and Restated Bylaws of the Company in effect as of May , 2010 ("Bylaws," and together with the Certificate of Incorporation, the "Governing Documents");

- (5) the Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [];
- (6) the Certificate of Designation of Series B-2 11.5% Convertible Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [];
- (7) the Certificate of Designation of Series C Convertible Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [];
- (8) the form of Certificate of Amendment to the Certificate of Incorporation of USEC Inc. attached to the Securities Purchase Agreement as Exhibit L (the "Charter Amendment");
- (9) the Warrant to Purchase Class B Common Stock or Series C Preferred Stock of USEC Inc. (the "Warrant"); and
- (10) such other documents as we have deemed necessary or appropriate as a basis for the opinions set forth below.

In our examination, we have assumed the legal capacity of all natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as certified, conformed, facsimile or photostatic copies, and the authenticity of the originals of such copies. In making our examination of documents executed or to be executed by parties other than the Company (each such party, an "Other Party"), we have assumed that such parties had the power, corporate or other, to enter into and perform all obligations thereunder and have also assumed the due authorization by all requisite action, corporate or other, and execution and delivery by such parties of such documents and that such documents constitute the valid and binding obligations of such parties. As to any facts material to the opinions expressed herein which we did not independently establish or verify, we have relied upon oral or written statements and representations of officers and other representatives and of public officials, including the facts and conclusions set forth therein.

I am admitted to practice in the State of New York and the District of Columbia. This opinion is limited to matters arising under the laws of the State of New York, the General Corporation Law of the State of Delaware and the federal laws of the United States of America. I do not express any opinion as to the laws of any other jurisdiction or as to the municipal laws or the laws, rules or regulations of any local agencies or governmental authorities of or within the State of New York, or in each case as to any matters arising thereunder or relating thereto. In giving the opinions in paragraph 1, I have assumed with your permission that to the extent any indenture, mortgage, lease, contract, agreement or instrument referenced therein is governed by laws other than the laws of the State of New York that such indenture, mortgage, lease, contract, agreement or instrument would be interpreted in the same manner as if they were governed by the laws of the State of New York.

Based upon the foregoing and subject to the limitations, qualifications, exceptions and assumptions set forth herein, I am of the opinion that:

- 1. The execution and delivery by the Company of each of the Transaction Agreements, the performance by the Company of its obligations thereunder and the issuance and sale at the First Closing of shares of Series B-1 12.75% Preferred Stock and the Warrants pursuant to the Securities Purchase Agreement by the Company on the date hereof do not constitute a violation of, or a default under any indenture, mortgage, lease, contract, agreement, instrument or other arrangement by which the Company is bound and listed as an exhibit to the Company's most recent (x) Annual Report on Form 10-K or (y) Quarterly Report on Form 10-Q filed with the United States Securities and Exchange Commission (collectively, the "SEC Exhibits"), except for violations or defaults which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- 2. The authorized capital stock of the Company consists of 250,000,000 shares of common stock, par value \$0.10 per share and 25,000,000 shares of preferred stock, par value \$1.00 per share, of which (i) shares are designated Series A Preferred Stock, (ii) shares are designated Series B-1 Preferred Stock, (iii) shares are designated Series B-2 Preferred Stock, and (iv) shares are designated Series C Preferred Stock.
 - 3. The Foreign Ownership Restrictions Determination has been duly adopted.
- 4. To my knowledge, there is no action, litigation or proceeding pending or threatened in writing by or before any Governmental Authority which is reasonably likely to materially and adversely affect the ability of the Company to perform its obligations under the Transaction Agreements or which seeks to prevent the consummation of the transactions contemplated by the Transaction Agreements.

5. The issuance and sale of the shares of Series B-1 Preferred Stock and the Warrants by the Company at the First Closing, and the issuance of shares of Class B Common Stock or Series C Preferred Stock issuable upon conversion of the Series B-1 Preferred Stock issued in the First Closing or the Series C Preferred Stock and upon the exercise of the Warrants issued in the First Closing are free of preemptive or similar rights arising under the SEC Exhibits.

This opinion is given as of the date hereof and I assume no obligation to update or supplement this opinion to reflect any facts or circumstances, which may hereafter come to my attention or any changes in laws which may hereafter occur. This opinion is solely for your benefit in connection with the above-referenced transaction and it may not be relied upon by any other person, and it may not be used for any other purpose, without my prior written consent, which may be granted or withheld in my sole discretion.

Very truly yours,

EXHIBIT H-2

FORM OF OPINION OF LATHAM & WATKINS LLP

[L&W Letterhead]

[Date]

To the Investors party to the Securities Purchase Agreement dated as of May 25, 2010, among USEC Inc. and such Investors

Re: Securities Purchase Agreement

Ladies and Gentlemen:

We have acted as special counsel to USEC Inc., a Delaware corporation (the "Company"), in connection with the contemplated issuance and sale of 125,000 shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, 75,000 shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share, and Warrants to purchase 12,500,000 shares of the Company's Class B Common Stock, par value \$0.10 per share, to be authorized pursuant to the Charter Amendment, and/or the Company's Series C Convertible Participating Preferred Stock, par value \$1.00 per share, pursuant to that certain Securities Purchase Agreement dated as of May 25, 2010 (the "Securities Purchase Agreement") among the Company and you. This letter is furnished pursuant to Section 4.2(k) of the Securities Purchase Agreement. Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings ascribed to them in the Securities Purchase Agreement.

As such counsel, we have examined such matters of fact and questions of law as we have considered appropriate for purposes of this letter, except where a specified fact confirmation procedure is stated to have been performed (in which case we have, with your consent, performed the stated procedure). We have examined, among other things, the following:

- (a) The Securities Purchase Agreement;
- (b) The Investor Rights Agreement dated as of , 2010 between the Company and you (the 'Investor Rights Agreement');
- (c) The Strategic Relationship Agreement dated as of May 25, 2010 between the Company and you (the 'Strategic Relationship Agreement," and together with the Securities Purchase Agreement and Investor Rights Agreement, the "Transaction Agreements");
- (d) The Rights Agreement dated as of April 24, 2001, by and between the Company, a Delaware corporation and Fleet National Bank, a [•] as the Rights Agreement (the "Rights Agreement") and the First Amendment to Rights Agreement dated May [•], 2010 by and between the Company and [•] (f/k/a) Fleet National Bank as the successor Rights Agent (the "Rights Agreement Amendment");
- (e) The Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on June 29, 1998, the Certificate of Amendment to the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2008, the Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2001 and the Certificate of Increase of Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on September 21, 2007 (collectively, the "Certificate of Incorporation" and the Amended and Restated Bylaws of the Company in effect as of May , 2010 ("Bylaws," and together with the Certificate of Incorporation, the "Governing Documents") and certain resolutions of the Board of Directors of the Company and certain committees thereof; and
- (f) The form of Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Company attached to the Securities Purchase Agreement as Exhibit B proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] to establish the designation of the Company's Series B-1 12.75% Preferred Stock ("Series B-1 Certificate of Designation").

(g) The form of Certificate of Designation of Series C Convertible Participating Preferred Stock of the Company attached to the Securities Purchase Agreement as Exhibit D proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] to establish the designation of the Company's Series C Convertible Participating Preferred Stock ("Series C Certificate of Designation").²

Except as otherwise stated herein, as to factual matters we have, with your consent, relied upon the foregoing, and upon oral and written statements and representations of officers and other representatives of the Company and others, including the representations and warranties of the Company in the Securities Purchase Agreement. We have not independently verified such factual matters.

Except as otherwise stated herein, we are opining herein as to the effect on the subject transaction only of the federal laws of the United States, the internal laws of the State of New York, and in paragraphs 1, 2, 4(i), 4(ii), 4(iii), 5, 6, 7 and 8 of this letter, the Delaware General Corporation Law, and we express no opinion with respect to the applicability thereto, or the effect thereon, of the laws of any other jurisdiction or, in the case of Delaware, any other laws, or as to any matters of municipal law or the laws of any local agencies within any state. Except as otherwise stated herein, our opinions herein are based upon our consideration of only those statutes, rules and regulations which, in our experience, are normally applicable to the sales of shares of convertible preferred stock and warrants in a private placement. Various issues pertaining to [] are addressed in the opinions of the Company's General Counsel and Morgan, Lewis & Bockius LLP, separately provided to you. We express no opinion with respect to those matters, and to the extent elements of those opinions are necessary to the conclusions expressed herein, we have, with your consent, assumed such matters.

Subject to the foregoing and the other matters set forth herein, as of the date hereof?

- 1. The Company is a corporation under the Delaware General Corporation Law, with corporate power and authority to enter into the Transaction Agreements and the Rights Agreement Amendment and perform its obligations thereunder. With your consent, based solely on certificates from public officials, we confirm that the Company is validly existing and in good standing under the laws of the State of Delaware.
- 2. The execution, delivery and performance of the Transaction Agreements and the Rights Agreement Amendment have been duly authorized by all necessary corporate action of the Company, and the Transaction Agreements have been duly executed and delivered by the Company.
- 3. Each of the Transaction Agreements constitutes a legally valid and binding obligation of the Company, enforceable against the Company in accordance with its terms.
- 4. The execution and delivery of the Transaction Agreements by the Company, the performance by the Company of its obligations thereunder and the issuance and sale by the Company pursuant to the Securities Purchase Agreement at the First Closing of shares of Series B-1 12.75% Preferred Stock (the "First Closing Preferred Shares")² and the Warrants (the "First Closing Warrants")^{10,11} on the date hereof do not:
 - (i) violate the provisions of the Governing Documents;
 - (ii) violate any United States federal or New York statute, rule or regulation applicable to the Company or the Delaware General Corporation Law; or
 - (iii) require any consents, approvals or authorizations to be obtained by the Company from, or any registrations, declarations or filings to be made by the Company with, any governmental authority, under any United States federal or New York statute, rule or regulation applicable to the Company or the Delaware General Corporation Law on or prior to the date hereof that have not been obtained or made.
- 5. The First Closing Preferred Shares have been duly authorized by all necessary corporate action of the Company, and, when issued to and paid for by you in accordance with the terms of the Securities Purchase Agreement and the Series B-1 Certificate of Designation, will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.
- 6. The shares of the Series C Preferred Stock¹² initially issuable upon conversion of the First Closing Preferred Shares or upon exercise of the First Closing Warrants have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, when issued to and, in the case of shares issuable upon exercise of the First Closing Warrants, paid for by, you in accordance with the terms of the Securities Purchase Agreement, the Series B-1 Certificate of Designation (in the case of shares of Series C Preferred Stock issuable upon conversion of First Closing Preferred Shares), the First Closing Warrants (in the case of shares of Series C Preferred Stock issuable upon exercise of the

First Closing Warrants) and the Series C Certificate of Designation, will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.

- 7. The shares of the Company's Ordinary Common Stock initially issuable upon conversion of the First Closing Preferred Shares have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, when issued in accordance with the terms of the Securities Purchase Agreement and the Series B-1 Certificate of Designation will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents. 13
- 8. The shares of the Company's Ordinary Common Stock initially issuable upon conversion of the shares of Series C Preferred Stock which are initially issuable upon conversion of First Closing Preferred Shares or upon exercise of the First Closing Warrants, have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, assuming that such shares of Series C Preferred Stock have been validly issued, when such shares of the Company's Ordinary Common Stock are issued in accordance with the terms of the Securities Purchase Agreement and the Series C Certificate of Designation, such shares of Ordinary Common Stock will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.
- 9. Assuming the representations and warranties of the parties thereto in the Transaction Agreements are true and assuming compliance by the parties thereto with their respective covenants and agreements set forth therein, no registration of the First Closing Preferred Shares or the First Closing Warrants under the Securities Act of 1933, as amended, is required for the purchase by you of such First Closing Preferred Shares or First Closing Warrants in the manner contemplated by the Securities Purchase Agreement. We express no opinion, however, as to when or under what circumstances any First Closing Preferred Shares or First Closing Warrants, or any securities issuable upon conversion of the First Closing Preferred Shares or upon exercise of the First Closing Warrants, may be reoffered or resold.

Our opinions are subject to:

- (a) the effect of bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or other similar laws relating to or affecting the rights or remedies of creditors;
- (b) the effects of general principles of equity, whether considered in a proceeding in equity or at law (including the possible unavailability of specific performance or injunctive relief), concepts of materiality, reasonableness, good faith and fair dealing, and the discretion of the court before which a proceeding is brought;
- (c) the invalidity under certain circumstances under law or court decisions of provisions for the indemnification of or contribution to a party with respect to a liability where such indemnification or contribution is contrary to public policy; and
- (d) we express no opinion with respect to (i) consents to, or restrictions upon, governing law or jurisdiction, venue, arbitration, remedies or judicial relief; (ii) advance waivers of claims, defenses, rights granted by law, or notice, opportunity for hearing, evidentiary requirements, statutes of limitation, trial by jury or at law, or other procedural rights; (iii) waivers of broadly or vaguely stated rights; (iv) covenants not to compete; (v) provisions for exclusivity, election, cumulation or limitation of rights or remedies; (vi) provisions authorizing or validating conclusive or discretionary determinations; (vii) grants of setoff rights; (viii) provisions for the payment of attorneys' fees where such payment is contrary to law or public policy; (ix) proxies, powers of attorney, and voting agreements or trusts; (x) provisions prohibiting, restricting or requiring consent to assignment or transfer of any right or property; (xi) provisions for liquidated damages, default interest, late charges, monetary penalties, makewhole premiums or other economic remedies to the extent such provisions are deemed to constitute a penalty; and (xii) the severability, if invalid, of provisions to the foregoing effect.

We express no opinion as to United States federal or state securities laws (except as set forth in paragraph 9 with respect to federal securities laws), tax laws, antitrust or trade regulation laws (including Hart-Scott-Rodino), insolvency or fraudulent transfer laws, antifraud laws, or compliance with fiduciary duty requirements, margin regulations, FINRA rules, stock exchange rules, pension or employee benefit laws, usury laws, environmental laws, any law or regulation concerning energy or nuclear materials (including, for the avoidance of doubt, any regulation or other legal guidance of the Department of Energy or the Nuclear Regulatory Commission), any law, regulation or legal guidance concerning classified or sensitive information, export control, anti-money laundering or anti-terrorism laws (including Office of Foreign Assets Control Regulations), or laws governing foreign investments in the United States (including CFIUS) (without limiting other laws excluded by customary practice).

With your consent, we have assumed (a) that the Transaction Agreements have been duly authorized, executed and delivered by, and constitute legally valid and binding obligations of, the parties thereto other than the Company, enforceable against each of them in accordance with their respective terms, (b) that the status of the Transaction Agreements as legally valid and binding obligations of the parties is not affected by any (i) breaches of, or defaults under, agreements or instruments, (ii) violations of statutes, rules, regulations or court or governmental orders, or (iii) failures to obtain required consents, approvals or authorizations from, or make required registrations, declarations or filings with, governmental authorities, provided that we make no such assumption to the extent we have specifically opined as to such matters with respect to the Company herein; and (c) that none of the Transaction Agreements has been amended, modified, supplemented or terminated since the original date of execution thereof and that no rights of any parties thereto have been waived by any action or inaction of theirs. With your consent, for purposes of our opinions in paragraphs 6,7 and 8, we have further assumed that, at the time the shares of a particular class or series of capital stock are issued, the Company has sufficient authorized shares of that class or series.

This letter is furnished only to you and is solely for your benefit in connection with the transactions referenced in the first paragraph. This letter may not be relied upon by you for any other purpose, or furnished to, assigned to, quoted to or relied upon by any other person, firm or other entity for any purpose, without our prior written consent, which may be granted or withheld in our discretion.

Very truly yours,

- The reference to Certificate of Incorporation in subsequent opinions shall include references to the Series B-1 and Series C Certificates of Designation as filed at the First Closing. The separate references to these docs will be deleted in this list.
- In connection with the Third Closing, the references to Series B-1 Preferred Stock will be revised to refer to the Series B-2 Preferred Stock.
- To be added in any opinion to be delivered subsequent to the Charter Amendment Approval having been obtained:
- The form of Charter Amendment attached to the Securities Purchase Agreement as Exhibit L proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] for the purpose of amending Article 4 of the Certificate of Incorporation to, among other things, create the Class B Common Stock. With your consent, we have assumed that on the date hereof, the Secretary of State of the State of Delaware has accepted the filing of the Charter Amendment.
- The first three opinions will be omitted at the Second and Third Closings, as will related assumptions and references.
- In connection with any opinion letter delivered at a subsequent closing, the definition of the series of Preferred Shares will be adjusted to reflect the series and number of shares being issued.
- In connection with any opinion letter delivered at the Second Closing, all references to the Warrants will be omitted. In connection with any opinion delivered at the Third Closing, all references to the Warrants shall be to the "Third Closing Warrants".
- In connection with any opinion letter delivered at a subsequent closing, the opinions given in paragraph 4 shall be limited to the issuance and sale of the securities being issued at such closing.

- In connection with any opinion letter delivered subsequent to the Charter Amendment Approval having been obtained, references to Series C Preferred Stock shall be changed to Class B Common Stock and references to the Series C Certificate of Designation shall be changed to the "Amended Certificate of Incorporation", which shall be defined to include the Charter Amendment.
- In connection with any opinion letter delivered subsequent to the Charter Amendment Approval having been obtained, paragraph 7 will be omitted.

EXHIBIT H-3

FORM OF OPINION OF MORGAN LEWIS & BOCKIUS LLP

Morgan, Lewis & Bockius llp 1111 Pennsylvania Avenue, NW Washington, DC 20004 Tel. 202.739.3000 Fax: 202.739.3001

www.morganlewis.com

Form of Regulatory Counsel Legal Opinion - First Closing

, 2010

Toshiba Corporation 1-1 Shibaura 1 - chome Minato-ku, Tokyo 105-8001 Japan

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502

c/o Morrison & Foerster LLP Shin - Marunouchi 1 - chome Chiyoda-ku, Tokyo 100-6529 Japan

c/o Baker Botts LLP 1299 Pennsylvania Avenue, NW Washington, DC 20004-2400 USA

Securities Purchase Agreement - USEC Inc., Toshiba Corporation and Babcock & Wilcox Investment Company -Re: First Closing

Ladies and Gentlemen:

We have acted as special counsel for <u>USEC Inc.</u> on certain regulatory matters as specified below in connection with the purchase by you of Series B-1 12.5% Preferred Stock, Series B-2 11.5% Preferred Stock and Warrants issued by USEC Inc., pursuant to the "Securities Purchase Agreement among USEC Inc., Toshiba Corporation, and Babcock & Wilcox Investment Company," dated as of May 25, 2010 (the "Purchase Agreement"), among USEC Inc. (the "Issuer") and Toshiba Corporation and Babcock & Wilcox Investment Company (the "Purchasers").

In rendering the opinions set forth below, we have reviewed copies of the following documents and instruments:

- (i) the Purchase Agreement;
- (ii) the Investor Rights Agreement;
- (iii) the Series B-1 Certificate of Designation;
- (iv) the Series B-2 Certificate of Designation;
- (v) the Series C Certificate of Designation;
- (vi) the Strategic Relationship Agreement;

- (vii) the Charter Amendment;
- (viii) the Foreign Ownership Restrictions Determination;
- (ix) the Warrants:

such documents and instruments identified in (ii) – (ix) above being as defined in the Purchase Agreement. Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Purchase Agreement.

The documents listed in clauses (i) through (ix) above are referred to herein as the <u>Transaction Documents</u>." In rendering the opinions set forth herein, we have, with your consent, relied only upon examination of the documents described above. As to any facts material to our opinions, we have made no independent investigation of such facts and have relied, to the extent that we deem such reliance proper, upon statements of public officials and officers or other representatives of the Issuer and the Purchasers and on the representations and warranties set forth in the Transaction Documents.

In rendering the opinions expressed below, we have assumed the legal capacity of all natural persons, the genuineness of all signatures, the authenticity of all documents submitted to us as originals, the conformity to authentic original documents of all documents submitted to us as copies, and the accuracy and completeness of all corporate, limited liability company, partnership and other entity records made available to us, which assumptions we have not independently verified.

Based upon the foregoing, and subject to the assumptions, qualifications, exceptions and limitations set forth herein, it is our opinion that:

- 1. The execution and delivery by the Issuer of the Transaction Documents to which it is a party in connection with the First Closing, and the performance by the Issuer of its obligations thereunder, including the issue and sale of the Preferred Stock and Warrants by the Issuer, will not:
 - (i) result in any violation by the Issuer of any Applicable Regulatory Law (as defined below),
 - (ii) result in any violation by the Issuer of any order, writ, judgment or decree known to us issued under Applicable Regulatory Laws.
 - "Applicable Regulatory Laws" means: the Atomic Energy Act of 1954, as amended, and the related rules, regulations or orders of the U.S. Nuclear Regulatory Commission ("NRC"); the U.S. Department of Energy (DOE) foreign ownership, control or influence requirements and restrictions; the Hart-Scott-Rodino Antitrust Improvements Act of 1976; and the requirements of the Committee on Foreign Investment in the United States (CFIUS) adopted pursuant to Section 721 of the Defense Production Act of 1950, as amended by the Foreign Investment and National Security Act of 2007, and as implemented by Executive Order 11858, as amended and 31 CFR Part 800.
- 2. All required Regulatory Approvals have been obtained for the issuance and sale of the Preferred Stock and Warrants by the Issuer and the performance by the Issuer of its obligations under each Transaction Document in connection with the First Closing, as applicable.
 "Regulatory Approvals" means any consent, clearance, approval, license, permit, registration, certification, authorization or validation of, or filing, recording or registration with, the NRC, DOE, the U.S. Department of Justice, the U.S. Federal Trade Commission, and CFIUS pursuant to any Applicable Regulatory Laws.
- 3. Other than the Regulatory Approvals set forth on Schedule B under the heading "First Closing" that have been obtained and the Regulatory Approval set forth in Section 2.9 (vi) of the Purchase Agreement that has been obtained, no Permit or order of, or filing with, any Governmental Authority is required in connection with the First Closing and the performance by the Issuer of its obligations under the Transaction Documents in connection therewith. No Permit or order of, or filing with, any Governmental Authority is required for the Transactions contemplated at the Second Closing and the Third Closing and the performance by the Issuer of its obligations under the Transaction Documents in connection therewith, except for the Regulatory Approvals set forth on Schedule B under the headings "Second Closing" and "Third Closing," respectively and the Regulatory Approvals set forth in Sections 2.9 (iv) and (vi) of the Purchase Agreement that have not yet been obtained.

The opinions expressed herein are limited to the matters discussed above; no opinion is implied or may be inferred

beyond such matters.

This opinion has been prepared in accordance with the customary practice of lawyers who regularly give, and lawyers who regularly advise recipients regarding, opinions of this kind.

We express no opinion as to the laws of any jurisdiction other than Applicable Regulatory Laws.

This letter is rendered as of the date set forth above. We expressly disclaim any obligation to update this letter after such date.

This letter is given solely for your benefit in connection with the transactions contemplated by the Transaction Documents and may not be furnished to, or relied upon by, any other person or for any other purpose without our prior written consent.

Sincerely,

EXHIBIT H-4

FORM OF OPINION OF SPECIAL COUNSEL TO INVESTOR

EXHIBIT H-4

OPINION OF SPECIAL COUNSEL TO INVESTOR

- 1) The Investor is a corporation under , with corporate power and authority to enter into the [Transaction Agreements] and perform its obligations thereunder. With your consent, based solely on certificates from public officials, we confirm that the Investor is validly existing and in good standing under the laws of the State of Delaware.
- 2) The execution, delivery and performance of the [Transaction Agreements] have been duly authorized by all necessary corporate action of the Investor, and the [Transaction Agreements] have been duly executed and delivered by the Investor.
- 3) Each of the [Transaction Agreements] constitutes a legally valid and binding obligation of the Investor, enforceable against the Investor in accordance with its terms.
- 4) The execution and delivery of the [Transaction Agreements] by the Investor, the performance by the Investor of its obligations thereunder and the purchase at the First Closing of shares of Series B-1 12.75% Preferred Stock and the Warrants pursuant to the Securities Purchase Agreement by the Investor on the date hereof do not:
 - (a) violate the provisions of the [governing documents]14;
 - (b) violate any United States federal or New York statute, rule or regulation applicable to the Investor, or the Delaware General Corporation Law15 other than U.S securities laws and such other laws as may customarily be excepted from opinions of this type; or
 - (c) result in the breach of or a default under any [as to B&W, credit or debt instrument of B&W or its Affiliates that is filed as an exhibit to the Form 10 of Babcock & Wilcox Company, as amended prior to the date hereof]
- to include the Investor's charter documents and bylaws, unless, for Toshiba, the nature of its jurisdiction would dictate otherwise. For B&W, this opinion should also cover the form of restated certificate of incorporation and restated bylaws.

Appropriate analog for Toshiba to be provided.

EXHIBIT I-1

FORM OF OPINION OF USEC INC. INTERNAL COUNSEL

 Peter B. Saba
 (301) 564-3327 phone

 General Counsel
 (301) 564-3205 fax

[date], 2010

Toshiba Corporation 1-1 Shibaura 1-chome Minato-ku, Tokyo 105-8001 Japan

And

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502

Re: Securities Purchase Agreement

Ladies and Gentlemen:

I am the General Counsel of USEC Inc., a Delaware corporation (the "Company"). This opinion is being delivered to you pursuant to Section 4.2(k) of that certain Securities Purchase Agreement dated as of May 25, 2010 among the Company, Toshiba Corporation and Babcock & Wilcox Investment Company (the "Securities Purchase Agreement") in connection with the contemplated issuance and sale of 125,000 shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, 75,000 shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share, and Warrants to purchase 12,500,000 shares of the Company's Class B Common Stock, par value \$0.10 per share, to be authorized pursuant to the Charter Amendment, and/or the Company's Series C Preferred Stock, par value \$1.00 per share, pursuant to the Securities Purchase Agreement. Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings ascribed to them in the Securities Purchase Agreement.

In rendering this opinion, I or attorneys on my staff have examined originals or copies (including facsimile transmissions), certified or otherwise identified to us as true and correct copies of originals, of the following documents:

- (1) the Securities Purchase Agreement;
- (2) the Investor Rights Agreement dated as of , 2010 between the Company and you (the "Investor Rights Agreement");
- (3) the Strategic Relationship Agreement dated as of May 25, 2010 between the Company and you (the "Strategic Relationship Agreement," and together with the Securities Purchase Agreement and Investor Rights Agreement, the "Transaction Agreements");
- (4) the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on June 29, 1998, the Certificate of Amendment to the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2008, the Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2001, the Certificate of Increase of Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of Delaware on September 21, 2007, the Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [], the Certificate of Designation of Series B-2 11.5% Convertible Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [] and the Certificate of Designation of Series C Convertible

Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [] (collectively, the "Certificate of Incorporation") and the Amended and Restated Bylaws of the Company in effect as of May , 2010 ("Bylaws," and together with the Certificate of Incorporation, the "Governing Documents");

- (5) the form of Certificate of Amendment to the Certificate of Incorporation of USEC Inc. attached to the Securities Purchase Agreement as Exhibit L (the "Charter Amendment");
- (6) the Warrant to Purchase Class B Common Stock or Series C Preferred Stock of USEC Inc. (the 'Warrant'); and
- (7) such other documents as we have deemed necessary or appropriate as a basis for the opinions set forth below.

In our examination, we have assumed the legal capacity of all natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as certified, conformed, facsimile or photostatic copies, and the authenticity of the originals of such copies. In making our examination of documents executed or to be executed by parties other than the Company (each such party, an "Other Party"), we have assumed that such parties had the power, corporate or other, to enter into and perform all obligations thereunder and have also assumed the due authorization by all requisite action, corporate or other, and execution and delivery by such parties of such documents and that such documents constitute the valid and binding obligations of such parties. As to any facts material to the opinions expressed herein which we did not independently establish or verify, we have relied upon oral or written statements and representations of officers and other representatives and of public officials, including the facts and conclusions set forth therein.

I am admitted to practice in the State of New York and the District of Columbia. This opinion is limited to matters arising under the laws of the State of New York, the General Corporation Law of the State of Delaware and the federal laws of the United States of America. I do not express any opinion as to the laws of any other jurisdiction or as to the municipal laws or the laws, rules or regulations of any local agencies or governmental authorities of or within the State of New York, or in each case as to any matters arising thereunder or relating thereto. In giving the opinions in paragraph 1, I have assumed with your permission that to the extent any indenture, mortgage, lease, contract, agreement or instrument referenced therein is governed by laws other than the laws of the State of New York that such indenture, mortgage, lease, contract, agreement or instrument would be interpreted in the same manner as if they were governed by the laws of the State of New York.

Based upon the foregoing and subject to the limitations, qualifications, exceptions and assumptions set forth herein, I am of the opinion that:

- 1. The issuance and sale at the Second Closing of shares of Series B-1 12.75% Preferred Stock pursuant to the Securities Purchase Agreement by the Company on the date hereof do not constitute a violation of, or a default under any indenture, mortgage, lease, contract, agreement, instrument or other arrangement by which the Company is bound and listed as an exhibit to the Company's most recent (x) Annual Report on Form 10-K or (y) Quarterly Report on Form 10-Q filed with the United States Securities and Exchange Commission (collectively, the "SEC Exhibits"), except for violations or defaults which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- 2. The authorized capital stock of the Company consists of 250,000,000 shares of common stock, par value \$0.10 per share and 25,000,000 shares of preferred stock, par value \$1.00 per share, of which (i) shares are designated Series A Preferred Stock, (ii) shares are designated Series B-1 Preferred Stock, (iii) shares are designated Series B-2 Preferred Stock, and (iv) shares are designated Series C Preferred Stock.
- 3. To my knowledge, there is no action, litigation or proceeding pending or threatened in writing by or before any Governmental Authority which is reasonably likely to materially and adversely affect the ability of the Company to perform its obligations under the Transaction Agreements or which seeks to prevent the consummation of the transactions contemplated by the Transaction Agreements.
- 4. The issuance and sale of the shares of Series B-1 Preferred Stock by the Company at the Second Closing, and the issuance of shares of Class B Common Stock or Series C Preferred Stock issuable upon conversion of the Series B-1 Preferred Stock issued in the Second Closing are free of preemptive or similar rights arising under the SEC Exhibits.

This opinion is given as of the date hereof and I assume no obligation to update or supplement this opinion to reflect any facts or circumstances, which may hereafter come to my attention or any changes in laws which may hereafter occur. This opinion is solely for your benefit in connection with the above-referenced transaction and it may not be relied upon by any other person, and it may not be used for any other purpose, without my prior written consent, which may be

granted or withheld in my sole discretion.

Very truly yours,

If the Charter Amendment is approved and filed, this Opinion will be modified as follows: "The authorized capital stock of the Company consists of 475,000,000 shares of common stock, par value \$0.10 per share, of which (a) 350,000,000 shares are designated Common Stock and (b) 125,000,000 shares are designated Class B Common Stock; and 25,000,000 shares of preferred stock, par value \$1.00 per share, of which (i) shares are designated Series A Preferred Stock, (ii) shares are designated Series B-1 Preferred Stock, (iii) shares are designated Series B-2 Preferred Stock, and (iv) shares are designated Series C Preferred Stock."

EXHIBIT I-2

FORM OF OPINION OF LATHAM & WATKINS LLP

[L&W Letterhead]

[Date]

To the Investors party to the Securities Purchase Agreement dated as of May 25, 2010, among USEC Inc. and such Investors

Re: Securities Purchase Agreement

Ladies and Gentlemen:

We have acted as special counsel to USEC Inc., a Delaware corporation (the "Company"), in connection with the contemplated issuance and sale of 125,000 shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, 75,000 shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share, and Warrants to purchase 12,500,000 shares of the Company's Class B Common Stock, par value \$0.10 per share, to be authorized pursuant to the Charter Amendment, and/or the Company's Series C Convertible Participating Preferred Stock, par value \$1.00 per share, pursuant to that certain Securities Purchase Agreement dated as of May 25, 2010 (the "Securities Purchase Agreement") among the Company and you. This letter is furnished pursuant to Section 4.2(k) of the Securities Purchase Agreement. Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings ascribed to them in the Securities Purchase Agreement.

As such counsel, we have examined such matters of fact and questions of law as we have considered appropriate for purposes of this letter, except where a specified fact confirmation procedure is stated to have been performed (in which case we have, with your consent, performed the stated procedure). We have examined, among other things, the following:

- (a) The Securities Purchase Agreement;
- (b) The Investor Rights Agreement dated as of , 2010 between the Company and you (the 'Investor Rights Agreement');
- (c) The Strategic Relationship Agreement dated as of May 25, 2010 between the Company and you (the 'Strategic Relationship Agreement," and together with the Securities Purchase Agreement and Investor Rights Agreement, the "Transaction Agreements");
- (d) The Rights Agreement dated as of April 24, 2001, by and between the Company, a Delaware corporation and Fleet National Bank, a [•] as the Rights Agreement (the "Rights Agreement") and the First Amendment to Rights Agreement dated May [•], 2010 by and between the Company and [•] (f/k/a) Fleet National Bank as the successor Rights Agent (the "Rights Agreement Amendment");
- (e) The Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on June 29, 1998, the Certificate of Amendment to the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2008, the Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2001 and the Certificate of Increase of Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on September 21, 2007 (collectively, the "Certificate of Incorporation" 2) and the Amended and Restated Bylaws of the Company in effect as of May , 2010 ("Bylaws," and together with the Certificate of Incorporation, the "Governing Documents") and certain resolutions of the Board of Directors of the Company and certain committees thereof, and
- (f) The form of Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Company attached to the Securities Purchase Agreement as Exhibit B proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] to establish the designation of the

Company's Series B-1 12.75% Preferred Stock ("Series B-1 Certificate of Designation"18).

(g) The form of Certificate of Designation of Series C Convertible Participating Preferred Stock of the Company attached to the Securities Purchase Agreement as Exhibit D proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] to establish the designation of the Company's Series C Convertible Participating Preferred Stock ("Series C Certificate of Designation").¹²

Except as otherwise stated herein, as to factual matters we have, with your consent, relied upon the foregoing, and upon oral and written statements and representations of officers and other representatives of the Company and others, including the representations and warranties of the Company in the Securities Purchase Agreement. We have not independently verified such factual matters.

Except as otherwise stated herein, we are opining herein as to the effect on the subject transaction only of the federal laws of the United States, the internal laws of the State of New York, and in paragraphs 1, 2, 4(i), 4(ii), 4(iii), 5, 6, 7 and 8 of this letter, the Delaware General Corporation Law, and we express no opinion with respect to the applicability thereto, or the effect thereon, of the laws of any other jurisdiction or, in the case of Delaware, any other laws, or as to any matters of municipal law or the laws of any local agencies within any state. Except as otherwise stated herein, our opinions herein are based upon our consideration of only those statutes, rules and regulations which, in our experience, are normally applicable to the sales of shares of convertible preferred stock and warrants in a private placement. Various issues pertaining to [] are addressed in the opinions of the Company's General Counsel and Morgan, Lewis & Bockius LLP, separately provided to you. We express no opinion with respect to those matters, and to the extent elements of those opinions are necessary to the conclusions expressed herein, we have, with your consent, assumed such matters.

Subject to the foregoing and the other matters set forth herein, as of the date hereof²⁰

- 1. The Company is a corporation under the Delaware General Corporation Law, with corporate power and authority to enter into the Transaction Agreements and the Rights Agreement Amendment and perform its obligations thereunder. With your consent, based solely on certificates from public officials, we confirm that the Company is validly existing and in good standing under the laws of the State of Delaware.
- 2. The execution, delivery and performance of the Transaction Agreements and the Rights Agreement Amendment have been duly authorized by all necessary corporate action of the Company, and the Transaction Agreements have been duly executed and delivered by the Company.
- 3. Each of the Transaction Agreements constitutes a legally valid and binding obligation of the Company, enforceable against the Company in accordance with its terms.
- 4. The execution and delivery of the Transaction Agreements by the Company, the performance by the Company of its obligations thereunder and the issuance and sale by the Company pursuant to the Securities Purchase Agreement at the First Closing of shares of Series B-1 12.75% Preferred Stock (the "First Closing Preferred Shares")²¹ and the Warrants (the "First Closing Warrants")²² on the date hereof do not:
 - (i) violate the provisions of the Governing Documents;
 - (ii) violate any United States federal or New York statute, rule or regulation applicable to the Company or the Delaware General Corporation Law; or
 - (iii) require any consents, approvals or authorizations to be obtained by the Company from, or any registrations, declarations or filings to be made by the Company with, any governmental authority, under any United States federal or New York statute, rule or regulation applicable to the Company or the Delaware General Corporation Law on or prior to the date hereof that have not been obtained or made.
- 5. The First Closing Preferred Shares have been duly authorized by all necessary corporate action of the Company, and, when issued to and paid for by you in accordance with the terms of the Securities Purchase Agreement and the Series B-1 Certificate of Designation, will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.
- 6. The shares of the Series C Preferred Stock initially issuable upon conversion of the First Closing Preferred Shares or upon exercise of the First Closing Warrants have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, when issued to and, in the case of shares issuable upon exercise of the First Closing Warrants, paid for by, you in accordance with the terms of the Securities Purchase Agreement, the Series B-1

Certificate of Designation (in the case of shares of Series C Preferred Stock issuable upon conversion of First Closing Preferred Shares), the First Closing Warrants (in the case of shares of Series C Preferred Stock issuable upon exercise of the First Closing Warrants) and the Series C Certificate of Designation, will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.

- 7. The shares of the Company's Ordinary Common Stock initially issuable upon conversion of the First Closing Preferred Shares have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, when issued in accordance with the terms of the Securities Purchase Agreement and the Series B-1 Certificate of Designation will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents 25
- 8. The shares of the Company's Ordinary Common Stock initially issuable upon conversion of the shares of Series C Preferred Stock which are initially issuable upon conversion of First Closing Preferred Shares or upon exercise of the First Closing Warrants, have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, assuming that such shares of Series C Preferred Stock have been validly issued, when such shares of the Company's Ordinary Common Stock are issued in accordance with the terms of the Securities Purchase Agreement and the Series C Certificate of Designation, such shares of Ordinary Common Stock will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.
- 9. Assuming the representations and warranties of the parties thereto in the Transaction Agreements are true and assuming compliance by the parties thereto with their respective covenants and agreements set forth therein, no registration of the First Closing Preferred Shares or the First Closing Warrants under the Securities Act of 1933, as amended, is required for the purchase by you of such First Closing Preferred Shares or First Closing Warrants in the manner contemplated by the Securities Purchase Agreement. We express no opinion, however, as to when or under what circumstances any First Closing Preferred Shares or First Closing Warrants, or any securities issuable upon conversion of the First Closing Preferred Shares or upon exercise of the First Closing Warrants, may be reoffered or resold.

Our opinions are subject to:

- (a) the effect of bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or other similar laws relating to or affecting the rights or remedies of creditors;
- (b) the effects of general principles of equity, whether considered in a proceeding in equity or at law (including the possible unavailability of specific performance or injunctive relief), concepts of materiality, reasonableness, good faith and fair dealing, and the discretion of the court before which a proceeding is brought;
- (c) the invalidity under certain circumstances under law or court decisions of provisions for the indemnification of or contribution to a party with respect to a liability where such indemnification or contribution is contrary to public policy; and
- (d) we express no opinion with respect to (i) consents to, or restrictions upon, governing law or jurisdiction, venue, arbitration, remedies or judicial relief; (ii) advance waivers of claims, defenses, rights granted by law, or notice, opportunity for hearing, evidentiary requirements, statutes of limitation, trial by jury or at law, or other procedural rights; (iii) waivers of broadly or vaguely stated rights; (iv) covenants not to compete; (v) provisions for exclusivity, election, cumulation or limitation of rights or remedies; (vi) provisions authorizing or validating conclusive or discretionary determinations; (vii) grants of setoff rights; (viii) provisions for the payment of attorneys' fees where such payment is contrary to law or public policy; (ix) proxies, powers of attorney, and voting agreements or trusts; (x) provisions prohibiting, restricting or requiring consent to assignment or transfer of any right or property; (xi) provisions for liquidated damages, default interest, late charges, monetary penalties, make-whole premiums or other economic remedies to the extent such provisions are deemed to constitute a penalty; and (xii) the severability, if invalid, of provisions to the foregoing effect.

We express no opinion as to United States federal or state securities laws (except as set forth in paragraph 9 with respect to federal securities laws), tax laws, antitrust or trade regulation laws (including Hart-Scott-Rodino), insolvency or fraudulent transfer laws, antifraud laws, or compliance with fiduciary duty requirements, margin regulations, FINRA rules, stock exchange rules, pension or employee benefit laws, usury laws, environmental laws, any law or regulation concerning energy or nuclear materials (including, for the avoidance of doubt, any regulation or other legal guidance of the Department of Energy or the Nuclear Regulatory Commission), any law, regulation or legal guidance concerning classified or sensitive information, export control, anti-money laundering or anti-terrorism laws (including Office of Foreign Assets Control Regulations), or laws governing foreign investments in the United States (including CFIUS) (without limiting other laws excluded by customary practice).

With your consent, we have assumed (a) that the Transaction Agreements have been duly authorized, executed and delivered by, and constitute legally valid and binding obligations of, the parties thereto other than the Company, enforceable against each of them in accordance with their respective terms, (b) that the status of the Transaction Agreements as legally valid and binding obligations of the parties is not affected by any (i) breaches of, or defaults under, agreements or instruments, (ii) violations of statutes, rules, regulations or court or governmental orders, or (iii) failures to obtain required consents, approvals or authorizations from, or make required registrations, declarations or filings with, governmental authorities, provided that we make no such assumption to the extent we have specifically opined as to such matters with respect to the Company herein; and (c) that none of the Transaction Agreements has been amended, modified, supplemented or terminated since the original date of execution thereof and that no rights of any parties thereto have been waived by any action or inaction of theirs. With your consent, for purposes of our opinions in paragraphs 6,7 and 8, we have further assumed that, at the time the shares of a particular class or series of capital stock are issued, the Company has sufficient authorized shares of that class or series.

This letter is furnished only to you and is solely for your benefit in connection with the transactions referenced in the first paragraph. This letter may not be relied upon by you for any other purpose, or furnished to, assigned to, quoted to or relied upon by any other person, firm or other entity for any purpose, without our prior written consent, which may be granted or withheld in our discretion.

Very truly yours,

- The reference to Certificate of Incorporation in subsequent opinions shall include references to the Series B-1 and Series C Certificates of Designation as filed at the First Closing. The separate references to these docs will be deleted in this list.
- In connection with the Third Closing, the references to Series B-1 Preferred Stock will be revised to refer to the Series B-2 Preferred Stock.
- To be added in any opinion to be delivered subsequent to the Charter Amendment Approval having been obtained:
- The form of Charter Amendment attached to the Securities Purchase Agreement as Exhibit L proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] for the purpose of amending Article 4 of the Certificate of Incorporation to, among other things, create the Class B Common Stock. With your consent, we have assumed that on the date hereof, the Secretary of State of the State of Delaware has accepted the filing of the Charter Amendment.
- The first three opinions will be omitted at the Second and Third Closings, as will related assumptions and references.
- In connection with any opinion letter delivered at a subsequent closing, the definition of the series of Preferred Shares will be adjusted to reflect the series and number of shares being issued.

 In connection with any opinion letter delivered at the Second Closing all references to the Warments will be arrived. In connection with any opinion delivered at the Second Closing all references to the Warments will be arrived.
- In connection with any opinion letter delivered at the Second Closing, all references to the Warrants will be omitted. In connection with any opinion delivered at the Third Closing, all references to the Warrants shall be to the "Third Closing Warrants".
- In connection with any opinion letter delivered at a subsequent closing, the opinions given in paragraph 4 shall be limited

- to the issuance and sale of the securities being issued at such closing.

 In connection with any opinion letter delivered subsequent to the Charter Amendment Approval having been obtained, references to Series C Preferred Stock shall be changed to Class B Common Stock and references to the Series C Certificate of Designation shall be changed to the "Amended Certificate of 24 Incorporation", which shall be defined to include the Charter Amendment.

 In connection with any opinion letter delivered subsequent to the Charter Amendment Approval having been obtained, paragraph 7 will be omitted.

EXHIBIT I-3

FORM OF OPINION OF MORGAN LEWIS & BOCKIUS LLP

Morgan, Lewis & Bockius llp 1111 Pennsylvania Avenue, NW Washington, DC 20004 Tel. 202.739.3000 Fax: 202.739.3001

www.morganlewis.com

Form of Regulatory Counsel Legal Opinion - Second Closing

, 2010

Toshiba Corporation 1-1 Shibaura 1 - chome Minato-ku, Tokyo 105-8001 Japan

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502

c/o Morrison & Foerster LLP Shin - Marunouchi 1 - chome Chiyoda-ku, Tokyo 100-6529 Japan

c/o Baker Botts LLP 1299 Pennsylvania Avenue, NW Washington, DC 20004-2400 USA

Securities Purchase Agreement - USEC Inc., Toshiba Corporation and Babcock & Wilcox Investment Company -Re: Second Closing

Ladies and Gentlemen:

We have acted as special counsel for <u>USEC Inc.</u> on certain regulatory matters as specified below in connection with the purchase by you of Series B-1 12.5% Preferred Stock, Series B-2 11.5% Preferred Stock and Warrants issued by USEC Inc., pursuant to the "Securities Purchase Agreement among USEC Inc., Toshiba Corporation, and Babcock & Wilcox Investment Company," dated as of May 25, 2010 (the "Purchase Agreement"), among USEC Inc. (the "Issuer") and Toshiba Corporation and Babcock & Wilcox Investment Company (the "Purchasers").

In rendering the opinions set forth below, we have reviewed copies of the following documents and instruments:

- (i) the Purchase Agreement;
- (ii) the Investor Rights Agreement;
- (iii) the Series B-1 Certificate of Designation;
- (iv) the Series B-2 Certificate of Designation;
- (v) the Series C Certificate of Designation;

- (vi) the Strategic Relationship Agreement;
- (vii) the Charter Amendment;
- (viii) the Foreign Ownership Restrictions Determination;

such documents and instruments identified in (ii) – (viii) above being as defined in the Purchase Agreement. Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Purchase Agreement.

The documents listed in clauses (i) through (viii) above are referred to herein as the <u>Transaction Documents</u>." In rendering the opinions set forth herein, we have, with your consent, relied only upon examination of the documents described above. As to any facts material to our opinions, we have made no independent investigation of such facts and have relied, to the extent that we deem such reliance proper, upon statements of public officials and officers or other representatives of the Issuer and the Purchasers and on the representations and warranties set forth in the Transaction Documents.

In rendering the opinions expressed below, we have assumed the legal capacity of all natural persons, the genuineness of all signatures, the authenticity of all documents submitted to us as originals, the conformity to authentic original documents of all documents submitted to us as copies, and the accuracy and completeness of all corporate, limited liability company, partnership and other entity records made available to us, which assumptions we have not independently verified.

Based upon the foregoing, and subject to the assumptions, qualifications, exceptions and limitations set forth herein, it is our opinion that:

- The issuance and sale of the Preferred Stock by the Issuer in connection with the Second Closing and the performance by the Issuer of its
 obligations thereunder and under the Purchase Agreement, will not:
 - (i) result in any violation by the Issuer of any Applicable Regulatory Law (as defined below),
 - (ii) result in any violation by the Issuer of any order, writ, judgment or decree known to us issued under Applicable Regulatory Laws.
 - "Applicable Regulatory Laws" means: the Atomic Energy Act of 1954, as amended, and the related rules, regulations or orders of the U.S. Nuclear Regulatory Commission ("NRC"); the U.S. Department of Energy (DOE) foreign ownership, control or influence requirements and restrictions; the Hart-Scott-Rodino Antitrust Improvements Act of 1976; and the requirements of the Committee on Foreign Investment in the United States (CFIUS) adopted pursuant to Section 721 of the Defense Production Act of 1950, as amended by the Foreign Investment and National Security Act of 2007, and as implemented by Executive Order 11858, as amended and 31 CFR Part 800.
- 2. All required Regulatory Approvals have been obtained for the issuance and sale of the Preferred Stock by the Issuer and the performance by the Issuer of its obligations under each Transaction Document in connection with the Second Closing, as applicable.
 - "Regulatory Approvals" means any consent, clearance, approval, license, permit, registration, certification, authorization or validation of, or filing, recording or registration with, the NRC, DOE, the U.S. Department of Justice, the U.S. Federal Trade Commission, and CFIUS pursuant to any Applicable Regulatory Laws.
- 3. Other than the Regulatory Approvals set forth on Schedule B under the heading "Second Closing" that have been obtained and the Regulatory Approvals set forth in Sections 2.9 (iv) and (vi) of the Purchase Agreement that have been obtained, no Permit or order of, or filing with, any Governmental Authority is required in connection with the Second Closing and the performance by the Issuer of its obligations under the Transaction Documents in connection therewith. No Permit or order of, or filing with, any Governmental Authority is required for the Transactions contemplated at the Third Closing and the performance by the Issuer of its obligations under the Transaction Documents in connection therewith, except for the Regulatory Approvals set forth on Schedule B under the heading "Third Closing" and the Regulatory Approvals set forth in Sections 2.9 (iv) and (vi) of the Purchase Agreement that have not been obtained.

The opinions expressed herein are limited to the matters discussed above; no opinion is implied or may be inferred

beyond such matters.

This opinion has been prepared in accordance with the customary practice of lawyers who regularly give, and lawyers who regularly advise recipients regarding, opinions of this kind.

We express no opinion as to the laws of any jurisdiction other than Applicable Regulatory Laws.

This letter is rendered as of the date set forth above. We expressly disclaim any obligation to update this letter after such date.

This letter is given solely for your benefit in connection with the transactions contemplated by the Transaction Documents and may not be furnished to, or relied upon by, any other person or for any other purpose without our prior written consent.

Sincerely,

EXHIBIT I-4

FORM OF OPINION OF SPECIAL COUNSEL TO INVESTORS

EXHIBIT I-4

OPINION OF SPECIAL COUNSEL TO INVESTOR

- 1) The Investor is a corporation under , with corporate power and authority to enter into the [Transaction Agreements] and perform its obligations thereunder. With your consent, based solely on certificates from public officials, we confirm that the Investor is validly existing and in good standing under the laws of the State of Delaware.
- 2) The execution, delivery and performance of the [Transaction Agreements] have been duly authorized by all necessary corporate action of the Investor, and the [Transaction Agreements] have been duly executed and delivered by the Investor.
- 3) Each of the [Transaction Agreements] constitutes a legally valid and binding obligation of the Investor, enforceable against the Investor in accordance with its terms.
- 4) The execution and delivery of the [Transaction Agreements] by the Investor, the performance by the Investor of its obligations thereunder and the purchase at the First Closing of shares of Series B-1 12.75% Preferred Stock and the Warrants pursuant to the Securities Purchase Agreement by the Investor on the date hereof do not:
 - (a) violate the provisions of the [governing documents]26;
 - (b) violate any United States federal or New York statute, rule or regulation applicable to the Investor, or the Delaware General Corporation Law22 other than U.S securities laws and such other laws as may customarily be excepted from opinions of this type; or
 - (c) result in the breach of or a default under any [as to B&W, credit or debt instrument of B&W or its Affiliates that is filed as an exhibit to the Form 10 of Babcock & Wilcox Company, as amended prior to the date hereof]
- to include the Investor's charter documents and bylaws, unless, for Toshiba, the nature of its jurisdiction would dictate otherwise. For B&W, this opinion should also cover the form of restated certificate of incorporation and restated bylaws.

Appropriate analog for Toshiba to be provided.

EXHIBIT J-1

FORM OF OPINION OF USEC INC. INTERNAL COUNSEL

 Peter B. Saba
 (301) 564-3327 phone

 General Counsel
 (301) 564-3205 fax

[date], 2010

Toshiba Corporation 1-1 Shibaura 1-chome Minato-ku, Tokyo 105-8001 Japan

And

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502

Re: Securities Purchase Agreement

Ladies and Gentlemen:

I am the General Counsel of USEC Inc., a Delaware corporation (the "Company"). This opinion is being delivered to you pursuant to Section 4.2(k) of that certain Securities Purchase Agreement dated as of May 25, 2010 among the Company, Toshiba Corporation and Babcock & Wilcox Investment Company (the "Securities Purchase Agreement") in connection with the contemplated issuance and sale of 125,000 shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, 75,000 shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share, and Warrants to purchase 12,500,000 shares of the Company's Class B Common Stock, par value \$0.10 per share, to be authorized pursuant to the Charter Amendment, and/or the Company's Series C Preferred Stock, par value \$1.00 per share, pursuant to the Securities Purchase Agreement. Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings ascribed to them in the Securities Purchase Agreement.

In rendering this opinion, I or attorneys on my staff have examined originals or copies (including facsimile transmissions), certified or otherwise identified to us as true and correct copies of originals, of the following documents:

- (1) the Securities Purchase Agreement;
- (2) the Investor Rights Agreement dated as of , 2010 between the Company and you (the "Investor Rights Agreement");
- (3) the Strategic Relationship Agreement dated as of May 25, 2010 between the Company and you (the "Strategic Relationship Agreement," and together with the Securities Purchase Agreement and Investor Rights Agreement, the "Transaction Agreements");

the State of Delaware on [] and the Certificate of Designation of Series C Convertible Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on [] (collectively, the "Certificate of Incorporation") and the Amended and Restated Bylaws of the Company in effect as of May , 2010 ("Bylaws," and together with the Certificate of Incorporation, the "Governing Documents");

- (5) the form of Certificate of Amendment to the Certificate of Incorporation of USEC Inc. attached to the Securities Purchase Agreement as Exhibit L (the "Charter Amendment");
- (6) the Warrant to Purchase Class B Common Stock or Series C Preferred Stock of USEC Inc. (the "Warrant"); and
- (7) such other documents as we have deemed necessary or appropriate as a basis for the opinions set forth below.

In our examination, we have assumed the legal capacity of all natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as certified, conformed, facsimile or photostatic copies, and the authenticity of the originals of such copies. In making our examination of documents executed or to be executed by parties other than the Company (each such party, an "Other Party"), we have assumed that such parties had the power, corporate or other, to enter into and perform all obligations thereunder and have also assumed the due authorization by all requisite action, corporate or other, and execution and delivery by such parties of such documents and that such documents constitute the valid and binding obligations of such parties. As to any facts material to the opinions expressed herein which we did not independently establish or verify, we have relied upon oral or written statements and representations of officers and other representatives and of public officials, including the facts and conclusions set forth therein.

I am admitted to practice in the State of New York and the District of Columbia. This opinion is limited to matters arising under the laws of the State of New York, the General Corporation Law of the State of Delaware and the federal laws of the United States of America. I do not express any opinion as to the laws of any other jurisdiction or as to the municipal laws or the laws, rules or regulations of any local agencies or governmental authorities of or within the State of New York, or in each case as to any matters arising thereunder or relating thereto. In giving the opinions in paragraph 1, I have assumed with your permission that to the extent any indenture, mortgage, lease, contract, agreement or instrument referenced therein is governed by laws other than the laws of the State of New York that such indenture, mortgage, lease, contract, agreement or instrument would be interpreted in the same manner as if they were governed by the laws of the State of New York.

Based upon the foregoing and subject to the limitations, qualifications, exceptions and assumptions set forth herein, I am of the opinion that:

- 1. The issuance and sale at the Third Closing of shares of Series B-2 11.5% Preferred Stock and the Warrants pursuant to the Securities Purchase Agreement by the Company on the date hereof do not constitute a violation of, or a default under any indenture, mortgage, lease, contract, agreement, instrument or other arrangement by which the Company is bound and listed as an exhibit to the Company's most recent (x) Annual Report on Form 10-K or (y) Quarterly Report on Form 10-Q filed with the United States Securities and Exchange Commission (collectively, the "SEC Exhibits"), except for violations or defaults which would not, individually or in the aggregate, reasonably be expected to result in a Material Adverse Effect.
- 2. The authorized capital stock of the Company consists of 250,000,000 shares of common stock, par value \$0.10 per share and 25,000,000 shares of preferred stock, par value \$1.00 per share, of which (i) shares are designated Series A Preferred Stock, (ii) shares are designated Series B-1 Preferred Stock, (iii) shares are designated Series B-2 Preferred Stock, and (iv) shares are designated Series C Preferred Stock.
- 3. To my knowledge, there is no action, litigation or proceeding pending or threatened in writing by or before any Governmental Authority which is reasonably likely to materially and adversely affect the ability of the Company to perform its obligations under the Transaction Agreements or which seeks to prevent the consummation of the transactions contemplated by the Transaction Agreements.
- 4. The issuance and sale of the shares of Series B-2 Preferred Stock and the Warrants by the Company at the Third Closing, and the issuance of shares of Class B Common Stock issuable upon conversion of the Series B-2 Preferred Stock issued in the Third Closing and upon the exercise of the Warrants issued in the Third Closing are free of preemptive or similar rights arising under the SEC Exhibits.

This opinion is given as of the date hereof and I assume no obligation to update or supplement this opinion to reflect

any facts or circumstances, which may hereafter come to my attention or any changes in laws which may hereafter occur. This opinion is solely for your benefit in connection with the above-referenced transaction and it may not be relied upon by any other person, and it may not be used for any other purpose, without my prior written consent, which may be granted or withheld in my sole discretion.

Very truly yours,

If the Charter Amendment is approved and filed, this Opinion will be modified as follows: "The authorized capital stock of the Company consists of 475,000,000 shares of common stock, par value \$0.10 per share, of which (a) 350,000,000 shares are designated Common Stock and (b) 125,000,000 shares are designated Class B Common Stock; and 25,000,000 shares of preferred stock, par value \$1.00 per share, of which (i) shares are designated Series A Preferred Stock, (ii) shares are designated Series B-1 Preferred Stock, (iii) shares are designated Series B-2 Preferred Stock, and (iv) shares are designated Series C Preferred Stock."

EXHIBIT J-2

FORM OF OPINION OF LATHAM & WATKINS LLP

[L&W Letterhead]

[Date]

To the Investors party to the Securities Purchase Agreement dated as of May 25, 2010, among USEC Inc. and such Investors

Re: Securities Purchase Agreement

Ladies and Gentlemen:

We have acted as special counsel to USEC Inc., a Delaware corporation (the "Company"), in connection with the contemplated issuance and sale of 125,000 shares of the Company's Series B-1 12.75% Convertible Preferred Stock, par value \$1.00 per share, 75,000 shares of the Company's Series B-2 11.5% Convertible Preferred Stock, par value \$1.00 per share, and Warrants to purchase 12,500,000 shares of the Company's Class B Common Stock, par value \$0.10 per share, to be authorized pursuant to the Charter Amendment, and/or the Company's Series C Convertible Participating Preferred Stock, par value \$1.00 per share, pursuant to that certain Securities Purchase Agreement dated as of May 25, 2010 (the "Securities Purchase Agreement") among the Company and you. This letter is furnished pursuant to Section 4.2(k) of the Securities Purchase Agreement. Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings ascribed to them in the Securities Purchase Agreement.

As such counsel, we have examined such matters of fact and questions of law as we have considered appropriate for purposes of this letter, except where a specified fact confirmation procedure is stated to have been performed (in which case we have, with your consent, performed the stated procedure). We have examined, among other things, the following:

- (a) The Securities Purchase Agreement;
- (b) The Investor Rights Agreement dated as of , 2010 between the Company and you (the "Investor Rights Agreement");
- (c) The Strategic Relationship Agreement dated as of May 25, 2010 between the Company and you (the "Strategic Relationship Agreement," and together with the Securities Purchase Agreement and Investor Rights Agreement, the "Transaction Agreements");
- (d) The Rights Agreement dated as of April 24, 2001, by and between the Company, a Delaware corporation and Fleet National Bank, a [•] as the Rights Agent (the "*Rights Agreement*") and the First Amendment to Rights Agreement dated May [•], 2010 by and between the Company and [•] (f/k/a) Fleet National Bank as the successor Rights Agent (the "*Rights Agreement Amendment*");
- (e) The Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on June 29, 1998, the Certificate of Amendment to the Certificate of Incorporation of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2008, the Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on April 25, 2001 and the Certificate of Increase of Certificate of Designation, Preferences and Rights of Series A Junior Participating Preferred Stock of the Company, in the form filed with the Secretary of State of the State of Delaware on September 21, 2007 (collectively, the "Certificate of Incorporation") and the Amended and Restated Bylaws of the Company in effect as of May , 2010 ("Bylaws," and together with the Certificate of Incorporation, the "Governing Documents") and certain resolutions of the Board of Directors of the Company and certain committees thereof, and
- (f) The form of Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Company attached to the Securities Purchase Agreement as Exhibit B proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] to establish the designation of the

Company's Series B-1 12.75% Preferred Stock ("Series B-1 Certificate of Designation" 20).

(g) The form of Certificate of Designation of Series C Convertible Participating Preferred Stock of the Company attached to the Securities Purchase Agreement as Exhibit D proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] to establish the designation of the Company's Series C Convertible Participating Preferred Stock ("Series C Certificate of Designation").²¹

Except as otherwise stated herein, as to factual matters we have, with your consent, relied upon the foregoing, and upon oral and written statements and representations of officers and other representatives of the Company and others, including the representations and warranties of the Company in the Securities Purchase Agreement. We have not independently verified such factual matters.

Except as otherwise stated herein, we are opining herein as to the effect on the subject transaction only of the federal laws of the United States, the internal laws of the State of New York, and in paragraphs 1, 2, 4(i), 4(ii), 4(iii), 5, 6, 7 and 8 of this letter, the Delaware General Corporation Law, and we express no opinion with respect to the applicability thereto, or the effect thereon, of the laws of any other jurisdiction or, in the case of Delaware, any other laws, or as to any matters of municipal law or the laws of any local agencies within any state. Except as otherwise stated herein, our opinions herein are based upon our consideration of only those statutes, rules and regulations which, in our experience, are normally applicable to the sales of shares of convertible preferred stock and warrants in a private placement. Various issues pertaining to [] are addressed in the opinions of the Company's General Counsel and Morgan, Lewis & Bockius LLP, separately provided to you. We express no opinion with respect to those matters, and to the extent elements of those opinions are necessary to the conclusions expressed herein, we have, with your consent, assumed such matters.

Subject to the foregoing and the other matters set forth herein, as of the date hereof³²

- 1. The Company is a corporation under the Delaware General Corporation Law, with corporate power and authority to enter into the Transaction Agreements and the Rights Agreement Amendment and perform its obligations thereunder. With your consent, based solely on certificates from public officials, we confirm that the Company is validly existing and in good standing under the laws of the State of Delaware.
- 2. The execution, delivery and performance of the Transaction Agreements and the Rights Agreement Amendment have been duly authorized by all necessary corporate action of the Company, and the Transaction Agreements have been duly executed and delivered by the Company.
- 3. Each of the Transaction Agreements constitutes a legally valid and binding obligation of the Company, enforceable against the Company in accordance with its terms.
- 4. The execution and delivery of the Transaction Agreements by the Company, the performance by the Company of its obligations thereunder and the issuance and sale by the Company pursuant to the Securities Purchase Agreement at the First Closing of shares of Series B-1 12.75% Preferred Stock (the "First Closing Preferred Shares")³³ and the Warrants (the "First Closing Warrants")³⁴ on the date hereof do not:
 - (i) violate the provisions of the Governing Documents;
 - (ii) violate any United States federal or New York statute, rule or regulation applicable to the Company or the Delaware General Corporation Law; or
 - (iii) require any consents, approvals or authorizations to be obtained by the Company from, or any registrations, declarations or filings to be made by the Company with, any governmental authority, under any United States federal or New York statute, rule or regulation applicable to the Company or the Delaware General Corporation Law on or prior to the date hereof that have not been obtained or made.
- 5. The First Closing Preferred Shares have been duly authorized by all necessary corporate action of the Company, and, when issued to and paid for by you in accordance with the terms of the Securities Purchase Agreement and the Series B-1 Certificate of Designation, will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.
- 6. The shares of the Series C Preferred Stock initially issuable upon conversion of the First Closing Preferred Shares or upon exercise of the First Closing Warrants have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, when issued to and, in the case of shares issuable upon exercise of the First Closing Warrants, paid for by, you in accordance with the terms of the Securities Purchase Agreement, the Series B-1

Certificate of Designation (in the case of shares of Series C Preferred Stock issuable upon conversion of First Closing Preferred Shares), the First Closing Warrants (in the case of shares of Series C Preferred Stock issuable upon exercise of the First Closing Warrants) and the Series C Certificate of Designation, will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.

- 7. The shares of the Company's Ordinary Common Stock initially issuable upon conversion of the First Closing Preferred Shares have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, when issued in accordance with the terms of the Securities Purchase Agreement and the Series B-1 Certificate of Designation will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents 37.
- 8. The shares of the Company's Ordinary Common Stock initially issuable upon conversion of the shares of Series C Preferred Stock which are initially issuable upon conversion of First Closing Preferred Shares or upon exercise of the First Closing Warrants, have been duly authorized and reserved for issuance by all necessary corporate action of the Company, and, assuming that such shares of Series C Preferred Stock have been validly issued, when such shares of the Company's Ordinary Common Stock are issued in accordance with the terms of the Securities Purchase Agreement and the Series C Certificate of Designation, such shares of Ordinary Common Stock will be validly issued, fully paid and non-assessable and free of preemptive rights arising from the Governing Documents.
- 9. Assuming the representations and warranties of the parties thereto in the Transaction Agreements are true and assuming compliance by the parties thereto with their respective covenants and agreements set forth therein, no registration of the First Closing Preferred Shares or the First Closing Warrants under the Securities Act of 1933, as amended, is required for the purchase by you of such First Closing Preferred Shares or First Closing Warrants in the manner contemplated by the Securities Purchase Agreement. We express no opinion, however, as to when or under what circumstances any First Closing Preferred Shares or First Closing Warrants, or any securities issuable upon conversion of the First Closing Preferred Shares or upon exercise of the First Closing Warrants, may be reoffered or resold.

Our opinions are subject to:

- (a) the effect of bankruptcy, insolvency, reorganization, preference, fraudulent transfer, moratorium or other similar laws relating to or affecting the rights or remedies of creditors;
- (b) the effects of general principles of equity, whether considered in a proceeding in equity or at law (including the possible unavailability of specific performance or injunctive relief), concepts of materiality, reasonableness, good faith and fair dealing, and the discretion of the court before which a proceeding is brought;
- (c) the invalidity under certain circumstances under law or court decisions of provisions for the indemnification of or contribution to a party with respect to a liability where such indemnification or contribution is contrary to public policy; and
- (d) we express no opinion with respect to (i) consents to, or restrictions upon, governing law or jurisdiction, venue, arbitration, remedies or judicial relief; (ii) advance waivers of claims, defenses, rights granted by law, or notice, opportunity for hearing, evidentiary requirements, statutes of limitation, trial by jury or at law, or other procedural rights; (iii) waivers of broadly or vaguely stated rights; (iv) covenants not to compete; (v) provisions for exclusivity, election, cumulation or limitation of rights or remedies; (vi) provisions authorizing or validating conclusive or discretionary determinations; (vii) grants of setoff rights; (viii) provisions for the payment of attorneys' fees where such payment is contrary to law or public policy; (ix) proxies, powers of attorney, and voting agreements or trusts; (x) provisions prohibiting, restricting or requiring consent to assignment or transfer of any right or property; (xi) provisions for liquidated damages, default interest, late charges, monetary penalties, make-whole premiums or other economic remedies to the extent such provisions are deemed to constitute a penalty; and (xii) the severability, if invalid, of provisions to the foregoing effect.

We express no opinion as to United States federal or state securities laws (except as set forth in paragraph 9 with respect to federal securities laws), tax laws, antitrust or trade regulation laws (including Hart-Scott-Rodino), insolvency or fraudulent transfer laws, antifraud laws, or compliance with fiduciary duty requirements, margin regulations, FINRA rules, stock exchange rules, pension or employee benefit laws, usury laws, environmental laws, any law or regulation concerning energy or nuclear materials (including, for the avoidance of doubt, any regulation or other legal guidance of the Department of Energy or the Nuclear Regulatory Commission), any law, regulation or legal guidance concerning classified or sensitive information, export control, anti-money laundering or anti-terrorism laws (including Office of Foreign Assets Control Regulations), or laws governing foreign investments in the United States (including CFIUS) (without limiting other laws excluded by customary practice).

With your consent, we have assumed (a) that the Transaction Agreements have been duly authorized, executed and delivered by, and constitute legally valid and binding obligations of, the parties thereto other than the Company, enforceable against each of them in accordance with their respective terms, (b) that the status of the Transaction Agreements as legally valid and binding obligations of the parties is not affected by any (i) breaches of, or defaults under, agreements or instruments, (ii) violations of statutes, rules, regulations or court or governmental orders, or (iii) failures to obtain required consents, approvals or authorizations from, or make required registrations, declarations or filings with, governmental authorities, provided that we make no such assumption to the extent we have specifically opined as to such matters with respect to the Company herein; and (c) that none of the Transaction Agreements has been amended, modified, supplemented or terminated since the original date of execution thereof and that no rights of any parties thereto have been waived by any action or inaction of theirs. With your consent, for purposes of our opinions in paragraphs 6,7 and 8, we have further assumed that, at the time the shares of a particular class or series of capital stock are issued, the Company has sufficient authorized shares of that class or series.

This letter is furnished only to you and is solely for your benefit in connection with the transactions referenced in the first paragraph. This letter may not be relied upon by you for any other purpose, or furnished to, assigned to, quoted to or relied upon by any other person, firm or other entity for any purpose, without our prior written consent, which may be granted or withheld in our discretion.

Very truly yours,

- The reference to Certificate of Incorporation in subsequent opinions shall include references to the Series B-1 and Series C Certificates of Designation as filed at the First Closing. The separate references to these docs will be deleted in this list.
- In connection with the Third Closing, the references to Series B-1 Preferred Stock will be revised to refer to the Series B-2 Preferred Stock.
- To be added in any opinion to be delivered subsequent to the Charter Amendment Approval having been obtained:
- () The form of Charter Amendment attached to the Securities Purchase Agreement as Exhibit L proposed to be delivered to the Secretary of State of the State of Delaware for filing on [] for the purpose of amending Article 4 of the Certificate of Incorporation to, among other things, create the Class B Common Stock. With your consent, we have assumed that on the date hereof, the Secretary of State of the State of Delaware has accepted the filing of the Charter Amendment.
- The first three opinions will be omitted at the Second and Third Closings, as will related assumptions and references.
- In connection with any opinion letter delivered at a subsequent closing, the definition of the series of Preferred Shares will be adjusted to reflect the series and number of shares being issued.

 In connection with any opinion letter delivered at the Second Closing all references to the Warrante will be arrived. In connection with any opinion delivered at the Second Closing all references to the Warrante will be arrived.
- In connection with any opinion letter delivered at the Second Closing, all references to the Warrants will be omitted. In connection with any opinion delivered at the Third Closing, all references to the Warrants shall be to the "Third Closing Warrants".
- In connection with any opinion letter delivered at a subsequent closing, the opinions given in paragraph 4 shall be limited

to the issuance and sale of the securities being issued at such closing.

In connection with any opinion letter delivered subsequent to the Charter Amendment Approval having been obtained, references to Series C Preferred Stock shall be changed to Class B Common Stock and references to the Series C Certificate of Designation shall be changed to the "Amended Certificate of Incorporation", which shall be defined to include the Charter Amendment.

In connection with any opinion letter delivered subsequent to the Charter Amendment Approval having been obtained, paragraph 7 will be omitted.

EXHIBIT J-3

FORM OF OPINION OF MORGAN LEWIS & BOCKIUS LLP

Morgan, Lewis & Bockius LLP 1111 Pennsylvania Avenue, NW Washington, DC 20004 Tel. 202.739.3000 Fax: 202.739.3001 www.morganlewis.com

Form of Regulatory Counsel Legal Opinion – Third Closing

, 2010

Toshiba Corporation 1-1 Shibaura 1 – chome Minato-ku, Tokyo 105-8001 Japan

Babcock & Wilcox Investment Company 800 Main Street Lynchburg, VA 24502 USA

c/o Morrison & Foerster LLP Shin – Marunouchi 1 – chome Chiyoda-ku, Tokyo 100-6529 Japan

c/o Baker Botts LLP 1299 Pennsylvania Avenue, NW Washington, DC 20004-2400 USA

Re: Securities Purchase Agreement – USEC Inc., Toshiba Corporation and Babcock & Wilcox Investment Company – Third Closing

Ladies and Gentlemen:

We have acted as special counsel for <u>USEC Inc.</u> on certain regulatory matters as specified below in connection with the purchase by you of Series B-1 12.5% Preferred Stock, Series B-2 11.5% Preferred Stock and Warrants issued by USEC Inc., pursuant to the "Securities Purchase Agreement among USEC Inc., Toshiba Corporation, and Babcock & Wilcox Investment Company," dated as of May 25, 2010 (the "<u>Purchase Agreement</u>"), among USEC Inc. (the "Issuer") and Toshiba Corporation and Babcock & Wilcox Investment Company (the "Purchasers").

In rendering the opinions set forth below, we have reviewed copies of the following documents and instruments:

- (i) the Purchase Agreement;
- (ii) the Investor Rights Agreement;
- (iii) the Series B-1 Certificate of Designation;
- (iv) the Series B-2 Certificate of Designation;
- (v) the Series C Certificate of Designation;
- (vi) the Strategic Relationship Agreement;

- (vii) the Charter Amendment;
- (viii) the Foreign Ownership Restrictions Determination;
- (ix) the Warrants;

such documents and instruments identified in (ii) - (ix) above being as defined in the Purchase Agreement. Capitalized terms used but not defined herein shall have the meanings ascribed to them in the Purchase Agreement.

The documents listed in clauses (i) through (ix) above are referred to herein as the <u>"Transaction Documents."</u> In rendering the opinions set forth herein, we have, with your consent, relied only upon examination of the documents described above. As to any facts material to our opinions, we have made no independent investigation of such facts and have relied, to the extent that we deem such reliance proper, upon statements of public officials and officers or other representatives of the Issuer and the Purchasers and on the representations and warranties set forth in the Transaction Documents.

In rendering the opinions expressed below, we have assumed the legal capacity of all natural persons, the genuineness of all signatures, the authenticity of all documents submitted to us as originals, the conformity to authentic original documents of all documents submitted to us as copies, and the accuracy and completeness of all corporate, limited liability company, partnership and other entity records made available to us, which assumptions we have not independently verified.

Based upon the foregoing, and subject to the assumptions, qualifications, exceptions and limitations set forth herein, it is our opinion that:

- 1. The issuance and sale of the Preferred Stock and Warrants by the Issuer in connection with the Third Closing and the performance by the Issuer of its obligations thereunder and under the Purchase Agreement, will not:
 - (i) result in any violation by the Issuer of any Applicable Regulatory Law (as defined below),
 - (ii) result in any violation by the Issuer of any order, writ, judgment or decree known to us issued under Applicable Regulatory Laws.
 - "Applicable Regulatory Laws" means: the Atomic Energy Act of 1954, as amended, and the related rules, regulations or orders of the U.S. Nuclear Regulatory Commission ("NRC"); the U.S. Department of Energy (DOE) foreign ownership, control or influence requirements and restrictions; the Hart-Scott-Rodino Antitrust Improvements Act of 1976; and the requirements of the Committee on Foreign Investment in the United States (CFIUS) adopted pursuant to Section 721 of the Defense Production Act of 1950, as amended by the Foreign Investment and National Security Act of 2007, and as implemented by Executive Order 11858, as amended and 31 CFR Part 800.
- 2. All required Regulatory Approvals have been obtained for the issuance and sale of the Preferred Stock and Warrants by the Issuer and the performance by the Issuer of its obligations under each Transaction Document in connection with the Third Closing, as applicable.
 - "Regulatory Approvals" means any consent, clearance, approval, license, permit, registration, certification, authorization or validation of, or filing, recording or registration with, the NRC, DOE, the U.S. Department of Justice, the U.S. Federal Trade Commission, and CFIUS pursuant to any Applicable Regulatory Laws.
- 3. Other than the Regulatory Approvals set forth on Schedule B under the heading "Third Closing" that have been obtained and the Regulatory Approvals set forth in Sections 2.9 (iv) and (vi) of the Purchase Agreement that have been obtained, no Permit or order of, or filing with, any Governmental Authority is required in connection with the Third Closing and the performance by the Issuer of its obligations under the Transaction Documents in connection therewith.

The opinions expressed herein are limited to the matters discussed above; no opinion is implied or may be inferred beyond such matters.

This opinion has been prepared in accordance with the customary practice of lawyers who regularly give, and lawyers who regularly advise recipients regarding, opinions of this kind.

We express no opinion as to the laws of any jurisdiction other than Applicable Regulatory Laws.

This letter is rendered as of the date set forth above. We expressly disclaim any obligation to update this letter after such date.

This letter is given solely for your benefit in connection with the transactions contemplated by the Transaction Documents and may not be furnished to, or relied upon by, any other person or for any other purpose without our prior written consent.

Sincerely,

EXHIBIT J-4

FORM OF OPINION OF SPECIAL COUNSEL TO INVESTORS

EXHIBIT J-4

OPINION OF SPECIAL COUNSEL TO INVESTOR

- 1) The Investor is a corporation under , with corporate power and authority to enter into the [Transaction Agreements] and perform its obligations thereunder. With your consent, based solely on certificates from public officials, we confirm that the Investor is validly existing and in good standing under the laws of the State of Delaware.
- 2) The execution, delivery and performance of the [Transaction Agreements] have been duly authorized by all necessary corporate action of the Investor, and the [Transaction Agreements] have been duly executed and delivered by the Investor.
- 3) Each of the [Transaction Agreements] constitutes a legally valid and binding obligation of the Investor, enforceable against the Investor in accordance with its terms.
- 4) The execution and delivery of the [Transaction Agreements] by the Investor, the performance by the Investor of its obligations thereunder and the purchase at the First Closing of shares of Series B-1 12.75% Preferred Stock and the Warrants pursuant to the Securities Purchase Agreement by the Investor on the date hereof do not:
 - (a) violate the provisions of the [governing documents]38;
 - (b) violate any United States federal or New York statute, rule or regulation applicable to the Investor, or the Delaware General Corporation Law²⁰ other than U.S securities laws and such other laws as may customarily be excepted from opinions of this type; or
 - (c) result in the breach of or a default under any [as to B&W, credit or debt instrument of B&W or its Affiliates that is filed as an exhibit to the Form 10 of Babcock & Wilcox Company, as amended prior to the date hereof]
- to include the Investor's charter documents and bylaws, unless, for Toshiba, the nature of its jurisdiction would dictate otherwise. For B&W, this opinion should also cover the form of restated certificate of incorporation and restated bylaws.

Appropriate analog for Toshiba to be provided.

EXHIBIT K

FORM OF PRESS RELEASE

FOR IMMEDIATE RELEASE:

May 25, 2010

Toshiba and Babcock & Wilcox Announce Definitive Agreement to Invest in USEC

Goal Is to Enable Successful Deployment of Leading Edge Uranium Enrichment Technology for Commercial Nuclear Power
Plants

-Boost for Nuclear Renaissance; Support for Ohio-based American Centrifuge Project-

WASHINGTON AND TOKYO – (May 25, 2010) In a sign that the future of nuclear energy is bright, USEC Inc. (NYSE: USU), a global nuclear fuel and technology company, today announced that nuclear industry leaders Toshiba Corporation (TOKYO: 6502) and Babcock & Wilcox Investment Company, an affiliate of The Babcock & Wilcox Company (B&W), have signed a definitive agreement to make an investment in USEC over three phases, each of which is subject to satisfaction of specific preconditions, including regulatory approvals. The \$200 million investment, shared equally between Toshiba and B&W, will strengthen the deployment of the American Centrifuge Plant and create key new business opportunities throughout the nuclear fuel cycle. Both companies, with their deep experience in the nuclear field, consider the investment agreement as an important strategic initiative.

"We have decided to make this investment in American know-how and American technology in order to produce more uranium fuel for the growing worldwide nuclear power market with high confidence in USEC as a leading supplier of low enriched uranium. Toshiba has intensified its front-end supply chain by executing various programs. Along with these efforts, the investment in USEC

will strengthen the nuclear fuel supply chain," said Yasuharu Igarashi, corporate senior vice president of Toshiba. "The nuclear renaissance is moving forward and this investment will help power its growth by securing the supply of uranium fuel for existing and potential customers."

Brandon Bethards, president and CEO of B&W, said, "This agreement will allow B&W to explore broader energy opportunities globally through expanded strategic relationships with Toshiba and USEC, as well as further enhance our already-strong relationship with the U.S. Department of Energy. This initiative further solidifies our position as a key contributor to a successful nuclear renaissance."

USEC Inc. is a leading supplier of enriched uranium fuel for commercial nuclear power plants. USEC 2009 revenues were more than \$2 billion.

"This is a new day for USEC in the era of nuclear power, and we're heartened by these votes of confidence in USEC and the American Centrifuge Project from two global leaders in the nuclear power industry," said John Welch, USEC president and CEO. "The agreement accelerates our

-more-

Toshiba and Babcock & Wilcox Announce Investments in USEC

Page 2 of 2

recent momentum and addresses a key issue in our sustained drive to deploy the innovative American Centrifuge technology. The strategic relationship between our companies will also create new business opportunities across the entire nuclear fuel cycle."

The agreement comes at a time when the federal government is considering issuing a loan guarantee to support completion of USEC's American Centrifuge Project, a next-generation uranium enrichment facility to be built in Piketon, Ohio. The project is based on the use of highly efficient centrifuges to produce uranium fuel. USEC is operating a cascade of centrifuge machines in a commercial plant configuration, collecting data and experience in preparation for transitioning into full commercial operation. USEC employees in several states are also involved in the design, development, testing and manufacturing of the centrifuge machines. If successful in obtaining a loan guarantee, the American Centrifuge Project could create at the peak of construction nearly 8,000 jobs, with almost half located in Ohio.

The investment is subject to applicable regulatory reviews.

###

Contacts:

USEC Media Paul Jacobson (301) 564-3399 jacobsonp@usec.com

Investors
Steven Wingfield
(301) 564-3354

Toshiba

Hiroki Yamazaki +81-3-3457-2105 www.toshiba.co.jp/contact/media.htm Babcock & Wilcox

Jud Simmons (434) 522-6462 hjsimmons@babcock.com

Jay Roueche (281) 870-5462

EXHIBIT L

FORM OF CHARTER AMENDMENT

FORM OF

CERTIFICATE OF AMENDMENT

TO THE

CERTIFICATE OF INCORPORATION

OF

USEC INC.

USEC Inc., a corporation organized and existing under the General Corporation Law of the State of Delaware (the 'Corporation'), in accordance with the provisions of Section 242 thereof, DOES HEREBY CERTIFY:

FIRST: By unanimous written consent, in lieu of a meeting, pursuant to Section 141(f) of the General Corporation Law of the State of Delaware, on May 24, 2010 the Board of Directors of the Corporation duly adopted resolutions setting forth a proposed amendment to the Certificate of Incorporation of the Corporation, as amended, declaring said amendment to be advisable and directing such amendment to be submitted to the stockholders of the Corporation for approval at a special meeting of stockholders. The resolution setting forth the proposed amendment is as follows:

RESOLVED, that Article FOURTH of the Certificate of Incorporation be, and it hereby is, amended in its entirety to read as follows, subject to the approval of the stockholders of the Corporation:

"FOURTH: Capital Stock

- A. <u>Authorized Shares</u>. The total number of shares of stock which the Corporation is authorized to issue is 500,000,000 shares. The authorized capital stock is divided into 25,000,000 shares of preferred stock, par value \$1.00 per share ("Preferred Stock"), 350,000,000 shares of common stock, par value \$.10 per share ("Common Stock") and 125,000,000 shares of Class B Common Stock, par value \$.10 per share ("Class B Common Stock"). Subject to the rights of any series of Preferred Stock, the number of authorized shares of any of the Common Stock, Class B Common Stock or Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority in voting power of the stock of the Corporation entitled to vote thereon irrespective of the provisions of Section 242(b)(2) of the DGCL (or any successor provision thereto), and no vote of the holders of any of the Common Stock, Class B Common Stock or Preferred Stock voting separately as a class shall be required therefor.
- B. <u>Preferred Stock</u>. The shares of Preferred Stock of the Corporation may be issued from time to time in one or more series thereof, the shares of each series thereof to have such voting powers, full or limited, or no voting powers, and such designations, preferences and relative, participating, optional or other special rights, and qualifications, limitations or restrictions thereof, as are stated and expressed herein or in the resolution or resolutions providing for the issue of such series, adopted by the Board of Directors as hereinafter provided.

Authority is hereby expressly granted to the Board of Directors of the Corporation, subject to the provisions of this Article FOURTH and to the limitations prescribed by the DGCL, to authorize the issue of one or more series of Preferred Stock and with respect to each such series to fix by resolution or resolutions providing for the issue of such series the voting powers, full or limited, if any, of the shares of such series and the designations, preferences and relative, participating, optional or other special rights, and qualifications, limitations or restrictions thereof. The authority of the Board of Directors with respect to each series thereof shall include, but not be limited to, the determination or fixing of the following:

- (i) the maximum number of shares to constitute such series, which may subsequently be increased or decreased by resolution of the Board of Directors unless otherwise provided in the resolution providing for the issue of such series, the distinctive designation thereof and the stated value thereof if different than the par value thereof;
- (ii) the dividend rate of such series, the conditions and dates upon which such dividends shall be payable, the relation which such dividends shall bear to the dividends payable on any other class or classes of stock or any other series of any class of stock of the Corporation, and whether such dividends shall be cumulative or noncumulative:

- (iii) whether the shares of such series shall be subject to redemption, in whole or in part, and, if made subject to such redemption, the times, prices and other terms and conditions of such redemption, including whether or not such redemption may occur at the option of the Corporation or at the option of the holder or holders thereof or upon the happening of a specified event;
 - (iv) the terms and amount of any sinking fund established for the purchase or redemption of the shares of such series;
- (v) whether or not the shares of such series shall be convertible into or exchangeable for shares of any other class or classes of any stock or any other series of any class of stock of the Corporation, and, if provision is made for conversion or exchange, the times, prices, rates, adjustments, and other terms and conditions of such conversion or exchange;
 - (vi) the extent, if any, to which the holders of shares of such series shall be entitled to vote with respect to the election of directors or otherwise;
 - (vii) the restrictions, if any, on the issue or reissue of any additional Preferred Stock;
 - (viii) the rights of the holders of the shares of such series upon the dissolution of, or upon the subsequent distribution of assets of, the Corporation; and
- (ix) the manner in which any facts ascertainable outside the resolution or resolutions providing for the issue of such series shall operate upon the voting powers, designations, preferences, rights and qualifications, limitations or restrictions of such series.
- C. Common Stock and Class B Common Stock. The Common Stock and Class B Common Stock shall have the following powers, preferences and rights and qualifications, limitations and restrictions:
 - (i) As used in this Section C. of this Article FOURTH, the following terms shall have the following meanings:
- (a) "Affiliate" shall mean any Person controlling, controlled by or under common control with any other Person. For purposes of this definition, "control" (including "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of securities, partnership or other ownership interests, by contract or otherwise.
- (b) "Aggregate Outstanding Value" shall mean, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock, plus, (2) for each share of Series C Preferred Stock then held by the Permitted Holders, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price (as defined in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable) upon which the Permitted Holders' acquisition of such share was calculated, plus, (3) for each share of Common Stock then held by the Permitted Holders, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Common Stock purchased in the market, the Base Price upon which the Permitted Holder's acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock, which have been added to the Liquidation Preference.
- (c) "Automatic Redemption" shall mean an automatic redemption pursuant to Section 7(g) of the Series B-1 Certificate of Designation subsequent to a Conversion Election (as defined in the Series B-1 Certificate of Designation), Section 8(c) of the Series B-1 Certificate of Designation or Section 8(c) of the Series B-2 Certificate of Designation.
- (d) "Automatic Redemption Adjustment" shall mean, for purposes of determining the Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value, that if an Automatic Redemption has been effected prior to the date of determining such values, (1) the aggregate amount of the Liquidation Preference, as of the date of redemption, of a Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Aggregate Outstanding Value and Permitted Holder Outstanding Value and (2) the aggregate amount of the Liquidation Preference, as of the date of redemption, of such Permitted Holder's Series B Preferred Stock (excluding shares issued as a Dividend) redeemed in connection with the Automatic Redemption shall be added to such Permitted Holder's Original Issue Value and Permitted Holder Original Issue Value; provided, however, that, if at any time after any Automatic Redemption, such Permitted Holder's Deemed Holder Percentage is less than 8%, then such adjustment to the

Aggregate Outstanding Value, the Permitted Holder Outstanding Value, the Original Issue Value and the Permitted Holder Original Issue Value shall not be made.

- (e) "B&W" shall mean Babcock & Wilcox Investment Company, a Delaware corporation.
- (f) "Beneficially Own" shall mean "beneficially own" as defined in Rule 13d-3 promulgated under Section 13(d) of the Exchange Act or any successor provisions thereto, and "Beneficial Ownership" shall have a correlative meaning.
 - (g) "Change of Control" shall mean the occurrence of any of the following:
- (1) any Person shall Beneficially Own, directly or indirectly, through a merger, business combination, purchase, or other transaction or series of transactions, shares of the Corporation's capital stock entitling such Person at such time to exercise 50% or more of the total voting power of the issued and outstanding capital stock having ordinary voting power to elect a majority of the Board of Directors, other than as a result of an acquisition of such stock by the Corporation, any of the Corporation's Subsidiaries or any of the Corporation's employee benefit plans (for purposes of this subsection (1), "Person" shall include any syndicate or group that would be deemed to be a "person" under Section 13(d)(3) of the Exchange Act);
- (2) the Corporation (A) merges or consolidates with or into any other Person, another Person merges with or into the Corporation, or the Corporation conveys, sells, transfers or leases all or substantially all of the Corporation's assets to another Person or (B) engages in any recapitalization, reclassification or other transaction in which all or substantially all of the Common Stock is exchanged for or converted into cash, securities or other property, in each case other than a merger or consolidation:
 - (a) that does not result in a reclassification, conversion, exchange or cancellation of the Corporation's outstanding Common Stock;
 - (b) that is effected solely to change the Corporation's jurisdiction of incorporation and results in a reclassification, conversion or exchange of outstanding shares of the Common Stock solely into shares of any class or series of common stock of the surviving entity; or
 - (c) where the issued and outstanding capital stock having voting power to vote generally to elect a majority of the Board of Directors outstanding immediately prior to such transaction is converted into or exchanged for such voting stock of the surviving or transferee Person constituting a majority of the outstanding shares of such voting stock of such surviving or transferee Person (immediately after giving effect to such issuance).
- (h) "Closing Deadline Failure" shall mean, unless waived in writing (1) by the Corporation if such Closing Deadline Failure is as a result of breach by a Permitted Holder, (2) by the Permitted Holders if such Closing Deadline Failure is as a result of breach by the Corporation, or (3) by the Permitted Holders and the Corporation if such Closing Deadline Failure is not as a result of a breach by the Permitted Holders or the Corporation, either, (A) with respect to the Second Closing (as defined in the Securities Purchase Agreement), that the Second Closing shall not have occurred by June 30, 2011 and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.2 thereof, or (B) with respect to the Third Closing (as defined in the Securities Purchase Agreement), that the Third Closing shall not have occurred by the Third Closing Termination Date (as defined in the Securities Purchase Agreement) and the Securities Purchase Agreement shall have been terminated pursuant to Section 10.3 thereof.
- (i) "Code" shall mean the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder by the Internal Revenue Service.
- (j) "Deemed Holder Percentage" shall mean, as to any Permitted Holder, the percentage resulting from the following calculation, (1)(A) the number of shares of Common Stock equal to the quotient of (w) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on the outstanding shares of Series B Preferred Stock held by such Permitted Holder from and including the immediately preceding Dividend Payment Date (as defined in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable) to, but excluding, the date of conversion and (x) the Base Price for the date of such calculation, plus (B) the number of outstanding of shares of (y) Series C Preferred Stock multiplied by 1000 plus, (z) if then outstanding, Class B Common Stock, in each case held by such Permitted Holder divided by (2)(A) the total number of shares of Common Stock equal to the quotient of (v) the Liquidation Preference plus an amount per share equal to the accrued but unpaid Dividends not previously added to the Liquidation Preference on all outstanding shares of Series B

Preferred Stock from and including the immediately preceding Dividend Payment Date to, but excluding, the date of conversion and (w) the Base Price for the date of such calculation, *plus* (B) the total number of all outstanding shares of (x) Series C Preferred Stock multiplied by 1000*plus* (y) if then outstanding, Class B Common Stock, *plus* (z) Common Stock.

- (k) "Dividend" shall have the meaning ascribed to it in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable.
 - (1) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (m) "Governmental Authority" shall mean any foreign governmental authority, the United States of America, any state of the United States and any political subdivision of any of the foregoing, and any agency, instrumentality, department, commission, board, bureau, central bank, authority, court, arbitral body or other tribunal, in each case whether executive, legislative, judicial, regulatory or administrative, having jurisdiction over any of the Permitted Holders, the Corporation, any of the Corporation's Subsidiaries or their respective Property.
 - (n) "Initial Liquidation Preference" shall mean \$1,000 per share of Series B Preferred Stock.
- (o) "Investor Rights Agreement" shall mean that certain Investor Rights Agreement, dated [—], 2010, among the Corporation, Toshiba and B&W, as amended from time to time.
- (p) "Liquidation Preference" shall have the meaning ascribed to it in the Series B-1 Certificate of Designation, the Series B-2 Certificate of Designation or the Series C Certificate of Designation, as applicable.
- (q) "Original Issuance Date" shall mean, with respect to each share of Class B Common Stock issued to the Permitted Holders, the date on which such share was issued by the Corporation.
- (r) "Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, the aggregate Initial Liquidation Preference of all the shares of Series B Preferred Stock issued to the Permitted Holders excluding those shares issued as a Dividend.
 - (s) "Permit" shall mean any approval, authorization, certificate, consent, license or permit of or from any Governmental Authority.
- (t) "Permitted Holder Material Breach" shall mean a material breach of the Securities Purchase Agreement or the Investor Rights Agreement by any Permitted Holder.
- (u) "Permitted Holder Original Issue Value" shall mean, subject to the Automatic Redemption Adjustment, if any, for any Permitted Holder, the aggregate Initial Liquidation Preference of all shares of Series B Preferred Stock issued to such Permitted Holder excluding those shares issued as a Dividend.
- (v) "Permitted Holder Outstanding Value" shall mean, as to any Permitted Holder, at any time and from time to time and subject to the Automatic Redemption Adjustment, if any, (1) the Original Issue Value of all of the outstanding shares of Series B Preferred Stock then held by such Permitted Holder, plus, (2) for each share of Series C Preferred Stock then held by a Permitted Holder, excluding those shares of Series C Preferred Stock issued upon exercise of the Warrants, the Base Price upon which such Permitted Holder's acquisition of such share was calculated, plus, (3) for each share of Common Stock then held by such Permitted Holder, excluding those shares of Class B Common Stock issued upon exercise of the Warrants or Common Stock purchased in the market, the Base Price (as defined in the Series B-1 Certificate of Designation or the Series B-2 Certificate of Designation, as applicable) upon which such Permitted Holder's acquisition of such share was calculated, plus (4) the aggregate amount of accrued and unpaid Dividends on outstanding shares of Series B Preferred Stock, which have been added to the Liquidation Preference pursuant to Section 5(a) of the Series B-1 Certificate of Designation or Series B-2 Certificate of Designation, as applicable.
- (w) "Permitted Holders" shall mean (1) [Toshiba U.S. Co.] and its Wholly-Owned Affiliates, (2) B&W and its Wholly-Owned Affiliates, (3) a special purpose entity jointly and wholly controlled by Toshiba and B&W and (4) Westinghouse Electric Company, LLC, to the extent it is controlled by Toshiba or a Permitted Holder described under (1) above; provided, however, that each Permitted Holder must be a U.S. Person.
- (x) "Person" shall mean any individual, corporation, company, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization, Governmental Authority or any other entity.

- (y) "Property" shall mean any interest in any kind of property or asset, whether real, personal or mixed, tangible or intangible.
- (z) "Qualified Director" shall mean any individual reasonably acceptable to the Nominating and Governance Committee of the Board of Directors.
- (aa) "Regulatory Bodies" shall mean the DOE and the U.S. Nuclear Regulatory Commission, and any successor Governmental Authorities thereto.
- (bb) "Securities Purchase Agreement" shall mean that certain Securities Purchase Agreement, dated May 25, 2010, among the Corporation, Toshiba and B&W, as amended from time to time.
 - (cc) "Series B Preferred Stock" shall mean the Series B-1 12.75% Preferred Stock together with the Series B-2 11.5% Preferred Stock.
- (dd) "Series B-1 12.75% Preferred Stock" shall mean the series of preferred stock of the Corporation, par value \$1.00 per share, designated as "Series B-1 12.75% Convertible Preferred Stock."
- (ee) "Series B-1 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-1 12.75% Convertible Preferred Stock of the Corporation as filed with the Secretary of State of the State of Delaware.
- (ff) "Series B-2 11.5% Preferred Stock" shall mean the series of preferred stock of the Corporation, par value \$1.00 per share, designated as "Series B-2 11.5% Convertible Preferred Stock."
- (gg) "Series B-2 Certificate of Designation" shall mean that certain Certificate of Designation of Series B-2 11.5% Convertible Preferred Stock of the Corporation as filed with the Secretary of State of the State of Delaware.
- (hh) "Series C Certificate of Designation" shall mean that certain Certificate of Designation of Series C Convertible Participating Preferred Stock of the Corporation as filed with the Secretary of State of the State of Delaware.
- (ii) "Series C Preferred Stock" shall mean the series of preferred stock of the Corporation, par value \$1.00 per share, designated as "Series C Convertible Participating Preferred Stock."
- (jj) "Subsidiary" of any Person shall mean any corporation, partnership, joint venture, limited liability company, trust or estate of which (or in which) more than 50% of (1) the issued and outstanding capital stock having ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether at the time capital stock of any other class or classes of such corporation shall or might have voting power upon the occurrence of any contingency), (2) the interest in the capital or profits of such partnership, joint venture or limited liability company or (3) the beneficial interest in such trust or estate is at the time directly or indirectly owned or controlled by such Person, by such Person and one or more of its other Subsidiaries or by one or more of such Person's other Subsidiaries. Notwithstanding the foregoing, American Centrifuge Manufacturing, LLC, a Delaware limited liability company, shall not be considered a Subsidiary of B&W or the Corporation.
- (kk) "Third-Party Transfer" shall mean an irrevocable Transfer in compliance with Section C.(viii) of this Article FOURTH of all legal ownership, Voting Control and Beneficial Ownership of any share or shares of Class B Common Stock to a Person other than a Permitted Holder or its Affiliates.
 - (ll) "Toshiba" shall mean Toshiba Corporation, a corporation organized under the laws of Japan.
- (mm) "Transfer" shall have the meaning ascribed to such term in Article ELEVENTH. Notwithstanding the preceding sentence or Article ELEVENTH, for purposes of this Article FOURTH, "Transfer" shall include the transfer of, or entering into any agreement, arrangement or understanding with respect to, Voting Control over a share or shares of Class B Common Stock.
- (nn) "<u>U.S. Person</u>" shall mean any person that is treated as a "United States Person" under Code Section 7701(a)(30) and that provides an IRS Form W-9 (or successor form), evidencing a complete exemption from United States withholding tax (including backup withholding tax), on or before the time at which it acquires securities pursuant to this Certificate of Incorporation.

(00) "Voting Control" shall mean, with respect to a share or shares of Class B Common Stock, the power, whether exclusive or shared, revocable or irrevocable, to vote or direct the voting of such share or shares of Class B Common Stock, by proxy, voting agreement or otherwise.

- (pp) "Warrants" shall mean those warrants to purchase Class B Common Stock or Series C Preferred Stock issued by the Corporation to the Permitted Holders pursuant to the Securities Purchase Agreement.
- (qq) "Wholly-Owned Affiliate" shall mean, as to any Person, any Affiliate that, directly or indirectly, is wholly-owned and controlled (other than by contract) by a Person, or any other Affiliate to which the Corporation, in its sole discretion, consents.

(ii) Voting Rights.

- (a) The holders of the shares of Class B Common Stock shall not be entitled to vote, except as otherwise provided herein or required by applicable law.
- (b) Effective at the time no Series B Preferred Stock shall be outstanding, the number of directors constituting the Board of Directors shall be increased by two Persons and the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of any other Common Stock and any other series of Preferred Stock, shall be entitled to elect two Qualified Directors to the Board of Directors (each such director, an "<u>Initial Investor Director</u>") until the earliest to occur of (i) a Closing Deadline Failure as a result of a Permitted Holder Material Breach at a time when the Securities Purchase Agreement is terminable pursuant to Sections 10.2(d) and 10.3(d) thereof, (ii) a Change of Control or (iii) such time as the Permitted Holders' Aggregate Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of the Original Issue Value or, (y) after December 31, 2016, 50% of the Original Issue Value, whereupon at any such time (A) the right of such holders to elect the Initial Investor Directors shall cease, (B) the term of office of the Initial Investor Directors shall immediately and automatically terminate, (C) the Initial Investor Directors will no longer be qualified to serve and (D) the number of directors constituting the Board of Directors shall be immediately and automatically reduced by two Persons.
- (c) Effective as of the first Original Issuance Date and at such time as when the Permitted Holders do not have the right to elect the Initial Investor Directors pursuant to Section C.(ii)(b)(iii) of this Article FOURTH and any Permitted Holder's Permitted Holder Outstanding Value is greater than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, the number of directors constituting the Board of Directors shall be increased by one Person and the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of Common Stock and any other series of Preferred Stock, shall be entitled to elect one Qualified Director to the Board of Directors (such director, the "Investor Director") until the earliest to occur of (i) an event described in Section C.(ii)(b)(i) or (ii) of this Article FOURTH or (ii) such time as each Permitted Holder's Permitted Holder Outstanding Value is equal to or less than (x) prior to or on December 31, 2016, 75% of such Permitted Holder's Permitted Holder Original Issue Value or (y) after December 31, 2016, 50% of such Permitted Holder's Permitted Holder Original Issue Value, whereupon at any such time (A) the right of the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock to elect the Investor Director shall cease, (B) the term of office of the Investor Director shall immediately and automatically terminate, (C) the Investor Director will no longer be qualified to serve and (D) the number of directors constituting the Board of Directors shall be immediately and automatically reduced by one Person.
- (d) For the avoidance of doubt, except for the increase or decrease in the number of directors provided for herein, nothing in this Section C.(ii) of this Article FOURTH shall prohibit the Board of Directors from fixing the number of directors constituting the Board of Directors pursuant to the By-Laws.
- (e) Subject to the provisions of this Section C.(ii) of this Article FOURTH, each Initial Investor Director or the Investor Director, as applicable, shall serve until the next annual meeting of the stockholders of the Corporation and until his or her successor is elected and qualified in accordance with this Section C. (ii) of this Article FOURTH and the By-Laws, unless any such Initial Investor Director or the Investor Director, as applicable, is earlier removed in accordance with the By-Laws, resigns or is otherwise unable to serve; provided, however, that only the holders of a majority of the voting power of the outstanding Class B Common Stock and the Series C Preferred Stock may remove any such Initial Investor Director or the Investor Director, as applicable, without cause at any time, and the holders of a majority of the voting power of the outstanding shares of the capital stock of the Corporation entitled to vote on the matter may remove any such Initial Investor Director or the Investor Director, as applicable, with cause at any time. Subject to the provisions of this Section C.(ii) of this Article FOURTH, in the event any Initial Investor Director or the Investor Director, as

applicable, is removed, resigns or is unable to serve as a member of the Board of Directors, the holders of a majority of the voting power of the outstanding Class B Common Stock and Series C Preferred Stock, voting together as a separate class to the exclusion of the holders of any other Common Stock and any other series of Preferred Stock, shall have the right to fill such vacancy. Each Initial Investor Director or the Investor Director, as applicable, may only be elected to the Board of Directors by the holders of the Class B Common Stock and Series C Preferred Stock in accordance with this Section C.(ii) of this Article FOURTH, and each such Initial Investor Director's or the Investor Director's seat, as applicable, shall otherwise remain vacant.

- (f) Each holder of Common Stock shall be entitled to one vote for each share of Common Stock held of record by such holder as of the applicable record date on any matter that is submitted to a vote of the stockholders of the Corporation; provided, however, that, except as otherwise required by law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock) that relates solely to the terms of one or more outstanding series of Preferred Stock or Class B Common Stock if the holders of such affected class or series are entitled, either separately or together with the holders of one or more other such class or series, to vote thereon pursuant to this Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock) or pursuant to the DGCL.
- (g) Notwithstanding Section C.(ii)(a) of this Article FOURTH, the holders of Series C Preferred Stock and Class B Common Stock shall be entitled to vote together with the holders of Common Stock (and any other class or series of capital stock entitled to vote on the matter with the Common Stock) as a single class with respect to any transactions involving a merger of the Corporation or sale of substantially all of the Corporation's assets, which must be submitted to the Corporation's stockholders pursuant to the DGCL; provided, however, that each holder of Class B Common Stock shall be entitled to (A) one vote for each outstanding share of Class B Common Stock held of record by such holder as of the applicable record date, but only to the extent that the aggregate voting power of all of the outstanding Series C Preferred Stock and Class B Common Stock does not exceed 20% of the total voting power of all outstanding shares of all classes and series of capital stock entitled to vote thereon or (B) if pursuant to clause (A) the aggregate voting power of all of the outstanding Series C Preferred Stock and Class B Common Stock would exceed 20% of the total voting power of all outstanding shares of all classes and series of capital stock entitled to vote on the matter, such fraction of one vote for (i) each one-one thousandth (1/1000) of a share of Series C Preferred Stock and (ii) each share of Class B Common Stock held of record by such holder as of the applicable record date such that the aggregate voting power of all of the outstanding Series C Preferred Stock and Class B Common Stock equaled 20% of the total voting power of all outstanding shares of all classes and series of capital stock entitled to vote thereon.
- (h) Notwithstanding Section C.(ii)(a) of this Article FOURTH, the vote or consent of the holders of at least a majority of the outstanding shares of Class B Common Stock, voting together as a separate class to the exclusion of the holders of the Common Stock and the Preferred Stock then outstanding and entitled to vote thereon, given in person or by proxy, either in writing without a meeting or by vote at any meeting called for the purpose, shall be necessary for effecting or validating any amendment, alteration or repeal of any provision of this Certificate of Incorporation (by merger, consolidation or otherwise) so as to adversely affect any of the powers, preferences, qualifications, limitations, restrictions and relative participating, optional or other rights of Class B Common Stock.
- (iii) <u>Dividends and Distributions</u>. Subject to applicable law and the rights, if any, of the holders of any outstanding series of Preferred Stock or any class or series of capital stock having a preference over or the right to participate with the Common Stock and Class B Common Stock with respect to the payment of dividends, dividends may be declared by the Board of Directors and paid on the Common Stock and Class B Common Stock out of the assets of the Corporation that are by law available therefor at such times and in such amounts as the Board of Directors in its discretion shall determine. Except as otherwise required by the DGCL, in any circumstance where the Corporation may declare dividends or otherwise make distributions (including, without limitation, any distribution on liquidation, dissolution or winding-up of the Corporation) on the Common Stock or Class B Common Stock, the Corporation shall declare the same per share dividends or make the same per share distributions, as the case may be, on such other class of stock; provided, however, that if any such dividends or distributions are declared with respect to the Common Stock in the form of additional shares of Common Stock (or rights to acquire Common Stock), such dividends or distributions are declared with respect to Class B Common Stock in the form of an equivalent number of shares of Class B Common Stock (or rights to acquire Class B Common Stock), such dividends or distributions are declared with respect to Class B Common Stock in the form of an equivalent number of shares of Common Stock (or rights to acquire Class B Common Stock), such dividends or distributions shall be made with respect to the Common Stock in the form of an equivalent number of shares of Common Stock (or rights to acquire Common Stock).
- (iv) Voluntary or Involuntary Liquidation. Subject to the rights of the holders of any series of Preferred Stock outstanding at any time, in the event of any liquidation, dissolution or winding up of the affairs of the Corporation, whether voluntary or involuntary, holders of the Common Stock and the holders of Class B Common Stock shall be entitled

to share equally, on a per share basis, in all assets of the Corporation of whatever kind available for distribution to the holders of the Common Stock and Class B Common Stock.

- (v) <u>Subdivision or Combination</u>. If the Corporation in any manner subdivides or combines the outstanding shares of any of the Common Stock, Class B Common Stock or Series C Preferred Stock, then the outstanding shares of the Common Stock, Class B Common Stock or Series C Preferred Stock, as applicable, will be subdivided or combined in the same manner.
- (vi) Equal Status. Except as expressly provided in this Article FOURTH, shares of Common Stock and Class B Common Stock shall have the same rights, powers, preferences and restrictions and rank equally, share ratably and be identical in all respect as to all matters. In any merger, consolidation, reorganization or other business combination, the consideration received per share by the holders of the Common Stock and the holders of the Class B Common Stock in such merger, consolidation, reorganization or other business combination shall be identical; provided, however, that if such consideration consists, in whole or in part, of shares of capital stock of, or other equity interests in, the Corporation or any other corporation, partnership, limited liability company or other entity, then the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of such shares of capital stock or other equity interests may differ to the extent that the designation and the powers, preferences and relative, participating, optional and other rights and the qualifications, limitations and restrictions of the shares of Common Stock and the Class B Common Stock differ as provided herein (including, without limitation, with respect to the voting rights and conversion provisions hereof) if and to the extent necessary due to regulatory requirements or restrictions applicable to the entity surviving such merger, consolidation, reorganization or other business combination that are similar in nature to those applicable to the Corporation; and provided, further, that if the holders of the Common Stock or the holders of the Class B Common Stock are granted the right to elect to receive one of two or more alternative forms of consideration, the foregoing provision shall be deemed satisfied if holders of the other class are granted identical election rights, subject to the preceding proviso.

(vii) Automatic Conversion.

- (a) Subject to Section C.(viii) of this Article FOURTH, a share of Class B Common Stock shall be automatically converted, without any action on the part of the Corporation (other than the subsequent exchange of Class B Common Stock certificates for Common Stock certificates or, in the case of uncertificated shares of Class B Common Stock, upon receipt of proper transfer instructions from the registered holder of the shares of Class B Common Stock or by his, her or its attorney lawfully constituted in writing, and upon payment of all necessary transfer taxes and compliance with appropriate procedures for transferring shares in uncertificated form), or any holder of Class B Common Stock or any other Person, into one fully paid and nonassessable share of Common Stock upon a Third-Party Transfer of such share.
- (b) In the event of any automatic conversion pursuant to the terms of Section C.(vii)(a) of this Article FOURTH, the conversion shall be deemed to have been effected upon such Third-Party Transfer (the "Class B Common Stock Automatic Conversion Time,"). At Class B Common Stock Automatic Conversion Time, the certificates that represented the shares of Class B Common Stock that were so converted immediately prior to such conversion (the "Converted Class B Common Stock") shall, automatically and without further action, represent the same number of fully paid and non-assessable shares of Common Stock. Permitted Holders of the Converted Class B Common Stock shall deliver their certificates, duly endorsed in blank or accompanied by proper instruments of transfer, in form reasonably satisfactory to the Corporation, duly executed by such Permitted Holder or such Permitted Holder's authorized attorney to the principal office of the Corporation (or such other office or agency (including the transfer agent, if applicable) of the Corporation as it may designate by notice in writing to the registered Permitted Holder at the address of such Permitted Holder appearing on the books of the Corporation), together with a written notice stating the name or names (with addresses) and denominations in which the certificate or certificates representing such shares of Common Stock are to be issued and including instructions for delivery thereof. Upon such delivery, the Corporation or its agent shall promptly issue and deliver at such stated address to such holder of shares of Common Stock a certificate or certificates representing the number of shares of Common Stock to which such holder is entitled by reason of such conversion, and shall cause such shares of Common Stock to be registered in the name of such holder. The Person entitled to receive the shares of Common Stock issuable upon such conversion shall be treated for all purposes as the record holder of such shares of Common Stock at and as of Class B Common Stock Automatic Conv

(viii) Restrictions.

(a) Notwithstanding anything in this Certificate of Incorporation to the contrary and unless the

Corporation, in its sole discretion, otherwise agrees in writing, Permitted Holders may not transfer shares of Class B Common Stock if such Transfer would require approvals from or filings with any Regulatory Body in order not to adversely affect the Permits or regulatory status of the Corporation or its Subsidiaries, unless such approvals and/or filings have been made and received; <u>provided, however</u>, this Section C.(viii)(a) of this Article FOURTH shall not apply to any transfer where the transferee received Common Stock pursuant to the terms hereof.

- (b) Notwithstanding anything in this Certificate of Incorporation to the contrary and unless the Corporation, in its sole discretion, otherwise agrees in writing, the conversion of Class B Common Stock for Common Stock shall also be subject to the requirements of Section 9.2 of the Securities Purchase Agreement
 - (c) Any purported conversion or Transfer of Class B Common Stock in violation of these restrictions shall be null and voidab initio.
- (ix) <u>Legends</u>. All certificates representing shares of Class B Common Stock shall bear a legend or other restriction substantially to the following effect (it being agreed that if such shares are not certificated, other appropriate restrictions shall be implemented to give effect to the following).

"THIS SECURITY WAS ORIGINALLY ISSUED IN A TRANSACTION EXEMPT FROM REGISTRATION UNDER THE UNITED STATES SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR AS MAY BE HELD BY A PERSON DEEMED AN "AFFILIATE" (AS DEFINED IN RULE 144 UNDER THE SECURITIES ACT) OF THE ISSUER OF THIS SECURITY, AND THIS SECURITY MAY NOT BE OFFERED, SOLD OR OTHERWISE TRANSFERRED IN THE ABSENCE OF SUCH REGISTRATION OR AN APPLICABLE EXEMPTION THEREFROM. THE HOLDER OF THIS SECURITY AGREES FOR THE BENEFIT OF THE COMPANY THAT (A) THIS SECURITY MAY BE OFFERED, RESOLD, PLEDGED OR OTHERWISE TRANSFERRED, ONLY (I) IN A TRANSACTION NOT INVOLVING A PUBLIC OFFERING, (II) PURSUANT TO ANY OTHER EXEMPTION FROM THE REGISTRATION REQUIREMENTS OF THE SECURITIES ACT, INCLUDING RULE 144 UNDER THE SECURITIES ACT (IF AVAILABLE), (III) PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR (IV) TO THE COMPANY OF ANY OF ITS SUBSIDIARIES, IN EACH OF CASES (I) THROUGH (IV) IN ACCORDANCE WITH ANY APPLICABLE SECURITIES LAWS OF ANY STATE OF THE UNITED STATES, AND (B) THE HOLDER WILL NOTIFY ANY SUBSEQUENT PURCHASER OF THIS SECURITY FROM IT OF THE RESALE RESTRICTIONS REFERRED TO IN (A) ABOVE. THIS SECURITY MAY ONLY BE TRANSFERRED IN ACCORDANCE WITH THE TERMS OF THE CERTIFICATE OF INCORPORATION OF USEC INC. (THE "COMPANY"), AS AMENDED.

THIS SECURITY IS SUBJECT TO THE RESTRICTIONS (INCLUDING THE VOTING AND TRANSFER RESTRICTIONS) SET FORTH IN ARTICLES FOURTH AND ELEVENTH OF USEC INC.'S CERTIFICATE OF INCORPORATION, AS AMENDED.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON TRANSFER, CONVERSION AND REDEMPTION) STATED IN, AND ARE TRANSFERABLE ONLY IN ACCORDANCE WITH, THE PROVISIONS OF SECTION 9 OF THE SECURITIES PURCHASE AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA CORPORATION ("TOSHIBA") AND BABCOCK & WILCOX INVESTMENT COMPANY ("B&W"), DATED AS OF MAY 25, 2010.

THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO THE OBLIGATIONS AND RESTRICTIONS (INCLUDING RESTRICTIONS ON THE DISPOSITION OF SECURITIES) STATED IN THE PROVISIONS OF SECTION 4.7 OF THE INVESTOR RIGHTS AGREEMENT BY AND AMONG THE COMPANY, TOSHIBA AND B&W, DATED AS OF , 2010."

(x) Written Consent. Any action as to which a class vote of the holders of Class B Common Stock, or the holders of Series C Preferred Stock and Class B Common Stock voting together, is required pursuant to the terms of this

Certificate of Incorporation or the Securities Purchase Agreement may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation.

- (xi) Reservation of Stock. The Corporation shall at all times reserve and keep available, free from preemptive rights, out of its authorized but unissued shares of Common Stock, or shares of Common Stock held in treasury by the Corporation, solely for issuance upon the conversion of Class B Common Stock, such number of shares of Common Stock as shall from time to time be issuable upon conversion of all the shares of Class B Common Stock then outstanding. All shares of Common Stock delivered upon conversion of Class B Common Stock shall have been duly authorized and validly issued and shall be fully paid and nonassessable, and shall be free from preemptive rights and free of any lien or adverse claim.
- (xii) <u>Taxes</u>. The Corporation shall pay any and all taxes that may be payable in respect of the issue or delivery of shares of Common Stock on conversion of Class B Common Stock. The Corporation shall not, however, be required to pay any tax which may be payable in respect of any Transfer involved in the issue and delivery of shares of Common Stock in a name other than that in which Class B Common Stock so converted was registered, and no such issue or delivery shall be made unless and until the Person requesting such issue has paid to the Corporation the amount of any such tax, or has established to the satisfaction of the Corporation that such tax has been paid."

SECOND: Thereafter, at a special meeting of stockholders of the Corporation duly called and held upon notice in accordance with Section 222 of the General Corporation Law of the State of Delaware, the affirmative vote of a majority of the voting power of all the shares of capital stock of the Corporation entitled to vote generally in the election of directors, voting together as a single class, as required by Article TWELFTH of the Certificate of Incorporation, was obtained in favor of such amendment in accordance with Section 242 of the General Corporation Law of the State of Delaware.

THIRD: That said amendment was duly adopted in accordance with the provisions of Section 242 of the General Corporation Law of the State of Delaware.

[THIS SPACE LEFT BLANK INTENTIONALLY]

| officer this [—] day of [—], 201[—]. | |
|--------------------------------------|-------------------------|
| | USEC INC. |
| | Ву: |
| | Name: [—] Title: [—] |

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Amendment to be duly executed and acknowledged by its undersigned duly authorized

SIGNATURE PAGE TO THE CERTIFICATE OF AMENDMENT TO THE CERTIFICATE OF INCORPORATION OF USEC INC.

[McDermott International Logo]

777 N. ELDRIDGE PARKWAY HOUSTON, TEXAS 77079

], 2010

To Stockholders of McDermott International, Inc.:

We are pleased to inform you that, on [], 2010, the board of directors of McDermott International, Inc., or McDermott, approved the spin-off of The Babcock & Wilcox Company, or B&W, an indirect, wholly owned subsidiary of McDermott, through a pro rata stock distribution to holders of McDermott common stock. B&W is a leading technology innovator in power generation systems, a specialty manufacturer of nuclear components and a premier service provider in its segments, with a rich legacy spanning over 140 years. We believe the spin-off, which will create two distinct businesses with separate publicly traded shares and, for each company, a dedicated board of directors and management, will better enable our offshore oil and gas construction business, which is conducted principally through J. Ray McDermott, S.A. and its subsidiaries, and our power generation systems and government operations businesses, which is conducted principally through B&W and its subsidiaries, to focus exclusively on realizing their own opportunities and addressing their distinct challenges. In addition, we believe that the two companies, each with its own financial and operating characteristics, will improve transparency for investors and may appeal to a wider combined investor base. For these reasons, we believe the spin-off is in the best interests of McDermott's stockholders.

As a result of the spin-off, McDermott stockholders will receive one share of B&W common stock for every [] shares of McDermott common stock held as of 5:00 p.m. New York City time on [], 2010, the record date. The distribution of B&W shares is expected to occur on [], 2010. Stockholder approval of the spin-off is not required, and you do not need to take any action to receive shares of B&W common stock in the spin-off. The distribution, which is subject to several customary conditions, will be issued in book-entry form only, which means that no physical stock certificates representing interests in B&W will be issued. A book-entry account statement reflecting your ownership of shares of B&W common stock will be mailed to you, or your brokerage account will be credited for the shares on or about [], 2010.

McDermott intends for the distribution of B&W common stock in the spin-off to be tax free for stockholders. To that end, McDermott has obtained a private letter ruling regarding the spin-off from the Internal Revenue Service. In addition, it is a condition to completing the spin-off that McDermott shall have received an opinion of counsel that the distribution of B&W common stock to McDermott stockholders will qualify as a distribution described in Section 355 of the Internal Revenue Code of 1986, as amended, for United States federal income tax purposes. The private letter ruling is, and the opinion will be, subject to certain qualifications and limitations. You should, of course, consult your own tax advisor as to the particular consequences of the spin-off to you, including the applicability and effect of any U.S. federal, state and local and foreign tax laws, which may result in the spin-off being taxable to you. The spin-off is also subject to other conditions, including necessary regulatory approvals.

If you sell your shares of McDermott common stock prior to or on the distribution date, you may also be selling your right to receive shares of B&W common stock. You are encouraged to consult with your financial advisor regarding the specific implications of selling your McDermott common stock prior to or on the distribution date.

Following the spin-off, McDermott common stock will continue to trade on the New York Stock Exchange under the ticker symbol "MDR" and B&W common stock will trade on the New York Stock Exchange under the ticker symbol "BWC." You do not need to take any action to receive your shares of B&W common stock. You do not need to pay any consideration for your shares of B&W common stock or surrender or exchange your shares of McDermott common stock.

We encourage you to read the enclosed information statement, which is being mailed to all McDermott stockholders. It describes the spin-off in detail and contains important information about B&W, including financial statements.

We believe the spin-off is a positive event for McDermott's stockholders, and we look forward to your continued support as a stockholder of McDermott. We remain committed to working on your behalf to build long-term stockholder value.

Sincerely,

Ronald C. Cambre Chairman of the Board John A. Fees Chief Executive Officer

[B&W Logo]

The Harris Building 13024 Ballantyne Corporate Place, Suite 700 Charlotte, North Carolina 28277

], 2010

To Stockholders of The Babcock & Wilcox Company:

It is my pleasure to welcome you as a stockholder of The Babcock & Wilcox Company. Following the spin-off, we will continue as a leading technology innovator in power generation systems, a specialty manufacturer of nuclear components and a premier service provider in our segments, with a rich legacy spanning over 140 years. We design, engineer, manufacture, construct and maintain power generation systems and environmental control systems, primarily for large utility and industrial customers, as well as provide related aftermarket parts and services. For our largest customer, the U.S. Government, we supply nuclear components for defense programs; manage, operate and maintain large government-owned facilities; and provide services for major capital construction and environmental projects. During the period from 2007 through 2009, we generated an average of over \$3 billion in annual revenues, with approximately \$350 million in average annual operating income. We employ approximately 13,000 people worldwide, not including approximately 10,000 joint venture employees.

Our strategy as an independent company will be to maximize the profitability of our operations by (1) continuing to serve our existing customer base, including the U.S. Government, unencumbered by recent changes in the rules under the Federal Acquisition Regulations, (2) increasing the contribution of international projects to our results and (3) pursuing opportunities to reorient and expand our portfolio to take advantage of significant growth opportunities in nuclear and renewable energy, as well as clean coal. We believe that our strengths, including our experienced management team, our existing customer and joint venture relationships, our manufacturing facilities and our operating expertise will enable us to achieve our goals. We are very excited about our prospects and believe that, as an independent company, we can more effectively focus on our operations and growth strategies.

Our common stock has been approved for listing on the New York Stock Exchange under the symbol "BWC."

We thank you in advance for your support as a holder of B&W common stock, and I invite you to learn more about B&W by reviewing the enclosed information statement.

Sincerely,

Brandon C. Bethards President and Chief Executive Officer The Babcock & Wilcox Company

Information contained herein is subject to completion or amendment. A registration statement on Form 10 relating to these securities has been filed with the Securities and Exchange Commission under the Securities Exchange Act of 1934, as amended.

Subject to Completion, dated June 4, 2010

EXHIBIT 99.1

INFORMATION STATEMENT

The Babcock & Wilcox Company

Common Stock (par value \$0.01 per share)

This information statement is being furnished in connection with the distribution by McDermott International, Inc., or McDermott, to its stockholders of all the outstanding shares of common stock of The Babcock & Wilcox Company, or B&W. As of the date of this information statement, McDermott owns all of B&W's outstanding common stock.

On [], 2010, after consultation with financial and other advisors, McDermott's board of directors approved the distribution of 100% of McDermott's interest in B&W. You, as a holder of McDermott common stock, will be entitled to receive one share of B&W common stock for every [] shares of McDermott common stock held as of 5:00 p.m., New York City time, on the record date, [], 2010. The distribution date for the spin-off will be [], 2010.

You will not be required to pay any cash or other consideration for the shares of B&W common stock that will be distributed to you or to surrender or exchange your shares of McDermott common stock in order to receive shares of B&W common stock in the spin-off. The distribution will not affect the number of shares of McDermott common stock that you hold. No approval by McDermott stockholders of the spin-off is required or being sought. You are not being asked for a proxy and you are requested not to send a proxy.

As discussed under "The Spin-Off—Trading of McDermott Common Stock After the Record Date and Prior to the Distribution," if you sell your shares of McDermott common stock in the "regular way" market after the record date and on or prior to the distribution date, you also will be selling your right to receive shares of B&W common stock in connection with the spin-off. You are encouraged to consult with your financial advisor regarding the specific implications of selling your shares of McDermott common stock on or prior to the spin-off.

In reviewing this information statement, you should carefully consider the matters described under the caption "Risk Factors" beginning on page 14.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved these securities or determined if this information statement is truthful or complete. Any representation to the contrary is a criminal offense.

This information statement does not constitute an offer to sell or the solicitation of an offer to buy any securities.

McDermott first mailed this information statement to its stockholders on or about [], 2010.

The date of this information statement is [], 2010.

TABLE OF CONTENTS

| | Page |
|---|------|
| QUESTIONS AND ANSWERS ABOUT THE SPIN-OFF | 1 |
| SUMMARY | 6 |
| SUMMARY HISTORICAL AND PRO FORMA COMBINED FINANCIAL INFORMATION | 12 |
| RISK FACTORS | 14 |
| CAUTIONARY STATEMENT CONCERNING FORWARD-LOOKING INFORMATION | 35 |
| THE SPIN-OFF | 37 |
| <u>CAPITALIZATION</u> | 49 |
| DIVIDEND POLICY | 50 |
| SELECTED HISTORICAL COMBINED FINANCIAL DATA | 51 |
| UNAUDITED PRO FORMA COMBINED FINANCIAL DATA | 52 |
| MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS | 58 |
| QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK | 79 |
| BUSINESS | 81 |
| CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS | 103 |
| RELATIONSHIP WITH MCDERMOTT AFTER THE SPIN-OFF | 104 |
| MANAGEMENT_ | 110 |
| EXECUTIVE COMPENSATION | 119 |
| SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT | 163 |
| DESCRIPTION OF CAPITAL STOCK | 164 |
| INDEMNIFICATION OF DIRECTORS AND OFFICERS | 169 |
| INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM | 170 |
| WHERE YOU CAN FIND MORE INFORMATION | 170 |
| INDEX TO FINANCIAL STATEMENTS | 171 |

Unless we otherwise state or the context otherwise indicates, all references in this information statement to "B&W," "our company," "we," "our," "ours" or "us" mean The Babcock & Wilcox Company and its subsidiaries as of the distribution date, and all references to "McDermott" mean McDermott International, Inc. and its subsidiaries, other than, for all periods following the spin-off, B&W.

The transaction in which B&W will be separated from McDermott and become an independent, publicly traded company is referred to in this information statement alternatively as the "distribution" or the "spin-off."

This information statement is being furnished solely to provide information to McDermott stockholders who will receive shares of B&W common stock in connection with the spin-off. It is not provided as an inducement or encouragement to buy or sell any securities. This information statement describes our business, the relationship between McDermott and us, and how the spin-off affects McDermott and its stockholders, and provides other information to assist you in evaluating the benefits and risks of holding or disposing of our common stock that you will receive in the spin-off. You should be aware of certain risks relating to the spin-off, our business and ownership of our common stock, which are described under the heading "Risk Factors." You should not assume that the information contained in this information statement is accurate as of any date other than the date set forth on the front cover. Changes to the information contained in this information statement is accurate as of any date other than the date set forth on the front cover. Changes to the information contained in this information statement, unless we are required by applicable securities laws to do so.

QUESTIONS AND ANSWERS ABOUT THE SPIN-OFF

O: What is the spin-off?

A: The spin-off involves McDermott's pro rata distribution to its stockholders of all the shares of our common stock. Following the spin-off, we will be a separate, publicly traded company, and McDermott will not retain any ownership interest in our company. You do not have to pay any consideration or give up any of your shares of McDermott common stock to receive shares of our common stock in the spin-off.

Q: Why is McDermott separating B&W from McDermott's business?

McDermott's board and management believe the benefits of the spinoff will include: the improved positioning of each of McDermott and B&W to accelerate growth; a more efficient allocation of capital for each company; the establishment of a distinct publicly traded stock for each company, which may be used as "currencies" to facilitate future acquisitions; the elimination of the risk to the combined businesses posed by recent modifications to the rules under the U.S. Federal Acquisition Regulations ("FAR") that limit the U.S. Government's ability to contract with "inverted" companies and their subsidiaries; and sharpened management focus and strategic vision and closer alignment of management incentives with stockholder value creation. Because of the likelihood that McDermott would be considered an "inverted" company under the recent modifications to the FAR (and, therefore, we would be considered a subsidiary of an "inverted" company), those modifications had a significant impact on the McDermott board's decision to effect the spin-off at this time. In approving the spin-off, McDermott's board of directors considered the fact that we derive a substantial amount of our revenues and profits from services provided to the U.S. Government, and the fact that any significant restriction on our ability to pursue future U.S. Government contract work consistent with our past operations would have a material adverse effect on the financial condition, results of operations and cash flows of the combined companies. The independent operation of the two businesses, with B&W as a Delaware corporation, will permit us to continue to serve the needs of our U.S. Government customers without interruption as a result of these rule

modifications, while allowing McDermott the opportunity to pursue its independent business strategy. For more information, see "The Spin-Off—Reasons for the Spin-Off."

Q: What is being distributed in the spin-off?

A: McDermott will distribute one share of B&W common stock for every [] shares of McDermott common stock outstanding as of the record date for the spin-off. Approximately [] million shares of our common stock will be distributed in the spin-off, based upon the number of shares of McDermott common stock outstanding on [], 2010. The shares of our common stock to be distributed by McDermott will constitute all of the issued and outstanding shares of our common stock. For more information on the shares being distributed in the spin-off, see "Description of Capital Stock."

Q: What is the record date for the spin-off, and when will the spin-off occur?

A: The record date is [], 2010, and ownership is determined as of 5:00 p.m. New York City time on that date. Shares of B&W common stock will be distributed on [], 2010, which we refer to as the distribution date.

Q: As a holder of shares of McDermott common stock as of the record date, what do I have to do to participate in the spin-off?

A: Nothing. You will receive one share of B&W common stock for every

[] shares of McDermott common stock held as of the record date and retained through the distribution date. You may also participate in the spin-off if you purchase McDermott common stock in the "regular way" market and retain your McDermott shares through the distribution date. See "The Spin-Off—Trading of McDermott Common Stock After the Record Date and Prior to the Distribution."

Q: If I sell my shares of McDermott common stock before or on the distribution date, will I still be entitled to receive B&W shares in the spin-off?

A: If you sell your shares of McDermott common stock prior to or on the distribution date, you

1

may also be selling your right to receive shares of B&W common stock. See "The Spin-Off—Trading of McDermott Common Stock After the Record Date and Prior to the Distribution." You are encouraged to consult with your financial advisor regarding the specific implications of selling your McDermott common stock prior to or on the distribution date.

Q: How will fractional shares be treated in the spin-off?

A: Any fractional share of our common stock otherwise issuable to you will be sold on your behalf, and you will receive a cash payment with respect to that fractional share. For an explanation of how the cash payments for fractional shares will be determined, see "The Spin-Off—Treatment of Fractional Shares."

Q: Will the spin-off affect the number of shares of McDermott I currently hold?

A: The number of shares of McDermott common stock held by a stockholder will be unchanged. The market value of each McDermott share, however, will decline to reflect the impact of the spin-off.

Q: What are the U.S. federal income tax consequences of the distribution of our shares of common stock to U.S. stockholders?

A: McDermott has obtained a private letter ruling from the Internal Revenue Service (the "IRS") and intends to obtain an opinion of Baker Botts L.L.P., in each case, substantially to the effect that, for U.S. federal income tax purposes, the spin-off will qualify under Section 355 of the Internal Revenue Code of 1986, as amended (the "Code"), and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code. The private letter ruling is, and the tax opinion will be, subject to certain qualifications and limitations.

Assuming the spin-off so qualifies, for U.S. federal income tax purposes, no gain or loss will be recognized by you, and no amount will be included in your taxable income (other than with respect to cash received in lieu of fractional shares), as a result of the spin-off. Certain U.S. federal income tax consequences of the spin-off are described in more detail under "The Spin-Off—Material U.S. Federal Income Tax Consequences of the Spin-Off."

Q: How will I determine the tax basis I will have in the shares of stock I receive in the spin-off?

A: Generally, your aggregate basis in the stock you hold in McDermott and the shares of our common stock received in the spin-off will equal the aggregate basis of McDermott common stock held by you immediately before the spin-off. This aggregate basis should be allocated between your McDermott common stock and our common stock you receive in the spin-off in proportion to the relative fair market value of each immediately after the distribution. You should consult your tax advisor about how this allocation will work in your situation (including a situation where you have purchased McDermott shares at different times or for different amounts) and regarding any particular consequences of the spin-off to you, including the application of state, local, and foreign tax laws. Certain material U.S. federal income tax consequences of the spin-off are described in more detail under "The Spin-Off—Material U.S. Federal Income Tax Consequences of the Spin-Off."

Q: Is the spin-off tax free to non-U.S. stockholders?

A: Non-U.S. stockholders will generally not be subject to U.S. tax on the spin-off. However, non-U.S. stockholders may be subject to tax on the spin-off in jurisdictions outside the U.S. The foregoing is for general information purposes and does not constitute tax advice. Stockholders should consult their own tax advisors regarding the particular consequences of the spin-off to them

Q: Will I receive a stock certificate for B&W shares distributed as a result of the spin-off?

A: Registered holders of McDermott common stock who are entitled to participate in the spin-off will receive a book-entry account statement reflecting their ownership of B&W common stock. For additional information, registered stockholders in the United States should contact McDermott's transfer agent, Computershare Trust Company, N.A., at 1 (800) 446-2617, or through its website at www.computershare.com. Stockholders from outside the United States may call (781) 575-2723. See "The Spin-Off—When and How You Will Receive Your Shares."

Q: What if I hold my shares through a broker, bank or other nominee?

A: McDermott stockholders who hold their shares through a broker, bank or other nominee will have their brokerage account credited with shares of B&W common stock. For additional information, those stockholders should contact their broker or bank directly.

Q: What if I have stock certificates reflecting my shares of McDermott common stock? Should I send them to the transfer agent or to

A: No, you should not send your stock certificates to the transfer agent or to McDermott. You should retain your McDermott stock certificates.

Q: What are the conditions to the spin-off?

A: The spin-off is subject to a number of conditions, including, among others, (1) the continued effectiveness of the private letter ruling received from the IRS on May 21, 2010 and an opinion of counsel, in each case, substantially to the effect that, for U.S. federal income tax purposes, the spin-off will qualify under Section 355 of the Code and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code, and (2) the SEC declaring effective the

registration statement of which this information statement forms a part. However, even if all of the conditions have been satisfied, McDermott may amend, modify or abandon any and all terms of the distribution and the related transactions at any time prior to the distribution date. See "The Spin-Off—Spin-Off Conditions and Termination."

Q: Will B&W incur any debt prior to or at the time of the spin-off?

A: We do not currently anticipate borrowing any amounts under our revolving credit facility prior to or at the time of the spin-off. See "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources."

Q: Are there risks to owning shares of B&W common stock?

A: Yes. B&W's business is subject both to general and specific business risks relating to its operations. In addition, the spin-off involves specific risks, including risks relating to B&W being an independent, publicly traded company. See "Risk Factors."

Q: Does B&W plan to pay cash dividends?

A: No. We do not currently plan to pay a regular dividend on our common stock following the spin-off. Our credit agreement includes restrictions on our ability to pay dividends. The declaration and amount of future dividends, if any, will be determined by our board of directors and will depend on our financial condition, earnings, capital requirements, financial covenants, industry practice and other factors our board of directors deems relevant. See "Dividend Policy."

Q: Will the B&W common stock trade on a stock market?

A: Currently, there is no public market for our common stock. Subject to the consummation of the spin-off, our common stock has been approved for listing on the New York Stock Exchange under the symbol "BWC." We anticipate that limited trading in shares of B&W common stock will begin on a "when-issued" basis on or shortly before the record date and will continue up to and including through the distribution date and that "regularway" trading in shares of B&W common stock will begin on the first trading day following the distribution date. The "when-issued" trading market will be a market for shares of B&W common stock that will be distributed to McDermott stockholders on the distribution date. If you owned shares of McDermott common stock at the close of business on the record date, you would be entitled to shares of our common stock distributed pursuant to the spin-off. You may trade this entitlement to shares of B&W common stock, without the shares of McDermott common stock you own, on the "when-issued" market.

We cannot predict the trading prices for our common stock before, on or after the distribution date.

- Q: What will happen to McDermott stock options, restricted shares, restricted stock units, deferred stock units and performance shares?
- A: We currently expect that, subject to approval of the compensation committee of McDermott's board of directors, equity-based compensation awards will generally be treated as follows:
 - Each outstanding option to purchase shares of McDermott common stock that is held by a director, officer or employee of McDermott who will remain a director, officer or employee of McDermott and will not be a director, officer or employee of B&W immediately after the spin-off will be replaced with an adjusted option to purchase McDermott common stock. Each of those adjusted options will reflect adjustments that will be generally intended to preserve the intrinsic value of the original option and the ratio of the exercise price to the fair market value of the stock subject to the option. To the extent the options being adjusted are vested, the adjusted options will also be vested.
 - Each outstanding option to purchase shares of McDermott common stock that is held by a person who is or will be a director of B&W (but not of McDermott) or by a person who is or will be an officer or employee of B&W immediately after the spin-off will be replaced with substitute options to purchase shares of B&W common stock. Each of those substitute options will have terms that will be generally intended to preserve the intrinsic value of the original option and the ratio of the exercise price to the fair market value of the stock subject to the option. To the extent the options being replaced are vested, the substitute options will also be vested.
 - Each outstanding option to purchase shares of McDermott common stock that is held by a (1) person who is a director of McDermott and who will remain on the board of directors of McDermott and will also serve on the board of directors of B&W immediately after the spin-off or (2) person who is one of a small number of current officers of B&W who will not serve as officers of B&W after the Distribution Date

- (the "Former B&W Officers") will be replaced with both an adjusted McDermott stock option and a substitute B&W stock option. Both options, when combined, will have terms that will be generally intended to preserve the intrinsic value of the original option and the ratio of the exercise price to the fair market value of the stock subject to the option. All McDermott stock options held by directors of McDermott as of the distribution date will have previously become vested pursuant to their existing terms. Accordingly, all adjusted or substitute options issued to McDermott directors will also be vested.
- The McDermott restricted stock awards, restricted stock unit awards and deferred stock unit awards of officers or employees of McDermott who will remain officers or employees of McDermott and will not become officers or employees of B&W immediately after the spin-off will be replaced with adjusted McDermott awards, each of which will generally preserve the value of the original award. As of the distribution date, none of McDermott's directors will hold any restricted stock awards, restricted stock unit awards or deferred stock unit awards.
- The McDermott restricted stock awards, restricted stock unit awards and deferred stock unit awards of persons who are or will be officers or employees of B&W immediately after the spin-off will be converted into substitute B&W awards, each of which will generally preserve the value of the original award.
- The McDermott restricted stock awards, restricted stock unit awards and deferred stock unit awards of the Former B&W Officers will be converted into rights to receive (1) an equivalent number of unrestricted shares of McDermott common stock and (2) [] shares of B&W common stock for each such share of McDermott common stock, in accordance with the spin-off distribution.
- The McDermott performance share awards of officers or employees of McDermott who will remain officers or employees of McDermott and will not become officers or

employees of B&W immediately after the spin-off will be converted into unvested rights to receive the value of deemed target performance in unrestricted shares of McDermott common stock, with the same vesting terms as the original awards.

- The McDermott performance share awards of persons who are or will be officers or employees of B&W immediately after the spin-off will be converted into unvested rights to receive the value of deemed target performance in unrestricted shares of B&W common stock, with the same vesting terms as the original awards.
- The McDermott performance share awards of the Former B&W officers will be converted into rights to receive the value of deemed target performance in unrestricted shares of McDermott common stock and B&W common stock, in accordance with the spin-off distribution.

For additional information on the treatment of McDermott equity-based compensation awards, see "The Spin-Off-Treatment of Stock-Based Awards "

What will the relationship between McDermott and B&W be following the spin-off?

After the spin-off, McDermott will not own any shares of B&W common stock, and each of McDermott and B&W will be an independent, publicly traded company with its own management team and board of directors. However, in connection with the spin-off, we have entered into a number of agreements with McDermott that will govern the spin-off and allocate responsibilities for obligations arising before and after the spin-off, including, among others, obligations relating to our employees and taxes. See "Relationship with McDermott After the Spin-Off."

Q: Will I have appraisal rights in connection with the spin-off?

No. Holders of shares of McDermott common stock are not entitled to appraisal rights in connection with the spin-off.

Q: Who is the transfer agent for your common stock?

Computershare Trust Company, N.A. 250 Royall Street Canton, Massachusetts 02021-1011

Q: Who is the distribution agent for the spin-off?

Computershare Trust Company, N.A. 250 Royall Street Canton, Massachusetts 02021-1011

Q: Whom can I contact for more information?

If you have questions relating to the mechanics of the distribution of B&W shares, you should contact the distribution agent:

Computershare Trust Company, N.A. 250 Royall Street

Canton, Massachusetts 02021-1011

Telephone: 1 (800) 446-2617

Before the spin-off, if you have questions relating to the spin-off, you should contact McDermott's corporate secretary at:

McDermott International, Inc. 777 N. Eldridge Pkwy. Houston, Texas 77079 Attention: Corporate Secretary Telephone: (281) 870-5901

SUMMARY

The following is a summary of some of the information contained in this information statement. It does not contain all the details concerning us or the spin-off, including information that may be important to you. We urge you to read this entire document carefully, including the risk factors, our historical and pro forma combined financial statements and the notes to those financial statements.

Unless the context requires otherwise or we specifically indicate otherwise, the terms "B&W," "our company," "we," "our," "ours" and "us" refer to The Babcock & Wilcox Company, a company incorporated under the laws of the state of Delaware, and its subsidiaries as of the distribution date; and the term "McDermott" refers to McDermott International, Inc., a publicly traded Panamanian corporation, and its subsidiaries (excluding us and any of our subsidiaries).

We describe in this information statement the business to be held by us after the spin-off as if it were our business for all historical periods described. However, we are an entity that will not have independently conducted any operations before the spin-off. References in this document to our historical assets, liabilities, products, business or activities generally refer to the historical assets, liabilities, products, business or activities of the B&W business as it was conducted as part of McDermott and its subsidiaries before the spin-off. Our historical combined financial results as part of McDermott contained in this information statement may not be indicative of our financial results in the future as an independent company or reflect what our financial results would have been had we been an independent company during the periods presented.

Except as otherwise indicated or unless the context otherwise requires, the information included in this information statement assumes the completion of the separation of B&W from McDermott and the related distribution of our common stock.

Our Company

We are currently a wholly owned subsidiary of McDermott. Following the spin-off, we will be an independent, publicly traded company. McDermott will not retain any ownership interest in our company. Our assets and businesses primarily consist of those that McDermott reports as its Power Generation Systems and Government Operations segments in its consolidated financial statements.

B&W is a leading technology innovator in power generation systems, a specialty manufacturer of nuclear components and a premier service provider in its segments, with an operating history of more than 140 years. We provide a variety of products and services to customers in the power and other steam-using industries, including electric utilities and other power generators, industrial customers in various other industries, and the U.S. Government. We operate in two business segments: Power Generation Systems and Government Operations. Through our Power Generation Systems segment, we design, engineer, manufacture, supply, construct and maintain power generation systems and environmental control systems, primarily for large utility and industrial customers, and we provide related aftermarket parts and services. Through our Government Operations segment, we manufacture nuclear components and provide various services to the U.S. Government, including uranium processing, environmental site restoration services, and management and operating services for various U.S. Government-owned facilities. These services are provided to the U.S. Department of Energy (the "DOE"), including the National Nuclear Security Administration (the "NNSA"), the Office of Science and the Office of Environmental Management.

We participate in the ownership of a variety of entities with third parties, primarily through corporations, limited liability companies and partnerships, which we refer to as "joint ventures." Our Government Operations segment manages and operates complex, high-consequence nuclear and national security operations for the DOE and the NNSA through joint ventures. We employ approximately 13,000 people worldwide, not including approximately 10,000 joint venture employees.

For the year ended December 31, 2009, we generated revenues of approximately \$2.9 billion and operating income of approximately \$270 million. For the year ended December 31, 2008, we generated revenues of approximately \$3.4 billion and operating income of approximately \$444.5 million.

In connection with the spin-off, we and McDermott are entering into certain agreements, including a master separation agreement, a tax sharing agreement and an employee matters agreement, under which we and McDermott will, among other things, indemnify each other against certain liabilities arising from our respective businesses. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us."

Following the spin-off, the parent company in our group of companies will be The Babcock & Wilcox Company, which was incorporated in Delaware as a subsidiary of McDermott. The address of our principal executive offices is The Harris Building, 13024 Ballantyne Corporate Place, Suite 700, Charlotte, North Carolina 28277. Our main telephone number is (704) 625-4900.

Summary of the Spin-Off

The following is a brief summary of the terms of the spin-off. Please see "The Spin-Off" for a more detailed description of the matters described below.

Distributing company

McDermott, which is the parent company of B&W. After the distribution, McDermott will not retain any shares of our common stock.

Distributed company

B&W, which is currently a wholly owned subsidiary of McDermott. After the distribution, B&W will be an independent, publicly traded company.

Distribution ratio

Each holder of McDermott common stock will receive one share of our common stock for every [] shares of McDermott common stock held on the record date. Approximately [] million shares of our common stock will be distributed in the spin-off, based upon the number of shares of McDermott common stock outstanding on [], 2010. The shares of our common stock to be distributed by McDermott, together with the shares of our restricted stock issued to our employees in connection with the spin-off, will constitute all of the issued and outstanding shares of our common stock. For more information on the shares being distributed in the spin-off, see "Description of Capital Stock."

Fractional shares

The transfer agent identified below will aggregate fractional shares into whole shares and sell them on behalf of stockholders in the open market at prevailing market prices and distribute the proceeds pro rata to each McDermott stockholder who otherwise would have been entitled to receive a fractional share in the spin-off. You will not be entitled to any interest on the amount of payment made to you in lieu of a fractional share. See "The Spin-Off—Treatment of Fractional Shares."

Distribution procedures

On or about the distribution date, the distribution agent identified below will distribute the shares of our common stock to be distributed by crediting those shares to book-entry accounts established by the transfer agent for persons who were stockholders of McDermott as of 5:00 p.m., New York City time, on the record date. Shares of B&W common stock will be issued only in book-entry form. No paper stock certificates will be issued. You will not be required to make any payment or surrender or exchange your shares of McDermott common stock or take any other action to receive your shares of our common stock. However, as discussed below, if you sell shares of McDermott common stock in the "regular way" market between the record date and the distribution date, you will be selling your right to receive the associated shares of our common stock in the distribution. Registered stockholders will receive additional information from the transfer agent shortly after the distribution date. Beneficial stockholders will receive information from their brokerage firms.

Distribution agent, transfer agent and registrar for our shares Computershare Trust Company, N.A. of common stock

Record date 5:00 p.m., New York City time, on [], 2010.

Distribution date [], 2010.

Trading prior to or on the distribution date

It is anticipated that, beginning shortly before the record date, McDermott's shares will trade in two

markets on the New York Stock Exchange, a "regular way" market and an "ex-distribution" market. Investors will be able to purchase McDermott shares without the right to receive shares of our common stock in the "ex-distribution" market for McDermott common stock. Any holder of McDermott common stock who sells McDermott shares in the "regular way" market on or before the distribution date will also be selling the right to receive shares of our common stock in the spin-off. You are encouraged to consult with your financial advisor regarding the specific implications of

selling McDermott common stock prior to or on the distribution date.

Assets and liabilities of the distributed company

We and McDermott have entered into a master separation agreement that contains the key provisions relating to the separation of our business from McDermott and the distribution of our shares of

retaining to the separation of our business from McDermott and the distribution of our shares of common stock. The master separation agreement identifies the assets to be transferred, liabilities to be assumed and contracts to be assigned either to us by McDermott or by us to McDermott in the spin-off and describe when and how these transfers, assumptions and assignments will occur. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—

Master Separation Agreement."

Relationship with McDermott after the spin-off

We and McDermott have entered into agreements to define various continuing relationships between

McDermott and us in various contexts. In particular, we have entered into transition services agreements under which we and McDermott will provide each other certain transition services on an interim basis. We and a subsidiary of McDermott have also entered into an agreement providing for the sharing of taxes incurred before and after the distribution, various indemnification rights with respect to tax matters and restrictions to preserve the tax-free status of the distribution to McDermott.

See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us."

Under the terms of the tax sharing agreement, we will generally be responsible for any taxes imposed

Under the terms of the tax sharing agreement, we will generally be responsible for any taxes imposed on us or McDermott and its subsidiaries in the event that certain transactions related to the

spin-off were to fail to qualify for tax-free treatment. However, if these transactions were to fail to qualify for tax-free treatment because of actions or failures to act by McDermott or its subsidiaries, a subsidiary of McDermott would be responsible for all such taxes. Please see "The Spin-Off—Material U.S. Federal Income Tax Consequences of the Spin-Off." Please see also "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Tax Sharing Agreement." Under the master separation agreement, we will also indemnify McDermott and its remaining subsidiaries against various claims and liabilities relating to the past operation of our business. Please see "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Master Separation Agreement."

U.S. federal income tax consequences

McDermott has obtained a private letter ruling from the IRS and expects to obtain an opinion of counsel, in each case, substantially to the effect that, for U.S. federal income tax purposes, the spin-off will qualify under Section 355 of the Code and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code. Certain United States federal income tax consequences of the spin-off are described in more detail under "The Spin-Off—Material U.S. Federal Income Tax Consequences of the Spin-Off."

Conditions to the spin-off

We expect that the spin-off will be effective on [], 2010, provided that the conditions set forth under the caption "The Spin-Off—Spin-Off Conditions and Termination" have been satisfied in McDermott's sole and absolute discretion. However, even if all of the conditions have been satisfied, McDermott may amend, modify or abandon any and all terms of the distribution and the related transactions at any time prior to the distribution date.

Reasons for the spin-off

McDermott's board and management believe the benefits of the spinoff will include: the improved positioning of each of McDermott and B&W to accelerate growth; a more efficient allocation of capital for each company; the establishment of a distinct publicly traded stock for each company, which may be used as "currencies" to facilitate future acquisitions; the elimination of the risk to the combined businesses posed by recent modifications to the rules under the FAR that limit the U.S. Government's ability to contract with "inverted" companies and their subsidiaries; and sharpened management focus and strategic vision and closer alignment of management incentives with stockholder value creation. Because of the likelihood that McDermott would be considered an "inverted" company under the recent modifications to the FAR (and, therefore, we would be considered a subsidiary of an "inverted" company), those modifications had a significant impact on the McDermott board's decision to effect the spin-off at this time. In approving the spin-off, McDermott's board of directors considered the fact that we derive a substantial amount of our revenues and profits from services

Stock exchange listing

Incurrence of debt

Risk factors

provided to the U.S. Government, and the fact that any significant restriction on our ability to pursue future U.S. Government contract work consistent with our past operations would have a material adverse effect on the financial condition, results of operations and cash flows of the combined companies. The independent operation of the two businesses, with B&W as a Delaware corporation, will permit us to continue to serve the needs of our U.S. Government customers without interruption as a result of these rule modifications, while allowing McDermott the opportunity to pursue its independent business strategy. For more information, see "The Spin-Off—Reasons for the Spin-Off."

Currently there is no public market for our common stock. Our common stock has been approved for listing on the New York Stock Exchange, or the NYSE, under the symbol "BWC." We anticipate that trading will commence on a "when-issued" basis on or shortly before the record date. "When-issued" trading refers to a transaction made conditionally because the security has been authorized but not yet issued. On the first trading day following the distribution of our shares of common stock in the spin-off, "when-issued" trading in respect of our common stock will end and "regular way" trading will begin. "Regular way" trading refers to trading after a security has been issued and typically involves a transaction that settles on the third full business day following the date of the transaction. We cannot predict the trading prices for our common stock following the spin-off. In addition, McDermott common stock will remain outstanding and will continue to trade on the NYSE.

You should review the risks relating to the spin-off, our industry and our business and ownership of our common stock described in "Risk Factors."

SUMMARY HISTORICAL AND PRO FORMA COMBINED FINANCIAL INFORMATION

The following table presents our summary historical combined financial information and unaudited pro forma combined financial information. We derived the historical combined statement of operations information for each of the years ended December 31, 2009, 2008 and 2007, and the balance sheet information as of December 31, 2009 and 2008, from our audited combined financial statements included in this information statement. We derived the historical condensed combined statement of operations information for the three months ended March 31, 2010 and 2009, and the balance sheet information as of March 31, 2010 and 2009, from our unaudited condensed combined financial statements included in this information statement. We derived the historical combined statement of operations information for the years ended December 31, 2006 and 2005, and the balance sheet information as of December 31, 2007, 2006 and 2005, from our unaudited combined financial statements not included in this information statement. The summary historical combined financial information reflects the effects of certain assets to be transferred to, and liabilities to be assumed by, McDermott in connection with the spin-off, and does not reflect the effects of certain liabilities to be assumed by B&W in connection with the spin-off. See "Unaudited Pro Forma Combined Financial Data."

The unaudited pro forma combined financial information set forth below is based on available information and assumptions that we believe are factually supportable. The information has been prepared on a combined basis using historical results of operations and bases of assets and liabilities of The Babcock & Wilcox Operations of McDermott International, Inc. and includes intercompany allocations of expenses. The costs to operate our business as an independent public entity may exceed the historical allocations of expenses related to areas that include, but are not limited to, litigation and other legal matters, compliance with the Sarbanes-Oxley Act and other corporate governance matters, insurance and claims management and the related cost of insurance, as well as general overall purchasing power. These possible increased costs are not included in the unaudited pro forma combined financial information, as their impact on our results of operations cannot be reasonably estimated. The unaudited pro forma combined financial information also does not reflect any significant changes that may occur after the spin-off in our financing plans and cost structure, including expenditures related to our growth initiatives.

The unaudited pro forma combined statement of operations information for the year ended December 31, 2009 and the three months ended March 31, 2010 includes adjustments to give effect to:

- the deemed transfer by B&W to McDermott of certain assets that will not be held by B&W;
- the deemed assumption by B&W of certain liabilities that were previously liabilities of McDermott or other subsidiaries of McDermott; and
- · the deemed assumption by McDermott or other subsidiaries of McDermott of certain liabilities that were previously liabilities of B&W;

in each case as if such transfers occurred on January 1, 2009. The unaudited pro forma combined balance sheet information as of March 31, 2010 includes adjustments to give effect to these deemed transactions as if they had occurred on March 31, 2010.

You should read the summary historical and pro forma combined financial information in conjunction with our combined financial statements and the accompanying notes, "Selected Historical Combined Financial Data," "Unaudited Pro Forma Combined Financial Data" and "Management's Discussion and Analysis of Financial Condition and Results of Operations," included in this information statement. The financial information presented below is not necessarily indicative of our future performance or what our financial position and results of operations would have been had we operated as an independent public company during the periods presented or, in the case of the unaudited pro forma information, had the transactions reflected in the pro forma adjustments actually occurred as of the dates assumed.

| | | Pro F | orma | 1 | Historical | | | | | | | |
|--|-----------------------|-------------------|------------|---------------|---------------------------------|-------------|-------------------------|------------------------|--------------|--------------|---------|---------|
| | Three Months Ended | | Year Ended | | Three Months Ended March 31, | | Year Ended December 31, | | | | | |
| | _ | March 31, 2010 | | 2009 | 2010 | 2009 | 2009 | 2008 (In thousands) | 2007 | 2006(1) | 20 | 05(1) |
| Statement of Operations Information: | | | | | | | | () | | | | |
| Revenues | \$ | 662,388 | \$ | 2,854,632 | \$ 662,388 | \$ 785,053 | \$2,854,632 | \$3,398,574 | \$3,199,944 | \$2,517,606 | \$ 6 | 01,042 |
| Costs and expenses | | 635,397 | | 2,630,631 | 637,661 | 700,090 | 2,640,162 | 3,005,883 | 2,907,958 | 2,348,569 | 5 | 60,223 |
| Equity in income of investees | | 14,019 | | 55,094 | 14,019 | 10,345 | 55,094 | 51,792 | 45,647 | 39,809 | | 37,705 |
| Operating income | | 41,010 | | 279,095 | 38,746 | 95,308 | 269,564 | 444,483 | 337,633 | 208,846 | | 78,524 |
| Other income (expense) | | (3,134) | | (15,527) | (8,634) | (6,062) | (37,263) | (11,744) | (23,285) | 7,990 | | (2,439) |
| Income before provision for income taxes | | 37,876 | | 263,568 | 30,112 | 89,246 | 232,301 | 432,739 | 314,348 | 216,836 | | 76,085 |
| Provision for (benefit from) income taxes | | 20,328 | | 99,319 | 13,256 | 35,710 | 84,381 | 108,885 | 99,018 | 84,826 | (| 17,666 |
| Net income attributable to The Babcock & Wilcox | | | | | | | | | | | | |
| Operations of McDermott International, Inc. | \$ | 17,535 | \$ | 164,093 | \$ 16,843 | \$ 53,484 | \$ 147,764 | \$ 323,766 | \$ 215,250 | \$ 131,976 | \$ | 93,751 |
| Other Data: | | | | | | | | | | | | |
| Depreciation and amortization | \$ | 16,812 | \$ | 72,212 | \$ 16,812 | \$ 16,263 | \$ 72,212 | \$ 45,985 | \$ 41,672 | \$ 32,484 | \$ | 14,493 |
| Capital expenditures | | 18,981 | | 93,725 | 18,981 | 20,029 | 93,725 | 63,014 | 60,709 | 43,203 | | 27,109 |
| | | Pro Forma | | | Historical | | | | | | | |
| | As of | | | As of As of 1 | | larch 31, | | As of December 31, | | | | |
| | 1 | March 31, | | ecember 31, | 2010 | 2009 | 2009 | 2008 | 2005 | 2006(1) | 2005(1) | |
| | _ | 2010 | _ | 2009 | 2010 | 2009 | 2009 | (In thousands) | 2007 | 2006(1) | | 05(-) |
| Balance Sheet Information: | | | | | | | | | | | | |
| Working capital | \$ | 96,027 | \$ | 93,992 | \$ (51,440) | \$ (32,128) | \$ 71,539 | \$ (90,040) | \$ (192,874) | \$ (297,286) | \$ | 429 |
| Total assets | | 2,403,245 | | 2,435,119 | 2,588,769 | 2,498,580 | 2,603,859 | 2,506,841 | 2,149,636 | 2,222,818 | 5 | 08,772 |
| Notes payable and current maturities of long-term debt | | 4,993 | | 6,432 | 4,993 | 4,161 | 6,432 | 9,021 | 6,599 | 257,492 | | 4,250 |
| Long-term debt | | 5,343 | | 4,222 | 5,343 | 5,915 | 4,222 | 6,109 | 10,609 | 15,242 | | 12,975 |
| Other Data: | | | | | | | | | | | | |
| Backlog | \$ | 4,571,972 | \$ | 4,740,060 | \$4,571,972 | \$4,918,507 | \$4,740,060 | \$5,358,740 | \$5,064,226 | \$3,494,477 | \$1,7 | 72,258 |
| | | | | | | | | | | | | |

⁽¹⁾ Results for the year ended December 31, 2006 include approximately ten months for the principal operating subsidiaries of our Power Generation Systems segment, which were included in our results effective February 22, 2006. We did not include the results of operations of these entities in our financial statements from February 22, 2000 through February 22, 2006 due to the Chapter 11 proceedings involving several of our subsidiaries. On February 22, 2000, Babcock & Wilcox Power Generation Group, Inc. (formerly known as The Babcock & Wilcox Company) ("B&W PGG") and certain of its subsidiaries (collectively, the "Debtors") filed a voluntary bankruptcy petition in the U.S. Bankruptcy Court for the Eastern District of New Orleans to reorganize under Chapter 11 of the U.S. Bankruptcy Code. The Debtors took this action as a means to determine and comprehensively resolve their asbestos liability as a result of asbestos-containing boilers and other products the Debtors sold, installed or serviced in prior decades. On February 22, 2000, the Debtors' operations became subject to the jurisdiction of the Bankruptcy Court, consistent with the requirements of Accounting Standards Codification ("ASC") 810-10-15-10 (a), (1) (ii), we deconsolidated the results of the Debtors in our financial statements on February 22, 2000 and presented our investment in the Debtors on the cost basis of accounting. Subsequent to February 22, 2000, we wrote this investment down to zero as impaired. On February 22, 2006, the Debtors emerged from bankruptcy. The results of the Debtors were reconsolidated into our financial statements effective February 22, 2006 and for all periods after such date. We accounted for this reconsolidation in a manner similar to the acquisition of the non-controlling interest in a subsidiary.

RISK FACTORS

You should carefully consider each of the following risks and all of the other information contained in this information statement. Some of these risks relate principally to our spin-off from McDermott, while others relate principally to our business and the industry in which we operate or to the securities markets generally and ownership of our common stock.

If any of these risks develop into actual events, our business, financial condition, results of operations or cash flows could be materially adversely affected by any of these risks, and, as a result, the trading price of our common stock could decline.

Risks Relating to Our Industry and Our Business

We derive substantial revenues from electric power generating companies and other steam-using industries, with demand for our products and services depending on capital expenditures in these historically cyclical industries. Additionally, recent legislative and regulatory developments relating to greenhouse gas emissions are impacting plans for new coal-fired power plants within the United States.

The demand for power generation products and services depends primarily on the capital expenditures of electric power generating companies, paper companies and other steam-using industries. These capital expenditures are influenced by such factors as:

- · prices for electricity, along with the cost of production and distribution;
- · prices for natural resources such as coal and natural gas;
- demand for electricity, paper and other end products of steam-generating facilities;
- · availability of other sources of electricity, paper or other end products;
- · requirements of environmental legislation and regulation, including potential requirements applicable to carbon dioxide emissions;
- · levels of capacity utilization at operating power plants, paper mills and other steam-using facilities;
- · requirements for maintenance and upkeep at operating power plants and paper mills to combat the accumulated effects of wear and tear;
- ability of electric generating companies and other steam users to raise capital; and
- total costs of electricity production using boilers compared to total costs using gas turbines and other alternative forms of generation.

A material decline in capital expenditures by electric power generating companies, paper companies and other steam-using industries over a sustained period of time could materially and adversely affect the demand for our power generation products and services and, therefore, our financial condition, results of operations and cash flows

U.S. coal-fired power plants have been scrutinized by environmental groups and government regulators over the emissions of potentially harmful pollutants. In addition to recent legislation at the state level, the U.S. Congress is considering legislation that would limit greenhouse gas emissions, including carbon dioxide. In April 2007, the U.S. Supreme Court ruled that the U.S. Environmental Protection Agency (the "EPA") has some authority to regulate greenhouse gases under the Clean Air Act. In 2009, the EPA published a final rule requiring the reporting of greenhouse gas emissions from specified large sources in the United States beginning in 2011 for emissions occurring in 2010 and a final rule finding that current and projected concentrations of six key greenhouse gases in the atmosphere threaten public health and welfare of current and future generations. In addition, in 2009, the EPA issued a "Mandatory Reporting of Greenhouse Gases" final rule. As a result of these

and other developments, some plans for coal-fired power plants have been cancelled or suspended. We cannot predict the ultimate impact that legislative and regulatory developments in this area will have on the overall demand for our products and services.

We rely on U.S. Government contracts for a large percentage of our revenue, and some of those contracts are subject to continued appropriations by Congress and may be terminated or delayed if future funding is not made available. In addition, the U.S. Government may not renew our existing contracts.

For the year ended December 31, 2009, we relied on U.S. Government contracts for approximately 33% of our revenue. Government contracts are subject to various uncertainties, restrictions and regulations, including oversight audits, which could result in withholding or delaying of payments to us. In addition, some of our large, multi-year contracts with the U.S. Government are subject to annual funding determinations and the continuing availability of Congressional appropriations. Although multi-year operations may be planned in connection with major procurements, Congress generally appropriates funds on a fiscal-year basis even though a program may continue for several years. Consequently, programs often are only partially funded initially, and additional funds are committed only as Congress makes further appropriations. As a result, we are subject to ongoing uncertainties associated with U.S. Government budget restraints and other factors affecting government funding.

The U.S. Government typically can terminate or modify any of its contracts with us either for its convenience or if we default by failing to perform under the terms of the applicable contract. A termination arising out of our default could expose us to liability and have an adverse effect on our ability to compete for future contracts and orders. If any of our contracts reflected in backlog are terminated by the U.S. Government, our backlog would be reduced by the expected value of the remaining work under such contracts. In addition, on those contracts for which we are teamed with others and are not the prime contractor, the U.S. Government could terminate a prime contract under which we are a subcontractor, irrespective of the quality of our products and services as a subcontractor. The reduction or termination of funding, or changes in the timing of funding, for a U.S. Government program in which we provide products or services would result in a reduction or loss of anticipated future revenues attributable to that program and could have a negative impact on our results of operations.

We also have several significant contracts with the U.S. Government that are subject to periodic renewal and rebidding through a competitive process. If the U.S. Government fails to renew these contracts, our results of operations and cash flows would be adversely affected.

As a result of these and other factors, the termination of one or more of our significant government contracts, our suspension from government contract work, the failure of the U.S. Government to renew our existing contracts or the disallowance of the payment of our contract costs could have a material adverse effect on our financial condition, results of operations and cash flows.

Demand for our products and services is vulnerable to economic downturns and reductions in private sector and government spending.

Demand for our products and services has been, and we expect that demand will continue to be, subject to significant fluctuations due to a variety of factors beyond our control, including economic and industry conditions. The global economic downturn that began in 2008 continued throughout 2009 and into 2010. Since 2008, concerns over inflation, energy costs, geopolitical issues, the availability and cost of credit and other market factors have contributed to increased volatility and diminished expectations for the global economy and expectations of slower global economic growth for the foreseeable future. Volatile oil prices, low business and consumer confidence and high unemployment have accompanied the global economic downturn.

As a result of the ongoing economic downturn, some of our customers have delayed, curtailed or cancelled proposed and existing projects and may continue to do so, thus decreasing the overall demand for our products and services and adversely affecting our results of operations. We have experienced and expect to continue to

experience delays or deferrals of proposed projects. For example, for the year ended December 31, 2009, our Power Generation Systems segment experienced a decline in revenues compared to 2008 and an overall slowdown in project awards. In light of current macroeconomic conditions, revenues from our Power Generation Systems segment may continue to decline in 2010.

In addition, our Government Operations segment depends on U.S. Government funding, particularly funding levels at the DOE. Significant changes in the level of funding (for example, the annual budget of the DOE) or specifically mandated levels for individual programs that are important to our business could have an unfavorable impact on us. In addition, if Congress does not pass annual appropriations bills in a timely fashion, spending delays under our U.S. Government contracts may result. Any reduction in the level of U.S. Government funding, particularly at the DOE, may result in, among other things, a reduction in the number and scope of projects put out for bid by the U.S. Government or the curtailment of existing U.S. Government programs, either of which may result in a reduction in the number of contract award opportunities available to us, a reduction of activities at DOE sites and an increase in costs, including the costs of obtaining contract awards.

In addition, our customers may find it more difficult to raise capital in the future due to limitations on the availability of credit, increases in interest rates and other factors affecting the federal, municipal and corporate credit markets. Also, our customers may demand more favorable pricing terms and find it increasingly difficult to timely pay invoices for our services, which would impact our future cash flows and liquidity. Inflation or significant changes in interest rates could reduce the demand for our products and services. Any inability to timely collect our invoices may lead to an increase in our accounts receivables and potentially to increased write-offs of uncollectible invoices. If the economy remains weak or uncertain, or customer spending declines further, then our backlog, revenues, net income and overall financial condition could deteriorate.

Our backlog is subject to unexpected adjustments and cancellations.

There can be no assurance that the revenues projected in our backlog will be realized or, if realized, will result in profits. Because of project cancellations or changes in project scope and schedule, we cannot predict with certainty when or if backlog will be performed. In addition, even where a project proceeds as scheduled, it is possible that contracted parties may default and fail to pay amounts owed to us or poor project performance could increase the cost associated with a project. Delays, suspensions, cancellations, payment defaults, scope changes and poor project execution could materially reduce or eliminate the revenues and profits that we actually realize from projects in backlog.

Reductions in our backlog due to cancellation or modification by a customer or for other reasons may adversely affect, potentially to a material extent, the revenues and earnings we actually receive from contracts included in our backlog. Many of the contracts in our backlog provide for cancellation fees in the event customers cancel projects. These cancellation fees usually provide for reimbursement of our out-of-pocket costs, revenues for work performed prior to cancellation and a varying percentage of the profits we would have realized had the contract been completed. However, we typically have no contractual right upon cancellation to the total revenues reflected in our backlog. Projects may remain in our backlog for extended periods of time. If we experience significant project terminations, suspensions or scope adjustments to contracts reflected in our backlog, our financial condition, results of operations and cash flows may be adversely impacted.

We are subject to risks associated with contractual pricing in our industries, including the risk that, if our actual costs exceed the costs we estimate on our fixed-price contracts, our profitability will decline, and we may suffer losses.

We are engaged in highly competitive industries, and we have priced a substantial number of our projects on a fixed-price basis. Our actual costs could exceed our projections. We attempt to cover the increased costs of anticipated changes in labor, material and service costs of long-term contracts, either through estimates of cost increases, which are reflected in the original contract price, or through price escalation clauses. Despite these

attempts, however, the cost and gross profit we realize on a fixed-price contract could vary materially from the estimated amounts because of supplier, contractor and subcontractor performance, changes in job conditions, variations in labor and equipment productivity and increases in the cost of labor and raw materials, particularly steel, over the term of the contract. These variations and the risks generally inherent in our industries may result in actual revenues or costs being different from those we originally estimated and may result in reduced profitability or losses on projects. Some of these risks include:

- difficulties encountered on our large-scale projects related to the procurement of materials or due to schedule disruptions, equipment performance failures or other factors that may result in additional costs to us, reductions in revenue, claims or disputes;
- our inability to obtain compensation for additional work we perform or expenses we incur as a result of customer change orders or our customers providing deficient design or engineering information or equipment or materials;
- · requirements to pay liquidated damages upon our failure to meet schedule or performance requirements of our contracts; and
- difficulties in engaging third-party subcontractors, equipment manufacturers or materials suppliers or failures by third-party subcontractors, equipment manufacturers or materials suppliers to perform could result in project delays and cause us to incur additional costs.

Our use of percentage-of-completion method of accounting could result in volatility in our results of operations.

We recognize revenues and profits under our long-term contracts on a percentage-of-completion basis. Accordingly, we review contract price and cost estimates periodically as the work progresses and reflect adjustments proportionate to the percentage of completion in income in the period when we revise those estimates. To the extent these adjustments result in a reduction or an elimination of previously reported profits with respect to a project, we would recognize a charge against current earnings, which could be material. Our current estimates of our contract costs and the profitability of our long-term projects, although reasonably reliable when made, could change as a result of the uncertainties associated with these types of contracts, and if adjustments to overall contract costs are significant, the reductions or reversals of previously recorded revenue and profits could be material in future periods.

Maintaining adequate bonding and letter of credit capacity is necessary for us to successfully bid on and win various contracts.

In line with industry practice, we are often required to post standby letters of credit and surety bonds to support contractual obligations to customers as well as other obligations. These letters of credit and bonds generally indemnify customers should we fail to perform our obligations under the applicable contracts. If a letter of credit or bond is required for a particular project and we are unable to obtain it due to insufficient liquidity or other reasons, we will not be able to pursue that project. We utilize bonding facilities, but, as is typically the case, the issuance of bonds under each of those facilities is at the surety's sole discretion. In addition, we have limited capacity under our credit facility for letters of credit. Moreover, due to events that affect the insurance and bonding and credit markets generally, bonding and letters of credit may be more difficult to obtain in the future or may only be available at significant additional cost. There can be no assurance that letters of credit or bonds will continue to be available to us on reasonable terms, particularly given our pro forma tangible net worth as of March 31, 2010. Our inability to obtain adequate letters of credit and bonding and, as a result, to bid on new work could have a material adverse effect on our business, financial condition and results of operations. As of March 31, 2010, we had \$248.4 million in letters of credit issued under our credit facilities and \$112.9 million in surety bonds outstanding.

Volatility and uncertainty of the credit markets may negatively impact our ability to obtain financing.

We intend to finance our existing operations and initiatives with cash and cash equivalents, cash flows from operations, and potential borrowings, including under our revolving credit facility. In the past several years, the credit markets and the financial services industry have been experiencing a period of unprecedented turmoil and upheaval characterized by the bankruptcy, failure, collapse or sale of various financial institutions and an unprecedented level of intervention from the U.S. federal government. These circumstances and events have led to a scarcity of credit, tighter lending standards and higher interest rates on loans. While we cannot predict the ultimate outcome on us, continued turmoil in the credit markets could have a material adverse effect on us. If adverse national and international economic conditions continue or deteriorate further, it is possible that we may not be able to fully draw upon our revolving credit facility and we may not be able to obtain financing on favorable terms.

Our debt obligations could restrict our operations.

The terms of our credit agreement impose various restrictions and covenants on us that could have adverse consequences, including:

- limiting our ability to react to changing economic, regulatory and industry conditions;
- · limiting our ability to compete and our flexibility in planning for, or reacting to, changes in our business and the industry;
- · limiting our ability to pay dividends to our stockholders; and
- · limiting our ability to borrow additional funds.

Our business strategy includes acquisitions to support our growth. Acquisitions of other businesses can create certain risks and uncertainties.

We intend to pursue growth through the acquisition of businesses or assets that we believe will enable us to strengthen our existing businesses and expand into new industries and regions. We may be unable to continue this growth strategy if we cannot identify suitable businesses or assets, reach agreement on potential strategic acquisitions on acceptable terms or for other reasons. Moreover, business acquisitions involve certain risks, including:

- difficulties relating to the assimilation of personnel, services and systems of an acquired business and the assimilation of marketing and other operational capabilities;
- · challenges resulting from unanticipated changes in customer relationships subsequent to acquisition;
- additional financial and accounting challenges and complexities in areas such as tax planning, treasury management, financial reporting and internal
 controls:
- · assumption of liabilities of an acquired business, including liabilities that were unknown at the time the acquisition transaction was negotiated;
- · diversion of management's attention from day-to-day operations;
- · failure to realize anticipated benefits, such as cost savings and revenue enhancements;
- · potentially substantial transaction costs associated with business combinations; and
- · potential impairment of goodwill or other intangible assets resulting from the overpayment for an acquisition.

Acquisitions may be funded by the issuance of additional equity or debt financing, which may not be available on attractive terms. Moreover, to the extent an acquisition transaction financed by non-equity consideration results in goodwill, it will reduce our tangible net worth, which might have an adverse effect on potential credit and bonding capacity.

Additionally, an acquisition may bring us into businesses we have not previously conducted and expose us to additional business risks that are different than those we have historically experienced.

Our business strategy also includes development and commercialization of new technologies to support our growth, which requires significant investment and involves various risks and uncertainties.

Our future growth will depend on our ability to continue to innovate by developing and commercializing new product and service offerings. Investments in new technologies involve varying degrees of uncertainties and risk. Commercial success depends on many factors, including the levels of innovation, the development costs and the availability of capital resources to fund those costs, the levels of competition from others developing similar or other competing technologies, our ability to obtain or maintain government permits or certifications, the effectiveness of production, distribution and marketing efforts, and the costs to customers to deploy and provide support for the new technologies. We may not achieve significant revenue from new product and service investments for a number of years, if at all. Moreover, new products and services may not be profitable, and, even if they are profitable, our operating margins from new products and services may not be as high as the margins we have experienced historically. In addition, new technologies may not be patentable and, as a result, we may face increased competition.

Among our opportunities involving new technologies, we are developing the B&W mPowerTM reactor, a small modular reactor designed with the flexibility to provide between 125 MW to 1,000 MW of electrical power generation (in increments of 125 MW), and the capability to operate for a four- to five-year cycle without refueling. The development, general and administrative and capital costs to develop and commercialize this technology will require a substantial amount of investment over a period of years. We expect that funding will be authorized at major milestones based on measurable and demonstrated progress, and the funding requirements may vary significantly from period to period. We intend to continue with our plan to seek third-party funding and/or participation to pursue the development and commercialization of this technology; however, we can provide no assurance that such third-party funding or participation will be provided. Commercialization of this technology will require certification from the Nuclear Regulatory Commission ("NRC"), which we intend to seek in time to begin deploying this technology as early as 2020. There can be no assurance that we will be successful in addressing all the technological challenges to developing and commercializing this technology or in obtaining the required NRC certification. Furthermore, while there currently are various small reactor competitors with limited capability, the potential exists for other competitors to emerge with competing technologies, in some cases with funding readily available, and we can provide no assurance that those competitors will not develop and commercialize similar or superior technologies sooner than we can or at a significant cost or price advantage.

Our operations are subject to operating risks, which could expose us to potentially significant professional liability, product liability, warranty and other claims. Our insurance coverage may be inadequate to cover all of our significant risks or our insurers may deny coverage of material losses we incur, which could adversely affect our profitability and overall financial condition.

We engineer, construct and perform services in large industrial facilities where accidents or system failures can have significant consequences. Risks inherent in our operations include:

- accidents resulting in injury or the loss of life or property;
- · environmental or toxic tort claims, including delayed manifestation claims for personal injury or loss of life;
- pollution or other environmental mishaps;
- adverse weather conditions:
- mechanical failures;
- · property losses;
- · business interruption due to political action in foreign countries or other reasons; and
- · labor stoppages.

Any accident or failure at a site where we have provided products or services could result in significant professional liability, product liability, warranty and other claims against us, regardless of whether our products or services caused the incident. We have been, and in the future we may be, named as defendants in lawsuits asserting large claims as a result of litigation arising from events such as those listed above.

We endeavor to identify and obtain in established markets insurance agreements to cover significant risks and liabilities. Insurance against some of the risks inherent in our operations is either unavailable or available only at rates or on terms that we consider uneconomical. Also, catastrophic events customarily result in decreased coverage limits, more limited coverage, additional exclusions in coverage, increased premium costs and increased deductibles and self-insured retentions. Risks that we have frequently found difficult to cost-effectively insure against include, but are not limited to, business interruption, property losses from wind, flood and earthquake events, nuclear hazards, war and confiscation or seizure of property in some areas of the world, pollution liability, liabilities related to occupational health exposures (including asbestos), professional liability/errors and omissions coverage, the failure, misuse or unavailability of our information systems, the failure of security measures designed to protect our information systems, and liability related to risk of loss of our work in progress and customer-owned materials in our care, custody and control. Depending on competitive conditions and other factors, we endeavor to obtain contractual protection against certain uninsured risks from our customers. When obtained, such contractual indemnification protection may not be as broad as we desire or may not be supported by adequate insurance maintained by the customer. Such insurance or contractual indemnity protection may not be sufficient or effective under all circumstances or against all hazards to which we may be subject. A successful claim for which we are not insured or for which we are underinsured could have a material adverse effect on us. Additionally, disputes with insurance carriers over coverage may affect the timing of cash flows and, if litigation with the carrier becomes necessary, an outcome unfavorable to us may have a material adverse effect on our results of operations.

Through two limited liability companies, we are also involved in management and operating activities for the U.S. Government at the Y-12 National Security Complex and Pantex Plant facilities. These activities involve, among other things, handling nuclear devices and their components from the aging stockpile of the U.S. Government. Most insurable liabilities arising from these sites are not protected in our corporate insurance program. Instead, we rely on government contractual agreements, some insurance purchased specifically for the sites and certain specialized self-insurance programs funded by the U.S. Government. The U.S. Government has historically fulfilled its contractual agreement to reimburse for insurable claims, and we expect it to continue this process during our administration of these two facilities. However, it should be noted that, in most situations, the U.S. Government is contractually obligated to pay subject to the availability of authorized government funds. The reimbursement obligation of the U.S. Government is also conditional, and provisions of the relevant contract or applicable law may preclude reimbursement.

In January 2010, we received notice from our existing nuclear liability underwriter that it intended to cancel our nuclear liability insurance, effective as of May 10, 2010, for our licensed facility in Erwin, Tennessee, which we acquired in December 2008 through our acquisition of Nuclear Fuel Services, Inc. The underwriter stated that the reason for the cancellation was a material change in risk that increased the risk of loss, and that this assessment was based on new information regarding the commercial development line at the facility. After the occurrence of the two incidents at the facility described under "Our nuclear operations subject us to various environmental, regulatory, financial and other risks" and prior to suspension of operations at the facility, the underwriter conducted a special inspection of the facility. We believe the underwriter's decision to cancel the policy was based on its assessment of insured risk following that inspection. The cancellation occurred after the suspension of operations at the facility. We have obtained replacement coverage from a new nuclear insurance underwriter to cover the facility, which became effective as of April 1, 2010. There can be no assurance, however, that any of our insurance coverages will be renewable upon the expiration of the coverage period or that future coverage will be available at preferred or required limits, which could increase our overall risk exposure for certain commercial activities at the Erwin facility.

We have a captive insurance company subsidiary which provides us with various insurance coverages. Claims as a result of our operations could adversely impact the ability of our captive insurance company subsidiary to respond to all claims presented.

Additionally, upon the February 22, 2006 effectiveness of the settlement relating to the Chapter 11 proceedings involving several of our subsidiaries, most of our subsidiaries contributed substantial insurance rights providing coverage for, among other things, asbestos and other personal injury claims, to an asbestos personal injury trust. With the contribution of these insurance rights to the asbestos personal injury trust, we may have underinsured or uninsured exposure for non-derivative asbestos claims or other personal injury or other claims that would have been insured under these coverages had the insurance rights not been contributed to the asbestos personal injury trust.

Our nuclear operations subject us to various environmental, regulatory, financial and other risks.

Our operations in designing, engineering, manufacturing, supplying, constructing and maintaining nuclear fuel and nuclear power equipment and components subject us to various risks, including:

- potential liabilities relating to harmful effects on the environment and human health resulting from nuclear operations and the storage, handling and disposal of radioactive materials;
- · unplanned expenditures relating to maintenance, operation, security, upgrades and repairs required by the NRC;
- · limitations on the amounts and types of insurance commercially available to cover losses that might arise in connection with nuclear operations; and
- potential liabilities arising out of a nuclear incident, whether or not it is within our control.

Our nuclear operations are subject to various safety-related requirements imposed by the U.S. Government, the DOE and the NRC. In the event of non-compliance, these agencies might increase regulatory oversight, impose fines or shut down our operations, depending upon the assessment of the severity of the situation. Revised security and safety requirements promulgated by these agencies could necessitate substantial capital and other expenditures.

In December 2009, our subsidiary Nuclear Fuel Services, Inc., which we acquired in December 2008, implemented a suspension of some operations at its Erwin, Tennessee manufacturing facility while implementing organizational, facility and management changes to enhance safety controls and processes. We suspended the operations in consultation with the NRC following the occurrence of two separate incidents that we reported to the NRC in October and November 2009. The October 2009 incident involved the generation of excessive heat and a hazardous gas at a specialized cleaning station. The incident was caused by the processing of a small amount of uranium-aluminum material in nitric acid. This incident resulted in some damage to piping, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. The November 2009 incident involved a fire in a glovebox enclosure. The incident was caused by a reaction between fluorine gas in a cylinder of uranium and the fiberglass backing of the teflon lining of a vent hose attached to the cylinder within the glovebox enclosure. This incident resulted in damage to a hose and a faceplate within the glove box enclosure, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. Inspections conducted separately by the NRC, our then existing nuclear liability underwriter and us revealed specific modifications necessary to improve Nuclear Fuel Services, Inc.'s overall safety performance. Although the NRC did not issue any directive or order to us to suspend these operations, we believe the NRC likely would have done so had we not voluntarily suspended the operations. Although we were already in the process of implementing safety, cultural and procedural improvements at Nuclear Fuel Services, Inc., we have refined and accelerated those changes and have developed other specific enhancements in consultation with the NRC, as confirmed in the NRC's January 7, 2010 confirmatory action letter to Nuclear Fue

The suspended operations include production operations, the commercial development line and the highly enriched uranium down-blending facility. These operations are in the process of being brought back on line through a phased restart, which we commenced in March 2010 following completion of a third-party review and NRC review of the modifications that were implemented. The production operations are now fully operational. We began bringing the highly enriched uranium down-blending facility back on line, through a phased restart.

in May 2010, and we expect that the facility will be fully operational in June 2010. We expect to bring the commercial development line back on line by the end of January 2011. There can be no assurance that we will not have to suspend our operations in the future to implement additional changes to enhance our safety controls and processes in order to comply with applicable laws and regulations.

Limitations or modifications to indemnification regulations of the U.S. or foreign countries could adversely affect our business.

The Price-Anderson Act partially indemnifies the nuclear industry against liability arising from nuclear incidents in the United States, while ensuring compensation for the general public. The Price-Anderson Act comprehensively regulates the manufacture, use and storage of radioactive materials, while promoting the nuclear industry by offering broad indemnification to commercial nuclear power plant operators and DOE contractors. Because we provide nuclear fabrication and other services to the DOE relating to its nuclear devices facilities and other programs and the nuclear power industry in the ongoing maintenance and modifications of its nuclear power plants, including the manufacture of equipment and other components for use in such nuclear power plants, we may be entitled to some of the indemnification protections under the Price-Anderson Act against liability arising from nuclear incidents in the United States. The indemnification authority under the Price-Anderson Act was extended through December 2025 by the Energy Policy Act of 2005. We also provide nuclear fabrication and other services to the nuclear power industry in Canada. Canada's Nuclear Liability Act generally conforms to international conventions and is conceptually similar to the Price-Anderson Act in the United States. Accordingly, indemnification protections and the possibility of exclusions under Canada's Nuclear Liability Act are similar to those under the Price-Anderson Act in the United States.

The Price-Anderson Act's indemnification provisions may not apply to all liabilities that we might incur while performing services as a contractor for the DOE and the nuclear power industry. If an incident or evacuation is not covered under the Price-Anderson Act's indemnification provisions, we could be held liable for damages, regardless of fault, which could have an adverse effect on our results of operations and financial condition. In connection with the international transportation of toxic, hazardous and radioactive materials, it is possible for a claim to be asserted which may not fall within the indemnification provided by the Price-Anderson Act. If such indemnification authority is not applicable in the future, our business could be adversely affected if the owners and operators of nuclear power plants fail to retain our services in the absence of commercially adequate insurance and indemnification.

Moreover, because we manufacture nuclear components for the U.S. Government's defense program, we may be entitled to some of the indemnification protections afforded by Public Law 85-804 for certain of our nuclear operations risks. Public Law 85-804 authorizes certain agencies of the U.S. Government, such as the DOE and the U.S. Department of Defense, to indemnify their contractors against unusually hazardous or nuclear risks when such action would facilitate the national defense. However, because the indemnification protections afforded by Public Law 85-804 are granted on a discretionary basis, situations could arise where the U.S. Government elects not to offer such protections. In such situations, our business could be adversely affected by either our inability to obtain commercially adequate insurance or indemnification or our refusal to pursue such operations in the absence of such protections.

We are subject to government regulations that may adversely affect our future operations.

Many aspects of our operations and properties are affected by political developments and are subject to both domestic and foreign governmental regulations, including those relating to:

- · constructing and manufacturing power generation products and nuclear components;
- · currency conversions and repatriation;
- · clean air and other environmental protection legislation;

- · taxation of foreign earnings and earnings of expatriate personnel; and
- · use of local employees and suppliers.

In addition, a substantial portion of the demand for our products and services is from electric power generating companies and other steam-using customers. The demand for power generation products and services can be influenced by state and federal governmental legislation setting requirements for utilities related to operations, emissions and environmental impacts. The legislative process is unpredictable and includes a platform that continuously seeks to increase the restrictions on power producers. Potential legislation limiting emissions from power plants, including carbon dioxide, could affect our markets and the demand for our products and services related to power generation.

We cannot determine the extent to which our future operations and earnings may be affected by new legislation, new regulations or changes in existing regulations.

Our businesses require us to obtain, and to comply with, national, state and local government permits and approvals.

Our businesses are required to obtain, and to comply with, national, state and local government permits and approvals. Any of these permits or approvals may be subject to denial, revocation or modification under various circumstances. Failure to obtain or comply with the conditions of permits or approvals may adversely affect our operations by temporarily suspending our activities or curtailing our work and may subject us to penalties and other sanctions. Although existing licenses are routinely renewed by various regulators, renewal could be denied or jeopardized by various factors, including:

- · failure to provide adequate financial assurance for decommissioning or closure;
- · failure to comply with environmental and safety laws and regulations or permit conditions;
- local community, political or other opposition;
- executive action; and
- legislative action.

In addition, if new environmental legislation or regulations are enacted or implemented, or existing laws or regulations are amended or are interpreted or enforced differently, we may be required to obtain additional operating permits or approvals. Our inability to obtain, and to comply with, the permits and approvals required for our businesses could have a material adverse effect on us.

We rely on intellectual property law and confidentiality agreements to protect our intellectual property. We also rely on intellectual property we license from third parties. Our failure to protect our intellectual property rights, or our inability to obtain or renew licenses to use intellectual property of third parties, could adversely affect our business.

Our success depends, in part, on our ability to protect our proprietary information and other intellectual property. Our intellectual property could be challenged, invalidated, circumvented or rendered unenforceable. In addition, effective intellectual property protection may be limited or unavailable in some foreign countries where we operate.

Our failure to protect our intellectual property rights may result in the loss of valuable technologies or adversely affect our competitive business position. We rely significantly on proprietary technology, information, processes and know-how that are not subject to patent or copyright protection. We seek to protect this information through trade secret or confidentiality agreements with our employees, consultants, subcontractors or other parties, as well as through other security measures. These agreements and security measures may be

inadequate to deter or prevent misappropriation of our confidential information. In the event of an infringement of our intellectual property rights, a breach of a confidentiality agreement or divulgence of proprietary information, we may not have adequate legal remedies to protect our intellectual property. Litigation to determine the scope of intellectual property rights, even if ultimately successful, could be costly and could divert management's attention away from other aspects of our business. In addition, our trade secrets may otherwise become known or be independently developed by competitors.

In some instances, we have augmented our technology base by licensing the proprietary intellectual property of third parties. In the future, we may not be able to obtain necessary licenses on commercially reasonable terms.

Our operations involve the handling, transportation and disposal of radioactive and hazardous materials, and environmental laws and regulations and civil liability for contamination of the environment or related personal injuries may result in increases in our operating costs and capital expenditures and decreases in our earnings and cash flow.

Our operations involve the handling, transportation and disposal of radioactive and hazardous materials, including nuclear devices and their components. Failure to properly handle these materials could pose a health risk to humans or wildlife and could cause personal injury and property damage (including environmental contamination). If an accident were to occur, its severity could be significantly affected by the volume of the materials and the speed of corrective action taken by emergency response personnel, as well as other factors beyond our control, such as weather and wind conditions. Actions taken in response to an accident could result in significant costs.

Governmental requirements relating to the protection of the environment, including solid waste management, air quality, water quality, the decontamination and decommissioning of nuclear manufacturing and processing facilities and cleanup of contaminated sites, have had a substantial impact on our operations. These requirements are complex and subject to frequent change. In some cases, they can impose liability for the entire cost of cleanup on any responsible party without regard to negligence or fault and impose liability on us for the conduct of others or conditions others have caused, or for our acts that complied with all applicable requirements when we performed them. Our compliance with amended, new or more stringent requirements, stricter interpretations of existing requirements or the future discovery of contamination may require us to make material expenditures or subject us to liabilities that we currently do not anticipate. Such expenditures and liabilities may adversely affect our business, financial condition, results of operations and cash flows. In addition, some of our operations and the operations of predecessor owners of some of our properties have exposed us to civil claims by third parties for liability resulting from alleged contamination of the environment or personal injuries caused by releases of hazardous substances into the environment. See "Business—Governmental Regulations and Environmental Matters."

In our contracts, we seek to protect ourselves from liability associated with accidents, but there can be no assurance that such contractual limitations on liability will be effective in all cases or that our or our customers' insurance will cover all the liabilities we have assumed under those contracts. The costs of defending against a claim arising out of a nuclear incident or precautionary evacuation, and any damages awarded as a result of such a claim, could adversely affect our results of operations and financial condition.

We maintain insurance coverage as part of our overall risk management strategy and due to requirements to maintain specific coverage in our financing agreements and in many of our contracts. These policies do not protect us against all liabilities associated with accidents or for unrelated claims. In addition, comparable insurance may not continue to be available to us in the future at acceptable prices, or at all.

We conduct a portion of our operations through joint venture entities, over which we may have limited control.

We currently have equity interests in several significant joint ventures and may enter into additional joint venture arrangements in the future. We do not manage all of these entities. Even in those joint ventures that we

manage, we are often required to consider the interests of our joint venture partners in connection with decisions concerning the operations of the joint ventures. In any case, differences in views among the joint venture participants may result in delayed decisions or disputes. We also cannot control the actions of our joint venture participants, and we sometimes have joint and several liabilities with our joint venture partners under the applicable contracts for joint venture projects. These factors could potentially harm the business and operations of a joint venture and, in turn, our business and operations.

Operating through joint ventures in which we are minority holders results in us having limited control over many decisions made with respect to projects and internal controls relating to projects. These joint ventures may not be subject to the same requirements regarding internal controls and internal control over financial reporting that we follow. As a result, internal control problems may arise with respect to the joint ventures that could adversely affect our ability to respond to requests or contractual obligations to customers or to meet the internal control requirements to which we are otherwise subject.

In addition, our arrangements involving joint ventures may restrict us from gaining access to the cash flows or assets of these entities. In some cases, our joint ventures have governmentally imposed restrictions on their abilities to transfer funds to us.

If our co-venturers fail to perform their contractual obligations on a project or if we fail to coordinate effectively with our co-venturers, we could be exposed to legal liability, loss of reputation and reduced profit on the project.

We often perform projects jointly with third parties. For example, we enter into contracting consortia and other contractual arrangements to bid for and perform jointly on large projects. Success on these joint projects depends in part on whether our co-venturers fulfill their contractual obligations satisfactorily. If any one or more of these third parties fail to perform their contractual obligations satisfactorily, we may be required to make additional investments and provide additional services in order to compensate for that failure. If we are unable to adequately address any such performance issues, then our customer may exercise its right to terminate a joint project, exposing us to legal liability, loss of reputation and reduced profit.

Our collaborative arrangements also involve risks that participating parties may disagree on business decisions and strategies. These disagreements could result in delays, additional costs and risks of litigation. Our inability to successfully maintain existing collaborative relationships or enter into new collaborative arrangements could have a material adverse effect on our results of operations.

Employee, agent or partner misconduct or our overall failure to comply with laws or regulations could weaken our ability to win contracts, lead to the suspension of our operations and result in reduced revenues and profits.

Misconduct, fraud, non-compliance with applicable laws and regulations, or other improper activities by one or more of our employees, agents or partners could have a significant negative impact on our business and reputation. Such misconduct could include the failure to comply with government procurement regulations, regulations regarding the protection of classified information, regulations regarding the pricing of labor and other costs in government contracts, regulations on lobbying or similar activities, regulations pertaining to the internal controls over financial reporting and various other applicable laws or regulations. For example, we regularly provide services that may be highly sensitive or that are related to critical national security matters; if a security breach were to occur, our ability to procure future government contracts could be severely limited. The precautions we take to prevent and detect these activities may not be effective, and we could face unknown risks or losses.

We are routinely audited and reviewed by the U.S. Government and its agencies. These agencies review our performance under our contracts, our cost structure and our compliance with applicable laws, regulations and standards, as well as the adequacy of, and our compliance with, our internal control systems and policies. Systems

that are subject to review include our purchasing systems, billing systems, property management and control systems, cost estimating systems, compensation systems and management information systems. Any costs found to be improperly allocated to a specific contract will not be reimbursed or must be refunded if already reimbursed. If an audit or review uncovers improper or illegal activities, we could be subject to civil and criminal penalties and administrative sanctions, including termination of contracts, forfeiture of profits, suspension of payments, fines, loss of security clearance and suspension or debarment from contracting with the U.S. Government. In addition, we could suffer serious reputational harm if allegations of impropriety were made against us.

We could be adversely affected by violations of the U.S. Foreign Corrupt Practices Act.

The U.S. Foreign Corrupt Practices Act ("FCPA") generally prohibits companies and their intermediaries from making improper payments to non-U.S. officials. Our training program and policies mandate compliance with the FCPA. We operate in some parts of the world that have experienced governmental corruption to some degree, and, in certain circumstances, strict compliance with anti-bribery laws may conflict with local customs and practices. Although we have procedures and controls in place to monitor internal and external compliance, if we are found to be liable for violations of the FCPA (either due to our own acts or our inadvertence, or due to the acts or inadvertence of others), we could suffer from civil and criminal penalties or other sanctions.

We may not be able to compete successfully against current and future competitors.

Some of our competitors or potential competitors have greater financial or other resources than we have. Our operations may be adversely affected if our current competitors or new market entrants introduce new products or services with better features, performance, prices or other characteristics than those of our products and services. Furthermore, we operate in industries where capital investment is critical. We may not be able to obtain as much purchasing and borrowing leverage and access to capital for investment as an independent public company, which may impair our ability to compete against competitors or potential competitors.

The loss of the services of one or more of our key personnel, or our failure to attract, assimilate and retain trained personnel in the future, could disrupt our operations and result in loss of revenues.

Our success depends on the continued active participation of our executive officers and key operating personnel. The unexpected loss of the services of any one of these persons could adversely affect our operations.

Our operations require the services of employees having the technical training and experience necessary to obtain the proper operational results. As such, our operations depend, to a considerable extent, on the continuing availability of such personnel. If we should suffer any material loss of personnel to competitors or be unable to employ additional or replacement personnel with the requisite level of training and experience to adequately operate our business, our operations could be adversely affected. While we believe our wage rates are competitive and our relationships with our employees are satisfactory, a significant increase in the wages paid by other employers could result in a reduction in our workforce, increases in wage rates, or both. If either of these events occurred for a significant period of time, our financial condition, results of operations and cash flows could be adversely impacted.

Negotiations with labor unions and possible work stoppages and other labor problems could divert management attention and disrupt operations. In addition, new collective bargaining agreements or amendments to agreements could increase our labor costs and operating expenses.

A significant number of our employees are members of labor unions. Although we have recently negotiated and renewed nearly all of our current union contracts for multi-year terms without incident, if we are unable to negotiate acceptable new contracts with our unions in the future, we could experience strikes or other work stoppages by the affected employees. If any such strikes or other work stoppages were to occur, we could experience a significant disruption of operations. In addition, negotiations with unions could divert management

attention. New union contracts could result in increased operating costs, as a result of higher wages or benefit expenses, for both union and nonunion employees. If nonunion employees were to unionize, we would experience higher ongoing labor costs.

Pension and medical expenses associated with our retirement benefit plans may fluctuate significantly depending on changes in actuarial assumptions, future market performance of plan assets, future trends in health care costs and legislative or other regulatory actions.

A substantial portion of our current and retired employee population is covered by pension and post-retirement benefit plans, the costs and funding requirements of which depend on our various assumptions, including estimates of rates of return on benefit-related assets, discount rates for future payment obligations, rates of future cost growth and trends for future costs. Variances from these estimates could have a material adverse effect on us. In addition, funding requirements for benefit obligations of our pension and post-retirement benefit plans are subject to legislative and other government regulatory actions. As of December 31, 2009, our defined benefit pension and post-retirement benefit plans were underfunded by approximately \$869 million. See Note 7 to the combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement.

Our internal controls may not be sufficient to achieve all stated goals and objectives.

Our internal controls and procedures were developed through a process in which our management applied its judgment in assessing the costs and benefits of such controls and procedures, which, by their nature, can provide only reasonable assurance regarding the control objectives. You should note that the design of any system of internal controls and procedures is based in part upon various assumptions about the likelihood of future events, and we cannot assure you that any design will succeed in achieving its stated goals under all potential future conditions, regardless of how remote.

Our Government Operations segment relies on several single-source suppliers, which could, under certain circumstances, adversely affect our revenues and operating results.

Our Government Operations segment relies on several single-source suppliers for materials used in its products. If the supply of a single-sourced material is delayed or ceases, we may not be able to produce the related product in a timely manner or in sufficient quantities, if at all, which could adversely affect our revenues and operating results. In addition, a single-source supplier of a key component could potentially exert significant bargaining power over price, quality, warranty claims, or other terms relating to the single-sourced materials.

Systems and information technology interruption could adversely impact our ability to operate.

In 2010, we expect to replace current key financial and human resources legacy systems with enterprise systems. This implementation subjects us to inherent costs and risks associated with replacing and changing these systems, including potential disruption of our internal control structure, substantial capital expenditures, demands on management time and other risks of delays or difficulties in transitioning to new systems or of integrating new systems into our current systems. Our systems implementations may not result in productivity improvements at the levels anticipated, or at all. In addition, the implementation of new technology systems may cause disruptions in our business operations. This disruption and any other information technology system disruptions and our ability to mitigate those disruptions, if not anticipated and appropriately mitigated, could have a material adverse effect on us.

Our international operations are subject to political, economic and other uncertainties not generally encountered in our domestic operations.

We derive a portion of our revenues from international operations. Our international operations are subject to political, economic and other uncertainties not generally encountered in our U.S. operations. These include:

- risks of war, terrorism and civil unrest;
- · expropriation, confiscation or nationalization of our assets;

- · renegotiation or nullification of our existing contracts;
- changing political conditions and changing laws and policies affecting trade and investment;
- overlap of different tax structures; and
- · risk of changes in foreign currency exchange rates.

Various foreign jurisdictions have laws limiting the right and ability of foreign subsidiaries and joint ventures to pay dividends and remit earnings to affiliated companies. Our international operations sometimes face the additional risks of fluctuating currency values, hard currency shortages and controls of foreign currency exchange.

War, other armed conflicts or terrorist attacks could have a material adverse effect on our business.

The wars in Iraq and Afghanistan and terrorist attacks and unrest have caused and may continue to cause instability in the world's financial and commercial markets and have significantly increased political and economic instability in some of the geographic areas in which we operate. Threats of war or other armed conflict may cause further disruption to financial and commercial markets. In addition, continued unrest could lead to acts of terrorism in the United States or elsewhere, and acts of terrorism could be directed against companies such as ours. Also, acts of terrorism and threats of armed conflicts in or around various areas in which we operate could limit or disrupt our markets and operations, including disruptions from evacuation of personnel, cancellation of contracts or the loss of personnel or assets. Armed conflicts, terrorism and their effects on us or our markets may significantly affect our business and results of operations in the future.

Risks Relating to the Spin-Off

We may be unable to achieve some or all of the benefits that we expect to achieve from our separation from McDermott.

As an independent public company, we believe that we will be able to more effectively focus on our operations and growth strategies than we could as a subsidiary of McDermott. However, by separating from McDermott there is a risk that our results of operations and cash flows may be susceptible to greater volatility due to fluctuations in our business levels and other factors that may adversely affect our operating and financial performance. In addition, as part of the McDermott group of companies, we have enjoyed certain benefits from McDermott's financial resources, including substantial borrowing capacity and capital for investment. As a result of the fact that McDermott's other operations will no longer be available to offset any volatility in our results of operations and cash flows and McDermott's financial and other resources will no longer be available to us, we may not be able to achieve some or all of the benefits that we expect to achieve as an independent public company.

Our historical combined and pro forma financial information are not necessarily indicative of our future financial condition, future results of operations or future cash flows nor do they reflect what our financial condition, results of operations or cash flows would have been as an independent public company during the periods presented.

The historical combined financial information we have included in this information statement does not reflect what our financial condition, results of operations or cash flows would have been as an independent public company during the periods presented and is not necessarily indicative of our future financial condition, future results of operations or future cash flows. This is primarily a result of the following factors:

- our historical combined financial results reflect allocations of expenses for services historically provided by McDermott, and those allocations may be significantly lower than the comparable expenses we would have incurred as an independent company;
- · our cost of debt and other capitalization may be significantly different from that reflected in our historical combined financial statements;
- the historical combined financial information does not reflect the changes that will occur in our cost structure, management, financing arrangements and business operations as a result of our separation from McDermott, including the costs related to being an independent company; and

• the historical combined financial information does not reflect the effects of certain liabilities that will be assumed by our company and does reflect the effects of certain assets that will be transferred to, and liabilities that will be assumed by, McDermott.

The pro forma adjustments are based on available information and assumptions that we believe are factually supportable. However, it is possible that our assumptions may prove not to be accurate. In addition, our unaudited pro forma combined financial statements do not give effect to certain on-going additional costs that we expect to incur in connection with being an independent public company. Accordingly, our unaudited pro forma combined financial statements do not reflect what our financial condition or results of operations would have been as an independent public company and are not necessarily indicative of our future financial condition or future results of operations. Please refer to "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Unaudited Pro Forma Combined Financial Data" and our historical combined financial statements and the notes to those statements included in this information statement.

We do not have a recent history of operating as an independent public company, we may encounter difficulties in making the changes necessary to operate as an independent public company, and we may incur greater costs as an independent public company.

We have historically used McDermott's infrastructure to support our business functions, including the following functions:

- · accounting and financial reporting;
- · information technology and communications;
- legal
- · human resources and employee benefits;
- · tax administration; and
- treasury and corporate finance.

Following the spin-off, we will no longer have access to McDermott's infrastructure, and we will need to establish our own. We expect to incur costs in 2010 to establish the necessary infrastructure to enable us to establish these business functions. We currently pay McDermott for these services on a cost-allocation basis. Following the spin-off, McDermott will continue to provide some of these services to us on a transitional basis, pursuant to transition services agreements we have entered into with McDermott. For more information regarding the transition services agreements, see "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Transition Services Agreements." At the end of this transition period, we will need to perform these functions ourselves or hire third parties to perform these functions on our behalf. The costs associated with performing or outsourcing these functions may exceed those charged by McDermott when we were part of McDermott or during the transition period. A significant increase in the costs of performing or outsourcing these functions could adversely affect our business, financial condition, results of operations and cash flows.

Our accounting and other management systems and resources may not be adequately prepared to meet the financial reporting and other requirements to which we will be subject following the spin-off and may strain our resources.

Our financial results previously were included within the consolidated results of McDermott, and our reporting and control systems were appropriate for those of subsidiaries of a public company. However, we were not directly subject to reporting and other requirements of the Securities Exchange Act of 1934, as amended, which we refer to as the Exchange Act. As a result of the spin-off, we will be directly subject to reporting and other obligations under the Exchange Act, including the requirements of Section 404 of the Sarbanes-Oxley Act of 2002, which will require, beginning with the filing of our Annual Report on Form 10-K for the year ending December 31, 2011, annual management assessments of the effectiveness of our internal controls over financial

reporting and a report by our independent registered public accounting firm addressing these assessments. These reporting and other obligations will place significant demands on our management and administrative and operational resources, including accounting resources.

To comply with these requirements, we anticipate that we will need to upgrade our systems, including information technology, implement additional financial and management controls, reporting systems and procedures and hire additional accounting and finance staff. We expect to incur additional annual expenses related to these steps and, among other things, directors and officers liability insurance, director fees, SEC reporting, transfer agent fees, increased auditing and legal fees and similar expenses, which expenses may be significant. If we are unable to upgrade our financial and management controls, reporting systems, information technology and procedures in a timely and effective fashion, our ability to comply with our financial reporting requirements and other rules that apply to reporting companies under the Exchange Act could be impaired. Any failure to achieve and maintain effective internal controls could have an adverse effect on our business, financial condition, results of operations and cash flows.

We will be subject to continuing contingent liabilities of McDermott following the spin-off.

After the spin-off, there will be several significant areas where the liabilities of McDermott may become our obligations. For example, under the Code and the related rules and regulations, each corporation that was a member of our consolidated tax reporting group during any taxable period or portion of any taxable period ending on or before the effective time of the spin-off is jointly and severally liable for the federal income tax liability of our entire consolidated tax reporting group for that taxable period. We have entered into a tax sharing agreement with a subsidiary of McDermott that allocates the responsibility for prior period taxes of our consolidated tax reporting group between us and McDermott and its subsidiaries. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Tax Sharing Agreement." However, if the subsidiary of McDermott were unable to pay, we could be required to pay the entire amount of such taxes. Other provisions of federal law establish similar liability for other matters, including laws governing tax-qualified pension plans as well as other contingent liabilities.

The spin-off could result in substantial tax liability.

McDermott has obtained a private letter ruling from the IRS substantially to the effect that, for U.S. federal income tax purposes, the spin-off will qualify under Section 355 of the Code and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code. The continued effectiveness of the private letter ruling is a condition to the completion of the spin-off. If the factual assumptions or representations made in the private letter ruling request are inaccurate or incomplete in any material respect, then we will not be able to rely on the ruling. Furthermore, the IRS will not rule on whether a distribution such as the spin-off satisfies certain requirements necessary to obtain tax-free treatment under section 355 of the Code. Rather, the private letter ruling is based on representations by McDermott that those requirements have been satisfied, and any inaccuracy in those representations could invalidate the ruling.

The spin-off is also conditioned on McDermott's receipt of an opinion of Baker Botts L.L.P., in form and substance satisfactory to McDermott, substantially to the effect that, for U.S. federal income tax purposes, the spin-off will qualify under Section 355 of the Code and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code. The opinion will rely on, among other things, the continuing validity of the private letter ruling and various assumptions and representations as to factual matters made by McDermott and us which, if inaccurate or incomplete in any material respect, would jeopardize the conclusions reached by such counsel in its opinion. The opinion will not be binding on the IRS or the courts, and there can be no assurance that the IRS or the courts will not challenge the conclusions stated in the opinion or that any such challenge would not prevail.

Neither we nor McDermott are aware of any facts or circumstances that would cause the assumptions or representations relied on in the private letter ruling or that will be relied on in the opinion to be inaccurate or

incomplete in any material respect. If, notwithstanding receipt of the private letter ruling and opinion, the spin-off were determined not to qualify under Section 355 of the Code, each U.S. holder of McDermott common stock who receives shares of our common stock in the spin-off would generally be treated as receiving a taxable distribution of property in an amount equal to the fair market value of the shares of our common stock received. That distribution would be taxable to each such stockholder as a dividend to the extent of McDermott's current and accumulated earnings and profits. For each such stockholder, any amount that exceeded McDermott's earnings and profits would be treated first as a non-taxable return of capital to the extent of such stockholder's tax basis in its shares of McDermott common stock with any remaining amount being taxed as a capital gain. In addition, if certain related preparatory transactions were to fail to qualify for tax-free treatment, we would be treated as if we had sold part of our assets (which will be retained by McDermott) in a taxable sale for fair market value and McDermott could be treated as receiving such assets from us as a taxable dividend. See "The Spin-Off—Material U.S. Federal Income Tax Consequences of the Spin-Off."

Under the terms of the tax sharing agreement we have entered into in connection with the spin-off, we will generally be responsible for any taxes imposed on us or McDermott and its subsidiaries in the event that the spin-off and/or certain related preparatory transactions were to fail to qualify for tax-free treatment. However, if the spin-off and/or certain related preparatory transactions were to fail to qualify for tax-free treatment because of actions or failures to act by McDermott or its subsidiaries, a subsidiary of McDermott would be responsible for all such taxes. If we are liable for taxes under the tax sharing agreement, that liability could have a material adverse effect on us. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Tax Sharing Agreement."

Potential liabilities associated with certain obligations under the tax sharing agreement cannot be precisely quantified at this time.

Under the terms of the tax sharing agreement we have entered into in connection with the spin-off, we will generally be responsible for all taxes attributable to us or any of our subsidiaries, whether accruing before, on or after the date of the spin-off. We have also agreed to be responsible for all taxes imposed on us or McDermott and its subsidiaries in the event the spin-off and/or certain related preparatory transactions were to fail to qualify for tax-free treatment. However, if the spin-off and/or certain related preparatory transactions were to fail to qualify for tax-free treatment because of actions or failures to act by McDermott or its subsidiaries, a subsidiary of McDermott would be responsible for all such taxes. Our liabilities under the tax sharing agreement could have a material adverse effect on us. At this time, we cannot precisely quantify the amount of liabilities we may have under the tax sharing agreement and there can be no assurances as to their final amounts. For a more detailed discussion, see "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Tax Sharing Agreement."

Under some circumstances, we could be liable for any resulting adverse tax consequences from engaging in significant strategic or capital raising transactions.

Even if the spin-off otherwise qualifies as a tax-free distribution under section 355 of the Code, certain preparatory transactions related to the spin-off may result in significant U.S. federal income tax liabilities to us under Section 355(e) or applicable provisions of the Code if 50% or more of McDermott's stock or our stock (in each case, by vote or value) is treated as having been acquired, directly or indirectly, by one or more persons as part of a plan (or series of related transactions) that includes the spin-off. Any acquisitions of McDermott stock or our stock (or similar acquisitions), or any understanding, arrangement or substantial negotiations regarding such an acquisition of McDermott stock or our stock (or similar acquisitions), within two years before or after the spin-off are subject to special scrutiny. The process for determining whether an acquisition triggering those provisions has occurred is complex, inherently factual and subject to interpretation of the facts and circumstances of a particular case.

Under the terms of the tax sharing agreement we have entered into in connection with the spin-off, we are generally liable for any such tax liabilities. However, a subsidiary of McDermott will be required to indemnify us against any such tax liabilities that result from actions taken or failures to act by McDermott or its

subsidiaries. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Tax Sharing Agreement." As a result of these rules and contractual provisions, we may be unable to engage in strategic or capital raising transactions that our stockholders might consider favorable, or to structure potential transactions in the manner most favorable to us, without certain adverse tax consequences.

Potential indemnification liabilities to McDermott pursuant to the master separation agreement could materially adversely affect our company.

The master separation agreement with McDermott provides for, among other things, the principal corporate transactions required to effect the spin-off, certain conditions to the spin-off and provisions governing the relationship between our company and McDermott with respect to and resulting from the spin-off. For a description of the master separation agreement, see "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Master Separation Agreement." Among other things, the master separation agreement provides for indemnification obligations designed to make our company financially responsible for substantially all liabilities that may exist relating to our business activities, whether incurred prior to or after the spin-off, as well as those obligations of McDermott assumed by us pursuant to the master separation agreement. If we are required to indemnify McDermott under the circumstances set forth in the master separation agreement, we may be subject to substantial liabilities.

In connection with our separation from McDermott, McDermott will indemnify us for certain liabilities. However, there can be no assurance that the indemnity will be sufficient to insure us against the full amount of such liabilities, or that McDermott's ability to satisfy its indemnification obligation will not be impaired in the future.

Pursuant to the master separation agreement, McDermott has agreed to indemnify us for certain liabilities. However, third parties could seek to hold us responsible for any of the liabilities that McDermott has agreed to retain, and there can be no assurance that the indemnity from McDermott will be sufficient to protect us against the full amount of such liabilities, or that McDermott will be able to fully satisfy its indemnification obligations. Moreover, even if we ultimately succeed in recovering from McDermott any amounts for which we are held liable, we may be temporarily required to bear these losses.

The terms of our separation from McDermott and the related agreements were determined in the context of a related-party transaction, and thus may not be comparable to terms that would be obtained in a transaction between unaffiliated parties.

Transactions and agreements entered into with McDermott on or before the closing of the spin-off present conflicts between our interests and those of McDermott. These transactions and agreements include the following:

- agreements related to the separation of our business from McDermott that provide for, among other things, the assumption by us of liabilities related to our business or subsidiaries, the assumption by McDermott of liabilities unrelated to our business, our respective rights, responsibilities and obligations with respect to taxes and tax benefits and the terms of our various interim and ongoing relationships, as described under "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us"; and
- administrative support services provided by McDermott to us, as well as by us to McDermott, and other transactions with McDermott, as described under "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Transition Services Agreements."

Because the terms of our separation from McDermott and these agreements are being entered into in the context of a related-party transaction, these terms may not be comparable to terms that would be obtained in a transaction between unaffiliated parties. We may not be able to resolve potential conflicts, and even if we do, the resolutions may be less favorable than if we were dealing with an unaffiliated third party. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us."

Several members of our board and management may have conflicts of interest because of their ownership of shares of common stock of McDermott or because of their continuing service on the McDermott board of directors.

Several members of our board and management own shares of common stock of McDermott and/or options to purchase common stock of McDermott because of their current or prior relationships with McDermott. In addition, all of the current members of our board of directors were, and two of our directors will continue to serve as, members of the McDermott board of directors. This share ownership and the continuing role as a director of both companies by these two directors could create, or appear to create, potential conflicts of interest when our directors and executive officers are faced with decisions that could have different implications for our company and McDermott. See "Management."

Risks Relating to Ownership of Our Common Stock

Because there has not been any public market for our common stock, the market price and trading volume of our common stock may be volatile and you may not be able to resell your shares of our common stock at or above the initial market price of our common stock following the spin-off.

Prior to the spin-off, there will have been no trading market for our common stock. We cannot assure you that an active trading market will develop or be sustained for our common stock after the spin-off, nor can we predict the prices at which our common stock will trade after the spin-off. The market price of our common stock could fluctuate significantly due to a number of factors, many of which are beyond our control, including:

- · fluctuations in our quarterly or annual earnings results or those of other companies in our industry;
- failures of our operating results to meet the estimates of securities analysts or the expectations of our stockholders or changes by securities analysts in their estimates of our future earnings;
- · announcements by us or our customers, suppliers or competitors;
- · changes in laws or regulations which adversely affect our industry or us;
- · changes in accounting standards, policies, guidance, interpretations or principles;
- · general economic, industry and stock market conditions;
- · future sales of our common stock; and
- the other factors described in these "Risk Factors" and other parts of this information statement.

Substantial sales of our common stock could cause our stock price to decline and issuances by us may dilute your ownership in our company.

Any sales of substantial amounts of our common stock in the public market after the spin-off, or the perception that these sales might occur, could lower the market price of our common stock and impede our ability to raise capital through the issuance of equity securities. Although we have no actual knowledge of any plan or intention on the part of any 5% or greater stockholder to sell our common stock following the spin-off, it is possible that some McDermott stockholders, including possibly some of McDermott's largest stockholders, will sell the shares of our common stock they receive in the spin-off for various reasons. For example, such investors may not believe that our business profile or level of market capitalization as an independent company fits their investment objectives. Further, if we issue additional equity securities to raise additional capital, your ownership interest in our company will be diluted and the value of your investment may be reduced.

We have no plans to pay regular dividends on our common stock, so you may not receive funds without selling your shares of our common stock.

We have no current intent to pay a regular dividend. The credit agreement relating to our revolving credit facility includes restrictions on our ability to pay dividends. Our board of directors will determine the payment of future dividends on our common stock, if any, and the amount of any dividends in light of applicable law, contractual restrictions limiting our ability to pay dividends, our earnings and cash flows, our capital requirements,

our financial condition, and other factors our board of directors deems relevant. Accordingly, you may have to sell some or all of your shares of our common stock in order to generate cash flow from your investment.

Provisions in our corporate documents and Delaware law could delay or prevent a change in control of our company, even if that change may be considered beneficial by some stockholders.

The existence of some provisions of our certificate of incorporation and bylaws and Delaware law could discourage, delay or prevent a change in control of our company that a stockholder may consider favorable. These include provisions:

- providing that our board of directors fixes the number of members of the board;
- providing for the division of our board of directors into three classes with staggered terms;
- · limiting who may call special meetings of stockholders;
- · prohibiting stockholder action by written consent, thereby requiring stockholder action to be taken at a meeting of the stockholders;
- establishing advance notice requirements for nominations of candidates for election to our board of directors or for proposing matters that can be acted on by stockholders at stockholder meetings;
- establishing supermajority vote requirements for certain amendments to our certificate of incorporation and bylaws;
- · limiting the right of stockholders to remove directors;
- authorizing a large number of shares of common stock that are not yet issued, which would allow our board of directors to issue shares to persons friendly
 to current management, thereby protecting the continuity of our management, or which could be used to dilute the stock ownership of persons seeking to
 obtain control of us; and
- authorizing the issuance of "blank check" preferred stock, which could be issued by our board of directors to increase the number of outstanding shares and thwart a takeover attempt.

In addition, following the spin-off, we will be subject to Section 203 of the Delaware General Corporation Law, which may have an anti-takeover effect with respect to transactions not approved in advance by our board of directors, including discouraging takeover attempts that might result in a premium over the market price for shares of our common stock.

We believe these provisions protect our stockholders from coercive or otherwise unfair takeover tactics by requiring potential acquirors to negotiate with our board of directors and by providing our board of directors with more time to assess any acquisition proposal, and are not intended to make our company immune from takeovers. However, these provisions apply even if the offer may be considered beneficial by some stockholders and could delay or prevent an acquisition that our board of directors determines is not in the best interests of our company and our stockholders. See "Description of Capital Stock—Anti-Takeover Effects of Provisions of our Certificate of Incorporation and Bylaws."

We may issue preferred stock that could dilute the voting power or reduce the value of our common stock.

Our certificate of incorporation authorizes us to issue, without the approval of our stockholders, one or more classes or series of preferred stock having such designation, powers, preferences and relative, participating, optional and other special rights, including preferences over our common stock respecting dividends and distributions, as our board of directors generally may determine. The terms of one or more classes or series of preferred stock could dilute the voting power or reduce the value of our common stock. For example, we could grant holders of preferred stock the right to elect some number of our directors in all events or on the happening of specified events or the right to veto specified transactions. Similarly, the repurchase or redemption rights or liquidation preferences we could assign to holders of preferred stock could affect the residual value of the common stock. See "Description of Capital Stock—Preferred Stock."

CAUTIONARY STATEMENT CONCERNING FORWARD-LOOKING INFORMATION

This information statement includes forward-looking statements. You can identify our forward-looking statements by words such as "anticipate," "estimate," "expect," "forecast," "goal," "intend," "plan," "predict," "project," "seek," "target," "could," "may," "should" or "would" or other similar expressions that convey the uncertainty of future events or outcomes. When considering these forward-looking statements, you should keep in mind the risk factors and other cautionary statements contained in this information statement.

Forward-looking statements include, but are not limited to, statements that relate to, or statements that are subject to risks, contingencies or uncertainties that relate to:

- · the expected benefits of the spin-off;
- · our business strategies;
- · future levels of revenues, operating margins, income from operations, net income or earnings per share;
- anticipated levels of demand for our products and services;
- · future levels of capital, environmental or maintenance expenditures;
- the success or timing of completion of ongoing or anticipated capital or maintenance projects;
- expectations regarding the acquisition or divestiture of assets and businesses;
- · our ability to obtain surety bonding capacity;
- · our ability to obtain appropriate insurance and indemnities;
- · timing for bringing the operations of Nuclear Fuel Services, Inc. back online;
- · the potential effects of judicial or other proceedings on our business, financial condition, results of operations and cash flows; and
- · the anticipated effects of actions of third parties such as competitors, or federal, foreign, state or local regulatory authorities, or plaintiffs in litigation.

We have based our forward-looking statements on our current expectations, estimates and projections about our industries and our company. We caution that these statements are not guarantees of future performance and you should not rely unduly on them, as they involve risks, uncertainties and assumptions that we cannot predict. In addition, we have based many of these forward-looking statements on assumptions about future events that may prove to be inaccurate. While our management considers these assumptions to be reasonable, they are inherently subject to significant business, economic, competitive, regulatory and other risks, contingencies and uncertainties, most of which are difficult to predict and many of which are beyond our control. Accordingly, our actual results may differ materially from the future performance that we have expressed or forecast in our forward-looking statements. Differences between actual results and any future performance suggested in our forward-looking statements could result from a variety of factors, including the following:

- · general economic and business conditions and industry trends;
- · general developments in the industries in which we are involved;
- · decisions on spending by the U.S. Government and electric power generating companies;
- the highly competitive nature of our businesses;
- · cancellations of and adjustments to backlog and the resulting impact from using backlog as an indicator of future earnings;
- · the ability of our suppliers to deliver raw materials in sufficient quantities and in a timely manner;
- · volatility and uncertainty of the credit markets;

- · our ability to comply with covenants in our credit agreement and other debt instruments and availability, terms and deployment of capital;
- the continued availability of qualified personnel;
- · the operating risks normally incident to our lines of business, including the potential impact of liquidated damages;
- · changes in, or our failure or inability to comply with, government regulations;
- · adverse outcomes from legal and regulatory proceedings;
- · the impact of potential regional, national and/or global requirements to significantly limit or reduce greenhouse gas and other emissions in the future;
- · changes in, and liabilities relating to, existing or future environmental regulatory matters;
- · rapid technological changes;
- · the realization of deferred tax assets, including through a reorganization McDermott completed in December 2006;
- the consequences of significant changes in interest rates and currency exchange rates;
- our ability to realize expected benefits from the spin-off;
- a determination by the IRS that the spin-off or certain related transactions should be treated as a taxable transaction;
- · our different capital structure as an independent company, including our access to capital, credit ratings, debt and ability to raise additional financing;
- · difficulties we may encounter in obtaining regulatory or other necessary approvals of any strategic transactions;
- · the risks associated with integrating acquired businesses;
- · social, political and economic situations in foreign countries where we do business;
- the possibilities of war, other armed conflicts or terrorist attacks;
- · the effects of asserted and unasserted claims;
- · our ability to obtain surety bonds, letters of credit and financing;
- our ability to maintain builder's risk, liability, property and other insurance in amounts and on terms we consider adequate and at rates that we consider economical;
- · the aggregated risks retained in our captive insurance subsidiary; and
- the impact of the loss of insurance rights as part of the final settlement in 2006 of the Chapter 11 proceedings involving several of our subsidiaries.

We believe the items we have outlined above are important factors that could cause estimates in our financial statements to differ materially from actual results and those expressed in the forward-looking statements made in this information statement. We have discussed many of these factors in more detail elsewhere in this information statement. These factors are not necessarily all the factors that could affect us. Unpredictable or unanticipated factors we have not discussed in this information statement could also have material adverse effects on actual results of matters that are the subject of our forward-looking statements. Neither we nor McDermott undertake any obligation to update the forward-looking statements included in this information statement to reflect events or circumstances after the date of this information statement, unless we are required by applicable securities laws to do so. We advise our security holders that they should (1) be aware that factors not referred to above could affect the accuracy of our forward-looking statements and (2) use caution and common sense when considering our forward-looking statements.

THE SPIN-OFF

General

The board of directors of McDermott regularly reviews the various operations conducted by McDermott to ensure that resources are deployed and activities are pursued in the best interest of its stockholders. On December 7, 2009, McDermott announced that its board of directors had approved in principle a plan involving the pro rata distribution of all of our shares of common stock to McDermott's stockholders in a tax-free spin-off. The spin-off is subject to final approval by the McDermott board of directors, which approval is subject to, among other things, the conditions described below under "—Spin-Off Conditions and Termination."

- receive one share of our common stock for every [] shares of McDermott common stock owned by such holder; and
- retain such holder's shares of McDermott common stock.

McDermott stockholders will not be required to pay for shares of our common stock received in the spin-off or to surrender or exchange shares of McDermott common stock in order to receive our common stock or to take any other action in connection with the spin-off. No vote of McDermott stockholders will be required or sought in connection with the spin-off, and McDermott stockholders will have no appraisal rights in connection with the spin-off

Reasons for the Spin-Off

McDermott's board and management believe that our separation from McDermott will provide the following benefits:

- improved positioning for each company to accelerate growth based on its distinct corporate strategy, market opportunities, free cash flow and customer relationships;
- more efficient allocation of capital, which will allow each company to develop an independent investment program without the constraints of a holding company, conglomerate structure;
- · establishment of distinct publicly traded stock for each company, which may be used as "currencies" to facilitate future acquisitions;
- elimination of the risk to the combined businesses posed by recent modifications to the rules under the FAR that limit the U.S. Government's ability to contract with "inverted" companies and their subsidiaries; and
- · sharpened management focus and strategic vision and closer alignment of management incentives with stockholder value creation.

As discussed below, the recent modifications to the rules under the FAR had a significant impact on the McDermott board's decision to effect the spin-off at this time.

Improved positioning for each company to accelerate growth

McDermott's board of directors and senior management believe that McDermott and B&W will be better positioned to accelerate their growth as two separate, publicly traded companies as compared to the current

combined company structure. McDermott's board of directors believes that opportunities for growth will result from the improved ability of each company to focus on and implement its distinct corporate strategy, take advantage of available market opportunities on a more timely basis, reinvest its cash flows within the business as its board of directors and management deem appropriate, and develop stronger and broader relationships with its customers.

As part of a combined company, each company's growth initiatives have effectively competed with the other company's investment opportunities in terms of available funding, resources, prioritization and time permitted for review and approval. The businesses of McDermott and B&W have typically operated autonomously. However, due to the different cyclicality characteristics of the McDermott and B&W businesses, each company has not had the ability to reinvest its cash flows in internal growth as quickly as may have been desirable for its business on a standalone basis. For the same reason, the capital structure of the combined company has been relatively conservative. While this capital structure was appropriate for the combined company, it had a negative impact on the ability of each company to feach company to said the strategy of the combined company was to maintain a balance between its businesses and not become overly concentrated in either one of these businesses. However, this strategy reduced the ability of each company to pursue its own growth initiatives.

More efficient allocation of capital

As discussed above, as part of the combined company, McDermott and B&W effectively competed for capital resources. In addition, the combined company maintained an appropriate level of liquidity for that type of structure. As a result, the flexibility of each company to invest capital in its business in a time and manner as its separate strategy would dictate was limited. McDermott's board of directors and senior management believe that the separation of the offshore oil and gas construction business from the power generation systems and government operations businesses will enable a more efficient allocation of capital by providing each separate company's board of directors and management the ability to reinvest that company's free cash flow, utilize cash and investments and access the capital markets, if needed, in a manner responsive to the needs of that specific company, rather than the needs of the combined company. We expect that this more efficient allocation of capital will provide the separated companies with greater flexibility to pursue their strategic initiatives within their capital-intensive industries.

Establishment of distinct publicly traded stock "currencies" to facilitate future acquisitions

As separate offshore oil and gas construction and power generation systems and government operations companies, we believe each company's stock can also serve as a more attractive acquisition "currency" to potential targets than the existing McDermott common stock. In the case of an acquisition of a business in which part of the consideration is to be paid in stock, we believe that the existing investors in the acquisition target company would generally prefer to receive stock of another company in the same sector as the target, rather than stock of an integrated company that has exposure to other sectors that they may not prefer or as to which they may not have sufficient familiarity to remain as long-term investors.

Elimination of the risk to the combined businesses posed by recent modifications to the rules under the FAR

On July 1, 2009, the U.S. Civilian Agency Acquisition Council and the U.S. Defense Acquisition Regulations Council (the "FAR Councils"), acting under the FAR, adopted interim rules, with an immediate effective date, that established contracting procedures to implement a mandate in the Omnibus Appropriations Act, 2009, that generally prohibits U.S. federal agencies from awarding new contracts to inverted companies and their subsidiaries with U.S. federal appropriated funds for fiscal year 2009 and certain prior fiscal years. The current version of the proposed U.S. federal appropriations legislation for 2010 contains restrictions on contracting with inverted companies similar to those in the 2009 Omnibus Appropriations Act.

As a result of a 1982 reorganization transaction, pursuant to which McDermott International, Inc., a corporation organized under the laws of the Republic of Panama, became the parent entity within the McDermott group of companies, McDermott International, Inc. would likely be considered an "inverted" company under the interim rules under the FAR, and, if so, B&W would be considered a subsidiary of an "inverted" company. The interim rules are not clear as to their impact on joint ventures, including those in which we are the majority or controlling owner. As of the date of this information statement, it is uncertain as to when the FAR Councils will adopt the final version of the interim rules and whether and the extent to which the final rules may differ from the interim rules. Although we believe the interim rules do not impact any of our existing contracts and should not adversely impact our ability to enter into subcontracts relating to government projects, the interim rules will impact our ability to pursue new contract awards directly with the U.S. Government, unless we are able to obtain waivers, which are available only on a very limited basis, or change our status as a subsidiary of an "inverted" company.

In approving the spin-off, McDermott's board of directors considered the fact that we derive a substantial amount of our revenues and profits from services provided to the U.S. Government, and the fact that any significant restriction on our ability to pursue future U.S. Government contract work consistent with our past operations would have a material adverse effect on the financial condition, results of operations and cash flows of the combined companies.

McDermott's board of directors also considered the fact that various tax legislative proposals intending to eliminate some perceived tax advantages of companies that have legal domiciles outside the United States but operate in the United States through one or more subsidiaries have repeatedly been introduced in the U.S. Congress. McDermott's board of directors and management considered that, if legislation were to be enacted in this area, the combined companies could be subject to a substantial increase in corporate income taxes, which could have a material adverse effect on the financial condition, results of operations and cash flows of the combined companies.

In light of these considerations, McDermott's board of directors, with input from its independent financial and legal advisors, engaged in a fresh, comprehensive review of the combined company and its businesses, focusing on the best way for us to continue serving the U.S. Government's defense and nuclear operations objectives, while enhancing long-term value for McDermott's stockholders. As a result of the spin-off, B&W will be an independent company still incorporated in Delaware, but without a foreign-organized parent entity. As such, we believe B&W will be able to compete for new contracts with the U.S. Government without restriction from the interim rules under the FAR and without any concern of the possible impact of future tax legislation aimed at companies that have legal domiciles outside the United States but operate in the United States through one or more subsidiaries. Furthermore, the independent operation of the two businesses, with B&W as a Delaware corporation, will permit us to continue to serve the needs of our U.S. Government customers without interruption as a result of these rule modifications, while allowing McDermott the opportunity to pursue its independent business strategy. Taking these considerations into account, the modifications to the rules under the FAR had a significant impact on the McDermott board's decision to effect the spin-off at this time.

Sharpened management focus and strategic vision and closer alignment of management incentives with stockholder value creation

McDermott's board of directors and management also took into account the fact that each company has different business strategies and offers significantly different business opportunities for growth. They determined that the spin-off should allow the management team of each company to improve its focus on its strategic priorities and make business and operational decisions that are in the best interest of its operations, taking into consideration the different challenges and opportunities and different financial profiles and capital needs pertinent to its business. As separate companies, each will be able independently to prioritize allocation of resources and capital in support of its business strategies. As a combined company, our projects have effectively competed with other investment opportunities within McDermott. As separate companies, each of McDermott and our company will no longer have to compete for investment capital with the other, and each would be in a position to pursue a growth strategy to optimize its own operations. By eliminating the necessary time and

resources required to resolve conflicting business priorities and strategic needs, the two businesses will be better able to compete through quicker decision making, more efficient deployment of capital and corporate resources and enhanced responsiveness to market demands.

In addition, McDermott's board of directors and management concluded that the separation of the two businesses through the spin-off will allow each company to provide incentive compensation to its key employees in the form of equity-based incentive compensation that is more closely aligned with the performance of each business. By separating the two companies, management of each company should be in an improved position to attract employees with the correct skill set, to motivate them appropriately and to retain them for the long term.

McDermott's board of directors also considered a number of potentially negative factors in evaluating the spin-off, including:

- the potential loss of synergies from operating as one company and potential increased costs;
- · potential loss of joint purchasing power;
- potential disruptions to the businesses as a result of the spin-off;
- the limitations placed on our company as a result of the tax sharing agreement and other agreements we have entered into with McDermott in connection with the spin-off;
- · risks of being unable to achieve the benefits expected to be achieved by the spin-off;
- the risk that the spin-off might not be completed;
- · the costs of the spin-off; and
- the risk that the trading price of a share of McDermott common stock after the spin-off plus the trading price of the []/[] of a share of B&W common stock distributed for each share of McDermott common stock will, in the aggregate, be less than the trading price of a share of McDermott common stock before the spin-off.

Except in connection with its overall determination with respect to our company as a viable, stand-alone public company following the spin-off, McDermott's board of directors did not consider existing or reasonably likely future litigation against B&W for personal injury, property damage or wrongful death claims as a significant factor in its deliberations relating to the spin-off.

The McDermott board of directors concluded that the potential long-term benefits of the spin-off outweighed these factors.

In view of the wide variety of factors considered in connection with the evaluation of the spin-off and the complexity of these matters, McDermott's board of directors did not find it useful to, and did not attempt to, quantify, rank or otherwise assign relative weights to the factors considered. The individual members of the McDermott board of directors likely may have given different weights to different factors.

Results of the Spin-Off

We and McDermott are parties to a number of agreements that govern the spin-off and our future relationship. For a more detailed description of these agreements, please see "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us."

You will not be required to make any payment for the shares of B&W common stock you receive, nor will you be required to surrender or exchange your shares of McDermott common stock or take any other action in order to receive the shares of B&W common stock to which you are entitled. The spin-off will not affect the number of outstanding shares of McDermott common stock or any rights of McDermott stockholders, although it will affect the market value of the outstanding McDermott common stock.

Manner of Effecting the Spin-Off

When and How You Will Receive B&W Shares

On the distribution date, McDermott will release its shares of B&W common stock for distribution by Computershare Trust Company, N.A., the distribution agent. The distribution agent will cause the shares of B&W common stock to which you are entitled to be registered in your name or in the "street name" of your bank or brokerage firm.

"Street Name" Holders. Many McDermott stockholders hold McDermott common stock through an account with a bank or brokerage firm. If this applies to you, that bank or brokerage firm is the registered holder that holds the shares on your behalf. For stockholders who hold their shares of McDermott common stock in an account with a bank or brokerage firm, the B&W common stock distributed to you will be registered in the "street name" of your bank or broker, who in turn will electronically credit your account with the B&W shares that you are entitled to receive in the distribution. We anticipate that banks and brokers will generally credit their customers' accounts with B&W common stock on or shortly after the distribution date. We encourage you to contact your bank or broker if you have any questions regarding the mechanics of having shares of B&W common stock credited to your account.

Registered Holders. If you are the registered holder of shares of McDermott common stock and hold your shares of McDermott common stock either in physical form or in book-entry form, the shares of B&W common stock distributed to you will be registered in your name and you will become the holder of record of that number of shares of B&W common stock. Our distribution agent will send you a statement reflecting your ownership of our common stock.

Direct Registration System. As part of the spin-off, we will be adopting a direct registration system for book-entry share registration and transfer of our common stock. The shares of our common stock to be distributed in the spin-off will be distributed as uncertificated shares registered in book-entry form through the direct registration system. No certificates representing your shares will be mailed to you in connection with the spin-off. Under the direct registration system, instead of receiving stock certificates, you will receive a statement reflecting your ownership interest in our shares. Contact information for our transfer agent and registrar is provided under "Questions and Answers About the Spin-Off." The distribution agent will begin mailing book-entry account statements reflecting your ownership of shares promptly after the distribution date. You can obtain more information regarding the direct registration system by contacting our transfer agent and registrar.

Treatment of Fractional Shares

The transfer agent will not deliver any fractional shares of our common stock in connection with the spin-off. Instead, the transfer agent will aggregate all fractional shares and sell them on behalf of those holders who otherwise would be entitled to receive a fractional share. We anticipate that these sales will occur as soon as practicable after the distribution date. Those holders will then receive a cash payment in the form of a check in an amount equal to their pro rata share of the total net proceeds of those sales. If you physically hold McDermott stock certificates, your check for any cash that you may be entitled to receive instead of fractional shares of our common stock will be mailed to you separately. We expect that checks will generally be distributed to stockholders within one to two weeks after the distribution date. Broker selling expenses in connection with these sales will be paid by McDermott.

It is expected that all fractional shares held in street name will be aggregated and sold by brokers or other nominees according to their standard procedures. You should contact your broker or other nominee for additional details.

None of McDermott, our company or the transfer agent will guarantee any minimum sale price for the fractional shares of our common stock. Neither we nor McDermott will pay any interest on the proceeds from the sale of fractional shares. The receipt of cash in lieu of fractional shares will generally be taxable to the recipient stockholders. See "—Material U.S. Federal Income Tax Consequences of the Spin-Off."

Market for Our Common Stock

There is currently no public market for our common stock. A condition to the spin-off is the listing on the NYSE of our common stock. Our common stock has been approved for listing on the NYSE under the symbol "BWC." We anticipate that trading of our common stock will commence on a "when-issued" basis on or shortly before the record date. When-issued trading refers to a sale or purchase made conditionally because the security has been authorized but not yet issued. On the first trading day following the distribution date, when-issued trading with respect to our common stock will end and "regular way" trading will begin. Regular way trading refers to trading after a security has been issued and typically involves a transaction that settles on the third full business day following the date of the transaction. Neither we nor McDermott can assure you as to the trading price of our common stock after the spin-off or as to whether the trading price of a share of McDermott common stock after the spin-off or as to whether the trading price of will not, in the aggregate, be less than the trading price of a share of McDermott common stock before the spin-off. The trading price of our common stock is likely to fluctuate significantly, particularly until an orderly market develops. See "Risk Factors—Risks Relating to Ownership of Our Common Stock." In addition, we cannot predict any change that may occur in the trading price of McDermott's common stock as a result of the spin-off.

Trading of McDermott Common Stock After the Record Date and Prior to the Distribution

Beginning on or shortly before the record date and through the distribution date, there will be two concurrent markets in which to trade McDermott common stock: a "regular way" market and an

"ex-distribution" market. Shares of McDermott common stock that trade in the "regular way" market will trade with an entitlement to shares of our common stock distributed in connection with the spin-off. Shares that trade in the "ex-distribution" market will trade without an entitlement to shares of our common stock distributed in connection with the spin-off. Therefore, if you owned shares of McDermott common stock at 5:00 p.m., New York City time, on the record date and sell those shares in the "regular way" market on or prior to the distribution date, you also will be selling your right to receive the shares of our common stock that would have been distributed to you in connection with the spin-off. If you sell those shares of McDermott common stock in the "ex-distribution" market prior to or on the distribution date, you will still receive the shares of our common stock that were to be distributed to you in connection with the spin-off as a result of your ownership of the shares of McDermott common stock.

We expect to have approximately [] million shares of our common stock outstanding immediately after the spin-off, based upon the number of shares of McDermott common stock outstanding on [], 2010. The shares of our common stock distributed to McDermott stockholders will be freely transferable, except for shares received by persons who may be deemed to be our "affiliates" under the Securities Act of 1933, as amended, or the Securities Act, and except for shares issued as restricted stock under our incentive plan. Persons who may be deemed to be our affiliates after the spin-off generally include individuals or entities that control, are controlled by, or are under common control with us, and may include some or all of our directors and executive officers. Our affiliates will be permitted to sell their shares of B&W common stock only pursuant to an effective registration statement under the Securities Act or an exemption from the registration requirements of the Securities Act, such as the exemption afforded by Rule 144.

Treatment of Stock-Based Awards

In connection with the spin-off, we currently expect that, subject to the approval of the compensation committee of McDermott's board of directors, McDermott's outstanding equity-based compensation awards will generally be treated as follows:

- Each outstanding option to purchase shares of McDermott common stock that is held by a director, officer or employee of McDermott who will remain a director, officer or employee of McDermott and will not be a director, officer or employee of B&W immediately after the spin-off will be replaced with an adjusted option to purchase McDermott common stock. Each of those adjusted options will reflect adjustments that will be generally intended to preserve the intrinsic value of the original option and the ratio of the exercise price to the fair market value of the stock subject to the option by adjusting the number of shares purchasable and the exercise price, by reference to the "regular way" closing price for McDermott common stock as of the distribution date and the volume-weighted-average trading price of the McDermott common stock in the first five trading days following the distribution date. The replacement options will generally be made subject to the same terms and conditions as the options being replaced. To the extent the options being replaced are vested, the replacement options will also be vested.
- Each outstanding option to purchase shares of McDermott common stock that is held by a person who is or will be a director of B&W (but not of McDermott) or by a person who is or will be an officer or employee of B&W immediately after the spin-off will be replaced with substitute options to purchase shares of B&W common stock. Each of those substitute options will have terms that will be generally intended to preserve the intrinsic value of the original option and the ratio of the exercise price to the fair market value of the stock subject to the option by adjusting the number of shares purchasable and the exercise price, by reference to the opening prices of a share of McDermott common stock and a share of B&W common stock on the first trading day following the distribution date. The substitute options will generally be made subject to the same terms and conditions as the options being replaced. To the extent the options being replaced are vested, the substitute options will also be vested.
- Each outstanding option to purchase shares of McDermott common stock that is held by a (1) person who is a director of McDermott and who will remain on the board of directors of McDermott and will

- The McDermott restricted stock awards, restricted stock unit awards and deferred stock unit awards of officers or employees of McDermott who will remain officers or employees of McDermott and will not become officers or employees of B&W immediately after the spin-off will be replaced with adjusted McDermott awards, each of which will generally preserve the value of the original award. The adjusted awards will generally be made subject to the same terms and conditions as the awards being replaced. As of the distribution date, none of McDermott's directors will hold any restricted stock awards, restricted stock unit awards, deferred stock unit awards or performance share awards.
- The McDermott restricted stock awards, restricted stock unit awards and deferred stock unit awards of persons who are or will be officers or employees of B&W immediately after the spin-off will be converted into substitute B&W awards, each of which will generally preserve the value of the original award. The adjusted awards will generally be made subject to the same terms and conditions as the awards being replaced.
- The McDermott performance share awards of officers or employees of McDermott who will remain officers or employees of McDermott and will not become officers or employees of B&W immediately after the spin-off will be converted into unvested rights to receive the value of deemed target performance in unrestricted shares of McDermott common stock, with the same vesting terms as the original awards.
- The McDermott performance share awards of persons who are or will be officers or employees of B&W immediately after the spin-off will be converted
 into unvested rights to receive the value of deemed target performance in unrestricted shares of B&W common stock, with the same vesting terms as the
 original awards.
- The McDermott performance share awards of the Former B&W Officers will be converted into rights to receive the value of deemed target performance in unrestricted shares of McDermott common stock and B&W common stock, in accordance with the spin-off distribution.

Any former employees of McDermott or B&W who hold outstanding stock options, shares of restricted stock, restricted stock units, deferred stock units or performance shares which are unvested will have their stock options, shares of restricted stock, restricted stock units, deferred stock units or performance shares similarly adjusted or replaced as a result of the spin-off.

In the case of adjusting McDermott options or granting substitute B&W options, the conversion formula may result in fractional shares or fractional cents. Any fractional shares subject to adjusted McDermott options

and substitute B&W options will be disregarded, and the number of shares subject to such options will be rounded down to the next lower whole number of shares. The exercise price for such options will be rounded up to the next higher whole cent. There are no stock appreciation rights currently outstanding under the McDermott incentive plans.

Spin-Off Conditions and Termination

We expect that the spin-off will be effective on [], 2010, provided that, among other things:

- the SEC has declared effective our registration statement on Form 10, of which this information statement is a part, under the Exchange Act, with no stop order in effect with respect to the Form 10, and this information statement shall have been mailed to McDermott's stockholders;
- the actions and filings necessary under securities and blue sky laws of the states of the United States and any comparable laws under any foreign jurisdictions have been taken and become effective;
- no order, injunction, decree or regulation issued by any court or agency of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the spin-off is in effect;
- · our common stock has been approved for listing on the New York Stock Exchange, subject to official notice of issuance;
- the private letter ruling McDermott obtained from the IRS with respect to the tax treatment of the spin-off has not been revoked or modified by the IRS in any material respect, and McDermott has received an opinion from its tax counsel regarding the tax treatment of the spin-off as of the distribution date (see "—Material U.S. Federal Income Tax Consequences of the Spin-Off" for more information regarding the private letter ruling and opinion of tax counsel);
- each of the ancillary agreements related to the spin-off shall have been entered into before the spin-off and shall not have been materially breached by any party thereto;
- all material governmental approvals and material consents necessary to consummate the spin-off have been received and continue to be in full force and
 effect; and
- no other events or developments have occurred that, in the judgment of the board of directors of McDermott, in its sole and absolute discretion, would result in the spin-off having a material adverse effect on McDermott or its stockholders.

McDermott may waive one or more of these conditions in its sole and absolute discretion, and the determination by McDermott regarding the satisfaction of these conditions will be conclusive. The fulfillment of these conditions will not create any obligation on McDermott's part to effect the distribution, and McDermott has reserved the right to amend, modify or abandon any and all terms of the distribution and the related transactions at any time prior to the distribution date.

Material U.S. Federal Income Tax Consequences of the Spin-Off

The following is a summary of material U.S. federal income tax consequences relating to the spin-off. This summary is based on the Code, related U.S. Treasury regulations, and interpretations of the Code and the U.S. Treasury regulations by the courts and the IRS, in effect as of the date of this information statement, and all of which are subject to change, possibly with retroactive effect. This summary does not discuss all the tax considerations that may be relevant to McDermott stockholders in light of their particular circumstances, nor does it address the consequences to McDermott stockholders subject to special treatment under the U.S. federal income tax laws (such as non-U.S. persons, insurance companies, dealers or brokers in securities or currencies, tax-exempt organizations, financial institutions, mutual funds, pass-through entities and investors in such entities, holders who hold their shares as a hedge or as part of a hedging, straddle, conversion, synthetic security, integrated investment or other risk-reduction transaction or who are subject to alternative minimum tax or

holders who acquired their shares upon the exercise of employee stock options or otherwise as compensation). In addition, this summary does not address the U.S. federal income tax consequences to those McDermott stockholders who do not hold their McDermott common stock as a capital asset. Finally, this summary does not address any state, local or foreign tax consequences.

MCDERMOTT STOCKHOLDERS ARE URGED TO CONSULT THEIR OWN TAX ADVISORS CONCERNING THE U.S. FEDERAL, STATE AND LOCAL AND FOREIGN TAX CONSEQUENCES OF THE SPIN-OFF TO THEM.

The spin-off is conditioned on the continued effectiveness of a private letter ruling obtained from the IRS and McDermott's receipt of an opinion from Baker Botts L.L.P. (which opinion will rely, in part, on the continued effectiveness of the private letter ruling), in each case substantially to the effect that (1) the spin-off will qualify under Section 355 of the Code and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code, and (2) the spin-off and certain related transactions will further qualify for tax-free treatment to McDermott and to us.

Assuming the spin-off and such related transactions meet the conditions described above:

- the spin-off and certain related transactions will not result in any U.S. taxable income, gain or loss to McDermott or to us, other than with respect to any intercompany items or excess loss accounts required to be taken into account under Treasury regulations relating to consolidated returns;
- no gain or loss will be recognized by (and no amount will be included in the taxable income of) McDermott stockholders on their receipt of shares of our common stock in the spin-off;
- the holding period of shares of our common stock received by each McDermott stockholder will include the holding period at the time of the spin-off for the McDermott common stock on which the spin-off is made;
- the tax basis of the McDermott common stock held by each McDermott stockholder immediately before the spin-off will be allocated between that McDermott common stock and our common stock received, including any fractional share of our stock deemed received in the spin-off, in proportion to the relative fair market value of each on the date of the spin-off; and
- a McDermott stockholder who receives cash for a fractional share of our common stock will generally recognize capital gain or loss measured by the
 difference between the amount of cash received and the basis of the fractional share interest in our common stock to which the stockholder would otherwise
 be entitled

U.S. Treasury Regulations also generally provide that if a McDermott stockholder holds different blocks of McDermott common stock (generally shares of McDermott common stock purchased or acquired on different dates or at different prices), the aggregate basis for each block of McDermott common stock purchased or acquired on the same date and at the same price will be allocated, to the greatest extent possible, between the shares of our common stock received in the spin-off in respect of such block of McDermott common stock and such block of McDermott common stock, in proportion to their respective fair market values, and the holding period of the shares of our common stock received in the spin-off in respect of such block of McDermott common stock will include the holding period of such block of McDermott common stock. If a McDermott stockholder is not able to identify which particular shares of our common stock are received in the spin-off with respect to a particular block of McDermott common stock, for purposes of applying the rules described above, the stockholder may designate which shares of our common stock are received in the spin-off in respect of a particular block of McDermott common stock, provided that such designation is consistent with the terms of the spin-off. Holders of McDermott common stock are urged to consult their own tax advisors regarding the application of these rules to their particular circumstances.

Although a private letter ruling from the IRS generally is binding on the IRS, if the factual representations or assumptions made in the private letter ruling request are inaccurate or incomplete in any material respect, we

will not be able to rely on the private letter ruling. Furthermore, the IRS will not rule on whether a distribution satisfies certain requirements necessary to qualify under Section 355 of the Code or to obtain tax-free treatment to McDermott and its stockholders under Section 355 of the Code. Rather, the private letter ruling is based on representations by McDermott that these requirements have been satisfied, and any inaccuracy in such representations could invalidate the private letter ruling. In addition to the continued effectiveness of the private letter ruling from the IRS, McDermott has made it a condition to the spin-off that McDermott obtain an opinion of Baker Botts L.L.P. substantially to the effect that (1) the spin-off will qualify under Section 355 of the Code and certain transactions related to the spin-off will qualify under Sections 355 and/or 368 of the Code, and (2) the spin-off and certain related transactions will further qualify for tax-free treatment to McDermott and to us. The opinion will address those matters upon which the IRS will not rule and will rely on the private letter ruling as to matters covered by the private letter ruling. In addition, the opinion will be based on, among other things, certain assumptions and representations made by McDermott and us, which if incorrect or inaccurate in any material respect would jeopardize the conclusions reached by such counsel in its opinion. The opinion will not be binding on the IRS or the courts and will be subject to other qualifications and limitations.

Notwithstanding receipt by McDermott of the private letter ruling and opinion of counsel, the IRS could assert that the spin-off and/or certain related transactions do not qualify for tax-free treatment for U.S. federal income tax purposes. If the IRS were successful in making such an assertion, we and our initial public stockholders could be subject to significant U.S. federal income tax liability. In general, (1) with respect to the spin-off, our initial public stockholders could be treated as if they had received a taxable distribution from McDermott in an amount equal to the fair market value of our common stock that was distributed to them and (2) with respect to certain related transactions, we would be treated as if we had sold part of our assets (which will be retained by McDermott) in a taxable sale for fair market value and McDermott could be treated as receiving such assets from us as a taxable dividend. For example, even if the spin-off were otherwise to qualify under Section 355 of the Code, certain related preparatory transactions may be taxable to us (but not to McDermott or our initial stockholders) under Section 355(e) of the Code, if the spin-off were later deemed to be part of a plan (or series of related transactions) pursuant to which one or more persons acquire, directly or indirectly, stock representing a 50% or greater interest in McDermott or us. For this purpose, any acquisitions of McDermott common stock or of our common stock within the period beginning two years before the spin-off and ending two years after the spin-off are presumed to be part of such a plan, although we or McDermott may be able to rebut that presumption.

In connection with the spin-off, we and a subsidiary of McDermott have entered into a tax sharing agreement pursuant to which we will agree to be responsible for certain liabilities and obligations following the spin-off. Under the terms of the tax sharing agreement, in the event that certain transactions related to the spin-off were to fail to qualify for tax-free treatment, we would generally be responsible for all of the tax imposed on us or McDermott and its subsidiaries resulting from such failure. However, if these related transactions were to fail to qualify for tax-free treatment because of actions or failures to act by McDermott or its subsidiaries, a subsidiary of McDermott would be responsible for all such tax. See "Relationship with McDermott After the Spin-Off—Agreements Between McDermott and Us—Tax Sharing Agreement."

Under U.S. Treasury regulations, each McDermott stockholder who, immediately before the distribution, owns at least 5% of the total outstanding common stock of McDermott must attach to such stockholder's U.S. federal income tax return for the year in which the spin-off occurs a statement setting forth certain information relating to the spin-off. In addition, all stockholders are required to retain permanent records relating to the amount, basis and fair market value of our stock which they receive and to make those records available to the IRS on request of the IRS.

THE FOREGOING IS A SUMMARY OF MATERIAL U.S. FEDERAL INCOME TAX CONSEQUENCES OF THE SPIN-OFF UNDER CURRENT LAW AND IS FOR GENERAL INFORMATION ONLY. THE FOREGOING DOES NOT PURPORT TO ADDRESS ALL U.S. FEDERAL INCOME TAX CONSEQUENCES OR TAX CONSEQUENCES THAT MAY ARISE UNDER THE TAX LAWS OF OTHER JURISDICTIONS OR THAT MAY APPLY TO PARTICULAR

CATEGORIES OF STOCKHOLDERS. EACH MCDERMOTT STOCKHOLDER SHOULD CONSULT ITS OWN TAX ADVISOR AS TO THE PARTICULAR TAX CONSEQUENCES OF THE SPIN-OFF TO SUCH STOCKHOLDER, INCLUDING THE APPLICATION OF U.S. FEDERAL, STATE, LOCAL AND FOREIGN TAX LAWS, AND THE EFFECT OF POSSIBLE CHANGES IN TAX LAWS THAT MAY AFFECT THE TAX CONSEQUENCES DESCRIBED ABOVE.

Reason for Furnishing this Information Statement

This information statement is being furnished solely to provide information to McDermott stockholders who will receive shares of B&W common stock in the spin-off. It is not to be construed as an inducement or encouragement to buy or sell any of our securities or any McDermott securities. We believe that the information contained in this information statement is accurate as of the date set forth on the front cover. Changes may occur after that date and neither McDermott nor we undertake any obligation to update the information, except to the extent applicable securities laws require us to do so.

CAPITALIZATION

The following table sets forth our capitalization (1) on a historical basis as of March 31, 2010, and (2) on a pro forma basis to reflect the spin-off. This table should be read in conjunction with "Selected Historical Combined Financial Information," "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Unaudited Pro Forma Combined Financial Information," and our combined financial statements and corresponding notes included elsewhere in this information statement.

| | March 3 | 31, 2010 |
|---|------------|-----------|
| | (in thou | sands) |
| | Historical | Pro Forma |
| Long-term debt (including current portion) | \$ 7,257 | \$ |
| Stockholders' equity | | |
| Preferred stock, \$0.01 par value; 75,000,000 shares authorized pro forma; no shares issued and outstanding pro forma | _ | _ |
| Common stock, \$0.01 par value; 325,000,000 shares authorized pro forma; [] shares issued and outstanding pro forma | _ | [](1) |
| Parent equity | 153,639 | |
| Total capitalization | \$160,896 | \$ |

(1) Represents the expected distribution of approximately [] shares of our common stock to holders of McDermott common stock based on the number of shares of McDermott common stock outstanding on [], 2010.

DIVIDEND POLICY

We do not currently plan to pay a regular dividend on our common stock following the spin-off. The declaration of any future cash dividends and, if declared, the amount of any such dividends, will be subject to our financial condition, earnings, capital requirements, financial covenants and other contractual restrictions and to the discretion of our board of directors. The credit agreement relating to our revolving credit facility includes restrictions on our ability to pay dividends. Our board of directors may take into account such matters as general business conditions, industry practice, our financial condition and performance, our future prospects, our cash needs and capital investment plans, income tax consequences, applicable law and such other factors as our board of directors may deem relevant. For a discussion of the covenants contained in our credit agreement, please see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources."

SELECTED HISTORICAL COMBINED FINANCIAL DATA

The following table presents our selected historical combined financial data. We derived the historical combined statement of operations information for each of the years ended December 31, 2009, 2008 and 2007, and the balance sheet information as of December 31, 2009 and 2008, from our audited combined financial statements included in this information statement. We derived the historical condensed combined statement of operations information for the three months ended March 31, 2010 and 2009, and the balance sheet information as of March 31, 2010 and 2009, from our unaudited condensed combined financial statements included in this information statement. We derived the historical combined statement of operations information for the years ended December 31, 2006 and 2005, and the balance sheet information as of December 31, 2007, 2006 and 2005, from our unaudited combined financial statements not included in this information statement. The selected historical combined financial data reflects the effects of certain assets to be transferred to, and liabilities to be assumed by, McDermott in connection with the spin-off, and does not reflect the effects of certain liabilities to be assumed by, B&W in connection with the spin-off. See "Unaudited Pro Forma Combined Financial Data."

You should read the selected historical combined financial data in conjunction with our combined financial statements and the accompanying notes and "Management's Discussion and Analysis of Financial Condition and Results of Operations" included in this information statement. The financial information may not be indicative of our future performance and does not necessarily reflect what our financial position and results of operations would have been had we operated as an independent public company during the periods presented, including changes that will occur in our operations as a result of our spin-off from McDermott.

Three Months Ended

| | | Marc | h 31, | Year Ended December 31, | | | | |
|---|--------|----------|-------------|-------------------------|---------------|---------------|--------------|--------------|
| | 2 | 2010 | 2009 | 2009 | 2008 | 2007 | 2006(1) | 2005(1) |
| | | | | | (In thousands | | | |
| Statement of Operations Information: | | | | | | | | |
| Revenues | \$ 6 | 662,388 | \$ 785,053 | \$ 2,854,632 | \$3,398,574 | \$ 3,199,944 | \$ 2,517,606 | \$ 601,042 |
| Costs and expenses | 6 | 537,661 | 700,090 | 2,640,162 | 3,005,883 | 2,907,958 | 2,348,569 | 560,223 |
| Equity in income of investees | | 14,019 | 10,345 | 55,094 | 51,792 | 45,647 | 39,809 | 37,705 |
| Operating income | | 38,746 | 95,308 | 269,564 | 444,483 | 337,633 | 208,846 | 78,524 |
| Other income (expense) | | (8,634) | (6,062) | (37,263) | (11,744) | (23,285) | 7,990 | (2,439) |
| Income before provision for income taxes | | 30,112 | 89,246 | 232,301 | 432,739 | 314,348 | 216,836 | 76,085 |
| Provision for (benefit from) income taxes | | 13,256 | 35,710 | 84,381 | 108,885 | 99,018 | 84,826 | (17,666) |
| Net income attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | \$ | 16,843 | \$ 53,484 | \$ 147,764 | \$ 323,766 | \$ 215,250 | \$ 131,976 | \$ 93,751 |
| Other Data: | | | | | | | | |
| Depreciation and amortization | \$ | 16,812 | \$ 16,263 | \$ 72,212 | \$ 45,985 | \$ 41,672 | \$ 32,484 | \$ 14,493 |
| Capital expenditures | | 18,763 | 20,029 | 93,725 | 63,014 | 60,709 | 43,203 | 27,109 |
| | | | | | | | | |
| | | As of Ma | arch 31, | | A | s of December | - , | |
| | 2 | 2010 | 2009 | 2009 | 2008 | 2007 | 2006(1) | 2005(1) |
| | | | | (In thousands) | | | | |
| Balance Sheet Information: | | | | | | | | |
| Working capital | | (51,440) | \$ (32,128) | | \$ (90,040) | \$ (192,874) | \$ (297,286) | |
| Total assets | 2,5 | 588,769 | 2,498,580 | 2,603,859 | 2,506,841 | 2,149,636 | 2,222,818 | 508,772 |
| Notes payable and current maturities of long-term debt | | 4,993 | 4,161 | 6,432 | 9,021 | 6,599 | 257,492 | 4,250 |
| Long-term debt | | 5,343 | 5,915 | 4,222 | 6,109 | 10,609 | 15,242 | 12,975 |
| Other Data: | | | | | | | | |
| Backlog | \$ 4,7 | 751,972 | \$4,918,507 | \$4,740,060 | \$ 5,358,740 | \$ 5,064,226 | \$ 3,494,477 | \$ 1,772,258 |

⁽¹⁾ Results for the year ended December 31, 2006 include approximately ten months for the principal operating subsidiaries of our Power Generation Systems segment, which were included in our results effective February 22, 2006. We did not include the results of operations of these entities in our financial statements from February 22, 2000 through February 22, 2006 due to the Chapter 11 proceedings involving several of our subsidiaries. On February 22, 2000, Babcock & Wilcox Power Generation Group, Inc. (formerly known as The Babcock & Wilcox Company) ("B&W PGG") and certain of its subsidiaries (collectively, the "Debtors") filed a voluntary bankruptcy petition in the U.S. Bankruptcy Court for the Eastern District of New Orleans to reorganize under Chapter 11 of the U.S. Bankruptcy Code. The Debtors took this action as a means to determine and comprehensively resolve their asbestos liability as a result of asbestos-containing boilers and other products the Debtors sold, installed or serviced in prior decades. On February 22, 2000, the Debtors' operations became subject to the jurisdiction of the Bankruptcy Court. Because we lost control of the Debtors while their operations were subject to the jurisdiction of the Bankruptcy Court. Consistent with the requirements of ASC 810-10-15-10 (a), (1) (ii), we deconsolidated the results of the Debtors in our financial statements on February 22, 2000 and presented our investment in the Debtors on the cost basis of accounting. Subsequent to February 22, 2000, we wrote this investment down to zero as impaired. On February 22, 2006, the Debtors emerged from bankruptcy. The results of the Debtors were reconsolidated into our financial statements effective February 22, 2006 and for all periods after such date. We accounted for this reconsolidation in a manner similar to the acquisition of the non-controlling interest in a subsidiary.

UNAUDITED PRO FORMA COMBINED FINANCIAL DATA

The following unaudited pro forma combined balance sheet as of March 31, 2010 and the unaudited pro forma combined statements of income for the year ended December 31, 2009 and the three months ended March 31, 2010 are based on a combined reporting entity comprised of the assets and liabilities used in managing and operating the Power Generation Systems and Government Operations segments of McDermott and two captive insurance companies, which will be contributed to B&W in connection with the spin-off. See "Selected Historical Combined Financial Information." This unaudited pro forma combined financial information should be read in conjunction with "Management's Discussion and Analysis of Financial Condition and Results of Operations" and the combined financial statements and the accompanying notes included in this information statement.

Our unaudited pro forma combined statements of income for the year ended December 31, 2009 and the three months ended March 31, 2010 include adjustments to give effect to:

- the deemed transfer by B&W to McDermott of certain assets that will not be held by B&W;
- · the deemed assumption by B&W of certain liabilities that were previously liabilities of McDermott or other subsidiaries of McDermott; and
- the deemed assumption by McDermott or other subsidiaries of McDermott of certain liabilities that were previously liabilities of B&W;

in each case as if such transfers occurred on January 1, 2009. The unaudited pro forma combined balance sheet as of March 31, 2010 includes adjustments to give effect to these deemed transactions as if they had occurred on March 31, 2010. The unaudited pro forma combined statements of income and balance sheet also give effect to the issuance by us to McDermott, in connection with certain transactions relating to the spin-off, of [] shares of our common stock, and the distribution of such shares to the holders of McDermott common stock.

The unaudited pro forma combined financial information set forth below is based on available information and assumptions that we believe are factually supportable. The information has been prepared on a combined basis using historical results of operations and bases of assets and liabilities of The Babcock & Wilcox Operations of McDermott International, Inc. and includes intercompany allocations of expenses. The costs to operate our business as an independent public entity may exceed the historical allocations of expenses related to areas that include, but are not limited to, litigation and other legal matters, compliance with the Sarbanes-Oxley Act and other corporate governance matters, insurance and claims management and the related cost of insurance, as well as general overall purchasing power. These possible increased costs are not included in the unaudited pro forma combined financial information, as their impact on our results of operations cannot be reasonably estimated. The unaudited pro forma combined financial information also does not reflect any significant changes that may occur after the spin-off in our financing plans and cost structure, including expenditures related to our growth initiatives.

The unaudited pro forma financial information is not necessarily indicative of our future performance or what our results of operations and financial position would have been if we had operated as an independent public company during the periods presented or if the transactions reflected in the pro forma adjustments had actually occurred as of the dates assumed.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEETS As of March 31, 2010

| | The Babcock & Wilcox Operations of McDermott International, Inc. | Pro Forma <u>Adj</u> ustments (In thousands) | The Babcock & Wilcox Operations of McDermott International, Inc. Pro Forma |
|--|---|--|--|
| ASSETS | | | |
| Current Assets: | | ********** | |
| Cash and cash equivalents | \$ 401,124 | \$(100,000) ^(a) | \$ 301,124 |
| Restricted cash and cash equivalents | 17,821 | | 17,821 |
| Investments | 14 | _ | 14 |
| Accounts receivable – trade, net | 302,506 | | 302,506 |
| Accounts receivable – other | 42,496 | _ | 42,496 |
| Contracts in progress | 299,166 | _ | 299,166 |
| Inventories | 95,009 | _ | 95,009 |
| Deferred income taxes | 99,109 | (400)(b) | 99,109 |
| Other current assets | 25,259 | (488) ^(b) | 24,771 |
| Total Current Assets | 1,282,504 | (100,488) | 1,182,016 |
| Property, Plant and Equipment | 953,715 | (24,592) | 929,123 |
| Less accumulated depreciation | 522,504 | (5,172) | 517,332 |
| Net Property, Plant and Equipment | 431,211 | (19,420)(b) | 411,791 |
| Investments | 70,909 | (1,805) ^(b) | 69,104 |
| Goodwill | 269,950 | | 269,950 |
| Deferred Income Taxes | 265,048 | (19,976) ^(b) | 245,072 |
| Investments in Unconsolidated Affiliates | 77,965 | | 77,965 |
| Note Receivable from Affiliate | 43,835 | (43,835)(c) | |
| Other Assets | 147,347 | | 147,347 |
| TOTAL | \$2,588,769 | <u>\$(185,524)</u> | \$2,403,245 |

See accompanying notes to unaudited pro forma condensed combined financial statements.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEETS As of March 31, 2010

| | The Babcock & Wilcox Operations of McDermott International, Inc. | Pro Forma <u>Adjustments</u> (In thousands) | The Babcock & Wilcox Operations of McDermott International, Inc. Pro Forma |
|--|---|---|--|
| LIABILITIES AND PARENT EQUITY | | | |
| Current Liabilities: | | | |
| Notes payable and current maturities of long-term debt | \$ 4,993 | \$ — | \$ 4,993 |
| Accounts payable | 183,360 | (7,394) ^(b) | 175,966 |
| Accounts payable to McDermott International, Inc. | 142,967 | (142,967)(c) | 212.416 |
| Accrued employee benefits Accrued liabilities—other | 299,026 | (86,610) ^(b) | 212,416 |
| Advance billings on contracts | 85,468 494,579 | (10,984) ^(b) | 74,484 494,579 |
| Advance onings on contracts Accrued warranty expense | 119,531 | _ | 119,531 |
| Income taxes payable | 4,020 | | 4,020 |
| Total Current Liabilities | 1,333,944 | (247,955) | 1,085,989 |
| Long-Term Debt | 5,343 | <u>(217,955</u>) | 5,343 |
| Accumulated Postretirement Benefit Obligation | 106,329 | | 106,329 |
| Environmental Liabilities | | | |
| | 48,523 | (20.00.04) | 48,523 |
| Pension Liability | 550,562 | $(28,806)^{(b)}$ | 521,756 |
| Notes Payable to Affiliate | 320,588 | $(320,588)^{(c)}$ | |
| Other Liabilities | 69,841 | 27,041 ^(d) | 96,882 |
| Commitments and Contingencies | | | |
| Equity | | | |
| Total Parent Equity | 153,639 | (153,639)(c),(e) | _ |
| Capital in excess of par value | <u> </u> | 538,423 (c),(e) | 538,423 |
| Parent Equity—The Babcock & Wilcox Operations of McDermott International, Inc. | | | |
| TOTAL | \$2,588,769 | \$(185,524) | \$2,403,245 |
| | | | |

See accompanying notes to unaudited pro forma condensed combined financial statements.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENTS OF INCOME For the Three Months Ended March 31, 2010

| | The Babcock & Wilcox Operations of McDermott International, Inc. | Pro Forma Adjustments (In thousands) | The Babcock & Wilcox Operations of McDermott International, Inc. Pro Forma |
|---|---|--|--|
| Revenues | \$ 662,388 | <u>\$ —</u> | \$ 662,388 |
| Costs and Expenses: | | | |
| Cost of operations | 544,951 | _ | 544,951 |
| Selling, general and administrative expenses | 92,723 | $(2,264)^{(f)}$ | 90,459 |
| (Gains) losses on asset disposals and impairments—net | (13) | | (13) |
| Total Costs and Expenses | 637,661 | (2,264) | 635,397 |
| Equity in Income of Investees | 14,019 | | 14,019 |
| Operating Income | 38,746 | 2,264 | 41,010 |
| Other Income (Expense): | | | |
| Interest income | 449 | (274) ^(g) | 175 |
| Interest expense | (5,993) | 5,774 (g) | (219) |
| Other expense—net | (3,090) | | (3,090) |
| Total Other Expense | (8,634) | 5,500 | (3,134) |
| Income before Provision for Income Taxes | 30,112 | 7,764 | 37,876 |
| Provision for Income Taxes | 13,256 | 7,072 (h) | 20,328 |
| Net Income | 16,856 | 692 | 17,548 |
| Less: Net Income attributable to Noncontrolling Interest | (13) | | (13) |
| Net Income Attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | \$ 16,843 | 692 | \$ 17,535 |
| Pro Forma Earnings per Share: | | | |
| Basic | | | \$ — |
| Diluted | | | \$ — |
| Weighted Average Shares used in Calculating Pro Forma Earnings per Share: | | | |
| Basic | | | _ |
| Diluted | | | _ |

See accompanying notes to unaudited pro forma condensed combined financial statements.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. UNAUDITED PRO FORMA COMBINED STATEMENT OF INCOME For the Year Ended December 31, 2009

| | The Babcock & Wilcox Operations of McDermott International, Inc. | Pro Forma Adjustments (In thousands) | The Babcock & Wilcox Operations of McDermott International, Inc. Pro Forma |
|---|--|--------------------------------------|--|
| Revenues | \$2,854,632 | <u>\$</u> | \$2,854,632 |
| Costs and Expenses: | | | |
| Cost of operations | 2,235,377 | _ | 2,235,377 |
| Selling, general and administrative expenses | 403,559 | (9,531) ^(f) | 394,028 |
| Losses on asset disposals and impairments—net | 1,226 | | 1,226 |
| Total Costs and Expenses | 2,640,162 | (9,531) | 2,630,631 |
| Equity in Income of Investees | 55,094 | | 55,094 |
| Operating Income | 269,564 | 9,531 | 279,095 |
| Other Income (Expense): | | | |
| Interest income | 3,439 | (1,854) ^(g) | 1,585 |
| Interest expense | (24,590) | 23,590 ^(g) | (1,000) |
| Other expense—net | (16,112) | | (16,112) |
| Total Other Expense | (37,263) | 21,736 | (15,527) |
| Income before Provision for Income Taxes | 232,301 | 31,267 | 263,568 |
| Provision for Income Taxes | 84,381 | 14,938 ^(h) | 99,319 |
| Net Income | 147,920 | 16,329 | 164,249 |
| Less: Net Income Attributable to Noncontrolling Interest | (156) | | (156) |
| Net Income Attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | <u>\$ 147,764</u> | \$ 16,329 | \$ 164,093 |
| Pro Forma Earnings per Share: | | | |
| Basic | | | \$ — |
| Diluted | | | \$ — |
| Weighted Average Shares used in Calculating Pro Forma Earnings per Share: | | | |
| Basic | | | _ |
| Diluted | | | _ |

See accompanying notes to unaudited pro forma financial statements.

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL STATEMENTS

Balance Sheet

- (a) Reflects an estimated cash distribution by The Babcock & Wilcox Operations of McDermott International, Inc. to McDermott to maintain appropriate working capital and liquidity levels. The estimated cash distribution is based on a preliminary determination by the McDermott board of directors that resulted from its review of the McDermott consolidated cash position and the respective foreseeable working capital and liquidity requirements of B&W and McDermott following the spin-off. The amount will be finally determined by the McDermott board of directors, as a fixed amount, in connection with its final approval of the spin-off.
- (b) Reflects asset and liabilities that will not be held by the Babcock & Wilcox Operations of McDermott International, Inc. after the spin-off, principally pension liabilities and property, plant and equipment that will be transferred to McDermott. The pension liabilities reflected in the pro forma adjustments are related to certain legacy employees of subsidiaries included in McDermott's offshore oil and gas construction segment. In order to match this obligation with the appropriate group, the subsidiary with these liabilities has been transferred to McDermott in connection with the spin-off.
- (c) Reflects the extinguishment or settlement of receivable and payable amounts with certain McDermott subsidiaries. All of the settlements of these receivable and payable amounts were in the form of the forgiveness of the amounts owed, with the exceptions being a cash payment totaling approximately \$43 million that a Canadian subsidiary of McDermott made to a subsidiary of ours and the transfer of a \$278 million promissory note and \$43 million in cash from a subsidiary of ours to a subsidiary of McDermott.
- (d) Reflects non-current tax liabilities of McDermott to be assumed by The Babcock & Wilcox Operations of McDermott International, Inc. in connection with the spin-off.
- (e) Reflects the expected distribution of approximately [] million shares of B&W common stock and the reclassification of McDermott's investment in our company to common stock in an amount equal to the aggregate par value of the shares of our common stock to be distributed in the spin-off, and the remainder to be capital in excess of par value.

Statements of Income

- (f) Reflects the reduction of pension expense associated with pension plan assets and liabilities that will be transferred to a McDermott subsidiary as a result of the spin-off.
- (g) Reflects the elimination of interest income and expense related to notes receivable and payable amounts with certain subsidiaries of McDermott that will be settled in connection with the spin-off.
 - (h) Reflects the income tax impacts of the adjustments described in (a) and (b) above and the balance sheet adjustments described above.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis of our financial condition and results of operations should be read in conjunction with "Selected Historical Combined Financial Information," "Unaudited Pro Forma Combined Financial Information" and the financial statements and accompanying notes appearing in this information statement. This discussion and analysis contains forward-looking statements that involve risks and uncertainties. Actual results may differ materially from those anticipated in these forward-looking statements as a result of certain factors, including those set forth under "Risk Factors" and elsewhere in this information statement. See "Cautionary Statement Concerning Forward-Looking Information." Unless the context requires otherwise or we specifically indicate otherwise, when used in this Management's Discussion and Analysis of Financial Condition and Results of Operations, the terms "we," "our," "ours" and "us" refer to The Babcock & Wilcox Operations of McDermott International, Inc. The historical financial information discussed below does reflect the effects of certain assets to be transferred to, and liabilities to be assumed by, McDermott in connection with the spin-off, and does not reflect the effects of certain liabilities to be assumed by B&W in connection with the spin-off. See "Unaudited Pro Forma Combined Financial Information."

General

In general, we operate in capital-intensive industries and rely on large contracts for a substantial amount of our revenues. We are currently exploring growth strategies across our segments through acquisitions to expand and complement our existing businesses. As we pursue these opportunities, we expect they would be funded by cash on hand, external financing (including debt), equity or some combination thereof.

Overview—Power Generation Systems

We expect the backlog of our Power Generation Systems segment of approximately \$2.0 billion at March 31, 2010 to produce revenues of approximately \$790 million in 2010, not including any change orders or new contracts that may be awarded during the year. Through this segment, we are actively bidding on and, in some cases, beginning preliminary work on projects that we expect will be awarded to us in 2010 subject to successful contract negotiations. These projects are not currently reflected in backlog. Our liquidity position for this segment remains strong, and we expect it to remain so throughout 2010.

Our Power Generation Systems segment's overall activity depends mainly on the capital expenditures of electric power generating companies and other steamusing industries. This segment's products and services are capital intensive. As such, customer demand is heavily affected by the variations in customers' business cycles and by the overall economies of the countries in which they operate.

The recent worldwide credit and economic environment, as well as uncertainty regarding environmental regulations, has adversely affected the utility industry. We have experienced and expect to continue to experience delays or deferrals of proposed projects. For example, for the year ended December 31, 2009, our Power Generation Systems segment experienced a decline in revenues compared to 2008 and an overall slowdown in project awards. In light of current macroeconomic conditions, revenues from our Power Generation Systems segment may continue to decline in 2010.

According to the International Energy Agency, consumption of electricity worldwide is expected to nearly double in the next 25 years. While we cannot predict what impact potential future legislation and regulations concerning CO_2 and other emissions will have on our results of operations, it is possible such legislation could favorably impact the environmental retrofit and service businesses of our Power Generation Systems segment.

On June 26, 2009 the U.S. House of Representatives passed the American Clean Energy and Security Act of 2009, H.R.2454, 111th Cong. 1st Sess. (commonly referred to as the "Waxman-Markey Bill"). This legislation

would require industry in the United States to reduce emissions of greenhouse gasses by the year 2050 by 83% from a baseline level of 2005. Other countries and certain states within the United States have also passed or are considering legislation to mitigate climate change by restricting the emissions of greenhouse gasses, while the EPA has initiated a rule-making process to reduce the emission of greenhouse gasses. It is unknown at this time whether or when the Waxman-Markey Bill or any similar legislation may become law. When using fossil fuels, our boiler products typically emit carbon dioxide. Were the Waxman-Markey Bill to become law, we believe that owners of power plants would respond first by reducing utilization rates and eventually by retiring fossil-fueled boilers. Future decisions to retire boilers would impact our business in a variety of ways, including the servicing and retrofitting of operating power plants. The need to replace retired generating capacity with cleaner technologies would also create business opportunities for us. To generate energy while minimizing the emission of greenhouse gasses, we are actively researching and developing a range of products, including:

- · non-carbon technologies, such as nuclear power plants and solar receivers for concentrating solar power plants;
- low-carbon technologies that enable clean use of fossil fuels, such as oxy-fuel combustion and regenerable solvent absorption technologies to scrub carbon dioxide from exhaust gasses; and
- carbon-neutral technologies, such as biomass-fueled boilers and gasifiers, which use a renewable resource where the growing biomass re-absorbs the
 carbon dioxide emitted during energy production.

At this time, we cannot predict the timing or extent of additional limits on emissions of greenhouse gasses, nor their specific impacts on our business.

Overview—Government Operations

We expect the backlog of our Government Operations segment of approximately \$2.8 billion at March 31, 2010 to produce revenues of approximately \$730 million in 2010, not including any change orders or new contracts that may be awarded during the year. Our liquidity position for this segment remains strong, and we expect it to remain so throughout 2010.

The revenues of our Government Operations segment are largely a function of defense spending by the U.S. Government. As a supplier of major nuclear components for certain U.S. Government programs, we are a significant participant in the defense industry. With our specialized capabilities of full life-cycle management of special nuclear materials, facilities and technologies, our Government Operations segment is well-positioned to continue to participate in the continuing cleanup, operation and management of the nuclear sites and weapons complexes maintained by the DOE.

In December 2009, our subsidiary Nuclear Fuel Services, Inc., which we acquired in December 2008, implemented a suspension of some operations at its Erwin, Tennessee manufacturing facility while implementing organizational, facility and management changes to enhance safety controls and processes. We suspended the operations in consultation with the NRC following the occurrence of two separate incidents that we reported to the NRC in October and November 2009. The October 2009 incident involved the generation of excessive heat and a hazardous gas at a specialized cleaning station. The incident was caused by the processing of a small amount of uranium-aluminum material in nitric acid. This incident resulted in some damage to piping, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. The November 2009 incident involved a fire in a glovebox enclosure. The incident was caused by a reaction between fluorine gas in a cylinder of uranium and the fiberglass backing of the teflon lining of a vent hose attached to the cylinder within the glovebox enclosure. This incident resulted in damage to a hose and a faceplate within the glove box enclosure, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. Inspections conducted separately by the NRC, our existing nuclear liability underwriter and us revealed specific modifications necessary to improve Nuclear Fuel Services, Inc.'s overall safety performance. Although the NRC

did not issue any directive or order to us to suspend these operations, we believe the NRC likely would have done so had we not voluntarily suspended the operations. Although we were already in the process of implementing safety, cultural and procedural improvements at Nuclear Fuel Services, Inc., we have refined and accelerated those changes and have developed other specific enhancements in consultation with the NRC, as confirmed in the NRC's January 7, 2010 confirmatory action letter to Nuclear Fuel Services, Inc.

The suspended operations include production operations, the commercial development line and the highly enriched uranium down blending facility. These operations are in the process of being brought back on line through a phased restart, which we commenced in March 2010 following completion of a third-party review and NRC review of the modifications that were implemented. The production operations are now fully operational. We began bringing the highly enriched uranium down-blending facility back on line, through a phased restart, in May 2010, and we expect that the facility will be fully operational in June 2010. We expect to bring the commercial development line back on line by the end of January 2011. There can be no assurance that we will not have to suspend our operations in the future to implement additional changes to enhance our safety controls and processes in order to comply with applicable laws and regulations.

Critical Accounting Policies and Estimates

Our financial statements and accompanying notes are prepared in accordance with accounting principles generally accepted in the United States. Preparing financial statements requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue and expenses. These estimates and assumptions are affected by management's application of accounting policies. We believe the following are our most critical accounting policies that we apply in the preparation of our financial statements. These policies require our most difficult, subjective and complex judgments, often as a result of the need to make estimates of matters that are inherently uncertain.

Contracts and Revenue Recognition. We determine the appropriate accounting method for each of our long-term contracts before work on the project begins. We generally recognize contract revenues and related costs on a percentage-of-completion method for individual contracts or combinations of contracts under the guidelines of the Financial Accounting Standards Board ("FASB") Topic Revenue Recognition. The use of this method is based on our experience and history of being able to prepare reasonably dependable estimates of the cost to complete our projects. Under this method, we recognize estimated contract revenue and resulting income based on costs incurred to date as a percentage of total estimated costs. Certain costs may be excluded from the cost-to-cost method of measuring progress, such as significant resulting periodic allocation of income. Total estimated costs, and resulting contract income, are affected by changes in the expected cost of materials and labor, productivity, scheduling and other factors. Additionally, external factors such as weather, customer requirements and other factors outside of our control may affect the progress and estimated cost of a project's completion and, therefore, the timing of revenue and income recognition. We routinely review estimates related to our contracts, and revisions to profitability are reflected in the quarterly and annual earnings we report.

For contracts as to which we are unable to estimate the final profitability except to assure that no loss will ultimately be incurred, we recognize equal amounts of revenue and cost until the final results can be estimated more precisely. For these deferred profit recognition contracts, we recognize revenue and cost equally and only recognize gross margin when probable and reasonably estimable, which we generally determine to be when the contract is approximately 70% complete. We treat long-term construction contracts that contain such a level of risk and uncertainty that estimation of the final outcome is impractical except to assure that no loss will be incurred as deferred profit recognition contracts.

Fixed-price contracts are required to be accounted for under the completed-contract method if we are unable to reasonably forecast cost to complete at start-up. For example, if we have no experience in performing the type of work on a particular project and were unable to develop reasonably dependable estimates of total costs to complete, we would follow the completed-contract method of accounting for such projects. Our management's policy is not to enter into fixed-price contracts without an accurate estimate of cost to complete. However, it is

possible that in the time between contract execution and the start of work on a project, we could lose confidence in our ability to forecast cost to complete based on intervening events, including, but not limited to, experience on similar projects, civil unrest, strikes and volatility in our expected costs. In such a situation, we would use the completed-contract method of accounting for that project. We did not enter into any contracts that we accounted for under the completed contract method during 2009, 2008 or 2007.

For all contracts, if a current estimate of total contract cost indicates a loss on a contract, the projected loss is recognized in full when determined.

Although we continually strive to improve our ability to estimate our contract costs and profitability, adjustments to overall contract costs due to unforeseen events could be significant in future periods. We recognize claims for extra work or for changes in scope of work in contract revenues, to the extent of costs incurred, when we believe collection is probable and can be reasonably estimated. We recognize income from contract change orders or claims when formally agreed with the customer. We reflect any amounts not collected as an adjustment to earnings. We regularly assess the collectibility of contract revenues and receivables from customers.

Property, Plant and Equipment. We carry our property, plant and equipment at depreciated cost, reduced by provisions to recognize economic impairment when we determine impairment has occurred. Factors that impact our determination of impairment include forecasted utilization of equipment and estimates of cash flow from projects to be performed in future periods. Our estimates of cash flow may differ from actual cash flow due to, among other things, technological changes, economic conditions or changes in operating performance. Any changes in such factors may negatively affect our business segments and result in future asset impairments.

We depreciate our property, plant and equipment using the straight-line method, over estimated economic useful lives of eight to 40 years for buildings and two to 28 years for machinery and equipment. We expense the costs of maintenance, repairs and renewals that do not materially prolong the useful life of an asset as we incur them.

Investments in Affiliates. We use the equity method of accounting for affiliates in which our investment ownership ranges from 20% to 50%, unless significant economic or governance considerations indicate that we are unable to exert significant influence, in which case the cost method is used. The equity method is also used for affiliates in which our investment ownership is greater than 50% but we do not have a controlling interest. Currently, all of our significant investments in affiliates that are not included in our combined results are recorded using the equity method. Affiliates in which our investment ownership is less than 20% and where we are unable to exert significant influence are carried at cost.

Self-Insurance. We have wholly owned insurance subsidiaries that provide employer's liability, general and automotive liability and workers' compensation insurance and, from time to time, builder's risk insurance within certain limits to our companies. We may also have business reasons in the future to have these insurance subsidiaries accept other risks which we cannot or do not wish to transfer to outside insurance companies. When estimating our self insurance liabilities, we consider a number of factors, including historical claims experience and trend lines, projected growth patterns, inflation and exposure forecasts. The assumptions we make with respect to each of these factors represent our judgment as to the most probable cumulative impact of each factor on our future obligations. Our calculation of self insurance liabilities requires us to apply judgment to estimate the ultimate cost to settle reported claims and claims incurred but not yet reported as of the balance sheet date. We engage the services of an actuarial firm to assist us in the calculation of our liabilities for self insurance. While the actual outcome of insured claims could differ significantly from estimated amounts, these loss estimates and accruals recorded in our financial statements for claims have historically been reasonable in light of the actual amount of claims paid. Provisions for exposure to self insurance claims and the related payments of claims have historically not had a material adverse impact on our consolidated financial position, results of operations and cash flows, and we do not expect these provisions to have a material impact on our self-insurance programs in the future.

Pension Plans and Postretirement Benefits. We estimate income or expense related to our pension and postretirement benefit plans based on actuarial assumptions, including assumptions regarding discount rates and expected returns on plan assets. We determine our discount rate based on a review of published financial data and discussions with our actuary regarding rates of return on high-quality, fixed-income investments currently available and expected to be available during the period to maturity of our pension obligations. Based on historical data and discussions with our actuary, we determine our expected return on plan assets based on the expected long-term rate of return on our plan assets and the market-related value of our plan assets. Changes in these assumptions can result in significant changes in our estimated pension income or expense and our combined financial condition. We revise our assumptions on an annual basis based upon changes in current interest rates, return on plan assets and the underlying demographics of our workforce. These assumptions are reasonably likely to change in future periods and may have a material impact on future earnings. Effective December 31, 2009, we adopted the disclosure provisions of FASB Topic 715, Compensation—Retirement Benefits. In accordance with this provision, we have disclosed additional information about our assets set aside to fund our pension and postretirement benefit obligations. See Note 7 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement for information on our pension and postretirement benefit plans.

Loss Contingencies. We estimate liabilities for loss contingencies when it is probable that a liability has been incurred and the amount of loss is reasonably estimable. We provide disclosure when there is a reasonable possibility that the ultimate loss will exceed the recorded provision or if such loss is not reasonably estimable. We are currently involved in some significant litigation, as discussed in Note 9 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement. We have accrued our estimates of the probable losses associated with these matters. However, our losses are typically resolved over long periods of time and are often difficult to estimate due to the possibility of multiple actions by third parties. Therefore, it is possible future earnings could be affected by changes in our estimates related to these matters.

Goodwill. FASB Topic Intangibles—Goodwill and Other, requires us to perform periodic testing for impairment. It requires a two-step impairment test to identify potential goodwill impairment and measure the amount of a goodwill impairment loss. The first step of the test compares the fair value of a reporting unit with its carrying amount, including goodwill. If the carrying amount of a reporting unit exceeds its fair value, the second step of the goodwill impairment test is performed to measure the amount of the impairment loss, if any. The second step compares the implied fair value of the reporting unit's goodwill with the carrying amount of that goodwill. Each year, we evaluate goodwill at each reporting unit to assess recoverability, and impairments, if any, are recognized in earnings. An impairment loss would be recognized in an amount equal to the excess of the carrying amount of the goodwill over the implied fair value of the goodwill. We determined that both the income and market valuation approaches provide inputs into the estimate of the fair value of our reporting units, which would be considered by market participants. Under the income valuation approach, we employed a discounted cash flow model to estimate the fair value of each reporting unit. This model requires the use of significant estimates and assumptions regarding future revenues, costs, margins, capital expenditures, changes in working capital, terminal year growth rate and cost of capital. Our cash flow models are based on our forecasted results for the applicable reporting units. Actual results could differ from our projections. Under the market valuation approach, we employed the guideline publicly traded company method, which indicates the fair value of the equity of each reporting unit by comparing it to publicly traded companies in similar lines of business. After identifying and selecting guideline companies, we analyzed their business and financial profiles for relative similarity. Factors such as size, growth, risk and profitabili

Asset Retirement Obligations and Environmental Clean-up Costs. We accrue for future decommissioning of our nuclear facilities that will permit the release of these facilities to unrestricted use at the end of each facility's life, which is a requirement of our licenses from the NRC. In accordance with the FASB Topic Asset

Retirement and Environmental Obligations, we record the fair value of a liability for an asset retirement obligation in the period in which it is incurred. In estimating fair value, we use present values of cash flows expected to be incurred in settling our obligations. To the extent possible, we perform a marketplace assessment of the cost and timing of performing the retirement activities. We apply a credit-adjusted risk-free interest rate to our expected cash flows in our determination of fair value. When we initially record such a liability, we capitalize a cost by increasing the carrying amount of the related long-lived asset. Over time, the liability is accreted to its present value each period, and the capitalized cost is depreciated over the useful life of the related asset. Upon settlement of a liability, we will settle the obligation for its recorded amount or incur a gain or loss. This topic applies to environmental liabilities associated with assets that we currently operate and are obligated to remove from service. For environmental liabilities associated with assets that we no longer operate, we have accrued amounts based on the estimated costs of clean-up activities, net of the anticipated effect of any applicable cost-sharing arrangements. We adjust the estimated costs as further information develops or circumstances change. An exception to this accounting treatment relates to the work we perform for one facility for which the U.S. Government is obligated to pay all the decommissioning costs.

Deferred Taxes. We record a valuation allowance to reduce our deferred tax assets to the amount that is more likely than not to be realized. We believe that the deferred tax asset recorded as of December 31, 2009 is more likely than not realizable through carrybacks, future reversals of existing taxable temporary differences and future taxable income. If we were to subsequently determine that we would be able to realize deferred tax assets in the future in excess of our net recorded amount, an adjustment to deferred tax assets would increase earnings for the period in which such determination was made. We will continue to assess the adequacy of the valuation allowance on a quarterly basis. Any changes to our estimated valuation allowance could be material to our combined financial condition and results of operations. Effective January 1, 2007, we adopted the provision of FASB Topic *Income Taxes*.

Warranty. We account for warranty costs to satisfy contractual warranty requirements as an accrued estimated expense recognized in conjunction with the associated revenue on the related contracts for our Power Generation Systems and Government Operations segments. In addition, we make specific provisions where we expect the actual warranty costs to significantly exceed the accrued estimates. Factors that impact our estimate of warranty costs include prior history of warranty claims and our estimates of future costs of materials and labor. Our future warranty provisions may vary from what we have experienced in the past.

Stock-Based Compensation. We account for stock-based compensation in accordance with FASB Topic Compensation—Stock Compensation. Under the fair value recognition provisions of this statement, the cost of employee services received in exchange for an award of equity instruments is measured at the grant date based on the fair value of the award. Stock-based compensation expense is recognized on a straight-line basis over the requisite service periods of the awards, which is generally equivalent to the vesting term. We use a Black-Scholes model to determine the fair value of certain share-based awards, such as stock options. The use of a Black-Scholes model requires highly subjective assumptions, such as assumptions about the volatility of our stock price and our expected dividend yield.

Business Combinations. Effective January 1, 2009, we became subject to the provisions of the FASB Topic Business Combinations. This topic broadens the fair value measurements and recognition of assets acquired, liabilities assumed and interests transferred as a result of business combinations. It also provides disclosure requirements to assist users of the financial statements in evaluating the nature and financial effects of business combinations.

For further discussion of recently adopted accounting standards, see Note 1 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement.

Three Months Ended March 31, 2010 Compared to Three Months Ended March 31, 2009

The Babcock & Wilcox Operations of McDermott International, Inc. (Combined)

Revenues:

Combined revenues decreased approximately 16%, or \$122.7 million, to \$662.4 million in the three months ended March 31, 2010 compared to \$785.1 million for the corresponding period in 2009 due to decreases in revenues from both of our business segments. Our Power Generation Systems segment experienced a 22% reduction in revenues largely attributable to reductions in our utility steam and system fabrication operations and fabrication repair and retrofit of existing facilities operations. Our Government Operations segment experienced a 1.5% reduction in revenues attributable primarily to lower volumes in the manufacture of components for a commercial uranium enrichment project and lower volumes in our commercial manufacturing operations.

Operating income including selling, general and administrative expenses, losses/gains on asset disposals and impairments-net, and equity in income of investees:

Combined operating income decreased \$56.6 million to \$38.7 million in the three months ended March 31, 2010 from \$95.3 million for the corresponding period in 2009 due to reduced segment operating income in both of our segments. The operating income of our Power Generation Systems and Government Operations segments decreased \$48.9 million and \$9.8 million, respectively, in the three months ended March 31, 2010 compared to the corresponding period of 2009. Our equity in income of investees, which is included in operating income, increased by \$3.7 million in the three months ended March 31, 2010 compared to the corresponding period of 2009, primarily attributable to improvements in our joint venture in China. We experienced an increase in our combined selling, general and administrative expenses totaling \$3.7 million in the three months ended March 31, 2010 compared to the corresponding period of 2009 attributable primarily to a \$1.7 million increase in selling and marketing expense in the 2010 period compared to the corresponding period of 2009.

Other income (expense):

Interest income decreased \$0.8 million to \$0.4 million in the three months ended March 31, 2010, primarily due to decreases in average cash equivalents and investments and prevailing interest rates.

Interest expense increased \$0.7 million to \$6.0 million in the three months ended March 31, 2010, primarily due to interest expense on a note payable to an affiliate.

Other expense - net increased by \$1.1 million to \$3.1 million in the three months ended March 31, 2010 primarily due to foreign currency exchange losses.

Provision for income taxes:

For the three months ended March 31, 2010 our provision for income taxes decreased \$22.4 million to \$13.3 million, while income before provision for income taxes decreased \$59.1 million to \$30.1 million. Our effective tax rate for the three months ended March 31, 2010 was approximately 44.02% as compared to 40.0% for the three months ended March 31, 2009. The rate increase was primarily attributable to a higher mix of U.S. earnings and losses in certain of our non-U.S. operations, which were benefited at lower rates.

Income before provision for income taxes, provision for income taxes and effective tax rates for our U.S. and non-U.S. jurisdictions were as shown below for the three months ended March 31:

| | Incom | e from | | | | |
|-------------|------------|--------------|----------|--------------|----------|--------|
| | Conti | nuing | | | | |
| | Opera | tions | | | | |
| | before Pro | ovision for | Provis | sion for | | |
| | Income | Income Taxes | | Income Taxes | | x Rate |
| | 2010 | 2009 | 2010 | 2009 | 2010 | 2009 |
| | (In thou | usands) | (In tho | usands) | | |
| ites | \$31,830 | \$75,572 | \$12,973 | \$ 30,830 | 40.76% | 40.79% |
| ited States | (1,718) | 13,674 | 283 | 4,880 | (16.47)% | 35.69% |
| | \$30,112 | \$89,246 | \$13,256 | \$35,710 | 44.02% | 40.00% |
| | | | | | | |

Power Generation Systems

Revenues decreased approximately 22%, or \$118.9 million, to \$409.7 million in the three months ended March 31, 2010, compared to \$528.6 million for the corresponding period of 2009. This decrease was primarily attributable to a \$61.6 million decrease in revenues from our utility steam and system fabrication operations and a \$46.8 million decrease in revenues from our fabrication, repair and retrofit of existing facilities operations in the three months ended March 31, 2010 compared to the corresponding period of 2009. The main driver for these decreases in revenues was a significant decrease in orders in 2009 compared to 2008, and particularly in the last six months of 2009. These reductions in orders booked followed a Federal Appeals Court's overturning of the Clean Air Interstate Rule, the Clean Air Mercury Rule and the Industrial Boiler Rule on Maximum Achievable Control Technology. The overturning of these rules and regulations, along with the uncertainty concerning any new legislation or replacement rules or regulations, has caused many of our major customers, principally electric utilities, to delay making substantial capital expenditures for new plants, as well as upgrades to existing power plants. In addition, considerations surrounding greenhouse gas limits under consideration by the U.S. Congress and the EPA have delayed the construction of new coal-fired power plants in the United States. We currently believe, as the EPA finalizes the revised rules and regulations (which we anticipate may be released in final form by mid 2011), our U.S. customers will increase their expenditures on environmental equipment to bring their operating, coal-fired power plants into compliance with the new emissions limits. When the final EPA rules are released, we expect that our bookings from new environmental equipment for those existing plants will begin to increase with revenue and operating income to follow normal project cycles. It is also possible that some of our customers may elect to close down their least efficient coal-fired boilers, and, as a result, our revenues and operating income from upgrade and retrofit activities from those customers may decline. We also experienced lower revenues from our boiler auxiliary equipment operations (\$15.5 million), our industrial boilers operations (\$8.6 million), and our replacement parts operations (\$8.4 million) in the three months ended March 31, 2010 compared to the corresponding period of 2009. These decreases were partially offset by increased revenues from our nuclear service operations (\$16.6 million), nuclear steam generator operations (\$4.2 million), and field service operations (\$2.4 million).

Operating income decreased \$48.9 million to \$9.3 million in the three months ended March 31, 2010, compared to \$58.2 million in the corresponding period of 2009. This decrease was primarily attributable to lower margins in our utility steam and system fabrication operations and our fabrication, repair and retrofit of existing facilities operations, primarily attributable to the effects of timing on profitable projects completed in the three months ended March 31, 2009 compared to the corresponding period of 2010 and to the decreases in revenues discussed above. We also experienced reductions in gross profit in our boiler auxiliary equipment and replacement parts operations totaling approximately \$6.0 million in 2010 compared to the corresponding period of 2009, primarily attributable to the decreases in revenues discussed above. In addition, we experienced lower margins in our nuclear steam generator operations in the three months ended March 31, 2010, which included an additional provision for warranty obligations totaling approximately \$3.7 million, compared to the corresponding period of 2009. We also experienced an increase in research and development expenses totaling approximately

\$7.6 million in the three months ended March 31, 2010 compared to the corresponding period of 2009, primarily attributable to research and development activities associated with our B&W mPower™ initiative. In addition, we experienced increases in our selling, general and administrative expenses totaling \$5.2 million in the three months ended March 31, 2010 compared to the corresponding period of 2009. These adverse impacts on segment operating income were partially offset by improvements in our industrial boilers and nuclear service operations.

In addition, equity in income of investees increased \$2.9 million from \$1.6 million in the three months ended March 31, 2009 to \$4.5 million in the three months ended March 31, 2010, primarily attributable to our joint venture in China.

Government Operations

Revenues decreased approximately 1.5%, or \$3.8 million, to \$253.3 million in the three months ended March 31, 2010 compared to \$257.1 million for the corresponding period of 2009, primarily attributable to lower volumes in the manufacture of components for a commercial uranium enrichment project (\$20.1 million), and lower volumes in our commercial manufacturing operations (\$9.2 million), substantially all of which was attributable to our Nuclear Fuel Services, Inc. subsidiary. These decreases were partially offset by increased volumes in the manufacture of nuclear components for certain U.S. Government programs and recovery work.

Operating income decreased \$9.8 million to \$36.0 million in the three months ended March 31, 2010, compared to \$45.8 million for the corresponding period of 2009, primarily attributable to the shutdown at the Nuclear Fuel Services, Inc. facility. We suspended the operations in consultation with the NRC following the occurrence of two separate incidents that we reported to the NRC in October and November 2009. The October 2009 incident involved the generation of excessive heat and a hazardous gas at a specialized cleaning station. The incident was caused by the processing of a small amount of uranium-aluminum material in nitric acid. This incident resulted in some damage to piping, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. The November 2009 incident involved a fire in a glovebox enclosure. The incident was caused by a reaction between fluorine gas in a cylinder of uranium and the fiberglass backing of the teflon lining of a vent hose attached to the cylinder within the glovebox enclosure. This incident resulted in damage to a hose and a faceplate within the glove box enclosure, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. Inspections conducted separately by the NRC, our existing nuclear liability underwriter and us revealed specific modifications necessary to improve Nuclear Fuel Services, Inc.'s overall safety performance. The suspended operations include production operations, the commercial development line and the highly enriched uranium down-blending facility. These operations are in the process of being brought back on line through a phased restart, which we commenced in March 2010 following completion of a third-party review and NRC review of the modifications that were implemented. The production operations are now fully operational. We began bringing the highly enriched uranium down-blending facility back on line, through a phased restart, in May 2010, and we expect that the facility will be fully operational in June 2010. We expect to bring the commercial development line back on line by the end of January 2011. That line represented less than 0.5% of our combined revenues in 2009. The impact of the shutdown was approximately \$9.0 million in the three months ended March 31, 2010. We also experienced lower contract productivity and lower volumes in the manufacture of components for a commercial uranium enrichment project in the three months ended March 31, 2010 compared to the corresponding period of 2009. In addition, equity in income of investees increased \$0.8 million to \$9.5 million in the three months ended March 31, 2010, compared to \$8.7 million in the corresponding period of 2009, primarily attributable to higher fees earned at two of our government sites, partially offset by increased expenses associated with one of our joint ventures in Tennessee.

Corporate

Unallocated corporate expenses decreased \$2.1 million to \$6.5 million for the three months ended March 31, 2010, as compared to \$8.6 million for the corresponding period of 2009, primarily attributable to reduced compensation expense due to reductions in corporate headcount totaling \$2.7 million and lower legal expenses incurred in the three months ended March 31, 2010 compared to the corresponding period of 2009.

Year Ended December 31, 2009 Compared to Year Ended December 31, 2008

The Babcock & Wilcox Operations of McDermott International, Inc. (Combined)

Revenues:

Combined revenues decreased approximately 16%, or \$544.0 million, to \$2.9 billion in the year ended December 31, 2009, compared to \$3.4 billion in the year ended December 31, 2009. Our Power Generation Systems segment revenues decreased approximately 28% in the year ended December 31, 2009, as compared to 2008, primarily attributable to lower revenues in our utility steam and system fabrication operations and fabrication, repair and retrofit of existing facilities operations. Our Government Operations segment revenues increased approximately 21% in the year ended December 31, 2009, compared to 2008, primarily attributable to the acquisition of Nuclear Fuel Services, Inc.

Operating income including selling, general and administrative expenses, losses/gains on asset disposals and impairments—net, and equity in income of investees:

Combined operating income decreased \$174.9 million from \$444.5 million in 2008 to \$269.6 million in 2009. Our Power Generation Systems segment operating income decreased by \$157.5 million in the year ended December 31, 2009, compared to the year ended December 31, 2008, primarily attributable to the decrease in revenues discussed above. Our Government Operations segment operating income increased by \$4.3 million in the year ended December 31, 2009, as compared to 2008, primarily attributable to higher volumes of manufacturing activity. We also experienced increases in selling, general and administrative expenses, which are included in operating income, totaling approximately \$66.0 million in the year ended December 31, 2009 compared to 2008, primarily attributable to a \$14.4 million increase in selling and marketing expenses in 2009, and \$36.3 million from our acquisition of Nuclear Fuel Services, Inc. Our equity in income of investees, which is included in operating income, increased \$3.3 million to \$55.1 million in the year ended December 31, 2009 primarily attributable to improved results from our Power Generation Systems segment's joint venture in China. We also realized a decrease in gains on asset disposals in the year ended 2009 compared to 2008 attributable to a gain we recognized in 2008 on the sale of our former Dumbarton, Scotland facility.

Other income (expense):

Interest income decreased \$11.9 million to \$3.4 million in the year ended December 31, 2009 compared to 2008, primarily due to decreases in average cash equivalents and investments and prevailing interest rates.

Interest expense increased \$1.9 million in the year ended December 31, 2009 compared to 2008, primarily due to interest expense on a note payable to an affiliate, as discussed in Note 5 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement.

Other expense—net increased \$11.8 million to a loss of \$16.1 million in the year ended December 31, 2009, primarily due to higher currency exchange losses in 2009.

Provision for income taxes:

For the year ended December 31, 2009, our provision for income taxes decreased \$24.5 million to \$84.4 million, while income before provision for income taxes decreased \$200.4 million to \$232.3 million. Our effective tax rate was approximately 36% for 2009, as compared to 25% for 2008. The increase in the effective rate was primarily attributable to non-recurring tax benefits which we recognized from the release of state valuation allowances and as a result of audit activity in 2008.

Income before provision for income taxes, provision for income taxes and effective tax rates for our U.S. and non-U.S. jurisdictions were as shown below:

| | Incom | e from | | | | |
|-------------------|------------|----------------|-----------|------------|--------------------|--------|
| | Continuing | Operations | | | | |
| | before Pro | ovision for | Provi | sion for | | |
| | Income | Income Taxes | | e Taxes | Effective Tax Rate | |
| | 2009 | 2008 | 2009 | 2008 | 2009 | 2008 |
| | (In tho | (In thousands) | | ousands) | | |
| United States | \$ 164,347 | \$ 351,398 | \$ 69,279 | \$ 89,904 | 42.15% | 25.58% |
| Non-United States | 67,954 | 81,341 | 15,102 | 18,981 | 22.22% | 23.34% |
| Total | \$ 232,301 | \$ 432,739 | \$ 84,381 | \$ 108,885 | 36.32% | 25.16% |
| | | | | | | |

We are subject to U.S. federal income tax at a rate of 35% on our U.S. operations plus the applicable state income taxes on our profitable U.S. subsidiaries. Our non-U.S. earnings are subject to tax at various tax rates and under various tax regimes, including deemed profits tax regimes.

During the year ended December 31, 2009, we recorded a reduction in liabilities under FASB TopicIncome Taxes of approximately \$5.4 million, including estimated tax-related interest and penalties.

See Note 4 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement for further information on taxes.

Power Generation Systems

Revenues decreased approximately 28%, or \$724.4 million, to \$1,824.5 million in the year ended December 31, 2009, compared to \$2,548.9 million in 2008. This decrease was primarily attributable to a 43% (or \$491.9 million) decrease we experienced in revenues from our utility steam and system fabrication operations and decreased revenues in our fabrication, repair and retrofit of existing facilities operations (\$171.1 million) in 2009. The main driver for these decreases in revenues was a significant decrease in orders booked starting in the final six months of 2008. These reductions in orders booked followed a Federal Appeals Court's overturning of the Clean Air Interstate Rule, the Clean Air Mercury Rule and the Industrial Boiler Rule on Maximum Achievable Control Technology. The overturning of these rules and regulations, along with the uncertainty concerning any new legislation or replacement rules or regulations, has caused many of our major customers, principally electric utilities, to delay making substantial capital expenditures for new plants, as well as upgrades to existing power plants. In addition, considerations surrounding greenhouse gas limits under consideration by the U.S. Congress and the EPA have delayed the construction of new coal-fired power plants in the United States. We currently believe, as the EPA finalizes the revised rules and regulations (which we anticipate may be released in final form by mid 2011), our U.S. customers will increase their expenditures on environmental equipment to bring their operating, coal-fired power plants into compliance with the new emissions limits. When the final EPA rules are released, we expect that our bookings from new environmental equipment for those existing plants will begin to increase with revenue and operating income to follow normal project cycles. It is also possible that some of our customers may elect to close down their least efficient coal-fired boilers, and, as a result, our revenues and operating income from upgrade and retrofit activities from those customers may decline. We also experienced lower revenues in 2009 compared to 2008 in our replacement parts operations (\$23.3 million), our industrial boilers operations (\$16.9 million), our nuclear steam generator operations (\$16.7 million) and our boiler auxiliary equipment operations (\$13.8 million) due to reduced demand for electricity and lower capital expenditures by our customers. These decreases were partially offset by increased revenues from our operations and maintenance operations (\$10.3 million) and nuclear service operations (\$4.1 million).

Operating income decreased \$157.5 million to \$157.9 million in the year ended December 31, 2009, compared to \$315.4 million in 2008, primarily attributable to lower volumes in our utility steam and system fabrication operations related to the decreases in revenues discussed above, and a bonus on a contract completed in 2008, for a total reduction of \$65.4 million in gross profit. In addition, we experienced lower gross profits in our fabrication, repair and retrofit of existing facilities operations and in our replacement parts operations, which

were also primarily attributable to the decreases in revenues discussed above. We also experienced lower margins and volume in our nuclear steam generator operations. The lower margins in our nuclear steam generator operations were attributable primarily to increased contract costs on the manufacture and delivery of two replacement once through steam generators, and our lower volumes were attributable to the decrease in revenues discussed above. In addition, we incurred an increase in our pension plan expense totaling approximately \$40.2 million in 2009 compared to 2008, primarily from the amortization of losses on our pension plan incurred in 2008. We also experienced an increase in selling, general and administrative expenses totaling approximately \$19.1 million in 2009 compared to 2008, primarily attributable to an increase of approximately \$11 million in selling and marketing expenses in 2009 compared to 2008. These decreases were partially offset by improved margins in our utility steam and system fabrication operations and our fabrication, repair and retrofit of existing facilities operations primarily attributable to contract cost improvements and the completion of projects in 2009, as well as higher volumes in our operations and maintenance operations attributable to the increase discussed above. Gains (losses) on asset disposals and impairments—net decreased \$9.9 million in the year ended December 31, 2009, primarily attributable to the gain we recognized in 2008 on the sale of our former Dumbarton, Scotland facility. Equity in income of investees increased \$3.6 million to \$14.0 million for the year ended December 31, 2009 compared to \$10.4 million in 2008, primarily attributable to our joint venture in China.

Government Operations

Revenues increased approximately 21%, or \$181.0 million, to \$1,032.0 million in the year ended December 31, 2009 compared to \$851.0 million in 2008, primarily attributable to our acquisition of Nuclear Fuel Services, Inc. (\$160.7 million) and additional volumes in the manufacture of nuclear components for certain U.S. Government programs and recovery work (\$20.7 million). These improvements were partially offset by lower volumes in the manufacture of components for a commercial uranium enrichment project, engineering and laboratory services and lower revenues from our management and operating contracts at several government sites.

Operating income increased \$4.3 million to \$154.5 million in the year ended December 31, 2009 compared to \$150.2 million in 2008, primarily attributable to additional volumes in the manufacture of nuclear components for certain U.S. Government programs and recovery work, and our acquisition of Nuclear Fuel Services, Inc., which contributed \$3.3 million of segment operating income. These improvements were partially offset by increased pension expense of \$39.5 million in 2009 compared to 2008 from the amortization of losses on our pension plan assets in 2008. In addition, we experienced lower volumes related to a commercial uranium enrichment project and lower revenues from our management and operating contracts at several government sites.

Corporate

Unallocated corporate expenses increased \$21.7 million in the year ended December 31, 2009 to \$42.8 million from \$21.1 million in 2008, primarily attributable to increased pension plan expense of \$16.3 million in 2009 compared to 2008 from the amortization of losses on pension plan assets experienced in 2008. We also experienced an increase in information technology expenses in the year ended December 31, 2009 compared to 2008.

Year Ended December 31, 2008 Compared to Year Ended December 31, 2007

The Babcock & Wilcox Operations of McDermott International, Inc. (Combined)

Revenues

Combined revenues increased approximately 6%, or \$198.6 million, to \$3.4 billion in the year ended December 31, 2008, compared to \$3.2 billion in the year ended December 31, 2007. Our Power Generation Systems segment revenues increased approximately 2% in the year ended December 31, 2008, as compared to 2007. Our Government Operations segment revenues increased approximately 23% in the year ended December 31, 2008, compared to 2007, primarily attributable to higher volumes in the manufacture of nuclear components for certain U.S. Government programs and for a commercial uranium enrichment project.

Operating income including selling, general and administrative expenses, losses/gains on asset disposals and impairments—net, and equity in income of investees:

Combined operating income increased \$106.9 million from \$337.6 million in the year ended December 31, 2007 to \$444.5 million in 2008. Our Power Generation Systems segment operating income increased by \$81.3 million in the year ended December 31, 2008, as compared to the year ended December 31, 2007, primarily attributable to favorable cost improvements on a significant number of its projects. Our Government Operations segment operating income increased by \$27.3 million in the year ended December 31, 2008, as compared to the year ended December 31, 2007, primarily attributable to the higher volumes of manufacturing activity described above. We experienced an increase in losses (gains) on asset disposals and impairments—net, included in operating income, totaling net gains of \$8.0 million in 2009 compared to 2008 primarily from a gain we recognized on the sale of our former Dumbarton, Scotland facility. Our equity in income of investees, which is included in operating income, increased by \$6.2 million in 2009 compared to 2008 primarily attributable to improved results in our joint ventures in Idaho, Tennessee and Louisiana, partially offset by lower results from our joint venture in China attributable to higher costs for materials. Our combined selling, general and administrative expenses increased \$48.0 million in 2008 to \$337.6 million compared to \$289.6 million in 2007 primarily attributable to information technology expense, support services for our utility steam and system fabrication operations and fabrication, repair and retrofit of existing facilities operations and increased selling and marketing expenses.

Other income (expense):

Interest income decreased \$9.9 million to \$15.3 million in the year ended December 31, 2008, primarily due to a decrease in average cash equivalents and investments and prevailing interest rates.

Interest expense decreased \$15.4 million to \$22.7 million in the year ended December 31, 2008, primarily due to interest during the year ended December 31, 2007 on the B&W PGG term loan that was terminated in April 2007 and lower amortization and costs under our credit facilities.

Other expense—net decreased \$6.1 million to \$4.2 million in the year ended December 31, 2008, primarily due to a provision for bad debt expense recorded during 2007.

Provision for income taxes:

For the year ended December 31, 2008, our provision for income taxes increased \$9.9 million to \$108.9 million, while income before provision for income taxes increased \$118.4 million to \$432.7 million. Our effective tax rate was approximately 25% for the year ended December 31, 2008, as compared to 32% for the year ended December 31, 2007. The decrease in the effective tax rate was primarily attributable to certain tax assets and benefits totaling approximately \$61.8 million, which we recognized in 2008 from the release of state valuation allowances and as a result of audit activity.

Income before provision for income taxes, provision for income taxes and effective tax rates for our U.S. and non-U.S. jurisdictions were as shown below:

| Incom | e from | | | | |
|------------|--|---|--|---|---|
| Continuing | Operations | | | | |
| before Pro | before Provision for | | | Effective Tax | |
| Income | Income Taxes | | Income Taxes | | e |
| 2008 | 2007 | 2008 | 2007 | 2008 | 2007 |
| (In thou | (In thousands) | | (In thousands) | | |
| \$351,398 | \$256,319 | \$ 89,904 | \$ 84,314 | 25.58% | 32.89% |
| 81,341 | 58,029 | 18,981 | 14,704 | 23.34% | <u>25.34</u> % |
| \$432,739 | \$314,348 | \$108,885 | \$99,018 | 25.16% | 31.50% |
| | Continuing before Pro Income 2008 (In the \$351,398 81,341 | Income Taxes 2008 2007 (In thousands) \$351,398 \$256,319 81,341 58,029 | Continuing Operations before Provision for Income Taxes 2008 2007 (In thousands) (In thousands | Continuing Operations before Provision for Income Taxes 2008 2007 (In thousands) (In thousands) (In thousands) (In thousands) (In thousands) (81,341 58,029 18,981 14,704 | Continuing Operations before Provision for Income Taxes 2008 2007 2008 2007 (In thousands) (In thousands) \$351,398 \$256,319 \$89,904 \$84,314 25.58% 81,341 58,029 18,981 14,704 23.34% |

We are subject to U.S. federal income tax at a rate of 35% on our U.S. operations plus the applicable state income taxes on our profitable U.S. subsidiaries. Our non-U.S. earnings are subject to tax at various tax rates and under various tax regimes, including deemed profits tax regimes.

Power Generation Systems

Revenues increased approximately 2%, or \$42.1 million, to \$2,548.9 million in the year ended December 31, 2008, compared to \$2,506.8 million in the year ended December 31, 2007. In 2008, we experienced a 23% increase in revenues from our fabrication, repair and retrofit of existing facilities operations (or \$127.2 million), primarily resulting from increases in orders booked in 2007 and timing of scheduled activities on projects throughout 2008. In addition, we experienced increases in our nuclear service operations (\$58.3 million), boiler auxiliary equipment operations (\$22.3 million), industrial boilers operations (\$17.2 million) and replacement parts operations (\$12.9 million), primarily resulting from timing of scheduled activities on projects. These increases were partially offset by decreased revenues from our utility steam and system fabrication operations (\$206.5 million), due primarily to the absence in 2008 of approximately \$178 million in revenues recognized from our termination and settlement agreement executed with TXU Corp. ("TXU") on the cancellation of five contracts to supply TXU supercritical, coal-fired boilers and selective catalytic reduction systems.

Operating income increased \$81.3 million to \$315.4 million in the year ended December 31, 2008, compared to \$234.1 million in 2007, primarily attributable to improved margins in our utility steam and system fabrication operations and increased volume and margins in our fabrication, repair and retrofit of existing facilities and replacement parts operations. These increased margins were largely the result of favorable cost improvements on a significant number of our projects and a bonus award of approximately \$13 million on a contract we completed in 2008. In addition, we experienced increased volumes in our nuclear service operations. We also experienced an approximately \$11 million decrease in our pension plan expense in 2008 compared to 2007, attributable to an improvement in the performance of our pension plan assets in 2008. Partially offsetting these increases were lower volumes and margins in our replacement nuclear steam generator operations and lower margins in our nuclear service operations. We also experienced \$21.0 million in higher selling, general and administrative expenses (net of pension expense), including higher stockbased compensation expense totaling \$2.9 million, in the year ended December 31, 2008. Gains (losses) on asset disposals and impairments—net increased \$9.6 million for the year ended December 31, 2008, primarily attributable to cost increases for materials at our joint venture in China

Government Operations

Revenues increased approximately 23%, or \$157.0 million, to \$851.0 million in the year ended December 31, 2008 compared to \$694.0 million in the year ended December 31, 2007, primarily attributable to higher volumes in the manufacture of nuclear components for certain U.S. Government programs (\$61.0 million), including increased contract procurement activities and additional volumes from Marine Mechanical Corporation, which we acquired in May 2007. In addition, we experienced higher volumes in the manufacture of nuclear components for a commercial uranium enrichment project (\$79.5 million) and higher revenues in our management and operating ("M&O") contracts.

Operating income increased \$27.3 million to \$150.2 million in the year ended December 31, 2008 compared to \$122.9 million in 2007, primarily attributable to the higher volumes in the manufacture of nuclear components for certain U.S. Government programs discussed above. In addition, we experienced higher volumes related to the commercial uranium enrichment project referenced above. We also experienced an approximately \$12.3 million decrease in our pension expense in 2008 compared to 2007, attributable to an improvement in the performance of our pension plan assets in 2008. These improvements were partially offset by the completion of

subcontract at a DOE clean-up site in Ohio during 2007 and the completion of M&O contracts at certain government sites. We also experienced higher selling, general and administrative expenses totaling approximately \$14.0 million in 2008 compared to 2007, primarily due to increased bid and proposal costs. Gains on asset disposals and impairments—net decreased by \$1.6 million in the year ended December 31, 2008, attributable to the gain we recorded on the sale of our investment in a research and development venture during the year ended December 31, 2007. Equity in income from investees increased \$10.1 million to \$41.4 million in the year ended December 31, 2008 compared to \$31.3 million in 2007, primarily attributable to increased profitability from our joint ventures in Idaho, Tennessee and Louisiana.

Corporate

Unallocated corporate expenses increased \$1.7 million in the year ended December 31, 2008 to \$21.1 million from \$19.4 million in the year ended December 31, 2007, primarily attributable to increased departmental expenses and higher expenses associated with our development of a global human resources management system. These increases were partially offset by lower pension plan expense in the year ended December 31, 2008.

Effects of Inflation and Changing Prices

Our financial statements are prepared in accordance with generally accepted accounting principles in the United States, using historical U.S. dollar accounting ("historical cost"). Statements based on historical cost, however, do not adequately reflect the cumulative effect of increasing costs and changes in the purchasing power of the dollar, especially during times of significant and continued inflation.

In order to minimize the potential negative impact of inflation on our operations, we attempt to cover the increased cost of anticipated changes in labor, material and service costs, either through an estimate of those changes, which we reflect in the original price, or through price escalation clauses in our contracts.

Liquidity and Capital Resources

Our overall liquidity position, which we generally define as our unrestricted cash and investments plus amounts available for borrowings under our credit facility, continued to remain strong in 2009. Our liquidity position at December 31, 2009 increased by approximately \$147 million to \$820 million from December 31, 2008, mainly due to factors discussed below in connection with the changes in our cash flows from operating, investing and financing activities. We experienced an increase in cash generated from our operating activities in 2009 compared to 2008. The major components of our net cash provided by operating activities were improvements in our net contracts in progress and advance billings components of working capital, improvement in our accounts receivable position, and lower funding of our pension plans. In some years, significant liquidity has been provided by our advance billings on contracts in progress, as a result of payments received from customers. As our customer cash advances are used in project execution and are not replaced by advances on new projects, our liquidity position is reduced. We experienced this condition in the year ended December 31, 2008, when we realized a use of cash from net contracts in progress and advance billings totaling approximately \$234 million.

Our overall liquidity position decreased by approximately \$60 million to \$760 million at March 31, 2010 compared to \$820 million at December 31, 2009. This decrease was largely attributable to our cash used in operating and investing activities. We experienced an increased use of cash from our net contracts in progress and advance billings components of working capital and from our pension liability, accumulated postretirement benefit obligation and accrued employee benefits in the three months ended March 31, 2010. We also experienced an increased use of cash in investing activities for capital expenditures and an acquisition in the three months ended March 31, 2010.

2010 Credit Facility

On May 3, 2010, Babcock & Wilcox Investment Company ("BWICO"), one of our affiliates, entered into a credit agreement (the "Credit Agreement") with a syndicate of lenders and letter of credit issuers and Bank of America, N.A., as administrative agent.

The Credit Agreement replaced the prior \$400.0 million senior secured revolving credit facility of B&W PGG and the prior \$135.0 million senior unsecured revolving credit facility of our subsidiary BWX Technologies, Inc. ("BWXT"). All amounts outstanding under B&W PGG's and BWXT's previous senior revolving credit facilities have been repaid, and all letters of credit outstanding under those prior credit facilities are now deemed issued under the Credit Agreement. BWICO is the initial borrower under the Credit Agreement. Upon completion of the spin-off, B&W will become the borrower under the Credit Agreement and BWICO, as a subsidiary of B&W, will become a guarantor under the Credit Agreement.

The Credit Agreement provides for revolving credit borrowings and issuances of letters of credit in an aggregate outstanding amount of up to \$700.0 million, and the credit facility is scheduled to mature on May 3, 2014. The proceeds of the Credit Agreement are available for working capital needs and other general corporate purposes. As of May 3, 2010, BWICO had no outstanding borrowings and approximately \$255.3 million of outstanding letters of credit under the Credit Agreement, leaving BWICO with approximately \$444.7 million available capacity for additional borrowings and letters of credit under the Credit Agreement.

The Credit Agreement includes procedures for additional financial institutions to become lenders, or for any existing lender to increase its commitment thereunder, subject to an aggregate maximum of \$850.0 million for all revolving loan and letter of credit commitments under the Credit Agreement.

BWICO's obligations under the Credit Agreement are guaranteed by substantially all of BWICO's wholly owned domestic subsidiaries. Upon completion of the spin-off, substantially all of B&W's wholly owned domestic subsidiaries that are not already guarantors under the BWICO Credit Agreement will become guarantors. Obligations under the Credit Agreement are secured by first-priority liens on certain assets owned by BWICO and the guarantors (other than BWXT and its subsidiaries). Upon completion of the spin-off, B&W and its wholly owned domestic subsidiaries that become guarantors under the Credit Agreement will grant liens on certain assets owned by them. If the corporate rating of B&W and its subsidiaries from Moody's is Baa3 or better (with a stable outlook or better), the corporate family rating of B&W and its subsidiaries from S&P is BBB- or better (with a stable outlook or better), and certain other conditions are met, the liens securing obligations under the Credit Agreement will be released, subject to reinstatement upon the terms set forth in the Credit Agreement.

The Credit Agreement requires only interest payments on a quarterly basis until maturity. The borrower under the Credit Agreement may prepay all loans under the Credit Agreement at any time without premium or penalty (other than customary LIBOR breakage costs), subject to certain notice requirements.

Loans outstanding under the Credit Agreement bear interest at the borrower's option at either the Eurodollar rate plus a margin ranging from 2.50% to 3.50% per year or the base rate (the highest of the Federal Funds rate plus 0.50%, the 30-day Eurodollar rate plus 1.0%, or the administrative agent's prime rate) plus a margin ranging from 1.50% to 2.50% per year. The applicable margin for revolving loans varies depending on the credit ratings of the Credit Agreement. The borrower under the Credit Agreement is charged a commitment fee on the unused portions of the Credit Agreement, and that fee varies between 0.375% and 0.625% per year depending on the credit ratings of the Credit Agreement. Additionally, the borrower under the Credit Agreement is charged a letter of credit fee of between 2.50% and 3.50% per year with respect to the amount of each financial letter of credit issued under the Credit Agreement, and a letter of credit fee of between 1.25% and 1.75% per year with respect to the amount of each performance letter of credit issued under the Credit Agreement, in each case depending on the credit ratings of the Credit Agreement. The borrower under the Credit Agreement also pays customary issuance fees and other fees and expenses in connection with the issuance of letters of credit under the Credit Agreement.

In connection with entering into the Credit Agreement, BWICO paid certain upfront fees to the lenders thereunder, and BWICO paid certain arrangement and other fees to the arrangers and agents of the Credit Agreement.

Based on the current credit ratings of the Credit Agreement, the applicable margin for Eurodollar-rate loans is 2.50%, the applicable margin for base-rate loans is 1.50%, the letter of credit fee for financial letters of credit is 2.50%, the letter of credit fee for performance letters of credit is 1.25%, and the commitment fee for unused portions of the Credit Agreement is 0.375%. The Credit Agreement does not have a floor for the base rate or the Eurodollar rate.

The Credit Agreement includes financial covenants that will be tested on a quarterly basis, based on the rolling four-quarter period that ends on the last day of each fiscal quarter, commencing with the fiscal quarter ending June 30, 2010. The maximum permitted leverage ratio (as defined in the Credit Agreement) is 2.50 to 1.00. The minimum consolidated interest coverage ratio (as defined in the Credit Agreement) is 4.00 to 1.00.

In addition, the Credit Agreement contains various covenants that, among other restrictions, limit BWICO's and its subsidiaries' (and after the spin-off, B&W's and its subsidiaries') ability to:

- · incur or assume indebtedness;
- grant or assume liens;
- make acquisitions or engage in mergers;
- · sell, transfer, assign or convey assets;
- make investments;
- · repurchase equity and make dividends and certain other restricted payments;
- · change the nature of its business;
- · engage in transactions with affiliates;
- enter into certain burdensome agreements;
- · enter into speculative hedging contracts;
- enter into sale and leaseback transactions;
- · make capital expenditures; and
- prior to the completion of the spin-off, change certain terms of the spin-off described in the Credit Agreement.

The Credit Agreement generally includes customary events of default for a secured credit facility. If an event of default relating to bankruptcy or other insolvency events with respect to the borrower occurs under the Credit Agreement, all obligations under the Credit Agreement will immediately become due and payable. If any other event of default exists under the Credit Agreement, the lenders may accelerate the maturity of the obligations outstanding under the Credit Agreement and exercise other rights and remedies. In addition, if any event of default exists under the Credit Agreement, the lenders may commence foreclosure or other actions against the collateral under the Credit Agreement.

If any default occurs under the Credit Agreement, or if the borrower is unable to make any of the representations and warranties in the Credit Agreement, the borrower will be unable to borrow funds or have letters of credit issued under the Credit Agreement.

Power Generation Systems

Bank Guarantees (Foreign Operations)

Certain foreign subsidiaries of B&W PGG had credit arrangements with various commercial banks for the issuance of bank guarantees. The aggregate value of all such bank guarantees as of March 31, 2010 was \$16.8 million.

Surety Bonds

MII, B&W PGG and another subsidiary of ours have jointly executed general agreements of indemnity in favor of various surety underwriters relating to surety bonds those underwriters issued in support of B&W PGG's contracting activity. As of March 31, 2010, bonds issued under such arrangements totaled approximately \$112.9 million. Any claim successfully asserted against any such surety by one or more of the bond obligees would likely be recoverable from MII, B&W PGG and our other subsidiary under such indemnity agreement (and we have agreed to indemnify MII from and against any such recovery from it). Due to events that affect the insurance and bonding and credit markets generally, bonding and letters of credit may be more difficult to obtain in the future or may only be available at significant additional cost. There can be no assurance that letters of credit or bonds will continue to be available to us on reasonable terms, particularly given our pro forma tangible net worth as of March 31, 2010.

Based on the liquidity position of our Power Generation Systems segment, we believe this segment has sufficient cash and letter of credit and borrowing capacity to fund its operating requirements for at least the next 12 months.

Warranty Claim (Power Generation Systems Segment)

One of our Canadian subsidiaries has received notice of a warranty claim on one of its projects on a contract executed in 1998. This situation relates to a condition referred to as "tube fretting," which results in tube wear as a result of relative movement of the tube at the point of contact with the tube support plates within a nuclear steam generator. Data collection and analysis can only be performed at specific time periods when the power plant is scheduled to be off-line for maintenance. We also received a notice from the customer during October 2008, and, during November 2008, we responded to the notice by disagreeing with the matters stated in the claim and disputing the claim. This project included a limited-term performance bond totaling approximately \$140 million for which we entered into an indemnity arrangement with the surety underwriters. It is possible that our subsidiary may incur warranty costs in excess of amounts provided for as of March 31, 2010. It is also possible that a claim could be initiated by our subsidiary's customer against the surety underwriter should certain events occur. If such a claim were successful, the surety could seek to recover from our subsidiary the costs incurred in satisfying the customer claim. If the surety seeks recovery from our subsidiary, we believe that our subsidiary would have adequate liquidity to satisfy its obligations. However, the ultimate resolution of this possible claim is uncertain, and an adverse outcome could have a material adverse impact on our combined financial condition, results of operations or cash flows.

Government Operations

Letters of Credit (Nuclear Fuel Services, Inc.)

At March 31, 2010, Nuclear Fuel Services, Inc., a subsidiary of BWXT, had \$3.6 million in letters of credit issued by various commercial banks on its behalf. The obligations to the commercial banks issuing such letters of credit are secured by cash, short-term certificates of deposit and certain real and intangible assets.

Based on the liquidity position of our Government Operations segment, we believe this segment has sufficient cash and letter of credit and borrowing capacity to fund its operating requirements for at least the next 12 months.

Other

Pension Plan

We recorded a \$36.9 million reduction in parent equity at December 31, 2009, compared to a reduction of \$325.8 million in parent equity at December 31, 2008. The substantial reduction at December 31, 2008 was primarily attributable to the volatility of the stock market in 2008. While the performance of our pension assets improved in 2009, the reduction in the discount rate for our major domestic plans from 6.25% to 6.00% increased our pension plan obligations, resulting in a net reduction in equity at December 31, 2009. As of December 31, 2009, our defined benefit pension and post-retirement benefit plans were underfunded by approximately \$869 million. See Note 7 to the combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement. A significant portion of those obligations is subject to cost recovery from the U.S. Government under contracts relating to operations that we have been involved in for many years. While we do not consider it likely, if those contracts with the U.S. Government were terminated, the amount and timing of our recovery for pension and other post-retirement benefit plans could be impacted. For a general discussion of our contracts with the U.S. Government, see "Business—Contracts."

Cash, Cash Equivalents and Investments

At March 31, 2010, we had restricted cash and cash equivalents totaling \$17.8 million, \$9.7 million of which was held in restricted foreign accounts, \$3.8 million of which was held as cash collateral for letters of credit, \$4.0 million of which was held for future decommissioning of facilities and \$0.3 million of which was held to meet reinsurance reserve requirements of our captive insurance companies. It is possible that a significant portion of restricted cash at March 31, 2010 will not be released within the next 12 months.

At March 31, 2010, we had investments with a fair value of \$70.9 million. Our investment portfolio consists primarily of investments in government obligations and other highly liquid money market instruments.

March 31, 2010 Compared to December 31, 2009

In aggregate, our cash and cash equivalents, restricted cash and cash equivalents and investments decreased by \$68.4 million to \$489.9 million at March 31, 2010 from \$558.3 million at December 31, 2009, primarily due to cash used in operating activities and purchases of property, plant and equipment.

Our working capital, excluding cash and cash equivalents and restricted cash and cash equivalents, decreased by \$57.2 million to a negative \$470.4 million at March 31, 2010 from a negative \$413.2 million at December 31, 2009, primarily due to the increase in the net amount of contracts in progress and advance billings on contracts.

Three Months Ended March 31, 2010 Compared to Three Months Ended March 31, 2009

Our net cash used in operations was \$45.3 million in the three months ended March 31, 2010, compared to cash provided by operations of \$40.5 million for the three months ended March 31, 2009. This change was primarily attributable to a decrease in our net contracts in progress and advance billings on contracts and our accounts payable position.

Our net cash used in investing activities increased by \$25.8 million to \$26.8 million in the three months ended March 31, 2010 from \$1.0 million in the corresponding period of 2009. This increase in net cash used in investing activities was primarily attributable to acquisitions in the three months ended March 31, 2010 and a decrease in proceeds from the sale of available securities in the three months ended March 31, 2010 compared to the corresponding period of 2009.

Our net cash provided by financing activities increased by \$6.5 million to \$1.8 million in the three months ended March 31, 2010 from \$4.7 million used in financing activities in the three months ended March 31, 2009.

This change was primarily attributable to payments on long-term debt made in the three months ended March 31, 2009 and higher excess tax benefits related to stock-based compensation in the three months ended March 31, 2010.

December 31, 2009 Compared to December 31, 2008

In aggregate, our cash and cash equivalents, restricted cash and cash equivalents and investments increased by approximately \$152.5 million to \$558.3 million at December 31, 2009 from \$405.8 million at December 31, 2008, primarily due to (1) cash provided by operations related primarily to our improvements in accounts receivables and reduced funding requirements under our pension plans, and (2) proceeds from the sale of available for sale securities.

Our working capital, excluding cash and cash equivalents and restricted cash and cash equivalents, decreased by approximately \$36.4 million to a negative \$413.2 million at December 31, 2009 from a negative \$376.8 million at December 31, 2008, primarily due to increases in accrued employee benefits.

Year Ended December 31, 2009 Compared to Year Ended December 31, 2008

Our net cash provided by operating activities was approximately \$252.8 million in the year ended December 31, 2009, compared to approximately \$60.1 million in the year ended December 31, 2008. This increase was primarily attributable to improvements in our accounts receivable position and reduced funding requirements under our pension plans in 2009 compared to 2008.

Our net cash used in investing activities decreased by approximately \$237.0 million to approximately \$67.6 million in the year ended December 31, 2009 from approximately \$304.6 million in the year ended December 31, 2008. This decrease in net cash used in investing activities was primarily attributable to higher funding requirements for acquisitions of businesses in 2008 compared to 2009, and an increase in proceeds from the sale of available-for-sale securities in 2009.

Our net cash provided by (used in) financing activities changed by approximately \$65.6 million to net cash used in financing activities of \$5.6 million in the year ended December 31, 2009 from net cash provided by financing activities of \$60.0 million in the year ended December 31, 2008, primarily due to an increase in borrowings in 2008 on a note payable to an affiliate (see Note 5 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement).

Contractual Obligations

Our cash requirements as of December 31, 2009 under current contractual obligations are as follows:

| | Total | than 1 Year | 1-3 Years | 3-5 Years | After 5 Years |
|--------------------------|-----------|----------------|----------------|--------------|------------------|
| | Total | Itai | (In thousands) | | 3 I cars |
| Long-term debt principal | \$ 7,588 | \$ 3,366 | \$ 4,222 | _ | _ |
| Operating leases | \$ 14,821 | \$ 5,577 | \$6,916 | \$ 2,328 | _ |

We have excluded notes payable to affiliates aggregating approximately \$320.5 million from the table above as those notes will be extinguished or settled in connection with the spin-off. In addition, we have excluded \$3.1 million of short-term line of credit borrowings made by one of our subsidiaries, as there is no stated contractual payment date. We have included this amount in notes payable and current maturities of long-term debt on our combined balance sheets. This facility is renewable annually and the interest rate associated with this line of credit was 5.3% per annum at December 31, 2009.

In addition to the contractual obligations summarized above, on May 25, 2010, we entered into a securities purchase agreement with USEC Inc. ("USEC") and Toshiba Corporation ("Toshiba"), pursuant to which we and

Toshiba have each agreed to make a phased strategic investment in USEC of up to \$100 million. Our investment is to be made over three phases, each of which is subject to satisfaction of specific preconditions, including regulatory approvals. USEC is a leading supplier of enriched uranium for commercial nuclear power plants. USEC will use the proceeds from our investment for general corporate purposes and for investment in its American Centrifuge Project, a next-generation uranium enrichment facility to be built in Piketon, Ohio. The project is based on the use of highly efficient centrifuges to produce enriched uranium. USEC is currently operating a group of centrifuge machines in a commercial plant configuration, collecting data and experience in preparation for transitioning to full commercial operation. We provide significant products and services to USEC in connection with the American Centrifuge Project, and we have entered into a memorandum of understanding to form a joint venture with USEC to manufacture and assemble the centrifuge machine for the project. Each of the three phases of investment is contingent on USEC meeting specified milestones in arranging and obtaining at least \$2 billion in funding under the DOE's loan guarantee program for front-end nuclear facilities. We anticipate making our first investment, in the amount of \$37.5 million, in the third quarter of 2010, which we intend to fund using cash on hand or cash generated by our operating activities.

Our contingent commitments under letters of credit and bank guarantees currently outstanding expire as follows:

| | Less than | 1-3 | 3-5 | After |
|-----------|------------|----------------|-----------|---------|
| Total | 1 Year | Years | Years | 5 Years |
| | <u> </u> | (In thousands) | <u></u> - | |
| \$377,004 | \$ 231,247 | \$ 131,439 | \$ 14,128 | \$ 190 |

We expect cash requirements totaling approximately \$58 million for contributions to our pension plans and \$15 million for contributions to our other postretirement benefit plans in 2010.

In accordance with the provisions of FASB Topic *Income Taxes*, we have recorded a \$15.5 million liability as of December 31, 2009 for unrecognized tax benefits and the payment of related interest and penalties. Due to the uncertainties related to these tax matters, we are unable to make a reasonably reliable estimate as to when cash settlement with a taxing authority will occur. However, we do not anticipate making any cash payments on these liabilities over the next 12 months.

QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Our exposure to market risk from changes in interest rates relates primarily to our cash equivalents and our investment portfolio, which primarily consists of investments in U.S. Government obligations and highly liquid money market instruments denominated in U.S. dollars. We are averse to principal loss and seek to ensure the safety and preservation of our invested funds by limiting default risk, market risk and reinvestment risk. All our investments in debt securities are classified as available-for-sale.

We have exposure to changes in interest rates applicable to borrowings under our revolving credit facility (see "Management's Discussion and Analysis of Financial Condition and Results of Operations—Liquidity and Capital Resources"). We have no material future earnings or cash flow exposures from changes in interest rates on our other long-term debt obligations, as substantially all of these obligations have fixed interest rates.

We have operations in foreign locations, and, as a result, our financial results could be affected by factors such as changes in foreign currency exchange rates or weak economic conditions in those foreign markets. In order to manage the risks associated with foreign currency exchange rate fluctuations, we attempt to hedge those risks with foreign currency derivative instruments. Historically, we have hedged those risks with foreign currency forward contracts. We have recently hedged some of those risks with foreign currency option contracts. We do not enter into speculative derivative positions.

Interest Rate Sensitivity

The following tables provide information about our financial instruments that are sensitive to changes in interest rates. The tables present principal cash flows and related weighted-average interest rates by expected maturity dates.

| | Principal Amount by Expected Maturity | | | | | | | |
|---------------------------|---------------------------------------|----------|-------------|-----------|-----------|------------|-----------|-------------------|
| | | | | (Dollars | in thousa | nds) | | |
| At December 31, 2009: | | | Years Endin | g Decembe | r 31, | | | Fair Value at |
| | 2010 | 2011 | 2012 | 2013 | 2014 | Thereafter | Total | December 31, 2009 |
| Investments | \$ 70,169 | \$ 2,877 | _ | _ | _ | \$ 883 | \$ 73,929 | \$ 73,554 |
| Average Interest Rate | 0.52% | 0.53% | | | | 2.22% | | |
| Long-term Debt—Fixed Rate | \$ 3,366 | \$ 4,222 | _ | _ | _ | _ | \$ 7,588 | \$ 7,588 |
| Average Interest Rate | 1.06% | 0.75% | | | | | | |
| At December 31, 2008: | | | Years Endin | g Decembe | r 31. | | | Fair Value at |
| 11 December 51, 2000 | 2009 | 2010 | 2011 | 2012 | 2013 | Thereafter | Total | December 31, 2008 |
| Investments | \$101,839 | \$15,324 | \$ 825 | _ | _ | \$ 1,162 | \$119,150 | \$ 119,084 |
| Average Interest Rate | 2.20% | 2.72% | 4.98% | _ | _ | 2.78% | | |
| Long-term Debt—Fixed Rate | \$ 4,250 | _ | _ | _ | _ | _ | \$ 4,250 | \$ 4,250 |
| Average Interest Rate | 6.80% | | | | | | | |

Exchange Rate Sensitivity

The following table provides information about our foreign currency forward contracts and options outstanding at December 31, 2009 and presents such information in U.S. dollar equivalents. The table presents notional amounts and related weighted-average exchange rates by expected (contractual) maturity dates and constitutes a forward-looking statement. These notional amounts generally are used to calculate the contractual payments to be exchanged under the contract. The average contractual exchange rates are expressed using market convention, which is dependent on the currencies being bought and sold under the forward contract.

Danish Krone

Forward Contracts to Purchase Foreign Currencies in U.S. Dollars (Dollars in thousands)

| | Year Ending | | Fair Value at | | Average Contractual | |
|---|-------------|---------------|-------------------|--------------|---------------------|--|
| Foreign Currency | Decer | mber 31, 2010 | December 31, 2009 | | Exchange Rate | |
| Canadian Dollars | \$ | 62,324 | \$ | (1,892) | 1.0239 | |
| Pound Sterling (selling Euros) | \$ | 5,976 | \$ | (53) | 0.8861 | |
| Japanese Yen (selling Canadian Dollars) | \$ | 5,262 | \$ | 758 | 101.8480 | |
| Pound Sterling (selling Canadian Dollars) | \$ | 1,181 | \$ | (31) | 1.7440 | |
| Thai Baht | \$ | 901 | \$ | 9 | 33.7944 | |
| Euros | \$ | 328 | \$ | (15) | 1.3650 | |
| | Ye | ear Ending | Fair | Value at | Average Contractual | |
| Foreign Currency | Decer | mber 31, 2011 | Decemb | ber 31, 2009 | Exchange Rate | |
| Canadian Dollars | \$ | 34,612 | \$ | (1,102) | 1.0139 | |
| Japanese Yen (selling Canadian Dollars) | \$ | 5,354 | \$ | 622 | 98.3035 | |
| Euros | \$ | 3,198 | \$ | (63) | 1.4633 | |
| | | | | | | |

Foreign Currency Option Contracts to Purchase Foreign Currencies in U.S. Dollars (Dollars in thousands)

\$

1,252

(1)

5.2195

| | Ye | ar Ending | Fair V | /alue at | |
|------------------|-------|---------------|---------|-------------|-------------|
| Foreign Currency | Decer | nber 31, 2010 | Decembe | er 31, 2009 | Strike Rate |
| Canadian Dollars | \$ | 95,250 | \$ | 4,366 | 1.0637 |
| Japanese Yen | \$ | 12,930 | \$ | 307 | 90.3500 |
| Euros | \$ | 2,405 | \$ | 74 | 1.4343 |

BUSINESS

Overview

B&W is currently a wholly owned subsidiary of McDermott. B&W is a successor to a business founded in 1867, which was acquired by McDermott in 1978. Following the spin-off, B&W will be an independent, publicly traded company. McDermott will not retain any ownership interest in B&W. B&W's assets and businesses primarily consist of those that McDermott reports as its Power Generation Systems and Government Operations segments in its financial statements. We participate in the ownership of a variety of entities with third parties, primarily through corporations, limited liability companies and partnerships, which we refer to as "joint ventures."

B&W is a leading technology innovator in power generation systems, a specialty manufacturer of nuclear components and a premier service provider in its segments, with an operating history of more than 140 years. We provide a variety of products and services to customers in the power and other steam-using industries, including electric utilities and other power generators, industrial customers in various other industries, and the U.S. Government. We operate in two business segments: Power Generation Systems and Government Operations. Through our Power Generation Systems segment, we design, engineer, manufacture, supply, construct and maintain power generation systems and environmental control systems, primarily for large utility and industrial customers, and we provide related aftermarket parts and services. Through our Government Operations segment, we manufacture nuclear components and provide various services to the U.S. Government, including uranium processing, environmental site restoration services, and management and operating services for various U.S. Government-owned facilities. These services are provided to the DOE, including the NNSA, the Office of Nuclear Energy, the Office of Science and the Office of Environmental Management.

Our Government Operations segment manages and operates nuclear facilities and associated plant infrastructure; manufactures components and assembles/dismantles nuclear devices; constructs large capital facilities; provides safeguards and security for inventory and assets; supports and conducts research and development for advanced energy technology; and manages the environmental programs for the DOE and the NNSA through joint ventures. We employ approximately 13,000 people worldwide, not including approximately 10,000 joint venture employees.

The main drivers for our businesses are:

- · demand for electricity, paper and other end products of steam-generating facilities;
- fossil and nuclear plant life cycles and the frequency with which plants need to replace or upgrade components;
- requirements for environmental improvements, including those imposed by legislative and regulatory developments aimed at reducing emissions of mercury, nitrogen oxides, sulfur dioxide, particulate matter, carbon dioxide and other environmental pollutants from power plants;
- · requirements for plant efficiency improvements and improvements in reliability;
- levels of funding for the dismantlement and life extension for nuclear/non-nuclear components, research and development of energy technologies, and environmental restoration services we provide to the DOE, other federal agencies and commercial clients; and
- · levels of demand for other products and services to the U.S. Government.

For the year ended December 31, 2009, we generated revenues of approximately \$2.9 billion and operating income of approximately \$270 million. For the year ended December 31, 2008, we generated revenues of approximately \$3.4 billion and operating income of approximately \$444.5 million. For financial information about our business segments, see Notes 1 and 14 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement, which presents revenue, operating income, depreciation and amortization expense and capital expenditures for the years ended December 31, 2009, 2008 and 2007 and identifiable assets and goodwill by business segment as of December 31, 2009 and 2008.

In February 2000, several of B&W's subsidiaries filed for voluntary Chapter 11 protection, due to mounting asbestos-related claims tied to asbestos-containing commercial and utility boilers and other products they sold, installed or serviced in prior decades. From the date of the filing of the Chapter 11 petition until their emergence from Chapter 11 in 2006, McDermott treated those subsidiaries as deconsolidated subsidiaries for financial reporting purposes. The final settlement with representatives of asbestos claimants in the Chapter 11 proceedings, which was effected in a plan of reorganization consummated as of February 22, 2006, involved McDermott's and B&W's contribution of rights under various insurance policies, cash and a promissory note, with an aggregate value of approximately \$2 billion, to a settlement trust, the channeling of asbestos-related personal injury claims to that trust and McDermott's retention of the ownership of the B&W subsidiaries involved in the Chapter 11 proceedings.

Our Competitive Strengths

Leading Market Positions

We are a proven leader in the design, engineering, manufacture, supply, construction and maintenance of steam generating and environmental equipment for power providers in the United States and in other parts of the world. As of March 31, 2010, our power generation systems and equipment supplied more than 300,000 MW of installed capacity in more than 800 utilities in over 90 countries. This extensive experience with steam generating equipment has helped us establish a leading position in the replacement nuclear steam generator market in the United States and worldwide, including replacement of units originally manufactured by other suppliers. We manufacture nuclear and non-nuclear components; manage, operate and maintain large government-owned facilities; and deliver major capital construction and environmental projects for the DOE and NNSA. We have received a number of industry awards recognizing our leadership position, technical innovation and customer service, including the National Safety Council's Occupational Excellence Achievement Award (Y-12), the 2009 National Safety Council Industry Leader Award (Y-12), DOE/NNSA Pollution Prevention (P2) Award (Pantex and Y-12), Closing the Circle Awards (recognizes federal facilities and employees for innovative practices and programs that have improved environmental performance and conditions at federal sites), Malcolm Baldrige National Quality Award given by the President of the United States (DynMcDermott for management and operation of the Strategic Petroleum Reserve), and Occupational Hazards Magazine Award (Pantex named to the 2007 list of America's Safest Companies). For our Power Generation Systems operations, Power Engineering magazine named Wisconsin Public Service Co.'s new Weston 4 project the "Best Coal-Fired Project of the Year for 2008." The boiler and air quality control systems for this 530 MW supercritical unit were designed, supplied and erected by our Power Generation Systems segment.

Investments in Advanced Technologies

We have invested in advanced technologies, through both research and development and our knowledge-centered technical professionals, to meet the challenge of growing power generation demand for energy sources that are lower pollutant emitting sources, or sources that have lower carbon footprints, fitting broadly into the category of "clean energy." Our "clean energy" technologies under development include carbon capture systems for coal-fired power plants; technologies for the development of higher efficiency coal-utilization systems to reduce emissions; technology to enable a greater use of biomass fuels for power generation; nuclear energy; and solar power generation technology.

We have built a new research center, rehabilitated a 30 MW pilot plant for oxy-combustion technology testing and built a new pilot plant for the development of solvent-based carbon dioxide capture systems for retrofit applications to existing power plants. One carbon capture technology we are developing with a third party, oxy-combustion, has been advanced through laboratory and pilot plant testing. We have applied for funding from the DOE to move this technology into a scale-plant phase. We intend to begin extensive field tests on our solvent-based carbon dioxide capture systems late in 2010 or early in 2011. We have also established a bubbling fluidized bed combustion technology for use in converting biomass fuels to thermal energy in accordance with environmental performance criteria. B&W has installed an advanced solar receiver in California

at a client-owned, concentrating solar power testing site to further validate this new technology on a commercial scale. This solar thermal power plant technology uses small, flat mirrors that track the sun and reflect the sun's heat to B&W's receiver, which boils water to create steam. This steam powers a traditional turbine and generator to produce clean electricity.

We are currently pursuing the development of several nuclear technologies with programs that include the development of advanced nuclear fuel for high temperature gas-cooled reactors; the supply of reactor fuels for university and other research programs; the development of aqueous reactor technology to support the production of medical isotopes for use in nuclear medicine imaging tests; and the design and development of the B&W mPowerTM reactor, a modular, scalable nuclear steam supply system.

Proven Track Record

We have served the global power industry for over 140 years. We have a rich history and a legacy of "industry firsts." We designed and installed the first utility boiler, and we developed the first supercritical boiler. We also have a history of developing new products to meet changing regulations and mandates. We designed and built the first limestone-based wet scrubber for sulfur dioxide control. In addition to our history of leadership in the fossil-fuel-fired boiler market, we have been involved in the field of nuclear power for more than 50 years, providing nuclear heat exchangers, nuclear plant services and more than 300 nuclear steam generators to customers around the world. We manufactured the reactor for the first nuclear ship, the NS *Savannah*. We have also been supplying nuclear power components to the U.S. Government since the 1955-launching of the U.S.S. *Nautilus*, the world's first nuclear-powered vessel.

High Quality Manufacturing Facilities

We operate a network of 13 major manufacturing facilities around the world. Our Government Operations segment's facilities are equipped with advanced, state-of-the-art equipment and capable of performing highly complex manufacturing operations. These facilities include the only two commercial facilities that have Category I licenses from the NRC, which enable us to own, store and process both high and low enriched uranium. We have performed work under the auspices of the International Atomic Energy Agency. Our Power Generation Systems segment has factories in North America, Europe and Asia equipped to provide products and services to power plants. Three of our manufacturing facilities are accredited with the American Society of Mechanical Engineers Nuclear Component Certification, or "ASME N-Stamp." Only a small number of North American suppliers of large, heavy walled nuclear components and vessels have this certification.

Demonstrated Success with Large Delivered and Erected Projects

We have demonstrated success in executing large, delivered and erected projects, both at new power plants and as retrofits at existing facilities. We specialize in heavy mechanical erection and maintenance of equipment for steam generation and environmental improvement. Since 2000, we have erected more than 50 major pieces of equipment at units with electric generating capacity exceeding 39,000 MW. For those customers who purchase our combined design-build offering, we utilize proven project execution processes and provide single-point accountability for our scope throughout the execution of the project. Leveraging our engineer-to-construct design experience early in the life of a project optimizes constructability, resulting in lower costs and improved certainty of schedule. Our successful experience with modular fabrication and assembly can increase safety, improve quality, reduce field labor costs and shorten project span time.

Diversified and Strong Client Base

We have a broad customer base, consisting of a wide range of utilities, independent power producers and businesses in various other industries around the world, as well as the U.S. Government, our largest customer. In 2009, in addition to the U.S. Government, other major customers included:

- · Allegheny Energy, Inc.
- · American Electric Power Company, Inc.
- · Bruce Power, L.P.

- · FirstEnergy Corp.
- · Public Service Company of New Mexico
- · United States Enrichment Corporation

In 2009, only two customers represented, individually, over five percent of our combined revenue.

Experienced Management Team

Our management team boasts a wealth of industry experience and a history of working together as a cohesive team. This substantial experience has allowed our management to develop the knowledge and skills necessary to manage highly complex manufacturing and project management operations.

Highly Qualified Employees

We have a large number of scientists, engineers, technicians and other highly skilled employees involved in all aspects of our business operations. A large percentage of our salaried employees have college degrees in technical areas. In our Government Operations segment, a substantial number of our employees have security clearances from the NRC, the DOE and/or the U.S. Department of Defense, and have developed the experience necessary to meet the challenges associated with handling and processing highly enriched nuclear material. We provide regular, ongoing training to our employees, and we believe that maintaining high skill levels is a source of pride for our employees.

Our Business Strategies

Pursue Growth Opportunities in the Conventional Power Generation Market

We intend to pursue growth opportunities in the fossil power generation market and to expand our operational presence in regions around the world where we expect to see continuing growth and increasing needs for service. We believe significant growth opportunities exist as a result of many factors, including:

- economic growth, particularly in emerging economies, where we expect demand for new power generation capacity to continue to rise as economic recovery continues;
- continued development of environmental regulation and concern, which we expect to continue to drive demand for cleaner power generation technologies, as well as environmental control systems and other devices that abate emissions from existing plants;
- increasing demand for refurbishment and maintenance of aging plants, which, in addition to the requirements of increasing environmental regulation, is being driven by the desire of power producers to reduce operating costs, enhance operating efficiency, improve reliability and extend the useful lives of their existing plants:
- the continuing effects of past deregulation and liberalization initiatives relating to electricity markets in the United States, which has led to increased competition, which, in turn, has driven demand for installation of generation equipment designed to increase efficiency and reduce costs, as well as to make plants more environmentally friendly; and
- the need for more complete solutions for owners of power facilities, including engineer-procure-construct commercial arrangements and operations and maintenance services.

Leverage our Experience and Knowledge in the Nuclear Power Market to Pursue Opportunities from Renewed Interest in Nuclear Technology

Since the 1950s, we have supplied the nuclear industry with more than 1,300 large heavy components worldwide. With ASME N-Stamp certified manufacturing operations in both the U.S. and Canada, we have the skills, knowledge base and infrastructure for engineering and fabricating pressure vessels, reactors, steam generators, heat exchangers and electro-mechanical devices. We intend to leverage our extensive experience in designing, engineering, manufacturing, constructing and servicing nuclear power equipment to pursue growth opportunities in the market for nuclear technology. Technical and safety improvements related to nuclear reactors have resulted in a renewed interest in nuclear technology as a substitute for fossil and other fuels. We believe this nuclear technology "renaissance" also stems from the recognition that nuclear power is likely to play an important role in establishing global energy security in the 21st century. Public opinion, in the United States and elsewhere, regarding the merits of nuclear energy production, has improved over the past decade. This improvement has been attributable to, among other things, advances in nuclear reactor design to improve operational safety and reliability.

Among other opportunities in the nuclear power industry, we are developing the B&W mPowerTM reactor, a modular nuclear reactor designed to provide scalability to build plants ranging from 125 MW to 1,000 MW of electric generating capacity (in increments of 125 MW) for a four- to five-year operating cycle without refueling. We intend to seek NRC certification of the B&W mPowerTM reactor in time to begin deploying this technology as early as 2020.

We are developing, in conjunction with a unit of Covidien PLC and the DOE, a homogenous aqueous phase reactor to supply medical isotopes for use in nuclear medicine imaging tests. The reactor uses low-enriched fuel to address the U.S. Government nuclear non-proliferation mandate associated with the current medical isotope supply chain. We are currently in the licensing phase with the NRC and the U.S. Food and Drug Administration and have received grant funding from the DOE to accelerate the project.

Continue to Serve the Nuclear Power Needs of the U.S. Government

For more than five decades, we have established a solid reputation as a trusted supplier to U.S. Government defense programs. In support of these programs, we intend to continue processing special nuclear materials and providing precision manufactured nuclear components.

Continue to Diversify our Business Portfolio into Clean Energy Technologies

We intend to continue to diversify our business portfolio into clean energy technologies by investing in research and development of clean energy technologies, including the advancement of those technologies we already have under development and the development of new technologies. Development of B&W's clean energy portfolio is focused on modular nuclear reactors, carbon capture from fossil combustion, biomass and energy-from-waste plants and concentrated solar plants. Commercialization to clean energy solutions involves active B&W initiatives ranging from continuing research, to field deployment to new commercial offerings in our backlog.

Continue Leveraging our Successes in Managing and Operating DOE Sites to Grow our Business within the DOE Complex and in Other Markets

Over the past six decades, the DOE has developed one of the largest government-owned operations in the United States, with responsibility for research, development, testing, operations and production of nuclear devices and a variety of nuclear-related research programs. We have been actively engaged in assisting the DOE in these activities, principally through various management and operations services as well as highly complicated decontamination and decommissioning activities at operational and former weapons production sites. We believe that our core competencies and our experience, expertise and longstanding relationships with customers, including

the NNSA, the Office of Nuclear Energy, the Office of Science and the Office of Environmental Management, position us to grow our business within the DOE complex. Additionally, although we have not been successful in recent years in our efforts to expand our management and operations activities outside the DOE complex, we believe this expertise should provide a platform to seek contracts from other U.S. and foreign government agencies to manage and operate similar facilities and production operations and to support specialized and classified component manufacturing, facility and infrastructure improvement and environmental restoration projects.

Maintain our Commitment to Safety

We are committed to promoting a safe workplace. We value the health and safety of each employee, and we seek to provide each of our employees with a workplace that is free of accidents and injuries. Through our Target Zero safety process, we are dedicated to preventing accidents and their associated costs by averting, eliminating or mitigating unsafe acts and conditions, and by responding properly to natural disasters and emergency situations. We are implementing a Human Performance Initiative to provide our employees with an additional set of tools and improvements designed to reduce human error and improve workplace safety. We believe that this initiative will promote behaviors throughout our organization that support safe and reliable operations. A safe work environment benefits our employees, our customers and our business partners and is a key element to our future success.

Drive Operational Excellence to Continually Support Profitable Growth

We use a continuous improvement process methodology in our operations designed to enhance productivity and increase customer satisfaction through industry-proven methodologies.

Selectively Pursue Acquisitions

Our management team has demonstrated its ability to identify complementary business acquisition opportunities, consummate acquisitions on favorable terms and integrate acquired businesses. We will continue to evaluate potential acquisitions, with the aim of increasing earnings while maintaining financial discipline.

Maintain Financial Flexibility

We intend to maintain financial flexibility through our access to various liquidity sources. As of the distribution date, we anticipate having access to liquidity from several sources, including our revolving credit facility with a borrowing and letter of credit capacity of up to \$700 million.

Power Generation Systems

Our Power Generation Systems segment includes the business and operations of Babcock & Wilcox Power Generation Group, Inc. ("B&W PGG") and Babcock & Wilcox Nuclear Energy, Inc. and their respective subsidiaries. Through this segment, we supply boilers fired with fossil fuels, such as coal, oil and natural gas, or renewable fuels such as biomass, municipal solid waste and concentrated solar energy. In addition, we supply commercial nuclear steam generators and components, environmental equipment and components, and related services to customers in different regions around the world. We design, engineer, manufacture, construct and service large utility and industrial power generation systems, including boilers used to generate steam in electric power plants, pulp and paper making, chemical and process applications and other industrial uses.

Through this segment, we have served the global power industry for over 140 years. We have a rich history and a legacy of "industry firsts." These include:

- the introduction of the first commercial, supercritical pressure steam cycle unit—the universal pressure boiler (1957);
- the manufacture of the reactor for the world's first commercial nuclear vessel, the NSSavannah (1961);

- the receipt of the first order for the design and supply of nuclear steam generators for a Canadian deuterium uranium, or CANDU, plant (1965);
- the installation of the first wet flue gas desulfurization system (1973);
- the introduction of the world's largest dry scrubbing system (1982);
- the design and construction of the world's largest waste-to-energy facility (1989);
- the erection of a universal pressure boiler and wet scrubber at the William H. Zimmer station, which was the first commercial nuclear plant to be converted to a coal-fired plant and the world's largest single-boiler emission control system (1991);
- · the receipt of the first order in the industry for once-through-design replacement nuclear steam generators (1999);
- the completion of a 550 MW boiler replacement in world-record time—less than 22 months—and two large selective catalytic reduction systems for a major Northeastern United States utility (2001); and
- the operation of the first coal-fired 30 MWt oxy-combustion system, to concentrate carbon dioxide, as a step to abate emissions of greenhouse gases (2008).

The following table sets forth the location and principal use of this segment's operating facilities that we own or lease.

| Location | Principal Use | Owned/Leased (Lease Expiration) |
|--------------------------------|---------------|---------------------------------|
| West Point, Mississippi | Manufacturing | Owned |
| Lancaster, Ohio | Manufacturing | Owned |
| Cambridge, Ontario, Canada | Manufacturing | Owned |
| Esbjerg, Denmark | Manufacturing | Owned |
| Jingshan, Hubei, China | Manufacturing | Owned |
| Guadalupe, Mexico | Manufacturing | Leased (through 2024) |
| Melville, Saskatchewan, Canada | Manufacturing | Owned |
| Barberton, Ohio | Manufacturing | Owned |
| Dumbarton, Scotland | Manufacturing | Owned |

We have granted liens on our ownership interests in each of the facilities we own that are located in the United States, to secure our obligations to the lenders under the credit agreement relating to our revolving credit facility. We have also granted liens on the facility in Denmark to secure our obligations to the lenders under a separate credit agreement maintained by our subsidiary that operates that facility.

This segment also has administrative offices and a research and development facility in Barberton, Ohio, administrative offices in Lynchburg, Virginia (leased through 2011) and a warehouse/service center in Copley, Ohio. Our Power Generation Systems segment has access to the Government Operations segment's manufacturing facilities in Barberton, Ohio, Euclid, Ohio and Mount Vernon, Indiana for the manufacture of commercial nuclear components.

Through our manufacturing facilities, we specialize in the fabrication of products used in the power generation industry and various other industries and the provision of related services, including:

- · heavy-pressure equipment for energy conversion, such as boilers fueled by coal, oil, bitumen, natural gas, solid municipal waste, biomass and other fuels;
- environmental control systems, including both wet and dry scrubbers for flue gas desulfurization, modules for selective catalytic reduction of the oxides of nitrogen, equipment to capture particulate matter, such as fabric-filter baghouses and wet and dry electrostatic precipitators, and similar devices;
- engineered-to-order services, products and systems for energy conversion worldwide and related auxiliary equipment, such as burners, pulverizers, soot blowers and ash handling systems;

- · steam generators, heat exchangers, reactor vessel closure heads and electromechanical devices for commercial nuclear power plants; and
- power plant equipment and related heavy mechanical erection services.

We support operating plants with a wide variety of additional services, including the installation of new systems and replacement parts, engineering services, construction, maintenance and field technical services, such as condition assessments and inventory services to help customers respond quickly to plant interruptions. Our wide range of construction and installation services include erection of utility and industrial boiler plants, installations of cogeneration facilities and installations of pollution control equipment, such as selective catalytic reduction systems and flue gas desulfurization scrubbers. We also provide power through cogeneration, refusefueled power plants and other independent power-producing facilities and participate in this market as contractors for engineer-procure-construct services, as equipment suppliers, as operations and maintenance contractors and as an owner.

Although it has been over 30 years since a new nuclear power plant commenced construction in the United States, we expect to participate in commercial nuclear projects and related opportunities in the future. Among our opportunities in the nuclear power industry, we are developing the B&W mPowerTM reactor, a small modular reactor designed with the flexibility to provide between 125 MW to 1,000 MW of electrical power generation (in increments of 125 MW), and the capability to operate for a four- to five-year operating cycle without refueling. We intend to seek NRC certification of the B&W mPowerTM reactor in time to begin deploying this technology as early as 2020. We intend to build the nuclear reactor modules in our manufacturing facilities and transport them to customer sites for installation. The modular and scalable design of the B&W mPowerTM reactor should allow us to match the generation needs of our customers with the proven performance of existing light water reactor technology. We believe the B&W mPowerTM reactor will reduce risks associated with deploying nuclear power and become a flexible, cost-effective solution for increasing electricity needs.

Our Power Generation Systems segment's overall activity depends mainly on the capital expenditures of electric power generating companies and other steam-using industries. Several factors influence these expenditures, including:

- · prices for electricity, along with the cost of production and distribution;
- · prices for coal and natural gas and other sources used to produce electricity;
- the demand for electricity, paper and other end products of steam-generating facilities;
- the availability of other sources of electricity, paper or other end products;
- · requirements for environmental improvements;
- · the impact of potential regional, state, national and/or global requirements to significantly limit or reduce greenhouse gas emissions in the future;
- · levels of capacity utilization at operating power plants, paper mills and other steam-using facilities;
- · requirements for maintenance and upkeep at operating power plants and paper mills to combat the accumulated effects of wear and tear;
- the ability of electric generating companies and other steam users to raise capital; and
- · total costs of electricity production using boilers compared to total costs using gas turbines and other alternative forms of generation.

Our Power Generation Systems segment's products and services are capital intensive. As such, customer demand is heavily affected by the variations in our customers' business cycles and by the overall economies of the countries in which they operate.

Government Operations

Our Government Operations segment includes the business and operations of Babcock & Wilcox Nuclear Operations Group, Inc., Babcock & Wilcox Technical Services Group, Inc. and their respective subsidiaries. Through this segment, we manufacture nuclear components and provide various services to the U.S. Government, including uranium processing, environmental site restoration services, and management and operating services for various U.S. Government-owned facilities. These services are provided to the DOE, including the NNSA, the Office of Nuclear Energy, the Office of Science and the Office of Environmental Management.

We have over 50 years of experience in the ownership and operation of large nuclear development, production and reactor facilities. This segment's principal operations include:

- · providing precision manufactured nuclear components for U.S. Government defense programs;
- · managing and operating nuclear production facilities;
- · managing and operating environmental management sites;
- · managing spent nuclear fuel and transuranic waste for the DOE;
- · providing critical skills and resources for DOE sites; and
- · developing and deploying next generation technology in support of U.S. Government programs.

The following table sets forth the location and principal use of this segment's major properties that we own or lease.

| Location | Principal Use | Owned/Leased (Lease Expiration) |
|-----------------------|---------------|---------------------------------|
| Lynchburg, Virginia | Manufacturing | Owned |
| Barberton, Ohio | Manufacturing | Owned |
| Euclid, Ohio | Manufacturing | Owned / Leased(1) |
| Mount Vernon, Indiana | Manufacturing | Owned |
| Erwin, Tennessee | Manufacturing | Owned |

(1) We acquired the Euclid facilities through a bond/lease transaction facilitated by the Cleveland Cuyahoga County Port Authority (the "Port"), whereby we acquired a ground parcel and the Port issued bonds, the proceeds of which were used to acquire, improve and equip the facilities, including the acquisition of a larger facility and a 40-year prepaid ground lease for a smaller facility. We are leasing the facilities from the Port with an expiration date of 2014, subject to certain extension options.

In addition, we lease (through 2011) administrative offices in Lynchburg, Virginia.

Through this segment's manufacturing facilities, we specialize in the design and manufacture of close-tolerance and high-quality equipment for nuclear applications. In addition, we are a leading manufacturer of critical nuclear components, fuels and assemblies for government and commercial uses. We have supplied nuclear components for DOE programs since the 1950s, and we are the largest domestic supplier of research reactor fuel elements for colleges, universities and national laboratories. We also convert or downblend high-enriched uranium into low-enriched fuel for use in commercial reactors to generate electricity. In addition, we have over 100 years of experience in supplying heavy fabrications for industrial use, including components for defense applications.

We work closely with the DOE supported non-proliferation program. Currently, this program is assisting in the development of a high-density, low-enriched uranium fuel required for high-enriched uranium test reactor conversions. We have also been a leader in the receipt, storage, characterization, dissolution, recovery and purification of a variety of uranium-bearing materials. All phases of uranium downblending and uranium recovery are provided at our Lynchburg, Virginia and Erwin, Tennessee sites

We manage and operate complex, high-consequence nuclear and national security operations for the DOE and the NNSA, primarily through our joint ventures. In addition to these joint ventures, Babcock & Wilcox Technical Services Clinch River, LLC was awarded a contract from USEC, Inc. in 2007 to manufacture classified metal parts for the American Centrifuge Program.

We have an experienced staff of design and manufacturing engineers capable of performing full scope, prototype design work coupled with manufacturing integration. The design, engineering and other capabilities of our Government Operations segment include:

- · steam separation equipment design and development;
- · thermal-hydraulic design of reactor plant components;
- structural component design for precision manufacturing;
- · materials expertise in high-strength, low-alloy steels and nickel-based materials;
- · material procurement of tubing, forgings and weld wire; and
- · metallographic and chemical analysis.

Our Government Operations segment's operations are generally capital intensive on the manufacturing side. The demand for nuclear components by the U.S. Government determines a substantial portion of this segment's backlog. We expect that orders for nuclear components will continue to be a significant part of backlog for the foreseeable future; however, such orders are subject to defense department budget constraints.

Acquisitions

The following is a brief description of some of our recent acquisitions:

Marine Mechanical Corporation. In May 2007, our Government Operations segment completed its acquisition of Marine Mechanical Corporation, which designs, manufactures and supplies electro-mechanical equipment used by the U.S. Government. This business complements our other government-related nuclear activities.

The Intech Group of Companies. On July 15, 2008, our Power Generation Systems segment completed its acquisition of the Intech group of companies ("Intech"). Intech consists of Intech, Inc., Ivey-Cooper Services, L.L.C. and Intech International Inc. Intech, Inc. provides nuclear inspection and maintenance services, primarily for the U.S. market. Ivey-Cooper Services, L.L.C. provides non-destructive inspection services to fossil-fueled power plants, as well as chemical, pulp and paper, and heavy fabrication facilities. Intech International Inc. provides non-destructive testing, field engineering and repair and specialized tooling services, primarily for the Canadian nuclear power generation industry.

Delta Power Services, LLC. In August 2008, our Power Generation Systems segment completed its acquisition of Delta Power Services LLC, an operation and maintenance services provider for the U.S. power generation industry. This business complements our current role in the fossil fuel and biomass markets. It also helps us to further diversify into the natural gas-fired power generation market.

Nuclear Fuel Services, Inc. On December 31, 2008, our Government Operations segment completed its acquisition of Nuclear Fuel Services, Inc., a provider of specialty nuclear fuels and related services. This business enhances our position as a leading provider of nuclear manufacturing and services for government and commercial markets.

Instrumentacion y Mantenimiento de Calderas, S.A. In September 2009, our Power Generation Systems segment acquired certain assets of Instrumentacion y Mantenimiento de Calderas, S.A. This acquisition provides us with additional manufacturing capabilities in support of our domestic services and regional industrial boiler manufacturing businesses.

Götaverken Miljö AB. In January 2010, our Power Generation Systems segment completed its acquisition of Götaverken Miljö AB, a flue gas cleaning and energy recovery company based in Gothenburg, Sweden. This business complements our B&W Vølund subsidiary, which is one of the world's leading suppliers of equipment and technologies designed to convert municipal solid waste and biomass into thermal energy.

GE Energy Businesses. On April 2, 2010, we acquired the electrostatic precipitator aftermarket and emissions monitoring business units of GE Energy, a division of General Electric Company, including its electrostatic precipitator replacement parts and mechanical components product lines, performance-enhancing hardware, controls and software, remote diagnostics equipment and emissions monitoring products and services. These products and services are used by a wide variety of power generation and industrial customers to monitor and control particulates and other emissions from power plants, factories and other facilities. These business units maintain offices in the United States in Kansas City, Missouri, Folkston, Georgia, Newport News, Virginia and Hatfield, Pennsylvania, as well as locations in Germany, India, Russia and China.

We continue to evaluate accelerated growth opportunities achievable through acquisition or consolidation, in addition to pursuing internal growth strategies.

Contracts

We execute our contracts through a variety of methods, including fixed-price, cost-plus and cost-reimbursable or some combination of those methods. Contracts are usually awarded through a competitive bid process, primarily based on price. However, other factors that customers may consider include plant or equipment availability, technical capabilities of equipment and personnel, efficiency, safety record and reputation.

Fixed-price contracts are for a fixed amount to cover all costs and any profit element for a defined scope of work. Fixed-price contracts entail more risk to us because they require us to predetermine both the quantities of work to be performed and the costs associated with executing the work.

We have contracts that extend beyond one year. Most of our long-term contracts have provisions for progress payments. We attempt to cover anticipated increases in labor, material and service costs of our long-term contracts either through an estimate of such charges, which is reflected in the original price, or through risk-sharing mechanisms, such as escalation or price adjustments for items such as labor and commodity prices.

We generally recognize our contract revenues and related costs on a percentage-of-completion basis. Accordingly, we review contract price and cost estimates periodically as the work progresses and reflect adjustments in profit proportionate to the percentage of completion in the period when we revise those estimates. To the extent that these adjustments result in a reduction or an elimination of previously reported profits with respect to a project, we would recognize a charge against current earnings, which could be material.

Many of our contracts with the U.S. Government are subject to annual funding determinations. In addition, contracts between the U.S. Government and its prime contractors usually contain standard provisions for termination at the convenience of the U.S. Government or the prime contractor. The contracts for the management and operation of U.S. Government facilities are awarded through a complex and protracted procurement process. These contracts are generally structured as five-year contracts with five-year renewal options, which are exercisable by the customer, or include provisions whereby the contract duration can be extended as a result of superior performance. These are cost-reimbursement contracts with a U.S. Government credit line with little corporate-funded working capital required. These contracts include a fee primarily based on performance, which is evaluated annually. As a U.S. Government contractor, we are subject to federal regulations under which our right to receive future awards of new federal contracts would be unilaterally suspended or barred if we were convicted of a crime or indicted based on allegations of a violation of specific federal statutes. In addition, some of our contracts with the U.S. Government require us to provide advance

notice in connection with any contemplated sale or shut down of the relevant facility. In each of those situations, the U.S. Government has an exclusive right to negotiate a mutually acceptable purchase of the facility.

Our arrangements with customers frequently require us to provide letters of credit, bid and performance bonds or guarantees to secure bids or performance under contracts. While these letters of credit, bonds and guarantees may involve significant dollar amounts, historically, there have been no material payments to our customers under these arrangements.

In the event of a contract deferral or cancellation, we generally would be entitled to recover costs incurred, settlement expenses and profit on work completed prior to deferral or termination. Significant or numerous cancellations could adversely affect our business, financial condition, results of operations and cash flows.

Backlog

Backlog represents the dollar amount of revenue we expect to recognize in the future from contracts awarded and in progress. Not all of our expected revenue from a contract award is recorded in backlog for a variety of reasons, including projects awarded and completed within the same fiscal quarter.

Backlog is not a measure defined by generally accepted accounting principles, and our methodology for determining backlog may not be comparable to the methodology used by other companies in determining their backlog amounts. Backlog may not be indicative of future operating results, and projects in our backlog may be cancelled, modified or otherwise altered by customers.

We generally include expected revenue of contracts in our backlog when we receive written confirmation from our customers. We generally do not include expected revenue of contracts related to our joint ventures in our backlog.

Our backlog at March 31, 2010 and December 31, 2009 was as follows:

| | March 31 2010 | March 31, 2010 | | er 31,) |
|--------------------------|------------------|-------------------|-------------|-------------|
| | | (Dollars in | ı millions) | |
| Power Generation Systems | \$1,969 | 41% | \$1,974 | 42% |
| Government Operations | 2,783 | 59% | 2,766 | 58% |
| Total Backlog | <u>\$4,752</u> | 100% | \$4,740 | 100% |

Of the March 31, 2010 backlog, we expect to recognize revenues as follows:

| | 2010 | (In millions) | Thereafter |
|--------------------------|---------|---------------|------------|
| Power Generation Systems | \$ 790 | \$ 540 | \$ 639 |
| Government Operations | 730 | 770 | 1,283 |
| Total Backlog | \$1,520 | \$1,310 | \$ 1,922 |

As of March 31, 2010, our backlog with the U.S. Government, primarily attributable to our Government Operations segment, was \$2.8 billion (of which \$35 million had not yet been funded), or approximately 59% of our total combined backlog. We do not include the value of our management and operating contracts in backlog.

During the year ended December 31, 2009, the U.S. Government awarded new orders of approximately \$0.8 billion to us, primarily in our Government Operations segment. New awards from the U.S. Government are typically received by our Government Operations segment during the fourth quarter of each year.

We believe the current worldwide credit and economic environment and short-term uncertainty regarding environmental regulations has affected our customers in the electric utility industry more than our other customers. While we have not experienced significant delays on existing projects in our Power Generation Systems' backlog, we have experienced increasing delays in expected bookings on new planned projects.

Competition

The competitive environments in which each segment operates are described below:

Power Generation Systems. With more than 140 years of experience, we are in a strong position to provide some of the most advanced steam generating equipment, emissions control equipment and services. Having supplied worldwide capacity of more than 300,000 megawatts and some of the world's largest and most efficient steam generating systems, we have the experience and technical capability to reliably convert a wide range of fuels to steam. Our strong, installed base in North America yields competitive advantages in after-market services, although this share of the market is pressured by lower level suppliers. Through this segment, we compete with a number of domestic and foreign-based companies specializing in steam-generating systems technology, equipment and services, including Alstom S.A., Doosan Babcock, Babcock Power, Inc., Foster Wheeler Ltd., Mitsubishi Heavy Industries and Hitachi, Ltd.; a variety of engineering and construction companies with respect to the installation of steam-generating systems; a number of additional companies in the markets for environmental control equipment and related specialized industrial equipment and in the independent power-producing business; and other suppliers of replacement parts, repair and alteration services and other services required to retrofit and maintain existing steam systems.

Government Operations. We have specialized capabilities that have allowed us to be a valued supplier of nuclear components for the U.S. Government since the 1950s. Also, through this segment, we are engaged in a highly competitive business through our management and operation of U.S. Government facilities. Many of our government contracts are bid as a joint venture, with one or more companies, in which we may have a minority position. The performance of the prime or lead contractor can impact our reputation and our future competitive position with respect to that particular project and customer. Competitors in the delivery of goods and services to the U.S. Government and the operation of U.S. Government facilities include Bechtel National, Inc., URS Corporation, CH2M Hill, Inc., Fluor Corporation, Lockheed Martin Corporation, Jacobs Engineering Group, Inc., EnergySolutions, Inc. and Northrop Grumman Corporation.

Joint Ventures

We participate in the ownership of a variety of entities with third parties, primarily through corporations, limited liability companies and partnerships, which we refer to as "joint ventures." Our Government Operations segment manages and operates nuclear facilities and associated plant infrastructure, manufactures components and assembles/dismantles nuclear devices, constructs large capital facilities, provides safeguards and security for inventory and assets, supports and conducts research and development for advanced energy technology and manages environmental programs for the DOE and the NNSA through joint ventures. We generally account for our investments in joint ventures under the equity method of accounting. Our significant joint ventures are described below.

Power Generation Systems

• Ebensburg Power Company & Ebensburg Investors Limited Partnership. Through these entities, which were formed by subsidiaries within our Power Generation Systems segment and ESI Energy, Inc. in 1992, we own an interest in and operate a combined solid waste and cogeneration facility located in Cambria County near Ebensburg, Pennsylvania. This facility uses bituminous waste coal for its primary fuel and sells generated electricity to a utility and steam to a hospital.

- Halley & Mellowes Pty. Ltd. Diamond Power International, Inc., one of our wholly owned subsidiaries, owns an interest in this Australian company, which was formed in 1984. Halley & Mellowes Pty. Ltd. sells soot blowers, boiler cleaning equipment, valves and material handling equipment, all of which are complementary to Diamond Power's product lines.
- Babcock & Wilcox Beijing Company, Ltd. We own equal interests in this entity with Beijing Jingcheng Machinery Electric Holding Company, Ltd.
 Babcock & Wilcox Beijing Company, Ltd. was formed in 1986 and is located in Beijing, China. Its main activities are the design, manufacturing, production and sale of various power plant and industrial boilers. It operates the largest heavy drum shop in northern China. This entity expands our markets internationally and provides additional capacity to our Power Generation Systems segment's boiler business.
- Agreement for Indian Joint Venture with Thermax Ltd. In March 2010, one of our subsidiaries and Thermax Ltd., a boiler manufacturer based in India, entered into an agreement to form a joint venture to build subcritical and highly efficient supercritical boilers and pulverizers for the Indian utility boiler market. We intend to license to the joint venture our technology for subcritical boilers 300 MW and larger, highly efficient supercritical boilers and coal pulverizers. The joint venture is also expected to construct a new 3,000 MW pressure parts facility in India. We expect the joint venture to be formed in the second quarter of 2010, subject to the satisfaction of certain customary closing conditions.

Government Operations

- Pantex Plant. Through Babcock & Wilcox Technical Services Pantex, L.L.C., a limited liability company we formed with Honeywell International Inc. and Bechtel National, Inc., we manage and operate the Pantex Plant for the DOE. The Pantex Plant is located on a 16,000-acre NNSA site near Amarillo, Texas. Key operations at this facility include evaluating, retrofitting and repairing nuclear devices; dismantling and sanitizing nuclear devices; developing, testing and fabricating high-explosive components; and handling and storing plutonium pits.
- Y-12 National Security Complex. Through Babcock & Wilcox Technical Services Y-12, L.L.C., a limited liability company we formed with Bechtel National, Inc. and in which we own a majority interest, we manage and operate the Y-12 Complex for the DOE. The Y-12 Complex is located on an 811-acre NNSA site in Oak Ridge, Tennessee. Operations at the site focus on the production, refurbishment and dismantlement of nuclear devices, storage of nuclear material and the prevention of the proliferation of weapons of mass destruction.
- Strategic Petroleum Reserve. Since 1993, this facility has been managed and operated by DynMcDermott Petroleum Operations Company, an entity we co-own with DynCorp International, International-Matex Tank and Terminals and Jacobs Engineering Group, Inc. The Strategic Petroleum Reserve stores an emergency supply of crude oil at four sites in huge underground salt caverns along the Texas and Louisiana Gulf Coast.
- Los Alamos National Laboratory. Since 2006, the Los Alamos National Security, LLC, a limited liability company formed in 2005 with the University of California, Bechtel National, URS Corporation and B&W Technical Services Group, Inc., has managed and operated the Los Alamos National Laboratory, a premier national security research institution, delivering scientific and engineering solutions for the nation's most crucial and complex problems. Located in New Mexico, the Los Alamos National Laboratory is the foremost site for the U.S. Government's ongoing research and development on the measures necessary for certifying the safety and reliability of nuclear devices without the use of nuclear testing.
- Lawrence Livermore National Laboratory. Lawrence Livermore National Security, LLC, a limited liability company formed in 2006 with the University
 of California, Bechtel National, URS Corporation and B&W Technical Services Group, Inc., manages and operates Lawrence Livermore

National Laboratory located in Livermore, California. The laboratory serves as a national resource in science and engineering, focused on national security, energy, the environment and bioscience, with special responsibility for nuclear devices.

- Savannah River Liquid Waste Disposition Program. In July 2009 Savannah River Remediation, a limited liability company formed by URS Corporation, Bechtel National, CH2M Hill Constructors, Inc. and B&W Technical Services Group, Inc., became the liquid waste contractor for the DOE's Savannah River Site located in Aiken, South Carolina. The objective of the liquid waste contract is to achieve closure of the Savannah River Site liquid waste tanks in compliance with the Federal Facilities Agreement, utilizing the Defense Waste Processing Facility and Saltstone Facility.
- Nevada Test Site. In 2006, National Security Technologies, LLC ("NSTec"), a limited liability company formed by Northrop Grumman Corporation,
 AECOM, CH2M Hill and Nuclear Fuel Services, Inc. (one of our wholly owned subsidiaries) began management and operation at the Nevada Test Site and
 its related facilities and laboratories for the DOE. Located in Las Vegas, Nevada, NSTec works on projects for other federal agencies such as the Defense
 Threat Reduction Agency, NASA, the NRC and the U.S. Air Force, Army, and Navy. Missions include defense experimentation and stockpile stewardship,
 homeland security and defense applications and environmental management.
- Isotek Systems. Isotek Systems, LLC is a limited liability company formed by EnergySolutions, Inc., Nuclear Fuel Services, Inc. and Burns and Roe
 Enterprises, Inc. Isotek received a contract in 2003 from the DOE to down blend enriched uranium-233 and extract isotopes that show great promise in the
 treatment of deadly cancers at the Oak Ridge National Laboratory in Oak Ridge, Tennessee. This contract is part of an initiative to clean up Cold War
 legacy sites.

Foreign Operations

Our Government Operations segment generates substantially all of its revenues from customers within the United States. Our Power Generations Systems segment revenues, net of intersegment revenues, and income derived from operations located outside of the United States, as well as the approximate percentages of our total combined revenues and total combined segment income, respectively, for each of the last three years were as follows (dollars in thousands):

| | Rev | enues | Segmen | t Income |
|------------------------------|-----------|---------------------|----------|---------------------|
| | · | Percent of Total | | Percent of Total |
| | Amount | Combined | Amount | Combined |
| Year ended December 31, 2009 | \$483,408 | 17% | \$72,311 | 23% |
| Year ended December 31, 2008 | \$526,080 | 15% | \$72,197 | 15% |
| Year ended December 31, 2007 | \$411,459 | 13% | \$49,122 | 14% |

For additional information on the geographic distribution of our revenues, see Note 14 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement.

Customers

We provide our products and services to a diverse customer base, including utilities and other power producers, businesses in various process industries, such as pulp and paper mills, petrochemical plants, oil refineries and steel mills, and the U.S. Government. Our five largest customers, as a percentage of our total combined revenues, during the years ended December 31, 2009 and 2008 were as follows:

| Year Ended December 31, 2009: | |
|---------------------------------------|-----|
| U.S. Government | 33% |
| American Electric Power Company, Inc. | 6% |
| FirstEnergy Corp. | 3% |
| Bruce Power, L.P. | 3% |
| United States Enrichment Corporation | 2% |
| Year Ended December 31, 2008: | |
| U.S. Government | 22% |
| American Electric Power Company, Inc. | 7% |
| First Energy Corp. | 4% |
| Allegheny Energy, Inc. | 3% |
| Public Service Company of New Mexico | 3% |

The U.S. Government is the primary customer of our Government Operations segment, comprising 92% and 89% of segment revenues for the years ended December 31, 2009 and 2008, respectively.

Our non-U.S. Government customers that account for a significant portion of revenues in one year may represent an immaterial portion of revenues in subsequent years.

Raw Materials and Suppliers

Our operations use raw materials, such as carbon and alloy steels in various forms and components and accessories for assembly, which are available from numerous sources. We generally purchase these raw materials and components as needed for individual contracts. Our Power Generation Systems segment does not depend on a single source of supply for any significant raw materials. Our Government Operations segment relies on several single-source suppliers for materials used in its products. We believe these suppliers are viable, and we and the U.S. Government expend significant effort to maintain the supplier base for our Government Operations segment.

Although shortages of some raw materials have existed from time to time, no serious shortage exists at the present time.

Employees

At March 31, 2010, we employed approximately 13,000 persons worldwide, not including approximately 10,000 joint venture employees. Approximately 4,700 of our employees were members of labor unions at March 31, 2010. Many of our operations are subject to union contracts, which we customarily renew periodically. We consider our relationships with our employees to be satisfactory.

Patents and Technology Licenses

We currently hold a large number of U.S. and foreign patents and have numerous patent applications pending. We have acquired patents and technology licenses and granted technology licenses to others when we have considered it advantageous for us to do so. Although in the aggregate our patents and technology licenses are important to us, we do not regard any single patent or technology license or group of related patents or technology licenses as critical or essential to our business as a whole. In general, we depend on our technological capabilities and the application of know-how, rather than patents and technology licenses, in the conduct of our various businesses.

Research and Development Activities

We conduct our principal research and development activities at our research and development facility in Barberton, Ohio and at our various manufacturing plants and engineering and design offices. Our Barberton facility is a world-class institution for the advancement and development of energy conversion and combustion systems, environmental emissions control processes, materials selection and manufacturing technologies for the power generation industry. Our research and development activities are related to the development and improvement of new and existing products and equipment, as well as conceptual and engineering evaluation for translation into practical applications. We charge to cost of operations the costs of research and development unrelated to specific contracts as incurred. Substantially all of these costs are in our Power Generation Systems segment and include costs related to the development of carbon capture and sequestration technology and our modular and scalable nuclear reactor business, B&W mPowerTM. Our research and development activities cost approximately \$53.2 million, \$38.5 million and \$34.8 million in the years ended December 31, 2009, 2008 and 2007, respectively. We expect to continue significant spending on research and development projects, as we continue development of our carbon capture and sequestration technology and the B&W mPowerTM reactor technology and other commercial nuclear projects. Contractual arrangements for customer-sponsored research and development can vary on a case-by-case basis and include contracts, cooperative agreements and grants. Of our total research and development expenses, our customers paid for approximately \$28.1 million, \$20.8 million and \$18.3 million in the years ended December 31, 2009, 2008 and 2007, respectively.

Hazard Risks and Insurance

Our operations present risks of injury to or death of people, loss of or damage to property, and damage to the environment. We have created loss control systems to assist us in the identification and treatment of the hazard risks presented by our operations, and we endeavor to make sure these systems are effective.

As loss control measures will not always be successful, we seek to establish various means of funding losses and liability related to incidents or occurrences. We primarily seek to do this through contractual protections, including waivers of consequential damages, indemnities, caps on liability, liquidated damage provisions, and access to the insurance of other parties. We also procure insurance, operate our own captive insurance company and/or establish funded or unfunded reserves. However, none of these methods will eliminate all risks.

Depending on competitive conditions, the nature of the work, industry custom and other factors, we may not be successful in obtaining adequate contractual protection from our customers and other parties against losses and liabilities arising out of or related to the performance of our work. The scope of the protection may be limited, may be subject to conditions and may not be supported by adequate insurance or other means of financing. In addition, we sometimes have difficulty enforcing our contractual rights with others following a material loss.

Similarly, insurance for certain potential loses or liabilities may not be available or may only be available at a cost or on terms we consider not to be economical. Insurers frequently react to market losses by ceasing to write or severely limiting coverage for certain exposures. Risks that we have frequently found difficult to cost-effectively insure against include, but are not limited to, business interruption, property losses from wind, flood and earthquake events, nuclear hazards, war and confiscation or seizure of property in some areas of the world, pollution liability, liabilities related to occupational health exposures (including asbestos), liability related to our executives participating in the management of certain outside entities, professional liability/errors and omissions coverage, the failure or unavailability of our information systems, and liability related to risk of loss of our work in progress and customer-owned materials in our care, custody and control. In cases where we place insurance, we are subject to the credit worthiness of the relevant insurer(s), the available limits of the coverage, our retention under the relevant policy, exclusions in the policy and gaps in coverage.

Our operations in designing, engineering, manufacturing, constructing and servicing nuclear power equipment and components for our commercial nuclear utility customers subject us to various risks, including,

without limitation, damage to our customers' property and third-party claims for personal injury, death and property damage. To protect against liability for damage to a customer's property, we endeavor to obtain waivers of liability and subrogation from the customer and its insurer and are usually named as an additional insured under the utility customer's nuclear property policy. We also attempt to cap our overall liability in our contracts. To protect against liability from claims brought by third parties, we are insured under the utility customer's nuclear liability policies and have the benefit of the indemnity and limitation of any applicable liability provision of the Price-Anderson Act. The Price-Anderson Act limits the public liability of manufacturers and operators of licensed nuclear facilities and other parties who may be liable in respect of, and indemnifies them against, all claims in excess of a certain amount. This amount is determined by the sum of commercially available liability insurance plus certain retrospective premium assessments payable by operators of commercial nuclear reactors. For those sites where we provide environmental remediation services, we seek the same protection from our customers as we do for our other nuclear activities. The Price-Anderson Act, as amended, includes a sunset provision and requires renewal each time that it expires. Contracts that were entered into during a period of time that Price-Anderson was in full force and effect continue to receive the benefit of the Price-Anderson Act's nuclear indemnity. The Price-Anderson Act is set to expire on December 31, 2025. We also provide nuclear fabrication and other services to the nuclear power industry in Canada. Canada's Nuclear Liability Act generally conforms to international conventions and is conceptually similar to the Price-Anderson Act in the United States. Accordingly, indemnification protections and the possibility of exclusions under Canada's Nuclear Liability Act are similar to those under the Price-Ander

Although we currently do not own or operate any nuclear reactors, we have some coverage under commercially available nuclear liability and property insurance for our facilities that are currently licensed to possess special nuclear materials. Substantially all of our Government Operations segment contracts involving nuclear materials are covered by and subject to the nuclear indemnity provisions of either the Price-Anderson Act or Public Law 85-804. However, to the extent the value of the nuclear materials in our care, custody or control exceeds the commercially available limits of our insurance, we potentially have underinsured risk of loss for such nuclear material.

Our Government Operations segment participates in the management and operation of various U.S. Government facilities. This participation is customarily accomplished through the participation in joint ventures with other contractors for any given facility. These activities involve, among other things, handling nuclear devices and their components from the aging stockpile of the U.S. Government. Insurable liabilities arising from these sites are rarely protected by our corporate insurance program. Instead, we rely on government contractual agreements, insurance purchased specifically for a site and certain specialized self-insurance programs funded by the U.S. Government. The U.S. Government has historically fulfilled its contractual agreement to reimburse its contractors for covered claims, and we expect it to continue this process during our participation in the administration of these facilities. However, in most of these situations in which the U.S. Government is contractually obligated to pay, the payment obligation is subject to the availability of authorized government funds. The reimbursement obligation of the U.S. Government is also conditional, and provisions of the relevant contract or applicable law may preclude reimbursement.

Our wholly owned captive insurance subsidiary provides workers' compensation, employer's liability, commercial general liability and automotive liability insurance to support our operations. We may also have business reasons in the future to have our insurance subsidiary accept other risks which we cannot or do not wish to transfer to outside insurance companies. These risks may be considerable in any given year or cumulatively. Our insurance subsidiary has not provided significant amounts of insurance to unrelated parties. Claims as a result of our operations could adversely impact the ability of our insurance subsidiary to respond to all claims presented.

Additionally, upon the February 22, 2006 effectiveness of the settlement relating to the Chapter 11 proceedings involving several of our subsidiaries, most of our subsidiaries contributed substantial insurance rights to the asbestos personal injury trust, including rights to (1) certain pre-1979 primary and excess insurance

coverages and (2) certain of our 1979-1986 excess insurance coverage. These insurance rights provided coverage for, among other things, asbestos and other personal injury claims, subject to the terms and conditions of the policies. The contribution of these insurance rights was made in exchange for the agreement on the part of the representatives of the asbestos claimants, including the representative of future claimants, to the entry of a permanent injunction, pursuant to Section 524(g) of the U.S. Bankruptcy Code, to channel to the asbestos trust all asbestos-related claims against our subsidiaries and former subsidiaries arising out of, resulting from or attributable to their operations, and the implementation of related releases and indemnification provisions protecting those subsidiaries and their affiliates from future liability for such claims. Although we are not aware of any significant, unresolved claims against our subsidiaries and former subsidiaries that are not subject to the channeling injunction and that relate to the periods during which such excess insurance coverage related, with the contribution of these insurance rights to the asbestos personal injury trust, it is possible that we could have underinsured or uninsured exposure for non-derivative asbestos claims or other personal injury or other claims that would have been insured under these coverages had the insurance rights not been contributed to the asbestos personal injury trust.

Governmental Regulations and Environmental Matters

General

Many aspects of our operations and properties are affected by political developments and are subject to both domestic and foreign governmental regulations, including those relating to:

- · constructing and equipping electric power and other industrial facilities;
- · possessing and processing special nuclear materials;
- workplace health and safety;
- · currency conversions and repatriation;
- · taxation of foreign earnings and earnings of expatriate personnel; and
- · protecting the environment.

We are required by various governmental and quasi-governmental agencies to obtain certain permits, licenses and certificates with respect to our operations. The kinds of permits, licenses and certificates required in our operations depend upon a number of factors.

We cannot determine the extent to which new legislation, new regulations or changes in existing laws or regulations may affect our future operations.

Environmental

Our operations and properties are subject to a wide variety of increasingly complex and stringent foreign, federal, state and local environmental laws and regulations, including those governing discharges into the air and water, the handling and disposal of solid and hazardous wastes, the remediation of soil and groundwater contaminated by hazardous substances and the health and safety of employees. Sanctions for noncompliance may include revocation of permits, corrective action orders, administrative or civil penalties and criminal prosecution. Some environmental laws provide for strict, joint and several liability for remediation of spills and other releases of hazardous substances, as well as damage to natural resources. In addition, companies may be subject to claims alleging personal injury or property damage as a result of alleged exposure to hazardous substances. Such laws and regulations may also expose us to liability for the conduct of or conditions caused by others or for our acts that were in compliance with all applicable laws at the time such acts were performed.

These laws and regulations include the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended ("CERCLA"), the Clean Air Act, the Clean Water Act, the Resource

Conservation and Recovery Act and similar laws that provide for responses to, and liability for, releases of hazardous substances into the environment. These laws and regulations also include similar foreign, state or local counterparts to these federal laws, which regulate air emissions, water discharges, hazardous substances and waste and require public disclosure related to the use of various hazardous substances. Our operations are also governed by laws and regulations relating to workplace safety and worker health, primarily, in the United States, the Occupational Safety and Health Act and regulations promulgated thereunder.

We are currently in the process of investigating and remediating some of our operating sites. We have recorded a total reserve for these sites of \$50.3 million. The sites include our manufacturing facilities at Lynchburg, Virginia (\$11.6 million reserve, primarily relating to the decommissioning at a nuclear analytical laboratory) and Erwin, Tennessee (\$37.8 million reserve for decontamination and decommissioning of legacy operations conducted prior to our acquisition of Nuclear Fuel Services, Inc.). Additional sites include our research center at Alliance, Ohio (\$0.3 million reserve), as well as discontinued operations at Canmore Mine, Alberta, Canada (\$0.4 million reserve), Kiski Valley, Pennsylvania (\$0.1 million reserve) and Maxey Flats, Kentucky (\$0.1 million reserve). Each of these sites either used or possessed hazardous materials in a manner that required remediation. Although we have recorded reserves in connection with certain of these matters, due to the uncertainties associated with environmental remediation, we cannot assure you that the actual costs resulting from these remediation matters will not exceed the recorded reserves.

Our compliance with U.S. federal, state and local environmental control and protection regulations resulted in pretax charges of approximately \$11.1 million in the year ended December 31, 2009. In addition, compliance with existing environmental regulations necessitated capital expenditures of \$0.7 million in the year ended December 31, 2009. We expect to spend another \$5.5 million on such capital expenditures over the next five years. We cannot predict all of the environmental requirements or circumstances that will exist in the future but anticipate that environmental control and protection standards will become increasingly stringent and costly. Based on our experience to date, we do not currently anticipate any material adverse effect on our business or financial condition as a result of future compliance with existing environmental laws and regulations. However, future events, such as changes in existing laws and regulations or their interpretation, more vigorous enforcement policies of regulatory agencies or stricter or different interpretations of existing laws and regulations, may require additional expenditures by us, which may be material. Accordingly, we can provide no assurance that we will not incur significant environmental compliance costs in the future.

We have been identified as a potentially responsible party at various cleanup sites under CERCLA. CERCLA and other environmental laws can impose liability for the entire cost of cleanup on any of the potentially responsible parties, regardless of fault or the lawfulness of the original conduct. Generally, however, where there are multiple responsible parties, a final allocation of costs is made based on the amount and type of wastes disposed of by each party and the number of financially viable parties, although this may not be the case with respect to any particular site. We have not been determined to be a major contributor of wastes to any of these sites. On the basis of our relative contribution of waste to each site, we expect our share of the ultimate liability for the various sites will not have a material adverse effect on our financial condition, results of operations or cash flows in any given year.

Environmental remediation projects have been and continue to be undertaken at certain of our current and former plant sites. During the fiscal year ended March 31, 1995, one of our subsidiaries decided to close its nuclear manufacturing facilities in Parks Township, Armstrong County, Pennsylvania (the "Parks Facilities") and proceeded to decommission the facilities in accordance with its then-existing license from the NRC. The facilities were subsequently transferred to another subsidiary of ours in the fiscal year ended March 31, 1998, and, during the fiscal year ended March 31, 1999, that subsidiary reached an agreement with the NRC on a plan that provided for the completion of facilities dismantlement and soil restoration by 2001 and license termination in 2003. An application to terminate the NRC license for the Parks Township facility was filed, and the NRC terminated the license in 2004 and released the facility for unrestricted use. For a discussion of certain civil litigation we are involved in concerning the Parks Facilities, see "—Legal Proceedings—"PADEP Notice."

We perform significant amounts of work for the U.S. Government under both prime contracts and subcontracts and operate certain facilities that are licensed to possess and process special nuclear materials. As a result of these activities, we are subject to continuing reviews by governmental agencies, including the EPA and the NRC

The NRC's decommissioning regulations require our Government Operations segment to provide financial assurance that it will be able to pay the expected cost of decommissioning each of its facilities at the end of its service life. We will continue to provide financial assurance aggregating \$33.7 million during the year ending December 31, 2010 with existing letters of credit for the ultimate decommissioning of all of these licensed facilities, except two. These two facilities, which represent the largest portion of our eventual decommissioning costs, have provisions in their government contracts pursuant to which substantially all of our decommissioning costs and financial assurance obligations are covered by the DOE, including the costs to complete the decommissioning projects underway at the Erwin facility.

The demand for power generation products and services can be influenced by state and federal governmental legislation setting requirements for utilities related to operations, emissions and environmental impacts. The legislative process is unpredictable and includes a platform that continuously seeks to increase the restrictions on power producers. Potential legislation limiting emissions from power plants, including carbon dioxide, could affect our markets and the demand for our products and services in our Power Generation Systems segment.

At December 31, 2009 and 2008, we had total environmental reserves, including provisions for the facilities discussed above, of \$50.3 million and \$38.0 million, respectively. Of our total environmental reserves at December 31, 2009 and 2008, \$2.5 million and \$5.0 million, respectively, were included in current liabilities. Inherent in the estimates of those reserves and recoveries are our expectations regarding the levels of contamination, decommissioning costs and recoverability from other parties, which may vary significantly as decommissioning activities progress. Accordingly, changes in estimates could result in material adjustments to our operating results, and the ultimate loss may differ materially from the amounts we have provided for in our combined financial statements.

Legal Proceedings

PADEP Notice

The Department of Environmental Protection of the Commonwealth of Pennsylvania ("PADEP") advised us in March 1994 that it would seek monetary sanctions and remedial and monitoring relief related to the Parks Facilities. The relief sought is related to potential groundwater contamination resulting from previous operations at the facilities. These facilities are currently owned by a subsidiary in our Government Operations segment. PADEP has advised us that it does not intend to assess any monetary sanctions, provided our Government Operations segment continues its remediation program for the Parks Facilities. Whether additional nonradiation contamination remediation will be required at the Parks Facility remains unclear. Results from sampling completed by our Government Operations segment have indicated that such remediation may not be necessary. Our Government Operations segment continues to evaluate closure of the groundwater issues pursuant to applicable Pennsylvania law.

Apollo and Parks Township Litigation

Since January 2010, five suits involving approximately 25 plaintiffs have been filed against B&W PGG, Babcock & Wilcox Technical Services Group, Inc., formerly known as B&W Nuclear Environmental Services, Inc. (together with B&W PGG, the "B&W Parties") and Atlantic Richfield Company ("ARCO") in the United States District Court for the Western District of Pennsylvania (the "Apollo and Parks Township Litigation"). The plaintiffs in the Apollo and Parks Township Litigation allege, among other things, that they suffered personal

injuries and property damage as a result of alleged radioactive and non-radioactive releases relating to the operation, remediation and/or decommissioning of two former nuclear fuel processing facilities located in Apollo and Parks Township, Pennsylvania. Those facilities previously were owned by Nuclear Materials and Equipment Company, a former subsidiary of ARCO ("NUMEC") which was acquired by B&W PGG. The plaintiffs in the Apollo and Parks Township Litigation seek compensatory and punitive damages. It is possible that additional, similar lawsuits will be filed.

At the time of ARCO's sale of NUMEC to B&W PGG, B&W PGG received an indemnity and hold harmless agreement from ARCO from claims or liabilities arising as a result of pre-closing NUMEC or ARCO actions.

We intend to vigorously defend these matters, and believe that in the event of liability, if any, the claims alleged in the Apollo and Parks Township Litigation will be resolved within the limits of coverage of our insurance policies and/or the ARCO indemnity.

B&W PGG settled approximately 245 personal injury and wrongful death claims, as well as approximately 125 property damage claims, alleging injury and damage as a result of alleged releases relating to these two facilities. The aggregate settlement amount for these claims was \$52.5 million, which was paid during 2009, all of which had been previously accrued. In connection with that settlement, the B&W Parties are pursuing recovery in the matter of The Babcock & Wilcox Company et al. v. American Nuclear Insurers et al. (the "ANI Litigation") from their insurer, American Nuclear Insurers and Mutual Atomic Energy Liability Underwriters, of the amounts paid in settlement of that prior action. The ANI Litigation is pending before the Court of Common Pleas of Allegheny County, Pennsylvania. No trial date has been set in the matter.

Iroquois Falls Litigation

An action entitled Iroquois Falls Power Corp. v. Jacobs Canada Inc., et al., was filed in the Superior Court of Justice, in Ontario, Canada, on June 1, 2005. Iroquois Falls Power Corp. ("Iroquois") seeks damages of approximately \$14 million (Canadian) as a result of an alleged breach by one of our former subsidiaries in connection with the supply and installation of heat recovery steam generators. McDermott Incorporated, an indirect, wholly owned subsidiary of McDermott, or MI, provided a guarantee to certain obligations of the former subsidiary. MI and two bonding companies with whom McDermott entered into an indemnity arrangement were also named as defendants. In March 2007, the Superior Court granted summary judgment in favor of all defendants and dismissed all claims of Iroquois, which appealed the ruling. Subsequently, the Court of Appeals for Ontario upheld the summary judgment, but sent the case back to the Superior Court of Justice to allow Iroquois an opportunity to amend its complaint to assert new claims. The Superior Court of Justice, however, denied Iroquois' request to amend its complaint and assert new claims against the defendants based on a breach of contractual warranty. Iroquois appealed the Superior Court's decision and, in June 2009, the Court of Appeals for Ontario reversed the decision and sent the case back to the Superior Court for Iroquois to file an amended complaint on those new claims. In January 2010, our notice to appeal the Court of Appeals' decision was dismissed by the Supreme Court.

Chapter 11 Proceedings

In February 2000, several of B&W's subsidiaries filed for voluntary Chapter 11 protection, due to mounting asbestos-related claims tied to asbestos-containing commercial and utility boilers and other products they sold, installed or serviced in prior decades. From the date of the filing of the Chapter 11 petition until their emergence from Chapter 11 in 2006, McDermott treated those subsidiaries as deconsolidated subsidiaries for financial reporting purposes. The final settlement with representatives of asbestos claimants in the Chapter 11 proceedings, which was effected in a plan of reorganization consummated as of February 22, 2006, involved McDermott's and B&W's contribution of rights under various insurance policies, cash and a promissory note, with an aggregate value of approximately \$2 billion, to a settlement trust, the channeling of asbestos-related personal injury claims to that trust and McDermott's retention of the ownership of the B&W subsidiaries involved in the Chapter 11 proceedings.

CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS

In connection with the spin-off, we have entered into several agreements with McDermott to define our ongoing relationship with McDermott after the spin-off. These agreements, among other things, allocate responsibility for obligations arising before and after the distribution date, including, among others, obligations relating to our employees, various transition services and taxes. For more information about those agreements and our historical relationship with McDermott, see "Relationship with McDermott After the Spin-Off."

Ms. Salomone served as the President and Chief Executive Officer of Marine Mechanical Corporation ("MMC") before we acquired MMC in 2007. Prior to our acquisition, the shares of Class A common stock of MMC were owned by an employee stock ownership trust (the "ESOP"), and Ms. Salomone was a participant in the ESOP. In addition, among other former MMC employees, Ms. Salomone owned shares of Class B common stock of MMC and certain options to purchase shares of Class B common stock of MMC. In connection with the acquisition, we deposited a portion of the acquisition price for the Class A shares, the Class B shares and the options in an escrow account, as a means of providing a fund for resolution of any post-closing claims under the indemnification provisions of the acquisition agreement. As of March 31, 2010, the balance of the escrow account was approximately \$5.6 million. Under the terms of the escrow arrangement, and subject to certain exceptions, any remaining balance in the escrow account as of May 2011 is scheduled to be released to the ESOP participants and former holders of the Class B shares and the options, including Ms. Salomone, after such date. Ms. Salomone has an approximate 5.2% interest in the funds in the escrow account based on her participation interest in the ESOP and her former ownership of Class B shares and options.

Related Person Transactions Policies and Procedures

We expect that our board will adopt a policy, which will be made available on our website, that will require all employees (including our Named Executive Officers) who have, or whose immediate family members have, any direct or indirect financial or other participation in any business that competes with, supplies goods or services to, or is a customer of ours, to disclose to us and receive written approval from our Corporate Ethics and Compliance department prior to transacting such business. We refer to any such transaction as a related person transaction.

Our employees are expected to make reasoned and impartial decisions in the workplace. As a result, approval of certain business will be denied if we believe that the employee's interest in such business could influence decisions relative to our business, or have the potential to adversely affect our business or the objective performance of the employee's work.

Our Corporate Ethics and Compliance department implements our Code of Business Conduct and related policies and the Governance Committee of our Board is responsible for overseeing our Ethics and Compliance Program, including compliance with our Code of Business Conduct. Our Code of Business Conduct provides that all employees (including Named Executive Officers) must ensure that business decisions they make and actions they take on behalf of our company are not influenced by personal considerations or personal relationships and will require appropriate disclosures of potential conflicts of interest.

Additionally, our Governance Committee is responsible for reviewing the professional occupations and associations of our Board members and reviews transactions between B&W and other companies with which our Board members are affiliated.

RELATIONSHIP WITH MCDERMOTT AFTER THE SPIN-OFF

Historical Relationship with McDermott

We are currently a wholly owned subsidiary of McDermott. As a result of this historic parent-subsidiary relationship, in the ordinary course of our business, we and our subsidiaries have received various services provided by McDermott and some of its other subsidiaries, including accounting, treasury, tax, legal, risk management, public affairs, human resources, procurement, information technology and other services. Our historical financial statements include allocations by McDermott of a portion of its overhead costs related to those services. These cost allocations have been determined on a basis that we and McDermott consider to be reasonable reflections of the use of those services.

McDermott's Distribution of Our Stock

McDermott will be our sole stockholder until completion of the spin-off. In the spin-off, McDermott is distributing its entire equity interest in us to its stockholders in a transaction that is intended to be tax free to McDermott and its U.S. stockholders. The spin-off will be subject to a number of conditions, some of which are more fully described above under "The Spin-Off—Spin-Off Conditions and Termination."

Agreements Between McDermott and Us

In the discussion that follows, we have described the material provisions of agreements we have entered into with McDermott. The description of those agreements is not complete and is qualified by reference to the terms of the agreements, the forms of which have been filed as exhibits to the registration statement on Form 10 of which this information statement is a part. We encourage you to read the full text of those agreements. Those agreements were developed under the oversight of the Restructuring Committee of the McDermott board of directors, which was formed to oversee an orderly and appropriate separation of our company from McDermott. Because the terms of our separation from McDermott and these agreements are being entered into in the context of a related-party transaction, these terms may not be comparable to terms that would be obtained in a transaction between unaffiliated parties.

Master Separation Agreement

The master separation agreement between McDermott and us governs the separation of our businesses from McDermott, the subsequent distribution of our shares to McDermott stockholders and other matters related to McDermott's relationship with us.

The Separation. To effect the separation, McDermott will effect a series of transactions which will cause us to succeed to the assets of our business as described in this information statement (which assets may include stock or other equity interests of McDermott subsidiaries). We will also succeed to, and have agreed to perform and fulfill, the liabilities described below. In particular, the master separation agreement generally provides that, upon completion of the separation:

- · we will directly or indirectly hold:
 - all of the assets owned by McDermott or any of its subsidiaries which are reflected on our most recent pro forma combined balance sheet set forth in this information statement, or subsequently-acquired or created assets that would have been reflected on a later-dated balance sheet, and
 - all other assets held by us, McDermott, or any of our respective subsidiaries used primarily in or that primarily relate to our business on or prior to the date of the spin-off, subject to certain exceptions;
- we will be subject to:
 - all outstanding liabilities reflected on our most recent pro forma, as adjusted combined balance sheet set forth in this information statement, or subsequently-incurred or accrued liabilities that would have been reflected on a later-dated balance sheet,

- liabilities to the extent relating to, arising out of or resulting from our post-spin-off operations, or any assets owned by us or our subsidiaries as of
 or after the spin-off, and
- liabilities we have assumed under the master separation agreement and other ancillary agreements.

The master separation agreement provides that capital stock, assets or liabilities that cannot legally be transferred or assumed prior to the spin-off will be transferred or assumed as soon as practicable following receipt of all necessary consents of third parties and regulatory approvals. In any such case, the master separation agreement provides that the party retaining such capital stock, assets or liabilities will hold the capital stock or assets in trust for the use and benefit of, or retain the liabilities for the account of, the party entitled to the capital stock, assets or liabilities (at the expense of that party), until the transfer or assumption can be completed. The party retaining the capital stock, assets or liabilities will also take any action reasonably requested by the other party in order to place the other party in the same position as would have existed if the transfer or assumption had been completed. Based on the third-party consents and approvals we have already received, we do not anticipate that these provisions will be relied on for any material items.

Except as set forth in the master separation agreement, no party is making any representation or warranty as to the companies, capital stock, assets or liabilities transferred or assumed as a part of the separation and any assets that may be transferred will be transferred on an "as is, where is" basis. As a result, we and McDermott each have agreed to bear the economic and legal risks that any conveyances of capital stock or assets are insufficient to vest good and marketable title to such capital stock or assets, as the case may be, in the party who should have title under the master separation agreement. The master separation agreement also provides that the spin-off is subject to the conditions described under "The Spin-Off—Spin-Off Conditions and Termination."

Surety Instruments and Guarantees. The master separation agreement requires that we and McDermott use our commercially reasonable efforts to terminate (or have us or one of our subsidiaries substituted for McDermott, or McDermott or one of its subsidiaries substituted for us, as applicable, under) all existing guarantees by one party of obligations relating to the business of the other party, including financial, performance and other guarantee obligations. We also have agreed with McDermott was will use our commercially reasonable efforts to have us or one of our subsidiaries substituted for McDermott under letters of credit or other surety instruments issued by third parties for the account of McDermott or any of its subsidiaries but on behalf of our business. In the event we are unable to be substituted under existing surety instruments or issue replacement surety instruments, we will otherwise reach agreement with McDermott prior to the spin-off to satisfy any requirements for surety instruments and will indemnify McDermott for the value of the applicable instruments.

Insurance. The master separation agreement provides that, to the extent permitted by the terms of the applicable policy, we and our directors, officers and employees will continue to have (as successors-in-interest) all rights we and they had immediately prior to the spin-off, with respect to events that occurred prior to the spin-off, as a subsidiary, affiliate, division, director, officer or employee of McDermott under any McDermott insurance policy with a third-party carrier. McDermott will have no liability if any insurance policy is terminated, is not renewed or extended, or is insufficient or unavailable.

Intellectual Property. In connection with the separation, we will generally assign to McDermott all intellectual property rights to the extent not exclusively used in our business, and McDermott will grant us a perpetual, nonexclusive, royalty-free license in all such intellectual property. We will retain all rights in our trademarks, and in intellectual property exclusively used in our business and other specified intellectual property.

Access to Information. Subject to applicable confidentiality provisions and other restrictions, we and McDermott will each give each other any information in our possession that the requesting party reasonably needs (1) to comply with requirements imposed on the requesting party by a governmental authority, (2) for use in any proceeding to satisfy audit, accounting, insurance claims, regulatory, litigation or other similar requirements, (3) to comply with its obligations under the master separation agreement and other ancillary agreements or (4) for any other significant business need as mutually determined in good faith by the parties.

Indemnification and Release. In general, under the master separation agreement, we have agreed to indemnify McDermott and its representatives and affiliates against certain liabilities from third-party claims to the extent relating to, arising out of or resulting from:

- our failure to discharge any of our liabilities or any of our agreements;
- the operation of our business, whether before or after the spin-off; and
- any untrue statement or alleged untrue statement of a material fact or material omission or alleged material omission in the registration statement of which this information statement is a part, other than certain information relating to McDermott.

We have also agreed to indemnify McDermott and its representatives and affiliates against 50% of the amount of liabilities from any third-party claims asserted after the distribution date to the extent relating to certain discontinued operations conducted prior to the distribution date that are not directly attributable or related to either McDermott's offshore oil and gas construction business or our business.

In general, under the master separation agreement, McDermott has agreed to indemnify us and our representatives and affiliates against certain liabilities from third-party claims to the extent relating to, arising out of or resulting from:

- · the failure of McDermott to discharge any liability of McDermott or any McDermott agreement that is not transferred to us;
- the operation of McDermott's business (other than our business), whether before or after the spin-off; and
- any untrue statement or alleged untrue statement of a material fact or material omission or alleged material omission in the registration statement of which this information statement is part, only for certain information relating to McDermott.

McDermott has also agreed to indemnify us and our representatives and affiliates against 50% of the amount of liabilities from any third-party claims asserted after the distribution date to the extent relating to certain discontinued operations conducted prior to the distribution date that are not directly attributable or related to either McDermott's offshore oil and gas construction business or our business.

Under the master separation agreement, we generally release McDermott and its affiliates, agents, successors and assigns, and McDermott generally releases us and our affiliates, agents, successors and assigns, from any liabilities between us or our subsidiaries on the one hand, and McDermott or its subsidiaries on the other hand, arising from acts or events occurring on or before the spin-off, including acts or events occurring in connection with the separation or distribution. This release will include the release of a net intercompany payable in an approximate amount of \$200 million as of December 31, 2009 currently owed to B&W by certain subsidiaries of ours that will be transferred to McDermott in connection with the spin-off. One of our subsidiaries also owes approximately \$320 million under a promissory note to another subsidiary of McDermott (and, as a result of certain pre-spin restructuring transactions, including the transfer of that subsidiary to McDermott, our consolidated group will be effectively relieved of any continuing obligations relating to that promissory note). The general release does not apply to obligations under the master separation agreement or any ancillary agreement or to specified ongoing contractual arrangements.

Termination. The master separation agreement provides that it may be terminated at any time before the spin-off by McDermott in its sole discretion.

Tax Sharing Agreement

We and a subsidiary of McDermott (the "McDermott Party") have entered into a tax sharing agreement that will govern the respective rights, responsibilities and obligations of McDermott and its subsidiaries, and us with respect to taxes and tax benefits, the filing of tax returns, the control of audits and other tax matters. References in this summary description of the tax sharing agreement to the terms "tax" or "taxes" mean taxes as well as any interest, penalties, additions to tax or additional amounts in respect of such taxes.

The results of U.S. operations of McDermott and/or certain of its subsidiaries are currently reflected in our consolidated return for U.S. federal income tax purposes and/or certain consolidated, combined and unitary returns for state, local and foreign tax purposes. However, for periods (or portions thereof) beginning after the transaction in which McDermott and its subsidiaries are separated from us (the "Separation Transaction"), McDermott and its subsidiaries generally will not join with us in the filing of any federal, state, local or foreign consolidated, combined or unitary tax returns.

Under the tax sharing agreement, for any tax period (or portion thereof), we will generally be responsible for paying all U.S. federal, state, local and foreign income taxes that are imposed on us and our subsidiaries, and the McDermott Party will be responsible for paying all U.S. federal, state, local and foreign income taxes that are imposed on McDermott and its subsidiaries. However, for tax periods (or portions thereof) ending on or prior to the date of the Separation Transaction, we will also be responsible for paying all U.S. federal, state, local and foreign income taxes relating to an intercompany borrowing by one of our affiliates.

We and the McDermott Party generally will not be required to reimburse one another for the use by one party or its affiliates of tax benefits attributable to the other party or its affiliates. However, we will have an obligation to reimburse the McDermott Party, and the McDermott Party will have an obligation to reimburse us, for:

- tax benefits arising as a result of an audit or other required adjustment occurring after the date of the Separation Transaction that are attributable to one party or its affiliates and used by the other party or its affiliates during tax periods (or portions thereof) ending on or prior to the date of the Separation Transaction; and
- tax benefits arising after the date of the Separation Transaction that are attributable to one party or its affiliates and that are carried back under applicable tax law and used by the other party or its affiliates during tax periods (or portions thereof) ending on or prior to the date of the Separation Transaction.

Notwithstanding the tax sharing agreement, under U.S. Treasury Regulations, each member of a U.S. consolidated group is severally liable for the U.S. federal income tax liability of each other member of such consolidated group. Accordingly, with respect to periods in which certain subsidiaries of McDermott have been included in our U.S. consolidated group, we and our subsidiaries could be liable to the U.S. Government for any U.S. federal income tax liability incurred, but not discharged, by those McDermott subsidiaries. However, if any such liability were imposed, we would generally be entitled to be indemnified by the McDermott Party for tax liabilities attributable to it under the tax sharing agreement.

We will be responsible for preparing and filing all consolidated, combined and unitary tax returns that include us or our subsidiaries and McDermott or its subsidiaries (regardless of when the tax period ends), or that include solely us or our subsidiaries, and the McDermott Party will be responsible for preparing and filing all consolidated, combined and unitary tax returns that include solely McDermott or its subsidiaries. The party that files a tax return will have the authority to respond to and appear in all tax proceedings, including tax audits, involving any taxes or any deemed adjustment to taxes reported on such tax return, except with respect to (1) those portions of such return that relate exclusively to the other party and (2) certain specified issues over which we will have exclusive audit control. The tax sharing agreement further provides for cooperation between

us and the McDermott Party with respect to tax matters, the exchange of information and the retention of records that may affect the tax liabilities of the parties to the agreement.

To the extent permitted by applicable tax law, we and the McDermott Party will treat any payments made under the tax sharing agreement as a capital contribution or distribution (as applicable) immediately prior to the Separation Transaction, and accordingly, as not includible in the taxable income of the recipient.

In the event that the Separation Transaction and/or certain related transactions were to fail to qualify for tax-free treatment, we would generally be responsible for all of the tax imposed on us or McDermott and its subsidiaries resulting from such failure. However, if such failure resulted from actions or failures to act by McDermott or its subsidiaries, the McDermott Party would be responsible for such tax.

THIS SUMMARY IS QUALIFIED BY REFERENCE TO THE FULL TEXT OF THE TAX SHARING AGREEMENT, A FORM OF WHICH HAS BEEN FILED AS AN EXHIBIT TO THE REGISTRATION STATEMENT ON FORM 10 OF WHICH THIS INFORMATION STATEMENT IS A PART.

Employee Matters Agreement

We and McDermott have entered into an employee matters agreement, which provides that each of McDermott and B&W will have responsibility for its own employees and compensation plans. The agreement contains provisions concerning benefit protection for McDermott employees, treatment of holders of McDermott stock options, restricted stock, restricted stock units, deferred stock units and performance shares, and cooperation between us and McDermott in the sharing of employee information and maintenance of confidentiality.

Treatment of Retirement, Health and Welfare Plans. In general, our employees currently participate in various retirement, health and welfare, and other employee benefit plans. Following the spin-off, we anticipate that our employees will generally continue to participate in the same plans or will participate in similar plans and arrangements that we will establish and maintain. Pursuant to the employee matters agreement, effective as of the distribution date, we and McDermott will each retain responsibility for our respective employees and compensation plans.

In several locations outside the United States, it likely will not be feasible to establish retirement or welfare plans due to the small number of employees at those locations. In those situations, we will establish alternative compensation or benefit programs to comply with our obligations to affected employees.

Treatment of Stock-Based Awards. The employee matters agreement also provides for the adjustments and replacement awards to be made with respect to options to purchase shares of McDermott common stock, McDermott shares of restricted stock, restricted stock units, deferred stock units and performance shares, as described under "The Spin-Off—Treatment of Stock-Based Awards."

Entitlement to Tax Deductions. We will be entitled to claim all tax deductions for compensation arising after the spin-off from the exercise of substitute B&W options, the vesting of B&W restricted stock, restricted stock unit, deferred stock unit and performance share awards held by current or former employees of ours, and McDermott will not claim any such deduction. McDermott will be entitled to claim all tax deductions for compensation arising after the spin-off from the exercise of adjusted McDermott options or substitute B&W options, the vesting of McDermott restricted stock, restricted stock unit, deferred stock unit and performance share awards or B&W restricted stock, restricted stock unit, deferred stock unit and performance share awards held by current or former McDermott employees, and we will not claim any such deduction.

Transition Services Agreements

We and McDermott have entered into transition services agreements under which we and McDermott will provide and/or make available various administrative services and assets to each other, for specified periods beginning on the distribution date. The services McDermott will provide us will include:

- · accounting services
- · executive consulting services;
- human resources services;
- information technology services;
- · investor relations services;
- legal services;
- · risk management services;
- · tax services; and
- treasury services.

The services we will provide to McDermott will include:

- · accounting services;
- human resources services;
- · information technology services;
- legal services;
- risk management services;
- · tax services; and
- treasury services.

In consideration for such services, we and McDermott will each pay fees to the other for the services provided, and those fees will generally be in amounts intended to allow the party providing services to recover all of its direct and indirect costs incurred in providing those services.

The personnel performing services under the transition services agreements will be employees and/or independent contractors of the party providing the service and will not be under the direction or control of the party to whom the service is being provided.

The transition services agreements will also contain customary mutual indemnification provisions.

Any extension or renewal of services under any transition services agreement beyond the period specified with respect to such service will be subject to the mutual agreement of McDermott and us.

MANAGEMENT

Directors and Executive Officers

Under Delaware law, the business and affairs of B&W will be managed under the direction of its board of directors. The B&W certificate of incorporation and bylaws provide that the number of directors may be fixed by the board from time to time. As of the distribution date, the board of directors of B&W is expected to consist of the individuals listed below (with their ages as of June 30, 2010). The present principal occupation or employment and five-year employment history of each individual (other than Mr. Bethards, our President and Chief Executive Officer, whose employment history follows the list of our executive officers below) follows the list below. Each of the individuals listed below is a citizen of the United States.

| Name | Age |
|-------------------------|-----|
| Brandon C. Bethards | 62 |
| John A. Fees. | 52 |
| Robert W. Goldman | 68 |
| Stephen G. Hanks | 60 |
| Oliver D. Kingsley, Jr. | 67 |
| D. Brad McWilliams | 68 |
| Richard W. Mies | 66 |

Mr. Fees has been Chief Executive Officer of McDermott since October 2008. He joined McDermott in 1979 and served as President and Chief Executive Officer of B&W from January 2007 to October 2008; President and Chief Operating Officer of BWX Technologies, Inc., a subsidiary of B&W, from September 2002 to January 2007; and President, General Manager of BWXT Services, Inc., a subsidiary of BWX Technologies, Inc., from September 1997 to November 2002. His earlier positions at subsidiaries of B&W include Vice President and General Manager. Mr. Fees has been a director of McDermott since October 2008.

Since October 2002, Mr. Goldman has served as an independent financial consultant. Previously, Mr. Goldman worked for Conoco Inc., an international, integrated energy company and predecessor to ConocoPhillips, from 1988 to 2002, most recently as Senior Vice President, Finance and Chief Financial Officer from 1998 to 2002. He formerly served as the Vice President, Finance of the World Petroleum Council from 2002 to 2008. He has also served as a director of El Paso Corporation since 2003, Parker Drilling Company since 2005 and Tesoro Corporation since 2004. Mr. Goldman has been a director of McDermott since November 2005.

From November 2007 until his retirement in January 2008, Mr. Hanks was President of the Washington Division of URS Corporation, an engineering, construction and technical services company, and he also served as a member of URS Corporation's Board of Directors during that time. Previously, from June 2001 to November 2007 he was President and CEO of Washington Group International, Inc., an integrated engineering, construction and management services company which was acquired by URS Corporation in 2007, and also served on its Board of Directors. Mr. Hanks has also served as a director of Lincoln Electric Holdings, Inc. since 2006. Mr. Hanks has been a director of McDermott since May 2009.

Until his retirement in November 2004, Mr. Kingsley served as President and Chief Operating Officer of Exelon Corporation, an integrated utility company, from May 2003, Senior Executive Vice President from February 2002 and President and Chief Nuclear Officer from October 2000. Mr. Kingsley also served as President and Chief Executive Officer of Exelon's subsidiary, Exelon Generation, from February 2000 to November 2004 and as President and Chief Nuclear Officer of Unicom Corporation, an integrated electric utility company, from November 1997 to October 2000. Mr. Kingsley has also served as a director of FPL Group, Inc. since 2007 and is the Associate Dean for Special Projects at the Sam Ginn College of Engineering, Auburn University. Mr. Kingsley has been a director of McDermott since November 2004.

From April 1995 until his retirement in April 2003, Mr. McWilliams was Senior Vice President and Chief Financial Officer of Cooper Industries Ltd., a worldwide manufacturer of electrical products, tools and hardware. He was Vice President of Cooper Industries from 1982 until April 1995. Mr. McWilliams previously served as a director of Kronos Incorporated from 1993 to 2005. Mr. McWilliams has been a director of McDermott since July 2003.

Admiral Mies is a Retired Admiral, United States Navy. He served in the U.S. Navy for 35 years, including most recently as Commander in Chief of the U.S. Strategic Command for the U.S. Air Force and U.S. Navy strategic nuclear forces from 1998 until his retirement from the Navy in 2002. Following his retirement from the Navy until 2007, he served as Senior Vice President of Science Applications International Corporation, a provider of scientific and engineering applications for national security, energy, environment, critical infrastructure and health. Since 2007, he has been Chief Executive Officer and President of The Mies Group, Ltd. (a consulting firm). He has also served as a director of Exelon Corporation since 2009 and Mutual of Omaha Insurance Company since 2002. Mr. Mies has been a director of McDermott since August 2008.

We expect that Messrs. Hanks and McWilliams will continue to serve on the board of directors of McDermott after the completion of the spin-off, while the other persons named above are expected to resign from the McDermott board of directors upon completion of the spin-off.

The individuals listed below (with their ages as of June 30, 2010) are expected to be executive officers of B&W as of the distribution date. The business address of each of the individuals listed below is c/o The Babcock & Wilcox Company, The Harris Building, 13024 Ballantyne Corporate Place, Suite 700, Charlotte, North Carolina 28277. Each of the individuals listed below is a citizen of the United States.

| Name | Age | Position |
|---------------------|-----|--|
| Vangel Athanas | 54 | Vice President of Human Resources |
| Brandon C. Bethards | 62 | President and Chief Executive Officer |
| David S. Black | 48 | Chief Accounting Officer |
| James D. Canafax | 39 | Senior Vice President, General Counsel and Corporate Secretary |
| S. Robert Cochran | 57 | President of Babcock & Wilcox Technical Services Group, Inc. |
| Richard L. Killion | 62 | President and Chief Operating Officer of Babcock & Wilcox Power Generation Group, Inc. |
| Christofer M. Mowry | 48 | President of Babcock & Wilcox Nuclear Energy, Inc. |
| Winfred D. Nash | 66 | President of Babcock & Wilcox Nuclear Operations Group, Inc. |
| Mary P. Salomone | 50 | Chief Operating Officer |
| Michael S. Taff | 48 | Senior Vice President and Chief Financial Officer |

Vangel Athanas has served as our Vice President of Human Resources since joining B&W in August 2008. Prior to joining B&W, Mr. Athanas served with Computer Sciences Corporation, an information technology outsourcing and services provider, from 1995 until August 2008, including his most recent position as the Vice President Human Resources of World Sourcing Services from April 2007 to August 2008. Mr. Athanas previously served as the Director Human Resources, Global Transformation Solutions of Computer Sciences Corporation from April 2004 to April 2007. Prior to that, Mr. Athanas was employed by the Electric Boat Division of General Dynamics from 1979 to 1995 in increasingly responsible human resources roles.

Brandon C. Bethards has served as our President and Chief Executive Officer since November 2008, having previously served as Interim Chief Executive Officer since September 2008. He joined Babcock & Wilcox Power Generation Group, Inc. ("B&W PGG") in the early 1970s and served most recently as President of B&W PPG from January 2007 to October 2008 and Senior Vice President and General Manager of B&W PPG's Fossil Power Division from February 2001 to January 2007. His earlier positions with B&W PPG included Vice President of Business Development, General Manager, District Engineer and Field Service Engineer.

David S. Black will serve as our Chief Accounting Officer upon the effectiveness of the spin-off. Mr. Black has served as our Vice President and Controller since January 2007 and Vice President and Controller of BWXT from September 2003 to January 2007. He joined BWXT in 1991 as General Accounting Manager for the Nuclear Environmental Services Division. Other positions he held with B&W include Financial Services Manager for the ASD Service Center Division, Controller for B&W Federal Services, Inc., and Controller for BWXT Services, Inc.

James D. Canafax will serve as our Senior Vice President, General Counsel and Corporate Secretary upon the effectiveness of the spin-off. Mr. Canafax has been an Assistant General Counsel of McDermott since January 2007. Mr. Canafax joined McDermott in July 2001 as Counsel and Contracts Manager for J. Ray McDermott, Inc. In July 2004, he was named Senior Counsel, Legal Manager and Assistant Secretary for J. Ray McDermott, Inc. and served in this position through December 2006. Prior to joining J. Ray McDermott, Inc., he was with Jones, Walker, Waechter, Poitevent, Carrère & Denègre L.L.P.

S. Robert Cochran has served as President of Babcock & Wilcox Technical Services Group, Inc. since July 2006. He joined our company as Senior Vice President of Strategic Development for BWXT, in which position he served from June 2005 through July 2006. Prior to joining B&W, Mr. Cochran served as Senior Vice President of Tyco Infrastructure and President of Kaiser Group International, Inc.

Richard L. Killion was named President and Chief Operating Officer of B&W PGG in November 2008. He previously served as Interim President of B&W PGG since September 2008. He joined B&W in 1970 and previously served as Vice President and General Manager of B&W PGG's Fossil Power Division, where he served from May 2007 to September 2008, Vice President of B&W's Fossil Steam Generating Systems from June 2006 to May 2007 and Director of Global Joint Ventures from May 2002 to June 2006. His earlier positions with B&W PGG include General Manager of Babcock & Wilcox Beijing Company Ltd.

Christofer M. Mowry has served as President of Babcock & Wilcox Nuclear Energy, Inc. since April 2010. From June 2009 until April 2010 he served as President of Babcock & Wilcox Modular Nuclear Energy, LLC, which is now a subsidiary of Babcock & Wilcox Nuclear Energy, Inc. He joined B&W in August 2008, serving as Senior Vice President and Chief Business Development Officer until June 2009. Prior to joining B&W, Mr. Mowry served as President and Chief Operating Officer of Welding Services, Inc., an energy services company located in Atlanta, Georgia, from January 2005 through June 2008. Previously, he held various global management positions with GE Energy from June 1995 until January 2005, after spending 11 years with the utility PECO Energy.

Winfred D. Nash has served as President of Babcock & Wilcox Nuclear Operations Group, Inc. ("B&W NOG") since November 2007. He joined B&W in the late 1960s and previously served as President of our Nuclear Operations Division from September 2006 to November 2007 and as Vice President and General Manager of BWXT from March 2001 to September 2006. His earlier positions within our Nuclear Government Operations group included a major leadership role with Operations, Quality Control and Program Management.

Mary P. Salomone has served as our Chief Operating Officer since January 2010. Most recently she served as Manager of Business Development for B&W NOG from January 2009 until January 2010. She also served as Manager of Strategic Acquisitions for B&W NOG from January 2008 to January 2009. From 1998 through December 2007, Ms. Salomone served as an officer of Marine Mechanical Corporation, which was acquired by us in May 2007, including as President and Chief Executive Officer from 2001 through December 2007. Ms. Salomone previously served with two of B&W's operating divisions, Nuclear Equipment Division and Fossil Power Division, from 1982 until 1998, in a variety of positions. These positions included Manager of Navy Contracts, Project Manager and Manager of Quality Assurance Engineering.

Michael S. Taff will serve as our Senior Vice President and Chief Financial Officer upon the effectiveness of the spin-off. Mr. Taff has served as Senior Vice President and Chief Financial Officer of McDermott since April 2007. He joined McDermott in June 2005 as its Vice President and Chief Accounting Officer. Prior to joining McDermott, Mr. Taff served as Vice President and Chief Financial Officer of HMT Inc., an engineering

and construction company, from June 2004 to June 2005 and as Vice President and Corporate Controller of Philip Services Corporation, a provider of industrial, environmental, transportation and container services, from September 1994 to May 2004.

Board of Directors

As of the distribution date, we expect that our board of directors will consist of 7 directors. The New York Stock Exchange requires that a majority of our board of directors qualify as "independent" according to the rules and regulations of the SEC and the New York Stock Exchange. We intend to comply with those requirements.

Upon completion of the spin-off, our directors will be divided into three classes serving staggered three-year terms. Class I directors will have an initial term expiring in 2011, Class II directors will have an initial term expiring in 2012 and Class III directors will have an initial term expiring in 2013. Class I will be comprised of Messrs. Bethards, Kingsley and McWilliams, Class II will be comprised of Messrs. Goldman and Hanks, and Class III will be comprised of Mr. Fees and Admiral Mies.

At each annual meeting of stockholders, directors will be elected to succeed the class of directors whose terms have expired. This classification of our board of directors could have the effect of increasing the length of time necessary to change the composition of a majority of the board of directors. Following this classification of the board, in general, at least two annual meetings of stockholders will be necessary for stockholders to effect a change in a majority of the members of the board of directors.

Committees of Our Board of Directors

Upon completion of the spin-off, the committees of our board of directors are expected to consist of an Audit Committee, a Governance Committee, a Compensation Committee and an Environmental, Health and Safety Committee.

Each of the Audit, Governance and Compensation Committees will be comprised entirely of independent nonmanagement directors. Our board of directors will adopt a written charter for each committee, which will be posted on our Web site prior to the distribution date. The expected members of the committees are identified in the following table.

| | Board Committee | | | | | | | | | | | |
|-------------------------|-----------------|--------------|------------|-------------------------------------|--|--|--|--|--|--|--|--|
| <u>Director</u> | Audit | Compensation | Governance | Environmental, Health and Safety | | | | | | | | |
| John A. Fees | | | | | | | | | | | | |
| Robert W. Goldman | | | | | | | | | | | | |
| Stephen G. Hanks | | | | | | | | | | | | |
| Oliver D. Kingsley, Jr. | | | | | | | | | | | | |
| D. Brad McWilliams | | | | | | | | | | | | |
| Richard W. Mies | | | | | | | | | | | | |

We expect that Mr. Goldman will be the chairman of the Audit Committee; Mr. Kingsley will be the chairman of the Compensation Committee; Mr. McWilliams will be the chairman of the Governance Committee; and Admiral Mies will be the chairman of the Environmental, Health and Safety Committee.

Audit Committee. The Audit Committee's role will be financial oversight. Our management will be responsible for preparing financial statements, and our independent registered public accounting firm will be responsible for auditing those financial statements. The Audit Committee will not be providing any expert or special assurance as to our financial statements or any professional certification as to the independent registered public accounting firm's work.

The Audit Committee will be directly responsible for the appointment, compensation, retention and oversight of our independent registered public accounting firm. The committee, among other things, will also

review and discuss our audited financial statements with management and the independent registered public accounting firm.

Our board has determined that Messrs. Goldman, Hanks, Kingsley and McWilliams and Admiral Mies each qualify as an "audit committee financial expert" within the definition established by the SEC.

Compensation Committee. The Compensation Committee will have overall responsibility for our officer compensation plans, policies and programs and will have the authority to engage and terminate any compensation consultant or other advisors to assist the committee in the discharge of its responsibilities. Please see the "Executive Compensation—Compensation Discussion and Analysis" for information about our executive officer compensation.

Governance Committee. The Governance Committee, in addition to other matters, will recommend to our board of directors: (1) the qualifications, term limits and nomination and election procedures relating to our directors; (2) nominees for election to our board of directors; and (3) compensation of nonmanagement directors.

The Governance Committee has determined that a candidate for election to our board of directors must meet specific minimum qualifications. The Governance Committee considers each candidate's:

- · record of integrity and ethics in his/her personal and professional life;
- · record of professional accomplishment in his/her field;
- · preparedness to represent the best interests of our stockholders;
- · personal, financial or professional interest in any competitor of ours; and
- preparedness to participate fully in board activities, including active membership on at least one board committee and attendance at, and active participation in, meetings of the board and the committee(s) of which he or she is a member, and lack of other personal or professional commitments that would, in the Governance Committee's sole judgment, interfere with or limit his or her ability to do so.

In addition, the Governance Committee also considers it desirable that candidates possess the following qualities or skills:

- · each candidate should contribute positively to the collaborative culture among board members; and
- · each candidate should possess professional and personal experiences and expertise relevant to our businesses and industries.

While B&W does not have a specific policy addressing board diversity, the board recognizes the benefits of a diversified board and believes that any search for potential director candidates should consider diversity as to gender, ethnic background and personal and professional experiences.

The Governance Committee will solicit ideas for possible candidates from a number of sources—including members of the board, our senior level executives and individuals personally known to the members of the board.

Any stockholder may nominate one or more persons for election as one of our directors at an annual meeting of stockholders if the stockholder complies with the notice, information and consent provisions contained in our bylaws.

The Governance Committee will consider candidates identified through the processes described above and will evaluate each of them, including incumbents, based on the same criteria. The Governance Committee will also take into account the contributions of incumbent directors as board members and the benefits to us arising from their experience on the board. Although the Governance Committee will consider candidates identified by stockholders, the Governance Committee has sole discretion whether to recommend those candidates to the board.

Each expected B&W director brings a strong and unique background and set of skills to the Board, giving the Board as a whole competence and experience in a wide variety of areas. The following experiences and qualifications were considered in concluding each individual would be an appropriate member of the board of directors of B&W:

Mr. Bethards has been affiliated with B&W for 36 years, having started with the predecessor to B&W PGG in the early 1970s. He has held a number of management positions within the B&W organization, including most recently as President and Chief Executive Officer. His tenure with the company, experience in the power industry and in depth knowledge of the operations and culture of B&W make him highly qualified to serve our stockholders.

Mr. Fees' service as Chief Executive Officer of McDermott and member of McDermott's board of directors makes him well qualified to serve as Chairman of the Board of B&W. Prior to becoming McDermott's Chief Executive Officer in 2008, Mr. Fees led a distinguished career at B&W for nearly 30 years. During his time with B&W, Mr. Fees held numerous management positions within B&W's Government Operations segment, including President and Chief Operating Officer, and served as B&W's President and Chief Executive Officer. Mr. Fees' experiences provide him with extensive knowledge of B&W's businesses and strong leadership skills. Mr. Fees also currently serves on the Board of Directors of the Nuclear Energy Institute.

Mr. Goldman has an extensive background in corporate finance. While serving as Chief Financial Officer of Conoco, Inc., Mr. Goldman played vital roles in the initial public offering and split-off of Conoco, Inc. from DuPont and the subsequent merger of Conoco and Phillips Petroleum. Mr. Goldman chairs the finance committee of El Paso Corporation and the governance committee of Tesoro Corporation while serving on the audit committees of both companies and the compensation committee of Parker Drilling Company. He is a member of the Executive Committee of the Board of Trustees of Kenyon College and chairs its Budget, Finance and Audit Committee. He also chaired McDermott's Finance Committee and served as a member of its Audit Committee. This experience will benefit him as the expected Chairman of B&W's Audit Committee and as a member of the Audit Committee.

Mr. Hanks served as Chief Executive Officer and a member of the board of Washington Group International for eight years. That company was acquired by URS Corporation, a publicly traded engineering, construction and technical services firm and one of the companies in our Custom Peer Group. He has extensive financial experience, having served as Chief Financial Officer of Morrison Knudsen Corporation, and he currently serves on the audit committee of Lincoln Electric Holdings, Inc.

Mr. Kingsley brings vast knowledge of the nuclear power generation business to the board of B&W. He serves on the board of FPL, Inc., a provider of electricity related services, and previously served as President and Chief Operating Officer of Exelon Corporation, an integrated utility company and a customer and supplier to one of our principal operating subsidiaries. He previously served as President of the World Association of Nuclear Operators. Throughout his career, Mr. Kingsley has had direct operating and oversight responsibility for major U.S. nuclear power plants. He is a member of the National Academy of Engineering and has received numerous awards given in the United States for nuclear excellence and achievement. His experience in the field of nuclear power operations, maintenance and engineering is valuable to us as a strategic resource for our operations. He currently serves on the nuclear and audit committees of FPL, Inc.

Mr. McWilliams has an extensive background in public accounting and is a Certified Public Accountant and licensed attorney. He served as Chief Financial Officer of Cooper Industries for nine years and in other financial and legal functions for over 22 years. In addition, he served for 12 years as chairman of the audit committee of another public company.

Admiral Mies' distinguished career as a nuclear submariner in the U.S. Navy and as a former Senior Vice President of Science Applications International Corporation has provided him with extensive experience in

nuclear operations, executive leadership and U.S. Government procurement activities. He served as the Senior Operations Commander of the U.S. Submarine Force and as the Commander in Chief of U.S. Strategic Command. He brings to the Board an extensive understanding of the U.S. Government, the single largest customer of B&W. He serves on the boards of Exelon Corporation and Mutual of Omaha and has extensive audit and governance committee experience.

Our Governance Committee will have primary oversight responsibility for our compliance and ethics program, excluding certain oversight responsibilities assigned to the Audit Committee. In conjunction with the Compensation Committee, the Governance Committee will oversee the annual evaluation of our Chief Executive Officer.

Director Compensation

Our nonemployee directors will receive compensation for their service on the board. We expect the cash compensation for our nonemployee directors will consist of the following:

- · an annual retainer of \$45,000 (prorated for partial terms); and
- a fee of \$2,500 for each board meeting personally attended, \$1,750 for each committee meeting personally attended and \$1,000 for each board and committee meeting attended by telephone.

We expect the chairs of board committees, the Non-Executive Chairman and the Lead Director will receive additional annual retainers as follows (pro-rated for partial terms):

- the chair of the Audit Committee: \$20,000;
- the chair of the Compensation Committee: \$15,000;
- the chair of the Governance Committee: \$10,000;
- the chair of the Environmental, Health and Safety Committee: \$15,000;
- the Non-Executive Chairman: \$175,000; and
- the Lead Director: \$10,000.

In addition, we also expect our nonemployee directors will receive annual stock-based awards with a value of \$110,000, subject to future adjustments as the Governance Committee and our board of directors may approve. In addition, Mr. Fees will receive a grant of stock-based awards with a value of \$110,000 following the completion of the spin-off.

Our nonemployee directors will also be reimbursed for any expenses associated with attending board or committee meetings.

Executive Compensation

Our executive compensation program is described in "Executive Compensation—Compensation Discussion and Analysis."

Compensation Policies and Practices and Risk

The McDermott Compensation Committee has concluded that risks arising from McDermott's compensation policies and practices for McDermott employees (including our employees) are not reasonably

likely to have a materially adverse effect on McDermott. In reaching this conclusion, the McDermott Compensation Committee considered the policies and practices in the following paragraph.

The McDermott Compensation Committee regularly reviews the design of significant compensation programs with the assistance of its compensation consultant. We believe these compensation programs motivate and retain our employees while allowing for appropriate levels of business risk through the following features:

- Reasonable Compensation Programs—Using the elements of total direct compensation, the McDermott Compensation Committee seeks to provide compensation opportunities for employees targeted at or near the median compensation of comparable positions in the market. As a result, we believe the total direct compensation of employees provides a reasonable and appropriate mix of cash and equity, annual and longer-term incentives, and performance metrics
- Emphasize Long-Term Incentive Over Annual Incentive Compensation—Long-term incentive compensation typically makes up a larger percentage of a participating employee's target total direct compensation than annual incentive compensation. Incentive compensation helps drive performance and align the interests of employees with those of shareholders. However, tying a significant portion of an employee's total direct compensation to long-term incentives (which typically vest over a period of three or more years) also helps to promote longer-term perspectives regarding company performance.
- Long-Term Incentive Compensation Subject to Forfeiture—The McDermott Compensation Committee may terminate any outstanding stock award if the recipient (1) is convicted of a misdemeanor involving fraud, dishonesty or moral turpitude or a felony, or (2) engages in conduct that adversely affects or may reasonably be expected to adversely affect McDermott's business reputation or economic interests.
- Linear and Capped Incentive Compensation Payouts—The McDermott Compensation Committee establishes financial performance goals which are used to plot a linear payout formula for annual and long-term incentive compensation, eliminating payout "cliffs" between the established performance goals. The maximum payout for both the annual and long-term incentive compensation is capped at 200% percent of target.
- Use of Multiple and Appropriate Performance Metrics—Utilizing diversified performance measures helps prevent compensation opportunities from being overly weighted toward the performance result of a single measure. In general, McDermott's incentive programs are historically based on a mix of financial and individual goals. Additionally, McDermott's primary financial performance metric has been a mix of consolidated and segment operating income. Compared to other financial metrics, operating income is a measure of the profitability of McDermott's business, which helps drive accountability at McDermott's operating segments, thereby reducing risks related to incentive compensation by putting the focus on quality of revenues not quantity.
- Stock Ownership Guidelines—McDermott's executive officers and directors are subject to share ownership guidelines, which also helps promote longer-term perspectives and align the interests of McDermott's executive officers and directors with those of McDermott's shareholders.

We expect that the B&W Compensation Committee will establish similar policies and practices.

Compensation Committee Interlocks and Insider Participation

None of our executive officers have served as members of a compensation committee (or if no committee performs that function, the board of directors) of any other entity that has an executive officer serving as a member of our board of directors.

Security Ownership of Executive Officers and Directors

We are currently wholly owned by McDermott. Immediately following the spin-off, McDermott will not own any shares of our common stock. None of our officers or directors own any shares of our common stock, but

those who own unrestricted shares of McDermott common stock will be treated the same as any other holder of McDermott common stock in any distribution by McDermott and, accordingly, will receive shares of our common stock in the spin-off. As described under "—Treatment of Stock-Based Awards," our executive officers and directors will receive stock options, shares of restricted stock, restricted stock units and deferred stock units in respect of our common stock following the closing of the spin-off.

The following table sets forth the McDermott common stock, stock options, shares of restricted stock, restricted stock units and deferred stock units held by our directors and our executive officers, as of [], and the number of shares of our common stock that will be held by our directors and our executive officers immediately upon completion of the spin-off, assuming there are no changes in each person's holdings of McDermott securities since [] and based on our estimates as of [assuming the distribution of one share of our common stock for every [shares of McDermott common stock, with no fractional shares:

> McDermott Restricted Stock. Restricted Stock Units McDermott Common McDermott and Deferred Stock Owned (1) Stock Options(2) Stock Units(3)

Shares of B&W Common Stock(4)

Name John A. Fees Robert W. Goldman Stephen G. Hanks Oliver D. Kingsley, Jr. D. Brad McWilliams Richard W. Mies Brandon C. Bethards Michael S. Taff Richard L. Killion Christofer M. Mowry Winfred D. Nash All directors and executive officers (16 individuals)

- Each director and executive officer holds less than 1% of McDermott's outstanding common stock.
- Reflects shares of McDermott common stock that can be acquired within 60 days upon the exercise of stock options.
- Reflects shares of restricted stock. Also includes shares underlying restricted stock units that can be acquired within 60 days.

 The number of B&W shares listed in the table reflects the substitution of B&W equity awards in the spin-off in respect of options to purchase shares of McDermott stock, McDermott shares of restricted stock units and deferred stock units and within 60 days upon the exercise of stock options; shares of restricted stock; and shares underlying restricted stock units and deferred stock units that can be acquired within 60 days.

EXECUTIVE COMPENSATION

Compensation Discussion and Analysis

The following Compensation Discussion and Analysis, or CD&A, provides information relevant to understanding the 2009 compensation of our executive officers identified in the Summary Compensation Table below under the caption "Compensation of Executive Officers," whom we refer to as our Named Executives. The following discussion also contains statements regarding future individual and company performance targets and goals. These targets and goals are disclosed in the limited context of our compensation programs and should not be understood to be statements of management's expectations or estimates of results or other guidance. We caution investors not to apply these statements to other contexts.

Summary

The Babcock & Wilcox Company is currently a wholly owned subsidiary of McDermott, which will, following the spin-off, own the subsidiaries that currently conduct the operations of McDermott's Power Generation Systems and Government Operations segments. For purposes of this CD&A, unless the context otherwise indicates or requires, we use "B&W" to refer to McDermott's Power Generation Systems and Government Operations segments, collectively. Our executive compensation program has been designed by the Compensation Committee of McDermott's Board of Directors ("McDermott's Compensation Committee"). Historically, including during 2009, McDermott's senior management, in consultation with our senior management, has supported McDermott's Compensation Committee in administering the compensation matters related to our Named Executives. Following the spin-off, our executive compensation program will be administered by the Compensation Committee of our Board of Directors (the "B&W Compensation Committee"). The responsibilities of the B&W Compensation Committee are described in "Management—Committees of Our Board of Directors" above.

For purposes of this CD&A, we refer to the following persons as our Named Executives:

- Brandon C. Bethards, President and Chief Executive Officer, B&W
- Michael S. Taff, Senior Vice President and Chief Financial Officer, McDermott and B&W
- · Richard L. Killion, President and Chief Operating Officer, Babcock & Wilcox Power Generation Group, Inc.
- Christofer M. Mowry, President, B&W Nuclear Energy, Inc.
- Winfred D. Nash, President, Babcock & Wilcox Nuclear Operations Group, Inc.

Prior to the spin-off, all of our Named Executives except for Mr. Taff have been employees of B&W. Mr. Taff is an employee of McDermott Incorporated, a wholly owned subsidiary of McDermott, and serves as the Senior Vice President and Chief Financial Officer of both McDermott and B&W. In connection with the spin-off, he is expected to become an employee of B&W and continue as its Senior Vice President and Chief Financial Officer. Since this CD&A focuses primarily on B&W's compensation programs and decisions affecting 2009 compensation of our Named Executives, compensation information for Mr. Taff relates to his role as an officer of McDermott.

Overview of Compensation Programs and Objectives

Philosophy and Objectives. Our compensation programs are based on McDermott's belief that the ability to attract, retain and motivate qualified employees to develop, expand and execute sound business opportunities throughout McDermott and its subsidiaries, including B&W, is essential to its success. To that end, McDermott's Compensation Committee, with the assistance of its compensation consultant, designs and administers compensation programs with the participation of McDermott's and our management that generally seek to provide compensation that:

- · incentivizes and rewards short- and long-term performance, continuity of service and individual contributions; and
- · promotes retention of well-qualified executives, while aligning the interests of our executives with those of McDermott's shareholders.

For 2009, McDermott's Compensation Committee targeted reasonable and competitive compensation for our Named Executives, with a significant portion of that compensation being performance-based. Specifically, for our Named Executives:

- 2009 target total direct compensation was within 10%-15% of the median compensation of officers in comparable positions in the market applicable to them:
- 60% of target long-term incentive compensation for 2009 was performance-based; and
- performance-based compensation accounted for over 50% of 2009 target total direct compensation.

The performance-based compensation that paid out in 2009 also reflected the commitment of McDermott's Compensation Committee to pay for performance. B&W's cumulative operating income of over \$1.1 billion for the three-year period ended December 31, 2009 represents the largest amount of operating income in any three-year period since B&W was acquired by McDermott in 1978. As a result, our Named Executives earned payments at the maximum level under performance shares granted in 2006, which vested in 2009. Our Named Executives were also eligible to earn near the maximum annual incentive compensation payout based on McDermott's consolidated and combined Power Generation Systems and Government Operations segments, as applicable, operating income for 2009. As designed by McDermott's Compensation Committee, individual performance helped differentiate among the Named Executives' annual incentive compensation payouts for 2009.

Compensation Consultant. In designing its executive compensation program, McDermott's Compensation Committee engaged an outside compensation consultant, Hewitt Associates, LLC, or Hewitt. Hewitt has served as the consultant to McDermott's Compensation Committee on executive and director compensation since October 2007. During 2009, Hewitt advised McDermott's Compensation Committee on all principal elements of McDermott's compensation programs and attended meetings of McDermott's Compensation Committee and participated in executive sessions without members of McDermott's management present. In 2009, Hewitt provided advice and analysis on the design, structure and level of executive and director compensation. It also provided McDermott's Governance Committee and its Board advice and analysis regarding nonmanagement director compensation. In connection with Hewitt's services to McDermott's Compensation Committee, in 2009 Hewitt sought and received input from McDermott's executive management on various matters and worked with McDermott's executive management to formalize proposals for McDermott's Compensation Committee. In 2009, Hewitt did not perform any other services for McDermott or us, other than assisting McDermott's management in preparing the performance graph included in McDermott's annual report on Form 10-K. Hewitt had been providing that service to McDermott's management for several years prior to serving as McDermott's Compensation Committee's consultant, and the fees for that service amounted to less than \$2,000 in 2009. Following the spin-off, we expect that our senior management will continue to support the administration of the compensation programs on behalf of B&W's Compensation Committee in a manner and scope similar to that which has been provided by McDermott's management.

On February 1, 2010, Hewitt spun-off its executive compensation business into a separate company known as Meridian Compensation Partners LLC. Since the spin-off, Meridian, rather than Hewitt, has been advising McDermott's Compensation Committee with respect to the matters described above.

Elements. With the objectives outlined above in mind, McDermott's Compensation Committee approves annual compensation for Named Executives principally consisting of the following three elements:

- annual base salary;
- · annual incentives; and
- · long-term incentives.

Collectively, we refer to these elements as the "total direct compensation" of a Named Executive.

Annual base salary provides a fixed level of compensation that helps attract and retain highly qualified executives. Annual incentive and long-term incentive compensation are the principal performance-based components of a Named Executive's compensation. The annual incentive element is cash-based compensation generally designed to incentivize a Named Executive to achieve performance goals relative to the then-current fiscal year. Historically, long-term incentives have generally been in the form of McDermott equity-based awards and were designed to retain and closely align the interests of Named Executives with McDermott's shareholders. Similar to annual incentive compensation, performance-based long-term incentive compensation is designed to promote the achievement of performance goals, only over a longer period—typically of three or more years.

As we discuss in more detail below, McDermott's Compensation Committee also administers several plans as part of our post-employment compensation arrangements designed to reward long-term service and performance.

Target Total Direct Compensation. McDermott's Compensation Committee seeks to provide reasonable and competitive compensation. As a result, it targets the elements of total direct compensation for our Named Executives generally within 10%-15% of the median compensation of our market for comparable positions. Throughout this CD&A, we refer to compensation that is within 10%-15% of market median as "market range" compensation.

McDermott's Compensation Committee may set elements of total direct compensation above or below the market range to account for a Named Executive's performance and experience, and other factors or situations that are not typically captured by looking at standard market data and practices that McDermott's Compensation Committee deems relevant to the appropriateness and/or competitiveness of a Named Executive's compensation.

When making decisions regarding individual compensation elements, McDermott's Compensation Committee also considers the effect on the Named Executive's target total direct compensation and target total cash-based compensation (annual base salary and annual incentives), as applicable. The goal of McDermott's Compensation Committee is to establish target compensation for each element it considers appropriate to support the compensation objectives that, when combined, create a target total direct compensation award for each Named Executive that is reasonable and competitive.

Defining Market Range Compensation—Benchmarking. To identify median compensation for each element of total direct compensation, McDermott's Compensation Committee relies on "benchmarking"—reviewing the compensation of our Named Executives relative to the compensation paid to similarly situated executives at companies we consider our peers. As a result, the annual base salary, target annual incentive compensation and target long-term incentive compensation as a whole for each of the Named Executives is benchmarked. However, the specific performance metrics and performance levels used within elements of annual and long-term incentive compensation are designed for the principal purpose of supporting our strategic and financial goals and/or driving the creation of shareholder value, and, as a result, are not generally benchmarked.

At the request of McDermott's Compensation Committee, Hewitt conducted a market compensation analysis and provided advice regarding the median compensation of the three elements of total direct compensation for McDermott's officers, including the Named Executives. Using 2008 survey data from its proprietary compensation database (adjusted approximately 2.9% upwards by Hewitt for the purpose of bringing the market data current), Hewitt collected information from companies generally reflecting the size, scope and complexity of the business and executive talent at McDermott and B&W. To account for the size of our operations relative to peer companies, Hewitt used regression analysis to adjust the market information on peer companies based on revenue. To account for the diversity of geography and industry among McDermott's operations, Hewitt analyzed information from two principal peer groups, the Corporate Group and the Babcock & Wilcox Group, In this CD&A references to "market" or "our market" are references to the compensation of companies within the Corporate Group or Babcock & Wilcox Group, as applicable.

Babcock & Wilcox Group. With assistance from McDermott's management, Hewitt compiled this group as the primary benchmark for McDermott's executives at its Power Generation Systems and Government Operations segments, including Messrs. Bethards, Killion, Mowry and Nash. The group consisted of 31 engineering, construction and/or governments operations companies that are representative of our Power Generation Systems and Government Operations segments. The component companies within this group included:

Ameron International Corporation Chicago Bridge & Iron Company N.V.

Cooper Industries Plc
Curtiss-Wright Corporation
Dover Corporation
Eaton Corporation
ESCO Technologies Inc.
Flowserve Corporation
Foster Wheeler AG

General Dynamics Corporation Granite Construction, Inc.

Honeywell International, Inc.

Hubbell, Inc.

Illinois Tool Works Inc. Ingersoll-Rand Co. Ltd. ITT Corporation Joy Global, Inc.

Lockheed Martin Corporation Martin Marietta Materials, Inc. Northrop Grumman Corporation

Raytheon Company Rockwell Collins, Inc.

The Shaw Group Inc. Terex Corporation Textron, Inc.

Thomas & Betts Corporation USG Corporation Valmont Industries, Inc. Vulcan Materials Company Walter Industries Inc.

Washington Group International Inc.

Corporate Group. With assistance from our management, Hewitt compiled this group as the primary benchmark for McDermott's executives at its corporate segment, including Mr. Taff. The group consisted of the companies comprising the Babcock & Wilcox Group in addition to 13 additional companies with operations in engineering, construction, and/or energy that constitute a mixture of companies that are representative of McDermott's three operating segments: Offshore Oil & Gas Construction, Power Generation Systems and Government Operations. The component companies of this group included:

Alliant Techsystems Inc. Ameron International Corporation Anadarko Petroleum Corporation

Baker Hughes, Inc.
BJ Services Company
Cameron International, Inc.
Chesapeake Energy Corporation.
Chicago Bridge & Iron Company N.V.
Cooper Industries Plc

Curtiss-Wright Corporation Devon Energy Corporation Dover Corporation Eaton Corporation

El Paso Corporation ESCO Technologies Inc. Flowserve Corporation FMC Technologies, Inc. Foster Wheeler AG

General Dynamics Corporation Granite Construction Incorporated

Halliburton Company Honeywell International, Inc.

Hubbell Inc.

Illinois Tool Works Inc. Ingersoll-Rand Plc ITT Corporation Joy Global Inc. KBR, Inc.

Lockheed Martin Corporation Martin Marietta Materials, Inc. Noble Corporation

Northrop Grumman Corporation Pioneer Natural Resources Company

Raytheon Company Rockwell Collins, Inc. The Shaw Group Inc. The Williams Companies, Inc.

Terex Corporation

Textron Inc.

Thomas & Betts Corporation

USG Corporation
Valmont Industries, Inc.
Vulcan Materials Company
Walter Industries, Inc.

In addition to these peer groups, Hewitt supplemented the market data with other publicly available compensation information related to companies in a peer group identified by McDermott's management and Hewitt in October 2007, which we refer to as the "Custom Peer Group." The Custom Peer Group consists of nine engineering and construction companies and is the same group McDermott uses in the performance graph included in its annual report on Form 10-K. The Custom Peer Group is comprised of the following companies: Cal Dive International, Inc.; Chicago Bridge & Iron Company N.V.; Fluor Corporation; Foster Wheeler AG; Jacobs Engineering Group Inc.; KBR, Inc.; Oceaneering International, Inc.; The Shaw Group Inc.; and URS Corporation. Of the Custom Peer Group companies, Chicago Bridge & Iron Company N.V., Foster Wheeler AG and The Shaw Group Inc. are also companies within the Corporate Group and Babcock & Wilcox Group, while KBR, Inc. is also a member of the Corporate Group. Compensation information for Custom Peer Group companies was based on information reported by those companies in publicly available Securities and Exchange Commission filings. The information available was largely limited to the five highest paid positions at the company and generally based on 2007 compensation. As a result, in 2009, the Custom Peer Group was not weighted in the calculation to determine median compensation, except to the extent any of its component companies was also a component company in the Babcock & Wilcox Group or the Corporate Group. Accordingly, McDermott's Compensation Committee relied on the Corporate Group and the Babcock & Wilcox Group, as applicable, to determine the market range of 2009 compensation for our Named Executives, and the Custom Peer Group had no effect on the market range of compensation.

Total Direct Compensation

2009 Overview. The 2009 target total direct compensation for each of our Named Executives was within the market range of target total direct compensation. The chart below shows the 2009 target total direct compensation by element for each Named Executive. Because some compensation elements are performance-based, Named Executives are capable of earning compensation above or below the market range for similarly situated executives in our market.

2009 Target Total Direct Compensation Summary

| | | | Annual | Long-Term | Total Direct |
|-----------------|-------|---------------|---------------|---------------|-----------------|
| | | | Incentive* | Incentive* | Compensation* |
| Named Executive | Annua | l Base Salary | (% of Salary) | (% of Salary) | (% of Market**) |
| B.C. Bethards | \$ | 526,200 | 70% | 286% | 98% |
| M.S. Taff | \$ | 505,000 | 70% | 348% | 102% |
| R. L. Killion | \$ | 330,000 | 55% | 168% | 99% |
| C.M. Mowry | \$ | 355,000 | 45% | 100% | 108% |
| W.D. Nash | \$ | 348,000 | 50% | 144% | 111% |

^{*} When making decisions as to the elements of a Named Executive's total direct compensation, McDermott's Compensation Committee considers the dollar value of annual incentive and long-term incentive compensation, but typically awards these elements as percentages of annual base salary. This is primarily because McDermott's market generally targets these elements on a percentage-of-salary basis. See "— Long-Term Incentive Compensation—Analysis of 2009 Equity Grants" below for a discussion of how equity grants are valued.

While McDermott's Compensation Committee does not set a specific target allocation among the elements of total direct compensation, it believes that a significant portion of a Named Executive's total direct compensation should be performance-based. On average, performance-based compensation accounted for approximately 51% of a Named Executive's 2009 target total direct compensation and 60% of his long-term incentive compensation.

Annual Base Salary

2009 Base Salaries. In January 2009, McDermott's Compensation Committee was provided with (1) the recommendation of McDermott's Chief Executive Officer as to 2009 base salaries for Messrs. Bethards and Taff,

^{**} Market = Median target annual incentive compensation based on the benchmark applicable to the executive. 100% represents median compensation.

(2) the recommendation of Mr. Bethards, made in consultation with McDermott's Chief Executive Officer, as to the 2009 base salaries for Messrs. Killion, Mowry and Nash, and (3) Hewitt's analysis of market compensation (as adjusted by Hewitt to January 2009). Hewitt's market analysis compared the recommended base salary and the target total direct compensation of each officer (assuming 2009 salary recommendations were approved) to the median compensation of the applicable market.

The 2009 base salaries for our Named Executives were as follows:

| | 2009 Annual | |
|-----------------|-------------|----------------------|
| Named Executive | Base Salary | Percent of Market(1) |
| B.C. Bethards | \$ 526,200 | 100% |
| M.S. Taff | \$ 505,000 | 88% |
| R.L. Killion | \$ 330,000 | 95% |
| C.M. Mowry | \$ 355,000 | 108% |
| W.D. Nash | \$ 348,000 | 110% |

(1) Market = Median annual base salary based on the benchmark applicable to the executive. 100% represents median compensation

McDermott's Compensation Committee sought to, and we expect B&W's Compensation Committee will continue to, position each Named Executive in the market range (+/- 10-15%) of median base salaries indicated by market data. In addition to market data, in response to the slowdown in global economic conditions that began in 2008, McDermott's Compensation Committee sought to limit increases in 2009 base salaries for its officers, including our Named Executives, to an average increase of 3-4%, with limited exceptions based on individual circumstances, in line with the median increase among companies in our market. Mr. Bethards' base salary did not change because he received a substantial increase in October 2008 in connection with his promotion and McDermott's Compensation Committee believed his salary was appropriate relative to the market. Mr. Taff, whose base salary was substantially below market in 2008, received an above-average increase in base salary to bring his base salary within the market range for his position. Messrs. Killion and Nash each received a slightly higher than average increase in base salary to reflect, in Mr. Killion's case, internal pay equity among officers, and, in Mr. Nash's case, the increased scope of operations for which he was responsible as a result of B&W's acquisition of Nuclear Fuel Services, Inc. in December 2008.

Annual Incentive Compensation

2009 Overview and Target Compensation. McDermott's Compensation Committee administers the annual incentive compensation program under its Executive Incentive Compensation Plan, which we refer to as the EICP.

The EICP is a cash incentive plan designed to motivate and reward officers and key employees of McDermott and its subsidiaries, including our Named Executives, for their contributions to business goals and other factors that McDermott believes drive its earnings and/or creates shareholder value.

The 2009 target EICP compensation for our Named Executives was as follows:

| | Target EICP | |
|-----------------|---------------------------|----------------------|
| Named Executive | (% of annual base salary) | Percent of Market(1) |
| B.C. Bethards | 70% | 86% |
| M.S. Taff | 70% | 89% |
| R.L. Killion | 55% | 92% |
| C.M. Mowry | 45% | 105% |
| W.D. Nash | 50% | 94% |

(1) Market = Median target annual incentive compensation based on the benchmark applicable to the executive. 100% represents median compensation

McDermott's Compensation Committee set 2009 target EICP compensation for our Named Executives at the same percentage of base salary as it established for 2008 target EICP compensation, with the exception of

Messrs. Taff and Nash. The 2008 target EICP percentage for those two executives was below or at the limit of the 2009 market range of competitive compensation. As a result, Mr. Taff's 2009 target EICP compensation was increased from 55% of his annual base salary to bring him within the 2009 market range and was set at a level commensurate with other McDermott executive officers. Similarly, Mr. Nash's 2009 target EICP compensation was increased from 45% of his annual base salary to more closely align his 2009 EICP target with market-median annual incentive compensation. Additionally, for Mr. Nash, consideration was also given to the increased scope of operations for which he was responsible as a result of B&W's acquisition of Nuclear Fuel Services, Inc. in December 2008.

2009 EICP Performance Goals. EICP compensation consists of a financial performance component and an individual performance component. The 2009 target EICP was split between these two components as follows:

- 70% of target EICP was attributable to financial performance; and
- 30% of target EICP was attributable to individual performance.

Financial performance is the largest factor in determining EICP compensation because McDermott's Compensation Committee generally considers it to be more objective and to more directly influence the creation of shareholder value, as compared to individual performance. McDermott's Compensation Committee, however, recognizes that individual performance can serve an important role in helping promote the achievement of strategic, non-financial goals. To reward significant individual contributions, McDermott's Compensation Committee increased the representation of the individual component from 15% to 30% of target EICP for 2009, as compared to 2008. However, to maintain the emphasis on financial performance, financial results determined the amount eligible to be paid under EICP compensation (irrespective of the level of individual performance attained). For example, if target financial performance was attained, then a Named Executive would be eligible to receive up to a target payout under the financial component and individual component. No EICP compensation would be earned (including for individual performance) unless the threshold level of financial performance was attained and the maximum EICP compensation a Named Executive could earn in 2009 was 200% of target EICP.

2009 Financial Goals. The 2009 financial goals consisted of operating income levels related to McDermott and/or B&W operations relevant to each Named Executive. In some instances a Named Executive may have operating income goals for more than one operation, in which case the financial goals were weighted toward the operations over which the Named Executive had primary responsibility. McDermott's Compensation Committee considers operating income an appropriate financial measure to use for compensation purposes, because it is the primary driver of net income, which McDermott's Compensation Committee expects to drive McDermott's stock price. In comparison to net income, however, operating income is more directly influenced by the revenues generated and costs incurred as a result of management action and is more readily attributable to McDermott's operating segments.

Within the financial performance component, McDermott's Compensation Committee historically established three levels of operating income goals. These levels would determine the threshold, target and maximum amounts that would be paid under the financial component of the EICP, with target level being based on management's internal projections of operating income and threshold and maximum levels set as a percentage of the target level. McDermott's Compensation Committee designs incentive plans to drive target level performance and does not believe that compensation should be earned for performance substantially below that level. As a result, no EICP compensation would be earned (including for individual performance) unless a threshold level of financial performance was attained. McDermott's Compensation Committee believes that Named Executives should be rewarded for superior financial performance. It therefore establishes a maximum level performance goal to incentivize higher performance, but caps the payout to maximize returns to shareholders for performance above the maximum payout level and reduce risk related to incentive compensation.

The performance range between the threshold level and the maximum level was relatively narrow prior to 2009. For instance, in 2008, the operating income goals ranged between 75% of target operating income at the

threshold payout level and 108% of target operating income at the maximum payout level. With operating income goals set within a range of 33% determining a payout range of 200%, relatively small changes in performance results produced large variations in EICP payouts.

For 2009, McDermott's Compensation Committee sought to increase the performance range between threshold and maximum level goals, making the maximum payment more difficult relative to target and reducing the minimum performance required to be attained before any EICP compensation would be earned. In addition, McDermott's Compensation Committee established a range for target level performance comprised of three separate operating income goals. The intended effect of these changes was to reduce the significance of the impact of minor variations in financial results, thereby reducing the risk related to the EICP.

McDermott's Compensation Committee considered a number of performance goals recommended by management, including threshold level goals as low as 50% of target operating income and maximum level goals as high as 150% of target operating income. With the advice of Hewitt, McDermott's Compensation Committee set the threshold level operating income goal at 70% of target and the maximum level operating income goal at 120% of target. The operating income goals at the target level ranged from 95% of the target goal to 105% of the target goal.

As a result, a Named Executive would have been eligible to earn the following amounts under the 2009 EICP based on attaining the following levels of operating income:

- 25% of target EICP compensation at threshold level operating income;
- 92.5% of target EICP compensation at target (min) level operating income;
- 100% of target EICP compensation at target level operating income;
- 107.5% of target EICP compensation at target (max) level operating income; and
- 200% of target EICP compensation at maximum level operating income.

For other levels of operating income between threshold and maximum, the percentage paid would have been determined by linear interpolation using the two neighboring pre-established performance levels and percentage payout. No payment would have been earned under the EICP for 2009 if operating income results had been below the threshold level.

In 2009, the B&W operating units produced operating income of approximately \$312.4 million, which exceeded the target level performance goal but not the maximum level. B&W's 2009 operating income included expense relating to the mPower™ modular nuclear reactor initiative. Relative to annual/incentive compensation, the mPower™ initiative is longer-term in nature. To avoid adversely impacting a Named Executive's EICP compensation for strategic investments, the performance was adjusted to exclude approximately \$13 million of the mPower™ expenses. As a result, for purposes of calculating 2009 EICP payments, B&W's operating income was approximately \$325.7 million, which exceeded the target level performance goal but not the maximum level.

2009 Individual Goals. Individual goals established for each Named Executive were tailored to the individual's position and focused on supporting strategic initiatives and achieving common goals. McDermott's Chief Executive Officer established the individual goals for Messrs. Bethards and Taff, while the remaining Named Executives proposed their respective individual goals which were approved by Mr. Bethards in consultation with McDermott's Chief Executive Officer. The Named Executives' individual goals and target weightings are set forth in the tables below.

2009 Annual Incentive Compensation. The following is a description of the performance goals and the evaluation relating to each Named Executive's 2009 annual incentive compensation:

Brandon C. Bethards

• Financial Goals. McDermott's Compensation Committee believed that as an executive officer of McDermott during 2009, a portion of Mr. Bethards' annual incentive compensation should be based on McDermott's consolidated financial results. However, as Chief Executive Officer of B&W, he was directly responsible for the operations of McDermott's Power Generation Systems and Government Operations segments. As a result, 50% of his 2009 target EICP was based on the operating income from those two segments and 20% was based on McDermott's consolidated operating income. The financial goals and payout percentage for 2009 are outlined in the table below.

| Performance | Target | | | | | | | |
|--|--------|-----------|---------|---------|---------|----------|---------|------------|
| Measures | EICP | Threshold | | Target | | Maximum | 2009 | Payout |
| (\$ in millions) | | 25% | 92.5% | 100% | 107.5% | 200% | Result | Percentage |
| Operating Income (B&W) | 50% | \$ 191.7 | \$260.2 | \$273.9 | \$287.6 | \$ 328.7 | \$325.7 | 195% |
| Operating Income (McDermott ⁽¹⁾) | 20% | \$ 241.9 | \$328.2 | \$345.5 | \$362.8 | \$ 414.6 | \$546.5 | 200% |

- (1) Consolidated operating income levels equal the sum of McDermott's segment operating income less unallocated corporate operating expenses.
 - Individual Goals. The individual goals and target weighting for Mr. Bethards were as follows:
 - achieve specific levels of health, safety and environmental performance averages relating to three metrics at our Power Generation Systems and Government Operations segments (10%);
 - successfully manage the completion of the initial phase of a strategic global financial implementation project as it relates to B&W entities (10%);
 and
 - achieve successful integration of a specified acquisition as defined by the integration plan milestones (10%).
 - Evaluation of 2009 Performance and Payment of 2009 EICP Compensation. For purposes of evaluating B&W financial performance for 2009, B&W's operating income results for 2009 excluded approximately \$13 million of expense related to B&W's mPower™ initiative. Mr. Bethards earned a 195% payout on the financial component related to B&W operating income and a 200% payout on the financial component related to McDermott operating income, or a weighted blended payout percentage of 196%. Based on these financial results, he was eligible to receive 2009 annual incentive compensation up to 196% of his 2009 target compensation, depending on individual performance. Mr. Bethards met or exceeded his individual goals, except with respect to two goals which in each case were only partially achieved, resulting in the achievement of 21.67% of the 30% target EICP attributable to individual performance. As a result, Mr. Bethards earned 180% of his 2009 target EICP compensation, or \$663,012.

Michael S. Taff

Financial Goals. The financial performance component of Mr. Taff's 2009 annual incentive compensation was based entirely on McDermott's consolidated operating income. The financial goals and payout percentage for 2009 are outlined in the table below.

| Performance | Target | | | | | | | |
|--|--------|-----------|---------|---------|---------|----------|---------|------------|
| Measure | EICP | Threshold | | Target | | Maximum | 2009 | Payout |
| (\$ in millions) | | 25% | 92.5% | 100% | 107.5% | 200% | Actual | Percentage |
| Operating Income (McDermott ⁽¹⁾) | 70% | \$ 241.9 | \$328.2 | \$345.5 | \$362.8 | \$ 414.6 | \$546.5 | 200% |

- (1) Consolidated operating income levels equal the sum of McDermott's segment operating income less unallocated corporate operating expenses.
 - Individual Goals. The individual goals and target weighting for Mr. Taff were as follows:
 - achieve specific levels of consolidated health, safety and environmental performance averages relating to three metrics at McDermott (10%);
 - · develop and implement a plan to address the BWXT credit facility, which was scheduled to mature in 2010 (10%); and
 - develop a strategic multi-year plan regarding information technology (10%).
 - Evaluation of 2009 Performance and Payment of 2009 EICP Compensation. Mr. Taff earned a 200% payout on the financial component related to consolidated McDermott operating income and, depending on his individual performance, was eligible to receive 2009 annual incentive compensation up to 200% of his 2009 target compensation. Mr. Taff met or exceeded all of his individual goals. As a result, he earned 200% of his 2009 target EICP compensation, or \$707,000.

Richard L. Killion

• Financial Goals. Mr. Killion is President and Chief Operating Officer of Babcock & Wilcox Power Generation Group, Inc. ("B&W PGG"), our principal subsidiary under which our Power Generation Systems segment is organized. As an executive officer of B&W, a portion of Mr. Killion's annual incentive compensation was based on B&W's consolidated financial results. However, as President of B&W PGG, he is directly responsible for the operations of B&W's Power Generation Systems segment and, as a result, 40% of his 2009 target EICP was based on that segment's operating income and 30% was based on B&W's combined operating income. The financial goals and payout percentage for 2009 are outlined in the table below.

| Performance | Target | | | | | | | |
|----------------------------|--------|-----------|---------|---------|---------|----------|---------|------------|
| Measures | EICP | Threshold | | Target | | Maximum | 2009 | Payout |
| (\$ in millions) | % | 25% | 92.5% | 100% | 107.5% | 200% | Result | Percentage |
| Operating Income (B&W PGG) | 40% | \$ 113.9 | \$143.0 | \$150.5 | \$158.0 | \$ 176.4 | \$183.3 | 200% |
| Operating Income (B&W) | 30% | \$ 191.7 | \$260.2 | \$273.9 | \$287.6 | \$ 328.7 | \$325.7 | 195% |

- Individual Goals. The individual goals and target weighting for Mr. Killion were as follows:
 - achieve specific levels of health, safety and environmental performance averages relating to three metrics at our Power Generation Systems segment (10%);
 - successfully complete the design and implementation plan for enterprise accounting and purchasing software at B&W PGG (10%); and
 - complete two acquisitions for B&W PGG (10%).

• Evaluation of 2009 Performance and Payment of 2009 EICP Compensation. For purposes of evaluating B&W financial performance for 2009, B&W's operating income results for 2009 excluded approximately \$13 million of expense related to B&W's mPower™ initiative. Mr. Killion earned a 200% payout on the financial component related to B&W PGG operating income and a 195% payout on the financial component related to B&W's combined operating income, or a weighted blended payout percentage of 198%. Based on these financial results, he was eligible to receive 2009 annual incentive compensation up to 198% of his 2009 target compensation, depending on individual performance. Mr. Killion met or exceeded his individual goals, except with respect to one goal which was only partially achieved, resulting in the achievement of 26.67% of the 30% target EICP attributable to individual performance. As a result, Mr. Killion earned 191.3% of his 2009 target EICP compensation, or \$347,210.

Christofer M. Mowry

• Financial Goals. At the time financial and individual performance goals were set for 2009, Mr. Mowry served as Senior Vice President, Business Development for B&W. As a result, his financial performance component of 2009 EICP consisted entirely of B&W's combined operating income. The financial goals and payout percentage for 2009 are outlined in the table below.

| Performance | Target | | | | | | | |
|------------------------|--------|-----------|---------|---------|---------|----------|---------|------------|
| Measures | EICP | Threshold | | Target | | Maximum | 2009 | Payout |
| (\$ in millions) | % | 25% | 92.5% | 100% | 107.5% | 200% | Result | Percentage |
| Operating Income (B&W) | 70% | \$ 191.7 | \$260.2 | \$273.9 | \$287.6 | \$ 328.7 | \$325.7 | 195% |

- Individual Goals. The individual goals and target weighting for Mr. Mowry were as follows:
 - achieve specific levels of health, safety and environmental performance averages relating to three metrics at our Power Generation Systems and Government Operations segments (10%);
 - operate B&W's business development department within approved budget (5%);
 - successfully implement process for B&W's strategic plan (5%); and
 - provide business development leadership to strategic initiatives (10%).
- Evaluation of 2009 Performance and Payment of 2009 EICP Compensation. For purposes of evaluating B&W financial performance for 2009, B&W's operating income results for 2009 excluded approximately \$13 million of expense related to B&W's mPower™ initiative. Mr. Mowry earned a 195% payout on the financial component related to B&W combined operating income and, depending on individual performance, was eligible to receive 2009 annual incentive compensation up to 195% of his 2009 target compensation. Mr. Mowry met or exceeded his individual goals, except with respect to one goal which was only partially achieved, resulting in the achievement of 26.67% of the 30% target EICP attributable to individual performance. As a result, Mr. Mowry earned 188.5% of his 2009 target EICP compensation, or \$301,129.

Winfred D. Nash

• Financial Goals. Mr. Nash is President of Babcock & Wilcox Nuclear Operations Group, Inc. ("B&W NOG"), one of the principal subsidiaries under which our Government Operations segment is organized. As an executive officer of B&W, a portion of Mr. Nash's annual incentive compensation was based on B&W's combined financial results. However, as President of B&W NOG, he is directly responsible for B&W NOG's operations and, as a result, 40% of his 2009 target EICP was based on B&W NOG's operating income and 30% on B&W's combined operating income. The financial goals and payout percentage for 2009 are outlined in the table below.

| Performance | Target | | | | | | | |
|----------------------------|--------|-----------|---------|---------|---------|----------|---------|------------|
| Measures | EICP | Threshold | | Target | | Maximum | 2009 | Payout |
| (\$ in millions) | % | 25% | 92.5% | 100% | 107.5% | 200% | Result | Percentage |
| Operating Income (B&W NOG) | 40% | \$ 94.4 | \$118.6 | \$124.8 | \$131.0 | \$ 143.6 | \$145.5 | 200% |
| Operating Income (B&W) | 30% | \$ 191.7 | \$260.2 | \$273.9 | \$287.6 | \$ 328.7 | \$325.7 | 195% |

- Individual Goals. The individual goals and target weighting for Mr. Nash were as follows:
 - achieve specific levels of health, safety and environmental performance averages relating to three metrics at B&W NOG (10%);
 - successfully develop a financial modeling tool for acquisition analysis (10%); and
 - achieve successful integration of a specified acquisition as defined by the integration plan milestones (10%).
- Evaluation of 2009 Performance and Payment of 2009 EICP Compensation. For purposes of evaluating B&W financial performance for 2009, B&W's operating income results for 2009 excluded approximately \$13 million of expense related to B&W's mPower™ initiative. Mr. Nash earned a 200% payout on the financial component related to B&W's combined operating income, or a weighted blended payout percentage of 198%. Based on these financial results, he was eligible to receive 2009 annual incentive compensation up to 198% of his 2009 target compensation, depending on individual performance. Mr. Nash met or exceeded his individual goals, except with respect to two goals which were only partially achieved, resulting in the achievement of 15% of the 30% target EICP attributable to individual performance. As a result, Mr. Nash earned 168.2% of his 2009 target EICP compensation, or \$292,668.

Long-Term Incentive Compensation

McDermott's Compensation Committee believes that the interests of its shareholders are best served when a significant percentage of compensation is comprised of equity and other long-term incentives that appreciate in value contingent upon increases in the value of McDermott's common stock and other performance measures that reflect improvements in McDermott's business fundamentals. Therefore, long-term incentive compensation represents the single largest element of our Named Executives' total direct compensation.

Analysis of 2009 Equity Grants

Mix of 2009 Equity. In 2006 and 2007, the long-term incentive compensation of our officers, including the Named Executives, consisted entirely of performance shares. These performance shares generally provided for vesting three years from the date of grant in an amount between 0% and 150% of the number of shares granted, depending on the level of cumulative consolidated operating income achieved during the vesting period. In 2008, McDermott's Compensation Committee, with the advice of Hewitt, incorporated time-vested restricted stock into long-term incentive compensation, to promote retention. However, performance shares still constituted 75% of our Named Executives' long-term incentive compensation for 2008. As a result, at the time the Compensation Committee considered 2009 long-term incentive compensation, performance shares dominated the outstanding long-term incentive compensation of Named Executives.

In addition, Hewitt's 2009 market analysis of long-term incentive compensation indicated that many companies were reexamining their long-term incentive mix. Specifically, Hewitt noted that more companies were using stock options for rewarding performance based on absolute stock price improvement, restricted stock for retention and performance-based stock to encourage a focus on financial or operational drivers.

As a result, John A. Fees, Chief Executive Officer of McDermott, recommended changing the mix and allocation of equity award types used in connection with 2009 long-term incentive compensation. With the advice of Hewitt and to maintain the competitiveness of each Named Executive's compensation, McDermott's Compensation Committee approved the use of a mix of performance shares, restricted stock units and stock options in 2009 for officers, including our Named Executives. To drive performance, McDermott's Compensation Committee determined to issue at least a majority of long-term incentive compensation in the form of performance-based compensation. For 2009, the performance-based long-term compensation consisted of performance shares and stock options. McDermott's Compensation Committee allocated 2009 target long-term incentive compensation as follows:

- 20% performance shares;
- · 40% stock options; and
- 40% restricted stock units.

Similar to prior years, the 2009 performance shares generally vest three years from the date of grant. However, McDermott's Compensation Committee made two principal changes in the 2009 performance shares. First, to be consistent with annual incentive compensation, the 2009 performance shares vest in an amount between 0% and 200% of the target shares granted. Second, McDermott's Compensation Committee added the concept of relative shareholder return as a performance condition, in addition to McDermott's consolidated operating income. McDermott's Compensation Committee continues to believe that operating income is an appropriate financial metric for use in both annual and long-term incentive compensation because of its potential to drive stock price and accountability at operating segments; however, in 2008 McDermott witnessed a disconnect in performance results. With approximately \$570 million of operating income, McDermott produced its second highest level of operating income in at least a decade, although the price of its stock declined. To tie performance shares more directly with shareholder return, McDermott's Compensation Committee designed the 2009 performance shares to vest (1) one-half based on the level of McDermott's cumulative operating income attained over the three-year period ending December 31, 2011 and (2) one-half based on the total shareholder return of our stock relative to the Custom Peer Group over the same period.

On the recommendation of McDermott's management and with the advice of Hewitt, the McDermott Compensation Committee set cumulative operating income for the target and maximum vesting levels at amounts that represent 6% and 10% annual growth from the target operating income goal for McDermott used in connection with 2009 annual incentive compensation. Consistent with annual incentive compensation, no performance shares will vest based on operating income if the performance results are below 70% of the target three-year cumulative operating income goal. In addition, the performance shares that vest based on shareholder return will vest as follows:

- 25% will vest if McDermott's total shareholder return is at the 25th percentile (threshold level) relative to the total shareholder return of the Custom Peer Group;
- 100% will vest if McDermott's total shareholder return is at the 50h percentile (target level) relative to the total shareholder return of the Custom Peer Group; and
- 200% will vest if McDermott's total shareholder return is at the 100th percentile (maximum level) relative to the total shareholder return of the Custom Peer Group.

The percentage that would vest between threshold, target and maximum levels is interpolated based on those performance levels. However, regardless of percentile, 200% would vest if McDermott's total shareholder return

for the three-year period ranks either first or second relative to the total shareholder return of the Custom Peer Group over the same period. No performance shares would vest based on shareholder return if McDermott's total shareholder return relative to the Custom Peer Group falls below the 25% threshold performance level.

For more information regarding the 2009 performance shares, restricted stock units and stock options, see the Grants of Plan-Based Awards table under "Compensation of Executive Officers—Estimated Future Payouts Under Equity Incentive Plan Awards." For information regarding how McDermott's outstanding equity-base compensation awards will be treated in connection with the spin-off, see "The Spin-Off—Treatment of Stock-Based Awards" above.

Value of 2009 Long-Term Incentive Compensation. The 2009 target long-term incentive compensation for our Named Executives was as follows:

| | Target LTI | |
|-----------------|---------------------------|----------------------|
| Named Executive | (% of annual base salary) | Percent of Market(1) |
| B.C. Bethards | 286% | 100% |
| M.S. Taff | 348% | 125% |
| R.L. Killion | 168% | 106% |
| C.M. Mowry | 100% | 108% |
| W.D. Nash | 144% | 116% |

(1) Market = Median target long-term incentives based on the benchmark applicable to the executive.

As a result of market conditions in 2008, the value of Named Executives' outstanding long-term incentive compensation dropped considerably. In February 2009, the expected value of the long-term incentive compensation granted between 2006 and 2008 was approximately 35% of the target grant date value. For certain officers, including one Named Executive (Mr. Taff), the expected value of their outstanding long-term incentive compensation was even lower due to the particular mix of their respective equity following recent promotions within McDermott. McDermott's Compensation Committee was concerned that long-term incentive compensation would not effectively promote its intended purpose at values substantially below target. McDermott's Compensation Committee therefore increased the target 2009 long-term incentive compensation of Mr. Taff by 9% over the amount initially recommended to be granted in 2009. As a result, Mr. Taff's target 2009 long-term incentive compensation exceeded the market range; however, his total direct compensation was within the market range.

Sizing Long-Term Incentive Compensation. McDermott's Compensation Committee generally determines the size of equity-based grants as a percentage of a Named Executive's annual base salary, rather than granting a targeted number of shares. The number of performance shares, restricted stock units and stock options granted can be expressed through the following formula:

value of target long-term incentive(\$)/FMV(\$).

The dollar value of the target equity award for each Named Executive was derived by multiplying the Named Executive's target percentage by his 2009 base salary. The fair market value of one performance share, restricted stock unit and stock option was determined by Hewitt near the time of grant and generally reflected a discount from the market price of McDermott's common stock as a result of the vesting conditions and restrictions on transfer. Hewitt used a Black Scholes model to value stock options. For the long-term incentive compensation granted in February 2009, the fair market value of McDermott's common stock as of the date the grants were calculated (based on the closing price of McDermott's common stock on the New York Stock Exchange) was \$13.21, compared to the discounted value of \$10.52 for one performance share, \$12.177 for one restricted stock unit and \$8.581 for an option to acquire one share of McDermott common stock. Because the long-term incentive compensation grants vest over three years, the number of shares calculated were rounded down to the nearest multiple of three. To illustrate, consider the 28,611 performance shares granted to our Chief

Executive Officer in 2009. The dollar value of Mr. Bethards' target 2009 long-term incentive compensation was \$1,504,932 (\$526,200 base salary multiplied by 286% target long-term incentive). Performance shares accounted for 20% of his target long-term incentive compensation granted in 2009, or \$300,986. At a fair market value of \$10.52/share, 20% of Mr. Bethards' target long-term compensation grant amounted to 28,611 performance shares.

Timing of Equity Grants

To avoid timing equity grants ahead of the release of material nonpublic information, McDermott's Compensation Committee generally approves stock option and other equity awards effective as of the first day of the next open trading window following the meeting at which the grants are approved, which is generally the third day following the filing of McDermott's annual report on Form 10-K or quarterly report on Form 10-Q with the Securities and Exchange Commission. This practice was followed for all long-term incentive compensation grants to Named Executives in 2009.

Perquisites

Perquisites are not factored into the determination of the total direct compensation of our Named Executives, because they are typically provided to Named Executives on an exception basis.

McDermott owns a fractional interest in two aircraft through an aircraft management company, which McDermott uses for business purposes and makes available to our Named Executives for limited personal use. When McDermott permits the personal use of aircraft by a Named Executive, McDermott has a choice regarding the amount of income tax imputed to the Named Executive for that use. Under current Internal Revenue Service rules, McDermott may impute to the Named Executive the actual cost incurred by McDermott for the flight or an amount based on Standard Industry Fare Level ("SIFL") rates set by the U.S. Department of Transportation. Imputing income based on SIFL rates usually results in less income tax liability to the Named Executive but higher income taxes to McDermott due to limitations on deducting aircraft expenses that exceed the income imputed to employees. To minimize McDermott's cost of permitting the personal use of the aircraft, McDermott imputes income for personal use of aircraft to our Named Executives in an amount that results in the least amount of tax burden for McDermott.

As required by applicable Securities and Exchange Commission rules, McDermott calculates compensation in respect of personal use of corporate aircraft based on its "incremental cost." McDermott computes incremental cost for personal use of aircraft based on the actual cost incurred by McDermott for the flight, including:

- the cost of fuel:
- · a usage charge equal to the hourly rate charged by McDermott's flight operator multiplied by the flight time;
- · "dead head" costs, if applicable, of flying aircraft without passengers to and from locations; and
- · the dollar amount of increased income taxes McDermott incurs as a result of disallowed deductions under IRS rules.

Since the aircraft are used primarily for business travel, incremental costs generally exclude fixed costs such as the purchase price of our interests in the aircraft, aircraft management fees, depreciation, maintenance and insurance. McDermott's cost for flights, whether business or personal, is not affected by the number of passengers. As a result, McDermott does not assign any amount, other than the amount of any disallowed deduction, when computing incremental costs for the presence of guests accompanying a Named Executive on such flights. While McDermott does not generally incur any additional cost, this travel may result in imputed income to the Named Executive and disallowed deductions on McDermott's income taxes. McDermott will reimburse the Named Executive for the travel expenses of a guest accompanying a Named Executive, including

the provision of a gross-up for any imputed income, when the presence of that guest is related to the underlying business purpose of the trip. McDermott also provides our Named Executives with a tax gross-up for income incurred in connection with a relocation with McDermott or one of its affiliated companies.

Post-Employment Compensation

Retirement Plans

Overview. McDermott provides retirement benefits through a combination of qualified defined benefit pension plans, which we refer to as the "Retirement Plans," and a qualified defined contribution 401(k) Plan, which we refer to as the "Thrift Plan," for most of our U.S. employees, including our Named Executives. McDermott maintains four Retirement Plans, including the following three which are relevant to our Named Executives:

- the Commercial Operations Retirement Plan for the benefit of the U.S. employees of McDermott's Power Generation Systems segment;
- · the Government Operations Retirement Plan for the benefit of the employees of McDermott's Government Operations segment; and
- · the McDermott Retirement Plan for the benefit of the employees of MI.

We sponsor the Commercial Operations Retirement Plan through B&W PGG. McDermott sponsors the Government Operations and McDermott Retirement Plans through MI. In connection with the spin-off, we will assume the Government Operations Retirement Plan and certain assets and liabilities of the McDermott Retirement Plan.

In addition to the broad-based qualified plans described above, McDermott maintains unfunded, nonqualified excess retirement plans, which we refer to as the "Excess Plans." The Excess Plans cover a small group of highly compensated employees, including some of our Named Executives (Messrs. Bethards, Killion and Nash), whose ultimate benefits under the applicable Retirement Plan are reduced by Internal Revenue Code limits on the amount of benefits which may be provided under qualified plans, and on the amount of compensation which may be taken into account in computing benefits under qualified plans. Benefits under the excess plans are paid from McDermott's general assets. We sponsor the Commercial Operations Excess Plan through B&W PGG and the Government Operations Excess Plan through our subsidiary, Babcock & Wilcox Investment Company. McDermott sponsors the McDermott Excess Plan through MI. In connection with the spin-off, we will assume certain liabilities of the McDermott Excess Plan. See the "Pension Benefits" table under "Compensation of Executive Officers" below for more information regarding the Retirement Plans.

Recent Changes to Retirement Plans. Over the past several years, McDermott has reassessed its retirement plans due to the volatility, cost and complexity associated with defined benefit plans and evolving employee preferences. As a result, McDermott has taken steps to shift away from traditional defined benefit plans and toward a defined contribution approach. In 2006, McDermott closed the McDermott Commercial Operations and Government Operations Retirement Plans to new salaried participants and froze benefit accruals for existing salaried participants with less than five years of credited service as of March 31, 2006, subject to specific annual cost-of-living increases. In lieu of future defined benefit plan accruals under those plans, McDermott further amended the Thrift Plan to provide an automatic cash contribution to the Thrift Plan accounts of affected employees and new hires in an amount between 3% and 8% of the employee's base pay, plus overtime pay, expatriate pay and commissions, based on their length of service. Messrs. Mowry and Taff were affected by these changes. Neither Mr. Mowry nor Mr. Taff participate in a Retirement Plan or an Excess Plan because neither executive met the eligibility requirements at the time that the plan applicable to the executive was closed to new participants. In 2007, McDermott offered salaried participants in the McDermott Commercial Operations and Government Operations Retirement Plans with between five and 10 years of credited service as of January 1, 2007 the one-time irrevocable choice between (1) continuing to accrue future benefits under the Retirement Plan

or (2) freezing their Retirement Plan accrued benefit as of March 31, 2007, subject to annual cost-of-living increases, and receiving an automatic service-based cash contribution to their Thrift Plan account instead. No Named Executives were affected by the changes in 2007.

Supplemental Plans. In 2005, as part of McDermott's determination to move away from defined benefit plans, McDermott's management recommended that McDermott's Board of Directors and McDermott's Compensation Committee terminate its then-existing nonqualified, defined benefit supplemental executive retirement plan. In its place, McDermott's Board of Directors and Compensation Committee established a new supplemental executive retirement plan, which we refer to as the "SERP," to help maintain the competitiveness of its post-employment compensation as compared to the market. The SERP is an unfunded, nonqualified defined contribution plan that provides participants with benefits based on the participant's notional account balance at the time of retirement or termination. Employees must generally be an officer of McDermott or one of its subsidiaries for two to three years before being eligible to participate in the SERP. As a result, Messrs. Bethards, Taff and Nash are participants in the SERP, but Messrs. Killion and Mowry are not.

Annually, McDermott credits a participant's notional account with an amount equal to 5% of the participant's prior-year base salary and annual bonus paid in the prior year. McDermott's Compensation Committee has designated deemed mutual fund investments to serve as indices for the purpose of determining notional investment gains and losses to each participant's account. Each participant allocates the annual notional contribution among the various deemed investments. SERP benefits are based on the participant's vested notional account balance at the time of retirement or termination. In order to take advantage of grandfathering provisions in Internal Revenue Code Section 457A, in 2009, the Compensation Committee amended the SERP to vest each participant's account as of December 31, 2008. Of the Named Executives with a SERP account balance as of the end of 2008, only Mr. Taff's account was not fully vested. At the time of the amendment, Mr. Taff was 80% vested in his SERP balance. See the Nonqualified Deferred Compensation table under "Compensation of Executive Officers" below for more information regarding the SERP and McDermott's contributions to our Named Executives' SERP accounts. In connection with the spin-off, we expect our Board or Directors will adopt a nonqualified supplemental executive retirement plan substantially similar to McDermott's SERP.

Employment and Severance Arrangements

Employment and Separation Agreements. Except for change-in-control agreements and retention agreements, we do not currently have any employment or severance agreements with any of our Named Executives.

Change-in-Control Agreements. McDermott believed that providing change-in-control agreements to key executives protects shareholders' interests by helping to assure management continuity and focus through and beyond a change in control. Accordingly, McDermott's Compensation Committee has offered change-in-control agreements to its senior management since 2005. Two of our Named Executives, Messrs. Bethards and Taff, have change-in-control agreements with McDermott. These agreements generally provide a cash severance payment of two times the sum of the Named Executive's annual base salary and target EICP, a cash payment equal to two years of medical benefits and an additional tax gross-up in the event of any excise tax liability. Additionally, McDermott's change-in-control agreements contain what is commonly referred to as a "double trigger," that is, they provide benefits only upon an involuntary termination or constructive termination of the executive officer within one year following a change in control. See the "Potential Payments Upon Termination or Change in Control" tables under "Compensation of Executive Officers" below and the accompanying disclosures for more information regarding the change-in-control agreements with our Named Executives, as well as other plans and arrangements that have different trigger mechanisms that relate to a change in control.

In connection with the spin-off, the change-in-control agreements between McDermott and Messrs. Bethards and Taff will terminate immediately following the distribution date.

Retention Agreements. In connection with McDermott's announcement to spin off B&W as an independent, publicly traded company, McDermott entered into retention agreements with 17 key members of its management, including Messrs. Bethards and Taff, to retain key employees and executives through the completion of the separation of B&W from McDermott.

Generally, the retention agreements provide either a retention or severance payment to the Named Executives in connection with the spin-off. The retention agreements generally provide a retention payment in the form of a restricted stock grant made near the time of the spin-off, that would vest one year following the separation, equal to 100% of the sum of the Named Executive's annual base salary plus target annual incentive compensation. That amount represented one-half of the severance payment that otherwise would be provided under each Named Executive's retention agreement in the event of a qualifying termination. With respect to Mr. Taff, one-third of his retention payment will be payable in cash on the effective date of the separation, in recognition of his agreement to serve as the Chief Financial Officer of B&W following the spin-off.

Although the spin-off would not constitute a change in control for purposes of the Change-in-Control Agreements discussed above or other McDermott compensation plans, McDermott's Compensation Committee determined that the need to maintain continuity of management and personnel that exists under a change in control scenario equally exists in connection with the planned spin-off of B&W. As a result, the retention agreements provide for severance payments that would generally be the same as the severance payments that would be made in connection with a qualifying termination on or following a change in control (other than tax gross-ups). Accordingly, the retention agreements provide for a cash severance payment of two times the sum of the Named Executive's annual base salary and target EICP, prorated target annual incentive compensation and a cash payment equal to two years of medical benefits as well as the full vesting of outstanding long-term incentive grants and SERP balance. The only payment provided for under the retention agreement not otherwise payable in a change in control is the potential early vesting of the Named Executive's Thrift Plan account. Under the terms of the Thrift Plan, unvested balances would become vested in the event a participant is involuntarily terminated in connection with a reduction in force. Because involuntary terminations for reasons other than cause in connection with the proposed separation generally would be considered to be associated with a reduction in force, McDermott's Compensation Committee determined to add the vesting of Thrift Plan accounts to the severance benefits to avoid any ambiguity on that point. The Thrift Plan normally vests after three years of service. All of our Named Executives who are party to a retention agreement are fully vested in their Thrift Plan accounts.

We will assume McDermott's obligations under the retention agreements with our Named Executives as of the distribution date, and those retention agreements will terminate one year following the spin-off, among other reasons.

See the "Potential Payments Upon Termination or Change in Control" tables under "Compensation of Executive Officers—Retention Agreements" below and the accompanying disclosures for more information regarding the retention agreements with our Named Executives.

Compensation of Executive Officers

The following table summarizes the prior three years' compensation of our Chief Executive Officer, our Chief Financial Officer and our three highest paid executive officers who did not serve as our CEO and CFO during 2009. We refer to these persons as our Named Executives. All of the information included in these tables reflects compensation earned by the individuals for service with McDermott. All references in the following tables to stock relate to awards of stock granted by McDermott. Such amounts do not necessarily reflect the compensation such persons will receive following the spin-off, which could be higher or lower, because historical compensation was determined by McDermott and future compensation levels will be determined based on the compensation policies, programs and procedures to be established by our compensation committee.

Summary Compensation Table

| Name and Principal Position | <u>Year</u> | Salary | Во | onus ⁽¹⁾ | Ā | Stock Awards ⁽²⁾ | Ā | Option Awards ⁽³⁾ | | Non-Equity Incentive Plan Compensation ⁽⁴⁾ | N Co | Change in Pension Value and lonqualified Deferred ompensation Earnings ⁽⁵⁾ | _c | All Other ompensation ⁽⁶⁾ | | Total |
|--|----------------------|--|------------------|---------------------|----------------|---------------------------------|----------------|---------------------------------|----------------|---|----------------|---|----------------|---|----------|-------------------------------------|
| B.C. Bethards President & Chief Executive Officer | 2009 2008 2007 | \$ 526,200 \$ 438,675 \$ 390,000 | \$ \$ \$ | 0 10,000 0 | \$ \$ \$ | 840,544 1,207,512 999,148 | \$ \$ \$ | 473,450 0 0 | \$ \$ \$ | 663,012 509,298 460,278 | \$ \$ \$ | 305,160 158,014 186,071 | \$ \$ \$ | 65,693 54,831 43,249 | \$ | 2,874,059 2,378,330 2,078,746 |
| M.S. Taff Senior Vice President & Chief Financial Officer | 2009 2008 2007 | \$ 505,000 \$ 440,000 \$ 374,999 | \$ \$ 1 \$ | 0 10,000 0 | \$ \$ \$ | 980,544 1,671,638 742,610 | \$ \$ \$ | 552,314 0 0 | \$ \$ \$ | 707,000 141,207 387,563 | | N/A N/A N/A | \$ \$ \$ | 59,315 45,757 34,211 | \$ | 2,804,173 2,408,602 1,539,383 |
| R.L. Killion President & Chief Operating Officer, B&W Power Generation Group | 2009 2008 2007 | \$ 330,000 \$ 272,850 \$ 219,600 | \$ \$ \$ | 0 0 0 | \$ \$ \$ | 309,627 551,098 337,550 | \$ \$ \$ | 174,413 0 0 | \$ \$ \$ | 347,210 256,350 168,790 | \$ \$ \$ | 305,071 182,469 151,738 | \$ \$ \$ | 7,546 6,957 6,594 | | 1,473,867 1,269,724 884,272 |
| C.M. Mowry ⁽⁷⁾ President, B&W Nuclear Energy, Inc. | 2009 2008 2007 | \$ 355,000 \$ 137,880 N/A | \$ \$ | 0 0 N/A | \$ | 198,286 400,649 N/A | \$ | 111,686 0 N/A | \$ \$ | 301,129 156,161 N/A | | N/A N/A N/A | \$ \$ | 12,921 4,759 N/A | \$ \$ | , . |
| W.D. Nash President, B&W Nuclear Operations Group | 2009 2008 2007 | \$ 348,000 \$ 330,800 \$ 315,000 | \$ \$ \$ | 0 0 0 | \$ \$ \$ | 279,320 673,920 465,819 | \$ \$ \$ | 157,324 0 0 | \$ \$ \$ | 292,668 292,808 276,413 | \$ \$ \$ | 76,599 6,272 28,297 | \$ \$ \$ | 37,872 33,271 30,546 | \$ | 1,191,783 1,337,071 1,116,075 |

- See "Bonus" below for a discussion of the amounts included in this column
- (2)
- See "Stock Awards" below for a discussion of the amounts included in this column. See "Option Awards" below for a discussion of the amounts included in this column. (3)
- See "Non-Equity Incentive Plan Compensation" below for a discussion of the amounts included in this column.

 See "Change in Pension Value and Nonqualified Deferred Compensation Earnings" below for a discussion of the amounts included in this column. (5)
- See "All Other Compensation" below for a discussion of the amounts included in this column.
- No compensation information is provided for Mr. Mowry for 2007 because he joined our company in August 2008. The information for 2008 reflects his compensation from August 2008 through December 2008.

Bonus. The amounts reported in the "Bonus" column are attributable to discretionary bonus awards.

Stock Awards. The amounts reported in the "Stock Awards" column represent the aggregate grant date fair value of stock awards granted in the applicable year and computed in accordance with Financial Accounting Standards Board Accounting Standards Codification ("FASB ASC") Topic 718. Under FASB ASC Topic 718, the fair value of the stock awards is determined on the date of grant and is not remeasured. Grant date fair values were determined using the closing price of McDermott's common stock on the date of grant for restricted stock, restricted stock units and performance shares. The grant date fair value of performance shares included in the stock awards is based on target-

level performance, which McDermott determined was the probable outcome of performance conditions at the time of grant. Assuming the maximum performance levels were probable, the aggregate grant date fair values of the stock awards would be as follows:

Grant Date Fair Value Assuming Maximum Performance

| Name | Year | | Value Assuming Maximum formance Level |
|---------------------------------------|------|----|--|
| B.C. Bethards | 2009 | \$ | 1,148,685 |
| | 2008 | \$ | 1,580,801 |
| | 2007 | \$ | 1,498,722 |
| M.S. Taff | 2009 | \$ | 1,339,993 |
| | 2008 | \$ | 2,326,078 |
| | 2007 | \$ | 1,113,915 |
| | | | |
| R.L. Killion | 2009 | \$ | 423,132 |
| | 2008 | \$ | 728,002 |
| | 2007 | \$ | 506,325 |
| C.M. Mowry | 2009 | \$ | 270,984 |
| · · · · · · · · · · · · · · · · · · · | 2008 | \$ | 400,649 |
| | 2007 | * | N/A |
| | | _ | |
| W.D. Nash | 2009 | \$ | 381,710 |
| | 2008 | \$ | 937,697 |
| | 2007 | \$ | 698,729 |

See the "Grants of Plan-Based Awards" table for more information regarding the stock awards McDermott granted in 2009.

Option Awards. The amounts reported in the "Option Awards" column represent the aggregate grant date fair value of all option awards granted in the applicable year computed in accordance with FASB ASC Topic 718. Under FASB ASC Topic 718, the fair value of stock options is determined on the date of grant using an option-pricing model and is not remeasured. McDermott used a Black-Scholes option-pricing model for measuring the fair value of stock options. The determination of the fair value of an award on the date of grant using an option-pricing model requires various assumptions, such as the expected life of the award and stock price volatility. For a discussion of the valuation assumptions, please see the information set forth under the heading "Stock Options" in Note 8, "Stock Plans," to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement.

See the "Grants of Plan-Based Awards" table for more information regarding the option awards McDermott granted in 2009.

Non-Equity Incentive Plan Compensation. The amounts reported in the "Non-Equity Incentive Plan Compensation" column are attributable to the McDermott annual incentive awards earned in fiscal years 2007, 2008 and 2009, but paid in 2008, 2009 and 2010, respectively. See the "Grants of Plan-Based Awards" table for more information regarding the annual incentive awards earned in 2009.

Change in Pension Value and Nonqualified Deferred Compensation Earnings. The amounts reported in the "Change in Pension Value and Nonqualified Deferred Compensation Earnings" column represent the changes in actuarial present values of the accumulated benefits under McDermott's defined benefit plans, determined by comparing the prior completed fiscal year-end amount to the covered fiscal year-end amount.

All Other Compensation. The amounts reported for 2009 in the "All Other Compensation" column are attributable to the following:

All Other Compensation

| | | | | | S | ervice- | | | |
|---------------|------|--------------|-------|-----------|-----|------------|-----|--------|-------------|
| | | | Based | | | | | | |
| | | | | | | Thrift | Tax | Gross- | |
| | SERP | Contribution | Thr | ift Match | Cor | itribution | τ | Jps | Perquisites |
| B.C. Bethards | \$ | 44,948 | \$ | 4,906 | | _ | \$ | 1,144 | \$ 14,695 |
| M.S. Taff | \$ | 41,378 | \$ | 7,358 | \$ | 7,358 | \$ | 3,221 | _ |
| R.L. Killion | | _ | \$ | 7,311 | | _ | \$ | 235 | _ |
| C.M. Mowry | | _ | \$ | 4,903 | \$ | 7,354 | \$ | 664 | _ |
| W.D. Nash | \$ | 30,361 | \$ | 6,641 | | _ | \$ | 870 | _ |

SERP. See the "Nonqualified Deferred Compensation" table for more information regarding the McDermott SERP contributions made in 2009.

Thrift Match and Service-Based Thrift Contribution. We refer to McDermott's defined contribution plan as the Thrift Plan. For information regarding the Thrift Plan matching contributions and service-based Thrift Plan contributions, see "Compensation Discussion and Analysis—Post-Employment Compensation—Retirement Plans" above.

Tax Gross-Ups. The tax gross-ups reported for 2009 under "All Other Compensation" are attributable to the following:

- Mr. Bethards: Mr. Bethards received tax gross-ups associated with income imputed to him as a result of his spouse accompanying him on business travel
 and as a result of a gift received at a management meeting.
- · Mr. Taff: Mr. Taff received tax gross-ups associated with income imputed to him as a result of his spouse accompanying him on business travel.
- · Mr. Killion: Mr. Killion received a tax gross-up associated with income imputed to him as a result of a gift received at a management meeting.
- Mr. Mowry: Mr. Mowry received tax gross-ups associated with income imputed to him as a result of his spouse accompanying him on business travel and
 as a result of a gift received at a management meeting.
- Mr. Nash: Mr. Nash received tax gross-ups associated with income imputed to him as a result of his spouse accompanying him on business travel and as a
 result of a gift received at a management meeting and a Christmas gift basket.

Perquisites. Perquisites and other personal benefits received by a Named Executive are not included if their aggregate value does not exceed \$10,000. For Mr. Bethards, the values of the perquisites and other personal benefits reported for 2009 are as follows:

• Mr. Bethards: \$12,474 is attributable to the costs of providing him relocation assistance in connection with his move from Ohio to Virginia following his appointment as Chief Executive Officer of B&W. The remainder is attributable to the cost of club dues and the cost of promotional merchandise in connection with a board of directors meeting.

Grants of Plan-Based Awards

The following Grants of Plan-Based Awards table provides additional information about stock awards and equity and non-equity incentive plan awards granted by McDermott to our Named Executives during the year ended December 31, 2009.

| | | | | ted Possible Von-Equity I lan Awards | ncentive | | ed Future quity Incer Awards ⁽²⁾ | itive Plan | All Other Stock Awards: | All Other Option Awards: | Exercise | | ant Date |
|---------------|--|--|-----------|--|------------|-----------------|---|------------------|--|---|---|----------------|--|
| Name | Grant Date | Committee Action Date | Threshold | Target | Maximum | Threshold (#) | Target | Maximum (#) | Number of Shares of Stock or Units ⁽³⁾ | Number of Securities Underlying Options ⁽⁴⁾ | or Base Price of Option Awards | of s | nir Value Stock and Option wards ⁽⁵⁾ |
| B.C. Bethards | 02/26/09 03/05/09 03/05/09 03/05/09 | 02/26/09 02/26/09 02/26/09 02/26/09 | \$ 64,460 | \$368,340 | \$ 736,680 | 7,152 — — | 28,611 — — | 57,222 — — | 49,434 — | | \$ 10.93 | \$ \$ \$ | 308,140 532,404 473,450 |
| M.S. Taff | 02/26/09 03/05/09 03/05/09 03/05/09 | 02/26/09 02/26/09 02/26/09 02/26/09 | \$ 61,863 | \$353,500 | \$ 707,000 | 8,343 | 33,375 | 66,750 — — | 57,669 — | — 81,834 | _ \$ 10.93 | \$ \$ \$ | 359,449 621,095 552,314 |
| R.L. Killion | 02/26/09 03/05/09 03/05/09 03/05/09 | 02/26/09 02/26/09 02/26/09 02/26/09 | \$ 31,763 | \$181,500 | \$ 363,000 | 2,634 | 10,539 | 21,078 — — | 18,210 — | | _ _ \$ 10.93 | \$ \$ \$ | — 113,505 196,122 174,413 |
| C.M. Mowry | 02/26/09 03/05/09 03/05/09 03/05/09 | 02/26/09 02/26/09 02/26/09 02/26/09 | \$ 27,956 | \$159,750 | \$ 319,500 | 1,687 — — | 6,750 | 13,500 | 11,661 | 16,548 | \$ 10.93 | \$ \$ \$ | 72,698 125,589 111,686 |
| W.D. Nash | 02/26/09 03/05/09 03/05/09 03/05/09 | 02/26/09 02/26/09 02/26/09 02/26/09 | \$ 30,450 | \$174,000 | \$ 348,000 | 2,376 | 9,507 — — | 19,014 — — | 16,428 — | 23,310 | _ _ \$ 10.93 | \$ \$ \$ | 102,390 176,930 157,324 |

- See "Estimated Possible Payouts Under Non-Equity Incentive Plan Awards" below for a discussion of the amounts included in this column.
- See "Estimated Future Payouts Under Equity Incentive Plan Awards" below for a discussion of the amounts included in this column. See "All Other Stock Awards" below for a discussion of the amounts included in this column.
- (3)
- See "All Other Option Awards" below for a discussion of the amounts included in this column
- See "Grant Date Fair Value of Stock and Option Awards" below for a discussion of the amounts included in this column.

Estimated Possible Payouts Under Non-Equity Incentive Plan Awards

McDermott's compensation committee administers the Executive Incentive Compensation Plan, an annual cash incentive program, which we refer to as the EICP. The payment amount, if any, of an EICP award is determined based on: (1) the attainment of short-term financial goals; (2) the attainment of short-term individual goals; and (3) the exercise of the McDermott compensation committee's discretionary authority. Each year, McDermott's compensation committee establishes financial goals. McDermott's Chief Executive Officer established the individual goals for our Named Executives.

The financial goals contain threshold, target and maximum performance levels which, if achieved, result in payments of 25%, 100% and 200% of the financial component, respectively. If the threshold financial goal is not achieved, no amount is paid on an EICP award under the financial component. For purposes of evaluating McDermott's performance under the financial performance component, McDermott's compensation committee may adjust McDermott's results prepared in accordance with U.S. generally accepted accounting principles ("GAAP") for unusual, nonrecurring or other items in the McDermott compensation committee's discretion. Payment is made on an EICP award under the individual component based on the attainment of the Named Executive's individual goals as determined and evaluated by McDermott's Chief Executive Officer. In addition, the McDermott compensation committee may decrease an EICP award in its discretion. The maximum EICP award a Named Executive can earn is 200% of his target EICP award.

The amounts shown reflect grants of 2009 EICP awards made by McDermott. The McDermott compensation committee established target EICP awards expressed as a percentage of the Named Executive's 2009 base salary. The amount shown in

the "target" column represents the value of the target EICP award determined by multiplying the target percentage established for each Named Executive by the Named Executive's 2009 base salary. For 2009, the target percentage of base salary for each Named Executive was as follows: 70% for Mr. Bethards, 70% for Mr. Taff, 55% for Mr. Killion, 45% for Mr. Mowry and 50% for Mr. Nash. The amount shown in the "maximum" column represents the maximum amount payable under the EICP, which is 200% of the target amount shown. The amount shown in the "threshold" column represents the amount payable under the EICP assuming the threshold level of the financial goals, but no individual goal, is attained and the McDermott compensation committee did not exercise any discretion over the EICP award. The financial goal represents 70% of the target EICP award. Attaining only the threshold level, or 25%, of the financial goal results in an EICP payment of 17.50% of the target EICP award. See "Compensation Discussion and Analysis—Annual Incentive Compensation" above for more information about the 2009 EICP awards and performance goals.

In connection with the spin-off, we expect our Board of Directors will adopt an annual incentive compensation plan substantially similar to McDermott's EICP. Assuming that is the case, for the 2010 EICP compensation of each current officer of B&W who will be an officer of B&W immediately following the spin-off, including Messrs. Bethards, Killion, Mowry and Nash, the officer's 2010 EICP compensation would be attributable to individual and financial performance. The financial performance would be based on pre-established levels of operating income related to B&W operations relevant to the officer. For the 2010 EICP compensation of each of the McDermott officers who are not current officers of B&W but who will be officers of B&W immediately following the spin-off, including Mr. Taff, the officer's 2010 EICP compensation would be divided based on the officer's service with McDermott prior to the effective date of the spin-off and the officer's service with B&W following the effective date of the spin-off. In respect of an officer's service with McDermott during 2010, the officer would earn the target amount of his or her 2010 EICP compensation for the annual base salary earned through the distribution date. In respect of the officer's service with B&W during 2010, the officer would earn additional 2010 EICP compensation based on his or her annual base salary earned following the distribution date through the end of 2010 attributable to individual and financial performance. Similar to current B&W officers, the financial performance with respect to these officers would also be based on pre-established levels of operating income related to B&W operations relevant to the officer.

Estimated Future Payouts Under Equity Incentive Plan Awards

The amounts shown reflect grants of performance shares under McDermott's 2001 D&O Plan made by McDermott. Each grant represents a right to receive one share of McDermott common stock for each vested performance share. The amount of performance shares that vest, if any, is scheduled to be determined on the third anniversary of the date of grant based (1) one-half on McDermott's cumulative operating income between January 1, 2009 and December 31, 2011, and (2) one-half on McDermott's total shareholder return relative to its custom peer group during the same period. For purposes of evaluating McDermott's cumulative operating income, McDermott's compensation committee may adjust McDermott's results prepared in accordance with GAAP for unusual, non-recurring or other items in the McDermott compensation committee's discretion. The amounts shown in the "target" column represent the number of performance shares granted, which will vest if both the target level of cumulative operating income is attained and McDermott's total shareholder return ranks in the 50th percentile relative to McDermott's custom peer group. The amounts shown in the "maximum" column represent the number of performance shares that will vest, which is 200% of the amount granted, if both the maximum level of cumulative operating income is attained and McDermott's total shareholder return ranks first or second in total shareholder return relative to McDermott's custom peer group. The amounts shown in the "threshold" column represent the number of performance shares that will vest, which is 25% of the amount granted, if both the minimum level of cumulative operating income is attained and McDermott's total shareholder return ranks in the 25th percentile relative to McDermott's total shareholder return ranks in the 25th percentile relative to McDermott's total shareholder return ranks in less than the McDermott's total shareholder return ranks in less than the McDermott's total description of how the McDermott performance shares will be treated in conne

All Other Stock Awards

The amounts shown reflect grants of restricted stock units made by McDermott under McDermott's 2001 D&O Plan. Each restricted stock unit represents the right to receive one share of McDermott common stock and generally is scheduled to vest in one-third increments on the first, second and third anniversaries of the date of grant. Upon vesting, the restricted stock units are converted into shares of McDermott common stock. See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how the McDermott restricted stock units will be treated in connection with the spin-off.

All Other Option Awards

The amounts shown reflect grants of stock options made by McDermott under McDermott's 2001 D&O Plan. Each grant represents the right to purchase at the exercise price shares of McDermott common stock over a period of seven years. The stock options are generally scheduled to vest and become exercisable in one-third increments on the first, second and third anniversaries of the date of grant. See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how the options to purchase McDermott stock will be treated in connection with the spin-off.

Grant Date Fair Value of Stock and Option Awards

The amounts included in the "Grant Date Fair Value of Stock and Option Awards" column represent the full grant date fair values of the equity awards computed in accordance with FASB ASC Topic 718. Under FASB ASC Topic 718, the fair value of equity awards is determined on the date of grant and is not remeasured. Grant date fair values are determined using the closing price of McDermott's common stock on the date of grant for restricted stock, restricted stock units and performance shares, and an option-pricing model for stock options. McDermott used a Black-Scholes option-pricing model for measuring the fair value of stock options granted. The determination of the fair value of an award on the date of grant using an option-pricing model requires various assumptions, such as the expected life of the award and stock price volatility. For more information regarding the compensation expense related to the awards, and a discussion of valuation assumptions utilized in option pricing, see the information set forth under the heading "Stock Options" in Note 8, "Stock Plans," to our combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement.

Outstanding Equity Awards at Fiscal Year-End

The following Outstanding Equity Awards at Fiscal Year-End table summarizes the equity awards McDermott has made to our Named Executives which were outstanding as of December 31, 2009.

| | | | Opti | ion Awards ⁽¹⁾ | | | Stock Awards(2) | | | | | | |
|---|--|--|--|--|-----------------------------|------------------------------|--|-------------------|--|---|---|---|--|
| Name | Grant Date | Number of Securities Underlying Unexercised Options Exercisable | Number of Securities Underlying Unexercised Options Unexercisable | Equity Incentive Plan Awards: Number of Securities Underlying Unexercised Unearned Options | Option Exercise Price | Option Expiration Date | Number of Shares or Units of Stock That Have Not Vested | Shares Stock T | et Value of or Units of That Have Vested ⁽³⁾ | Equity Incentive Plan Awards: Number of Unearned Shares, Units or Other Rights That Have Not Vested | Pla M Pay of Sh: (Ri | Equity Incentive an Awards: Market or yout Value 'Unearned ares, Units or Other ights That Have Not Vested(3) | |
| B.C. Bethards Stock Options Performance Shares Restricted Stock Performance Shares Restricted Stock RSU Performance Shares | 03/05/09 05/10/07 03/03/08 03/03/08 11/10/08 03/05/09 03/05/09 | _ | 70,149 | _ | \$10.9300 | 03/05/16 | <u></u> | \$ \$ | 1,066,044 62,906 — 416,165 1,186,910 | 3,545 — — 57,222 | \$ \$ | 85,115 — 1,373,900 | |
| M.S. Taff Stock Options Stock Options DSU Performance Shares Restricted Stock Performance Shares RSU Performance Shares | 06/08/05 03/05/09 06/08/05 05/10/07 03/03/08 03/03/08 03/05/09 03/05/09 | 23,000 | 81,834 | _ | \$ 7.1933 \$10.9300 | 06/08/15 03/05/16 | 4,500 33,000 4,593 — 57,669 | \$ \$ \$ | 108,045 792,330 110,278 — 1,384,633 | 6,215 — 66,750 | \$ \$ | 149,222 — 1,602,668 | |
| R.L. Killion Stock Options Stock Options Performance Shares Restricted Stock Performance Shares Restricted Stock Restricted Stock Performance Shares | 03/20/00 03/05/09 05/10/07 03/03/08 03/03/08 11/10/08 03/03/08 03/05/09 | 320 | 25,842 | Ξ | \$ 3.1354 \$10.9300 | 03/20/10 03/05/16 | 15,000 1,240 — 6,780 18,210 | \$ | 360,150 29,772 — 162,788 437,222 | 1,680 21,078 | \$ \$ | 40,337 — — 506,083 | |
| C.M. Mowry Stock Options Restricted Stock RSU Performance Shares | 03/05/09 08/14/08 03/05/09 03/05/09 | _ | 16,548 | - | \$10.9300 | 03/05/16 | 7,513 11,661 | | 180,387 279,981 | 13,500 | \$ | 324,135 | |
| W.D. Nash Stock Options DSU Performance Shares Restricted Stock Performance Shares RSU Performance Shares | 03/05/09 05/12/05 05/10/07 03/03/08 03/03/08 03/05/09 03/05/09 | _ | 23,310 | _ | \$10.9300 | 03/05/16 | 1,853 | \$ \$ \$ | 63,242 497,007 44,491 — 394,436 | 2,505 — 19,014 | \$ \$ | 60,145 — 456,526 | |

⁽²⁾

See "Option Awards" below for a discussion of the amounts reported in this column including the vesting dates of unexercisable option awards.

See "Stock Awards" below for a discussion of the amounts reported in this column including the vesting dates of outstanding stock awards.

Market values in these columns are based on the closing price of McDermott's common stock as of December 31, 2009 (\$24.01), as reported on the New York Stock Exchange.

Option Awards. Information presented in the "Option Awards" columns relates to options to purchase shares of McDermott common stock held by our Named Executives as of December 31, 2009. All unexercisable options generally vest in three equal installments on the first, second and third anniversaries of the grant date, as follows:

 Grant Date
 One-third of grant vests on:

 03/05/09
 March 5, 2010, 2011 and 2012

See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how the options to purchase McDermott stock will be treated in connection with the spin-off.

Stock Awards. Information presented in the Stock Awards columns relates to awards of restricted stock, restricted stock units, deferred stock units and performance shares made by McDermott and held by our Named Executives as of December 31, 2009. The awards reported in the "Equity Incentive Plan Awards" columns consist entirely of stock awards subject to performance-based conditions as of December 31, 2009. The awards reported in the "Number of Shares or Units of Stock that have not Vested" column reflect stock awards subject to service-based vesting, including performance shares whose performance conditions have been satisfied as of December 31, 2009.

Restricted Stock Awards. Shares of restricted stock are generally scheduled to vest in one-third increments on the first, second and third anniversaries of the grant date. The vesting schedule of the restricted stock outstanding as of December 31, 2009 is as follows:

| Grant Date | One-third of grant vests on: |
|------------|------------------------------|
| 03/03/08 | March 3, 2010 and 2011 |
| 08/14/08 | August 14, 2010 and 2011 |
| 11/10/08 | November 10, 2010 and 2011 |

See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how shares of McDermott restricted stock will be treated in connection with the spin-off.

Restricted Stock Units. Restricted stock units represent the right to receive one share of McDermott common stock for each vested restricted stock unit. Restricted stock units outstanding as of December 31, 2009 are generally scheduled to vest in one-third increments on the first, second and third anniversaries of the grant date, as follows:

 Grant Date
 One-third of grant vests on:

 03/05/09
 March 5, 2010, 2011 and 2012

See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how the McDermott restricted stock units will be treated in connection with the spin-off.

Deferred Stock Units. Deferred stock units are settled in cash in an amount equal to the number of vested units multiplied by the average of the highest and lowest price of McDermott's common stock on the date of vesting. Deferred stock units are generally scheduled to vest in five equal installments on each anniversary of the date of grant. The vesting schedule of the deferred stock units outstanding as of December 31, 2009 is as follows:

 Grant Date
 One-fifth of grant vests on:

 05/12/05
 May 12, 2010

 06/08/05
 June 8, 2010

 08/09/05
 August 9, 2010

See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how McDermott deferred stock units will be treated in connection with the spin-off.

Performance Shares. Performance shares represent the right to receive one share of McDermott common stock for each performance share that vests. The number of performance shares that vest depends on the attainment of specified performance goals. The number and value of performance shares reported is based on achieving threshold performance levels, unless the previous year's performance level exceeds threshold, in which case the number and value of performance shares reported is based on attaining the next higher performance level. The number and value of the 2007 performance shares reported are based on attaining the maximum performance level, or 150% of the performance shares granted. The number and value of the 2008 performance shares reported are based on attaining the maximum performance level, or 25% of the performance shares granted. The number and value of the 2009 performance shares reported are based on attaining the maximum performance level, or 200% of the performance shares granted. See the "Grants of Plan-Based Awards" table for more information about performance shares. Performance shares are generally scheduled to vest on the third anniversary of the date of grant, as follows:

| Grant Date | Vests per attainment of performance levels on: | | | | | | |
|------------|--|--|--|--|--|--|--|
| 05/10/07 | May 10, 2010 | | | | | | |
| 03/03/08 | March 3, 2011 | | | | | | |
| 03/05/09 | March 5, 2012 | | | | | | |

See "The Spin-Off—Treatment of Stock-Based Awards" for a description of how the McDermott performance shares will be treated in connection with the spin-off.

Option Exercises and Stock Vested

The following Option Exercises and Stock Vested table provides additional information about the value realized by our Named Executives on exercises of McDermott option awards and vesting of McDermott stock awards during the year ended December 31, 2009.

| | Option | Awards | Stock Awards | | | |
|---------------|-----------------|----------------|----------------|----------------|--|--|
| | Shares | | Shares | | | |
| | Acquired | Value Realized | Acquired | Value Realized | | |
| Name | on Exercise (#) | on Exercise | on Vesting (#) | on Vesting | | |
| B.C. Bethards | 0 | N/A | 32,477 | \$631,050.07 | | |
| M.S. Taff | 0 | N/A | 31,547 | \$567,563.97 | | |
| R.L. Killion | 1,000 | \$ 19,774.70 | 8,105 | \$162,762.85 | | |
| C.M. Mowry | 0 | N/A | 3,757 | \$ 88,327.07 | | |
| W.D. Nash | 0 | N/A | 41,061 | \$825,452.24 | | |

Option Awards. Each stock option exercise reported in the Option Exercises and Stock Vested table was effected as a simultaneous exercise and sale. The value realized on exercise was calculated based on the difference between the exercise prices of the stock options and the prices at which the shares were sold.

Stock Awards. For each Named Executive, the number of shares acquired on vesting reported in the Option Exercises and Stock Vested table represents the aggregate number of shares that vested during 2009 in connection with awards of McDermott performance shares, restricted stock, restricted stock units and/or deferred stock units. The awards of deferred stock units reflected in this table are payable entirely in cash. As a result, no shares of McDermott stock were actually acquired upon the vesting of these deferred stock units. See the "Outstanding Equity Awards at Fiscal Year End" table for more information on the settlement of deferred stock unit awards. The following table sets forth the amount of McDermott shares attributable to McDermott performance shares, restricted stock, restricted stock units and deferred stock units, for each Named Executive:

| | Perform | ance Shares | Restri | cted Stock | Deferred Stock Units | | | |
|---------------|---------------------|------------------------------|------------------------|------------------------------|------------------------|------------------------------|--|--|
| | Number of Shares | | Number of Shares | | Number of Shares | | | |
| Name | Acquired on Vesting | Value Realized on Vesting | Acquired on Vesting | Value Realized on Vesting | Acquired on Vesting | Value Realized on Vesting | | |
| B.C. Bethards | 22,500 | \$ 407,025 | 9,977 | \$ 224,025 | 0 | N/A | | |
| M.S. Taff | 24,750 | \$ 447,728 | 2,297 | \$ 22,614 | 4,500 | \$ 97,223 | | |
| R.L. Killion | 4,095 | \$ 74,079 | 4,010 | \$ 88,684 | 0 | N/A | | |
| C.M. Mowry | 0 | N/A | 3,757 | \$ 88,327 | 0 | N/A | | |
| W.D. Nash | 37,500 | \$ 765,938 | 927 | \$ 9,126 | 2,634 | \$ 50,388 | | |

The number of McDermott shares acquired in connection with the vesting of McDermott performance shares, restricted stock awards and restricted stock units includes 11,160, 7,298, 2,630, 1,219 and 12,938 shares withheld by McDermott at the election of Messrs. Bethards, Taff, Killion, Mowry and Nash, respectively, to satisfy the minimum statutory withholding tax due upon vesting.

Pension Benefits

The following Pension Benefits table shows the present value of accumulated benefits payable to each of our Named Executives under McDermott's qualified and nonqualified pension plans.

| Name | Plan Name | Number of Years Credited Service | Present Value of Accumulated Benefit | Payments During 2009 |
|---------------|---|--|--|----------------------------|
| B.C. Bethards | B&W Governmental Operations Qualified Retirement Plan | 36.000 | \$ 1,148,331 | \$ 0 |
| | B&W Governmental Operations Excess Plan | 36.000 | \$ 1,307,901 | \$ 0 |
| M.S. Taff | N/A | N/A | N/A | N/A |
| | N/A | N/A | N/A | N/A |
| R.L. Killion | B&W Commercial Operations Qualified Retirement Plan | 39.583 | \$ 1,278,742 | \$ 0 |
| | B&W Commercial Operations Excess Plan | 39.583 | \$ 318,609 | \$ 0 |
| C.M. Mowry | N/A | N/A | N/A | N/A |
| | N/A | N/A | N/A | N/A |
| W.D. Nash | B&W Governmental Operations Qualified Retirement Plan | 40.333 | \$ 1,195,345 | \$ 0 |
| | B&W Governmental Operations Excess Plan | 40.333 | \$ 941,030 | \$ 0 |

Overview of Qualified Plans. McDermott maintains retirement plans that are funded by trusts and cover certain eligible regular full-time employees of McDermott and its subsidiaries, described below in the section entitled "Participation and Eligibility," except certain nonresident alien employees who are not citizens of a European Community country or who do not earn income in the United States, Canada or the United Kingdom.

- Mr. Bethards and Mr. Nash participate in the Retirement Plan for Employees of Babcock & Wilcox Government Operations (the "B&W Government Operations Qualified Retirement Plan") for the benefit of the eligible employees of The Babcock & Wilcox Company and our Governmental Operations segment;
- Mr. Killion participates in the Retirement Plan for Employees of Babcock & Wilcox Commercial Operations (the "B&W Commercial Operations Qualified Retirement Plan") for the benefit of eligible employees of Babcock & Wilcox Power Generation Group, Babcock & Wilcox Nuclear Generation Group and our Power Generation Systems segment; and
- Due to their respective employment dates, Messrs. Taff and Mowry do not participate in our defined benefit plans. For more information on our retirement plans, see "Compensation Discussion and Analysis—Post-Employment Compensation—Retirement Plans."

We sponsor the Commercial Operations Retirement Plan through B&W PGG. McDermott sponsors the Government Operations and McDermott Retirement Plans through MI. In connection with the spin-off, we will assume the Government Operations Retirement Plan and certain assets and liabilities of the McDermott Retirement Plan.

Participation and Eligibility. Generally, employees over the age of 21 years, who were hired before April 1, 2005, are eligible to participate in the B&W Governmental Operations Qualified Retirement Plan or the B&W Commercial Operations Qualified Retirement Plan.

- For participants with less than five years of service as of March 31, 2006—Benefit accruals were frozen as of that date. Affected employees now receive annual service-based company cash contributions to their Thrift Plan account.
- For participants with more than five but less than ten years of service as of January 1, 2007—If a participant made an election to do so, benefit accruals were frozen as of March 31, 2007, with the electing participants now receiving annual service-based company cash contributions to their Thrift Plan accounts.

• Frozen accrued benefits of affected employees under these plans will increase annually in line with increases in the Consumer Price Index, up to a maximum of 8% and a minimum of 1%, for each year the employee remains employed. For further discussion on the service-based company cash contributions under the Thrift Plan, see "Compensation Discussion and Analysis—Post-Employment Compensation—Retirement Plans."

Benefits. Benefits under these plans are calculated under one of two formulas:

- For participating employees originally hired by McDermott's Power Generation Systems or Government Operations segment ("Tenured Employees") before April 1, 1998—benefits are based on years of credited service and final average cash compensation (including bonuses and commissions); and
- (2) For participating employees hired before April 1, 1998 who are not Tenured Employees, and for participating employees hired on or after April 1, 1998—benefits are based on years of credited service, final average cash compensation (excluding bonuses and commissions) and anticipated social security benefits. Final average cash compensation is based on each employee's average annual earnings during the 60 successive months out of the 120 successive months before retirement in which such earnings were highest.

The present value of accumulated benefits reflected in the "Pension Benefits" table above is based on a 6.00% discount rate and the 1994 Group Annuity Mortality Table projected to 2005.

Retirement and Early Retirement. Under each of these plans, normal retirement is age 65. The normal form of payment is a single-life annuity or a 50% joint and survivor annuity, depending on the employee's marital status when payments are scheduled to begin. Early retirement eligibility and benefits under these plans depend on the employee's date of hire. Messrs. Bethards, Killion and Nash are each currently eligible for early retirement.

For Tenured Employees hired before April 1, 1998 (which includes Messrs. Bethards, Killion and Nash):

- an employee is eligible for early retirement if the employee has completed at least 15 years of credited service and attained the age of 50; and
- early retirement benefits are based on the same formula as normal retirement, but the pension benefit is unreduced if the sum of the employee's age and years of service equals 75 or greater at the date benefits commence; otherwise the pension benefit is reduced 4% for each "point" less than 75.

For employees hired on or after April 1, 1998:

- · an employee is eligible for early retirement after completing at least 15 years of credited service and attaining the age of 55; and
- early retirement benefits are based on the same formula as normal retirement, but the pension benefit is generally reduced 0.4% for each month that benefits commence before age 62.

Overview of Nonqualified Plans. To the extent benefits payable under these qualified plans are limited by Section 415(b) or 401(a)(17) of the Internal Revenue Code, pension benefits will be paid directly by McDermott's applicable subsidiaries under the terms of unfunded excess benefit plans (the "Excess Plans") maintained by them. Effective January 1, 2006, the Excess Plans were amended to limit the annual bonus payments taken into account in calculating the Tenured Employees' Excess Plan benefits to the lesser of the actual bonus paid or 25% of the prior year's base salary.

 Messrs. Bethards and Nash participate in the Restoration of Retirement Income Plan for Certain Participants in the Retirement Plan for Employees of Babcock & Wilcox Governmental Operations; and

• Mr. Killion participates in the Restoration of Retirement Income Plan for Certain Participants in the Retirement Plan for Employees of Babcock & Wilcox Commercial Operations.

We sponsor the Commercial Operations Excess Plan through B&W PGG and the Government Operations Excess Plan through our subsidiary, Babcock & Wilcox Investment Company. McDermott sponsors the McDermott Excess Plan through MI. In connection with the spin-off, we will assume certain liabilities of the McDermott Excess Plan.

Nonqualified Deferred Compensation

The following Nonqualified Deferred Compensation table summarizes our Named Executives' compensation under McDermott's nonqualified supplemental retirement plan. The compensation shown in this table is entirely attributable to the McDermott International, Inc. New Supplemental Employee Retirement Plan, or SERP, established January 1, 2005.

| | | | | | | | Agg | regate | | |
|-----------------------------|-----------|--------------|-------|------------------|------|-----------------|-------|----------|-----|-------------------|
| | Exe | cutive | | Registrant | Aggı | regate Earnings | With | drawals/ | Agg | regate Balance at |
| Name | Contribut | ions in 2009 | Contr | ibutions in 2009 | | in 2009 | Distr | ibutions | | 12/31/09 |
| B.C. Bethards | \$ | 0 | \$ | 44,948.00 | \$ | 51,718.71 | \$ | 0 | \$ | 446,531.52 |
| M.S. Taff | \$ | 0 | \$ | 41,378.00 | \$ | 34,000.95 | \$ | 0 | \$ | 139,978.13 |
| R.L. Killion ⁽¹⁾ | | N/A | | N/A | | N/A | | N/A | | N/A |
| C.M. Mowry ⁽¹⁾ | | N/A | | N/A | | N/A | | N/A | | N/A |
| W.D. Nash | \$ | 0 | \$ | 30,361.00 | \$ | 89,855.65 | \$ | 0 | \$ | 291,636.80 |

(1) No information is provided for Messrs. Killion or Mowry because neither is a participant in the SERP.

McDermott's SERP is an unfunded, defined contribution retirement plan for officers of McDermott and its operating segments selected to participate by McDermott's compensation committee. Benefits under the SERP are based on the participating officer's vested percentage in his notional account balance at the time of retirement or termination. An officer generally vests in his SERP account 20% each year, subject to accelerated vesting for death, disability and termination without cause or termination within 24 months following a change in control.

In connection with the spin-off, we expect our Board of Directors will adopt a nonqualified supplemental executive retirement plan substantially similar to McDermott's SERP.

Executive Contributions in 2009. Employee contributions are not permitted under McDermott's SERP.

Employer Contributions in 2009. McDermott makes annual contributions to participating employees' notional accounts equal to a percentage of the employee's prior-year compensation. Under the terms of the SERP, the contribution percentage does not need to be the same for each participant. Additionally, McDermott's compensation committee may make a discretionary contribution to a participant's account at any time.

For 2009, McDermott's contributions equaled 5% of the Named Executives' base salaries and annual incentive compensation awards paid in 2008. No discretionary contributions were made in 2009. All of McDermott's 2009 contributions are included in the Summary Compensation Table above as "All Other Compensation."

Aggregate Earnings in 2009. The amount reported in this table as earnings represents hypothetical accrued gains during 2009 on each Named Executive's account. The accounts are "participant-directed" in that each participating officer personally directs the investment of contributions made on his behalf. As a result, any accrued gains or losses are attributable to the performance of the Named Executive's notional mutual fund investments.

No amount of the earnings shown is reported as compensation in the Summary Compensation Table.

Aggregate Balance at 12/31/09. The balance of a participating officer's account consists of contributions made by McDermott and hypothetical accrued gains or losses. The balances shown represent the accumulated account values (including gains and losses) for each Named Executive as of December 31, 2009.

The balances shown include contributions from previous years which have been reported as compensation to the Named Executives in the Summary Compensation Table for those years—to the extent a Named Executive was included in the Summary Compensation Table during those years. The amounts and years reported are as follows:

| Named Executive | Year | An | ount Reported |
|-----------------|------|----|---------------|
| B.C. Bethards | 2008 | \$ | 31,042.00 |
| | 2007 | | N/A |
| M.S. Taff | 2008 | \$ | 31,125.00 |
| | 2007 | \$ | 20,705.85 |

In May 2009, McDermott's compensation committee amended the SERP to vest SERP balances unvested on December 31, 2008 (including future gains and losses thereon). Amounts allocated on or after January 1, 2009 vest pursuant to the participant's vested percentage based upon years of service. Accordingly, as of January 1, 2010, each Named Executive who participates in the SERP is 100% vested in his SERP balance shown, except Mr. Taff. Mr. Taff, who did not begin participating in McDermott's SERP until 2006, is 92.19% vested in his SERP balance shown (100% vested in amounts allocated to his SERP account as of December 31, 2008 and 80% vested in amounts allocated to his SERP account during 2009).

Potential Payments Upon Termination or Change in Control

The following tables show potential payments to our Named Executives under existing contracts, agreements, plans or arrangements, whether written or unwritten, for various scenarios under which a payment would be due (assuming each is applicable) involving a change in control or termination of employment of each of our Named Executives, assuming a December 31, 2009 termination date and, where applicable, using the closing price of McDermott's common stock of \$24.01 (as reported on the New York Stock Exchange) as of December 31, 2009. These tables do not reflect amounts that would be payable to the Named Executives pursuant to benefits or awards that are already vested.

The amounts reported in the below tables for McDermott stock options, restricted stock, restricted stock units, deferred stock units and performance shares represent the value of unvested and accelerated shares or units, as applicable, calculated by:

- for stock options: multiplying the number of accelerated options by the difference between the exercise price and \$24.01 (the closing price of McDermott's common stock on December 31, 2009, as reported on the New York Stock Exchange);
- for restricted stock, restricted stock units and performance shares: multiplying the number of accelerated shares or units by \$24.01 (the closing price of McDermott's common stock on December 31, 2009, as reported on the New York Stock Exchange); and
- for deferred stock units (which represent a right to receive a cash payment equal to the product of the number of vested units and the average of the highest and lowest sales price of McDermott's common stock on the vesting date): multiplying the number of accelerated units by \$24.17 (the average of the highest and lowest price of McDermott's common stock on December 31, 2009, as reported on the New York Stock Exchange).

Estimated Value of Benefits to Be Received Upon Involuntary Termination Without Cause

The following table shows the estimated value of payments and other benefits due the Named Executives assuming their involuntary termination without cause as of December 31, 2009.

| | B.C. Bethards | | M.S. Taff | R.L. Killion | | C.M. Mowry | | W.D. Nash | |
|---|---------------|--------------|--------------------|--------------|-----------|------------|-----------|-----------|-----------|
| Severance Payments | \$ | 1,789,080.00 | \$ 1,717,000.00 | \$ | 88,846.15 | \$ | 10,240.38 | \$ | 93,692.31 |
| EICP | \$ | 368,340.00 | \$ 353,500.00 | | _ | | _ | | _ |
| Supplemental Executive Retirement Plan (SERP) | | _ | \$ 10,934.53 | | N/A | | N/A | | _ |
| Benefits | \$ | 21,738.24 | \$ 35,837.50 | | _ | | _ | | _ |
| Thrift Plan | | _ | _ | | _ | | _ | | _ |
| Stock Options | \$ | 917,548.92 | \$ 1,070,388.72 | | _ | | _ | | _ |
| (unvested and accelerated) | | | | | | | | | |
| Restricted Stock | \$ | 479,071.53 | \$ 110,277.93 | | _ | | _ | | _ |
| (unvested and accelerated) | | | | | | | | | |
| Restricted Stock Units | \$ | 1,186,910.34 | \$ 1,384,632.69 | | _ | | _ | | _ |
| (unvested and accelerated) | | | | | | | | | |
| Deferred Stock Units | | _ | _ | | _ | | _ | | _ |
| (unvested and accelerated) | | | | | | | | | |
| Performance Shares | \$ | 1,027,411.91 | \$ 1,398,222.35 | | _ | | _ | | _ |
| (unvested and accelerated) | | | | | | | | | |
| Tax Gross-Up | | | | _ | | | | _ | |
| Total | \$ | 5,790,100.94 | \$ 6,080,793.72 | \$ | 88,846.15 | \$ | 10,240.38 | \$ | 93,692.31 |

Severance Payment. With the exception of Messrs. Bethards and Taff, the amounts reported represent payments under the Severance Plan for Employees of McDermott Incorporated and Participating Subsidiary and Affiliated Companies assuming the Named Executives' involuntary termination without cause. Through this severance plan, full-time employees of McDermott and participating subsidiaries are entitled to receive a severance benefit in the event their employment is terminated because of the elimination of a previously required position or previously required service, or due to the consolidation of departments, abandonment of plants or offices, or technological change or declining business activities, where such termination is intended to be permanent. The amount of severance benefit is determined based on the length of service and the employee's base salary. In general, an eligible employee is entitled to a severance benefit of one half week of base salary for each year of service, subject to a maximum of 14 weeks of pay. For more information on the amounts paid to Messrs. Bethards and Taff, see "Retention Agreements" below.

SERP. The amount reported represents Mr. Taff's SERP balance as of December 31, 2009 that would become vested on an involuntary termination not for cause. As discussed under "Compensation Discussion & Analysis—Post-Employment Compensation—Retirement Plans—Supplemental Plans," Mr. Taff was 100% vested in amounts allocated to his account as of December 31, 2008, and was 80% vested in amounts allocated to his account during 2009. Messrs. Bethards and Nash were 100% vested in their SERP balance. Messrs. Killion and Mowry do not participate in the SERP. Under the SERP, termination for "cause" means:

- the overt and willful disobedience of orders or directives issued to a person who participates in the SERP (a "Participant") that are within the scope of his duties, or any other willful and continued failure of a Participant to perform substantially his duties with McDermott (occasioned by reason other than physical or mental illness or disability) after a written demand for substantial performance is delivered to the Participant by McDermott's compensation committee or the Chief Executive Officer of McDermott which specifically identifies the manner in which McDermott's compensation committee or the Chief Executive Officer of McDermott believes that the Participant has not substantially performed his or her duties, after which the Participant will have 30 days to defend or remedy such failure to substantially perform his or her duties;
- · the willful engaging by the Participant in illegal conduct or gross misconduct which is materially and demonstrably injurious to McDermott; or
- the conviction of the Participant with no further possibility of appeal, or plea of nolo contendere by the Participant to, any felony or crime of falsehood.

For more information on the amount reported for Mr. Taff, see "Retention Agreements" below.

Retention Agreements

The amounts reported for Messrs. Bethards and Taff assume the Named Executive was terminated on December 31, 2009 in connection with our separation from McDermott. In contemplation of the spin-off, on December 10, 2009, McDermott entered into retention agreements with certain key members of our management, including Messrs. Bethards and Taff (the "Retention Agreements"). The Retention Agreements provide either a retention or severance payment to these employees in connection with the disposition of all or substantially all of the stock or assets of B&W or J. Ray McDermott, S.A. (whether by a spin-off, sale or otherwise), which are referred to as a "restructuring transaction" in the Retention Agreements. In the event the applicable employee is terminated prior to the first anniversary of the effective date of the restructuring transaction, either (1) by the employer company for any reason other than cause or disability or (2) by the employee for good reason, the employer company is required to pay the Named Executive a cash severance payment, an EICP payment, 200% of the annual cost of coverage for medical, dental and vision benefits, and an amount equal to the unvested portion of the Named Executive's Thrift Plan account. The Named Executive will also receive accelerated vesting with respect to his outstanding equity awards.

Under the Retention Agreements, "cause" is defined as:

- the willful and continued failure of the employee to perform substantially the employee's duties (occasioned by reason other than physical or mental illness or disability of the employee) after a written demand for substantial performance is delivered to the employee by the compensation committee of the employer or the Chief Executive Officer of the employer which specifically identifies the manner in which the compensation committee or the Chief Executive Officer believes that employee has not substantially performed his or her duties, after which the employee will have 30 days to defend or remedy such failure to substantially perform his or her duties;
- the willful engaging by employee in illegal conduct or gross misconduct which is materially and demonstrably injurious to the employer; or
- the conviction of the employee with no further possibility of appeal, or plea of nolo contendere by the employee to, any felony or crime of falsehood.

Severance Payment. The severance payment that may be made to each Named Executive in connection with his termination of employment following a restructuring transaction is a cash payment equal to 200% of the sum of his annual base salary prior to termination and his EICP target award applicable to the year in which the termination occurs. Severance payments under a restructuring transaction are in lieu of any payment under any severance policy generally applicable to the salaried employees of McDermott. For a hypothetical termination as of December 31, 2009, the severance payment would have been calculated based on the following base salary and target EICP awards:

- Mr. Bethards: \$526,200 base salary and \$368,340 target EICP (70% of his base salary); and
- Mr. Taff: \$505,000 base salary and \$353,500 target EICP (70% of his base salary).

EICP Payment. The EICP is an annual cash-based performance incentive plan under which payments are made in the year following the year in which performance is measured. For example, 2009 EICP awards are paid in 2010 for performance achieved in 2009. As a result, depending on the timing of the termination relative to the payment of an EICP award, a Named Executive could receive up to two EICP payments in connection with termination resulting from a restructuring transaction, as follows:

- If an EICP award for the year prior to termination is paid to other EICP participants after the date of the Named Executive's termination, the Named Executive would be entitled to a cash payment equal to the product of the Named Executive's EICP target percentage and the Named Executive's annual base salary for the applicable period. No such payment would have been due a named Executive on a December 31, 2009 termination, because the 2008 EICP awards had already been paid prior to the Named Executive's termination date.
- The Named Executive would be entitled to a prorated EICP payment based upon the Named Executive's target award for the year in which the termination occurs and the number of days in which the executive was employed with McDermott during that year. Based on a hypothetical December 31, 2009 termination, each Named Executive would have been entitled to an EICP payment equal to 100% of his 2009 target EICP. Mr. Bethards' 2009 target EICP was \$368,340 and Mr. Taff's 2009 target EICP was \$353,500.

SERP. The Retention Agreements provide for full vesting in a Named Executive's SERP account balance upon termination under a restructuring transaction. The amount reported represents Mr. Taff's SERP balance as of December 31, 2009 that would become vested upon his termination under a restructuring transaction. Mr. Bethards was 100% vested in his SERP balance as of December 31, 2009.

Benefits. The amounts reported represent two times the full annual cost of coverage for medical, dental and vision benefits provided to the Named Executive and the Named Executive's covered dependents for the year ended December 31, 2009.

Equity Awards. The amounts reported for McDermott stock options, restricted stock, restricted stock units and performance shares represent the value of unvested and accelerated stock options, shares or units, as applicable, for the 2008 and 2009 grants of such awards. The amounts reported for the 2008 and 2009 performance share grants represent the value of such grants calculated as to the initial grant of the shares.

Estimated Value of Benefits to Be Received Upon Normal Retirement

The following table shows the estimated value of payments and other benefits due the Named Executives assuming their retirement as of December 31, 2009. This table does not reflect amounts that would be payable to the Named Executives pursuant to benefits or awards that are already vested (for example, pension benefits). See the "Pension Benefits" table above for more information on pension benefits.

| | B.C. Bethards | M.S. Taff | R.L. Killion | C.M. Mowry | W.D. Nash |
|---|---------------|-----------|--------------|------------|--------------|
| Severance Payments | _ | _ | _ | _ | _ |
| EICP | _ | _ | _ | _ | _ |
| Supplemental Executive Retirement Plan (SERP) | _ | _ | N/A | N/A | _ |
| Stock Options | _ | _ | _ | _ | _ |
| (unvested and accelerated) | | | | | |
| Restricted Stock | \$119,767.88 | _ | \$ 48,140.05 | _ | \$ 11,122.63 |
| (unvested and accelerated) | | | | | |
| Restricted Stock Units | _ | _ | _ | _ | _ |
| (unvested and accelerated) | | | | | |
| Deferred Stock Units | _ | _ | _ | _ | \$ 63,663.78 |
| (unvested and accelerated) | | | | | |
| Performance Shares | \$467,700.39 | _ | \$173,294.58 | _ | \$245,060.47 |
| (unvested and accelerated) | | | | | |
| Tax Gross-Up | | | | | |
| Total | \$587,468.27 | \$ 0.00 | \$221,434.63 | \$ 0.00 | \$319,846.88 |

SERP. Under the terms of the SERP, participants are eligible for retirement upon attaining age 65. Mr. Nash is eligible for retirement under the terms of the SERP, but was 100% vested as of December 31, 2009.

Equity Awards. Generally, the terms of McDermott stock and option award grants define retirement as a voluntary termination of employment after attaining age 60 and completing 10 years of service with McDermott. Under this definition, the Named Executives eligible for retirement as of a December 31, 2009 were Messrs. Bethards, Killion and Nash. The outstanding grants of restricted stock and restricted stock units and the unvested grants of stock options provide for accelerated vesting of a percentage of the Named Executive's outstanding shares and units if his retirement date is on or after the first anniversary of the grant date, and a greater percentage of accelerated vesting if his retirement date is on or after the second anniversary of the grant date. The outstanding grants of deferred stock units vest 100% upon termination due to normal retirement under a funded or unfunded retirement plan of McDermott or any subsidiary. Mr. Nash is eligible for accelerated vesting of deferred stock units.

Estimated Value of Benefits to Be Received Upon Termination Due to Death or Disability

The following table shows the value of payments and other benefits due the Named Executives assuming their death or disability as of December 31, 2009.

| | B.C. Bethards | M.S. Taff | R.L. Killion | C.M. Mowry | W.D. Nash |
|---|--------------------|--------------------|--------------------|------------------|--------------------|
| Severance Payments | _ | _ | _ | _ | _ |
| EICP | _ | _ | _ | _ | _ |
| Supplemental Executive Retirement Plan (SERP) | _ | \$ 10,934.53 | N/A | N/A | _ |
| Stock Options | \$ 917,548.92 | \$ 1,070,388.72 | \$ 338,013.36 | \$ 216,447.84 | \$ 304,894.80 |
| (unvested and accelerated) | | | | | |
| Restricted Stock | \$ 479,071.53 | \$ 110,277.93 | \$ 192,560.20 | \$ 180,387.13 | \$ 44,490.53 |
| (unvested and accelerated) | | | | | |
| Restricted Stock Units | \$ 1,186,910.34 | \$ 1,384,632.69 | \$ 437,222.10 | \$ 279,980.61 | \$ 394,436.28 |
| (unvested and accelerated) | | | | | |
| Deferred Stock Units | _ | \$ 108,765.00 | _ | _ | \$ 63,663.78 |
| (unvested and accelerated) | | | | | |
| Performance Shares | \$ 1,738,107.91 | \$ 1,926,442.35 | \$ 654,488.59 | \$ 162,067.50 | \$ 800,181.27 |
| (unvested and accelerated) | | | | | |
| Tax Gross-Up | | | | | |
| Total | \$ 4,321,638.70 | \$ 4,611,441.22 | \$ 1,622,284.25 | \$ 838,883.08 | \$ 1,607,666.66 |

SERP. The amount reported represents Mr. Taff's SERP balance as of December 31, 2009 that would become vested on death or disability. Mr. Bethards and Mr. Nash were 100% vested in their SERP balance as of December 31, 2009. Messrs. Killion and Mowry do not participate in the SERP.

Equity Awards. Under the terms of the McDermott awards outstanding for each Named Executive as of December 31, 2009, all unvested stock awards become vested and all unvested option awards become vested and exercisable on death or disability.

Estimated Value of Benefits to Be Received Upon Change in Control

The following table shows the estimated value of payments and other benefits due the Named Executives assuming a change in control and termination as of December 31, 2009.

| | B.C. Bethards | | M.S. Taff | | R.L. Killion | | C.M. Mowry | | W.D. Nash |
|---|--------------------|----|--------------|----|--------------|----|--------------|----|--------------|
| Severance Payments | \$ 1,789,080.00 | \$ | 1,717,000.00 | | _ | | _ | | _ |
| EICP | \$ 368,340.00 | \$ | 353,500.00 | | _ | | _ | | _ |
| Supplemental Executive Retirement Plan (SERP) | _ | \$ | 10,934.53 | | N/A | | N/A | | _ |
| Benefits | \$ 21,738.24 | \$ | 35,837.50 | | _ | | _ | | _ |
| Stock Options | \$ 917,548.92 | \$ | 1,070,388.72 | \$ | 338,013.36 | \$ | 216,447.84 | \$ | 304,894.80 |
| (unvested and accelerated) | | | | | | | | | |
| Restricted Stock | \$ 479,071.53 | \$ | 110,277.93 | \$ | 192,560.20 | \$ | 180,387.13 | \$ | 44,490.53 |
| (unvested and accelerated) | | | | | | | | | |
| Restricted Stock Units | \$ 1,186,910.34 | \$ | 1,384,632.69 | \$ | 437,222.10 | \$ | 279,980.61 | \$ | 394,436.28 |
| (unvested and accelerated) | | | | | | | | | |
| Deferred Stock Units | _ | \$ | 108,765.00 | | _ | | _ | \$ | 63,663.78 |
| (unvested and accelerated) | | | | | | | | | |
| Performance Shares | \$ 2,780,406.02 | \$ | 2,991,886.10 | \$ | 1,027,579.98 | \$ | 324,135.00 | \$ | 1,194,113.34 |
| (unvested and accelerated) | | | | | | | | | |
| Tax Gross-Up | \$ 2,164,376.03 | \$ | 2,002,180.42 | | _ | | | | _ |
| Total | \$ 9,707,471.08 | \$ | 9,785,402.89 | \$ | 1,995,375.64 | \$ | 1,000,950.58 | \$ | 2,001,598.73 |

McDermott has change-in-control agreements with various officers, including Messrs. Bethards and Taff. Generally, under these agreements, if a Named Executive is terminated within one year following a change in control either (1) by McDermott for any reason other than cause or death or disability; or (2) by the Named Executive for good reason, McDermott is required to pay the Named Executive a cash severance payment, an EICP payment and, if applicable, a tax gross-up payment. In addition to these payments, the Named Executive would be entitled to various accrued benefits earned through the date of termination, such as earned but unpaid salary, earned but unused vacation and reimbursements.

Under these agreements, a "change in control" occurs if:

- a person (other than a McDermott employee benefit plan or a corporation owned by McDermott stockholders in substantially the same proportion as their ownership of McDermott voting shares) becomes the beneficial owner of 30% or more of the combined voting power of McDermott's then outstanding voting stock;
- during any period of two consecutive years, individuals who at the beginning of such period constitute McDermott's Board of Directors, and any new
 director whose election or nomination by McDermott's Board was approved by at least two-thirds of the directors of McDermott's Board then still in office
 who either were directors at the beginning of the period or whose election or nomination was previously approved, cease to constitute a majority of
 McDermott's Board;
- McDermott's stockholders approve: (1) a merger or consolidation of McDermott with another company, other than a merger or consolidation which would result in McDermott's voting securities outstanding immediately prior thereto continuing to represent at least 50% of the voting stock of McDermott or such surviving entity outstanding immediately after such merger or consolidation; (2) a plan of complete liquidation of McDermott; or (3) an agreement for the sale or disposition by McDermott of all or substantially all of McDermott's assets; or
- · any other set of circumstances is deemed by McDermott's Board in its sole discretion to constitute a change in control.

Severance Payment. The severance payment made to each Named Executive in connection with a change in control is a cash payment equal to 200% of the sum of his annual base salary prior to termination and his EICP target award applicable to the year in which the termination occurs. For a hypothetical termination as of December 31, 2009, the severance payment following a change-in-control event would have been calculated the same as a termination under a restructuring transaction. For more information, see "Retention Agreements" above.

EICP Payment. The amounts reported represent the EICP payments made to the Named Executives in connection with a change in control. The amount paid is calculated the same as a termination under a restructuring transaction. For more information, see "Retention Agreements" above.

SERP. The amount reported represents Mr. Taff's SERP balance as of December 31, 2009 that would become vested in connection with a termination of employment following a change in control. Mr. Taff was 100% vested in amounts allocated to his account as of December 31, 2008 and was 80% vested in amounts allocated to his account during 2009. The amount shown would be due to Mr. Taff if he was terminated without cause within one year after a change in control. Messrs. Bethards and Nash are 100% vested in their SERP balance. Messrs. Killion and Mowry do not participate in the SERP. Under the SERP, a "change in control" occurs under the same circumstances described above with respect to McDermott's change in control agreements.

Benefits. The amounts reported represent two times the full annual cost of coverage for medical, dental and vision benefits provided to the Named Executive and the Named Executive's covered dependents for the year ended December 31, 2009.

Tax Gross-Up. If any payment is subject to the excise tax imposed by section 4999 of the Internal Revenue Code of 1986, as amended, McDermott would reimburse the affected Named Executive for all excise taxes imposed under section 4999 and any income and excise taxes that are payable as a result of such reimbursement. The calculation of the section 4999 gross-up amount in the above table is based upon a section 4999 excise tax rate of 20%, a 35% federal income tax rate, a 1.45% Medicare tax rate and a state income tax rate of 7.75% for Mr. Bethards.

Equity Awards. Under the terms of the McDermott awards outstanding, all unvested restricted stock, restricted stock units and deferred stock units would become vested on a change in control, regardless of whether there is a subsequent termination of employment. All unvested stock options would become vested and exercisable on a change in control, regardless of whether there is a subsequent termination of employment. Performance shares are subject to accelerated vesting on a change in control, regardless of whether there is a subsequent termination of employment. Under McDermott's 2001 D&O Plan, a "change in control" occurs under the same circumstances described above with respect to McDermott's change-in-control agreements.

New Compensation Plans

2010 Long-Term Incentive Plan of The Babcock & Wilcox Company

We intend to adopt the 2010 Long-Term Incentive Plan of The Babcock & Wilcox Company (the "2010 LTIP"), which is expected to be approved by McDermott as our sole shareholder before the spin-off. The following is a general description of the 2010 LTIP.

Administration. The 2010 LTIP will be administered by the Compensation Committee of our Board of Directors. The Compensation Committee will select the participants and determine the type or types of awards and the number of shares, units or options to be granted to each participant under the 2010 LTIP. All or part of an award may be subject to conditions established by the Compensation Committee, which may include continued

service with our company, achievement of specific business objectives, increases in specified indices, attainment of specified growth rates or other measures of performance. The Compensation Committee will have full and final authority to implement the 2010 LTIP and may, from time to time, adopt rules and regulations in order to carry out the terms of the 2010 LTIP. As permitted by law, the Compensation Committee may delegate its duties under the 2010 LTIP to our chief executive officer and other senior officers.

Eligibility. Members of the Board of Directors and employees of our company and its subsidiaries (including former employees of subsidiaries and their respective subsidiaries), as well as consultants, will be eligible to participate in the 2010 LTIP. The Compensation Committee will select the participants for the 2010 LTIP. Any participant may receive more than one award under the 2010 LTIP.

Shares Available for Issuance through the 2010 LTIP. 10,000,000 shares have been approved for issuance under the 2010 LTIP, including shares to be issued in connection with the spin-off.

The 2010 LTIP provides for a number of forms of stock-based compensation, as further described below. The 2010 LTIP also permits the reuse or reissuance under the 2010 LTIP of shares of common stock underlying canceled, expired, terminated or forfeited awards granted under the 2010 LTIP. The Compensation Committee shall make appropriate adjustments in the number and kind of shares that may be issued, the number and kind of shares subject to outstanding awards, the exercise or other applicable price and other value determinations applicable to outstanding awards under the 2010 LTIP to reflect any amendment to the 2010 LTIP, stock split, stock dividend, recapitalization, merger, consolidation, reorganization, combination or exchange of shares or other similar event.

Types of Awards Under the 2010 LTIP. The Compensation Committee may award to participants incentive and nonqualified stock options, restricted stock, restricted stock units, performance shares and performance units. The forms of awards are described in greater detail below.

Stock Options. The Compensation Committee will have discretion to award incentive stock options and nonqualified stock options. A stock option is a right to purchase a specified number of shares of common stock at a specified exercise price. An incentive stock option is intended to qualify as such under Section 422 of the Code. Under the 2010 LTIP, no participant may be granted options during any fiscal year that are exercisable for more than 1,200,000 shares of our common stock. The exercise price of an option may not be less than the fair market value of a share of our common stock on the date of grant. Subject to the specific terms of the 2010 LTIP, the Compensation Committee will have discretion to determine the number of shares, the exercise price, the terms and conditions of exercise and such additional limitations on and terms of option grants as it deems appropriate.

Options granted to participants under the 2010 LTIP will expire at such times as the Compensation Committee determines at the time of the grant, but no option will be exercisable later than seven years after the date of grant. Each option award agreement will set forth the extent to which the participant will have the right to exercise the option following termination of the participant's employment. The termination provisions will be determined within the discretion of the Compensation Committee, need not be uniform among all participants and may reflect distinctions based on reasons for termination of employment.

Upon the exercise of an option granted under the 2010 LTIP, the exercise price is payable in full to us (1) in cash; (2) if permitted, by tendering shares of our common stock having a fair market value at the time of exercise equal to the total option price; (3) if permitted, by a combination of (1) and (2); or (4) by any other method approved by the Compensation Committee in its sole discretion.

Restricted Stock. The Compensation Committee also is authorized to award restricted shares of common stock under the 2010 LTIP on such terms and conditions as it shall establish. Although recipients will own and have the right to vote restricted shares from the date of grant, they will not have the right to sell or otherwise transfer the shares during the applicable period of restriction or until earlier satisfaction of other conditions

imposed by the Compensation Committee in its sole discretion. The award agreement will specify the periods of restriction, any restrictions based on achievement of specific performance objectives, restrictions under applicable federal or state securities laws and such other terms it deems appropriate. Unless the Compensation Committee otherwise determines, participants will receive dividends on their shares of restricted stock. The Compensation Committee in its discretion may apply any restrictions to the dividends that it deems appropriate.

Each award agreement for restricted stock will set forth the extent to which the participant will have the right to retain unvested shares of restricted stock following termination of the participant's employment. These provisions will be determined in the sole discretion of the Compensation Committee, need not be uniform among all participants and may reflect distinctions based on reasons for termination of employment.

Restricted Stock Units. An award of a restricted stock unit constitutes an agreement by us to deliver shares of our common stock or to pay an amount equal to the fair market value of a share of our common stock for each restricted stock unit to a participant in the future. Restricted stock units may be granted by the Compensation Committee on such terms and conditions as it may establish. The restricted stock unit award agreement will specify the vesting period or periods, any specific performance objectives and such other conditions as may apply to the award. During the applicable vesting period, participants will have no voting rights with respect to the shares of common stock underlying a restricted stock unit grant. However, participants shall, unless the Compensation Committee otherwise determines, receive dividend equivalents on the shares underlying their restricted stock unit grant in the form of cash or additional restricted stock units if a dividend is paid with respect to shares of our common stock.

Each award agreement for restricted stock units will set forth the extent to which the participant will have the right to retain unvested restricted stock units following termination of employment. These provisions will be determined in the sole discretion of the Compensation Committee, need not be uniform among all participants and may reflect distinctions based on reasons for termination of employment.

No more than 1,200,000 shares of common stock may be granted in the form of awards of restricted stock and restricted stock units to any participant in any fiscal year.

Performance Units and Performance Shares. Performance units and performance shares are forms of performance awards that are subject to the attainment of one or more pre-established performance goals during a designated performance period. Performance units and performance shares may be granted by the Compensation Committee at any time in such amounts and on such terms as the Compensation Committee determines. Each performance unit will have an initial value that is established by the Compensation Committee at the time of grant.

Each performance unit and performance share will have an initial value equal to the fair market value of a share of our common stock on the date of grant. The Compensation Committee in its discretion will determine the applicable performance period and will establish performance goals for any given performance period. When the performance period expires, the holder of performance units or performance shares will be entitled to receive a payout on the performance units or performance shares earned over the performance period based on the extent to which the performance goals have been achieved.

Participants holding performance units or performance shares are not entitled to receive dividend units for dividends declared with respect to shares of our common stock underlying such awards. Payment of performance units and performance shares may be made in cash or in shares of common stock that have an aggregate fair market value equal to the earned performance units or performance shares on the last day of the applicable performance period. Each award agreement will set forth the extent to which the participant will have the right to receive a payout of these performance units or performance shares following termination of the participant's employment. The termination provisions will be determined in the sole discretion of the Compensation Committee, need not be uniform among all participants and may reflect distinctions based on reasons for termination of employment.

No more than 1,200,000 shares of common stock may be granted in the form of awards of performance shares to any participant in any fiscal year. No more than \$6,000,000 may be paid in cash to any participant with respect to performance units granted in any fiscal year, as valued on the date of each grant.

Performance Measures. The Compensation Committee may grant awards under the 2010 LTIP to eligible employees subject to the attainment of specified performance measures. The number of performance-based awards granted to an employee in any year will be determined by the Compensation Committee in its sole discretion, subject to the limitations set forth in the 2010 LTIP. The value of each performance-based award will be determined based on the achievement of pre-established, objective performance goals during each performance period. The duration of a performance period is set by the Compensation Committee. A new performance period may begin every year, or at more frequent or less frequent intervals, as determined by the Compensation Committee.

The Compensation Committee will establish the objective performance goals applicable to the valuation of performance-based awards granted in each performance period, the performance measures that will be used to determine the achievement of those performance goals and any formulas or methods to be used to determine the value of the performance-based awards.

Performance measures will be defined by the Compensation Committee on a consolidated, segment, group or division basis and/or in comparison to one or more peer groups or indices. Performance measures selected by the Compensation Committee will be based on one or more of the following: cash flow; cash flow return on capital; cash flow return on assets; cash flow return on equity; net income; return on capital; return on assets; return on equity; share price; carnings per share; earnings before interest and taxes; earnings before interest, taxes, depreciation and amortization; total and relative shareholder return; operating income; return on net assets; gross or operating margins; safety and economic value added. Following the end of a performance period, the Compensation Committee will determine the value of the Performance-based awards granted for the period based on its determination of the degree of attainment of the pre-established objective performance goals. The Compensation Committee will also have discretion to reduce (and, if intended to qualify as performance-based compensation under Section 162(m) of the Code, not to increase) the value of a performance-based award to "Covered Employees," as defined in Section 162(m) of the Code.

Deferrals. The Compensation Committee will have the discretion to provide for the deferral of an award or to permit participants to elect to defer payment of some or all types of awards in a manner consistent with the requirements of Section 409A of the Code.

Change in Control. The treatment of outstanding awards upon the occurrence of a change in control (as defined in the 2010 LTIP) will be determined in the sole discretion of the Compensation Committee and will be described in the applicable award agreements and need not be uniform among all awards granted under the 2010 LTIP.

Amendments. The 2010 LTIP may be modified, altered, suspended or terminated by the Board of Directors at any time and for any purpose that the Board of Directors deems appropriate, but no amendment to the 2010 LTIP shall adversely affect any outstanding awards without the affected participant's consent, and stockholder approval may be required by applicable law, regulation or stock exchange requirements.

Transferability. Except as otherwise specified in a participant's award agreement, no award granted pursuant to, and no right to payment under, the 2010 LTIP will be assignable or transferable by a 2010 LTIP participant except by will or by the laws of descent and distribution or pursuant to a qualified domestic relations order, and any right granted under the 2010 LTIP will be exercisable during a participant's lifetime only by the participant or by the participant's guardian or legal representative.

Duration of the 2010 LTIP. The 2010 LTIP will remain in effect until all options and rights granted thereunder have been satisfied or terminated pursuant to the terms of the 2010 LTIP and all performance periods for performance-based awards granted thereunder have been completed. However, in no event will any award be granted under the 2010 LTIP on or after [], 2020.

The foregoing description of the 2010 LTIP does not purport to be complete and is qualified by reference to the 2010 LTIP.

The Babcock & Wilcox Company Executive Compensation Plan

We intend to adopt The Babcock & Wilcox Company Executive Incentive Compensation Plan (the "B&W EICP") before the spin-off. The following is a general description of the B&W EICP.

The B&W EICP will be a cash incentive plan administered by the Compensation Committee of our Board of Directors. Except as prohibited by law, the Compensation Committee may generally delegate its duties under the B&W EICP to our chief executive officer and other executive officers.

All of our full-time salaried employees will be eligible to participate in the B&W EICP. Our Chief Executive Officer will automatically participate. Actual participation in the EICP by other employees will be based upon recommendations by our Chief Executive Officer, subject to approval by the Compensation Committee.

The Compensation Committee will establish, for each plan year, performance goals and award opportunities, in writing, which correspond to various levels of achievement of the preestablished performance goals based upon any combination of corporate, segment, group, divisional or individual goals. The award opportunity for any participant who is included in the group of "covered employees" (as defined in Section 162(m) of the Code) will be based on the following performance criteria:

- the award to be paid upon meeting preestablished targeted performance results;
- · the potential final award in relation to the various levels of achievement of the preestablished performance goals; and
- · company, segment, group or division performance in relation to the preestablished performance goals.

Performance measures that may be used to determine award opportunities for Named Executive Officers are generally limited to cash flow, cash flow return on capital, cash flow return on equity, net income, operating income, return on capital, return on assets, return on equity and economic value added (each as defined in the B&W EICP) and other measures consistent with deductibility under Code Section 162(m). Once established, performance goals normally cannot be changed during the plan year. However, the Compensation Committee may adjust performance goals to account for changes in accounting principles, unusual, nonrecurring events and extraordinary items, to the extent not inconsistent with Code Section 162(m). The Compensation Committee shall have the authority to reduce or eliminate final awards, based upon any criteria it deems appropriate. In addition, the Compensation Committee may use such other performance measures, including subjective measures, and make adjustments to performance goals during the plan year, if the Compensation Committee determines that it is advisable for an award not to qualify for the performance-based exception from the deductibility limitations of Code Section 162(m).

Following the end of each plan year, awards will be computed for each plan participant. Final individual awards may vary above or below the target award, based on the level of achievement of the preestablished performance goal. The B&W EICP will place a limit of \$3,000,000 on payouts to any one plan participant in any given plan year for tax deductibility reasons but grant authority to the Compensation Committee to exceed those limits in its discretion. The Board of Directors may amend the B&W EICP from time to time.

The foregoing description of the B&W EICP does not purport to be complete and is qualified by reference to the B&W EICP.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

Before the spin-off, all of the outstanding shares of our common stock are and will be owned beneficially and of record by McDermott. The following table sets forth information with respect to the expected beneficial ownership of our common stock immediately following completion of the spin-off by:

- · each stockholder who is expected following the spin-off to beneficially own more than 5% of our common stock;
- each executive officer named in the Summary Compensation Table;
- · each of our directors; and
- · all of our executive officers and directors as a group.

We have based the percentage of class amounts set forth below on each indicated person's beneficial ownership of McDermott common stock as of [], 2010, unless we indicate some other basis for the share amounts, and based on the distribution of one share of our common stock for every [] shares of McDermott common stock outstanding. To the extent our directors and executive officers own unrestricted shares of McDermott common stock at the time of the spin-off, they will participate in the distribution of shares of common stock in the spin-off on the same terms as other holders of McDermott common stock. Following the spin-off, we will have an aggregate of approximately [] million shares of common stock outstanding, based on approximately [] million shares of McDermott common stock outstanding on [], 2010. The number of shares beneficially owned by each stockholder, director or officer is determined according to the rules of the SEC and the information is not necessarily indicative of beneficial ownership for any other purpose. The mailing address for each of the directors and executive officers is c/o The Babcock & Wilcox Company, The Harris Building, 13024 Ballantyne Corporate Place, Suite 700, Charlotte, North Carolina 28277.

Shares of Common Stock to

| | be Beneficia | |
|--|--------------|-----------|
| Name of Beneficial Owner | Number | Percent |
| T. Rowe Price Associates, Inc. | | 12.44%(1) |
| PRIMECAP Management Company | | 5.68%(2) |
| BlackRock Inc. | | 5.17%(3) |
| John A. Fees | | |
| Robert W. Goldman | | |
| Stephen G. Hanks | | |
| Oliver D. Kingsley, Jr. | | |
| D. Brad McWilliams | | |
| Richard W. Mies | | |
| Brandon C. Bethards | | |
| Michael S. Taff | | |
| Richard L. Killion | | |
| Christofer M. Mowry | | |
| Winfred D. Nash | | |
| All executive officers and directors as a group (16 persons) | | |

- (1) As reported on Schedule 13G/A with respect to ownership of McDermott common stock filed with the SEC on February 12, 2010. The Schedule 13G/A reports beneficial ownership of 28,706,971 shares of McDermott's common stock, sole voting power over 7,314,005 shares and sole dispositive power over 28,706,971 shares.
- McDermott's common stock, sole voting power over 7,314,005 shares and sole dispositive power over 28,706,971 shares.

 (2) As reported on Schedule 13G/A with respect to ownership of McDermott common stock filed with the SEC on February 11, 2010. The Schedule 13G/A reports beneficial ownership of 13,114,160 shares of McDermott's common stock and sole voting power over 7,472,760 shares and sole dispositive power over 13,114,160 shares.
- (3) As reported on Schedule 13G with respect to ownership of McDermott common stock filed with the SEC on January 29, 2010. The Schedule 13G reports beneficial ownership of 11,944,038 shares of McDermott's common stock and sole voting power and sole dispositive power over 11,944,038 shares.

DESCRIPTION OF CAPITAL STOCK

Introduction

In the discussion that follows, we have summarized selected provisions of our certificate of incorporation and bylaws relating to our capital stock. This summary is not complete. This discussion is subject to the relevant provisions of Delaware law and is qualified in its entirety by reference to our restated certificate of incorporation and our amended and restated bylaws. You should read the provisions of our restated certificate of incorporation and our amended and restated bylaws as currently in effect for provisions that may be important to you. We have filed copies of those documents with the SEC, and they are incorporated by reference as exhibits to the registration statement on Form 10 of which this information statement forms a part. See "Where You Can Find More Information."

Authorized Capital Stock

Our authorized capital stock consists of:

- · 325,000,000 shares of common stock; and
- 75,000,000 shares of preferred stock, issuable in series.

Common Stock

Each share of our common stock entitles its holder to one vote in the election of each director and on all other matters voted on generally by our stockholders, other than any matter that (1) solely relates to the terms of any outstanding series of preferred stock or the number of shares of that series and (2) does not affect the number of authorized shares of preferred stock or the powers, privileges and rights pertaining to the common stock. No share of our common stock affords any cumulative voting rights. This means that the holders of a majority of the voting power of the shares voting for the election of directors can elect all directors to be elected if they choose to do so. Our board of directors may grant holders of preferred stock, in the resolutions creating the series of preferred stock, the right to vote on the election of directors or any questions affecting our company.

Holders of our common stock will be entitled to dividends in such amounts and at such times as our board of directors in its discretion may declare out of funds legally available for the payment of dividends. We currently intend to retain our entire available discretionary cash flow to finance the growth, development and expansion of our business and do not anticipate paying any cash dividends on the common stock in the foreseeable future. Any future dividends will be paid at the discretion of our board of directors after taking into account various factors, including:

- · general business conditions;
- · industry practice;
- · our financial condition and performance;
- · our future prospects;
- · our cash needs and capital investment plans;

- · our obligations to holders of any preferred stock we may issue;
- · income tax consequences; and
- · the restrictions Delaware and other applicable laws and our credit arrangements then impose.

In addition, the credit agreement relating to our revolving credit facility restricts our ability to pay cash dividends.

If we liquidate or dissolve our business, the holders of our common stock will share ratably in all our assets that are available for distribution to our stockholders after our creditors are paid in full and the holders of all series of our outstanding preferred stock, if any, receive their liquidation preferences in full.

Our common stock has no preemptive rights and is not convertible or redeemable or entitled to the benefits of any sinking or repurchase fund. All shares of common stock to be distributed in connection with the spin-off will be fully paid and nonassessable.

Our shares of common stock have been approved for listing on the New York Stock Exchange under the symbol "BWC."

The transfer agent and registrar for our common stock is Computershare Trust Company, N.A.

Preferred Stock

At the direction of our board of directors, without any action by the holders of our common stock, we may issue one or more series of preferred stock from time to time. Our board of directors can determine the number of shares of each series of preferred stock, the designation, powers, preferences and relative, participating, optional or other special rights, and the qualifications, limitations or restrictions applicable to any of those rights, including dividend rights, voting rights, conversion or exchange rights, terms of redemption and liquidation preferences, of each series.

Undesignated preferred stock may enable our board of directors to render more difficult or to discourage an attempt to obtain control of our company by means of a tender offer, proxy contest, merger or otherwise, and thereby to protect the continuity of our management. The issuance of shares of preferred stock may adversely affect the rights of our common stockholders. For example, any preferred stock issued may rank prior to the common stock as to dividend rights, liquidation preference or both, may have full or limited voting rights and may be convertible into shares of common stock. As a result, the issuance of shares of preferred stock, or the issuance of rights to purchase shares of preferred stock, may discourage an unsolicited acquisition proposal or bids for our common stock or may otherwise adversely affect the market price of our common stock or any existing preferred stock.

Limitation on Directors' Liability

Delaware law authorizes Delaware corporations to limit or eliminate the personal liability of their directors to them and their stockholders for monetary damages for breach of a director's fiduciary duty of care. The duty of care requires that, when acting on behalf of the corporation, directors must exercise an informed business judgment based on all material information reasonably available to them. Absent the limitations Delaware law authorizes, directors of Delaware corporations are accountable to those corporations and their stockholders for monetary damages for conduct constituting gross negligence in the exercise of their duty of care. Delaware law enables Delaware corporations to limit available relief to equitable remedies such as injunction or rescission. Our certificate of incorporation limits the liability of our directors to us and our stockholders to the fullest extent Delaware law permits. Specifically, no director will be personally liable for monetary damages for any breach of the director's fiduciary duty as a director, except for liability:

- · for any breach of the director's duty of loyalty to us or our stockholders;
- · for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law;

- for unlawful payments of dividends or unlawful stock repurchases or redemptions as provided in Section 174 of the Delaware General Corporation Law;
- · for any transaction from which the director derived an improper personal benefit.

This provision could have the effect of reducing the likelihood of derivative litigation against our directors and may discourage or deter our stockholders or management from bringing a lawsuit against our directors for breach of their duty of care, even though such an action, if successful, might otherwise have benefited us and our stockholders. Our bylaws provide indemnification to our officers and directors and other specified persons with respect to their conduct in various capacities. See "Indemnification of Directors and Officers."

Statutory Business Combination Provision

As a Delaware corporation, we are subject to Section 203 of the Delaware General Corporation Law. In general, Section 203 prevents an "interested stockholder," which is defined generally as a person owning 15% or more of a Delaware corporation's outstanding voting stock or any affiliate or associate of that person, from engaging in a broad range of "business combinations" with the corporation for three years following the date that person became an interested stockholder unless:

- before that person became an interested stockholder, the board of directors of the corporation approved the transaction in which that person became an interested stockholder or approved the business combination;
- on completion of the transaction that resulted in that person's becoming an interested stockholder, that person owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, other than stock held by (1) directors who are also officers of the corporation or (2) any employee stock plan that does not provide employees with the right to determine confidentially whether shares held subject to the plan will be tendered in a tender or exchange offer; or
- following the transaction in which that person became an interested stockholder, both the board of directors of the corporation and the holders of at least two-thirds of the outstanding voting stock of the corporation not owned by that person approve the business combination.

Under Section 203, the restrictions described above also do not apply to specific business combinations proposed by an interested stockholder following the announcement or notification of designated extraordinary transactions involving the corporation and a person who had not been an interested stockholder during the previous three years or who became an interested stockholder with the approval of a majority of the corporation's directors, if a majority of the directors who were directors prior to any person's becoming an interested stockholder during the previous three years, or were recommended for election or elected to succeed those directors by a majority of those directors, approve or do not oppose that extraordinary transaction.

Anti-Takeover Effects of Provisions of our Certificate of Incorporation and Bylaws

Some of the provisions of our certificate of incorporation and bylaws discussed below may have the effect, either alone or in combination with Section 203 of the Delaware General Corporation Law, of making more difficult or discouraging a tender offer, proxy contest, merger or other takeover attempt that our board of directors opposes but that a stockholder might consider to be in its best interest.

Our certificate of incorporation provides that our stockholders may act only at an annual or special meeting of stockholders and may not act by written consent. Our bylaws provide that only a majority of our board of directors or the chairman of our board of directors may call a special meeting of our board of directors or our stockholders.

Our certificate of incorporation provides for a classified board of directors. Except for directors that our preferred stockholders may elect, our board of directors is divided into three classes, with the directors of each class as nearly equal in number as possible. At each annual meeting of our stockholders, the term of a different class of our directors will expire. As a result, we contemplate that our stockholders will elect approximately one-third of our board of directors each year. Our board of directors believes that a classified board structure facilitates continuity and stability of leadership and policy by helping ensure that, at any given time, a majority of our directors will have prior experience as directors of our company and will be familiar with its business and operations. This will, in the view of our board of directors, permit more effective long-term planning and help create long-term value for our stockholders. Board classification could, however, prevent a party who acquires control of a majority of our outstanding voting stock from obtaining control of our board of directors until the second annual stockholders' meeting following the date that party obtains that control. This system of electing and removing directors may discourage a third party from making a tender offer or otherwise attempting to obtain control of us, because it generally makes it more difficult for stockholders to replace a majority of the directors.

Our certificate of incorporation provides that the number of directors will be fixed exclusively by, and may be increased or decreased exclusively by, our board of directors from time to time, but will not be less than three. Our certificate of incorporation provides that directors may be removed only with cause or upon a board determination (as such terms are defined in our certificate of incorporation) and, in either case, by a vote of at least 80% of the voting power of our outstanding voting stock. A vacancy on our board of directors may be filled by a vote of a majority of the directors in office, and a director appointed to fill a vacancy serves for the remainder of the term of the class of directors in which the vacancy occurred. These provisions will prevent our stockholders from removing incumbent directors without cause and filling the resulting vacancies with their own nominees.

Our bylaws contain advance notice and other procedural requirements that apply to stockholder nominations of persons for election to our board of directors at any annual or special meeting of stockholders and to stockholder proposals that stockholders take any other action at any annual meeting. In the case of any annual meeting, a stockholder proposing to nominate a person for election to our board of directors or proposing that any other action be taken must give our corporate secretary written notice of the proposal not less than 90 days and not more than 120 days before the anniversary of the date of the immediately preceding annual meeting of stockholders. These stockholder proposal deadlines are subject to exceptions if the pending annual meeting date is more than 30 days prior to or more than 30 days after the anniversary of the immediately preceding annual meeting. If the chairman of our board of directors or a majority of our board of directors calls a special meeting of stockholders for the election of directors, a stockholder proposing to nominate a person for that election must give our corporate secretary written notice of the proposal not earlier than 120 days prior to that special meeting and not later than the last to occur of (1) 90 days prior to that special meeting or (2) the 10th day following the day we publicly disclose the date of the special meeting. Our bylaws prescribe specific information that any such stockholder notice must contain. These advance notice provisions may have the effect of precluding a contest for the election of our directors or the consideration of stockholder proposals if the proper procedures are not followed, and of discouraging or deterring a third party from conducting a solicitation of proxies to elect its own slate of directors or to approve its own proposal, without regard to whether consideration of those nominees or proposals might be harmful or beneficial to us and our stockholders.

Our certificate of incorporation provides that our stockholders may adopt, amend and repeal our bylaws at any regular or special meeting of stockholders by a vote of at least 80% of the voting power of our outstanding voting stock, provided the notice of intention to adopt, amend or repeal the bylaws has been included in the notice of that meeting. Our certificate of incorporation also confers on our board of directors the power to adopt, amend or repeal our bylaws with the affirmative vote of a majority of the directors then in office.

As discussed above under "—Preferred Stock," our certificate of incorporation authorizes our board of directors, without the approval of our stockholders, to provide for the issuance of all or any shares of our

preferred stock in one or more series and to determine the designation, powers, preferences and relative, participating, optional or other special rights, and the qualifications, limitations or restrictions applicable to any of those rights, including dividend rights, voting rights, conversion or exchange rights, terms of redemption and liquidation preferences, of each series. The issuance of shares of our preferred stock or rights to purchase shares of our preferred stock could discourage an unsolicited acquisition proposal. In addition, under some circumstances, the issuance of preferred stock could adversely affect the voting power of our common stockholders.

In addition to the purposes described above, these provisions of our certificate of incorporation and bylaws are also intended to increase the bargaining leverage of our board of directors, on behalf of our stockholders, in any future negotiations concerning a potential change of control of our company. Our board of directors has observed that certain tactics that bidders employ in making unsolicited bids for control of a corporation, including hostile tender offers and proxy contests, have become relatively common in modern takeover practice. Our board of directors considers those tactics to be highly disruptive to a corporation and often contrary to the overall best interests of its stockholders. In particular, bidders may use these tactics in conjunction with an attempt to acquire a corporation at an unfairly low price. In some cases, a bidder will make an offer for less than all the outstanding capital stock of the target company, potentially leaving stockholders with the alternatives of partially liquidating their investment at a time that may be disadvantageous to them or retaining an investment in the target company under substantially different management with objectives that may not be the same as the new controlling stockholder. The concentration of control in our company that could result from such an offer could deprive our remaining stockholders of the benefits of listing on the New York Stock Exchange and public reporting under the Exchange Act.

While our board of directors does not intend to foreclose or discourage reasonable merger or acquisition proposals, it believes that value for our stockholders can be enhanced by encouraging would-be acquirers to forego hostile or coercive tender offers and negotiate with the Board terms that are fair to all stockholders. Our board of directors believes that the provisions described above will (1) discourage disruptive tactics and takeover attempts at unfair prices or on terms that do not provide all stockholders with the opportunity to sell their stock at a fair price and (2) encourage third parties who may seek to acquire control of our company to initiate such an acquisition through negotiations directly with our board of directors. Our Board also believes these provisions will help give it the time necessary to evaluate unsolicited offers, as well as appropriate alternatives, in a manner that assures fair treatment of our stockholders. Our board of directors recognizes that a takeover might in some circumstances be beneficial to some or all of our stockholders, but, nevertheless, believes that the benefits of seeking to protect its ability to negotiate with the proponent of an unfriendly or unsolicited proposal to take over or restructure our company outweigh the disadvantages of discouraging those proposals.

INDEMNIFICATION OF DIRECTORS AND OFFICERS

Section 145 of the Delaware General Corporation Law provides that a corporation may indemnify directors and officers as well as other employees and individuals against expenses including attorneys' fees, judgments, fines and amounts paid in settlement in connection with various actions, suits or proceedings, whether civil, criminal, administrative or investigative (other than an action by or in the right of the corporation, such as a derivative action), if they acted in good faith and in a manner they reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, if they had no reasonable cause to believe their conduct was unlawful. A similar standard is applicable in the case of any actions by or in the right of the corporation, except that indemnification only extends to expenses, including attorneys' fees, incurred in connection with the defense or settlement of such actions, and the statute requires court approval before there can be any indemnification where the person seeking indemnification has been found liable to the corporation. The statute provides that it is not exclusive of other indemnification that may be granted by a corporation's certificate of incorporation, bylaws, agreement, a vote of stockholders or disinterested directors or otherwise.

Our certificate of incorporation provides that we will indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may be amended, any person who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that such person or a person for whom such person is the legal representative, is or was a director or officer of us or, while a director or officer of us, is or was serving at our request as a director, officer, employee or agent of another corporation or of a partnership, joint venture, limited liability company, trust or other enterprise, including service with respect to employee benefit plans, against all liability and losses suffered and expenses (including attorneys' fees) incurred by such person in connection with such action, suit or proceeding. Our certificate of incorporation also provides that we will pay the expenses incurred by a director or officer in defending any such proceeding in advance of its final disposition, subject to such person providing us with specified undertakings. Notwithstanding the foregoing, our certificate of incorporation provides that we shall be required to indemnify or make advances to a person in connection with a proceeding (or part thereof) initiated by such person only if the proceeding (or part thereof) was authorized by our board of directors. These rights are not exclusive of any other right that any person may have or may acquire under any statute, provision of our certificate of incorporation, bylaws, agreement, vote of stockholders or disinterested directors or otherwise. No amendment, modification or repeal of those provisions will in any way adversely affect any right or protection under those provisions of any person in respect of any act or omission occurring prior to the time of such amendment, modification or repeal.

Our certificate of incorporation also permits us to secure and maintain insurance on behalf of any of our directors, officers, employees or agents and each person who is, or was, serving at our request as a director, officer, employee or agent of another corporation, partnership, joint venture, limited liability company, trust or other enterprise for any liability asserted against and incurred by such person in any such capacity. We intend to obtain directors' and officers' liability insurance providing coverage to our directors and officers.

INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

The combined financial statements and financial statement schedules of The Babcock & Wilcox Operations of McDermott International, Inc. as of December 31, 2009 and 2008 and for the three years ended December 31, 2009 included in this information statement have been audited by Deloitte & Touche LLP, an independent registered public accounting firm, as stated in their report appearing herein, which report expresses an unqualified opinion on the financial statements and financial statement schedules and includes an explanatory paragraph referring to the basis of presentation of the combined financial statements.

WHERE YOU CAN FIND MORE INFORMATION

We have filed with the SEC a registration statement on Form 10 under the Exchange Act relating to the common stock being distributed in the spin-off. This information statement forms a part of that registration statement but does not contain all of the information set forth in the registration statement and the exhibits and schedules to the registration statement. For further information relating to us and the shares of our common stock, reference is made to the registration statement, including its exhibits and schedules. Statements made in this information statement relating to any contract or other document are not necessarily complete and you should refer to the exhibits attached to the registration statement for copies of the actual contract or document. You may review a copy of the registration statement, including its exhibits and schedules, at the SEC's Public Reference Room, located at 100 F Street, NE, Washington, D.C. 20549 or on the SEC's Web site at http://www.sec.gov. You may obtain a copy of the registration statement from the SEC's Public Reference Room upon payment of prescribed fees. Please call the SEC at (800) SEC-0330 for further information on the operation of the Public Reference Room.

As a result of the spin-off, we will become subject to the information and reporting requirements of the Exchange Act and, in accordance with the Exchange Act, we will file periodic reports, proxy statements and other information with the SEC. Those periodic reports, proxy statements and other information will be available for inspection and copying at the SEC's Public Reference Room and the SEC's Web site at http://www.sec.gov.

We intend to furnish holders of our common stock with annual reports containing combined financial statements prepared in accordance with U.S. generally accepted accounting principles and audited and reported on, with an opinion expressed, by an independent registered public accounting firm.

We plan to make available free of charge on our Web site, our Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and amendments to those reports as soon as reasonably practicable after we electronically file or furnish such materials to the SEC. All of these documents will be made available free of charge on our Web site, www.babcock.com. The information on our Web site is not, and shall not be deemed to be, a part of this information statement or incorporated into any other filings we make with the SEC.

No person is authorized to give any information or to make any representations with respect to the matters described in this information statement other than those contained in this information statement or in the documents incorporated by reference in this information statement and, if given or made, such information or representation must not be relied upon as having been authorized by us or McDermott. Neither the delivery of this information statement nor consummation of the spin-off shall, under any circumstances, create any implication that there has been no change in our affairs or those of McDermott since the date of this information statement, or that the information in this information statement is correct as of any time after its date.

INDEX TO FINANCIAL STATEMENTS

| | Number |
|--|--------|
| Combined financial statements for The Babcock & Wilcox Operations of McDermott International, Inc. for the years ended December 31, 2009, 2008 and | |
| 2007. | |
| Report of Independent Registered Public Accounting Firm | F-1 |
| Combined Balance Sheets | F-2 |
| Combined Statements of Income | F-4 |
| Combined Statements of Comprehensive Income (Loss) | F-5 |
| Combined Statements of Parent Equity (Deficit) | F-6 |
| Combined Statements of Cash Flows | F-7 |
| Notes to Combined Financial Statements | F-8 |
| Condensed Combined financial statements for The Babcock & Wilcox Operations of McDermott International, Inc. for the three months ended March 31, | |
| 2010 and 2009 (Unaudited). | |
| Condensed Combined Balance Sheets (Unaudited) (March 31, 2010 and December 31, 2009) | F-55 |
| Condensed Combined Statements of Income (Unaudited) | F-57 |
| Condensed Combined Statements of Comprehensive Income (Unaudited) | F-58 |
| Condensed Combined Statements of Parent Equity (Deficit) (Unaudited) | F-59 |
| Condensed Combined Statements of Cash Flows (Unaudited) | F-60 |
| Notes to Condensed Combined Financial Statements (Unaudited) | F-61 |

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors of McDermott International, Inc.:

We have audited the accompanying combined balance sheets of The Babcock & Wilcox Operations of McDermott International, Inc. (the "Company") as of December 31, 2009 and 2008, and the related combined statements of income, comprehensive income (loss), parent equity (deficit), and cash flows for each of the three years in the period ended December 31, 2009. Our audits also included the financial statement schedules listed in the index at Item 15. These financial statements and financial statement schedules are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements and financial statement schedules based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, such combined financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2009 and 2008, and the results of its operations and its cash flows for each of the three years in the period ended December 31, 2009, in conformity with accounting principles generally accepted in the United States of America. Also, in our opinion, such financial statement schedules, when considered in relation to the basic combined financial statements taken as a whole, present fairly in all material respects the information set forth therein.

As discussed in Note 1 to the combined financial statements, the accompanying combined financial statements have been prepared from the separate records maintained by the Company and may not necessarily be indicative of the conditions that would have existed or the results of operations if the Company had been operated as an unaffiliated entity. Portions of certain expenses include allocations of corporate items applicable to McDermott International, Inc. as a whole.

/S/ DELOITTE & TOUCHE LLP

Houston, Texas March 12, 2010

| | Decen | nber 31, |
|---|--------------|--------------|
| | 2009 | 2008 |
| | (In the | ousands) |
| ASSETS | | |
| Current Assets: | | |
| Cash and cash equivalents | \$ 469,468 | \$ 279,646 |
| Restricted cash and cash equivalents (Note 1) | 15,305 | 7,116 |
| Investments (Note 11) | 14 | 50,642 |
| Accounts receivable—trade, net | 319,861 | 395,025 |
| Accounts receivable—other | 39,289 | 39,153 |
| Contracts in progress | 245,998 | 210,719 |
| Inventories (Note 1) | 98,644 | 125,560 |
| Deferred income taxes (Note 4) | 96,680 | 95,045 |
| Other current assets | 21,456 | 19,928 |
| Total Current Assets | 1,306,715 | 1,222,834 |
| Property, Plant and Equipment | 945,298 | 840,727 |
| Less accumulated depreciation | 515,237 | 469,902 |
| Net Property, Plant and Equipment | 430,061 | 370,825 |
| Investments (Note 11) | 73,540 | 68,442 |
| Goodwill (Note 1) | 262,866 | 263,152 |
| Deferred Income Taxes (Note 4) | 270,002 | 306,369 |
| Investments in Affiliates (Note 3) | 68,327 | 61,627 |
| Note Receivable from Affiliate (Note 5) | 42,573 | 36,619 |
| Other Assets | 149,775 | 176,973 |
| TOTAL | \$ 2,603,859 | \$ 2,506,841 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. COMBINED BALANCE SHEETS

| | Decem | nber 31, |
|--|--------------|--------------|
| | 2009 | 2008 |
| | (In tho | usands) |
| LIABILITIES AND PARENT EQUITY (DEFICIT) | | |
| Current Liabilities: | | |
| Notes payable and current maturities of long-term debt | \$ 6,432 | \$ 9,021 |
| Accounts payable | 178,350 | 261,507 |
| Accounts payable to McDermott International, Inc. (Note 5) | 112,053 | 60,831 |
| Accrued employee benefits | 198,195 | 174,538 |
| Accrued liabilities—other | 74,700 | 116,203 |
| Advance billings on contracts | 537,448 | 559,150 |
| Accrued warranty expense | 115,055 | 119,722 |
| Income taxes payable | 12,943 | 11,902 |
| Total Current Liabilities | 1,235,176 | 1,312,874 |
| Long-Term Debt (Note 6) | 4,222 | 6,109 |
| Accumulated Postretirement Benefit Obligation (Note 7) | 105,484 | 107,468 |
| Environmental Liabilities (Note 9) | 47,795 | 32,944 |
| Pension Liability (Note 7) | 699,117 | 665,871 |
| Notes Payable to Affiliate (Note 5) | 320,568 | 318,200 |
| Other Liabilities | 70,791 | 132,923 |
| Commitments and Contingencies (Note 9) | | |
| Total Parent Equity (Deficit) | 120,706 | (69,548) |
| TOTAL | \$ 2,603,859 | \$ 2,506,841 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. COMBINED STATEMENTS OF INCOME

| | Ye | 1, | |
|---|--------------|----------------|--------------|
| | 2009 | 2008 | 2007 |
| | | (In thousands) | |
| Revenues | \$ 2,854,632 | \$ 3,398,574 | \$ 3,199,944 |
| Costs and Expenses: | | | |
| Cost of operations | 2,235,377 | 2,677,935 | 2,620,013 |
| Selling, general and administrative expenses | 403,559 | 337,551 | 289,551 |
| Losses (gains) on asset disposals and impairments—net | 1,226 | (9,603) | (1,606) |
| Total Costs and Expenses | 2,640,162 | 3,005,883 | 2,907,958 |
| Equity in Income of Investees | 55,094 | 51,792 | 45,647 |
| Operating Income | 269,564 | 444,483 | 337,633 |
| Other Income (Expense): | | | |
| Interest income | 3,439 | 15,286 | 25,220 |
| Interest expense | (24,590) | (22,740) | (38,094) |
| Other expense—net | (16,112) | (4,290) | (10,411) |
| Total Other Expense | (37,263) | (11,744) | (23,285) |
| Income before Provision for Income Taxes | 232,301 | 432,739 | 314,348 |
| Provision for Income Taxes (Note 4) | 84,381 | 108,885 | 99,018 |
| Net Income | 147,920 | 323,854 | 215,330 |
| Less: Net Income attributable to Noncontrolling Interest | (156) | (88) | (80) |
| Net Income Attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | \$ 147,764 | \$ 323,766 | \$ 215,250 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. COMBINED STATEMENTS OF COMPREHENSIVE INCOME (LOSS)

| | Year Ended December 31, | | |
|--|-------------------------|--------------------|-----------|
| | 2009 | 2008 | 2007 |
| | | (In thousands) | |
| Net Income | \$147,920 | \$ 323,854 | \$215,330 |
| Other Comprehensive Income (Loss): | | | |
| Currency translation adjustments: | | | |
| Foreign currency translation adjustments | 22,103 | (24,717) | 13,471 |
| Unrealized gains on derivative financial instruments: | | | |
| Unrealized gains (losses) on derivative financial instruments | 8,134 | (14,702) | 2,872 |
| Reclassification adjustment for (gains) losses included in net income | (1,551) | 752 | (2,243) |
| Unrecognized gains on benefit obligations: | | | |
| Unrecognized gains (losses) arising during the period | (36,872) | (325,846) | 29,627 |
| Amortization of losses included in net income | 54,962 | 21,217 | 31,798 |
| Amortization of losses included in retained earnings | _ | _ | 704 |
| Unrealized gains on investments: | | | |
| Unrealized gains (losses) arising during the period | (339) | 82 | 24 |
| Reclassification adjustment for gains (losses) included in net income | 30 | (147) | (23) |
| Other Comprehensive Income (Loss) | 46,467 | (343,361) | 76,230 |
| Total Comprehensive Income (Loss) | 194,387 | (19,507) | 291,560 |
| Comprehensive Income attributable to Noncontrolling Interest | (199) | (80) | (80) |
| Comprehensive Income (Loss) Attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | \$194,188 | <u>\$ (19,587)</u> | \$291,480 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. COMBINED STATEMENTS OF PARENT EQUITY (DEFICIT) (In thousands)

| | Accumulated Other Comprehensive (Loss) | Parent Equity | Non- Controlling Interest | Total Parent Equity (Deficit) |
|--|---|------------------|---------------------------------|----------------------------------|
| Balance December 31, 2006 | \$ (325,875) | \$ 49,752 | \$ 316 | \$ (275,807) |
| Net income | _ | 215,250 | 80 | 215,330 |
| Adoption of FASB Topic 740 (Note 5) | _ | (13,015) | _ | (13,015) |
| Adoption of FASB Topic 715 (Note 7) | 704 | (1,687) | _ | (983) |
| Adoption of FASB Topic 360 | _ | 96 | _ | 96 |
| Amortization of benefit plan costs | 31,798 | _ | _ | 31,798 |
| Unrealized gain on benefit obligations | 29,627 | _ | _ | 29,627 |
| Unrealized loss on investments | 1 | _ | _ | 1 |
| Foreign currency translation adjustments | 13,471 | _ | _ | 13,471 |
| Unrealized gain on derivatives | 629 | _ | _ | 629 |
| Stock-based compensation charges | _ | (32,471) | _ | (32,471) |
| Distributions to noncontrolling interests | | | (65) | (65) |
| Balance December 31, 2007 | \$ (249,645) | \$217,925 | \$ 331 | (31,389) |
| Net income | | 323,766 | 88 | 323,854 |
| Amortization of benefit plan costs | 21,217 | _ | _ | 21,217 |
| Unrecognized losses on benefit obligations | (325,846) | _ | _ | (325,846) |
| Unrealized gain on investments | (65) | _ | _ | (65) |
| Foreign currency translation adjustments | (24,709) | _ | (8) | (24,717) |
| Unrealized loss on derivatives | (13,950) | _ | _ | (13,950) |
| Stock-based compensation charges | _ | (18,601) | _ | (18,601) |
| Distributions to noncontrolling interests | | | (51) | (51) |
| Balance December 31, 2008 | \$ (592,998) | \$523,090 | \$ 360 | (69,548) |
| Net income | | 147,764 | 156 | 147,920 |
| Amortization of benefit plan costs | 54,962 | _ | _ | 54,962 |
| Unrecognized losses on benefit obligations | (36,872) | _ | _ | (36,872) |
| Unrealized gain on investments | (309) | _ | _ | (309) |
| Foreign currency translation adjustments | 22,060 | _ | 43 | 22,103 |
| Unrealized loss on derivatives | 6,583 | _ | _ | 6,583 |
| Stock-based compensation charges | _ | (4,077) | _ | (4,077) |
| Distributions to noncontrolling interests | | | (56) | (56) |
| Balance December 31, 2009 | \$ (546,574) | \$666,777 | \$ 503 | \$ 120,706 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. COMBINED STATEMENTS OF CASH FLOWS

| | | Year Ended December 31, | | |
|--|---------------------|-------------------------|--------------------|--|
| | 2009 | (In thousands) | 2007 | |
| CASH FLOWS FROM OPERATING ACTIVITIES: | | (in thousands) | | |
| Net income | \$147,920 | \$ 323.854 | \$ 215,330 | |
| Non-cash items included in net income: | Ψ117,520 | Ψ 323,031 | Ψ 213,330 | |
| Depreciation and amortization | 72,712 | 45,985 | 41,672 | |
| Income of investees, net of dividends | (4,450) | (2,117) | (3,142) | |
| Losses (gains) on asset disposals and impairments—net | 1,226 | (9,603) | (1,606) | |
| Provision for deferred taxes | 29,535 | 23,730 | 11,881 | |
| Excess tax benefits from FAS 123(R) stock-based compensation | 1,813 | (13,381) | (22,228) | |
| Amortization of pension and postretirement costs | 86,637 | 33,735 | 47,370 | |
| Other, net | (6,586) | (527) | (3,524) | |
| Changes in assets and liabilities, net of effects from acquisition and divestitures: | | | | |
| Accounts receivable | 71,316 | (12,901) | 67,249 | |
| Accounts payable and other | (40,555) | (4,076) | (15,934) | |
| Income tax receivable | 11,619 | 43,024 | 254,329 | |
| Inventories Not contracts in progress and advance billings | 29,262 | (21,760) | (14,315) | |
| Net contracts in progress and advance billings Income taxes | (63,879) 10,740 | (234,648) 25,808 | 118,460 (6,894) | |
| Accrued and other current liabilities | (52,556) | 22,503 | (14,763) | |
| Pension liability, accumulated postretirement benefit obligation and accrued employee benefits | (32,330) | (154,743) | (70,844) | |
| Other, net | (41,553) | (4,795) | 77 | |
| NET CASH PROVIDED BY OPERATING ACTIVITIES | 252,823 | 60,088 | 603,118 | |
| | 232,623 | 00,088 | 003,118 | |
| CASH FLOWS FROM INVESTING ACTIVITIES: | (0.100) | (2,000) | 2 111 | |
| (Increase) decrease in restricted cash and cash equivalents Purchases of property, plant and equipment | (8,189) (93,725) | (2,000) (63,014) | 3,111 (60,709) | |
| Acquisition of businesses, net of cash acquired | (8,497) | (190,878) | (71,499) | |
| (Increase) decrease in available-for-sale securities | 45,230 | (59,200) | (26,406) | |
| (morease) decrease in available for sale securities | 13,230 | (37,200) | (20,100) | |
| Increase in note receivable from affiliate | _ | _ | (42,378) | |
| Proceeds from asset disposals | 245 | 9,978 | 3,033 | |
| Investments in affiliates | (2,700) | | | |
| Other | 1 | 551 | (35) | |
| NET CASH USED IN INVESTING ACTIVITIES | (67,635) | (304,563) | (194,883) | |
| CASH FLOWS FROM FINANCING ACTIVITIES: | | | | |
| Payment of short-term borrowings and long-term debt | (6,155) | (4,768) | (254,746) | |
| Increase (decrease) in note payable to affiliate | 2,368 | 49,950 | (86,800) | |
| Debt issuance costs | _ | _ | (2,123) | |
| Increase in short-term borrowing | | 1,460 | | |
| Excess tax benefits from FAS 123(R) stock-based compensation | (1,813) | 13,381 | 22,228 | |
| Other | | (1) | | |
| NET CASH PROVIDED BY (USED IN) FINANCING ACTIVITIES | (5,600) | 60,022 | (321,441) | |
| EFFECTS OF EXCHANGE RATE CHANGES ON CASH | 10,234 | (9,867) | 5,316 | |
| NET INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS | 189,822 | (194,320) | 92,110 | |
| CASH AND CASH EQUIVALENTS AT BEGINNING OF PERIOD | 279,646 | 473,966 | 381,856 | |
| CASH AND CASH EQUIVALENTS AT END OF PERIOD | \$469,468 | \$ 279,646 | \$ 473,966 | |
| | ψτο, του | Ψ 277,040 | Ψ 175,200 | |
| SUPPLEMENTAL DISCLOSURES OF CASH FLOW INFORMATION: Cash paid (received) during the period for: | | | | |
| Interest (net of amount capitalized) | \$ 24.728 | \$ 27,520 | \$ 43.973 | |
| Income taxes (net of refunds) | \$ (29,663) | \$ 12,256 | \$ 43,973 | |
| medic dates (net of felulus) | φ (27,003) | Ψ 12,230 | φ(272,020) | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS

NOTE 1—SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The Company, Basis of Presentation and Principles of Combination

The Babcock & Wilcox Operations of McDermott International, Inc. (the "Company") represent a combined reporting entity comprised of the assets and liabilities used in managing and operating the Power Generation Systems and Government Operations segments of McDermott International, Inc. ("MII") in addition to two captive insurance companies which will be contributed to The Babcock & Wilcox Company in conjunction with the planned spin-off. On December 7, 2009, MII announced a plan to separate MII into two independent, publicly traded companies that will be effected through a spin-off transaction that is intended to be tax-free to MII.

The accompanying combined financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America. All intercompany balances and transactions have been eliminated in the combined financial statements. We use the equity method to account for investments in entities that we do not control, but over which we have significant influence. We generally refer to these entities as "joint ventures." Income from investees net of dividends received reported on our combined statement of cash flows is different from equity in income of investees reported on our combined statement of income due to the timing of cash dividends received from our equity method investees. We record our share of proportionate income from our investees on a monthly basis based on the net income of our investees. Cash dividends are declared by these investees periodically and will usually not match our equity in income from investees in any given period. We present the notes to our combined financial statements on the basis of continuing operations, unless otherwise stated.

The combined results of operations, financial position and cash flows reflected in the accompanying combined financial statements may not be indicative of the future performance of the combined businesses and do not necessarily reflect what its combined results of operations, financial position and cash flows would have been had the combined businesses operated as an independent public company during the periods presented, including changes in its operations and capitalization as a result of the separation and distribution from MII.

Certain corporate and general and administrative expenses, including those related to executive management, tax, accounting, legal and treasury services, and certain employee benefits, have been allocated based on a level of effort calculation. Level of effort is a methodology requiring employees to estimate time spent for each business segment on the various functions described above. We prepare budgets for the various departments and allocate costs from corporate to the appropriate segment based on the time factor calculated and departmental costs. Management believes such allocations are reasonable. However, the associated expenses reflected in the accompanying combined statements of income may not be indicative of the actual expenses that would have been incurred had the combined businesses been operating as an independent public company for the periods presented. Following the separation and distribution from MII, B&W will perform these functions using internal resources or purchased services, certain of which may be provided by MII during a transitional period pursuant to a transition services agreement. Refer to Note 5—Related Party Transactions for a detailed description of transactions with other affiliates of MII.

In presenting the combined financial statements, management makes estimates and assumptions that affect the amounts reported and related disclosures. Estimates, by their nature, are based on judgment and available information. Accordingly, actual results could differ from those estimates.

As used in these footnote disclosures, the following terms have the meanings set forth below:

- MII (or McDermott) means McDermott International, Inc., a Panamanian corporation;
- BHI means Babcock & Wilcox Holdings, Inc., a Delaware corporation;

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

- MI means McDermott Incorporated, a Delaware corporation and a subsidiary of BHI, and its consolidated subsidiaries;
- · BWICO means Babcock & Wilcox Investment Company, a Delaware corporation and a subsidiary of MI, and its consolidated subsidiaries;
- · BWXT means BWX Technologies, Inc., a Delaware corporation and a subsidiary of BWICO, and its consolidated subsidiaries;
- B&W PGG means Babcock & Wilcox Power Generation Group, Inc., a Delaware corporation and a subsidiary of BWICO, and its consolidated subsidiaries; and
- · JRM means J. Ray McDermott, S.A., a Panamanian corporation and a subsidiary of MII, and its consolidated subsidiaries.

Unless the context otherwise indicates, "we," "us" and "our" mean The Babcock & Wilcox Operations of McDermott International, Inc.

We operate in two business segments: Power Generation Systems; and Government Operations.

Our Power Generation Systems segment includes the business and operations of Babcock & Wilcox Power Generation Group, Inc. ("B&W PGG") and Babcock & Wilcox Nuclear Energy, Inc. and their respective subsidiaries. Through this segment, we supply boilers fired with fossil fuels, such as coal, oil and natural gas, or renewable fuels such as biomass, municipal solid waste and concentrated solar energy. In addition, we supply commercial nuclear steam generators and components, environmental equipment and components, and related services to customers in different regions around the world. We design, engineer, manufacture, supply, construct and service large utility and industrial power generation systems, including boilers used to generate steam in electric power plants, pulp and paper making, chemical and process applications and other industrial uses.

Our Government Operations segment includes the business and operations of Babcock & Wilcox Nuclear Operations Group, Inc., Babcock & Wilcox Technical Services Group, Inc. and their respective subsidiaries. Through this segment, we manufacture nuclear components and provide various services to the U.S. Government, including uranium processing, environmental site restoration services, and management and operating services for various U.S. Government-owned facilities. These services are provided to the DOE, including the NNSA, the Office of Nuclear Energy, the Office of Science and the Office of Environmental Management.

Use of Estimates

We use estimates and assumptions to prepare our financial statements in conformity with GAAP. These estimates and assumptions affect the amounts we report in our financial statements and accompanying notes. Our actual results could differ from these estimates. Variances could result in a material effect on our financial condition and results of operations in future periods.

Investments

Our investments, primarily government obligations and other highly liquid money market instruments, are classified as available-for-sale and are carried at fair value, with the unrealized gains and losses, net of tax, reported as a component of accumulated other comprehensive loss. We classify investments available for current operations in the balance sheet as current assets, while we classify investments held for long-term purposes as noncurrent assets. We adjust the amortized cost of debt securities for amortization of premiums and accretion of discounts to maturity. That amortization is included in interest income. We include realized gains and losses on

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

our investments in other income (expense). The cost of securities sold is based on the specific identification method. We include interest on securities in interest income.

Foreign Currency Translation

We translate assets and liabilities of our foreign operations, other than operations in highly inflationary economies, into U.S. dollars at current exchange rates, and we translate income statement items at average exchange rates for the periods presented. We record adjustments resulting from the translation of foreign currency financial statements as a component of other comprehensive loss. We report foreign currency transaction gains and losses in income. We have included in other income (expense) transaction losses of \$(5.4) million, \$(0.5) million and \$(1.3) million for the years ended December 31, 2009, 2008 and 2007, respectively.

Contracts and Revenue Recognition

We generally recognize contract revenues and related costs on a percentage-of-completion method for individual contracts or combinations of contracts based on work performed, man hours, or a cost-to-cost method, as applicable to the product or activity involved. Under this method, we recognize estimated contract revenue and resulting income based on costs incurred to date as a percentage of total estimated costs. Certain costs may be excluded from the cost-to-cost method of measuring progress, such as significant costs for materials and major third-party subcontractors, if it appears that such exclusion would result in a more meaningful measurement of actual contract progress and resulting periodic allocation of income. We include revenues and related costs so recorded, plus accumulated contract costs that exceed amounts invoiced to customers under the terms of the contracts, in contracts in progress. We include in advance billings on contracts billings that exceed accumulated contract costs and revenues and costs recognized under the percentage-of-completion method. Most long-term contracts contain provisions for progress payments. We expect to invoice customers for all unbilled revenues. We review contract price and cost estimates periodically as the work progresses and reflect adjustments proportionate to the percentage-of-completion in income in the period when those estimates are revised. For all contracts, if a current estimate of total contract costs indicates a loss on a contract, the projected loss is recognized in full when determined.

For contracts as to which we are unable to estimate the final profitability except to assure that no loss will ultimately be incurred, we recognize equal amounts of revenue and cost until the final results can be estimated more precisely. For these deferred profit recognition contracts, we recognize revenue and cost equally and only recognize gross margin when probable and reasonably estimable, which we generally determine to be when the contract is approximately 70% complete. We treat long-term construction contracts that contain such a level of risk and uncertainty that estimation of the final outcome is impractical, except to assure that no loss will be incurred, as deferred profit recognition contracts.

Our policy is to account for fixed-price contracts under the completed-contract method if we believe that we are unable to reasonably forecast cost to complete at start-up. Under the completed-contract method, income is recognized only when a contract is completed or substantially complete.

Variations from estimated contract performance could result in material adjustments to operating results for any fiscal quarter or year. We include claims for extra work or changes in scope of work to the extent of costs incurred in contract revenues when we believe collection is probable.

We are usually entitled to financial settlements relative to the individual circumstances of deferrals or cancellations of long-term contracts. We do not recognize those settlements or claims for additional compensation until we reach final settlements with our customers. In June 2006, B&W PGG was awarded separate contracts to supply eight supercritical, coal-fired boilers and selective catalytic reduction ("SCR") systems as part of TXU

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Corp.'s ("TXU") solid-fuel power generation program in Texas. TXU subsequently suspended activity on five of the eight boilers and SCR systems. The five suspended contracts were formally cancelled by execution of a termination and settlement agreement dated April 13, 2007. During the year ended December 31, 2007, B&W PGG received cash payments totaling \$124 million from TXU, which completed the \$243 million termination settlement obligation from TXU. See Note 6 regarding the performance guarantee provided by MII.

The following amounts represent retainages on contracts:

| | December 31, | | | |
|---|--------------------|----|---------|--|
| | 2009 | | 2008 | |
| | (In thousands) | | | |
| Retainages expected to be collected within one year | \$ 94,561 | \$ | 96,686 | |
| Retainages expected to be collected after one year | 16,919 | | 31,007 | |
| Total Retainages | \$ 111,480 | \$ | 127,693 | |

We have included in accounts receivable—trade retainages expected to be collected in 2010. Retainages expected to be collected after one year are included in other assets. Of the long-term retainages at December 31, 2009, we anticipate collecting \$15.5 million in 2011, \$1.2 million in 2012 and \$0.2 million in 2013.

Comprehensive Loss

The components of accumulated other comprehensive loss included in parent equity (deficit) is as follows:

| | | December 31, | | | |
|---|-------|----------------|-------|---------|--|
| | 2 | 2009 | | 2008 | |
| | | (In thousands) | | | |
| Net unrealized loss on investments | \$ | (375) | \$ | (66) | |
| Currency translation adjustments | 3 | 31,728 | | 9,668 | |
| Net unrealized loss on derivative financial instruments | (| (5,758) | (| 12,341) | |
| Unrecognized losses on benefit obligations | (57 | 72,169) | (5) | 90,259) | |
| Accumulated Other Comprehensive Loss | \$(54 | 16,574) | \$(5) | 92,998) | |

Warranty Expense

We accrue estimated expense to satisfy contractual warranty requirements when we recognize the associated revenue on the related contracts. In addition, we make specific provisions where we expect the actual warranty costs to significantly exceed the accrued estimates. Such specific provisions could have a material effect on our combined financial condition, results of operations and cash flows.

The following summarizes the changes in the carrying amount of accrued warranty:

| | Ŋ | Year Ended December 31, | | |
|--------------------------------|-----------|-------------------------|-----------|--|
| | 2009 | 2008 | 2007 | |
| | | (In thousands) | | |
| Balance at beginning of period | \$119,722 | \$100,650 | \$ 72,261 | |
| Additions and adjustments | 13,475 | 27,031 | 40,472 | |
| Charges | (18,142) | (7,959) | (12,083) | |
| Balance at end of period | \$115,055 | \$119,722 | \$100,650 | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Asset Retirement Obligations and Environmental Clean-up Costs

We accrue for future decommissioning of our nuclear facilities that will permit the release of these facilities to unrestricted use at the end of each facility's life, which is a requirement of our licenses from the U.S. Nuclear Regulatory Commission (the "NRC"). In accordance with the FASB Topic Asset Retirement and Environmental Obligations, we record the fair value of a liability for an asset retirement obligation in the period in which it is incurred. When we initially record such a liability, we capitalize a cost by increasing the carrying amount of the related long-lived asset. Over time, the liability, which is included in environmental liabilities on our combined balance sheets, is accreted to its present value each period, and the capitalized cost is depreciated over the useful life of the related asset. Upon settlement of a liability, we will settle the obligation for its recorded amount or incur a gain or loss. This topic applies to environmental liabilities associated with assets that we currently operate and are obligated to remove from service. For environmental liabilities associated with assets that we no longer operate, we have accrued amounts based on the estimated costs of clean-up activities for which we are responsible, net of any cost-sharing arrangements. We adjust the estimated costs as further information develops or circumstances change. An exception to this accounting treatment relates to the work we perform at facilities for which the U.S. Government is obligated to pay all of the decommissioning costs.

Substantially all of our asset retirement obligations relate to the remediation of our nuclear analytical laboratory and the Nuclear Fuel Services, Inc. facilities. The following table reflects our asset retirement obligations:

| | Ye | Year Ended December 31, | |
|---|-----------|-------------------------|----------|
| | 2009 | (In thousands) | 2007 |
| Balance at beginning of period | \$ 25,647 | \$ 9,328 | \$ 8,395 |
| Acquisition of Nuclear Fuel Services, Inc. (Note 2) | 1,627 | 15,281 | |
| Additions | 300 | _ | _ |
| Accretion | 1,917 | 1,038 | 933 |
| Balance at end of period | \$ 29,491 | \$ 25,647 | \$ 9,328 |

Research and Development

Research and development activities are related to development and improvement of new and existing products and equipment and conceptual as well as engineering evaluation for translation into practical applications. We charge to cost of operations the costs of research and development unrelated to specific contracts as incurred. Research and development activities totaled \$78.3 million, \$56.2 million and \$51.3 million in the years ended December 31, 2009, 2008 and 2007, respectively, which include \$25.1 million, \$17.7 million and \$16.5 million, respectively, related to amounts paid for by our customers. Substantially all of these costs are in our Power Generation Systems segment and include costs related to the development of carbon capture and sequestration technologies and our modular and scalable reactor, B&W mPowerTM. The net expenses recognized in the years ended December 31, 2009, 2008 and 2007 totaled approximately \$53.2 million, \$38.5 million and \$34.8 million, respectively.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Inventories

We carry our inventories at the lower of cost or market. We determine cost principally on the first-in, first out basis, except for certain materials inventories of our Power Generation Systems segment, for which we use the last-in, first-out ("LIFO") method. We determined the cost of approximately 17% of our total inventories using the LIFO method at December 31, 2009 and 2008, respectively, and our total LIFO reserve at December 31, 2009 and 2008 was approximately \$6.6 million and \$7.0 million, respectively. Inventories are summarized below:

| | Decem | iber 31, |
|----------------------------|-----------|------------|
| | 2009 | 2008 |
| | (In tho | usands) |
| Raw Materials and Supplies | \$ 71,207 | \$ 92,770 |
| Work in Progress | 6,381 | 12,157 |
| Finished Goods | 21,056 | 20,633 |
| Total Inventories | \$ 98,644 | \$ 125,560 |

Property, Plant and Equipment

We carry our property, plant and equipment at depreciated cost, less any impairment provisions.

We depreciate our property, plant and equipment using the straight-line method over estimated economic useful lives of eight to 40 years for buildings and two to 28 years for machinery and equipment. Our depreciation expense was \$62.6 million, \$41.3 million and \$38.8 million for the years ended December 31, 2009, 2008 and 2007, respectively.

We expense costs of maintenance, repairs and renewals that do not materially prolong the useful life of an asset as we incur them.

Goodwill

The following summarizes the changes in the carrying amount of goodwill (in thousands):

| | r Generation Systems | Government Operations | Total |
|--|-----------------------------|--------------------------|-----------|
| Balance at December 31, 2007 | \$ 77,079 | \$ 51,931 | \$129,010 |
| Acquisition of Nuclear Fuel Services, Inc. (Note 2) | _ | 123,542 | 123,542 |
| Acquisition of the Intech group of companies (Note 2) | 8,151 | _ | 8,151 |
| Acquisition of Delta Power Services, LLC (Note 2) | 3,683 | _ | 3,683 |
| Currency translation adjustments | (1,234) | | (1,234) |
| Balance at December 31, 2008 | \$ 87,679 | \$ 175,473 | \$263,152 |
| Adjustments related to the acquisition of Nuclear Fuel Services, Inc. (Note 2) | _ | (8,066) | (8,066) |
| B&W de Monterrey asset acquisition (Note 2) | 7,442 | _ | 7,442 |
| Currency translation adjustments | 338 | | 338 |
| Balance at December 31, 2009 | \$ 95,459 | \$ 167,407 | \$262,866 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Other Intangible Assets

We report our other intangible assets in other assets. We amortize those intangible assets which have definite lives to operating expense, using the straight-line method.

Other assets include the following other intangible assets:

| | Ye | Year Ended December 31, | |
|---------------------------------|-----------|-------------------------|----------|
| | 2009 | 2008 | 2007 |
| Amortized intangible assets: | | (In thousands) | |
| Gross cost: | | | |
| Customer relationships | 30,950 | 31,150 | 19,790 |
| Acquired backlog | 18,720 | 17,280 | 9,540 |
| Trade name | 10,680 | 3,820 | 1,770 |
| Unpatented technology | 5,600 | 5,600 | _ |
| Patented technology | 4,440 | 4,060 | _ |
| All other | 8,585 | 10,983 | 7,737 |
| Total | 78,975 | 72,893 | 38,837 |
| Accumulated amortization: | | | |
| Customer relationships | (3,726) | (1,985) | (660) |
| Acquired backlog | (7,746) | (3,407) | (1,363) |
| Trade name | (2,371) | (683) | (236) |
| Unpatented technology | (837) | (277) | _ |
| Patented technology | (888) | _ | _ |
| All other | (5,163) | (4,459) | (3,994) |
| Total | (20,731) | (10,811) | (6,253) |
| Net amortized intangible assets | \$ 58,244 | \$ 62,082 | \$32,584 |
| Unamortized intangible assets: | | | |
| NRC category 1 license | \$ 43,830 | \$ 42,370 | \$ — |
| Trademarks and trade names | 1,305 | 7,395 | 1,305 |
| Total | \$ 45,135 | \$ 49,765 | \$ 1,305 |

The following summarizes the changes in the carrying amount of other intangible assets:

| | Ye | Year Ended December 31, | |
|--------------------------------|-----------|-------------------------|----------|
| | 2009 | 2008 | 2007 |
| | <u></u> - | (In thousands) | |
| Balance at beginning of period | \$111,847 | \$ 33,889 | \$ 5,605 |
| Business acquisitions (Note 2) | _ | 82,630 | 31,100 |
| Amortization expense | (9,975) | (4,672) | (2,816) |
| Other | 1,507 | | |
| Balance at end of period | \$103,379 | \$111,847 | \$33,889 |
| | | | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The estimated amortization expense for the next five fiscal years is as follows (in thousands):

| Year Ending December 31, | Amount |
|--------------------------|----------------|
| 2010 | Amount \$9,906 |
| 2011 | \$9,736 |
| 2012 | \$7,364 |
| 2013 | \$4,951 |
| 2014 | \$3,863 |

Other Non-Current Assets

We have included deferred debt issuance costs in other assets. We amortize deferred debt issuance cost as interest expense over the life of the related debt. Following are the changes in the carrying amount of these assets:

| | Y | Year Ended December 31, | | |
|--------------------------------------|----------|-------------------------|----------|--|
| | 2009 | 2009 2008 | | |
| | | (In thousands) | | |
| Balance at beginning of period | \$ 4,169 | \$ 6,440 | \$10,819 | |
| Additions | _ | _ | 2,123 | |
| Interest expense—debt issuance costs | (2,271) | (2,271) | (6,502) | |
| Balance at end of period | \$ 1,898 | \$ 4,169 | \$ 6,440 | |

Capitalization of Interest Cost

We capitalize interest in accordance with the FASB Topic *Interest*. We incurred total interest of \$27.4 million, \$24.7 million and \$38.6 million in the years ended December 31, 2009, 2008 and 2007, respectively, of which we capitalized \$2.8 million, \$1.9 million and \$0.5 million in the years ended December 31, 2009, 2008 and 2007, respectively.

Cash and Cash Equivalents

Our cash equivalents are highly liquid investments, with maturities of three months or less when we purchase them, which we do not hold as part of our investment portfolio.

We record current cash and cash equivalents as restricted when we are unable to freely use such cash and cash equivalents for our general operating purposes.

At December 31, 2009, we had restricted cash and cash equivalents totaling \$15.3 million, \$7.4 million of which was held in restricted foreign accounts, \$3.5 million of which was held as cash collateral for letters of credit, \$4.0 million of which was held for future decommissioning of facilities, and \$0.4 million of which was held to meet reinsurance reserve requirements of our captive insurance companies. It is possible that a portion of restricted cash at December 31, 2009 will not be released within the next 12 months.

Derivative Financial Instruments

Our worldwide operations give rise to exposure to market risks from changes in foreign exchange rates. We use derivative financial instruments to reduce the impact of changes in foreign exchange rates on our operating results. We use these instruments primarily to hedge our exposure associated with revenues or costs on our long-

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

term contracts that are denominated in currencies other than our operating entities' functional currencies. We do not hold or issue financial instruments for trading or other speculative purposes.

We enter into derivative financial instruments primarily as hedges of certain firm purchase and sale commitments denominated in foreign currencies. We record these contracts at fair value on our combined balance sheets. Depending on the hedge designation at the inception of the contract, the related gains and losses on these contracts are either deferred in parent equity (deficit) (as a component of accumulated other comprehensive loss) until the hedged item is recognized in earnings or offset against the change in fair value of the hedged firm commitment through earnings. Any ineffective portion of a derivative's change in fair value is immediately recognized in earnings. The gain or loss on a derivative financial instrument not designated as a hedging instrument is also immediately recognized in earnings. Gains and losses on derivative financial instruments that require immediate recognition are included as a component of other income (expense)—net in our combined statements of income.

Self-Insurance

We have certain wholly owned insurance subsidiaries that provide employer's liability, general and automotive liability and workers' compensation insurance and, from time to time, builder's risk insurance (within certain limits) to our companies. We may also, in the future, have these insurance subsidiaries accept other risks that we cannot or do not wish to transfer to outside insurance companies. Included in other liabilities on our combined balance sheets are reserves for self-insurance totaling \$56.8 million and \$60.6 million at December 31, 2009 and 2008, respectively.

Loss Contingencies

We estimate liabilities for loss contingencies when it is probable that a liability has been incurred and the amount of loss is reasonably estimable. We provide disclosure when there is a reasonable possibility that the ultimate loss will exceed the recorded provision or if such loss is not reasonably estimable. We are currently involved in some significant litigation, as discussed in Note 9. We have accrued our estimates of the probable losses associated with these matters. However, our losses are typically resolved over long periods of time and are often difficult to estimate due to various factors, including the possibility of multiple actions by third parties. Therefore, it is possible future earnings could be affected by changes in our estimates related to these matters.

Stock-Based Compensation

We expense stock-based compensation in accordance with FASB Topic Compensation—Stock Compensation. Under this topic, the fair value of equity-classified awards, such as restricted stock, performance shares and stock options, is determined on the date of grant and is not remeasured. Grant date fair values for restricted stock and performance shares are determined using the closing price of MII's common stock on the date of grant. Grant date fair values for stock options are determined using a Black-Scholes option-pricing model ("Black-Scholes"). The determination of the fair value of a share-based payment award on the date of grant using an option-pricing model requires the input of highly subjective assumptions, such as the expected life of the award and stock price volatility. For liability-classified awards, such as cash-settled deferred stock units and performance units, fair values are determined at grant date using the closing price of MII's common stock and are remeasured at the end of each reporting period through the date of settlement.

Under the provisions of this topic, we recognize expense based on the grant date fair value, net of an estimated forfeiture rate, for all share-based awards granted on a straight-line basis over the requisite service

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

periods of the awards, which is generally equivalent to the vesting term. This topic requires compensation expense to be recognized, net of an estimate for forfeitures, such that compensation expense is recorded only for those awards expected to vest. We review the estimate for forfeitures periodically and record any adjustments deemed necessary for each reporting period. If our actual forfeiture rate is materially different from our estimate, the stock-based compensation expense could be significantly different from what we have recorded in the current period.

Additionally, this topic amended the FASB Topic *Statement of Cash Flows* to require excess tax benefits to be reported as a financing cash flow, rather than as a reduction of taxes paid. These excess tax benefits result from tax deductions in excess of the cumulative compensation expense recognized for options exercised and other equity-classified awards.

See Note 8 for a further discussion of stock-based compensation.

Recently Adopted Accounting Standards

In June 2009, the FASB issued Statement of Financial Accounting Standards ("SFAS") No. 168—The FASB Accounting Standards Codification and the Hierarchy of Generally Accepted Accounting Principles a replacement of FASB Statement No. 162. SFAS No. 168 made the FASB Accounting Standards Codification (the "Codification") the single source of U.S. GAAP used by nongovernmental entities in preparation of financial statements, except for rules and interpretive releases of the Securities and Exchange Commission (the "SEC") under authority of federal securities laws, which are the sources of authoritative accounting guidance for SEC registrants. The Codification is meant to simplify user access to all authoritative accounting guidance by reorganizing U.S. GAAP pronouncements into roughly 90 accounting topics within a consistent structure; its purpose is not to create new accounting and reporting guidance. The Codification supersedes all existing non-SEC accounting and reporting standards and was effective for us beginning July 1, 2009. Following SFAS No. 168, the FASB will not issue new standards in the form of Statements, FASB Staff Positions, or Emerging Issues Task Force Abstracts; instead, it will issue Accounting Standards Updates. The FASB will not consider Accounting Standards Updates as authoritative in their own right; these updates will serve only to update the Codification, provide background information about the guidance, and provide the bases for conclusions on the change(s) in the Codification. In the discussion that follows, references in "italics" relate to Codification Topics and Subtopics, and their descriptive titles, as appropriate.

In May 2009, the FASB issued the Topic Subsequent Events. This section of the Codification incorporates specific accounting and disclosure requirements for subsequent events into U.S. generally accepted accounting principles. The adoption of these provisions did not have a material impact on our combined financial statements.

In April 2009, the FASB issued a revision to the Topic *Business Combinations*. This revision amends and clarifies the Topic *Business Combinations* to address subsequent measurement and accounting for, and disclosure of, assets and liabilities arising from contingencies in a business combination. On January 1, 2009, we adopted the provisions of this update. The adoption of these provisions did not have a material impact on our combined financial statements.

In April 2009, the FASB issued a revision to the Topic *Financial Instruments*. This revision requires disclosures about fair value of financial instruments in notes to interim financial statements as well as annual financial statements. The notes to our financial statements reflect this revision.

In April 2008, the FASB issued a revision to the Topic Intangibles—Goodwill and Other. This revision requires companies estimating the useful life of a recognized intangible asset to consider their historical

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

experience in renewing or extending similar arrangements or, in the absence of historical experience, to consider assumptions that market participants would use about renewals or extensions as adjusted for the entity-specific factors in this topic. On January 1, 2009, we adopted this revision. The adoption of these provisions did not have a material impact on our combined financial statements.

In March 2008, the FASB issued a revision to the Topic *Derivatives and Hedging*. This revision requires enhanced disclosures about derivative and hedging activities and is effective for financial statements issued for fiscal years and interim periods beginning after November 15, 2008. On January 1, 2009, we adopted this revision for our disclosures about derivative instruments and hedging activities. The notes to our combined financial statements reflect this revision.

In December 2007, the FASB issued a revision to the Topic *Consolidation*. This revision establishes accounting and reporting standards pertaining to ownership interests in subsidiaries held by parties other than the parent, the amount of net income attributable to the parent and to the noncontrolling interest, changes in a parent's ownership interest and the valuation of any retained noncontrolling equity investment when a subsidiary is deconsolidated. It also establishes disclosure requirements that clearly identify and distinguish between the interests of the parent and the interests of the noncontrolling owners. On January 1, 2009, we adopted this revision. We have presented noncontrolling interest as a separate component of parent equity as of December 31, 2009 and 2008.

In December 2007, the FASB issued a revision to the Topic *Business Combinations*. This revision broadens the guidance of this Topic, extending its applicability to all transactions and events in which one entity obtains control over one or more other businesses. It broadens the fair value measurements and recognition of assets acquired, liabilities assumed and interests transferred as a result of business combinations. It also provides disclosure requirements to enable users of the financial statements to evaluate the nature and financial effects of business combinations. On January 1, 2009, we adopted the provisions of this revision. The adoption of these provisions did not have a material impact on our combined financial statements.

New Accounting Standards

In June 2009, the FASB issued a revision to the Topic *Consolidation*. This revision expands the scope of this topic and amends guidance for assessing and analyzing variable interest entities. This revision will be effective for fiscal years beginning after November 15, 2009. We do not expect this revision to have a material impact on our combined financial statements.

NOTE 2—BUSINESS ACQUISITIONS

Instrumentacion y Mantenimiento de Calderas, S.A.

In September 2009, our Power Generation Systems segment acquired certain assets of Instrumentacion y Mantenimiento de Calderas, S.A. In connection with this acquisition, we recorded goodwill of \$7.4 million and property, plant and equipment of \$4.2 million. We accounted for this acquisition as a business combination.

Nuclear Fuel Services, Inc.

On December 31, 2008, our Government Operations segment completed its acquisition of Nuclear Fuel Services, Inc. ("NFS") for \$157.1 million, net of cash acquired. NFS is a provider of specialty nuclear fuels and related services and is a leader in the conversion of Cold War-era government stockpiles of highly enriched

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

uranium into commercial-grade nuclear reactor fuel. NFS also owns and operates a nuclear fuel fabrication facility licensed by the NRC in Erwin, Tennessee and has approximately 790 employees. In connection with the acquisition of NFS, we initially recorded goodwill of \$123.5 million, none of which will be deductible for tax purposes. We also initially recorded other intangible assets of \$63.4 million. Those intangible assets now consist of the following (dollar amounts in thousands):

| | | Amortization |
|------------------------|-----------|--------------|
| | Amount | Period |
| NRC category 1 license | \$ 43,830 | Indefinite |
| Backlog | \$ 9,180 | 4 years |
| Tradename | \$ 6,860 | 6 years |
| Patented technology | \$ 4,440 | 5 years |
| Non-compete agreement | \$ 610 | 5 years |

During 2009, we finalized our purchase price allocation for the Nuclear Fuel Services, Inc. acquisition, which we completed on December 31, 2008. The purchase price adjustments included a reduction in goodwill of \$8.1 million, an increase in property, plant and equipment of \$16.2 million and an increase in environmental reserves of \$13.5 million.

The Intech Group of Companies

On July 15, 2008, our Power Generation Systems segment completed its acquisition of the Intech group of companies ("Intech") for \$20.2 million. Intech consists of Intech, Inc., Ivey-Cooper Services, L.L.C. and Intech International Inc. Intech, Inc. provides nuclear inspection and maintenance services, primarily for the U.S. market. Ivey-Cooper Services, L.L.C. provides non-destructive inspection services to fossil-fueled power plants, as well as chemical, pulp and paper, and heavy fabrication facilities. Intech International Inc. provides non-destructive testing, field engineering and repair and specialized tooling services, primarily for the Canadian nuclear power generation industry. In connection with the acquisition of Intech, we recorded goodwill of \$8.2 million. We also recorded other intangible assets of \$10.0 million. Those intangible assets consist of the following (dollar amounts in thousands):

| | | Amortization |
|------------------------|---------|--------------|
| | Amount | Period |
| Unpatented technology | \$5,600 | 10 years |
| Customer relationships | \$2,600 | 10 years |
| Tradename | \$1,800 | 10 years |

Delta Power Services, LLC

On August 1, 2008, our Power Generation Systems segment completed its acquisition of Delta Power Services, LLC ("DPS") for \$13.5 million. DPS is a provider of operation and maintenance services for the U.S. power generation industry. Headquartered in Houston, Texas, DPS has approximately 190 employees at eight gas, biomass or coal-fired power plants in Virginia, California, Texas, Florida and Michigan. In connection with the acquisition of DPS, we recorded goodwill of \$3.7 million. We also recorded other intangible assets of \$9.3 million. Those intangible assets consist of the following (dollar amounts in thousands):

| | | Amortization |
|------------------------|----------|----------------|
| | Amount | Period |
| Customer relationships | \$ 8,760 | 1.4 - 20 years |
| Tradename | \$ 250 | 25 years |
| Non-compete agreement | \$ 240 | 3 years |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Marine Mechanical Corporation

On May 1, 2007, our Government Operations segment completed its acquisition of Marine Mechanical Corporation ("MMC") for \$71.5 million in cash. We recorded goodwill of \$39.0 million in connection with this acquisition, none of which will be deductible for tax purposes. Headquartered in Euclid, Ohio, MMC designs, manufactures and supplies electro-mechanical equipment used by the U.S. Government. In addition to the goodwill, we recorded identifiable intangible assets of \$31.1 million. Those intangible assets consist of the following (amounts in thousands):

| | | Amortization |
|------------------------|-----------|--------------|
| | Amount | Period |
| Customer relationships | \$ 19,790 | 20 years |
| Backlog | \$ 9,540 | 4.7 years |
| Tradename | \$ 1,770 | 5 years |

NOTE 3—EQUITY METHOD INVESTMENTS

We have investments in entities that we account for using the equity method. The undistributed earnings of our equity method investees were \$44.2 million and \$37.6 million at December 31, 2009 and 2008, respectively.

Summarized below is combined balance sheet and income statement information, based on the most recent financial information, for investments in entities we accounted for using the equity method:

| | Dece | mber 31, |
|--------------------------------------|------------|------------|
| | 2009 | 2008 |
| | (In the | ousands) |
| Current assets | \$ 360,992 | \$ 254,503 |
| Noncurrent assets | 78,559 | 124,357 |
| Total Assets | \$ 439,551 | \$ 378,860 |
| Current liabilities | \$ 234,451 | \$ 158,266 |
| Noncurrent liabilities | 65,223 | 73,855 |
| Owners' equity | 139,877 | 146,739 |
| Total Liabilities and Owners' Equity | \$ 439,551 | \$ 378,860 |

| Year Ended December 31, | | | | | |
|-------------------------|----------------|--|--|---|---|
| | 2009 | | 2008 | | 2007 |
| | <u> </u> | (1 | n thousands) | | |
| \$ | 2,199,032 | \$ | 2,083,809 | \$ | 1,887,132 |
| \$ | 180,205 | \$ | 182,120 | \$ | 156,527 |
| \$ | 131,530 | \$ | 139,197 | \$ | 121,859 |
| | 13,504 | _ | 15,947 | | 15,916 |
| \$ | 118,026 | \$ | 123,250 | \$ | 105,943 |
| | \$ \$ \$ | \$ 2,199,032 \$ 180,205 \$ 131,530 13,504 | 2009 \$ 2,199,032 \$ \$ 180,205 \$ \$ 131,530 \$ 13,504 | 2009 2008 (In thousands) \$ 2,199,032 \$ 2,083,809 \$ 180,205 \$ 182,120 \$ 131,530 \$ 139,197 13,504 15,947 | 2009 2008 (In thousands) \$ 2,199,032 \$ 2,083,809 \$ 180,205 \$ 182,120 \$ 131,530 \$ 139,197 13,504 15,947 |

Revenues of equity method investees include \$1,551.9 million, \$1,584.6 million and \$1,519.9 million of reimbursable costs recorded by limited liability companies in which our Government Operations segment owns an interest at December 31, 2009, 2008 and 2007, respectively. Our investment in equity method investees was \$0.8 million less than our underlying equity in net assets of those investees based on stated ownership

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

percentages at December 31, 2009. These differences were primarily related to the timing of distribution of dividends and various adjustments under GAAP.

The provision for income taxes is based on the tax laws and rates in the countries in which our investees operate. There is no expected relationship between the provision for income taxes and income before provision for income taxes. The taxation regimes vary not only by their nominal rates, but also by the allowability of deductions, credits and other benefits. For some of our U.S. investees, U.S. income taxes are the responsibility of the respective owners.

Reconciliation of net income per combined income statement information to equity in income from investees per our combined statements of income is as follows:

| | Year Ended December 31, | | |
|--|-------------------------|----------------|----------|
| | 2009 | 2008 | 2007 |
| | | (In thousands) | |
| Equity income based on stated ownership percentages | \$56,286 | \$56,420 | \$50,620 |
| All other adjustments due to amortization of basis differences, timing of GAAP adjustments and other adjustments | (1,192) | (4,628) | (4,973) |
| Equity in income from investees net of tax | \$55,094 | \$51,792 | \$45,647 |

Our transactions with affiliates included the following:

| | Ye | Year Ended December 31, | | | |
|--------------------|-----------|-------------------------|-----------|--|--|
| | 2009 | 2008 | 2007 | | |
| | | (In thousands) | | | |
| Sales to | \$ 35,802 | \$ 23,096 | \$ 9,404 | | |
| Purchases from | \$ 12,222 | \$ 39,963 | \$ 42,536 | | |
| Dividends received | \$ 50,644 | \$ 49,676 | \$ 41,844 | | |

Our accounts receivable-trade, net includes receivables from these affiliates of \$4.8 million and \$1.3 million at December 31, 2009 and 2008, respectively. Accounts payable includes payables to these affiliates of \$0.4 million and \$3.3 million, respectfully, at December 31, 2009 and 2008.

NOTE 4—INCOME TAXES

We conduct business globally, and as a result, we or one or more of our subsidiaries file income tax returns in the U.S. federal jurisdiction and in many state and foreign jurisdictions. In the normal course of business, we are subject to examination by taxing authorities throughout the world, including such major jurisdictions as Canada and the United States. With few exceptions, we are no longer subject to non-U.S. tax examinations for years prior to 2006.

The IRS has completed audits of the years 2001 through 2006 for the combined group that includes BWICO, and these years are awaiting review by the Joint Committee on Taxation. We anticipate review by the Joint Committee on Taxation of the group's tax years ended 2001 through 2006 to be resolved within the next 12 months.

State income tax returns are generally subject to examination for a period of three to five years after filing the respective returns. With few exceptions, we do not have any state returns under examination for years prior to 2002.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Effective January 1, 2007, we adopted the provisions of FASB Topic *Income Taxes* regarding the treatment of uncertain tax positions. As a result of this adoption, we recognized a charge of approximately \$13.0 million to our accumulated deficit component of parent equity at January 1, 2007. A reconciliation of unrecognized tax benefits for the years ended December 31, 2009, 2008 and 2007 was as follows (in thousands):

| | Yea | Year Ended December 31, | | | |
|--|-----------------|-------------------------|-----------|--|--|
| | 2009 | 2008 | 2007 | | |
| Balance at beginning of year | \$14,521 | \$ 28,858 | \$ 41,108 | | |
| Increases based on tax positions taken in the current year | 1,322 | 3,688 | 391 | | |
| Increases based on tax positions taken in the prior years | 216 | 353 | 2,922 | | |
| Decreases based on tax positions taken in the prior years | (91) | (4,525) | (13,250) | | |
| Decreases due to settlements with tax authorities | (4,126) | (13,441) | (2,313) | | |
| Decrease due to lapse of applicable statute of limitations | <u> </u> | (412) | | | |
| Balance at end of year | <u>\$11,842</u> | \$ 14,521 | \$ 28,858 | | |
| | | | | | |

Approximately \$10.6 million of the balance of unrecognized tax benefits at December 31, 2009 would reduce our effective tax rate if recognized. The remaining balance relates to positions for which the ultimate deductibility is highly certain but for which there is uncertainty about the timing of such deductibility. Because of the impact of deferred tax accounting, other than interest and penalties, the disallowance of the shorter deductibility period would not affect the annual effective tax rate but would accelerate the payment of cash to the taxing authority to an earlier period.

During 2009, we attempted to settle several years of outstanding audits with the Commonwealth of Virginia by making payments under a special tax amnesty program. The payments approximated the unrecognized tax benefits, resulting in no adjustment to tax expense. We have not yet received formal notification that the audits are closed.

As part of the adoption of FASB Topic *Income Taxes*, we began to recognize interest and penalties related to unrecognized tax benefits in our provision for income taxes. During the year ended December 31, 2009, we recorded additional accruals of \$0.4 million and made payments of \$3.2 million, resulting in a year-end liability of approximately \$3.6 million for the payment of tax-related interest and penalties. At December 31, 2008 and 2007, we recorded liabilities of approximately \$6 million and \$11.6 million, respectively, for the payment of tax-related interest and penalties. The \$5.2 million change during the year ended December 31, 2008 was attributable to the settlement of certain audits and the reassessment of related tax positions.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Deferred income taxes reflect the net tax effects of temporary differences between the financial and tax bases of assets and liabilities. Significant components of deferred tax assets and liabilities as of December 31, 2009 and 2008 were as follows:

| | Decem | ber 31, |
|--|-----------|-----------|
| | 2009 | 2008 |
| | (In tho | usands) |
| Deferred tax assets: | | |
| Pension liability | \$206,757 | \$226,802 |
| Accrued warranty expense | 41,324 | 44,662 |
| Accrued vacation pay | 11,843 | 11,935 |
| Accrued liabilities for self-insurance (including postretirement health care benefits) | 53,121 | 53,862 |
| Accrued liabilities for executive and employee incentive compensation | 35,363 | 43,959 |
| Net operating loss carryforward | 6,273 | 4,619 |
| Environmental and products liabilities | 11,655 | 31,272 |
| Long-term contracts | 19,334 | 18,303 |
| State tax net operating loss carryforward | 44,781 | 40,876 |
| Foreign tax credit carryforward | 16,062 | 19,178 |
| Other | 13,143 | 19,253 |
| Total deferred tax assets | 459,656 | 514,721 |
| Valuation allowance for deferred tax assets | (28,701) | (32,227) |
| Deferred tax assets | 430,955 | 482,494 |
| Deferred tax liabilities: | | |
| Property, plant and equipment | 28,503 | 27,839 |
| Investments in joint ventures and affiliated companies | 3,528 | 3,822 |
| Intangibles | 32,351 | 35,011 |
| Other | 4,651 | 4,786 |
| Total deferred tax liabilities | 69,033 | 71,458 |
| Net deferred tax assets | \$361,922 | \$411,036 |

Income from continuing operations before provision for income taxes was as follows:

| | Yes | Year Ended December 31, | | | |
|---|------------|-------------------------|------------|--|--|
| | 2009 | 2008 | 2007 | | |
| | | (In thousands) | | | |
| U.S. | \$ 164,347 | \$ 351,398 | \$ 256,319 | | |
| Other than U.S. | 67,954 | 81,341 | 58,029 | | |
| Income from continuing operations before provision for income taxes | \$ 232,301 | \$ 432,739 | \$ 314,348 | | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The provision for income taxes consisted of:

| | Y | Year Ended December 31, | | | |
|----------------------------|----------|-------------------------|----------|--|--|
| | 2009 | 2009 2008 | | | |
| | | (In thousands) | | | |
| Current: | | | | | |
| U.S.—federal | \$26,689 | \$ 53,691 | \$64,160 | | |
| U.S.—state and local | 14,749 | 16,063 | 7,304 | | |
| Other than U.S. | 13,408 | 15,401 | 15,673 | | |
| Total current | 54,846 | 85,155 | 87,137 | | |
| Deferred: | | · | | | |
| U.S.—Federal | 32,330 | 49,965 | 12,163 | | |
| U.S.—State and local | (4,489) | (29,815) | 687 | | |
| Other than U.S. | 1,694 | 3,580 | (969) | | |
| Total deferred | 29,535 | 23,730 | 11,881 | | |
| Provision for income taxes | \$84,381 | \$108,885 | \$99,018 | | |

The following is a reconciliation of the U.S. statutory federal tax rate (35%) to the combined effective tax rate:

| | Year Ended December 31, | | | |
|--|-------------------------|-------|-------|--|
| | 2009 | 2008 | 2007 | |
| U.S. federal statutory (benefit) rate | 35.0% | 35.0% | 35.0% | |
| State and local income taxes | 3.4 | 3.8 | 0.8 | |
| Non-U.S. operations | (4.1) | (2.5) | (1.0) | |
| Non-deductible business expense | _ | 0.5 | 1.0 | |
| Valuation allowance for deferred tax assets | (0.3) | (7.0) | 0.1 | |
| Audit settlements | 0.3 | (4.7) | (4.0) | |
| Other | 2.0 | 0.1 | (0.4) | |
| Effective tax rate attributable to continuing operations | 36.3% | 25.2% | 31.5% | |

At December 31, 2009, we had a valuation allowance of \$28.7 million for deferred tax assets, which we expect cannot be realized through carrybacks, future reversals of existing taxable temporary differences and our estimate of future taxable income. We believe that our remaining deferred tax assets are realizable through carrybacks, future reversals of existing taxable temporary differences and our estimate of future taxable income. Any changes to our estimated valuation allowance could be material to our combined financial statements.

We have foreign net operating loss carryforwards of approximately \$13.8 million available to offset future taxable income in foreign jurisdictions. The foreign net operating losses have a valuation allowance of \$1.4 million against the related deferred taxes. We have U.S. federal net operating loss carryforwards of approximately \$6.9 million, which carry a \$1.4 million valuation allowance. These net operating loss carryforwards are scheduled to expire in years 2010 to 2026. We have state net operating losses of approximately \$1 billion available to offset future taxable income in various states. The state net operating loss carryforwards begin to expire in the year 2010. We are carrying a valuation allowance of \$19.5 million against the deferred tax asset related to the state loss carryforwards.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

We would be subject to withholding taxes if we were to distribute earnings from certain of our foreign subsidiaries. For the year ended December 31, 2009, the undistributed earnings of these subsidiaries is \$239.0 million. Unrecognized deferred income tax liabilities, including withholding taxes, of approximately \$15.5 million would be payable upon distribution of these earnings. We have provided \$0.7 million of taxes on earnings we intend to remit. All other earnings are considered permanently reinvested.

NOTE 5—RELATED PARTY TRANSACTIONS

We are a party to transactions with subsidiaries, divisions and controlled joint ventures of MII and other MII subsidiaries in the normal course of operations. These transactions include the following:

| | , | Year Ended Decemb | ber 31, |
|---|------|-------------------|---------|
| | | 2009 | 2008 |
| | | (In thousands) |) |
| Purchases | \$ | 251 \$ | 404 |
| Interest income | \$ | 1,217 \$ | 2,604 |
| Interest expense | \$ 2 | 23,590 \$ | 19,645 |
| Corporate administrative expense allocated to MII and JRM | \$: | 52,142 \$ | 34,967 |

Our accounts payable to McDermott International, Inc. consist of items of a normal recurring nature, including, but not limited to, the items disclosed above. These balances are generally settled within 60 days of the invoice date.

At December 31, 2009, MII had an outstanding performance guarantee for a contract executed by B&W PGG with TXU. The total contract value of this project was approximately \$138.5 million.

Our note receivable from an affiliate is a 2007 loan agreement between Babcock & Wilcox Canada, Ltd., a subsidiary of B&W PGG, as lender, and J. Ray McDermott Canada Holding, Ltd. ("JRMCHL"), as borrower, for up to 60 million Canadian Dollars. The actual amount funded to JRMCHL under this agreement was \$45.2 million. Interest is set at the Canadian Prime Rate and is due quarterly. The loan agreement terminates on June 30, 2011, and the entire amount outstanding under the loan agreement, including accrued interest, is scheduled to become due and payable at that time. Either party may terminate the loan agreement by providing 30 days' written notice to the other party, in which case the entire amount outstanding under the loan agreement, including accrued interest, will become due and payable at the end of that 30-day period. JRMCHL may repay the loan at any time, without penalty. During the year ended December 31, 2007, JRMCHL made a principal payment of \$2.9 million to Babcock & Wilcox Canada, Ltd. No subsequent principal payments have been made. We expect this loan to be settled in connection with the proposed spin-off.

Our note payable to an affiliate relates to a 2006 loan agreement with McDermott Kft., Budapest, Zug branch, the registered branch of McDermott Group Financing Limited Liability Company, whereby we borrowed \$355 million. The loan bears interest at a rate of 7.28% per annum at December 31, 2009 and is payable quarterly. The loan is scheduled to mature on December 21, 2011. Principal is due in full on the maturity date but may be paid in whole or in part in advance without penalty. During the year ended December 31, 2009, we had no borrowings or repayments under this facility. We expect this loan to be settled in connection with the proposed spin-off.

On September 17, 2009, we entered into a loan agreement with MII, whereby we can borrow up to \$5.0 million in connection with the acquisition of Instrumentacion y Mantenimiento de Calderas, S.A. Interest is set at the Eurodollar deposit rate and is due quarterly. The loan agreement is scheduled to terminate on December 31, 2014, and the entire amount outstanding under the loan agreement, including the accrued interest, is scheduled to

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

become due and payable at that time. Either party may terminate the loan agreement by providing 30 days' written notice to the other party, in which case the entire amount outstanding under the loan agreement, including the accrued interest, will become due and payable at the end of that 30-day period. During the year ended December 31, 2009, we had \$2.4 million in borrowings under this facility. We expect this loan to be settled in connection with the proposed spin-off.

NOTE 6—LONG-TERM DEBT AND NOTES PAYABLE

| | Decem | ber 31, |
|--|---------------|-------------|
| | 2009 | 2008 |
| Long town debt consists of | (In thou | usands) |
| Long-term debt consists of: Unsecured Debt: | | |
| | ¢5.016 | ¢11 540 |
| Other notes payable through 2012 Secured Debt: | \$5,916 | \$11,548 |
| | 4.650 | 4 0 4 5 |
| Power Generation Systems—various notes payable | 1,659 | 1,945 |
| Other | 13 | <u> 177</u> |
| | 7,588 | 13,670 |
| Less: Amounts due within one year | 3,366 | 7,561 |
| Long-term debt | 4,222 | \$ 6,109 |
| Notes payable and current maturities of long-term debt consist of: | - | |
| Short-term lines of credit | 3,066 | \$ 1,460 |
| Current maturities of long-term debt | 3,366 | 7,561 |
| Total | 6,432 | \$ 9,021 |
| Weighted average interest rate on short-term borrowing | 5.3% | 7.2% |

Our short-term lines of credit represent borrowings by one of our subsidiaries. We have included this amount in notes payable and current maturities of long-term debt on our combined balance sheets. This facility is renewable annually and the interest rate associated with this line of credit was 5.3% per annum at December 31, 2009.

Maturities of long-term debt during the three years subsequent to December 31, 2009 are as follows: 2010, \$3.4 million; 2011, \$0.1 million; 2012, \$4.2 million.

Power Generations Systems Credit Facility

On February 22, 2006, B&W PGG entered into a senior secured credit facility with a syndicate of lenders (the "B&W PGG Credit Facility"). As amended to date, this facility provides for borrowings and issuances of letters of credit in an aggregate amount of up to \$400 million and matures on February 22, 2011. The proceeds of the B&W PGG Credit Facility are available for working capital needs and other similar corporate purposes of B&W PGG.

B&W PGG's obligations under the B&W PGG Credit Facility are unconditionally guaranteed by all of our domestic subsidiaries included in B&W PGG and secured by liens on substantially all the assets of those subsidiaries, excluding cash and cash equivalents.

The B&W PGG Credit Facility only requires interest payments on a quarterly basis until maturity. Amounts outstanding under the B&W PGG Credit Facility may be prepaid at any time without penalty.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Loans outstanding under the revolving credit subfacility bear interest at either the Eurodollar rate plus a margin ranging from 1.00% to 1.75% per year or the base rate plus a margin ranging from 0.00% to 0.75% per year. The applicable margin for revolving loans varies depending on credit ratings of the B&W PGG Credit Facility. B&W PGG is charged a commitment fee on the unused portion of the B&W PGG Credit Facility, and that fee varies between 0.25% and 0.375% per year depending on credit ratings of the B&W PGG Credit Facility. Additionally, B&W PGG is charged a letter of credit fee of between 1.00% and 1.75% per year with respect to the amount of each letter of credit issued under the B&W PGG Credit Facility. An additional 0.125% per year fee is charged on the amount of each letter of credit issued under the B&W PGG Credit Facility.

The B&W PGG Credit Facility contains customary financial covenants, including maintenance of a maximum leverage ratio and a minimum interest coverage ratio within B&W PGG and covenants that, among other things, restrict the ability of B&W PGG to incur debt, create liens, make investments and acquisitions, sell assets, pay dividends, prepay subordinated debt, merge with other entities, engage in transactions with affiliates and make capital expenditures. The capital expenditure annual limits allow us to roll forward amounts not spent under the limits from one year to the next. However, the amount rolled forward must be spent entirely in the next year and may not be rolled forward again to future years. At December 31, 2009, B&W PGG was in compliance with all of the covenants set forth in the B&W PGG Credit Facility.

As of December 31, 2009, there were no outstanding borrowings, but letters of credit issued under the B&W PGG Credit Facility totaled \$199.2 million. At December 31, 2009, there was \$200.8 million available for borrowings or to meet letter of credit requirements under the B&W PGG Credit Facility. If there had been borrowings under this facility, the applicable interest rate at December 31, 2009 would have been 3.25% per year.

Bank Guarantees (Foreign Operations)

Certain foreign subsidiaries of B&W PGG had credit arrangements with various commercial banks for the issuance of bank guarantees. The aggregate value of all such bank guarantees as of December 31, 2009 was \$16.5 million.

Surety Bonds

In June 2008, MII, B&W PGG and BHI jointly executed a general agreement of indemnity in favor of a surety underwriter relating to surety bonds that underwriter issued in support of the contracting activity of our Power Generation Systems segment. As of December 31, 2009, surety bonds issued under this arrangement totaled approximately \$98.5 million. Any claim successfully asserted against the surety by one or more of the bond obligees would likely be recoverable from any of MII, B&W PGG and BHI under the indemnity agreement.

Government Operations Credit Facility

On December 9, 2003, BWXT entered into a senior unsecured credit facility with a syndicate of lenders (the "BWXT Credit Facility"), which is currently scheduled to mature March 18, 2010. This facility provides for borrowings and issuances of letters of credit in an aggregate amount of up to \$135 million. The proceeds of the BWXT Credit Facility are available for working capital needs and other general corporate purposes of BWXT. We believe we will be successful in obtaining an extension on this facility until we enter into a new, replacement credit facility. If we are not able to obtain an extension, we would consider other alternatives, which could include cash collateralizing letters of credit outstanding under this facility.

The BWXT Credit Facility only requires interest payments on a quarterly basis until maturity. Amounts outstanding under the BWXT Credit Facility may be prepaid at any time without penalty.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Loans outstanding under the BWXT Credit Facility bear interest at either the Eurodollar rate plus a margin ranging from 1.25% to 1.75% per year or the base rate plus a margin ranging from 0.25% and 0.75% per year. The applicable margin for revolving loans varies depending on the leverage ratio of BWXT as of the last day of the preceding fiscal quarter. BWXT is charged an annual commitment fee of 0.375%, which is payable quarterly. Additionally, BWXT is charged a letter of credit fee of between 1.25% and 1.75% per year with respect to the amount of each letter of credit issued, depending on the leverage ratio of BWXT as of the last day of the preceding fiscal quarter. An additional 0.125% per year fee is charged on the amount of each letter of credit issued.

The BWXT Credit Facility contains customary financial and nonfinancial covenants and reporting requirements. The financial covenants require maintenance of a maximum leverage ratio, a minimum fixed charge coverage ratio and a maximum debt to capitalization ratio within BWXT. At December 31, 2009, BWXT was in compliance with all of the covenants set forth in the BWXT Credit Facility.

At December 31, 2009, there were no borrowings outstanding, but letters of credit issued under the BWXT Credit Facility totaled \$59.0 million. At December 31, 2009, there was \$76.0 million available for borrowings or to meet letter of credit requirements under the BWXT Credit Facility. If there had been borrowings under this facility, the applicable interest rate at December 31, 2009 would have been 3.50% per year.

Letters of Credit (Nuclear Fuel Services, Inc.)

At December 31, 2009, NFS, had approximately \$3.7 million in letters of credit issued by various commercial banks on its behalf. The obligations to the commercial banks issuing such letters of credit are secured by cash, short-term certificates of deposit and certain real and intangible assets.

NOTE 7—PENSION PLANS AND POSTRETIREMENT BENEFITS

We have historically provided defined benefit retirement benefits, primarily through noncontributory pension plans, for most of our regular employees. As of 2006, our retirement plans for U.S.-based employees were closed to new entrants for our corporate employees and were closed to new salaried plan entrants for salaried employees of BWXT and B&W PGG and their respective subsidiaries. Effective December 31, 2007, the salaried retirement plan acquired with MMC in May 2007 was closed to new entrants and benefit accruals were frozen for existing participants who were not vested as of December 31, 2007. Effective October 31, 2008, the salaried nourly retirement plans acquired with MMC were merged into the retirement plan for employees of BWXT. Effective December 31, 2008, we acquired the retirement plans and postretirement benefit plans of our subsidiary NFS. Effective December 31, 2009, the salaried retirement plan acquired with NFS was closed to new entrants and benefit accruals were frozen for existing participants who were not vested as of December 31, 2009.

We do not provide retirement benefits to certain non-resident alien employees of foreign subsidiaries. Retirement benefits for salaried employees who accrue benefits in a defined benefit plan are based on final average compensation and years of service, while benefits for hourly paid employees are based on a flat benefit rate and years of service. Our funding policy is to fund the plans as recommended by the respective plan actuaries and in accordance with the Employee Retirement Income Security Act of 1974, as amended, or other applicable law. The Pension Protection Act of 2006 replaces the current funding provisions for single-employer defined benefit plans. Funding provisions under the Pension Protection Act accelerate funding requirements to ensure full funding of benefits accrued. The Pension Protection Act became effective in 2008 and had no impact on our combined financial condition or cash flows for 2008.

Effective December 31, 2009, we adopted the disclosure provisions of FASB Topic 715, Compensation—Retirement Benefits. In accordance with the provisions of this topic, we have disclosed additional information about our assets set aside to fund our pension and postretirement benefit obligations.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Effective December 31, 2007, we adopted the measurement date provision of SFAS No. 158, Employers' Accounting for Defined Benefit Pension and Other Postretirement Plans, for our plans that were not on a calendar year measurement. In accordance with this provision, we recorded a reduction in retained earnings of \$1.7 million, net of a related tax benefit of \$0.8 million.

We make available other benefits which include postretirement health care and life insurance benefits to certain salaried and union retirees based on their union contracts. Certain subsidiaries provide these benefits to unionized and salaried future retirees.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Obligations and Funded Status

| | Year | Benefits Ended | Other I Year I | Ended |
|--|---------------------|--------------------|----------------------|--------------|
| | 2009 | nber 31, 2008 | 2009 | 2008 |
| | | (In thou | | 2000 |
| Change in benefit obligation: | | | | |
| Benefit obligation at beginning of period | \$2,398,965 | \$ 2,355,231 | \$ 141,864 | \$ 103,476 |
| Service cost | 36,565 | 35,651 | 958 | 282 |
| Interest cost | 147,910 | 140,169 | 8,713 | 5,562 |
| Acquisitions | _ | 94,082 | _ | 45,080 |
| Plan participants' contributions | 287 | 283 | 120 | |
| Amendments | 6,141 | 100 | _ | _ |
| Settlements | (2,054) | (1,216) | _ | |
| Actuarial (gain) loss | 80,860 | (56,357) | 2,463 | 1,971 |
| Foreign currency exchange rate changes | 23,154 | (36,882) | 964 | (1,447) |
| Benefits paid | (139,641) | (132,096) | (9,488) | (13,060) |
| Benefit obligation at end of period | <u>\$2,552,187</u> | \$ 2,398,965 | \$ 145,594 | \$ 141,864 |
| Change in plan assets: | | | | |
| Fair value of plan assets at beginning of period | \$1,695,445 | \$ 2,040,794 | \$ 27,079 | \$ — |
| Actual return on plan assets | 168,558 | (399,896) | 3,346 | _ |
| Acquisitions | <u> </u> | 67,321 | _ | 27,079 |
| Plan participants' contributions | 287 | 283 | 120 | _ |
| Company contributions | 49,475 | 157,787 | 9,736 | 13,060 |
| Foreign currency exchange rates | 24,324 | (38,748) | _ | _ |
| Benefits paid | (139,641) | (132,096) | (9,488) | (13,060) |
| Fair value of plan assets at the end of period | \$1,798,448 | \$ 1,695,445 | \$ 30,793 | \$ 27,079 |
| Funded status | \$ (753,739) | \$ (703,520) | \$(114,801) | \$ (114,785) |
| Amounts recognized in the balance sheet consist of: | <u> </u> | <u>+ (++++++</u>) | <u>+ (111,0011</u>) | <u> </u> |
| Accrued employee benefits | \$ (58,000) | \$ (43,530) | \$ (9,317) | \$ (7.317) |
| Accumulated postretirement benefit obligation | \$ (38,000) | \$ (43,330) — | (105,484) | (107,468) |
| Pension liability | (696,584) | (663,146) | (105,404) | (107,400) |
| Intangible asset | 845 | 3,156 | _ | |
| Net amount recognized | \$ (753,739) | \$ (703,520) | (114,801) | \$ (114,785) |
| č | <u>\$ (733,739)</u> | \$ (703,320) | (114,801) | \$ (114,763) |
| Amounts recognized in accumulated other comprehensive loss: | | | | |
| Net actuarial loss | \$ 776,128 | \$ 814,486 | \$ 13,182 | \$ 14,844 |
| Prior service cost | 21,746 | 13,113 | 473 | 399 |
| Unrecognized transition obligation | | | 1,055 | 889 |
| Total before taxes | <u>\$ 797,874</u> | \$ 827,599 | \$ 14,710 | \$ 16,132 |
| Supplemental information: | | | | |
| Plans with accumulated benefit obligation in excess of plan assets | | | | |
| Projected benefit obligation | \$2,448,685 | \$ 2,323,021 | N/A | N/A |
| Accumulated benefit obligation | \$2,341,477 | \$ 2,213,377 | \$ 145,594 | \$ 141,864 |
| Fair value of plan assets | \$1,694,101 | \$ 1,616,345 | \$ 30,793 | \$ 27,079 |
| Plans with plan assets in excess of accumulated benefit obligation | | | | |
| Projected benefit obligation | \$ 103,502 | \$ 75,944 | N/A | N/A |
| Accumulated benefit obligation | \$ 92,812 | \$ 67,798 | \$ — | \$ — |
| Fair value of plan assets | \$ 104,347 | \$ 79,100 | \$ — | \$ — |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

| | | Pension Benefits | | • | Other Benefits | |
|--|------------|------------------|---------------|----------|----------------|----------|
| | | Year Ended | | | Year Ended | |
| | | December 31, | | | December 31, | |
| | 2009(1) | 2008 | 2007 | 2009 | 2008 | 2007 |
| | <u> </u> | <u> </u> | (In thousands |) | | |
| Components of net periodic benefit cost: | | | | | | |
| Service cost | \$ 36,565 | \$ 35,651 | \$ 36,679 | \$ 958 | \$ 282 | \$ 331 |
| Interest cost | 147,910 | 140,169 | 135,893 | 8,713 | 5,562 | 5,988 |
| Expected return on plan assets | (136,211) | (170,866) | (154,939) | (1,504) | _ | |
| Amortization of transition obligation | _ | _ | _ | 257 | 281 | 273 |
| Amortization of prior service cost | 3,119 | 2,757 | 3,085 | 66 | 73 | 71 |
| Recognized net actuarial loss | 81,484 | 29,177 | 42,224 | 1,711 | 1,447 | 1,717 |
| Net periodic benefit cost | \$ 132,867 | \$ 36,888 | \$ 62,942 | \$10,201 | \$7,645 | \$ 8,380 |

⁽¹⁾ Excludes approximately \$2.1 million of income attributable to settlement of previously recorded unfunded pension liabilities.

Additional Information

| | Pension | Pension Benefits Other Be | | | | |
|---|-----------|---------------------------|--------------|-----------------|--|-------|
| | Year | Year Ended | | Year Ended Year | | Ended |
| | Decen | ıber 31, | December 31, | | | |
| | 2009 | 2008 | 2009 | 2008 | | |
| | <u> </u> | (In thousands) | | | | |
| Increase in accumulated other comprehensive loss due to actuarial losses—before taxes | \$ 54,875 | \$ 513,477 | \$612 | \$ 1,971 | | |

We have recognized in the current fiscal year, and expect to recognize in the next fiscal year, the following amounts in other comprehensive loss as components of net periodic benefit cost:

| | Year 1 | Recognized in the Year Ended December 31, 2009 | | Year Ended Year | | e Recognized in the Year Ending ecember 31, 2010 | |
|---|----------|--|-----------|-----------------|--|--|--|
| | Pension | Other | Pension | Other | | | |
| | | (In thousands) | | | | | |
| Pension cost in accumulated other comprehensive loss: | | | | | | | |
| Net actuarial loss | \$81,484 | \$1,711 | \$ 81,612 | \$ 1,460 | | | |
| Prior service cost | 3,119 | 66 | 3,486 | 73 | | | |
| Transition obligation | | 257 | | 281 | | | |
| | \$84,603 | \$2,034 | \$ 85,098 | \$ 1,814 | | | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Assumptions

| | Pension Benefits | | Other Benefits | |
|---|------------------|-------|----------------|-------|
| | 2009 | 2008 | 2009 | 2008 |
| Weighted average assumptions used to determine net periodic benefit obligations at December 31: | | | | |
| Discount rate | 6.03% | 6.32% | 5.57% | 6.15% |
| Rate of compensation increase | 3.77% | 3.96% | _ | _ |
| Weighted average assumptions used to determine net periodic benefit cost for the years ended | | | | |
| December 31: | | | | |
| Discount rate | 6.32% | 6.21% | 6.11% | 5.74% |
| Expected return on plan assets | 8.19% | 8.32% | _ | _ |
| Rate of compensation increase | 4.20% | 3.94% | _ | _ |

The expected rate of return on plan assets assumption is based on the long-term expected returns for the investment mix of assets currently in the portfolio. In setting this rate, we use a building-block approach. Historic real return trends for the various asset classes in the plan's portfolio are combined with anticipated future market conditions to estimate the real rate of return for each class. These rates are then adjusted for anticipated future inflation to determine estimated nominal rates of return for each class. The expected rate of return on plan assets is determined to be the weighted average of the nominal returns based on the weightings of the classes within the total asset portfolio. We have been using an expected return on plan assets assumption of 8.5% for the majority of our existing pension plan assets which is consistent with the long-term asset returns of the portfolio.

Our existing other benefit plans are unfunded, with the exception of the NFS postretirement benefit plans. These plans provide health benefits to certain salaried and hourly employees, as well as retired employees, of NFS. Approximately 84% of total assets for these postretirement benefit plans are contributed into a Voluntary Employees' Beneficiary Association ("VEBA") trust.

| | 2009(1) | 2008(1) |
|--|-------------|-------------|
| Assumed health-care cost trend rates at December 31 | | |
| Health-care cost trend rate assumed for next year | 8.30%-8.50% | 8.50%-8.60% |
| Rates to which the cost trend rate is assumed to decline (ultimate trend rate) | 4.50% | 4.50% |
| Year that the rate reaches ultimate trend rate | 2016-2028 | 2016-2028 |

⁽¹⁾ Assumed health-care cost trend rates and years that the ultimate trends are reached vary among our postretirement benefit arrangements.

Assumed health-care cost trend rates have a significant effect on the amounts we report for our health-care plan. A one-percentage-point change in our assumed health-care cost trend rates would have the following effects:

| | One- | One- |
|--|-------------|----------------|
| | Percentage- | Percentage- |
| | Point | Point |
| | Increase | Decrease |
| | | (In thousands) |
| Effect on total of service and interest cost | \$ 718 | \$ (623) |
| Effect on postretirement benefit obligation | \$ 10.616 | \$ (9.281) |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Investment Goals

General

The overall investment strategy of the pension trusts is to achieve long-term growth of principal, while avoiding excessive risk and to minimize the probability of loss of principal over the long term. The specific investment goals that have been set for the pension trusts in the aggregate are (1) to ensure that plan liabilities are met when due and (2) to achieve an investment return on trust assets consistent with a reasonable level of risk.

Allocations to each asset class for both domestic and foreign plans are reviewed periodically and rebalanced, if appropriate, to assure the continued relevance of the goals, objectives and strategies. The pension trusts for both our domestic and foreign plans employ a professional investment advisor and a number of professional investment managers whose individual benchmarks are, in the aggregate, consistent with the plan's overall investment objectives. The goals of each investment manager are (1) to meet (in the case of passive accounts) or exceed (for actively managed accounts) the benchmark selected and agreed upon by the manager and the trust and (2) to display an overall level of risk in its portfolio that is consistent with the risk associated with the agreed upon benchmark.

The investment performance of total portfolios, as well as asset class components, is periodically measured against commonly accepted benchmarks, including the individual investment manager benchmarks. In evaluating investment manager performance, consideration is also given to personnel, strategy, research capabilities, organizational and business matters, adherence to discipline and other qualitative factors that may impact the ability to achieve desired investment results.

Domestic Plans

We sponsor the following domestic defined benefit plans:

- · Retirement Plan for Employees of McDermott Incorporated and Participating Subsidiary and Affiliated Companies;
- Retirement Plan for Employees of Babcock & Wilcox Commercial Operations;
- · Retirement Plan for Employees of Babcock & Wilcox Government Operations; and
- Nuclear Fuel Services, Inc. Retirement Plan for Salaried Employees and Nuclear Fuel Services, Inc. Retirement Plan for Hourly Employees acquired with NFS (the "NFS Plans").

With the exception of the NFS Plans, the assets of the domestic pension plans were commingled for investment purposes and held by the Trustee, The Bank of New York Mellon, in the McDermott Incorporated Master Trust (the "Master Trust") through December 31, 2009. Effective January 1, 2010, the NFS Plans have been merged into the Master Trust. For the years ended December 31, 2009 and 2008, the investment return (loss) on domestic plan assets of the Master Trust (before deductions for management fees) was approximately 9.3% and (19.8)%, respectively. These percentages exclude the NFS Plans.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The following is a summary of the domestic pension plans' asset allocations at December 31, 2009 and 2008 by asset category.

| | 2009 | 2008 |
|-------------------------------------|------------|------------|
| Asset Category: | | |
| Equity Securities | 28% | 16% |
| Fixed Income | 27% | 32% |
| Commingled and Mutual Funds | 17% | 13% |
| U.S. Government Securities | 10% | 17% |
| Partnerships with Security Holdings | 11% | 11% |
| Real Estate | 4% | 6% |
| Other | <u>3</u> % | <u>5</u> % |
| Total | 100% | 100% |

The target allocation for 2010 for the domestic plans, by asset class, is as follows:

| Asset Class: | |
|----------------|-------|
| Public Equity | 42.5% |
| Private Equity | 10.0% |
| Fixed Income | 38.0% |
| Real Estate | 5.0% |
| Other | 4.5% |

Foreign Plans

B&W PGG sponsors foreign plans through certain foreign subsidiaries. These plans are the plans of Babcock & Wilcox Canada, Ltd. (the "Canadian Plans") and the Diamond Power Specialty Limited Retirement Benefits Plan (the "Diamond UK Plan").

The weighted average asset allocations of these plans at December 31, 2009 and 2008 by asset category were as follows:

| | 2009 | 2008 |
|-------------------|------|------|
| Asset Category: | | |
| Fixed Income | 77% | 53% |
| Equity Securities | 17% | 45% |
| Other | 6% | 2% |
| Total | 100% | 100% |

The target allocation for 2010 for the foreign plans, by asset class, is as follows:

| | Canadian <u>Plans</u> | Diamond UK Plan |
|---------------|--------------------------|--------------------|
| Asset Class: | | |
| U.S. Equity | 15% | 10% |
| Global Equity | 50% | 45% |
| Fixed Income | 35% | 45% |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Fair Value

Effective December 31, 2009 with the additional disclosure provisions of FASB Topic 715, Compensation—Retirement Benefits, the following is a summary of total investments for our plans measured at fair value at December 31, 2009. See Note 13 for a detailed description of fair value measurements and the hierarchy established for valuation inputs.

| | _ | 12/31/09 | _ | Level 1 (In thou | sands | Level 2 | Level 3 |
|-------------------------------------|----|-----------|----|---------------------|-------|---------|---------------|
| Pension and Other Benefits: | | | | (III tilou | sanus | •, | |
| Fixed Income | \$ | 519,313 | \$ | 15,690 | \$ | 503,468 | \$ 155 |
| Equities | | 476,709 | | 476,500 | | _ | 209 |
| Commingled and Mutual Funds | | 377,242 | | 74,438 | | 93,905 | 208,899 |
| U.S. Government Securities | | 155,832 | | 155,832 | | _ | _ |
| Partnerships with Security Holdings | | 179,358 | | _ | | 2,318 | 177,040 |
| Real Estate | | 70,048 | | _ | | _ | 70,048 |
| Cash and Accrued Items | | 50,739 | | 50,152 | | 493 | 94 |
| Total Assets | \$ | 1,829,241 | \$ | 772,612 | \$ | 600,184 | \$ 456,445 |

The following is a summary of the changes in the plans' Level 3 instruments measured on a recurring basis for the year ended December 31, 2009 (in thousands):

| Balance at beginning of period | \$ 483,153 |
|--------------------------------|------------|
| Issuances and acquisitions | 29,758 |
| Dispositions | (64,124) |
| Realized loss | (2,348) |
| Unrealized gain | 10,006 |
| Balance at end of period | \$ 456,445 |

Our pension plan assets are included in a master trust. The master trust invests in certain entities that do not have a quoted market price but calculate net asset value per share or its equivalent. We review these net asset values to determine if any adjustments are required. Our review considers available market data, relevant index returns, preliminary estimates from investees, directional guidance from investees and other data obtained through research, consultation with third-party advisors and direct inquiry to our investees. Based on these factors and any other relevant information we are able to obtain through consultation with third-party advisors, we estimate the value of each investment, including all level 3 investments. If using such estimated values rather than the latest investee net asset values results in a material reduction in the total value of the trust assets, we adjust the reported net asset values to the revised estimated values. For our purposes, a material reduction is 5% or more

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Cash Flows

| | Domesti | Domestic Plans | | Plans |
|---|---------------------|-------------------|---------------------|-------------------|
| | Pension Benefits | Other Benefits | Pension Benefits | Other Benefits |
| Expected employer contributions to trusts of defined benefit plans: | | | | |
| 2010 | \$ 57,419 | N/A | \$ 581 | N/A |
| Expected benefit payments: | | | | |
| 2010 | \$ 141,387 | \$ 14,331 | \$ 13,941 | \$ 699 |
| 2011 | \$ 149,199 | \$ 14,203 | \$ 14,126 | \$ 738 |
| 2012 | \$ 157,658 | \$ 13,758 | \$ 11,924 | \$ 712 |
| 2013 | \$ 164,723 | \$ 13,335 | \$ 12,657 | \$ 703 |
| 2014 | \$ 171,382 | \$ 13,009 | \$ 13,897 | \$ 706 |
| 2015-2019 | \$ 918,338 | \$ 52,133 | \$ 87,944 | \$ 3,419 |

The expected employer contributions to trusts for 2010 are included in current liabilities at December 31, 2009.

Defined Contribution Plans

We participate in the McDermott International, Inc. Supplemental Executive Retirement Plan ("SERP Plan"), which is a defined contribution plan. We recorded income (expense) related to the SERP Plan of approximately \$0.8 million, \$(1.0) million and \$0.9 million in the years ended December 31, 2009, 2008 and 2007, respectively.

We also have provided benefits under the Thrift Plan for Employees of McDermott Incorporated and Participating Subsidiary and Affiliated Companies ("Thrift Plan"). The Thrift Plan generally provides for matching employer contributions of 50% of participants' contributions up to 6 percent of compensation. These matching employer contributions are typically made in shares of MII common stock. In lieu of benefit accruals under the respective retirement plans, the Thrift Plan also provides for service-based contributions to some of our employees. Amounts charged to expense for employer contributions under the Thrift Plan totaled \$17.6 million, \$15.3 million and \$12.5 million in the years ended December 31, 2009, 2008 and 2007, respectively.

Multiemployer Plans

One of our subsidiaries contributes to various multiemployer plans. The plans generally provide defined benefits to substantially all unionized workers in this subsidiary. Amounts charged to pension cost and contributed to the plans were \$20.6 million, \$30.4 million and \$32.6 million in the years ended December 31, 2009, 2008 and 2007, respectively.

NOTE 8—STOCK PLANS

At December 31, 2009, certain of our officers and employees participated in benefit plans of MII, which involve the issuance of MII common stock. Where required, disclosures have been adjusted for MII's stock splits effected in the form of a stock dividend in September 2007 and May 2006.

2009 McDermott International, Inc. Long-Term Incentive Plan

In May 2009, MII's shareholders approved the 2009 McDermott International, Inc. Long-Term Incentive Plan (the "2009 LTIP"). Executive officers, key employees and consultants are eligible to participate in the plan.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The Compensation Committee of MII's Board of Directors selects the participants for the plan. The plan provides for a number of forms of stock-based compensation, including incentive and non-qualified stock options, restricted stock, restricted stock units and performance shares and performance units, subject to satisfaction of specific performance goals. Shares approved under MII's 2001 Directors and Officers Long-Term Incentive Plan (the "2001 LTIP") that were not awarded as of the date of approval of the 2009 LTIP, or shares that are subject to awards that are cancelled, terminated, forfeited, expired or settled in cash in lieu of shares, are available for issuance under the 2009 LTIP. Options to purchase shares are granted at not less than 100% of the fair market value (average of the high and low trading price) on the date of grant, become exercisable at such time or times as determined when granted and expire not more than seven years after the date of grant.

2001 Directors and Officers Long-Term Incentive Plan

In May 2009, MII's shareholders approved the 2009 LTIP. As a result, no further awards will be issued under the 2001 LTIP. Executive officers, key employees and consultants were eligible to participate in the 2001 LTIP. The Compensation Committee of MII's Board of Directors selected the participants for the plan. The plan provided for a number of forms of stock-based compensation, including nonqualified stock options, incentive stock options, stock appreciation rights, restricted stock, deferred stock units, performance shares and performance units, subject to satisfaction of specific performance goals. Options to purchase shares were granted at not less than 100% of the fair market value (average of the high and low trading price) on the date of grant, became exercisable at such time or times as determined when granted and expire not more than seven years after the date of the grant. Options granted prior to May 2009 expire not more than ten years after the date of the grant. Shares of common stock available to be awarded under the 2001 LTIP are available under the terms of the 2009 LTIP and have been included in the amount available for grant discussed above

In the event of a change in control of MII, all of these stock-based compensation programs have provisions that may cause restrictions to lapse and accelerate the exercisability of outstanding options.

Total compensation expense recognized for the years ended December 31, 2009, 2008 and 2007 was as follows:

| | | npensation | Tax | Net |
|--------------------------------------|-----------|------------|----------------------|----------|
| | <u>I</u> | Expense | Benefit | Impact |
| | | | (In thousands) | |
| | | Year F | Ended December 31, 2 | :009 |
| Stock options | \$ | 2,799 | \$ (999) | \$ 1,800 |
| Restricted stock | | 3,052 | (1,106) | 1,946 |
| Restricted stock units | | 3,870 | (1,392) | 2,478 |
| Performance shares | | 14,248 | (5,162) | 9,086 |
| Performance and deferred stock units | | 1,922 | (686) | 1,236 |
| TOTAL | <u>\$</u> | 25,891 | <u>\$(9,345)</u> | \$16,546 |
| | | | <u></u> | |
| | | Year E | Ended December 31, 2 | 008 |
| Stock options | \$ | 498 | \$ (178) | \$ 320 |
| Restricted stock | | 2,202 | (805) | 1,397 |
| Performance shares | | 20,404 | (7,244) | 13,160 |
| Performance and deferred stock units | | 1,732 | (618) | 1,114 |
| TOTAL | \$ | 24,836 | \$(8,845) | \$15,991 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

| | Con | npensation | Tax | Net |
|--------------------------------------|-----|------------|--------------------|----------|
| | 1 | Expense | Benefit | Impact |
| | | | (In thousands) | |
| | | Year F | Ended December 31, | 2007 |
| Stock options | \$ | 1,569 | \$ (560) | \$ 1,009 |
| Restricted stock | | 47 | (17) | 30 |
| Performance shares | | 14,028 | (4,957) | 9,071 |
| Performance and deferred stock units | | 5,122 | (1,825) | 3,297 |
| TOTAL | \$ | 20,766 | <u>\$(7,359)</u> | \$13,407 |

As of December 31, 2009, total unrecognized estimated compensation expense related to nonvested awards was \$21.5 million, net of estimated tax benefits of \$12.0 million. The components of the total gross unrecognized estimated compensation expense of \$33.5 million and their expected weighted-average periods for expense recognition are as follows (amounts in millions; periods in years):

| | Amount | Weighted- Average Period |
|--------------------------------------|---------|--------------------------------|
| | | |
| Stock options | \$ 8.7 | 1.1 |
| Restricted stock | \$ 3.9 | 1.1 |
| Restricted stock units | \$ 8.0 | 1.3 |
| Performance shares | \$ 12.7 | 0.6 |
| Performance and deferred stock units | \$ 0.2 | 0.4 |

Stock Options

The following table summarizes activity for MII's stock options for the year ended December 31, 2009 (share data in thousands):

| | Number of Shares | Weighted- Average Exercise Price | Weighted- Average Remaining Contractual Term | Aggregate Intrinsic Value (in millions) |
|--------------------------------|------------------------|---|--|---|
| Outstanding, beginning of year | 576 | \$ 3.99 | | |
| Granted | 1,213 | 11.73 | | |
| Exercised | (202) | 3.85 | | |
| Cancelled/expired/forfeited | (5) | 22.69 | | |
| Transferred to affiliate | (0) | | | |
| Outstanding, end of year | 1,582 | \$ 9.92 | 5.4 Years | \$ 22.6 |
| Exercisable, end of year | 374 | \$ 4.07 | 2.7 Years | \$ 7.6 |

The aggregate intrinsic value included in the table above represents the total pretax intrinsic value that would have been received by the option holders had all option holders exercised their options on December 31, 2009. The intrinsic value is calculated as the total number of option shares multiplied by the difference between the closing price of MII's common stock on the last trading day of each period and the exercise price of the options. This amount changes based on the fair market value of MII's common stock.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The weighted average fair value of stock options granted in the year ended December 31, 2009 was \$11.73. There were no stock options granted in the years ended December 31, 2008 and 2007. The total fair value of options vested was \$1.3 million and \$2.6 million during the years ended December 31, 2008 and 2007, respectively. No options vested during the year ended December 31, 2009.

During the years ended December 31, 2009, 2008 and 2007, the total intrinsic value of stock options exercised was \$3.8 million, \$54.5 million and \$105.6 million, respectively. The actual tax benefits realized related to stock options exercised during the years ended 2009 and 2008 was \$1.7 million and \$11.6 million, respectively.

Restricted Stock

Non-vested restricted stock awards as of December 31, 2009 and changes during the year ended December 31, 2009 were as follows (share data in thousands):

| | Number of Shares | Weighted- Average Grant Date Fair Value |
|-------------------------------|------------------------|--|
| Non-vested, beginning of year | 243 | \$ 37.86 |
| Granted | <u> </u> | _ |
| Restrictions lapsed | (82) | 15.55 |
| Cancelled/forfeited | (2) | 52.65 |
| Transferred to affiliate | 3 | 56.25 |
| Nonvested, end of year | 162 | \$ 37.86 |

Restricted Stock Units

Non-vested restricted stock awards as of December 31, 2009 and changes during the year ended December 31, 2009 were as follows (share data in thousands):

| | Number of Shares | Weighted- Average Grant Date Fair Value |
|-------------------------------|------------------------|--|
| Non-vested, beginning of year | _ | \$ — |
| Granted | 1,114 | 11.40 |
| Restrictions lapsed | _ | _ |
| Cancelled/forfeited | (9) | 21.54 |
| Transferred to affiliate | <u>—</u> | _ |
| Nonvested, end of year | 1,105 | \$ 11.36 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Performance Shares

Non-vested performance share awards as of December 31, 2009 and changes during the year ended December 31, 2009 were as follows (share data in thousands):

| | Number of Shares | Weighted- Average Grant Date Fair Value |
|-------------------------------|------------------------|--|
| Non-vested, beginning of year | 1,615 | \$ 33.06 |
| Granted | 708 | 15.00 |
| Vested | (977) | 23.12 |
| Cancelled/forfeited | (5) | 43.68 |
| Transferred to affiliate | 9 | 32.08 |
| Non-vested, end of year | 1,350 | \$ 30.72 |

For awards made in 2007 and 2008, the actual number of shares earned by each participant is dependent upon achievement of certain consolidated operating income targets over the three-year performance periods. The awards actually earned will range from zero to 150% of the targeted number of performance shares, to be determined upon completion of the three-year performance period.

For awards made in 2009, the actual number of shares earned by each participant is dependent upon (1) achievement of certain consolidated operating income targets and (2) total shareholder return relative to our peers over the three-year performance periods. The awards actually earned will range from zero to 200% of the targeted number of performance shares, to be determined upon completion of the three-year performance period.

The intrinsic value of performance shares vested during the year ended December 31, 2009 was \$18.0 million. No performance shares vested during the years ended December 31, 2008 and 2007.

Performance and Deferred Stock Units

Nonvested performance and deferred stock unit awards as of December 31, 2009 and changes during the year ended December 31, 2009 were as follows (share data in thousands):

| | Number of Units | Aggregate Intrinsic Value (in millions) |
|------------------------------|-----------------------|---|
| Nonvested, beginning of year | 157 | |
| Granted | _ | |
| Vested | (84) | |
| Cancelled/forfeited | (1) | |
| Transferred to Affiliate | _ | |
| Nonvested, end of year | <u>72</u> | \$ 1.7 |

The aggregate intrinsic value included in the table above represents the total pretax intrinsic value recorded as a liability at December 31, 2009 in the combined balance sheets. During the years ended December 31, 2009, 2008 and 2007, we paid \$1.5 million, \$4.1 million and \$3.3 million, respectively, for the settlement of vested performance and deferred stock units.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

NOTE 9—COMMITMENTS AND CONTINGENCIES

Investigations and Proceedings

On January 29, 2010, Michelle McMunn, Cara D. Steele and Yvonne Sue Robinson filed suit against B&W PGG, Babcock & Wilcox Technical Services Group, Inc., formerly known as B&W Nuclear Environmental Services, Inc. (together with B&W PGG, the "B&W Parties") and Atlantic Richfield Company ("ARCO") in the United States District Court for the Western District of Pennsylvania (the "McMunn Litigation"). The plaintiffs in the McMunn Litigation allege, among other things, that they suffered personal injuries and property damage as a result of alleged radioactive and non-radioactive releases relating to the operation, remediation and/or decommissioning of two former nuclear fuel processing facilities located in Apollo and Parks Township, Pennsylvania. Those facilities previously were owned by Nuclear Materials and Equipment Company, a former subsidiary of ARCO ("NUMEC") which was acquired by B&W PGG. The plaintiffs in the McMunn Litigation seek compensatory and punitive damages.

At the time of ARCO's sale of NUMEC to B&W PGG, B&W PGG received an indemnity and hold harmless agreement from ARCO from claims or liabilities arising as a result of pre-closing NUMEC or ARCO actions.

We intend to vigorously defend this matter, and believe that in the event of liability, if any, the claims alleged in the McMunn Litigation will be resolved within the limits of coverage of our insurance policies and/or the ARCO indemnity.

B&W PGG settled approximately 245 personal injury and wrongful death claims, as well as approximately 125 property damage claims, alleging injury and damage as a result of alleged releases relating to these two facilities. The aggregate settlement amount for these claims was \$52.5 million, which was paid during 2009, all of which had been previously accrued. In connection with that settlement, the B&W Parties are pursuing recovery in the matter of *The Babcock & Wilcox Company et al. v. American Nuclear Insurers et al.* (the "ANI Litigation") from their insurer, American Nuclear Insurers and Mutual Atomic Energy Liability Underwriters, of the amounts paid in settlement of that prior action. The ANI Litigation is pending before the Court of Common Pleas of Allegheny County, Pennsylvania. No trial date has been set in the matter.

An action entitled *Iroquois Falls Power Corp. v. Jacobs Canada Inc., et al.,* was filed in the Superior Court of Justice, in Ontario, Canada, on June 1, 2005. Iroquois Falls Power Corp. ("Iroquois") seeks damages of approximately \$14 million (Canadian) as a result of an alleged breach by one of our former subsidiaries in connection with the supply and installation of heat recovery steam generators. MI, which provided a guarantee to certain obligations of the former subsidiary, and two bonding companies with whom MII entered into an indemnity arrangement, were also named as defendants. In March 2007, the Superior Court granted summary judgment in favor of all defendants and dismissed all claims of Iroquois, which appealed the ruling. Subsequently, the Court of Appeals for Ontario upheld the summary judgment, but sent the case back to the Superior Court of Justice to allow Iroquois an opportunity to amend its complaint to assert new claims. The Superior Court of Justice, however, denied Iroquois' request to amend its complaint and assert new claims against the defendants based on a breach of contractual warranty. Iroquois appealed the Superior Court's decision

and, in June 2009, the Court of Appeals for Ontario reversed the decision and sent the case back to the Superior Court for Iroquois to file an amended complaint on those new claims. In January 2010, our notice to appeal the Court of Appeals' decision was dismissed by the Supreme Court.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Other

Warranty Claim

One of our Canadian subsidiaries has received notice of a warranty claim on one of its projects on a contract executed in 1998. This situation relates to technical issues concerning components associated with nuclear steam generators. Data collection and analysis can only be performed at specific time periods when the power plant is scheduled to be off-line for maintenance. We also received a notice from the customer during October 2008, and, during November 2008, we responded to the notice by disagreeing with the matters stated in the claim and disputing the claim. This project included a limited-term performance bond totaling approximately \$140 million for which we entered into an indemnity arrangement with the surety underwriters. It is possible that our subsidiary may incur warranty costs in excess of amounts provided for as of December 31, 2009. It is also possible that a claim could be initiated by our subsidiary's customer against the surety underwriter should certain events occur. If such a claim were successful, the surety could seek to recover from our subsidiary the costs incurred in satisfying the customer claim. If the surety seeks recovery from our subsidiary, we believe that our subsidiary would have adequate liquidity to satisfy its obligations. However, the ultimate resolution of this possible claim is uncertain, and an adverse outcome could have a material adverse impact on our combined financial condition, results of operations and cash flows.

Suspended Operations

In December 2009, our subsidiary Nuclear Fuel Services, Inc., which we acquired in December 2008, implemented a suspension of some operations at its Erwin, Tennessee manufacturing facility while implementing organizational, facility and management changes to enhance safety controls and processes. We suspended the operations in consultation with the NRC following the occurrence of two separate incidents that we reported to the NRC in October and November 2009. The October 2009 incident involved the generation of excessive heat and a hazardous gas at a specialized cleaning station. The incident was caused by the processing of a small amount of uranium-aluminum material in nitric acid. This incident resulted in some damage to piping, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. The November 2009 incident involved a fire in a glovebox enclosure. The incident was caused by a reaction between fluorine gas in a cylinder of uranium and the fiberglass backing of the teflon lining of a vent hose attached to the cylinder within the glovebox enclosure. This incident resulted in damage to a hose and a faceplate within the glove box enclosure, but did not result in any injuries to personnel or offsite releases of chemical or radioactive material. Inspections conducted separately by the NRC, our existing nuclear liability underwriter and us revealed specific modifications necessary to improve Nuclear Fuel Services, Inc.'s overall safety performance. Although the NRC did not issue any directive or order to us to suspend these operations, we believe the NRC likely would have done so had we not voluntarily suspended the operations. Although we were already in the process of implementing safety, cultural and procedural improvements at NFS, we have refined and accelerated those changes and have developed other specific enhancements in consultation with the NRC, as confirmed in the NRC's January 7, 2010 confirmatory action letter to Nuclear Fuel Services, Inc.

The suspended operations include production operations, the commercial development line and the highly enriched uranium down-blending facility. There can be no assurance that we will not have to suspend our operations in the future to implement additional changes to enhance our safety controls and processes in order to comply with applicable laws and regulations.

Additionally, due to the nature of our business, we are, from time to time, involved in routine litigation or subject to disputes or claims related to our business activities, including, among other things:

- performance- or warranty-related matters under our customer and supplier contracts and other business arrangements; and
- · workers' compensation claims, premises liability claims and other claims.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Based upon our prior experience, we do not expect that any of these other litigation proceedings, disputes and claims will have a material adverse effect on our combined financial condition, results of operations or cash flows.

Environmental Matters

We have been identified as a potentially responsible party at various cleanup sites under the Comprehensive Environmental Response, Compensation, and Liability Act, as amended ("CERCLA"). CERCLA and other environmental laws can impose liability for the entire cost of cleanup on any of the potentially responsible parties, regardless of fault or the lawfulness of the original conduct. Generally, however, where there are multiple responsible parties, a final allocation of costs is made based on the amount and type of wastes disposed of by each party and the number of financially viable parties, although this may not be the case with respect to any particular site. We have not been determined to be a major contributor of wastes to any of these sites. On the basis of our relative contribution of waste to each site, we expect our share of the ultimate liability for the various sites will not have a material adverse effect on our financial condition, results of operations or cash flows in any given year.

The Department of Environmental Protection of the Commonwealth of Pennsylvania ("PADEP") advised us in March 1994 that it would seek monetary sanctions and remedial and monitoring relief related to the Parks Facilities. The relief sought is related to potential groundwater contamination resulting from previous operations at the facilities. These facilities are currently owned by a subsidiary in our Government Operations segment. PADEP has advised us that it does not intend to assess any monetary sanctions, provided our Government Operations segment contamination remediation will be required at the Parks Facility remains unclear. Results from sampling completed by our Government Operations segment have indicated that such remediation may not be necessary. Our Government Operations segment continues to evaluate closure of the groundwater issues pursuant to applicable Pennsylvania law.

We perform significant amounts of work for the U.S. Government under both prime contracts and subcontracts and operate certain facilities that are licensed to possess and process special nuclear materials. As a result of these activities, we are subject to continuing reviews by governmental agencies, including the U.S. Environmental Protection Agency and the NRC.

The NRC's decommissioning regulations require our Government Operations segment to provide financial assurance that it will be able to pay the expected cost of decommissioning each of its facilities at the end of its service life. We will continue to provide financial assurance aggregating \$33.7 million during the year ending December 31, 2010 with existing letters of credit for the ultimate decommissioning of all of these licensed facilities, except two. These two facilities, which represent the largest portion of our eventual decommissioning costs, have provisions in their government contracts pursuant to which substantially all of our decommissioning costs and financial assurance obligations are covered by the U.S. Department of Energy, including the costs to complete the decommissioning projects underway at the facility in Erwin. Tennessee.

At December 31, 2009 and 2008, we had total environmental reserves (including provisions for the facilities discussed above) of \$50.3 million and \$38.0 million, respectively. Of our total environmental reserves at December 31, 2009 and 2008, \$2.5 million and \$5.0 million, respectively, were included in current liabilities. Inherent in the estimates of those reserves and recoveries are our expectations regarding the levels of contamination, decommissioning costs and recoverability from other parties, which may vary significantly as decommissioning activities progress. Accordingly, changes in estimates could result in material adjustments to our operating results, and the ultimate loss may differ materially from the amounts that we have provided for in our combined financial statements.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

Operating Leases

Future minimum payments required under operating leases that have initial or remaining non-cancellable lease terms in excess of one year at December 31, 2009 are as follows (in thousands):

| Fiscal Year Ending December 31, | Amount |
|---------------------------------|---------|
| 2010 | \$5,577 |
| 2011 | \$3,955 |
| 2012 | \$2,961 |
| 2013 | \$1,351 |
| 2014 | \$ 977 |
| Thereafter | _ |

Total rental expense for the years ended December 31, 2009, 2008 and 2007 was \$9.8 million, \$9.2 million and \$8.9 million respectively. These expense amounts include contingent rentals and are net of sublease income, neither of which is material.

Surety Bonds

In June 2008, MII, B&W PGG and BHI jointly executed a general agreement of indemnity in favor of a surety underwriter relating to surety bonds that underwriter issued in support of B&W PGG's contracting activity. As of December 31, 2009, bonds issued under this arrangement totaled approximately \$98.5 million. Any claim successfully asserted against the surety by one or more of the bond obligees would likely be recoverable from MII, B&W PGG and BHI under the indemnity agreement.

Proposed Unfavorable Tax Adjustments

We were advised in 2006 by the IRS of proposed unfavorable tax adjustments related to the 2001 through 2003 tax years. We reviewed the IRS positions and disagreed with certain proposed adjustments. Accordingly, we filed a protest with the IRS regarding the resolution of these issues, and the process has proceeded through an appeals hearing with an IRS appellate conferee. We have provided for any amounts that we believe will ultimately be payable for these proposed adjustments. However, the ultimate resolution of these proposed adjustments are uncertain, and an adverse outcome could have a material adverse impact on our combined financial condition, results of operations and cash flows.

We perform significant amounts of work for the U.S. Government under both prime contracts and subcontracts. As a result, various aspects of our operations are subject to continuing reviews by governmental agencies.

We maintain liability and property insurance against such risk and in such amounts as we consider adequate. However, certain risks are either not insurable or insurance is available only at rates that we consider uneconomical.

NOTE 10—FINANCIAL INSTRUMENTS WITH CONCENTRATIONS OF CREDIT RISK

Our primary customers are the U.S. Government and companies in the electric power generation industry (including several government-owned utilities and independent power producers). These concentrations of customers may impact our overall exposure to credit risk, either positively or negatively, in that our customers may be similarly affected by changes in economic or other conditions. In the years ended December 31, 2009 and 2008, the U.S. Government accounted for approximately 33% and 22%, respectively, of our total revenues.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

We and some of our customers operate worldwide and are therefore exposed to risks associated with the economic and political forces of various countries and geographic areas. Approximately 16% of our trade receivables at December 31, 2009 are due from foreign customers. See Note 14 for additional information about our foreign operations. We generally do not obtain any collateral for our receivables.

We believe that our provision for possible losses on uncollectible accounts receivable is adequate for our credit loss exposure. At December 31, 2009 and 2008, the allowance for possible losses that we deducted from accounts receivable—trade on the accompanying balance sheet was \$3.3 million and \$2.7 million, respectively.

NOTE 11—INVESTMENTS

The following is a summary of our available-for-sale securities at December 31, 2009:

| | | Gross | Gross | Estimated |
|--|-----------|------------|-----------------|-----------|
| | Amortized | Unrealized | Unrealized | Fair |
| | Cost | Gains | Losses | Value |
| | | (In the | ousands) | |
| U.S. Treasury securities and obligations of U.S. Government agencies | \$ 63,997 | \$ 44 | \$ (4) | \$64,037 |
| Money market instruments and short-term investments | 461 | 2 | _ | 463 |
| Asset-backed securities and collateralized mortgage obligations | 1,169 | _ | (429) | 740 |
| Corporate and foreign government bonds and notes | 8,302 | 14 | (2) | 8,314 |
| Total | \$ 73,929 | \$ 60 | <u>\$ (435)</u> | \$73,554 |

The following is a summary of our available-for-sale securities at December 31, 2008:

| | Amortized Cost | Unrealized Gains (In the | Unrealized Losses ousands) | Estimated Fair Value |
|--|-------------------|--------------------------|----------------------------|-------------------------|
| U.S. Treasury securities and obligations of U.S. Government agencies | \$ 72,858 | \$ 658 | \$ — | \$ 73,516 |
| Money market instruments and short-term investments | 28,003 | 77 | _ | 28,080 |
| Asset-backed securities and collateralized mortgage obligations | 1,501 | _ | (548) | 953 |
| Corporate and foreign government bonds and notes | 16,788 | | (253) | 16,535 |
| Total | \$119,150 | \$ 735 | <u>\$ (801)</u> | \$119,084 |

Proceeds, gross realized gains and gross realized losses on sales of available-for-sale securities were as follows:

| | | 01033 | 01033 |
|------------------------------|-----------|----------|----------|
| | | Realized | Realized |
| | Proceeds | Gains | Losses |
| Year Ended December 31, 2009 | \$140,203 | \$ — | \$ 30 |
| Year Ended December 31, 2008 | \$367,109 | \$ 148 | \$ — |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

NOTE 12—DERIVATIVE FINANCIAL INSTRUMENTS

Our worldwide operations give rise to exposure to market risks from changes in foreign exchange rates. We use derivative financial instruments (primarily foreign currency forward-exchange contracts) to reduce the impact of changes in foreign exchange rates on our operating results. We use these instruments primarily to hedge our exposure associated with revenues or costs on our long-term contracts and other cash flow exposures that are denominated in currencies other than our operating entities' functional currencies. We do not hold or issue financial instruments for trading or other speculative purposes.

We enter into derivative financial instruments primarily as hedges of certain firm purchase and sale commitments denominated in foreign currencies. We record these contracts at fair value on our combined balance sheets. Depending on the hedge designation at the inception of the contract, the related gains and losses on these contracts are either deferred in parent equity (deficit) as a component of accumulated other comprehensive loss, until the hedged item is recognized in earnings, or offset against the change in fair value of the hedged firm commitment through earnings. The ineffective portion of a derivative's change in fair value and any portion excluded from the assessment of effectiveness are immediately recognized in earnings. The gain or loss on a derivative instrument not designated as a hedging instrument is also immediately recognized in earnings. Gains and losses on derivative financial instruments that require immediate recognition are included as a component of other income (expense) – net in our combined statements of income.

We have designated all of our forward contracts as cash flow hedges. The hedged risk is the risk of changes in functional-currency-equivalent cash flows attributable to changes in spot exchange rates of firm commitments related to long-term contracts. We exclude from our assessment of effectiveness the portion of the fair value of the forward contracts attributable to the difference between spot exchange rates and forward exchange rates. Ineffective portions of our forward contracts are recorded in other income (expense) – net on our combined statements of income. At December 31, 2009, we had deferred approximately \$5.8 million of net losses on these derivative financial instruments in accumulated other comprehensive loss. We expect to recognize approximately \$3.5 million of this amount in the next 12 months.

At December 31, 2009, all of our derivative financial instruments consisted of foreign currency forward-exchange contracts, and foreign currency options. The notional value of our forward contracts totaled \$120.4 million at December 31, 2009, with maturities extending to December 2011. These instruments consist primarily of contracts to purchase or sell Euros or Canadian Dollars. The fair value of these contracts totaled (\$1.8) million, all of which are Level 2 in nature (See Note 13). The fair value of our foreign currency option contracts totaled \$4.7 million at December 31, 2009, and is included in other current assets on our combined balance sheets. We are exposed to credit-related losses in the event of nonperformance by counterparties to derivative financial instruments. However, when possible, we enter into International Swaps and Derivative Association, Inc. agreements with our hedge counterparties to mitigate this risk. We also attempt to mitigate this risk by using major financial institutions with high credit ratings and limit our exposure to hedge counterparties based on their credit ratings. The counterparties to all of our derivative financial instruments are financial institutions included in our credit facilities. Our hedge counterparties have the benefit of the same collateral arrangements and covenants as described under these facilities.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The following tables summarize our derivative financial instruments at December 31, 2009:

| | Asset Derivatives December 31, 2009 | | Liability Derivatives December 31, 2009 | | | |
|---|---|---------------|--|--------------------------|----------------------------|--------|
| | Balance Sheet Account | Fair Value | | Balance Sheet Account | Fair Value | |
| | | (In t | thousands) | | | |
| Derivatives designated as hedging instruments: | | | | | | |
| Foreign-exchange contracts | Accounts receivable-other | \$ | 1,582 | Accounts payable | \$ | 3,350 |
| Derivatives not designated as hedging instruments: | | | | | | |
| Foreign currency options | Other current assets | \$ | 4,747 | Accounts payable | \$ | _ |
| The Effect of Derivative Instruments on the S (I | statements of Financial Performance Decemb in thousands) | ber 31, 20 | 009 | | | |
| | | | | | Twelve Months | |
| | | | | | Ended December 31, 2009 | |
| Derivatives Designated as Hedges: | | | | 2000 | JC1 U. | , 2005 |
| Cash Flow Hedges: | | | | | | |
| Foreign Exchange Contracts: | | | | | | |
| Amount of gain recognized in other comprehensive income | | | | \$ | 1 | 0,979 |
| Income (loss) reclassified from accumulated other comprehe | nsive loss into income: effective por | tion | | | | |
| Location | | | | | | |
| Revenues | | | | \$ | | 71 |
| Cost of operations | | | | \$ | | 1,499 |
| Other—net | | | | \$ | | 624 |
| Gain (loss) recognized in income: portion excluded from effective | ectiveness testing | | | | | |
| Location | | | | | | |
| Other—net | | | | \$ | (- | 4,335) |

NOTE 13—FAIR VALUES OF FINANCIAL INSTRUMENTS

As discussed in Note 1, we adopted FASB Topic, Fair Value Measurements and Disclosure, January 1, 2008 for fair value measurement of financial instruments and recurring fair value measurements of nonfinancial assets and liabilities. This topic defines fair value, establishes a framework for measuring fair value and expands disclosures about fair value measurements.

This topic defines fair value as the price that would be received to sell an asset or paid to transfer a liability (an exit price) in an orderly transaction between market participants at the measurement date. This topic also sets forth the disclosure requirements regarding fair value and establishes a hierarchy for valuation inputs that emphasizes the use of observable inputs when measuring fair value. A financial instrument's categorization within the fair value hierarchy is based upon the lowest level of input that is significant to the fair value measurement. The fair value hierarchy established by this topic is broken down as follows:

Level 1—inputs are based upon quoted prices for identical instruments traded in active markets.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

- Level 2—inputs are based upon quoted prices for similar instruments in active markets, quoted prices for similar or identical instruments in inactive
 markets and model-based valuation techniques for which all significant assumptions are observable in the market or can be corroborated by observable
 market data for substantially the full term of the assets and liabilities.
- Level 3—inputs are generally unobservable and typically reflect management's estimates of assumptions that market participants would use in pricing the
 asset or liability. The fair values are therefore determined using model-based techniques that include option pricing models, discounted cash flow models
 and similar valuation techniques.

The following sections describe the valuation methodologies we use to measure the fair values of our available-for-sale securities and derivatives.

Derivatives

Level 2 derivative assets and liabilities primarily include over-the-counter forwards. These currently consist of foreign exchange rate derivatives. Where applicable, the value of these derivative assets and liabilities is computed by discounting the projected future cash flow amounts to present value using market-based observable inputs, including foreign exchange forward and spot rates, interest rates and counterparty performance risk adjustments.

At December 31, 2009, we had forward contracts outstanding to purchase or sell foreign currencies, primarily Euros and Canadian Dollars, with a total notional value of \$120.4 million and a total fair value of \$(1.8) million. In addition, we had foreign currency options outstanding at December 31, 2009 with a total fair value of \$4.7 million.

Available-for-Sale Securities

Investments other than derivatives primarily include U.S. Government and agency securities, money market funds, mortgage-backed securities, corporate notes and bonds and certificates of deposit.

In general, and where applicable, we use a pricing service that principally uses a composite of observable prices and quoted prices in active markets for identical assets or liabilities to determine fair value. This pricing methodology applies to our Level 2 investments.

Fair Value Measurements

The following is a summary of our available-for-sale securities measured at fair value at December 31, 2009:

| 12/31/09 | Level I | Level 2 | Level 3 |
|----------|----------------------------------|--|---|
| <u> </u> | (In thou | | |
| \$ 463 | \$ — | \$ 463 | \$ — |
| 64,037 | 63,005 | 1,032 | _ |
| 740 | _ | 572 | 168 |
| 8,314 | | 8,314 | |
| \$73,554 | \$63,005 | \$10,381 | \$ 168 |
| | \$ 463 64,037 740 8,314 | \$ 463 \$ — 64,037 63,005 740 — 8,314 — | \$ 463 \$ — \$ 463 64,037 63,005 1,032 740 — 572 8,314 — 8,314 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The following is a summary of our available-for-sale securities measured at fair value at December 31, 2008:

| | 12/31/08 | Level 1 | Level 2 | Level 3 |
|---|------------|-----------|-----------|---------|
| | | (In thous | | |
| Commercial paper | \$ 19,080 | \$ — | \$ 19,080 | \$ — |
| Certificates of deposit | 9,000 | _ | 9,000 | _ |
| U.S. Government and agency securities | 73,516 | 68,452 | 5,064 | _ |
| Asset-backed/mortgage-backed securities | 953 | _ | 786 | 167 |
| Foreign government bonds | 839 | _ | 839 | _ |
| Corporate notes and bonds | 15,696 | | 15,696 | |
| Total | \$ 119,084 | \$ 68,452 | \$ 50,465 | \$ 167 |
| | | | | |

Changes in Level 3 Instrument

The following is a summary of the changes in our Level 3 instrument measured on a recurring basis for the years ended December 31, 2009 and 2008 (in thousands):

| 2009 | 2008 |
|--|-------|
| Balance at beginning of period \$ 167 | 386 |
| Total realized and unrealized gains (losses): 54 | (150) |
| Included in other income (expense) | 1 |
| Included in other comprehensive income 54 | (151) |
| Purchases, issuances, and settlements — | _ |
| Principal repayments (53) | (69) |
| Balance at end of period \$\\\\$ 168 | S 167 |

Other Financial Instruments

We used the following methods and assumptions in estimating our fair value disclosures for financial instruments:

Cash and cash equivalents and restricted cash and cash equivalents: The carrying amounts that we have reported in the accompanying combined balance sheets for cash and cash equivalents approximate their fair values.

Investments: We estimate the fair values of investments based on quoted market prices. For investments for which there are no quoted market prices, we derive fair values from available yield curves for investments of similar quality and terms.

Long- and short-term debt: We base the fair values of debt instruments on quoted market prices. Where quoted prices are not available, we base the fair values on the present value of future cash flows discounted at estimated borrowing rates for similar debt instruments or on estimated prices based on current yields for debt issues of similar quality and terms.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

The estimated fair values of our financial instruments are as follows:

| | December | December 31, 2009 | | 31, 2008 |
|--------------------------------------|------------|-------------------|-------------|-------------|
| | Carrying | | | <u> </u> |
| | Amount | Fair Value | Amount | Fair Value |
| | | (In thousands) | | |
| Balance Sheet Instruments | | | | |
| Cash and cash equivalents | \$469,468 | \$469,468 | \$279,646 | \$279,646 |
| Restricted cash and cash equivalents | \$ 15,305 | \$ 15,305 | \$ 7,116 | \$ 7,116 |
| Investments | \$ 73,554 | \$ 73,554 | \$119,084 | \$119,084 |
| Debt | \$ 10,654 | \$ 10,782 | \$ 15,130 | \$ 15,221 |
| Forward contracts | \$ (1,769) | \$ (1,769) | \$ (17,000) | \$ (17,000) |
| Foreign currency options | \$ 4,747 | \$ 4,747 | \$ — | \$ — |

NOTE 14—SEGMENT REPORTING

Our reportable segments are Power Generation Systems and Government Operations, as described in Note 1. The operations of our segments are managed separately and each has unique technology, services and customer class.

We account for intersegment sales at prices that we generally establish by reference to similar transactions with unaffiliated customers. Reportable segments are measured based on operating income exclusive of general corporate expenses, contract and insurance claims provisions, legal expenses and gains (losses) on sales of corporate assets. Other reconciling items to income before provision for income taxes are interest income, interest expense, minority interest and other income (expense) —net.

Segment Information for the Years Ended December 31, 2009, 2008 and 2007

1. Information about Operations in our Different Industry Segments:

| | Yes | Year Ended December 31, | | | |
|---|-------------|-------------------------|-------------|--|--|
| | 2009 | 2008 | 2007 | | |
| | · | (In thousands) | <u> </u> | | |
| REVENUES(1): | | | | | |
| Power Generation Systems | \$1,824,450 | \$2,548,855 | \$2,506,814 | | |
| Government Operations | 1,032,023 | 851,019 | 694,029 | | |
| Adjustments and Eliminations | (1,841) | (1,300) | (899) | | |
| | \$2,854,632 | \$3,398,574 | \$3,199,944 | | |
| (1) Segment revenues are net of the following intersegment transfers and other adjustments: | | | | | |
| Power Generation Systems Transfers | \$ 2 | \$ 55 | \$ 123 | | |
| Government Operations Transfers | 1,839 | 1,245 | 776 | | |
| | \$ 1,841 | \$ 1,300 | \$ 899 | | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

| | Yea | Year Ended December 31, | | |
|--|------------|-------------------------|-----------|--|
| | 2009 | 2008 | 2007 | |
| | | (In thousands) | | |
| OPERATING INCOME: | | | | |
| Power Generation Systems | \$157,869 | \$315,397 | \$234,066 | |
| Government Operations | 154,542 | 150,232 | 122,941 | |
| | \$312,411 | \$465,629 | \$357,007 | |
| Unallocated Corporate ⁽¹⁾ | (42,847) | (21,146) | (19,374) | |
| Total Operating Income ⁽²⁾ | \$269,564 | \$444,483 | \$337,633 | |
| Other Income (Expense): | | | | |
| Interest income | 3,439 | 15,286 | 25,220 | |
| Interest expense | (24,590) | (22,740) | (38,094) | |
| Other expense—net | (16,112) | (4,290) | (10,411) | |
| Total Other Expense | (37,263) | (11,744) | (23,285) | |
| Income before Provision for Income Taxes | 232,301 | 432,739 | 314,348 | |
| Unallocated corporate includes general corporate overhead not allocated to segments. Included in operating income is the following: | | | | |
| Gains (Losses) on Asset Disposal and Impairments—Net: | | | | |
| Power Generation Systems | \$ (272) | \$ 9,606 | \$ (25) | |
| Government Operations | (171) | _ | 1,631 | |
| Unallocated Corporate | (783) | (3) | | |
| | \$ (1,226) | \$ 9,603 | \$ 1,606 | |
| Equity in Income of Investees: | | · | | |
| Power Generation Systems | \$14,043 | \$10,411 | \$14,359 | |
| Government Operations | 41,051 | 41,381 | 31,288 | |
| | \$55,094 | \$51,792 | \$45,647 | |
| | | | | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

| | | Year Ended December 31, | | | | |
|---|----|-------------------------|-----|--------------|----|-----------|
| | _ | 2009 | | 2008 | | 2007 |
| | | | (Iı | n thousands) | | |
| SEGMENT ASSETS: | | | | | | |
| Power Generation Systems | \$ | 1,429,475 | \$ | ,, | \$ | 1,483,704 |
| Government Operations | _ | 945,806 | | 711,628 | | 479,026 |
| Total Segment Assets | | 2,375,281 | | 2,235,116 | | 1,962,730 |
| Corporate Assets | | 228,578 | | 271,725 | | 186,906 |
| Total Assets | \$ | 2,603,859 | \$ | 2,506,841 | \$ | 2,149,636 |
| CAPITAL EXPENDITURES: | | | | | | |
| Power Generation Systems | \$ | 32,148 | \$ | 33,896 | \$ | 40,218 |
| Government Operations | | 45,062 | | 16,348 | | 14,117 |
| Segment Capital Expenditures | | 77,210 | | 50,244 | | 54,335 |
| Corporate Capital Expenditures | | 16,515 | | 12,770 | | 6,374 |
| Total Capital Expenditures | \$ | 93,725 | \$ | 63,014 | \$ | 60,709 |
| DEPRECIATION AND AMORTIZATION: | | | | | | |
| Power Generation Systems | \$ | 17,859 | \$ | 22,080 | \$ | 21,266 |
| Government Operations | | 51,588 | | 22,445 | | 19,269 |
| Segment Depreciation and Amortization | | 69,447 | | 44,525 | | 40,535 |
| Corporate Depreciation and Amortization | | 3,265 | | 1,460 | | 1,137 |
| Total Depreciation and Amortization | \$ | 72,712 | \$ | 45,985 | \$ | 41,672 |
| INVESTMENT IN AFFILIATES: | | | | | | |
| Power Generation Systems | \$ | 64,666 | \$ | 57,701 | \$ | 50,919 |
| Government Operations | _ | 3,661 | | 3,926 | | 3,983 |
| Total Investment in Affiliates | \$ | 68,327 | \$ | 61,627 | \$ | 54,902 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

2. Information about our Product Lines:

| | Y | Year Ended December 31, | | |
|--|--------------|-------------------------|--------------|--|
| | 2009 | 2008 | 2007 | |
| | | (In thousands) | | |
| REVENUES: | | | | |
| Power Generation Systems: | | | | |
| Original Equipment Manufacturers' Operations | 673,920 | 1,185,305 | 1,371,427 | |
| Aftermarket Goods and Services | 778,075 | 974,730 | 829,185 | |
| Nuclear Equipment Operations | 174,785 | 187,312 | 137,864 | |
| Boiler Auxiliary Equipment | 124,363 | 138,192 | 115,855 | |
| Operations and Maintenance | 70,444 | 60,171 | 54,854 | |
| Eliminations/Other | 2,863 | 3,145 | (2,371) | |
| | 1,824,450 | 2,548,855 | 2,506,814 | |
| Government Operations: | | | | |
| Nuclear Component Program | \$ 885,507 | \$ 705,442 | \$ 619,154 | |
| Commercial Operations | 98,189 | 89,857 | 3,853 | |
| Nuclear Environmental Services | 33,702 | 40,352 | 51,703 | |
| Management & Operation Contracts of U.S. Government Facilities | 13,174 | 15,779 | 18,776 | |
| Contract Research | _ | 46 | 1,877 | |
| Other Government Operations | 833 | 821 | 708 | |
| Eliminations | 618 | (1,278) | (2,042) | |
| | 1,032,023 | 851,019 | 694,029 | |
| Eliminations | (1,841) | (1,300) | (899) | |
| | \$ 2,854,632 | \$ 3,398,574 | \$ 3,199,944 | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO COMBINED FINANCIAL STATEMENTS—(Continued)

3. Information about our Operations in Different Geographic Areas:

| | | Year Ended December 31, | | |
|---|--------------|-------------------------|--------------|--|
| | 2009 | 2008 | 2007 | |
| Th W 1700 at | | (In thousands) | | |
| ENUES(3): | | | | |
| United States | \$ 2,300,039 | | \$ 2,746,834 | |
| Canada | 267,567 | / | 227,233 | |
| China | 54,192 | , | 32,595 | |
| Sweden | 40,852 | | 41,754 | |
| Japan | 26,904 | | 3,362 | |
| Denmark | 21,965 | , | 36,382 | |
| Belgium | 2,666 | 20,423 | 17,416 | |
| Germany | 15,734 | 12,893 | 8,815 | |
| Ireland | 10,336 | 1,083 | 251 | |
| Norway | 21,996 | 11,467 | 1,457 | |
| United Kingdom | 6,746 | 10,899 | 7,050 | |
| Indonesia | 8,834 | 7,090 | 25,056 | |
| India | 5,229 | 5,794 | 2,556 | |
| France | 8,351 | 5,323 | 9,595 | |
| Brazil | 2,131 | 4,199 | 4,536 | |
| Other Countries | 61,090 | 56,252 | 35,052 | |
| | \$ 2,854,632 | \$ 3,398,574 | \$ 3,199,944 | |
| We allocate geographic revenues based on the location of the customer's operations. | | | | |
| PROPERTY, PLANT AND EQUIPMENT, NET: | | | | |
| United States | \$ 360,076 | \$ 313,006 | 3 235,274 | |
| Canada | 40,058 | 33,524 | 41,146 | |
| United Kingdom | 11,234 | 10,661 | 14,587 | |
| Denmark | 8,852 | 8,549 | 8,943 | |
| Other Countries | 9,841 | 5,085 | 4,792 | |
| | \$ 430,061 | | 304,742 | |

4. Information about our Major Customers:

In the years ended December 31, 2009, 2008 and 2007, the U.S. Government accounted for approximately 33%, 22% and 21%, respectively, of our total revenues. We have included these revenues in our Government Operations segment.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. CONDENSED COMBINED BALANCE SHEETS (UNAUDITED)

| | March 31, 2010 (In tho | December 31, 2009 |
|---|------------------------------|----------------------|
| ASSETS | , | |
| Current Assets: | | |
| Cash and cash equivalents | \$ 401,124 | \$ 469,468 |
| Restricted cash and cash equivalents (Note 1) | 17,821 | 15,305 |
| Investments | 14 | 14 |
| Accounts receivable—trade, net | 302,506 | 319,861 |
| Accounts receivable—other | 42,496 | 39,289 |
| Contracts in progress | 299,166 | 245,998 |
| Inventories (Note 1) | 95,009 | 98,644 |
| Deferred income taxes | 99,109 | 96,680 |
| Other current assets | 25,259 | 21,456 |
| Total Current Assets | 1,282,504 | 1,306,715 |
| Property, Plant and Equipment | 953,715 | 945,298 |
| Less accumulated depreciation | 522,504 | 515,237 |
| Net Property, Plant and Equipment | 431,211 | 430,061 |
| Investments | 70,909 | 73,540 |
| Goodwill (Note 1) | 269,950 | 262,866 |
| Deferred Income Taxes | 265,048 | 270,002 |
| Investments in Unconsolidated Affiliates | 77,965 | 68,327 |
| Note Receivable from Affiliate | 43,835 | 42,573 |
| Other Assets | 147,347 | 149,775 |
| TOTAL | \$ 2,588,769 | \$ 2,603,859 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. CONDENSED COMBINED BALANCE SHEETS (UNAUDITED)

| | March 31, 2010 | December 31, 2009 |
|--|-------------------|----------------------|
| LIABILITIES AND PARENT EQUITY | (111 till) | usanus) |
| Current Liabilities: | | |
| Notes payable and current maturities of long-term debt | \$ 4,993 | \$ 6,432 |
| Accounts payable | 183,360 | 178,350 |
| Accounts payable to McDermott International, Inc. | 142,967 | 112,053 |
| Accrued employee benefits | 299,026 | 198,195 |
| Accrued liabilities—other | 85,468 | 74,700 |
| Advance billings on contracts | 494,579 | 537,448 |
| Accrued warranty expense | 119,531 | 115,055 |
| Income taxes payable | 4,020 | 12,943 |
| Total Current Liabilities | 1,333,944 | 1,235,176 |
| Long-Term Debt | 5,343 | 4,222 |
| Accumulated Postretirement Benefit Obligation | 106,329 | 105,484 |
| Environmental Liabilities | 48,523 | 47,795 |
| Pension Liability | 550,562 | 699,117 |
| Notes Payable to Affiliate | 320,588 | 320,568 |
| Other Liabilities | 69,841 | 70,791 |
| Commitments and Contingencies | | |
| Total Parent Equity | 153,639 | 120,706 |
| TOTAL | \$ 2,588,769 | \$ 2,603,859 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. CONDENSED COMBINED STATEMENTS OF INCOME (UNAUDITED)

| | Three Mor Mare | oths Ended |
|---|-------------------|------------|
| | 2010 | 2009 |
| December | | usands) |
| Revenues | \$662,388 | \$785,053 |
| Costs and Expenses: | | |
| Cost of operations | 544,951 | 610,914 |
| Selling, general and administrative expenses | 92,723 | 88,969 |
| (Gains) losses on asset disposals and impairments—net | (13) | 207 |
| Total Costs and Expenses | 637,661 | 700,090 |
| Equity in Income of Investees | 14,019 | 10,345 |
| Operating Income | 38,746 | 95,308 |
| Other Income (Expense): | | |
| Interest income | 449 | 1,239 |
| Interest expense | (5,993) | (5,261) |
| Other expense—net | (3,090) | (2,040) |
| Total Other Expense | (8,634) | (6,062) |
| Income before Provision for Income Taxes | 30,112 | 89,246 |
| Provision for Income Taxes | 13,256 | 35,710 |
| Net Income | 16,856 | 53,536 |
| Less: Net Income attributable to Noncontrolling Interest | (13) | (52) |
| Net Income Attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | <u>\$ 16,843</u> | \$ 53,484 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. CONDENSED COMBINED STATEMENTS OF COMPREHENSIVE INCOME (UNAUDITED)

| | Three Mon Marc | |
|---|-------------------|----------|
| | 2010 | 2009 |
| | (In tho | |
| Net Income | \$16,856 | \$53,536 |
| Other Comprehensive Income (Loss): | | |
| Currency translation adjustments: | | |
| Foreign currency translation adjustments | 728 | (4,676) |
| Unrealized (losses) gains on derivative financial instruments: | | |
| Unrealized gains (losses) on derivative financial instruments | 947 | (1,545) |
| Realized gains (losses) on derivative financial instruments | 609 | (876) |
| Amortization of benefit plan costs | 13,629 | 13,055 |
| Unrealized gains (losses) on investments: | | |
| Unrealized gains (losses) arising during the period, net of taxes | 16 | (359) |
| Realized gains recognized during the period | 115 | |
| Other Comprehensive Income | 16,044 | 5,599 |
| Total Comprehensive Income | 32,900 | 59,135 |
| Comprehensive Income attributable to Noncontrolling Interest | (16) | (43) |
| Comprehensive Income Attributable to The Babcock & Wilcox Operations of McDermott International, Inc. | \$32,844 | \$59,092 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. CONDENSED COMBINED STATEMENTS OF PARENT EQUITY (DEFICIT) (UNAUDITED) (In thousands)

| | Accumulated Other Comprehensive Loss | Parent Equity | Non- Controlling Interest | Total Parent Equity (Deficit) |
|--|---|------------------|---------------------------------|-------------------------------------|
| Balance December 31, 2009 | \$ (546,574) | \$666,777 | \$ 503 | \$ 120,706 |
| Net income | _ | 16,843 | 13 | 16,856 |
| Amortization of benefit plan costs | 13,629 | _ | _ | 13,629 |
| Unrealized gains on investments | 131 | _ | _ | 131 |
| Foreign currency translation adjustments | 725 | _ | 3 | 728 |
| Unrealized gain on derivatives | 1,556 | _ | _ | 1,556 |
| Stock-based compensation charges | | 33 | | 33 |
| Balance March 31, 2010 | \$ (530,533) | \$683,653 | \$ 519 | \$ 153,639 |
| Balance December 31, 2008 | \$ (592,998) | \$523,090 | \$ 360 | \$ (69,548) |
| Net income | _ | 49,584 | 52 | 53,536 |
| Amortization of benefit plan costs | 13,055 | _ | _ | 13,055 |
| Unrealized losses of investments | (359) | _ | _ | (359) |
| Foreign currency translation adjustments | (4,667) | _ | (9) | (4,676) |
| Unrealized loss on derivatives | (2,421) | _ | _ | (2,421) |
| Stock-based compensation charges | | (193) | | (193) |
| Balance March 31, 2009 | \$ (587,390) | \$572,481 | \$ 403 | \$ (10,606) |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. CONDENSED COMBINED STATEMENTS OF CASH FLOWS (UNAUDITED)

| Inventories 2,372 14,556 | | Three Mor | |
|--|--|------------------|-----------|
| CASH FLOWS FROM OPERATING ACTIVITIES \$ 16,856 \$ 53,536 Non-cash items included in net income: Depreciation and amortization 16,812 16,263 Income of investees, net of dividends (8,758) (2,287) (Gains) losses on asset disposals and impairments—net (13) 207 Excess tax benefits from ASC 718 stock-based compensation (1,908) (123) Amortization of pension and postretirement costs 21,420 20,602 Other, net 20,002 20,002 Changes in assets and liabilities, net of effects from acquisition and divestitures: Texas asset and liabilities, net of effects from acquisition and divestitures: 36,336 33,867 Income tax receivable 15,624 34,724 Accounts payable and other 36,336 (33,867) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) 42,488 Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) | | | |
| Net Income \$ 16,856 \$ 53,536 Non-cash items included in net income: 50,000 50,000 10,000 | A CONTRACTOR OF THE CONTRACTOR | (In tho | isands) |
| Non-cash items included in net income: Depreciation and amortization 16,812 16,263 Income of investees, net of dividends (8,758) (2,287 Gains) losses on asset disposals and impairments—net (13) 207 Excess tax benefits from ASC 718 stock-based compensation (1,908) (123) Amortization of pension and postretirement costs 21,420 20,602 Other, net 723 (1,027 Changes in assets and liabilities, net of effects from acquisition and divestitures: Accounts receivable 15,624 34,724 Accounts payable and other 36,836 (33,867 Income tax receivable (313) (1,140 Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575 43,230 Income taxes (12,053) (2,488 Accrued and other current liabilities Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548 Other, net (2,533 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,532 40,495 CASH FLOWS FROM INVESTING ACTIVITIES (2,516 (8,098 Purchases of property, plant and equipment (18,763 (20,029 Acquisition of businesses, net of cash acquired (8,165 — | | ф 16.056 | Ø 52 52 6 |
| Depreciation and amortization 16,812 16,263 Income of investees, net of dividends (8,758) (2,287 (Gains) losses on asset disposals and impairments—net (13) 207 Excess tax benefits from ASC 718 stock-based compensation (1,908) (12,33) Amortization of pension and postretirement costs 21,420 20,602 Other, net 723 (1,027) Changes in assets and liabilities, net of effects from acquisition and divestitures: 15,624 34,724 Accounts receivable 15,624 34,724 Accounts payable and other 36,836 (33,867) Income tax receivable (31)3 (1,140) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 12,307 12,307 12,307 | | <u>\$ 16,856</u> | \$ 53,536 |
| Income of investees, net of dividends (8,758) (2,287) (Gains) losses on asset disposals and impairments—net (13) 207 Excess tax benefits from ASC 718 stock-based compensation (1,908) (123) Amortization of pension and postretirement costs 21,420 20,602 Other, net 723 (1,027) Changes in assets and liabilities, net of effects from acquisition and divestitures: 36,836 33,867 Accounts receivable 15,624 34,724 Accounts payable and other 36,836 (33,867) Income tax receivable (313) (1,140) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (2,516) (8,098) Purchases of property, plant and equi | | 46040 | 4 5 0 50 |
| (Gains) losses on asset disposals and impairments—net (13) 207 Excess tax benefits from ASC 718 stock-based compensation (1,908) (123) Amortization of pension and postretirement costs 20,602 Other, net 723 (1,027) Changes in assets and liabilities, net of effects from acquisition and divestitures: | | | |
| Excess tax benefits from ASC 718 stock-based compensation (1,908) (123) Amortization of pension and postretirement costs 21,420 20,602 Other, net 723 (1,027) Changes in assets and liabilities, net of effects from acquisition and divestitures: The counts receivable 15,624 34,724 Accounts payable and other 36,836 (33,867) (313) (1,140) Income tax receivable (313) (1,140) (1,2053) (2,148) Net contracts in progress and advance billings (96,575) (43,230) (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 19,010 (46,548) Other, net (2,533) 12,307 (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,548) 40,495 CASH FLOWS FROM INVESTING ACTIVITIES: (2,516) (8,098) Purchases of property, plant and equipment (18,763) (20,029) Acquisition of businesses, net of cash acquired (8,165) — | | | |
| Amortization of pension and postretirement costs 21,420 20,602 Other, net 723 (1,027) Changes in assets and liabilities, net of effects from acquisition and divestitures: | | | |
| Other, net 723 (1,027) Changes in assets and liabilities, net of effects from acquisition and divestitures: 36,824 34,724 Accounts receivable 36,836 (33,867) Income tax receivable (313) (1,140) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,532) 40,495 CASH FLOWS FROM INVESTING ACTIVITIES: (2,516) (8,098) Purchases of property, plant and equivalents (2,516) (8,098) Purchases of property, plant and equipment (18,763) (20,029) Acquisition of businesses, net of cash acquired (8,165) — | | | (/ |
| Changes in assets and liabilities, net of effects from acquisition and divestitures: Accounts receivable 15,624 34,724 Accounts payable and other 36,836 (33,867) Income tax receivable (313) (1,140) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,532) 40,495 CASH FLOWS FROM INVESTING ACTIVITIES: (2,516) (8,098) Purchases of property, plant and equipment (18,763) (20,029) Acquisition of businesses, net of cash acquired (8,165) — | | | , |
| Accounts receivable 15,624 34,724 Accounts payable and other 36,836 (33,867) Income tax receivable (313) (1,140) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,532) 40,495 CASH FLOWS FROM INVESTING ACTIVITIES: (2,516) (8,098) Purchases of property, plant and equipment (2,516) (8,098) Acquisition of businesses, net of cash acquired (8,165) — | | 123 | (1,027) |
| Accounts payable and other 36,836 (33,867) Income tax receivable (313) (1,140) Inventories 2,372 14,556 Net contracts in progress and advance billings (96,575) (43,230) Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,532) 40,495 CASH FLOWS FROM INVESTING ACTIVITIES: (2,516) (8,098) Increase in restricted cash and cash equivalents (2,516) (8,098) Purchases of property, plant and equipment (8,165) — Acquisition of businesses, net of cash acquired (8,165) — | | 15 624 | 34 724 |
| Income tax receivable | | | / |
| Inventories 2,372 14,556 | | | (1,140) |
| Income taxes (12,053) (2,488) Accrued and other current liabilities 13,117 19,010 Pension liability, accumulated postretirement benefit obligation and accrued employee benefits (47,139) (46,548) Other, net (2,533) 12,307 NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES (45,532) 40,495 CASH FLOWS FROM INVESTING ACTIVITIES: Increase in restricted cash and cash equivalents (2,516) (8,098) Purchases of property, plant and equipment (18,763) (20,029) Acquisition of businesses, net of cash acquired (8,165) — | Inventories | | |
| Accrued and other current liabilities Pension liability, accumulated postretirement benefit obligation and accrued employee benefits Other, net NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES CASH FLOWS FROM INVESTING ACTIVITIES: Increase in restricted cash and cash equivalents Purchases of property, plant and equipment Acquisition of businesses, net of cash acquired 13,117 19,010 (46,548) (47,139) (46,548) (2,533) 12,307 (45,532) 40,495 (28,098) (2,516) (8,098) (18,763) (20,029) (8,165) | Net contracts in progress and advance billings | (96,575) | (43,230) |
| Pension liability, accumulated postretirement benefit obligation and accrued employee benefits Other, net NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES CASH FLOWS FROM INVESTING ACTIVITIES: Increase in restricted cash and cash equivalents Purchases of property, plant and equipment Acquisition of businesses, net of cash acquired (8,165) (46,548) (2,533) (24,495) (45,532) (46,548) (47,139) (47,139) (47,13 | Income taxes | (12,053) | (2,488) |
| Other, net NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES CASH FLOWS FROM INVESTING ACTIVITIES: Increase in restricted cash and cash equivalents Purchases of property, plant and equipment Acquisition of businesses, net of cash acquired (8,165) (2,533) (4,532) (40,495) (8,098) (18,763) (20,029) | | | |
| NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES CASH FLOWS FROM INVESTING ACTIVITIES: Increase in restricted cash and cash equivalents Purchases of property, plant and equipment Acquisition of businesses, net of cash acquired (8,165) (40,495) (8,098) (18,763) (20,029) | Pension liability, accumulated postretirement benefit obligation and accrued employee benefits | (/ / | (46,548) |
| CASH FLOWS FROM INVESTING ACTIVITIES: Increase in restricted cash and cash equivalents Purchases of property, plant and equipment Acquisition of businesses, net of cash acquired (8,165) (8098) (8,065) | Other, net | (2,533) | 12,307 |
| Increase in restricted cash and cash equivalents (2,516) (8,098) Purchases of property, plant and equipment (18,763) (20,029) Acquisition of businesses, net of cash acquired (8,165) — | NET CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES | _(45,532) | 40,495 |
| Purchases of property, plant and equipment (18,763) (20,029) Acquisition of businesses, net of cash acquired (8,165) — | CASH FLOWS FROM INVESTING ACTIVITIES: | | |
| Acquisition of businesses, net of cash acquired (8,165) — | Increase in restricted cash and cash equivalents | (2,516) | (8,098) |
| | | (18,763) | (20,029) |
| Decrease in available for sale securities 2.730 27.005 | | (8,165) | _ |
| | Decrease in available-for-sale securities | 2,739 | 27,095 |
| Proceeds from asset disposals 79 25 | 1 | 79 | |
| Other (39) | Other | | (39) |
| NET CASH USED IN INVESTING ACTIVITIES (26,626) (1,046) | NET CASH USED IN INVESTING ACTIVITIES | (26,626) | (1,046) |
| CASH FLOWS FROM FINANCING ACTIVITIES: | CASH FLOWS FROM FINANCING ACTIVITIES: | | |
| Payment of short-term borrowings and long-term debt (150) (4,825) | Payment of short-term borrowings and long-term debt | (150) | (4,825) |
| Increase in note payable to affiliate 20 — | | 20 | _ |
| Excess tax benefits from ASC 718 stock-based compensation 1,908 123 | | 1,908 | |
| | | | (32) |
| NET CASH PROVIDED BY (USED IN) FINANCING ACTIVITIES 1,778 (4,734) | NET CASH PROVIDED BY (USED IN) FINANCING ACTIVITIES | 1,778 | (4,734) |
| EFFECTS OF EXCHANGE RATE CHANGES ON CASH 2,036 (6,050) | EFFECTS OF EXCHANGE RATE CHANGES ON CASH | 2,036 | (6,050) |
| NET INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS (68,344) 28,655 | NET INCREASE (DECREASE) IN CASH AND CASH EQUIVALENTS | (68,344) | 28,655 |
| CASH AND CASH EQUIVALENTS AT BEGINNING OF PERIOD 469,468 279,646 | CASH AND CASH EOUIVALENTS AT BEGINNING OF PERIOD | 469,468 | 279,646 |
| CASH AND CASH EQUIVALENTS AT END OF PERIOD \$401,124 \$308,311 | | | |
| SUPPLEMENTAL DISCLOSURES OF CASH FLOW INFORMATION: | SUPPLEMENTAL DISCLOSURES OF CASH FLOW INFORMATION: | | |
| Cash paid (received) during the period for: | | | |
| Interest (net of amount capitalized) \$ 5,993 \$ 5,380 | | \$ 5,993 | \$ 5,380 |
| Income taxes (net of refunds) \$ 5,502 \$ 8,103 | | \$ 5,502 | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)

NOTE 1—SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The Company, Basis of Presentation and Principles of Combination

The Babcock & Wilcox Operations of McDermott International, Inc. (the "Company") represent a combined reporting entity comprised of the assets and liabilities used in managing and operating the Power Generation Systems and Government Operations segments of McDermott International, Inc. ("MII") in addition to two captive insurance companies which will be combined and contributed to The Babcock & Wilcox Company ("B&W") in conjunction with the planned spin-off of B&W by MII. On December 7, 2009, MII announced a plan to separate B&W from MII through a spin-off transaction that is intended to be tax-free to MII.

The accompanying condensed combined financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America. We have presented our condensed combined financial statements in U.S. Dollars in accordance with the interim reporting requirements of Form 10 and Rule 10-01 of Regulation S-X. We have included all adjustments, consisting only of normal recurring adjustments, which are, in the opinion of management, necessary for a fair presentation. Financial information and disclosures normally included in our financial statements prepared annually in accordance with accounting principles generally accepted in the United States of America have been condensed or omitted. Readers of these financial statements should, therefore, refer to the combined financial statements for the years ended December 31, 2009, 2008 and 2007 included in this information statement. All intercompany balances and transactions have been eliminated in the condensed combined financial statements. We use the equity method to account for investments in entities that we do not control, but over which we have significant influence. We generally refer to these entities as "joint ventures." We present the notes to our condensed combined financial statements on the basis of continuing operations, unless otherwise stated.

The combined results of operations, financial position and cash flows reflected in the accompanying condensed combined financial statements may not be indicative of the future performance of the combined businesses and do not necessarily reflect what its combined results of operations, financial position and cash flows would have been had the combined businesses operated as an independent public company during the periods presented, including changes in its operations and capitalization as a result of the separation and distribution from MII.

Certain corporate and general and administrative expenses, including those related to executive management, tax, accounting, legal and treasury services, and certain employee benefits, have been allocated based on a level of effort calculation. Our management believes such allocations are reasonable. However, the associated expenses reflected in the accompanying combined statements of income may not be indicative of the actual expenses that would have been incurred had the combined businesses been operating as an independent public company for the periods presented. Following the separation and distribution from MII, B&W will perform these functions using internal resources or services provided by third parties, certain of which may be provided by MII during a transitional period pursuant to a transition services agreement.

In presenting the condensed combined financial statements, our management makes estimates and assumptions that affect the amounts reported and related disclosures. Estimates, by their nature, are based on judgment and available information. Accordingly, actual results could differ from those estimates.

As used in these footnote disclosures, the following terms have the meanings set forth below:

- MII (or McDermott) means McDermott International, Inc., a Panamanian corporation;
- BHI means Babcock & Wilcox Holdings, Inc., a Delaware corporation;
- MI means McDermott Incorporated, a Delaware corporation and a subsidiary of BHI, and its consolidated subsidiaries;

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

- BWICO means Babcock & Wilcox Investment Company, a Delaware corporation and a subsidiary of MI, and its consolidated subsidiaries;
- · BWXT means BWX Technologies, Inc., a Delaware corporation and a subsidiary of BWICO, and its consolidated subsidiaries;
- B&W PGG means Babcock & Wilcox Power Generation Group, Inc., a Delaware corporation and a subsidiary of BWICO, and its consolidated subsidiaries;
- · B&W NE means Babcock & Wilcox Nuclear Energy. Inc., a Delaware corporation and a subsidiary of BWICO, and its consolidated subsidiaries; and
- JRM means J. Ray McDermott, S.A., a Panamanian corporation and a subsidiary of MII, and its consolidated subsidiaries.

Unless the context otherwise indicates, "we," "us" and "our" mean The Babcock & Wilcox Operations of McDermott International, Inc.

We operate in two business segments: Power Generation Systems; and Government Operations.

Our Power Generation Systems segment includes the business and operations of B&W PGG and B&W NE. Through this segment, we supply boilers fired with fossil fuels, such as coal, oil and natural gas, or renewable fuels such as biomass, municipal solid waste and concentrated solar energy. In addition, we supply commercial nuclear steam generators and components, environmental equipment and components, and related services to customers in different regions around the world. We design, engineer, manufacture, supply, construct and service large utility and industrial power generation systems, including boilers used to generate steam in electric power plants, pulp and paper making, chemical and process applications and other industrial uses.

Our Government Operations segment includes the business and operations of Babcock & Wilcox Nuclear Operations Group, Inc., Babcock & Wilcox Technical Services Group, Inc. and their respective subsidiaries. Through this segment, we manufacture nuclear components and provide various services to the U.S. Government, including uranium processing, environmental site restoration services, and management and operating services for various U.S. Government-owned facilities. These services are provided to the DOE, including the National Nuclear Security Administration, the Office of Nuclear Energy, the Office of Science and the Office of Environmental Management.

Comprehensive Loss

The components of accumulated other comprehensive loss included in parent equity (deficit) are as follows:

| | March 31, 2010 | December 31, 2009 |
|---|-------------------|----------------------|
| | (In thous | sands) |
| Net unrealized loss on investments | \$ (244) | \$ (375) |
| Currency translation adjustments | 32,453 | 31,728 |
| Net unrealized loss on derivative financial instruments | (4,202) | (5,758) |
| Unrecognized losses on benefit obligations | (558,540) | (572,169) |
| Accumulated Other Comprehensive Loss | \$(530,533) | \$ (546,574) |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

Inventories

The components of inventories are as follows:

| | March 31, 2010 | December 31, 2009 |
|----------------------------|-------------------|----------------------|
| | | ousands) |
| Raw Materials and Supplies | \$64,308 | \$ 71,207 |
| Work in Progress | 7,925 | 6,381 |
| Finished Goods | 22,776 | 21,056 |
| Total Inventories | \$95,009 | \$ 98,644 |

Restricted Cash and Cash Equivalents

At March 31, 2010, we had restricted cash and cash equivalents totaling \$17.8 million, \$9.7 million of which was held in restricted foreign accounts, \$3.8 million of which was held as cash collateral for letters of credit, \$4.0 million of which was held for future decommissioning of facilities, and \$0.3 million of which was held to meet reinsurance reserve requirements of our captive insurance companies.

Warranty Expense

We accrue estimated expense to satisfy contractual warranty requirements when we recognize the associated revenue on the related contracts. In addition, we make specific provisions where we expect the actual warranty costs to significantly exceed the accrued estimates. Such provisions could have a material effect on our combined financial condition, results of operations and cash flows.

The following summarizes the changes in the carrying amount of accrued warranty:

| | Three Mon | ths Ended |
|--------------------------------|-----------|-----------|
| | Marc | h 31, |
| | 2010 | 2009 |
| | (Unau | dited) |
| | (In thou | isands) |
| Balance at beginning of period | \$115,055 | \$119,722 |
| Additions and adjustments | 7,372 | 2,016 |
| Charges | (2,896) | (2,530) |
| Balance at end of period | \$119,531 | \$119,208 |

Research and Development

Research and development activities are related to the development and improvement of new and existing products and equipment as well as conceptual and engineering evaluation for translation into practical applications. We charge to cost of operations the costs of research and development unrelated to specific contracts as incurred. For the three months ended March 31, 2010 and 2009, our net research and development expense included in cost of operations totaled \$17.1 million and \$10.0 million, respectively. Substantially all of these costs are in our Power Generation Systems segment and include costs related to the development of carbon capture and sequestration technologies and our modular and scalable reactor, B&W mPowerTM.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

Recently Adopted Accounting Standards

In January 2010, the Financial Accounting Standards Board ("FASB") issued a revision to the topic Fair Value Measurements and Disclosures. This revision adds new fair value disclosures for certain transfers of investments between Level 1 and Level 2 measurements and clarifies existing disclosures regarding valuation techniques. On January 1, 2010, we adopted this revision. The adoption of this revision did not have an impact on our condensed combined financial statements.

In January 2010, the FASB issued a revision to the topic Consolidation. This revision clarifies the scope of partial sale and deconsolidation provisions related to acquisitions and noncontrolling interests. On January 1, 2010, we adopted this revision. The adoption of this revision did not have an impact on our condensed combined financial statements.

In June 2009, the FASB issued a revision to the topic Consolidation. This revision was subsequently amended in December 2009 and February 2010. These revisions expand the scope of this topic and amend guidance for assessing and analyzing variable interest entities. These revisions were effective on January 1, 2010 and had no impact on our condensed combined financial statements.

New Accounting Standards

In January 2010, the FASB issued a revision to the topic Fair Value Measurements and Disclosures. This revision sets forth new rules on providing enhanced information for Level 3 measurements. The disclosure provisions under this revision will be effective for fiscal years beginning after December 15, 2010, in both interim and annual disclosures. The adoption of this revision is not expected to have an impact on our condensed combined financial statements.

NOTE 2—BUSINESS ACQUISITIONS

In January 2010, B&W PGG acquired all the outstanding equity interests in Götaverken Miljö AB, a flue gas cleaning and energy recovery company based in Gothenburg, Sweden. In connection with this acquisition, we recorded goodwill of approximately \$7.8 million. We accounted for this acquisition as a business combination.

NOTE 3—PENSION PLANS AND POSTRETIREMENT BENEFITS

Components of net periodic benefit cost included in net income are as follows:

| | Pension Three Mon Marc | ths Ended | Other I Three Mor Marc | ths Ended |
|---------------------------------------|------------------------------|-----------|------------------------------|-----------|
| | 2010 | 2009 | 2010 | 2009 |
| | · <u></u> | (In thous | ands) | |
| Service cost | \$ 9,627 | \$ 9,020 | \$ 267 | \$ 231 |
| Interest cost | 37,342 | 36,745 | 1,963 | 2,168 |
| Expected return on plan assets | (35,566) | (33,826) | (411) | (376) |
| Amortization of prior service cost | 894 | 686 | 18 | 15 |
| Amortization of transition obligation | _ | _ | 70 | 59 |
| Recognized net actuarial loss | 20,098 | 19,438 | 341 | 403 |
| Net periodic benefit cost | \$ 32,395 | \$ 32,063 | \$ 2,248 | \$ 2,500 |

Included in accrued employee benefits on our condensed combined balance sheet at March 31, 2010 is \$209.3 million of current liability related to our pension plans, which we expect to fund within the next 12 months.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

NOTE 4—COMMITMENTS AND CONTINGENCIES

On March 19, 2010, Jessi Ann Casella, William E. Uhing and Kenneth B. Cyphert filed suit against B&W PGG, Babcock & Wilcox Technical Services Group, Inc., formerly known as B&W Nuclear Environmental Services, Inc. and Atlantic Richfield Company in the United States District Court for the Western District of Pennsylvania (the "Casella Litigation"). The plaintiffs in the Casella Litigation allege, among other things, that they suffered personal injuries and property damage as a result of alleged radioactive and non-radioactive releases relating to the operation, remediation and/or decommissioning of the same two former nuclear fuel processing facilities located in Apollo and Parks Township, Pennsylvania involved in the McMunn Litigation described in our combined financial statements for the years ended December 31, 2009, 2008 and 2007.

NOTE 5—DERIVATIVE FINANCIAL INSTRUMENTS

Our worldwide operations give rise to exposure to market risks from changes in foreign exchange rates. We use derivative financial instruments (primarily foreign currency forward-exchange contracts) to reduce the impact of changes in foreign exchange rates on our operating results. We use these instruments primarily to hedge our exposure associated with revenues or costs on our long-term contracts and other cash flow exposures that are denominated in currencies other than our operating entities' functional currencies. We do not hold or issue financial instruments for trading or other speculative purposes.

We enter into derivative financial instruments primarily as hedges of certain firm purchase and sale commitments denominated in foreign currencies. We record these contracts at fair value on our condensed combined balance sheets. Depending on the hedge designation at the inception of the contract, the related gains and losses on these contracts are either deferred in parent equity (deficit) as a component of accumulated other comprehensive loss until the hedged item is recognized in earnings, or offset against the change in fair value of the hedged firm commitment through earnings. The ineffective portion of a derivative's change in fair value and any portion excluded from the assessment of effectiveness are immediately recognized in earnings. The gain or loss on a derivative instrument not designated as a hedging instrument is also immediately recognized in earnings. Gains and losses on derivative financial instruments that require immediate recognition are included as a component of other income (expense)—net in our condensed combined statements of income.

We have designated all of our forward contracts as cash flow hedges. The hedged risk is the risk of changes in functional-currency-equivalent cash flows attributable to changes in spot exchange rates of forecasted transactions related to long-term contracts and certain capital expenditures. We exclude from our assessment of effectiveness the portion of the fair value of the forward contracts attributable to the difference between spot exchange rates and forward exchange rates. Ineffective portions of our forward contracts are recorded in other income (expense)—net on our combined statements of income. At March 31, 2010, we had deferred approximately \$4.2 million of net losses on these derivative financial instruments in accumulated other comprehensive loss. We expect to recognize substantially all of this amount in the next 12 months.

At March 31, 2010, all of our derivative financial instruments consisted of foreign currency forward-exchange contracts and foreign currency options. The notional value of our forward contracts totaled \$143.2 million at March 31, 2010, with maturities extending to June 2013. These instruments consist primarily of contracts to purchase or sell Euros or Canadian Dollars. The fair value of these contracts was in a net liability position totaling \$1.0 million at March 31, 2010 and was Level 2 in nature. The fair value of our foreign currency options totaled \$4.8 million at March 31, 2010 and was included in other current assets on our combined balance sheets. The foreign currency options were Level 2 in nature (See Note 6).

We are exposed to credit-related losses in the event of nonperformance by counterparties to derivative financial instruments. However, when possible, we enter into International Swaps and Derivative Association,

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

Inc. agreements with our hedge counterparties to mitigate this risk. We also attempt to mitigate this risk by using major financial institutions with high credit ratings and limit our exposure to hedge counterparties based on their credit ratings. The counterparties to all of our derivative financial instruments are financial institutions included in our credit facilities described in Note 6 to our combined financial statements for the years ended December 31, 2009, 2008 and 2007. Our hedge counterparties have the benefit of the same collateral arrangements and covenants as those described with respect to our credit facilities.

The following tables summarize our derivative financial instruments:

| | Asset and | Liability Derivatives |
|-----------------------------------|-------------------|-----------------------|
| | March 31, 2010 | December 31, 2009 |
| | (In | thousands) |
| Derivatives Designated as Hedges: | | |
| Foreign Exchange Contracts: | | |
| Location | | |
| Accounts receivable-other | \$ 668 | \$ 1,582 |
| Accounts payable | \$ 1,597 | \$ 3,350 |

The Effects of Derivative Instruments on our Financial Statements

Three Months

| | Ended M | farch 31, |
|--|-----------|-----------|
| | 2010 | 2009 |
| | (In tho | usands) |
| Derivatives Designated as Hedges: | | |
| Cash Flow Hedges: | | |
| Foreign Exchange Contracts: | | |
| Amount of gain (loss) recognized in other comprehensive income | \$ 1,226 | \$(2,562) |
| Loss reclassified from accumulated other comprehensive loss into income: effective portion | | |
| Location | | |
| Revenues | \$ 459 | \$ 577 |
| Cost of operations | \$ 154 | \$(1,871) |
| Other-net | \$ 331 | \$ 37 |
| Gain (loss) recognized in income: portion excluded from effectiveness testing | | |
| <u>Location</u> | | |
| Other-net | \$(1,564) | \$(1,081) |
| Derivatives Not Designated as Hedges: | | |
| Currency Options: | | |
| Gain (loss) recognized in income | | |
| Location | | |
| Other-net | \$ 34 | \$ — |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

NOTE 6—FAIR VALUES OF FINANCIAL INSTRUMENTS

The following is a summary of our available-for-sale securities measured at fair value at March 31, 2010:

| | 03/31/10 | Level 1 | Level 2 | Level 3 |
|---|-----------|-----------|----------|---------|
| | | (In thous | ands) | |
| Certificates of deposit | \$ 461 | \$ — | \$ 461 | \$ — |
| U.S. Government and agency securities | 65,581 | 65,380 | 201 | |
| Asset-backed/mortgage-backed securities | 570 | _ | 570 | |
| Corporate notes and bonds | 4,311 | | 4,311 | |
| Total | \$ 70,923 | \$ 65,380 | \$ 5,543 | \$ — |
| | | | | |

The following is a summary of our available-for-sale securities measured at fair value at December 31, 2009:

| | 12/31/09 | Level 1 | Level 2 | Level 3 |
|---|-------------------|-----------|-----------|---------------------------------------|
| | | (In thou | isands) | · · · · · · · · · · · · · · · · · · · |
| Certificates of deposit | \$ 463 | \$ — | \$ 463 | \$ — |
| U.S. Government and agency securities | 64,037 | 63,005 | 1,032 | _ |
| Asset-backed/mortgage-backed securities | 740 | _ | 572 | 168 |
| Corporate notes and bonds | 8,314 | | 8,314 | |
| Total | \$ 73,55 <u>4</u> | \$ 63,005 | \$ 10,381 | \$ 168 |

Changes in Level 3 Instrument

The following is a summary of the changes in our Level 3 instrument measured on a recurring basis for the three months ended March 31, 2010 (in thousands):

| | | rch 31, 2010 |
|---|----|-----------------|
| Balance at beginning of period | \$ | 168 |
| Total realized and unrealized gains (losses): | | 119 |
| Included in other income (expense) | | _ |
| Included in other comprehensive income | | 119 |
| Purchases, issuances, and settlements | | (283) |
| Principal repayments | _ | (4) |
| Balance at end of period | \$ | |

Other Financial Instruments

We used the following methods and assumptions in estimating our fair value disclosures for financial instruments:

Cash and cash equivalents and restricted cash and cash equivalents: The carrying amounts that we have reported in the accompanying combined balance sheets for cash and cash equivalents approximate their fair values.

Investments: We estimate the fair values of investments based on quoted market prices. For investments for which there are no quoted market prices, we derive fair values from available yield curves for investments of similar quality and terms.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

Long- and short-term debt: We base the fair values of debt instruments on quoted market prices. Where quoted prices are not available, we base the fair values on the present value of future cash flows discounted at estimated borrowing rates for similar debt instruments or on estimated prices based on current yields for debt issues of similar quality and terms.

The estimated fair values of our financial instruments are as follows:

| | March 3 | March 31, 2010 | | December 31, 2009 | |
|--------------------------------------|------------|----------------|------------|-------------------|--|
| | Carrying | | Carrying | <u> </u> | |
| | Amount | Fair Value | Amount | Fair Value | |
| | · <u> </u> | (In thousands) | | | |
| Balance Sheet Instruments | | | | | |
| Cash and cash equivalents | \$401,124 | \$401,124 | \$469,468 | \$469,468 | |
| Restricted cash and cash equivalents | \$ 17,821 | \$ 17,821 | \$ 15,305 | \$ 15,305 | |
| Investments | \$ 70,923 | \$ 70,923 | \$ 73,554 | \$ 73,554 | |
| Debt | \$ 10,336 | \$ 10,413 | \$ 10,654 | \$ 10,782 | |
| Forward contracts | \$ (929) | \$ (929) | \$ (1,769) | \$ (1,769) | |
| Foreign currency options | \$ 4,782 | \$ 4,782 | \$ 4,747 | \$ 4,747 | |

NOTE 7—STOCK-BASED COMPENSATION

| | | pensation xpense | Tax <u>Benefit</u> (In thousands) onths Ended March 31. | Net Impact |
|--------------------------------------|----|-----------------------------------|---|------------|
| Stock Options | \$ | 448 | \$ (162) | \$ 286 |
| Restricted Stock | Ψ | 551 | (202) | 349 |
| Performance Shares | | 2,284 | (828) | 1,456 |
| Performance and Deferred Stock Units | | 541 | (193) | 348 |
| Total | \$ | 3,824 | \$(1,385) | \$ 2,439 |
| | | Three Months Ended March 31, 2009 | | |
| Stock Options | \$ | 179 | \$ (64) | \$ 115 |
| Restricted Stock | | 757 | (274) | 483 |
| Performance Shares | | 4,598 | (1,677) | 2,921 |
| Performance and Deferred Stock Units | | 406 | (144) | 262 |
| Total | \$ | 5,940 | <u>\$(2,159)</u> | \$ 3,781 |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

NOTE 8—SEGMENT REPORTING

An analysis of our operations by segment is as follows:

| | | Three Months Ended March 31, | |
|---|-------------------|---------------------------------|--|
| | 2010 | 2009 | |
| | (In t | housands) | |
| REVENUES(1): | # 400 = 04 | 0.500.550 | |
| Power Generation Systems | \$409,731 | \$528,573 | |
| Government Operations | 253,251 | 257,105 | |
| Adjustments and Eliminations | (594) | (625) | |
| | \$662,388 | \$785,053 | |
| (1) Segment revenues are net of the following intersegment transfers and other adjustments: | | | |
| Power Generation Systems Transfers | \$ 4 | s — | |
| Government Operations Transfers | 590 | 625 | |
| Continuent operations framework | \$ 594 | \$ 625 | |
| | 3 394 | \$ 023 | |
| | 2010 | 2009 | |
| OPERATING INCOME | | | |
| Power Generation Systems | \$ 9,300 | \$ 58,159 | |
| Government Operations | 35,953 | 45,752 | |
| | \$45,253 | 103,911 | |
| Unallocated Corporate(1) | (6,507) | (8,603) | |
| Total Operating Income ⁽²⁾ | \$38,746 | \$ 95,308 | |
| Other Income (Expense): | | | |
| Interest income | 449 | 1,239 | |
| Interest expense | (5,993) | (5,261) | |
| Other expense—net | _(3,090) | (2,040) | |
| Total Other Expense | (8,634) | (6,062) | |
| Income before Provision for Income Taxes | \$30,112 | \$ 89,246 | |
| Unallocated corporate includes general corporate overhead not allocated to segments. | | | |
| Included in operating income is the following: | | | |
| Gains (Losses) on Asset Disposal and Impairments—Net: | | | |
| Power Generation Systems | \$ (12) | \$ 12 | |
| Government Operations | 25 | _ | |
| Unallocated Corporate | <u> </u> | (219) | |
| | \$ 13 | \$ (207) | |
| Equity in Income of Investees: | | | |
| Power Generation Systems | \$ 4,541 | \$ 1,643 | |
| Government Operations | 9,478 | 8,702 | |
| | \$14,019 | \$ 10,345 | |
| | ,0 | , | |

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

NOTE 9—RELATED PARTY

We have transactions with MII and its other subsidiaries occurring in the normal course of operations. These transactions included the following:

| | Three Mor | Three Months Ended | | |
|---|-----------|--------------------|--|--|
| | Marc | March 31, | | |
| | 2010 | 2009 | | |
| | (In the | (In thousands) | | |
| Purchases | \$ 357 | \$ 206 | | |
| Interest income | \$ 259 | \$ 358 | | |
| Interest expense | \$ 5,774 | \$ 5,057 | | |
| Corporate administrative expense allocated to MII and JRM | \$ 9,201 | \$ 11,274 | | |

Our accounts payable with MII consist of items of a normal recurring nature, including, but not limited to, the items disclosed above. The balances are generally settled within 60 days of the invoice date.

Our note receivable from affiliate is a loan executed by Babcock & Wilcox Canada, Ltd., a subsidiary of B&W PGG, with J. Ray McDermott Canada Holding, Ltd. ("JRMCHL") for up to 60 million Canadian Dollars. The actual amount funded to JRMCHL under this agreement was \$45.2 million. Interest is set at the Canadian Prime Rate and is due quarterly. The loan agreement is scheduled to terminate on June 30, 2011, and the entire amount outstanding under the loan agreement, including accrued interest, is scheduled to become due and payable at that time. Either party may terminate the loan agreement by providing 30 days' written notice to the other party, in which case the entire amount outstanding under the loan agreement, including accrued interest, would be due and payable at the end of that 30-day period. JRMCHL may repay the loan at any time, without penalty. No payments were made during the three months ended March 31, 2010. We expect this loan will be settled in connection with the spin-off.

Our note payable relates to a loan agreement with McDermott Kft., Budapest, Zug branch, the registered branch of McDermott Group Financing Limited Liability Company, whereby we borrowed \$355 million in connection with a reorganization of various subsidiaries of McDermott. The loan bears interest at a rate of 7.25% and is payable quarterly. The loan matures on December 21, 2011. The loan principal is due in full on the maturity date but may be paid in whole or in part in advance without penalty. During the three months ended March 31, 2010 we had no new borrowings or repayments under this facility. We expect this loan will be settled in connection with the spin-off.

On September 17, 2009, we entered into a loan agreement with MII whereby we may borrow up to \$5.0 million in connection with the acquisition of Instrumentacion y Mantenimiento de Calderas, S.A. Interest is set at the Eurodollar deposit rate and is due quarterly. The loan agreement is scheduled to terminate on December 31, 2014, and the entire amount outstanding under the loan agreement, including the accrued interest, is scheduled to become due and payable at that time. Either party may terminate the loan agreement by providing 30 days' written notice to the other party, in which case the entire amount outstanding under the loan agreement, including the accrued interest, will become due and payable at the end of that 30-day period. We expect this loan will be settled in connection with the proposed spin-off.

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. NOTES TO CONDENSED COMBINED FINANCIAL STATEMENTS (UNAUDITED)—(Continued)

NOTE 10—CREDIT FACILITY

On May 3, 2010, BWICO entered into a credit agreement with a syndicate of lenders and credit issuers. The facility provides an aggregate outstanding amount of up to \$700.0 million for revolving credit borrowings and issuances of letters of credit and expires on May 3, 2014. Upon completion of the spin-off, we will become the borrower under the credit agreement, and BWICO will become a guarantor under the credit agreement. The facility replaces our existing facilities.

Schedule II

THE BABCOCK & WILCOX OPERATIONS OF McDERMOTT INTERNATIONAL, INC. VALUATION AND QUALIFYING ACCOUNTS

| | Additions | | | |
|---|-------------------------|-------------------------|---------------------|-------------------|
| | Balance at Beginning | Charged to Costs and | Charged to Other | Balance at End |
| Description | of Period | Expenses(1) | Accounts | of Period |
| Valuation Allowance for Deferred Tax Assets | | | | |
| Year Ended December 31, 2009 | \$ (32,227) | \$ 3,238 | \$ 288 | \$(28,701) |
| Year Ended December 31, 2008 | \$ (85,384) | \$ 53,496 | \$ (339) | \$(32,227) |
| Year Ended December 31, 2007 | \$(122,215) | \$(36,831) | _ | \$(85,384) |

⁽¹⁾ Net of reductions and other adjustments, all of which are charged to costs and expenses.